

SOUTHEASTERN WISCONSIN REGIONAL PLANNING COMMISSION

KENOSHA COUNTY Leon T. Dreger

Thomas J. Gorlinski Sheila M. Siegler

MILWAUKEE COUNTY

Daniel J. Diliberti William Ryan Drew Patrick Marchese

OZAUKEE COUNTY

Leroy A. Bley Thomas H. Buestrin, Treasurer Elroy J. Schreiner

WAUKESHA COUNTY

Duane H. Bluemke Robert F. Hamilton Paul G. Vrakas

SOUTHEASTERN WISCONSIN REGIONAL PLANNING COMMISSION STAFF

Kurt W. Bauer, PE, AICP, RLS Executive Director
Philip C. Evenson, AICP Assistant Director
Kenneth R. Yunker, PE Assistant Director
Robert P. Biebel, PE Chief Environmental Engineer
Leland H. Kreblin, RLS Chief Planning Illustrator
Elizabeth A. Larsen Administrative Officer
Donald R. Martinson, PE Chief Transportation Engineer
John R. Meland Chief Economic Development Planner
Thomas D. Patterson Geographic Information Systems Manager
Bruce P. Rubin Chief Land Use Planner
Roland O. Tonn, AICP Chief Community Assistance Planner

Special acknowledgment is due Mr. Albert A. Beck and Mr. Christopher N. Jakubiak, Principal Planners; Mr. Robert E. Beglinger, Principal Engineer; Mr. John L. Zastrow, Principal Specialist; Mr. John L. Forslund, Senior Planner; Ms. Kathryn E. Sobottke, Planner; Mr. John M. Hagen, Engineer; and Ms. Mary L. Gundrum, Research Analyst, for their contributions to this report.

RACINE COUNTY

David B. Falstad, Chairman Martin J. Itzin Jean M. Jacobson, Secretary

WALWORTH COUNTY

John D. Ames Anthony F. Balestrieri Allen L. Morrison, Vice-Chairman

WASHINGTON COUNTY

Lawrence W. Hillman Daniel S. Schmidt Patricia A. Strachota

Chairman	Wisconsin Regional Planning Commission
Kurt W. Bauer*	Executive Director, Southeastern
Secretary	Wisconsin Regional Planning Commission
Edward A. Beimborn	Professor of Civil
	Engineering and Mechanics,
	University of Wisconsin-Milwaukee
Peter W. Beitzel	Vice President of International Trade,
1	fransportation, and Business Development,
Metropo	olitan Milwaukee Association of Commerce
John M. Bennett*	City Engineer, City of Franklin
Richard A. Bolte	Director of Transportation,
	Waukesha County
Benjamin J. Coopman, Jr	Highway Commissioner,
	Walworth County
Tyrone P. Dumas	Director of Public Works,
	Milwaukee County
Joel P. Ettinger	Regional Administrator, Region V,
	U. S. Department of Transportation,
	Federal Transit Administration
Raymond A. Forgianni, Jr	Director, Department of City
	Development, City of Kenosha
Michael J. Glasheen	Transit Planner, City of Racine
George E. Hall	Chief, Boundary Review Section,
	Bureau of Intergovernmental Relations,
	Wisconsin Department of Administration
Michael L. Hansen	Chairman, Wisconsin Coach Lines, Inc.
Thomas A. Howells	President, Wisconsin Motor
	Carriers Association
Dennis M. Johnson	City Engineer, City of West Allis
Robert C. Johnson	. Transit Coordinator, City of Waukesha
Stephen J. Hiniker	Environmental Policy Coordinator,
	City of Milwaukee
David A. Kuemmel	Associate Professor,
	Civil and Environmental
	Engineering, Marquette University
Thomas P. Kujawa*	Managing Director,
	Milwaukee County Transit System
Glenn Lampark	Deputy Public Works Director,
	Racine County
Joseph McCarthy	Director of Transportation,
	City of Kenosha
William A. Muth, Jr.	Director of Public Works,
	City of Brookfield
Robert R. Packee*	District Director, Wisconsin
	Department of Transportation, District 2
Frederick J. Patrie	Director of Public Works,
	Kenosha County
James E. St. John*	Division Administrator,
	U. S. Department of Transportation,
	Federal Highway Administration
William C. Schaefer*	Transportation and
	Land Use Specialist, Citizens
	for a Better Environment
Franklin B. Scharrer	Highway Commissioner,
	Washington County
Philip J. Scherer	Executive Director, Transportation
	Development Association of Wisconsin
Mariano A. Schifalacqua	City Engineer, City of Milwaukee
Roger L. Schrantz	Administrator, Division of
	Planning and Budget, Wisconsin
	Department of Transportation
Donald F. Theiler	Director, Bureau of Air Management,
V	Visconsin Department of Natural Resources
Rodney Vanden Noven	Director of Public Works,
	City of Waukesha
Sylvester N. Weyker	Highway Commissioner,
	Ozaukee County

*Member, Subcommittee on Public Involvement

Special acknowledgement is due the following former members of the Committee and individuals who regularly served as alternate members during the course of the planning program: John A. Erickson, City Engineer, City of Milwaukee; Donald K. Holland, City Administrator, City of Kenosha; Milton L. Howell, Jr., Director of Public Works, City of New Berlin; Kenneth S. Kinney, Strategic Planning Director, Department of Administration, City of Milwaukee; George A. Mitchell, Director, Department of Public Works and Development, Milwaukee County; D. Michael Mucha, Acting City Engineer, City of Mequon; Donald A. Roensch, Administrator, City of Mequon; John R. Duffe, Transportation and Air Quality Planner, Wisconsin Department of Natural Resources; Thomas L. Frank, Planning and Research Engineer, Federal Highway Administration, U. S. Department of Transportation; Douglas Gerleman, Program Development Office, Region V, Federal Transit Administration, U. S. Department of Transportation; George Gundersen, Director, Bureau of System Planning, Division of Planning and Budget, Wisconsin Department of Transportation; Lawrence P. Kaplan, Engineer in Charge-Transportation, City of Milwaukee; Kenneth J. Warren, Director, Transit and Taffic Engineering, Milwaukee County Department of Public Works; Thomas A. Winkel, District Chief Planning Engineer, District 2, Wisconsin Department of Transportation.

PLANNING REPORT NUMBER 41

÷

A REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN: 2010

Prepared by the

Southeastern Wisconsin Regional Planning Commission P. O. Box 1607 Old Courthouse 916 N. East Avenue Waukesha, Wisconsin 53187-1607

The preparation of this publication was financed in part through planning funds provided by the Wisconsin Department of Transportation and the U. S. Department of Transportation, Federal Highway and Federal Transit Administrations.

December 1994

Inside Region \$15.00 Outside Region \$30.00 (This page intentionally left blank)



STATEMENT OF THE CHAIRMAN

The Commission has completed a third-generation regional transportation system plan for Southeastern Wisconsin, advancing the vision for transportation system development into the 21st century. The recommendations contained in the plan as herein documented were developed in the Commission's continuing regional land use-transportation planning effort, which since 1963 has provided the seven-county Region with sound transportation system plans. The Commission's transportation planning efforts have for almost 35 years provided a forum in which changing social and economic conditions, as well as new legal initiatives and regulations attendant to transportation system development, can be publicly evaluated and conscientiously reflected in recommended transportation system improvements for the Region.

In 1966, after careful evaluation and intensive public review of alternatives, the Commission adopted its first-generation regional land use and transportation system plans as guides to land use change and transportation facility development through the design year 1990. Those plans, with supporting data, analyses, forecasts, and objectives, were documented in SEWRPC Planning Report No. 7. In 1978, the Commission adopted its second-generation regional land use and transportation system plans, extending the design year to 2000. These plans were documented in SEWRPC Planning Report No. 25 and reaffirmed the basic concepts underlying the initial plans.

Planning for land use and transportation system development in Southeastern Wisconsin has remained fully integrated in this, the third cycle of transportation systems planning. About 24 months ago, the Commission adopted a third-generation regional land use plan, for the design year 2010. Like the previous plans, the new land use plan promotes a compact, centralized regional settlement pattern. New urban uses are recommended to be located only in areas covered by soils suitable for such use; in areas that may be efficiently served by basic urban services and facilities, including public sanitary sewer, public water supply, and mass transit; and in areas free of special hazards, such as erosion and flooding. The plan also seeks to preserve all primary environmental corridor lands and most prime agricultural lands in the Region. The new regional transportation system plan has been designed to serve the demand for travel generated under the adopted year 2010 regional land use plan and thereby helps to implement that plan.

Significant social and economic changes have occurred in the Region since the adoption of the second-generation transportation system plan. These changes have affected the demand for travel that the transportation system plan must serve and include: significant increases in employment levels and in female participation in the labor force, significant declines in average household size and in private automobile operating costs, an increase in private automobile ownership and use, and continued land development practices that foster urban sprawl and automobile dependency. The transportation effects of these changes have included, among others, increasing trip lengths and traffic congestion, and decreasing transit ridership. The new plan has been designed to moderate travel demand to the extent practicable in order to deal with these transportation effects of social and economic change in the Region.

Accordingly, the plan seeks to reverse the trends of declining transit ridership and accelerating peak-period automobile use and traffic congestion. The plan recommends specific land use planning and site-design measures to promote more efficient and cost-effective urban development that makes travel through a variety of transportation modes both safe and practical, the completion of the long-planned Milwaukee area freeway traffic management system, and traffic engineering measures and parking regulations to ensure that full and efficient use is made of existing facilities before consideration is given to arterial street and highway system expansion. The plan proposes to develop true regional rapid and express transit systems, providing for the eventual development of commuter-rail passenger service, light-rail facilities, and busway and high-occupancy vehicle facilities within the Region. The plan also provides for a significant increase in local transit service levels, with accompanying increases in paratransit service levels in conformance with the Federal Americans with Disabilities Act of 1990 (ADA), and for the continuation of local shared-ride taxi services in small urban areas of the Region. Recognizing that, despite an emphasis on transportation system management and public transit, most passengers and goods will continue to be moved on the regional highway system, the plan provides for the rehabilitation of key arterial facilities and the expansion of highway capacity in selected corridors to relieve traffic congestion and to provide for a spacing of arterial facilities that lends proper structure to planned urban development.

The recommendations for transportation system development contained in the new plan were arrived at through a process which, in keeping with past Commission transportation planning practices, permitted the consideration of a full array of measures to resolve anticipated transportation problems prior to proposing any expansion of the arterial street and highway system. The plan design process employed by the Commission was fully consistent with the Federal rules guiding the implementation of the Intermodal Surface Transportation Efficiency Act of 1991 and the Clean Air Act Amendments of 1990. The plan has been found to conform in its methodology, data, and recommendations with the intents and purposes of the State Implementation Plan for Air Quality, as is required by Federal law.

The aforementioned social and economic changes and Federal legislation combine to underscore the need to prepare a new regional transportation system plan. This need has been met. It is, therefore, hoped that government, business, industry, and interested citizen groups and individuals within the Region will take an active interest in the recommendations contained in the new design year 2010 regional transportation system plan. It is further hoped that, upon carefully reviewing the soundness and practicality of those recommendations, concerned units and agencies of government, business, industry, and interested citizens will support plan adoption and implementation.

Respectfully submitted,

6 Jatiled

David B. Falstad Chairman

(This page intentionally left blank)

TABLE OF CONTENTS

Page

Chapter I–INTRODUCTION	1
Need for Regional Planning	1
The Regional Planning Commission	3
Basic Functions	3
Organization	4
The Region	4
Commission Work Programs to Date	8
Inventory	8
Plan Design	9
Promotion of Plan Implementation	9
Continuing Regional Land	-
Use-Transportation Study	9
Evolution of the Continuing Study	10
Objectives of the Continuing	
Land Use-Transportation Study	11
Public Participation	11
Organizational Structure	12
Scheme of Presentation	12
Chapter II—BASIC PRINCIPLES	
AND CONCEPTS	15
Scope of the Regional	
Transportation System Plan	15
Basic Principles	16
The Transportation System Planning Process	18
Study Design	18
Formulation of Objectives and Standards	19
Inventory	19
Analyses and Forecast	20
Plan Design	20
Plan Test and Evaluation	20
Plan Selection and Adoption	21
Summary	21
Chapter III—MONITORING AND	
IMPLEMENTATION ACTIVITIES	
RELATED TO THE ADOPTED YEAR	
2000 REGIONAL IRANSPORTATION SVSTEM DI AN	02
Introduction	20
Plan Adoption	20 92
Plan Amendments	20
Plan Refinements	25
Plan-Related Technical Studies	28
Plan Description: Adopted Regional	
Transportation System Plan	29
Arterial Street and Highway System	29
Jurisdictional Realignments	29
Functional Improvements	32
Surface Arterial Improvements	32
Freeway Improvements	33
Arterial Street and Highway System—2010	35
Public Transit	35
Rapid Transit Service, "Light" Rail	38
Rapid Transit Service, Bus on Freeways	- 38
Rapid Transit Service, Commuter Rail	39
Express and Local Bus Service	39
	30

Plan Design: Socio-Economic Forecasts	40
Population and Economic Activity	40
Population	41
Households	41
Employment	42
Automobile Availability	42
Truck Availability	44
Plan Design: Land Use Distribution	46
Conversion of Rural Land	
to Urban Development	46
Location of Incremental	
Urban Development	52
Major Regional Centers	53
Industrial Contors	55
Commercial Conters	55
Outdoor Bograption Contarg	61
Plan Implementation	61
Arterial Streets and Highways	61
Invited and Inginways	61
Functional Improvementa	66
Punctional Improvements	70
	70
Iransportation System Management	73
Plan Evaluation: Forecast and	
Actual Trip Making Levels	74
Internal Person Trips	74
Internal Vehicle Trips	74
Vehicle Miles of Travel (VMT)	76
Public Transit Ridership	77
Evaluation of the Forecasts	77
Summary and Conclusions	79
Chapter IV—INVENTORY	
OF TRANSPORTATION	
FACILITIES AND SERVICES	85
Introduction	85
Classification and Supply of	
Arterial Streets and Highways	85
Functional Classification	85
Jurisdictional Classification	88
Utilization of Arterial Streets and Hoghways	90
System Utilization	93
Relationship of System Use	
to Design Capacity	97
Classification, Supply, and	
Utilization of Public Transit	103
System Classification	103
System Identification	108
Rapid Transit	109
Express Transit	109
Local Transit	109
Land Area and Ponylation	
Served by Local Public Transit	112
Transit Provision and Utilization	118
Public Transit Ridershin	191
Supply and Use of Intermedal	121
Darking Facilities	101
	1.77
Dublic Then alt Daultin	121

Page

Carpool Parking Freeway Traffic Management System Summary	124 130 130
AND DATTERNS	107
AND PATTERNS	107
The 1001 Degional Inventory	191
of Trouch Moion Elements	197
A courses Checks	120
Socia Economia A course ou Chaeles	100
Troval A courses Checks	149
Inventory Findings	144
Overtity of Total Travel	140
Internal Parcon Travel	140
Internal Porcon Trin Production	147
Relationship of Vahiela Availability	1/8
Relationship of Household Size	1/8
Relationship of Household	140
Annual Income	1/9
Mode of Internal Person Trins	1/0
Trin Length	156
Salacted Characteristics of Users of	100
Various Modes of Internal Person Travel	156
Mass Transit Trin Production	158
Purposes of Internal Trips	169
Auto Occupancy by	102
Selected Trin Purpose	169
Relationship of Traval	102
Mode and Trin Purpose	163
Hourly Patterns of Internal Person Travel	163
Daily Patterns in Person Travel	163
Vehicle Availability and Internal	100
Vehicle Trip Production	165
Internal Commercial Truck Trip	
Production by Truck Type	165
Characteristics of Internal Commercial	
Truck Use within the Region on	
an Average Weekday	168
Hourly Distribution of	
Internal Vehicle Travel	168
Location of Travel	169
Characteristics of Travel in Selected	
Major Central Business Districts	173
Major Commercial and Industrial	
Center Trip Attractors	189
Trip Patterns	190
County-to-County	190
Travel between Subareas	193
External Trip Production	196
Special Surveys	198
Mass Transit User Survey	198
Personal Opinion Survey	201
Traffic Congestion Severity	202
in Last 10 Very	909
In Last 10 reals	202 202
Worst Thing about Traffic Congestion	202 202
Actions to Allowinto Traffic Congestion	200 900
Travel Reduction to Most Fodoral	200
Clean Air Act Requirements	204

-	
Pag	e
	-

Factors Needed to Influence	
Carpool and Transit Use	204
Funding Sources for County	
and Local Arterials	206
Limitations Believed to Result from	
Lack of Public Transportation	206
Special Origin-Destination Surveys	
on Intercity and Interregional	
Transportation Services	206
Quantity of Interregional	
Passanger Travel	208
Round Tring	200
Frequency of Trip	200
Trequency of Trip	010
Trip Purpose by Carrier Type	210
Locations of Origins and Destinations	210
Mode of Travel before	
Boarding the Carrier	211
Selected Socio-Economic Characteristics	211
Summary and Conclusions	212
Chapter VI—ANTICIPATED REGIONAL	
GROWTH AND CHANGE, THE ADOPTED	
REGIONAL LAND USE PLAN, AND	
ALTERNATIVE FUTURES FOR	
SOUTHEASTERN WISCONSIN	219
Introduction	219
Alternative Future Growth Scenarios	221
Economic Growth Scenarios	221
Population Growth Scenarios	222
Population Projections	224
Household Projections	225
Employment Projections	220
Demonal Income Projections	220 991
I and Has Damand Projections	095
Land Use Demand Projections	200
Adopted Regional Land Use Flan	200
	240
Residential Development	
and Redevelopment	240
Commercial Development	240
Major Commercial Centers	240
Local Commercial Development	243
Industrial Development	243
Major Industrial Centers	243
Local Industrial Development	245
Governmental and Institutional Land Use	945
Transportation Communication.	240
	240
and Utility Land Use	245
and Utility Land Use	245 247 249
and Utility Land Use	243 247 249 251
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural	245 247 249 251
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural open Other Open Land Uses	245 247 249 251 251
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households	245 247 249 251 251 251
and Utility Land Use	243 247 249 251 251 253 257
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Duble Service and destination	243 247 249 251 251 253 257
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Public Sanitary Sewer and Wattor Surging Severing	243 247 249 251 251 253 257
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Public Sanitary Sewer and Water Supply Service	243 247 249 251 251 253 257 257
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Public Sanitary Sewer and Water Supply Service Alternative Land Use Plans	243 247 249 251 253 253 257 257 258
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Public Sanitary Sewer and Water Supply Service Alternative Land Use Plans Design Year Population and	243 247 249 251 251 253 257 258 258
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Public Sanitary Sewer and Water Supply Service Alternative Land Use Plans Design Year Population and Economic Activity Levels	243 247 249 251 251 253 257 257 258 261
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Employment Distribution Public Sanitary Sewer and Water Supply Service Alternative Land Use Plans Design Year Population and Economic Activity Levels Planned Distribution of Population	243 247 249 251 253 257 257 258 261
and Utility Land Use Open Space: Recreational Land Use Open Space: Environmental Corridors Open Space: Agricultural and Other Open Land Uses Distribution of Population and Households Distribution Space: Agricultural and Other Open Land Uses Distribution of Population and Households Public Sanitary Sewer and Water Supply Service Alternative Land Use Plans Design Year Population and Economic Activity Levels Planned Distribution of Population and Economic Activity	243 247 249 251 253 257 257 258 261 261

Land Use	262
Major Centers	262
Major Commercial Centers	269
Major Industrial Centers	271
Primary Environmental Corridors	271
Prime Agricultural Lands	271
Public Sanitary Sewer and	
Water Supply Services	271
Distribution of Population	
and Households	273
Employment Distribution	278
Summary	281
Chapter VII—TRAVEL	~~~
SIMULATION MODELS	285
Introduction	285
The Role of Quantitative Analysis	285
Basic Modeling Concepts	287
Classification of Travel	290
Geographic Aggregation System	291
Format of Presentation	292
Transportation-Specific Land Use Forecasts	295
Personal Vehicles	295
Trip Generation	296
Basic Concepts	296
Validation of Second-Generation	
Trip Generation Procedures	298
Third-Generation Trip	
Generation Procedures	301
Trip Distribution	306
Basic Concents	306
Validation of Second Concration	000
Trip Distribution Procedures	308
Third-Generation Trin	
Distribution Procedures	200
Model Split	910
Basis Concents	010 010
Validation of Second Conception	910
Model Salit Dress dures	010
Modal Split Procedures	310
Infra-Generation Modal Split Procedures	311
Traffic Assignment	314
Iraffic Assignment Calibration	325
Travel Simulation Model Validation	325
Total Travel Demand–Model Validation	326
Transit Travel Demand	326
Highway Travel Demand	332
Issues Relating to Travel Simulation Models	335
Modifications to Travel Simulation Models	340
Vehicle Ownership	341
Trip Generation	341
Trip Distribution	342
Air Quality-Related Issues	342
Summary and Conclusions	342
•	
Chapter VIII—OBJECTIVES,	
PRINCIPLES, AND	
STANDARDS	345
Introduction	345

Basic Concepts and Definitions

Chapter IX—ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM MANAGEMENT PLANS	363
Introduction	363
Plan Design Methodology	363
Deficiency Determination	364
Plan Synthesis	366
Sources of Design Solutions	267
Consideration of Federal	007
Descriptions in Disa Design	967
No Duild Transportation System Dian	201
Determination C. N. Duild	307
Determination of a No-Build	000
Iransportation System	368
Arterial Street and Highway System	368
Transit System	369
Transportation System Management	369
Assignment of Travel to	
Highway and Transit Networks	374
Personal Vehicle Availability	374
Person Trip Generation	374
Mode of Travel	378
System Performance	379
System Deficiencies	379
Consideration of Travel	
Demand Management	383
Pricing Measures	392
Parking Measures	393
Land Use Measures	393
Work Schedule Changes	394
Other Measures	394
Alternative Transportation	
System Management Plans	394
Comprehensive Freeway	
Traffic Management	395
Traffic Management	395
Intelligent Vehicle	
ighway Systems (IVHS)	395
Land Use Planning and	
Site Design Measures	396
Pricing Measures	396
Public Transit Improvement	
and Expansion	397
Ranid Transit	399
Express Transit	404
Local Transit	404
Transit Force	404
Summary of Alternative Transportation	100
System Management Plans	105
Testing of Alternative Transnortation	100
System Management Plans	407
Assignment of Travel to	-101
Highway and Transit Notworks	407
Darsonal Vahiala Availability	407
Person Trin Generation	408
Mode of Travel	408

Specific Development Objectives

Principles and Standards

Overriding Considerations

345

346

346

347

348

348

409

System Deficiencies	411
Conclusions from the Testing	
of Alternative Transportation	
System Management Plans	414
Summary and Conclusions	415
No-Build Transportation System Plan	416
Alternative Transportation	
System Management Plans	417
Testing of Alternative Transportation	
System Management Plans	417
Chapter X—ALTERNATIVE PLAN	
COMPARISON AND EVALUATION	419
Introduction	419
Alternative Regional	
Transportation System Plan 1	419
Transportation System Management	419
Pricing	423
Land Use Planning and	
Site Design Measures	423
Areawide Promotional Measures	423
Traffic Management	423
Public Transit System	
Maintenance and Improvement	424
Rapid Transit Service	424
Express Transit	426
Local Transit Service	426
Transit Fares	427
Arterial Street and Highway System	
Maintenance and Improvement	427
New Arterial Facilities	428
New Arterial Facilities	428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways	428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway	428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks	428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vabiele Availability	428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Berson Trip Conception	428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel	428 428 428 428 428 428 428 428
New Arterial Facilities	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost	428 428 428 428 428 428 428 428 428 434 435 440
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional	 428 428 428 428 428 428 428 434 435 440
New Arterial Facilities	428 428 428 428 428 428 428 428 434 435 440 440
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management	428 428 428 428 428 428 428 428 428 434 435 440 440 442
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design	428 428 428 428 428 428 428 428 428 434 435 440 442 442
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional	428 428 428 428 428 428 428 428 428 434 435 440 442 442 442
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures	428 428 428 428 428 428 428 428 428 434 435 440 442 442 442 442
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Dublic Transport System	428 428 428 428 428 428 428 428 434 435 440 442 442 442 442 444
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System	428 428 428 428 428 428 428 428 428 434 435 440 440 442 442 442 442
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement	428 428 428 428 428 428 428 428 428 434 435 440 440 442 442 442 444 444
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service	428 428 428 428 428 428 428 428 428 435 440 440 442 442 442 444 444
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Land Uransit Service	428 428 428 428 428 428 428 428 428 435 440 440 442 442 442 442 444 444 445
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Express Transit Service Local Transit Service	428 428 428 428 428 428 428 428 428 435 440 440 442 442 442 442 444 444 445 445 445
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares	428 428 428 428 428 428 428 428 435 440 440 442 442 442 442 442 444 445 445 445 446
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Local Transit Service Arterial Street and Highway System	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares Arterial Street and Highway System Maintenance and Improvement New Arterial Street in the formation of the service	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Deficiencies System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares Arterial Street and Highway System Maintenance and Improvement New Arterial Facilities	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Deficiencies System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares Arterial Street and Highway System Maintenance and Improvement New Arterial Facilities Widening of Existing Arterial Facilities	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Deficiencies System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares Arterial Street and Highway System Maintenance and Improvement New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways	428 428 428 428 428 428 428 428 428 435 440 440 442 442 442 442 442 444 445 445 445 446 446 446 446 446
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Performance System Deficiencies System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares Arterial Street and Highway System Maintenance and Improvement New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway	428 428 428 428 428 428 428 428 428 428
New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks Vehicle Availability Person Trip Generation Mode of Travel System Deficiencies System Development Cost System Development Cost Alternative Regional Transportation System Plan 3 Transportation System Management Land Use Planning and Site Design Areawide Promotional Measures Traffic Management Public Transit System Maintenance and Improvement Rapid Transit Service Local Transit Service Transit Fares Arterial Street and Highway System Maintenance and Improvement New Arterial Facilities Widening of Existing Arterial Facilities Preserving Arterial Streets and Highways Assignment of Travel to Highway and Transit Networks	428 428 428 428 428 428 428 428 428 428

Person Trip Generation	450
Mode of Travel	450
System Performance	452
System Deficiencies	454
System Development Cost	456
Alternative Transportation System	
Plan Evaluation, Satisfaction of	
Objectives, and Standards	456
Objective No. 1: Serve Regional	
Land Use Pattern Efficiently	456
Objective No. 2: Minimize Costs	461
Objective No. 3: Provide a Multimodal	
Transportation System	469
Objective No. 4: Minimize Disruption	471
Objective No. 5: Protect Natural Environment.	
Promote Public Health, and Achieve	
Ambient Air Quality Standards	473
Objective No. 6: Facilitate Movement	
between Component Parts of the Region	475
Objective No. 7: Reduce Accident Exposure	478
Objective No. 8: Minimize	110
Energy Consumption	179
Objective No. 0. Facilitate	415
Linkod Trinmoking	170
Linked Inpinaking	410
Saustaction of Objectives and Standards	401
Comparison of Alternative Flan Costs	400
to Estimated Available Revenues	482
Advisory Committee Recommendation	105
for Preliminary Recommended Plan	485
Summary and Conclusions	486
Summary and Conclusions	486
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION	486
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM DI AN	486
Summary and Conclusions Chapter XI—RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN	486 487
Summary and Conclusions	486 487 487
Summary and Conclusions	486 487 487
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation System Plan	486 487 487 488
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Operation Systems Management	486 487 487 488 488
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway The Construction System	486 487 487 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Output Description	486 487 487 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions	486 487 487 488 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management	486 487 487 488 488 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS)	486 487 487 488 488 488 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures	486 487 487 488 488 488 488 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures	486 487 487 488 488 488 488 488 488 488 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit	486 487 487 488 488 488 488 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit	486 487 487 488 488 488 488 488 488 488 488
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing New Facilities	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing New Facilities System Improvement: Widening	486 487 487 488 488 488 488 488 488 489 489 489 489
Summary and Conclusions Chapter XIRECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing New Facilities System Improvement: Widening Existing Facilities	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing New Facilities System Improvement: Widening Existing Facilities System Preservation: Maintaining	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing New Facilities System Improvement: Widening Existing Facilities System Preservation: Maintaining Existing Facilities	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Arterial Street and Highway System System Expansion: Constructing New Facilities System Improvement: Widening Existing Facilities System Preservation: Maintaining Existing Facilities Plan Performance and Costs	486 487 487 488 488 488 488 488 488 488 489 489 489
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Local Transit System Expansion: Constructing New Facilities System Improvement: Widening Existing Facilities System Preservation: Maintaining Existing Facilities Plan Performance and Costs Anticipated Impact on Travel Attendant	486 487 487 488 488 488 488 488 488 489 489 489 489
Summary and Conclusions Chapter XI-RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN Introduction The Preliminary Recommended Year 2010 Regional Transportation System Plan Transportation Systems Management Comprehensive Freeway Traffic Management System Curb-Lane Parking Restrictions Traffic Management Intelligent Transportation Systems (ITS) Areawide Carpooling Promotion Measures Land Use Measures Public Transit Rapid Transit Local Transit Local Transit System Expansion: Constructing New Facilities System Improvement: Widening Existing Facilities System Preservation: Maintaining Existing Facilities Plan Performance and Costs Anticipated Impact on Travel Attendant to a 100 Percent Increase in Perceived	486 487 487 488 488 488 488 488 488 489 489 489 489

Public Reaction to the	
Preliminary Recommended Plan	509
County Jurisdictional Highway System	000
Planning Advisory Committee and	
Transportation System Planning	
and Programming Advisory	
Committee Meetings	510
Public Hearing Comments	510
r ublic Hearing Comments	F11
Advisory Committee Regenerate Dublic	911
Comment on Dualing in any Dia	
Comment on Freiminary Plan	515
Final Recommended Regional	
Transportation System Plan	518
Iransportation Systems Management	518
Transit Systems Management and	
Service Enhancement Measures	527
Public Transit Maintenance	
and Improvement	527
Rapid Transit	527
Express Transit	530
Local Transit	532
Relationship between the East-West Corridor	
Transit Study and the Recommended Plan	533
Arterial Street and Highway System	
Maintenance and Improvement	533
System Expansion: Constructing	
New Facilities	534
System Improvement: Widening	
Existing Facilities	543
System Preservation: Maintaining	040
Existing Facilities	519
System Dreasynation: The Milwaylee	040
Area Encourses Sustan	F 40
Dien Deufermennen and O	543
Plan Performance and Costs	543
Analysis of Sensitivity of the	
Recommended Regional Transportation	
System Plan to Alternative Future	
Regional Growth Scenarios	548
Summary	550
Public Reaction to Preliminary	
Recommended Plan and Advisory	
Committee Response	551
Recommended Year 2010 Regional	
Transportation System Plan	552
Recommended Plan Performance and Costs	553
Analysis of the Sensitivity of the	
Recommended Plan to Alternative	
Future Growth Scenarios	554
Anticipated Impact on Travel Attendant	•••
to a 100 Percent Increase in	
Automobile Operating Costs	554
Concluding Remarks	554
Chapter XII—PLAN	
IMPLEMENTATION	557
Introduction	557
Basic Principles and Concepts	557
Plan Implementation Organizations	558
Local-Level Agencies	558
Local Plan Commissions	558
Boards of Public Works	558
- out to be a waite if CIAB	000

Committees of the County Boards:	
Highway Trongit and Dublic Works	EEO
Highway, Iransit, and Fublic works	220
Transit Commissions and Boards	559
Areawide Agencies	559
Cooperative Contract Commissions	559
Metropolitan Transit Authority	559
Regional Transportation Authority	559
Regional Planning Commission	560
State-Level Agencies	560
Wisconsin Department	
of Transportation	560
Wisconsin Department	000
of Notional Passources	561
	901
wisconsin Department	
of Administration	561
Wisconsin Department	
of Development	561
University of Wisconsin-Extension	561
Federal-Level Agencies	561
U. S. Department of Transportation	561
U. S. Department of Transportation	
Federal Highway Administration	562
US Department of Transportation	002
Edoual Transit Administration	FCO
Federal Transit Administration	002
U. S. Environmental Protection Agency	562
Plan Adoption and Integration	562
Subsequent Adjustment of the Plan	564
Plan Implementation Recommendations	564
Transportation System Management	564
Public Transit	565
Arterial Streets and Highways	569
Jurisdictional Recommendations	569
Kenosha County	570
Milwaykoo County	570
	570
	570
	570
Walworth County	573
Washington County	581
Waukesha County	581
Jurisdictional Transfers	581
Functional Improvement	
Recommendations	581
Kenosha County	584
Milwaukee County	508
Orankas County	500
Desine County	090
	599
Walworth County	599
Washington County	600
Waukesha County	601
Functional Improvements	601
Recommended Revisions to Proposed	
National Highway System	602
Plan Implementation Priorities	604
Funding Plan Implementation	611
State Trunk Highway	
Revenues and Expenditures	611
County and Logal Highway and Transit	011
Dounty and Local Highway and Iransit	610
Concluding Decore	013
Concluding Recommendations	616
Detailed Implementation Planning	619
Major Investment Studies	619

Ρ	a	g	e
-	-	-	_

Transit Development Planning	630
Arterial Street and Highway Planning	633
Management Systems	633
Special Studies	634
Summary and Conclusions	636
Local Level	636
County Boards of Supervisors	636
Common Councils, Village	
Boards, and Town Boards	637
Areawide Level	637
Regional Planning Commission	637
State Level	638
Wisconsin Department	
of Transportation	638
Wisconsin Department	
of Natural Resources	639
Wisconsin Department	
of Administration	639
Wisconsin Department	
of Development	639
University of Wisconsin-Extension	639
Federal Level	639
U. S. Department of Transportation,	
Federal Highway Administration	639
U.S. Department of Transportation,	
Federal Transit Administration	639
U. S. Environmental Protection Agency	640
General Considerations	640

Chapter XIII—SUMMARY	
AND CONCLUSIONS	643
Introduction	643
Inventory Findings	644
Year 2000 Regional	
Plan Implementation	644
Regional Transportation	
System Facilities and Use	645
Travel Habits and Patterns	646
Objectives Principles and Standards	646
Anticipated Regional	010
Growth and Change	647
Design and Evaluation of Alternative Plans	647
Design and Evaluation of Alternative Flans	011
Dian and Public Participation	650
Final Recommended Regional	000
Transportation System Plan	650
Transportation System Fian	000
Management Flowert	651
Dublis There aid Mainden an an	001
Public Transit Maintenance	0E 1
and Improvement Element	001
Arterial Street and	
Highway Maintenance	
and Improvement Element	653
Performance and Costs	<u> </u>
of Recommended Plan	654
Plan Implementation	655
Conclusion	657

LIST OF APPENDICES

Appendix

Page

Α	Urban Spra	wl Regional Land Use Plan Alternative: 2010	661
	Table A-1	Total Population in the Region by County: 1985, 2010 Adopted Land	
		Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	663
	Table A-2	Total Households in the Region by County: 1985, 2010 Adopted Land	
		Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	663
	Table A-3	Total Employment in the Region by County: 1985, 2010 Adopted Land	
		Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	666
	Table A-4	Total Land Use in the Region by County: 1985, 2010 Adopted Land	
		Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	666
	Table A-5	Total Urban Lands in the Region by County: 1985, 2010 Adopted Land	
		Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	667
	Table A-6	Total Prime Agricultural Lands in the Region by County: 1985, 2010 Adopted	
		Land Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	667
	Table A-7	Total Primary Environmental Corridor Area in the Region by County: 1985, 2010	
		Adopted Land Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	668
	Table A-8	Population Density in the Region: Selected Years 1850-1985, 2010 Adopted	
		Land Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	668
	Table A-9	Existing and Proposed Developed Area and Population Served by Public	
		Sanitary Sewer and Water Supply Service in the Region: 1985, 2010 Adopted	
		Land Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	669
	Table A-10	Existing and Proposed Major Commercial Centers in the Region: 1985, 2010	
		Adopted Land Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	670
	Table A-11	Existing and Proposed Major Industrial Centers in the Region: 1985, 2010	
		Adopted Land Use Plan, and 2010 Existing Trends Urban Sprawl Alternative	671

Appendix

В

С

Page

Map A-1Existing Trends Urban Sprawl Land Use Plan for Southeastern Wisconsin: 2010Map A-2Comparison of 2010 Adopted Land Use Plan and 2010 Existing Trends Urban Sprawl AlternativeConsideration in the Regional Transportation System Planning Process for Southeastern Wisconsin of the Fifteen Specific Metropolitan Planning Considerations Set Forth in the Federal Intermodal Surface Transportation Efficiency Act (ISTEA)Accessibility Provided to Employment Opportunities and Major Activity Centers through Travel on Arterial Streets and Highways and Public Transit in the Region in 2010 under the No-Build Alternative and Alternative Plans 1 and 3Table C-1Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Arterial Streets and Highways: 1991 and 2010 Alternative Transportation System PlansTable C-2Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Arterial Streets and Highways: 1991 and 2010 Alternative Transportation System PlansTable C-2Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Transit: 1991 and 2010 Alternative Transportation System Plans	61 61 67 67 67
Consideration in the Regional Transportation System Planning Process for Southeastern Wisconsin of the Fifteen Specific Metropolitan Planning Considerations Set Forth in the Federal Intermodal Surface Transportation Efficiency Act (ISTEA) Accessibility Provided to Employment Opportunities and Major Activity Centers through Travel on Arterial Streets and Highways and Public Transit in the Region in 2010 under the No-Build Alternative and Alternative Plans 1 and 3 Table C-1 Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Arterial Streets and Highways: 1991 and 2010 Alternative Transportation System Plans Table C-2 Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Arterial Streets and Highways: 1991 and 2010 Alternative Transportation System Plans column the Selected Activity Centers through Travel Time Standards to Employment and Selected Activity Centers through Travel on Transit: 1991 and 2010 Alternative Transportation System Plans	6' 6' 66
 Accessibility Provided to Employment Opportunities and Major Activity Centers through Travel on Arterial Streets and Highways and Public Transit in the Region in 2010 under the No-Build Alternative and Alternative Plans 1 and 3	67 67
Table C-1Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Arterial Streets and Highways: 1991 and 2010 Alternative Transportation System PlansTable C-2Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Transit: 1991 and 2010 Alternative Transportation System Plans	6' 6(
Table C-2Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity Centers through Travel on Transit: 1991 and 2010 Alternative Transportation System Plans	60
Map C-1 Areas Meeting the Travel Time Standard for Employment by Arterial Street and Highway: 2010 No-Build Alternative Plan	6′
Map C-2 Areas Meeting the Travel Time Standard for Major Retail and Service Centers by Arterial Street and Highway: 2010 No-Build Alternative Plan	6′
Map C-3 Areas Meeting the Travel Time Standard for Major Medical Centers by Arterial Street and Highway: 2010 No-Build Alternative Plan	6
Map C-4 Areas Meeting the Travel Time Standard for Major Recreation Centers by Arterial Street and Highway: 2010 No-Build Alternative Plan	6
Map C-5 Areas Meeting the Travel Time Standard for Major Educational Centers by Arterial Street and Highway: 2010 No-Build Alternative Plan	6
Terminals by Arterial Street and Highway: 2010 No-Build Alternative Plan	6
by Transit: 2010 No-Build Alternative Plan	6
and Service Centers by Transit: 2010 No-Build Alternative Plan	6
Centers by Transit: 2010 No-Build Alternative Plan	6
Centers by Transit: 2010 No-Build Alternative Plan	6
Centers by Transit: 2010 No-Build Alternative Plan	6
Terminals by Transit: 2010 No-Build Alternative Plan Image: Constraint of the Constraint of	6
Map C-14 Areas Meeting the Travel Time Standard for Major Retail and Service	6
Map C-15 Areas Meeting the Travel Time Standard for Major Medical Centers by Arterial Street and Highway: 2010 Alternative Plan 1	6
Map C-16 Areas Meeting the Travel Time Standard for Major Recreation Centers by Arterial Street and Highway: 2010 Alternative Plan 1	6
Map C-17 Areas Meeting the Travel Time Standard for Major Educational Centers by Arterial Street and Highway: 2010 Alternative Plan 1	6
Map C-18Areas Meeting the Travel Time Standard for Scheduled Air TransportTerminals by Arterial Street and Highway: 2010 Alternative Plan 1	6
Map C-19 Areas Meeting the Travel Time Standard for Employment by Transit: 2010 Alternative Plan 1	~
Map C-20 Areas Meeting the Travel Time Standard for Major Retail and Service Centers by Transit: 2010 Alternative Plan 1	0

Appendix

Page

Map C-21	Areas Meeting the Travel Time Standard for Major Medical Contern by Transit 2010 Alternative Plan 1	680
Man C-22	Areas Meeting the Travel Time Standard for Major	009
	Recreation Centers by Transit: 2010 Alternative Plan 1	689
Map C-23	Areas Meeting the Travel Time Standard for Major	
	Educational Centers by Transit: 2010 Alternative Plan 1	690
Map C-24	Areas Meeting the Travel Time Standard for Scheduled Air	
Man Clor	Transport Terminals by Transit: 2010 Alternative Plan 1	690
Map C-25	Areas Meeting the Travel Time Standard for Employment	691
Man C-26	Areas Meeting the Travel Time Standard for Major Retail and Service	001
1.1up 0 20	Centers by Arterial Street and Highway: 2010 Alternative Plan 3	691
Map C-27	Areas Meeting the Travel Time Standard for Major Medical	
	Centers by Arterial Street and Highway: 2010 Alternative Plan 3	692
Map C-28	Areas Meeting the Travel Time Standard for Major Recreation	400
Mar. (190	Centers by Arterial Street and Highway: 2010 Alternative Plan 3	692
Map C-29	Areas Meeting the Travel Time Standard for Major Educational	603
Man C-30	Areas Meeting the Travel Time Standard for Scheduled Air Transport	000
map 0.00	Terminals by Arterial Street and Highway: 2010 Alternative Plan 3	693
Map C-31	Areas Meeting the Travel Time Standard for	
-	Employment by Transit: 2010 Alternative Plan 3	694
Map C-32	Areas Meeting the Travel Time Standard for Major Retail	
16 0.00	and Service Centers by Transit: 2010 Alternative Plan 3	694
Map C-33	Areas Meeting the Travel Time Standard for Major	605
Man C 34	Aroos Meeting the Travel Time Standard for Major	090
Map C-04	Recreation Centers by Transit: 2010 Alternative Plan 3	695
Map C-35	Areas Meeting the Travel Time Standard for Major	
	Educational Centers by Transit: 2010 Alternative Plan 3	696
Map C-36	Areas Meeting the Travel Time Standard for Scheduled Air	
	Transport Terminals by Transit: 2010 Alternative Plan 3	696
Environme	ntal Assessment of the Recommended Regional Transportation System Plan	697
Table D-1	Transit System Operating Characteristics Facilities and Performance: 1991 and	
Table D-1	2010 No-Build and Recommended Regional Transportation System Plan	700
Table D-2	Arterial Street and Highway System Preservation, Improvement,	
	and Expansion by Arterial Facility Type by County: 2010 Final	
	Recommended Regional Transportation System Plan	701
Table D-3	Comparison of Land-Taking Requirements for Transportation System	
	Improvements in the Region by County: 2010 No-Build and	700
Table D 4	Comparison of Motor Fuel Consumption by Vahieles	102
Table D-4	Traveling in the Region by Fuel Type: 2010 No-Build and	
	Recommended Regional Transportation System Plans	704
Table D-5	Characteristics of the No-Build and Recommended Regional	
	Transportation System Plans Which Relate to Energy Consumption	705
Table D-6	Air Pollutant Emissions from Vehicles on the Arterial Street and	
	Highway System in the Region: 1991 and 2010 No-Build and	-
m -1.1. m 7	Recommended Regional Transportation System Plans	705
Table D-7	Costs in the Region over the Deriod 1004 2010: No Ruild and	
	Recommended Regional Transportation System Plans	706
Table D-8	Total Population Served by Public Transit by Area: 1991 and 2010	
	No-Build and Recommended Regional Transportation System Plans	707
Table D-9	Comparison of Urbanized Area Population Meeting Travel Time Standards	
	to Employment Opportunities and Selected Activity Centers through Travel	
	on Transit and on Arterial Streets and Highways: 2010 No-Build and	-
	Recommended Regional Transportation System Plans	707

D

LIST OF TABLES

Table

Chapter II

Page

1	Transportation-Specific Inventories Conducted for Use in the	
	Third-Generation Regional Transportation System Plan Preparation	19
	Chapter III	
_		
2	The Evolving Regional Transportation System Plan for Southeastern Wisconsin	26
3	Technical Reports Prepared in Support of the Adopted Plan 2000 As Amended	28
4	Planned Distribution of Arterial Street and Highway Mileage	
	in the Region by Jurisdictional Classification: 2000	30
5	Planned Miles of Jurisdictional Realignments in the Region by County: 2000	32
6	Planned Miles of Functional Improvements to the Arterial Street and	
	Highway System in the Region by Jurisdictional System and County: 2000	33
7	Planned Distribution of Arterial Street and Highway Mileage	
	in the Region by Jurisdictional Classification: 2010	35
8	Planned Miles of Functional Improvements to the Arterial Street and	
	Highway System in the Region by Jurisdictional System and County: 2010	36
9	Operating Characteristics of Public Transit Service in the	
	Region by Urbanized Area: 1972 and Adopted 2000 Plan	36
10	Actual and Forecast Population Levels in the Region by County: 1970 and 1990	42
11	Actual and Forecast Household Levels in the Region by County: 1970 and 1990	44
12	Actual and Forecast Employment Levels in the Region by County: 1970 and 1990	46
13	Actual and Forecast Automobile Availability in the Region by County: 1970 and 1990	48
14	Actual and Forecast Motor Truck Availability in the Begion: 1970 and 1990	48
15	Actual and Forecast Personal Vehicle Availability in the Region: 1990	49
16	Irban Land Conversion in the Region: 1970.1985	51
17	Planned and Actual Proportion of New Housing	•+
11	Inits by Residential Land Use Category, 1970, 1985	51
18	Adval and Planned Change in the Number of Hausing Units in	01
10	the Derice Nu Decidential L and Lie Cotegowy 1070 and 1095	59
10	In exercising by Residential Land Use Category: 1970 and 1965	52
19	Stotue of Moior Industrial Content in the Region 1000	56
20	Status of Major Industrial Centers in the Region: 1990	50
21	Status of Major Commercial Centers in the Region: 1390	60
22	Status of Major Outdoor Recreation Centers in the Region: 1990	02
23	Implementation Status of Jurisdictional Realignments Recommended in the	
	Adopted Year 2000 Regional Transportation System Plan by County: 1991	64
24	Implementation Status of Planned Functional Improvements to the Arterial	
~-	Street and Highway System as Set Forth in the Design Year 2000 Plan: 1991	68
25	Implementation Status of Planned Freeways in the Region: 1978-1991	70
26	Summary Implementation Status of Total Planned Functional	
	Improvements to the Arterial Street and Highway System: 1991	70
27	Actual and Forecast Annual Revenue Vehicle-Miles of Public	
	Transit Service in the Region by Urbanized Area: 1972 and 1991	73
28	Operating Characteristics of Public Fixed-Route Transit Service in the	
	Region by Urbanized Area: 1972, 1991, and Year 2000 Adopted Plan	73
29	Actual and Forecast Internal Person Trips, Vehicle Trips,	
	and Vehicle Miles of Travel in the Region: 1972 and 1991	76
30	Actual and Forecast Annual Public Transit Revenue	
	Passengers in the Region by Urbanized Area: 1972 and 1991	77
31	Arterial Street and Highway System Traffic Congestion on an	
	Average Weekday in the Region: Historical and Planned Systems	80
	Chapter IV	

32	Distribution of Total Street and Highway Mileage	
	within the Region by County: 1963, 1972, and 1991	87
33	Change in Arterial Street and Highway Mileage: 1972-1991 and 1963-1991	88
34	Distribution of Highway Mileage by Functional Classification and by County	90
35	Existing Distribution of Arterial Street and Highway Mileage within	
	the Region by County and Jurisdictional Classification: 1991	93
36	Arterial Vehicle-Miles of Travel within the Region by County: 1963, 1972, and 1991	96

37	Arterial Vehicle-Miles of Travel within the Region on	
0.	an Average Weekday by County: 1963, 1972, and 1991	98
38	Volume to Canacity Batios for the Arterial Street and Highway	
00	System within the Region by County: 1963, 1972, and 1991	105
39	Distinguishing Characteristics of Commuter, Heavy, and Light Rail	108
40	Local Fixed-Route Transit Areal Extent. Population Served. and	
	Round-Trip Route Miles by Urbanized Area: 1963, 1972, and 1991	116
41	Area and Population Served by Local Fixed-Route Transit: 1963, 1972, and 1991	117
42	Average Daily Fixed-Route Transit Vehicle-Miles Provided	
	within the Region by Urbanized Area: 1963, 1972, and 1991	117
43	Land Area and Population Served by Fixed-Route Local Transit Meeting	
	SEWRPC Minimum Service Frequency Standards by Urbanized Area: 1991	118
44	Utilization of Fixed-Route Transit Service on an Average	
	Weekday within the Region by Urbanized Area: 1991	119
45	Ranking of Similar Urbanized Areas by Annual	
	Percent Utilization of Public Transit Service: 1990	120
46	Seats Provided and Passengers Carried by Fixed-Route	
	Public Transit by Central Business District	123
47	Annual Fixed-Route Public Transit Ridership within	
	the Region by Urbanized Area: 1963, 1972, 1991	123
48	Annual Rides Per Capita on Fixed-Route Public Transit Service	
	within the Region by Urbanized Area: 1963, 1972, and 1991	123
49	Supply and Use of Parking at Public Transit Parking Lots within	
	the Region by Rapid Transit Corridor: Fourth Quarter 1991	126
50	Supply and Use of Parking at Carpool Parking Lots within	
	the Region by Travel Corridor: Fourth Quarter 1991	128
51	Summary of Changes in the Provision and Utilization of Arterial Streets	
	and Highways and Public Transit within the Region: 1963, 1972, and 1991	134

Chapter V

52	The 1991 Regional Travel Inventory: Comparison of the Distribution of Households by	
	Household Size and County from 1990 Federal Census and 1991 Resident Household Survey	140
53	The 1991 Regional Travel Inventory: Comparison of the Number of Vehicles Available	
	by County from the 1990 Federal Census and 1991 Resident Household Survey Data	141
54	The 1991 Regional Travel Inventory: Comparison of the Distribution	
	of the Percentage of Households by Income and by County from the	
	1990 Federal Census and 1991 Resident Household Survey Data	141
55	The 1991 Regional Travel Inventory: Comparison of Southeastern Wisconsin Truck	
	Registrations and Truck Availability as Determined by the 1991 Truck Survey	142
56	The 1991 Regional Travel Inventory: Comparison of Estimated Southeastern	
	Wisconsin Cordon Line Crossing: Estimated Average Weekday Traffic	
	Volumes from Traffic Counts and from Travel Surveys	143
57	The 1991 Regional Travel Inventory: Comparison of Estimates of Average Weekday	
	Vehicular Traffic Crossing the Kenosha, Milwaukee, and Racine Screenlines Using	
	Unadjusted Travel Survey Data from Traffic Counts and Travel Survey Data	143
58	The 1991 Regional Travel Inventory: Comparison of Estimate Vehicle-Miles of Travel on an	
	Average Weekday in Selected Areas of the Region Using Unadjusted Travel Survey Data	145
59	The 1991 Regional Travel Inventory: Comparison of Estimated Average Weekday	
	Vehicular Traffic Crossing the Kenosha, Milwaukee, and Racine Screenlines	145
60	The 1991 Regional Travel Inventory: Comparison of Estimated Average	
	Weekday Traffic Crossing the Kenosha, Milwaukee, and Racine Cut Lines	145
61	The 1991 Regional Travel Inventory: Comparison of Estimated Vehicle-Miles	
	of Travel on an Average Weekday in Selected Areas of the Region	146
62	Average Weekday Person and Vehicle Trips by Survey Type: 1963, 1972, and 1991	147
63	Comparison of Historic Regional Internal Person Trips, Households,	
	Employment, and Population: 1963, 1972, and 1991	148
64	Average Weekday Internal Person Trips per Household in the	
	Region by Automobile Availability: 1963, 1972, and 1991	149
65	Average Weekday Internal Person Trips per Household	
	in the Region by Household Size: 1963, 1972, and 1991	149

66	Average Weekday Internal Person Trips per Household	
	in the Region by Income Group: 1963, 1972, and 1991	154
67	Distribution of Average Weekday Internal Person Trips	
	in the Region by Mode of Travel: 1963, 1972, and 1991	155
68	Average Trip Lengths and Times for Internal Person Trips	
	in the Region by Trip Purpose: 1963, 1972, and 1991	156
69	Percentage Distribution of Selected Characteristics of Tripmakers	
	by Mode of Travel in the Region: 1963, 1972, and 1991	157
70	Average Weekday Internal Transit Trips per Transit Tripmaking Household in the	
	Urbanizing Areas of the Region by Automobile Availability 1963 1972 and 1991	159
71	Average Weekday Internal Transit Trins per Transit Trinmaking Household in	100
• •	the Linhanizing Areas of the Region by Household Size, 1963, 1972, and 1901	160
79	Average Wookday Intersol Tring nor Transit Tring align Household by	100
12	the light method in the ration by Internet Council 162 1070 and 1001	161
73	Distribution of Avenue Wooldow Intermed Progen Trice in the	101
10	Distribution of Average weekday internal Person Trips in the	100
774	Region by Trip Furpose at Destination: 1963, 1972, and 1991	162
74	Average Automobile Occupancy of Average Weekday Internal Trips	* * *
-	in the Region by Selected Trip Purpose: 1963, 1972, and 1991	163
75	Vehicle Availability and Average Weekday Internal Vehicle Trips	
	in the Region by Vehicle Type: 1963, 1972, and 1991	170
76	Commercial-Use Truck Availability and Average Weekday Internal	
	Truck Trips in the Region by Type: 1963, 1972, and 1991	171
77	Selected Tripmaking Characteristics of Commercial-Use	
	Trucks Garaged in the Region: 1963, 1972, and 1991	171
78	Average Weekday Person Trip Destinations in Selected Central Business	
	Districts of the Region by Mode of Travel: 1963, 1972, and 1991	186
79	Average Weekday Vehicle Trip Destinations in Selected Central Business	
	Districts of the Region by Mode of Travel: 1963, 1972, and 1991	187
80	Average Weekday Internal Person Trip Destinations in Selected Central Business	
	Districts of the Region by Trip Purpose and Mode of Travel: 1963, 1972, and 1991	188
81	Average Weekday Person Trips, Excluding School Trips, between,	
	and within. Counties in the Region: 1963, 1972, and 1991	193
82	Average Weekday Person Trips, Excluding School Trips, between	100
02	and within Subaras in the Region: 163, 105, and 1001	195
83	Average Wedday External Derron and Vabia Tring	150
00	in the Region by Direction, 102, 1079 and 1001	106
84	In the Region by Direction. 1900, 1972, and 1991	190
04	Average weekday External Automobile and rersonal-Use Truck Driver Trips	107
05	and venicle Occupancy in the Region by 1mp Purpose: 1963, 1972, and 1991	197
69	Distribution of Average Weekday External Commercial-Use Truck Trips	100
0.0	in the Region by Destination and Purpose: 1963, 1972, and 1991	198
86	Percentage Distribution of Average Weekday Bus Passenger Trips	
	in the Region by Trip Purpose and Transit System: 1972 and 1991	199
87	Percentage Distribution of Average Weekday Bus Passenger Travel in the Region	
	by Transit System and Selected Characteristics of Transit Users: 1972 and 1991	200
88	Percentage Distribution of Average Weekday Bus Passenger Travel in	
	the Region by Transit System by 1991 Constant Dollars: 1972 and 1991	201
89	The 1991 Personal Opinion Survey: Percentage Distribution of	
	Responses Regarding Perceived Traffic Congestion in the Region	202
90	The 1991 Personal Opinion Survey: Percentage Distribution of Responses	
	Regarding Severity of Traffic Congestion during the Past 10 Years	202
91	The 1991 Personal Opinion Survey: Percentage Distribution of	
	Responses Regarding Severity of Traffic Congestion on Freeways	
	and Streets in Milwaukee County and Surrounding Counties	203
92	The 1991 Personal Oninion Survey: Percentage Distribution of	
	Responses Reparding Pareivad Effects of Traffic Congestion	203
93	The 1991 Personal Opinion Survey Percentage Distribution of Regeneras Regarding	400
	Support for Possible Actions to Alleviate Traffic Congestion in the Region	204
94	The 1991 Personal Opinion Survey Percentage Distribution of Support	204
J-1	for Dessible Actions to Reduce Automobile Trevel to and from Wayl	<u>مە</u> د
95	The 1001 Demonal Opinion Survey Demonstore Distribution of Fasters Considered	200
<i>3</i> 0	Necessary before Chapting to Corneal or Use There at her there. Division a Alart	00"
	recessary before Choosing to Carpool or Use Transit Kather than Driving Alone	205

96	The 1991 Personal Opinion Survey: Percentage Distribution of Responses	
	Regarding Choices for Possible Funding Sources for County and Local Arterials	206
97	Percentage Distribution of Responses Regarding Activities Believed	
	to Be Limited by a Lack of Public Transportation: 1972 and 1991	207
98	Number of Interregional Passenger Trips on Intercity Modes in the Region: 1972 and 1993	209
99	Percentage Distribution of Round Trips in the Region on an	
	Average Weekday by Carrier Type: 1972, 1989, 1991, and 1993	210
100	Percentage Distribution of Frequency of Interregional	
	Tripmaking by Surface Carrier Type: 1972, 1991, and 1993	211
101	Air Travel Frequency of Enplaning Passengers	
	Using General Mitchell International Airport: 1989	211
102	Percentage Distribution of Average Weekday Bus and Rail	
	Passenger Trips by Trip Purpose: 1972, 1991, and 1993	212
103	Overall Travel Purpose of Enplaning Passengers Using	
	General Mitchell International Airport: 1871 and 1989	212
104	Percentage Distribution of Location of Trip Origins and Destinations	
	for Interregional Rail, Bus, and Air Passengers: 1989, 1991, and 1993	213
105	Percentage Distribution, by Carrier Type, of Mode of Travel Used by	
	Interregional Passengers before Boarding: 1971, 1972, 1989, 1989, and 1993	213
106	Percentage Distribution of Selected Socio-Economic Characteristics of	
	Interregional Rail, Bus, and Air Passengers by Carrier Type: 1989, 1991, and 1993	214

Chapter VI

107	Population Levels in the Region by County: Actual	99 <i>4</i>
100	Howesholds in the Parison by Country Actual 1980 and 1990 and Projected 1990, 2000, and 2010	224
100	Fundament Level in the Region by County. Actual 1300 and 1300 and 1500 and 1500, 2000, and 2010	221
109	1980 and 1980 and Projected 1990 2000 and 2010	230
110	Employment I evide in the Parion by Maior Industry Group: Actual 1980 and Projected 2010	232
110	Employment Levels in the Region by Major Industry Group. Actual 1966 and 196	202
111	Percent Distribution of Employment by Major Industry	232
110	Group in the Region: Actual 1960 and 110 ected 2010	234
112	Actual and Frojected reisonal income Levels in the Region. 1950-2010	204
110	Projected Land Use Demand in the Region: 1905/2010	207
114	Existing and Frojosed Land Use in the Region, 1355	941
115	and 2010 Recommended Regional Land Use rial	271
115	Existing and Proposed Residential Land Use in the Region	949
116	by County: 1965 and 2010 Recommended Land Use Flan	472
110	Existing and Proposed Commercial Land Use in the Region	949
110	by County: 1985 and 2010 Recommended Land Ose Flan	242
117	Selected Characteristics of Flanned Major Commercial	944
110	Centers in the Region: 2010 Recommended Land Use Flan	244
118	Existing and Proposed Industrial Land Use in the Region	945
110	by County: 1985 and 2010 Recommended Land Use Plan	240
119	Selected Characteristics of Planned Major Industrial	946
1.00	Centers in the Region: 2010 Recommended Land Use Plan	. 240
120	Existing and Proposed Governmental and Institutional Land Use in	949
101	the Region by County: 1985 and 2010 Recommended Land Use Plan.	240
121	Existing and Proposed Transportation, Communication, and Utility Land	010
100	Use in the Region by County: 1985 and 2010 Recommended Land Use Plan	240
122	Existing and Proposed Recreational Land Use in the Region	950
	by County: 1985 and 2010 Recommended Land Use Plan	200
123	Selected Characteristics of Major Public Outdoor Recreation	050
	Centers in the Region: 2010 Recommended Land Use Plan	202
124	Primary Environmental Corridor Area in the Region	050
105	by County: 2010 Recommended Land Use Plan	203
125	Existing and Proposed Agricultural Land Use in the Region	0E 4
100	by County: 1985 and 2010 Recommended Land Use Plan	204
126	Existing and Proposed Open Land Uses in the Region by	054
105	County: 1985 and 2010 Recommended Land Use Plan	204
127	Existing and Proposed Population Distribution in the Region	050
	by County: 1985 and 2010 Recommended Land Use Plan	200

128	Existing and Proposed Household Distribution in the Region by County: 1985 and 2010 Recommended Land Use Plan	256
129	Incremental Urban Land Use, Population, and Households in the Region by County: 1985 and 2010 Recommended Land Use Plan	200
130	Population Density in the Region: Selected Years, 1850-1985, and 2010 Recommended Land Use Plan	258
1,31	Existing and Proposed Employment Distribution in the Region by County, 1985 and 2010 Recommended Land Use Plan	259
132	Existing and Proposed Developed Area and Population Served by Public Sanitary Sewer and Water Supply Service in the Region: 1985 and 2010 Recommended Land Use Plan	259
133	Existing and Proposed Developed Area and Population Served by Public Sanitary Sewer and Water Supply Service in the Begion by County: 1985 and 2010 Recommended Land Use Plan	261
134	Existing and Proposed Land Use in the Region: 1985, 2010 Recommended	263
135	Existing and Proposed Urban Land Use in the Region: 1985, 2010 Recommended	268
136	Existing and Proposed Major Commercial Centers in the Region: 1985, 2010 Recommended Lond Use Plan, and 2010 Alternative Futures Land Use Plans	200
137	Existing and Proposed Major Industrial Centers in the Region: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	200
138	Existing and Proposed Prime Agricultural Land in the Region: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	274
139	Area and Population Served by Public Sanitary Sewer and Water Supply Service in the Region: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	274
140	Existing and Proposed Population in the Region: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	275
141	Percentage Distribution of Population in the Region by County: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	276
142	Existing and Proposed Households in the Region: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	277
143	Percentage Distribution of Households in the Region by County: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	278
144	Population Density in the Region: Selected Years 1850-1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	279
145	Existing and Proposed Employment in the Region: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	280
146	Percentage Distribution of Employment in the Region by County: 1985, 2010 Recommended Land Use Plan, and 2010 Alternative Futures Land Use Plans	281
	Chapter VII	
147	Trip Classification and Travel Simulation Procedure	291
148	Percentage of Household by Vehicles Available and Household Size: 1991	297
149	Comparison of Observed and Estimated Internal Person Trip Generation	200
150	Within the Southeastern Wisconsin Acgion: 1972 and 1991	200
150	Internal Person Trip Production Cross-classification Models, Home-Dased work Trips	302
150	Internal Person Trip Froduction Cross-classification Models: Home Deced Shopping Trips	202
152	Internal Person Trip Production Cross-Classification Models, Nonhome Deced Unite Trips	202
154	Comparison of Observed and Patimeted Augment Trin Langth	000
104	for Total Person Travel within the Region: 1972 and 1991	309
155	Comparison of Average Weekday Transit Ridership in the Milwaukee	91.0
156	Comparison of 1984 Observed and 1984 Model-Estimated	012
157	Average weekday Iransit nuersnip by 11me Period	010
159	Comparison of 1972 and 1991 Survey-Estimated Automobile Occupancy by Trip Purpose	01Z
100	Mode-Onoice Models	915 915
160	venicle Occupancy by Trip Purpose: Automobile Person Trips per Venicle Trip	010
161	1 ype of Data Encoded for Links in HighWay NetWork	310 917
169	Footors to Convert Average Weekdey Transit Travel by Twin Durness to Travel by Time Devied	01/ 201
163	Factors to Convert Average Weekday Transit Travel by Trip r utpose to Travel by Thile Period	041 202
100	r actions to convert Average weekuay mignway maver to reak-renou maver	040

164	Percent of Boarding and Alighting Passengers by Mode of Access/Egress at Public Transit Stations	324
165	Comparison of Observed and Model-Estimated Average Weekday Internal Person Trip Generation within the Southeastern Wisconsin Region: 1990	326
166	Total Person Trip Table: 1991 Survey Estimate and Model Estimate	328
167	Comparison of Average Weekday Transit Ridership in the Milwaukee	
	Urbanized Area: 1990 Observed versus 1990 Model-Estimated Figures	332
168	Comparison of Average Weekday Unlinked Transit Passenger Trips (boarding passengers)	
	Carried by Transit Operators Serving the Four-County Milwaukee Metropolitan	
	Statistical Area: Actual 1990 Operator Counts and 1990 Model Estimates	332
169	Comparison of 1990 Observed and 1990 Model-Estimated Average	
	Weekday Transit Ridership by Time Period for Transit Operators	000
170	Serving the Four-County Milwaukee Metropolitan Statistical Area	3 32
170	Transit Ridership on Selected Bus Routes within the Region: 1990	333
171	Comparison of Estimated Vehicle Miles of Travel on an Average Weekday in the Region: 1990	333
	Chapter VIII	
		050
172	Transportation System Development Objectives, Principles, and Standards	350
173	Comparison of Transportation System Development Standards: Design	960
	Year 2000 and Revised Year 2010 Regional Transportation System Plans	300
	Chapter IX	
174	Arterial Street and Highway Facilities in the Region by Arterial Facility	
	Type and by County: 1991 and 2010 No-Build Transportation System Plan	371
175	Transit System Operating Characteristics and Facilities in the	974
1 7 0	Region: 1991 and 2010 No-Build Transportation System Plan	314
176	Selected Socio-Economic and Transportation Unaracteristics of	378
177	Distribution of Internal Person Trins Made by Household Resident in the	010
τ <i>ί</i> ι	Region by Trin Purpose 1991 and 2010 No-Build Transportation System Plan	378
178	Distribution of Internal Person Trips Made by Household Residents in the	
1.0	Region by Mode of Travel: 1991 and 2010 No-Build Transportation System Plan	379
179	Distribution of Internal Transit Trips Made by Household Residents in the	
	Region by Trip Purpose: 1991 and 2010 No-Build Transportation System Plan	380
180	Distribution of Internal Automobile Person Trips Made by Household Residents in	
	the Region by Trip Purpose: 1991 and 2010 No-Build Transportation System Plan	380
181	Distribution of Internal Automobile Driver Trips Made by Household Residents in	260
100	the Region by Trip Purpose: 1991 and 2010 No-Build Transportation System Flan	300
182	Classe 1001 and 2010 No Build Transportation System Plan	381
183	Vabicla Miles of Travel on the Arterial Street and Highway System in the	001
100	Region by County: 1991 and 2010 No-Build Transportation System Plan	382
184	Transit System Performance in the Region: 1991 and 2010 No-Build Transportation System Plan	383
185	Arterial Mileage in the Region Operating under, at, and over Design	
	Capacity: 1991 and 2010 No-Build Transportation System Plan	384
186	Urbanized Area Population Meeting Travel Time Standards to Employment	
	and Selected Activity Centers through Travel on Arterial Streets and	
	Highways: 1991 and 2010 No-Build Transportation System Plan	386
187	Urbanized Area Population Meeting Travel Time Standards to Employment and Selected Activity	0.07
100	Centers through Travel on Transit: 1991 and 2010 No-Build Transportation System Plans	301
188	Transportation System Demand Management Measures	300
180	roughdany Applicable with Southeastern Wisconsin Region	000
103	and 2010 Alternative Transportation System Management Plan 1	399
190	Transit System Operating Characteristics and Facilities in the Region: 1991	200
100	and 2010 Alternative Transportation System Management Plan 2	401
191	Transit System Operating Characteristics and Facilities in the Region: 1991	
	and 2010 Alternative Transportation System Management Plan 3	401
192	Summary of Alternative Transportation System Management Plans	406

Page

193	Selected Socio-Economic and Transportation Characteristics of the Region: 1991	
	and 2010 No-Build and Alternative Transportation System Management Plans	407
194	Distribution of Internal Person Trips within the Region by Mode: 1991 and	
	2010 No-Build and Alternative Transportation System Management Plans	408
195	Distribution of Vehicle Trips within the Region by Vehicle Class: 1991 and	
	2010 No-Build and Alternative Transportation System Management Plans	409
196	Vehicle-Miles of Travel on the Arterial Street and Highway System within the Region by	
	County: 1991 and 2010 No-Build and Alternative Transportation System Management Plans	410
197	Transit System Performance in the Region: 1991 and 2010 No-Build	
	and Alternative Transportation System Management Plans	410
198	Population Served by Local Public Transit in the Region by Urbanized Area: 1991	
	and 2010 No-Build and Alternative Transportation System Management Plans	411
199	Arterial Mileage Operating under, at, and over Design Capacity: 1991 and	
	2010 No-Build Alternative and Transportation System Management Plans	412
200	Urbanized Area Population Meeting Travel Time Standards to Employment and	
	Selected Activity Centers through Travel on Arterial Streets and Highways: 1991	
	and 2010 No-Build and Alternative Transportation System Management Plans	413
201	Urbanized Area Population Meeting Travel Time Standards to Employment	
	and Selected Activity Centers through Travel on Transit: 1991 and 2010	
	No-Build and Alternative Transportation System Management Plans	414
202	Comparison of Major Transportation Impacts of the Year	
	2010 Alternative Transportation System Management Plans	415

Chapter X

203	Summary of Major Plan Elements Comprising the No-Build and Alternative	
	Regional Transportation System Plans for Southeastern Wisconsin: 2010	420
204	Transit System Operating Characteristics and Facilities in the	
	Region: 1991 and 2010 Alternative Transportation System Plan 1	424
205	Arterial Street and Highway Facilities in the Region by County: 1991	
	and 2010 Alternative Transportation System Plan 1	430
206	Vehicle Availability within the Region: 1991 and 2010 Alternative Transportation System Plan 1	432
207	Distribution of Internal Person Trips Made by Household Residents in the Region by Trip	
	Purpose: 1991 and 2010 Adopted Land Use Plan and Alternative Transportation System Plan 1	432
208	Distribution of Internal Person Trips Made by Household Residents in the Region by Mode of	
	Travel: 1991 and 2010 Adopted Land Use Plan and Alternative Transportation System Plan 1	432
209	Distribution of Internal Automobile Driver Trips Made by Household	
	Residents in the Region by Trip Purpose; 1991 and 2010 Adopted	
	Land Use Plan and Alternative Transportation System Plan 1	433
210	Distribution of Internal Automobile Person Trips Made by Household	
	Residents in the Region by Trip Purpose; 1991 and 2010 Adopted	
	Land Use Plan and Alternative Transportation System Plan 1	433
211	Distribution of Internal Transit Trips Made by Household Residents in the Region by Trip	
	Purpose: 1991 and 2010 Adopted Land Use Plan and Alternative Transportation System Plan 1	434
212	Distribution of Total Vehicle Trips in the Region by Vehicle Class: 1991 and	
	2010 Adopted Land Use Plan and Alternative Transportation System Plan 1	435
213	Vehicle-Miles of Travel on the Arterial Street and Highway System in the	
	Region by County: 1991 and 2010 Alternative Transportation System Plan 1	436
214	Transit System Performance in the Southeastern Wisconsin	
	Region: 1991 and 2010 Alternative Transportation System Plan 1	437
215	Population Served by Public Transit in the Region: 1991	
	and 2010 Alternative Transportation System Plan 1	437
216	Traffic Congestion on the Arterial Street and Highway System in the Region by County: 1991	438
217	Traffic Congestion on the Arterial Street and Highway System in	
	the Region by County: 2010 No-Build Transportation System Plan	440
218	Traffic Congestion on the Arterial Street and Highway System in the	
	Region by County: 2010 Alternative Transportation System Plan 1	442
219	Transit System Operating Characteristics and Facilities in the	
	Region: 1991 and 2010 Alternative Transportation System Plan 3	444
220	Arterial Street and Highway Facilities in the Region by County: 1991	
	and 2010 Alternative Transportation System Plan 3	448

221	Vehicle Availability within the Region: 1991 and 2010 Alternative Transportation System Plan 3	450
222	by Trip Purpose: 1991 and 2010 Alternative Transportation System Plan 3	451
223	Distribution of Internal Person Trips Made by Household Residents in the Region by Mode of Travel: 1991 and 2010 Alternative Transportation System Plan 3	451
224	Distribution of Internal Automobile Driver Trips Made by Household Residents in the Region by Trip Purpose: 1991 and 2010 Alternative Transportation System Plan 3	451
225	Distribution of Internal Automobile Person Trips Made by Household Residents in the Region by Trip Purpose: 1991 and 2010 Alternative Transportation System Plan 3	452
226	Distribution of Internal Transit Trips Made by Household Residents in the Region by Trip Purpose: 1991 and 2010 Alternative Transportation System Plan 3	453
227	Distribution on Total Vehicle Trips in the Region by Vehicle Class: 1991 and 2010 Alternative Transportation System Plan 3	453
228	Vehicle-Miles of Travel on the Arterial Street and Highway System in the Region by County: 1991 and 2010 Alternative Transportation System Plan 3	454
229	Transit System Performance in the Region: 1991 and 2010 Alternative Transportation System Plan 3	455
230	Population Served by Public Transit in the Region 1991	455
231	Traffic Congestion on the Arterial Street and Highway System in the	400
232	Region by County: 2010 Alternative Transportation System Plan 3	400
233	and 2010 No-Build and Alternative Transportation System Plans	459
	and Selected Activity Centers through Travel on Transit 1991 and 2010 No-Build and Alternative Transportation System Plans	460
234	Transportation System Capital, Operation and Maintenance, and User Costs in the Region over the Period 1994-2010: No-Build and Alternative Regional Transportation System Plans	468
235	Comparison of the User and System Costs and Benefit-Cost Ratios: 2010 Alternative Transportation System Plans	469
236	Total Population Served by Public Transit: 2010 No-Build and Alternative Transportation System Plans	470
237	Number of Jobs Served by Rapid Transit: 2010 No-Build and Alternative Transportation System Plans	471
238	Comparison of Land-Taking Requirements for Transportation System Improvements in the Region by County: 2010 No-Build and Alternative Transportation System Plans	472
239	Loss of Primary Environmental Corridors in the Region to Improvement and Expansion of Arterial Street and Highway Facilities: 2010 Alternative Transportation System Plans 1 and 3	474
240	Average Weekday Air-Pollutant Emissions in the Region by Transportation System Plan: 1990 and 2010 No-Build and Alternative Transportation System Plans	475
241	Loss of Prime Agricultural Land in the Region to Improvement and Expansion of Arterial Street and Highway Facilities: 2010 Alternative Transportation System Plans 1 and 3	475
242	Comparison of the Amount of Total Travel in the Region: 1991 and 2010 No-Build and Alternative Transportation System Plans	476
243	Arterial Street and Highway Facilities Operating under, at, and over Design Capacity in the Barion: 2010 No.Build and Alternative Transportation System Plans	477
244	Distribution of Passenger-Miles of Travel in the Region by Mode and Facility Type: 1991 and 2010 No. Build and Alternative Transportation System Plans	479
245	Traffic Accident Experience and Cost on the Transportation System 1 the Bacion: 1001 and 2010 No. Build and Alternative Transportation System Plans	480
246	Comparison of Motor Fuel Consumption by Vehicles Traveling in the Bogion: 1991 and 2010 No Build and Alternative Travenortation System Plans	482
247	Summary of Relative Ability of the No-Build and Alternative Transportation System Plans to Meet Transportation Development Objectives	483
248	Average Annual Costs and Revenues Associated with No-Build and Alternative Transportation System Plans in the Region: 1995 through 2010	484
	Chanter XI	-01

Chapter XI

249	Transit System Operating Characteristics in the	
	Region: 1991 and 2010 Preliminary Recommended Plan	 490

250	Arterial Street and Highway System Preservation, Improvement, and Expansion by Arterial	
	Facility Type by County: 2010 Preliminary Recommended Transportation System Plan	503
251	Summary of Transportation Performance Characteristics: 1991 and	
	2010 Preliminary Recommended Transportation System Plan	504
252	Average Annual Public Costs and Revenues Associated with the Preliminary	
	Recommended Transportation System Plan: 1995 through 2010	506
253	Comparison of the Performance of Alternative Plans: 2010 Preliminary	
	Recommended Transportation System Plan and Plan with Pricing Measures	508
254	Transit System Operating Characteristics in the Region: 1991	
	and 2010 Final Recommended Transportation System Plan	528
255	Arterial Street and Highway System Preservation, Improvement, and Expansion by Arterial	
	Facility Type by County: 2010 Final Recommended Regional Transportation System Plan	534
256	Summary of Transportation Performance Characteristics of the Final Recommended	
	Regional Transportation System Plan for Southeastern Wisconsin: 2010	545
257	Average Annual Costs and Revenues Associated with the Final Recommended	
	Regional Transportation System Plan: 1995 through 2010	546
258	Demographic and Economic Characteristics: 1970, 1980, 1990, and 2010	
	Adopted Regional Land Use Plan and Alternative Future Scenarios	549
259	Travel Characteristics in the Region: 1970, 1980, 1990, and 2010	
	Adopted Regional Land Use Plan and Alternative Future Scenarios	549

Chapter XII

260	Distribution of Arterial Street and Highway Mileage within in the Region	
	by County and Jurisdictional Classification: 2010 Recommended Plan	570
261	Changes in Jurisdictional Responsibility of Arterial Streets and Highways in Kenosha	
	County under the Recommended Regional Transportation System Plan: 2010	580
262	Changes in Jurisdictional Responsibility for Arterial Streets and Highways in Milwaukee	
	County under the Recommended Regional Transportation System Plan: 2010	583
263	Changes in Jurisdictional Responsibility for Arterial Streets and Highways in Ozaukee	
	County under the Recommended Regional Transportation System Plan: 2010	586
264	Changes in Jurisdictional Responsibility for Arterial Streets and Highways in Racine	
	County under the Recommended Regional Transportation System Plan: 2010	588
265	Changes in Jurisdictional Responsibility for Arterial Streets and Highways in Walworth	
	County under the Recommended Regional Transportation System Plan: 2010	589
266	Changes in Jurisdictional Responsibility for Arterial Streets and Highways in Washington	
	County under the Recommended Regional Transportation System Plan: 2010	592
267	Changes in Jurisdictional Responsibility for Arterial Streets and Highways in Waukesha	
	County under the Recommended Regional Transportation System Plan: 2010	596
268	Functional Improvements to the Arterial Street and Highway System in Kenosha	
	County under the Final Recommended Regional Transportation System Plan: 2010	599
269	Functional Improvements to the Arterial Street and Highway System in Milwaukee	
	County under the Final Recommended Regional Transportation System Plan: 2010	600
270	Functional Improvements to the Arterial Street and Highway System in Ozaukee	
	County under the Final Recommended Regional Transportation System Plan: 2010	601
271	Functional Improvements to the Arterial Street and Highway System in Racine	
	County under the Final Recommended Regional Transportation System Plan: 2010	602
272	Functional Improvements to the Arterial Street and Highway System in Walworth	
	County under the Final Recommended Regional Transportation System Plan: 2010	603
273	Functional Improvements to the Arterial Street and Highway System in Washington	
	County under the Final Recommended Regional Transportation System Plan: 2010	604
274	Functional Improvements to the Arterial Street and Highway System in Waukesha	
	County under the Final Recommended Regional Transportation System Plan: 2010	605
275	Recommended Changes in the National Highway System Derived from the Recommended	
	2010 Regional Transportation System Plan for Southeastern Wisconsin	607
276	Number and Percent of Additional Lane-Miles of Arterial Streets and Highways in the	
	Region by County and Implementation Priority Schedule: 1996, 2001, 2007, and 2010	610
277	Number and Percent of Additional Revenue Vehicle-Miles of Transit Service in the Region	
	by Type of Service and Implementation Priority Schedule: 1996, 2001, 2007, and 2010	611
278	Average Annual Costs and Revenues Associated with Implementation of the State Trunk	
<i>.</i>	Highway Preservation, Improvement, and Expansion Element of the Recommended	
	Regional Transportation System Plan for Southeastern Wisconsin: 1995 through 2010	612

Figure

279	Average Annual Costs and Revenue Associated with Implementation of the County and Local Trunk Highway Preservation, Improvement, and Expansion Element of the Recommended	
	Regional Transportation System Plan for Southeastern Wisconsin by County: 1995 through 2010	614
280	Average Annual Costs and Revenue Associated with Implementation	
	of the Transit Element of the Recommended Regional Transportation	
	System Plan for Southeastern Wisconsin by County: 1995 through 2010	615
281	Summary of Additional Local Revenue Required to Implement the	
	Recommended Regional Transportation System Plan for Southeastern	
	Wisconsin by County: Average Annual, 1995 through 2010	617
282	Summary of Additional Local Revenue Required to Implement the Recommended Regional	
	Transportation System Plan for Southeastern Wisconsin and to Relieve the Property	
	Tax of Costs Associated with Arterial Streets and Highways by County	618
	Chapter XIII	
283	Funding Implementation of the Major Elements of the Recommended	
	Regional Transportation Plan for Southeastern Wisconsin: 2010	655

LIST OF FIGURES

Page

Chapter I

$\frac{1}{2}$	SEWRPC Organizational Structure	5 13
-	Chapter II	
3	Land Use-Transportation Planning Method	18
	Chapter III	
4	Actual and Forecast Regional and County Population: 1950-2000	43
5	Actual and Forecast Regional and County Households: 1950-2000	45
6	Actual and Forecast Regional and County Employment: 1950-2000	47
7	Actual and Forecast Automobile Availability in the Region: 1960-1990	48
8	Actual and Forecast Truck Availability in the Region: 1960-1990	48
ğ	Motor Trucks in the Region: 1991	49
10	Actual and Planned Urban Population Density in the Region: 1850-2000	53
11	Actual and Forecast Annual Revenue Vehicle-Miles	
	of Public Transit Service in the Region: 1972-2000	72
12	Actual and Forecast Average Weekday Internal	
+ -	Person and Vehicle Trips in the Region: 1972-2000	76
13	Actual and Forecast Average Weekday Vehicle Miles	
10	of Arterial Travel in the Region: 1963-2000	76
14	Actual and Forecast Public Transit Ridership in the Region: 1950-2000	77
15	Public Transit Ridership in the Region by Urbanized Areas: 1950-2000	78
16	Milwaukee Urbanized Area Freeway Flyer Ridership: 1964-1991	78
	Chapter IV	
17	Example of Highway Network Man	93
18	Vehicle-Miles of Travel per Arterial Mile within the Region by County: 1991	97
19	Arterial Vehicle Miles of Travel within the Region	
10	on an Average Weekday 1963 1972 and 1991	97
20	Relation of Freeway Mileage and Vehicle-Miles of Travel to Total	
<i>4</i> 0	TRANTON OF FROM WITH A DIMONDAL TIMOD OF FRANCING TOMA	

	Arterial Mileage and Vehicle-Miles of Travel: 1963, 1972, and 1991	99
21	Typical Urban Arterial Operating at, and over, Design	
	Capacity during Morning Peak-Traffic Period	102
22	Conceptual Levels of Service on the Freeway System	104
23	Percentage of Total Arterial Mileage Operating at, and over, Design Capacity: 1963, 1972, 1991	106
24	Classification of Existing Public Transportation	107

Figure

Page

25	Example of Transit Network	109
26	Percent Utilization of Fixed-Route Transit Service on an Average	
	Weekday within the Region by Urban Area: 1963, 1972, 1991	119
27	Weekday Bus Seats Provided and Passengers Carried to	
	and from the Kenosha Central Business District: 1991	122
28	Weekday Bus Seats Provided and Passengers Carried to	
	and from the Milwaukee Central Business District: 1991	122
29	Weekday Bus Seats Provided and Passengers Carried to	
	and from the Racine Central Business District: 1991	122
30	Historic Public Transit Ridership in Urbanized Areas within the Region	124
31	Public Transit Parking Lot Use: 1972-1991	127
32	Carpool Parking Lot Use: 1978-1991	130

Chapter V

33	Percentage Change in Internal Person Trips,	
	Households, Employment, and Population: 1963-1991	148
34	Percentage Changes in Vehicle Availability, Person Trips, Population, and	
	Mass Transit Revenue Passengers in the Region: 1963, 1972, and 1991	155
35	Percentage Distribution of Average Weekday Internal Person Trips in	
	the Region by Trip Purpose and Mode of Travel: 1963, 1972, and 1991	164
36	Hourly Variation of Average Weekday Internal Person Trips	
	in the Region by Trip Purpose at Destination: 1963	166
37	Hourly Variation of Average Weekday Internal Person Trips	
	in the Region by Trip Purpose at Destination: 1972	166
38	Hourly Variation of Average Weekday Internal Person Trips	
	in the Region by Trip Purpose at Destination: 1991	167
39	Daily Variation of Average Weekday Internal Person Trips	
	in the Region by Mode of Travel: 1963, 1972, and 1991	168
40	Daily Variation of Average Weekday Internal Person Trips	
	in the Region by Trip Purpose: 1963, 1972, and 1991	169
41	Hourly Variation of Average Weekday Internal Truck	
	and Taxi Trips in the Region by Vehicle Type: 1963	172
42	Hourly Variation of Average Weekday Internal Truck	
	and Taxi Trips in the Region by Vehicle Type: 1972	172
43	Hourly Variation of Average Weekday Internal Truck	
	and Taxi Trips in the Region by Vehicle Type: 1991	173
	Ob anton XII	
	Chapter VI	
44	Historical and Projected White Total Fertility Rate for the Region: 1960-2010	222
45	Historical and Projected Nonwhite Total Fertility Rate for the Region: 1960-2010	223
46	Projected Levels of Net Migration for the Region under High-Growth,	

45	Historical and Projected Nonwhite Total Fertility Rate for the Region: 1960-2010	223
46	Projected Levels of Net Migration for the Region under High-Growth,	
	Intermediate-Growth, and Low-Growth Future Economic Conditions	223
47	Population Levels in the Region by County: Actual and Projected: 1950-2010	225
48	Actual and Alternative Future Regional and County Population Levels: 1950-2010	226
49	Household Levels in the Region by County: Actual and Projected 1950-2010	228
50	Actual and Alternative Future Regional and County Household Levels: 1950-2010	229
51	Employment Levels in the Region by County: Actual and Projected: 1950-2010	231
52	Actual and Alternative Future Regional and County Employment Levels: 1960-2010	233
53	Actual and Projected Personal Income Levels in the Region: 1950-2010	235
54	Actual and Projected Mean Household Income in the Region: 1950-2010	235
55	Urban Population Density in the Region: Actual	
	1850-1985 and 2010 Recommended Land Use Plan	258
56	Existing and Proposed Urban Land Use in the Region: 1985, 2010 Recommended	
	Land Use Plan, and 2010 Alternative Futures Land Use Plans	268
57	Existing and Proposed Population in the Region: 1985, 2010 Recommended	
	Land Use Plan, and 2010 Alternative Futures Land Use Plans	275
58	Existing and Proposed Households in the Region: 1985, 2010 Recommended	
	Land Use Plan, and 2010 Alternative Futures Land Use Plans	277
59	Existing and Proposed Employment in the Region: 1985, 2010 Recommended	
	Land Use Plan, and 2010 Alternative Futures Land Use Plans	280

Chapter VII

60	Internal Person Trip Attraction Cross-Classification Models	305
61	Travel Time Friction Factors for Internal Total Person Trips in the Region: 1991	309
62	Travel Time to Volume/Capacity Ratio Relationships: 24-Hour and Peak-Hour	322
63	Home-Based Work Trips, Home-Based Other Trips,	
	Home-Based Shopping Trips, Nonhome-Based Trips	327

LIST OF MAPS

Page

Chapter I

1	The Southeastern Wisconsin Region	ι.												•												• •			, .		•	6
2	The Regional Setting in the Midwest	t	•••	•	•	•••	•	•••	•	•••	•	••	•	•	• •	•	•••	•	•	••	•	•••	•	•••	•	•••	• •	• •	••	••	•	7

Chapter III

3	Adoption of the Year 2000 Regional Transportation System Plan and Related Amendments	24
4	Planned Functional and Jurisdictional Arterial Street and Highway	
	System in the Region: Adopted Plan 2000 as Amended	31
5	Planned Functional Improvements to the Arterial Street and	
	Highway System in the Region: Adopted Plan 2000 as Amended	34
6	Planned Public Transit System in the Region: Adopted Plan 2000 as Amended	37
7	Regional Land Use Plan for Southeastern Wisconsin: 2000	50
8	Incremental Urban Development in the Region: 1970-1985	54
9	Major Industrial Centers in the Region: 1990	57
10	Major Commercial Centers in the Region: 1990	60
11	Major Outdoor Recreation Centers in the Region: 1990	63
12	Implementation Status of the Planned Jurisdictional Highway System Changes as	
	Set Forth in the Adopted Design Year 2000 Regional Transportation System Plan: 1991	65
13	Implementation Status of the Planned Functional Improvements to	
	the Arterial Street and Highway System as Set Forth in the Adopted	
	Design Year 2000 Regional Transportation System Plan: 1991	67
14	Implementation Status of the Planned Transit Element	
	of the Adopted Plan 2000 as Amended: 1993	71
15	Implementation Status of Planned Provision of Carpool Parking Lots	
	in Adopted Year 2000 Regional Transportation System Plan: 1991	75
16	Arterial Street and Highway System Traffic Congestion	
	on an Average Weekday in the Region: 1991	81

Chapter IV

		~~
17	Arterial Street and Highway System in the Region: 1991	89
18	Functionally Classified Arterial Street and Highway System in the Region: 1991	91
19	Jurisdictionally Classified Arterial Street and Highway System in the Region: 1991	92
20	Arterial Street and Highway Utilization in the Region: 1963	94
21	Arterial Street and Highway Utilization in the Region: 1972	95
22	Arterial Street and Highway Utilization in the Region: 1991	95
23	Congestion on the Arterial Street and Highway System in the Region: 1963	100
24	Congestion on the Arterial Street and Highway System in the Region: 1972	100
25	Congestion on the Arterial Street and Highway System in the Region: 1991	101
26	Rapid and Express Fixed-Route Public Transit in the Region: 1963	110
27	Rapid and Express Fixed-Route Public Transit in the Region: 1972	110
28	Rapid and Express Fixed-Route Public Transit in the Region: 1991	111
29	Local Fixed Route Public Transit Service in the Kenosha Urbanized Area: 1991	113
30	Local Fixed-Route Public Transit Service in the Milwaukee Urbanized Area: 1991	114
31	Local Fixed-Route Public Transit Service in the Racine Urbanized Area: 1991	115
32	Utilization of Public Transit Parking Lots in the Region: 1991	125
33	Utilization of Carpool Parking Lots in the Region: 1991	129
34	Existing Milwaukee Area Freeway Traffic Management System: 1991	131

Мар

260

264

265

266

267

270

272

Chapter V

35	The 1991 Regional Travel Inventory: Travel Inventory	
	and Analysis Areas and Accuracy Check Locations	144
36	Average Weekday Internal Person Trips per Household in the Region: 1963	150
37	Average Weekday Internal Person Trips per Household in the Region: 1972	150
38	Average Weekday Internal Person Trips per Household in the Region: 1991	151
39	Vehicles Available per Household in the Region: 1963	152
40	Vehicles Available per Household in the Region:1972	152
41	Vehicles Available per Household in the Region: 1991	153
42	Average Weekday Internal Person Trip Destinations in the Region: 1963	174
43	Average Weekday Internal Person Trip Destinations in the Region: 1972	174
44	Average Weekday Internal Person Trip Destinations in the Region: 1991	175
45	Average Weekday Mass Transit Trip Destinations in the Region: 1963	176
46	Average Weekday Mass Transit Trip Destinations in the Region: 1972	176
47	Average Weekday Mass Transit Trip Destinations in the Region: 1991	177
48	Average Weekday Truck Trip Destinations in the Region: 1963	178
49	Average Weekday Truck Trip Destinations in the Region: 1972	178
50	Average Weekday Truck Trip Destinations in the Region: 1991	179
51	Truck Garaging Locations in the Region: 1963	180
52	Truck Garaging Locations in the Region: 1972	180
53	Truck Garaging Locations in the Region: 1991	181
54	Average Weekday External Trip Desire Lines in the Region: 1963	182
55	Average Weekday External Trip Desire Lines in the Region: 1972	182
56	Average Weekday External Trip Desire Lines in the Region: 1991	183
57	Boundaries of the Milwaukee Central Business District and Core Area	184
58	Boundaries of the Racine Central Business District	185
59	Boundaries of the Kenosha Central Business District	185
60	Change in Number of Trips Attracted to Major Commercial Centers in the Region: 1972 and 1991	190
61	Change in Number of Trips Attracted to Major Industrial Centers in the Region: 1972 and 1991	191
62	Average Weekday Person Trips hetween Counties in the Region: 1963, 1972, and 1991	192
63	Average Weekday Person Trips between Subareas in the Region: 1963, 1972, and 1991	194
	Chapter VI	
64	Adopted Land Use Plan for the Southeastern Wisconsin Region: 2010	239
65	Major Commercial Centers in the Region: 2010 Recommended Land Use Plan	243
66	Major Industrial Centers in the Region: 2010 Recommended Land Use Plan	247
67	Major Governmental and Institutional Centers in the Region: 2010 Recommended Land Use Plan	249
68	Major Transportation and Utility Centers in the Region: 2010 Recommended Land Use Plan	249
69	Major Outdoor Recreation Centers in the Region: 2010 Recommended Land Use Plan	253
70	Prime Agricultural Lands in the Region: 2010 Recommended Land Use Plan	255
71	Pronosed Public Sanitary Sawar and Water Sunnly Sarvice	-00

71	Proposed Public Sanitary Sewer and Water Supply Service
	Areas in the Region: 2010 Recommended Land Use Plan
72	High-Growth Decentralized Plan for the Southeastern Wisconsin Region: 2010
73	High-Growth Centralized Plan for the Southeastern Wisconsin Region: 2010
74	Intermediate-Growth Decentralized Plan for the Southeastern Wisconsin Region: 2010
75	Low-Growth Decentralized Plan for the Southeastern Wisconsin Region: 2010
76	Major Commercial Centers in the Region: 2010 Recommended Land
	Use Plan and 2010 Alternative Futures Land Use Plans
77	Major Industrial Centers in the Region: 2010 Recommended Land
	Use Plan and 2010 Alternative Futures Land Use Plans

Chapter VII

78	Traffic Analysis Zones and Planning Analysis Areas within the Southeastern Wisconsin Region	293
79	Supplemental Subareas Used for Travel Pattern	
	Analysis within the Southeastern Wisconsin Region	294
80	Areas of Trip Production Analysis for Cross-Classification Model Development and Application	304
81	Major Local and Express Bus Routes Operated within Milwaukee County: 1991	334
82	Comparison of Simulated Average Weekday Traffic Volumes and Traffic Volume	
	Counts at Selected Locations on the Arterial Street and Highway System: 1990	336

Map

Page

83	Comparison of Estimated Average Weekday Traffic Volumes at Selected Locations on the Arterial Street and Highway System in the East-West Travel Corridor in Southeastern Wisconsin: 1991	337
	Chapter IX	
81	Arterial Street and Highway System in the Region: 2010 No Build Transportation System Plan	370
85	Public Transit System in the Region: 2010 No-Build Transportation System Plan	373
86	Area Served by Local Fixed-Route Transit in the Kenosha Urbanized	
00	Area: 1991 and 2010 No-Build Transportation System Plan	375
87	Area Served by Local Fixed-Route Transit in the Milwaukee Urbanized	
	Area: 1991 and 2010 No-Build Transportation System Plan	376
88	Area Served by Local Fixed-Route Transit in the Racine Urbanized	
	Area: 1991 and 2010 No-Build Transportation System Plan	377
89	Arterial Mileage in the Region Operating under, at, and over	
	Design Capacity: 2010 No-Build Transportation System Plan	385
90	Areas Meeting the Travel Time Standards for Employment Opportunities by Arterial	
	Street and Highway: 1991 and 2010 No-Build Transportation System Plan	388
91	Areas Meeting Travel Time Standards for Employment Opportunities through	
	Travel on Transit: 1991 and 2010 No-Build Transportation System Plan	389
92	Public Transit System in the Region: 2010 Alternative Plan 1	398
93	Public Transit System in the Region: 2010 Alternative Plan 2	400
94	Public Transit System in the Region: 2010 Alternative Plan 3	402
	Chapter X	
05	Extent of Potential Busway and Light Bail/Express	
30	Bus Cuideway Eacilities in the Milwarkee Area	425
96	Arterial Street and Highway System in the Region, 2010 Alternative Plan 1	429
90 97	Traffic Congestion on the Arterial Street and Highway System in the Region: 1991	439
91	Traffic Congestion on the Arterial Street and Highway System in the Region 1991	
50	in the Region: 2010 No Build Transportation System Plan	441
99	Traffic Congestion on the Arterial Street and Highway System	
55	in the Region 2010 Alternative Transportation System Plan 1	443
100	Arterial Street and Highway System in the Region: 2010 Alternative Plan 3	447
101	Traffic Congestion on the Arterial Street and Highway System	
101	in the Region: 2010 Alternative Transportation System Plan 3	457
102	Areas Meeting the Travel Time Standard for Employment by Arterial Street and Highway	462
103	Areas Meeting the Travel Time Standard for Employment by Transit	464
104	Accessibility to Land Use Activities in the Region	466
	Chanter XI	
		401
105	Regional Public Transit System: 2010 Preliminary Recommended Transportation System Plan	491
106	Arterial Street and Highway System: 2010 Preliminary	
	Recommended Transportation System Plan	495
107	Functional Improvements to the Arterial Street and Highway System in Kenosha	
	County: 2010 Preliminary Recommended Transportation System Plan	496
108	Functional Improvements to the Arterial Street and Highway System in Milwaukee	107
	County: 2010 Preliminary Recommended Transportation System Plan	497
109	Functional Improvements to the Arterial Street and Highway System in Ozaukee	400
	County: 2010 Preliminary Recommended Transportation System Plan	498
110	Functional Improvements to the Arternal Street and Highway System in Racine	400
	County: 2010 Preliminary Recommended Transportation System Plan	499
111	Functional Improvements to the Arterial Street and Highway System in Walworth	500
110	County: 2010 Preliminary Recommended Transportation System Plan	500
112	r unculorial improvements to the Arterial Street and Highway System in Washington	501
110	County, 2010 Freiminary Recommended Transportation System Flan in Waylocka	001
113	r unculonal improvements to the Arterial Street and righway System in Waukesha	509
114	Traffic Congestion on the Arterial Street and Highway System in the Region: 9010	004
114	Preliminary Recommended Regional Transnortation System Plan	507
115	Location of Anticinated Curb-Lane Parking Restrictions on	507
110	Arterial Streets and Highways in Kenosha County: 2010	520
		~ - *

Мар

116	Location of Anticipated Curb-Lane Parking Restrictions on	
	Arterial Streets and Highways in Milwaukee County: 2010	521
117	Location of Anticipated Curb-Lane Parking Restrictions on	
	Arterial Streets and Highways in Ozaukee County: 2010	522
118	Location of Anticipated Curb-Lane Parking Restrictions on	
	Arterial Streets and Highways in Racine County: 2010	523
119	Location of Anticipated Curb-Lane Parking Restrictions on	
	Arterial Streets and Highways in Walworth County: 2010	524
120	Location of Anticipated Curb-Lane Parking Restrictions on	
	Arterial Streets and Highways in Washington County: 2010	525
121	Location of Anticipated Curb-Lane Parking Restrictions on	
	Arterial Streets and Highways in Waukesha County: 2010	526
122	Public Transit System: 2010 Final Recommended Regional Transportation System Plan	529
123	Extent of Potential Busway and Light-Rail/Express Bus Guideway Facilities in the	
	Milwaukee Area: 2010 Final Recommended Regional Transportation System Plan	531
124	Arterial Street and Highway System in the Region: 2010 Final	
	Recommended Regional Transportation System Plan	535
125	Functional Improvements to the Arterial Street and Highway System in Kenosha	
	County: 2010 Final Recommended Regional Transportation System Plan	536
126	Functional Improvements to the Arterial Street and Highway System in Milwaukee	
	County: 2010 Final Recommended Regional Transportation System Plan	537
127	Functional Improvements to the Arterial Street and Highway System in Ozaukee	
	County: 2010 Final Recommended Regional Transportation System Plan	538
128	Functional Improvements to the Arterial Street and Highway System in Racine	
	County: 2010 Final Recommended Regional Transportation System Plan	539
129	Functional Improvements to the Arterial Street and Highway System in Walworth	
	County: 2010 Final Recommended Regional Transportation System Plan	540
130	Functional Improvements to the Arterial Street and Highway System in Washington	010
200	County 2010 Final Becommended Regional Transportation System Plan	541
191	Functional Improvements to the Arterial Street and Highway System in Waykeshe	041
101	County 2010 Final Recommended Regional Transportation System In Watkesha	549
132	Traffic Congestion on the Asterial Street and Highway System I han	044
102	Registry 2010 Final Recommended Registral Transportation System Day	517
		047

Chapter XII

133	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Kenosha County: 2010	571
134	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Milwaukee County: 2010	572
135	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Ozaukee County: 2010	574
136	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Racine County: 2010	575
137	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Walworth County: 2010	576
138	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Washington County: 2010	577
139	Proposed Jurisdiction of Arterial Streets and Highways Included in the Recommended	
	Regional Transportation Plan as Applied to Waukesha County: 2010	578
140	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Kenosha County	579
141	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Milwaukee County	582
142	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Ozaukee County	585
143	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Racine County	587
144	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Walworth County	591
145	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Washington County	594

Мар

146	Proposed Changes in Jurisdictional Responsibility for Arterial Streets and Highways	
	under the Recommended Regional Transportation Plan as Applied to Waukesha County	595
147	The Proposed National Highway System within Southeastern Wisconsin: June 1994	608
148	Recommended Additions to, and Deletions from, the Proposed	
	National Highway System within Southeastern Wisconsin	608
149	Recommended National Highway System within Southeastern Wisconsin	609
150	Transportation Facilities under Consideration in the Current Wisconsin Department	
	of Transportation East-West Travel Corridor Study (Major Investment Study)	621
151	Transportation Facilities Proposed to Be Considered	
	in a North Travel Corridor Major Investment Study	622
152	Transportation Facilities Proposed to Be Considered	
	in a Northwest Travel Corridor Major Investment Study	624
153	Transportation Facilities Proposed to Be Considered	
	in a South Travel Corridor Major Investment Study	625
154	Transportation Facilities Proposed to Be Considered	
	in a Southwest Travel Corridor Major Investment Study	626
155	Transportation Facilities Proposed to Be Considered in a	
	Milwaukee Crosstown Travel Corridor Major Investment Study	627
156	Transportation Facilities Proposed to Be Considered in a	
	Mitchell Field Travel Corridor Major Investment Study	627
157	Transportation Facilities Proposed to Be Considered in a	
	Northridge Travel Corridor Major Investment Study	628
158	Transportation Facilities Proposed to Be Considered in a	
	Future East-West Travel Corridor Major Investment Study	629
159	Transportation Facilities Proposed to Be Considered in a	
	Burlington-Antioch Travel Corridor Major Investment Study	631
160	Transportation Facilities Proposed to Be Considered in a	
	Walworth-Fox Lake Travel Corridor Major Investment Study	632

Chapter I

INTRODUCTION

This report documents a third-generation regional transportation system plan for the Southeastern Wisconsin Region. The report presents the major findings and recommendations arrived at through a reappraisal of the second-generation regional transportation system plan.¹ That plan, adopted by the Commission in 1978, in turn revised and updated the initial transportation system plan for the Region, adopted in 1966.² The new plan presented herein reflects changing conditions within the Region since the adoption of the second-generation plan and extends the plan design year to 2010.

Some significant trends affecting the movement of people and goods have emerged since the completion of the second-generation regional transportation system plan. The Southeastern Wisconsin Region, as most urbanizing areas in the United States, has since 1978 witnessed declining public transit ridership, declining costs of private automobile operation, and the continuation of local land development practices that foster automobile dependency. Federal responses to the effects of these and other trends have included the passage of the Clean Air Act Amendments of 1990 and the Intermodal Surface Transportation Efficiency Act of 1991. The trends and both pieces of legislation require a public reexamination of the plans and policies guiding transportation system planning and development. The Commission's continuing regional land use-transportation study has

¹SEWRPC Planning Report No. 25, <u>A Regional</u> Land Use-Transportation Plan for Southeastern <u>Wisconsin-2000</u>, Volume One, <u>Inventory Findings</u>, April 1975; and Volume Two, <u>Alternative</u> and Recommended Plans, May 1978.

²SEWRPC Planning Report No. 7, <u>The Regional</u> <u>Land Use-Transportation Study</u>, Volume One, <u>Inventory Findings: 1963</u>, May 1965; Volume Two, <u>Forecasts and Alternative Plans: 1990</u>, June 1966; and Volume Three, <u>Recommended</u> <u>Regional Land Use-Transportation Plans: 1990</u>, November 1966. long provided a framework within which changing conditions can be considered in the planning process.

The continuing land use-transportation study comprises an integral part of the broader mission and work program of the Regional Planning Commission. The Commission is charged by law with the function and duty of making and adopting a comprehensive plan for the physical development of the Region. The continuing land use-transportation study, as its name implies, is directed toward the preparation of two key elements of a long-range, areawide, comprehensive plan for the physical development of the Region: a land use plan and a transportation system plan. In 1992, the Commission adopted a third-generation regional land use plan, extending the design year to 2010. The land use plan provides the basis for the preparation of all other elements of the regional comprehensive plan and the new regional land use plan serves as the foundation for the completion of the companion third-generation regional transportation system plan.

This introductory chapter describes the need for regional planning and the status of the regional planning effort within Southeastern Wisconsin. This introduction to the Commission's broader regional planning program is intended to facilitate a better understanding of the transportation system plan reappraisal process and of its findings and recommendations as presented herein.

Although the plan presented in this report supersedes the previously adopted regional transportation system plans detailed in SEWRPC Planning Reports No. 7 and No. 25, the earlier plans and reports continue to have intrinsic value as important sources of historical data about the development of the land use pattern and transportation system within Southeastern Wisconsin.

NEED FOR REGIONAL PLANNING

Regional, or areawide, planning has become increasingly accepted as a necessary governmental function in the large metropolitan areas of the United States. This acceptance is based, in part, on a growing awareness that problems of physical and economic development and of environmental deterioration transcend the geographic limits and fiscal capabilities of local units of government. It has also been recognized that sound resolution of areawide problems requires the cooperation of all units and agencies of government concerned and of private interests as well.

As used by the Commission, the term "region" means an area larger than a county but smaller than a state, united by economic interest and geography, and by common developmental and environmental problems. A regional basis is necessary to provide a meaningful technical approach to the proper planning and design of public works systems such as highway and transit, and sewerage and water supply, and of park and open space facilities. A regional basis is also essential to provide a sound approach to the resolution of such environmental problems as flooding, air and water pollution, natural resource base deterioration, and changing land use.

Private as well as public interests are vitally affected by areawide developmental and environmental problems and by proposed solutions to these problems; it appears neither desirable nor possible for any one level or agency of government to impose the decisions required to resolve these kinds of problems. Such decisions can better come from consensus among the public and private interests concerned, based on a common interest in the welfare of the entire Region. Regional planning is necessary to promote this consensus and the necessary cooperation among urban and rural; local, State and Federal; and public and private interests. In this light, regional planning is not a substitute for Federal, State, or local public planning or for private planning. Rather, regional planning is a vital supplement to such planning.

The Wisconsin regional planning enabling act declares that regional planning commissions are to be entirely advisory in nature. More specifically, Section 66.945(8)(a) of the Wisconsin Statutes provides that the "... functions of the regional planning commission shall be solely advisory to ..." the governments and governmental officials which comprise and serve the geographic region in question. In a 1977 case,³

the Wisconsin Supreme Court, in addressing the powers of regional planning commissions in Wisconsin, stated "... their functions are merely advisory ..." Similarly, in a 1979 case,⁴ the Wisconsin Supreme Court stated that "The Dane County Planning Commission is merely an advisory body. Its determinations are recommendations only."

The advisory nature of regional planning commissions is at times called into question when such commissions take on planning responsibilities under special designations provided under Federal law. Under Section 208 of the Federal Clean Water Act, for example, certain regional planning commissions, including the Southeastern Wisconsin Regional Planning Commission, have been designated by the Governor as an areawide water quality management planning agency for a given geographical area. In carrying out the planning responsibilities attendant to that designation, the regional planning commissions prepare and adopt water quality management plans which are intended to be considered and subsequently approved by the Wisconsin Department of Natural Resources as the regulatory agency of the State. Under Federal and State law, those plans, once they are adopted by the Federal and State agencies concerned, can take on great significance in terms of Federal and State regulatory decision making and, hence, are frequently viewed as more than advisory plans. In those situations, however, it is the approval of the plans by the State regulatory agency concerned, and not by the regional planning commission, which transforms the plan from an advisory document to one having regulatory significance. Moreover, recent court cases in Wisconsin⁵ have made it

⁴<u>Kegonsa Joint Sanitary District v. City of</u> <u>Stoughton</u>, 87 Wis. 2d 131, 154, 274 N.W. 2d 598 (1979).

⁵Doerr v. Racine County, Case No. 92-0034, Court of Appeal; decided December 9, 1992; ordered unpublished, January 26, 1993; <u>Dane</u> County Regional Planning Commission v. Wisconsin Department of Natural Resources, Case No. 91-CV-2990, Circuit Court, Dane County.

³Tanck v. Dane County Regional Planning Commission, 81 Wis. 2d 76, 85, 260 N.W. 2d 18 (1977).

very clear that in performing their responsibilities under the Section 208 gubernatorial designation, regional planning commissions remain advisory in nature, transmitting, through plans, recommendations pertaining to water quality to the Federal and State regulatory agencies. It is those agencies which are empowered to accept, modify, or reject the planning commission recommendations. Anyone aggrieved by an action of the regulatory agency which is undertaken in reliance upon a plan recommendation must pursue legal recourse against the regulatory agency and not the advisory planning body.

Another special designation often assigned to regional planning commissions is that of the Metropolitan Planning Organization (MPO) under Federal transportation legislation. That designation is similarly gubernatorial in nature and involves preparing and adopting transportation plans and improvement programs to meet Federal requirements. Under Federal law, the MPO designation means that the regional planning commission provides a "forum for cooperative transportation decision making." While the metropolitan transportation plan is required to be adopted only by the MPO, clearly that plan is intended under Federal law to be a consensus plan that has come about through a cooperative transportation planning process. By Federal law, the plan is to be provided to the Governor. Moreover, by Federal law, the State transportation plan must incorporate and be consistent with the metropolitan transportation plan. Taking into account the policy-making functions of the Governor and the Wisconsin Department of Transportation in terms of transportation decision making in the State, the metropolitan transportation plan adopted by the Commission as the MPO for Southeastern Wisconsin is, as a practical matter, advisory to the Governor and the Department and, indeed, given Federal requirements, may be expected to be significantly shaped through the cooperative transportation decision-making process operated by the MPO by the policy directives of the Governor reflected through the Department.

A review of the Federal legislation and rules attendant to transportation improvement programming also indicates an advisory role by the Commission as the MPO. While the Commission must initially approve a multi-year transportation program, that program must then be approved by the Governor. In practice, the Governor delegates his responsibility for such approval to the Secretary of the Wisconsin Department of Transportation. As a practical matter, then, the Commission's recommendations as the MPO regarding programming are advisory to the Governor and the Department of Transportation.

Given the foregoing, then, the regional planning program in Southeastern Wisconsin emphasizes the promotion of close cooperation among the various governmental agencies concerned with land use development and with the development and operation of supporting public works facilities. The Commission believes that the highest form of areawide planning combines accurate data and competent technical work with the active participation of knowledgeable and concerned public officials and private citizens in the formulation of plans that address clearly identified problems. Such planning is intended to lead not only to a more efficient regional development pattern, but also to a more desirable environment in which to live and work.

THE REGIONAL PLANNING COMMISSION

The Southeastern Wisconsin Regional Planning Commission was established in 1960 under Section 66.945 of the Wisconsin Statutes as the official areawide planning agency for the highly urbanized southeastern region of the State. The Commission was created to provide basic information and planning services necessary to solve problems which transcend the corporate boundaries and fiscal capabilities of the local units of government comprising the Region. The financial support for the Commission's work programs is provided by county tax levies apportioned on the basis of equalized valuation. These basic funds are supplemented by State and Federal aid.

Basic Functions

The Commission conceives regional planning as having three basic functions. The first involves the collection, analysis, and dissemination of basic planning and engineering data on a uniform, areawide basis in order that better development decisions can be made in both the public and private sectors. The Commission believes that the establishment and utilization of such data can in and of itself contribute to better development decision making within the Region. The second function involves the preparation of a framework of long-range areawide plans for the physical development of the Region. This function is mandated by State enabling legislation. While the scope and content of these plans can extend to all phases of regional development, the Commission believes that emphasis should be place on the preparation of plans for land use and supporting transportation, utility, and community facilities. The third function involves the provision of a center for the coordination of day-to-day planning and plan implementation activities of all of the units and levels of government operating within the Region. Through this function, the Commission seeks to integrate regional and local plans and planning efforts and thereby to promote regional plan implementation.

Organization

The Commission consists of 21 members, who serve without pay, three from each of the seven member counties. One Commissioner from each county is appointed by the county board and is an elected county board supervisor. The remaining two from each county are appointed by the Governor, one from a list prepared by the county board, one on the Governor's own motion. The organizational structure of the Commission and its relationship to the constituent units and agencies of government comprising or operating within the Region is shown in Figure 1.

The full Commission meets at least four times a year and is responsible for establishing overall policy, adopting the annual budget, and adopting regional plan elements. The Commission has four standing committees: Executive, Administrative, Planning and Research, and Intergovernmental and Public Relations. The Executive Committee meets monthly to oversee the work effort of the Commission and is empowered to act for the Commission in all matters except the adoption of the budget and the adoption of the regional plan elements. The Administrative Committee meets monthly to oversee the routine but essential housekeeping activities of the Commission. The Planning and Research Committee meets as necessary to review all of the technical work carried out by the Commission staff and its consultants. The Intergovernmental and Public Relations Committee serves as the Commission's principal arm in the communication process with the constituent county boards.

The Committee meets as necessary to consider intergovernmental problems. The Commission is assisted in its work by 28 technical, citizen, and intergovernmental coordinating and advisory committees. These committees include both elected and appointed public officials and interested citizens with knowledge in the Commission work areas. The committees perform a significant function in both the formulation and the execution of the Commission work programs.

THE REGION

The Southeastern Wisconsin Region, as shown on Map 1, is comprised of Kenosha, Milwaukee, Ozaukee, Racine, Walworth, Washington, and Waukesha Counties. These seven counties have an area of about 2.689 square miles, or about 5 percent of the total land and inland water area of Wisconsin, and a total resident population of about 1.82 million people. About 37 percent of the population of the State lives within these seven counties, which contain three of the 13 metropolitan statistical areas wholly or partially located in Wisconsin. The Region contains about 41 percent of the tangible wealth of the State, as measured by equalized property valuation, and represents the greatest wealth-producing area of the State, providing about 39 percent of all employment in the State. The Region contains 154 general purpose local units of government, and encompasses all or parts of 11 major watersheds.

Geographically the Region is located in a relatively good position with regard to continued growth and development. It is bounded on the east by Lake Michigan, which provides an ample supply of fresh water for both domestic and industrial use and is an integral part of a major international transportation network. It is bounded on the south by the rapidly expanding northeastern Illinois metropolitan region and on the west and north by the fertile agricultural lands and desirable recreational areas of the rest of the State of Wisconsin. As shown on Map 2, many of the most important industrial areas and heaviest population concentrations in the Midwest lie within 250 miles of the Region; over 32 million people reside within this radius.

Map 1 shows the boundaries of both the urbanized areas and urban areas within the Region. Urbanized areas are delineated by the U.S. Bureau of the Census based on resident popula-

Figure 1





Source: SEWRPC.



The Southeastern Wisconsin Region, consisting of Kenosha, Milwaukee, Ozaukee, Racine, Walworth, Washington, and Waukesha Counties, encompasses an area of about 2,689 square miles, or about 5 percent of the total area of the State. The Region has a resident population of about 1.82 million persons, or about 37 percent of the population of the State, and provides about 981,400 jobs, or about 39 percent of the total employment in the State. There are 154 general-purpose local units of government in the seven-county Region.

Source: U. S. Bureau of the Census, U. S. Department of Transportation, and SEWRPC.
Map 2

THE REGIONAL SETTING IN THE MIDWEST



Many of the most important industrial areas and the largest population and employment concentrations in the Midwest are located within 250 miles of the Southeastern Wisconsin Region. More than 32 million people, or more than one-seventh of the entire population of the United States, live within the 250-mile radius.

Source: SEWRPC.

tion and population density, and are intended to represent the more intensively developed urban cores of the metropolitan areas. Metropolitan areas as delineated by the U.S. Bureau of the Census consist of one or more counties containing one or more urbanized areas, and are intended to represent socio-economic entities centered on the urbanized areas concerned. There are three urbanized areas within the Region: Kenosha, Milwaukee, and Racine. By definition each has a resident population of over 50,000 and a population density of at least 1,000 persons per square mile. Urban areas are designated by the Regional Planning Commission pursuant to guidelines issued by the U.S. Department of Transportation for use in functionally classifying streets and highways; determining the eligibility of proposed transportation improvements for Federal funding; defining areas within which outdoor advertising is to be federally regulated; and to aid in the reporting of highway performance and related data. Urban areas are also intended to represent the more intensively developed urban cores of metropolitan areas; and all urbanized areas adjusted to incorporate logical transportation facility segments thereby avoiding the needless fragmentation of the functional systems concerned.

COMMISSION WORK PROGRAMS TO DATE

Since its creation, the Regional Planning Commission has diligently pursued its three basic functions of areawide inventory, plan design, and promotion of plan implementation through intergovernmental cooperation and coordination, although the relative emphasis placed upon these functions has changed somewhat over time. Initially, major emphasis in the Commission's work program was on the inventory function, with increasing attention being placed over the years on the plan design and on the intergovernmental coordination functions.

Inventory

With respect to the inventory function, the Commission's planning program, as conducted since 1961, has resulted in the creation of a data bank containing in a readily usable form the basic planning and engineering information required for sound, areawide planning. The data assembled in the regional data bank include, among others, definitive data on streamflows; floodlands; surface and groundwater quality; woodlands, wetlands, and wildlife habitat; sites having scenic, scientific, cultural, and recreational value; soils; existing and proposed land uses; travel habits and patterns; transportation system capacity and utilization; existing and proposed utility service areas; and the demographic and economic base and structure of the Region. The data base also includes an extensive topographic and cadastral base mapping and horizontal and vertical survey control file.

Data in the regional planning data bank have been assembled both through the collation of data collected by other agencies and through inventory efforts conducted by the Commission itself. Data collated from the efforts of other agencies include highway and transit facility capacity, use, and service levels; transportation terminal facility capacity; automobile and truck availability; and population and economic activity levels. The data inventoried by the Commission is assembled through inventory efforts ranging from aerial photography, largescale topographic and cadastral base mapping, and control survey programs through extensive land use, woodland, wetland, wildlife habitat, potential park site, and public utility system inventories to massive surveys of travel characteristics, detailed operational soil surveys, and streamflow gauging and water quality monitoring efforts.

The regional planning data bank is supported by an extensive data conversion, filing, and retrieval capability which permits the basic data to be readily manipulated and tabulated by various geographic areas, ranging in size from the Region as a whole down through natural watersheds, counties, and minor civil divisions to planning analysis areas, census enumeration districts and tracts, traffic analysis zones, U.S. Public Land Survey sections and quartersections, and, for certain data, urban blocks and block faces. Of increasing importance in the regional planning data bank is the Commission's automated geographic information systems capability. Key regional map files, including land use inventory and soil survey maps, have been digitized, allowing for automated map reproduction and related data analysis functions. The Commission's planning data bank provides valuable points of departure for all Commission work efforts and is, moreover, available for use by the constituent agencies and units of government and by the private sector.

Plan Design

With respect to the plan design function, the Commission has placed great emphasis upon the development of a comprehensive plan for the physical development of the Region in the belief that such a plan is essential if land use development is to be properly coordinated with development of supporting transportation, utility, and community facility systems; if the development of each of these individual functional systems is to be coordinated with the development of each of the others; and if serious and costly developmental and environmental problems are to be avoided and a safer, more healthful and attractive, as well as more efficient, regional settlement pattern is to be achieved. Under the Commission's approach, the preparation, adoption, and use of the comprehensive plan are considered to be the primary objective of the planning process; all planning and plan implementation efforts are related to the comprehensive plan.

The comprehensive plan not only provides an official framework for coordinating and guiding growth and development within a multijurisdictional urbanizing region, but also provides a good conceptual basis for the application of systems engineering skills to the growing problems of such a region. The comprehensive regional plan also provides the essential framework for more detailed physical development planning at the county, community, and neighborhood levels.

Because the scope and complexity of areawide development problems prohibit the preparation of an entire comprehensive plan at one time, the Commission has determined to proceed with the preparation of individual plan elements which together comprise the required comprehensive plan. By the end of 1992, the adopted regional plan consisted of 24 individual plan elements.

Four of the elements of the regional comprehensive plan are land use-related: the regional land use plan, the regional housing plan, the regional library facilities and services plan, and the regional park and open space plan. Eight of the plan elements relate to transportation. These consist of the regional transportation plan, including highway and transit elements, the regional airport system plan, the transportation systems management plan, the elderly and handicapped transportation plan, and detailed transit development plans for the Kenosha, Racine, Waukesha, and West Bend urban areas. Ten of the adopted plan elements fall within the broad functional area of environmental planning. These consist of the regional water quality management plan, the regional wastewater sludge management plan, the regional air quality attainment and maintenance plan, and comprehensive watershed development plans for the Root, Fox, Milwaukee, Menomonee, Kinnickinnic, and Pike River watersheds and for the Oak Creek watershed.

The final two plan elements consist of comprehensive community development plans for the Kenosha and Racine urbanized areas.

Certain of the aforementioned plan elements, namely, the regional transportation system plan, the regional airport system plan, and the regional water quality management plan, are second-generation plans. The regional land use plan, which serves as the basis for the preparation of all the other comprehensive plan elements, is a third-generation planning effort. Many of the plan elements described above have been refined and detailed through formal plan amendments.

Promotion of Plan Implementation

With respect to the promotion of plan implementation, the Commission carries on an active community assistance planning program, in which functional guidance and advice on planning problems are provided to local units of government. The Commission's assistance to local units of government often consists of interpreting regional planning studies in the light of local conditions, so that the findings and recommendations of these studies may be incorporated into local development plans and plan implementation programs. Six local planning guides have been prepared under this program to provide information helpful in the preparation of local plans and plan implementation ordinances. The subjects of these guides are land subdivision control, official mapping, zoning, organization of local planning agencies, floodland and shoreland development, and the use of soils data in development planning and control.

CONTINUING REGIONAL LAND USE-TRANSPORTATION STUDY

The continuing regional land use-transportation study provides an organizing framework for the Commission to prepare the land use- and transportation-related elements of the regional comprehensive plan for Southeastern Wisconsin. This organizing framework is adaptable to changing conditions as study designs, which describe the major work elements to be completed under the continuing study, are periodically updated and revised. The periodic update and revision of the study designs is intended to ensure that changes in socio-economic conditions, in legal requirements, and in the state of the art of planning technology are brought to bear on the land use and transportation planning process.

Evolution of the Continuing Study

The land use-transportation study was initially undertaken in 1963 with the primary objective being the preparation of long-range land use and transportation system plans. That objective was fully met in 1966, when the Commission adopted and published the first regional land use plan and regional transportation system plan for Southeastern Wisconsin.

Federal laws enacted in the early 1960s underscored the importance of the Commission's emerging framework in guiding areawide land use and transportation planning. The 1962 Federal-Aid Highway Act required that, in order to be eligible for continued Federal aid for new highway construction, all urbanized areas in the United States had to have underway a continuing, comprehensive, areawide transportation planning process. The 1964 Federal Urban Mass Transportation Act established similar areawide planning prerequisites for Federal aid in support of transit system development. Therefore, even prior to the completion of the initial land usetransportation study, the Commission, its constituent local units of government, and the affected State and Federal agencies considered establishing a continuing land use and transportation planning effort in Southeastern Wisconsin.

The Commission published a Study Design in August 1967, describing the major work elements to be undertaken under such a continuing land use-transportation planning study. The continuing study was conducted in accordance with that design from January 1967 through December 1969. A second Study Design, updating the first, was prepared in December 1970 and revised in December 1971. The continuing land use-transportation study has been conducted in accordance with that design through December 1991. Major work elements completed under the continuing study include:

- The second-generation regional land use and transportation system plans for the design year 2000;
- Jurisdictional highway system plans for each of the seven counties comprising the Region;
- Detailed plans for integrating the stub ends of the Lake Freeway South, Lake Freeway North, Park Freeway East, Park Freeway West, Stadium Freeway South, and Stadium Freeway North into the surface arterial system;
- A rapid transit system plan for the greater Milwaukee area;
- Transportation systems management plans for the Kenosha, Milwaukee and Racine areas;
- Transit system development plans for the Kenosha, Racine, Waukesha and West Bend areas;
- Elderly and handicapped transportation plans for the entire Region, with particular emphasis on the Kenosha, Milwaukee, Racine, Waukesha, and West Bend areas;
- A freeway traffic management plan for the greater Milwaukee area; and
- The annual preparation of a transportation improvement program for the Kenosha, Milwaukee, and Racine urbanized areas.

With the adoption of the Federal Intermodal Surface Transportation Efficiency Act of 1991, a renewed importance was attached by the Federal government to the metropolitan transportation planning and programming process. This renewed emphasis was further strengthened by the requirements of the Federal Clean Air Act Amendments of 1990. It was, therefore, considered essential to prepare a new Study Design for the continuing land use-transportation planning effort in Southeastern Wisconsin. The third Study Design, updating the second, was prepared by the Commission in 1992. The reappraisal of the second-generation regional transportation system plan, presented herein, is the first major work element to be completed under the new study design. The new study design was reviewed and approved, on February 1, 1993, by the Commission's Technical Coordinating and Advisory Committee on Regional Transportation System Planning.

Objectives of the Continuing Land Use-Transportation Study

The objectives designed to guide the Commission, its constituent units of local government, and the concerned State and Federal agencies in the conduct of the continuing land usetransportation study are as follows:

- 1. To bring the policy provisions of the Federal Intermodal Surface Transportation Efficiency Act of 1991 to bear on systemwide transportation planning in the Region, to meet the planning and programming requirements of the Act so as to continue to qualify the constituent State and local units of government for Federal aids in partial support of the development of highway and transit facilities, and to assist in meeting the planning review requirements of Gubernatorial Order 29 which, in effect, continues the procedures established under Section 204 of the Federal Demonstration Cities and Metropolitan Development Act and the U.S. Bureau of the Budget Circular Memorandum A-95 issued pursuant to the Federal Intergovernmental Cooperation Act.
- 2. Continuously to update and revise data collected in, and forecasts prepared under, the initial regional land use-transportation study so that the full value of these data and forecasts can be realized and development decisions within the Region made intelligently on the basis of current, factual information.
- 3. Periodically to update and revise the plans prepared under the initial study effort in light of changing conditions within the Region.
- 4. To provide for the continued integration of the land use and transportation planning efforts within the Region with other elements of the comprehensive areawide planning effort.

- 5. To help the Wisconsin Department of Natural Resources meet the planning requirements attendant to the management of mobile sources of pollution as set forth in the Federal Clean Air Act Amendments of 1990.
- 6. Finally, and perhaps most importantly, to convert the plans prepared under the initial and continuing study efforts into action programs for plan implementation.

PUBLIC PARTICIPATION

From the very origin of the Southeastern Wisconsin Regional Planning Commission, it was recognized that the regional community and its elected and appointed representatives in government service must be involved in the regional planning program. Indeed, the Commission membership itself consists of a combination of elected local governmental officials and citizen members. Thus, by intent, policy, and organizational structure, the Commission has tried to be responsive to its constituents.

In addition to providing public participation by conducting extensive public attitudinal and behavioral surveys, the Commission has developed an intricate plan formulation and review procedure specifically designed to gain the advice and consent of concerned elected officials and citizen leaders. This procedure focuses primarily on the use of advisory committees, informal public informational meetings, and formal public hearings.

Very early in its existence the Commission recognized that any comprehensive regional planning program covers such a broad spectrum of related governmental and private development programs and interests that no agency, whatever its function or authority, could "go it alone" in such a planning program. The basic Commission organizational structure, therefore, provides for the extensive use of advisory committees the to promote necessary intergovernmental and interagency coordination, to broaden the technical knowledge and experience at the disposal of the Commission, and to involve elected and appointed public officials and knowledgeable citizen leaders more actively in the regional planning effort.

Accordingly, the third regional transportation system planning program was carried out under the guidance of the Commission's Technical Coordinating and Advisory Committee on Regional Transportation System Planning. Membership on that Committee included representatives from the U.S. Department of Transportation, Federal Transit Administration and Federal Highway Administration; from the Wisconsin Departments of Transportation and Natural Resources; from the university community; from municipal and county planning, transportation, and public works departments; and from private utilities, transportation enterprises, and environmental groups. A complete membership list of the Technical Coordinating and Advisory Committee is provided on the inside cover of this report.

In addition to the Advisory Committee, public participation in the planning process was achieved through the publication of informational materials, including three issues of the Commission Newsletter, a transportation planning forum jointly sponsored by the Wisconsin Department of Transportation and the Commission, a regional planning conference for the purpose of presenting the final recommended transportation system plan and eliciting public comment regarding it, and a series of public informational meetings and public hearings. These public involvement activities and others were intended to provide an opportunity for the general public to become familiar with the plan reevaluation process and to allow individuals and groups to affect the decision-making process through comments and questions. The public involvement procedures for the third-generation regional planning process are documented in SEWRPC Memorandum Report No. 87, Public Involvement in the Transportation System Planning and Programming Processes: Year 2010 Regional Transportation System Plan. To assist in implementing the public involvement procedures, the Commission acquired the assistance of the University of Wisconsin Cooperative Extension Service.

ORGANIZATIONAL STRUCTURE

As shown in Figure 2, the primary responsibility for the reevaluation of the present transportation plan and the preparation of a new regional transportation plan rested with the Commission's Transportation Planning Division. This division reports directly to the Executive Director of the Commission, who in turns reports to the Commission. The Transportation Planning Division was directly supported in its efforts by three service divisions of the Commission staff: Administrative Services, Cartographic and Graphic Arts, and Information Systems; and indirectly supported by the remaining functional planning divisions: Community Assistance Planning, Economic Development Planning, Environmental Planning, and Land Use Planning.

SCHEME OF PRESENTATION

The major findings and recommendations of the third regional transportation planning program are documented in this report. Following this introductory chapter, Chapter II sets forth the basic principles underlying the regional transportation plan reevaluation and outlines the steps in the planning process. Chapter III presents an overview of the adopted year 2000 regional transportation system plan, together with a summary of the progress to date in the implementation of that plan. Chapters IV and V present new benchmark data and historic trend data essential to the transportation plan reappraisal process. Specifically, Chapter IV presents data pertinent to the supply and use of existing transportation facilities in the Region, while Chapter V presents findings from the third major inventory of travel habits and patterns. conducted by the Commission in 1991.

The remaining chapters are concerned with anticipated growth and change in the Region, presenting forecasts of travel demand as well as the process of developing, evaluating, and selecting the recommended regional transportation system plan. Chapter VI presents forecasts of anticipated regional growth and change and describes the adopted 2010 regional land use plan. Chapter VII describes the development and application of transportation planning models used by the Commission to forecast, simulate, and analyze changes in travel demand, and air quality impacts. Chapter VIII reviews the objectives adopted as part of the year 2000 regional transportation plan and sets forth a revised set of objectives and supporting principles and standards. Chapter IX presents alternative transportation system plans. Chapter X presents a comparison of alterative plans and an evaluation of these plans in light of the

Figure 2

ORGANIZATIONAL STRUCTURE FOR THE REGIONAL TRANSPORTATION SYSTEM PLAN REAPPRAISAL



Source: SEWRPC.

revised objectives and standards. Chapter XI presents the recommended design year 2010 regional transportation plan. Chapter XII describes the actions which should be taken by the various concerned units and agencies of government to facilitate implementation of the regional transportation plan. Finally, Chapter XIII summarizes the report, restating the major findings, conclusions, and recommendations of the reappraisal effort.

It should be noted that this report can only summarize briefly the volume of information assembled in the extensive data collection, analysis, and forecasting phases of the plan reevaluation effort. Although the reproduction of these data in conventional form is impossible due to their magnitude and complexity, data from the files are available to member units and agencies of government and to the general public, upon specific request. This report, therefore, serves an additional purpose of indicating the type of data available from the Commission which may be of value in assisting Federal, State, and local units and agencies of government and private investors in making environmentally and socially sound decisions concerning community development and the provision of infrastructure within the Region.

This planning report is supplemented by certain other Commission publications. These include the SEWRPC technical reports and technical record series, which deal with certain phases of the work in greater depth and detail than is possible in this final report. These additional publications set forth the procedures and findings of the major work elements of the plan reevaluation, including the results and findings of the Commission's socio-economic inventories and forecasts.

Chapter II

BASIC PRINCIPLES AND CONCEPTS

A sound approach to the preparation of a regional transportation system plan requires careful consideration of the proper scope of the regional transportation system plan, the formulation of basic principles on which the planning process is based, and the clear identification of the major steps of the planning process itself.

This chapter describes the approach taken by the Commission in preparing the thirdgeneration regional transportation system plan. Specifically, this chapter defines the terms "transportation system," and "transportation system development," thereby identifying the scope of the regional transportation system planning process; sets forth the basic principles on which the transportation system planning process is based; and describes the major steps of the planning process.

SCOPE OF THE REGIONAL TRANSPORTATION SYSTEM PLAN

The "transportation system" is defined by the Commission as the functionally related surface transportation facilities and management measures that enable the intraregional and interregional movement of people and goods. The transportation system is considered down to, but not including, the level of the neighborhood and the major activity center. Transportation facilities, the physical components of the system, include arterial streets and highways,¹ transit facilities, bicycle and pedestrian facilities,² and all related terminal facilities considered on a regional scale. The transit facilities include all modes found in previous Commission planning reports to have potential application as costeffective modes within the Region, including, more specifically, bus on busway, bus on reserved lanes, bus in mixed traffic on freeways and surface arterials, bus and other highoccupancy vehicles on high-occupancy vehicle lanes, and light and commuter rail. A previous Commission study found heavy rail to be inapplicable within the Region as a cost-effective mode. Transportation facilities such as interregional railway yards and terminals, airports, and seaports are considered in the transportation system planning process only as major trip generators which must be adequately served by arterial street and highway and transit facilities.

This definition of the transportation system emphasizes that this system for moving people and goods within and through the Region, is, like all systems, more than the sum of its physical elements. Transportation system management measures, which regulate the movement and attempt to control the demand on transportation facilities, are considered integral components of the transportation system. The rapid change in development conditions within the Region requires that flexibility be incorporated into system design. Transportation man-

¹For the purpose of regional transportation planning in Southeastern Wisconsin, the following functionally classified highways are herein collectively defined as the "regional arterial highway system": urban principal arterials, urban minor arterials, rural principal arterials, rural minor arterials, and rural major collector streets and highways. This classification is consistent with the functional reclassification of arterial streets and highways required by the Federal Intermodal Surface Transportation Efficiency Act of 1991.

²Bicycle and pedestrian facilities have been included in the definition of "transportation system" since the adoption of the secondgeneration regional transportation system plan. The expansion of the definition underscores the need to incorporate these modes of transportation specifically into the general transportation specifically into the general transportatives to the more capital- and energy-intensive modes of surface travel. The Commission is in the process of preparing a regional bicycle and pedestrian facilities plan as an additional transportation-related element of the regional comprehensive plan. That plan will be documented in a separate Commission report.

agement measures promote such flexibility, enabling the system to respond better to changes in travel demand and to external forces which may affect the use and operation of the transportation system, such as changing motor fuel costs and constraints imposed by environmental protection requirements.

In the context of the regional transportation system plan, "transportation system development" refers to the process by which the system is purposely altered to realize its potential in achieving related objectives. This may mean that the system is modified to accommodate existing and future travel patterns better, or is modified to change travel patterns. Transportation system development refers to the physical expansion of the existing system, as well as to the physical and nonphysical strategies to make more meaningful use of existing facilities. Such nonphysical strategies may include freeway traffic management systems to better detect, confirm, and remove incidents; freeway and advisory information services to provide commuters with information about accidents and other disruptions of traffic flow; and freeway operational control measures, including the use of freeway on-ramp meters, to grant preferential access to high-occupancy vehicles.

Transportation system plans must be increasingly integrated into the larger urban complex to achieve broad regional development objectives, such as air quality improvement. Such integration is vital if the system is to capitalize on opportunities such as advances in telecommunications and to respond to socioeconomic changes such as smaller household sizes, greater female labor-force participation, and the growth of new major activity centers. If the transportation system is to be effectively fitted to changing travel demand, transportation policy makers must be cognizant of the changes that affect that demand.

Transportation system planning, therefore, must be carried out as an integral part of the comprehensive planning effort as described in Chapter I; transportation system plans must be based upon long-range, areawide land use plans. Only within the framework of such areawide land use plans can the planning, design, construction, and maintenance of the transportation system within an urbanizing region be purposefully directed. Accordingly, the third-generation regional transportation system plan, as herein presented, is based upon the design year 2010 regional land use plan. That plan follows the precedent set in the initial regional land use plan for Southeastern Wisconsin by calling for a centralized development pattern directing urban development into areas that are environmentally and economically suitable for such development. The data, forecasts, analyses, objectives, alternatives considered, and the actual plan, designed and adopted, in the land use planning process indicate the structure of the transportation system development proposals contained herein.

An understanding of the term "land use" is, therefore, vital to an understanding of the principles underlying the transportation system planning process. Within the context of regional planning, the term "land use" is defined as the human activities which, grouped together, form the overall generalized pattern of urban and rural development considered at a regional scale. Particular emphasis is placed on those aspects of land use which either through their individual or aggregate effects interact strongly with the need for major utility, recreational, and transportation facilities. These include such large landconsuming uses as agriculture; regional parks and open space reserves; woodlands, wetlands and surface waters; residential uses; and commercial and industrial centers. These also include major concentrations of land use activities, such as regional shopping, office, and institutional centers and major transportation terminals. Regional land use plans spatially allocate the demand for these aspects of human activity to achieve regional development objectives.

BASIC PRINCIPLES

The Commission views the planning process as cyclical in nature, alternating between areawide system planning and local project planning. With respect to transportation system planning, under this concept transportation-related proposals are initially advanced at the areawide, systems, level of planning and then an attempt is made to implement the proposals through local project planning. If, for whatever reasons, a particular facility construction or management proposal advanced at the areawide systems planning level cannot be implemented at the project level, that determination is taken into account in the next cycle of systems planning. The specific planning process applied in the transportation plan reappraisal effort is based on seven basic principles. These are:

- 1. Transportation system planning must be regional in scope. Travel patterns develop over an entire urban region without regard to corporate limits. Thus, transportation planning cannot be accomplished successfully within the confines of a single municipality or even a single county if that municipality or county is a part of a larger urban complex. The regional surface transportation system, which is composed of arterial streets and highways, transit facilities and services, bicycle and pedestrian facilities, and related terminal facilities, as well as transportation system management measures, must form a single integrated system over the entire Region, a system which can adequately serve changing regional travel patterns.
- 2. Transportation system planning must be conducted concurrently with, and cannot be separated from, land use planning. The land use pattern determines the amount and spatial distribution of travel to be accommodated by the transportation system and the ability of various modes of transportation to serve travel demand costeffectively. In turn, the transportation system is one of the most important determinants of the land use pattern, forming the basic framework for all urban development today. Although detailed land use patterns are primarily of local concern and properly subject to local planning and control, the aggregate effects of the spatial distribution of land use activities are regional in scope and interact strongly with the need for regional utility, recreation, and transportation facilities.
- 3. Highway and transit systems must be planned together. Each mode of transportation should be assigned that part of the total travel demand which it is best suited to carry. In order for the system to be most effective, arterial street and highway systems and related terminal facilities should function in a coordinated manner.
- 4. Transportation facilities and management measures must be planned as an integrated

system. The capacities of each link in the system must be carefully fitted to traffic loads and the effects of each proposed facility and management measure on the remainder of the system must be quantitatively tested. Bicycle and pedestrian facilities must be integrated into the regional transportation system planning process.

- 5. Transportation system planning must recognize the existence of a limited natural resource base to which urban and rural development must be properly adjusted to ensure a pleasant and habitable environment. Land, water, and air resources are limited and subject to grave misuse through improper land use and transportation system development. Such misuse can lead to serious and costly environmental problems that may be difficult or impossible to correct.
- 6. Transportation system planning must recognize the role of transportation in the achievement of personal and community goals. Access to good transportation, including a choice of modes, facilitates the freedom to choose between a variety of places to live, work, shop, and recreate. The role of transportation in making accessible environmentally sound economic, cultural, and educational opportunities, thus in promoting sound social and economic development, must be recognized in the transportation system planning process.
- 7. Transportation systems planning must recognize the importance of properly relating the regional transportation system to the State and national systems. The planning for the interregional movement of people and goods, particularly by railway, pipeline, and waterway, is primarily the responsibility of the State and Federal levels of government. Decisions made at these levels of government affect the scale and timing of regional transportation system development and the availability of capital funds to implement regional transportation system improvements. Therefore, coordination in the planning process with the State and Federal levels of the government becomes essential to the attainment of a balanced, integrated, and workable regional transportation system.

THE TRANSPORTATION SYSTEM PLANNING PROCESS

On the basis of the foregoing principles, the **Regional Planning Commission employs a seven**step planning process by which the Region and its primary functional relationships can be accurately described both graphically and numerically: the complex movement of people and vehicles over highway and transit facilities simulated; and the effect of different courses of action with respect to regional transportation system development evaluated. The seven steps in the planning process are: 1) study design, 2) formulation of objectives and standards, 3) inventory, 4) analyses and forecast, 5) plan design, 6) plan test and evaluation, and 7) plan selection and adoption. Plan implementation, although a step beyond the foregoing planning process, is considered throughout the process so that realization of the plans may be fostered.

Study Design

Every planning program must embrace a formal structure or study design so that the program can be carried out in a logical and consistent manner. This study design must: specify the content of the fact-gathering operations, define the geographic area for which data will be gathered and plans prepared, outline the manner in which the data collected are to be processed and analyzed, specify requirements for forecasts and for forecast accuracy, and define the nature of the plans to be prepared and the criteria for their evaluation and adoption. The study design for the third-generation regional planning program was set forth in general terms in Study Design for the Continuing Regional Land Use-Transportation Study, approved by the Commission's Technical Coordinating and Advisory Committee on Regional Transportation System Planning in February 1993.

In the approach to transportation system planning, as set forth in the various publications, the transportation system is designed to serve and support the planned land use pattern. The design and test of the transportation system plans, therefore, rest upon the basic concept that travel within a metropolitan area is an orderly and measurable occurrence directly related to land use. Based on this concept, techniques are used which make it possible to calculate future travel demand as a function of land use.

Figure 3

LAND USE-TRANSPORTATION PLANNING PROCESS



Source: SEWRPC.

By considering the future distribution of land use as the major factor in projecting travel patterns, integrated transportation systems can be designed to serve not only the existing travel patterns but also the entirely new travel patterns which evolve out of regional growth and change. Figure 3 illustrates the process followed in the formation of the third-generation regional transportation system plan and the relationship of that method to the third-generation regional land use plan.

TRANSPORTATION-SPECIFIC INVENTORIES CONDUCTED FOR USE IN THE THIRD-GENERATION REGIONAL TRANSPORTATION SYSTEM PLAN PREPARATION

Inventory of Transportation Facilities and Network Utilization	Inventory of Transportation Movement and Behavioral Factors Affecting Travel Habits and Patterns
Highway Facilities and Service Levels	Screenline Survey
Transit Facilities and Service Levels	Home Interview Survey
Transportation Terminal Facilities	Truck and Taxi Survey
Automobile Availability	External Surveys
Truck Availability	Mass Transit Survey
Source: SEWRPC.	· · · · · · · · · · · · · · · · · · ·

Formulation of Objectives and Standards

In its most basic sense, planning is a rational process for establishing and meeting objectives. The formulation of objectives is, therefore, an essential task to be undertaken before plans can be prepared. The objectives which are chosen guide the preparation of alternative plans and, when converted to standards, provide the criteria for evaluating and selecting from among the alternatives.

Since objectives provide the logical basis for the design of plans, formulation of sound objectives is a crucial step in the planning process. In order to be useful in plan design, the objectives must not only be stated clearly and be sound logically, but must also be related in a demonstrable way to alternative system development proposals. Only if the objectives are clearly relatable to system development and subject to objective test can a choice be made from among alternative plans in order to select that plan which best meets the agreed-upon objectives.

Inventory

Reliable basic planning and engineering data, collected on a uniform, areawide basis, are absolutely essential to the formulation of workable system development plans. Consequently, inventory becomes the first operational step in any planning process, growing out of the study design. The crucial nature of factual information in the planning process should be evident, since no intelligent forecasts can be made, or alternative courses of action selected, without knowledge of the current state of the system being planned.

The sound formation of a regional transportation system plan requires that factual data be developed on the existing and potential demand for transportation between various points inside the Region and outside the Region, on the relative demand for alternative modes of transportation, and on the major determinants of these demands, as well as on the existing and potential supply of transportation system capacity.

The transportation-specific inventories can be grouped under two headings: "Existing Transportation Facilities and Network Utilization," and "Existing Transportation Movement and Behavioral Factors Affecting Travel Habits and Patterns." The specific inventories under each of these major headings are shown in Table 1.

Because transportation system planning must be conducted concurrently with, and cannot be separated from, land use planning, the data derived from land use-related inventories is crucial. Accordingly, the sound formation of transportation system plans also requires that data be developed on the existing land use pattern, on the potential demand for each of the major land use categories, on the major determinants of these demands, and on existing local development objectives and constraints, as well on the underlying natural resource base and its ability to support land use development.

The major transportation and land use inventories considered together must encompass all the various factors which influence transportation system development. The data collected in the inventories must be pertinent to:

- 1. Describing existing conditions with respect to transportation system development and identifying existing problems and problem patterns with respect thereto;
- 2. Forecasting future transportation system needs in light of existing land use conditions and planned land use patterns;

- 3. Formulating alternative transportation system development plans to complement the adopted land use plan; and
- 4. Testing and evaluating alternative transportation system development plans.

Analyses and Forecast

Inventories provide factual information about the present situation, but analyses and forecasts are necessary to provide estimates of future needs for resources, land, and transportation. Analyses of the information provided by the inventories are required to provide an understanding of the existing situation, the future trends of change in that situation, and the factors influencing these trends. Particularly important among the analytical relationships established are those which link population and economic activity levels to the demand for land and transportation.

Future needs must be estimated from a sequence of interlocking forecasts founded in the results of the planning analyses. Economic activity and population forecasts set the general scale of future growth, which, in turn, is translated into future natural resources, land use, and travel demands. These future demands can then be scaled against the existing supply and plans formulated to meet deficiencies or to make more effective use of existing supply.

Plan Design

Plan design, or synthesis, is the process of fusing separate but related elements determined to be needed into the existing system to be planned. The most well-conceived objectives, the most sophisticated results of data collection and analyses, and the most accurate forecasts are of little value if they do not ultimately result in sound plans to meet system development objectives. The outputs of each of the three planning operations—formulation of objectives and standards, inventory, and forecast—become inputs to the problem of plan design.

Plan Test and Evaluation

If plans, formulated in the design stage of the planning process, are to prove practical and thereby capable of realization in terms of actual transportation system development, some measures must be applied to test alternative plans quantitatively in advance of their adoption and implementation. Traffic simulation models, developed and refined over the past three decades and applied in the planning process, make it possible to determine the existing and potential travel demand on any proposed transportation network. The complete sequence of simulation is applied in four stages.

- 1. Trip generation, in which the total number of future trips generated in each subarea of the Region is determined, using the relationships found to exist between land use and travel from analyses of inventory data.
- 2. Modal split, in which these future trips are divided into those using transit and those using personal vehicles, using a modal split model.
- 3. Trip distribution, in which the originating trips generated are allocated to destination zones; then interzonal travel desire lines are established for both transit and highway travel, using a trip distribution model.
- 4. Traffic assignment, in which the interzonal trips are then assigned to existing and proposed transit and highway facility networks, using a traffic assignment model.

Using the four-step simulation procedure, it is possible to test and verify the workability and efficiency of any transportation system development proposal. The last step in the procedure, traffic assignment, will reveal those areas where travel is anticipated to exceed, or to be within. the system capacity. The trip assignment procedure also provides the basis for determining which modifications to the system will ultimately result in a practical and efficient transportation system plan for which development costs can be calculated. Trip assignment also permits user benefits to be calculated for benefitcost analyses. Finally, such assignment provides a more precise basis for the application of standards so that the degree to which each alternative transportation system plan meets the chosen objectives can be better determined.

The degree to which the alternative regional transportation system plans serve the adopted regional land use plan is also evaluated in the test and evaluation step. During the travel simulation procedure, described above, forecasts of travel are made on the basis of the adopted regional land use plan. Specifically, these travel forecasts are based on the levels and spatial distribution of the population, household, and economic activity envisioned in the land use plan. The evaluation of alternative transportation system proposals in light of the land use plan may suggest the need for changes in the land use plan, as shown by the feedback loop in Figure 3.

Plan Selection and Adoption

The last step in the seven-step planning process is the selection and adoption of a final recommended plan from among the alternative plans considered. In this step, the alternative plan deemed most capable of meeting the stated objectives and standards is chosen for adoption by the Commission. The Commission has developed an extensive program to involve elected and appointed officials, citizen leaders, and the general public in plan selection and adoption. The public involvement program, as described in Chapter I, includes the use of advisory committees, public informational meetings, and formal public hearings. The program allows elected officials, technicians, and members of the general public concerned with regional development to participate in the selection and refinement of a final regional transportation system plan. After refinement, as warranted by the public review process, the plan is considered for adoption by the Regional Planning Commission. Upon adoption by the Commission, the plan is certified to concerned units and agencies of government for adoption and implementation.

SUMMARY

This chapter has set forth the scope of the thirdgeneration regional transportation system plan, the seven basic principles underlying the planning process, and the seven procedural steps followed by the Commission in that process. Though experience in preparing the first and second regional transportation system plans has proved the general scope, basic principles, and seven-step planning process to be essentially sound, certain modifications were made.

The definition of "transportation system" was expanded to underscore the multi-modal approach necessitated by sound planning and the integral role transportation management measures have come to play in transportation system operation. The regional transportation system is defined by the Commission as the functionally related surface transportation facilities, including arterial street and highway, mass transit, bicycle, and pedestrian facilities and system management measures that enable the intraregional and interregional movement of people and goods.

Basic principles one through five, as outlined in this chapter, were expounded in the initial and second-generation regional transportation planning efforts and modified only slightly, in keeping with the refined definition of the term "transportation system." Principle number six was added to the original five in recognition of the increasing importance of good transportation, including a choice among modes, to the achievement of personal and collective goals and thus to community development. Principle number seven was added to reflect the importance of functionally relating the regional transportation system to the State and national transportation systems and, consequently, the importance of coordinating planning endeavors among various levels of government.

The process applied in the earlier transportation planning efforts was again activated in the preparation of the year 2010 regional transportation system plan. That process, consisting of the seven steps of study design, formulation of objectives and standards, inventory, analyses and forecasts, plan design, plan test and evaluation, and plan selection, remains a coherent and rational approach to systemwide transportation planning. (This page intentionally left blank)

Chapter III

MONITORING AND IMPLEMENTATION ACTIVITIES RELATED TO THE ADOPTED YEAR 2000 REGIONAL TRANSPORTATION SYSTEM PLAN

INTRODUCTION

The year 2000 regional transportation system plan has functioned as an important guide to transportation system development in the Region since its adoption by the Commission in 1978. It has been amended through planning executed at the local and subregional levels under the continuing land use-transportation study. This chapter describes the year 2000 regional transportation system plan as amended, provides an evaluation of the forecasts underlying the design of the plan, and describes the progress toward plan implementation. The information presented in this chapter provides an important basis for assessing the validity of the adopted plan and for identifying needed changes.

To the extent necessary for a proper evaluation of the regional transportation plan, this chapter provides some information on the changes in trip making and traffic congestion that have occurred within the Region from 1972 through 1991. More complete presentations of the changes in trip making and traffic congestion are provided in other chapters of this report. To the extent necessary for a proper evaluation of the status of regional transportation plan implementation, this chapter also presents some information on changes in urban development that have occurred within the Region from 1972 to 1991. For a more comprehensive review and analysis of such land use changes since 1970. and hence of the implementation of the year 2000 regional land use plan, the reader should refer to SEWRPC Planning Report No. 40, A Regional Land Use Plan for Southeastern Wisconsin-2010. The reader may also wish to refer directly to SEWRPC Planning Report No. 25, in which the year 2000 regional land use plan is documented.

PLAN ADOPTION

The Regional Planning Commission formally adopted the second-generation regional transportation system plan June 1, 1978. The plan, with supporting data, analyses, forecasts, and objectives, was fully documented in SEWRPC Planning Report No. 25. The Commission acted to adopt the plan pursuant to Section 66.945(9) and (10) of the Wisconsin Statutes. This plan was certified to concerned Federal and State agencies of government and to the constituent county and local units of government for consideration, adoption, and implementation.

The Commission's adoption of the plan was preceded by an extensive series of public informational meetings and public hearings held throughout the Region. The preliminary regional transportation system plan was the subject of four such hearings held by the Commission in the fall of 1977. Prior to those hearings, the Commission prepared and widely distributed the September-October 1977 issue of the SEWRPC <u>Newsletter</u> (Vol. 17, No. 5), which presented in summary form the preliminary plan recommendations. Extensive coverage of the plan was given by the public information media, including a major two-page feature story in the Thanksgiving Day issue of The Milwaukee Journal.

An essential action before the implementation of the regional transportation system plan is the adoption of the plan by the constituent units of local government and by the governmental agencies having plan implementation authority. As of January 1, 1993, the year 2000 regional transportation system plan had been adopted by all seven County Boards within the Region and the by local governing bodies of 11 cities, seven villages, and seven towns (see Map 3). The plan was also acknowledged by such government agencies as the U.S. Department of Transportation, Federal Highway Administration and Urban Mass Transit Administration (the latter now the Federal Transit Administration), and the Wisconsin Department of Transportation.

PLAN AMENDMENTS

The second-generation regional transportation system plan, as a long-range development plan, was designed with an approximate 20-year planning horizon. This planning horizon was selected to reflect the economic life of transportation facilities proposed for construction during

ADOPTION OF THE YEAR 2000 REGIONAL TRANSPORTATION SYSTEM PLAN AND RELATED AMENDMENTS



An essential action preceding the implementation of the regional transportation system plan is the adoption of the plan by the constituent units of local government and by the governmental agencies having plan implementation authority. As of January 1, 1993, the year 2000 regional transportation system plan had been adopted by all seven county boards within the Region and by the local governing bodies of eleven cities, seven villages, and seven towns. The plan was also formally acknowledged by the U. S. Department of Transportation, Federal Highway and Urban Mass Transit Administrations, the latter now the Federal Transit Administration, and the Wisconsin Department of Transportation.

Source: SEWRPC.

the initial stages of the plan. Since the adoption of the plan design year 2000 regional transportation system plan, the continuing regional land use-transportation study has provided the means for preparing county jurisdictional highway system plan amendments and integrating these county plan amendments into the regional transportation system plan, thereby also amending the regional plan. In the operation of the subregional-to-systemwide planning cycles it became apparent that the design year of the regional plan was no longer a practical target for some of the jurisdictional highway system and corridor plans, particularly those prepared after 1985, since the economic life of the facilities included in such plans falls beyond the year 2000 design target. In and after 1985 it also became possible for the Commission to avail itself of more current land use and travel data. Consequently, the design year for the capital improvements proposed in the most recent jurisdictional highway plans for Walworth and Ozaukee Counties, proposed for the Burlington Bypass in Racine County, and for several major travel corridor plans was extended beyond the year 2000, to the year 2010.¹ These extensions of the design year were made by the Commission with the understanding that the planned improvements would be integrated into the third-generation regional transportation system plan during its preparation, provided such improvements advanced the objectives of the new regional plan.

In describing the arterial street and highway component of the adopted regional transportation system plan and in evaluating its implementation, it is therefore necessary to distinguish between the regional plan, essentially as adopted in 1978, and the regional plan as it stood amended through 1992, with some highway improvement proposals advanced to the

¹See: Amendment to the Walworth County Jurisdictional Highway System Plan-2010, March 1992; Amendment to the Ozaukee County Jurisdictional Highway System Plan-2010, January 1993; Amendment to the Racine County Jurisdictional Highway System Plan-2000 (Recommendations for the proposed City of Burlington Bypass-2010), December 1990; SEWRPC Community Assistance Planning Report No. 200 A Land Use and Transportation System Development Plan for the IH 94 South Freeway Corridor, Kenosha, Milwaukee, and Racine Counties, January 1992; SEWRPC Community Assistance Planning Report No. 151 A Transportation System Plan for the Blue Mound Road (USH 18) Corridor, December 1987. Currently under preparation by the Commission is a land use and transportation system development plan for the IH 94 West Freeway Corridor.

design year 2010. An evaluation of the implementation of the plan as it stood amended through 1992 would indicate unrealistic progress toward implementation, only a very few years having elapsed since the adoption of the latest concerned jurisdictional highway system and corridor plans, thus precluding meaningful opportunity for plan implementation. The needed implementation evaluation will therefore focus on the improvements set forth in the plan design year 2000 plan as adopted in 1978, while a description of the planned improvements will be provided as set forth under both conditions.

Plan Refinements

The Commission's continuing regional land usetransportation study has also provided the means for refining the adopted design year 2000 regional transportation system plan through the preparation and adoption of several significant plan elements (see Table 2). The description of these plan elements is intended to provide an appropriate context for understanding the true breadth of the year 2000 plan recommendations.

In 1980, pursuant to Federal regulations, the Commission adopted a transportation systems management plan element. That plan element is set forth in SEWRPC Community Assistance Planning Report No. 50, A Transportation Systems Management Plan for the Kenosha, Milwaukee, and Racine Urbanized Areas in Southeastern Wisconsin. The plan recommended the implementation of such management measures as a freeway traffic management system, improved transit services, a system of park-ride and park-pool lots, and the promotion of carpooling and vanpooling. In addition, this plan element called for specific studies for integrating the "stub ends" of uncompleted freeways into the arterial street and highway system, for the examination of the potential for work time rescheduling to reduce peak-period travel demands, for the examination of taxi fares and regulation, and for the examination of parking rate structures to discourage the use of singleoccupancy motor vehicles for commuting within the Region.

On the basis of studies derived from the transportation system management plan recommendations, the second-generation regional transportation system plan was subsequently amended to provide specifically for the integration into the arterial street network of the "stub ends" of the Lake Freeway-South, the Lake Freeway-North, the Park Freeway-East, the Stadium Freeway-South, the Stadium Freeway-North, and the Park Freeway-West. These planned subregional projects were completed between 1981 and 1985, thereby refining the adopted regional transportation system plan.

The adopted year 2000 regional transportation system plan was further refined through the completion and adoption in 1982 of SEWRPC Planning Report No. 33, <u>A Primary Transit</u> <u>System Plan for the Milwaukee Area</u>. Specific recommendations concerning the provision of rapid transit service in the greater Milwaukee area were set forth in that document. The report explicitly recognized the potential for light rail and commuter rail transit services in certain major travel corridors.

In 1983, the Commission completed SEWRPC Planning Report No. 34, A Transportation System Plan for the Milwaukee Northwest Side/ Ozaukee County Study Area. The plan identified both short-range and long-range highway and transit improvements which would be required for that portion of the Region in the absence of previously planned freeway improvements. Short-range plan recommendations included: the integration of the freeway "stub ends" at the Park Freeway-West and Stadium Freeway-North into the surface arterial system, the abatement of congestion and accident problems along 20 selected segments of the surface arterial street system, and the improvement of public transit. Long-range plan recommendations included: freeway improvements on IH 43 North, freeway traffic management improvements, surface arterial street extensions and widening projects, and major public transit improvements, including the provision of rapid transit service in the corridor either through express buses or light rail operating on exclusive transitways and reserved lanes.

In 1987, at the request of Milwaukee County, the Commission completed a major transit plan implementation study, an alternatives analysis, of the Milwaukee northwest travel corridor. That study was designed to address in detail whether the public transit service to be provided in the corridor, as recommended in the earlier plan amendment, should be provided through express bus or light rail transit. The results of this study are set forth in SEWRPC Community Assistance Planning Report No. 150, <u>A Rapid Transit</u> Facility Plan for the Milwaukee Northwest

THE EVOLVING REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN

Plan Element	Plan Document	Date of Adoption
Regional Transportation System Plan ^a	Planning Report No. 25, <u>A Regional Land Use Plan and a</u> <u>Regional Transportation Plan for Southeastern Wisconsin:</u> <u>2000</u> , Volume One, <u>Inventory Findings</u> ; Volume Two, <u>Alternative and Recommended Plans</u>	June 1, 1978
Amendment—Lake Freeway South Corridor	Amendment to the Regional Transportation Plan-2000, Lake Freeway South Corridor	June 18, 1981
Amendment Milwaukee Area Primary Transit System	Planning Report No. 33, <u>A Primary Transit</u> System Plan for the Milwaukee Area	June 17, 1982
Amendment – Racine County	Amendment to the Regional Transportation Plan-2000, Racine County	December 2, 1982
Amendment-Waukesha County	Amendment to the Regional Transportation Plan—2000, Waukesha County	December 2, 1982
Amendment—Milwaukee Northwest Side/Ozaukee County	Planning Report No. 34, <u>A Transportation System Plan for the</u> Milwaukee Northwest Side/Ozaukee County Study Area	September 8, 1983
Amendment—Lake Freeway-North/ Park Freeway-East	Amendment to the Regional Transportation Plan—2000, Lake Freeway North/Park Freeway East	December 1, 1983
Amendment—Stadium Freeway South Corridor	<u>Amendment to the Regional Transportation Plan-2000,</u> <u>Stadium Freeway South Corridor</u>	March 11, 1985
Amendment – Waukesha County	Amendment to the Regional Transportation Plan-2000, Waukesha County	June 20, 1988
Amendment Washington County	Amendment to the Washington County Jurisdictional Highway System Plan-2000	June 20, 1990
Amendment—Racine County ^b	Amendment to the Racine County Jurisdictional Highway System Plan-2000	December 5, 1990
Amendment – Kenosha County	Amendment to the Regional Transportation Plan-2000, Kenosha County	December 5, 1990
Amendment – Walworth County ^C	Amendment to the Walworth County Jurisdictional Highway System Plan-2010	March 4, 1992
Amendment-Ozaukee County ^C	Amendment to the Ozaukee County Jurisdictional Highway System Plan – 2010	January 18, 1993
Amendment—IH 94 South Corridor ^C	Community Assistance Planning Report No. 200, <u>A Land Use and Transportation System Development Plan</u> for the IH 94 South Freeway Corridor, Kenosha, Milwaukee, and Racine Counties	January 15, 1992
Racine Area Transit Development Plan	Community Assistance Planning Report No. 79, <u>Racine Area</u> Transit System Plan and Program: 1984-1 <u>988</u>	September 12, 1974
Waukesha County Transit Plan	Community Assistance Planning Report No. 105, Waukesha County Transit Plan: 1988-1992	d
City of Waukesha Transit Development Plan	Community Assistance Planning Report No. 154, <u>A Transit</u> Development Plan for the City of Waukesha: 1988-1992	June 20, 1990
City of West Bend Transit Development Plan	Community Assistance Planning Report No. 189, <u>A Transit</u> <u>Feasibility Study and Development Plan for the City of</u> <u>West Bend: 1992-1996</u>	March 4, 1992
Kenosha Area Transit Development Plan ^e	Community Assistance Planning Report No. 183, <u>Kenosha Transit System Development Plan: 1991-1995,</u> <u>City of Kenosha, Wisconsin</u>	June 17, 1992
Transportation Systems Management Plan	Community Assistance Planning Report No. 50, <u>A Transportation Systems Management Plan for the</u> <u>Kenosha, Milwaukee, and Racine Urbanized Areas in</u> <u>Southeastern Wisconsin:1981</u>	December 4, 1980
Amendment—Milwaukee Northwest Side/Ozaukee County	Planning Report No. 34, <u>A Transportation System Plan for</u> the Milwaukee Northwest Side/Ozaukee County Study Area	September 8, 1983
Amendment-Milwaukee Area	Planning Report No. 39, <u>A Freeway Traffic</u> <u>Management System Plan for the Milwaukee Area</u>	December 5, 1988

Table 2 (continued)

Plan Element	Plan Document	Date of Adoption
Elderly-Handicapped Transportation Plan ^f	Planning Report No. 31, <u>A Regional Transportation Plan</u> for the Transportation Handicapped in Southeastern Wisconsin: 1978-1982	April 13, 1978
Amendment-Racine Area	SEWRPC Resolution No. 78-17	December 7, 1978
Amendment - Milwaukee County	Memorandum Report No. 58, <u>A Paratransit Service Plan</u> for Disabled Persons—Milwaukee County Transit System	January 15, 1992
Amendment – Kenosha Area	Memorandum Report No. 59, <u>Paratransit Service Plan</u> for Disabled Persons—City of Kenosha Transit System	January 15, 1992
Amendment—Racine Area	Memorandum Report No. 60, <u>Paratransit Service Plan</u> for Disabled Persons—City of Racine Transit System	January 15, 1992
Amendment-City of Waukesha	Memorandum Report No. 61, <u>Paratransit Service Plan for</u> <u>Disabled Persons—City of Waukesha Transit System Utility</u>	January 15, 1992
AmendmentWaukesha County	Memorandum Report No. 62, <u>Paratransit Service Plan</u> for Disabled Persons—Waukesha County Transit System	January 15, 1992

^a The regional transportation plan is a second-generation plan. The initial regional transportation plan was adopted by the Commission on December 1, 1966, and documented in SEWRPC Planning Report No. 7, <u>Land Use-Transportation Study</u>, Volume Three, <u>Recommended Regional Land Use</u> <u>and Transportation Plans – 1990</u>, and was subsequently amended by the adoption on June 4, 1970, of the Milwaukee County jurisdictional highway system plan documented in SEWRPC Planning Report No. 11, <u>Jurisdictional Highway System Plan for Milwaukee County</u>; the adoption on March 2, 1972, of the Milwaukee area transit plan set forth in the document entitled <u>Milwaukee Area Transit Plan</u>; the adoption on March 4, 1973, of the Walworth County jurisdictional highway system plan documented in SEWRPC Planning Report No. 15, <u>Jurisdictional Highway System Plan for Walworth County</u>; the adoption on March 7, 1974, of the Ozaukee County jurisdictional highway system plan documented in SEWRPC Planning Report No. 17, <u>Jurisdictional Highway System Plan for Ozaukee County</u>; the adoption on June 5, 1975, of the Waukesha County jurisdictional highway system plan documented in SEWRPC Planning Report No. 18, <u>Jurisdictional Highway System Plan for Waukesha County</u>; the adoption on September 11, 1975, of the Washington County jurisdictional highway system plan documented in SEWRPC Planning Report No. 23, <u>Jurisdictional Highway System Plan for Washington County</u>; the adoption on September 11, 1975, of the Kashington County; indictional highway system plan documented in SEWRPC Planning Report No. 23, <u>Jurisdictional Highway System Plan for Washington County</u>; the adoption on September 11, 1975, of the Racine County jurisdictional highway system plan documented in SEWRPC Planning Report No. 24, <u>Jurisdictional Highway System Plan for Kenosha County</u>; and the adoption on December 4, 1975, of the Racine County jurisdictional highway system plan documented in SEWRPC Planning Report No. 22, <u>Jurisdictional Highway System Plan for Kenosha County</u>; and th

^bThe design year for the proposed City of Burlington Bypass included in the <u>Amendment to the Racine County Jurisdictional Highway System</u> <u>Plan—2000</u>, adopted in December 1990, was advanced to the design year 2010 with the understanding that this arterial street and highway improvement would be integrated into the third-generation transportation system plan during the preparation of the plan.

^CThe design year of the proposed arterial street and highway system improvements was advanced to the design year 2010 with the understanding that such improvements would be integrated into the third-generation regional transportation system plan during the preparation of the plan.

^dThe <u>Waukesha County Transit Plan: 1988-1992</u> was completed by the Commission staff in September 1988 but neither the Waukesha County Mass Transit Advisory Committee nor the Regional Planning Commission moved to adopt the plan.

^eThe Kenosha area transit development plan is a third-generation plan. The initial plan was adopted by the Commission on June 3, 1976, and documented in SEWRPC Community Assistance Planning Report No. 7, <u>Kenosha Area Transit Development Plan: 1976-1980</u>. The second-generation plan was adopted on March 11, 1985, and documented in SEWRPC Community Assistance Planning Report No. 101, <u>Kenosha Area Transit System Plan and Program: 1984-1988</u>.

^f The 1992 amendments to the 1978 elderly-handicapped transportation plan supersede a series of earlier amendments to the 1978 plan. These earlier amendments are as follows: 1) an amendment adopted by the Commission on June 20, 1980, and documented in SEWRPC Community Assistance Planning Report No. 39, <u>A Public Transit System Accessibility Plan, Volume Two, Milwaukee Urbanized Area/Milwaukee County;</u> 2) three amendments adopted by the Commission on September 11, 1980, and documented in SEWRPC Community Assistance Planning Report No. 39, <u>A Public Transit System Accessibility Plan</u>, respectively, in Volume One, <u>Kenosha Urbanized Area</u>; Volume Three, <u>Racine Urbanized Area</u>; and Volume Four, <u>Milwaukee Urbanized Area/Waukesha County;</u> 3) an amendment adopted by the Commission on June 18, 1981, and documented in the <u>Amendment to the Public Transit Accessibility Plan</u> for the Milwaukee Urbanized Area/Waukesha Transit System; 4) five amendments adopted by the Commission on December 7, 1987, and documented, respectively, in SEWRPC Memorandum Report No. 17, <u>A Public Transit Program for Handicapped Persons – City of Waukesha Transit System</u>; SEWRPC Memorandum Report No. 22, <u>A Public Transit Program for Handicapped Persons – City of Waukesha</u>; SEWRPC Memorandum Report No. 22, <u>A Public Transit Program for Handicapped Persons – City of Severe</u>; SEWRPC Memorandum Report No. 22, <u>A Public Transit Program for Handicapped Persons – City of Waukesha</u>; SEWRPC Memorandum Report No. 22, <u>A Public Transit Program for Handicapped Persons – City of Waukesha</u>; SEWRPC Memorandum Report No. 23, <u>A Public Transit Program for Handicapped Persons – City of Kenosha</u>; SEWRPC Memorandum Report No. 23, <u>A Public Transit Program for Handicapped Persons – City of Kenosha Transit System</u>; and SEWRPC Memorandum Report No. 23, <u>A Public Transit Program for Handicapped Persons – City of Kenosha Transit System</u>; and SEWRPC Memorandum Report No. 23, <u>A Public Transit Program for Handicapped Persons – City of Kenosha Transit System</u>; and SEWRP

Source: SEWRPC.

Component Area	Report Document	Date Completed
Transportation System Management	Technical Report No. 20, <u>Carpooling in</u> <u>the Metropolitan Milwaukee Area</u> ^a	March 1977
Public Transit	Technical Report No. 23, <u>Transit Related</u> <u>Socioeconomic, Land Use, and Transportation</u> <u>Conditions and Trends in the Milwaukee Area</u>	December 1980
Public Transit	Technical Report No. 24, <u>State-of-the-Art</u> of Primary Transit System Technology	February 1981
Transportation System Management	Technical Report No. 27, <u>Milwaukee Area</u> <u>Work Time Rescheduling Study</u>	August 1981
Public Transit	Technical Report No. 26, <u>Milwaukee Area</u> <u>Alternative Primary Transit System Plan</u> <u>Preparation, Test, and Evaluation</u>	March 1982
Transportation System Management	Technical Report No. 28, <u>Evaluation of the</u> <u>Milwaukee Area Rideshare Program: 1979-1982</u>	May 1983

TECHNICAL REPORTS PREPARED IN SUPPORT OF THE ADOPTED PLAN 2000 AS AMENDED

^aAn evaluation of the effectiveness of the Metropolitan Milwaukee Area Carpooling Program in achieving its stated objectives conducted concurrently with the preparation of the year 2000 regional transportation system plan.

Source: SEWRPC.

<u>Corridor</u>. On the basis of the study findings, the Milwaukee County Board and County Executive selected the express bus alternative as the preferred method of providing rapid transit service in the northwest travel corridor. Implementation of the selected alternative was then undertaken.

Within the framework of the second-generation regional transportation plan, more detailed short-range transit development plans have also been completed, adopted, and kept up to date for the Kenosha, Racine, and Waukesha urban areas. In addition, a new short-range transit development plan has recently been completed and adopted for the West Bend urban area.

Finally, the Commission also has completed and adopted a transportation plan for elderly and disabled persons. The original plan element, adopted in 1978, is updated periodically to maintain consistency with changing Federal laws and regulations. In 1992, the elderly and handicapped transportation plan, as it applies to the public transit service areas of the Region, was updated through the preparation and adoption of paratransit service plans. These service plans are intended to meet the requirements of the Federal Americans with Disabilities Act.

Plan-Related Technical Studies

The Commission also completed a series of studies since the adoption of the secondgeneration regional transportation system plan which are intended to provide information pertinent to the preparation of plans aimed at refining the adopted year 2000 plan. These studies, documented in SEWRPC Technical Reports, are listed in Table 3. The first such study, completed in December 1980, is documented in Technical Report No. 23, Transit-Related Socioeconomic, Land Use, and Transportation Conditions and Trends in the Milwaukee Area. That study and a companion study documented in Technical Report No. 24, State-of-the-Art of Primary Transit System Technology, combined to provide the information needed by the Commission in preparing the primary transit system plan for the Milwaukee Area. In August of 1981, the Commission completed Technical Report No. 27, Milwaukee Area Work Time Rescheduling Study. The study concerned found that the potential to reduce peak-period congestion was significant if a substantial proportion of work-related travel could be shifted from peak morning and afternoon travel hours. The study also found that the practicability of achieving the shift in work-related travel was questionable. Accordingly, the study recommended that no areawide work time rescheduling effort be undertaken at that time.

PLAN DESCRIPTION: ADOPTED REGIONAL TRANSPORTATION SYSTEM PLAN

The adopted year 2000 regional transportation system plan consisted of three functionally related components: arterial streets and highways, including freeways; public transit facilities and services; and transportation system management measures.

Conceptually, the plan was designed to alleviate the need for expanded highway facilities by reducing automobile travel through expanded public transit and transportation system management measures. The management measures were designed to make more effective use of existing capacity before considering commitments to expanding arterial street and highway system capacity. Only the residual travel demand not captured by improved public transit and transportation system management measures would be addressed through arterial street and highway capacity improvements. The three components of the plan together would serve to alleviate existing and future traffic congestion on the arterial street and highway system, provide a more modally balanced and environmentally sound transportation system, and efficiently and effectively serve the land use pattern envisioned in the adopted regional land use plan. A description of the recommendations related to each system component follows.

Arterial Street and Highway System

The arterial street and highway system consists of surface arterials and freeways. Surface arterials generally permit marginal access to abutting land uses, although such access may be limited, and generally have at-grade intersections with other arterial and nonarterial roadways. Freeways are high-capacity, high-speed, divided arterial facilities and have complete control of access, with access provided only at grade-separated interchanges. The description of the arterial street and highway system component of the regional transportation system plan is intended to serve two purposes:

- 1. To identify the recommendations of the plan essentially as adopted in 1978, thereby providing the basis for later evaluating the implementation of the plan and
- 2. To present the most current plan recommendations coming out of recent county jurisdictional highway system plan amendments.

The arterial street and highway element of the adopted year 2000 regional transportation system plan recommended that the State, county, and local highway agencies jurisdictionally realign and functionally improve the arterial street and highway system in the Region. The planned functional and jurisdictional arterial street and highway system, as recommended in the adopted year 2000 plans,² is set forth in Table 4 and shown on Map 4. The adopted plan envisioned a regional arterial street and highway system totaling about 3,578 miles by the year 2000.

Jurisdictional Realignments: The jurisdictional realignment proposals were intended to achieve an equitable distribution of arterial street and highway development and maintenance costs among the various levels and agencies of government concerned. Criteria for recommending jurisdictional realignments were related to the

²See SEWRPC Planning Report No. 25, A Regional Land Use Plan and a Regional Transportation Plan for Southeastern Wisconsin: 2000, as well as the seven-county jurisdictional highway system plans set forth in SEWRPC Planning Reports No. 11, A Jurisdictional Highway System Plan for Milwaukee County; No. 15, A Jurisdictional Highway System Plan for Walworth County; No. 17, A Jurisdictional Highway System Plan for Ozaukee County; No. 18, A Jurisdictional Highway System Plan for Waukesha County; No. 22, A Jurisdictional Highway System Plan for Racine County; No. 23, A Jurisdictional Highway System Plan for Washington County; and No. 24, A Jurisdictional Highway System Plan for Kenosha County.

	Sta	ite	County		Local		To	otal
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent
Kenosha	112.2	31.6	198.5	56.0	44.0	12.4	354.7	100.0
Milwaukee	247.0	32.0	195.0	25.3	329.0	42.7	771.0	100.0
Ozaukee	89.5	29.1	165.7	53.9	52.0	17.0	307.2	100.0
Racine	178.8	41.5	173.2	40.2	79.0	18.3	431.0	100.0
Walworth	224.2	46.4	244.3	50.6	14.7	3.0	483.2	100.0
Washington	158.6	33.4	239.7	50.5	76.1	16.1	474.4	100.0
Waukesha	223.3	29.5	425.9	56.3	107.7	14.2	756.9	100.0
Region	1,233.6	34.5	1,642.3	45.9	702.5	19.6	3,578.4	100.0

PLANNED DISTRIBUTION OF ARTERIAL STREET AND HIGHWAY MILEAGE IN THE REGION BY JURISDICTIONAL CLASSIFICATION: 2000^a

^aAdopted Year 2000 Regional Transportation System Plan, as amended by the removal of the Stadium Freeway-South, Lake Freeway-North, Park Freeway-East, and by the substitution of the Lake Arterial for the Lake Freeway-South; and prior to its amendment by the following: <u>Amendment to the Walworth County Jurisdictional Highway System</u> <u>Plan-2010</u>, March 1992; <u>Amendment to the Ozaukee County Jurisdictional Highway System Plan-2010</u>, January 1993; <u>Amendment to the Racine County Jurisdictional Highway System Plan-2000</u> (Recommendations for the proposed City of Burlington Bypass-2010), December 1990; SEWRPC Community Assistance Planning Report No. 200, <u>A Land</u> <u>Use and Transportation System Development Plan for the IH 94 South Freeway Corridor, Kenosha, Milwaukee, and Racine Counties</u>, January 1992; SEWRPC Community Assistance Planning Report No. 151, <u>A Transportation System Plan</u> for the Blue Mound Road (USH 18) Corridor, December 1987.

Source: SEWRPC.

kinds of trips served by, and the operational characteristics of, the facilities. Applying such criteria, the plan recommended that State trunk highways serve the longest trips, access the most important land uses, and carry the highest traffic volumes. Concomitantly, the plan recommended that county trunk highways serve trips of intermediate length and carry intermediate traffic volumes, and that local trunk arterials serve the shortest trips, access locally-oriented land uses, and carry the lightest traffic volumes.

The jurisdictional realignment element of the plan recommended abolishing the "connecting street" concept³ with respect to State trunk highways, thus providing for continuous State trunk highways, with attendant maintenance and traffic control responsibilities, through incorporated urban areas. Similarly, the plan recommended that county trunk highways penetrate incorporated communities and thereby provide continuous routes through those communities.

The recommended distribution of the arterial street and highway mileage among the three jurisdictional classifications is set forth in Table 4. Under the plan, State trunk highways would total 1,234 miles, or about 34 percent, of the planned regional arterial system. County trunk highways would total about 1,642 miles, or an additional 46 percent, of the regional system. Local arterial streets and highways would constitute the remaining 702 miles, or about 20 percent, of the regional system. This recommended distribution of arterial street and highway mileage would be achieved through the planned transfer of arterial miles between the jurisdictional classifications as set forth in Table 5. The adopted plan recommended the following with regard to jurisdictional realignments:

1. That approximately 123 miles be transferred to the State trunk system from the county trunk system and the local arterial system.

³Connecting streets and highways are defined as the marked routes of State trunk highways over the municipal streets and highways where the municipality has the responsibility for maintenance and operation.

PLANNED FUNCTIONAL AND JURISDICTIONAL ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION: ADOPTED PLAN 2000 AS AMENDED



The design year 2000 regional transportation system plan recommended that the State, county, and local highway agencies jurisdictionally realign and functionally improve the arterial streets and highways within the Region. The adopted plan envisioned a regional arterial street and highway system that would total about 3,574 miles by the year 2000.

	County							Begion
Type of Transfer	Kenosha	Milwaukee	Ozaukee	Racine	Walworth	Washington	Waukesha	Total
Transfers to State Trunk System From County Trunk System From Local System	6.5 0.8	11.0 10.2	4.3 4.0	28.2 4.4	19.0 2.6	8.0 4.2	18.9 1.2	95.9 27.4
Total	7.3	21.2	8.3	32.6	21.6	12.2	20.1	123.3
Transfers to County Trunk System From State Trunk System From Local System	34.7 20.0	54.7 86.4	86.4 29.5	36.7 37.3	30.3 59.2	46.9 53.9	53.4 78.6	343.1 364.9
Total	54.7	141.1	115.9	74.0	89.5	100.8	132.0	708.0
Transfers to Local Arterial System From State Trunk System From County Trunk System	0.0 13.0	12.8 15.1	0.0 2.9	4.4 11.3	1.2 1.0	8.7 5.4	18.5 91.6	45.6 140.3
Total	13.0	27.9	2.9	15.7	2.2	14.1	110.1	185.9
Transfers to Local Nonarterial System From State Trunk System From County Trunk System	0.0 97.2	1.1 3.3	0.0 13.7 ·	2.1 21.4	9.0 26.6	3.7 52.8	4.1 134.5	20.0 349.5
Total	97.2	4.4	13.7	23.5	35.6	56.5	138.6	369.5

PLANNED MILES OF JURISDICTIONAL REALIGNMENTS IN THE REGION BY COUNTY: 2000

Source: SEWRPC.

- 2. That approximately 708 miles be transferred to the county trunk system from the State trunk system and the local arterial system.
- 3. That approximately 186 miles be transferred to the local arterial system from the State trunk system and the county trunk system.
- 4. That approximately 370 miles be transferred to the local nonarterial system from the State trunk system and the county truck system.

<u>Functional Improvements</u>: The year 2000 adopted regional transportation system plan recommended three types of functional improvements to the arterial street and highway system: system preservation, consisting of the resurfacing and reconstruction necessary to properly maintain existing arterial roadways; system improvement, consisting of the widening of existing facilities to provide additional traffic capacity; and system expansion, consisting of the construction of new arterial facilities. The functional improvements as set forth in the adopted regional plan are summarized in Table 6 and shown on Map 5. As indicated in Table 6, about 2,941 miles, or 82 percent, of the total arterial street and highway system would require only preservation; about 401 miles, or about 11 percent, would require improvement; and about 236 miles, or about 7 percent would constitute new facilities. The most significant planned surface arterial and freeway improvements are listed below.

Surface Arterial Improvements: The major surface arterial improvements recommended under the plan included, among others, the following: in Kenosha County, the construction of the Lake Arterial from the Racine County line to the Wisconsin-Illinois state line, the improvement of STH 165 from its present four lane section to STH 32, and the improvement of STH 31 from CTH S to the Racine County line; in Milwaukee County, the construction of the Lake Arterial from Carferry Drive to the Racine County line and the improvement of Rawson Avenue from about S. 60th Street to USH 41; in Ozaukee County, the improvement of CTH N/ Wauwatosa Road from STH 167 to STH 60; in Racine County, the construction of the Lake Arterial from the Milwaukee County line to the Kenosha County line and the improvement of STH 36 from the City of Burlington to Waukesha County; in Walworth County, the construc-

PLANNED MILES OF FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY JURISDICTIONAL SYSTEM AND COUNTY: 2000^a

Planned Miles of Eurocional Improvement										
	State Trunk Highways ^b County Trunk Highways									
County	Preservation	Improvement	Expansion	Preservation	Improvement	Expansion				
Kenosha	58.6	36.9	16.7	165.0	26.0	7.5				
Milwaukee	200.6	31.1	15.3	158.2	33.8	3.0				
Ozaukee	49.2	21.4	18.9	163.7	2.0	0.0				
Racine	125.6	36.8	16.4	149.5	16.0	7.7				
Walworth	162.5	13.7	48.0	235.8	4.6	3.9				
Washington	90.5	48.2	19.9	224.1	4.4	11.2				
Waukesha	157.1	49.0	17.2	377.0	36.4	12.5				
Region	844.1	237.1	152.4	1,473.3	123.2	45.8				

Planned Miles of Functional Improvement									
		Local Arterials		Ent	ire Arterial Syste	em	Miles of Functional		
County	Preservation	Improvement	Expansion	Preservation	Improvement	Expansion	Improvement		
Kenosha	37.9	2.4	3.7	261.5	65.3	27.9	354.7		
Milwaukee	300.5	23.5	5.0	659.3	88.4	23.3	771.0		
Ozaukee	48.0	0.0	4.0	260.9	23.4	22.9	307.2		
Racine	67.3	6.6	5.1	342.4	59.4	29.2	431.0		
Walworth	5.0	0.4	9.3	403.3	18.7	61.2	483.2		
Washington	66.5	0.3	9.3	381.1	52.9	40.4	474.4		
Waukesha	98.8	7.2	1.7	632.9	92.6	31.4	756.9		
Region	624.0	40.4	38.1	2,941.4	400.7	236.3	3,578.4		

^aAdopted Year 2000 Regional Transportation System Plan, as amended by the removal of the Stadium Freeway-South, Lake Freeway-North, Park Freeway-East, and by the substitution of the Lake Arterial for the Lake Freeway-South; and prior to its amendment by the following: <u>Amendment to the Walworth County Jurisdictional Highway System Plan – 2010</u>, March 1992; <u>Amendment to the Ozaukee County Jurisdictional Highway System Plan – 2010</u>, January 1993; <u>Amendment to the Racine County Jurisdictional Highway System Plan – 2000</u> (Recommendations for the proposed City of Burlington Bypass – 2010), December 1990; SEWRPC Community Assistance Planning Report No. 200, <u>A Land Use and Transportation System</u> <u>Development Plan for the IH 94 South Freeway Corridor, Kenosha, Milwaukee, and Racine Counties</u>, January 1992; SEWRPC Community Assistance Planning Report No. 151, <u>A Transportation System Plan for the Blue Mound Road (USH 18) Corridor</u>, December 1987.

^bState trunk highways include both freeways and surface arterials.

Source: SEWRPC.

tion of STH 120 bypass from CTH BB to STH 36; in Washington County, the improvement of USH 45 from the Village of Kewaskum to the City of West Bend and the improvement of STH 33 from USH 41 to the City of West Bend; in Waukesha County, the construction of the City of Waukesha Bypass from STH 59 to IH 94, the construction of the CTH SS extension from CTH G to CTH T, and the improvement of STH 59 from Milwaukee County to STH 164. <u>Freeway Improvements</u>: The freeway improvements recommended under the adopted plan included the West Bend Freeway (USH 45); the conversion of USH 41 in Washington County from an expressway to a freeway; the completion of the STH 16 freeway to, and bypass around, the City of Oconomowoc; the extension of USH 12 northerly through Walworth County, with a bypass of the City of Whitewater and an eventual connection outside the Region with



PLANNED FUNCTIONAL **IMPROVEMENTS TO THE** ARTERIAL STREET AND HIGHWAY PLAN 2000 AS AMENDED



The design year 2000 regional transportation system plan recommended three types of functional improvements to arterial streets and highways within the Region: system preservation, system improvement, and system expansion. The plan called for the preservation of about 2,941 miles, or 82 percent, of the total planned system; for the improvement of about 401 miles, or 11 percent, of the planned system; and for the construction of 236 miles of additional arterial facilities.

	Sta	State County		L	ocal	Tot	Total	
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent
Kenosha	114.9	32.5	195.0	55.2	43.1	12.3	353.0	100.0
Milwaukee	247.0	32.0	195.0	25.3	329.0	42.7	771.0	100.0
Ozaukee	80.1	25.5	161.5	51.4	72.4	23.1	314.0	100.0
Racine	156.6	35.7	177.6	40.5	104.1	23.8	438.3	100.0
Walworth	223.0	46.0	239.8	49.5	22.1	4.5	484.9	100.0
Washington	158.6	33.4	239.7	50.5	76.1	16.1	474.4	100.0
Waukesha	223.3	29.5	425.9	56.3	107.7	14.2	756.9	100.0
Region	1,203.7	33.5	1,634.5	45.5	754.5	21.0	3,592.5	100.0

PLANNED DISTRIBUTION OF ARTERIAL STREET AND HIGHWAY MILEAGE IN THE REGION BY JURISDICTIONAL CLASSIFICATION: 2010

Source: SEWRPC.

IH 90-94; and the widening of IH 43 from the Silver Spring interchange to Bender Road. New freeway interchanges recommended in the plan included: Highland Road on IH 43 in the City of Mequon and Drexel Avenue and Puetz Road on IH 94 in the City of Oak Creek.

The freeway improvements proposed are relatively modest compared to the improvements contained in the initial regional transportation system plan adopted in 1966. Approximately 141 miles of the once-proposed 436 mile regional freeway system were eliminated or converted to surface arterial facilities in the reappraisal of that initial transportation system plan or under amendments to the year 2000 plan. The adopted plan proposes to carry the traffic that the eliminated freeways would have carried on surface arterials and to reduce vehicular travel through both transportation system management measures and improved and expanded public transit service.

Arterial Street and Highway System-2010

As previously indicated, in the recent preparation of amendments to three county jurisdictional highway system plans, the Commission advanced the design year for highway facility improvements to 2010. The extension of the design year beyond 2000 was dictated by the needs of long-range planning. It is important to note, however, that the improvements proposed in the concerned county jurisdictional highway system plan amendments were consistent with the objectives and standards underlying the adopted year 2000 plan. That plan was the prevailing guide to regional transportation system development in Southeastern Wisconsin.

The planned functional and jurisdictional arterial street and highway system, as defined by the most recent amendments to the county jurisdictional highway system improvements, is set forth in Table 7. In its current form, the arterial street and highway component of the adopted plan envisions a regional arterial street and highway system totalling nearly 3,593 miles. The functional improvements currently proposed to be made to the arterial street and highway system are set forth in Table 8.

Public Transit

The adopted regional transportation system plan recommended improvements to, and expansion of, the urban fixed-route public transit system in the Region (see Table 9 and Map 6). The plan envisioned that by the year 2000 public transit service would consist of a fleet of 1,104 transit vehicles operating over 2,940 round-trip route miles. In 1972, the base year for the plan, the total public transit vehicle fleet consisted of 565 buses operating over 1,201 round-trip route miles.

With the planned service improvements, it was envisioned that the public transit share of total trips made within the Region could be increased from 4 percent in 1972 to about 6 percent in 2000.

Recommendations were provided in the plan for three levels of public transit facilities and services: rapid, express, and local.

PLANNED MILES OF FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY JURISDICTIONAL SYSTEM AND COUNTY: 2010

Planned Miles of Functional Improvement										
	State	e Trunk Highways	sa	Cou	nty Trunk Highwa	ays				
County	Preservation	Improvement	Expansion	Preservation	Improvement	Expansion				
Kenosha	76.4	32.8	5.7	167.2	19.6	8.2				
Milwaukee	200.6	31.1	15.3	158.2	33.8	3.0				
Ozaukee	39.4	40.7	0.0	151.2	8.3	2.0				
Racine	101.7	29.4	25.5	157.1	18.5	2.0				
Walworth	157.9	39.1	26.0	233.1	4.2	2.5				
Washington	103.6	48.2	6.8	225.0	3.9	10.8				
Waukesha	157.1	12.5								
Region	836.7	270.3	96.5	1,468.8	124.7	41.0				

Planned Miles of Functional Improvement									
		Local Arterials		Enti	Entire Arterial System				
County	Preservation	Improvement	Expansion	Preservation	Improvement	Expansion	Improvement		
Kenosha	37.9	2.4	2.8	281.5	54.8	16.7	353.0		
Milwaukee	300.5	23.5	5.0	659.3	88.4	23.3	771.0		
Ozaukee	66.6	0.5	5.3	257.2	49.5	7.3	314.0		
Racine	93.8	2.7	7.6	352.6	50.6	35.1	438.3		
Walworth	16.6	0.2	5.3	407.6	43.5	33.8	484.9		
Washington	67.0	0.0	9.1	395.6	52.1	26.7	474.4		
Waukesha	98.8	7.2	1.7	632.1	92.6	31.4	756.9		
Region	681.2	36.5	36.8	2,985.9	431.5	174.3	3,592.5		

^aState trunk highways include both freeways and surface arterials.

Source: SEWRPC.

Table 9

OPERATING CHARACTERISTICS OF PUBLIC TRANSIT SERVICE IN THE REGION BY URBANIZED AREA: 1972 AND ADOPTED 2000 PLAN

Transit Service Characteristic	Kenosha Urbanized Area	Milwaukee Urbanized Area	Racine Urbanized Area	Total Region
Existing 1972 Round-Trip Route Miles	59	1,061	81	1,201
Peak-Period Vehicle Requirements	12	442	10	464
Total Fleet (including spares)	18	533	14	565
Adopted Year 2000 Plan Round-Trip Route Miles	147	2,641	153	2,941
Peak-Period Vehicle Requirements Buses Light Rail Commuter Rail	42 	872 24 27	38 	952 24 27
Subtotal	42	923	38	1,003
Total Fleet (including spares)	46	1,016	42	1,104

Source: SEWRPC.



The year 2000 regional transportation system plan recommends a system of about 56 miles of rapid transit routes for motor bus and light rail operations. The plan recommends the development of three commuter rail lines on a demonstration basis including one line from downtown Milwaukee via the Cities of St. Francis, Cudahy, South Milwaukee, Oak Creek, Racine, and Kenosha into the Chicago area and the significant expansion into all seven counties, and improvement of, freeway flyer service. The year 2000 plan also recommends the extension of local transit service throughout the greater Milwaukee area and within the Kenosha, Racine, and Waukesha urbanized areas to serve existing and planned high- and medium-density urban development.

- Rapid transit service is intended to facilitate relatively fast and convenient transportation along heavily traveled corridors and between major activity centers and highdensity residential communities. Rapid transit service is characterized by relatively high average operating speeds and relatively low accessibility, with stops or station spacings located one-half mile or more apart. Rapid transit service can be provided by commuter rail and "heavy" rail trains operating over exclusive, grade-separated rights-of-way, or by buses operating over exclusive, grade-separated busways. Rapid transit service can also be provided by buses operating in mixed traffic on operationally controlled freeways and by "light" rail vehicles operating over exclusive, though not necessary fully grade-separated, rights-of-way.
- Express transit service can be provided by buses and "light" rail vehicles operating in mixed traffic or over reserved lanes of arterial streets and highways and in mixed traffic over freeways, with stops generally located less than 1,200 feet apart at intersecting transit routes, intersecting arterial streets, and major traffic generators. Express transit is intended to serve moderate-length trips. Express transit service provides a greater degree of accessibility at somewhat slower operating speeds than rapid transit, and may provide "feeder" service to the rapid transit system.
- Local transit service is characterized by a high degree of accessibility and low operating speeds. Local service can be provided by buses or "light" rail vehicles operating in mixed traffic over arterial and collector streets, with stops located no more than 1,200 feet apart. Such service can be provided by motor buses, electric trackless trolley buses, or "light" rail vehicles.

The public transit recommendations are shown in graphic summary form on Map 6 and are as follows:

Rapid Transit Service, "Light" Rail: With the demapping of the previously proposed Park Freeway West and Stadium Freeway North, the plan recommended that, subject to confirmation in a detailed corridor study meeting Federal requirements attendant to major transit investments, a high level of transit service in the Milwaukee northwest travel corridor be provided by bus or "light" rail that would extend from downtown Milwaukee to the Northridge Shopping Center. Within downtown Milwaukee, the service would operate over Wisconsin Avenue, which the plan recommended be converted to a transit mall between N. 10th Street and N. Prospect Avenue. This mall could serve both "light" rail vehicle and buses.

Rapid Transit Service, Bus on Freeways: The plan recommended that existing "freeway flyer" rapid transit and express bus service in the Milwaukee area be expanded and improved, with service extended to the south to the Racine and Kenosha areas, to the southwest to Mukwonago and East Troy, to the northwest to West Bend, and to the north to Port Washington. Throughout the Milwaukee urban area, the plan recommended that bus-on freeway service be expanded to become all-day service rather than peakperiod only service. In all, the plan called for 24 bus-on freeway routes, totaling 1,057 round-trip route miles in length. The system would be served by 56 transit stations, 50 of which would have park-ride lots. Given a companion recommendation, discussed below, to institute a comprehensive freeway traffic management system, the plan envisioned that express bus service over freeways would provide truly rapid transit service providing a competitive alternative to the automobile. The plan also envisioned a high level of coordination between the rapid transit and express bus service and feeder and local bus services in order to provide access to major travel generators other than the Milwaukee central business district.

The plan also recognized that it might be desirable to institute "light" rail rapid transit service eventually in certain additional corridors within Milwaukee County in addition to the northwest corridor. Five additional "light" rail service corridors were identified: 1) downtown Milwaukee to the Mitchell Field/Cudahy area, 2) downtown Milwaukee to the Southridge Shopping Center area, 3) downtown Milwaukee to the Milwaukee County Institutions and Mayfair Shopping Center area, 4) downtown Milwaukee to the Timmerman Field area, and 5) downtown Milwaukee to the University of Wisconsin-Milwaukee/Shorewood area. The plan recommended that in-depth corridor studies be undertaken to examine the feasibility and desirability of "light" rail service in lieu of bus service within these corridors.

Rapid Transit Service, Commuter Rail: The plan recommended that commuter railway service be provided on a trial, or demonstration, basis. Three specific routes were identified in the plan as good candidates for such service: downtown Milwaukee to Grafton, over former Milwaukee Road trackage; downtown Milwaukee to Oconomowoc, also over former Milwaukee Road trackage; and downtown Milwaukee to Racine and Kenosha, primarily along the Chicago & North Western Transportation Company trackage. Of the three, the potential service to Racine and Kenosha was found to hold the most promise for a successful demonstration.

Express and Local Bus Service: The plan recommended that express and local bus service be provided throughout the Milwaukee urbanized area and in the Racine and Kenosha urbanized areas as well. Within the Milwaukee area, the plan identified certain express, or limited-stop, routes operating over arterial streets. The plan recommended that the public transit operators undertake detailed short-range transit planning that would identify the specific local service routes to be provided. In total, public transit service would be extended in such a way as to provide service to an aggregate of about 500 square miles, serving about 75 percent of the regional population.

Transportation System Management

The transportation system management element of the adopted year 2000 regional transportation system plan was aimed at reducing singleoccupant vehicle use by providing incentives for the use of ridesharing and public transit and increasing the capacity of existing arterial streets at low capital cost with minimal disruption of traffic. The proposals related to transportation management are summarized below.

1. The expansion of freeway traffic management to improve freeway traffic flow by encouraging ridesharing and transit use, to reduce congestion due to accidents and disabled vehicles, to encourage motorists to make better route decisions through advisory information, and to control peaks in freeway demand. The proposed freeway traffic management system would reduce travel time, vehicle operating costs, and traffic incidents on the freeways.

Elements of the proposed freeway traffic management system include: incident management to better detect, confirm, and remove incidents; advisory information to provide pertinent information to freeway users; and areawide freeway operational control through the use of centrally monitored and regulated freeway on-ramp meters. Systemwide freeway operational control would permit on-ramp metering rates to be determined by a centralized freeway flow monitoring system enabling freeway operation speeds to be maintained at 35 to 45 miles per hour. The proposed freeway operational control would provide high-occupancy vehicles exclusive bypass of on-ramp meters to encourage ridesharing and transit use.

- 2. The establishment of carpool parking lots in outlying parts of the Milwaukee metropolitan area as demand warranted. Six possible highway interchange locations were identified. In 1978, the year the plan was adopted, there were six such lots in the Region. The plan also recommended an aggressive program to promote carpooling in the Milwaukee area.
- 3. The conduct of studies to determine the extent to which existing automobile parking policies influence the choice of travel mode to downtown Milwaukee. The plan proposed reversing the current parking rate structure, which encourages long-term parking for work-related trip making. The plan also proposed providing preferential parking in facilities for high-occupancy vehicles to promote ridesharing.
- 4. The prohibition of curb parking, as necessary, on arterial streets during peak hours of travel to ensure that arterial street capacity was effectively used before commitments were made to improve arterial streets. Over all, the plan recommended that curb parking restrictions be eventually imposed on about 600 miles of arterial streets and highways, or about 17 percent of the total planned arterial street and highway system.

5. The use of traffic engineering and access management to ensure that available arterial street capacity was effectively utilized before commitments were made to improve surface arterials. Proposed measures included: traffic signalization, rightand left-turn lanes at intersections, and driveway and local street spacing.

PLAN DESIGN: SOCIO-ECONOMIC FORECASTS

The adopted regional transportation system plan proceeded from the design of the year 2000 regional land use plan. In the design of the land use plan, forecasts of population, households, and employment were developed and used to determine the required scale of future urban development. The amount of urban land determined to be needed to accommodate forecast population and economic activity levels was then distributed to U. S. Public Land Survey quarter-sections⁴ to achieve the planned centralized urban development pattern.

The same general forecasts employed in designing the regional land use plan, population, households, employment, and the distribution of urban development, as well the forecast of automobile availability, were used as inputs into the trip generation model⁵ to determine the level of trip making the land use plan would foster within and between various subareas of the Region. The subareas, traffic analysis zones, are aggregations of the quarter-sections used in the design of the regional land use plan. The travel simulation procedure was used to forecast the levels of trip making likely to result from the planned land use pattern. In the design of the transportation system plan, an appropriate mix of transportation system development proposals necessary to accommodate the forecast travel patterns was then advanced.

Consequently, before examining the status of transportation system development since the adoption of the regional plan, it is necessary to evaluate the major socio-economic forecast variables used in the design of the plan. The conformance or departure of these forecast variables to or from actual conditions serves as an important qualifier in the evaluation of the continued validity of the adopted regional transportation system plan.

Population and Economic Activity

The basic purpose for making a comparison between forecast and actual conditions with respect to the levels and distribution of population and economic activity is to determine the extent to which the scale of urban development in the Region approximates the scale envisioned. Should it be found that the actual levels of population, employment, and households, which determine the overall scale of urban development, greatly differ from the levels forecast, then the basic validity of the adopted land use and transportation system plans may be brought into question. The basic validity of the plans would be brought into question whether the observed differences in the determinants of the scale of development were significantly smaller or larger than forecast. For example, should the actual population size, number of jobs, and number of households in particular be found to significantly exceed the levels forecast, then it should be expected that the planned scale of urban development would be exceeded. This in turn could lead to severe traffic congestion and other problems related to insufficient arterial capacity and public transit service.

The Commission socio-economic forecasts are tempered by an envisioned modification of the market forces that have driven land use decentralization in the Region since the early part of the twentieth century. The Commission recognizes that the private search for, and public accommodation of, the highest returns on private land investment can result in a scattered and inefficient urban settlement pattern. Therefore, in the year 2000 regional land use plan, the Commission adjusted the forecast population and employment levels at the county level to design a land use plan with a more centralized distribution of population, employment, and related land use and transportation system development than an extrapolation of existing trends would indicate.

⁴A U. S. Public Land Survey quarter-section consists of a rectilinear area containing approximately 160 acres with all sides approximating one-half mile.

⁵See Chapter VII.

For this reason the second-generation regional land use plan was termed a "controlled centralization" land use plan. Thus, there is a normative aspect to the Commission socio-economic forecasts, which at the county level seek to moderate the decentralization trends in the Region. The normative nature of the socioeconomic forecasts required that more resident population be allocated to Milwaukee County than would be the case if the observed decentralization trends were simply projected to continue.

The surveillance activities undertaken by the Commission with respect to population, households, and employment in the Region from 1970 through 1990 resulted in the basic findings set forth below:

Population: The population forecast had envisioned an increase in the regional population of about 286,300 persons, or an estimated 16 percent, from 1970 through 1990, as set forth in Table 10 and shown in Figure 4. By 1990, the actual population in the Region exhibited a slight increase of about 54,300 persons, or about 3 percent, over the 1970 base-year population. Thus, the actual 1990 regional population was lower than the 1990 forecast population by about 232,000 persons, or 11 percent. Evaluation of the factors that determine population levels, births, deaths, and migration, indicated that the variance between estimated and forecast levels resulted primarily from a rate of population outmigration in excess of the rate assumed in the population forecast. This excessive outmigration may be attributed in large part to the severe economic recession which occurred in the Region between 1979 and 1983.

At the county level, Kenosha, Ozaukee, Walworth, Washington, and Waukesha Counties all experienced population increases over the 20-year period. The population level in Racine County remained relatively stable during that time, showing an almost 3 percent increase, while the population of Milwaukee County decreased by 95,000 persons, or 9 percent. For each county, however, the actual 1990 population level was lower than the forecast level, the variance ranging from about 7 percent in Milwaukee County to nearly 24 percent in Ozaukee County.

While the overall regional population level by 1990 was less than forecast, the relative distribution of the population in the Region was remarkably close to the forecast distribution. For example, the proportion of the regional population residing in Milwaukee County, which was 60 percent in 1970, was forecast to decrease to 50 percent in 1990. Milwaukee County's proportion of the actual regional population in 1990 was 53 percent. In relatively fast-growing Waukesha County, where 13 percent of the regional population resided in 1970, monitoring revealed that the County housed 17 percent of the regional population in 1990. This may be compared with a forecast share for 1990 of 17 percent.

Households: The household forecast had envisioned an increase in the number of households in the Region of 131,300, or 24 percent, by 1990, as set forth in Table 11 and shown in Figure 5. By 1990, the actual number of households in the Region had increased over the 1970 base year by about 139,500, a 26 percent increase. Thus, on a regional scale, the 1990 household level exceeded that forecast by about 8,200, or 1 percent. The fact that the number of households increased as forecast, despite relatively stable population levels over the 20-year period, can be explained by the significant decrease in average household size, which was larger than forecast. Indeed, the actual number of persons per household, which stood at about 3.20 in 1970, had declined to about 2.68 by 1990. This can be compared to the 1990 forecast household size of about 3.06 persons per household.

At the county level, the monitoring indicated that each county in the Region experienced an increase in the number of households over the 20-year period. Actual household levels in 1990 were slightly higher than forecast in Racine, Walworth, and Waukesha Counties, and slightly lower than forecast in Kenosha, Ozaukee, and Washington Counties. The actual level of households in Milwaukee County deviated from the forecast level by less than one-half percent, indicating nearly complete conformance with the forecast. The variance in actual and forecast household levels was greatest in Waukesha County, almost 8 percent.

Review of the data in Table 11 indicated that the relative distribution of the number of households in the Region was remarkably close to the forecast distribution. Milwaukee County's share of all households in the Region, which stood at 63 percent in 1970, was forecast to decrease to

	Existing 1970		Forecast 1990				Actual 1990				Difference between	
			Change 1970-1990		Total		Change 1970-1990		Total		Actual and Forecast 1990 Population Levels	
County	Total	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent
Kenosha	117,900	6.7	40,200	34.1	158,100	7.7	10,300	8.7	128,200	7.1	-29,900	-18.9
Milwaukee	1,054,300	60.1	-27,800	-2.6	1,026,500	50.3	-95,000	-9.0	959,300	53.0	-67,200	-6.5
Ozaukee	54,500	3.1	41,400	76.0	95,900	4.7	18,300	33.6	72,800	4.0	-23,100	-24.1
Racine	170,800	9.7	32,100	18.8	202,900	9.9	4,300	2.5	175,100	9.7	-27,800	-13.7
Walworth	63,500	3.6	23,400	36.7	86,900	4.3	11,500	18.1	75,000	4.1	-11,900	-13.7
Washington	63,800	3.6	53,100	83.2	116,900	5.7	31,500	49.4	95,300	5.3	-21,600	-18.5
Waukesha	231,300	13.2	123,900	53.6	355,200	17.4	73,400	31.7	304,700	16.8	-50,500	-14.2
Region	1,756,100	100.0	286,300	16.3	2,042,400	100.0	54,300	3.1	1,810,400	100.0	-232,000	-11.4

ACTUAL AND FORECAST POPULATION LEVELS IN THE REGION BY COUNTY: 1970 AND 1990

Source: U. S. Bureau of the Census and SEWRPC.

almost 56 percent by 1990. Monitoring data showed that the estimated proportion in 1990 was 55 percent. Similarly, Waukesha County's share of regional households stood at about 12 percent in 1970 and was forecast to increase to about 15 percent by 1990. Monitoring data revealed that the estimated level in 1990 was just under 16 percent.

Employment: The employment forecast had envisioned an increase by 1990 in the number of jobs in the Region of nearly 170,700, or 23 percent, as set forth in Table 12 and shown in Figure 6. By 1990, despite a severe economic recession within the Region, the estimated number of jobs in the Region had increased over the 1970 base year by about 236,600, a 31 percent increase. Thus, on a regional scale, the 1990 employment level exceeded that forecast by about 65,900 jobs, or 7 percent. Given a relatively stable regional population, the increase in regional employment reflects in part an increasing participation rate of the total population, particularly among women, in the labor force between 1970 and 1990. In fact, the female labor participation rate in the Region increased from 45 percent in 1970 to 64 percent by 1990.

At the county level, the monitoring indicated that each county in the Region experienced an increase in total employment over the 20-year period. The largest relative increases occurred in Walworth, Ozaukee, Washington, and Waukesha Counties, marking a continuing decentralization of economic activity within the Region. Employment in Waukesha County more than doubled between 1970 and 1990, increasing by 125 percent. The increases in employment among the counties between 1970 and 1985 varied somewhat from that forecast. Employment levels increased substantially faster than forecast in Waukesha and Washington Counties and slightly faster than forecast in Milwaukee County, Ozaukee County, and Walworth County, where employment grew by 2, 3, and almost 5 percent, respectively. In Kenosha and Racine Counties, employment increased somewhat slower, about 6 and 3 percent less than forecast, respectively.

An indication that economic activity is becoming increasingly decentralized is the change in the distribution of regional employment among the counties. For example, Milwaukee County's share of the regional employment, which stood at 67 percent in 1970, was forecast to decrease to 61 percent by 1990. Monitoring data showed that the County's actual proportion had declined to 58 percent. By contrast, Waukesha County's share of regional employment, which stood at 10 percent in 1970, was forecast to increase to 14 percent by 1990. Monitoring data showed that the actual share in 1990 had reached 17 percent.

<u>Automobile Availability</u>: The automobile availability forecast had envisioned an increase in the number of automobiles in the Region of 242,700, or 38 percent, by 1990, as set forth in Table 13 and shown in Figure 7. By 1990, the number of automobiles available to residents in the Region had actually increased by 293,300, an increase of 46 percent, over the 1970 level. Thus, on a regional scale, the 1990 automobile availability level exceeded the forecast level by 50,600 automobiles, or about 6 percent. Several factors have contributed to this variance including an increase in the number of smaller size households and multiple ownership of automobiles by
Figure 4





Source: U. S. Bureau of the Census, Wisconsin Department of Administration, and SEWRPC.

	Existing	1970		Foreca	st 1990			Actua	1990		Difference between	
			Change 19	70-1990	0 Total		Change 1970-1990		Total		1990 Household Levels	
County	Total	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent
Kenosha	35.500	6.6	14,100	39.7	49,600	7.4	11.500	32.4	47.000	7.0	-2,600	-5.2
Milwaukee	338,600	63.1	33,000	9.7	371,600	55.6	34,400	10.2	373,000	55.2	1,400	0.4
Ozaukee	14,800	2.8	12,300	83.1	27,100	4.1	10,900	73.6	25,700	3.8	-1,400	-5.2
Racine	49,800	9.3	11,700	23.5	61,500	9.2	13,900	27.9	63,700	9.4	2,200	3.6
Walworth	18,500	3.4	7,300	39.5	25,800	3.9	9,100	49.2	27,600	4.1	1,800	7.0
Washington	17,400	3.2	16,200	93.1	33,600	5.0	15,600	89.7	33,000	4.9	-600	-1.8
Waukesha	61,900	11.5	36,700	59.3	98,600	14.8	44,100	71.2	106,000	15.7	7,400	7.5
Region	536,500	100.0	131,300	24.4	667,800	100.0	139,500	26.0	676,000	100.0	8,200	1.2

ACTUAL AND FORECAST HOUSEHOLD LEVELS IN THE REGION BY COUNTY: 1970 AND 1990

Source: U. S. Bureau of the Census and SEWRPC.

households. The increase in automobiles available is significant in light of the slower than forecast population growth experienced in the Region.

At the county level, the monitoring indicated that each county in the Region had experienced an increase in total automobiles available over the 20-year period. The largest relative increases occurred in Ozaukee and Washington Counties, the two counties experiencing the most rapid increases in resident population levels and number of households during the same time period. In both counties, the number of automobiles available more than doubled between 1970 and 1990. Though automobile availability increased substantially in Ozaukee and Washington Counties, it did not attain the levels forecast. Actual 1990 levels were below forecast levels by 11 and 10 percent, respectively. By contrast, monitoring indicated that in Milwaukee County automobile availability exceeded the level forecast by nearly 15 percent, the greatest variance between actual and forecast automobile availability levels among the counties in the Region.

While the level of automobiles available in the Region in 1990 was 6 percent greater than forecast, the relative distribution of automobiles among the counties was remarkably close to the forecast distribution. Indeed, the only real deviation between the actual and forecast proportion of automobiles occurred in Milwaukee County, about 7 percent.

<u>Truck Availability</u>: The motor truck availability forecast had envisioned an increase in the number of trucks available to truck operators in the Region of about 56,300, or about 79 percent, by 1990 as set forth in Table 14 and shown in Figure 8. By 1990, the actual number of trucks available in the Region had increased by about 106,200, or by about 148 percent, over the 1970 base level. Thus, on a regional scale, the actual number of trucks in 1990 exceeded the forecast level by 49,930, or 39 percent. About 78 percent of all trucks in 1990, or 139,230, were classified as light trucks, having a net weight of less than 8,000 pounds. The remaining 22 percent, or 38,700, were classified as municipal or heavy trucks, having a net weight of greater than 8,000 pounds and two axles or more with dual wheels.

The difference in the actual and forecast level of trucks available in the Region may be attributed in part to the increasing use of light trucks as personal vehicles. The 1991 truck survey, conducted as part of the third major inventory of travel in the Region, indicated that light trucks used solely as personal vehicles accounted for 48 percent of all trucks in 1991 (see Figure 9). This may be compared to 23 percent in 1972, and to 9 percent in 1963. The increasing use of light trucks as personal vehicles has the same implications for transportation system planning as the increasing use of automobiles.

Indeed, the model used in the second-generation planning process to forecast automobile availability attempted to reflect the household decision process of owning a personal vehicle, regardless of its type, automobile or personal-use truck. In this regard, Table 15 sets forth an estimate of the actual and the forecast levels of personal vehicle availability. When personal-use trucks, whose number was estimated on the basis of the findings of the home interview survey conducted by the Commission in 1991, are added to the number of automobiles available in the Region in 1990, the resulting personal vehicle availability level slightly exceeds one million and is nearly 15 percent greater than the forecast level. Such consideration of personal-use trucks provides an estimate of personal vehicle

Figure 5

ACTUAL AND FORECAST REGIONAL AND COUNTY HOUSEHOLDS: 1950-2000



Source: U. S. Bureau of the Census and SEWRPC.

2000

2000

ACTUAL AND FORECAST EMPLOYMENT LEVELS IN THE REGION BY COUNTY: 1970 AND 1990

	Existing	1970	Forecast 1990					Actual	•	Difference between		
			Change 19	70-1990	Total		Change 1970-1990		Total		1990 Employment Levels	
County	Total	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent
Kenosha	40,000	5.3	9,200	23.0	49,200	5.3	6,500	16.3	46,500	4.7	-2,700	-5.5
Milwaukee Ozaukee	507,100 19,800	67.3 2.6	58,900 11,500	11.6 58.1	566,000 31,300	61.2 3.4	71,100 12,400	14.0 62.6	578,200 32,200	58.4 3.3	12,200 900	2.2
Racine	62,700 24,500	8.3 3.3	21,600 11,000	34.4 45.0	84,300 35,500	9.2 3.8	19,500 12,600	31.1 51.4	82,200 37,100	8.3 3.7	-2,100 1,600	-2.5 4.5
Washington Waukasha	23,100	3.1 10.1	7,700	33.3 66.4	30,800	3.3 13.8	18,700	81.0 125.2	41,800 172,300	4.2 17.4	11,000	35.7 35.3
Region	753,700	100.0	170,700	22.6	924,400	100.0	236,600	31.4	990,300	100.0	65,900	7.1

Source: U. S. Bureau of Economic Analysis; Wisconsin Department of Industry, Labor and Human Relations; and SEWRPC.

availability that more closely reflects the magnitude of traffic on the arterial streets and highways of the Region, than does the level of automobile availability alone.

PLAN DESIGN: LAND USE DISTRIBUTION

The foregoing evaluation of the forecasts made for the design of the transportation system plan indicates that while the population of the Region has not increased as forecast, the other determinants of the scale of urban development, households and employment, have increased substantially in accord with the forecasts and well within forecast accuracy tolerances. The close conformity between the actual and forecast number of households is particularly significant, since the household represents a basic consuming unit which generates much of the demand for urban land and transportation facilities and services.

The socio-economic forecasts are used in the plan design step of the planning process to determine the scale of future urban development and specifically the amount of rural land to be converted to urban land uses. Therefore, the conformity of the socio-economic forecasts to actual conditions should suggest conformity between the planned scale of urban development and the actual scale during the same period of time. Monitoring of land use data between 1970 and 1985 has, in fact, revealed such conformity, the actual scale of urban development within the Region exceeded the forecast scale by about 15 square miles, or by about 3 percent.

Pertinent to any transportation system plan reappraisal, however, is the distribution of the forecast land use development. As previously stated, the distribution of urban development is used as an input to the traffic models to forecast trip production and distribution. Thus, an evaluation of the pattern of urban development since the base year of the regional land use plan is pertinent to the transportation system plan reevaluation. The distribution of urban development stands as one of the factors potentially contributing to any variance in actual and forecast levels of trip production and distribution.

Accordingly, this section reports the results of the monitoring of residential land use development activity in the Region over the period 1970 through 1985 and over the period 1970 through 1990 with respect to major regional centers. The bulk of the information presented regarding residential urban development activity is based on the detailed land use inventory maintained by the Commission. The land use inventory is a precise accounting of land cover, identifying all land which has been developed for, and is actually in, urban use. Under the land use inventory, urban land use includes all residential, commercial, industrial, governmental, intensive recreational, and transportation development located within the Region.

Conversion of Rural Land

to Urban Development

As noted above, to accommodate forecast increases in population, households and employment by 1985, the regional land use plan envisioned that nearly 91 square miles of rural land would be converted to urban use. The adopted year 2000 regional land use plan is shown on Map 7. As shown in Table 16, actual rural land conversion during the 15-year period totaled about 105 square miles, almost 15 square miles, or about 3 percent, more than forecast.

Figure 6



Source: U. S. Bureau of Economic Analysis; Wisconsin Department of Industry, Labor and Human Relations; and SEWRPC.

47

ACTUAL AND FORECAST AUTOMOBILE AVAILABILITY IN THE REGION BY COUNTY: 1970 AND 1990

	Existing	g 1970		Forecast 1990				Actual 1990				Difference between	
			Change 1970-1990		Total		Change 1970-1990		Total		Automobile Availability		
County	Total	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent	Number	Percent of Region	Number	Percent	
Kenosha	43,900	6.9	30,500	69.5	74,400	8.5	23,000	52.4	66,900	7.2	-750	-10.1	
Milwaukee	367,700	58.0	28,300	7.7	396,000	42.1	86,400	23.5	454,100	49.0	-58,100	14.7	
Ozaukee	21,000	3.3	26,400	125.7	47,400	5.4	21,200	101.0	42,200	4.5	-5,200	-11.0	
Racine	62,400	9.8	27,900	44.7	90,300	10.3	29,300	47.0	91,700	9.9	-1,400	1.6	
Walworth	24,700	4.0	17,200	69.6	41,900	4.8	18,000	72.9	42,700	4.6	800	1.9	
Washington	23,500	3.7	33,300	141.7	56,800	6.5	27,500	117.0	51,000	5.5	-5,800	-10.2	
Waukesha	90,800	14.3	79,100	87.1	169,900	19.4	87,900	96.8	178,700	19.3	8,800	5.2	
Region	634,000	100.0	242,700	38.3	876,700	100.0	293,300	46.3	927,300	100.0	50,000	5.8	

Source: U. S. Bureau of Economic Analysis; Wisconsin Department of Industry, Labor and Human Relations; and SEWRPC.

Table 14

ACTUAL AND FORECAST MOTOR TRUCK AVAILABILITY IN THE REGION: 1970 AND 1990

	F	orecast 19	90		Actual 1990	Difference between Actual and Forecast 1990 Truck Availability		
Existing	Change 1970-1990		Total	Change 19	70-1990			Total
1970	Number Percent		Number	Number	Number Percent		Number	Percent
71,700	56,300	78.5	128,000	106,230	148.2	177,930	49,930	39.0

Source: SEWRPC.

Figure 7

ACTUAL AND FORECAST AUTOMOBILE AVAILABILITY IN THE REGION: 1960-1990



Source: SEWRPC.

Figure 8

ACTUAL AND FORECAST TRUCK AVAILABILITY IN THE REGION: 1960-1990



Source: SEWRPC.

	4 92		Difference between Actual and Forecast 1990 Personal Vehicle Availability			
Type of Personal Vehicle	Forecast ^a	Actual	Number	Percent		
Automobile		927,300				
Motor Truck		77,000 ^b	(जल्दा)			
Total	876,700	1,004,300	127,600	14.6		

ACTUAL AND FORECAST PERSONAL VEHICLE AVAILABILITY IN THE REGION: 1990

^aDocumented elsewhere in this report as automobile availability. The model used to prepare this forecast attempted to reflect the decision of households to own a vehicle, regardless of its type, auto or truck.

^bCommission estimate based upon 1991 Home Interview Survey assuming the personal use of automobiles between 1990 and 1991 increased at a rate equal to the rate of increase in the availability of light trucks in the Region (4.5 percent between 1990 and 1991).

Source: SEWRPC.



Source: Wisconsin Department of Transportation and SEWRPC.

Table 16 also shows the planned and actual allocation of urban development among residential and nonresidential land uses between 1970 and 1985. Under the regional land use plan, about 58 square miles of land were planned to be converted from rural to urban residential use by 1985. Monitoring indicated, however, that the actual increase in residential land totaled 66 square miles, about seven square miles, or 12 percent, greater than the planned increase. Thus, by 1985, the actual amount of residential development in the Region totaled nearly 289 square miles, exceeding the planned amount by nearly 3 percent. The distribution of that urban development among the residential land use categories is particularly important to the reevaluation of the adopted regional transportation system plan. The density and location of residential development to a large extent determines the potential effectiveness of vehicle trip reduction strategies as well as the viability of public transit service. Thus, the conversion of rural land to low-density residential development can preclude effective ridesharing programs and the cost effective provision of public transit service to areas so converted.

While the year 2000 regional land use plan recommended that new residential development occur primarily at medium densities, between 1970 and 1985 the Region experienced substantial residential development at low densities. In fact, low-density residential development accounted for almost 73 percent of the overall increase in residential land between 1970 and 1985. As shown in Table 16, the plan recommended that less than 12 square miles of land be converted to low-density residential use, defined as from 0.7 to 2.2 housing units per net residential acre. The monitoring showed, however, that 41 square miles were actually so converted. By contrast, the plan envisioned that by 1985 nearly 32 square miles of rural land would be converted to medium-density residential use, defined as from 2.3 to 6.9 housing units per net residential acre. Yet, less than 17 square miles were actually so converted. The conversion of rural land to low-density residential develop-

Map 7 REGIONAL LAND USE PLAN FOR SOUTHEASTERN WISCONSIN: 2000



The second-generation, year 2000, regional land use plan, like the first-generation, design year 1990, plan, sought to influence the operation of the urban land market within the Region in three important ways in order to achieve a more helpful and attractive, as well as more efficient, regional settlement pattern. First, the plan recommended that intensive urban development occur only in those areas of the Region which are covered by soils suitable for such development, which are not subject to such special hazards as flooding and shoreline erosion, and which can be readily served by essential municipal facilities and services. Second, the plan recommended the preservation in essentially natural open use of the remaining primary environmental corridors, those linear areas in the landscape encompassing the most important features of the natural resource space. Third, the plan recommended the retention in agricultural and related use of almost all of the remaining prime agricultural lands, the most productive farmlands and farm units in the Region.

URBAN LAND CONVERSION IN THE REGION: 1970-1985

	Actu	Actual 1970 Planned 198				1985 Actua			al 1985		Difference between	
			Change 1	Change 1970-1985		otal	Change 1	970-1985	Total		Urban Land Area	
Urban Land Category	Square Miles	Percent of Total or Subtotal	Square Miles	Percent	Square Miles	Percent of Total or Subtotal	Square Miles	Percent	Square Miles	Percent of Total or Subtotal	Square Miles	Percent
Nonresidentiai ^a	228.1	50.6	32.3	14.2	260.4	48.1	39.5	17.3	267.6	48.1	7.2	2.8
Residential High-Density ^b Medium-Density ^c Low-Density ^d Suburban-Density ^e	42.1 67.9 106.5 6.5	18.9 30.4 47.8 2.9	0.9 31.7 11.5 14.1	2.1 46.7 10.8 216.9	43.0 99.6 118.0 20.6	15.3 35.4 42.0 7.3	1.3 16.7 41.4 6.1	3.1 24.6 38.9 93.8	43.4 84.6 147.9 12.6	15.0 29.3 51.3 4.4	0.4 -15.0 29.9 -8.0	0.9 -15.1 25.3 -38.8
Subtotal	223.0	49.4	58.2	26.1	281.2	51.9	65.5	29.4	288.5	51.9	7.3	2.6
Total	451.1	100.0	90.5	20.1	541.6	100.0	105.0	23.3	556.1	100.0	14.5	2.7

^aIncludes those urban land use categories, other than residential, for which planned increments were identified under the year 2000 regional land use plan, namely: commercial; industrial; governmental and institutional; public recreational; and transportation, communication, and utilities. Private recreation lands and unused urban lands are not reflected in this table. Private recreation lands and unused urban land together encompassed 54.3 square miles in 1970 and 49.7 square miles in 1985.

^bDefined as 7.0 to 17.9 housing units per net residential acre.

^CDefined as 2.3 to 6.9 housing units per net residential acre.

^dDefined as 0.7 to 2.2 housing units per net residential acre.

^eDefined as 0.2 to 0.6 housing units per net residential acre.

Source: SEWRPC.

Table 17

PLANNED AND ACTUAL PROPORTION OF NEW HOUSING UNITS BY RESIDENTIAL LAND USE CATEGORY: 1970-1985

Residential Land	Planned 1985	Actual 1985	Difference
Use Category	Proportion (percent)	Proportion (percent)	
High-Density ^a	6.3	7.5	1.2
	76.4	64.1	-12.3
	9.6	26.9	17.3
	7.7	1.5	-6.2
Total	100.0	100.0	

^aDefined as 7.0 to 17.9 housing units per net residential acre.

^bDefined as 2.3 to 6.9 housing units per net residential acre.

^cDefined as 0.7 to 2.2 housing units per net residential acre.

^dDefined as 0.2 to 0.6 housing units per net residential acre.

Source: SEWRPC.

ment was most pronounced and had its greatest areal impact in Waukesha and Washington Counties.

Although a substantially greater amount of residential land was developed at low densities, the majority of housing units constructed during the 15-year period were accommodated at higher densities. As indicated in Table 17, over 70 percent of all housing units built between 1970 and 1985 were at medium or high residential densities. This is particularly significant to the evaluation of the continued viability of the regional transportation system plan because the concentration of occupied housing units enables the economic provision of public transit as well as the effective use of ridesharing strategies. Thus, if trip production in the Region increased

ACTUAL AND PLANNED CHANGE IN THE NUMBER OF HOUSING UNITS IN THE REGION BY RESIDENTIAL LAND USE CATEGORY: 1970 AND 1985

	Existing	1970	Planned 1985				Actual 1985				Difference between	
			Change 1970-1985		Total		Change 1970-1985		Total		Actual and Planned 1985 Housing Units	
Residential Land Use Category	Number	Percent of Total	Number	Percent	Number	Percent of Total	Number	Percent	Number	Percent of Total	Number	Percent
High-Density ^a Medium-Density ^b Low-Density ^c Suburban-Density ^d	319,930 139,490 74,770 2,300	59.6 26.0 14.0 0.4	6,070 73,090 9,160 7,390	1.9 52.4 12.3 321.3	326,000 212,580 83,930 9,690	51.6 33.6 13.3 1.5	8,080 68,760 28,850 1,650	2.5 49.3 38.6 71.7	328,010 208,250 103,620 3,950	51.0 32.3 16.1 0.6	2,010 -4,330 19,690 -5,740	0.6 -2.0 23.5 -59.2
Region	536,490	100.0	95,710	17.8	632,200	100.0	107,340	20.0	643,830	100.0	11,630	1.8

^aDefined as 7.0 to 17.9 housing units per net residential acre.

^bDefined as 2.3 to 6.9 housing units per net residential acre.

^cDefined as 0.7 to 2.2 housing units per net residential acre.

^dDefined as 0.2 to 0.6 housing units per net residential acre.

Source: SEWRPC.

significantly faster than forecast despite the development of most new households at planned residential densities, other factors must be considered to explain high levels of trip making. Such explanations may include basic lifestyle changes as well as greater than anticipated lowdensity residential development. As shown in Table 17, the actual proportion of all new housing units accommodated at low density, 27 percent, exceeded the planned proportion by 17 percentage points. By contrast, the actual proportion of all new units accommodated at medium density, 64 percent, was 12 percent below the planned proportion.

Consistent with the findings stated above, the monitoring indicated that more new housing units were provided at low residential densities over the 15-year period than were planned. Table 18 shows that the land use plan recommended an increase in the number of housing units provided in the low-density category of 12 percent by 1985. By 1985, the actual number of housing units provided at low residential densities had increased by 32 percent. During the same 15-year period, a shortfall of 2 percent in the number of housing units provided in the medium-density category occurred. Table 18 also shows that nearly 2 percent more housing units were provided in the Region between 1970 and 1985 than had been planned.

The conversion of rural land to low-density residential development is a significant factor in

evaluating the validity of the adopted plan. To understand the affect of low-density urban development on trip making within the Region more fully, the location of new urban development must be examined.

Location of Incremental Urban Development

A study of the location of urban development requires that an alternative method of land use analysis be used, an approach different from that used to provide the land use data presented above. Under this method, termed urban growth ring analysis, urban development is defined as those areas wherein houses and other buildings have been constructed in relatively compact groups or where a closely spaced network of minor streets has been constructed, thus indicating a concentration of residential, commercial, industrial, or some other urban uses. Urban growth ring analysis identifies the limits or boundaries of concentrations of urban growth and is thus useful in graphically representing the location of new urban development and determining changes in overall urban density in the Region.

Monitoring data with respect to the location of incremental urban growth indicated that nearly half of all urban development between 1970 and 1985 occurred in a diffused pattern, making travel in much of the newly developed areas almost totally automobile-dependent. Of the nearly 139 square miles of incremental urban development identified through the Commis-

		Increment	al Urban Deve	lopment 1970-1	985 (square	miles)	
County	Extent of Urban	Located Recommende Developm Regional Lar	in Areas ed for Urban ent in the nd Use Plan	Located in A Recommende Developme Regional Lan	Areas Not d for Urban ent in the d Use Plan		Extent of Urban
	Development 1970 (square miles)	Amount	Percent of Total	Amount	Percent of Total	Total	Development 1985 (square miles)
Kenosha	26.4	5.4	65.1	2.9	34.9	8.3	34.7
Milwaukee	149.8	17.2	99.4	0.1	0.6	17.3	167.1
Ozaukee	19.0	6.5	60.7	4.2	39.3	10.7	29.7
Racine	33.9	6.5	58.6	4.6	41.4	11.1	45.0
Walworth	22.1	3.6	33.6	7.1	66.4	10.7	32.8
Washington	14.1	5.8	25.8	16.7	74.2	22.5	36.6
Waukesha	72.3	27.4	47.0	30.9	53.0	58.3	130.6
Region	337.6	72.4	52.1	66.5	47.9	138.9	476.5

INCREMENTAL URBAN DEVELOPMENT IN THE REGION BY COUNTY: 1970-1985

Source: SEWRPC.

sion's growth ring analysis, approximately 66 square miles, or 48 percent, took place in a scattered fashion over much of Southeastern Wisconsin. The extent of this urban development, concentrated particularly in Waukesha and Washington Counties, is set forth in Table 19 and is shown in red on Map 8.

The remaining 72 square miles of incremental urban development, identified through the Commission's growth ring analysis, occurred within urban service areas as recommended in the regional land use plan. They are shown in green on Map 8. Such development can be readily and economically provided with essential public services such as sanitary sewer, water supply, and transit. In this regard, a supplemental analysis of the provision of sanitary sewer service to new housing units has revealed that, between 1970 and 1985, 79 percent of all new housing units constructed were within planned sanitary sewer service area, marking a significant achievement in the implementation of the year 2000 regional land use plan.

As already noted, the increase in the levels of households and employment exerted an urban land demand of the approximate magnitude envisioned in the plan, the scale of urban development varying only slightly from the forecast. Since population growth was insignificant between 1970 and 1985, the actual urban population density declined from about 5,100 persons per square mile to about 3,900 persons

Figure 10

ACTUAL AND PLANNED URBAN POPULATION DENSITY IN THE REGION: 1850-2000



Source: SEWRPC.

per square miles by 1985, as shown in Figure 10. Consequently, a substantial area of land was converted from rural to urban use over the 15year period to serve essentially the same population level as existed within the Region in 1970.

Major Regional Centers

The regional land use plan included explicit recommendations concerning the location and size of three types of major regional centers of activity: industrial, commercial, and outdoor recreational. Major regional centers, especially in the case of industrial and commercial centers



Monitoring the location of incremental urban growth indicated that nearly one-half of all urban development between 1970 and 1985, measured in square miles, occurred in a scattered pattern, making travel in many of the newly developed areas almost totally automobile dependent. Of the nearly 139 square miles of incremental urban development identified through the Commission's growth ring analysis, approximately 66 square miles, or 48 percent, took place in a scattered fashion over much of the Southeastern Wisconsin Region, particularly in Waukesha and Washington Counties.

Source: SEWRPC.

because of large employment concentrations, are significant trip generators. Consequently, the planned scale of transportation system development and rapid transit service in particular would be affected by large changes in employment at such centers and large shifts in employment among such centers. The proposed and actual development of major regional centers over the period 1972 through 1990 is described below.

<u>Industrial Centers</u>: Major industrial centers are defined in the adopted regional land use plan as concentrations of employment having at least 3,500 "industrial" jobs, with the term "industrial" encompassing the manufacturing, wholesaling, and construction sectors. While this definition is used for regional land use planning purposes, for regional transportation planning purposes the Commission also takes into account total jobs at each major industrial center. Therefore, the status of employment at each major center in terms of both industrial and total jobs is reported herein.

As shown in Table 20 and on Map 9, the regional land use plan envisioned that by the year 2000, 22 major industrial centers would be provided to serve the Region. Of the 22 major centers, 17 were in existence in 1972 and all but one met the threshold industrial employment criterion in both 1972 and 1990. Of the five planned new centers, three, Milwaukee-Granville, Oak Creek, and Waukesha, had been established and met the threshold industrial employment criterion by 1990. Of the remaining two planned centers, one, Burlington, was under development but did not yet meet the threshold industrial employment criterion; the other, Kenosha-West, had not yet been established.

The monitoring data indicated significant declines in the industrial and total employment levels at a number of the older industrial centers which were thriving in 1972, including Kenosha-East, Milwaukee-Glendale, Milwaukee-Menomonee Valley East, Milwaukee-Near North, and West Allis-East. Industrial employment at the West Allis-East center declined by 6,200 jobs, or 67 percent, between 1972 and 1990, causing West Allis-East to lose its major center designation.

The significant declines in industrial employment at the older centers noted above may be attributed to two factors: the overall loss of manufacturing jobs within the Region during the severe economic recession of 1979 to 1983 and, more significantly, the general restructuring of the national economy away from heavy manufacturing dependent on rail and seaport transportation facilities. During the economic recession of 1979 to 1983, the broad employment trend away from manufacturing was most pronounced; about 80,000 industrial jobs were lost within the Region.

The monitoring data also indicated a significantly higher than planned employment increase at one center and the emergence of two major unplanned centers. Employment at the Butler Center, located in the Villages of Butler and Menomonee Falls and the adjacent Cities of Brookfield, Milwaukee, and Wauwatosa, increased by 41 percent beyond the planned level. With about 21,000 industrial and about 29,000 total jobs in 1990, the Butler center represents the largest industrial employment center in the Region. The Waukesha-North center and the Pewaukee center were developed as major industrial centers by 1990 in areas that were not called for in the regional land use plan.

In addition, monitoring information indicates that four additional major industrial centers may emerge at sites previously unplanned. Three of these centers are under development, one each in the Cities of Franklin, West Bend, and Hartford.⁶ A fourth potential industrial center, Pabst Farms along IH 94 in the Oconomowoc area of Waukesha County, has yet to receive the required local governmental approvals.

<u>Commercial Centers</u>: Major commercial centers are classified into two types: retail centers and office centers. A single center may be one or the other, or both, if both retail and office employment criteria are met. Major retail centers are defined as concentrations of employment having at least 2,000 jobs in the retail trade sector. Major office centers are defined as concentrations of employment having at least 3,500 jobs in the

⁶The industrial center at Hartford has been included as a planned major industrial center in the third-generation regional land use plan. See SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin</u> 2010, January 1992.

		,	ndustrial Em	ployment		Total Employment			
			Estima	ted 1990	Meets Griterion			Estimat	ed 1990
Major Industrial Center ^a	Existing 1972	Planned 2000	Number	Percent Change from 1972	for Designation as a Major Center in 1990 ^a	Existing 1972	Planned 2000	Number	Percent Change from 1972
Eviating in 1970									
Kenosha-East Cudahy-South Milwaukee	11,600 7,300	11,600 8,400	4,800 5,100	-59 -30	Yes Yes	14,600 9,800	15,300 11,800	7,100 8,900	-51 -9
Milwaukee-Glendale	17,800	18,200	8,500	-52	Yes	22,700	23,700	13,700	-40
Milwaukee-Menomonee Valley East	18,600	19,000	9,400	-49	Yes	23,500	24,600	17,000	-28
Milwaukee-Menomonee Valley West	5,300	5,400	6,400	21	Yes	8,200	8,600	9,500	16
Milwaukee-Near North	15,000	15,300	6,200	-59	Yes	17,300	18,000	9,600	-45
Milwaukee-Near South	12,600	12,900	9,900	-21	Yes	14,700	15,500	12,300	-16
Milwaukee-North	20,800	21,200	13,100	-37	Yes	25,300	26,200	16,400	-35
Milwaukee-South	4,100	4,200	4,500	10	Yes	5,400	5,600	6,100	-13
West Allis-East	9,300	9,500	3,100	-67	No	13,700	14,400	8,300	-39
West Allis-West	3,600	3,700	5,700	58	Yes	7,800	8,200	10,800	38
West Milwaukee	15,400	15,700	6,300	-59	Yes	18,800	19,500	8,700	54
Mt. Pleasant	3,500	9,400	4,700	34	Yes	3,500	9,600	4,800	37
Racine	12,500	12,800	9,700	-22	Yes	15,500	16,600	11,900	-23
West Bend	3,800	7,100	5,200	37	Yes	3,900	7,200	5,400	38
Butler	14,600	14,900	21,100	45	Yes	17,200	19,800	28,600	66
New Berlin	3,500	8,500	10,400	197	Yes	4,000	9,500	13,800	245
Planned 2000									11 11 1
Kenosha-West		4,500	600		No		4,800	900	
Milwaukee-Granville	1,500	15,500	13,400	793	Yes	1,800	16,000	18,600	933
Oak Creek	800	8,800	9,300	1,063	Yes	1,100	9,200	11,500	945
Burlington	1,200	4,700	2,500	103	No	2,700	7,000	4,700	74
Waukesha	3,000	8,000	5,500	83	Yes	4,200	10,100	7,900	88
Unplanned but Developed by 1990		i							
Pewaukee			6,200		Yes			7,800	
Waukesha-North			4,900		Yes			5,900	

STATUS OF MAJOR INDUSTRIAL CENTERS IN THE REGION: 1990

^aThe criterion selected as a basis for determining the status of major industrial centers is a minimum of 3,500 "industrial" jobs. The term "industrial" encompasses the manufacturing, wholesaling, and construction sectors as defined in the <u>Standard Industrial Classification Manual</u> of the U.S. Office of Management and Budget. The table above reports both industrial and total employment at the designated centers. For data collection and analysis purposes, each major center is defined as encompassing a select group of contiguous U.S. Public Land Survey one-quarter sections. For travel forecasting purposes each major center is defined by its location within one or more traffic analysis zones.

Source: SEWRPC.

"office and service" sectors, sectors which encompass the finance, insurance, real estate, and service industries, except government service.⁷

The status of the actual development of each of the 16 planned major commercial centers as of the year 1990 is summarized in Table 21 and on Map 10. By 1990, 12 of the 13 major commercial centers so designated in 1972 met one or both of the commercial center criteria. Mitchell Street, the exception, never met the office center employment criteria. Retail employment at this center fell below the 2,000 threshold level between 1972 and 1990.

Of the three planned new commercial centers, two, Northridge and Racine-West, had been established by 1990 and met the threshold employment levels for classification as major retail centers. In both cases, new shopping centers had been built as recommended in the plan. In the case of the third new commercial center, Oak Creek, a relatively minor development had been completed at the planned location, though employment at that center did not meet the levels needed for classification as either a major retail center or a major office center by 1990.

⁷Special account is taken of jobs in the government and utility sectors only in the four older central business districts; those in Kenosha, Racine, Waukesha, and West Bend, in view of their unique historic character.



The regional land use plan envisioned that by the year 2000, 22 major industrial centers would serve the Region. Of the 22 major centers, 17 were in existence in 1970, and all but one, the West Allis-East Center, met the threshold industrial employment criterion in both 1970 and 1990. Although these older industrial centers continue to meet threshold employment criteria, monitoring shows that employment levels at many of these centers have declined significantly since 1970. Of the five planned new centers, three, Milwaukee Granville, Oak Creek, and Waukesha, had been established and met the threshold industrial employment criterion by 1990. Two industrial centers, Pewaukee and Waukesha North, were developed to threshold by the year 2000.

			Retail	Trade Employment	
			Estir	nated 1990	Meets Criterion for
	Existing	Planned	-	Percent Change	Designation as a Major
Major Commercial Center	1972	2000	Number	from 1972	Retail Center in 1990 ^a
Existing in 1970					
Kenosha CBD	N/A	N/A	N/A	N/A	N/A
Bay Shore	2,300	2,300	2,400	4	Yes
Capitol Court	2,200	2,300	3,300	50	Yes
Mayfair	3,500	3,600	5,100	46	Yes
Milwaukee CBD	11,100	11,100	5,200	-52	Yes
Mitchell Street	3,200	3,300	1,100	-56	No
Southgate-					
Point Loomis	2,000	2,000	2,400	20	Yes
Southridge	2,600	2,600	3,700	42	Yes
West Allis	1,000	1,500	2,800	180	Yes
Racine CBD	N/A	N/A	N/A	N/A	N/A
West Bend CBD	N/A	N/A	N/A	N/A	N/A
Brookfield-Blue					
Mound Road	1,600	1,800	6,700	319	Yes
Waukesha CBD	N/A	N/A	N/A	N/A	N/A
Planned 2000					
Northridge		3,800	5,000	1	Yes
Oak Creek		2,900	300	- -	Νο
Racine-West		3,300	3,500		Yes

STATUS OF MAJOR COMMERCIAL CENTERS IN THE REGION: 1990

		Office and Service Employment										
			Estir	mated 1990	Meets Criterion for							
	Existing	Planned		Percent Change	Designation as a Major							
Major Commercial Center	1972	2000	Number	from 1972	Office Center in 1990 ^a							
Existing in 1970												
Kenosha CBD	3,000	3,300	3,700	23	Yes							
Bay Shore	N/A	N/A	N/A	N/A	N/A							
Capitol Court	N/A	N/A	N/A	N/A	N/A							
Mayfair	3,100	5,100	7,200	119	Yes							
Milwaukee CBD	54,900	55,000	65,900	-7	Yes							
Mitchell Street	N/A	N/A	N/A	N/A	N/A							
Southgate-												
Point Loomis	N/A	N/A	N/A	N/A	N/A							
Southridge	N/A	N/A	N/A	N/A	N/A							
West Allis	N/A	N/A	N/A	N/A	N/A							
Racine CBD	5,700	7,100	4,100	-39	Yes							
West Bend CBD	2,500	3,800	3,800	48	Yes							
Brookfield-Blue												
Mound Road	300	4,300	13,600	1,967	Yes							
Waukesha CBD	5,300	7,500	5,000	-13	Yes							
Planned 2000	-											
Northridge	N/A	N/A	N/A	N/A	N/A							
Oak Creek	N/A	N/A	N/A	N/A	N/A							
Racine-West	N/A	N/A	N/A	N/A	N/A							

Table 21 (continued)

		Total Employment						
			Estimated 1990					
	Existing	Planned		Percent Change				
Major Commercial Center	1972	2000	Number	from 1972	Estimated 1990			
Existing in 1970								
Kenosha CBD	4,500	4,800	4,600	2	4,600			
Bay Shore	4,500	4,600	3,900	-13	3,900			
Capitol Court	3,300	3,400	3,900	18	3,900			
Mayfair	8,100	10,200	13,600	68	13,600			
Milwaukee CBD	88,900	91,000	91,400	28	91,400			
Mitchell Street	5,500	5,800	4,100	-25	4,100			
Southgate-								
Point Loomis	2,600	2,700	3,700	42	3,700			
Southridge	2,800	4,600	4,700	68	4,700			
West Allis	1,700	2,300	6,300	271	6,300			
Racine CBD	7,100	18,600	5,200	-27	5,200			
West Bend CBD	5,300	6,900	6,100	15	6,100			
Brookfield-Blue								
Mound Road	2,200	6,400	24,900	1,032	24,900			
Waukesha CBD	7,500	10,100	6,200	-17	6,200			
Planned 2000	· .							
Northridge	'	4,600	6,500		6,500			
Oak Creek		3,000	1,100		1,100			
Racine-West		5,600	5,700		5,700			

NOTE: N/A indicates not applicable.

^aMajor commercial centers have been classified into two types, retail centers and office centers. The criterion selected as a basis for determining the status of major retail centers is a minimum of 2,000 jobs in the "retail trade" sector as defined in the <u>Standard Industrial Classification Manual</u> of the U. S. Office of Management and Budget. The criterion selected as a basis for determining the status of major office centers, except as noted below, is a minimum of 3,500 "office and service" jobs. The term "office and service" encompasses the finance, insurance, and real estate sector and the services sector, excluding educational services, as defined in the <u>Standard Industrial Classification Manual</u>. The four older, smaller central business districts in the Region, the Kenosha CBD, Racine CBD, Waukesha CBD, and West Bend CBD, are identified as major office centers on the basis of the 3,500 minimum job level, taking into account government and utility sector jobs together with office and service jobs. The above table reports retail trade employment and total employment for retail centers and office and service and total employment for office centers. Some centers meet both the retail and office criteria. For data collection and analysis purposes, each major center is defined as encompassing a select group of contiguous U. S. Public Land Survey one-quarter sections. For travel forecasting purposes each center is defined by its location within one or more traffic analysis zones.

Source: SEWRPC.

Monitoring data indicated that for most centers the actual 1990 total commercial employment levels closely approximated levels planned for the year 2000. However, substantial deviation between actual and planned employment levels was recorded at four major commercial centers: Mayfair, with 3,400 more jobs than planned; West Allis, with 4,000 more jobs than planned; Brookfield-Blue Mound Road, with 18,500 more jobs than planned; and the Waukesha central business district, with 3,900 fewer jobs than planned.

Monitoring data also indicated that seven additional major commercial centers may emerge at sites not proposed in the year 2000 regional land use plan (see Map 10). At five of these sites, Mequon, Milwaukee-Park Place, Pewaukee,



The year 2000 regional land use plan envisioned that 16 major commercial centers would serve the Region. Major commercial centers are classified into two types: retail centers and office centers. By 1990, all the major commercial centers that existed in 1970, with the exception of the Mitchell Street Center in the City of Milwaukee, met both an office and retail center employment criteria. Two newly planned centers, Northridge and Racine West, had been established by 1990 and met the threshold employment levels for classification as major retail centers. Monitoring data also indicated that seven additional major commercial centers may emerge at locations not proposed in the year 2000 regional land use plan.

Source: SEWRPC.

Kenosha-West, and Kenosha-Southwest, development of commercial office and/or retail centers was under way by 1990. At the sixth site, Milwaukee County Research Park in Wauwatosa, development plans have been announced and the lands committed by Milwaukee County as the site of an office park.⁸ At the seventh site, the Pabst Farms in the Oconomowoc area, development plans have recently been announced, but have not yet received required public approvals.

Outdoor Recreation Centers: Major public outdoor recreation centers, or regional parks, are defined in the adopted regional land use plan as multi-use outdoor recreation areas with a minimum site area of 250 acres. The regional land use plan recommended that 29 regional parks be provided to serve the Region by the year 2000 as shown in Table 22 and on Map 11. Of that total, 27 park sites were fully or partially acquired by 1970. At one of the two remaining sites, Cliffside in Racine County, additional acquisition was required to bring the site up to the threshold 250acre size by 1990. Major park status was achieved at the other planned regional park, Monches in Waukesha County, when, in 1991, the County acquired an additional 90 acres.

Table 22 shows that two additional planned regional park sites, Sugar Creek in Walworth County and Paradise Valley in Washington County, had not been acquired by 1990. Finally, the monitoring indicated that an additional major park site, not included in the year 2000 regional land use plan, had been acquired by 1990. That site, Brookfield-Mitchell, approximated 260 acres in 1990, and was acquired and planned by the City of Brookfield for major park status.⁹

⁹The Brookfield-Mitchell site has been included as a planned major outdoor recreation center in the third-generation regional land use plan. See SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin–</u> 2010, January 1992.

PLAN IMPLEMENTATION

The foregoing description of urban development in the Region since 1970 is intended to assist in evaluating the travel forecasts made for the design of the adopted regional transportation system plan. The consistency between the forecast and actual travel levels to be described later must be viewed in light of the extent to which urban development occurred in a dispersed fashion contrary to land use plan recommendations. A sound evaluation of the travel forecasts also requires an evaluation of the state of implementation of the regional transportation system plan itself.

As previously indicated, the adopted regional transportation system plan contained recommendations in three component areas: arterial streets and highways, public transit, and transportation management measures. Substantial implementation progress remains to be made in each of the component areas. As indicated previously, the adopted year 2000 plan recommended that before extensive improvements were to be made to the arterial street and highway system, the public transit system was to be greatly improved and expanded and the recommended transportation management measures were to be implemented to make the most efficient use of existing arterial street and highway capacity. Implementation of the public transit recommendations has been poor; the transportation management system, despite actions taken by the Wisconsin Department of Transportation, remains essentially underdeveloped. The status of transportation system implementation is described below.

Arterial Streets and Highways

The adopted plan called on the Federal, State, and constituent units of government both to realign jurisdictionally and functionally improve the arterial street and highway system.

Jurisdictional Realignment: The implementation status of the jurisdictional realignment element of the regional transportation plan by county is summarized in Table 23, and the changes made as of 1991 are shown on Map 12. Based on plan schedules, about one-half of the 1,387 miles planned for jurisdictional transfer should have been completed by 1990. A summary of the status of the planned realignments among each jurisdiction follows.

⁸The commercial centers at Mequon, Milwaukee-Park Place, Pewaukee, Kenosha West, and Milwaukee County Research Park have been included as planned major commercial centers in the third-generation regional land use plan. See SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin–</u> 2010, January 1992.

· · · · · · · · · · · · · · · · · · ·		·		
		Site Area		Meets Site Area Criterion for
	Existing	Planned	Estimated	Designation as a
Major Regional Park	1970	2000	1990	Major Park in 1990 ^a
Fully Acquired in 1970	-	-		
Brighton Dale	360	360	360	Yes
Petrifying Springs	360	360	360	Yes
Harrington Beach	640	640	640	Yes
Hawthorne Hills	290	290	290	Yes
Mee-Kwon	240	240	240	Yes
Ela	240	240	240	Yes
Johnson	360	360	360	Yes
Brown Deer	370	370	370	Yes
Dretzka	330	330	330	Yes
Greenfield	300	300	300	Yes
Lake Michigan-North	420	420	420	Yes
Lake Michigan-South	840	840	840	Yes
	310	310	310	Yes
Oakwood	280	280	280	Yes
Whitnall	640	640	640	Yes
Whitewater Lake	250	250	250	Yes
Menomonee	400	400	400	Yes
Minooka	300	300	300	Yes
Mukwonago	220	220	220	Yes
Naga-Waukee	420	420	420	Yes
Ottawa Lake	220	220	220	Yes
Planned 2000, Partially Acquired in 1970	260	260	260	X
	260	360	260	Yes
	310	440	310	Yes
	220	540	220	NO
Big Foot Beach	270	330	270	Yes
Pike Lake	680	740	680	Yes
Monches [~]	190	440	320	Yes
Planned 2000, Not Acquired by 1985				
Sugar Creek		310		No
Paradise Valley		450		No
Unplanned, Acquired by 1985				
Brookfield-Mitchell	·		533	Yes

STATUS OF MAJOR OUTDOOR RECREATION CENTERS IN THE REGION: 1990

^aA major regional park is defined as a publicly owned site of at least 250 acres that provides opportunities for a variety of resource-oriented outdoor recreational activities, such as camping, beach swimming, and golf. In the plan implementation study, attention was focused on the status of public land acquisition to accommodate the proposed parks, recognizing that facility development at the proposed sites could proceed at a later date. It should be noted that the Ela, Mee-Kwon, Mukwonago, and Ottawa Lake sites abut existing parkway lands or lands recommended for parkway acquisition. The area of the site proper in conjunction with such associated existing or proposed parkway lands exceeds 250 acres.

^bThe Monches Regional Park met the criteria for designation as a "major regional park" in 1991 when Waukesha County acquired a 90-acre addition to the existing site.

Source: SEWRPC.



The year 2000 regional land use plan envisioned a system of 29 regional parks providing opportunities for such recreational activities as camping, picnicking, and swimming. Of the 29 planned regional parks, 21 had been fully acquired by 1990 and five had been partially acquired by 1990. The remaining two planned recreational centers, Paradise Valley and Sugar Creek, had not been acquired by 1990.

IMPLEMENTATION STATUS OF JURISDICTIONAL REALIGNMENTS RECOMMENDED IN THE ADOPTED YEAR 2000 REGIONAL TRANSPORTATION SYSTEM PLAN BY COUNTY: 1991

Type of Transfer	Kenosha County	Milwaukee County	Ozaukee County	Racine County	Walworth County	Washington County	Waukesha County	Region Total
Transfers to State Trunk System								
From County Trunk System								
Planned (miles)	6.5	110	12	202	10.0		100	05.0
	0.5		4.3	20.2	19.0	8.0	18.9	95.9
	0.0	0.0	1.0	5.6	18.3	0.0	16.1	41.6
	0.0	0.0	37.2	19.9	96.3	0.0	85.2	43.4
From Local System		10.0						
	0.8	10.2	4.0	4.4	2.6	4.2	1.2	27.4
Completed (miles)	0.0	1.9	0.0	1.3	0.1	2.7	0.3	6.3
Percent Completed	0.0	18.6	0.0	29.5	3.8	64.3	25.0	23.0
							.	
Planned (miles)	7.3	22.2	8.3	32.6	21.6	12.2	20.1	123.3
Completed (miles)	0.0	1.9	1.6	6.9	18.4	2.7	16.4	47.9
Percent Completed	0.0	8.6	19.3	21.2	85.2	22.1	81.6	38.8
Transfers to County Trunk System From State Trunk System		×						
Planned (miles)	34.7	54.7	86.4	36.7	30.3	46.9	53.4	343.1
Completed (miles)	0.0	0.0	20.0	2.3	21.9	7.8	22.3	74.3
Percent Completed	0.0	0.0	23.1	5.7	72.3	16.6	41.8	21.7
From Local System								
Planned (miles)	20.0	86.4	29.5	37.3	59.2	53.9	78.6	364.9
Completed (miles)	0.0	9.0	1.5	0.1	0.0	0.0	20.5	31.1
Percent Completed	0.0	10.4	5.1	0.3	0.0	0.0	26.1	8.5
Total								
Planned (miles)	56.7	141.1	115.9	74.0	89.5	100.8	132.0	708.0
Completed (miles)	0.0	9.0	21.5	2.4	21.9	7.8	42.8	105.4
Percent Completed	0.0	6.4	18.6	3.2	24.5	7.7	32.4	14.9
				-				
Fram State Truck Sustem								-
Planned (miles)	0.0	12.0	0.0		10	0.7	10 5	45.0
	0.0	12.0	0.0	4.4	1.2	8.7	18.5	45.6
	0.0	0.5 FO.0	0.0	1.5	0.6	4.9	18.5	32.0
		50.8		10.0	-50.0	56.3	100.0	56.9
From County Trunk System	10.0							
	13.0	15.1	2.9	11.3	1.0	5.4	91.6	140.3
	0.0	3.2	2.1	2.5	0.0	0.0	88.6	96.4
	0.0	21.2	/2.4	22.1	0.0	0.0	96.7	68.7
	10.0			45 -				407 -
	13.0	27.9	2.9	15./	2.2	14.1	110.1	185.9
Completed (miles)	0.0	9.7	2.1	4.0	0.6	4.9	107.1	128.4
Percent Completed	0.0	34.8	72.4	25.5	27.3	34.6	97.3	69.1
Transfers to Local Nonarterial System	1							
From State Trunk System								
Planned (miles)	0.0	1.1	0.0	2.1	9.0	3.7	4.1	20.0
Completed (miles)	0.0	0.0	0.0	0.0	0.6	0.6	0.4	1.6
Percent Completed		0.0		0.0	6.7	16.2	9.8	8.0
From County Trunk System					l			-
Planned (miles)	97.2	3.3	13.7	21.4	26.6	52.8	134.5	349.5
Completed (miles)	0.0	0.0	0.0	2.5	0.0	0.0	78.5	81.0
Percent Completed	0.0	0.0	0.0	11.7	0.0	0.0	58.4	23.2
Total	1				-			
Planned (miles)	97.2	4.4	13.7	23.5	35.6	56.5	138.6	369.5
Completed (miles)	0.0	0.0	0.0	2.5	0.6	0.6	78.9	82.6
Percent Completed	0.0	0.0	0.0	10.6	1.7	1.1	56.9	22.4

Source: SEWRPC.



On the basis of the year 2000 regional transportation plan schedules, about one-half of planned jurisdictional transfers should have been completed by 1990. By 1990, nearly 48 arterial miles, or 39 percent of planned miles, had been transferred to the State trunk system; about 105 miles, or 15 percent of planned miles, had been transferred to the county trunk system; about 128 arterial miles, or 69 percent of planned miles, had been transferred to the local arterial system; and approximately 83 miles, or 22 percent of planned miles, had been transferred to the local nonarterial system from the State or county trunk systems.

- 1. Of the approximately 123 miles to be transferred to the State trunk system from the county trunk system and the local arterial system nearly 48 miles, or 39 percent, have been completed.
- 2. Of the approximately 708 miles to be transferred to the county trunk system from the State trunk system and the local arterial system 105 miles, or 15 percent, have been completed.
- 3. Of the approximately 186 miles to be transferred to the local arterial system from the State trunk system and the county trunk system 128 miles, or 69 percent, have been completed.
- 4. Of the approximately 370 miles to be transferred to the local nonarterial system from the State trunk system and the county truck system nearly 83 miles, or 22 percent, have been completed.

<u>Functional Improvements</u>: Progress, as of 1991, toward implementing the functional improvements to the arterial street and highway system as recommended in the adopted year 2000 plan is summarized by jurisdictional system and by county in Table 24 and shown on Map 13.

As already noted, system preservation is defined in the adopted regional transportation system plan as the resurfacing and reconstruction, for the same capacity, necessary to maintain existing arterial street and highways. The regional plan envisioned that by the year 2000 system preservation work would be completed on about 2.941 miles of the arterial street and highway system. As of 1991, work on 706 miles, or nearly 24 percent of the miles planned for system preservation, had actually been completed. The greatest implementation of system preservation within the Region, in both relative terms and in absolute miles, has been accomplished on the State trunk highways. System preservation work on nearly 44 percent of the 844 miles of State trunk highways planned for such work had been completed by 1991. This progress can be compared with about 16 percent of the 1,473 miles of county trunk highways, and about 15 percent of the approximately 624 miles of local arterials planned for system preservation.

As already noted, system improvement is defined in the adopted regional transportation

system plan as the widening of existing arterial streets and highways to provide additional traffic capacity. The plan envisioned that by the year 2000 system improvement work would have been completed on about 401 miles of the arterial streets and highways within the Region. As of 1991, work on 123 miles, or nearly 31 percent of the miles planned to be widened for added capacity, had been completed. Among the jurisdictional categories, the greatest relative implementation of system improvement occurred on local arterials as local governments, especially the City of Milwaukee, addressed the need to reconstruct arterial streets to improved rather than existing capacities. By 1991, system improvement work on about 52 percent of the planned local arterials had been completed while work on nearly 36 percent of the planned county trunk highways and about 25 percent of the planned State trunk highways had been completed.

As already noted, system expansion is defined in the adopted regional transportation system plan as the construction of new arterial streets and highways. The plan envisioned that by the year 2000, 236 miles of new roadway would have been added to the regional arterial street and highway system. As of 1991, construction of 69 miles, or 29 percent of the planned new miles, had been completed. The bulk of system expansion occurred at the State jurisdictional level. Indeed, nearly 93 percent of all new arterial mileage was jurisdictionally classified as State trunk highway. By 1991, 64 miles, or about 42 percent, of the planned State trunk additions were in place. compared with 6 percent of the planned county trunk miles and about 6 percent of the planned local arterial miles.

Of the 64 miles of new State highways constructed between 1978 and 1991, almost 13 miles consisted of the addition to USH 45, the West Bend freeway. Table 25 shows the implementation status of planned freeway expansions in the Region. Since the adoption of the secondgeneration regional transportation system plan, the planned West Bend freeway expansion, as well as the stub-end treatments of the Lake Freeway-North, the Park Freeway-East, the Stadium Freeway-North, the Park Freeway-West, and the Stadium Freeway-South, have all been completed. Construction of the Lake Arterial from the south end of the Daniel Hoan Memorial Bridge to E. Layton Avenue is currently under way.





The regional plan envisioned that by the year 2000 system preservation work would be completed on about 2,941 miles of the arterial street and highway system. As of 1990, work on 706 miles, or nearly 24 percent of the miles planned for system preservation, had been completed. The plan also envisioned that by the year 2000 system improvement work would have been completed on about 401 miles of the arterial street and highway system. As of 1990, work on 123 miles, or nearly 31 percent of the miles planned for system improvement, had been completed. Finally, the plan envisioned that by the year 2000 system improvement, had been completed. Finally, the plan envisioned that by the year 2000 system improvement, had been completed. Finally, the plan envisioned that by the year 2000 system. As of 1990, work on 123 miles, or nearly 31 percent of the miles planned for system improvement, had been completed. Finally, the plan envisioned that by the year 2000 system. As of 1990, construction of 69 miles, or 29 percent, of the planned new miles had been completed. According to plan schedules, work on all of the miles planned for system improvement and expansion should have been completed by 1990. However, systemwide monitoring has shown that work on only 29 percent of the miles planned for system improvement and system expansion had been completed, while work on only 25 percent of the miles planned for system preservation had been completed, while work on only 25 percent of the miles planned for system improvement and system expansion had been completed, while work on only 25 percent of the miles planned for system preservation had been completed.

IMPLEMENTATION STATUS OF PLANNED FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM AS SET FORTH IN THE DESIGN YEAR 2000 PLAN:^a 1991

		System	Preservation			
	S	ate Trunk Highways	b	Co	unty Trunk Highway	s
County	Planned Miles	Implemented Miles	Percent	Planned Miles	Implemented Miles	Percent
Kenosha Milwaukee Milwaukee Ozaukee Racine Naworth Walworth Nashington Washington Nashington	58.6 200.6 49.2 125.6 162.5 90.5 157.1	13.0 99.4 32.1 37.8 82.8 20.9	22.2 49.6 65.2 30.2 51.0 20.2	165.0 158.2 163.7 149.5 235.8 224.1	36.5 68.1 29.2 30.5 26.6 26.2	22.1 43.0 17.8 20.4 11.3 11.7
Region	844.1	372.9	55.3 44.2	377.0 1,473.3	21.4	5.7 16.2

		System	Preservation					
		Local Arterials			Total			
County	Planned Miles	Implemented Miles	Percent	Planned Miles	Implemented Miles	Percent		
Kenosha	37.9	3.0	7.9	261.5	52.4	20.0		
Milwaukee	300.5	67.6	22.5	659.3	235.1	35.7		
Ozaukee	48.0	3.2	6.7	260.9	64.5	24.7		
Racine	67.3	5.7	8.5	342.4	75.8	22.1		
Walworth	5.0	0.6	12.0	403.3	110.0	27.3		
Washington	66.5	5.4	8.1	381.1	52.4	13.7		
Waukesha	98.8	7.3	7.4	632.9	115.6	18.3		
Region	624.0	92.8	14.9	2,941.4	705.9	24.0		

System Improvement								
	St	ate Trunk Highways	b	Co	County Trunk Highways			
County	Planned Miles	Implemented Miles	Percent	Planned Miles	Implemented Miles	Percent		
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	36.9 31.1 21.4 36.8 13.7 48.2 49.0	13.5 9.3 5.8 8.3 1.3 0.0 19.9	36.6 29.9 27.1 22.6 9.5 0.0 40.6	26.0 33.8 2.0 16.0 4.6 4.4 36.4	12.3 10.1 0.0 2.2 0.3 0.5 18.7	47.3 29.9 0.0 13.8 6.5 11.4 51.4		
Region	237.1	58.1	24.5	123.2	44.1	35.8		

	System Improvement							
		Local Arterials			Total			
County	Planned Miles	Implemented Miles	Percent	Planned Miles	Implemented Miles	Percent		
Kenosha	2.4	1.5	62.5	65.3	27.3	41.8		
Milwaukee	23.5	15.5	66.0	88.4	34.9	39.5		
Ozaukee	0.0	0.0	0.0	23.4	5.8	24.8		
Racine	6.6	0.9	13.6	59.4	11.4	19.2		
Walworth	0.4	0.2	50.0	18.7	1.8	9.6		
Washington	0.3	0.3	100.0	52.9	0.8	1.5		
Waukesha	7.2	2.6	36.1	92.6	41.2	44.5		
Region	40.4	21.0	52.0	400.7	123.2	30.7		

Table 24 (continued)

System Expansion							
	St	ate Trunk Highways	b	Cou	inty Trunk Highway	'S	
County	Planned Miles	Implemented Miles	Percent	Planned Miles	Implemented Miles	Percent	
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	16.7 15.3 18.9 16.4 48.0 19.9 17 2	0.0 4.3 18.9 1.2 26.7 13.1 0.0	0.0 28.1 100.0 7.3 55.6 65.8 0.0	7.5 3.0 0.0 7.7 3.9 11.2 12.5	2.0 0.0 0.4 0.0 0.4 0.0	26.7 0.0 5.2 0.0 3.6 0.8	
Region	152.4	64.2	42.1	45.8	2.9	6.3	

System Expansion								
		Local Arterials			Total			
County	Planned Miles	Implemented Miles	Percent	Planned Miles	Implemented Miles	Percent		
Kenosha	3.7	0.0	0.0	27.9	2.0	7.2		
Milwaukee	5.0	1.4	28.0	23.3	5.7	24.5		
Ozaukee	4.0	0.7	17.5	22.9	19.6	85.6		
Racine	5.1	0.0	0.0	29.2	1.6	5.5		
Walworth	9.3	0.0	0.0	61.2	26.7	43.6		
Washington	9.3	0.2	2.2	40.4	13.7	33.9		
Waukesha	1.7	0.0	0.0	31.4	0.1	0.3		
Region	38.1	2.3	6.0	236.3	69.4	29.4		

^aAdopted Year 2000 Regional Transportation System Plan, as amended by the removal of the Stadium Freeway-South, Lake Freeway-North, Park Freeway-East, and by the substitution of the Lake Arterial for the Lake Freeway-South; and prior to its amendment by the following: <u>Amendment to the Walworth County Jurisdictional Highway System Plan – 2010</u>, March 1992; <u>Amendment to the Ozaukee</u> <u>County Jurisdictional Highway System Plan – 2010</u>, January 1993; <u>Amendment to the Racine County Jurisdictional Highway System</u> <u>Plan – 2000</u> (Recommendations for the proposed City of Burlington Bypass – 2010), December 1990; SEWRPC Community Assistance Planning Report No. 200, <u>A Land Use and Transportation System Development Plan for the IH 94 South Freeway Corridor, Kenosha,</u> <u>Milwaukee, and Racine Counties</u>, January 1992; SEWRPC Community Assistance Planning Report No. 151, <u>A Transportation System Plan</u> for the Blue Mound Road (USH 18) Corridor, December 1987.

^bState trunk highways include both freeways and arterials.

Source: SEWRPC.

Significant variation has occurred among the counties in the implementation of the planned functional improvements to the arterial street and highway system, as shown in Table 24. With respect to system preservation, the greatest degree of implementation occurred in Milwaukee County, where work on 235 miles, or about 36 percent, of the planned miles had been completed. The least degree of implementation of system preservation occurred in Washington County, where work on 52 miles, or 14 percent, of the planned miles had been completed. With regard to system improvement, the greatest implementation, in terms of miles completed, occurred in Waukesha County, where work on 41 miles was completed. In terms of percent completed, in Kenosha County, work on 42 percent of the planned 65 miles had been completed. The poorest record of implementation of system improvement occurred in Washington County, where work on less than one mile of the 53 planned miles had been completed. With respect to system expansion, the greatest degree of implementation, in terms of miles completed,

IMPLEMENTATION STATUS OF PLANNED FREEWAYS IN THE REGION: 1978-1991

Freeway Facility	Planned 1978 (miles)	Constructed 1991 (miles)
USH 12 STH 16 USH 41 USH 45-West Bend	17.0 5.4 21.0 12.7	0.0 0.0 0.0 12.7
Total	56.1	12.7

Source: SEWRPC.

occurred in Walworth County, where about 27 miles were added to the arterial system. In terms of percent completed, in Ozaukee County nearly 86 percent of the 23 planned miles were added to the arterial system. The poorest implementation of system expansion occurred in Waukesha County, where less than one mile of the approximately 31 planned miles had been added to the arterial system as recommended in the adopted plan.

In summary, the pace of functionally improving the arterial street and highway system has fallen behind schedule. Indeed, on the basis of plan schedules, work on about one-half of the miles planned for system improvement and expansion should have been completed by 1990. However, systemwide monitoring indicated that work on only about 30 percent of the miles planned for system improvement and system expansion, those functional improvements designed to add capacity to the arterial street and highway system, had been completed. With respect to system preservation, work on only 24 percent of the planned miles had been completed by 1991. A summary of the implementation of the combined functional improvements occurring in each county, is shown in Table 26. By 1991, work on only one in four miles in the Region planned for functional improvement had been completed.

Public Transit

Overall, the adopted plan envisioned a significantly expanded and improved transit system. In particular, it was envisioned that the rapid transit element would serve a wide variety of trips and provide a competitive and attractive

Table 26

SUMMARY IMPLEMENTATION STATUS OF TOTAL PLANNED FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM: 1991

	Entire Arterial Street and Highway System					
		Implei	mented			
County	Planned Miles	Miles	Percent			
Kenosha	354.7	81.7	23.0			
Milwaukee	771.0	275.7	35.8			
Ozaukee	307.2	89.9	29.3			
Racine	431.0	88.8	20.6			
Walworth	483.2	138.5	28.7			
Washington	474.4	66.9	14.1			
Waukesha	756.9	156.9	20.7			
Region	3,578.4	898.4	25.1			

Source: SEWRPC.

alternative to the automobile. By 1991, the number of vehicle-miles of transit service provided had not increased over the 1972 levels; and the average speed of a transit trip had remained about the same as in 1972. In part for these reasons, the increase in transit use envisioned in the plan has not been achieved. Indeed, transit use has declined from about 4 percent of all trip making in the Region in 1972 to 3 percent in 1991. The implementation status of the transit recommendations is shown in graphic summary form on Map 14 and is described below.

As already noted, the proposal to provide "light" rail rapid transit in the northwest corridor of Milwaukee County was evaluated in depth in the 1988 alternative analysis conducted by the Commission for Milwaukee County. This study resulted in a decision by the Milwaukee County Board and County Executive that a high level of transit service in this corridor be provided initially, not by "light" rail, but by express bus. Milwaukee County, with Federal financial assistance, implemented the express bus service, called "Metrolink," in 1992.

Although no action has been taken to implement the commuter rail recommendation contained in the adopted plan, significant steps have been taken to implement the express bus on freeway recommendations. Express bus service over freeways is currently provided by 15 routes operating over a total of about 519 round-trip route miles. Of the existing routes, 14 routes



In total, the adopted plan recommended that public transit serve an aggregate area of 500 square miles and about 75 percent of the regional population. Significant steps have been taken to implement the freeway flyer rapid transit bus service. The plan recommended the operation of 24 freeway flyer routes and the provision of 56 public transit stations. Freeway flyer rapid transit is currently provided over 15 routes and serves 25 transit stations. In addition, since the adoption of the year 2000 plan, local public transit service has been extended within the Milwaukee, Racine, and Kenosha urbanized areas and shared-ride taxicab service has been provided in the urban areas of Whitewater, Hartford, and West Bend. Currently, public transit service is provided to an aggregate area of about 225 square miles and about 69 percent of the total resident population of the Region.

operating over a total of about 492 round-trip route miles have been implemented in accordance with plan recommendations. The remaining existing route has been implemented to provide special service for the University of Wisconsin-Milwaukee. All the express bus routes currently in operation, with the exception of one route in Milwaukee County, provide service only during weekday peak periods. The exception operates between the Northridge Shopping Center and the Milwaukee central business district over W. Brown Deer Road and IH 43 and provides service from about 5:30 a.m. until 7:00 p.m. each weekday.

The 15 existing routes serve 25 transit stations within Milwaukee and Waukesha Counties. A total of 19 of the stations served, including 17 with park-ride lots and two without parking lots, have been publicly constructed. Of these 19 publicly constructed transit stations, 16 have been located essentially as proposed by the express bus-on-freeway element of the plan. Two stations, Northridge and Timmerman Field, were envisioned under the adopted plan to be served ultimately by "light" rail transit, but are currently being served by an express bus route. The USH 41 and Pilgrim Road public transit station has been developed outside plan recommendations to meet current needs. The remaining six of the 25 existing stations served include four at privately owned retail establishments located close to where public facilities were proposed under the plan, three at private shopping center lots with parking and one at a retail store without parking. The remaining two stations served are in private shopping center lots at locations which do not correspond to any proposed station locations under the plan. In addition to the 19 public transit stations which are currently served by express bus service, two public transit stations which are not currently being served by any express bus routes have also been constructed at locations recommended under the plan, USH 45 and Good Hope Road and MATC in Mequon, and are currently being utilized as carpool parking facilities.

Since adoption of the system plan, local public transit service has continued to be provided and extended within the Milwaukee, Racine, and Kenosha urbanized areas. In addition, public transit in the form of shared-ride taxicab service has been provided in three smaller urban areas of the Region, Whitewater, Hartford, and West

Figure 11



Source: SEWRPC.

Bend. Public transit service is now provided to an aggregate of about 225 square miles, or about 8 percent of the total area of the Region, and to about 69 percent of the total resident population of the Region.

An important quantitative measure of the degree of transit plan implementation is the level of vehicle-miles of transit service provided. The plan envisioned that by 1991 transit operators in the Region would together increase the number of vehicle-miles of service by about 77 percent, from about 20 million in 1972 to nearly 36 million vehicle-miles in 1991. As shown in Table 27 and Figure 11, the actual level of revenue vehicle-miles provided remained relatively stable. Thus, the actual 1991 level was below the planned level by about 43 percent. The provision of vehicle-miles of service peaked in 1981 at a level of about 24 million.

The plan also envisioned that by the year 2000 the public transit fleet would consist of over 1,100 vehicles operating over about 2,296 roundtrip route miles. The operating characteristics of public transit service provided within each urbanized area in the Region are set forth in Table 28. In 1972, the base year for the adopted plan, the transit fleet consisted of a total of 565 buses operating over 1,201 round-trip route miles. As of 1991, the public transit fleet consisted of 662 buses operating over 2,066 round-

ACTUAL AND FORECAST ANNUAL REVENUE VEHICLE-MILES OF PUBLIC TRANSIT SERVICE IN THE REGION BY URBANIZED AREA: 1972 AND 1991

	Existing	Forecast 1991 (millions)		Actual 1991 (millions) ^a		Difference between Actual and Forecast 1991 Annual Revenue Vehicle-Miles	
Urbanized Area	1972 (millions)	Change 1972-1991	Total	Change 1972-1991	Total	Number (millions)	Percent
Kenosha Milwaukee Racine	0.3 19.4 0.5	1.5 12.8 1.4	1.8 32.2 1.8	0.4 -1.1 0.8	0.7 18.3 1.3	-1.1 -13.9 -0.6	-59.1 -43.2 -31.1
Region	20.2	15.6	35.8	0.1	20.3	-15.5	-43.3

^aExcludes service on shared-ride taxicab systems outside the three urbanized areas within the Region.

Source: SEWRPC.

Table 28

OPERATING CHARACTERISTICS OF PUBLIC FIXED-ROUTE TRANSIT SERVICE IN THE REGION BY URBANIZED AREA: 1972, 1991, AND YEAR 2000 ADOPTED PLAN

	Kenosha Urbanized Area			Milwaukee Urbanized Area			Racine Urbanized Area			Total Region		
Transit Service Characteristic	Existing 1972	Existing 1991	Planned 2000	Existing 1972	Existing 1991	Planned 2000	Existing 1972	Existing 1991	Planned 2000	Existing 1972	Existing 1991	Planned 2000
Round-Trip Route Miles	59	171	147	1,061	1,954	2,641	81	171	153	1,201	2,296	2,941
Peak-Period Vehicle Requirements Buses	12	28 	42	442 	471 	872 24 27	10 	31 	38 	464 	530 	952 24 27
Subtotal	12	28	42	442	471	923	10	31	38	464	530	1,003
Total Fleet (including spares)	18	35	46	533	585	1,016	14	42	42	565	662	1,104

Source: SEWRPC.

trip miles. As shown in Table 28, substantial progress has been made in the provision of round-trip route miles of service.

Transportation System Management

The recommended transportation system management element has been only modestly implemented since the plan was adopted and substantial implementation actions related to freeway traffic management remain to be accomplished. The recommended freeway traffic management system for the Milwaukee area is intended to furnish freeway users with information concerning traffic conditions with the potential of interrupting traffic flow and to provide for freeway operational control. Recommendations regarding the provision of such advisory information and incident management measures remain to be implemented.

The operational control element of the freeway management system is intended to regulate and monitor freeway access during peak-period travel hours and thereby reduce freeway traffic congestion, improve traffic flow, and facilitate the movement of transit and other highoccupancy vehicles. The Wisconsin Department of Transportation has recently begun the installation of additional freeway on-ramp metering equipment and the replacement of old on-ramp meter controllers with equipment utilizing microprocessor-based equipment. The upgraded equipment has the capability to store trafficrelated data in support of systemwide freeway management. Currently 26 on-ramp meters are in place at various freeway access points in the Milwaukee area, 18 of which utilize this updated technology. However, the number and extent of the meters is currently insufficient to permit control of freeway conditions systemwide. The adopted plan also called for the exclusive bypass of such on-ramps meters for busses and other high-occupancy vehicles. Currently, buses are granted preferential access to the freeway system at four on-ramp locations and exclusive access at two locations.

With respect to carpool facilities and promotion, significant progress has been made in the provision of carpool parking facilities throughout the Region. However the envisioned utilization of the facilities, as discussed in Chapter IV, and of car and van pooling has not been achieved. The adopted year 2000 plan recommended the provision of off-street carpool parking facilities at key freeway interchanges as demand warranted. There were six such facilities in 1978 and in 1991 there were 18 (see Map 15). Three of 12 new lots are located at interchanges specifically suggested in the year 2000 plan: STH 60 and CTH P; USH 41 and CTH Y; and STH 16 and CTH C.

A continuing program of providing computer matching of commuters interested in carpooling is also in place. However, the recommended program to market carpooling aggressively to area commuters and employers and to encourage the provision of preferential parking to carpool participants has not operated since 1981.

With respect to the recommended parking measures, action has been taken only recently by the City of Milwaukee to study some of the recommendations, including a change in the parking rate structure to discourage long-term parking, and adjusting the supply of parking in downtown Milwaukee. From 1968 to 1988, contrary to plan recommendations, the parking supply in downtown Milwaukee increased by approximately 34 percent, from 35,300 spaces to 47,200 spaces, substantially faster than the increase in employment in downtown Milwaukee. The percentage of total travel to, from, and within downtown Milwaukee by public transit is estimated to have declined over that same 20-year period from approximately 25 percent to 15 percent.

PLAN EVALUATION: FORECAST AND ACTUAL TRIP MAKING LEVELS

The surveillance activities undertaken by the Commission with respect to the travel forecasts used in the design of the adopted regional transportation system plan revealed the following.

Internal Person Trips

The term "internal person trips" is defined by the Commission as the total number of trips made by any transportation mode within the Region on an average weekday. Internal person trips thus includes trips made as a passenger in a personal vehicle. The forecast is derived in the first step of the regional traffic simulation procedure, trip generation, through the quantification of relationships between land use and transportation. The forecast variables evaluated include population, households, employment, land use distribution, and automobile availability. These were used as inputs in this modeling procedure.

The forecast of internal person trips in the Region underlying the regional transportation plan had envisioned an increase in the number of internal person trips of about 20 percent by 1991 as shown in Table 29 and Figure 12. By 1991, the number of internal person trips had risen by about one million trips, or 23 percent, over the 1972 base level. Thus, at the regional level, the actual number of internal person trips exceeded the forecast level by about 155,000 trips, or by about 3 percent.

Internal Vehicle Trips

The term "internal vehicle trips" is defined by the Commission as the total number of vehicle trips made on an average weekday within the Region. The forecast level is derived in the third step of the traffic simulation procedure, modal choice, in which the total number of person trips, forecast in the trip generation step, is divided among the travel modes. Variables affecting the forecast of internal vehicle trips include, in addition to those affecting internal person trips, in-vehicle travel time, out-of-vehicle travel times, and cost differences between the automobile and transit modes.

The forecast of internal vehicle trips underlying the regional transportation system plan had envisioned an increase of 596,700 trips, or about 21 percent, by 1991, as shown in Table 29 and Figure 13. By 1991, the actual level of internal



Significant progress has been made in the provision of carpool parking facilities throughout the Region. The adopted year 2000 plan called for the provision of such parking lots at key freeway interchanges as demand for ride-sharing warranted and identified six specific locations. There were six carpool parking facilities in place when the plan was adopted in 1978; currently there are 18. Three of the 12 new lots are located at interchanges specially identified in the adopted plan: STH 60 and CTH P, USH 41 and CTH Y, and STH 16 and CTH C.

ACTUAL AND FORECAST INTERNAL PERSON TRIPS, VEHICLE TRIPS, AND VEHICLE MILES OF TRAVEL IN THE REGION: 1972 AND 1991

Internal Trips		F	orecast 19	91	ŀ	Actual 199	Difference between Actual and Forecast 1991 Internal Trips		
		Change 1972-1991		Total	Change 197	2-1991			Total
	1972	Number	Percent	Number	Number	Percent	Number	Number	Percent
Person Trips	4,460,400	875,700	20	5,336,100	1,031,000	23	5,491,400	155,300	3
Vehicle Trips	2,884,700	596,700	21	3,481,400	1,176,100	40	4,060,800	579,400	17
Vehicle Miles of Travel	20,124,000	6,876,000	34	27,000,000	12,948,000	64	33,072,000	6,072,000	23

Source: SEWRPC.

Figure 12

ACTUAL AND FORECAST AVERAGE WEEKDAY INTERNAL PERSON AND VEHICLE TRIPS IN THE REGION: 1972-2000



Source: SEWRPC.

vehicle trips had increased by 1.2 million, or about 40 percent, over the 1972 level. Therefore, at the regional level, the actual number of internal vehicle trips exceeded the forecast level by nearly 17 percent.

Vehicle Miles of Travel (VMT)

"Vehicle miles of travel" is an estimate of the total miles driven on the arterial streets and

Figure 13

ACTUAL AND FORECAST AVERAGE WEEKDAY VEHICLE MILES OF ARTERIAL TRAVEL IN THE REGION: 1963-2000



Source: SEWRPC.

highways within the Region on an average weekday. The forecast of VMT is an output of the last step of the traffic simulation procedure, traffic assignment, in which total forecast vehicle trips are assigned to both existing and proposed arterial streets and highways and the resulting mileage calculated. Travel times between transportation analysis zones for both automobile and public transit travel directly affect the forecast of VMT.

The forecast of VMT underlying the regional transportation system plan had envisioned an increase of about 6.9 million vehicle miles by 1991, as shown in Table 29 and Figure 13. By 1991, the actual level of VMT had increased over the 1972 base-year level by 12.9 million or by

		Foreca	st 1991	Actua	l 1991 ^a	Difference between Actual and Forecast 1991 Revenue Passengers	
Urbanized Area	Existing 1972	Change 1972-1991	Total	Change 1972-1991	Total	Number	Percent
Kenosha	503,200	4,096,800	4,600,000	624,800	1,128,000	-3,472,000	-75
Milwaukee	52,722,100	42,277,900	95,000,000	-5,455,000	47,267,100	-47,732,900	-50
Racine	525,700	4,574,300	5,100,000	1,302,100	1,827,800	-3,272,200	-64
Region	53,751,000	50,849,000	104,600,000	-3,528,100	50,222,900	-54,437,100	-52

ACTUAL AND FORECAST ANNUAL PUBLIC TRANSIT REVENUE PASSENGERS IN THE REGION BY URBANIZED AREA: 1972 AND 1991

^aExcludes ridership on shared-ride taxicab services outside the three urbanized areas within the Region.

Source: SEWRPC.

nearly 64 percent to a level of 33 million miles. Thus, the actual number of vehicle miles of travel in the Region exceeded the forecast level by five million miles, or nearly 22 percent.

Public Transit Ridership

The forecast level of public transit ridership underlying the regional transportation system plan had envisioned an increase of nearly 51 million revenue passengers by 1991, as shown in Table 30 and Figure 14. By 1991, the actual level of transit ridership in the Region had fallen below the 1972 base-year level of 54 million revenue passengers by 3.5 million, or by about 7 percent. Thus, the actual level of public transit ridership in the Region was below the forecast level by about 54.4 million revenue passengers, or 52 percent.

At the urbanized area level, the Kenosha and Racine urbanized areas both experienced increases over the 1972 base-year ridership level. The 1991 ridership level in the Kenosha urbanized area was 600,000 passengers greater than the 1972 level of 500,000. Transit ridership in 1991 in the Racine urbanized area was similarly greater than in 1972, by over 1.3 million revenue passengers. The increase in transit ridership in the Kenosha and Racine urbanized areas was, however, not significant enough to offset the significant declines in ridership experienced in the Milwaukee urbanized area. In the Milwaukee urbanized area, transit ridership in 1991 was about 5.5 million revenue passengers below the 1972 level. In each of the three urbanized areas, 1991 ridership levels were far below forecast levels, as shown in Figure 15.

Figure 14

ACTUAL AND FORECAST PUBLIC TRANSIT RIDERSHIP IN THE REGION: 1950-2000



Source. SEWARC.

Ridership on the freeway flyer service totaled about 1.1 million revenue passengers in 1991, as shown in Figure 16, down from a peak ridership level in 1980 of about 1.98 million revenue passengers. The increase in freeway flyer ridership between 1977 and 1980, as shown in Figure 16, may be attributed in part to the increase in motor fuel prices during that period of time.

Evaluation of the Forecasts

The foregoing review of the travel forecasts underlying the year 2000 regional transportation system plan reveals that the actual levels of travel in the Region as indicated by internal vehicle trips, vehicle miles of travel, and public transit ridership had not conformed to the levels forecast. By 1991, internal vehicle trips and VMT **REGION BY URBANIZED AREAS: 1950-2000**



Source: SEWRPC.

had increased to levels significantly higher than forecast, while public transit ridership decreased to a level far below the ridership forecast.

The actual level of internal person trips, however, varied only slightly from the forecast level, indicating that the increase in trip making in the Region between 1972 and 1991 was overFigure 16

MILWAUKEE URBANIZED AREA FREEWAY FLYER RIDERSHIP: 1964-1991



whelmingly the product of increases in personal vehicle use. Furthermore, the data indicate that the number of vehicle trips had increased faster than the number of person trips, indicating that single occupancy vehicle trips increasingly constituted a larger share of total trips in the Region. An indication of this trend is the decline in the vehicle occupancy rate between 1972 and 1991. The average vehicle occupancy rate for all trip purposes within the Region decreased by 11 percent: from 1.42 persons per automobile in 1972 to 1.26 by 1991. This trend toward increasing vehicle use is in part due to basic lifestyle changes in the Region, including declining household sizes and increasing labor participation rates among women.

Because vehicle trip making and VMT exceeded forecast levels and because implementation of planned transportation improvements lagged, higher than anticipated levels of traffic congestion have occurred. By 1991 the mileage of arterial streets and highways experiencing congestion had exceeded by 233 miles the level of 46 miles planned for the year 2000. Historical and current levels of traffic congestion in the Region are shown in Table 31 and on Map 16. Traffic congestion occurs when arterial facilities carry traffic volumes exceeding their design capacity. Traffic congestion typically occurs during the morning and evening peak-period travel hour, or in some cases, during the threehour morning and evening peak-traffic periods.

At both the regional level and among the counties, except Kenosha County, the number of miles experiencing congestion had increased
between 1972 and 1991 (see Table 31). The adopted plan envisioned that by the year 2000 a total of 46 miles, or just over 1 percent of the entire system, would experience traffic congestion on an average weekday. As of 1991, however, 279 miles, or nearly 9 percent of the entire arterial street and highway system mileage, experienced traffic congestion on an average weekday (see Map 16). Monitoring data reveals a similar variance in the planned and actual number of congested miles at the county level particularly in Milwaukee, Waukesha, and Racine Counties, as summarized in Table 31.

SUMMARY AND CONCLUSIONS

The design year 2000 regional transportation system plan was adopted by the Regional Planning Commission in June 1978. It consisted of three components: an arterial street and highway, including freeway, component; a public transit component; and a transportation system management component. The transit and transportation system management components were designed to minimize the need to expand arterial street and highway facilities by reducing the demand for automobile travel and by increasing the efficiency and effectiveness of the arterial street and highway system.

The transportation system plan was based upon, and designed to serve, the design year 2000 regional land use plan, also adopted in 1978. The adopted land use plan recommended a centralized urban settlement pattern with new urban growth occurring in concentric rings outward from, and contiguous to, existing urban centers. The planned centralization would make the efficient provision of public transit possible, increase the efficacy of ridesharing strategies, lessen the need for costly arterial street and highway improvements, and prevent severe traffic congestion. In turn, implementation of the transportation system plan would assist in orderly and rational land use development by facilitating access, particularly multi-modal access, to major regional activity centers and residential communities within the Region.

The design of the regional transportation system plan was based on forecasts of population, households, employment, and on automobile and truck availability. While the resident population in the Region did not increase to the levels forecast, the other determinants of the general scale of urban development, the number of households and jobs, closely conformed to forecast levels. The close conformity between the actual and forecast number of households is especially significant since the number of households is a particularly important factor in the generation of travel demand. A summary of the evaluation of the forecasts is set forth below.

- 1. From 1970 to 1990, the population of the Region was forecast to increase by 16 percent, to over two million persons. By 1990, the regional population had increased to a level of 1.8 million persons, only about 3 percent over 1970 level. Consequently, the 1990 population was lower than forecast by about 232,000 persons, or about 11 percent.
- 2. From 1970 to 1990, the number of households within the Region was forecast to increase by about 24 percent, to about 667,800 households. By 1990, the number of households had increased to about 676,000, or about 26 percent, over the 1970 level. Consequently, the 1990 regional household level exceeded the forecast level by about 8,200 households, or about 1 percent.
- 3. From 1970 to 1990, the number of jobs within the Region was forecast to increase by about 23 percent, to about 921,400 jobs. By 1990, despite a severe economic recession from 1979 to 1983, regional employment had increased to a level of about 990,300 jobs, or by about 31 percent. Consequently, the 1990 regional employment level exceeded the forecast level by about 65,900 jobs, or about 7 percent.
- 4. From 1970 to 1990, the number of automobiles available within the Region was forecast to increase by about 38 percent, to about 876,700 automobiles. By 1990, the number of automobiles within the Region had increased to about 927,300, or by about 46 percent. Consequently, the 1990 regional automobile availability level exceeded the forecast level by about 50,000, or about 6 percent.
- 5. From 1970 to 1990, the number of trucks available in the Region was forecast to increase by about 79 percent, to about

	Arterial Facilities Carrying Average Weekday Traffic Volumes Exceeding Design Capacity and Experiencing Traffic Congestion									
			Planned System							
	19	1963 1972 1991						2000		
Facility Type	Mileage	Percent of Total System	Mileage	Percent of Total System	Mileage	Percent of Total System	Mileage	Percent of Total System		
Kenosha County Milwaukee County Ozaukee County Racine County Walworth County Washington County Waukesha County	13.5 116.3 8.3 13.6 5.3 0.0 34.8	4.8 14.7 3.1 3.9 1.3 0.0 5.0	22.0 61.0 5.5 20.3 4.8 9.1 42.9	7.9 8.3 2.2 5.8 1.2 2.7 6.6	15.1 121.9 13.3 34.3 3.1 19.5 72.0	4.8 15.7 4.6 9.9 0.7 4.9 10.0	3.2 31.6 1.9 0.0 0.0 0.0 9.2	0.9 4.1 0.6 0.0 0.0 0.0 1.2		
Total	191.8	6.0	165.6	5.3	279.2	8.5	45.9	1.3		

ARTERIAL STREET AND HIGHWAY SYSTEM TRAFFIC CONGESTION ON AN AVERAGE WEEKDAY IN THE REGION: HISTORICAL AND PLANNED SYSTEMS

Source: SEWRPC.

128,000. By 1990, the number of trucks had increased to about 177,900 trucks, about 148 percent over the 1970 level. Thus, the 1990 regional truck availability level exceeded the forecast level by about 49,900, or about 39 percent.

6. A more complete evaluation of the automobile availability forecast must consider the use of motor trucks as personal vehicles within the Region. The automobile availability forecast conducted for the secondgeneration plan attempted to estimate the future availability of personal vehicles in the Region regardless of their type, automobile or personal-use truck. In this regard, when the number of motor trucks estimated to be used as personal vehicles in 1990 is added to the number of automobiles in 1990, the resulting number of total personal vehicles exceeds the forecast by slightly more than one million, or by nearly 15 percent.

The design of the regional transportation system plan was also based on a forecast land use distribution within the Region, making an evaluation of urban development activity essential to transportation plan reappraisal. An evaluation of incremental urban development between 1970 and 1985 indicated that about 48 percent of all new residential development, or about 66 square miles, occurred in a highly diffused fashion throughout the Region; that nearly 15 square miles more land was converted from rural use to urban use than the plan recommended; that about 70 percent of all housing units were constructed at a medium or high residential density though the plan called for 83 percent; that 27 percent of all new housing units were constructed at a low residential density, though the plan recommended 10 percent; and that overall urban population density decreased from 5,100 persons per square mile in 1970 to 3,900 by 1985, though the plan envisioned a population density of 4,100 persons per square mile.

In addition, total employment at the older major industrial centers, generally concentrated within the Milwaukee, Racine and Kenosha urbanized areas, declined by 21 percent though the plan recommended that employment at such centers increase by about 13 percent. Two industrial centers, with a combined total of 13,700 jobs, were developed, contrary to plan recommendations. Currently, four additional centers dispersed throughout the Region have the potential to reach the 3,500 industrial employment level necessary for classification as major industrial centers.

The progress of implementing the regional transportation system plan is set forth below.



ARTERIAL STREET AND HIGHWAY SYSTEM TRAFFIC



The adopted plan envisioned that by the year 2000, 46 miles, or 1 percent of total arterial mileage within the Region, would operate over design capacity on an average weekday and thereby exhibit traffic congestion. As of 1991, however, because of significant increases in vehicle trip making and a failure to implement planned transportation improvements more fully, 279 miles, or nearly 9 percent of total arterial mileage, were operating over design capacity. Nearly 44 percent of this congested mileage, or 122 miles, was located within Milwaukee County.

- 1. By 1990, about one-half of the mileage recommended in the plan for jurisdictional realignment and capacity improvement and expansion should have been implemented. As of 1991, however, only 26 percent of the 1,387 miles recommended for jurisdictional realignment had been appropriately transferred; work had been completed on only 31 percent of the 637 miles recommended for either system improvement or system expansion. The greatest progress in implementing the combined system improvement and expansion recommendations had occurred on the State trunk highway system at about 31 percent, followed by the local arterial system at about 30 percent, and the county trunk highway system at about 28 percent. Over all, work on only 24 percent of the miles planned for system preservation had been completed by 1991. All completed jurisdictional transfers and functional improvements were made as recommended in the adopted plan.
- 2. The adopted plan envisioned that by the year 2000 the public transit fleet would consist of about 1,000 transit vehicles, operating over 2,941 round-trip route miles. By 1991, the fleet consisted of 662 buses, operating over 2,296 round-trip miles. The plan envisioned that by 1991 approximately 36 million revenue vehicle-miles of service would be provided within the Region. In 1991, only 20.3 million revenue vehicle-miles were provided, less than a 1 percent increase over 1972.
- 3. The adopted plan aimed at reducing singleoccupant automobile use by providing incentives for ridesharing and public transit use, thus minimizing the need for arterial capacity expansion. By 1990, the Wisconsin Department of Transportation had initiated implementation of the proposed freeway traffic management system by undertaking preliminary engineering. Significant progress had been made in the provision of carpool parking facilities throughout the Region. Some 18 carpool facilities, five more than planned, were in use by 1990. The utilization of the parking spaces at these facilities, discussed in Chapter IV, stood at 35 percent in 1991. The recommended program to market carpooling aggressively to area commuters

and employers and to encourage the provision of preferential parking to carpool participants has not been operated within the Region since 1981. Action has been taken only recently by the City of Milwaukee to study parking related recommendations, including changing the parking rate structure to discourage long-term parking and adjusting the supply of parking in downtown Milwaukee.

Several factors have contributed to the lack of implementation of the regional transportation system plan. Inadequate funding, with respect to county and local arterial improvements, in particular, has been a substantial impediment to plan implementation. Federal funding for highway improvements has declined and no significant replacement State program funding has been forthcoming. The necessary reliance on the property tax to fund county and local arterial preservation and improvement projects has resulted in a lower than planned level of highway implementation.

Funding constraints have also been a significant factor in the poor implementation of the public transit component of the plan. Federal funding has declined and local funding is limited because of its reliance on the property tax base, already stressed by competing local programs. In consequence of declining Federal funding, State funding has been increased. It is apparent that without such State funding the level of public transit service in the Region would have been reduced even beyond the relatively low existing levels.

Other reasons for lack of public transit plan implementation include the escalation of transit unit costs; the declining costs of motor fuel and the attendant decline in public interest in transit, which tends to rise and fall with motor fuel prices; increasing transit fares; and the lack of land use plan implementation. With respect to land use plan implementation, particularly significant trends which have affected public transit plan implementation and which are at variance with the regional land use plan include: the decline in employment at central city commercial and industrial centers and the converse increase in employment at outlying centers, the decline of residential density in the central cities, and substantial new development at less than the recommended medium density.

The funding to improve the arterial street and highway system and to improve and expand the public transit system as planned has not been available. Moreover, the factors which would promote increased transit ridership and service, higher fuel prices, a more concentrated land use pattern, lower transit fares, have not unfolded as assumed.

The regional transportation system plan was designed to accommodate forecast increase in travel within the Region. Despite lower than forecast population growth, increases in vehicle trip making and vehicle miles of travel substantially exceeded levels forecast, while transit use stagnated. A summary of the forecasts of travel is set forth below.

- 1. From 1972 to 1991, the number of person trips within the Region was forecast to increase by about 20 percent, to about 5.3 million. By 1991, the number of internal person trips had actually increased by 23 percent, to about 5.5 million trips. Consequently, the actual level of internal person trips exceeded the forecast by about 155,000 trips, or 3 percent.
- 2. From 1972 to 1991, the number of vehicle trips within the Region was forecast to increase by about 21 percent, to about 3.5 million. By 1991, the number of internal vehicle trips had actually increased by 40 percent, to almost 4.1 million trips. Consequently, the 1991 level of internal vehicle trips exceeded the forecast by nearly 600,000 trips, or 17 percent.
- 3. From 1972 to 1991, vehicle miles of travel in the Region was forecast to increase by about 34 percent, to 27 million. By 1991, VMT had actually increased by about 64 percent, to a level of 33 million vehicle miles. Consequently, 1991 VMT in the Region exceeded the forecast by six million vehicle miles, or nearly 22 percent.

4. From 1972 to 1991, public transit ridership in the Region was forecast to increase by nearly 95 percent, to about 105 million revenue passengers. By 1991, transit ridership had fallen by 3.5 million, to a level of 50 million revenue passengers, about 7 percent below the 1972 level. Consequently, the 1991 level of public transit ridership was below the forecast level by 54.4 million, or 52 percent.

The dispersion of urban development contrary to the regional land use plan and poor implementation of the regional transportation system plan have contributed to cause higher than forecast vehicle trip production and vehicle miles of travel and consequently substantial traffic congestion and attendant problems on arterial streets and highways within the Region. Indeed, by 1988, 267 miles of the arterial street and highway system, 211 miles more than envisioned for the year 2000 in the adopted plan, experienced traffic congestion on an average weekday.

In the design of the year 2000 regional transportation system plan, a chosen set of transportation system development proposals was deemed sufficient to accommodate forecast trip making levels and to meet agreed-upon system development objectives. Because actual trip making levels have proven to be far in excess of those forecast, the regional transportation system plan, as it now stands, is no longer adequate to achieve the development objectives advanced under that plan.

The knowledge gained through the study of the adopted regional transportation system plan serves as an important point of departure for the preparation of the third-generation regional transportation system plan. The information contained in this chapter, combined with an understanding of the inventory data and the changes anticipated within the Region, will permit the reformulation of development objectives and standards to guide transportation system development to the year 2010. (This page intentionally left blank)

Chapter IV

INVENTORY OF TRANSPORTATION FACILITIES AND SERVICES

INTRODUCTION

The regional transportation system forms the basic framework for both urban and rural development. It provides the accessibility necessary for development to occur and to remain vital and provides the mobility necessary for personal attainment of benefit from such development. The achievement of the basic social and economic objectives set forth in Chapter VIII hinges on a balanced, accessible, and efficient regional transportation system.

The transportation system is a major influence on the distribution of population and employment, and, therefore, a major influence on regional land use plan implementation. The transportation system can foster the achievement of objectives intrinsically related, not only to transportation, but also to other aspects of regional development as well, including urban development, environmental protection, and agricultural preservation. The ability to meet such broader objectives is related to both the supply and use of the transportation system.

The availability, location, and operation of transportation system facilities, services, and management measures also influence the path, mode, and frequency of travel. Such travel characteristics, in turn, influence the operation of the transportation system. Relationships between supply and utilization must therefore be understood. Such understanding necessitates comprehensive efforts to inventory the existing supply and the use of transportation facilities and services.

This chapter presents the findings of a definitive inventory of the supply of transportation system facilities and services, using 1991 as the base year. It focusses on the location, capacity, provision, and utilization of arterial street and highway facilities, public transit facilities and services, intermodal parking facilities, and the emerging freeway traffic management system of the Region. Such inventory data are necessary to formulate, calibrate, and validate traffic simulation models used in the plan test and evaluation phase of the regional transportation system planning process, as well as the sound design of alternative transportation system plans.

This chapter also describes the significant changes which have occurred in the supply and use of transportation facilities within the Region since 1972, the base year for the secondgeneration regional transportation system plan, and in some cases since 1963, the base year for the initial plan. A thorough examination of travel habits and patterns based on findings of the third major inventory of travel within the Region is set forth in Chapter V. Those findings and interpretations serve as the demand side of the transportation inventory equation. The focus in this chapter is on the supply side.

CLASSIFICATION AND SUPPLY OF ARTERIAL STREETS AND HIGHWAYS

Functional Classification

Functional classification is an important principle underlying sound regional transportation system planning. It defines the type and level of service which any particular street or highway should render and provides a means for defining direct and time-saving routes through the total street and highway system. Functional classification, therefore, is a prerequisite to the necessary simulation of existing and future travel patterns and thus to the design and evaluation of alternative regional transportation systems.

Individual streets and highways do not serve travel independently in any major way. Because most travel involves movement through a network of streets and highways, it becomes necessary to determine how this travel can be efficiently channelized within the total network. Functional classification defines the nature of this channelization. At least three functional classifications of streets and highways are recognized: land-access, collector, and arterial.

Only those facilities functionally classified as arterials are explicitly considered in the regional transportation system planning process. Knowledge of the location and function of land-access and collector streets and highways, however, is also essential to the design of an effective longrange regional transportation system plan. This is especially true of collector facilities, which are intended primarily to collect traffic from, and distribute traffic to, land-access streets, conveying such traffic to the arterial system. Knowledge of the supply and location of land-access and collector streets and highways is also vital to any comprehensive planning effort, since both types of facilities serve as a means of access to abutting property, provide important links in the urban stormwater drainage system, and provide the location for utilities serving urban and rural development.

Arterial streets and highways are intended to serve the through movement of fast or heavy traffic and to provide transportation service between two or more major subareas of the Region and between such subareas and areas outside of the Region. Arterials should form an integrated, areawide system, located and designed to carry imposed traffic loadings properly. Freeways, expressways, and certain parkways, as well as standard arterial streets and highways, have special design characteristics which qualify them for inclusion in the arterial system.¹

The functional classification of the existing regional street and highway system was accomplished through a comparative evaluation of four major factors: 1) traffic: traffic volume and type, operating speeds, and average trip length, 2) physical characteristics: horizontal and vertical alignment, pavement width, and pavement type, 3) system integration: system continuity, facility spacing, and legal regulations, and 4) land use service: the areawide significance of the land use activities to be served.

Existing facilities proposed to be classified as arterials were mapped and copies of the preliminarily proposed arterial street and highway system were sent to concerned State, county, and local units of government for review and comment. After the Regional Planning Commission incorporated necessary changes and additions, the existing arterial street and highway system was reviewed and approved by the Technical Coordinating and Advisory Committee on Regional Transportation System Planning.

The amount of arterial mileage within the Region has generally increased over time, from 3,188 miles in 1963 to 3,274 miles in 1991 (see Table 32). It is interesting to note, however, that between 1963 and 1972 the amount of arterial mileage declined by 70 miles, or 2 percent, to 3,118 miles. This net decline may be attributed to the expansion of the freeway system in the mid- to-late 1960s, which replaced the arterial function of paralleling streets and highways. Despite the existence of the freeways and promoted by the substantial dispersion of urban development within the Region, the amount of arterial mileage increased by 156 miles, or about 5 percent, between 1972 and 1991 (see Table 33).

Of the 156-mile increase in arterial mileage between 1972 and 1991, only 70 miles represented new arterial construction; the remaining 86 miles were converted to arterial status to accommodate increased travel demand. About 64 percent of the mileage added to the Region's arterial system between 1972 and 1991, or 101 miles, was located within Waukesha and Washington Counties. As indicated in Chapter III, the most significant urban development at variance with the regional land use plan recommendations also occurred in these two counties.

The existing arterial system, shown on Map 17, represents about 29 percent of the Region's total street and highway system, which in 1991 consisted of about 11,200 miles (see Table 32). Over time arterial mileage has decreased as a percentage of total street and highway mileage. This decline is due in part to the construction of nonarterial streets and highways to accommo-

 $^{^{1}}A$ freeway is defined as a divided arterial highway with full control of access and with grade separations at all intersections. An expressway is defined as a divided arterial highway with full or partial control of access and with grade separations at some, but not necessarily all, intersections. A parkway is defined as an arterial highway provided for noncommercial traffic with full or partial control of access, usually located within a ribbon of park-like development. The term parkway as defined herein should not be confused with park roads or drives, which are not intended to serve as arterials. Standard arterial streets and highways may be defined as arterials with atgrade intersections. Standard arterials may or may not permit direct access to, and egress from, abutting property.

DISTRIBUTION OF TOTAL STREET AND HIGHWAY MILEAGE WITHIN THE REGION BY COUNTY: 1963, 1972, AND 1991

	1963							
County	Arterial	Collector and Local/Land-Access	Total ^a	Arterial Mileage as a Percent of Total Mileage				
Kenosha	281.5	547.1	828.6	34.0				
Milwaukee	791.5	1,642.6	2,434.1	32.5				
Ozaukee	264.9	366.9	631.8	41.9				
Racine	351.3	632.4	983.7	35.7				
Walworth	399.7	824.2	1,223,9	32.7				
Washington	402.3	688.0	1,090.3	36.9				
Waukesha	697.0	1,054.0	1,751.0	39.8				
Total	3,188.2	5,755.2	8,943.4	35.6				

	1972							
County	Arterial	Collector and Local/Land-Access	Total ^a	Arterial Mileage as a Percent of Total Mileage				
Kenosha	287.1	593.4	880.5	32.6				
Milwaukee	795.7	1,851.7	2,647.4	30.1				
Ozaukee	253.5	466.7	720.2	35.2				
Racine	355.4	728.0	1,083.4	32.8				
Walworth	412.0	846.9	1,308.9	31.5				
Washington	344.8	821.1	1,165.9	29.6				
Waukesha	670.2	1,342.5	2,012.7	33.3				
Total	3,118.7	6,700.3	9,819.0	31.8				

	1991							
County	Arterial	Collector and Local/Land-Access	Total ^a	Arterial Mileage as a Percent of Total Mileage				
Kenosha	317.7	660.6	978.3	32.4				
Milwaukee	775.4	2,131.6	2,907.0	26.7				
Ozaukee	288.5	572.5	861.0	33.5				
Racine	347.9	816.4	1,164.3	30.0				
Walworth	429.2	996.4	1,425.6	30.1				
Washington	399.2	924.4	1,323.6	30.2				
Waukesha	716.3	1,824.6	2,540.9	28.2				
Total	3,274.2	7,926.5	11,200.7	29.2				

^a Total street and highway mileage does not include private streets and roads or roadways in public parks and on institutional lands.

Source: SEWRPC.

date new urban development in previously undeveloped areas. In 1963 and 1972, arterial mileage accounted for about 36 percent and about 32 percent, respectively, of the total street and highway mileage within the Region.

The arterial street and highway system can be further functionally classified in terms of eligibility for Federal funding in support of highway improvements. To identify such facilities, the Wisconsin Department of Transportation (Wis-DOT) utilized a national functional highway classification scheme developed by the U.S. Department of Transportation (USDOT), Federal Highway Administration (FHWA). As required by the Federal Intermodal Surface Transportation Efficiency Act of 1991 (ISTEA), the following categories were used in urban areas: "principal arterial, minor arterial, urban collector, and land-access;" and in rural areas: "principal arterial, minor arterial, major collector, minor collector, and land-access."

Arterial facilities classified under the Federal classification scheme as urban and rural principal arterials, urban and rural minor arterials, and rural major collectors were identified by the Commission in cooperation with concerned county and local units of government for inclusion in the 1991 existing regional arterial system. Table 34 shows the existing distribution of highway mileage among these classes. Those facilities classified as rural major collectors are included in the regional arterial system because they are vital to arterial system continuity. Indeed, as shown on Map 18, when rural major collectors transverse urban areas, they are classified under the Federal scheme as minor arterials.

Jurisdictional Classification

Streets and highways may be classified according to jurisdiction as well as function. Jurisdictional classification is important to plan implementation. It establishes which level of government, be it State, county, or local, has, or should have, responsibility for the design, construction, maintenance, and operation of each segment of the total street and highway system. Jurisdictional classification is intended to group into subsystems, in a logical fashion, all streets and highways under the jurisdiction of a given level of government.

Criteria for jurisdictionally classifying the streets and highways in the Region were developed by the Commission in cooperation with the Federal, State, and local units and agencies of government concerned under the individual county jurisdictional highway planning programs. The criteria relate to: 1) trip service: the average trip length on each segment during an average weekday, 2) land use service: the areawide significance of land use activities to be connected and served, and 3) facility operational characteristics attendant to system continuity: facility spacing, traffic volume, traffic mobility, and land access.

In applying the criteria it was determined that: 1) those facilities intended to provide a high level of traffic mobility, to serve trips of the longest length, to provide minimal land access, and to have regional or interregional continuity should

Table 33

CHANGE IN ARTERIAL STREET AND HIGHWAY MILEAGE: 1972-1991 AND 1963-1991

	Cha 1972-	nge 1991	Change 1963-1991		
County	Number	Percent	Number	Percent	
Kenosha	30.6	10.7	36.2	12.9	
Milwaukee	-20.3	-2.6	-16.1	-2.0	
Ozaukee	35.0	13.8	23.6	8.9	
Racine	-7.5	-2.1	-3.4	-1.0	
Walworth	17.2	4.2	29.5	7.4	
Washington	54.4	15.8	-3.1	-0.8	
Waukesha	46.1	6.9	19.3	2.8	
Total	155.5	5.0	86.0	2.7	

Source: SEWRPC.

be classified as State trunk highways, 2) those facilities intended to provide an intermediate level of traffic mobility and land access and to have intercommunity continuity should be classified as county trunk arterials, and 3) those facilities intended to provide a low level of traffic mobility, a high degree of arterial land access, and to have intracommunity continuity should be classified as local arterial streets.

In practice, jurisdictional classification does not always relate fully to the functional criteria set forth above. The existing jurisdictional highway subsystems, as shown in Map 19, are the result of a long evolutionary process influenced by many complex political, administrative, financial, and engineering considerations and constraints.

Table 35 sets forth the existing distribution of arterial street and highway mileage among each jurisdictional subsystem within the Region and within each county of the Region. In 1991, about 1,261 miles, or nearly 39 percent of the total arterial system, were classified as State trunk highways, including State connecting streets; about 1,079 miles, or 33 percent, were classified as county trunk highways; the remaining 934 miles, or about 29 percent of the arterial system, were classified as local arterials.

Table 35 also compares the jurisdictionally classified mileage within each county in the Region. Milwaukee County had the most arterial State trunk highway and connecting street mileage. About 251 miles, or about 32 percent of arterial mileage within Milwaukee County, was classified as part of the State highway system.



Arterial streets and highways are intended to serve the through movement of traffic, providing transportation service between two or more subareas of a region, or through a region. The remainder of the street system consists of land access and collector streets whose principal function is to provide access to abutting property and to provide connections between land access and arterial streets, respectively. In 1991, the arterial street and highway system in the Southeastern Wisconsin Region consisted of 3,274 miles of facilities, including freeways, expressways, certain parkways, and standard surface streets and highways. This arterial system comprised 29 percent of the 11,200-mile total street and highway system in the Region.

	Urt	ban				
County	Principal Arterials	Minor Arterials	Principal Arterials	Minor Arterials	Major Collector Streets and Highways	Total
Kenosha	45.5	71.6	35.9	35.5	129.2	317.7
Milwaukee	257.0	518.4				775.4
Ozaukee	38.1	111.4	39.3	9.7	90.0	288.5
Racine	63.4	71.0	61.9	47.8	103.8	347.9
Walworth	25.7	19.7	90.9	85.4	207.5	429.2
Washington	35.8	64.0	53.2	60.3	185.9	399.2
Waukesha	176.4	242.2	35.0	89.6	173.1	716.3
Region	641.9	1,098.3	316.2	328.3	889.5	3,274.2

DISTRIBUTION OF HIGHWAY MILEAGE BY FUNCTIONAL CLASSIFICATION AND BY COUNTY

Source: SEWRPC.

Connecting streets are marked and signed routes of State trunk highways routed over streets and highways for which local municipalities have the responsibility for maintenance and operation. The City of Milwaukee, the convergence point of many arterial facilities, has significant connecting street maintenance and operational responsibilities. Indeed, about 53 percent of the total connecting street mileage within the Region is located within Milwaukee County.

In 1991, Waukesha County had the highest mileage of arterial county trunk highway within the Region. About 320 miles, or nearly 45 percent of the total arterial mileage within Waukesha County, was part of the county highway system. Milwaukee County, the most urbanized county in the Region, had the most mileage of local arterial trunk highway. About 442 miles, or 57 percent of the total arterial mileage within Milwaukee County, was part of the local highway system.

UTILIZATION OF ARTERIAL STREETS AND HIGHWAYS

The regional arterial street and highway system was mapped and converted to network form for travel forecasting and system analyses purposes. This conversion was accomplished by assigning node numbers to all intersections and defining each segment between two nodes as a link in the system, as illustrated in Figure 17.

These forecasting and analyses necessitated significant inventory efforts. Data setting forth travel times on, and the length and capacity of, each link in the network have been maintained by the Commission since 1963 through extensive monitoring procedures. Data necessary to determine the traffic capacity of a link include, in addition to the length, average operating speed, pavement width, intersection approach gradient, type of facility, type of area through which the link passes, and the existence or nonexistence of intersection turning lanes. These data permitted the calculation of minimum travel-time paths through the network and the assignment of traffic to these minimum travel-time paths, duly considering capacity constraints. Such data and procedures enabled the accurate simulation of travel on the arterial street and highway system.

In order to provide a measure of the present utilization of the existing arterial street and highway system, the average weekday traffic volume was obtained for each link in the system. Programs to count traffic volume conducted by the Wisconsin Department of Transportation and by the County and City of Milwaukee on a regular basis provided much of the necessary current traffic volume data. In order to obtain complete data for the entire arterial network, however, these counts had to be supplemented by counts taken by other local municipalities and by the Commission itself. These programs, which included control and sampling counts for the arterial system as a whole, also included the



Under the Intermodal Surface Transportation Efficiency Act of 1991, existing streets and highways within the Region required functional classification according to a Federal classification scheme. Those streets and highways identified as existing arterials by the Commission in cooperation with concerned local units of government were further classified as follows under the Federal functional classification scheme: urban principal arterial, urban minor arterial, rural principal arterial, rural minor arterial, and rural major collector.

Map 19



Jurisdictional classification determines which level of government, State, county, or local, has, or should have, the responsibility for the design, construction, operation, and maintenance of streets and highways. In 1991, about 1,261 arterial miles, or nearly 39 percent of the arterial street and highway system, were classified as State trunk highways; about 1,079 arterial miles, or 33 percent of the arterial system, were classified as county trunk highways; and about 934 arterial miles, or 29 percent of the arterial system, were classified as local streets or highways.

Source: SEWRPC.

92

EXISTING DISTRIBUTION OF ARTERIAL STREET AND HIGHWAY MILEAGE WITHIN THE REGION BY COUNTY AND JURISDICTIONAL CLASSIFICATION: 1991

	State			County		Local		Total	
County	Trunk Highways	Connecting Streets	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent
Kenosha	106.1	12.7	37.4	139.9	44.0	59.0	18.6	317.7	100.0
Milwaukee	167.9	83.5	32.4	81.6	10.5	442.4	57.1	775.4	100.0
Ozaukee	90.4	10.3	34.9	96.9	33.6	90.9	31.5	288.5	100.0
Racine	139.7	19.2	45.7	124.5	35.8	64.5	18.5	347.9	100.0
Walworth	200.4	13.3	49.8	168.2	39.2	47.3	11.0	429.2	100.0
Washington	179.0	7.1	46.6	148.0	37.1	65.1	16.3	399.2	100.0
Waukesha	218.4	12.9	32.3	320.0	44.7	165.0	23.0	716.3	100.0
Region	1,101.9	159.0	38.5	1,079.1	33.0	934.2	28.5	3,274.2	100.0

Source: Wisconsin Department of Transportation and SEWRPC.

cordon and screen-line counts necessary to evaluate the accuracy of the origin and destination surveys. The comprehensive nature of the traffic counting programs made it possible to estimate with a high degree of accuracy the average weekday and peak-hour traffic flows on all segments of the arterial network and to calculate total vehicle-miles of travel (VMT) on the arterial street and highway system.

System Utilization

The average weekday arterial traffic flows within the Region in 1991 are graphically displayed on Map 22. The highest traffic volumes were carried on the freeways and certain standard arterials within the Milwaukee urbanized area. East-west arterials that carried significant traffic volumes included, among others, W. Good Hope Road (STH 74 and STH 100) between IH 45 and IH 43, W. Silver Spring Drive between USH 41 and IH 43, W. Capitol Drive (USH 190) between the Milwaukee-Waukesha County line and IH 43, and W. Blue Mound Road (USH 18) from N. Mayfair Road to Barker Road. North-south arterials that carried significant volumes in 1991 included, among others, N. 76th Street from W. Brown Deer Road to W. Silver Spring Road, and Mayfair Road (USH 100) from W. Capitol Drive to USH 45. Two of these heavily used facilities, N. Mayfair Road and W. Blue Mound Road, parallel freeways: USH 45 and IH 94, respectively.

Major changes occurred in the utilization of the arterial street and highway system between 1963,

Figure 17

EXAMPLE OF HIGHWAY NETWORK MAP



1972, and 1991, as a comparison of Maps 20, 21, and 22 indicates. These changes may be attributed in part to new urban development and in part to the redistribution of traffic as improvements were made in the arterial system itself. Between 1963 and 1972, for instance, the completion of



Average weekday traffic flows on the arterial street and highway system in the Region in 1963 are shown on the above map. This pattern of traffic flow is largely representative of pre-freeway conditions in the Region, since very few miles of urban freeway were open to traffic in 1963. In the Milwaukee metropolitan area, very heavy traffic volumes occurred on such standard surface arterials as W. Blue Mound Road, S. 27th Street, W. Capitol Drive, W. Fond du Lac Avenue, W. Appleton Avenue, W. Oklahoma Avenue, and STH 100, the former metropolitan bypass route.







The average weekday traffic flows on the regional arterial system in 1991 are shown on the above map. While little change has occurred in the geographic pattern of arterial use, traffic flows have increased significantly over time. Indeed, about 33 million vehiclemiles of travel were recorded on the arterial system in 1991, a 64 percent increase over the level recorded in 1972 and a 153 percent increase over the level recorded in 1963. In 1991, freeways, which comprised only 9 percent of total arterial mileage in the Region, carried about 39 percent of the total vehicle miles of travel. *Source: SEWRPC*.

ARTERIAL VEHICLE-MILES OF TRAVEL WITHIN THE REGION BY COUNTY: 1963, 1972, AND 1991

	1963		1972		1991	
County	Vehicle-Miles of Travel (thousands)	Percent of Total	Vehicle-Miles of Travel (thousands)	Percent of Total	Vehicle-Miles of Travel (thousands)	Percent of Total
Kenosha	938 7,348 484 1,125 685 696 1,796	7.2 56.2 3.7 8.6 5.2 5.3 13.8	1,428 10,695 850 1,813 873 1,151 3,314	7.1 53.2 4.2 9.0 4.3 5.7 16.5	2,500 14,391 1,942 2,966 1,913 2,379 6,981	7.6 43.4 5.9 9.0 5.8 7.2 21.1
Region	13,072	100.0	20,124	100.0	33,072	100.0

	Change: 1	972-1991	Change: 1963-1991		
County	Number	Percent	Number	Percent	
Kenosha	1,072	75.1	1,562	166.5	
Milwaukee	3,696	34.6	7,043	95.8	
Ozaukee	1,092	128.5	1,458	301.2	
Racine	1,153	63.6	1,841	163.6	
Walworth	1,040	119.1	1,228	179.3	
Washington	1,228	106.7	1,683	241.8	
Waukesha	3,667	110.7	5,185	288.7	
Region	12,948	64.3	20,000	153.0	

Source: SEWRPC.

important freeway segments resulted in significant reductions in traffic volumes on arterials paralleling the freeways and significant increases in traffic volumes on arterials connecting to freeway interchanges. Between 1972 and 1991, however, little change in the pattern of arterial utilization was evident. The magnitude of arterial street and highway use, however, increased significantly.

The magnitude of arterial use can be expressed in terms of total vehicle-miles of travel. On an average weekday in 1991, over 33 million VMT were recorded on the arterial street and highway system within the Region. As shown in Table 36, more vehicle-miles were traveled in Milwaukee County than in any other county of the Region. Milwaukee County alone accounted for 43 percent of the total VMT within the Region and exhibited the most intensive utilization of the existing arterial system, recording about 18,750 vehicle-miles of travel per mile of arterial street and highway (see Figure 18). Table 36 also shows the change in arterial vehicle-miles of travel within each county and for the Region. Between 1972 and 1991, VMT within the Region increased by nearly 13 million vehicle-miles, or by about 64 percent. Between 1963 and 1991, VMT within the Region increased by about 20 million vehicle-miles, or by about 153 percent (see Figure 19).

It is important to note that, while the absolute increase in VMT between 1972 and 1991 within Milwaukee County is significant, nearly 3.7 million, the County's percentage increase in VMT between 1972 and 1991, about 35 percent, is significantly smaller than in the other counties. Further, the percentage increase in VMT in the urbanized counties of Kenosha, Milwaukee, and Racine was smaller than that in the four other counties of the Region. Indeed, vehicle miles traveled in Ozaukee, Walworth, Washington, and Waukesha Counties more than doubled between 1972 and 1991. This finding is indicative of the dispersion of urban development

Figure 18



VEHICLE-MILES OF TRAVEL PER ARTERIAL MILE WITHIN THE REGION BY COUNTY: 1991

Source: SEWRPC.

throughout the Region and the consequent changes in vehicular travel patterns.

Table 37 shows vehicle-miles of travel on freeways and standard arterials, indicating an increasing utilization of freeways within the Region. In 1991, freeways, which constituted 295 miles, or about 9 percent of the total arterial system, carried about 35 percent of total arterial vehicle-miles of travel (see Figure 20), compared to 31 percent in 1972 and 11 percent in 1963. In 1991, freeways within the Region carried nearly 11.6 million vehicle-miles, 87 percent more than the approximately 6.2 million carried in 1972.

Relationship of System Use to Design Capacity

The volume-to-capacity (V/C) ratio exhibited by an arterial facility is a useful means of identifying and quantifying existing and probable

Figure 19

ARTERIAL VEHICLE-MILES OF TRAVEL WITHIN THE REGION ON AN AVERAGE WEEKDAY: 1963, 1972, AND 1991



Source: SEWRPC.

future imbalances between street use and supply. The V/C ratio is defined as the relationship between the average daily weekday traffic utilizing a particular section of the arterial system and the design capacity of that section. This relationship, when determined and evaluated for the entire arterial system, is useful in identifying major travel corridors where additional capacity, travel demand management, and/or public transit investment may be needed.

Volume-to-capacity ratios were computed for each link of the arterial network for the base years of the first- and second-generation plans and again for 1991. The results are indicated on Maps 23 through 25. In order to facilitate their presentation and subsequent analysis, the V/C ratios were grouped into three categories: under design capacity, a V/C ratio of 0.90 or less; at design capacity, a V/C ratio of 0.91 to 1.10; and over design capacity, a V/C ratio of 1.11 or more. A description of operational conditions on urban standard arterials, rural arterials, and freeways is provided below.

• In general, urban standard arterials operating under design capacity will have average

ARTERIAL VEHICLE-MILES OF TRAVEL WITHIN THE REGION ON AN AVERAGE WEEKDAY BY COUNTY: 1963, 1972, AND 1991

			1963	3		
	Freeway		Standard A	rterial	Total	
County	Vehicle-Miles of Travel Percent (thousands) of Total		Vehicle-Miles of Travel (thousands)	Percent of Total	Vehicle-Miles of Travel (thousands)	Percent of Total
KenoshaKenoshaMilwaukeeOzaukeeOzaukeeRacineRacineNalworthWalworthNashingtonWaukeshaNaukesha	204 531 20 203 345 159	21.7 7.2 9.1 18.0 0.0 49.6 8.7	734 6,817 464 922 685 351 1,637	78.3 92.8 95.9 82.0 100.0 50.4 91.1	938 7,348 484 1,125 685 696 1,796	100.0 100.0 100.0 100.0 100.0 100.0 100.0
Total	1,462	11.2	11,610	88.8	13,072	100.0

_			1972	2		
	FreewayVehicle-Milesof TravelPercent(thousands)of Total		Standard A	Arterial	Total	
County			Vehicle-Miles of Travel (thousands)	Percent of Total	Vehicle-Miles of Travel (thousands)	Percent of Total
Kenosha	382	26.8	1,046	73.2	1,428	100.0
Milwaukee	3,977	57.2	6,718	62.8	10,695	100.0
Ozaukee	223	26.2	627	73.8	850	100.0
Racine	415	12.9	1,398	77.1	1,813	100.0
Walworth	56	6.4	817	93.6	873	100.0
Washington	190	16.5	961	33.5	1,151	100.0
Waukesha	970	29.3	2,344	70.7	3,314	100.0
Total	6,213	30.9	13,911	69.1	20,124	100.0

			1991			
	FreewayVehicle-Milesof TravelPercent(thousands)of Total		Standard A	rterial	Total	
County			Vehicle-Miles of Travel (thousands)	Percent of Total	Vehicle-Miles of Travel (thousands)	Percent of Total
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	675 5,945 762 708 540 546 2,421	27.8 41.3 39.2 33.9 28.2 23.0 34.7	1,825 8,446 1,180 2,258 1,373 1,833 4,560	53.0 58.7 60.8 96.1 71.8 77.0 65.3	2,500 14,391 1,942 2,966 1,913 2,379 6,981	100.0 100.0 100.0 100.0 100.0 100.0 100.0
Total	11,597	35.1	21,475	64.9	33,072	100.0

RELATION OF FREEWAY MILEAGE AND VEHICLE-MILES OF TRAVEL TO TOTAL ARTERIAL MILEAGE AND VEHICLE-MILES OF TRAVEL: 1963, 1972, AND 1991







100

been absorbed without increasing traffic congestion to intolerable levels.



By 1972, the development of the regional freeway system had significantly altered traffic patterns and conditions within the Region. In the Milwaukee area, congestion on many important surface arterials, such as W. National Avenue, S. 27th Street, W. Blue Mound Road, STH 100, W. Fond du Lac Avenue, W. Appleton Avenue, and N. Port Washington Road, had been relieved by the construction of freeway facilities. The overwhelming majority of congested arterial mileage, 87 percent, was located in the Milwaukee, Racine, and Kenosha urbanized areas. In Milwaukee County about 17 percent of the arterial system was operating at or over design capacity, a significant reduction in traffic congestion due largely to the opening of freeway facilities. For the Region as a whole, arterial traffic congestion was held at about 1963 levels despite an 8 percent increase in population, a 40 percent increase in motor vehicle registrations, a 25 percent increase in trip generation, and a 54 percent increase in arterial vehicle-miles of travel. Without the regional freeway system, such increases in travel demand could not have



By 1991, the extra highway capacity provided by the expansion of the freeway system within Milwaukee County had largely been exceeded by significant peak-period traffic volumes and traffic congestion on paralleling surface arterial streets and highways, which had diminished upon freeway construction, returned. In 1991, about 279 arterial miles, or nearly 9 percent of the arterial street and highway system, were operating over design capacity; nearly 264 arterial miles, or 8 percent of the arterial system, were operating at design capacity. Within Milwaukee County, in the same year, nearly 30 percent of arterial mileage was operating at, or over, design capacity.

Figure 21

TYPICAL URBAN ARTERIAL OPERATING AT, AND OVER, DESIGN CAPACITY DURING MORNING PEAK-TRAFFIC PERIOD



The above photographs of the eastbound approach to the intersection of W. Capitol Drive with N. 27th Street depict a typical arterial street "at" design capacity and "over" design capacity operating conditions during the morning peak travel period. The upper photograph, depicting "at" design capacity operation, indicates a steady flow of traffic with some attendant restrictions on a driver's opportunity to change from one traffic lane to another, slightly reduced vehicle operating speed, and short delays at some intersections. The lower photograph, depicting "over" design capacity operation, indicates a heavier flow of traffic, precluding a driver's opportunity to change from one traffic lane to another, substantially reduced vehicle operating speeds, and extended delays at intersections caused by backups of vehicles at the intersection.

Source: SEWRPC.

speeds of 25 to 40 miles per hour (mph) and average delays at signalized intersections of five to 15 seconds. Standard arterials operating at design capacity will have average speeds of about 20 to 30 mph and average delays at signalized intersections of about 25 seconds. Urban standard arterial streets operating over design capacity will have substantial delays at signalized intersections. During peak traffic periods, vehicles may have to wait through more than one red traffic signal. The average delay to each vehicle at signalized intersections will be at least 35 seconds and may approach 120 seconds. The average travel speed along such urban arterials will generally be less than 15 to 20 mph. In addition, the potential for accidents is increased on arterials carrying traffic volumes over design capacity. Figure 21 shows a typical urban arterial operating at design capacity and over design capacity during the morning peak traffic period.

- In general, rural arterials (with a 55-mph speed limit) operating under design capacity will have average speeds of 50 to 55 mph, with minimal restrictions on lane changing or passing. Rural arterials operating at design capacity will have average speeds of 45 to 50 mph, with some restrictions on lane changing or passing. Rural arterials operating over design capacity will have average speeds of 30 to 45 mph, with significant restrictions on lane changing on multi-lane facilities and on passing on two-lane facilities.
- In general, freeways (with 55-mph speed limit) operating under design capacity will have average speeds of 55 mph and no restrictions on lane changing. Freeways operating at design capacity will have speeds of 45 to 55 mph and some restrictions on lane changing. Freeways operating over design capacity will have average speeds of 30 to 45 mph and significant restrictions on lane changing. Stop-and-go traffic at speeds below 30 mph may occur behind freeway sections operating over design capacity.

Designations of levels of service are used to measure qualitatively the operational characteristics of a given link in the arterial network. Under this approach, there are six levels of service corresponding to letters "A" through "F," "A" describing free-flow, unrestricted traffic conditions and "F" describing a breakdown in traffic flow, the point past which conceptually it becomes impossible to operate an additional vehicle on the facility. The three ranges of V/Cratios, previously indicated, generally correspond to levels of service designations in the following manner: facilities operating under design capacity represent levels of service "A" or "B," facilities operating at design capacity represent level of service "C," facilities operating over design capacity represent levels of service "D," "E," or "F." Figure 22 shows freeway traffic conditions under each of the six levels of service.

As shown on Map 25, most of the arterial mileage operating at or over design capacity within the Region is located in, and adjacent to, intensely developed urban areas. Indeed, about 44 percent of the arterial mileage within the Region operating over design capacity in 1991 was located within Milwaukee County and nearly 16 percent of the total arterial mileage within Milwaukee County was operating over design capacity during an average weekday in 1991 (see Table 38). A comparison of Maps 23 through 25 also indicates that traffic congestion has increasingly occurred on arterials located within small urban areas. As shown on Map 25, arterial facilities operating at or over design capacity in 1991 were located within the Cities of Cedarburg, Lake Geneva, Oconomowoc, and West Bend and within the Villages of Grafton and Hartland.

Traffic congestion is closely tied to the supply and use of arterial highway capacity. Between 1963 and 1972, arterial mileage operating at or over design capacity decreased by about 14 miles, or 4 percent, to 166 miles. This decrease in traffic congestion was primarily due to the construction of the freeway system and the additional arterial capacity freeways provided. Between 1972 and 1991, however, arterial mileage operating at and over design capacity increased significantly, as shown in Figure 23. Indeed, the number of miles operating at or over design capacity increased by 71 percent, from about 318 in 1972 to 543 in 1991. Between 1963 and 1991, the number of arterial miles operating at or over design capacity increased by 212 miles, or about 64 percent.

CLASSIFICATION, SUPPLY, AND UTILIZATION OF PUBLIC TRANSIT

If a balanced regional transportation system is to be designed, careful attention must be given to the existing public fixed-route transit facilities and services of areawide significance and to their potential use. Such facilities and services must be designed as integral parts of the regional transportation system. This section defines and classifies the various types of public transportation, identifies the urban fixed-route public transit system considered in the regional transportation system planning process and in the traffic simulation procedures, and describes the existing provision and utilization of public transit services within the Region.

System Classification

Public transportation may be defined as the transportation of relatively large groups of people by publicly, quasi-publicly, or privately owned

Figure 22

CONCEPTUAL LEVELS OF SERVICE ON THE FREEWAY SYSTEM

BASIC FREEWAY SEGMENTS



LEVEL-OF-SERVICE A



LEVEL-OF-SERVICE D



LEVEL-OF-SERVICE B



LEVEL-OF-SERVICE E



LEVEL-OF-SERVICE C

LEVEL-OF-SERVICE F

Source: Transportation Research Board, National Research Council, <u>Highway Capacity Manual</u>, Washington D. C., 1985.

VOLUME-TO-CAPACITY RATIOS^a FOR THE ARTERIAL STREET AND HIGHWAY SYSTEM WITHIN THE REGION BY COUNTY: 1963, 1972, AND 1991

	1963						
	Volume-to-Ca 0.00-	pacity Range 0.90	Volume-to-Capacity Range 0.91-1.10		Volume-to-Capacity Range Over 1.10		
County	Mileage	Percent of Total	Mileage	Percent of Total	Mileage	Percent of Total	Total Mileage
Kenosha	260.8	92.6	7.2	2.6	13.5	4.8	281.5
Milwaukee	589.8	74.5	85.4	10.8	116.3	14.7	791.5
Ozaukee	250.3	94.5	6.3	2.4	8.3	3.1	264.9
Racine	327.7	93.3	10.0	2.8	13.6	3.9	351.3
Walworth	390.5	97.7	3.9	1.0	5.3	1.3	399.7
Washington	401.8	99.9	0.5	0.1	0.0	0.0	402.3
Waukesha	635.6	91.2	26.6	3.8	34.8	5.0	697.0
Region	2,856.5	89.6	139.9	4.4	191.8	6.0	3,188.2

	1972						
	Volume-to-Capacity Range 0.00-0.90		Volume-to-Capacity Range 0.91-1.10		Volume-to-Capacity Range Over 1.10		
County	Mileage	Percent of Total	Mileage	Percent of Total	Mileage	Percent of Total	Total Mileage
Kenosha	250.4	87.2	14.7	5.1	22.0	7.7	287.1
Milwaukee	662.9	83.3	71.8	9.0	61.0	7.7	795.7
Ozaukee	237.9	93.8	10.1	4.0	5.5	2.2	253.5
Racine	316.0	88.9	19.1	5.4	20.3	5.7	355.4
Walworth	404.5	98.2	2.7	0.7	4.8	1.1	412.0
Washington	326.0	94.6	9.7	2.8	9.1	2.6	344.8
Waukesha	603.5	90.0	23.8	3.6	42.9	6.4	670.2
Region	2,801.2	89.8	151.9	4.9	165.6	5.3	3,118.7

	1991						
	Volume-to-Capacity Range 0.00-0.90		Volume-to-Capacity Range 0.91-1.10		Volume-to-Capacity Range Over 1.10		
County	Mileage	Percent of Total	Mileage	Percent of Total	Mileage	Percent of Total	Total Mileage
Kenosha	286.0	90.9	16.6	5.2	15.1	4.8	317.7
Milwaukee	543.2	70.1	110.3	14.2	121.9	15.7	775.4
Ozaukee	254.2	88.1	21.0	7.3	13.3	4.6	288.5
Racine	288.7	83.0	24.9	7.2	34.3	9.8	347.9
Walworth	411.1	95.8	15.0	3.5	3.1	0.7	429.2
Washington	356.5	89.5	23.2	5.8	19.5	4 .9	399.2
Waukesha	591.3	82.5	53.0	7.4	72.0	10.1	716.3
Region	2,731.0	83.4	264.0	8.1	279.2	8.5	3,274.2

^aThe significance of the volume-to-capacity ratio of the ranges used is:

0.00-0.90 - Under design capacity, fully adequate and safest operational level. 0.91-1.10 - At design capacity but still adequate. Over 1.10 - Over design capacity, congested at times.

PERCENTAGE OF TOTAL ARTERIAL MILEAGE OPERATING AT, AND OVER, DESIGN CAPACITY: 1963, 1972, 1991



Source: SEWRPC.

vehicles routed between or along significant concentrations of related trip origins and destinations. Public transportation is most commonly provided in the Region by motor bus, commercial airplane, and railway passenger train.

As shown in Figure 24, public transportation may be divided into two subcategories: fixedroute and nonfixed-route. Fixed-route public transportation operates relatively large vehicles over predetermined routes on regular schedules and provides service to the general public or special subgroups. Nonfixed-route public transportation provides service to the general public or to special subgroups on a demand-responsive basis and includes paratransit service provided to disabled persons. Demand-responsive service is characterized by the flexible routing and scheduling of relatively small vehicles to provide shared-occupancy door-to-door transportation.

Operation of fixed-route public transportation can be simulated by mathematical models developed and maintained by the Commission. Fixed-route passenger transportation modes provided within the urban areas of the Region are therefore explicitly considered in the regional transportation system planning process. Operation of nonfixed-route public transportation, however, cannot be readily mathematically simulated and is only implicitly considered in the planning process. Nonfixed-route public transportation is given more definitive consideration by the Commission in subregional planning efforts, particularly in the preparation of transit system development programs for the urban areas of the Region.

Fixed-route public transportation may be divided into common-carrier service and special-carrier service. Fixed-route common-carrier service is provided to the general public, while special-carrier service is provided to subgroups. Examples of fixed-route special-carrier public transportation include yellow school bus service operated by area school districts and the special U-Bus service operated by the University of Wisconsin-Milwaukee for its students and staff. Special-carrier service is considered only implicitly in the planning process.

As shown in Figure 24, fixed-route commoncarrier service may be divided into three categories: interregional, urban, and rural. Interregional public transportation provides service across regional boundaries to meet external travel demand and includes commercial air travel, railway passenger service, and interregional bus service. Rural public transportation provides service in rural communities, making accessible economic and social opportunities otherwise not readily available in such areas to persons without access to a personal vehicle. Urban public transportation. commonly referred to as public transit, provides service within and between urban areas of the Region to meet internal travel demand. Public transit is essential in any metropolitan area to meet the travel needs of persons unable to use personalized transportation and to provide an alternative mode of travel for certain trips within and between urban areas.

Urban public transit is explicitly considered in the regional transportation planning process, while interregional and rural services are not. Interregional service is considered only to the extent that terminal and intermodal facilities, such as airports and bus and railway stations, comprise major trip generators affecting internal travel demand and patterns. Interregional commercial air travel is explicitly considered by the Commission under a separate comprehensive regional airport system planning program, while

Figure 24





Source: SEWRPC.

interregional passenger railway and motor bus service is considered by the Wisconsin Department of Transportation under a separate Statewide planning program.

Urban public transit may be divided into rapid, express, local, and paratransit levels of service. Rapid transit is intended to facilitate relatively fast and convenient transportation along heavily traveled corridors and between major activity centers and high- and medium-density residential communities within the Region. Rapid transit has relatively high average operating speeds and relatively low accessibility, with station spacings, if any, one-half mile or more apart. Rapid transit service can be provided by commuter rail and heavy rail operating over exclusive, grade-separated rights-of-way or by motor buses operating over exclusive, gradeseparated busways. Rapid transit can also be provided by motor buses operating in mixed traffic on freeways and by light rail operating over exclusive, though not grade-separated, rights-of-way. All forms of rapid transit service are explicitly considered in the planning process. Table 39 presents the distinguishing characteristics of commuter, heavy, and light rail.

Express transit service is provided over arterial streets and highways with stops generally no less than 1,200 feet apart at intersecting transit routes, intersecting arterial streets, and major traffic generators. Express transit serves trips of moderate length and can be provided by motor bus or by light rail operating in mixed traffic on shared rights-of-way. Express transit service provides a greater degree of accessibility at somewhat slower operating speeds than rapid transit and may provide "feeder" service to the rapid transit system. Express transit service is also explicitly considered in the regional transportation system planning process.

Characteristic	Commuter Rail	Heavy Rail	Light Rail
Passenger Capacity	Seating capacity ranging from 88 to 157 passengers per car	From 180 to 250 passengers per car; seating capacity ranging from 44 to 62 passengers per car	From 75 to 80 passengers per car, or 155 to 180 passengers per two-car articulated unit; seating capacity ranging from 50 to 62 passengers per car, or 68 to 80 passengers per two-car articulated unit
Length of Trains	Typically three cars	Up to 10 cars	One to three cars
Passenger Loading	At grade-level or high-level platform	High-level platform	At grade-level or high-level platform
Power Supply	Electric-overhead wire or third rail; diesel-electric locomotives on self-propelled coaches	Electric-third rail	Electric-overhead wire
Right-of-Way	Usually fully grade-separated	Usually fully grade-separated	Usually not fully grade separated

DISTINGUISHING CHARACTERISTICS OF COMMUTER, HEAVY, AND LIGHT RAIL

Source: SEWRPC.

Local transit service is characterized by a high degree of accessibility and low operating speeds. Local service is provided over arterial and collector streets with stops no more than 1.200 feet apart. Such service can be provided by motor bus, electric trolleybus, or light rail vehicles and is explicitly considered in the regional transportation system planning process. Local transit also provides a passenger collection-circulation-distribution function within major activity centers. The collectioncirculation-distribution function of local transit service may include the use of motor buses, electric trolleybuses, vans, light rail vehicles, automated guideway vehicles, and other methods of moving people. This function of local transit service is explicitly considered by the Commission in the subregional transportation planning process, particularly in the preparation of transit system development programs for the Milwaukee, Kenosha, Racine, West Bend, and Waukesha urban areas.

Paratransit is operated within local transit service areas to meet the transportation needs of the elderly and of those persons who because of mental or physical disability are unable to avail themselves of conventional transit service. Such service is explicitly considered by the Commission in the preparation and annual update of paratransit service plans for each of the six urban public transit systems in the Region.

System Identification

The urban fixed-route, common-carrier, rapid, express, and local transit service provided in the Region, except the collection and distribution function of local transit, was the object of the inventory necessary to build the transit network used in travel forecasting and system analysis. In building the transit network, node numbers were assigned to all transfer and terminal points and each segment between two nodes was defined as a link.

The transit network, graphically represented in Figure 25, is more complex than the arterial street and highway network since links must be encoded to allow the special travel times associated with transit use to be simulated. Travel times involved in walking, waiting, boarding and exiting transit vehicles, and shuttling between auto and transit modes are accounted for by these special links. These data permit the identification and calculation of minimum travel time paths through the entire regional transit network and the assignment of traffic to these minimum time paths, thereby accurately simulating public transit travel within the Region. The public transit system considered in the construction of the transit network consists of six separate subsystems operated by two counties and three municipalities. The rapid, express, and local services provided by these six subsystems within the Region are described below.

Figure 25





<u>Rapid Transit</u>: Rapid transit service within the Region in 1991 consisted of 13 freeway flyer motor bus routes. Eleven routes were provided by Milwaukee County and operated by the Milwaukee County Transit System. Operation of the remaining two was initiated in 1981 by Waukesha County. The first route, between the Village of Menomonee Falls and the central business district (CBD) of Milwaukee, was operated for the County by the Milwaukee County Transit System. The other, between the City of Oconomowoc and the Milwaukee CBD was operated for the County by Wisconsin Coach Lines, Inc., a private transit operator (see Map 28). In 1963, the base year of the initial regional transportation system plan, rapid transit within the Region consisted of commuter rail service operated by the Milwaukee Road between the City of Watertown, located approximately 10 miles west of the City of Oconomowoc but outside the Region, and the Milwaukee CBD (see Map 26). Freeway flyer service was initiated in 1964 by the Milwaukee and Suburban Transport Corporation between the City of Wauwatosa and the Milwaukee central business district. By 1972, seven routes were operated by the Milwaukee and Suburban Transport Corporation over freeways in the Milwaukee urban area, as shown on Map 27.

Express Transit: Express transit service provided within the Region in 1991 is also shown on Map 28. In 1991, express transit service consisted of a total of seven motor bus routes, including three routes in the Oconomowoc-Waukesha-Milwaukee travel corridor operated for Waukesha County by Wisconsin Coach Lines, Inc., three routes in Milwaukee County operated by the Milwaukee County Transit System, and one route between the Milwaukee CBD and the Cities of Racine and Kenosha sponsored since 1984 by the City of Racine and operated by Wisconsin Coach Lines, Inc. The most frequent commuter service in the Region in 1991, as in 1972, was provided by Wisconsin Coach Lines, Inc., in the Waukesha-Milwaukee travel corridor, with 18 eastbound trips and 17 westbound trips per weekday.

As shown on Map 26, express transit service in 1963 consisted of eight bus routes operated in several travel corridors by Greyhound Lines and Wisconsin Coach Lines, Inc., and two routes operated in the Milwaukee urban area by the Milwaukee and Suburban Transport Corporation. Express transit service remained largely unchanged during the 1963 through 1972 period; however, service between Waukesha and Oconomowoc and between Milwaukee and East Troy was abandoned. As shown on Map 27, in 1972 express service was operated over five routes by Wisconsin Coach Lines, Inc., and over two routes operated by the Milwaukee and Suburban Transport Corporation.

Local Transit: Fixed-route local transit service was provided in 1991 within the Kenosha, Milwaukee, and Racine urbanized areas. Local transit in the Kenosha urbanized area was provided by the City of Kenosha Transit Com-

Map 26

RAPID AND EXPRESS FIXED-ROUTE PUBLIC TRANSIT IN THE REGION: 1963





RAPID AND EXPRESS FIXED-ROUTE PUBLIC TRANSIT IN THE REGION: 1972



Map 28

RAPID AND EXPRESS FIXED-ROUTE PUBLIC TRANSIT IN THE REGION: 1991



Source: SEWRPC.

Maps 26, 27, and 28 show the rapid and express transit systems in the Region as these systems existed in 1963, 1972, and 1991. Rapid transit in 1963 consisted solely of commuter railway passenger train service operated by the Milwaukee Road between Watertown, just west of the Region, and the Milwaukee central business district. By 1972, such service had been abandoned and seven freeway flyer motor bus routes were operated within the Milwaukee area. By 1991, rapid transit service in the Region was provided in a fairly dense network of 13 freeway flyer routes, largely within Milwaukee County. While the geographic extent of rapid transit service increased over time, the extent of express transit service was reduced. Express transit in 1963 consisted of 10 bus routes operated within most of the major regional travel corridors oriented to central Milwaukee. By 1972, express service between Waukesha and Oconomowoc and between Milwaukee and East Troy had been dropped and express service between Lake Geneva and Milwaukee cut back to Burlington and Milwaukee. By 1991, seven express transit routes were operated in the Region, primarily in the Milwaukee urban area. Only two regional Milwaukee-oriented travel corridors were served, however: from Milwaukee west to Oconomowoc and south to Racine and Kenosha.

mission, which operated service over seven fixed routes, radial in design and emanating from downtown Kenosha, with direct, nontransfer service from the CBD to all portions of the City and its immediate environs, including the University of Wisconsin-Parkside (see Map 29). The Kenosha transit system also operated two special shuttle routes, which provided local transit service to major commercial, recreational, and employment centers which have developed outside the regular Kenosha local transit service area. In 1991, the system provided service from 6:00 a.m. to 6:00 p.m. every day except Sunday, with approximate 30- to 60minute peak-period headways and 60-minute nonpeak-period headways. As shown on Map 29, peak-period headways exceeded the minimum frequency of service standard, 30 minutes, on two radial routes: one through the north side of the City and the other through the south side of the City.

Local transit service was provided in the Milwaukee urbanized area by the City of Waukesha Transit System Utility and the Milwaukee County Transit System. The fixed-route bus system operated by the City of Waukesha Transit System Utility, Waukesha METRO Transit, provided service over nine fixed radial routes. These nine routes began from downtown Waukesha and provided direct, nontransfer service from the CBD to all portions of the City and its immediate environs. As shown on Map 30, two of the routes served important traffic generators outside of the City: the Waukesha County Technical College in the Town of Pewaukee and the Goerke's Corners transit station in the Town of Brookfield. In 1991, the system provided service from 6:00 a.m. to 6:45 p.m. on weekdays and from 8:00 a.m to 6:00 p.m. on Saturdays at headways of approximately 30 to 60 minutes. Peak-period headways for Route No. 6, a radial route serving the southwest portion of the City of Waukesha, stood at 60 minutes.

As also shown on Map 30, the Milwaukee County Transit System provided local transit service in the Milwaukee urbanized area over 35 regular fixed routes: 15 radial routes emanating from downtown Milwaukee, 16 crosstown routes not serving downtown Milwaukee, and four feeder routes connecting to the crosstown and radial routes. In 1991, the system provided service seven days a week, typically from 5:00 a.m. to 1:00 a.m. Peak-period headways on most routes ranged from 10 to 20 minutes and nonpeak-period headways ranged from 15 to 30 minutes. Headways on three of the local transit routes operated by the Milwaukee County Transit System exceeded 30 minutes during weekday peak periods. Under contract with Waukesha County, the Milwaukee County Transit System also operated an extension of one local route from Milwaukee County to the Brookfield Square Shopping Center in Waukesha County.

Local public transit was provided in the Racine urbanized area by the City of Racine Belle Urban System, which operated service over 10 fixed routes. As shown on Map 31, eight of the 10 fixed routes were radial in design, emanating from the Racine CBD, and provided service to all portions of the City and to its immediate environs. The ninth route, a crosstown route, was routed to the west of downtown Racine. The tenth, a feeder route, served the Town of Caledonia and connected to two of the eight radial routes. In 1991, the system provided service from 5:30 a.m. to 7:00 p.m. on weekdays and from 7:00 a.m. to 6:00 p.m. on Saturdays. Peak-period headways approximated 20 to 45 minutes and nonpeak-period headways ranged from 30 to 45 minutes. Peak-period headways for the route serving the Town of Caledonia exceeded 30 minutes, as shown on Map 31.

Land Area and Population

Served by Local Public Transit

By definition, a local public transit service area encompasses the land and the population within one-quarter mile of a fixed local transit route. The location and extent of this area is a useful measure of the accessibility of local public transit service. The areal extent and resident population of the Kenosha, Milwaukee, and Racine local transit service areas and the number of round trip route miles operated within these areas is set forth in Table 40.

Between 1963 and 1991 the local transit service area within the Region increased significantly. Indeed, local transit operators extended transit routes into existing and newly developing areas at a rate of nearly 52 round-trip route miles per year. Over the span of 28 years, the number of round trip route miles increased from 856 in 1963 to 2,296 in 1991. As a result, over 47 additional square miles were brought within the boundaries of local transit service areas between 1963 and 1991.

Map 29



LOCAL FIXED-ROUTE PUBLIC TRANSIT SERVICE IN THE KENOSHA URBANIZED AREA: 1991

In 1991, local public transit service in the Kenosha urbanized area was provided by the City of Kenosha Transit Commission which operated seven fixed motor bus routes from downtown Kenosha to all portions of the City. Five of the seven routes met the SEWRPC minimally acceptable headway standard of 30 minutes during peak travel periods. In 1991, the Kenosha local transit service area comprised 21 square miles and encompassed a population of about 86,600.

LOCAL FIXED-ROUTE PUBLIC TRANSIT SERVICE IN THE MILWAUKEE URBANIZED AREA: 1991



In 1991, local public transit service in the Milwaukee urbanized area was provided by the Milwaukee County Transit System (MCTS), which operated 35 fixed routes, and by the City of Waukesha Transit System Utility, which operated 10 fixed routes. Peak-period headways on all but three routes provided by MCTS and on all but one provided by the City of Waukesha met the SEWRPC minimally acceptable 30-minute standard. In 1991, the local transit service area in the Milwaukee urbanized area comprised 158 square miles and encompassed a population of about 935,400.
Map 31

LOCAL FIXED-ROUTE PUBLIC TRANSIT SERVICE IN THE RACINE URBANIZED AREA: 1991



In 1991, local public transit service in the Racine urbanized area was provided by the City of Racine Belle Urban System, which operated 10 fixed motor bus routes: eight radial routes, one crosstown route, and one feeder route. Peak-period headways on all but the feeder route, which served the Town of Caledonia and connected to the eight radial routes, met the SEWRPC minimally acceptable 30-minute standard. In 1991, the Racine local transit service area comprised 27 square miles and encompassed a population of about 114,600.

Source: SEWRPC.

Urbanized	Area (square miles)				Population			Round-Trip Route Miles		
Area	1963	1972	1991	1963	1972	1991	1963	1972	1991	
Kenosha	10.9 136.9	15.9 147.5	20.9 158.0	71,900 1,025,800	81,500 1,026,000	86,600 935,400	55 716	59 1,061	171 1,954	
Racine	11.7	14.0	27.4	95,600	97,100	114,600	76	81	171	
Total	159.5	177.4	206.3	1,193,300	1,204,600	1,136,600	847	1,201	2,296	

LOCAL FIXED-ROUTE TRANSIT AREAL EXTENT, POPULATION SERVED, AND ROUND-TRIP ROUTE MILES BY URBANIZED AREA: 1963, 1972, AND 1991

	Area (square miles)				Population				Round-Trip Route Miles			
Urbanized	Change 1963-1991		Change 1972-1991		Change 1963-1991		Change 1972-1991		Change 1963-1991		Change 1972-1991	
Area	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Kenosha	10.0 21.1 15.7	91.7 15.4 134.2	5.0 10.5 13.4	31.4 7.1 95.7	14,700 -90,400 19,000	20.4 -8.8 19.9	5,100 -90,600 17,500	6.3 -8.8 18.0	116 1,238 95	210.9 172.9 125.0	112 893 90	189.8 84.2 111.1
Total	46.8	29.3	28.9	16.3	-56,700	-4.8	-68,000	-5.6	1,449	171.1	1,095	91.2

Source: SEWRPC.

Regionwide, the expansion of the local transit service areas did not translate into an expansion in the number of people served by local transit. Indeed, as shown in Table 40, about 56,700 fewer potential transit users resided within one-quarter mile of a local transit route in 1991 than in 1963. This phenomenon is indicative of the dispersion of population away from older urban areas in the Region, where transit service has traditionally been provided.

The dominance of the Milwaukee urbanized area in the consideration of regionwide transit statistics is particularly evident when considering resident population within local transit service areas. While the population served by local transit within the Region decreased by nearly 5 percent between 1963 and 1991, the population served within the Kenosha and Racine urbanized areas actually increased by about 20 percent in both areas. Between 1963 and 1991, through the extension of transit service and incremental urban development within the Kenosha and Racine urbanized areas, an additional 33,700 persons were added to the local transit service areas. During the same period, however, the local transit service area population within the Milwaukee urbanized area declined by about 90,400 persons.

Table 41 shows the percentage of land area and percentage of population served by local transit

in each of the urbanized areas in 1963, 1972, and 1991. Over the long term, between 1963 and 1991, the percentage of urbanized land area and population served has decreased. In 1963, 38 percent of the urbanized land area and 86 percent of the urbanized population in the Region was served by local transit. By 1991 these figures had decreased to about 35 percent and 79 percent, respectively.

Changes in transit service over time are evident in comparing the number of vehicle-miles provided and the number of route miles over which transit vehicles operated. Vehicle-miles may be defined as the total number of miles travelled by fixed-route transit vehicles. Thus, more frequent transit service will result in more vehicle-miles of service. Round-trip route miles may be defined as the total mileage of streets and highways over which transit vehicles operate on fixed routes. Therefore, if round trip route miles remain constant over time and vehicle-miles increase, transit service may be said to be improved, at least to be more frequent.

As previously indicated, between 1972 and 1991 each of the fixed-route local transit providers within the Region increased round-trip route miles in an attempt to serve broader areas. To maintain existing service levels, therefore, or to improve such levels, the same transit operators would have had to, at minimum, increase the

AREA AND POPULATION SERVED BY LOCAL FIXED-ROUTE TRANSIT: 1963, 1972, AND 1991

	Urbanized Area (square miles)			Local	Percent of Urbanized Area Served				
Urbanized Area	1963	1972	1991	1963	1972	1991	1963	1972	1991
Kenosha Milwaukee Racine	13.3 392.0 14.5	17.5 456.4 28.1	41.1 511.9 38.8	10.9 136.9 11.9	15.9 147.5 14.0	20.9 158.0 27.4	82.0 34.9 82.1	90.9 32.3 50.0	50.8 30.8 70.6
Total	419.8	502.0	590.8	159.5	177.4	206.3	38.0	35.3	34.9

	Urbanized Area Population			Loc A	Percent of Population Served				
Urbanized Area	1963	1972	1991	1963	1972	1991	1963	1972	1991
Kenosha	73,400	86,500	94,300	71,900	81,500	86,600	91.8	94.2	91.8
Milwaukee	1,216,500	1,267,400	1,226,300	1,025,800	1,026,000	935,400	76.2	81.0	76.2
Racine	95,900	115,200	121,800	95,600	97,100	114,600	94.1	84.3	94.1
Total	1,385,800	1,469,100	1,442,400	1,193,300	1,204,600	1,136,600	78.8	82.0	78.8

^aArea within one-quarter mile of a local bus route.

^bEstimate based upon population data from the U. S. Census allocated to U. S. Public Land Survey quarter-sections by Commission staff.

Source: U. S. Bureau of the Census and SEWRPC.

Table 42

			_	Change: 1	963-1991	Change: 1972-1991		
Urbanized Area	1963	1972	1991	Number	Percent	Number	Percent	
Kenosha	2,500	1,100	2,500	0	0.0	1,400	127.3	
Milwaukee	78,900	61,300	56,400	-22,500	-28.5	-4,900	-8.0	
Racine	3,500	1,600	4,400	900	25.7	2,800	175.0	
Total	84,900	64,000	63,300	-21,600	-25.4	-700	-1.1	

AVERAGE DAILY FIXED-ROUTE TRANSIT VEHICLE-MILES PROVIDED WITHIN THE REGION BY URBANIZED AREA: 1963, 1972, AND 1991

Source: SEWRPC.

provision of vehicle-miles. As shown in Table 42, this was the case in both the Kenosha and Racine local transit service areas. Within the Kenosha transit service area vehicle-miles increased from 1,100 in 1972 to 2,500 in 1991. Within the Racine transit service area vehiclemiles were increased from 1,600 in 1972 to 4,400 in 1991. By contrast, the provision of vehiclemiles within the Milwaukee urban area was reduced by about 8 percent, from 61,300 in 1972 to 56,400 in 1991, indicating an overall decline in local transit service. The magnitude of this reduction in the Milwaukee area was sufficient to reduce the regionwide total by about 1 percent.

Though the size and resident population of local service areas provides a measure of service provided, a more useful measure of the degree to which local transit serves a given area and its residents involves the application of plan design standards. If local transit is to provide a reasonable alternative to the personal vehicle and is to be integrated with express and rapid transit systems, it must meet certain minimum standards with respect to frequency of service. The

LAND AREA AND POPULATION SERVED BY FIXED-ROUTE LOCAL TRANSIT MEETING SEWRPC MINIMUM SERVICE FREQUENCY STANDARDS^a BY URBANIZED AREA: 1991

		Land Area		Population				
Urbanized Area	Area (square miles)	Area Served (square miles)	Percent of Urbanized Area Served	Population	Population Served	Percent of Urbanized Population Served		
Kenosha	41.1	18.9	46	94,292	76,500	81		
Milwaukee	511.9	151.7	30	1,226,293	926,000	76		
Racine	38.8	22.5	58	121,788	101,100	83		
Total	591.8	193.2	33	1,442,373	1,103,660	77		

^aThe SEWRPC standard for minimum frequency of service for local fixed-route public transit states that headways shall not exceed 30 minutes during peak periods, nor 60 minutes during nonpeak periods.

Source: U. S. Bureau of the Census and SEWRPC.

frequency of service standard, adopted as part of design year 2000 regional transportation system plan, stated that local transit headways² shall not exceed 30 minutes during weekday peak periods nor exceed 60 minutes during nonpeakperiods. Areas to which transit is provided with headways exceeding this standard are not considered by the Commission to be served.

Comparing existing local transit service with service frequency standards permits delineation of the land area and population provided with adequate levels of transit. The land area and population of each urbanized area within onequarter mile of a local transit route meeting the planning standard for frequency of service is set forth in Table 43. As shown in Table 43. 46 percent of the land area and 81 percent of the residential population of the Kenosha urbanized area was served by local transit in 1991, about 30 percent of the land area and 76 percent of the resident population of the Milwaukee urbanized area was served by local transit in 1991, and about 58 percent of the land area and 83 percent of the resident population of the Racine urbanized area was served by local transit in 1991. When the frequency of service standard is applied it becomes clear that the significantly expanded local transit service area in the Racine urbanized area, due to the provision of Route 10 in northern areas of the City of Racine, has not been accompanied by a frequency of service that could help attract new transit users to transit.

Transit Provision and Utilization

Capacity determinants are quite different for public transit systems from those for arterial street and highway systems. The capacity of a transit system depends on the type of facilities and vehicles, the passenger capacity of vehicles, and number of vehicles operating during a given period of time. Consequently, actual transit operating capacities can be readily increased or decreased by adding or subtracting from scheduled services. Transit system capacity is more flexible and less precisely quantifiable than the capacity of the arterial street and highway system.

Capacity is, therefore, less useful as a criterion for evaluating an existing transit system than it is for evaluating an existing arterial street and highway system. Service provided, in terms of seat-miles, is a more meaningful criterion for such an evaluation. This criterion is readily comparable to passenger miles, a measure of transit utilization both on an individual link and on a systemwide basis. The frequency of service provided, in terms of the number of units per hour or day on a given route, is a factor in the determination of seat-miles provided and is also an important measure in itself. Waiting time is

 $^{^{2}}$ The term "headway" is defined as the time interval between any two successive transit vehicles providing service on the same route in the same direction.

	Transit Servic	e Provided	Transit Ser		
Urbanized Area	Seat-Miles	Percent of Total	Passenger Miles	Percent of Total	Percent Utilization
Kenosha Milwaukee Racine	110,700 2,692,300 172,000	3.7 90.5 5.8	16,800 573,200 19,100	2.8 94.1 3.1	15.2 21.3 11.1
Total	2,975,000	100.0	609,100	100.0	20.5

UTILIZATION OF FIXED-ROUTE TRANSIT SERVICE ON AN AVERAGE WEEKDAY WITHIN THE REGION BY URBANIZED AREA: 1991

Source: SEWRPC.

a component of overall transit travel time. Both measures were utilized in the transit inventory presented herein.

Transit service provided in each of the urbanized areas within the Region, together with the utilization of this service, is summarized in Table 44. Over 90 percent of the seat-miles provided and 94 percent of the passenger miles recorded within the Region in 1991 were attributable to the operation of transit within the Milwaukee urbanized area. Consequently, the supply and use of transit within the Milwaukee urbanized area dominates any systemwide consideration of transit utilization.

Percent utilization is a measure of how well the existing supply of transit service is used. It is determined by dividing the number of passenger miles recorded within an area by the number of seat-miles provided within the area. In 1991, the Region overall achieved a daily percent utilization rate of about 21 percent, matching the percent utilization rate recorded in the Milwaukee urbanized area. The urbanized areas of Kenosha and Racine achieved a 15 percent and a 11 percent utilization rate, respectively.

As shown in Figure 26, between 1972 and 1991 the utilization of public transit within the Milwaukee and Kenosha urbanized areas increased from 17 to 21 percent, and from 9 to 15 percent, respectively. Conversely, the percent utilization of public transit in the Racine urbanized area decreased from 18 percent in 1972 to 11 percent in 1991. The decline evident in the Racine urbanized area is a consequence of several factors, including the 13-square-mile expansion

Figure 26

PERCENT UTILIZATION OF FIXED-ROUTE TRANSIT SERVICE ON AN AVERAGE WEEKDAY WITHIN THE REGION BY URBAN AREA: 1963, 1972, 1991



Source: SEWRPC.

of the local public transit service area between 1972 and 1991 and a required increase in route miles to serve that expanded area plus a change in the bus fleet which doubled the average vehicle capacity. In all three urbanized areas, the utilization of public transit is significantly below levels set in 1963; the comparable percent utilization rates by urbanized area were: Milwaukee, 30; Kenosha, 19; and Racine, 24.

In order to evaluate the regionwide percent utilization rates set forth in Table 44, the annual percent utilization of public transit in the

RANKING OF SIMILAR URBANIZED AREAS BY ANNUAL PERCENT UTILIZATION OF PUBLIC TRANSIT SERVICE: 1990

Urbanized Area	Population	Annual Seat-Miles (millions)	Annual Passenger Miles (millions)	Percent ^a Utilization
Cincinnati	1,212,675	553.8	168.5	30.4
Milwaukee	1,226,293	805.4	177.1	22.0
Fort Lauderdale/Hollywood/				
Pompano Beach	1,238,134	402.6	82.0	20.4
Portland/Vancouver	1,172,158	954.4	174.5	18.3
Kansas City	1,275,317	395.3	69.8	17.7
Riverside/San Bernadino	1,170,196	242.9	42.7	17.6
Norfolk/Virginia Beach/				
Newport News	1,323,098	339.8	44.8	13.2

Urbanized Area	Total Land Area (square miles)	Annual Seat-Miles (millions)	Annual Passenger Miles (millions)	Percent ^a Utilization
Baltimore	593	976.4	319.5	32.7
	512	553.8	168.5	30.4
	588	1,311.0	397.2	30.3
Milwaukee	512	805.4	177.1	22.0
Cleveland	636	1,035.1	200.0	19.4
	469	258.7	49.6	19.2
	508	261.6	45.1	17.2

Urbanized Area	Population Density (persons per square mile)	Annual Seat-Miles (millions)	Annual Passenger Miles (millions)	Percent ^a Utilization
Cincinnati	2,369	553.8	168.5	30.4
Memphis	2,420	281.5	65.7	23.4
Milwaukee	2,395	805.4	177.1	22.0
Albany/Schenectady/Troy	2,436	261.0	56.4	21.6
New Haven/Meriden	2,402	154.2	29.4	19.1
Wilmington	2,392	106.3	17.9	16.9
Fort Wayne	2,389	62.5	4.7	7.6

^aPercentile utilization is derived by dividing annual passenger miles by annual seat-miles.

Source: U. S. Department of Transportation, Federal Transit Administration, <u>Transit Profiles – Agencies in Urbanized Areas</u> <u>Exceeding 200,000 Population</u>, Washington, D. C., 1991; and SEWRPC.

Milwaukee urbanized area was compared to that of other urbanized areas in the United States of similar population size, land area size, and population density. As shown in Table 45, in terms of 1990 annual utilization of fixed-route motor bus transit, the Milwaukee urbanized area, with an annual utilization rate of 22 percent, ranked second highest among areas similar in population size; fourth highest among areas similar in total land area; and third highest among areas similar in population density. A cumulative ranking of all the 17 different urbanized areas listed in Table 45 indicates that in terms of providing annual seat-miles, the Milwaukee urbanized area ranked fifth highest, providing 805.4 million in 1990. In terms of annual passenger miles, the Milwaukee urbanized area ranked fourth highest, recording 177.1 million in 1990. The findings indicate that, compared to similar urbanized areas, the Milwaukee urbanized area provided a significantly high level of annual seat-miles and recorded a respectful level of transit use in terms of annual passenger miles.

Within the Region, a more detailed analysis of transit utilization was made anent the CBDs of the cities of Kenosha, Milwaukee, and Racine. The number of bus seats provided and passengers carried on each route entering and leaving the CBDs was summarized by direction of travel and hour of day. Results are displayed graphically in Figures 27 through 29 and presented, in aggregated form, in Table 46. It should be noted that these comparisons represent gross evaluations of service provided and utilized during selected times and do not reflect actual passenger loadings on individual routes. Consequently, the use of seats on certain routes entering or leaving the CBDs may be proportionately higher or lower than displayed.

The findings indicate that in 1991 a significantly higher ratio of service provided to service used existed on an average weekday in Kenosha and Racine than in Milwaukee. Indeed, as is shown in Table 46, in 1991 buses entering the CBDs of Kenosha and Racine provided, respectively, 4.8 and 4.1 seats per passenger carried, while in Milwaukee, inbound buses provided 2.3 seats for every inbound transit passenger. These findings indicate that public transit service is provided more efficiently to and from the Milwaukee CBD. Over all in 1991, there were approximately 2.5 seats available to every transit passenger entering or leaving the major central business districts within the Region.

Public Transit Ridership

Annual public transit ridership levels recorded in 1963, 1973, and 1991 within the Kenosha, Milwaukee, and Racine urbanized areas are set forth in Table 47. Public transit ridership within the Region has declined significantly over time. In 1963, over 94.5 million revenue passengers were carried on public transit within the three urbanized areas of the Region. In 1972, about 53.9 million revenue passengers were carried, about 43 percent fewer than in 1963. In 1991, about 50.2 million passengers were carried, 47 percent fewer than in 1963 and about 7 percent fewer than in 1972. It is interesting to note that the 1991 public transit ridership levels within the Kenosha and Racine urbanized areas were more than two and three times greater than the 1972 levels, respectively. The 1991 ridership level in the Milwaukee urbanized area, however, was nearly 11 percent lower than the 1972 level.

A similar pattern emerges when public transit rides per capita are considered. As shown in Table 48, rides per capita, that is, the number of annual revenue passengers per resident of the respective local transit service areas, increased significantly between 1972 and 1991 within the Kenosha and Racine urbanized areas and decreased slightly within the Milwaukee urbanized area. Though increases have occurred since 1972, rides per capita in each urbanized area are significantly below levels set in 1963.

The historical trends in transit ridership in the Kenosha, Milwaukee, and Racine urbanized areas are shown in Figure 30. Ridership within the Kenosha and Racine urbanized areas grew gradually from 1972 though 1980, leveled off, and declined slightly to current levels. Ridership within the Milwaukee urbanized area increased in the late 1970s until 1980 and has generally declined since then.

SUPPLY AND USE OF INTERMODAL PARKING FACILITIES

Intermodal parking facilities are an important element of the regional transportation system since they affect transportation modal choice, and hence traffic congestion, within certain travel corridors. Parking supply and demand at public transit lots and at carpool lots is monitored annually by the Wisconsin Department of Transportation and the Commission. Such parking facilities, unlike those located within the CBDs and at other major activity centers, serve primarily as intermodal transfer points and are explicitly considered in the regional transportation system planning process.

The role of intermodal parking facilities in the attainment of transportation development objectives, especially as related to traffic congestion, air quality, and energy conservation, is critical. Commuters utilizing public transit lots shift from personal transportation modes to rapid or express transit while commuters utilizing carpool lots typically shift to some form of ridesharing. These modal changes alleviate traffic congestion and reduce air pollutant emissions and fuel consumption. An adequate supply of

Figure 27

WEEKDAY BUS SEATS PROVIDED AND PASSENGERS CARRIED TO AND FROM THE KENOSHA CENTRAL BUSINESS DISTRICT: 1991



Source: SEWRPC.

Figure 28

WEEKDAY BUS SEATS PROVIDED AND PASSENGERS CARRIED TO AND FROM THE MILWAUKEE CENTRAL BUSINESS DISTRICT: 1991



Source: Milwaukee County Transit System and SEWRPC.





WEEKDAY BUS SEATS PROVIDED AND PASSENGERS CARRIED TO AND FROM THE RACINE CENTRAL BUSINESS DISTRICT: 1991

Source: SEWRPC.

SEATS PROVIDED AND PASSENGERS CARRIED BY FIXED-ROUTE PUBLIC TRANSIT BY CENTRAL BUSINESS DISTRICT

	Inbound Trips			Outbound Trips			
Central	Seats	Passengers	Ratio of Seats	Seats	Passengers	Ratio of Seats	
Business District	Provided	Carried	to Passengers	Provided	Carried	to Passengers	
Kenosha	6,644	1,383	4.81	6,732	1,421	4.74	
Milwaukee	81,800	35,837	2.28	81,652	35,972	2.27	
Racine	12,880	3,168	4.07	12,800	3,550	3.60	
Total	101,324	40,388	2.51	101,184	40,943	2.47	

Source: SEWRPC.

Table 47

ANNUAL FIXED-ROUTE PUBLIC TRANSIT RIDERSHIP WITHIN THE REGION BY URBANIZED AREA: 1963, 1972, 1991

	Annual Revenue Passengers			Change: 19	72-1991	Change: 1963-1991		
Urbanized Area	1963	1972	1991	Number	Percent	Number	Percent	
Kenosha	1,884,400	503,200	1,128,000	624,800	124.2	-756,400	-40.1	
Milwaukee	89,761,600	52,875,400	47,267,100	-5,608,300	-10.6	-42,494,500	-47.3	
Racine	2,902,000	525,700	1,827,800	1,302,100	247.7	-1,074,200	-37.0	
Total	94,548,000	53,904,300	50,222,900	-3,681,400	-6.8	-44,325,100	-46.9	

Source: SEWRPC.

Table 48

ANNUAL RIDES PER CAPITA ON FIXED-ROUTE PUBLIC TRANSIT SERVICE WITHIN THE REGION BY URBANIZED AREA: 1963, 1972, AND 1991

	Local Mass Transit Rides Per Capita							
Urbanized Area	1963	1972	1991	Percent Change 1963-1991	Percent Change 1972-1991			
Kenosha	26	6	13	-50.3	111.0			
Milwaukee Racine	88 30	52 5	51 16	-42.4 -47.5	-1.9 194.6			
Combined Urbanized Area Rides Per Capita	79	45	44	-44.4	-1.3			

Source: SEWRPC.

both types of facilities at appropriate locations supports the operation of the multi-modal transportation system within the Region.

Public Transit Parking

In 1991, freeway flyer service was provided to 18

parking lots, distributed about evenly among four major rapid transit corridors within the Region, as shown on Map 32. These intermodal facilities provided 3,740 parking spaces. An additional combined 335 parking spaces were provided at the Milwaukee Area Technical College transit station in Mequon and at the W. Good Hope Road transit station, although transit service was not provided to these sites in 1991. Consequently, any parking use recorded during 1991 at these sites may be attributed to carpool participants.

The utilization of parking at sites to which public transit service was provided in 1991 is set forth in Table 49 and is shown on Map 32. The greatest utilization of such parking occurred within the USH 41/45 rapid transit corridor, where about 58 percent of available parking spaces were used on an average weekday during the fourth quarter of 1991. The lowest utilization of freeway flyer parking occurred within the IH 43 North rapid transit corridor, where about 39 percent of available parking spaces were used on an average weekday during the fourth quarter of 1991.

The utilization of parking spaces at individual parking lots illustrated on Map 32 ranges from a high of 83 percent, at the Pilgrim Road public transit station in Menomonee Falls, to a low of 20 percent, at the Northridge public transit station in Milwaukee. In addition to the Pilgrim Road site, other parking lots with utilization rates greater than 60 percent, indicated on the accompanying map, are located at the following locations: Whitnall transit station, City of Hales Corners; State Fair Park, City of West Allis; and Phar-Mor shopping center, City of Brookfield.

As indicated in Table 49 public transit parking facilities within the Region are located at certain public transit stations and at certain shopping centers. On an average weekday during the fourth quarter of 1991, nearly 47 percent of the 3,125 parking spaces at public transit stations³ and about 46 percent of the 725 spaces available at shopping center lots were used. The utilization of spaces at public transit stations peaked in 1981, when about 71 percent of the approximately 2,560 parking spaces were used (see

Figure 30

HISTORIC PUBLIC TRANSIT RIDERSHIP IN URBANIZED AREAS WITHIN THE REGION





Figure 31). The utilization of spaces at shopping center lots peaked in 1980, when about 78 percent of the 1,095 spaces available were used.

Carpool Parking

In 1991, there were 16 carpool parking facilities located at key freeway interchanges within the

³The annual monitoring of the use of public transit parking at public transit stations indicated that on an average weekday during the entire year of 1991 about 46 percent of available spaces were used.



In 1991, public transit service was provide to 18 parking lots distributed among major travel corridors emanating from the City of Milwaukee. Together, the parking lots provided 3,850 parking spaces, of which 47 percent were determined to be used on an average weekday during the fourth quarter of 1991. The highest utilization of public transit parking spaces was recorded at the Pilgrim Road transit station in Menomonee Falls, where 83 percent of the available spaces were used on an average weekday.

SUPPLY AND USE OF PARKING AT PUBLIC TRANSIT PARKING LOTS WITHIN THE REGION BY RAPID TRANSIT CORRIDOR: FOURTH QUARTER 1991

	Available	Vehicles Parked	Percent of
Location ^a	Parking Spaces	(average weekday)	Spaces Used
IH 43-North Rapid Transit Corridor	100	20	20.0
Target Shopping Center (Brown Deer)	125	75	60.0
Brown Deer (River Hills)	360	137	38.1
Silver Mill Shopping Center (Milwaukee)	100	23	23.0
North Shore (Glendale)	175	79	45.1
Subtotal	860	334	38.8
Percent of Regionwide Total	22.3	18.5	
USH 41/45 Rapid Transit Corridor			
Pilgrim Road (Menomonee Falls)	65	54	83.1
Timmerman Field (Milwaukee)	140	42	30.0
Phar-Mor Shopping Center (Brookfield)	200	138	69.0
W. Watertown Plank Road (Wauwatosa)	200	96	48.0
State Fair Park (West Allis)	200	137	68.5
Subtotal	805	467	58.0
Percent of Regionwide Total	20.9	25.8	
IH 94/STH 16-West Rapid Transit Corridor			_
Olympia Shopping Center (Oconomowoc)	50	7 ^b	14.5 ^b
STH 67 and IH 94 (Summit)	80	46	57.5
Goerke's Corners (Brookfield)	250	107	42.8
Subtotal	380	160	42.1
Percent of Regionwide Total	9.9	8.9	
IH 43-South Rapid Transit Corridor			
W. Loomis Road (Greenfield)	415	96	23.1
Southridge Shopping Center (Greendale)	250	88	35.2
Whitnall (Hales Corners)	370	272	73.5
Subtotal	1,035	456	44.1
Percent of Regionwide Total	26.9	25.2	
IH 94-South Rapid Transit Corridor			
W. College Avenue (Milwaukee)	530	287	54.1
W. Holt Avenue (Milwaukee)	240	103	42.9
Subtotal	770	390	50.6
Percent of Regionwide Total	20.0	21.6	
Regionwide Total	3,850	1,807	46.9

NOTE: Public transit service was not provided during 1991 to the Milwaukee Area Technical College (Mequon) and the W. Good Hope Road public transit stations. In 1991, these public transit stations had a combined total of 335 parking spaces. In 1991, there were no freeway flyer parking lots located in the IH 794-South rapid transit corridor.

^aIn 1991, freeway flyer parking facilities, those parking lots served by rapid transit in 1991, were located at public transit stations and at shopping center parking lots within the Region.

^bCommission estimate.

Source: SEWRPC.



PUBLIC TRANSIT PARKING LOT USE: 1972-1991



Source: SEWRPC.

Region containing 1,112 parking spaces.⁴ Most of such parking, about 41 percent, was located within the IH 43-South travel corridor. About 29 percent of carpool parking was sited within the IH 94/STH 16-West travel corridor. The IH 43-North and USH 41/45 corridors combined to account for the remaining 30 percent of total carpool parking. No carpool parking facilities were located within the IH 94-South or the USH 12 travel corridors.

The utilization of the carpool parking within each travel corridor is set forth in Table 50 and shown on Map 33. The greatest utilization of carpool parking, about 46 percent, occurred within the IH 94/STH 16-West corridor. The lowest utilization of carpool parking, about 25 percent, occurred within the IH 43-South corridor. Regionwide, 32 percent of the 1,112 spaces available at carpool lots were used on an average weekday during the fourth quarter of 1991.⁵

The utilization of parking spaces at the individual carpool parking lots is also illustrated on

⁴Three public transit lots in the Region, USH 41 and Pilgrim Road, IH 94 and STH 67, and IH 94 and USH 18, functioned in part as carpool parking lots. Because public transit service was provided to these locations, they are listed in Table 49).

⁵The annual monitoring of the use of carpool parking indicated that on an average weekday during the entire year of 1991 about 33 percent of available spaces were used.

SUPPLY AND USE OF PARKING AT CARPOOL PARKING LOTS WITHIN THE REGION BY TRAVEL CORRIDOR: FOURTH QUARTER 1991

	Available	Vehicles Parked	Percent of
Location	Parking Spaces	(average weekday)	Spaces Used
IH 43-North Travel Corridor			·
IH 43 and STH 32 and STH 84 (Port Washington)	50	3	6.0
IH 43 and STH 57 (Saukville)	100	14	14.0
IH 43 and CTH C (Grafton)	50	34	68.0
STH 57 and STH 84 (Fredonia)	10	6	60.0
Subtotal	210	58	13.3
Percent of Regionwide Total	18.9	16.5	
USH 41/45 Travel Corridor			
USH 41 and CTH Y (Germantown)	100	20	20.0
STH 60 and CTH P (Jackson)	30	13	43.3
Subtotal	130	33	25.4
Percent of Regionwide Total	11.7	9.4	
IH 94/STH 16-West Travel Corridor			
STH 16 and CTH C (Nashotah)	50	15	30.0
STH 16 and STH 83 (Chenequa)	65	15	23.1
STH 16 and CTH P (Oconomowoc)	45	19	42.2
IH 94 and CTH CC (Delafield)	30	19	63.3
IH 94 and CTH G (Pewaukee)	50	22	44.0
IH 94 and STH 164 (Pewaukee)	80	57	71.3
Subtotal	320	147	45.9
Percent of Regionwide Total	28.8	41.9	. <u></u>
IH 43-South Travel Corridor			
IH 43 and STH 83 (Mukwonago)	95	42	44.2
IH 43 and STH 164 (Big Bend)	100	27	27.0
IH 43 and CTH Y (New Berlin)	60	18	30.0
IH 43 and CTH O (New Berlin)	197	27	13.7
Subtotal	452	114	25.2
Percent of Regionwide Total	40.6	32.5	
Regionwide Total	1,112	351	31.6

NOTE: In 1991, no carpool parking lots were located within the IH 794, IH 94 South, and USH 12 travel corridors.

During an average weekday in the fourth quarter of 1991, 21 of the combined total 335 parking spaces available at Milwaukee Area Technical College (Mequon) and the W. Good Hope Road public transit stations were recorded as being used, presumably by carpool participants, since transit service was not provided to these sites during 1991.

Source: SEWRPC.

Map 33. It ranges from a high of 71 percent at the IH 94 and STH 164 site in Pewaukee to a low of 6 percent at the IH 43/STH 32 and STH 84 site in Port Washington. In addition to the IH 94 and STH 164 site, other lots with utilization rates greater than 60 percent, indicated on Map 33, are located at the following major interchanges: IH 43 and CTH C in Grafton and IH 94 and CTH CC in Delafield. In addition to the IH 43/ STH 32 and STH 84 site, lots with utilization rates below 15 percent, indicated on the accompanying map, are located at the following major interchanges: IH 43 and STH 57 in Saukville and IH 43 and CTH 0 in New Berlin.



In 1991, there were 16 carpool parking lots distributed among major travel corridors emanating from the City of Milwaukee. Together, the parking lots provided 1,112 parking spaces, of which 32 percent were determined to be used on an average weekday during the fourth quarter of 1991. The highest utilization of carpool parking spaces was recorded at the IH 94 and STH 164 site in Pewaukee, where 71 percent of the available spaces were used on an average weekday.

Figure 32

CARPOOL PARKING LOT USE: 1978-1991



Source: SEWRPC.

Figure 32 shows the historic trend in the supply and use of parking at carpool lots. The peak in the use of such spaces, coinciding with the use of freeway flyer parking lots, occurred during the fourth-quarter of 1981 at about 49 percent, when 468 of the total 957 parking spaces were used on an average weekday. The use of carpool parking lots declined between 1981 and 1986 and increased slightly each successive year to its present level.

FREEWAY TRAFFIC MANAGEMENT SYSTEM

The existing freeway traffic management system is shown in graphic summary form on Map 34. In 1991, 21 freeway access points were equipped with on-ramp meters and corresponding traffic detectors in the Milwaukee area, 18 of which utilized microprocessor-based equipment. The metered on-ramps are located adjacent to freeway segments that experience the most severe traffic congestion during morning and evening weekday peak-traffic periods.

The principal objective of the freeway management system is to reduce the severity and duration of freeway traffic congestion. The onramp meters exercise control of freeway traffic volumes by restricting on-ramp traffic, thereby preventing groups of vehicles from attempting to merge into traffic simultaneously. The controller for each on-ramp meter can select one of three metering rates to accomplish the objective of smoothing traffic flow: one vehicle every six, nine, or 12 seconds, depending on the level of freeway congestion measured by the corresponding traffic detector at a location just "upstream" of the meter.

In 1991, preferential access was provided at four freeway access points and exclusive access was provided at two freeway access points to buses providing rapid and express transit service from the outlying park-and-ride lots and public transit stations to the Milwaukee CBD. Preferential access allows buses to bypass automobile traffic waiting at controlled ramps.

A comprehensive freeway management system for the Milwaukee area is not currently in place. The Commission recommended the development of such a system in 1988 with the publication of Planning Report No. 39, A Freeway Traffic Management System Plan for the Milwaukee Area. A comprehensive freeway management system would provide the means to monitor existing traffic flow rates systemwide and to regulate access to the freeway system based on current traffic conditions, that is, on a real-time basis. It would consist of a central traffic management center and include extensive incident management procedures, motorist advisory information and assistance services, and coordination of freeway controls with local street traffic controls. Preliminary engineering for the development of such a comprehensive freeway management system within the Milwaukee area is currently underway.

SUMMARY

This chapter has summarized the existing supply and utilization of the regional transportation system, arterial streets and highways,



Traffic entering the freeway system at a limited number of freeway on-ramps within the Milwaukee area freeway system is currently metered during the morning and evening peak periods. This map shows the locations of the 21 existing metered ramps, as well as the locations of the four existing ramp meters with bypasses for busses.

Source: Wisconsin Department of Transportation and SEWRPC.

fixed-route public transit, intermodal parking, and freeway traffic management. The chapter has also documented the changes that have occurred in the status and use of the system since 1972, the base year of the secondgeneration regional transportation system plan, and, in some cases, since 1963, the base year of the initial plan. A number of inventory findings with particular significance for transportation system development are evident. These include:

- 1. As of 1991 there were approximately 11,200 miles of street and highways of all types, local/land-access, collector, and arterial, open to traffic within the Region. Nearly 29 percent, or 3,274 miles of the total street and highway system functioned as arterials.
- 2. In 1991, approximately 33 million vehiclemile of travel were found to occur on the arterial street and highway system on an average weekday within the Region. Most of this arterial utilization occurred within the intensely developed portions of the Region, particularly along the travel corridors emanating from the Milwaukee CBD through the Milwaukee urbanized area. The most intensive use of the arterial system, about 19,000 vehicle-miles of travel per arterial mile, occurred in Milwaukee County. Freeways within the Region, which constituted 295 miles or about 9 percent of the total arterial system, carried 35 percent of total arterial system vehicle-miles of travel on an average weekday in 1991.
- 3. In 1991, approximately 543 miles, or about 17 percent of the total arterial street and highway system, were operating at or over design capacity. About 43 percent of this mileage, or 232 miles, was within Milwaukee County.
- 4. In 1991, fixed-route public transit service in the form of local, express, and rapid transit was provided within the Kenosha, Milwaukee, and Racine urbanized areas. The operation of public transit within the Milwaukee urbanized area accounted for about 90 percent of the total seat-miles and about 94 percent of the total passenger miles recorded within the Region. The annual 1990 utilization of public transit in

the Milwaukee urbanized area was recorded at 22 percent, indicating a slightly more effective provision of service than was provided in most urbanized areas of similar areal extent, population size, and population density within the United States. The 1991 daily percent utilization rates of public transit in the Kenosha, Milwaukee, and Racine urbanized areas were recorded at 15 percent, 21 percent, and 11 percent, respectively.

- 5. In 1991, nearly 33 percent of the urbanized land area and 77 percent of the urbanized area population was served by local public transit operating at or below 30-minute peak-period standard headways. In the Kenosha urbanized area 51 percent of the land area and 92 percent of the population was served by local transit. In the Milwaukee urbanized area 31 percent of the land area and 76 percent of the population was served by local public transit. In the Racine urbanized area 71 percent of the land area and 94 percent of the population was served by local transit.
- 6. Between 1972 and 1991, public transit ridership within the Region decreased by about 7 percent. This decline may be attributed to a 11 percent decrease in transit ridership within the Milwaukee urbanized area which was sufficient to offset concurrent ridership gains in the Kenosha and Racine urbanized areas. Between 1972 and 1991, public transit ridership within the Kenosha urbanized area more than doubled and ridership within the Racine urbanized area more than tripled.
- 7. On an average weekday during the fourth quarter of 1991, about 47 percent of the 3,850 parking spaces available to public transit patrons within the Region were used. Public transit parking lots were located at public transit stations and certain shopping centers along major rapid transit freeway corridors. In 1991, the use of such parking was found to be about 39 percent within the IH 43 North corridor, about 58 percent within the USH 41/45 corridor, about 42 percent within the IH 94/STH 16-West corridor, about 44 percent within IH 43 South corridor, and

about 51 percent within the IH 94-South corridor.

- 8. On an average weekday during the fourth quarter of 1991 about 32 percent of the 1,112 available parking spaces at carpool lots within the Region were used. In 1991, the use of available parking spaces at carpool lots was found to be about 13 percent within the IH 43-North corridor, about 25 percent within the USH 41/45 corridor, about 46 percent within the IH 94/STH 16-West corridor, and about 25 percent within the IH 43-South corridor. No carpool parking facilities were located within the IH 94 South, the IH 794, or the USH 12 freeway corridors.
- 9. In 1991, 21 freeway on-ramps within the Milwaukee area were equipped with metering devices to control freeway traffic volumes by monitoring and restricting onramp traffic flows. Preferential access to the freeway system was provided to buses at four freeway access points. Exclusive bypass of the meters was provided at two access points.

Significant changes have occurred in the provision and utilization of arterial street and highway and public transit facilities and services since 1963, the date of the Commission's first definitive travel related inventories. These changes are summarized in Table 51 and discussed below.

The total street and highway mileage within the Region increased between 1963 and 1991 by about 25 percent, from 8,943 miles to 11,199 miles. This increase may be attributed in part to the increase in urban development within the Region and the consequent construction of local/ land-access and collector facilities to serve that development. From 1963 to 1985 the amount of land devoted to urban land uses increased by 37 percent, while resident population increased by 4 percent, the number of households increased by 34 percent, and employment increased by 35 percent.

The portion of total street and highway system mileage made up of arterial facilities has declined over time. In 1963, nearly 36 percent of total street and highway mileage was functionally classified as arterial; in 1972, about 32 percent; and in 1991, about 29 percent. Between 1963 and 1972 arterial system mileage decreased by 70 miles, or about 2 percent, from 3,188 miles to 3,118 miles. This decline was due, in part, to freeway construction and the consequent return of certain arterial facilities to collector and land-access status and function. In terms of absolute miles, arterial mileage increased between 1972 and 1991 as approximately 70 new miles were constructed and 84 miles took on arterial functions, carrying heavier traffic volumes and providing needed system continuity for the through movement of traffic.

The utilization of the arterial street and highway system has increased. In 1963, about 13 million arterial vehicle-miles of travel were recorded in the Region on an average weekday. Vehiclemiles of travel increased to about 20 million vehicle-miles by 1972 and to about 33 million vehicle-miles by 1991. Arterial VMT increased 64 percent between 1972 and 1991, and 153 percent between 1963 and 1991.

The significant increase in arterial utilization served to increase traffic congestion within the Region. In 1963, about 332 arterial miles, or about 10 percent of the system, was operating at or over design capacity. Arterial mileage operating at or over design capacity decreased between 1963 and 1972, to 318 miles, in response to freeway construction and the added highway capacity freeways provided. Between 1972 and 1991, however, arterial mileage operating at or over design capacity increased by 71 percent, or 225 miles. By 1991, 543 miles, or nearly 17 percent of total arterial mileage, were approaching, or already experiencing, congested traffic conditions.

Table 51 also indicates that significant changes have occurred in the provision and utilization of public transit within the Region. Between 1963 and 1991 the land area within the Region provided with local transit service increased significantly. Local transit operators extended transit routes into newly developing areas, providing additional round trip route miles at a rate of nearly 52 round trip route miles per year. Over the span of 28 years, the number of round trip route miles increased by about 168 percent, from 856 in 1963 to 2,296 in 1991. As a result, about 47 additional square miles were brought into the local transit service areas between 1963 and 1991.

SUMMARY OF CHANGES IN THE PROVISION AND UTILIZATION OF ARTERIAL STREETS AND HIGHWAYS AND PUBLIC TRANSIT WITHIN THE REGION: 1963, 1972, AND 1991

				Change: 19	72-1991	Change: 19	63-1991
Measure of Provision and Utilization	1963	1972	1991	Number	Percent	Number	Percent
Arterial Streets and Highways Total Street and Highway Mileage ^a Arterial Street and Highway Mileage Arterial Street and Highway Mileage Arterial Mileage as a Percent of Total Mileage Arterial Wehicle Miles of Travel Arterial Mileage Operating at or over Design Capacity ^b	8,943.4 3,188.2 35.6 13,072,000 332	9,819.0 3,118.7 31.8 20,124,000 318	11,199.0 3,274.2 29.2 33,072,000 543	1,380 155.5 -2.6 12,948,000 225	14.1 5.0 64.3 70.8	2,255.6 86.0 -6.4 20,000,000 211	25.2 2.7 153.0 63.6
Public Transit ^C Total Local Transit Service Area (square miles) Kenosha Urbanized Area	159.5 10.9 136.9 11.7	177.4 15.9 147.5 14.0	206.3 20.9 158.0 27.4	28.9 5.0 10.5 13.4	16.3 31.4 7.1 95.7	46.8 10.0 21.1 15.7	29.3 91.7 15.4 134.2
Population Served by Local Transit	1,193,300 71,900 1,025,800 95,600	1,204,600 81,500 1,026,000 47,100	1,136,600 86,600 935,400 114,600	-68,000 5,100 -90,600 17,500	-5.6 6.3 -8.8 18.0	-56,700 14,700 -90,400 19,000	-4.8 20.4 -8.8 19.9
Total Round-Trip Route Miles Kenosha Urbanized Area Milwaukee Urbanized Area Racine Urbanized Area	847 55 716 76	1,201 59 1,061 81	2,296 171 1,954 171	1,095 112 893 90	91.2 189.8 84.2 111.1	1,449 116 1,238 95	171.1 210.9 172.9 125.0
Total Transit Vehicle-Miles (average weekday) Kenosha Urbanized Area Milwaukee Urbanized Area Racine Urbanized Area	84,900 2,500 78,900 3,500	64,000 1,100 61,300 1,600	63,300 2,500 56,400 4,400	-700 1,400 -4,900 2,800	-1.1 127.3	-21,600 0 -22,500 900	-25.4 0 -28.5 25.7
Total Seat-Miles (average weekday)	3,412,000 80,800 3,201,300 129,900	3,258,700 43,200 3,186,000 29,500	2,975,000 110,700 2,692,300 172,000	-283,700 67,500 -493,700 142,500	-8.7 156.3 -15.5 483.1	-437,000 29,900 -509,000 42,100	-12.8 37.0 -15.9 32.4
Total Passenger Miles (average weekday) Kenosha Urbanized Area Milwaukee Urbanized Area Racine Urbanized Area	1,013,600 15,100 966,700 31,800	560,700 3,900 551,500 5,300	609,100 16,800 573,200 19,100	48,400 12,900 21,700 13,800	8.6 330.8 3.9 260.4	404,500 1,700 -393,500 -12,700	-39.9 11.3 -68.6 -39.9
Percent Utilization of Public Transit ^d	29.7 18.7 30.2 24.4	17.2 9.0 17.3 18.0	20.5 15.2 21.3 11.1	3.3 6.2 4.0 -6.9		-9.2 -3.5 -8.9 -13.3	••• ••• ••
Annual Revenue Passengers	94,548,000 1,884,400 89,761,600 2,902,000	53,904,300 503,200 52,875,400 525,700	50,222,900 1,128,000 47,267,100 1,827,800	-3,681,400 624,800 -5,608,300 1,302,100	-6.8 124.2 -10.6 247.7	-44,325,100 -756,400 -42,494,500 1,074,100	-46.9 -40.1 -47.3 -37.0

^aIncludes streets and highways functionally classified as arterials, collectors, and local/land-access streets.

^bAt design capacity:volume-to-design capacity ratio of 0.91-1.10; over design capacity: volume-to-design capacity ratio over 1.10.

^CIncludes only fixed-route transit service. Does not include the fixed-route transit service provided in the City of Port Washington, discontinued in 1966.

^dPercent utilization, a measure of how well the supply of fixed-route transit service is used, is calculated by dividing the number of average weekday passenger-miles by the number of average weekday seat-miles.

Source: SEWRPC.

Over all, however, the expansion of the local transit service areas did not result in an increase in the number of people served by local transit. In fact, on a regionwide level, in 1991, 56,700 fewer persons resided within one-quarter mile of a local transit route, the distance that would enable a resident to access the transit system conveniently by walking, than did in 1963. In providing service to smaller, less concentrated populations over greater route mileage, transit providers within the Milwaukee urbanized area combined to operate fewer vehicles less often. Transit vehicle-miles provided within the Milwaukee urbanized area decreased by nearly 29 percent between 1963 and 1991.

The dominance of the Milwaukee urbanized area in the consideration of regionwide transit statistics is particularly evident when considering resident population served by local transit. While the population served by local transit within the Region decreased by 5 percent between 1963 and 1991, the population served within the Kenosha and Racine urbanized areas actually increased by about 20 percent in both areas. Between 1963 and 1991, transit providers within the Kenosha and Racine urbanized areas combined to add an additional 33,700 potential transit riders to the local transit service areas and increased transit vehicle-miles provided by about 15 percent.⁶ During the same period, the local transit service area population within the Milwaukee urbanized area declined by 90,400 persons, indicating the decentralization of urban development and resident population. These declines in the Milwaukee urbanized area were sufficient to offset the gains made in population served within the two smaller urbanized areas.

Public transit ridership has declined significantly within the Region, by nearly 47 percent between 1963 and 1991. Between 1972 and 1991, however, ridership within the Kenosha and Racine urbanized areas increased more than two-fold and three-fold, respectively, but did not attain levels set in 1963. Unlike the smaller urbanized areas, the ridership level in the Milwaukee area in 1991 was significantly below the level set in 1972, by 5.6 million.

⁶Between 1963 and 1991, the increased provision of transit vehicle-miles occurred only within the Racine urbanized area where a 26 percent increase occured. Within the Kenosha urbanized area, the provision of vehicle-miles was the same in both 1963 and 1991. (This page intentionally left blank)

Chapter V

TRAVEL HABITS AND PATTERNS

INTRODUCTION

One of the central concepts underlying the regional transportation study is that travel is an orderly, regular, and measurable occurrence, evidenced by recognizable travel patterns. Periodically, a complete and accurate inventory of existing travel is necessary to identify those patterns and disclose those aspects which demonstrate a high degree of repetitiveness. Such knowledge is a prerequisite to an understanding of probable future travel behavior and, therefore, to intelligent planning of future travel requirements. In this respect, the inventory of travel must provide a clear representation of total travel, while taking stock of, and describing in detail, each of its component parts.

Another fundamental concept underlying the regional transportation study is that land use and transportation are closely interrelated. A complete and accurate inventory of existing travel in the Region is, therefore, also necessary in order to determine the quantitative relationships existing between land use and travel, thereby providing a solid basis for the derivation of future travel demand from existing and proposed land use patterns.

Finally, an accurate inventory of travel is necessary to provide the basis for an understanding of the interactions between travel behavior and land use development. Such an understanding is essential to intelligent planning for future land use requirements.

This chapter describes the necessary travel inventories, the existing travel behavior within the Region, and the significant forces shaping regional travel habits and patterns, comparing the findings of the 1991 regional travel inventory with those of the 1972 and 1963 travel inventories. In addition to the major elements of the regional travel inventory described below, a set of special surveys was conducted as part of the inventory. These special surveys are also described in this chapter.

THE 1991 REGIONAL INVENTORY OF TRAVEL: MAJOR ELEMENTS

Findings of the 1963 and 1972 regional inventories of travel were described in SEWRPC Planning Report No. 7, Volume One, The Land Use-Transportation Study: Inventory Findings: 1963, May 1965, and SEWRPC Planning Report No. 25, A Regional Land Use Plan and a Regional Transportation Plan for Southeastern Wisconsin: 2000, Volume One, Inventory Findings, April 1975, respectively. The 1991 regional inventory of travel,[†] although not as extensive as the 1972 inventory, was more extensive than the 1963 study both in the number and kinds of surveys conducted and also in the scope and detail of information collected. Like the 1963 inventory, the 1991 inventory included comprehensive origin-destination surveys of average weekday travel, including resident household, truck, taxi, and external cordon surveys. In addition, the 1991 inventory, like the 1972 inventory, included surveys of mass transit, intercity motor bus, and intercity railway average weekday travel. Not included in the 1991 inventory were special origin-destination surveys of weekend travel, surveys of mass transit nonusers, and surveys of major traffic generators included in the 1972 inventory.

The 1991 survey of resident households was based on a sample of 17,500 households, or approximately 2.6 percent of the estimated total of 676,100 households in the Region. The sample was selected from electric utility residential customer account address lists. This large-scale sample provides a rich set of data, permitting the description and analysis of resident household travel both by subarea and between subareas of the Region. Information obtained through personal interviews from each sampled household included detailed data concerning: specific household characteristics, including the number of household members, number of vehi-

¹Although the most recent regional travel inventory was conducted from 1991 to 1993, this inventory has been designated the "1991" inventory for purposes of reference and of comparison to the 1963 and 1972 inventories. The special surveys described later in this chapter, however, are referred to by the years in which, respectively, they were conducted.

cles available, structure type of residence, and household income range; specific data for each household member, such as relationship to head of household, age, license-to-drive status, sex, and employment status; and, for each trip made by persons over the age of five on the assigned travel day, the origin and destination of trip, trip purpose, time of day, mode of travel, blocks walked at origin of trip and destination of trip, and, for automobile drivers, the number of passengers in the vehicle, parking location, type of parking, duration of parking, and cost of parking.

In addition, 1,400 samples, representing approximately 3.6 percent of the 38,800 residents of the Region living in group quarters, such as Huber law jail facilities, shelters, nursing homes, motels, hotels, and rooming houses, and schools and other institutions, were surveyed. The sample was drawn from a list of such facilities compiled by the Commission using telephone directories and consultations with various agencies of government. Residents who were groupquartered but who were severely restricted in their ability to travel were not surveyed. This group included residents of mental hospitals, prisons, and invalid homes.

The five major mass transit systems operating in the Region in 1991 were also surveyed. Each of the five systems was sampled at rates designed to permit analysis of the characteristics of existing transit system ridership. For the Kenosha area transit system, 800 samples were obtained, representing a 35 percent rate of return and a 17 percent sample of its estimated 4,800 average weekday riders. For the Milwaukee area transit system, 8,700 samples were obtained, representing a 30 percent rate of return and a 4 percent sample of its estimated 232,700 average weekday riders. For the Racine area transit system, 1,600 samples were obtained, representing a 50 percent rate of return and a 17 percent sample of its estimated 9,200 average weekday riders. For the City of Waukesha transit system, 600 samples were obtained, representing a 40 percent rate of return and a 24 percent sample of its estimated 2,500 average weekday riders. For the Waukesha County transit system, 500 samples were obtained, representing a 68 percent rate of return and a 66 percent sample of its estimated 760 average weekday riders.

Information obtained through mail-back survey forms included detailed data concerning specific household characteristics, including the location of each tripmaker's home, the number of household members, number of vehicles available, and household income range; specific data regarding each tripmaker, such as age, sex, license-to-drive status, and race; and for each trip, the origin and destination of the trip, trip purpose, time of day, transfer information, mode of travel to the bus stop, fares, round-trip frequency, and length of time using transit.

The 1991 regional travel inventory also included a truck survey. The truck survey was intended to provide information regarding the movement of freight and the delivery of services by commercial trucks. Personal travel made using trucks was collected and reported in the resident household and group-quarters surveys. The survey of truck travel was based on a sample of 2,500 commercial trucks, or approximately 3 percent of the estimated 87,500 commercial trucks² registered in the Region. Information obtained through a mail survey for each sampled truck included detailed data concerning: the business or industry of the truck owner; the truck garaging location, carrier type, odometer reading at the beginning and end of the travel day, and vehicle type; and for each trip made using the truck on the assigned travel survey day, the origin and destination of the trip, trip purpose, and time of day.

The taxi survey conducted as a part of the 1991 travel inventory included travel by limousines and specialized carriers providing service to elderly and disabled tripmakers, as well as travel by taxicabs. About 400 samples were obtained, representing about 33 percent of the approximately 1,200 private personalized transportation vehicles estimated to be operated in the Region. Information obtained through the mail-back survey from each sampled vehicle included detailed data concerning: the business and industry of the vehicle owner; the vehicle garaging location, carrier type, odometer reading at the beginning and end of the travel day,

 $^{^{2}}$ For purposes of the truck survey, the number of trucks available within the Region for commercial use was determined to be 87,500. The truck survey findings indicated that there were, in addition, 80,600 trucks available in the Region used for personal travel.

and vehicle type; and for each trip made on the assigned travel survey day, the origin and destination of the trip, trip purpose, and time of day.

In the external cordon survey, roadside interview stations were established on 44 major highways crossing the boundaries of the Region. At these stations, mail-back survey forms were distributed to 174,700 motorists between the hours of 6:00 a.m. and 6:00 p.m. in May 1992 and early June 1992. Approximately 47,100 usable survey forms were returned, representing more than 16 percent of the 288,700 regional boundary crossings by vehicles estimated to occur at the interview stations during an average weekday. Information obtained through the mail-back survey included: the vehicle used in making the trip, the garaging address of the vehicle, type of vehicle, and number of passengers carried; and, for trucks, the carrier type. For trips crossing the cordon line, data regarding the origin, destination, and purpose of each trip were also obtained.

As a part of the new regional inventory of travel, special origin-destination surveys were conducted in the spring of 1993 to provide information on persons entering, exiting, or traveling through the Region on motor buses and railway trains. In addition, a survey of enplaning passengers at General Mitchell International Airport was conducted in the fall of 1989. The survey of intercity motor bus travel was based on a sample of 1,000 riders or approximately 71 percent of the estimated 1,400 average weekday intercity motor bus riders. The survey of intercity railway passengers was based on a sample of 1,300 riders, or approximately 76 percent of the estimated 1,700 average weekday intercity railway passengers. Information obtained through these mail-back surveys included: specific household characteristics, such as location of residence, number of vehicles available in the household, and household income range; specific data regarding the tripmaker, such as occupation, age, and level of education; and data related to the trip, such as the origin, destination, and purpose of the trip; and, if part of a round trip, boarding location and frequency of the trip made. Surveyed riders were also asked to identify reasons for choosing their mode of travel and to state which alternative mode they would choose if the current service was not available.

The expanded data obtained in these surveys provided a complete representation of the total travel occurring within the Region on an average weekday in 1991. In each survey, careful attention was given to data collection scheduling to prevent any day-related bias in the information.

Accuracy Checks

Two distinct sets of accuracy checks were employed to determine the degree of accuracy and completeness of data obtained in the major travel surveys. In one set, data on socioeconomic characteristics obtained from the major surveys were compared with data from the 1990 Federal Census and other independent sources. In the other set of accuracy checks, vehicle trip volumes derived from travel surveys were compared to vehicle trip volumes obtained by classification counts made at screenlines, cut lines, and cordon lines.³ The level of vehiclemiles of travel derived from travel surveys was also compared to actual vehicle-miles of travel estimated from traffic counts.

Socio-Economic Accuracy Checks: Application of the socio-economic accuracy checks to the resident household survey indicated that the percentage distribution of households by household size as established by the survey was essentially the same as that identified by the 1990 Federal Census at a county as well as regional level. Table 52 sets forth a comparison of the distribution of households by household size within each county as measured by the 1990 Federal Census and as derived from the resident household

 $^{^{3}}A$ cut line is defined as an imaginary line extending through a selected portion of a geographic area for the purpose of comparing and analyzing data as estimated from traffic counts with data derived from travel surveys. A screenline is an imaginary line extending through a selected portion of a geographic area along natural or built barriers providing a limited number of crossing points established for the purpose of comparing and analyzing travel data as estimated from traffic counts with data derived from travel surveys. A cordon line is an imaginary line extending around a selected geographic area for the purpose of comparing and analyzing external travel data as estimated from traffic counts with such data as derived from travel surveys.

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF THE DISTRIBUTION OF HOUSEHOLDS BY HOUSEHOLD SIZE AND COUNTY FROM 1990 FEDERAL CENSUS AND 1991 RESIDENT HOUSEHOLD SURVEY

					Household Siz	0			
						.0		· · · · · · · · · · · · · · · · · · ·	1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -
		One Person			Two Persons			Three Persons	
	Number of	Number of Households			Households		Number of	Households	
County	1990 Federal Census	Resident Household Survey	Percent Difference ^a	1990 Federal Census	Resident Household Survey	Percent Difference ^a	1990 Federal Census	Resident Household Survey	Percent Difference ^a
Kenosha Milwaukee	10,923 109,528	10,993 109.569	0.6 0.0	14,750 116,694	14,788 116,863	0.3 0.1	8,436 60,421	8,424 60,265	-0.1 -0.3
Ozaukee	4,374	4,361	-0.3	8,627	8,651	0.3	4,877	4,854	-0.5
Walworth	6,609	6,624	0.0	9,256	9,254	0.0	4,651	4,661	0.2
Washington Waukesha	5,657 17,619	5,643 17,497	-0.2 -0.7	10,405 34,360	10,418 34,598	0.1 0.7	6,037 20,084	6,040 20,009	0.0 -0.4
Region	168,759	168,730	0.0	214,108	214,499	0.2	116,069	115,784	-0.2

			Househ	old Size			Total Number of	
		Four Persons		F	ive or More Pers	Households		
	Number of	Households		Number of	Households			· · · ·
County	1990 Federal Census	Resident Household Survey	Percent Difference ^a	1990 Federal Census	Resident Household Survey	Percent Difference ^a	1990 Federal Census	Resident Household Survey
Kenosha	7,638	7,565	-1.0	5,293	5,271	-0.4	47,040	47,040
Milwaukee	49,004	48,960	-0.1	37,401	37,390	0.0	373,048	373,048
Ozaukee	4,826	4,849	0.5	3,003	2,992	-0.4	25,707	25,707
Racine	10,852	10,944	0.8	7,256	7,291	0.5	63,736	63,736
Walworth	4,234	4,206	-0.7	2,875	2,880	0.2	27,625	27,625
Washington	6,575	6,596	0.3	4,333	4,310	-0.5	33,007	33,007
Waukesha	21,200	21,169	-0.1	12,727	12,717	-0.1	105,990	105,990
Region	104,329	104,288	0.0	72,888	72,851	-0.1	676,153	676,153

⁹ The percent difference is derived by dividing the result of subtracting the number of households reported in the 1990 Federal Census from the number of households reported in the resident household survey by the number of households in the 1990 Federal Census.

Source: U. S. Bureau of the Census and SEWRPC.

survey. As Table 52 shows, the county-level survey data on household size are within 0.3 percent of corresponding 1990 Federal Census data in more than one-half of the applicable cases. The largest difference between countylevel survey data and corresponding 1990 Federal Census data is in the case of four-person households in Kenosha County, in which case the difference is 1.0 percent. At the regional level, all differences between Census and survey data were 0.2 percent or less.

Table 53 sets forth comparisons of data on vehicle availability by county for 1991 as obtained, respectively, from 1990 Federal Census data and from the 1991 resident household survey. The comparison indicated that vehicle availability by county was accurately estimated by the survey, varying by 2.3 percent to 6.4 percent from Federal Census estimates. The total number of vehicles available in the Region, based on the 1991 residential household survey, was estimated to be about 1,142,500 vehicles, about 2.4 percent more than the 1,116,000 vehicles estimated to be available in 1991 using 1990 Federal Census data.

The distribution of annual household income as estimated by the travel survey was also compared with similar data estimated from the 1990 Census. Table 54 sets forth this comparison. At a regional level, the percentage of households in any given income range, based on the 1991 resident household survey, did not differ from the corresponding percentage of households based on 1990 Federal Census data by more than 3 percent.

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF THE NUMBER OF VEHICLES AVAILABLE BY COUNTY FROM THE 1990 FEDERAL CENSUS AND 1991 RESIDENT HOUSEHOLD SURVEY DATA

	1991 Estima	ate Based	1991 Res	1991 Resident Household Survey				
	on 1990 Fede	eral Census			Difference			
County	Number of Vehicles	Percent of Total	Number of Vehicles	Percent of Total	in Number of Vehicles	Percent Difference		
Kenosha	83,700	7.5	88,800	7.8	5,100	6.1		
Milwaukee	516,100	46.2	540,600	47.3	24,500	4.7		
Ozaukee	53,300	4.8	52,100	4.6	-1,200	-2.3		
Racine	113,400	10.2	120,700	10.6	7,300	6.4		
Walworth	51,900	4.7	54,000	4.7	2,100	4.0		
Washington	70,600	6.3	68,700	6.0	-1,900	-2.7		
Waukesha	227,000	20.3	217,600	19.0	-9,400	-4.1		
Region	1,116,000	100.0	1,142,500	100.0	26,500	2.4		

Source: U. S. Bureau of the Census and SEWRPC.

Table 54

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF THE DISTRIBUTION OF THE PERCENTAGE OF HOUSEHOLDS BY INCOME AND BY COUNTY FROM THE 1990 FEDERAL CENSUS AND 1991 RESIDENT HOUSEHOLD SURVEY DATA

	Kenosha Milwaukee			Ozaukee				Racine				
Household Income	1990 Federal Census	Resident Household Survey	Difference in Percent									
Under \$5,000	3.3	1.5	-1.8	4.8	2.5	-2.3	1.3	1.1	-0.2	3.3	0.9	-2.4
\$5,000 to 9,999	8.7	4.3	-4.4	11.9	8.2	-3.7	4.1	2.5	-1.6	8.7	5.0	-3.7
10,000 to 14,999	8.3	8.5	0.2	9.6	9.9	0.3	5.1	4.3	-0.8	8.2	6.9	-1.3
15,000 to 19,999	9.3	9.5	0.2	9.3	10.6	1.3	6.1	4.8	-1.3	8.4	8.5	0.1
20,000 to 24,999	9.6	11.4	1.8	9.1	12.0	2.9	6.2	7.7	1.5	8.3	10.1	1.8
25,000 to 29,999	9.5	9.9	0.4	8.5	10.4	1.9	7.2	7.8	0.6	7.8	11.0	3.2
30,000 to 34,999	8.8	9.9	1.1	8.0	10.2	2.2	7.3	7.1	-0.2	8.5	11.2	2.7
35,000 to 39,999	7.4	9.1	1.7	7.3	8.6	1.3	8.0	8.3	0.3	8.0	9.4	1.4
40,000 to 49,999	13.2	14.3	1.1	11.9	11.4	-0.5	14.4	16.2	1.8	13.9	13.7	-0.2
50,000 to 74,999	16.0	16.9	0.9	13.9	11.5	-2.4	22.3	23.5	1.2	17.4	16.2	-1.2
75,000 to 99,999	4.0	2.9	-1.1	3.3	2.7	-0.6	8.5	7.9	-0.6	4.6	4.3	-0.3
100,000 or Over	1.9	1.8	-0.1	2.4	2.0	-0.4	9.5	8.8	-0.7	2.9	2.8	-0.1
Total	100.0	100.0	0.0	100.0	100.0	0.0	100.0	100.0	0.0	100.0	100.0	0.0

	Walworth				Washingtor	า		Waukesha			The Region	l.
Household Income	1990 Federal Census	Resident Household Survey	Difference in Percent	1990 Federal Census	Resident Household Survey	Difference in Percent	1990 Federal Census	Resident Household Survey	Difference in Percent	1990 Federai Census	Resident Household Survey	Difference in Percent
Under \$5,000	3.6	2.2	-1.4	1.3	0.9	-0.4	1.4	1.0	-0.4	3.7	1.9	-1.8
\$5,000 to 9,999	8.5	10.1	1.6	5.0	2.7	-2.3	3.9	2.6	-1.3	9.3	6.4	-2.9
10,000 to 14,999	8.9	10.6	1.7	5.9	3.8	-2.1	4.5	3.9	-0.6	8.2	8.1	-0.1
15,000 to 19,999	10.3	8.8	-1.5	6.5	6.8	0.3	5.3	5.0	-0.3	8.4	9.0	0.6
20,000 to 24,999	9.5	7.7	-1.8	7.8	9.2	1.4	6.0	5.8	-0.2	8.4	10.4	2.0
25,000 to 29,999	8.6	7.6	-1.0	8.1	9.9	1.8	6.2	6.5	0.3	8.1	9.6	1.5
30,000 to 34,999	8.7	11.0	2.3	8.9	10.7	1.8	7.3	9.4	2.1	8.0	10.0	2.0
35,000 to 39,999	7.5	9.2	1.7	9.1	9.7	0.6	7.6	8.5	0.9	7.5	8.8	1.3
40,000 to 49,999	13.4	12.5	-0.9	17.8	18.2	0.4	16.0	16.0	0.0	13.2	13.1	-0.1
50,000 to 74,999	14.4	14.0	-0.4	21.1	19.0	-2.1	26.0	27.3	1.3	16.9	15.6	-1.3
75,000 to 99,999	4.0	3.9	-0.1	5.3	5.1	-0.2	9.0	8.8	-0.2	4.7	4.1	-0.6
100,000 or Over	2.5	2.5	0.0	3.1	3.9	0.8	6.8	5.1	-1.7	3.4	2.9	-0.5
Total	100.0	100.0	0.0	100.0	100.0	0.0	100.0	100.0	0.0	100.0	100.0	0.0

Source: U. S. Bureau of Census and SEWRPC.

	1992 Truck R	egistrations	1991 Truc Motor Truck	k Survey s Available	Difference		
Truck Classification	Number	Percent of Total	Number	Percent of Total	in Number of Vehicles	Percent Difference	
Light	135,600 30,700 3,400 7,200	76.6 17.4 1.9 4 1	129,700 28,400 3,100 6,900	77.2 16.9 1.8 4.1	-5,900 -2,300 -300 -300	-4.4 -7.5 -8.8 -4.2	
Total	176,900	100.0	168,100	100.0	-8,800	-5.0	

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF SOUTHEASTERN WISCONSIN TRUCK REGISTRATIONS AND TRUCK AVAILABILITY AS DETERMINED BY THE 1991 TRUCK SURVEY

Source: Wisconsin Department of Transportation and SEWRPC.

Estimates of truck availability, including estimates by type, as determined by the 1991 travel survey, were compared with corresponding estimates as derived from 1992 Wisconsin Department of Transportation, Division of Motor Vehicles, registration records. This comparison indicates a high degree of accuracy for the truck data derived from the survey, both within the Region and within its counties. Table 55 sets forth this comparison with regard to the Region. As Table 55 shows, the 1991 survey findings indicated that survey estimates of motor truck availability represented about 95 percent of the total trucks registered. Previous surveys conducted by the Commission indicated that motor truck availability estimates approximated 94 percent of all trucks registered within the Region. The remaining 6 percent was not available because those trucks were removed from the State or were in salvage yards, used vehicle lots, or similar storage.

The results of the accuracy checks on the resident household survey socio-economic characteristic data indicate that the survey data demonstrate a high degree of accuracy and completeness. Upon completion of the accuracy checks on the expanded housing and socio-economic characteristic data as derived from the resident household and other travel surveys, the second set of checks was performed to ascertain the degree of accuracy of, and to determine the need for adjustment to the trip information obtained in, these surveys. For this analysis, expanded trip data from the resident household, external truck, and taxi travel surveys were combined. <u>Travel Accuracy Checks</u>: To verify that travel into and out of the Region was adequately represented by the 1991 travel inventory, travel accuracy checks were first conducted at the external cordon of the Region. The findings shown in Table 56 indicated that the travel survey data did indeed accurately represent external travel and could be used without adjustment of external travel data as reported in the external cordon and resident household and truck surveys.

The next travel accuracy checks performed included the comparisons of the combined travel survey data with traffic counts of vehicle crossings at selected east-west screenlines and cut lines and estimates of actual vehicle-miles of travel. The principal check compared traffic count and classification data from all vehicle crossings of screenlines between 6 a.m. and 8 p.m. with vehicle trip volumes determined from the travel survey data to be crossing such screenlines. The three screenlines, which are shown on Map 35 and which paralleled natural or built barriers to minimize undetected crossings, bisected the Milwaukee, Racine, and Kenosha urbanized areas. In addition, the Milwaukee screenline, which roughly paralleled IH 94, extended across the Region from the Waukesha-Jefferson County line on the west to Lake Michigan on the east. The results of this comparison, shown in Table 57, indicate that the combined travel surveys underrepresented travel by about 14 percent on the Milwaukee screenline and by almost 20 percent on the Kenosha and Racine screenlines.

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF ESTIMATED SOUTHEASTERN WISCONSIN CORDON LINE CROSSING: ESTIMATED AVERAGE WEEKDAY TRAFFIC VOLUMES FROM TRAFFIC COUNTS AND FROM TRAVEL SURVEYS

External Cordon Line Segment	Estimated Average Weekday Traffic Volumes from Traffic Counts	Estimated Average Weekday Traffic Volumes from Travel Survey Data	Difference in Volumes	Percent Difference
Northern Region Boundary Western Region Boundary Southern Region Boundary	39,500 130,500 129,300	33,600 134,100 131,500	-5,900 3,600 2,200	-14.9 2.8 1.7
Total	299,300	299,200	-100	0.0

Source: SEWRPC.

Table 57

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF ESTIMATES OF AVERAGE WEEKDAY VEHICULAR TRAFFIC CROSSING THE KENOSHA, MILWAUKEE, AND RACINE SCREENLINES USING UNADJUSTED TRAVEL SURVEY DATA FROM TRAFFIC COUNTS AND TRAVEL SURVEY DATA

Screenline	Estimated Average Weekday Traffic Volumes from Traffic Counts	Estimated Average Weekday Traffic Volumes from Travel Survey Data	Difference in Volumes	Percent Difference
Kenosha	211,400	169,600	-41,800	-19.8
Milwaukee	819,200	701,200	-118,000	-14.4
Racine	195,700	158,700	-37,000	-18.9

Source: SEWRPC.

The other major travel accuracy check compared estimated vehicle-miles of travel, as derived from survey trip data, assigned to the arterial street and highway network to actual traffic estimates derived from traffic volume counts for the Kenosha, Milwaukee, and Racine urbanized areas; for the remainder of the Region, which consists of rural and small urban areas; and for the Region as a whole.

As shown in Table 58, estimates of vehicle-miles of travel derived from the assignment of the survey trip data were about 6 percent lower for the Region and about 7 percent lower for the Milwaukee urbanized area than the corresponding estimates of vehicle-miles of travel derived from traffic volume counts.

This underrepresentation of travel is similar to that observed in the 1963 and 1972 surveys and in travel origin and destination surveys conducted nationwide. The underrepresentation of travel occurs in the off-peak periods with respect to nonwork and nonschool travel, including trips made for shopping, personal business, social, and recreational purposes. The adjustment factors applied to the 1991 survey were similar to the factors applied to the 1963 and 1972 travel inventories in each of the urbanized areas to nonwork travel. Application of the adjustment factor to the 1991 travel data increased total internal vehicle trips by approximately 14 percent.

Following adjustment, survey data on estimated average weekday vehicular traffic crossing screenlines were compared with corresponding data derived from traffic counts. The results of the screenline accuracy checks on the adjusted travel data, as shown in Table 59, indicated that the simulated traffic volumes from the adjusted survey data represented 95 percent of the traffic volumes crossing the screenlines in Kenosha, 101 percent in Milwaukee, and 96 percent in Racine. As Table 60 shows, a comparison of survey-estimated average weekday traffic volumes crossing cut lines with comparable estimates derived from traffic counts indicated



Household characteristic data, vehicle information, and travel data were obtained by personal interviews with residents of 17,500 households and by mail-back survey forms from 12,200 mass transit users, 2,900 commercial-use truck and taxi operators, and 47,100 drivers of vehicles crossing regional boundaries. The above map shows the boundaries of the three urbanized areas selected for intensive survey data analysis, the location of the 44 roadside interview stations in the external cordon survey, and the screenlines and cut lines selected to check accuracy.

Source: SEWRPC.

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF ESTIMATE VEHICLE-MILES OF TRAVEL ON AN AVERAGE WEEKDAY IN SELECTED AREAS OF THE REGION USING UNADJUSTED TRAVEL SURVEY DATA

Area	Estimated Average Weekday Vehicle-Miles of Travel from Traffic Counts	Estimated Average Weekday Vehicle-Miles of Travel from Travel Survey Data	Difference in Vehicle Miles	Percent Difference
Kenosha Urbanized Area	1,009,400	1,024,600	15,200	1.5
Milwaukee Urbanized Area	20,334,400	18,952,900	-1,381,500	-6.8
Racine Urbanized Area	1,171,100	1,114,400	-56,700	-4.8
Area of Region outside				
Urbanized Area	9,687,200	9,178,100	-509,100	-5.3
Region	32,202,100	30,270,000	-1,932,100	-6.0

Source: SEWRPC.

Table 59

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF ESTIMATED AVERAGE WEEKDAY VEHICULAR TRAFFIC CROSSING THE KENOSHA, MILWAUKEE, AND RACINE SCREENLINES

Screenline	Estimated Average Weekday Traffic Volumes from Traffic Counts	Estimated Average Weekday Traffic Volumes from Adjusted Travel Survey Data	Difference in Volumes	Percent Difference
Kenosha	211,400	201,700	-9,700	-4.6
	819,200	826,500	7,300	0.9
	195,700	187,300	-8,400	-4.3

Source: SEWRPC.

Table 60

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF ESTIMATED AVERAGE WEEKDAY TRAFFIC CROSSING THE KENOSHA, MILWAUKEE, AND RACINE CUT LINES

Cut Line	Map 35 Reference	Estimated Average Weekday Traffic Volumes from Traffic Counts	Estimated Average Weekday Traffic Volumes from Adjusted Travel Survey Data	Difference in Volumes	Percent Difference	
Kenosha				1		
North	K-N	57,100	56,200	-900	-1.6°	
South	K-S	90,200	94,700	4,500	5.0	
West	K-W	67,500	71,100	3,600	5.3	
Milwaukee						
North	M-N	411,100	409,100	-2,000	-0.5	
South	M-S	351,000	345,200	-5,800	-1.7	
West	M-W	493,100	513,400	20,300	4.1	
Racine						
North	R-N	53,900	61,400	7,500	13.9	
South	R-S	114,000	113,500	-500	-0.4	
West	West R-W 89,500		98,400	8,900	9.9	
Rural	West	215,100	212,500	-2,600	-1.2	

Source: SEWRPC.

Area	Estimated Average Weekday Vehicle-Miles of Travel from Traffic Counts	Estimated Average Weekday Vehicle-Miles of Travel from Adjusted Travel Survey Data	Difference in Vehicle Miles	Percent Difference
Kenosha Urbanized Area	1,009,400	1,208,900	199,500	19.8
Milwaukee Urbanized Area	20,334,400	20,374,400	40,000	0.2
Racine Urbanized Area	1,171,100	1,225,200	54,100	4.6
Area of Region outside				
Urbanized Area	9,687,200	10,733,300	1,046,100	10.8
Region	32,202,100	33,541,800	1,339,700	4.2

THE 1991 REGIONAL TRAVEL INVENTORY: COMPARISON OF ESTIMATED VEHICLE-MILES OF TRAVEL ON AN AVERAGE WEEKDAY IN SELECTED AREAS OF THE REGION

Source: SEWRPC.

that simulated traffic volumes in Kenosha represented between 98 percent and 105 percent of counted volumes; in Milwaukee, between 98 percent and 104 percent; in Racine, between 100 percent and 114 percent; and in the rural area, 99 percent. A check on vehicle-miles of travel, as shown in Table 61, derived from the adjusted travel data indicated that vehicular travel simulated from survey data in the Kenosha urbanized area represented 120 percent of the corresponding estimate derived from traffic counts: in the Milwaukee urbanized area. 100 percent; in the Racine urbanized area, 105 percent; and in the rural and small urban areas, 111 percent. For the entire Region, simulated vehicle-miles of travel from the adjusted travel survey represented 104 percent of total vehicle-miles of travel estimated from traffic counts.

INVENTORY FINDINGS

Quantity of Total Travel

An estimated 5.91 million person trips⁴ were made within the Region on an average weekday in 1991, as shown on Table 62. This represents an increase of about 2.12 million person trips per weekday, or an increase of about 56 percent, since 1963 and an increase of 1.23 million person trips per weekday, or 26 percent, since 1972. Of these 5.91 million person trips, about 5.59 million, or 95 percent, were internal person trips. Internal trips have both trip origin and trip destination within the Region.

The 5.59 million internal person trips represent an increase of 2.0 million trips, or 55 percent, since 1963 and an increase of 1.1 million trips, or 24 percent, since 1972. The average number of internal person trips per capita on an average weekday increased from 2.2 in 1963 to 2.5 in 1972 and to 3.1 in 1991, while the average number of internal person trips per household on an average weekday increased from 7.3 in 1963 to 7.9 in 1972 and then decreased slightly to 7.8 in 1991.

In 1991, an estimated 4.89 million vehicle trips, consisting of automobile and truck trips, were made within the Region on an average weekday. This represents an increase of 2.33 million vehicle trips, or 91 percent, since 1963 and an increase of 1.48 million vehicle trips, or 43 percent, since 1972. Of the 4.89 million vehicle trips, about 4.62 million, or 94 percent, were internal vehicle trips. Internal vehicle trips thus increased by 2.15 million, or 87 percent, since 1963 and by 1.33 million, or 40 percent, since 1972. Vehicle trips made within the Region have increased faster than person trips, particularly since 1972, as a result of a decline in both

⁴A person trip is defined as a one-way journey between a point of origin and a point of destination by a person five years of age or older traveling as an automobile driver or as a passenger in an automobile, taxi, truck, motorcycle, school bus, or other mass transit carrier. To be considered, the trip must have been at least the equivalent of one full city block in length. A person trip also includes trips made by bicycle or walking when the trip purpose is work.

		Person Trips												
	1963		1972		1991		Change: 1963-1991		Change: 1972-1991					
Survey Type	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent	Number	Percent				
Home Interview	3,603,000	94.9	4,504,900	96.2	5,594,300	94.6	1,991,300	55.3	1,089,400	24.2				
External Cordon	191,700	5.1	176,900	3.8	317,400	5.4	125,700	65.6	140,500	79.4				
Total	3,794,700	100.0	4,681,800	100.0	5,911,700	100.0	2,117,000	55.8	1,229,900	26.3				

AVERAGE WEEKDAY PERSON AND VEHICLE TRIPS BY SURVEY TYPE: 1963, 1972, AND 1991

		Vehicle Trips											
	1963		1972		1991		Change: 1963-1991		Change: 1972-1991				
Survey Type	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent	Number	Percent			
Home Interview/ Truck and Taxi	2,466,400	96.0	3,290,300	96.3	4,620,300	94.4	2,153,900	87.3	1,330,000	40.4			
External Cordon	101,600	4.0	125,700	3.7	273,300	5.6	171,700	169.0	147,600	117.4			
Total	2,568,000	100.0	3,416,000	100.0	4,893,600	100.0	2,325,600	90.6	1,477,600	43.3			

Source: SEWRPC.

automobile occupancy and transit use. The percentage increase between 1963 and 1991 for vehicle trips was 91 percent, as compared with 56 percent for person trips. The percentage increase in vehicle trips between 1972 and 1991 was 43 percent, compared with 26 percent for person trips.

Table 62 also shows that an estimated 317,400 external person trips and 273,300 external vehicle trips were made in 1991. External trips have one end or both ends located outside of the Region. From 1963 to 1991, external travel increased by 125,700 person trips, or by about 66 percent; and by 171,700 vehicle trips, or by about 169 percent.

The increase in internal person trips is on par with increases in the number of households and jobs within the Region. As shown in Table 63 and Figure 33, between 1963 and 1991, internal person tripmaking increased by 55 percent, households increased by 46 percent, and employment increased by 56 percent. The increase in person trips, however, far exceeded the 9 percent increase in population during the same time period. These findings confirm that increases in households and employment were the major socio-economic factors behind increased tripmaking within the Region and that population increases had a negligible impact between 1963 and 1991. The significant increase in employment despite the small increase in population is indicative of the increased participation of women in the labor force. This increase in employment is also largely the cause of increased tripmaking in the Region.

Internal Person Travel

Internal Person Trip Production: A comparison of information shown on Maps 36, 37, and 38 indicates that average person trip production per household varied widely by subarea of the Region in 1963, 1972, and 1991. The lowest rates of household person trip production in each of those three years were in the central areas of the larger cities and in certain portions of the rural areas, where the average number of trips per household made on an average weekday was usually less than four. The highest rates were found in the suburban and rural urban-fringe areas, where the number of trips per household made on an average weekday was 12 or more. Average trip production per household between 1972 and 1991 increased particularly in many of the outlying areas of the Region as urban development occurred in these areas.

Internal Person Trips	1963	1972	1991	Percent Change 1963-1991	Percent Change 1972-1991
Internal Person Trips	3,603,300	4,504,900	5,594,300	55.3	24.2
Households ^a	491,400	567,700	714,900	45.5	
Employment (jobs)	630,900	748,800	981,400	55 <i>.</i> 6	31.1
	1,654,000	1,810,700	1,798,300	8.7	-0.7

COMPARISON OF HISTORIC REGIONAL INTERNAL PERSON TRIPS, HOUSEHOLDS, EMPLOYMENT, AND POPULATION: 1963, 1972, AND 1991

^aIncludes group-quartered population households: approximately 10,200 in 1963, 10,400 in 1972, and 38,800 in 1991.

Source: SEWRPC.

Figure 33



Source: SEWRPC.

Relationship of Vehicle Availability: A strong correlation exists between person trip production and the number of vehicles available to households. The 1991 survey findings indicated that about 1,142,500 vehicles were available in the Region. This represents an average of 1.60 vehicles per household, or about 620 vehicles per 1,000 residents. In 1963, 527,300 vehicles were available, an average of 1.07 vehicles per household and 319 vehicles per 1,000 residents. In 1972, 704,600 vehicles were available, an average of 1.24 vehicles per household and 389 vehicles per 1,000 residents (see Maps 39, 40, and 41). A comparison of Maps 40 and 41 shows that between 1972 and 1991 the number of vehicles available to households in rural and suburban areas increased significantly. By 1991, the vast majority of households in these areas each had two or more automobiles available.

Table 64 shows the relationship of vehicle availability to person trip production in the Region. Household person trip production increases sharply in relation to increased vehicle availability. It may also be noted that from 1963 to 1991, the percentage of households with two or more automobiles increased from 24 percent to 53 percent of all households, the percentage of households with zero automobiles declined from 18 percent to 13 percent of all households, and the percentage with one automobile declined from 58 percent to 34 percent of all households. This increase in household automobile availability contributed to the increase in person trips generated within the Region since 1963.

Relationship of Household Size: Person trip production within the Region is also related to the number of persons comprising the household. Table 65 indicates that in 1991 one-person households averaged about three weekday internal person trips per household, two-person households averaged about seven such trips per household, three-person households averaged about nine such trips per household, four-person households averaged about 12 such trips per household, and five-or-more-person households averaged about 15 such trips per household. The average number of such trips per household increased in each successively higher household size category in 1963, 1972, and 1991. The average trip generation rate for total households remained constant at about eight trips per household in both 1972 and 1991, however, the distribution of the number of households by household size changed markedly from 1972 to 1991. One-person households more than doubled in number, increasing from 98,700 in 1972 to

AVERAGE WEEKDAY INTERNAL PERSON TRIPS PER HOUSEHOLD IN THE REGION BY AUTOMOBILE AVAILABILITY: 1963, 1972, AND 1991

1		1	963			1972				1991				Person Trips		
]	Households		Person Trips		Hous	Households		Person Trips		Households		n Trips	per Household			
Autos Available	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	1963	1972	1991	
None	90,700	18.4	193,900	5.4	95,600	16.8	182,500	4.1	90,700	12.7	192,700	3.4	2.1	1.9	2.1	
One	284,600	57.9	2,107,800	58.5	279,200	49.2	1,964,900	43.6	243,500	34.1	1,328,600	23.8	7.4	7.0	5.5	
Two	103,000	21.0	1,123,900	31.2	161,300	28.4	1,851,000	41.1	281,200	39.3	2,813,800	50.3	10.9	11.5	10.0	
Three or																
More	13,100	2.7	177,400	4.9	31,600	5.6	506,500	11.2	99,500	13.9	1,259,200	22.5	13.5	16.0	12.7	
Total	491,400	100.0	3,603,000	100.0	567,700	100.0	4,504,900	100.0	714,900	100.0	5,594,300	100.0	7.3	7.9	7.8	

Source: SEWRPC.

Table 65

AVERAGE WEEKDAY INTERNAL PERSON TRIPS PER HOUSEHOLD IN THE REGION BY HOUSEHOLD SIZE: 1963, 1972, AND 1991

		1	963		1972				1991				Person Trips			
	Households		Person Trips		Households		Person Trips		Hous	Households		Person Trips		per Household		
Household Size	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	1963	1972	19 9 1	
One	60,000	12.2	114,600	3.2	98,700	17.4	234,800	5.2	206,500	28.9	622,100	11.1	1.9	2.4	3.0	
Two	136,300	27.7	685,000	19.0	164,200	28.9	903,200	20.0	215,000	30.1	1,539,900	27.5	5.0	5.5	7.2	
Three	87,800	17.9	668,400	18.6	92,400	16.3	762,100	16.9	116,100	16.2	1,081,400	19.3	7.6	8.2	9.3	
Four	84,000	17.1	807,100	22.4	86,600	15.3	904,700	20.1	104,400	14.6	1,290,100	23.1	9.6	10.4	12.4	
Five or																
More	123,300	25.1	1,327,900	36.8	125,800	22.2	1,700,100	37.7	72,900	10.2	1,060,800	19.0	10.8	13.5	14.6	
Total	491,400	100.0	3,603,000	100.0	567,700	100.0	4,504,900	100.0	714,900	100.0	5,594,300	100.0	7.3	7.9	7.8	

Source: SEWRPC.

206,500 in 1991, while five-or-more-person households decreased from 125,800 to 72,900 during the same period. It may also be noted that the average number of trips per household in each household size category increased between 1963 and 1972 and between 1972 and 1991.

Relationship of Household Annual Income: A third factor strongly correlated with person trip production is annual household income. Table 66 shows the 1991 and 1972 inventory findings, based upon data collected for the entire sevencounty Region, together with the 1963 findings, based upon data collected only in the three urbanized areas. Table 66 indicates that the average number of weekday internal person trips per household in each of the survey years generally increased in direct proportion to annual household income. Mode of Internal Person Trips: Freedom to select a particular mode of travel is not always present. Many households, for example, are located in areas not served by mass transit, and thus are highly dependent on the use of an automobile. Other households do not have an automobile available, and thus are dependent almost entirely on mass transit. The survey findings indicate that auto-oriented travel on an average weekday accounted for about 88 percent of internal person trips in 1963 and about 92 percent of internal person trips in both 1972 and 1991. The 1991 data indicate that autooriented travel is increasingly becoming driveronly travel. In 1963 and 1972, auto passenger travel accounted for 27 percent of total internal person trips, but in 1991 accounted for only 19 percent of such trips. As Table 67 shows, driver-only trips increased from 2.9 million, or

AVERAGE WEEKDAY INTERNAL PERSON TRIPS PER HOUSEHOLD IN THE REGION: 1963



Source: SEWRPC.

AVERAGE WEEKDAY INTERNAL PERSON TRIPS PER HOUSEHOLD IN THE REGION: 1972



Source: SEWRPC.


AVERAGE WEEKDAY INTERNAL PERSON TRIPS PER HOUSEHOLD IN THE REGION: 1991

Source: SEWRPC.

Residents of Southeastern Wisconsin made about 5.91 million person trips within the Region on an average weekday in 1991, an increase of 2.12 million trips, or 56 percent, over the 1963 level; and of 1.23 million trips, or 26 percent, over the 1972 level. The average number of internal person trips per household increased from 7.3 in 1963 to 7.9 in 1972 and decreased only slightly to 7.8 by 1991. As shown on Maps 36, 37, and 38, areas with above-average tripmaking rates have increasingly comprised a larger share of the Region over time. The significant expansion of high tripmaking areas is most evident in western Waukesha County and in much of Washington County. Areas with tripmaking rates lower than the average are shown in 1963, 1972, and 1991 to be concentrated in the central parts of major cities and in scattered rural locations.

VEHICLES AVAILABLE PER HOUSEHOLD IN THE REGION:1972



Map 39

VEHICLES AVAILABLE PER HOUSEHOLD IN THE REGION: 1963



Source: SEWRPC.

1.81

VEHICLES AVAILABLE PER HOUSEHOLD IN THE REGION: 1991



Source: SEWRPC.

In 1963, there were 527,300 personal vehicles, or 1.07 available vehicles per household, in the Region. In 1972, there were 704,600 personal vehicles, or 1.24 available vehicles per household. In 1991, there were 1,142,500 personal vehicles available, or 1.60 vehicles per household. Survey data indicate that the number of zero- and one-vehicle households has decreased over time, while the number of households with two or more vehicles has increased significantly. As shown on Maps 39, 40, and 41, while in 1963 and 1972 areas with households with two or more vehicles were concentrated in the suburban Milwaukee area, by 1991 such areas comprised nearly the entire Region, excluding only scattered rural locations and the central portions of major cities.

AVERAGE WEEKDAY INTERNAL PERSON TRIPS PER HOUSEHOLD IN THE REGION BY INCOME GROUP: 1963, 1972, AND 1991

	1963 ^a						
·	Households ^b		Person				
Income Range (1963 actual dollars)	Number	Percent of Total	Number	Percent of Total	Person Trips per Household		
Under 2,000	45,400 46,100 92,400 94,800 52,400 24,700 9,300 5,000	12.0 12.1 24.3 25.0 13.8 6.5 2.4 1.3	94,600 194,200 649,200 865,600 569,300 291,300 115,500 61,500	3.2 6.5 21.8 29.1 19.2 9.8 3.9 2.1	2.1 4.2 7.0 9.1 10.9 11.8 12.4 12.3		
Total	379,800	100.0	2,971,200	100.0	7.8		

	1972							
	Households ^C		Person					
Income Range (1972 actual dollars)	Number	Percent of Total	Number	Percent of Total	Person Trips per Household			
Under 2,000	14,700	3.8	37,600	1.1	2.6			
2,000 to 3,999	36,900	9.5	107,700	3.2	2.9			
4,000 to 5,999	31,000	7.9	163,000	4.8	5.3			
6,000 to 7,999	36,100	9.3	221,400	6.5	6.1			
8,000 to 9,999	53,700	13.8	428,200	12.5	8.0			
10,000 to 11,999	63,500	16.3	615,000	18.0	9.7			
12,000 to 14,999	69,000	17.7	772,300	22.6	11.2			
15,000 to 24,999	66,900	17.2	824,200	24.1	12.3			
25,000 to 49,999	15,300	3.9	214,600	6.3	14.0			
50,000 or Over	2,900	0.7	37,700	1.1	13.0			
Total	390,000	100.0	3,421,700	100.0	8.8			

	1991								
	House	eholds ^d	Person	Trips					
Income Range (1991 actual dollars)	Number	Percent of Total	Number	Percent of Total	Person Trips per Household				
Under 5,000	27,900	4.9	63,400	1.4	2.3				
5,000 to 9,999	40,900	7.1	140,100	3.1	3.4				
10,000 to 14,999	45,700	8.0	215,200	4.8	4.7				
15,000 to 19,999	49,800	8.7	292,200	6.5	5.9				
20,000 to 24,999	57,400	10.0	375,400	8.3	6.5				
25,000 to 29,999	53,100	9.3	403,700	8.9	7.6				
30,000 to 34,999	55,500	9.7	467,100	10.3	8.4				
35,000 to 39,999	48,100	8.4	434,600	9.6	9.0				
40,000 to 49,999	71,700	12.5	723,200	16.0	10.1				
50,000 to 74,999	85,500	14.9	955,800	21.1	11.2				
75,000 to 99,999	22,600	3.9	274,900	6.1	12.2				
100,000 or Over	15,800	2.8	178,000	3.9	11.3				
Total	574,000	100.0	4,523,600	100.0	7.9				

^aIncome data was collected only for the urbanized areas in 1963.

^bApproximately 8 percent of total households did not provide household annual income data.

^CApproximately 31 percent of total households did not provide household annual income data.

^dApproximately 20 percent of total households did not provide household annual income data.

Source: SEWRPC.

DISTRIBUTION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY MODE OF TRAVEL: 1963, 1972, AND 1991

Paraga Tring											
		-	Persor	n Irips			1.				
	19	63	19	72	19	91	Change: 1	Change: 1963-1991		972-1991	
		Percent		Percent		Percent					
Mode of Travel	Number	of Total	Number	of Total	Number	of Total	Number	Percent	Number	Percent	
Auto Driver	2,165,700	60.1	2,897,000	64.3	4,077,500	72.9	1,911,800	88.3	1,180,500	40.7	
Auto Passenger	985,100	27.4	1,227,400	27.2	1,054,400	18.8	69,300	7.0	-173,000	-14.1	
Mass Transit	323,900	9.0	186,200	4.1	178,000	3.2	-145,900	-45.0	-8,200	-4.4	
School Bus	119,900	3.3	173,800	3.9	229,000	4.1	109,100	91.0	55,200	31.8	
Other ^a	8,400	0.2	20,500	0.5	55,400	1.0	47,000	559.5	34,900	170.2	
Total	3,603,000	100.0	4,504,900	100.0	5,594,300	100.0	1,991,300	55.3	1,089,400	24.2	

^aIncludes taxi, motorcycle, rail, and intercity bus in 1963 and 1972; walk and bicycle trips are also included in 1991.

Source: SEWRPC.

70 percent of auto-oriented travel, in 1972 to nearly 4.1 million driver-only trips, or 79 percent of auto-oriented travel, in 1991, an increase of about 1.2 million driver-only trips, or 41 percent, since 1972.

This decline in automobile passenger trips and increase in driver-only trips is reflected in the average vehicle occupancy rate for all trip purposes, which decreased from 1.45 persons in 1963 to 1.42 persons in 1972 and to 1.26 persons in 1991.

The substantial growth in household vehicle availability, particularly with respect to multiple-vehicle households, along with the decline in average household size, has contributed to the increase in driver-only travel and to the decline in automobile passenger travel, and the attendant decline in average automobile occupancy. The decline in automobile occupancy is particularly evident with regard to work trips. Average vehicle occupancy for work trips has declined from 1.21 persons in 1963 to 1.17 persons in 1972 and to 1.06 persons in 1991.

Mass transit passenger travel in the Region, in contrast to the increases in auto travel, decreased from about 324,000 revenue passenger trips per average weekday in 1963 to 186,000 in 1972, or by about 43 percent, and to 178,000 in 1991, or by about 4 percent from the 1972 levels. Mass transit travel, which constituted about 9 percent of total regional travel in 1963, declined to only 4 percent of total regional travel in 1972, and to only 3 percent in 1991. This decline in transit use occurred despite a

Figure 34

PERCENTAGE CHANGES IN VEHICLE AVAILABILITY, PERSON TRIPS, POPULATION, AND MASS TRANSIT REVENUE PASSENGERS IN THE REGION: 1963, 1972, AND 1991



Source: SEWRPC.

60 percent increase in total person tripmaking on an average weekday in the Region, as shown in Figure 34.

Person trips by school bus within the Region increased substantially, from about 120,000 in 1963 to about 174,000 in 1972, or by 45 percent, and to about 229,000 in 1991, or by 32 percent from the 1972 levels. School bus trips in 1991 constituted a larger volume of internal person trips than did general mass transit trips.

	196	3	197	'2	199	1	Percent 0	Change	Percent C	Change
	Trip Le	ngth	Trip Le	ngth	Trip Le	ngth	1972-1	991	1963-1	1991
Trip Purpose	Minutes	Miles	Minutes	Miles	Minutes	Miles	Minutes	Miles	Minutes	Miles
Home-Based Work	18.5	6.1	16.1	7.5	16.9	9.1	5.0	21.3	-8.6	49.2
Home-Based Shopping	9.7	3.3	9.6	4.0	9.1	4.3	-5.2	7.5	-6.2	30.3
Home-Based Other	12.7	4.6	11.6	4.9	10.9	5.4	-6.0	10.2	-14.2	17.4
Nonhome-Based	13.0	4.0	12.4	4.9	11.6	5.7	-6.5	16.3	-10.8	42.5
Average	13.5	4.5	12.4	5.3	12.1	6.1	-2.4	15.1	-10.4	35.5

AVERAGE TRIP LENGTHS AND TIMES FOR INTERNAL PERSON TRIPS IN THE REGION BY TRIP PURPOSE: 1963, 1972, AND 1991

Source: SEWRPC.

Trips to work by bicycle or walking totaled 23,600 in 1963, 32,400 in 1972, and 39,200 in 1991, representing an increase of 37 percent between 1963 and 1972 and an additional increase of 21 percent between 1972 and 1991.

<u>Trip Length</u>: As Table 68 shows, the average length in minutes of home-based work trips within the Region increased by 5 percent between 1972 and 1991. The average length in minutes of internal person trips for all other purposes decreased by about 6 percent during this same period. In terms of trip length measured in miles, home-based work trip lengths increased by about 21 percent; trips for other purposes increased by between 8 and 16 percent between 1972 and 1991. Between 1972 and 1991 the trip length average for all purposes as measured in travel time decreased by about 2 percent but as measured in distance increased by about 15 percent.

Between 1963 and 1991, work trip lengths measured in travel time decreased by about 9 percent and the average length in minutes of trips for all other purposes decreased by between 6 and 14 percent. In terms of trip length measured in miles, work trip lengths increased by about 49 percent. The average length in miles of internal person trips for other purposes increased by between 17 and 43 percent during this same time period. Between 1963 and 1991 the trip length average for all purposes as measured in travel time decreased by about 10 percent, but as measured in distance, it increased by nearly 36 percent.

Selected Characteristics of Users of Various Modes of Internal Person Travel: In 1963 and 1972, over 60 percent of the driver-only trips within the Region on an average weekday were made by males, while over 60 percent of automobile passenger and mass transit trips were made by females (see Table 69). However, in 1991, driver-only trips were made almost equally by males and females, and automobile passenger and mass transit trips showed an increase in the number of males and a corresponding decrease in the number of females. Table 69 also indicates that of all age groups in each of the survey years, persons from ages 35 to 44 generated the highest percentage of driver-only trips, children from ages five to 14 generated the highest percentage of auto and school bus passenger trips, and people from ages 15 to 19 generated the highest percentage of mass transit passenger trips. The percentage of trips made in 1991 by persons 65 years of age or older as either auto drivers or auto passengers more than doubled from the percentage distributions found on the 1963 and 1972 surveys, reflecting the increasing activity of this age group.

The 1991 data show that since the 1963 and 1972 surveys, the percentage of mass transit passenger trips made by persons 45 years of age or older decreased, while the percentage of such trips made by persons under the age of 45 increased. Table 69 also shows a younger median-age group making trips as mass transit passengers in each succeeding survey year: in 1963, the median age group was 35 to 44; in 1972, the median age group was 25 to 29; and in 1991, the median age group was 20 to 24 years of age.

PERCENTAGE DISTRIBUTION OF SELECTED CHARACTERISTICS OF TRIPMAKERS BY MODE OF TRAVEL IN THE REGION: 1963, 1972, AND 1991

		Perc	ent of Trips by M	Node	
	N .	Auto	Auto	Transit	School Bus
Selected Characteristic	Year	Driver	Passenger	Passenger	Passenger
Sex					
Male	1963	69.9	31.9	36.8	49.0
	1972	60.7 E0.4	31.7	34.2	50.3
· · ·	1991	50.4	35.0	41.2	51.3
Female	1963	30.1	68.1	63.2	51.0
	1972	39.3 49.6	64.4	05.8 58.8	49.7
	1001	40.0	100.0	100.0	+0.7
Ισται	1963	100.0	100.0	100.0	100.0
·	1991	100.0	100.0	100.0	100.0
Five to 14	1963		31.2	9.6	75.4
	1972		30.3	12.2	77.4
	1991		39.4	19.1	85.3
15 to 19	1963	3.9	11.2	19.2	23.7
	1972	6.3	15.7	22.1	21.5
	1991	3.0	9.5	20.5	13.9
20 to 24	1963	8.6	7.6	9.8	
	1972	11.8	10.3	11.7	
	1991	5.3	3.9	11.9	
25 to 29	1963	11.5	6.6	4.6	
	1972	12.5	7.0	5.0	
	1991	9.0	4.4	8.9	
30 to 34	1963	12.6	5.8	4.6	
	1972	11.1	5.0	3.9	
	1991	13.3	4.8	8.2	
35 to 44	1963	28.1	13.7	13.6	
	1972	21.9	8.0	9.2	
	1991	27.0	9.0	11.4	
45 to 54	1963	19.9	11.2	14.7	
	1972	19.4	9.7	62	
EE to GA	1062	10.7	0.0	14.0	
55 10 64	1903	10.7	7.4	14.3	
	1991	12.7	8.5	5.6	
65 and Older	1963	<u> </u>	53	9.6	
	1972	6.2	6.5	10.8	
	1991	12.7	15.5	8.2	
Total	1963	100.0	100.0	100.0	99.1 ^b
	1972	100.0	100.0	100.0	98.9
	1991	100.0	100.0	100.0	99.2
				*	
				· · · ·	

Table 69 (continued)

		Perc	cent of Trips by N	Node	
Selected Characteristic	Year	Auto Driver	Auto Passenger ^a	Transit Passenger	School Bus Passenger
Other			_		
Licensed Drivers	1963	100.0	38.4	29.6	8.2
	1972	100.0	44.9	29.2	8.6
	1991	100.0	41.8	36.7	3.8
Going to and from Work as	1963	46.3	16.4	50.8	
Proportion of Total Purpose	1972	42.0	15.4	42.5	
Travel for Each Mode	1991	40.9	10.2	29.4	0.2
Going to and from School as	1963	1.2	10.2	21.1	99.6
Proportion of Total Purpose	1972	2.4	10.2	28.9	99.2
Travel for Each Mode	1991	2.9	19.3	41.0	99.5
Trips In Six Peak Hours as	1963	41.3	32.9	64.6	86.0
Proportion of Total Weekday	1972	42.1	35.9	62.4	79.0
Travel for Each Mode	1991	43.4	42.4	55.2	77.5
Going to and from Milwaukee CBD	1963	6.0	4.4	31.2	
as Proportion of Total Weekday	1972	5.1	4.5	30.7	
Travel for Each Mode	1991	6.8	5.1	28.4	3.3

^aIncludes truck and taxi passengers.

^bThe remaining 0.9 percent in 1963, 1.1 percent in 1972, and 0.8 percent in 1991 are distributed throughout the other age groups.

Source: SEWRPC.

As Table 69 also indicates, the number of auto passenger trips made by licensed drivers varied in each of the survey years, the number of licensed drivers using mass transit increased in 1991, and the number of such drivers using a school bus decreased. The number of work-based trips to total person trips continued to decrease slightly for auto drivers, decreasing by approximately one-third for both auto passenger and mass transit travel. While the number of schoolbased trips remained relatively constant in the auto driver category, substantial increases occurred in the auto passenger and mass transit passenger categories.

From 1963 to 1972 and from 1972 to 1991, the proportion of total person trips made in the six peak traffic hours of the day to total person trips increased for both auto drivers and auto passengers, while the proportions of such trips to total person trips for mass transit passengers and school bus passengers decreased. In 1991, peak-hour trips accounted for slightly more than 40 percent of total auto driver and auto passenger travel, 55 percent of mass transit trips, and nearly 78 percent of school bus passenger trips.

It is significant to note that while travel to and from the Milwaukee central business district (CBD) by mass transit declined from 36 percent of all trips made in the Milwaukee CBD in 1963 to 22 percent in 1972 and to 13 percent in 1991, the proportion of mass transit trips made to and from the Milwaukee CBD relative to all mass transit trips made throughout the Region remained relatively stable at about 30 percent between 1963 and 1991.

<u>Mass Transit Trip Production</u>: The total number of households making mass transit trips on an average weekday in the urbanized areas of the Region decreased from approximately 110,500 in 1963 to 73,200 in 1972 and to 62,500 in 1991. The relationships of mass transit trip production in

AVERAGE WEEKDAY INTERNAL TRANSIT TRIPS PER TRANSIT TRIPMAKING HOUSEHOLD IN THE URBANIZING AREAS OF THE REGION BY AUTOMOBILE AVAILABILITY: 1963, 1972, AND 1991

	1963							
Automobiles	House	holds	Transit	Transit Trips				
Available per Household	Number	Percent of Total	Number	Percent of Total	Trips per Household			
None	39,300 56,200 13,500 1,500	35.6 50.8 12.2 1.4	127,100 153,000 35,800 3,900	39.8 47.8 11.2 1.2	3.2 2.7 2.7 2.6			
Total	110,500	100.0	319,800	100.0	2.9			

	1972							
Automobiles	House	eholds	Transi	Average				
Available per Household	Number	Percent of Total	Number	Percent of Total	Trips per Household			
None	27,400 30,800 12,300 2,700	37.4 42.1 16.8 3.7	74,900 76,600 28,300 5,000	40.5 41.5 15.3 2.7	2.7 2.5 2.3 1.9			
Total	73,200	100.0	184,800	100.0	2.5			

	1991							
Automobiles	House	holds	Transit	Transit Trips				
Available per Household	Number	Percent of Total	Number	Percent of Total	Trips per Household			
None	23,100 17,500	36.9 28.0	80,300 46,600	45.1 26.2	3.5 2.7			
Two	15,300 6,600	24.5 10.6	36,100 15,000	20.3 8.4	2.4 2.3			
Total	62,500	100.0	178,000	100.0	2.8			

Source: SEWRPC.

the Region to automobile availability, household size, and income are set forth in Tables 70, 71, and 72, respectively.

As Table 70 shows, in each of the survey years the average number of transit trips per transit tripmaking household decreased as the number of automobiles owned by the household increased. Also, from 1963 to 1991, the number of zero- or one-automobile households making transit trips decreased, and the number of twoor-more-automobile households making transit trips increased. These decreases and increases are consistent with the decrease in total zero- or one- automobile households and the increase in total two-or-more-automobile households in the urbanizing areas in the Region.

As Table 71 shows, although the average number of transit trips per transit tripmaking household generally increased as household size increased, the average number of transit trips per person in these households generally decreased in relation to household size. The average number of transit trips per household and per person for transit tripmaking households of each household size have remained relatively constant over the past three decades.

AVERAGE WEEKDAY INTERNAL TRANSIT TRIPS PER TRANSIT TRIPMAKING HOUSEHOLD IN THE URBANIZING AREAS OF THE REGION BY HOUSEHOLD SIZE: 1963, 1972, AND 1991

					1					
		1963								
	Households		Transit	t Trips	Average	Average				
Household Size (persons)	Number	Percent of Total	Number	Percent of Total	Trips per Household	Trips per Person				
One	13,800	12.5	34,700	10.9	2.5	2.5				
Two	29,900	27.1	78,000	24.4	2.6	1.3				
Three	21,900	19.8	64,100	20.0	2.9	1.0				
Four	17,500	15.8	50,600	15.8	2.9	0.7				
Five or More	27,400	24.8	92,400	28.9	3.4	0.7				
Total	110,500	100.0	319,800	100.0	2.9	0.9				

	1972								
	House	Households		t Trips	Average	Average			
Household Size (persons)	Number	Percent of Total	Number	Percent of Total	Trips per Household	Trips per Person			
One	12,800	17.6	27,800	15.0	2.2	2.2			
Two	17,800	24.3	38,300	20.7	2.2	1.1			
Three	11,600	15.8	27,100	14.7	2.3	0.8			
Four	11,600	15.8	30,700	16.6	2.6	0.7			
Five or More	19,400	26.5	60,900	33.0	3.1	0.6			
Total	73,200	100.0	184,800	100.0	2.5	0.8			

	1991										
	House	eholds	Transi	t Trips	Average	Average					
Household Size (persons)	Number	Percent of Total	Number	Percent of Total	Trips per Household	Trips per Person					
One	12,600	20.2	32,400	18.2	2.6	2.6					
Two	10,500	16.8	27,000	15.2	2.6	1.3					
Three	12,900	20.6	34,200	19.2	2.7	0.9					
Four	13,000	20.8	36,100	20.3	2.8	0.7					
Five or More	13,500	21.6	48,300	27.1	3.6	0.7					
Total	62,500	100.0	178,000	100.0	2.8	0.8					

Source: SEWRPC.

Table 72 presents the distribution of households, transit trips, and average transit trips per household among six household annual-income ranges. The survey findings indicate that lowerincome households constituted a larger share of all households with members using public transit in 1991 than in 1963 and 1972. Indeed, households earning under \$14,000 per year, in 1991 dollars, comprised 32 percent of all transitusing households in 1991 as opposed to about 17 percent in 1963 and 15 percent in 1972. It follows, then, that transit trips made by members of lower-income households comprised a larger share of all transit trips made in 1991 than in 1963 and 1972. Indeed, members of households earning under \$14,000 per year (1991 dollars) made 38 percent of all transit trips in 1991, as opposed to about 17 percent in 1963 and 15 percent in 1972. As a result, although the average number of transit trips made by all

AVERAGE WEEKDAY INTERNAL TRIPS PER TRANSIT TRIPMAKING HOUSEHOLD BY THE URBANIZING AREAS OF REGION BY INCOME GROUP: 1963, 1972, AND 1991

	1963							
	Households		Transi	Transit Trips				
Annual Income Range (constant 1991 dollars)	Number ^a	Percent of Total	Number	Percent of Total	Trips per Household			
Under 14,000	16,600	16.8	47,700	16.7	2.9			
14,000 to 27,899	31,400	31.8	91,600	32.2	2.9			
27,900 to 41,899	31,600	32.0	91,200	32.0	2.9			
41,900 to 55,899	13,800	14.0	38,900	13.7	2.8			
55,900 to 76,799	3,900	4.0	11,000	3.9	2.8			
76,800 or Over	1,400	1.4	4,300	1.5	3.1			
Total	98,700	100.0	284,700	100.0	2.9			

		1972							
	Households		Trans	it Trips	Average				
Annual Income Range (constant 1991 dollars)	Number ^a	Percent of Total	Number	Percent of Total	Trips per Household				
Under 14,000	6,800	14.7	17,000	14.6	2.5				
14,000 to 27,899	9,800	21.2	26,500	22.7	2.7				
27,900 to 41,899	13,900	30.0	32,300	27.7	2.3				
41,900 to 55,899	8,900	19.2	23,700	20.3	2.7				
55,900 to 76,799	4,600	9.9	10,700	9.2	2.3				
76,800 or Over	2,300	5.0	6,400	5.5	2.8				
Total	46,300	100.0	116,600	100.0	2.5				

	1991							
	Households		Transi	Transit Trips				
Annual Income Range (constant 1991 dollars)	Number ^a	Percent of Total	Number	Percent of Total	Trips per Household			
Under 14,000	16,900 14,600 9,700 5,500 5,100 1,200	31.9 27.5 18.3 10.4 9.6 2.3	57,400 44,400 23,700 13,900 10,400 2,800	37.6 29.1 15.5 9.1 6.8 1.9	3.4 3.0 2.4 2.5 2.0 2.3			
Total	53,000	100.0	152,600	100.0	2.9			

^aApproximately 11 percent of transit tripmaking households in 1963, 37 percent in 1972, and 15 percent in 1991 did not provide household annual income data.

Source: SEWRPC.

households in the Region was recorded at 2.9 per weekday in both 1963 and 1991, lower-income households made more transit trips on an average day in 1991 than in 1963, and higherincome households made fewer transit trips on an average weekday in 1991 than in 1963. As shown in Table 72, transit trips made by persons from households earning incomes under \$14,000 per year (1991 dollars) increased by 21 percent between 1963 and 1991. Conversely, transit trips

DISTRIBUTION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY TRIP PURPOSE AT DESTINATION: 1963, 1972, AND 1991

			Person	Trips						
	19	33 1972		72	2 1991		1 Change: 19		Change: 1	972-1991
Trip Purpose	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent	Number	Percent
Home-Based Work	895,900	24.9	1,064,500	23.6	1,314,600	23.4	418,700	46.7	250,100	23.5
Home-Based Shopping	546,800	15.2	678,600	15.1	803,800	14.4	257,000	47.0	125,200	18.4
Home-Based Other	1,197,200	33.2	1,556,000	34.5	1,714,900	30.7	517,700	43.2	158,900	10.2
Nonhome-Based	649,600	18.0	787,400	17.5	1,133,900	20.3	484,300	74.6	346,500	44.0
School	313,500	8.7	418,400	9.3	627,100	11.2	313,600	100.0	208,700	49.9
Total	3,603,000	100.0	4,504,900	100.0	5,594,300	100.0	1,991,300	55.3	1,089,400	24.2

Source: SEWRPC.

made by persons from households earning \$27,900 or more per year (1991 dollars) decreased by 65 percent during the same period of time.

These findings indicate that the decline in transit ridership between 1963 and 1991 occurred primarily among persons from more affluent households, leaving the use of mass transit increasingly dominated by persons from lowerincome households. The reasons for this decline may include the movement of households from the traditional transit service areas in the densely urbanized parts of the Region, the significant increase in the number of persons living in poverty in those same areas, and cutbacks in transit service that promoted mode shifts among those able to own and operate a personal vehicle. The income characteristics of the existing users of public transit indicate that the significant percentage decreases in transit use which occurred between 1963 and 1991 may not be expected to continue into the future, since existing mass transit ridership is limited in its ability to make use of alternative modes of travel.

<u>Purposes of Internal Trips</u>: The activities of a household are usually centered around the home. It logically follows, therefore, that home-oriented travel may be expected to account for a large portion of total internal person travel on an average weekday. Trips which have either an origin or destination at home, which are known as "home-based" trips, were found to comprise about four-fifths of total average weekday internal person trips in each of the survey years. This indicates the importance of the home as a generator of trips. Thus, future facility requirements within the Region will be determined in large measure by the amount and location of future residential development.

The percentage distributions of the purposes of weekday internal person trips remained stable during the period from 1963 to 1991 (see Table 73). During this period, home-based work trips comprised between 23 and 25 percent of all such trips; home-based shopping trips, between 14 and 15 percent; home-based trips in other categories, between 31 and 35 percent; nonhomebased trips, between 18 and 20 percent; and school trips, between 9 and 11 percent. These percentage distributions remained stable over three decades despite substantial increases in the absolute numbers of trips in all categories. This stability demonstrates that some aspects of travel are indeed orderly and regular over time. In percentage terms, the largest increases in internal person trips between 1972 and 1991 were found in nonhome-based trips, which increased by 346,500, or 44 percent, and school trips, which increased by 208,700 trips, or 50 percent.

Average Auto Occupancy by Selected Trip <u>Purpose</u>: The overall average number of persons per auto, including the driver, remained relatively constant from 1963 to 1972, at 1.45 and 1.42, respectively. From 1972 to 1991, however, the overall occupancy rate decreased substantially, from 1.42 to 1.26 persons per auto. With the exception of school trips, only minor differences occurred within given trip purpose categories from 1963 to 1972. Average auto occupancy

		Average Auto Occupancy by Selected Trip Purpose (number of persons)									
Year	Home-Based Work	Home-Based Shopping	Home-Based Other	Nonhome Based	School	Total Travel					
1963	1.21	1.53	1.58	1.34	4.96	1.45					
1972	1.17	1.47	1.54	1.38	2.83	1.42					
1991	1.06	1.27	1.34	1.20	2.71	1.26					

AVERAGE AUTOMOBILE OCCUPANCY OF AVERAGE WEEKDAY INTERNAL TRIPS IN THE REGION BY SELECTED TRIP PURPOSE: 1963, 1972, AND 1991

Source: SEWRPC.

decreased significantly in every trip purpose classification from 1972 to 1991. The most notable decrease was for home-based work trips which, in 1991, showed an auto occupancy rate of 1.06 persons per auto (see Table 74).

<u>Relationship of Travel Mode and Trip Purpose</u>: The distributions of internal person trips by trip purpose and by mode of travel, as shown in Figure 35, remained relatively unchanged between 1963, 1972, and 1991. Figure 35 sets forth the percentage distributions of internal person trips by trip purpose and by mode of travel. As Figure 35 shows:

- Of total driver-only trips, the percentage distributions of trip purposes are markedly similar in all three survey years despite the increases in total driver-only trips recorded in 1972 and 1991.
- Of total auto trips with passengers, the percentage distributions of home-based work trips and the percentage distribution of home-based shopping trips decreased in both 1972 and 1991. The largest increase occurred in home-based school trips in 1991, from 9 percent in 1963 and 1972 to 16 percent in 1991.
- Of total bus passenger trips, the percentage of home-based work trips decreased from 47 percent in 1963 to 38 percent in 1972 and to 26 percent in 1991. The largest increases

occurred in school trips, from 20 percent in 1963 to 27 percent in 1972 and to 39 percent in 1991.

Hourly Patterns of Internal Person Travel: The hourly distributional patterns of internal person trips indicated that although total person trip volumes increased substantially on an average weekday from 1963 to 1972 and from 1972 to 1991, the regular ebb and flow of travel remained markedly similar both in the proportion of trips by trip purpose and in the proportion and times of peak periods (see Figures 36, 37, and 38). Approximately 32 percent of daily travel within the Region occurred in the two morning and two afternoon peak hours of the day in each survey year. Of these peak-hour movements, trips to and from work comprised 47 percent of the total in 1963, 44 percent in 1972, and 41 percent in 1991. These findings indicate that one of the primary transportation problems within the Region continues to be meeting the peak demand of the journeys to and from work.

Daily Patterns in Person Travel: The patterns shaped by the variations of daily volumes of total weekday internal person travel in each of the survey years are quite similar (see Figure 39). The patterns indicate that total internal person travel varied no more than 6 percent from the daily average for any weekday other than Friday. Total Friday person trip volumes were 14 percent above average in 1963, 9 percent above average in 1972, and 7 percent above average in 1991. Because driver-only trip volumes dominate person trip volumes, the patterns of those trips closely matched those of



PERCENTAGE DISTRIBUTION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY TRIP PURPOSE AND MODE OF TRAVEL: 1963, 1972, AND 1991

total person trips in each of the survey years. The patterns of mass transit trips were about the same in all three survey years, varying by no more than 8 percent from the weekday average in each of the three survey years.

The patterns of auto, truck, and taxi passenger trips had the largest variance from the weekday average in each of the survey years, with the travel on Friday rising to 25 percent above the average in both 1963 and in 1991 and to 19 percent above the average in 1972. The patterns of school bus passenger trips, although exhibiting differences by day of week in each of the survey years, varied by no more than 12 percent from the weekday average. Differences are probably random in nature, possibly occurring because of the randomness of such activities as scheduled field trips, teacher conferences, and other special school activities, as well as student absenteeism.

The reasons for daily variations in weekday internal person travel are indicated in Figure 40, which shows the daily variations by trip purpose. The large variations in tripmaking on Fridays in each of the survey years were largely caused by the significant increases in homebased shopping, nonhome-based, and homebased "other" trips. Home-based shopping trips were 38 percent, 19 percent, and 14 percent above the weekday average on Fridays in 1963, 1972, and 1991, respectively. Nonhome-based trips were 25 percent, 24 percent, and 16 percent above the weekday average on Fridays in 1963, 1972, and 1991, respectively. Home-based "other" trips were approximately 12 to 13 percent above the weekday average on Fridays in each of the survey years.

For home-based shopping and nonhome-based trips, the variance from the weekday average for Friday trips decreased from both 1963 to 1972 and from 1972 to 1991 (see Figure 40). The reasons for this decrease are unclear, but may be related to the changes in lifestyle which have accompanied the trend toward smaller households and increased rates of female participation in the labor force.

Vehicle Availability and Internal Vehicle Trip Production: In addition to the approximately 1,142,500 personal-use automobiles, trucks, and vans and the 45,000 motorcycles available to residents of the Region in 1991, there were about 87,500 commercial trucks and 900 limousines and specialized carriers available for use within the Region on an average weekday in 1991. Together, they constitute a total of 1,275,900 vehicles available in the Region on weekdays (see Table 75). From 1963 to 1972, total weekday vehicle availability in the Region increased from 586,300 vehicles to 803,300 vehicles, or by 37 percent; from 1972 to 1991, total weekday vehicle availability in the Region increased from 803,300 vehicles to 1,275,900 vehicles, or by 59 percent.

As Table 75 also indicates, on an average weekday in 1991, the 1,275,900 vehicles available within the Region made a total of 4,620,300

internal vehicle trips. This is an increase of 1,330,000 vehicle trips, or 40 percent, over the 1972 total of 3,290,300 internal vehicle trips. In turn, the 1972 total represents an increase of 823,900 vehicle trips, or 33 percent, over the 1963 total of 2,466,400 internal vehicle trips. These increases, which are substantially higher than the 25 percent increase in person trips from 1963 to 1972 and the 24 percent increase in person trips from 1972 to 1991, reflect the rapid 37 percent growth in vehicle availability from 1963 to 1972 and the 59 percent growth from 1972 to 1991. The increases in vehicle trips also reflect the decline in average vehicle occupancy per trip from 1.42 persons per vehicle in 1972 to 1.26 persons per vehicle in 1991.

Vehicles defined as being in personal use by resident households averaged 4.1 internal vehicle trips per vehicle on an average weekday and accounted for about 88 percent of total internal vehicle trips in 1963 and 1972. In 1991, they averaged 3.6 vehicle trips per weekday but still accounted for 88 percent of total internal vehicle trips. Commercial trucks accounted for 11 percent of total internal vehicle trips and had a trip rate of 5.9 trips per vehicle on an average weekday in 1991. In 1963 and 1972, by comparison, commercial trucks accounted for 12 percent and 11 percent, respectively, of total internal vehicle trips, and had a trip rate of 5.0 and 4.8 weekday trips per vehicle, respectively. In 1972, the 21,000 motorcycles available made 8,000 internal trips, a trip rate of 0.4 trip per vehicle. By 1991, the number of motorcycles had more than doubled to 45,000, but made only 6,300 trips, a trip rate of 0.1 trip per vehicle.

Internal Commercial Truck Trip Production by Truck Type: The number of trucks available within the Region increased from 58,500 in 1963 to 77,250 in 1972 and to 87,500 in 1991 (see Table 76). It should be noted that, because of the substantial increase in the use of light trucks as personal vehicles rather than as commercial vehicles, the 1991 definition of commercial trucks excludes trucks employed primarily for personal use. Such personal-use trucks were included with automobiles as personal-use vehicles. In 1963, personal-use trucks represented 5,100, or about 9 percent, of the total 58,500 trucks available; in 1972 they represented 18,100, or about 23 percent, of the total 77,250 trucks available. In 1991, personal-use trucks were estimated to total about 80,600 trucks, or about





HOME

BEGINNING HOUR TIME P. M.

HOURLY VARIATION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY TRIP PURPOSE AT DESTINATION: 1963

Source: SEWRPC.

A.M.



HOURLY VARIATION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY TRIP PURPOSE AT DESTINATION: 1991

Source: SEWRPC.

48 percent of the total 168,100 trucks available. There was a slight decrease in the number of light trucks available for commercial use in 1991 as compared to 1972, from 51,000 light trucks in 1972 to 49,100 such trucks in 1991. This decrease was due to a change in the definition which excluded personal-use light trucks. Medium trucks in commercial use increased from 22,850 to 28,400 during the period from 1972 to 1991, an increase of 24 percent. The number of heavy trucks in commercial use decreased from 3,400 in 1972 to 3,100 in 1991, a decrease of 9 percent. In 1991, for the first time, municipal trucks were classified separately. The 6,900 municipal trucks recorded in 1991 accounted for 8 percent of all commercial trucks.

Together, these light, medium, heavy, and municipal trucks in commercial use made an estimated total of 520,100 trips on an average weekday in 1991, representing an increase of 149,100 trips, or 40 percent, from 1972 to 1991. Trip rates for medium and heavy trucks showed increases in both 1972 and 1991, indicating greater daily use of these trucks, since their total numbers did not increase at the same rate as



DAILY VARIATION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY MODE OF TRAVEL: 1963, 1972, AND 1991

autos and trucks in personal use did during that period. The average number of trips per weekday for all trucks in commercial use was 5.0 trips per vehicle in 1963, 4.8 in 1972, and 5.9 in 1991 (see Table 76).

Characteristics of Internal Commercial Truck Use within the Region on an Average Weekday: The 1963, 1972, and 1991 inventories indicate that in the Region the average number of weekday trips per commercial truck has increased over time, along with the average distance traveled per trip. Table 77 shows that the average number of trips made per truck increased from about 5.0 in both 1963 and 1972 to 5.9 trips in 1991. The average miles traveled per truck trip increased from 4.9 miles per trip in 1963, to 7.3 miles per trip in 1972, and to 8.4 miles per trip in 1991, a 71 percent increase between 1963 and 1991 and a 15 percent increase between 1972 and 1991.

Hourly Distribution of Internal Vehicle Travel: The hourly distribution of commercial truck trips and trips by limousines and specialized carriers in 1991 indicates that most of such trips occurred during the business hours of the day. The peak hours for tripmaking were between 9:00 a.m. and 11:00 a.m. Trips made by light trucks were concentrated between 7:00 a.m. and 4:00 p.m., while trips made by medium and heavy trucks and municipal trucks were concentrated between 7:00 a.m. and 3:00 p.m. This decline in tripmaking by medium and heavy trucks after 3:00 p.m. in 1991 alleviates, to a degree, afternoon peak-period congestion. A decline in truck



DAILY VARIATION OF AVERAGE WEEKDAY INTERNAL PERSON TRIPS IN THE REGION BY TRIP PURPOSE: 1963, 1972, AND 1991

Source: SEWRPC.

tripmaking may be observed during the noon hour. The hourly distribution of truck and taxi tripmaking observed in 1991 is very similar to that observed in 1963 and 1972 (see Figures 41, 42, and 43).

Location of Travel

The amount of travel attracted to a given area is largely determined by the amount, type, and intensity of development in that area. Maps 42, 43, and 44 show the spatial distribution of internal person trip destinations within the Region on an average weekday for each of the survey years. The highest concentrations of internal person trip destinations on all three maps occur in the highly developed central business districts and major industrial and commercial areas of the larger urbanized areas of the Region. These concentrations include, among others, the Blue Mound Road corridor; the Milwaukee CBD; such major shopping centers as Brookfield Square, Northridge, Southridge, Regency Mall, and Mayfair; and the University of Wisconsin-Milwaukee campus. Significant increases in concentrations of person trip destinations in 1991, however, were found in many smaller communities, such as the Village of Jackson, City of Oconomowoc, Town of Richfield, Village of Union Grove, City of West Bend, and City of Whitewater, further demonstrating the decentralization of land use in Southeastern Wisconsin.

Maps 45, 46, and 47 show the spatial distribution of internal mass transit trip destinations within the Region on an average weekday for 1963,

VEHICLE AVAILABILITY AND AVERAGE WEEKDAY INTERNAL VEHICLE TRIPS IN THE REGION BY VEHICLE TYPE: 1963, 1972, AND 1991

	1963						
	Veh	icles	Vehicle	Vehicle Trips			
Type of Vehicle	Number	Percent of Total	Number	Percent of Total	Number of Trips		
Personal Vehicle ^a	527,300 58,500 500	89.9 10.0 0.1	2,166,000 293,400 7,000	87.8 11.9 0.3	4.1 5.0 14.0		
Total	586,300	100.0	2,466,400	100.0	4.2		

	1972							
	Veh	Vehicles		Trips	Average			
Type of Vehicle	Number	Percent of Total	Number	Percent of Total	Number of Trips			
Personal Vehicle ^a	704,600	87.7	2,897,000	88.0	4.1			
Commercial-Use Trucks ^b	77,250	9.6	371,000	11.4	4.8			
Taxi/Limousine/Specialized Carrier	450	0.1	14,300	0.4	31.8			
Motorcycle ^C	21,000	2.6	8,000	0.2	0.4			
Total	803,300	100.0	3,290,300	100.0	4.1			

	_		1991			
	Vehi	cles	Vehicle	Trips	Average	
Type of Vehicle	Number	Percent of Total	Number	Percent of Total	Number of Trips	
Personal Vehicle ^a	1,142,500 87,500	89.5 6.9	4,077,500 520,100	88.2 11.3	3.6 5.9	
Taxi/Limousine/Specialized Carrier Motorcycle ^c	900 45,000	0.1 3.5	16,400 6,300	0.4 0.1	18.2 0.1	
Total	1,275,900	100.0	4,620,300	100.0	3.6	

^aFor 1991, this category includes autos, personal-use trucks, and vans on home interview survey.

^bBy definition, this category, for 1963 and 1972, includes personal-use trucks as well as commercial-use trucks. In 1991, this category was defined to include only commercial-use trucks. In 1963 and 1972, personal-use trucks represented relatively modest proportions of total trucks: 5,100 of a total 58,500 trucks, or 9 percent, in 1963, and 18,100 of a total 77,250 trucks, or 23 percent, in 1972. In 1991, personal-use trucks represented 80,600 of a total 168,100 trucks, or 48 percent.

^CIn 1963, motorcycles were included with automobiles.

Source: SEWRPC.

			Trucks			Truck	Trips	
Type of Truck ^a	Year	Number	Percent of Total	Percent Change	Number	Percent of Total	Percent Change	Trips per Truck
Light	1963 1972 1991	33,800 51,000 49,100	57.8 66.0 56.1	50.9 -3.7	169,500 185,800 214,300	57.8 50.1 41.1	9.6 15.3	5.0 3.6 4.4
Medium	1963 1972 1991	20,500 22,850 28,400	35.0 29.6 32.5	 11.5 24.3	110,900 173,500 259,700	37.8 46.8 49.8	56.4 49.7	5.4 7.6 9.1
Heavy	1963 1972 1991	4,200 3,400 3,100	7.2 4.4 3.5	 -19.0 -8.8	13,000 11,700 17,500	4.4 3.1 3.6	-10.0 49.6	3.1 3.4 5.6
Municipal	1991	6,900	7.9		28,600	5.5		4.1
Total	1963 1972 1991	58,500 77,250 87,500	100.0 100.0 100.0	 32.1 13.3	293,400 371,000 520,100	100.0 100.0 100.0	26.4 40.2	5.0 4.8 5.9

COMMERCIAL-USE TRUCK AVAILABILITY AND AVERAGE WEEKDAY INTERNAL TRUCK TRIPS IN THE REGION BY TYPE: 1963, 1972, AND 1991

^aA light truck is defined as one having a gross weight of 10,000 pounds or less for farm trucks and 8,000 pounds or less for all other trucks. A medium truck is defined as one having a gross weight greater than 10,000 pounds but no more than 50,000 pounds for farm trucks and greater than 8,000 pounds but no more than 50,000 pounds for all other trucks. A heavy truck is defined as one having a gross weight of more than 50,000 pounds. Municipal trucks have no weight limitations.

Source: SEWRPC.

Table 77

				Change: 1963-1991		Change: 1972-1991	
Trip Characteristics	1963	1972	1991	Number	Percent	Number	Percent
Average Weekday Trips per Truck	5.0	4.8	5.9	0.9	18.0	1.1	22.9
Average Trip Length (miles)	4.9	7.3	8.4	3.5	71.4	1.1	15.1

SELECTED TRIPMAKING CHARACTERISTICS OF COMMERCIAL-USE TRUCKS GARAGED IN THE REGION: 1963, 1972, AND 1991

Source: SEWRPC.

1972, and 1991, respectively. These maps indicate that mass transit utilization has declined substantially from 1963 to 1972 and from 1972 to 1991. In 1991, mass transit trips were concentrated only in the Milwaukee CBD; at the University of Wisconsin-Milwaukee and Marquette University campuses; in an area bounded by S. 13th Street on the east, W. Greenfield Avenue on the south, S. 20th Street on the west and W. Pierce Street on the north; in an area bounded by N. Sherman Boulevard on the east, W. North Avenue on the south, N. 60th Street on the west, and W. Burleigh Street on the north; in an area bounded by N. 27th Street on the east, W. Hampton Avenue on the south, N. Sherman Boulevard on the west, and W. Silver Spring Drive on the north; and in an area bounded by N. 13th Street on the east, W. Burleigh Street on the south, N. 20th Street on the west, and W. Keefe Avenue on the north.



Source: SEWRPC.

Figure 42

HOURLY VARIATION OF AVERAGE

WEEKDAY INTERNAL TRUCK AND TAXI TRIPS



Source: SEWRPC.

Maps 48, 49, and 50, respectively, show the spatial distribution of internal truck trip destinations within the Region on an average weekday in 1963, when such destinations totaled 293,400; in 1972, when such destinations totaled 371,000; and in 1991, when such destinations totaled 520,100. The patterns shown on these maps indicate that internal truck trip destinations are most highly concentrated in the central areas of the larger urbanized areas of the Region and in major commercial and industrial areas within, and adjacent to, those larger urbanized areas. These maps also indicate that while substantial increases in the number of truck trip destinations

occurred from 1963 to 1972 and to 1991 in many rural areas as well as in the highly developed industrial and commercial areas of the Cities of Milwaukee, Racine, and Kenosha, the number of truck trip destinations within the central areas of these cities remained relatively unchanged.

Maps 51, 52, and 53 show the garaging locations of commercial trucks for, respectively, 1963, 1972, and 1991. Truck garaging locations, although concentrated in the urbanized areas, are dispersed throughout the rural areas of the Region. A comparison of garaging locations with truck trip destinations indicates that both



Source: SEWRPC.

the number of trucks and the number of trips has increased over time. However, the pattern of trips is concentrated in the more highly developed areas of the Region.

Maps 54, 55, and 56 show the person trip "desire lines" connecting the origins and destinations of external trips with the points of entry to, or exit from, the Region for each of the survey years. These maps also show the through movement of person travel as well as the internal-external travel interchanges. Comparisons of 1991 internal-external travel interchanges with internal-external travel in 1972 and 1963 show increased tripmaking from each of the points of entry or exit to more areas of the Region, reflecting the increase in tripmaking since 1972. Significantly, no large concentrations of trip destinations existed in any of the survey years in any single portion of the Milwaukee urbanized area. As in 1963 and 1972, the majority of through trips in 1991 followed north-south routes, the largest volumes utilizing a combination of IH 94 and either IH 43 or USH 41. Since the completion of IH 43, a greater number of 1991 north-south trips use IH 94 and IH 43 than in 1972, when a greater number of north-south trips used IH 94 and USH 41. Also, through trip points of origin and destination have increased in volume in 1991 from the southwest and the northwest.

Characteristics of Travel in

Selected Major Central Business Districts

Historically, the highly developed central business districts of the larger cities of the Region have been regarded as the financial, commercial, and cultural centers of the urbanized areas in which they are located. As such they create considerable interest with respect to growth and change. Because the degree of activity within a central business district in terms of employment and tripmaking is often equated with the wellbeing of the entire urbanized area it serves, special emphasis is placed here upon the examination of the trends from 1963 to 1972 and from 1972 to 1991 in the central business districts of Milwaukee, Racine, and Kenosha. The boundaries of these districts are shown, respectively, on Maps 57, 58, and 59. Map 57 also graphically shows the core area of the Milwaukee CBD. Tables 78, 79, and 80 each summarize survey data for each survey year and for each central business district. Table 78 presents person trips by mode of travel, Table 79 presents vehicle trips by mode of travel, and Table 80 presents person trip destinations by trip purpose and mode of travel.

Comparisons of the 1963, 1972, and 1991 inventory findings set forth in Table 78 indicate that the total number of person trip destinations on an average weekday increased in the Milwaukee CBD and decreased in the Kenosha and Racine central business districts. In spite of the fact that tripmaking on an average weekday decreased by 9,300 person trips, or 17 percent, in the core of the Milwaukee CBD between 1972 and 1991, there was an increase of 13,300 person

AVERAGE WEEKDAY INTERNAL PERSON TRIP DESTINATIONS IN THE REGION: 1972



Map 42

AVERAGE WEEKDAY INTERNAL PERSON TRIP DESTINATIONS IN THE REGION: 1963





Source: SEWRPC.





Source: SEWRPC.

The number of trips attracted to a given area of the Region is determined by the amount, kind, and intensity of land use development in that area. In all three survey years, the highest concentrations of person trip destinations were found in the highly developed urban areas of the Region. The 1991 survey findings, however, indicate both a larger dispersion of trip destinations throughout the Region and a more dense clustering of trip destinations in the smaller urban areas. The overall pattern produced by plotting the location of internal trip destinations closely resembles existing urban development patterns.

AVERAGE WEEKDAY MASS TRANSIT TRIP DESTINATIONS IN THE REGION: 1963



AVERAGE WEEKDAY MASS TRANSIT TRIP DESTINATIONS IN THE REGION: 1972



Source: SEWRPC.





Mass transit trip destinations within the Region on an average weekday decreased sharply, from about 323,900 in 1963 to 186,200 in 1972 and to 178,000 in 1991. By 1991, significant mass transit trip destinations were concentrated only in certain areas within the City of Milwaukee, including the central business district and the University of Wisconsin-Milwaukee and Marquette University campuses.

Source: SEWRPC.

AVERAGE WEEKDAY TRUCK TRIP DESTINATIONS IN THE REGION: 1972





Source: SEWRPC.

Source: SEWRPC.

AVERAGE WEEKDAY TRUCK TRIP DESTINATIONS IN THE REGION: 1963

Map 48



AVERAGE WEEKDAY TRUCK TRIP DESTINATIONS IN THE REGION: 1991



The number of internal truck trip destinations within the Region on an average weekday increased from about 293,400 in 1963 to 371,000 in 1972 and to 520,100 in 1991. As shown on Maps 48, 49, and 50, while in all three survey years truck trip destinations were most highly concentrated in the central business districts of the Cities of Kenosha, Milwaukee, and Racine, a greater dispersion of truck trip destinations throughout the Region is evident in 1991.

TRUCK GARAGING LOCATIONS IN THE REGION: 1972





Source: SEWRPC.

Source: SEWRPC.

TRUCK GARAGING LOCATIONS IN THE REGION: 1963

Map 51





TRUCK GARAGING LOCATIONS IN THE REGION: 1991



Maps 51, 52, and 53 graphically show the spatial distribution of commercial-use truck garaging locations within the Region in 1963, 1972, and 1991. There were about 58,500 such trucks garaged within the Region in 1963, 77,300 in 1972, and 87,500 in 1991. Much of the increase in commercial-use truck garaging locations since 1963 has occurred in the smaller urban areas of the Region.



AVERAGE WEEKDAY EXTERNAL TRIP DESIRE LINES IN THE REGION: 1963



Source: SEWRPC. 182



AVERAGE WEEKDAY EXTERNAL TRIP DESIRE LINES IN THE REGION: 1991

Maps 54, 55, and 56 show person-trip "desire lines" connecting the origins and destinations of external person trips with points of entry to and exit from the Region, as well as the movement of person trips directly through the Region. As indicated by the above maps, between 1972 and 1991, north-south through trips became more evenly distributed over the two primary routes serving such through traffic: IH 94-IH 43 and IH 94-USH 41. A factor affecting this change was the completion of IH 43 during the same time period.

Source: SEWRPC.

trips, or 10 percent, in the total district over that time period. In the Kenosha CBD, total person trips decreased from 30,700 in 1963 to 29,800 in 1972, or by 3 percent, and to 13,000 in 1991, a decrease of 56 percent from 1972 to 1991. In the Racine CBD, total person trips decreased from 27,500 in 1963 to 20,900 in 1972, or by 24 percent, and to 18,700 in 1991, a decrease of nearly 11 percent from 1972 to 1991.

The Milwaukee CBD showed a decrease in total person trips for an average weekday from 1963 to 1972 of 6,200, or 4 percent, and then an increase from 1972 to 1991 of 13,300, or 10 percent. As a result, there were 7,100 more trips in 1991 than in 1963. Trips made by auto passengers increased from 22,400, or 16 percent

of the total in 1963, to 28,300, or 21 percent of the total in 1972, and then declined to 19,800, or 13 percent of the total in 1991. Transit trips declined from 50,500, or 36 percent of the total, in 1963; to 29,000, or 22 percent of the total, in 1972; to 18,800, or 13 percent of the total, in 1991. During this same time period, person trips by auto drivers increased from 66,200, or 48 percent of the total in 1963, to 75,800, or 57 percent of the total in 1972, and to 103,200, or 70 percent of the total in 1991. Trips to and from the Milwaukee CBD were made mostly by automobiles: 84 percent in 1963, 86 percent in 1972 and 89 percent in 1991. The average auto occupancy of personal-use vehicles entering and leaving the Milwaukee CBD was 1.34 persons per vehicle in 1963; 1.37 in 1972; and 1.19 in 1991.

BOUNDARIES OF THE MILWAUKEE CENTRAL BUSINESS DISTRICT AND CORE AREA



Source: SEWRPC.

Trips made by trucks in commercial use to and from the Milwaukee CBD declined from 11,800 in 1963 to 11,100 in 1972, and then increased to 12,400 in 1991 (see Table 79). Trips made by all other modes, including school bus, taxi, and motorcycle, totaled 900 in 1963, 700 in 1972, and 1,400 in 1991. In addition, there were an estimated 3,900 walking and bicycle trips in 1991, resulting in a total of 5,300 trips made by other modes (see Table 78).

Table 80 sets forth internal person trip destinations by destination trip purpose and mode of travel to the Milwaukee, Kenosha, and Racine central business districts for 1963, 1972, and 1991. In the Milwaukee CBD, trips to work accounted for one-half or more of the trips in each of the survey years, increasing by 16,600 trips, or 25 percent, from 1972 to 1991. Also increasing from 1972 to 1991 were person trips to social and recreational activities, which increased by 4,800 trips, or 39 percent; homebound trips, 3,300 trips, or 77 percent; and schoolbound trips, 1,900 trips, or 24 percent. Trips made for personal business purposes increased from 25,300 in 1963 to 29,000 in 1972, an increase of 3,700, or 15 percent; and then decreased to 20,000 in 1991, a decrease of 9,000.

BOUNDARIES OF THE RACINE CENTRAL BUSINESS DISTRICT



BOUNDARIES OF THE KENOSHA CENTRAL BUSINESS DISTRICT



Maps 57, 58, and 59 show graphically the boundaries of the central business district of the Cities of Milwaukee, Racine, and Kenosha. Map 57 also shows the boundaries of the Milwaukee central business district core area.

Source: SEWRPC.

or 31 percent, between 1972 and 1991. Also decreasing were shopping trips, which decreased from 16,500 in 1963 to 12,400 in 1972, or by 25 percent; and to 7,500 in 1991, or by 40 percent from 1972 to 1991.

Tripmaking in the Kenosha CBD declined substantially from 29,800 person trips in 1972 to 13,000 person trips in 1991, a decline of 56 percent (see Table 78). This decline may be attributed to the effects of job reductions in the automotive industry that have occurred in Kenosha since 1972. In 1991, driver-only trips accounted for about 81 percent of total person trips; auto passenger trips accounted for about 16 percent; and all other modes, including transit, accounted for about 3 percent. Major declines in the number of trips by destination trip purpose from 1972 to 1991 are shown in Table 80 and include: work, from 5,700 trips to 4,900 trips, or 14 percent; shopping, from 5,500 trips to 1,000 trips, or 82 percent; personal business, from 11,600 trips to 3,200 trips, or 72 percent; social-recreational, from 2,700 trips to 1,800 trips, or 33 percent; and school, from 2,000 trips to 100 trips, or 95 percent. The decrease in school trips is due to the relocation of a high school during this time period.

In the Racine CBD, total person trips decreased from 27,500 in 1963 to 20,900 in 1972, a decrease of 24 percent; and to 18,700 in 1991, a further decrease of 11 percent (see Table 78). Vehicle trips decreased from 19,600 in 1963 to 15,900 in 1972, a decrease of 19 percent; and then increased to 16,400 in 1991, an increase of 3 percent from 1972 (see Table 79). In 1991, the primary mode choice of persons entering and

AVERAGE WEEKDAY PERSON TRIP DESTINATIONS IN SELECTED CENTRAL BUSINESS DISTRICTS OF THE REGION BY MODE OF TRAVEL: 1963, 1972, AND 1991

		1	Ailwaukee CBD		М	ilwaukee CBD Co	re
Type of Trip and Travel Mode	Year	Trips	Percent of Total	Percent Change	Trips	Percent of Total	Percent Change
Internal Auto Driver	1963 1972 1991	65,100 74,300 101,000	46.5 55.5 68.7	14.1 35 <i>.</i> 9	27,600 27,500 31,000	40.2 48.9 66.1	-0.4 12.7
Auto Passenger	1963 1972 1991	21,600 27,300 18,900	15.4 20.4 12.8	26.4 -30.8	10,700 12,100 6,000	15.6 21.5 12.8	 13.1 -50.4
Transit Passenger	1963 1972 1991	50,500 29,000 18,800	36.1 21.7 12.8	-42.6 -35.2	28,700 15,500 7,400	41.8 27.6 15.8	-46.0 -52.3
Other	1963 1972 1991	900 700 5,300 ^a	0.6 0.5 3.6	-22.2 657.1	300 100 1,400 ^b	0.4 0.2 3.0	-66.7 1,300.0
Subtotal	1963 1972 1991	138,100 131,300 144,000	98.6 98.1 97.9	 -4.9 9.7	67,300 55,200 45,800	98.1 98.2 97.7	-18.0 -17.0
External Auto Driver	1963 1972 1991	1,100 1,500 2,200	0.8 1.1 1.5	36.4 46.7	700 600 800	1.0 1.1 1.7	 -14.3 33.3
Auto Passenger	1963 1972 1991	800 1,000 900	0.6 0.7 0.6	25.0 -10.0	600 400 300	0.9 0.7 0.6	-33.3 -25.0
Subtotal	1963 1972 1991	1,900 2,500 3,100	1.4 1.9 2.1	31.6 24.0	1,300 1,000 1,100	1.9 1.8 2.3	 -23.1 10.0
Total	1963 1972 1991	140,000 133,800 147,100	100.0 100.0 100.0	-4.4 9.9	68,600 56,200 46,900	100.0 100.0 100.0	 -18.1 -16.5
			Kenosha CBD			Racine CBD	
Type of Trip and Travel Mode	Year	Trips	Percent of Total	Percent Change	Trips	Percent of Total	Percent Change
Internal Auto Driver	1963 1972 1991	19,200 19,600 9,700	62.5 65.8 74.6	2.1 -50.5	17,900 14,500 14,800	65.1 69.4 79.1	 -19.0 2.1
Auto Passenger	1963 1972 1991	8,600 9,000 1,800	28.0 30.2 13.8	4.7 -80.0	7,400 5,800 2,400	26.9 27.8 12.8	 -21.6 -58.6
Transit Passenger	1963 1972 1991	1,800 400 200	5.9 1.3 1.5	-77.8 -50.0	2,000 300 400	7.3 1.4 2.1	 -85.0 33.3
Other	1963 1972	200 400	0.7	100.0	100	0.4 1.0	100.0

1							
Type of Trip and Travel Mode	Year	Trips	Percent of Total	Percent Change	Trips	Percent of Total	Percent Change
Internal Auto Driver	1963 1972 1991	19,200 19,600 9,700	62.5 65.8 74.6	2.1 -50.5	17,900 14,500 14,800	65.1 69.4 79.1	-19.0 _2.1
Auto Passenger	1963 1972 1991	8,600 9,000 1,800	28.0 30.2 13.8	4.7 -80.0	7,400 5,800 2,400	26.9 27.8 12.8	-21.6 -58.6
Transit Passenger	1963 1972 1991	1,800 400 200	5.9 1.3 1.5	-77.8 -50.0	2,000 300 400	7.3 1.4 2.1	-85.0 33.3
Other	1963 1972 1991	200 400 200 ^c	0.7 1.3 1.5	100.0 -50.0	100 200 600 ^d	0.4 1.0 3.2	100.0 200.0
Subtotal	1963 1972 1991	29,800 29,400 11,900	97.1 98.7 91.5	-1.3 -59.5	27,400 20,800 18,200	99.6 99.5 97.3	-24.1 -12.5
External Auto Driver	1963 1972 1991	500 300 800	1.6 1.0 6.2	-40.0 166.7	100 100 300	0.4 0.5 1.6	0.0 200.0
Auto Passenger	1963 1972 1991	400 100 300	1.3 0.3 2.3	-75.0 200.0	e e _ 200	0.0 0.0 1.1	
Subtotal	1963 1972 1991	900 400 1,100	2.9 1.3 8.5	-55.6 175.0	100 100 500	0.4 0.5 2.7	400.0
Total	1963 1972 1991	30,700 29,800 13,000	100.0 100.0 100.0	-2.9 -56.4	27,500 20,900 18,700	100.0 100.0 100.0	-24.0 -10.5

^aIncludes 3,900 walking and bicycle trips.

^bIncludes 1,200 walking and bicycle trips.

^CIncludes 100 walking and bicycle trips.

^dIncludes 100 walking and bicycle trips.

^eTrips by these modes are fewer than 50. All trips are rounded to the nearest hundred.

Source: SEWRPC.
AVERAGE WEEKDAY VEHICLE TRIP DESTINATIONS IN SELECTED CENTRAL BUSINESS DISTRICTS OF THE REGION BY MODE OF TRAVEL: 1963, 1972, AND 1991

			Milwaukee CBE)	Mi	lwaukee CBD Co	re
Type of Trip and Travel Mode	Year	Trips	Percent of Total	Percent Change	Trips	Percent of Total	Percent Change
Internal Auto	1963 1972 1991	65,100 74,300 101,000	82.3 83.8 87.0	14.1 35.9	27,600 27,500 31,000	84.4 85.7 89.1	-0.4 12.7
Commercial-Use Truck	1963 1972 1991	11,600 10,900 12,100	14.7 12.3 10.4	-6.0 11.0	3,700 3,200 3,100	11.3 10.0 8.9	-13.5 -3.1
Taxi	1963 1972 1991	1,100 1,800 500	1.4 2.0 0.4	63.6 -72.2	700 800 100	2.2 2.5 0.3	 14.3 -87.5
Subtotal	1963 1972 1991	77,800 87,000 113,600	98.4 98.1 97.8	11.8 30.6	32,000 31,500 34,200	97.9 98.1 98.3	 -1.6 8.6
External Auto	1963 1972 1991	1,100 1,500 2,200	1.4 1.7 1.9	36.4 46.7	700 600 800	2.1 1.9 2.3	-14.3 33.3
Commercial-Use Truck	1963 1972 1991	200 200 300	0.2 0.2 0.3	0.0 50.0	a a 100	0.0 0.0 0.3	
Subtotal	1963 1972 1991	1,300 1,700 2,500	1.6 1.9 2.2	30.8 47.1	700 600 600	2.1 1.9 1.7	-14.3 0.0
Total	1963 1972 1991	79,100 88,700 116,100	100.0 100.0 100.0	12.1 30.9	32,700 32,100 34,800	100.0 100.0 100.0	 -1.8 8.4

			Kenosha CBD		Racine CBD			
Type of Trip and Travel Mode	Year	Trips	Percent of Total	Percent Change	Trips	Percent of Total	Percent Change	
Internal								
Auto	1963	19,200	91.4		17,900	91.3		
	1972	9,700	91.1 82.2	-50.5	14,500	91.2 90.2	2.1	
Commercial-Use Truck	1963	1,200	5.7	25.0	1,600	8.2 8.2	 -18.8	
	1991	1,200	10.2	-20.0	1,500	9.1	15.4	
Тахі	1963	100	0.5		0 0	0.0		
,	1972 1991	100	0.5 0.0	0.0 -100.0	u a	0.0 0.0		
Subtotal	1963	20,500	97.6	,	19,500	99.5		
-	1972 1991	21,200 10,900	98.6 92.4	3.4 -48.6	15,800 16,300	99.4 99.4	-19.0 3.2	
External		<i>2</i>					_	
Auto	1963	500	2.4		100	0.5		
	1972 1991	800	1.4 6.8	-40.0 166.7	300	0.6	200.0	
Commercial-Use Truck	1963		0.0		a	0.0		
	1972		0.0		a	0.0		
	1991	100	0.8			0.0		
Subtotal	1963	500	2.4	40.0	100	0.5		
	1991	900	7.6	200.0	100	0.6	0.0	
Total	1963	21,000	100.0		19,600	100.0		
	1972 1991	21,500 11,800	100.0 100.0	2.4 -45.1	15,900 16,400	100.0 100.0	-18.9 3.1	

^a Trips by these modes are less than 50. All numbers are rounded to the nearest hundred.

Source: SEWRPC.

AVERAGE WEEKDAY INTERNAL PERSON TRIP DESTINATIONS IN SELECTED CENTRAL BUSINESS DISTRICTS OF THE REGION BY TRIP PURPOSE AND MODE OF TRAVEL: 1963, 1972, AND 1991

			Milwaukee CBD: Mode of Travel to Selected CBD										
		Auto	Driver	Auto P	assenger	Transit	Passenger	Other		Total			
Destination Trip Purpose	Year	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total		
Home	1963 1972 1991	1,700 2,400 3,800	2.6 3.2 3.8	1,400 1,000 1,100	6.5 3.7 5.8	3,300 900 1,400	6.5 3.1 7.4	100 ^a 1,300	11.1 0.0 24.6	6,500 4,300 7,600	4.7 3.3 5.3		
Work	1963 1972 1991	36,000 40,000 63,900	55.3 53.9 63.3	7,600 10,800 6,600	35.2 39.6 34.9	26,500 14,500 8,900	52.5 50.0 47.3	100 200 2,700	11.1 28.6 51.0	70,200 65,500 82,100	50.8 49.9 57.0		
Shopping	1963 1972 1991	4,900 3,700 4,300	7.5 5.0 4.3	2,900 2,500 1,200	13.4 9.2 6.3	8,700 6,200 2,000	17.2 21.4 10.6	a a	0.0 0.0 0.0	16,500 12,400 7,500	12.0 9.4 5.2		
Personal Business	1963 1972 1991	13,800 18,100 13,400	21.2 24.4 13.3	4,800 7,600 3,400	22.2 27.8 18.0	6,400 3,200 2,800	12.7 11.0 14.9	300 100 400	33.4 14.3 7.5	25,300 29,000 20,000	18.3 22.1 13.9		
School	1963 1972 1991	1,700 3,600 5,200	2.6 4.8 5.1	500 800 1,200	2.3 2.9 6.3	2,600 3,300 2,900	5.2 11.4 15.4	200 100 400	22.2 14.3 7.5	5,000 7,800 9,700	3.6 5.9 6.7		
Social and Recreational	1963 1972 1991	7,000 6,500 10,400	10.8 8.7 10.2	4,400 4,600 5,400	20.4 16.8 28.7	3,000 900 800	5.9 3.1 4.4	200 300 500	22.2 42.8 9.4	14,600 12,300 17,100	10.6 9.4 11.9		
Total	1963 1972 1991	65,100 74,300 101,000	100.0 100.0 100.0	21,600 27,300 18,900	100.0 100.0 100.0	50,500 29,000 18,800	100.0 100.0 100.0	900 700 5,300	100.0 100.0 100.0	138,100 131,300 144,000	100.0 100.0 100.0		

			Kenosha CBD: Mode of Travel to Selected CBD											
		Auto	Driver	Auto P	assenger	Transit	Passenger	0	ther	Total				
Destination Trip Purpose	Year	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total			
Home	1963	600	3.1	200	2.3	100	5.6	^a	0.0	900	3.0			
	1972	1,100	5.6	700	7.8	a	0.0	100	25.0	1,900	6.5			
	1991	600	6.2	200	11.1	a	0.0	100	50.0	900	7.6			
Work	1963	6,200	32.3	1,100	12.8	300	16.6	a	0.0	7,600	25.5			
	1972	4,700	24.0	1,000	11.1	^a	0.0	a	0.0	5,700	19.4			
	1991	4,300	44.3	400	22.2	100	50.0	100	50.0	4,900	41.2			
Shopping	1963	3,800	19.8	2,800	32.6	200	11.1	a	0.0	6,800	22.8			
	1972	3,800	19.3	1,600	17.8	100	25.0	a	0.0	5,500	18.7			
	1991	600	6.2	300	16.7	100	50.0	0	0.0	1,000	8.4			
Personal Business	1963 1972 1991	6,800 7,700 2,800	35.4 39.3 28.9	2,000 3,800 400	23.2 42.2 22.2	400 100 ^a	22.2 25.0 0.0	a a a	0.0 0.0 0.0	9,200 11,600 3,200	30.9 39.4 26.9			
School	1963	300	1.6	800	9.3	700	38.9	200	100.0	2,000	6.7			
	1972	500	2.6	1,000	11.1	200	50.0	300	75.0	2,000	6.8			
	1991	100	1.0	0	0.0	0	0.0	0	0.0	100	0.8			
Social and Recreational	1963 1972 1991	1,500 1,800 1,300	7.8 9.2 13.4	1,700 900 500	19.8 10.0 27.8	100 ^a 0	5.6 0.0 0.0	a a 0	0.0 0.0 0.0	3,300 2,700 1,800	11.1 9.2 15.1			
Total	1963	19,200	100.0	8,600	100.0	1,800	100.0	200	100.0	29,800	100.0			
	1972	19,600	100.0	9,000	100.0	400	100.0	400	100.0	29,400	100.0			
	1991	9,700	100.0	1,800	100.0	200	100.0	200	100.0	11,900	100.0			

Table 80 (continued)

					Racine C	BD: Mode o	Travel to Sele	ected CBD			
		Auto	Driver	Auto P	assenger	Transit	Passenger	0	ther	Т	otal
Destination Trip Purpose	Year	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total	Trips	Percent of Total
Home	1963	500	2.8	300	4.1	100	5.0	a	0.0	900	3.3
	1972	600	4.1	300	5.2	50	16.7	a	0.0	950	4.6
	1991	500	3.4	0	0.0	100	25.0	100	16.7	700	3.8
Work	1963	5,800	32.4	1,100	14.9	800	40.0	a	0.0	7,700	28.1
	1972	4,900	33.8	1,200	20.7	50	16.7	_50	25.0	6,200	29.8
	1991	6,600	44.6	400	16.7	0	0.0	^a	0.0	7,000	38.5
Shopping	1963	3,500	19.6	2,000	27.0	400	20.0	a	0.0	5,900	21.5
	1972	2,600	0.0	1,100	19.0	100	33.3	a	0.0	3,800	18.3
	1991	1,200	8.1	100	4.2	100	25.0	a	0.0	1,400	7.7
Personal Business	1963 1972 1991	5,900 5,000 3,400	33.0 34.5 23.0	2,300 2,200 400	31.1 37.9 16.7	500 100 100	25.0 33.3 25.0	50 100 100	50.0 50.0 16.7	8,750 7,400 4,000	31.9 35.6 22.0
School	1963	200	0.0	100	1.3	a	0.0	a	0.0	300	1.1
	1972	400	2.8	100	1.7	a	0.0	50	25.0	550	2.6
	1991	200	1.4	100	4.2	a	0.0	300	50.0	600	3.3
Social and Recreational	1963 1972 1991	2,000 1,000 2,900	11.1 6.9 19.5	1,600 900 1,400	21.6 15.5 58.2	200 ^a 100	10.0 0.0 25.0	50 ^a 100	50.0 0.0 16.6	3,850 1,900 4,500	14.1 9.1 24.7
Total	1963	17,900	100.0	7,400	100.0	2,000	100.0	100	100.0	27,400	100.0
	1972	14,500	82.1	5,800	100.0	300	100.0	200	100.0	20,800	100.0
	1991	14,800	100.0	2,400	100.0	400	100.0	600	100.0	18,200	100.0

^aTrips by these vehicle types are fewer than 50.

Source: SEWRPC.

leaving the central business district was as an auto driver. Driver-only trips accounted for 15,100, or 81 percent, of the total person trips. Auto passenger travel accounted for 2,600 trips. or about 14 percent. Trips by all other modes, including transit, accounted for 1,000 person trips, or about 5 percent (see Table 78). Work trips increased from 6,200 in 1972 to 7,000 in 1991, a 13 percent increase, and trips made for social and recreational purposes increased from 1,900 in 1972 to 4,500 in 1991, an increase of 137 percent. This latter increase may be attributable to the development of Festival Park on the Racine lakefront. Shopping trips decreased from 5,900 in 1963 to 3,800 in 1972, a decrease of 36 percent, and to 1,400 in 1991, a decrease of 63 percent since 1972. Personal business trips decreased from about 8,800 person trips in 1963 to 7,400 in 1972, a decrease of 16 percent, and to 4,000 person trips in 1991, a decrease of 46 percent since 1972 (see Table 80).

Major Commercial and

Industrial Center Trip Attractors

Travel to a number of major commercial and industrial centers located within the densely populated urban areas of Milwaukee, Racine, and Kenosha Counties has declined, while travel to major centers located in outlying urban areas has increased significantly (see Maps 60 and 61).⁵ For instance, in 1972, the Kenosha and Racine CBD commercial centers attracted over seven times the number of trips attracted by the

⁵The travel data presented refer to total travel attracted to the commercial and industrial centers on an average weekday, with the areal extent of the centers as defined in Appendix E of SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin</u> 2010, January 1992.

Map 60

CHANGE IN NUMBER OF TRIPS ATTRACTED TO MAJOR COMMERCIAL CENTERS IN THE REGION: 1972 AND 1991



As the above maps show, travel to some of the major commercial centers located in the densely populated urban areas of Milwaukee, Racine, and Kenosha Counties has declined, while travel to major centers in outlying urban areas has increased significantly. In 1972, for instance, the Kenosha and Racine CBD centers attracted over seven times the number of trips attracted by the outlying Kenosha West and Regency Mall areas. By 1991, however, the two outlying areas attracted 6,400 more trips than the former CBD locations. The Regency Mall, Kenosha West, Northridge, Park Place, Pewaukee, and Mequon commercial centers each attracted over 100 percent more trips in 1991 than in 1972.

Source: SEWRPC.

outlying Kenosha West and Regency Mall areas. By 1991, however, the two outlying areas combined to attract 6,400 more trips than the former CBD locations. The Regency Mall, Kenosha West, Northridge, Park Place, Pewaukee, and Mequon commercial centers and the Waukesha North, Pewaukee, Pleasant Prairie, and Granville industrial centers attracted over 100 percent more trips in 1991 than in 1972. The largest commercial center trip attractor is the Milwaukee CBD center, with 220,700 daily trips, followed by the Blue Mound Road center, with 135,200 daily trips. The largest industrial center trip attractor is the West Allis West center, with 97,100 daily trips, followed by the Cudahy/South Milwaukee center, with 87,400 daily trips.

Trip Patterns

<u>County-to-County</u>: Map 62 and Table 81 show the magnitude of intra- and inter-county travel within the Region, excluding school trips, on an average weekday in 1963, 1972, and 1991. Several



CHANGE IN NUMBER OF TRIPS ATTRACTED TO MAJOR INDUSTRIAL CENTERS IN THE REGION: 1972 AND 1991



As the above maps show, travel to some of the major industrial centers located in the densely populated urban areas of Milwaukee, Racine, and Kenosha Counties has declined, while travel to major centers in outlying urban areas has increased significantly. The Waukesha North, Pewaukee, Pleasant Prairie, and Granville industrial centers each attracted over 100 percent more trips in 1991 than in 1972.

Source: SEWRPC.

important conclusions can be drawn from these data. First, travel internal to counties dominates total travel within the Region. However, there has been a small shift over time away from intracounty travel toward increased inter-county travel. In 1963, 91 percent of trips, excluding school trips, were intra-county, that is, they had both origin and destination within the same county, while 9 percent of trips were inter-county. In 1972, 87 percent of trips were intra-county, while 13 percent were inter-county. In 1991, 85 percent of the trips were intra-county, while 15 percent were inter-county. Second, the proportion of travel internal to the three urbanized counties, Kenosha, Milwaukee, and Racine, to total regional travel has decreased. As shown in Table 81, travel internal to the urbanized counties has decreased from 80 percent of all travel in 1963, to 70 percent in 1972, and to 60 percent in 1991. Third, the number of trips to and within Milwaukee County decreased from 65 percent of all trips in 1963 to 57 percent in 1972 and to

Map 62

AVERAGE WEEKDAY PERSON TRIPS BETWEEN COUNTIES IN THE REGION: 1963, 1972, AND 1991





The accompanying maps show the magnitude of inter-county travel, excluding school trips, made within the Region in 1963, 1972, and 1991. The changes that have occurred in the traditional suburb-to-central city commuting pattern is evident here. Travel to Milwaukee County from Waukesha County has increased by 20 percent between 1972 and 1991 and by 110 percent between 1963 and 1991, while travel in the reverse direction, to Waukesha County from Milwaukee County, has increased by 65 percent between 1972 and 1991 and 205 percent between 1963 and 1991.

A further indication of these changes in travel patterns is the fraction of total regional travel each county has generated over time. The number of trips made to and within Milwaukee County has decreased from 65 percent of all trips in 1963 to 57 percent in 1972 and to 52 percent in 1991. Meanwhile, the number of trips attracted to, and within, Waukesha County has increased from 10 percent of all trips in 1963 to 15 percent in 1972 and to 19 percent in 1991.

The proportion of travel generated in each of the other five counties of the Region has remained relatively stable. The number of trips made to, and within, Ozaukee, Walworth, and Washington Counties has remained near or below 5 percent of all trips in the Region. The number of trips attracted to, and made within, Kenosha and Racine Counties has remained at about 8 percent or less and 11 percent, respectively.

Source: SEWRPC.



⁹ TRIPS ARE BASED ON THE RESIDENT HOUSEHOLD SURVEYS AND INCLUDE TRIPS FOR ALL PURPOSES EXCEPT SCHOOL.

1991



 TRIPS ARE BASED ON THE RESIDENT HOUSEHOLD SURVEYS AND INCLUDE TRIPS FOR ALL PURPOSES EXCEPT SCHOOL.

AVERAGE WEEKDAY PERSON TRIPS, EXCLUDING SCHOOL TRIPS, BETWEEN, AND WITHIN, COUNTIES IN THE REGION: 1963, 1972, AND 1991

	Attraction County: 1963											
Production County	Kenosha	Milwaukee	Ozaukee	Racine	Walworth	Washington	Waukesha	Total				
Kenosha	240,900	1,700	0	14,200	1,100	0	100	258,000				
Milwaukee	2,300	2,054,900	11,300	8,700	1,400	4,400	60,200	2,143,200				
Ozaukee	0	20,500	42,700	0	0	1,000	600	64,800				
Racine	15,000	12,600	0	332,800	1,700	0	1,300	363,400				
Walworth	1,200	2,800	0	2,600	58,300	200	1,600	66,700				
Washington	300	8,100	2,200	0	200	51,700	5,500	68,000				
Waukesha	300	97,500	900	1,300	1,100	2,600	221,700	325,400				
Total	260,000	2,198,100	57,100	359,600	63,800	59,900	291,000	3,289,500				

		Attraction County: 1972												
Production County	Kenosha	Milwaukee	Ozaukee	Racine	Walworth	Washington	Waukesha	Total						
Kenosha	309,600	2,300	100	16,500	2,700	0	100	331,300						
Milwaukee	2,700	2,166,000	14,300	9,200	1,400	5,700	110,900	2,310,200						
Ozaukee	0	30,800	89,700	100	0	3,300	2,400	126,300						
Racine	18,900	20,800	0	388,100	3,100	100	2,900	433,900						
Walworth	800	2,200	0	5,000	117,900	0	2,700	128,600						
Washington	100	13,800	5,200	200	0	101,300	12,700	133,300						
Waukesha	100	170,900	2,200	1,800	2,600	4,600	440,900	623,100						
Total	332,200	2,406,800	111,500	420,900	127,700	115,000	572,600	4,086,700						

	Attraction County: 1991												
Production County	Kenosha	Milwaukee	Ozaukee	Racine	Walworth	Washington	Waukesha	Total					
Kenosha	297,400	6,100	100	33,100	1,500	0	700	338,900					
Milwaukee	3,300	2,256,900	28,800	15,700	2,600	12,500	183,500	2,503,300					
Ozaukee	200	53,000	147,200	200	100	5,300	5,500	211,500					
Racine	23,300	40,500	500	445,300	4,900	300	8,000	522,800					
Walworth	3,800	5,000	100	7,300	160,900	0	10,700	187,800					
Washington	100	33,000	9,800	300	100	190,000	30,000	263,300					
Waukesha	1,100	205,100	3,600	3,900	3,600	12,400	709,900	939,600					
Total	329,200	2,599,600	190,100	505,800	173,700	220,500	948,300	4,967,200					

Source: SEWRPC.

50 percent in 1991. Fourth, the number of trips to and within Waukesha County has increased from 10 percent of all trips in 1963, to 15 percent in 1972, and to 19 percent in 1991.

Travel between Subareas

The growth in regional internal person travel has occurred primarily in the suburban and exurban areas of the Region. Suburban areas are defined as developed urban portions of the Region located adjacent to, and surrounding, the Cities of Milwaukee, Racine, and Kenosha; exurban areas are defined as outlying portions of Ozaukee, Racine, Walworth, Washington, and Waukesha Counties. Map 63 and Table 82 show the magnitude of tripmaking excluding trips made for the purpose for attending school between the Region's major subareas in 1963, 1972, and 1991.

While average weekday travel, excluding school trips, with both trip origin and destination within the central Milwaukee urban area remained relatively stable at between 1.8 million and 1.9 million person trips from 1963 to 1991, such travel within the suburban areas of the Region increased substantially, from 277,700 person trips in 1963 to 547,300 person trips in 1972 and to 922,000 person trips in 1991. Average weekday travel with both trip origin and destination within the exurban area increased from 86,900 person trips in 1963 to 172,100

Map 63

AVERAGE WEEKDAY PERSON TRIPS BETWEEN SUBAREAS IN THE REGION: 1963, 1972, AND 1991



⁴ TRIPS ARE BASED ON THE RESIDENT HOUSEHOLD SURVEYS AND INCLUDE TRIPS FOR ALL PURPOSES EXCEPT SCHOOL.

For the purpose of analyzing trends in Regional tripmaking, the Region may be divided into subareas corresponding generally to the intensity of existing urban development. The three accompanying maps show the magnitude of tripmaking between the major subareas of the Region in 1963, 1972, and 1991. The growth in regional tripmaking has occurred primarily within the suburban and exurban subareas. Indeed, since 1972, the base year of the currently adopted transportation plan, 95 percent of the increase of 880,000 within the Region were attracted or produced within the suburban and exurban areas. During the same period of time, two-thirds of both the regional household and employment growth occurred within those areas. These trends were clearly foreseen and commented upon in the Commission's first-generation regional transportation system plan, adopted in 1966.

Source: SEWRPC.



TRIPS ARE BASED ON THE RESIDENT HOUSEHOLD SURVEYS AND INCLUDE TRIPS FOR ALL PURPOSES EXCEPT SCHOOL.







AVERAGE WEEKDAY PERSON TRIPS, EXCLUDING SCHOOL TRIPS, BETWEEN, AND WITHIN, SUBAREAS IN THE REGION: 1963, 1972, AND 1991

				Attr	action Zone: 1	963			
Production Zone	Central Milwaukee	Suburban	Exurban	Racine and Environs	Western Racine County	Kenosha and Environs	Western Kenosha County	Walworth County	Total
Central Milwaukee	1,883,400	106,500	8,300	3,700	3,000	1,800	400	1,300	2,008,400
Suburban	184,100	277,700	10,600	1,900	1,400	300	300	700	477,000
Exurban	14,000	14,300	86,900	100	100	100	0	600	116,100
Racine and Environs	5,300	1,100	100	302,600	2,900	12,400	300	400	325,100
Western Racine County	5,800	1,500	100	3,700	23,600	1,300	700	1,300	38,000
Kenosha and Environs	1,400	100	100	10,900	400	225,100	2,200	500	240,700
Western Kenosha County	300	0	0	400	2,500	4,400	9,200	600	17,400
Walworth County	2,700	1,400	600	300	2,300	600	600	58,300	66,800
Total	2,097,000	402,600	106,700	323,600	36,200	246,000	13,700	63,700	3,289,500

		Attraction Zone: 1972											
Production Zone	Central Milwaukee	Suburban	Exurban	Racine and Environs	Western Racine County	Kenosha and Environs	Western Kenosha County	Walworth County	Total				
Central Milwaukee	1,890,400	175,800	8,900	5,100	2,500	2,100	400	1,100	2,086,300				
Suburban	308,500	547,300	18,300	2,200	1,200	200	100	2,400	880,200				
Exurban	18,600	34,700	172,100	200	100	100	0	500	226,300				
Racine and Environs	10,800	1,400	0	330,900	3,400	15,500	300	500	362,800				
Western Racine County	8,400	3,100	200	6,800	46,900	1,400	1,600	2,700	71,100				
Kenosha and Environs	1,500	200	0	11,300	600	284,800	1,700	500	300,600				
Western Kenosha County	600	100	100	600	4,000	4,900	18,200	2,200	30,700				
Walworth County	2,100	2,600	300	400	4,600	500	300	117,900	128,700				
Total	2,240,900	765,200	199,900	357,500	63,300	309,500	22,600	127,800	4,086,700				

		Attraction Zone: 1991											
Production Zone	Central Milwaukee	Suburban	Exurban	Racine and Environs	Western Racine County	Kenosha and Environs	Western Kenosha County	Walworth County	Total				
Central Milwaukee Suburban Exurban Racine and Environs Western Racine County Kenosha and Environs Western Kenosha County Walworth County	1,792,300 364,100 34,600 17,900 12,500 4,700 800 4,700	319,300 922,000 94,100 10,600 7,400 900 600 10,000	11,100 36,000 312,700 700 400 0 0 1,100	8,100 5,900 300 358,700 11,600 24,900 2,200 1,000	3,300 2,300 200 6,500 68,500 900 5,100 6,300	2,400 900 100 16,200 2,000 245,600 11,400 900	300 800 1,900 3,300 7,600 32,800 2,900	2,200 2,900 1,300 700 4,200 600 900 160,900	2,139,000 1,334,900 443,400 413,200 109,900 285,200 53,800 187,800				
Total	2,231,600	1,364,900	362,000	412,700	93,100	279,500	49,700	173,700	4,967,200				

Source: SEWRPC.

person trips in 1972 and to 312,700 person trips in 1991. Between 1963 and 1991, the 644,300-trip increase in suburban travel represented about 38 percent of the total internal person trip growth of approximately 1.68 million, excluding school trips. The 225,800-trip increase in exurban travel represented about 13 percent of the growth in internal person travel, excluding school trips, within the Region. These changes were clearly foreseen and commented upon in the Commission's first-generation regional transportation system plan, completed in 1966. These findings are also consistent with the fact that nearly 66 percent of all growth in regional households and 68 percent of all growth in regional employment between 1963 and 1991 occurred in the suburban and exurban areas.

Average weekday travel from the Milwaukee central urban area to the suburban areas increased substantially as well, from 106,500 person trips in 1963 to 175,800 person trips in 1972 and to 319,300 person trips in 1991, or by about 200 percent between 1963 and 1991. This increase in reverse commuting represented about 13 percent of the growth in regional internal

		Auto	Driver ^a	Auto Pa	Auto Passenger ^a		Auto Person Trips			Truck Driver		Total Vehicle Trips		
Direction	Year	Number	Percent Change	Number	Percent Change	Number	Percent Change	Percent of All Directions	Number	Percent Change	Number	Percent Change	Percent of all Directions	
Inbound	1963 1972 1991	39,700 46,700 107,300	 17.6 129.8	47,900 33,500 40,300	-30.1 20.3	87,600 80,200 147,600	-8.4 84.0	45.7 45.3 46.5	7,100 10,900 19,300	53.5 77.1	46,800 57,600 126,600	23.1 119.8	46.1 45.8 46.3	
Outbound	1963 1972 1991	40,000 47,500 111,900	 18.8 135.6	48,100 35,900 41,800	25.4 16.4	88,100 83,400 153,700	 -5.3 84.3	46.0 47.2 48.4	7,200 10,700 19,800	48.6 85.0	47,200 58,200 131,700	23.3 126.3	46.4 46.3 48.2	
Through	1963 1972 1991	5,900 6,600 10,000	 11.9 51.5	10,100 6,700 6,100	 -33.7 -9.0	16,000 13,300 16,100	-16.9 21.1	8.3 7.5 5.1	1,700 3,300 5,000	94.1 51.5	7,600 9,900 15,000	30.3 51.5	7.5 7.9 5.5	
All Directions	1963 1972 1991	85,600 100,800 229,200	 17.8 127.4	106,100 76,100 88,200	 -28.3 15.9	191,700 176,900 317,400	 -7.7 79.4	100.0 100.0 100.0	16,000 24,900 44,100	55.6 77.1	101,600 125,700 273,300	23.7 117.4	100.0 100.0 100.0	

AVERAGE WEEKDAY EXTERNAL PERSON AND VEHICLE TRIPS IN THE REGION BY DIRECTION: 1963, 1972, AND 1991

^aIncludes personal-use trucks.

Source: SEWRPC.

person travel, excluding school trips, from 1963 to 1991. It is important to note that the growth in reverse commuting between 1963 and 1991 greatly exceeded the growth in the more traditional suburb-to-central city travel. During this time, travel from suburban areas to the Milwaukee central urban area increased by 98 percent, from 184,100 person trips in 1963 to 364,100 person trips in 1991.

External Trip Production

In addition to the 5.59 million internal person. trips and 4.62 million internal vehicle trips made within the Region on an average weekday in 1991, there were 317,400 auto person trips and 273,300 total vehicle trips entering, leaving, or passing through the Region. In each of the survey years, as indicated in Table 83, the numbers of external auto person and vehicle trips entering the Region were very similar to the respective numbers of such trips leaving the Region, ranging from 45 to 48 percent in the case of external auto person trips and from 46 to 48 percent in the case of external vehicle trips. External vehicle trips which passed through the Region, which remained at about 8 percent of all external vehicle trips between 1963 and 1972, decreased to about 6 percent of total external vehicle trips in 1991.

Although the extent of external travel by direction was similar for each of the survey years, comparisons of the ratio of external vehicle travel to external person travel and the corresponding mode use indicate significant changes in external travel. The most significant change was a decline in vehicle occupancy. External person trips decreased from 191,700 in 1963 to 176,900 in 1972, a decrease of 8 percent. External auto person trips then increased from 176,900 in 1972 to 317,400 in 1991, an increase of 79 percent. External vehicle trips, however, increased from 101,600 in 1963 to 125,700 in 1972, or by 24 percent, and from 125,700 in 1972 to 273,300 in 1991, an increase of 117 percent. Increases from 1972 to 1991 by mode are: auto driver and personal-use truck trips, 128,400, or 127 percent; truck driver trips, 19,200, or 77 percent. Auto and personal-use truck passenger trips, which totaled 106,100 in 1963, declined to 76,100 in 1972, or by 28 percent, and increased to 88,200 in 1991, or by 16 percent between 1972 and 1991.

Of the total external person trips, which consist of auto and personal-use truck driver and auto and personal-use truck passenger trips, driver trips accounted for 45 percent in 1963, 57 percent in 1972, and 72 percent in 1991. Of the total

AVERAGE WEEKDAY EXTERNAL AUTOMOBILE AND PERSONAL-USE TRUCK DRIVER TRIPS AND VEHICLE OCCUPANCY IN THE REGION BY TRIP PURPOSE: 1963, 1972, AND 1991

	1963							
	Auto Dri	ver Trips	Total Pers					
Trip Purpose	Number	Percent of Total	Number	Percent of Total	Vehicle Occupancy			
Home-Based WorkHome-Based ShoppingHome-Based OtherNonhome-BasedSchool	24,600 5,200 45,000 9,400 1,400	28.7 6.1 52.6 11.0 1.6	36,900 12,300 121,600 18,200 2,700	19.2 6.4 63.5 9.5 1.4	1.50 2.37 2.70 1.94 1.93			
Total	85,600	100.0	191,700	100.0	2.24			

	1972							
	Auto Dri	ver Trips	Total Per	son Trips				
Trip Purpose	Number	Percent of Total	Number	Percent of Total	Vehicle Occupancy			
Home-Based Work Home-Based Shopping Home-Based Other Nonhome-Based School	36,700 7,200 41,000 12,300 3,600	36.4 7.1 40.7 12.2 3.6	49,400 15,100 87,900 18,700 5,800	27.9 8.5 49.7 10.6 3.3	1.35 2.10 2.14 1.52 1.61			
Total	100,800	100.0	176,900	100.0	1.75			

	1991								
	Auto Dri	ver Trips ^a	Total Pers						
Trip Purpose	Number	Percent of Total	Number	Percent of Total	Vehicle Occupancy				
Home-Based Work	112,900	49.3	129,600	40.8	1.15				
Home-Based Shopping	15,700	6.8	26,200	8.2	1.67				
Home-Based Other	59,800	26.1	106,300	33.5	1.78				
Nonhome-Based	33,200	14.5	44,300	14.0	1.33				
School	7,600	3.3	11,000	3.5	1.45				
Total	229,200	100.0	317,400	100.0	1.38				

^aIncludes personal-use trucks.

Source: SEWRPC.

external vehicle trips, auto and personal-use truck driver trips accounted for 84 percent in 1963, 80 percent in 1972, and 84 percent in 1991.

As indicated in Table 84, the vehicle occupancy rate for external travel declined from 1963 to 1972 and from 1972 to 1991 for each trip purpose. Although the 1972 survey showed a decline in total external person trips from 1963, the 1991 data show increases in total person trips for each trip purpose. Table 84 also shows the increase in home-based work trips which cross the boundaries of the Region, reflecting both residents of the Region commuting to jobs outside the Region and nonresidents of the Region commuting to jobs within the Region.

	1963		19	1972		91	Percent Change	Percent Change	
Trip Purpose	Number	Percent	Number	Percent	Number	Percent	1963-1972	1972-1991	
Base of Operations Work-Connected Business Pick-Up/Delivery of Goods Customer Service	4,200 700 10,100 300	27.5 4.6 66.0 2.0	9,400 3,200 9,800 100	41.8 14.2 43.6 0.4	17,600 9,700 15,300 1,500	39.9 22.0 34.7 3.4	123.8 357.1 -3.0 -66.7	87.2 203.1 56.1 1,400.0	
Total	15,300	100.0	22,500	100.0	44,100	100.0	47.1	96.0	

DISTRIBUTION OF AVERAGE WEEKDAY EXTERNAL COMMERCIAL-USE TRUCK TRIPS IN THE REGION BY DESTINATION AND PURPOSE: 1963, 1972, AND 1991

Source: SEWRPC.

The volume of external commercial truck trips, as shown in Table 85, increased from 15,300 trips per day in 1963 to 22,500 trips per day in 1972, an increase of 47 percent. From 1972 to 1991, such trips increased from 22,500 trips per day to 44,100 trips per day, an increase of 96 percent. Trips made for the purpose of picking up and/or delivering goods, which averaged approximately 10,100 trips per day in 1963 and 9,800 trips per day in 1972, increased to 15,300 trips per day in 1991, an increase of about 5,500 trips per day, or 56 percent, since 1972. Trips made for work-connected business purposes increased from 3,200 trips per day in 1972 to 9,700 in 1991, an increase of 6,500 trips, or 203 percent. Trips made to return to bases of operations increased from 9,400 trips per day in 1972 to 17,600 trips per day in 1991, an increase of 8,200 trips, or about 87 percent. The increases in work-connected business trips plus the increase in trips related to pickup and/or delivery of goods, combined with the increase between 1972 and 1991 in commercial truck trips for the purpose of customer service, are indicative of an economy that is expanding beyond the borders of the Southeastern Wisconsin Region.

SPECIAL SURVEYS

In addition to the major elements of the regional travel inventory, a set of special surveys was conducted as part of this inventory. These special surveys included: a mass transit user survey, a personal opinion survey, and origin-destination surveys on intercity and interregional transportation services not recently surveyed.

Mass Transit User Survey

The Commission conducted special surveys of mass transit passengers on the transit systems

operated by the City of Kenosha Transit Commission, Milwaukee County Transit System, City of Racine Belle Urban System, City of Waukesha Transit System Utility—Waukesha Metro Transit, and Waukesha County. Transit passengers on the express transit route sponsored in 1991 by the City of Racine between the Cities of Milwaukee, Racine, and Kenosha were also surveyed. The principal purpose of these surveys was to obtain accurate descriptions of the socioeconomic and travel characteristics of the ridership of the overall regional mass transit system.

In conducting this survey, a prepaid, preaddressed, mail-back questionnaire was distributed on a sample basis to riders on each mass transit system during the survey period. The number of questionnaires returned and the percentage of the average weekday ridership for each system was: on the Kenosha system, 800 questionnaires returned, or 17 percent of 4,800 average weekday riders, on the Milwaukee system, 8,700 questionnaires, or 4 percent of the 232,700 average weekday riders; on the Racine system, 1,600 questionnaires, or 17 percent of 9,200 average weekday riders; on the Waukesha system, 600 questionnaires, or 24 percent of 2,500 average weekday riders; on the Milwaukee-Racine-Kenosha system, 100 questionnaires, or 37 percent of 270 average weekday riders; and on the Waukesha-Milwaukee system, 400 questionnaires, or 82 percent of 490 average weekday riders.

To permit the sample data to be expanded to represent total weekday ridership on each of the six systems, the number of passengers boarding each bus was recorded and an expansion factor was then determined by dividing the number of

PERCENTAGE DISTRIBUTION OF AVERAGE WEEKDAY BUS PASSENGER TRIPS IN THE REGION BY TRIP PURPOSE AND TRANSIT SYSTEM: 1972 AND 1991

		Transit System									
		Milwaukee			Racine			Waukesha			
Trip Purpose	1972	1991	Difference in Percent	1972	1991	Difference in Percent	1972	1991	Difference in Percent		
Home-Based Work Home-Based Shopping Home-Based Other Nonhome-Based School	56.8 6.5 12.5 4.7 19.5	26.4 9.6 17.3 7.0 39.7	-30.4 3.1 4.8 2.3 20.2	42.2 11.2 19.9 3.9 22.8	25.0 8.6 23.3 10.6 32.5	-17.2 -2.6 3.4 6.7 9.7	35.5 10.3 13.1 0.0 41.1	29.0 5.7 10.0 4.4 50.9	-6.5 -4.6 -3.1 4.4 9.8		
Total	100.0	100.0		100.0	100.0		100.0	100.0			

	Transit System								
	Kenosha				Wat	ukesha-Milwauk	see		
Trip Purpose	1972 ^b	1991	Difference in Percent	1991	1972	1991	Difference in Percent		
Home-Based Work	26.5	16.4	-10.1	53.8	72.0	71.2	-0.8		
Home-Based Shopping	12.3	8.0	-4.3	1.4	12.2	6.9	-5.3		
Home-Based Other	19.8	13.7	-6.1	14.0	4.1	4.8	0.7		
Nonhome-Based	3.7	5.1	1.4	9.1	4.6	5.4	0.8		
School	37.7	56.8	19.1	21.7	7.1	11.7	4.6		
Total	100.0	100.0		100.0	100.0	100.0			

^aService not provided in 1972.

^bExcludes school "trippers," or bus runs designed to accommodate school-aged children.

Source: SEWRPC.

boarding passengers by the number of completed questionnaires. An adjustment was made for passengers who transferred.

As Table 86 shows, home-based work and school trips constituted the major portion of bus passenger travel on each of the transit systems in 1972 and 1991. As a percentage of total trips, homebased work trips decreased on each system surveyed between 1972 and 1991. Taken together, home-based shopping and other nonwork and nonschool related home-based trips increased on the Milwaukee and Racine systems and decreased on the other systems during that time.

Table 87 presents the distribution of bus passenger travel on the 1972 and 1991 transit systems as reported in the surveys by sex, age, annual household income, and race. For comparison purposes, Table 88 presents 1972 and 1991 percentage distributions of bus passengers by income, using 1991 dollars. These tables indicate that:

- Female passengers made the majority of trips on all systems in both years. However, from 1972 to 1991, the proportion of male passengers increased on all systems existing in 1972, except the Waukesha-Milwaukee system.
- As in 1972, the largest portion of 1991 bus passenger trips consisted of passengers 16 through 24 years of age on the Milwaukee, Racine, and Kenosha systems. On the Waukesha system, the largest portion of bus passenger trips in 1972 was made by passengers 16 through 24 years of age. In 1991 the majority of passengers was one through 15 years of age. On the Waukesha-

PERCENTAGE DISTRIBUTION OF AVERAGE WEEKDAY BUS PASSENGER TRAVEL IN THE REGION BY TRANSIT SYSTEM AND SELECTED CHARACTERISTICS OF TRANSIT USERS: 1972 AND 1991

		Percent o	f Trips by Transit Syste	m: 1972				
Selected Characteristics	Milwaukee	Racine	Waukesha ^a	Kenosha ^a	Waukesha- Milwaukee			
Sex								
Male	27.7 72.3	20.1 79.9	7.7 92.3	29.1 70.9	38.2 61.8			
Total	100.0	100.0	100.0	100.0	100.0			
Age					· · · ·			
One to 15	7.2	10.5	4.0	11.2				
16 to 24	31.8	29.8	43.0	35.7	20.6			
25 to 34	12.4	8.9	2.9	4.9	11.2			
35 to 44	10.4	6.9	10.0	3.4	17.4			
45 to 54	15.9	15.7	18.1	13.4	19.2			
55 to 64	15.0	14.6	11.7	11.3	24.1			
65 or Older	7.3	13.6	10.3	20.1	^{7.5} 7.5			
Total	100.0	100.0	100.0	100.0	100.0			
Household Income	· · · ·			-				
(actual dollars)	· · · · · ·							
Under 4,000	20.3	26.8	7.6	25.4	6.6			
4,000 to 7,999	29.7	27.9	21.0	28.6	18.3			
8,000 to 11,999	26.9	23.3	42.3	19.9	26.8			
12,000 to 14,999	12.3	11.0	12.9	14.6	19.9			
15,000 to 19,999	7.0	7.3	15.2	6.3	21.6			
20,000 to 24,999	2.4	2.2		2.5	3.9			
25,000 or Over	1.4	1.5	1.0	2.7	2.9			
Total	100.0	100.0	100.0	100.0	100.0			
Race					·			
Black	12.3	8.8	· • • ·	2.4				
White	85.3	87.6	93.5	96.0	97.5			
Other	2.4	3.6	6.5	1.6	2.5			
Total	100.0	100.0	100.0	100.0	100.0			

		Percent of Trips by Transit System: 1991							
Selected Characteristics	Milwaukee	Racine	Waukesha	Kenosha	Waukesha- Milwaukee	Milwaukee- Racine-Kenosha ^b			
Sex Male Female	38.1 61.9	38.2 61.8	43.3 56.7	39.3 60.7	37.3 62.7	46.9 53.1			
Total	100.0	100.0	100.0	100.0	100.0	100.0			
Age One to 15 16 to 24 25 to 34 35 to 44 45 to 54 55 to 64 65 or Older	4.9 31.3 24.1 18.6 9.5 6.3 5.3	15.4 35.9 21.2 12.4 5.5 3.9 5.7	32.4 27.3 11.7 11.5 5.1 6.2 5.8	25.4 33.9 14.1 5.9 6.3 5.0 9.4	3.0 20.7 22.5 21.3 18.5 11.8 2.2	17.1 35.5 30.7 6.9 4.4 5.4			
Total	100.0	100.0	100.0	100.0	100.0	100.0			
Household Income (actual dollars) Under 5,000	10.7 14.1 13.3 12.2 11.8 8.9 7.6 6.0 6.5 8.9	22.2 18.5 14.6 7.4 9.2 6.0 6.4 4.6 4.2 6.9	15.7 12.7 12.5 13.2 6.9 5.4 7.0 4.3 6.5 15.8	15.7 17.0 15.5 7.4 8.0 5.3 5.4 6.1 6.6 13.0	2.6 3.9 6.3 5.5 7.6 9.5 8.1 7.3 14.3 34.9	10.9 10.3 12.7 15.8 9.3 13.4 8.9 7.8 10.9			
Iotal Race Black White Other Total	30.6 63.3 6.1	39.9 49.7 10.4	100.0 4.5 84.1 11.4	100.0 13.7 77.5 8.8	4.2 90.3 5.5	100.0 18.8 68.8 12.4			

^aExcludes school "trippers," or bus runs designed to accommodate school-aged children.

^bService not provided in 1972.

Source: SEWRPC.

PERCENTAGE DISTRIBUTION OF AVERAGE WEEKDAY BUS PASSENGER TRAVEL IN THE REGION BY TRANSIT SYSTEM BY 1991 CONSTANT DOLLARS: 1972 AND 1991

	Percent of Trips by Transit System										
Household Income (constant	Milwa	aukee	Rac	ine	< Wauk	kesha	Kend	osha	Milwaukee- Racine- Kenosha ^a	Wauk Milwa	esha- aukee
1991 dollars)	1972	1991	1972	1991	1972	1991	1972	1991	1991	1972	1991
Under 14,000	20.3	35.4	26.8	52.4	7.6	38.2	25.4	45.0	18.8	6.6	11.6
27,900 to 41,899	29.7	32.0 18.5	27.9	23.2 14.2	42.3	25.9 14.9	28.8 19.9	15.2	27.5	26.8	20.2
41,900 to 52,299 52,300 or Over	12.3 10.8	5.7 8.4	11.0 11.0	3.7 6.5	12.9 16.2	6.0 15.0	14.6 11.5	5.7 12.4	6.9 10.0	20.9 27.4	13.3 33.1
Total	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

^aService not provided in 1972.

Source: SEWRPC.

Milwaukee system, the largest portion of bus passenger trips in 1972 was made by passengers 55 through 64 years of age; in 1991 by passengers 25 through 34 years of age. No comparison was possible for the Milwaukee-Racine-Kenosha system, which did not provide service in 1972.

- The 1972 survey showed that approximately one-half of total bus passenger trips on the Milwaukee, Racine, Waukesha, and Kenosha systems was made by persons under 35 years of age. Generally, the median age of bus passengers in 1991 is lower. On the Kenosha and Racine systems, about one-half of the passenger trips in 1991 were made by persons under 25 years of age. On the Milwaukee system, the percentage of bus passenger trips made by persons under 35 years of age increased from 51 percent in 1972 to 60 percent in 1991. On the Waukesha-Milwaukee system, the percentage of bus passenger trips made by persons under 35 years of age increased from 32 percent in 1972 to 46 percent in 1991. On the Waukesha system, the percentage of bus passenger trips made by persons under 35 years of age rose from about 50 percent in 1972 to about 71 percent in 1991. Again, no comparison was possible for the Milwaukee-Racine-Kenosha system, which did not provide service in 1972.
- On all five systems existing in 1972, the proportion of black and other nonwhite passengers increased from 1972 to 1991.

• The average annual household income of bus passengers, in 1991 dollars, decreased from 1972 to 1991. As shown on Table 88, in 1972, the percentage of bus passenger trips made by persons with an annual household income of less than \$27,900 in 1991 dollars ranged from 25 percent to 55 percent, while in 1991 this percentage ranged from 32 percent to 76 percent. Table 87 presents income ranges for mass transit passengers by transit system in actual dollars for the 1972 and 1991 survey years, respectively.

The total data obtained in this survey are judged to be representative of daily urban and suburban transit tripmaking and of transit tripmakers. The data can be used in comparing present routing with potential routes desired by transit users for consideration in the provision of improved transit service as an element of the short-range transit development of the regional modal split model for the simulation of future travel conditions, and in calibrating transit system networks preparatory to alternative system plan development.

Personal Opinion Survey

As part of the 1991 home interview survey, a special survey was conducted to obtain the opinions, preferences, and attitudes of heads of households or spouses regarding certain aspects of transportation and housing. The salient findings of this survey are provided in the following sections. Where possible, these find-

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING PERCEIVED TRAFFIC CONGESTION IN THE REGION

Personal Opinion Concerning Severity of Traffic Congestion	Percent
Not a Problem	4.6
A Minor Problem during Peak-Traffic Periods	44.9
A Severe Problem during Peak-Traffic Periods	40.4
A Severe Problem during Peak-Traffic Periods and	
during Other Parts of the Day	8.8
No Response	1.3
Total	100.0

Source: SEWRPC.

Table 90

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING SEVERITY OF TRAFFIC CONGESTION DURING THE PAST 10 YEARS

Personal Opinion Concerning Change in Level of Traffic Congestion	Percent
Has Become Substantially Worse in the Last 10 Years	40.4
Has Become Somewhat Worse in the Last 10 Years	44.1
Is About the Same as 10 Years Ago	8.2
Has Become Less of a Problem over the Last 10 Years	2.5
There Is No Traffic Congestion Problem	2.7
No Reponse	2.1
Total	100.0

Source: SEWRPC.

ings are compared with the findings of a similar personal opinion survey conducted as part of the 1972 home interview survey.⁶

The personal opinion surveys conducted by the Commission directly questioned respondents about their attitude, not behavior, on certain travel- and housing-related issues. Preferences were expressed without regard to the practicality of satisfying those preferences in the face of economic and other realities. The 1991 survey reflects the attitudes of the more than 1,700 households responding to the survey.

<u>Traffic Congestion Severity</u>: Item A of the questionnaire asked each respondent to indicate his or her perception of the severity of traffic congestion in the Region. About 45 percent of the respondents indicated that traffic congestion was a minor problem during peak traffic periods, while 40 percent indicated that traffic congestion was a severe problem during peak traffic periods. An additional 9 percent of respondents indicated that traffic congestion was a severe problem not only during peak traffic periods but during other parts of the day (see Table 89).

Traffic Congestion Change in Last 10 Years: Item B of the questionnaire asked each respondent to indicate his or her perception of the change in traffic congestion over the last 10 years. Almost 85 percent of the respondents indicated that traffic congestion had become worse, 44 percent indicating that traffic congestion had become somewhat worse and 40 percent indicating that traffic congestion had become substantially worse. About 3 percent indicated that traffic congestion had moderated over the last 10 years, while 8 percent indicated that traffic congestion had remained unchanged (see Table 90).

Location of Worst Traffic Congestion: Item C of the questionnaire asked each respondent to identify in rank order the facilities in the Region exhibiting the worst traffic congestion problems. The percentage distributions of responding households are shown by facility type in Table 91. The facilities ranked worst by nearly 75 percent of the respondents were the Milwaukee County freeways. About 39 percent of the respondents ranked Milwaukee County streets the second worst.

⁶For a more detailed presentation of the findings of the 1972 survey, see SEWRPC Technical Report No. 13, <u>A Survey of Public Opinion in</u> Southeastern Wisconsin—1972, September 1974.

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING SEVERITY OF TRAFFIC CONGESTION ON FREEWAYS AND STREETS IN MILWAUKEE COUNTY AND SURROUNDING COUNTIES

		Resp							
Facilities	Worst	Second- Worst	Third- Worst	Fourth- Worst	Fifth- Worst	No Response	Total		
Milwaukee County Freeways	74.6	10.8	2.9	2.3	0.3	9.1	100.0		
Surrounding County Freeways	3.1	28.7	34.9	12.8	0.8	19.7	100.0		
Milwaukee County Streets	10.4	39.4	24.5	5.9	0.7	19.1	100.0		
Surrounding County Streets	3.4	5.5	13.3	53.2	1.3	23.3	100.0		
Other	2.2	0.7	1.2	0.5	5.6	89.8	100.0		

Source: SEWRPC.

Table 92

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING PERCEIVED EFFECTS OF TRAFFIC CONGESTION

	Responses (percent)							
Effects	Worst	Second- Worst	Third- Worst	Fourth- Worst	Fifth- Worst	Sixth- Worst	No Response	Total
Fuel Consumption/Air Pollution	31.9	15.1	11.4	12.8	17.0	0.6	11.2	100.0
Increased Traffic Accidents	20.7	22.1	15.5	16.4	12.8	0.4	12.1	100.0
Increased Travel Time	13.9	20.8	22.8	20.4	9.4	0.1	12.6	100.0
Unreliable Travel Time	11.3	16.4	20.8	18.0	18.9	0.6	14.0	100.0
Irritation of Stop-and-Go Driving	17.0	16.0	15.0	15.5	23.1	0.4	13.0	100.0
Other	1.3	0.4	0.3	0.1	0.2	3.2	94.5	100.0

Source: SEWRPC.

Worst Thing about Traffic Congestion: Item D of the questionnaire asked each respondent to identify in rank order the worst things about traffic congestion. About 32 percent of the respondents ranked increased motor fuel consumption and air pollutant emissions the worst. The second- through fifth-ranked responses to this question were not as clearly defined. Increased traffic accidents were ranked as second worst by 22 percent and worst by 21 percent. Increased travel time was ranked as third worst by 23 percent, second worst by 21 percent, and fourth worst by 20 percent. Unreliable travel time was ranked as third worst by 21 percent, fifth worst by 19 percent, and fourth worst by 18 percent. The irritation of stopand-go driving was ranked as fifth worst by 23 percent and worst by 17 percent. Although 32 percent of the respondents indicated that increased fuel costs and air pollutant emissions were the worst impacts of traffic congestion, the various other factors were also ranked in varying degrees of importance (see Table 92).

Actions to Alleviate Traffic Congestion: Item E of the questionnaire asked each respondent to indicate agreement or disagreement with various potential actions that could be taken to alleviate traffic congestion. The actions and the percentage of respondents indicating support for each action were: increased use of carpooling and public transit, 81 percent; extension of bus

Possible Actions	Yes	No	No Response	Total
Increase Use of Carpooling and Public Transit	81.2	12.5	6.3	100.0
Extend Bus Service to New Areas	77.0	17.9	5.1	100.0
Provide More Express and Freeway Flyer Bus Service	72.6	20.6	6.8	100.0
Provide Convenient Bike Lanes and Paths	56.3	36.0	7.7	100.0
Build Light Rail and Commuter Rail Transit	54.4	38.3	7.3	100.0
Increase Frequency of Bus Service	54.0	39.5	6.5	100.0
Reduce Bus Fare	49.0	43.2	7.8	100.0
Widen Existing and Build New Arterial Streets	44.6	46.9	8.5	100.0
Widen Existing Freeways	42.9	49.5	7.6	100.0
Build New Freeways	25.5	66.5	8.0	100.0
No Action Is Necessary	9.7	49.5	40.8	100.0
Increase the Cost of Operating an Automobile	6.8	86.3	6.9	100.0
Other	6.0	1.6	92.4	100.0

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING SUPPORT FOR POSSIBLE ACTIONS TO ALLEVIATE TRAFFIC CONGESTION IN THE REGION

Source: SEWRPC.

service to new areas, 77 percent; provision of more express and freeway flyer bus service, 73 percent; provision of convenient bike lanes and paths, 56 percent; building light-rail and commuter rail transit, 54 percent; and increasing frequency of bus service, 54 percent. The two actions respondents opposed the most included building of new freeways, 67 percent, and increasing the cost of operating an automobile, 86 percent (see Table 93).

Travel Reduction to Meet Federal Clean Air Act <u>Requirements</u>: Item F, Part 1, of the questionnaire asked the respondents to indicate agreement or disagreement with various actions that could be taken to reduce automobile travel, particularly work-related travel, to meet the requirements of the Federal Clean Air Act. The action approved most frequently, by 82 percent of the respondents, was to improve public transit to encourage more transit use, including provision of more available, faster, and more frequent bus transit. The action opposed most frequently, by 74 percent of the respondents, was the elimination of free employee parking to encourage more carpooling and transit use. All other specific actions were favored by a majority of the respondents, ranging from 55 to 62 percent (see Table 94).

Factors Needed to Influence Carpool and Transit Use: Item F, Part 2, of the questionnaire asked respondents to indicate agreement or disagreement with factors that would need to change before they would carpool or use transit if they currently drive alone to and from work. Of the six factors listed, as shown in Table 95, three were cited most frequently. Specifically, faster and more frequent public bus transit was cited by 60 percent; faster and more frequent public transit, including light-rail and commuter rail, by 50 percent; and carpool incentives, such as exclusive carpool freeway lanes and priority parking, by 50 percent. Only a small percentage. 16 and 14 percent, respectively, of respondents indicated that elimination of free workplace parking or substantially increased automobile costs would encourage them to consider alternatives to driving alone to work. Respondents may have reacted to these two proposed actions, not with respect to potential to change their travel

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF SUPPORT FOR POSSIBLE ACTIONS TO REDUCE AUTOMOBILE TRAVEL TO AND FROM WORK

			No	
Possible Actions	Yes	No	Response	Total
Improve Public Transit to Encourage More Transit Use, Including Provision of More Available, Faster, and More Frequent Bus Transit	81.8	12.8	5.4	100.0
Provide Incentives to Carpool and Transit Users, Such as Exclusive Carpool and Transit Lanes on Streets and Freeways	62.4	29.7	7.9	100.0
Encourage Employers to Provide Transit Subsidies to Promote More Transit Use	61.6	29.6	8.8	100.0
Improve Public Transit with Light Rail and Commuter Rail	57.6	34.9	7.5	100.0
Encourage Employers to Offer Four-Day or Three-Day Work Weeks	56.9	35.3	7.8	100.0
Provide Convenient Bike Lanes and Paths	56.5	34.6	8.9	100.0
Encourage Employers to Arrange Programs to Permit Employees to Work at Home	54.9	36.4	8.7	100.0
Eliminate Free Employee Parking to Encourage More Carpooling and Transit Use	18.1	74.2	7.7	100.0
Other	3.8	1.1	95.1	100.0

Source: SEWRPC.

Table 95

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF FACTORS CONSIDERED NECESSARY BEFORE CHOOSING TO CARPOOL OR USE TRANSIT RATHER THAN DRIVING ALONE

Factors	Yes	No	No Response	Total
Faster and More Frequent Public Bus Transit	60.1	25.0	14.9	100.0
Faster and More Frequent Public Transit, Including Light Rail and Commuter Rail	50.2	34.7	15.1	100.0
Carpool Incentives, Such as Exclusive Carpool Freeway Lanes and Priority Parking	50.1	33.1	16.8	100.0
Convenient Bike Lanes and Paths	40.0	41.6	18.4	100.0
Elimination of Free Parking at Your Workplace with Possible Charges of \$20 to \$30 per Month	16.0	67.4	16.6	100.0
Substantial Increases in the Cost of Operating an Auto, Such as Increased Gasoline Prices	14.4	70.5	15.1	100.0

Source: SEWRPC.

behavior, but, rather, with respect to their disapproval of such increases in the cost of operating an automobile.

Funding Sources for County and Local Arterials: Item G of the questionnaire asked respondents to indicate their choice for funding the cost of operating, maintaining, and improving county and local arterials. A 31 percent plurality favored a combination of sources, including a user fee of 6 cents per gallon of gasoline, a user fee of an additional \$40 per vehicle annually, and an increase in the sales tax of 0.25 percent. Approximately 23 percent stated that the cost should continue to be covered by the property tax, at an estimated rate of \$0.80 per \$1,000 of assessed value (see Table 96).

Limitations Believed to Result from Lack of Public Transportation: Item H of the questionnaire asked respondents to indicate whether the lack of adequate public transportation between their homes and certain areas of the Region prevented or severely limited household members from accepting employment, reaching shopping and recreational areas of their choice, conducting necessary personal business, or visiting friends or relatives. This was the only transportation-related question that was similar to a 1972 personal opinion question. Both the 1972 and 1991 data are shown in Table 97. For all categories of possible limitation, the percentages of respondents indicating a belief that a lack of public transportation limited personal options were lower in 1991 than in 1972, indicating that fewer respondents believed that members of their households were limited in their activities due to a lack of public transportation. These decreases may be due in part to the increase in personal vehicles available to residents of the Region.

Special Origin-Destination Surveys on Intercity and Interregional Transportation Services

As part of the 1991 regional inventory of travel, special origin-destination surveys were conducted in the spring of 1993 on all intercity and interregional public passenger transportation services that operated regularly scheduled weekday service within the Region and that had not been recently surveyed. These transportation services included the Metra commuter rail service operated between Kenosha and Chicago and intercity motor coach services operated by four carriers: Badger Coaches, Inc., Greyhound

Table 96

THE 1991 PERSONAL OPINION SURVEY: PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING CHOICES FOR POSSIBLE FUNDING SOURCES FOR COUNTY AND LOCAL ARTERIALS

	Funding Sources	Percent
Α.	Remain on the Property Tax (an estimated \$0.80 per \$1,000 of assessed value, or \$56 per \$70,000 house)	22.5
В.	Be Funded Instead by a User Fee, Such as Increased Motor Fuel Taxes (an estimated \$0.06 per gallon)	14.5
C.	Be Funded Instead by a User Fee, Such as Increased Annual Vehicle Registration Fees (an estimated additional \$40 per vehicle)	™
D.	Be Funded Instead by Increased Sales Tax (an estimated additional 0.25 percent)	9.1
E.	Some Combination of Sources B, C, and D	31.1
F.	Other	8.6
G.	No Response	5.4
	Total	100.0

Source: SEWRPC.

Lines, Inc., Lamers Bus Lines, Inc., and United Limo, Inc. Amtrak intercity passenger train service was surveyed in the spring of 1991.⁷ These survey data were combined with the highway travel surveys to provide a complete inventory of interregional surface travel by mode. In addition, a survey of enplaning passengers at General Mitchell International Airport, the sole scheduled-air-carrier airport serving Southeastern Wisconsin, was conducted in the fall of 1989.⁸

⁷See SEWRPC Memorandum Report No. 57, <u>Amtrak Milwaukee-Chicago Passenger Survey</u> Findings: June 1991, January 1993.

⁸See SEWRPC Technical Report No. 32, <u>General</u> <u>Mitchell International Airport Enplaning Pas</u>senger Survey Findings: 1989, August 1990.

	1972			1991				
Activities	Yes	No	No Response	Total	Yes	No	No Response	Total
Accepting Employment	11.3	78.1	10.6	100.0	9.7	84.5	5.8	100.0
Reaching Recreational Areas	12.0	76.6	11.4	100.0	8.4	84.8	6.8	100.0
Reaching Shopping Areas of Your Choice	15.8	73.7	10.5	100.0	12.2	82.4	5.4	100.0
Conducting Necessary Personal Business	8.3	80.2	11.5	100.0	7.1	86.8	6.1	100.0
Making Social Visits to Friends or Relatives	10.6	78.1	11.3	100.0	8.6	85.2	6.2	100.0

PERCENTAGE DISTRIBUTION OF RESPONSES REGARDING ACTIVITIES BELIEVED TO BE LIMITED BY A LACK OF PUBLIC TRANSPORTATION: 1972 AND 1991

Source: SEWRPC.

In each of the surveys, each boarding and through passenger was asked to complete and return a questionnaire. The questions that were used typically requested and elicited delineations of such socio-economic characteristics of the passenger as age, occupation, home address, household annual income, and auto availability; and to characteristics of the trip being made, such as trip origin and destination, trip purpose, mode of travel used to reach the boarding location, and frequency with which the particular trip was made. Depending upon the survey, other information was also obtained, such as whether the trip was part of a round trip and why the tripmaker chose to make this trip on the travel mode concerned.

The only commuter railway service operated in the Region in 1993 was Metra's Chicago & North Western Railway North Line between Kenosha and Chicago, with intermediate stops in the north shore suburbs of northeastern Illinois. Metra is the commuter railway service division of the Regional Transportation Authority, which serves northeastern Illinois. Service on this particular commuter railway route was provided by the Chicago & North Western Transportation Company under contract with Metra. On weekdays in 1993, this service consisted of nine commuter trains in each direction between Kenosha and Chicago. In 1972, weekday commuter railway service in Southeastern Wisconsin consisted of 24 trains, including nine trains in each direction between Kenosha and Chicago, as in 1993; two trains in each direction between the City of Lake Geneva and Chicago; and one train in each direction between the Village of Walworth and Chicago.

In the spring of 1993, scheduled intercity motor coach services were provided by four carriers: Badger Coaches, Inc.; Greyhound Lines, Inc.; Lamers Bus Lines, Inc.; and United Limo, Inc. Service provided on weekdays by Badger Coaches included six runs in each direction between Madison, downtown Milwaukee, and General Mitchell International Airport. Service provided by Greyhound in Southeastern Wisconsin is centered in Milwaukee, which the carrier uses as a regional hub at which passengers may transfer between buses. In 1993, Greyhound operated a total of 39 weekday bus runs in both directions between Milwaukee and Chicago; Milwaukee and Green Bay, continuing on to Calumet, Michigan; Milwaukee and Appleton, continuing on to Stevens Point and Rhinelander; and Milwaukee and Madison, continuing on to Minneapolis-St. Paul. Some of these bus runs made only limited stops and some made local stops. Also, some of these runs operate through Milwaukee and some originate or terminate in Milwaukee. Service provided on weekdays by Lamers Bus Lines included one run in each direction between Milwaukee, Appleton, and Wausau. Service provided on weekdays by United Limo included nine runs in each direction between downtown Milwaukee, General Mitchell International Airport, and Chicago O'Hare International Airport. Together, the four intercity motor coach carriers operated a combined total of 71 bus runs, all of which were surveyed.

In 1972, there were six intercity motor coach carriers providing service through the operation of 144 weekday bus runs in the Region. Of these 144 weekday runs, 96 were operated by Greyhound to Chicago, to various locations in Wisconsin and Upper Michigan, and to cities as far away as Seattle; 12 runs were operated by Wisconsin Coach Lines, Inc., between Milwaukee and Fond du Lac, Watertown, and Rockford, Illinois; 12 runs were operated by Badger Coaches between Milwaukee and Madison; 12 runs were operated by Tri-State Coach Lines, Inc., between Milwaukee and Chicago's O'Hare International Airport; eight runs were operated by Wisconsin-Michigan Coach Lines, Inc., between Milwaukee and Green Bay, Sister Bay, and Marshfield; and four runs were operated by Peoria-Rockford Bus Company between Milwaukee, Rockford, and Dixon, Illinois.

In 1993, intercity railway passenger service in Southeastern Wisconsin was provided by Amtrak over CP Rail System trackage, formerly Chicago, Milwaukee, St. Paul & Pacific Railroad Company trackage, with stops within the Region at Milwaukee and Sturtevant. When the most recent survey by the Commission of this service was conducted in the spring of 1991, Amtrak operated five weekday trains in each direction between Milwaukee and Chicago and one weekday train in each direction between Chicago, Milwaukee, St. Paul-Minneapolis, and Seattle.

In 1972, by comparison, Amtrak operated three weekday trains in each direction between Milwaukee and Chicago, two weekday trains in each direction between Milwaukee, Chicago, and St. Louis, and two weekday trains in each direction between Chicago, Milwaukee, St. Paul-Minneapolis, and Seattle.

In 1972, cross-lake car ferry service on Lake Michigan was operated by the Chesapeake & Ohio Railway Company between Milwaukee and Ludington, Michigan. This service, which carried passengers, automobiles, and railway freight cars and scheduled two weekday departures from each port during the summer season, was discontinued in 1984.

Scheduled air-carrier service to and from Milwaukee County's General Mitchell International Airport was provided by 16 airline companies in the fall of 1989 when the most recent survey of air-carrier service was conducted by the Commission. These airline companies included: American Airlines, Comair, Continental Airlines, Delta Air Lines, Eastern Airlines, Enterprise Airlines, Express Airlines, Midway Airlines, Midway Commuter, Midwest Express Airlines, Mesaba Airlines, Northwest Airlines, Skyway Airlines, Trans World Airlines, United Airlines, and USAir. In 1989, these carriers provided a total of 364 scheduled nonstop weekday flights between Mitchell International and 33 other cities or metropolitan areas. Cities with 10 or more nonstop weekday flights to or from Milwaukee included: Chicago, Detroit, Grand Rapids, Atlanta, Cincinnati, New York, Minneapolis-St. Paul, Appleton, Cleveland, Indianapolis, St. Louis, Madison, Green Bay, Boston, Columbus, and Wausau. Many of these flights continued on to other cities.

By comparison, in 1971, when a previous enplaning passenger survey was undertaken by the Commission, there were six airline companies providing a total of 250 scheduled nonstop weekday flights between Milwaukee and 32 other cities or metropolitan areas. These airline companies included: Air Michigan, Eastern Airlines, North Central Airlines, Northwest Airlines, Ozark Airlines, and United Airlines. Cities with 10 or more nonstop weekday flights to or from Milwaukee included: Chicago, New York, Madison, Minneapolis-St. Paul, Detroit, and Oshkosh. Between the 1971 and 1989 enplaning passenger surveys, airline companies have made dramatic changes to their route structures, resulting in most of the scheduled service from Milwaukee being concentrated in major, heavily traveled markets and to cities that serve as system hubs for the airline companies.

Quantity of Interregional Passenger Travel: The 1991 survey findings indicated that on an average weekday, 3,088 passengers were being carried on interregional surface public transportation modes, including Amtrak intercity railway passenger trains, Metra commuter railway trains, and intercity motor coaches serving Southeastern Wisconsin. In addition, 12,614 passengers were being carried on an average weekday on scheduled air-carrier flights between Milwaukee and other cities. Accordingly, a total of 15,702 interregional passenger trips were being made in the Region on public transportation on an average weekday, as shown in Table 98. Also, as shown in Table 98, there were 317,400 interregional automobile person trips made on an average weekday, accounting for 95 percent of all interregional public transportation and automobile trips on an average week-

NUMBER OF INTERREGIONAL PASSENGER TRIPS ON INTERCITY MODES IN THE REGION: 1972 AND 1993

	1972			1993			
Mode	Number	Percent without Auto	Percent with Auto	Number	Percent without Auto	Percent with Auto	
Intercity Motor Bus	1,297	14.3	0.7	1,341	8.5	0.4	
Metra Commuter Rail	267	2.9	0.2	748	4.8	0.2	
Amtrak Intercity Rail	560	6.2	0.3	999 ^a	6.4	0.3	
Chesapeake & Ohio							
Cross-Lake Car Ferry	749	8.2	0.4				
Air Carriers	6,225 ^b	68.4	3.3	12,614 ^C	80.3	3.8	
Private Automobile	176,900		95.1	317,400		95.3	
Total	185,998	100.0	100.0	333,102	100.0	100.0	

^aSurvey taken in 1991.

^bSurvey taken in 1971.

^CSurvey taken in 1989.

Source: SEWRPC.

day. Of the remaining interregional trips, 1,341 passengers were carried on the 71 intercity motor coach runs, an average of about 19 passengers per run; 748 passengers were carried on the 18 Metra commuter rail trains, an average of about 42 passengers per train; 999 passengers were carried on the 12 Amtrak intercity railway passenger trains, an average of about 83 passengers per train; and 12,614 passengers were carried on the 364 scheduled air-carrier flights, an average of about 35 passengers per flight. The numbers and percentages of passengers returning completed questionnaires in the surveys were: for intercity motor coach services, 883, or 66 percent; for Metra commuter rail trains, 426, or 57 percent; for Amtrak intercity rail trains, 878, or 88 percent; and for the scheduled air carriers, an average of 2,710 enplaning passengers per day over a five-day survey period. When this figure is doubled to account for both enplaning and deplaning passengers, it results in a projected total of 5,420, or 43 percent.

By comparison, it was estimated that on an average weekday in 1972, a total of 2,873 passengers were being carried on interregional surface transportation modes, including Amtrak intercity passenger trains, commuter railway trains, intercity motor coaches, and car ferries in Southeastern Wisconsin. Including the 6,225 passengers that were carried on scheduled aircarrier flights between Milwaukee and other cities, a total of 9,098 interregional passenger trips were made on public transportation in the Region on an average weekday in 1972. There were also 176,900 interregional automobile person trips, accounting for 95 percent of all interregional public transportation and automobile trips on an average weekday in 1972. Of the remaining interregional trips, 1,297 passengers were carried on intercity motor coach runs, 267 passengers were carried on commuter railway trains, 560 passengers were carried on Amtrak intercity railway passenger trains, 749 were carried on cross-lake car ferries; and 6,225 passengers were carried on scheduled air-carrier flights.

<u>Round Trips</u>: According to the most recent surveys, the proportion of interregional passenger trips made as a part of a round trip ranged from 60 percent on intercity motor coaches to 95 percent on scheduled air carriers, as shown on Table 99. This compares to the 1972 survey findings, where the percentages of passengers making round trips ranged from 84 percent on intercity motor coaches to 97 percent on commuter rail trains. Round trips by air cannot be determined from the 1971 air-carrier survey. It should be noted that although 35 percent of

		Carrier Type							
	. Am	trak	Motor Bus		Commuter Rail		Air Carrier		
Making Round Trips	1972	1991 ^a	1972	1993	1972	1993	1971 ^b	1989	
Yes	91.6	65.3	84.2	59.6	96.9	87.7	·	94.8	
No	8.4	34.7	15.8	40.4	3.1	12.3		5.2	
Total	100.0	100.0	100.0	100.0	100.0	100.0		100.0	

PERCENTAGE DISTRIBUTION OF ROUND TRIPS IN THE REGION ON AN AVERAGE WEEKDAY BY CARRIER TYPE: 1972, 1989, 1991, AND 1993

^aQuestion in 1991 concerning whether the survey trip was part of a round trip asked whether the trip was part of round trip on day surveyed.

^bRound-trip data were not collected during the 1971 survey.

Source: SEWRPC.

Amtrak passengers in 1991 indicated that their trips were not part of any round trip, the question on the Amtrak survey specifically asked if the trip was part of a round trip being taken on the same day. It is believed that for many of these Amtrak passengers, the surveyed trip was indeed part of a round trip being completed on another day. In part, this is due to long-distance travelers who accounted for 16 percent of all passengers, or nearly one-half of the total Amtrak passengers, who indicated that the surveyed trip was a one-way trip in the Milwaukee-Chicago corridor.

<u>Frequency of Trip</u>: In response to a question about the frequency with which the trip was made, a high degree of repetitiveness was found only in commuter railway travel, in which 56 percent of the passengers indicated that the trip was usually made 20 or more times per month (see Table 100). Amtrak, air-carrier, and intercity motor coach passengers are generally infrequent travelers, as about 60 to 80 percent of such passengers indicated that they travel by their respective modes less than once per month. The frequency of travel among scheduled aircarrier passengers is shown in Table 101.

<u>Trip Purpose by Carrier Type</u>: The distribution of travel by trip purpose reflects some of the differences in the characteristics of travel by the four current public interregional modes, as shown in Tables 102 and 103. For example, there was a greater diversity of trip purposes for motor coach and Amtrak railway passengers than for commuter railway and scheduled air-carrier passengers in each of the survey years. The majority of motor coach passengers, 63 percent, made trips for reasons other than work, school, or shopping, while the majority of passengers using commuter railway service, 73 percent, made work-related trips. On Amtrak railway service, 41 percent of the trips were found to be work-related; on the scheduled air carriers, 54 percent of the trips were work-related.

Locations of Origins and Destinations: The geographic distribution of the trip origins and destinations of interregional passengers is presented in Table 104. In those cases where the passenger did not furnish an ultimate origin or destination, the boarding or deboarding location is used. The distribution of passenger trip origins and destinations, combined with trip purposes, emphasizes the nature of the carrier as either commuter-oriented or as a long-distance carrier. Not surprisingly, about 99 percent of commuter rail passenger trip interchanges are between the Southeastern Wisconsin Region and the counties of northeastern Illinois, predominantly between Kenosha County and Cook County. This pattern of travel is also found on the Amtrak railway service, where at least 85 percent of the trip interchanges are between the Southeastern Wisconsin Region and the counties of northeastern Illinois, predominantly

· · · · · · · · · · · · · · · · · · ·		·	Carrie	r Туре		- · · ·
	Am	Amtrak Motor Bus			Commuter Rail	
Frequency of Travel	1972	1991	1972	1993	1972	1993
Less than One Trip per Month	65.5	59.0	59.9	66.8	9.5	19.4
One to Four Trips per Month	13.5	16.5	27.2	20.4	4.6	10.1
Five to Nine Trips per Month	10.0	6.5	9.3	6.6	6.6	4.2
10 to 19 Trips per Month	2.7	4.6	2.4	3,7	12.5	10.5
20 or More Trips per Month	8.3	13.4	1.2	2.5	66.8	55.8
Total	100.0	100.0	100.0	100.0	100.0	100.0

PERCENTAGE DISTRIBUTION OF FREQUENCY OF INTERREGIONAL TRIPMAKING BY SURFACE CARRIER TYPE: 1972, 1991, AND 1993

Source: SEWRPC.

between Milwaukee County and Cook County. Among the intercity surface modes, only on motor coaches is there a wide geographic diversification of trip origins and destinations, reflecting the use of motor bus service by passengers as a provider of long-distance service for trip purposes other than work. Because the scheduled air-carrier surveys conducted at Mitchell International included only enplaning passengers and not deplaning passengers, the data show air-carrier trip origins being heavily concentrated within Southeastern Wisconsin and air-carrier trip destinations entirely outside the Region.

Mode of Travel before Boarding the Carrier: The modes of travel used to reach the train, motor bus, or airplane also emphasize the differences in travel characteristics among the various modes, as shown in Table 105. The most frequently used mode used to reach the boarding locations of all carriers in each of the survey years was the private auto, which accounted for 50 to 67 percent of the access modes used. By carrier type, the next most frequently used modes were: Amtrak, taxi, walking, or another train, which accounted for 17 percent, 16 percent, and 9 percent, respectively; intercity motor coaches, walking, local bus, taxi, or intercity motor coach, which accounted for 10 percent, 10 percent, 9 percent, and 8 percent, respectively; commuter rail, walking and taxi, which accounted for 28 percent and 8 percent, respec-

Table 101

AIR TRAVEL FREQUENCY OF ENPLANING PASSENGERS USING GENERAL MITCHELL INTERNATIONAL AIRPORT: 1989

Number of Departures Made Annually at Mitchell International	Percent
One or Fewer	34.5
Two to Six	39.2
Seven to 12	11.1
13 to 25	8.0
26 to 50	5.5
More than 50	1.7
Total	100.0

Source: SEWRPC.

tively; and air carriers, other aircraft and rental car, which accounted for 12 percent and 11 percent, respectively. All other modes accounted for less than 5 percent of travel to the boarding location for each carrier type.

<u>Selected Socio-Economic Characteristics</u>: Selected socio-economic characteristics of interregional motor coach, railway train, and air passengers are shown in Table 106. Passengers using Amtrak had a median age of 38 and a median annual household income of \$55,000. A majority classified their employment as either executive-

	Carrier Type							
	Amtrak		Amtrak Motor B		Amtrak Motor Bus		Commu	uter Rail
Trip Purpose	1972	1991	1972	1993	1972	1993		
Home-Based Work	19.5	40.5	17.9	11.8	70.0	73.1		
Home-Based Shopping	1.6	2.2	2.8	0.2	4.1	1.3		
Home-Based Other	45.1	31.2	40.0	62.7	11.9	13.5		
Nonhome-Based	28.6	24.5	23.4	15.6	9.6	7.4		
School	5.2	1.6	15.9	9.7	4.4	4.7		
Total	100.0	100.0	100.0	100.0	100.0	100.0		

PERCENTAGE DISTRIBUTION OF AVERAGE WEEKDAY BUS AND RAIL PASSENGER TRIPS BY TRIP PURPOSE: 1972, 1991, AND 1993

Source: SEWRPC.

Table 103

OVERALL TRAVEL PURPOSE OF ENPLANING PASSENGERS USING GENERAL MITCHELL INTERNATIONAL AIRPORT: 1871 AND 1989

	Percent of Passengers Citing Trip Purpose		
Trip Purpose	1971	1989	
Work or Work-Related Business Personal Business School Social/Recreational/Vacation Other	50.7 6.7 2.3 38.5 1.8	54.1 8.7 1.6 32.4 3.2	
Total	100.0	100.0	

Source: SEWRPC.

managerial or professional-technical and had an educational level of four years of college or better. Intercity motor coach passengers had a median age of 32 and a median annual household income of \$22,000. Their principal occupations were reported to be student. craftsmanoperator-laborer, professional-technical, or retired. The majority had a high school education or had attended some college. Passengers on commuter railway trains had a median age of 35 and a median annual household income of \$46,000. The majority worked either in professional-technical, executive-managerial, or clerical-administrative support occupations and had at least four years of college. Air-carrier passengers had a median age of 41 and a

median annual household income of \$54,000 in 1989 dollars. No occupational or educational data were collected in the 1989 enplaning passenger survey. Data on automobile availability were not collected in the 1991 Amtrak survey or the 1989 enplaning passenger survey. However, 8 percent of passengers using commuter rail and 20 percent of passengers using motor coach came from households with no automobiles available.

SUMMARY AND CONCLUSIONS

The Commission travel surveys conducted for 1963, 1972, and 1991 clearly demonstrate that travel is an orderly, regular, and measurable occurrence, evidenced by recognizable travel patterns. Recognition of those aspects and patterns of travel which demonstrate a high degree of repetitiveness is a prerequisite to foreseeing future travel behavior and, consequently, to sound transportation system plan-The Commission's comprehensive ning. inventories of travel describe in detail each of the component parts of the total travel pattern of the Region. Knowledge of each of these parts is essential to an understanding of the total travel pattern within the Region, and, consequently, to the conceptual processes involved in establishing generalized norms of travel behavior.

This chapter has presented, in summary form, the basic findings of the 1991 Commission inventory of travel within the Region. In order to assess the changes occurring in travel habits

PERCENTAGE DISTRIBUTION OF LOCATION OF TRIP ORIGINS AND DESTINATIONS FOR INTERREGIONAL RAIL, BUS, AND AIR PASSENGERS: 1989, 1991, AND 1993

	Carrier Type							
	Amtrak: 1991		Motor Bus: 1993		Commuter Rail: 1993		Air Carrier: 1989	
State and County	Origins	Destinations	Origins	Destinations	Origins	Destinations	Origins	Destinations
Southeastern Wisconsin								
Kenosha	0.4	1.0	1.4	1.0	43.5	31.5	1.6	
Milwaukee	36.5	28.9	28.4	21.6	1.6	1.2	43.2	
Ozaukee	2.9	2.8	0.9	0.5	0.0	0.0	3.9	
Racine	4.0	3.0	1.6	2.3	10.2	9.2	4.6	
Walworth	0.1	0.3	0.0	0.1	0.3	0.8	1.9	
Washington	0.8	0.4	0.4	0.5	0.0	0.0	2.5	
Waukesha	8.5	8.2	2.5	2.7	0.3	0.3	16.7	
Subtotal	53.2	44.6	35.2	28.7	55.9	43.0	74.4	
Other Wisconsin Counties	3.5	4.1	19.2	26.8	0.0	0.0	14.9	5.1
Northeastern Illinois Counties	34.2	40.9	22.1	17.4	43.8	55.9	2.9	0.7
Other Illinois Counties	1.3	1.0	0.7	0.6	0.3	0.0		
All Other States and Countries	7.8	9.4	22.8	26.5	0.0	1.1	7.8	94.2
Total	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0

^aResponses are only from passengers enplaning at Milwaukee.

^bIncludes Cook, Du Page, Kane, Lake, McHenry, and Will Counties, Illinois.

^CIncludes the six northeastern Illinois counties plus the nine northwestern Illinois counties of Boone, Carroll, DeKalb, Jo Daviess, Lee, Ogle, Stephenson, Whiteside, and Winnebago.

^dIncludes all Illinois counties.

Source: SEWRPC.

Table 105

PERCENTAGE DISTRIBUTION, BY CARRIER TYPE, OF MODE OF TRAVEL USED BY INTERREGIONAL PASSENGERS BEFORE BOARDING: 1971, 1972, 1989, 1989, AND 1993

	Carrier Type								
	Amtrak		Motor Bus		Commuter Rail		Air Carrier ^a		
Mode of Travel	1972	1991	1972	1993	1972	1993	1971	1989	
Private Auto Rental Car Taxi Hotel Courtesy Car Local City Bus Intercity Bus Train "L" or Subway Walking	62.3 b 16.6 c 8.1 e 6.7 9 4.9	50.2 0.7 17.0 0.2 4.1 9.4 1.0 16.3	45.5 b 8.7 - c 22.5 - e 2.2 9 18.5	57.9 0.4 8.9 0.7 9.7 8.1 0.6 10.3	74.7 - b 2.5 - c 1.9 - e 	57.9 0.4 7.5 0.3 4.0 0.4 1.3 27.9	57.0 9.3 8.2 2.1 d 5.7 ^f 	67.2 10.9 3.4 2.6 - d 4.1 ^f	
Other	1.4	1.1	2.6	3.4	0.6	0.3	17.7''	11.8''	
Total	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	

^aResponses are only from passengers enplaning at Milwaukee.

^bIncluded with "Private Auto" in 1972.

^cIncluded with "Other" in 1972.

^dNegligible, but included with "Intercity Bus."

^eIncluded with "Local Bus" in 1972.

^fAlso includes airport limousines.

^gIncluded with "Train" in 1972.

^hIncludes passengers arriving by other scheduled air carriers, air taxis, and private airplanes.

Source: SEWRPC.

PERCENTAGE DISTRIBUTION OF SELECTED SOCIO-ECONOMIC CHARACTERISTICS OF INTERREGIONAL RAIL, BUS, AND AIR PASSENGERS BY CARRIER TYPE: 1989, 1991, AND 1993

	Carrier Type						
	Amtrak	Motor Bus	Commuter Rail	Air Carrier			
Selected Socio-Economic Characteristics	1991	1993	1993	1989			
Age Group							
15 or Younger	4.4	1.8	0.3	1.6			
16 to 24	11.6	28.9	14.2	7.0			
25 to 34	24.2	23.7	33.7	23.1			
35 to 44	29.2	16.8	30.6	26.4			
45 to 54	16.6	11.1	16.6	20.8			
55 to 64	8.8	6.6	2.2	13.0			
65 and Older	5.2	11.1	2.4	8.1			
Total	100.0	100.0	100.0	100.0			
Occupation ^a							
Executive/Managerial	27.4	4.5	23.3				
Clerical/Administrative Support	6.1	7.5	13.5				
Sales/Buyer	10.3	2.7	4.8	,			
Professional/Technical	24.4	12.2	31.6	· ^			
Craftsman/Operator/Laborer	3.8	15.9	7.6				
Self-Employed	5.3	11.2	5.3				
Teacher	3.0	2.7	1.8	·			
Homemaker	3.4	8.5	2.6				
Student	9.6	23.3	8.7				
Retired	5.4	11.5	0.8				
Other	1.3						
Total	100.0	100.0	100.0				
Education ^a				_			
Some High School	3.9	9.7	3.8				
High School or GED	8.9	28.2	14.7				
Some College	18.6	27.9	22.3				
Associate/Technical Degree	8.3	7.3	7.3				
Bachelor's Degree	34.7	13.9	32.9				
Graduate Degree	25.6	13.0	19.0	·			
Total	100.0	100.0	100.0				
Annual Household Income (actual dollars)							
Less than 10,000	5.6	23.2	6.4	3.1			
10,000 to 19,999	6.2	23.9	7.4	5.4			
20,000 to 29,999	11.6	16.0	10.7	10.3			
30,000 to 39,999	10.8	13.1	18.1	12.8			
40,000 to 49,999	11.4	9.5	13.2	14.1			
50,000 to 74,999	22.6	8.1 🦻	26.9	26.3			
75,000 to 99,999	11.7	3.6	12.9	12.0			
100,000 or More	20.1	2.6	4.4	16.0			
Total	100.0	100.0	100.0	100.0			
Number of Automobiles ^{a,b}							
None	- - ¹	19.5	8.0				
One		36.4	34.1				
Two or More		44.1	57.9	1997 - 19			
Total		100.0	100.0				

^aThis information was not collected in the 1989 enplaning passenger survey.

^bThis information was not collected in the 1991 Amtrak survey.

Source: SEWRPC.

and patterns within the Region over time, comparisons have been made between the findings of the 1991 inventory with those of earlier Commission travel inventories of 1963 and 1972. The findings of the travel inventories which bear special significance for regional transportation system plan development include the following:

- On an average weekday in 1991, about 5.9 million person trips were made within the Region. This represents increases from 1972 of 1.2 million person trips, or 26 percent; and from 1963 of 2.1 million person trips, or 56 percent. The increases in regional tripmaking reflect the increases in the number of households within the Region of 26 percent from 1972 to 1991, and of 46 percent from 1963 to 1991, as well as the increases in employment of 31 percent from 1972 to 1991, and of 56 percent from 1963 to 1991. The increases in person trips in the Region were substantially greater than the changes in the resident population of the Region, a 1 percent decrease from 1972 to 1991 and a 9 percent increase from 1963 to 1991. Future levels of households and employment, rather than of resident population, should be considered key indicators of potential future travel growth.
- The number of internal person trips per resident household of the Region has remained remarkably stable, at 7.3 trips per household in 1963, 7.9 in 1972 and 7.8 in 1991. The level of average weekday internal person trips per capita, however, has increased significantly, from 2.2 trips per capita in 1963 to 2.5 in 1972 and to 3.1 in 1991. The stability in the household trip rate is remarkable, considering the substantial socio-economic and land use changes which have occurred within the Region over the last 30 years. These changes include an increase in automobile availability per household from 1.1 in 1963, to 1.2 in 1972, and to 1.6 in 1991; a decline in the cost of operating an automobile with respect to motor fuel, measured in 1991 dollars, from 12.7 cents per mile in 1963 to 9.4 cents per mile in 1972 and to 5.5 cents per mile in 1991; and a reduction in travel time between the various parts of the Region from 1963 to 1991 of between 6 percent and 14 percent. The stability in household trip production over the 30-year period of substantial socio-

economic and land use changes in the Region indicates that continued stability in household trip production may be expected and that it may be unreasonable to expect that changes in land use patterns or site design, travel price, travel accessibility, or population lifestyles will materially affect the number of daily trips made per household within the Region.

- On an average weekday in 1991, nearly 4.9 million vehicle trips were made within the Region. This represents an increase of about 1.5 million vehicle trips, or 44 percent, from 1972 to 1991; and an increase of 2.3 million vehicle trips, or about 90 percent, from 1963 to 1991. The increase in vehicle trips from 1972 to 1991 is more substantial than the increase in person trips, specifically, an increase of 1.5 million vehicle trips and of 1.2 million person trips over the 19-year period. The principal factor contributing to the more rapid increase in vehicle trips is the decline in average vehicle occupancy observed in the surveys, from 1.42 persons per vehicle in 1972 to 1.26 persons per vehicle in 1991 with respect to all trips and, from 1.17 persons per vehicle in 1972 to 1.06 persons per vehicle in 1991 for work trips. Vehicle tripmaking should not increase significantly faster than person tripmaking in the future because vehicle occupancy can no longer experience declines of the magnitude exhibited historically.
- On an average weekday in 1991, 33 million vehicle-miles of travel occurred within the Region as a result of the 4.9 million vehicle trips. The historic increases in vehicle-miles of travel reported in Chapter IV of this report, from 20 million in 1972 to 33 million in 1991, or 65 percent, and from 13 million in 1963 to 33 million in 1991, or 154 percent, have been more rapid than the corresponding historic increases in total person tripmaking and vehicle tripmaking. The principal contributing factor to the more substantial increases in vehicle-miles of travel has been an increase in the average length of internal person trips from 5.3 miles in 1972 to 6.1 miles in 1991, an increase of about 18 percent; and from 4.5 miles in 1963 to 6.1 miles in 1991, an increase of about 35 percent. Thus, the 65 percent increase in

highway traffic and attendant congestion in the Region from 1972 to 1991 has been the result only in part of demographic and economic growth and change and related person tripmaking. Only about 50 percent of the growth in highway traffic over roughly the past 20 years may be attributed to increased tripmaking as a result of demographic and economic growth and change. The remaining 50 percent may be attributed equally to the decline in vehicle occupancy and the increase in trip length.

- Nearly all, about 95 percent, of the person and vehicle trips made within the Region on an average weekday in each survey year were made by residents of the Region. Therefore, the location and capacity of future transportation facilities must accommodate, to the degree possible, the patterns of travel of regional residents.
- The number of personal vehicles, automobiles and personal-use light trucks available to residents of the Region increased from about 527,000 in 1963 to 705,000 in 1972, an increase of 178,000 autos, or 34 percent, and to 1,142,500 in 1991, an increase of 437,500, or 62 percent from 1972 to 1991. The percentage of total households in the Region having two or more personal vehicles available increased from 24 percent in 1963 to 34 in 1972 and to 53 percent in 1991, while the percentage of total households having no personal vehicle available decreased from 18 percent in 1963 to 17 percent in 1972, and to 13 percent in 1991. In addition, while automobile travel increased from about 88 percent of all internal person travel in the Region in 1963 to about 92 percent in both 1972 and 1991, mass transit travel decreased from 9 percent of total internal person travel in 1963 to 4 percent in 1972 and to 3 percent in 1991. These findings indicate that an even greater amount of total person travel than that observed in 1991 may be expected to be made by personal vehicle in future years, barring significant improvement in the regional transit system and changes in public attitudes toward the use of transit and public preferences relating to the type and location of housing.
- While substantial increases in tripmaking were found in all other travel modes on an

average weekday, mass transit travel decreased sharply within the Region, from 324,000 trips in 1963, to 186,000 trips in 1972, and to 178,000 trips in 1991.

- In each of the three survey years, approximately 88 percent of total internal vehicle trips were made by automobile or taxi, and only about 11 to 12 percent were made by truck. The distribution of truck trips did not exhibit the same sharp concentrations in time of tripmaking as did automobile trips and did not coincide with the peak periods of automobile travel, either by hour of the day or day of the week. These findings indicate that, with respect to highway facilities, the transportation problem within the Region is primarily one of the movement of people rather than goods.
- Approximately 80 percent of total internal person trips within the Region on an average weekday in 1963, 1972, and 1991 consisted of trips made to or from places of residence. It is apparent, then, that future travel facility requirements within the Region will be determined largely by the amount and location of future residential development.
- Trips between home and work accounted for 25 percent of all internal person trips made within the Region on an average weekday in 1963, 24 percent in 1972, and 23 percent in 1991. The findings indicate that the location of future employment centers will be another important factor affecting travel demand within the Region. In 1991, school-related travel accounted for about 11 percent of all internal person trips within the Region; trips between home and shopping for about 15 percent; trips between home and other destinations for social, recreational, and personal business purposes for about 31 percent, and trips between nonhome origins and destinations for about 20 percent.
- The increase in tripmaking experienced within the Region over time has been a result of household and employment growth in the suburban and exurban portions of the Region, and of attendant tripmaking. Between 1963 and 1991, the 644,300-trip increase in suburban-area travel represented

about 38 percent of the total internal person trip growth of approximately 1.68 million, excluding school trips. The 225,800-trip increase in exurban-area travel represented about 13 percent of the growth in travel, excluding school trips, within the Region. These changes were clearly foreseen and commented upon in the Commission's firstgeneration regional transportation system plan, completed in 1966. These findings are also consistent with the fact that nearly 66 percent of all growth in regional households and 68 percent of all growth in regional employment between 1963 and 1991 occurred in the suburban and exurban areas of the Region.

It is important to note that travel habits and patterns, such as those reported here, are the direct result of literally millions of individual personal decisions often made on a daily basis. Each individual exercises his or her choice regarding available transportation alternatives, utilizing a wide array of criteria, the relative importance assigned to the criteria being a matter of personal values. The criteria that an individual might utilize in making a transportation decision may include timeliness, trip purpose, cost, convenience, and perceived personal safety. Certain factors involved defy quantification, such as, among others, pleasure in driving, prestige, and family and personal activity levels. Individual choice is influenced significantly, and may be constrained, by the availability, convenience, and cost of alternatives. The latter may be determined, or at least influenced, by public policies.

The recognition that travel patterns and traffic volumes are a result of individual choice and of the factors considered in making these choices is critical to understanding the implications and practicality of any proposal to significantly change travel habits and patterns. In subsequent chapters of this report, transportation system alternatives which would require that the availability, convenience, and cost of alternative travel modes be substantially altered, are proposed and evaluated. To the extent practicable, the implications of these alternatives with respect to the modification of travel behavior will be evaluated by using Commission travel simulation models. (This page intentionally left blank)

Chapter VI

ANTICIPATED REGIONAL GROWTH AND CHANGE, THE ADOPTED REGIONAL LAND USE PLAN, AND ALTERNATIVE FUTURES FOR SOUTHEASTERN WISCONSIN

INTRODUCTION

Change is a basic characteristics of the modern world; urban growth, decay, and renewal are among the most important aspects of this change. No modern nation, state, or region can escape the effects of urban change; no part of daily life can avoid being influenced in some way by forces rooted in this complex process. Since change is inevitable, the question facing public officials and citizen leaders of a region, such as Southeastern Wisconsin, is not whether urban growth, decay, and renewal will occur, but how much may be expected to occur, when, and how well it will be shaped and guided in the public interest. Changes in population size, composition, and distribution; in employment levels and personal income; and in land use are all inevitable. In any planning effort, then, it is necessary to forecast changes in those factors which lie outside the scope of plan design, but which influence such design. This chapter presents the results of attempts to forecast both the magnitude and direction of anticipated changes in these factors and also the alternative land use plans designed to accommodate such changes.

Many methods have been developed for forecasting change in a region such as Southeastern Wisconsin. Some of these methods are quite simple, some are highly complex, but all are ultimately based upon historical experience and, in general, rely on a combination of mathematical formulation and professional judgment.

The principal difference between forecasting methods is generally reflected in the differing emphasis upon these two basic elements. At one extreme, a method may involve little or no mathematical formulation and may depend almost entirely upon the exercise of professional judgment. Because the variables entering into these forecasts are most often not clearly defined, even in the minds of their authors, such forecasts are generally not capable of reduction to a precise procedure which can be expressed mathematically. At the other extreme, a method may depend almost entirely upon mathematical formulation and require little or no exercise of professional judgment. Such forecasts, founded as they are on a precise procedure, may be readily replicated once the rules of the procedure are established. These procedural rules may be called forecasting models; if expressed in mathematical terms, they may be designated as mathematical forecasting models.¹

It is important to understand that the forecasts based upon mathematical forecasting models are not necessarily more accurate than forecasts based largely upon experienced professional judgment. Forecasts based upon models, however, have two great advantages: they require that the underlying assumptions be explicitly stated and they permit the effects of differing underlying assumptions to be determined quantitatively.

It must also be recognized that no one can "predict" the future and that all forecasts, how ever made, involve uncertainty and, therefore, must always be used with great caution. Forecasts cannot take into account events which are unpredictable but may have a major effect upon future conditions. Such events include wars; epidemics; major social, political, and economic upheavals; and radical institutional changes. Moreover, both public and private decisions of a nature less radical than the foregoing can be made which may significantly affect the ultimate accuracy of any forecast. For these reasons, forecasting, like planning, must be a continuing process. As otherwise unforeseeable events

¹A model is a replica or representation of some other object or thing. Models are often used to simulate the operation of complex, real-world systems composed of a large number of interacting elements. A mathematical model uses a set of mathematical relationships (equations) to operate on a set of typical real data. The model is then used to simulate the operation of a real system, such as the flow of automobile traffic in a highway network. unfold, forecast results must be revised and, in turn, plans which are based on such forecasts must be reviewed and revised accordingly.

In more than three decades of long-range physical planning, the Commission has undertaken a number of studies of future population and economic activity. The traditional practice has been to prepare a number of projections of possible future population and economic activity levels, selecting from this range one set of forecast population and economic activity levels believed to be most likely to represent future conditions. The selected forecasts are then utilized in the development, test, and evaluation of alternative plans. This traditional approach was followed in the preparation of the first- and second-generation regional land use and transportation system plans.

The traditional approach to planning works well in periods of socio-economic stability, when historic trends can be anticipated to continue relatively unchanged over the plan design period. However, during periods of major change in social and economic conditions, when there is uncertainty as to whether historic trends will continue, an alternative to this traditional approach may be required. Surveillance activities carried out under the continuing regional planning program point to increasing uncertainty regarding future social and economic conditions in Southeastern Wisconsin. To deal with this uncertainty, the Commission has adopted an "alternative futures" approach to areawide, systems level planning.

The alternative futures approach to systems level planning has provided a broad basis for the design and evaluation of both land use plans and supporting systemwide facility plans such as the regional transportation system plan. It has involved the postulation of high-growth, low-growth, and intermediate-growth alternative future growth scenarios for the Region; the preparation of projections of population and employment for each scenario;² and the design of a recommended land use plan and alternative land use plans for each future growth scenario. Such alternative future plans have served to bracket conceptually the recommended land use plan and have permitted the evaluation of alternative transportation plans under various possible future land use conditions.

The first section of this chapter describes the high-growth, low-growth and intermediategrowth scenarios for Southeastern Wisconsin and sets forth the related projections of population, households, employment, and personal income. This section also presents estimated future land use requirements associated with the projected population and economic activity levels in the Region. The land use requirements associated with the intermediate-growth scenario served as the foundation for the design of the adopted 2010 land use plan.

The second section of this chapter describes the recommended regional land use plan, adopted by the Commission in September 1993. The plan is set forth in greater detail and its recommendations elaborated more fully in SEWRPC Planning Report No. 40, A Regional Land Use Plan for Southeastern Wisconsin: 2010. The adopted regional land use plan incorporates the basic concepts of the year 2000 land use plan and updates and extends that plan to a new design year. Like the year 2000 plan, the new plan recommends a relatively compact and centralized regional settlement pattern, with urban development occurring generally in concentric rings along the full periphery of, and outward from, existing urban centers. The overall scale and distribution of urban development distinguish the adopted plan from the alternative futures plans; the adopted plan may be termed an "intermediate-growth centralized land use plan."

The third section of this chapter summarizes the alternative land use plans considered, including a high-growth decentralized plan, a low-growth decentralized plan, an intermediate-growth decentralized plan, and a high-growth centralized plan.

²Information regarding the possible future population and economic activity levels in the Region through the year 2010 developed by the Commission using the alternative futures approach is presented in SEWRPC Technical Report No. 10 (2nd Edition), <u>The Economy of Southeastern Wisconsin</u>, 1984; and SEWRPC Technical Report No. 11 (2nd Edition), <u>The</u> Population of Southeastern Wisconsin, 1984.

ALTERNATIVE FUTURE GROWTH SCENARIOS

Under the alternative futures approach, three alternative future growth scenarios were postulated for Southeastern Wisconsin. The sets of conditions postulated for each "future" are intended to represent consistent, reasonable scenarios of future change in population and economic activity levels in the Region through the year 2010. Two scenarios, the "high-growth" scenario and the "low-growth" scenario, are intended to represent reasonable extremes, while the third scenario, the "intermediate-growth" scenario, is intended to represent a likely future. The general trends in resident population and economic activity levels envisioned under the respective scenarios are herein described.³

Economic Growth Scenarios

Among the many uncertainties surrounding future economic conditions in the Region, two appeared to be particularly pertinent as the Commission attempted to prepare long-range economic forecasts. The first uncertainty pertains to the relative strength of the manufacturing sector, historically the dominant sector of the regional economy, but one which has experienced a major decline over the past two decades. The second uncertainty pertains to the long-term impacts of the recession of 1979 through 1983, a recession which lasted longer and was more severe within the Region than any recession since the Great Depression of the 1930s. Because that recession did not display a pattern typical of a normal business cycle, it could not be assumed that the ensuing recovery would be typical; thus the possibility of major structural changes in the regional economy could not be dismissed. The alternative economic growth scenarios described herein embody varying assumptions regarding the ability of the regional economy, particularly the manufacturing sector, to recover from the 1979 through 1983 recession and again compete effectively with other regions of the nation and the world.

The economic changes that may be expected under a high-growth scenario represent a return to the type of growth that has historically occurred in the regional economy. Under this scenario, there would be no long-term damage to the regional economy as a result of the 1979 through 1983 recession and the economic recovery of the Region would be strong after 1985, with long-term growth rates recovering to levels at, or slightly below, national averages. This growth would be expected to result from the identification and exploitation of strengths in the regional economy, such as labor availability, a good vocational-technical educational system, land availability, and a high-quality infrastructure of railway, highway, seaport, airport, and sewerage and water systems. In addition, the traditional manufacturing interests that historically comprised the foundation of the regional economy would be successful in reducing production costs and increasing productivity through the application of advanced technologies to traditional manufacturing processes, thereby improving their competitive position. The trade and service sectors would, under this scenario, continue to grow at rapid rates, as they have done over the past several decades.

Under the intermediate-growth scenario, the recovery of the regional economy from the 1979 through 1983 recession would be delayed and would be initially weaker than the national recovery as the heavy industrial and manufacturing concerns that dominate the regional economy continue to close unprofitable plants and limit operations in the streamlining efforts necessary for survival during poor economic conditions. Under this scenario, the changes occurring during this contraction of the manufacturing employment group would ultimately lead to a stronger, though initially smaller, regional manufacturing economy as more efficient factory operations allow employers to expand and modernize existing plants. Employment in retail and wholesale trade would increase as the economy recovers. Employment

³The "high-growth," "intermediate-growth," and "low-growth" scenarios as described in this chapter are equivalent to the "optimistic," "intermediate," and "pessimistic" regional growth scenarios, respectively, described in SEWRPC Technical Report No. 10 (2nd Edition), <u>The Economy of Southeastern Wisconsin</u>, 1984, and SEWRPC Technical Report No. 11 (2nd Edition), <u>The Population of Southeastern Wisconsin</u>, 1984.

in the services sector would also increase, continuing recent trends seen at both the regional and national levels.

The economic conditions that may be expected under a low-growth scenario represent a departure from long-term trends under which the Region was able to maintain or to increase its relative share of national employment. Under the low-growth scenario, the recovery of the regional economy from the 1979 through 1983 recession would be a lengthy one, with regional employment remaining depressed. Over the long term, the Region would experience a continuation or even an acceleration of a trend first observed in the 1970s, when Southeastern Wisconsin began to experience a decline in its share of total national employment. This departure from historic trends is based on an assumed inability of area manufacturers to modernize their aging physical capital stock, resulting in declining productivity and continued erosion of product markets. Increased foreign competition in manufacturing industries and the continued migration of manufacturing employment to other regions of the United States would more than offset any employment gains that could be expected in such possible growth sectors as wholesale trade, retail trade, medical and other professional services, and finance, insurance, and real estate. Under this scenario, the lack of industrial expansion or rejuvenation could be expected to hold total employment levels in the Region at, or slightly below, the 1980 levels through the year 2010.

Population Growth Scenarios

The population of an area is constantly changing as a result of births, deaths, and the migration of persons into, or out of, the area. Area population projections may be prepared through explicit consideration of these components of population change. The process of developing alternative population growth scenarios for Southeastern Wisconsin involved a review of critical social and economic factors that could be expected to impact on mortality, fertility, and migration rates through the year 2010 and the establishment of a reasonable range of values for each component of population change. Each of the three population growth scenarios developed by the Commission reflects a combination of assumed fertility, mortality, and migration rates selected from the range of possible values. The

Figure 44

HISTORICAL AND PROJECTED WHITE TOTAL FERTILITY RATE FOR THE REGION: 1960-2010



Source: U. S. Bureau of the Census and SEWRPC.

assumed combinations of fertility, mortality, and migration rates were ultimately used to project the probable size of the population of the Region under a range of possible future conditions.

Among the population growth scenarios, the highest fertility rates were assumed for the highgrowth scenario and the lowest fertility rates for the low-growth scenario (see Figures 44 and 45). The overall range between the low and high fertility rates was relatively narrow, a joint product of changing societal preferences regarding family size, the increased availability of family planning, and the effectiveness of birth control. The narrow range only slightly diminishes the crucial nature of the assumed rates, however, because the differences in the rates are compounded over the course of several generations. With regard to the other determinant of natural increase in population, mortality rates, it should be noted that the same set of mortality rates was assumed for each population growth scenario.

The rates of migration assumed under the highgrowth, intermediate-growth, and low-growth scenarios were logically linked to anticipated


Source: U. S. Bureau of the Census and SEWRPC.

economic conditions, taking into account the number of jobs, unemployment rates, rates of dual job holding, and labor force participation rates envisioned under the corresponding economic growth scenarios. Significant net inmigration of population was postulated under the high-growth scenario as a result of the assumed strong recovery from the 1979 through 1983 recession and the assumed ability of the regional economy to compete with other areas (see Figure 46). Under the intermediate-growth scenario, it was assumed that the net outmigration experienced in the Region during the 1970s would gradually diminish in response to gradually improving economic conditions and that net in-migration of population into the Region would begin after the year 2000. Under the low-growth scenario, it was assumed that significant out-migration of population would occur over the entire 1980 to 2010 projection period in response to a stagnating regional economy, with the highest rates of out-migration anticipated during the 1980s.

The varying assumptions regarding rates of natural increase and net migration described above lead to significantly different scenarios of

PROJECTED LEVELS OF NET MIGRATION FOR THE REGION UNDER HIGH-GROWTH, INTERMEDIATE-GROWTH, AND LOW-GROWTH FUTURE ECONOMIC CONDITIONS



population change in the Region. A substantial increase, 31 percent, in the regional population between 1980 and 2010 is envisioned under the high-growth scenario, while a relatively modest population increase, 6 percent, is envisioned under the intermediate-growth scenario. A decrease in the regional population of about 14 percent is envisioned under the low-growth scenario.

Accompanying changes in population size would be changes in the number, size, and type of households. Under each growth scenario, the number of households would increase and average household size would decrease, although rates of change in the number of households and average household size would vary considerably for the three scenarios. The high-growth scenario assumes that "traditional" patterns of household composition will exist between 1980 and 2010 and that households consisting of a husband, wife, and children will constitute the dominant type, although the average number of children in the households will be lower than in the past. Under this scenario, the average household size in the Region would decrease from about 2.8 in 1980 to about 2.6 in 2010. The intermediate-growth

POPULATION LEVELS IN THE REGION BY COUNTY: ACTUAL 1980 AND 1990 AND PROJECTED 1990, 2000, AND 2010

							_		
				Population					
	Actual	Actual		Projected	I		Projected Change 1980-2010		
County	1980	1990	Scenario	1990	2000	2010	Number	Percent	
Kenosha	123,100	128,200	High-growth Intermediate-growth Low-growth	135,700 117,300 109,900	152,900 118,000 105,200	166,800 123,300 101,800	43,700 200 -21,300	35.5 0.2 -17.3	
Milwaukee	965,000	959,300	High-growth Intermediate-growth Low-growth	964,900 924,300 861,700	991,900 892,200 831,800	1,009,800 911,300 818,100	44,800 -53,700 -146,900	4.6 -5.6 -15.2	
Ozaukee	67,000	72,800	High-growth Intermediate-growth Low-growth	84,000 69,700 60,500	106,200 75,000 58,800	139,000 81,900 57,700	72,000 14,900 -9,300	107.5 22.2 -13.9	
Racine	173,100	175,100	High-growth Intermediate-growth Low-growth	188,400 165,200 152,900	206,000 166,000 146,800	224,700 171,800 139,600	51,600 -1,300 -33,500	29.8 -0.8 -19.4	
Walworth	71,500	75,000	High-growth Intermediate-growth Low-growth	85,600 78,200 66,900	106,200 85,600 65,000	129,700 89,900 63,700	58,200 18,400 -7,800	81.4 25.7 -10.9	
Washington	84,900	95,300	High-growth Intermediate-growth Low-growth	114,300 97,500 78,000	135,900 109,500 75,800	164,400 116,000 74,400	79,500 31,100 -10,500	93.6 36.6 -12.4	
Waukesha	280,200	304,700	High-growth Intermediate-growth Low-growth	353,800 302,000 267,700	424,800 336,000 264,400	481,700 378,000 261,800	201,500 97,800 -18,400	71.9 34.9 -6.6	
Region	1,764,800	1,810,400	High-growth Intermediate-growth Low-growth	1,926,700 1,754,200 1,597,600	2,123,900 1,782,300 1,547,800	2,316,100 1,872,200 1,517,100	551,300 107,400 -247,700	31.2 6.1 -14.0	

Source: U. S. Bureau of the Census and SEWRPC.

scenario assumes that traditional patterns of household composition will be less dominant and that single-parent families and single-person households will be more prevalent than under the high-growth scenario. The historic increase in single-parent and single-person households would, however, be moderate under this scenario. The average household size in the Region would decrease from about 2.8 in 1980 to about 2.4 in 2010. The low-growth scenario assumes that husband-wife households will continue to decrease as a proportion of total households and that single-parent and single-person households will increase as a proportion of total households. as they have done historically. Under this scenario, the average household size in the Region would decrease from about 2.8 in 1980 to about 2.1 in 2010.

Population Projections

Commission population projections for the Region and the constituent counties under the three regional growth scenarios described above are set forth in Table 107. The projected distribution of the population among the counties in the Region under the three growth scenarios is also shown in Figure 47.

Under the high-growth scenario, the resident population of the Region would increase by about 551,300 persons, or 31 percent, from 1,764,800 persons in 1980 to 2,316,100 persons in 2010. Under this scenario, Waukesha County could be expected to gain about 201,500 persons, an increase of 72 percent over the 30-year projection period. The other six counties in the Region would experience smaller, but neverthe-



POPULATION LEVELS IN THE REGION BY COUNTY: ACTUAL AND PROJECTED: 1950-2010

Source: U. S. Bureau of the Census and SEWRPC.

less significant, population increases, ranging from about 43,700 persons in Kenosha County to about 79,500 persons in Washington County.

Under the intermediate-growth scenario, the regional population would be expected to increase by about 107,400 persons, or about 6 percent, to a level of 1,872,200 in the year 2010. Waukesha County would also experience the largest absolute population increase, 97,800 persons, under this scenario. Ozaukee, Walworth, and Washington Counties would gain 14,900, 18,400, and 31,100 persons, respectively. The population levels of Kenosha and Racine Counties would not change significantly. The population of Milwaukee County would decrease by about 53,700 persons over the 30-year projection period, decreasing from 965,000 persons in

1980 to just under 900,000 persons in the year 2000, then increasing to about 911,300 in the year 2010.

The low-growth scenario envisions a year 2010 population of 1,517,100, a decrease of 247,700 persons, or 14 percent, from the 1980 level. Each county in the Region would experience a population decrease under this scenario, ranging from about 7,800 persons in Walworth County to about 146,900 persons in Milwaukee County. The substantial loss of population in Milwaukee County would account for 59 percent of the overall loss projected for the Region under this scenario.

It should be noted that since 1980 the actual population level of the Region has most closely approximated the level anticipated under the intermediate-growth scenario (see Figure 48). Recently released 1990 U.S. Census data indicate that the resident population of the Region stood at 1,810,400 persons in 1990, 3 percent, or 56,200 persons, greater than the level of 1,754,200 anticipated under the intermediategrowth scenario. The 1990 resident population was about 212,800 persons, or 13 percent, greater than the level of 1,597,600 envisioned under the low-growth scenario, and about 116,300 persons, or 6 percent, less than the level of 1,926,700 projected under the high-growth scenario. Actual population levels are compared with projected population levels by county in Figure 48.

Household Projections

Changes in the number of households have important implications for long-range land use and transportation system planning. The household creates much of the demand for various land uses and such public facilities and services as arterial streets and highways and public transit service. The projections of the number of households in the Region under the high-growth, intermediate-growth, and low-growth scenarios are set forth in Table 108. The projected changes in the distribution of households among counties in the Region is shown in Figure 49. As will become apparent, the anticipated changes in the number of households do not necessarily parallel the anticipated changes in total population under the respective scenarios, a result of assumed changes in household types and related changes in household size, previously described.

As indicated in Table 108, under the highgrowth scenario, the number of households in



Source: Wisconsin Department of Administration, U. S. Bureau of the Census, and SEWRPC. 226

HOUSEHOLDS IN THE REGION BY COUNTY: ACTUAL 1980 AND 1990 AND PROJECTED 1990, 2000, AND 2010

				Households				
	Actual	Actual			Projecteo 1980-	I Change 2010		
County	1980	1990	Scenario	1990	2000	2010	Number	Percent
Kenosha	43,100	47,000	High-growth Intermediate-growth Low-growth	48,200 43,000 41,600	55,300 45,500 43,500	61,400 50,200 46,300	18,300 7,100 3,200	42.5 16.5 7.4
Milwaukee	363,700	373,000	High-growth Intermediate-growth Low-growth	369,300 365,900 351,300	384,200 370,000 371,100	397,100 398,900 401,000	33,400 35,200 37,300	9.2 9.7 10.3
Ozaukee	21,800	25,700	High-growth Intermediate-growth Low-growth	27,900 23,900 21,400	35,900 27,000 22,700	48,000 31,200 24,600	26,200 9,400 2,800	120.2 43.1 12.8
Racine	59,400	63,700	High-growth Intermediate-growth Low-growth	65,700 59,400 56,700	73,300 62,700 59,600	81,300 68,400 62,200	21,900 9,000 2,800	36.9 15.2 4.7
Walworth	24,800	27,600	High-growth Intermediate-growth Low-growth	30,600 28,900 25,500	38,800 33,400 27,000	48,400 36,900 29,100	23,600 12,100 4,300	95.2 48.8 17.3
Washington	26,700	33,000	High-growth Intermediate-growth Low-growth	36,700 32,200 26,600	44,400 38,300 28,200	54,800 42,800 30,500	28,100 16,100 3,800	105.2 60.3 14.2
Waukesha	88,500	106,000	High-growth Intermediate-growth Low-growth	114,500 100,600 92,000	139,800 118,000 99,400	161,700 140,200 108,200	73,200 51,700 19,700	82.7 58.4 22.3
Region	628,000	676,100	High-growth Intermediate-growth Low-growth	692,900 653,900 615,100	771,700 694,900 651,500	852,700 768,600 701,900	224,700 140,600 73,900	35.8 22.4 11.8

Source: U. S. Bureau of the Census and SEWRPC.

the Region would increase by 224,700, or about 36 percent, from about 628,000 in 1980 to about 852,700 in 2010. The projected relative increase of 36 percent is slightly greater than the increase of about 31 percent in the total population projected under this scenario. As might be expected, Waukesha County, which has the largest projected absolute increase in total population, would have the largest absolute increase in households between 1980 and 2010, about 73,200. For the other six counties, the increases would range from about 18,300 in Kenosha to about 33,400 in Milwaukee.

Under the intermediate-growth scenario, the number of households in the Region would increase by about 140,600, or 22 percent, to a level of about 768,600 in the year 2010. The projected 22 percent increase in households is significantly greater than the 6 percent increase in total population projected under this scenario. Among the seven counties, the increases in households would range from about 7,100 in Kenosha to about 51,700 in Waukesha.

Under the low-growth scenario, the number of households would increase by about 73,900, or about 12 percent, to about 701,900 in the year 2010. This is in contrast to a projected 14 percent decrease in the total regional population between 1980 and 2010. It is a result of the assumed increase in smaller households, particularly single-person and single-parent households. Under this scenario, the increase in households would be about 37,300 in Milwaukee County, about 19,700 in Waukesha County, and under 5,000 in Kenosha, Ozaukee, Racine, Walworth and Washington Counties.

As shown in Figure 49, the actual increase in the number of households in the Region since 1980 has been between the levels anticipated under



HOUSEHOLD LEVELS IN THE REGION BY COUNTY: ACTUAL AND PROJECTED 1950-2010

Source: U. S. Bureau of the Census and SEWRPC.

the high-growth and intermediate-growth scenarios. The number of households in the Region in 1990 was anticipated to be 692,900 under the high-growth scenario, 653,900 under the intermediate-growth scenario, and 615,100 under the low-growth scenario. Recently released data from the 1990 Federal Census of Population and Housing indicate that the number of households in the Region was 676,100 in 1990, about 16,800 households, or 2 percent, below the level envisioned under the high-growth scenario. Consequently, the actual number of households exceeded the levels anticipated under the intermediate-growth and low-growth scenarios by 22,200, or 3 percent, and by 61,000, or 10 percent, respectively. Actual household levels are compared with projected household levels by county in Figure 50.

Employment Projections

Commission employment projections for the Region under the three regional growth scena-

rios are presented in Table 109. The projected change in the distribution of employment among the counties in the Region is shown in Figure 51. Under the high-growth scenario, total employment in the Region would increase by 42 percent, or 367,400 jobs, from 884,200 in 1980 to 1,251,600 in 2010. Each of the seven counties would experience a significant employment increase, ranging from about 24,500 jobs in Ozaukee County to 103,900 jobs in Waukesha County.

Under the intermediate-growth scenario, employment growth would occur at a more moderate rate, about 19 percent, to about 1,051,300 jobs by the year 2010, an increase of about 167,100 jobs over 1980. Employment in Waukesha County would increase by about 68,900 jobs; the increases for the other six counties would range from about 9,700 jobs in Milwaukee County to about 29,000 jobs in Racine County.

Under the low-growth scenario, employment levels, which dropped significantly during the 1979 to 1983 depression, would stay depressed throughout the 1980s and would increase only gradually thereafter. By 2010, employment in the Region would reach 870,900, about 13,300 jobs less than the 1980 level. Milwaukee County would lose about 63,300 jobs between 1980 and 2010 under this scenario, while Waukesha County would gain about 37,000 jobs. Employment changes in the other five counties would be relatively minor.

Each of the regional growth scenarios envisions continued change in the structure of the regional economy. In particular, under each of the growth scenarios, the historic dominance of manufacturing employment in the Region would continue to decrease, while the relative share of employment in service industries would increase.⁴ As indicated in Table 110, under the high-growth scenario, manufacturing employment would

⁴As presented in this chapter, manufacturing employment includes employment in manufacturing, construction, and wholesale trade. Service employment includes persons employed in services and finance, insurance, and real estate along with self-employed persons. Agricultural employment includes employment in agriculture, agricultural services, forestry, mining, and miscellaneous.

ACTUAL AND ALTERNATIVE FUTURE REGIONAL AND COUNTY HOUSEHOLD LEVELS: 1950-2010



Source: U. S. Bureau of the Census and SEWRPC.

EMPLOYMENT LEVELS IN THE REGION BY COUNTY: ACTUAL 1980 AND 1990 AND PROJECTED 1990, 2000, AND 2010

				Employment	t				
	Actual	Actual		Projected	ł		Projected Change 1980-2010		
County	1980	1990	Scenario	1990	2000	2010	Number	Percent	
Kenosha	50,100	46,500	High-growth Intermediate-growth Low-growth	56,300 50,200 45,400	65,600 54,200 47,100	75,100 61,000 48,800	25,000 10,900 -1,300	49.9 21.8 -2.6	
Milwaukee	542,300	578,200	High-growth Intermediate-growth Low-growth	573,000 518,000 484,300	599,800 530,000 481,800	625,800 552,000 479,000	83,500 9,700 -63,300	15.4 1.8 -11.7	
Ozaukee	25,600	32,200	High-growth Intermediate-growth Low-growth	31,600 26,400 23,600	40,000 31,400 25,200	50,100 36,800 26,100	24,500 11,200 500	95.7 43.8 2.0	
Racine	76,100	82,200	High-growth Intermediate-growth Low-growth	94,800 81,900 72,200	114,400 91,400 75,700	137,700 105,100 78,400	61,600 29,000 2,300	80.9 38.1 3.0	
Walworth	31,100	37,100	High-growth Intermediate-growth Low-growth	40,500 34,400 30,000	51,100 40,000 32,000	62,600 47,300 34,800	31,500 16,200 3,700	101.3 52.1 1 <u>1.9</u>	
Washington	31,400	41,800	High-growth Intermediate-growth Low-growth	41,500 36,100 31,600	53,300 42,800 35,300	68,800 52,600 39,200	37,400 21,200 7,800	119.1 67.5 24.8	
Waukesha	127,600	172,300	High-growth Intermediate-growth Low-growth	150,200 133,900 124,200	186,600 161,800 143,800	231,500 196,500 164,600	103,900 68,900 37,000	81.4 54.0 29.0	
Region	884,200	990,300	High-growth Intermediate-growth Low-growth	987,900 880,900 811,300	1,110,800 951,600 840,900	1,251,600 1,051,300 870,900	367,400 167,100 -13,300	41.6 18.9 -1.5	

Source: Wisconsin Department of Industry, Labor and Human Relations and SEWRPC.

increase by 29 percent, compared to a projected increase of about 66 percent in service employment. Under the intermediate-growth scenario, manufacturing employment would increase by only 7 percent, compared to a service employment increase of about 37 percent. Under the low-growth scenario, manufacturing employment would decrease by about 13 percent, while service employment would increase by 12 percent. As a result, by the year 2010, the relative share of service employment would be nearly equal to that of manufacturing under each of the growth scenarios (see Table 111).

As shown in Figure 52, during the latter half of the 1980s, actual employment in the Region approached levels anticipated under the highgrowth scenario. The employment level was anticipated to be 987,900 jobs in 1990 under the high-growth scenario, 880,900 jobs under the intermediate-growth scenario, and 811,300 jobs under the low-growth scenario. The actual 1990 level of 990,300 jobs is about 0.2 percent above the level anticipated under the high-growth scenario, about 12 percent above the level anticipated under the intermediate-growth scenario, and about 22 percent above the level anticipated under the low-growth scenario. Actual employment levels through 1990 are compared with those projected under the three regional growth scenarios for each of the seven counties in Figure 52. In reviewing these data, it is important to recognize that employment levels are subject to short-term fluctuation in response to



EMPLOYMENT LEVELS IN THE REGION BY COUNTY: ACTUAL AND PROJECTED: 1950-2010

Source: Wisconsin Department of Industry, Labor and Human Relations and SEWRPC.

business cycles, cycles which are apparent on the accompanying graphs. Long-term employment trends following the deep recession of 1979 through 1983 are not yet clearly defined.

Personal Income Projections

Changes in employment levels and in the types of jobs available may be expected to result in changes in personal income levels in the Region. Future income levels will have a direct bearing on the ability of the regional population to pursue personal preferences regarding housing types and location, recreation, and other lifestyle factors and, accordingly, may be expected to have a major impact on the evolving regional settlement pattern. Projections of personal income levels under the three alternative regional growth scenarios are presented in this section. For each scenario, the primary income projections made were those of per capita income. Projections of aggregate personal income were made by multiplying the projected per capita income by the projected population. Projected mean household income was derived from projected aggregate income levels and the projected number of households. It should be noted that the historic and projected income data presented herein are all expressed in constant 1990 dollars.

Historic and projected personal income levels in the Region are set forth in Table 112. As the table indicates, per capita income in the Region measured in constant 1990 dollars increased steadily, from \$7,702 in 1950 to \$10,285 in 1960, to \$12,896 in 1970, and to \$15,240 in 1980.⁵ As measured in constant dollars, the increase in per capita income was relatively stable during the 1950s. 1960s, and 1970s, showing average annual increases of \$258, \$261, and \$234, respectively. The average annual percentage increase, however, decreased in each decade, from 2.9 percent per year during the 1950s to 2.3 percent during the 1960s and to 1.7 percent during the 1970s. More recently, personal income levels in the Region have been adversely affected by the deep economic recession of 1979 through 1983. As indicated in Table 112, in 1985, per capita income in the Region stood at \$13,559, about \$1,681, or 11 percent, below the 1980 level.

Under the high-growth scenario, it is envisioned that a strong recovery from the 1979 through 1983 recession would be accompanied by a significant increase in personal income levels. During the second half of the 1980s, per capita income in the Region would recover from the

⁵The source of personal income data presented in this report is the U. S. Census Bureau. While income levels are generally identified by the census year in which they are collected, 1950, 1960, 1970, and 1980, and 1990, it should be noted that the income figures collected in the census are for the year preceding the census, that is, 1949, 1959, 1969, 1979, and 1989, respectively. The 1985 income data presented herein are estimates developed by the Census Bureau for 1985.

EMPLOYMENT LEVELS IN THE REGION BY MAJOR INDUSTRY GROUP: ACTUAL 1980 AND PROJECTED 2010

			Employme	nt		
	Actual	Actual	Projected		Projected 1980-2	Change 2010
Major Industry Group 1980 1990		Scenario	2010	Number	Percent	
Agriculture ^a	15,300	12,100	High-growth Intermediate-growth Low-growth	13,300 12,500 12,200	-2,000 -2,800 -3,100	-13.1 -18.3 -20.3
Manufacturing ^b	Manufacturing ^b 331,100 310,500		High-growth Intermediate-growth Low-growth	426,600 353,300 288,800	95,500 22,200 -42,300	28.8 6.7 -12.8
Retail Trade	tail Trade 131,900 155,700 High-gr Interme Low-gr		High-growth Intermediate-growth Low-growth	190,200 162,800 133,300	58,300 30,900 1,400	44.2 23.4 1.1
Transportation, Communication, and Utility	39,600	40,900	High-growth Intermediate-growth Low-growth	50,900 42,800 35,900	11,300 3,200 -3,700	28.5 8.1 -9.3
Government and Education	120,700	139,500	High-growth Intermediate-growth Low-growth	163,500 143,800 126,500	42,800 23,100 5,800	35.5 19.1 4.8
Services ^C 245,		331,600	High-growth Intermediate-growth Low-growth	407,100 336,100 274,200	161,500 90,500 28,600	65.8 36.8 11.6
Total	884,200	990,300	High-growth Intermediate-growth Low-growth	1,251,600 1,051,300 870,900	367,400 167,100 -13,300	41.6 18.9 -1.5

^aIncludes agriculture, agricultural services, forestry, mining and miscellaneous.

^bIncludes manufacturing, construction, and wholesale trade.

^cIncludes services; finance, insurance, and real estate; and self-employed.

Source: Wisconsin Department of Industry, Labor and Human Relations; U. S. Bureau of Economic Analysis; and SEWRPC.

Table 111

PERCENT DISTRIBUTION OF EMPLOYMENT BY MAJOR INDUSTRY GROUP IN THE REGION: ACTUAL 1980 AND PROJECTED 2010

	Actual	Actual		Projected: 2010	
Major Industry Group	1980	1990	High-Growth	Intermediate-Growth	Low-Growth
Agriculture ^a	1.7	1.2	1.1	1.2	1.4
Manufacturing ^b	37.4	31.4	34.1	33.6	33.2
Retail Trade	14.9	15.7	15.2	15.5	15.3
Transportation, Communication,			1.1		10 A
and Utility	4.5	4.1	4.1	4.1	4.1
Government and Education	13.7	14.1	13.0	13.7	14.5
Services ^c	27.8	33.5	32.5	31.9	31.5
Total	100.0	100.0	100.0	100.0	100.0

^aIncludes agriculture, agricultural services, forestry, mining, and miscellaneous.

^bIncludes manufacturing, construction, and wholesale trade.

^CIncludes services; finance, insurance, and real estate; and self-employed.

Source: Wisconsin Department of Industry, Labor and Human Relations; U. S. Bureau of Economic Analysis; and SEWRPC.



Source: Wisconsin Department of Industry, Labor and Human Relations and SEWRPC.

Income Category	Year	Per Capita Income	Mean Household Income
Actual Income	1950	\$ 7,702	\$26,954
	1960	10,285	34,738
	1970	12,896	42,216
	1980	15,240	42,832
	1985	13,559	36,723
	1990	14,351	38,541
Projected Income			
High-Growth Scenario	1990	\$16,400	\$45,600
	2000	20,400	56,100
	2010	24,400	66,100
Intermediate-Growth Scenario	1990	\$15,200	\$40,900
	2000	16,900	43,400
	2010	18,600	45,300
Low-Growth Scenario	1990	\$13,600	\$35,200
	2000	14,300	33,900
	2010	15,000	32,400

ACTUAL AND PROJECTED PERSONAL INCOME LEVELS IN THE REGION: 1950-2010

NOTE: All income data are presented in constant 1990 dollars.

Source: U. S. Bureau of the Census and SEWRPC.

reduced levels of the recession and by 1990 would exceed the 1980 level (see Figure 53). Between 1990 and the year 2010, per capita income measured in constant 1990 dollars would increase at the relatively high rate of about 2 percent per year, approximately the same growth rate experienced in the Region between 1960 and 1980, a period generally characterized by favorable economic growth within the Region. Under this scenario, per capita income in the Region would rise to about \$24,400 in the year 2010, an increase of about \$9,160, or 60 percent, over 1980. The mean household income would rise by just over 54 percent, from about \$42,800 in 1980 to about \$66,100 in 2010 (see Figure 54).

The intermediate-growth scenario also anticipates a recovery in personal income levels during the latter half of the 1980s, although the recovery would be slower than under the highgrowth scenario. Under the intermediate-growth scenario, it is envisioned that by 1990 per capita income would recover to the 1980 level. It is further envisioned that between 1990 and 2010, per capita income would increase by about 1 percent per year. This represents a continuation of the trend of declining relative increases in per capita income evident during the 1960s and 1970s. Even at the reduced rate of growth, however, per capita income would increase significantly over the long term. By the year 2010, per capita income would reach \$18,600, about \$3,400, or 22 percent, above the 1980 level. Mean household income would increase somewhat, to about \$45,300, an increase of about \$2,500, or 6 percent.

As previously noted, the low-growth scenario envisions a very slow recovery from the 1979 to 1983 recession. This scenario envisions only minimal long-term growth in total employment and a decrease in traditionally high-paying manufacturing jobs in the Region. These conditions may be expected to alter the long-term trend in personal income levels in the Region. Under this scenario, it is envisioned that per capita income would remain at depressed recessionary levels until 1990 and then, between 1990 and 2010, would increase slowly, at about half the average annual rate of increase anticipated under the intermediate-growth scenario, or about 0.5 percent. Under a low-growth scenario, then, per capita income would be expected to approxi-

ACTUAL AND PROJECTED PERSONAL INCOME LEVELS IN THE REGION: 1950-2010



Source: U. S. Bureau of the Census and SEWRPC.

mate \$15,000 in the year 2010, about the same as the 1980 level. Mean household income would approximate \$32,400 in 2010, a decrease of about 24 percent from the 1980 level.

It is interesting to compare projections of personal income prepared by the Bureau of Economic Analysis of the U.S. Department of Commerce with the personal income levels anticipated under the respective regional growth scenarios. The most recent Bureau projections were prepared in 1985 and reflect economic data available through 1983.6 Those projections envisioned a rapid recovery of personal income levels in the Region between 1983 and 1990. Over the long term, from 1990 through 2010, Bureau projections indicated that per capita income in the Region would increase at an average annual rate of 1.2 percent; the anticipated rate of increase is somewhat higher during the early part of this period and somewhat lower during the latter part. The projected increase of 1.2 percent per year between 1990 and 2010 is similar



ACTUAL AND PROJECTED MEAN HOUSEHOLD INCOME IN THE REGION: 1950-2010

Source: U. S. Bureau of the Census and SEWRPC.

to that envisioned under the intermediate-growth scenario, substantially lower than the 2 percent envisioned under the high-growth scenario, and substantially higher than the 0.5 percent envisioned under the low-growth scenario.

Land Use Demand Projections

Changes in the levels and distribution of population, households, and employment may be expected to generate additional demand for

⁶U. S. Bureau of Economic Analysis income projections for the Region are based on income projections for the Kenosha Metropolitan Statistical Area, which consists of Kenosha County; the Racine Metropolitan Statistical Area, which consists of Racine County; and the Milwaukee Metropolitan Statistical Area, which consists of Milwaukee, Ozaukee, Washington, and Waukesha Counties. Bureau projections are not available for Walworth County. urban land development in the Region. Projections of demand for the major categories of land use in the Region under the three regional growth scenarios are presented in Table 113. For the residential; governmental and institutional; recreational; and transportation, communication, and utility land use categories, the projected increases in land area between 1980 and 2010 are based upon the anticipated increases in the number of households under each scenario. For the industrial and commercial land use categories, the projected changes in land area are, for the most part, based upon the anticipated increases in related employment for each growth scenario (see footnotes in Table 113).

As indicated in Table 113, under the highgrowth scenario, it is envisioned that land in urban use within the Region would increase by about 159,400 acres, or 42 percent, from about 378,100 acres in 1980 to 537,500 acres in the year 2010. As a result, urban land uses would encompass about 31 percent of the total area of the Region, compared to about 22 percent in 1980. Growth in the two largest urban land use categories, residential and transportation, communication, and utility, would combine to account for about 128,700 acres, or about 81 percent of the overall increase in urban lands envisioned under the high-growth scenario.

Under the intermediate-growth scenario, urban lands would increase by about 98,000 acres, or about 26 percent, to about 476,100 acres in the year 2010. Urban lands would then comprise nearly 28 percent of the total area of the Region. The combined increase of 80,500 acres in residential land and transportation, communication, and utility land would constitute about 82 percent of the overall increase in urban lands anticipated under the intermediate-growth scenario.

A very modest increase in urban lands of about 50,900 acres, or about 14 percent, is anticipated under the low-growth scenario. Urban lands would total 429,000 acres, or about 25 percent of the Region, in the year 2010. About 83 percent of the overall increase in urban lands would consist of residential and transportation, communication, and utility lands.

The development of the alternative regional growth scenarios was based upon explicit consideration of a number of the social and economic factors that may be expected to affect

county and regional population and employment levels. These factors include birth, death, and migration rates; labor force participation rates; and the relative anticipated strengths of various sectors of the regional economy. To the extent practicable, assumptions regarding such factors were expressed quantitatively in the development of the growth scenarios. It should be recognized that there are many other factors which are not easily quantified and do not lend themselves to explicit consideration in the development of alternative growth scenarios and, particularly, in the projection of county population and employment levels. Among these factors are the quality of education, the levels and quality of public services, the quality of infrastructure systems, the quality of neighborhoods, relations between races and between socio-economic groups, and the incidence of crime. Trends in these factors may be expected to influence the degree of centralization or decentralization of population and employment within the Region. Thus, while these factors can be treated only implicitly in the development of the alternative future growth scenarios, they may have a significant bearing on future county population and employment levels.

ADOPTED REGIONAL LAND USE PLAN

A concerted effort was made in previous regional land use planning efforts to explore and evaluate the full range of practical future land development patterns available to the Region. In the first regional land use planning effort, undertaken in 1963, four different land use plan designs were prepared and evaluated. In the second effort, undertaken in 1972, two additional alternative designs were explored. Both efforts indicated that a controlled existing trend plan, emphasizing a centralized settlement pattern, would provide the best guide to shaping urban development in the Region.

In view of the extensive work in preparing and evaluating alternative land use designs conducted under the first and second regional land use planning efforts and the conclusive nature of the findings, it was determined that additional design alternatives would not be explored. Rather, it was determined that the basic concepts of the adopted year 2000 regional land use plan would be advanced to a new design year and incorporated into the new land use plan.

PROJECTED LAND USE DEMAND IN THE REGION: 1980-2010

	Actual	Actual	Projected: 2	010	Projected Change 1980-2010		
Land Use Category	1980	1985	Scenario	Acres	Acres	Percent	
Urban Residential ^a	179,831	184,603	High-growth Intermediate-growth Low-growth	268,250 235,157 208,911	88,419 55,326 29,080	49.2 30.8 16.2	
Commercial ^b	8,162	8,714	High-growth Intermediate-growth Low-growth	11,459 9,983 8,612	3,297 1,821 450	40.4 22.3 5.5	
Industrial ^b	11,171	12,080	High-growth Intermediate-growth Low-growth	20,859 15,728 13,471	9,688 4,557 2,300	86.7 40.8 20.6	
Transportation, Communication, and Utilities ^a	117,706	120,279	High-growth Intermediate-growth Low-growth	157,950 142,887 130,941	40,244 25,181 13,235	34.2 21.4 11.2	
Governmental and Institutional ^a	17,033	17,240	High-growth Intermediate-growth Low-growth	23,122 20,843 19,036	6,089 3,810 2,003	35.7 22.4 11.8	
Recreational ^a	24,309	25,564	High-growth Intermediate-growth Low-growth	35,904 31,564 28,122	11,595 7,255 3,813	47.7 29.8 15.7	
Unused Urban ^C	19,935	19,215	High-growth Intermediate-growth Low-growth	19,935 19,935 19,935	000	0.0 0.0 0.0	
Total Urban Land	378,147	387,695	High-growth Intermediate-growth Low-growth	537,479 476,097 429,028	159,332 97,950 50,881	42.1 25.9 13.5	
Agricultural and Other Open Land	1,342,969	1,333,418	High-growth Intermediate-growth Low-growth	1,183,637 1,245,019 1,292,088	-159,332 -97,950 -50,881	-11.9 -7.3 -3.8	
Total	1,721,116	1,721,113		1,721,116			

^aFor the residential, governmental and institutional, recreational, and transportation, communication, and utility land use categories, the projected changes in land area between 1980 and 2010 under the three regional growth scenarios reflect the increases in the number of households anticipated under the respective growth scenarios. For each category, the projected increases in land area were obtained by multiplying the projected increase in households between 1980 and 2010 for each growth scenario by the historic ratio of the increase in land area to the increase in households within the Region for the period 1963 to 1980.

^bFor the industrial and commercial land use categories, the projected changes in land area are based upon anticipated increases in related employment under the respective growth scenarios and land area to employee ratios used by the Commission in systems level land use planning. The land area to employee ratios for the commercial and industrial use categories were re-evaluated under the current regional land use planning program and found to properly reflect current commercial and industrial development patterns. The projected increase in commercial land for all three scenarios was obtained by multiplying the projected increase in related employment between 1980 and 2010 by the land area to employee ratio for commercial land. For both the high-growth and intermediate-growth scenarios, the projected increase in industrial land was obtained by multiplying the projected increase in related employment between 1985 and 2010 by the land area to employee ratio for commercial land. For both the high-growth and intermediate-growth scenarios, the projected increase in industrial land was obtained by multiplying the projected increase in related employment between 1985 and 2010 by the land area to employee ratio for industrial land. The industrial land projection for these two scenarios was based upon the anticipated industrial employment increase from 1985 to 2010, rather than 1980 to 2010, in order to reflect the increase in industrial employment is anticipated under the low-growth scenario between 1985 and 2010, industrial land would be expected to increase somewhat as a result of continuing change in the distribution of industrial activity in the Region. Accordingly, a nominal increase in industrial land, equal to about one-half of the increase in industrial land projected under the intermediate-growth scenario.

^cThe amount of unused urban land was assumed constant over the projection period.

Source: SEWRPC.

The new plan would, thus, update and extend recommendations of the previously adopted plan. As shown in graphic summary on Map 64, the plan recommends the promotion of compact, centralized land use development in the Region, with development generally occurring in concentric rings along the periphery of, and outward from, existing urban centers.

Like the year 2000 plan, the adopted year 2010 plan seeks to influence the operation of the urban land market in several important ways in order to achieve a more healthful, attractive, and efficient settlement pattern. In this regard, the plan recommends that new urban development occur primarily in those areas of the Region which are covered by soils suitable for such development and in those areas which can be readily served by essential municipal facilities and services, including public sanitary sewerage, water supply, and public transit facilities and services. The plan recommends the preservation in essentially natural, open uses of the identified environmental corridors and the preservation in agricultural and related uses of most of the remaining prime agricultural lands in the Region. The adopted year 2010 land use plan takes into account changes in land use that have taken place in the Region since the adoption of the year 2000 plan; the findings and recommendations of other local, county, and regional planning efforts since completed; and forecasts of population and economic activity levels within the Region through the year 2010.

As a practical matter, the preparation of any specific land use plan must be based on a single set of population and employment projections. It was the collective judgment of the Advisory Committee guiding the preparation of the new plan that future population and employment levels within the Region would be most closely approximated by the intermediate-growth scenario. Accordingly, the Committee directed that the new land use plan be prepared to accommodate the population and employment forecasts attendant to the intermediate-growth scenario, subject, however, to two modifications:

1. The forecast regional design year 2010 population and employment levels were adjusted at the county level to achieve a more centralized distribution of population, employment, and related urban land use development within the Region. Adjustments included the allocation of more population to Milwaukee County than initially forecast, with corresponding reductions in the design year population levels for Ozaukee, Walworth, Washington, and Waukesha Counties. In Kenosha and Racine Counties the planned population distribution was centralized around the Kenosha and Racine urbanized areas. The distribution of employment within the Region was similarly centralized.

2. Design year population and employment levels were adjusted to reflect the implications of new benchmark population and employment data, particularly data from the 1990 U.S. Census of Population and Housing. Such data indicated that population and employment growth was exceeding that envisioned under the intermediate-growth scenario in certain areas of the Region. Adjustments were made to accommodate reasonable growth and development between 1990 and 2010 in areas of the Region which had grown more rapidly than expected by 1990 and where additional growth is likely to occur over the next two decades. The most noteworthy adjustments were increases in the design year population levels for Kenosha and Racine Counties.⁷

As a result of these two adjustments, the new regional land use plan was designed to accommodate a year 2010 resident population of about 1,911,000 persons, about 38,800 persons, or 2 percent, more than envisioned under the intermediate-growth forecast, and a total employment of about 1,095,000 jobs, about 43,700 jobs, or 4 percent, more than initially forecast.

⁷The adjustment of design year population levels in the southern portion of the Region, particularly Kenosha County, was necessitated in part by a perceived "Illinois" influence, that is, workers in northeastern Illinois increasingly seeking residences in Wisconsin, a phenomenon not anticipated in the intermediate-growth scenario forecasts.

Map 64

ADOPTED LAND USE PLAN FOR THE SOUTHEASTERN WISCONSIN REGION: 2010



The adopted design year 2010 regional land use plan envisions a need to convert about 86 square miles of land from rural to urban use to accommodate an anticipated population increase of about 168,000 persons and an anticipated employment increase of about 223,000 jobs in the Region between 1985 and 2010. The plan seeks to promote a centralized regional settlement pattern, maintaining, to the extent practicable, population and employment levels of the central portions of the Kenosha, Milwaukee, and Racine urbanized areas. The plan seeks to encourage the location of new urban development only in areas which are covered by soils suitable for such use; which are not subject to special hazards such as flooding and erosion; and which can be readily provided with basic urban services and facilities, including public sanitary sewer, water supply, and mass transit services. The plan seeks to preserve all the remaining primary environmental corridors and most remaining prime agricultural lands in the Region. 239

Source: SEWRPC.

Plan Description⁸

Under the adopted land use plan for Southeastern Wisconsin, the population of the Region may be expected to reach a level of about 1,911,000 persons by the year 2010, an increase of about 168.300 persons over the 1985 population level. while employment may be expected to reach about 1,095,000 jobs by the year 2010, an increase of 223,100 jobs over the 1985 level. The plan proposes to accommodate this growth in population and employment through the conversion of approximately 86 square miles of rural to urban use by the year 2010. The future land use pattern proposed by the plan is shown on Map 64 and is summarized for the Region overall in Table 114 and for the individual counties and planning analysis areas in Appendices C and D of SEWRPC Planning Report No. 40.

Residential Development and Redevelopment

The adopted land use plan proposes to meet the residential needs of the growing regional population through the maintenance of existing urban areas and, as needed, the outward expansion of those areas. The conservation and redevelopment of existing urban residential areas is an important element of the adopted plan. Indeed, of the approximately 222,000 acres of urban residential land envisioned by the year 2010, 83 percent, or 185,000 acres, existed in 1985.

The bulk of the remaining 37,700 acres of required residential land would be developed at urban medium densities.⁹ As shown in Table 115, medium-density residential land would increase by about 30,800 acres, or 57 percent; high-density residential land would increase by 1,800 acres, or 7 percent; low-density residential land would increase by 3,000 acres, or 3 percent; and suburban-density land would increase by 1,300 acres, or 17 percent.

⁸The last detailed land use inventory for the Region was completed for the year 1985 and therefore the base year for socio-economic data presented in this section of the chapter is 1985.

⁹Under medium residential density, a singlefamily lot would typically be about one-quarter acre in size and a multiple-family development would typically average about 10 dwelling units per net acre. Among the seven counties in the Region, Waukesha County would experience the largest increase in urban residential land, 13,100 acres (see Table 115). For the other six counties in the Region, the proposed increase in urban residential land ranges from about 3,100 acres in Walworth County to about 4,600 acres in Kenosha County. As further indicated in Table 115, the plan proposes a modest increase of just over 700 acres in land to be developed for rural residential use.¹⁰

The plan encourages the development of new urban residential land in planned neighborhoods. Insofar as possible, each neighborhood should be bounded by arterial streets; major park, parkway, or institutional lands; bodies of water; or other natural or cultural features which serve to physically separate each unit from the surrounding units.

Commercial Development

Major Commercial Centers: The adopted land use plan proposes the development of approximately 1,300 acres of new commercial land within the Region, excluding related off-street parking. Such development would increase the total stock of commercial land in the Region to about 10,000 acres by the year 2010 (see Table 116). This increase would accommodate anticipated increases in retail and service employment and the demands associated with growth and redistribution of the population within the Region.

To accommodate planned increases in commercial employment and to facilitate the efficient provision of services and goods throughout the Region, the plan recommends the 14 existing commercial centers be retained and that five

¹⁰Rural residential development by definition implies a parcel size of at least five acres for single-family housing development. Frequently, particularly in woodland areas and areas of steep slope, only a small portion of the total parcel is developed as homesite and yard, the remainder is kept in a natural condition. Accordingly, the incremental area allocated to rural residential development under the recommended plan is based on the assumption that only one-fifth of each additional rural residential parcel would be developed as homesite and associated yard area.

	and the second						
	Existing	1985	Planned In 1985-	ncrement 2010	Total 2010		
Land Use Category	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total	
Urban Residential							
High-Density	27,797	1.6	1,817	6.5	29,614	1,7	
Medium-Density	54,153	3.1	30,776	56.8	84,929	4.9	
Low-Density	94,618	5.5	3,039	3.2	97,657	5.7	
Suburban-Density	8,035	0.5	1,344	16.7	9,379	0.6	
Subtotal	184,603	10.7	36,976	20.0	221,579	12.9	
Commercial	8,714	0.5	1,320	15.1	10,034	0.6	
Industrial	12,080	0.7	5,186	42.9	17,266	1.0	
and Utilities ^a	120,279	7.0	14,560	12.1	134,839	7.8	
Governmental and Institutional	17,240	1.0	1,042	6.0	18,282	1.1	
Recreational	25,564 ^b	1.5	4,089 ^C	16.0	29,653	1.7	
Unused Urban Land	19,215	1.1	-8,400	-43.7	10,815	0.6	
Urban Subtotal	387,695	22.5	54,773	14.1	442,468	25.7	
Rural							
Residential	a		721		721	e	
Agricultural	931,956	54.1	-40,487	-4.3	891,469	51.8	
Other Open Lands [†]	401,462	23.4	-15,007	-3.7	386,455	22.5	
Rural Subtotal	1,333,418	77.5	-54,773	-4.1	1,278,645	74.3	
Total	1,721,113	100.0	· O	0.0	1,721,113	100.0	

EXISTING AND PROPOSED LAND USE IN THE REGION: 1985 AND 2010 RECOMMENDED REGIONAL LAND USE PLAN

^aIncludes off-street parking areas.

^bIncludes net site area of public and nonpublic recreation sites.

^CIncludes only that net site area recommended for public recreation use.

^dIncluded in 1985 land use inventory as part of urban residential land use.

^eLess than 0.1 percent.

^fIncludes woodlands, water, wetlands, unused rural land, landfill sites, and quarries.

Source: SEWRPC.

new commercial centers be developed (see Map 65). As indicated in Chapter III of this report, two types of commercial centers are identified under the regional planning process: retail and office. To qualify as a major retail center, a site must accommodate at least 2,000 retail jobs. To qualify as a major office center a site must accommodate 3,500 office and servicerelated jobs. In 1985, the 14 major commercial centers located within the Region provided 164,300 jobs. As shown in Table 117, the adopted plan envisions that by the year 2010 employment at these centers will increase by about 58,500 jobs, or about 36 percent, to about 223,100 jobs. Four of the five proposed centers, located at Park Place in northwestern Milwaukee County, along IH 43 in the City of Mequon, near the Milwaukee

EXISTING AND PROPOSED RESIDENTIAL LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

		Residential Land Use											
Urban High-Density					U	rban Medi	um-Densit	у		Urban Low-Density			
	Existing 1985	Planned Increment Total 1985-2010 2010		Existing 1985	Plan Incre 1985-	med ment -2010	Total 2010	Existing 1985	Planned Increment 1985-2010		Total 2010		
County	(acres)	Acres	Percent	(acres)	(acres)	Acres	Percent	(acres)	(acres)	Acres	Percent	(acres)	
Kenosha Milwaukee	1,619 22,721	90 1,424	5.6 6.3	1,709 24,145	6,676 16,055	5,233 3,932	78.4 24.5	11,909 19,987	6,718 8,295	-723 -969	-10.8 -11.7	5,995 7,326	
Ozaukee Bacine	30 2,436	0 51	0.0 2.1	30 2.487	3,664 6,913	2,431 3.619	66.3 52.4	6,095 10,532	9,006	1,229	13.6 5.2	10,235	
Walworth	0	O	0.0	· 0	5,446	2,499	45.9	7,945	10,639	368	3.5	11,007	
Washington Waukesha	268 723	135 117	50.4 16.2	403 840	3,277 12,122	3,131 9,931	95.5 81.9	6,408 22,053	12,019 37,908	587 2,028	4.9 5.3	12,606 39,936	
Region	27,797	1,817	6.5	29,614	54,153	30,776	56.8	84,929	94,618	3,039	3.2	97,657	

						Residential	Land Use						
Suburban-Density						Total I	Jrban			Rural			
	Planned Increment 1985-2010		Total 2010	Total Existing		Planned Increment 1985-2010		Existing 1985 ^a	Planned Increment 1985-2010		Total 2010		
County	(acres)	Acres	Percent	(acres)	(acres)	Acres	Percent	(acres)	(acres)	Acres	Percent	(acres)	
Kenosha	307	21	6.8	328	15,320	4,621	.30.2	19,941		84		84	
Milwaukee	924	0	0.0	924	47,995	4,387	9.1	52,382		74	•••	74	
Ozaukee	994	43	4.3	1,037	13,694	3,703	27.0	17,397		55		55	
Racine	59	44	74.6	103	19,441	4,233	21.8	23,674		99		99	
Walworth	395	185	46.8	580	16,480	3,052	18.5	19,532		221		221	
Washington	512	16	3.1	528	16,076	3,869	24.1	19,945		68		68	
Waukesha	4,844	1,035	21.4	5,879	55,597	13,111	23.6	68,708		120		120	
Region	8,035	1,344	16.7	9,379	184,603	36,976	20.0	221,579	'	721		721	

^aIncluded in 1985 land use inventory as part of urban residential land use.

Source: SEWRPC.

Table 116

EXISTING AND PROPOSED COMMERCIAL LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

	Commercial Land Use ^a									
	Existin	g 1985	Planned 1985	Increment 5-2010	Total	2010				
County	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total				
Kenosha Milwaukee Ozaukee	615 3,454 470	7.1 39.6 5.4	227 261 77	36.9 7.6 16.4	842 3,715 547	8.4 37.0 5.4				
Racine Walworth Washington Waukesha	906 776 547 1.946	10.4 8.9 6.3 22.3	123 47 66 519	13.6 6.1 12.1 26.7	1,029 823 613 2,465	10.3 8.2 6.1 24.6				
Region	8,714	100.0	1,320	15.1	10,034	100.0				

^aExcludes related off-street parking areas.

Source: SEWRPC.



MAJOR COMMERCIAL CENTERS IN THE REGION: 2010 RECOMMENDED LAND USE PLAN

The recommended year 2010 regional land use plan envisions a total of 19 major commercial centers to serve the needs of the Region through the plan design year, including eight major retail centers, eight major office centers, and three major combined retail and office centers. Fourteen of these centers existed in 1985 and would be retained through the year 2010. Five new centers would be developed by the year 2010, including a major retail center located along IH 94 in Kenosha County and four new office centers located in the Cities of Mequon, Milwaukee, and Wauwatosa and the Town of Pewaukee. It is envisioned that by the year 2010 the 19 planned major commercial centers would accommodate a total of about 199,000 retail and service jobs, or about 38 percent of all such jobs in the Region.

Source: SEWRPC.

County Institutions grounds in the City of Wauwatosa, and near the intersection of IH 94 and CTH J in the Town of Pewaukee, would function primarily as office centers. The fifth proposed commercial center, Kenosha West, located near the intersection of IH 94 and STH 50 in Kenosha County, would function primarily as a retail center. The plan envisions that employment at these sites will increase from 3,300 in 1985 to 21,700 by the design year 2010.

By the year 2010, the 19 major commercial centers proposed under the recommended land use plan would accommodate a total of 199,300 retail and service jobs, or 38 percent of all retail and service employment within the Region. These centers would encompass about 1,700 acres of commercial land, or 17 percent of all commercial land in the Region.

Local Commercial Development: Under the adopted plan, almost 800 additional acres of land would be developed as neighborhood and community commercial areas between 1985 and 2010. By the year 2010, land devoted to neighborhood and community commercial use would total 8,300 acres, representing about 83 percent of all commercial land in the Region. This land would accommodate a total of 330,100 retail and service jobs, or 62 percent of all retail and service jobs in the Region in the year 2010.

Industrial Development

<u>Major Industrial Centers</u>: The plan proposes to add about 5,200 acres of industrial land in the Region, increasing the total stock of such land to almost 17,300 acres by the plan design year (see Table 118). This increase would meet the land requirements of the anticipated increases in, and redistribution of, manufacturing and wholesaling activity within the Region and would be so distributed as to protect and enhance the continued efficient operation of these important components of the economic base of the Region.

This is proposed to be accomplished through the development of planned industrial centers properly located with respect to the existing and proposed transportation systems, through the protection and enhancement of existing industrial areas, and through the efficient provision of adequate utility services. The plan provides adequate sites for industrial development which meet the full array of criteria for such development, including ready accessibility to high-speed, all-weather arterial highway facilities; soils suitable for industrial development; adequate power and water supply; sanitary sewer service and stormwater drainage; reasonable access to airport, seaport, and railway facilities, as appropriate; and ready access to the labor supply.

	Cente	r Type	Commer	cial Land Use in	n Acres	Т	otal Employme	nt
				Planned			Planned	
Major			Existing	Increment	Total	Existing	Increment	Total
Commercial Center	Retail	Office	1985	1985-2010	2010	1985	1985-2010	2010
Existing								
Kenosha CBD		X	40	5	45	4,600	700	5,300
Bay Shore	X		43	5	48	4,000	800	4,800
Capitol Court	X		25	0	25	3,400	500	3,900
Mayfair	X	X	82	2	84	13,200	200	13,400
Milwaukee CBD	X	X	160	7	167	82,500	30,600	113,100
Northridge	X		61	13	74	6,100	1,300	7,400
Southgate-Point Loomis	X		41	0	41	3,400	500	3,900
Southridge	X		64	10	74	4,900	700	5,600
West Allis	X		95	0	95	4,900	600	5,500
Racine CBD		X	51	6	57	4,700	800	5,500
Regency Mall	X		46	47	93	4,400	2,000	6,400
West Bend		X	71	11	82	5,100	700	5,800
Blue Mound Road	x	X	257	230	487	17,500	19,200	36,700
Waukesha CBD		х	49	1	50	5,600	200	5,800
Subtotal			1,085	337	1,422	164,300	58,800	223,100
Proposed								
Kenosha West	X		21	57	78	600	2,800	3,400
Park Place		X	13	45	58	200	4,500	4,700
Milwaukee County								
Research Park		X	0	44	44	10,700	4,700	15,400
Mequon		X	48	37	85	1,800	3,400	5,200
Pewaukee		x	6	38	44	300	3,400	3,700
Subtotal			88	221	309	13,600	18,800	32,400
Total	'		1,173	558	1,731	177,900	77,600	255,500

SELECTED CHARACTERISTICS OF PLANNED MAJOR COMMERCIAL CENTERS IN THE REGION: 2010 RECOMMENDED LAND USE PLAN

NOTES: To qualify as a major retail center, a site must accommodate at least 2,000 retail jobs. To qualify as a major office center, a site must accommodate at least 3,500 office and service-related jobs.

The total 1985 employment of 177,900 at the above sites includes 124,000 commercial jobs and 53,900 other jobs. The total year 2010 employment of 255,500 includes 199,300 commercial jobs and 56,200 other jobs. Commercial jobs include the retail; service; and finance, insurance, and real estate employment categories; and self-employed.

Land use and employment data are based upon aggregations of data for U. S. Public Land Survey quarter sections which approximate the major commercial centers. The specific quarter sections included in each planned major commercial center are shown in Appendix E of SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern</u> <u>Wisconsin—2010</u>. There is some "overlap" between certain planned major commercial and planned major industrial centers as approximated by quarter section, owing to the mixture of land uses. For the overlapping quarter sections, all employment other than industrial related has been reported in this table.

Source: SEWRPC.

As indicated in Chapter III of this report, to qualify as a major industrial centers a site must accommodate at least 3,500 industrial jobs. The proposed major industrial centers range in character from older industrial complexes in central-city areas, which have traditionally emphasized heavy manufacturing activity, to planned industrial parks in outlying areas of the Region. It should be noted that, both nationally and within the Region, new industrial centers are increasingly characterized by a mix of uses, a mix which may include service operations, research facilities, and office facilities in addition to manufacturing and wholesaling uses. The

····	Industrial Land Use ^a						
н. 	Existin	Existing 1985 Planned Increment 1985-2010		Increment -2010	Total 2010		
County	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total	
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	917 5,375 577 1,416 678 690 2,427	7.6 44.5 4.8 11.7 5.6 5.7 20.1	567 1,211 388 564 494 777 1,185	61.8 22.5 67.2 39.8 72.9 112.6 48.8	1,484 6,586 965 1,980 1,172 1,467 3,612	8.6 38.1 5.6 11.5 6.8 8.5 20.9	
Region	12,080	100.0	5,186	42.9	17,266	100.0	

EXISTING AND PROPOSED INDUSTRIAL LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

^aExcludes related off-street parking areas.

Source: SEWRPC.

developing industrial centers recommended under the year 2010 land use plan may thus be expected to accommodate an increasing diversity of industrial and industry-related uses and jobs.

The major industrial centers recommended under the year 2010 regional land use plan are identified on Map 66 and in Table 119. The 22 centers existing in 1985 encompassed a total of about 5,700 acres of industrial land, excluding off-street parking, and accommodated about 171,500 industrial jobs, or 60 percent of all industrial employment in the Region in 1985.

The plan proposes to retain all 22 existing sites as major industrial centers through the year 2010 and to expand certain of these centers. It is anticipated that by the year 2010 the 22 existing major centers would encompass an additional 1,800 acres of industrial land and accommodate an additional 29,400 industrial jobs.

The plan also proposes to add three new major industrial centers by the year 2010. The three proposed centers would be located in or near the Cities of Burlington and Hartford and the Village of Pleasant Prairie. Under the plan, it is anticipated that industrial employment at these sites would increase from 3,800 jobs in 1985 to 15,000 by the year 2010. Industrial land use at these three sites would increase from 200 acres in 1985 to 1,000 acres in 2010. By the year 2010, the 25 major industrial centers proposed under the adopted land use plan would accommodate a total of 212,100 industrial jobs, or 58 percent of all industrial employment in the Region. These centers would encompass about 8,500 acres of industrial land, or about 49 percent of all industrial land in the Region. The three sites included under the year 2010 plan but not under the year 2000 plan include the Pewaukee and Waukesha North industrial areas, which were developed after the preparation of the year 2000 plan, and the proposed new Hartford industrial center.

Local Industrial Development: Under the adopted plan, about 2,600 additional acres of land would be developed at smaller, communitylevel industrial areas. By the year 2010, land devoted to community-level industrial use would total 8,800 acres, representing 51 percent of all industrial land in the Region. This land would accommodate a total of 152,600 industrial jobs, or 42 percent of all industrial jobs in the Region.

Governmental and Institutional Land Use

As indicated in Table 120, the adopted land use plan proposes to add by the year 2010 about 1,000 acres of new governmental and institutional land to the existing stock of such land within the Region, resulting in a total of about 18,300 acres of governmental and institutional land by the plan design year. The additional

SELECTED CHARACTERISTICS OF PLANNED MAJOR INDUSTRIAL CENTERS IN THE REGION: 2010 RECOMMENDED LAND USE PLAN

	Industria	al Land Use in	Acres	Т	otal Employme	ent
		Planned			Planned	
Major	Existing	Increment	Total	Existing	Increment	Total
Industrial Center	1985	1985-2010	2010	1985	1985-2010	2010
Existing					· .	
Kenosha	219	-20	199	8,600	-2,500	6,100
Cudahy-South Milwaukee	240	21	261	10,200	500	10,700
Milwaukee-Glendale	308	32	340	12,900	1,200	14,100
Milwaukee-Granville	530	417	947	12,700	6,600	19,300
Milwaukee-Menomonee Valley East	357	-9	348	18,400	3,200	21,600
Milwaukee-Menomonee Valley West	125	0	125	8,600	400	9,000
Milwaukee-Near North	111	0	111	9,300	300	9,600
Milwaukee-Near South	319	20	339	13,000	600	13,600
Milwaukee-North	349	0	349	17,500	400	17,900
Milwaukee-South	96	10	106	6,200	500	6,700
Oak Creek	281	318	599	9,300	5,000	14,300
West Allis-East	211	2	213	8,200	1,900	10,100
West Allis-West	136	80	216	9,100	1,500	10,600
West Milwaukee	355	4	359	8,900	-1,300	7,600
Mt. Pleasant	216	144	360	5,200	2,000	7,200
Racine-East	332	14	346	12,300	1,400	13,700
West Bend-North	118	116	234	4,600	1,700	6,300
Butler	596	114	710	25,500	1,500	27,000
New Berlin	309	39	348	11,200	500	11,700
Pewaukee	136	279	415	4,900	5,500	10,400
Waukesha-North	103	107	210	6,300	2,700	9,000
Waukesha-South	225	135	360	7,000	2,400	9,400
Subtotal	5,672	1,823	7,495	229,900	36,000	265,900
Proposed	1	· · · · ·				
Pleasant Prairie	0	425	425	300	6,100	6,400
Burlington	169	119	288	4,600	1,700	6,300
Hartford	39	245	284	1,400	3,500	4,900
Subtotal	208	789	997	6,300	11,300	17,600
Total	5,880	2,612	8,492	236,200	47,300	283,500

NOTES: To qualify as a major industrial center, a site must accommodate at least 3,500 industrial jobs.

The total 1985 employment of 236,200 at the above sites includes 171,500 industrial jobs and 64,700 other jobs. The total year 2010 employment of 283,500 includes 212,100 industrial jobs and 71,400 other jobs.

Land use and employment data are based upon aggregations of data for U. S. Public Land Survey quarter sections which approximate the major industrial centers. The specific quarter sections included in each planned major industrial center are shown in Appendix E of SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin—2010</u>. There is some "overlap" between certain planned major industrial and planned major commercial centers as approximated by quarter section, owing to the mixture of land uses. For the overlapping quarter sections, only the industrial related employment has been reported in this table.

Source: SEWRPC.



MAJOR INDUSTRIAL CENTERS IN THE REGION: 2010 RECOMMENDED LAND USE PLAN

The recommended year 2010 regional land use plan envisions a total of 25 major industrial centers to serve the needs of the Region through the plan design year. Twenty-two of these centers existed in 1985 and would be retained through the year 2010. Three new centers are proposed, including centers located in or near the Cities of Burlington and Hartford and the Village of Pleasant Prairie. It is envisioned that by the year 2010, the 25 planned major industrial centers would accommodate about 212,000 industrial jobs, or about 58 percent of all industrial jobs in the Region.

Source: SEWRPC.

Major existing governmental and institutional centers to be retained under the plan, including county seats and State and Federal office buildings, medical complexes, universities, technical schools, major libraries, and major cultural centers, are shown on Map 67.¹¹ No new major governmental or institutional centers are envisioned, and additional development of existing major centers would be limited to that necessary to meet the needs of the growing population.

Transportation, Communication, and Utility Land Use

As indicated in Table 121, the adopted land use plan proposes to add approximately 14,600 acres of new transportation, communication, and utility land to the existing stock of such land within the Region. A total of about 134,800 acres of land would be devoted to transportation, communication, and utility uses by the year 2010, an increase of about 12 percent over the 1985 level. Most of the additional land would be required for rights-of-way for new or improved arterial, collector, and minor streets needed to serve new urban development or to provide adequate transportation service to existing urban development. Some of the additional land would be required for planned airport expansions recommended in the regional airport system plan. Minor amounts of land would also be required for the planned expansion of existing, or construction of new, public sanitary sewage treatment facilities, as recommended in the regional water quality management plan.

The plan envisions only two new major utility centers in the Region through the plan design year: a new public sewage treatment plant serving the Village of Wales and a new peakload electric power generation plant in the Town

governmental and institutional lands proposed under the plan would consist of such neighborhood and community uses as new schools, churches, hospitals and nursing homes, and public facilities, including police and fire stations and city, village, and town halls.

¹¹With respect to major educational centers, Map 67 shows only four-year universities and public technical colleges. It should be noted that there are numerous other public and private postsecondary educational institutions in the Region.

EXISTING AND PROPOSED GOVERNMENTAL AND INSTITUTIONAL LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

	Governmental and Institutional Land Use							
• •	Existing 1985		Planned 1985	Planned Increment 1985-2010		2010		
County	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total		
Kenosha	1,314	7.6	155	11.8	1,469	8.0		
Milwaukee	7,154	41.5	172	2.4	7,326	40.1		
Ozaukee	1,024	6.0	88	8.6	1,112	6.1		
Racine	1,813	10.5	96	5.3	1,909	10.4		
Walworth	1,259	7.3	75	6.0	1,334	7.3		
Washington	1,087	6.3	102	9.4	1,189	6.5		
Waukesha	3,589	20.8	354	9.9	3,943	21.6		
Region	17,240	100.0	1,042	6.0	18,282	100.0		

Source: SEWRPC.

Table 121

EXISTING AND PROPOSED TRANSPORTATION, COMMUNICATION, AND UTILITY LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

	Transportation, Communication, and Utility Land Use							
	Existin	g 1985	Planned I 1985	ncrement -2010	Total	Total 2010		
County	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total		
Kenosha	9,912	8.2	2,200	22.2	12,112	9.0		
Milwaukee	36,337	30.2	2,288	6.3	38,625	28.6		
Ozaukee	8,637	7.2	1,205	14.0	9,842	7.3		
Racine	12,973	10.8	1,523	11.7	14,496	10.8		
Walworth	14,603	12.1	1,085	7.4	15,688	11.6		
Washington	12,828	10.7	1,539	12.0	14,367	10.7		
Waukesha	24,989	20.8	4,720	18.9	29,709	22.0		
Region	120,279	100.0	14,560	12.1	134,839	100.0		

Source: SEWRPC.

of Paris.¹² The major transportation and utility centers envisioned under the recommended year

concluded that the least costly alternative for sewage treatment in the Village of North Prairie is the continued reliance on onsite sewage disposal systems.

It may be expected that the approach recommended in the facilities plan will be reflected in the removal from the regional plan of the proposal for a public sanitary sewerage system in the Village of North Prairie. In anticipation of this, the year 2010 regional land use plan does not reflect public sanitary sewer service for the Village.

¹²The adopted regional water quality management plan proposed the development of a public sanitary sewerage system to serve the Village of North Prairie by the year 2000. The plan envisioned that sewage would be treated at a new plant serving the Village. A detailed sewerage facilities plan, documented in Village of North Prairie Wastewater Facility Plan, Phase One, 1986; and Village of North Prairie Wastewater Facility Plan, Phase Two, 1989, has since 248

MAJOR GOVERNMENTAL AND INSTITUTIONAL CENTERS IN THE REGION 2010 RECOMMENDED LAND USE PLAN



The map above shows the locations of the major governmental and institutional centers, including county seats, major State and Federal office buildings, major medical complexes, universities, technical colleges, major public libraries, and major cultural centers, envisioned under the recommended regional land use plan through the plan design year 2010. No new major governmental or institutional centers are envisioned. Additional development at the existing major centers would be limited to that necessary to meet the needs of the growing population.

Source: SEWRPC.

2010 regional land use plan, including public sewage treatment plants, major electric power generation plants, major airports, major bus and railway passenger stations, and the Milwaukee seaport, are shown on Map 68.

Open Space: Recreational Land Use

As indicated in Table 122, under the plan, about 4,100 acres of land would be added to the

MAJOR TRANSPORTATION AND UTILITY CENTERS IN THE REGION 2010 RECOMMENDED LAND USE PLAN



The major transportation and utility centers envisioned under the recommended year 2010 regional land use plan are shown on this map. They include public sewage treatment plants, major electric power generation plants, major airports, major bus and railway passenger stations, and the Milwaukee seaport. The plan envisions the development of only two new major utility centers in the Region through the plan design year, a new public sewage treatment plan serving the Village of Wales and a new peak-load electric power generation plant in the Town of Paris. The plan envisions the expansion of certain of the existing major utility and transportation centers in accordance with the adopted regional water quality management and regional airport system plans.

Source: SEWRPC.

existing stock of recreational land use in the Region. This represents an increase of about 16 percent over the 1985 acreage. It should be noted that the data in Table 122 pertain to "intensive use" areas, that is, land actually developed or anticipated to be developed as outdoor recreation facility areas. It should also be noted that the additional recreational land indicated in Table 122 represents only the

			Recreationa	I Land Use ^a		
	Existing 1985		Planned 1985	Increment -2010 ^b	Total 2010	
County	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total
Kenosha	2,749	10.8	415	15.1	3,164	10.7
Milwaukee	7,206	28.2	932	12.9	8,138	27.4
Ozaukee	1,809	7.1	272	15.0	2,081	7.0
Racine	2,391	9.4	390	16.3	2,781	9.4
Walworth	3,541	13.8	258	7.3	3,799	12.8
Washington	1,874	7.3	605	32.3	2,479	8.4
Waukesha	5,994	23.4	1,217	20.3	7,211	24.3
Region	25,564	100.0	4,089	16.0	29,653	100.0

EXISTING AND PROPOSED RECREATIONAL LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

^aIncludes only that land intensively used for recreational purposes.

^bIncludes only that increment which is for public recreational use.

Source: SEWRPC.

increase in land devoted to public recreational use. The additional recreational areas called for under the plan are based in part on neighborhood development standards, which seek to provide adequate neighborhood park land in developing areas. The recreational land use recommendations also reflect specific park site acquisition and development proposals set forth in the county park and open space plans recently prepared by the Commission for each of the seven counties in the Region.¹³

The year 2010 regional land use plan proposes a system of 31 major parks of regional size and significance to serve the needs of the Region through the year 2010. Such parks have an area of at least 250 acres and provide opportunities for a variety of resource-oriented outdoor recreational activities. Twenty-nine of the 31 sites were recommended as major park sites under the year 2000 regional land use plan. Of the 29 previously recommended sites, only two, Sugar Creek in Walworth County and Paradise Valley in Washington County, have yet to be publicly acquired.

The plan recognizes the development of two major parks not identified in the year 2000 plan: Mitchell Park, an approximately 800-acre site located in the City and Town of Brookfield, and an approximately 400-acre unnamed site surrounding a major lake recently created from an abandoned quarry in the Village of Pleasant Prairie. Facility development at these sites as envisioned in local site plans would qualify both sites as major parks. The recommended major

¹³The park and open space plans of the seven counties are documented in SEWRPC Community Assistance Planning Report No. 131, A Park and Open Space Plan for Kenosha County, 1987; SEWRPC Community Assistance Planning Report No. 132, A Park and Open Space Plan for Milwaukee County, 1991; SEWRPC Community Assistance Planning Report No. 133, A Park and Open Space Plan for Ozaukee County, 1987; SEWRPC Community Assistance Planning Report No. 134, A Park and Open Space Plan for Racine County, 1988; SEWRPC Community Assistance Planning Report No. 135, A Park and Open Space Plan for Walworth County, 1991; SEWRPC Community Assistance Planning Report No. 136, A Park and Open Space Plan for Washington County, 1989; and SEWRPC Community Assistance Planning Report No. 137, A Park and Open Space Plan for Waukesha County, 1989.

park sites, along with existing major special-use outdoor recreation sites in the Region, are listed in Table 123 and shown on Map 69.

Open Space: Environmental Corridors

The most important elements of the natural resource base of the Region, including the best remaining woodlands; wetlands; prairies; wildlife habitats; surface waters and associated shorelands and floodlands; and historic, scenic, and scientific sites, have been found to occur combined in linear patterns throughout the Region. These linear patterns of prime natural resource concentrations have been termed primary environmental corridors. The preservation and protection of these environmental corridors in accordance with regional development objectives is considered essential to the maintenance of a wholesome environment within the Region and the preservation of the unique cultural and natural heritage of the Region, as well as of its natural beauty.

In 1985, primary environmental corridor lands in the Region encompassed about 299,600 acres, or 17 percent of the total area of the Region. These corridors are generally located along major stream valleys, along the Lake Michigan shoreline, around major inland lakes, and in the Kettle Moraine area of the Region. The adopted year 2010 land use plan, proposes the preservation of the existing primary environmental corridor lands in essentially natural, open uses. Under the plan, development within these corridors would be limited to that needed to accommodate required transportation and utility facilities, compatible outdoor recreational facilities, and, on a limited basis, rural-density residential use.

In addition to the preservation of existing primary environmental corridor lands, the land use plan envisions that certain adjacent floodland areas currently in agricultural or other open use would be restored to a wetland condition, thereby becoming part of the environmental corridor network. These lands, which together encompass about 3,600 acres, have been recommended for county or State acquisition for open space preservation purposes under the aforementioned county park and open space plans.

The proposed environmental corridor network recommended under the year 2010 regional land use plan, including the existing corridors and the proposed additional areas, is shown on Map 64. The planned environmental corridors encompass 303,200 acres, or about 18 percent of the total area of the Region (see Table 124).

Open Space: Agricultural

and Other Open Land Uses

There were approximately 1,333,400 acres, or 2,083 square miles, of open land within the Region in 1985, including almost 932,000 acres of agricultural land and almost 401,500 acres of other open lands. These rural lands serve at least two important functions in the Region. As a land use, they provide open areas which lend form and shape to urban development; provide invaluable opportunities for passive recreation; and serve to preserve, protect, and enhance certain elements of the natural resource base. As the site of an economic activity, these lands provide employment opportunities in agriculture and the extractive industries of forestry and mining.

Under the adopted land use plan, the expansion of urban activities into present rural areas would result in the conversion of about 54,800 acres, or about 86 square miles, of rural land, to urban land uses between 1985 and 2010. This conversion would occur at an average annual rate of about 2,200 acres, or about 3.4 square miles. In addition to the conversion of rural land to urban land uses, the plan envisions that about 700 additional acres of rural land would be developed for rural estate use.

As indicated in Table 125, much of the urban expansion proposed under the recommended land use plan, 40,500 acres, would take place on lands now in agricultural use and would result in a decrease of about 4 percent in the existing stock of agricultural land within the Region. Among the seven counties, the greatest decline in agricultural land, 13,400 acres, would occur in Waukesha County. For the other six counties, the required conversion of agricultural land would range from about 3,500 acres in Walworth County to about 6,100 acres in Kenosha County.

The adopted plan seeks to hold the conversion of prime agricultural lands to urban land uses to a minimum. Prime agricultural lands are well suited for highly productive agricultural use. Accordingly, the recommended year 2010 land use plan proposes to convert to urban use only those prime agricultural lands which were already committed to urban development

SELECTED CHARACTERISTICS OF MAJOR PUBLIC OUTDOOR RECREATION CENTERS IN THE REGION: 2010 RECOMMENDED LAND USE PLAN

	Land Use Area in Acres							
		Gross Area ^a			Net Area ^b			
		Planned			Planned			
Major Public Outdoor	Existing	Increment	Total	Existing	Increment	Total		
Recreation Center	1985	1985-2010	2010	1985	1985-2010	2010		
Parks					1. A.			
Existing Brighton Dala	260		260	196		100		
Petrifving Springs	358	0	358	187		187		
Silver Lake	258	100	358	24	28	52		
Brown Deer	365	0	365	268	0	268		
Dretzka	327	0	327	305		305		
Lake Michigan-North	295 415	0	295	192	63	255		
Lake Michigan-South	841	ŏ	841	323	120	443		
Lincoln	312	0	312	226	0	226		
	278	0	278	242	0	242		
Harrington Beach	640	0	640	315	64	315		
Hawthorne Hills	285	ŏ	285	210	04	210		
Mee-Kwon	239	Ō	239 ^C	165	5	170		
Cliffside	223	315	538	69	79	148		
Johnson	357	57	357	250	0	250		
Whitewater Lake	250	0	250	86		86		
Pike Lake	678	58	736	28	32	60		
Menomonee	397	0	397	159	26	185		
Minooka	297	0	297	105	0	105		
	222	157	222° 573	141	U. 50	141		
Ottawa Lake	220	0	220 ^c	95	59 0	95		
Subtotal	8,941	693	9,634	4.057	476	4.533		
Existing-Site Acquired						.,		
But Not Developed			- 14 - 14 - 14 - 14 - 14 - 14 - 14 - 14					
Bender	308	127	435	0	159	159		
	239	0	239	0	40	40		
	194	248	442	0	32	32		
Subtotal	/41	375	1,116	0	231	231		
Proposed – New Site	0	270	270	0	100	1.00		
Sugar Creek	0	370	370		120	120		
Paradise Valley	ŏ	450	450	ŏ	89	89		
Mitchell	0	813	813	0	100	100		
Subtotal	0	1,938	1,938	0	447	447		
Subtotal Parks	9,682	3,006	12,688	4,057	1,154	5,211		
Special-Use Sites Existing		,						
Bong Recreation Area	4,515	460	4,975	138	26	164		
Maier Festival Park	51	0	51	51	0	51		
Milwaukee County Stadium	102	0	102	26		26		
Mitchell Conservatory	61		61	53	9	93		
State Fair Park	214	ŏ	214	159	ŏ	159		
Old World Wisconsin	450	0	450	77	4	81		
Subtotal Special-Use Sites	5,563	460	6,023	588	39	627		
Total	15,245	3,466	18,711	4,645	1,193	5,838		

^aIncludes entire site area.

^bIncludes only that land intensively used for recreation purposes.

^cSite abuts existing parkway lands or lands recommended for parkway acquisition. The area of the site proper in conjunction with the associated parkway lands exceeds 250 acres.



MAJOR OUTDOOR RECREATION CENTERS IN THE REGION: 2010 RECOMMENDED LAND USE PLAN

The map above shows the major park sites and major special use outdoor recreation sites envisioned under the recommended regional land use plan. The plan proposes a total of 31 major parks to meet the needs of the Region through the year 2010. Twenty four of the sites for these parks were publicly owned and developed in 1985, while three sites had been publicly acquired for park purposes in 1985 but not yet developed. The plan envisions four new major parks at sites in the City and Town of Brookfield, in the Village of Pleasant Prairie, at the long-recommended Paradise Valley site in Washington County, and at the Sugar Creek site in Walworth County,

Source: SEWRPC.

because of their proximity to existing and expanding concentrations of urban uses and prior commitment of heavy capital investments and utility extensions. The identified prime agricultural lands encompassed about 670,100 acres, or about 72 percent of all land in agricultural use in the Region in 1985. As indicated in Table 125, the adopted land use plan proposes to convert only about 10,300 acres, or just over

PRIMARY ENVIRONMENTAL CORRIDOR AREA IN THE REGION BY COUNTY 2010 RECOMMENDED LAND USE PLAN

County	Acres ^a	Percent of Tota	
Kenosha	28,900	9.5	
Milwaukee	10,300	3.4	
Ozaukee	19,900	6.6	
Racine	23,800	7.8	
Walworth	65,500	21.6	
Washington	60,900	20.1	
Waukesha	93,900	31.0	
Region	303,200	100.0	

^aThe planned environmental corridor area includes 299,600 acres within the existing primary environmental corridor configuration in the Region in 1985 and 3,600 additional acres within adjacent floodland areas that are currently in agricultural and other open use that are recommended to be restored to a wetland condition by the plan design year.

Source: SEWRPC.

1 percent, of the remaining prime agricultural lands to urban use by the year 2010. These prime agricultural lands are shown on Map 70.

In addition to agricultural lands, there were 401,500 acres of other open land uses in the Region in 1985, including woodlands, water, wetlands, quarries, landfill sites, and unused rural land. As indicated in Table 126, under the recommended land use plan, a total of 15,000 acres, or about 4 percent of the remaining acreage of these other open lands, would be converted to urban use by the year 2010.

Distribution of Population and Households

Under the adjusted intermediate regional growth scenario, the resident population of the Region would increase by about 168,000 persons, or about 10 percent, from about 1,743,000 persons in 1985 to about 1,911,000 persons by the year 2010. Under the recommended plan, the year 2010 regional population would be distributed among the seven counties as shown in Table 127. As indicated in Table 127, under the plan, Waukesha County would gain about 78,000 persons, while Kenosha, Ozaukee, Racine, Walworth, and Washington Counties would experience population increases ranging from 12,000 to 27,000 persons. Milwaukee County would experience some population loss.

EXISTING AND PROPOSED AGRICULTURAL LAND USE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

		Agricultural Land Use								
		Тс	otal			Pr	ime			
	Existing 1985 Planned Increment 1985-2010 2010		Total 2010	Existing 1985	Planned Increment 1985-2010		Total 2010			
County	(acres)	Acres	Percent	(acres)	(acres)	Acres	Percent	(acres)		
Kenosha	106,165	-6,057	-5.7	100,108	76,471	-2,042	-2.7	74,429		
Milwaukee	21,128	-4,072	-19.3	17,056	1,351	0	0.0	1,351		
Ozaukee	92,650	-4,016	-4.3	88,634	73,335	-1,425	-1.9	71,910		
Racine	137,196	-4,576	-3.3	132,620	98,626	-526	-0.5	98,100		
Walworth	249,705	-3,510	-1.4	246,195	208,941	-917	-0.4	208,024		
Washington	168,134	-4,900	-2.9	163,234	108,256	-1,463	-1.4	106,793		
Waukesha	156,978	-13,356	-8.5	143,622	103,078	-3,886	-3.8	99,192		
Region	931,956	-40,487	-4.3	891,469	670,058	-10,259	-1.5	659,799		

Source: SEWRPC.

Table 126

EXISTING AND PROPOSED OPEN LAND USES IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

		Open Land Use ^a						
	Existing 1985		Planned I 1985	ncrement -2010	Total 2010			
County	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total		
Kenosha	40,038	10.0	-1,560	-3.9	38,478	10.0		
Milwaukee	17,267	4.3	-2,081	-12.1	15,186	3.9		
Ozaukee	30,514	7.6	-1,204	-3.9	29,310	7.6		
Racine	40,377	10.0	-1,704	-4.2	38,673	10.0		
Walworth	81,169	20.2	-1,385	-1.7	79,784	20.6		
Washington	77,029	19.2	-1,827	-2.4	75,202	19.5		
Waukesha	115,068	28.7	-5,246	-4.6	109,822	28.4		
Region	401,462	100.0	-15,007	-3.7	386,455	100.0		

^aIncludes woodlands, water, wetlands, unused rural land, landfill sites, and quarries.

Source: SEWRPC.

Under the plan, the number of households in the Region would increase from about 644,000 in 1985 to about 774,000 by the year 2010, an overall increase of about 130,000 households, or about 20 percent. As previously indicated, in relative terms, the number of households in the Region would continue to grow at a faster rate than the regional population. As indicated in Table 128, under the plan, each county would experience a significant increase in the number of households between 1985 and 2010, ranging from just under 8,000 additional households in Ozaukee County to about 39,000 additional households in Waukesha County. Despite a slight decrease in population, as noted above, Milwaukee County would gain almost 33,000 households between 1985 and 2010, second only to Waukesha County.



ILLINOIS

The recommended year 2010 regional land use plan proposes the preservation in agricultural use of most of the remaining prime agricultural lands in Region. Under the plan, conversion of prime agricultural lands to urban use would be limited to those areas which were already committed to urban development because of their proximity to existing and expanding concentrations of urban uses and the prior commitment of heavy capital investments and utility extensions. Under the plan about 10,300 acres, or just over 1 percent of the remaining prime agricultural lands in the Region, would be converted to urban use by the year 2010.

EXISTING AND PROPOSED POPULATION DISTRIBUTION IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

	1985 Population		Planned 1985	Increment -2010	2010 Population	
County	Number	Percent of Total	Number	Percent	Number	Percent of Total
Kenosha	121,100	7.0	26,800	22.1	147,900	7.7
Milwaukee	939,600	53.9	-5,600	-0.6	934,000	48.9
Ozaukee	67,500	3.9	12,300	18.2	79,800	4.2
Racine	169,200	9.7	16,800	9.9	186,000	9.7
Walworth	72,200	4.1	15,100	20.9	87,300	4.6
Washington	87,200	5.0	24,500	28.1	111,700	5.8
Waukesha	285,900	16.4	78,400	27.4	364,300	19.1
Region	1,742,700	100.0	168,300	9.7	1,911,000	100.0

Source: SEWRPC.

Table 128

EXISTING AND PROPOSED HOUSEHOLD DISTRIBUTION IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

County	1985 Households		Planned I 1985	ncrement -2010	2010 Households	
	Number	Percent of Total	Number	Percent	Number	Percent of Total
Kenosha	44,200	6.9	14,900	33.7	59,100	7.6
Milwaukee	368,200	57.2	32,800	8.9	401,000	51.8
Ozaukee	22,900	3.5	7,600	33.2	30,500	3.9
Racine	61,200	9.5	12,700	20.8	73,900	9.6
Walworth	25,600	4.0	10,000	39.1	35,600	4.6
Washington	28,500	4.4	13,100	46.0	41,600	5.4
Waukesha	93,200	14.5	39,400	42.3	132,600	17.1
Region	643,800	100.0	130,500	20.3	774,300	100.0

Source: SEWRPC.

As further indicated in Tables 127 and 128, the relative distribution of population and households among the seven counties would change somewhat, with Milwaukee and Waukesha Counties most affected. Under the plan, Milwaukee County's share of the regional population would decrease from 54 percent to 49 percent, while Waukesha County's share would increase from just over 16 percent to about 19 percent. Similarly, Milwaukee County's share of all households in the Region would decrease from 57 percent to 52 percent, while Waukesha County's share would increase from just over 14 percent to about 17 percent. For each of the other five counties in the Region, the relative share of the total population and households in the Region would change by 1 percent or less.

The land use plan proposes an amount of urban land use sufficient to accommodate the anticipated future population and household levels in the Region through the plan design year. The increase in the amount of urban land proposed is compared to the anticipated increases in

	Increment: 1985-2010							
	Urban La	and Area	a Population		House	ouseholds		
County	Acres Percent		Number	Percent	Number	Percent		
Kenosha	7,533	23.6	26,800	22.1	14,900	33.7		
Milwaukee	6,079	5.2	-5,600	-0.6	32,800	8.9		
Ozaukee	5,165	18.9	12,300	18.2	7,600	33.2		
Racine	6,181	15.3	16,800	9.9	12,700	20.8		
Walworth	4,674	12.3	15,100	20.9	10,000	39.1		
Washington	6,659	19.8	24,500	28.1	13,100	46.0		
Waukesha	18,482	18.6	78,400	27.4	39,400	42.3		
Region	54,773	14.1	168,300	9.7	130,500	20.3		

INCREMENTAL URBAN LAND USE, POPULATION, AND HOUSEHOLDS IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

Source: SEWRPC.

population and households in Table 129. In total, the recommended land use plan would accommodate the approximately 10 percent increase in population and 20 percent increase in households with an approximately 14 percent increase in urban land uses.

As indicated in Table 130 and Figure 55, the population density in the developed area of the Region under the adopted plan would decline from the 1985 level of about 3,600 persons per square mile to a year 2010 level of about 2,800 persons per square mile, continuing the trend toward declining densities evident in the Region since 1920. The rate of decline would be reduced, however, by implementation of the plan proposals to develop the majority of new residential land within the Region at medium, instead of low, densities and to provide such development with public sanitary sewer and water supply services.

Employment Distribution

Under the adjusted intermediate regional growth scenario, the total number of jobs in the Region would increase from about 872,000 in 1985 to about 1,095,000 in 2010, an increase of about 223,000 jobs, or 26 percent. Under the recommended plan, each county would gain a significant number of jobs between 1985 and 2010 (see Table 131). Employment in Milwaukee County would increase by about 85,000 jobs, more than in any other county. Employment in Waukesha County would increase by almost 59,000 jobs. Among the other five counties in the Region, planned employment increases would range from just under 12,000 jobs in Ozaukee County to almost 21,000 jobs in Kenosha County.

As indicated in Table 131, on a relative basis, employment would increase at a faster rate in outlying counties of the Region than in Milwaukee County. While the plan seeks to centralize employment to the extent practicable, Milwaukee County's share of the total regional employment would, nevertheless, continue to decline somewhat, from just over 60 percent in 1985 to about 56 percent in 2010. Conversely, Waukesha County's share of total regional employment would increase from about 16 percent to about 18 percent. For each of the other five counties, the share of total regional employment would change by less than 1 percent between 1985 and 2010.

Public Sanitary Sewer and Water Supply Service Under the adopted land use plan, all proposed new urban development within the Region would be served with public sanitary sewer and water supply facilities (see Map 71). In addition, public sanitary sewer and water supply service would be extended to certain urban areas lacking these facilities in 1985. In 1985, about 320 square miles, or about 67 percent of the total developed urban area of the Region, and about 1.51 million persons, or almost 87 percent of the resident population of the Region, were served by public sanitary sewer facilities (see Table 132). About 263 square miles, or about 55 percent of the

	Urban Population		Rural Population			Area (square miles)		Persons per Square Mile	
Year	Number	Percent of Total	Number	Percent of Total	Total Population	Urban	Total	Urban	Total
1850	28,623	25.2	84,766	74.8	113,389	4	2,689	7,156	42.2
1880	139,509	50.3	137,610	49.7	277,119	18	2,689	7,751	103.1
1900	354,082	70.6	147,726	29.4	501,808	37	2,689	9,570	186.6
1920	635,376	81.1	148,305	18.9	783,681	56	2,689	11,346	291.4
1940	991,535	92.9	76,164	7.1	1,067,699	90	2,689	11,017	397.1
1950	1,179,084	95.0	61,534	5.0	1,240,618	146	2,689	8,076	461.4
1963	1,634,200	97.6	40,100	2.4	1,674,300	282	2,689	5,795	622.6
1970	1,728,946	98.5	27,137	1.5	1,756,083	338	2,689	5,115	653.1
1980	1,749,238	99.1	15,558	0.9	1,764,796	444	2,689	3,940	656.3
1985	1,730,500	99.3	12,200	0.7	1,742,700	477	2,689	3,628	648.1
2010	1,902,800	99.6	8,200	0.4	1,911,000	668	2,689	2,849	710.7

POPULATION DENSITY IN THE REGION: SELECTED YEARS, 1850-1985, AND 2010 RECOMMENDED LAND USE PLAN

Source: U. S. Bureau of the Census and SEWRPC.

Figure 55

URBAN POPULATION DENSITY

IN THE REGION: ACTUAL 1850-1985



Source: SEWRPC.

developed area of the Region, and about 1.39 million persons, or about 80 percent of the resident population of the Region, were served by public water supply facilities. Under the adopted plan, about 570 square miles, or about 85 percent of the developed urban area, and about 1.74 million persons, or about 91 percent of the total population, would be served by public sanitary sewer facilities and public water supply facilities by the plan design year.

The developed urban area and the population levels which would be served by public sanitary sewer and water supply service under the adopted plan is summarized by county in Table 133. The developed area served by sanitary sewer and water supply service by the year 2010 would range from about 57 percent in Washington County to nearly 100 percent in Milwaukee County. The proportion of the resident population so served would range from a low of about 67 percent in Walworth and Washington Counties to a high of almost 100 percent in Milwaukee County.

ALTERNATIVE LAND USE PLANS

In view of the continuing uncertainty surrounding future social and economic conditions in the Region, it would be imprudent to dismiss the possibility of future growth and change in the Region occurring in a manner significantly at variance with that envisioned under the adopted plan. Accordingly, a determination was made to prepare alternative land use plans. Four alternative land use plans were prepared: a high-growth decentralized plan, a high-growth centralized plan, an intermediate-growth decentralized plan,
EXISTING AND PROPOSED EMPLOYMENT DISTRIBUTION IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

	1985 Em	ployment	Planned I 1985-	ncrement -2010	2010 Emp	loyment
County	Number	Percent of Total	Number	Percent	Number	Percent of Total
Kenosha	42,500	4.9	20,500	48.2	63,000	5.7
Milwaukee	527,300	60.5	85,400	16.2	612,700	56.0
Ozaukee	26,900	3.1	11,800	43.9	38,700	3.5
Racine	74,500	8.5	17,600	23.6	92,100	8.4
Walworth	28,100	3.2	12,400	44.1	40,500	3.7
Washington	31,300	3.6	16,600	53.0	47,900	4.4
Waukesha	141,300	16.2	58,800	41.6	200,100	18.3
Region	871,900	100.0	223,100	25.6	1,095,000	100.0

Source: SEWRPC.

Table 132

EXISTING AND PROPOSED DEVELOPED AREA AND POPULATION SERVED BY PUBLIC SANITARY SEWER AND WATER SUPPLY SERVICE IN THE REGION: 1985 AND 2010 RECOMMENDED LAND USE PLAN

	Existing	Service	Planned	Service	Total S	Service
	19	85	Incre	ment	20	10
Area and Population	Public Sanitary Sewer	Public Water Supply	Public Sanitary Sewer	Public Water Supply	Public Sanitary Sewer	Public Water Supply
Developed Area ^a Total Square Miles Square Miles Served Percent of Total Served	476.5 319.8 ^b 67.1	476.5 262.9 ^c 55.2	191.3 250.5	191.3 309.8 	667.8 570.3 ^d 85.4	667.8 572.7 ^d 85.8
Population Total Population Population Served Percent of Total Served	1,742,700	1,742,700	168,300	168,300	1,911,000	1,911,000
	1,507,800	1,389,700	231,300	352,800	1,739,100	1,742,500
	86.5	79.7			91.0	91.2

NOTE: Public sanitary sewer and water supply service areas presented in this table do not include lands that are located adjacent to, but outside, the Region, including 1.1 square miles of land in the Jefferson County portion of the Whitewater urban service area, 0.4 square mile of land in the Jefferson County portion of the Oconomowoc urban service area, and 0.4 square mile of land in the Dodge County portion of the Hartford urban service area.

^aBased on historic urban growth analysis; see Table 67 of SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for</u> <u>Southeastern Wisconsin—2010</u>.

^bDoes not include 57.1 square miles of land served with public sanitary sewer located outside the 1985 developed urban area.

^cDoes not include 30.1 square miles of land served with public water supply located outside the 1985 developed urban area.

^dDoes not include 4.9 square miles of land to be served with public sanitary sewer and water supply service located outside the planned 2010 developed urban area.



The recommended regional land use plan proposes to serve essentially all new urban development within the Region with public sanitary sewer and public water supply service. About 570 square miles, or 85 percent of the developed urban area of the Region, and about 1.74 million persons, or 91 percent of the total regional population, would be served with public sanitary sewer and water supply facilities by the year 2010. As shown above, public water supply would be provided in several outlying communities for which public sanitary sewer service is not planned.

EXISTING AND PROPOSED DEVELOPED AREA AND POPULATION SERVED BY PUBLIC SANITARY SEWER AND WATER SUPPLY SERVICE IN THE REGION BY COUNTY: 1985 AND 2010 RECOMMENDED LAND USE PLAN

								and the state	<u> </u>			,		
			1 - E		Existing 198	5		· · ·	10		$P^{(1)}$	Planned 20	10	
	-		Public Se	ewer Service			Public Water	Supply Servi	ice			Public Water Su	Sewer and pply Service ^a	
	Developed Area ^e	Dev Area	Developed Area Served Population Served Jare Percent iles of County Number of County				reloped Served ^C	Popul Ser	ation . ved	Developed Area	Dev Area	veloped Served ^d	Popul	ation ved
County	(square miles)	Square Miles	Percent of County	Number	Percent of County	Square Miles	Percent of County	Number	Percent of County	(square miles)	Square Miles	Percent of County	Number	Percent of County
Kenosha	34.7	24.9	71.8	101,800	84.1	17.6	50.7	86,700	71.6	67.5	63.5	94.1	135,300	91.5
Milwaukee	167.1	162.2	97.1	933,100	99.3	155.1	92.8	915,000	97.4	203.5	202.0	99.3	932,500	99.8
Ozaukee	29.7	17.0	57.2	50,700	75.1	8.1	27.3	33,800	50.1	45.2	39.5	87.4	66,400	83.2
Racine	45.0	33.7	74.9	144,300	85.3	24.2	53.8	126,500	74.8	69.2	61.3	88.6	167.300	89.9
Walworth	32.8	13.8	42.1	41,200	57.1	11.5	35.1	37,100	51.4	44.7	30.9	69.1	59,100	67.7
Washington	36.6	10.9	29.8	45,400	52.1	10.6	29.0	43,900	50.3	54.0	31.0	57.4	75,100	67.2
Waukesha	130.6	57.3	43.9	191,300	66. 9	35.8	27.4	146,700	51.3	183.7	142.1	77.4	303,400	83.3
Region	476.5	319.8	67.1	1,507,800	86.5	262.9	55.2	1,389,700	79.7	667.8	570.3	85.4	1,739,100	91.0

NOTE: Public sanitary sewer and water supply service areas presented in this table do not include lands that are located adjacent to, but outside, the Region, including 1.1 square miles of land in the Jefferson County portion of the Whitewater urban service area, 0.4 square mile of land in the Jefferson County portion of the Oconomowoc urban service area, and 0.4 square mile of land in the Dodge County portion of the Hartford urban service area.

^aDoes not include an area of 2.4 square miles and a population of 3,400 persons in those communities which, in the year 2010, would have public water supply systems but not public sanitary sewer service.

^bDoes not include 57.1 square miles of land served with public sanitary sewer located outside the 1985 developed urban area.

^cDoes not include 30.1 square miles of land served with public water supply located outside the 1985 developed urban area.

^dDoes not include 4.9 square miles of land to be served with public sanitary sewer and water supply service located outside the planned 2010 developed urban area.

^eBased on historic urban growth analysis; see Table 67 of SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin-2010</u>.

Source: SEWRPC.

and a low-growth decentralized plan. Such plans differ from the adopted year 2010 land use plan in terms of the overall scale of development to be accommodated and the distribution of such development within the Region.

Together, the alternative plans, along with the adopted plan, provide a range of possible future conditions with respect to the level and distribution of population and economic activity and attendant land use patterns in the Region. The plans therefore broaden the framework within which local level and systems level transportation planning and other major facility planning can take place. Within this framework, for example, proposals for major transportation improvements may be evaluated to determine how well they would perform under a range of possible future conditions. Through such sensitivity analyses more robust plan elements which may be expected to remain viable under greatly varying conditions can be identified.

Design Year Population and Economic Activity Levels

The alternative land use plans were designed to accommodate the population and employment levels envisioned under the respective scenarios. Thus, the high-growth decentralized and highgrowth centralized land use plans would accommodate a year 2010 resident population of about 2.316,100 persons and a total of about 1.251,600 jobs. The intermediate-growth decentralized plan would accommodate a year 2010 population of about 1,872,200 persons and 1,051,000 jobs. The low-growth decentralized plan would accommodate a year 2010 population of about 1,517,000 persons and 871,000 jobs. In comparison, the adopted land use plan would accommodate a year 2010 population of 1,911,000 persons and an employment total of 1,095,000 jobs.

Planned Distribution of

Population and Economic Activity

While the alternative land use plans differ from the adopted land use plan in the scale and distribution of population and economic activity, and, accordingly, in the amount of and location of new urban development, the alternative land use plans incorporate certain key normative aspects of the adopted plan. Specifically, the following design guidelines, used in the preparation of the adopted plan, were also used in the preparation of the alternative futures plans:

- 1. New urban development would emphasize medium densities and would, for the most part, be located in areas of the Region provided with centralized sanitary sewer and water supply services.
- 2. No new urban development would be allocated to the delineated primary environmental corridors, thereby preserving the best remaining elements of the natural resource base of the Region.
- 3. To the greatest extent practicable, no new urban development would be allocated to the delineated prime agricultural lands, thereby preserving highly productive lands for the continuing production of food and fiber.

The alternative land use plans, like the adopted plan, would preserve virtually all remaining primary environmental corridors in the Region. The alternative plans, like the adopted plan, also emphasize the provision of basic public utilities and services to new urban development and the preservation of prime agricultural lands. The alternative plans do differ from one another and from the recommended plan in the extent to which such objectives can be met, owing to differences in the scale and distribution of development to be accommodated.

It is important to recognize that the "decentralization" of population and economic activity under the high-growth, intermediate-growth, and low-growth decentralized plans was accommodated within the framework of the aforementioned plan design guidelines. Each of these plans envisions the decentralization of population and economic activity on a county basis, with Milwaukee County continuing to lose population and employment to Ozaukee, Walworth, Washington, and Waukesha Counties, and the population and employment of the Kenosha and Racine urbanized areas decentralizing to the outlying areas of Kenosha and Racine Counties. While thus emphasizing decentralization of population and economic activity on a county basis, each of these plans envisions that new urban development occurring in the outlying counties and outlying areas of Kenosha and Racine Counties would be concentrated in and around existing urban centers of those counties and be provided with basic urban services and facilities from and by those centers. These plans thus represent controlled decentralization and do not represent uncontrolled urban sprawl.

Plan Description

The land use patterns envisioned under the highgrowth decentralized, high-growth centralized, intermediate-growth decentralized, and lowgrowth decentralized alternative land use plans are shown on Maps 72 through 75. The balance of this chapter provides a description and comparison of these four alternative futures plans. For the purpose of comparing the various plans with the adopted plan, each of the tables accompanying the text below provides data on both the alternative plans and adopted plan.

Land Use: Of particular importance in a comparison of the plans is the anticipated change in urban land uses. As indicated in Table 134, under the high-growth decentralized plan, urban lands would increase by about 110,000 acres, or about 28 percent, from about 388,000 acres in 1985 to about 498,000 acres by the year 2010. Urban lands would increase by about 86,000 acres, or 22 percent, under the high-growth centralized plan; by about 67,000 acres, or about 17 percent, under the intermediate-growth decentralized plan; and by about 33,000 acres, or about 9 percent, under the low-growth decentralized plan. The amounts of urban land envisioned under the alternative futures plans and the recommended plan are set forth by county in Figure 56 and in Table 135.

<u>Major Centers</u>: Because of the differences in the level and distribution of population and employment to be accommodated, the alternative futures plans differ from each other and from the recommended year 2010 regional land use plan in terms of the number and location of planned major commercial and industrial centers.¹⁴

¹⁴Proposals for the major governmental and institutional, major transportation and utility, and major recreational centers are the same as those of the recommended land use plan.

EXISTING AND PROPOSED LAND USE IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

		Rec	ommended	Plan	Dec	Low-Growi centralized	th Plan	Inter Der	mediate-Gr centralized	rowth Plan	⊢ Dec	ligh-Growtl entralized I	n Plan	C	High-Grow entralized f	th Plan
	Existing 1985	Plann Increm 1985-2	ed ent 2010	Total 2010	Plann Increm 1985-2	ed ient 2010	Total 2010	Plann Incren 1985-2	ned nent 2010	Total 2010	Plann Increm 1985-2	ed ent 010	Total 2010	Plann Increm 1985-2	ied ient 2010	Total 2010
Land Use Category	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)
Urban Land Use																
Residential	184,603	36,976	20.0	221,579	19,170	10.4	203,773	43,927	23.8	228,530	72,269	39.1	256,872	56,960	30.9	241,563
Commercial	8,714	1,320	15.1	10,034	739	8.5	9,453	1,581	18.1	10,295	2,317	26.6	11,031	1,852	21.3	10,566
Industrial	12,080	5,186	42.9	17,266	4,005	33.2	16,085	6,650	55.0	18,730	9,586	79.4	21,666	9,041	74.8	21,121
Transportation,																
Communication,																
and Utilities ^a	120,279	14,560	12.1	134,839	8,994	7.5	129,273	16,947	14.1	137,226	27,736	23.1	148,015	22,377	18.6	142,656
Governmental and																
Institutional	17,240	1,042	6.0	18,282	591	3.4	17,831	1,168	6.8	18,408	2,019	11.7	19,259	1,639	9.5	18,879
Recreational	25,564 ^b	4,089 ^C	16.0	29,653	3,783 ^C	14.8	29,347	4,359 ^c	17.1	29,923	5,551 ^C	21.7	31,115	4,943 ^C	19.3	30,507
Unused Urban Land	19,215	-8,400	-43.7	10,815	-4,496	-23.4	14,719	-7,363	-38.3	11,852	-9,239	-48.1	9,976	-10,835	-56.4	8,380
Subtotal	387,695	54,773	14.1	442,468	32,786	8.5	420,481	67,269	17.4	454,964	110,239	28.4	497,934	85,977	22.2	473,672
Bural Land Use																
Residential	d	721		721	246		246	1,368		1,368	1.368		1,368	738		738
Agricultural	931.956	-40,487	-4.3	891,469	-24.763	-2.7	907,193	-52,123	-5.6	879,833	-90,242	-9.7	841.714	-67.863	-7.3	864.093
Other Open Lands ^e	401,462	-15,007	-3.7	386,455	-8,269	-2.1	393,193	-16,514	-4.1	384,948	-21,365	-5.3	380,097	-18,852	-4.7	382,610
Subtotal	1,333,418	-54,773	-4.1	1,278,645	-32,786	-2.5	1,300,632	-67,269	-5.0	1,266,149	-110,239	-8.3	1,223,179	-85,977	-6.4	1,247,441
Total	1,721,113	0	0.0	1,721,113	0	0.0	1,721,113	0	0.0	1,721,113	0	0.0	1,721,113	0	0.0	1,721,113

^aIncludes off-street parking areas.

^bIncludes net site area of public and nonpublic recreation sites.

^CIncludes only that net site area recommended for public recreation use.

^dIncluded in 1985 land use inventory as part of urban residential land use.

^eIncludes woodlands, water, wetlands, unused rural land, landfill sites, and quarries.

HIGH-GROWTH DECENTRALIZED PLAN FOR THE SOUTHEASTERN WISCONSIN REGION: 2010



The high-growth decentralized plan envisions a relatively high rate of growth in population and economic activity in the Region and a continuation of the trend observed over the past several decades toward decentralization of population, employment, and attendant urban development away from the older urban cities of the Region. The plan envisions the conversion of about 110,000 acres, or 172 square miles, of land from rural to urban use in order to accommodate the growth and redistribution of population and economic activity in the Region through the year 2010. Under this plan, the resident population of the Region would increase substantially, to a level of about 1,252,000 jobs by the year 2010, an increase of 380,000 jobs, or 44 percent, over 1985. Under the plan much of the proposed new urban development would occur in outlying areas of the Region.

HIGH-GROWTH CENTRALIZED PLAN FOR THE SOUTHEASTERN WISCONSIN REGION: 2010



The high-growth centralized land use envisions a relatively high rate of growth in population and economic activity along with a moderation of the past trends toward decentralization of population and economic activity within the Region. Under this plan, about 86,000 acres, or 134 square miles, of land would be converted from rural to urban use between 1985 and the year 2010. Like the high-growth decentralized plan, this plan would accommodate an increase of about 573,000 persons, or 33 percent, in the resident population of the Region and an increase of 380,000, or 44 percent, in total employment. In contrast to the high-growth decentralized plan, however, this plan emphasizes the maintenance and enhancement of the population and employment levels of the older, large urban centers of the Region and would result in a more centralized distribution of population and economic activity in the Region.

INTERMEDIATE-GROWTH DECENTRALIZED PLAN FOR THE SOUTHEASTERN WISCONSIN REGION: 2010



The intermediate-growth decentralized plan envisions a moderate increase in population and economic activity within the Region along with a continuation of the trend toward the decentralization of population, employment, and attendant urban development away from the older urban cities in favor of outlying areas. The intermediate-growth decentralized plan envisions the conversion of 67,000 acres, or 105 square miles, of land from rural to urban use in the Region between 1985 and the year 2010. Under this plan, the resident population of the Region would increase by about 129,000 persons, or about 7 percent, to a level of 1,872,000 by the year 2010, while total employment would increase by 179,000 jobs, or about 21 percent, to a level of 1,051,000.

LOW-GROWTH DECENTRALIZED PLAN FOR THE SOUTHEASTERN WISCONSIN REGION: 2010



ILLINOIS The low-growth decentralized plan envisions a declining regional population and stagnating economic activity along with the continued decentralization of population and economic activity away from the older urban areas of the Region. The low-growth decentralized plan would accommodate a plan design year 2010 population of about 1,517,000 persons, a decrease of 226,000 persons, or about 13 percent, from the 1985 level, and total regional employment of 871,000 jobs, about the same as the 1985 level. Despite the anticipated decrease in population, the number of households would increase by about 32,000, or by about 5 percent, as household sizes continue to decrease. The plan envisions the conversion of about 33,000 acres, or 51 square miles, of land from rural to urban use to accommodate the continued redistribution of population and employment and the modest increase in the number of households. *Source: SEWRPC.*

Figure 56



EXISTING AND PROPOSED URBAN LAND USE IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

Table 135

EXISTING AND PROPOSED URBAN LAND USE IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

								Urban La	nd Use ^a							
		Rec	:ommended	Plan	De	Low Growtl centralized	n Plan	Inte De	mediate-Gr centralized	owth Plan	De	High-Growth centralized f	i Plan		High-Growt Centralized F	th Plan
	Existing	Planı İncrer 1985-	ned ment 2010	Total 2010	Plan Increr 1985-	ned ment 2010	Total 2010	Plan Increr 1985-	ned nent 2010	Total 2010	Plan Increi 1985-	ned nent 2010	Total 2010	Plar Incre 1985	nned ment -2010	Total 2010
County	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)
Kenosha	31,971	7,533	23.6	39,504	2,739	8.6	34,710	3,997	12.5	35,968	9,430	29.5	41,401	9,013	28.2	40,984
Milwaukee	116,795	6,079	5.2	122,874	3,767	3.2	120,562	5,653	4.8	122,448	6,379	5.5	123,174	12,650	10.8	129,445
Ozaukee	27,292	5,165	18.9	32,457	4,126	15.1	31,418	6,773	24.8	34,065	13,926	51.0	41,218	7,466	27.4	34,758
Racine	40,340	6,181	15.3	46,521	1,762	4.4	42,102	4,746	11.8	45,086	10,430	25.9	50,770	9,747	24.2	50,087
Walworth	38,082	4,674	12.3	42,756	2,744	7.2	40,826	6,512	17.1	44,594	12,134	31.9	50,216	7,061	18.5	45,143
Washington	33,670	6,659	19.8	40,329	4,582	13.6	38,252	11,610	34.5	45,280	18,637	55.4	52,307	11.741	34.9	45,411
Waukesha	99,545	18,482	18.6	118,027	13,066	13.1	112,611	27,978	28.1	127,523	39,303	39.5	138,848	28,299	28.4	127.844
Region	387,695	54,773	14.1	442,468	32,786	8.5	420,481	67,269	17.4	454,964	110,239	28.4	497,934	85,977	22.2	473,672

⁸Urban land uses include residential; commercial; industrial; transportation, communication, and utility; governmental and institutional; and recreational land uses and unused urban lands.

						_							
,		Exis	ting	Recom	nended	Low-C	Growth tralized	Intermed Dece	iate-Growth ntralized	High-G Decen	Growth tralized	High-C Centr	irowth alized
		19	85	Plan:	2010	Plan:	2010	Plan	2010	Plan:	2010	Plan:	2010
County	Major Commercial Center	Retail	Office	Retail	Office	Retail	Office	Retail	Office	Retail	Office	Retail	Office
Kenosha	Kenosha CBD		x		x		x		x		x		x
	Kenosha-Southwest								X		X		X
	Kenosha-West			x		х		х		x		X	
Milwaukee	Bay Shore	x		x		х		x		x		×	•••
	Capitol Court	X		X						х		X	
	Mayfair	X	x	x	x	х	X	x	X	х	х	X	X
	Milwaukee CBD	x	x	x	x	X	×	х	X X	х	х	X	X
	Milwaukee County Research Park				x		x		X X		х		X
	Northridge	x		x		х		х		x		x	
	Oak Creek									X		x	
	Park Place				X		X		x		x		X
	Southgate-Point Loomis	х		x						x		x	
	Southridge	X		x		х		X		x		X	
	West Allis	x		×		х		x		x		x	••
Ozaukee	Mequon				x		x		x		x		x
Racine	Racine CBD		x		x		x		x		x		x
	Racine-West									X			
	Regency Mall	X		х		x		X		x		x	
Walworth	Delavan									x			
Washington	West Bend		x		x		x		x		×		x
Waukesha	Blue Mound Road	x	x	x	x	х	x	x	x	x	x	x	x
	Oconomowoc										X		
	Pewaukee				X				X		X		x
	Waukesha CBD		X		X		X		x		X		х

EXISTING AND PROPOSED MAJOR COMMERCIAL CENTERS IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

NOTE: To qualify as a major retail center, a site must accommodate at least 2,000 retail jobs. To qualify as a major office center, a site must accommodate at least 3,500 office and service-related jobs.

Source: SEWRPC.

<u>Major Commercial Centers</u>: The major commercial centers envisioned under the alternative futures land use plans and the recommended year 2010 land use plan are shown on Map 76 and listed in Table 136.

The high-growth decentralized plan envisions a total of 24 major commercial centers in the Region in the year 2010. The plan envisions the retention of the 14 existing major centers and the addition of 10 new centers, including four retail centers and six office centers. As shown on Map 76, a number of the proposed new sites would be located in outlying areas of the Region.

The other high-growth plan, the high-growth centralized plan, envisions a total of 21 major commercial centers. That plan envisions the retention of the 14 existing major centers and the addition of seven new centers, including two retail centers and five office centers, by the year 2010.

The intermediate-growth decentralized plan envisions a total of 18 major commercial centers. The plan envisions the retention of 12 of the 14 existing major centers and the addition of six new centers, including one retail center and five office centers. Under the intermediate-growth decentralized plan, with the continued decrease in population and employment levels in the central portion of Milwaukee County, two existing major centers, the Capitol Court shopping center and the Southgate-Point Loomis shopping center, would decline in importance and no longer function as major commercial centers. They would, however, continue to serve as community-level commercial centers.

The low-growth decentralized plan envisions a total of 16 major commercial centers. This plan envisions the retention of 12 of the 14 existing major centers and the addition of four new centers, including one retail center and three office centers. This plan, like the intermediate-



This map shows the locations of the major commercial centers envisioned under the alternative futures land use plans and the recommended year 2010 regional land use plan. Such sites include major retail centers accommodating retail employment of at least 2,000 jobs and major office centers accommodating office and service related employment of at least 3,500 jobs. Twenty-four major commercial centers are envisioned under the high-growth decentralized plan compared to 21 centers under the highgrowth centralized plan, 18 centers under the intermediate-growth decentralized plan, and 16 centers under the low-growth decentralized plan. The recommended regional land use plan envisions 19 major commercial centers in the Region by the year 2010.

growth decentralized plan, envisions that two existing centers in Milwaukee County, Capitol Court and Southgate-Point Loomis, would decline in importance and cease to function as major commercial centers.

<u>Major Industrial Centers</u>: The major industrial centers envisioned under the alternative futures plans and the recommended year 2010 regional land use plan are shown on Map 77 and listed in Table 137.

The high-growth decentralized plan envisions a total of 33 major industrial centers in the Region by the year 2010. The plan envisions the retention of all 22 existing major industrial centers and the addition of 11 new centers. Many of the new sites envisioned under the plan are located in outlying areas of the Region.

The high-growth centralized plan envisions a total of 27 major industrial centers in the Region. The plan envisions the retention of all 22 existing major industrial centers and the addition of five new centers, six fewer than would be added under the high-growth decentralized plan.

The intermediate-growth decentralized plan envisions a total of 22 major industrial centers. The plan envisions the retention of 18 of the 22 existing centers and the addition of four new centers. Under this plan, four existing centers, Kenosha, Milwaukee-South, West Allis-East,¹⁵ and West Milwaukee, would experience substantial employment losses and cease functioning as major industrial centers.

The low-growth decentralized plan envisions a total of 17 major industrial centers. The plan envisions the retention of 14 of the 22 existing major industrial centers and the addition of three new centers. Under the plan, eight existing centers, Kenosha, Cudahy-South Milwaukee, Milwaukee-Glendale, Milwaukee-Menomonee Valley West, Milwaukee-Near North, Milwaukee-South, West Allis-East, and West Milwaukee, would decline in importance and cease functioning as major centers. <u>Primary Environmental Corridors</u>: In the design of the alternative futures land use plans, as in the design of the adopted plan, no new urban development was allocated to the delineated primary environmental corridors.

<u>Prime Agricultural Lands</u>: Under each of the alternative futures plans, substantial amounts of agricultural land would be converted to urban use to accommodate expanding urban areas. In the design of the alternative futures plans, however, an attempt was made to minimize the loss of prime agricultural lands, while accommodating required urban development.

The anticipated losses of prime agricultural land under the alternative futures plans and under the recommended regional land use plan are indicated in Table 138. As indicated, from 1985 to 2010, the high-growth decentralized plan envisions the loss of about 34,900 acres, or about 5 percent of the remaining prime agricultural lands within the Region; the high-growth centralized plan envisions a loss of 20,400 acres, or 3 percent; the intermediate-growth decentralized plan envisions a loss of about 17,500 acres, or just under 3 percent; and the low-growth decentralized plan envisions a loss of 6,700 acres, or about 1 percent.

<u>Public Sanitary Sewer and Water Supply Services</u>: Under both of the alternative futures plans and the adopted land use plan, all proposed new urban development within the Region would be served with public sanitary sewer and water supply facilities. In addition, under each plan, public sanitary sewer and water supply service would be extended to certain existing urban areas lacking these facilities in 1985. Table 139 shows the land area and population served in the Region in 1985 and under the year 2010 adopted and alternative futures land use plans.

As indicated in Table 139, among the alternative futures plans, the high-growth decentralized plan envisions the greatest expansion of sanitary sewer and water supply service. Under that plan, about 734 square miles, or about 91 percent of the developed urban area, and about 2.17 million persons, or about 94 percent of the resident population of the Region, would be served with public sanitary sewer and public water supply facilities by the plan design year. Under the high-growth centralized plan, about 653 square miles, or 88 percent of the developed area of the Region, and about 2.14 million

¹⁵Monitoring of industrial employment data indicates that by 1990 industrial employment at the West Allis-East site had fallen below the threshold level of 3,500 jobs required for classification as a major industrial center.



Shown on this map are the locations of the major industrial centers envisioned under the alternative futures land use plans and the recommended year 2010 regional land use plan. Major industrial centers are defined as concentrations of industrial land having industrial related employment of at least 3,500 jobs. The high-growth decentralized plan envisions a total of 33 major industrial centers, compared to 27 centers under the high-growth centralized plan; 22 centers under the intermediate-growth decentralized plan; and 17 centers under the low-growth decentralized plan. The recommended regional land use plan envisions a total of 25 major industrial centers in the plan design year 2010.

EXISTING AND PROPOSED MAJOR INDUSTRIAL CENTERS IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

					· · · · · · · · · · · · · · · · · · ·		
				Low-Growth	Intermediate-Growth	High-Growth	High-Growth
		Existing	Recommended	Decentralized	Decentralized	Decentralized	Centralized
County	Major Industrial Center	1985	Plan: 2010	Plan: 2010	Plan: 2010	Plan: 2010	Plan: 2010
Kenosha	Kenosha	x	x			x	x
	Pleasant Prairie		x	x	x	×	x
Milwaukee	Cudahy-South Milwaukee	х	x		x	x	x
	Franklin			••		X	x
	Milwaukee-Glendale	x	x	• •	X	X	x
	Milwaukee-Granville	x	x	х	X X	X	x
	Milwaukee-Menomonee Valley East	X	X	х	X	х	x
	Milwaukee-Menomonee Valley West	x	x		×	x	x
	Milwaukee-Near North	x	x		X	x	x
	Milwaukee-Near South	x	X	x	x	x	x
	Milwaukee-North	X	x	x	X	X	x
	Milwaukee-South	x	x			x	x
	Oak Creek	x	X	x	x	X	х
	West Allis-East	x	X			x	×
	West Allis-West	х	x •	x	X	×	x
	West Milwaukee	x	X			x	x
Özaukee	Grafton					x	
Basing						~	
Racine				X		X	X
	Mt. Pleasant	×		l Š	X	X	X
		~	X .	│ ∧ .	X .	X	X
					**	X	
Walworth	Delavan	••				×	
	Elkhorn				x	x	x
Washington	Hartford		x	×	×	x	x
	Jackson		,			X	
	West Bend-North	X	X	X	X	· X	X .
	West Bend-South			'		×	
Waukesha	Butler	x	X	x	x	x	×
	New Berlin	x	x	x	X	X	X
	Oconomowoc			• • ·		X	
	Pewaukee	X	X	X	X	× ×	X
	Waukesha-North	X	x	X	X	X	X
	Waukesha-South	x	x	×	x	x	×

NOTE: To qualify as a major industrial center, a site must accommodate at least 3,500 industrial jobs.

Source: SEWRPC.

persons, or about 92 percent of the resident population, would be served by public sanitary sewer and water supply facilities by the plan design year. Under the intermediate-growth decentralized plan, about 617 square miles, or about 87 percent of the developed urban area of the Region, and about 1.70 million persons, or about 91 percent of the resident population, would be served with public sanitary sewer and water supply facilities. Under the low-growth decentralized plan, about 491 square miles, or 84 percent of the developed urban area, and about 1.37 million persons, or about 90 percent of the resident population, would be so served.

Distribution of Population and Households: Population levels for the Region and the constituent counties anticipated under the alternative futures plans and the adopted year 2010 land use plan are presented in Table 140 and Figure 57.

Under both the high-growth decentralized and high-growth centralized plans the resident population of the Region would increase by about 573,000 persons, or about 33 percent, from about 1,743,000 persons in 1985 to about 2,316,000 persons in the year 2010. The two highgrowth plans differ significantly, however, in terms of the distribution of the anticipated growth. Under the high-growth decentralized plan, each county in the Region except Milwaukee County would experience significant population growth between 1985 and 2010, with the increases ranging from about 46,000 persons in Kenosha County to about 244,000 persons in

EXISTING AND PROPOSED PRIME AGRICULTURAL LAND IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

	1							Prime Agr	cultural Lar	nd						
		Rec	ommended	Plan	De	Low-Grow centralized	th Plan	Inte De	rmediate-Gr centralized	owth Plan	Dec	High-Growt	h Plan	Ce	High-Growt entralized F	h 'lan
	Existing 1985	Planr Incren 1985-2	ned nent 2010	Total 2010	Plan Increr 1985-	ned nent 2010	Total 2010	Plann Incren 1985-:	ned nent 2010	Total 2010	Plann Incren 1985-:	ned nent 2010	Total 2010	Planr Increm 1985-2	ned nent 2010	Total 2010
County	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)	Acres	Percent	(acres)
Kenosha	76,471	-2,042	-2.7	74,429	-694	-0.9	75,777	-1,091	-1.4	75,380	-3,019	-3.9	73,452	-2,979	-3.9	73,492
Milwaukee	1,351	, O ,	0.0	1,351	0	0.0	1,351	0	0.0	1,351	0	0.0	1,351	0	0.0	1,351
Ozaukee	73,335	-1,425	-1.9	71,910	-1,130	-1.5	72,205	-2,105	-2.9	71,230	-5,987	-8.2	67,348	-2,273	-3.1	71,062
Racine	98,626	-526	-0.5	98,100	-207	-0.2	98,419	-537	-0.5	98,089	-1,552	-1.6	97,074	-1,189	-1.2	97,437
Walworth	208,941	-917	-0.4	208,024	-688	-0.3	208,253	-1,496	-0.7	207,445	-4,366	-2.1	204,575	-1,821	-0.9	207,120
Washington	108,256	-1,463	-1.4	106,793	-1,120	-1.0	107,136	-2,951	-2.7	105,305	-5,690	-5.3	102,566	-2,945	-2.7	105,311
Waukesha	103,078	-3,886	-3.8	99,192	-2,829	-2.7	100,249	-9,292	-9.0	93,786	-14,326	-13. 9	88,752	-9,187	-8.9	93,891
Region	670,058	-10,259	-1.5	659,799	-6,668	-1.0	663,390	-17,472	-2.6	652,586	-34,940	-5.2	635,118	-20,394	-3.0	649,664

Source: SEWRPC.

274

Table 139

AREA AND POPULATION SERVED BY PUBLIC SANITARY SEWER AND WATER SUPPLY SERVICE IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

	Existing 19	Service 85	Recomme 20	nded Plan 10	Low-C Decentra 20	Growth lized Plan 10	Intermedia Decentra 20	te-Growth lized Plan 10	High-G Decentra 20	Growth lized Plan 10	High-G Centraliz 20	Growth zed Plan 10
Area and Population	Public Sanitary Sewer	Public Water Supply										
Developed Area Total Square Miles ^a Square Miles Served Percent of Total Served	476.5 319.8 ^b 67.1	476.5 262.9 ^c 55.2	667.8 570.3 ^d 85.4	667.8 572.7 ^d 85.8	582.2 490.5 ^d 84.2	582.2 492.8 ^d 84.6	708.1 617.4 ^d 87.2	708.1 620.7 ^d 87.7	804.7 734.2 ^e 91.2	804.7 734.5 ^e 91.3	740.5 653.2 ^d 88.2	740.5 656.2 ^d 88.6
Population Total Population Population Served Percent of Total Served	1,742,700 1,507,800 86.5	1,742,700 1,389,700 79.7	1,911,000 1,739,100 91.0	1,911,000 1,742,500 91.2	1,517,100 1,365,800 90.0	1,517,100 1,369,000 90.2	1,872,200 1,700,800 90.8	1,872,200 1,706,500 91.1	2,316,100 2,172,300 93.8	2,316,100 2,172,700 93.8	2,316,100 2,136,700 92.3	2,316,100 2,141,800 92.5

NOTE: Public sanitary sewer and water supply service areas presented in this table do not include lands that are located adjacent to, but outside, the Region, including 1.1 square miles of land in the Jefferson County portion of the Whitewater urban service area, 0.4 square mile of land in the Jefferson County portion of the Oconomowoc urban service area, and 0.4 square mile of land in the Dodge County portion of the Hartford urban service area.

^aBased on historic urban growth; see Table 67 of SEWRPC Planning Report No. 40, <u>A Regional Land Use Plan for Southeastern Wisconsin—2010</u>.

^bDoes not include 57.1 square miles of land served with public sanitary sewer located outside the 1985 developed urban area.

^CDoes not include 30.1 square miles of land served with public water supply located outside the 1985 developed urban area.

^dDoes not include 4.9 square miles of land to be served with public sanitary sewer and water supply service located outside the planned 2010 developed urban area.

^eDoes not include 6.2 square miles of land to be served with public sanitary sewer and water supply service located outside the planned 2010 developed urban area.





EXISTING AND PROPOSED POPULATION IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

Source: SEWRPC.

Table 140

EXISTING AND PROPOSED POPULATION IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

								Popula	tion							
		Rec	commended	Plan	De	Low-Growt ecentralized	h Plan	Inte De	ermediate-Gr acentralized	owth Plan	De	High-Grown centralized	th Plan	c	High-Growi entralized F	h Man
County	Existing	Planr Incren 1985-	ned nent 2010	Total	Plans Incren 1985-	ned nent 2010	Total	Plani Incren 1985-	ned nent 2010	Total	Planı Incren 1985-	ned nent 2010	Total	Planr Incren 1985-:	ned nent 2010	Total
	1985	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010
Kenosha	121,100	26,800	22.1	147,900	-19,300	-15.9	101,800	2,200	1.8	123,300	45,700	37.7	166,800	45,700	37.7	166,800
Milwaukee	939,600	-5,600	-0.6	934.000	-193,500	-20.6	746,100	-108,500	-11.5	831,100	-18,700	-2.0	920,900	159,200	16.9	1,098,800
Ozaukee	67,500	12,300	18.2	79,800	100	0.1	67,600	25,500	37.8	93,000	83,800	124.1	151,300	38,900	57.6	106,400
Racine	169,200	16,800	9.9	186,000	-29,600	-17.5	139,600	2,600	1.5	171,800	55,500	32.8	224,700	55,500	32.8	224,700
Walworth	72.200	15,100	20.9	87,300	-2,100	-2.9	70,100	24,800	34.3	97,000	65,400	90.6	137,600	36,600	50.7	108,800
Washington	87,200	24,500	28.1	111.700	3,900	4.5	91,100	47,400	54.4	134,600	97,800	112.2	185,000	61,800	70.9	149,000
Waukesha	285,900	78,400	27.4	364,300	14,900	5.2	300,800	135,500	47.4	421,400	243,900	85.3	529,800	175,700	61.5	461,600
Region	1,742,700	168,300	. 9.7	1,911,000	-225,600	-12.9	1.517,100	129,500	7.4	1,872,200	573,400	32.9	2,316,100	573,400	32.9	2,316,100

			Percentage	Distribution of P	opulation within the Reg		
County	Existing 1985	Existing 1990	Recommended Plan: 2010	Low-Growth Decentralized Plan: 2010	Intermediate-Growth Decentralized Plan: 2010	High-Growth Decentralized Plan: 2010	High-Growth Centralized Plan: 2010
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	7.0 53.9 3.9 9.7 4.1 5.0 16.4	7.1 53.0 4.0 9.7 4.1 5.3 16.8	7.7 48.9 4.2 9.7 4.6 5.8 19.1	6.7 49.2 4.5 9.2 4.6 6.0 19.8	6.6 44.4 9.2 5.2 7.2 22.5	7.2 39.8 6.5 9.7 5.9 8.0 22.9	7.2 47.5 4.6 9.7 4.7 6.4 19.9
Region	100.0	100.0	100.0	100.0	100.0	100.0	100.0

PERCENTAGE DISTRIBUTION OF POPULATION IN THE REGION BY COUNTY: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

Source: SEWRPC.

Waukesha County. Milwaukee County would experience a decrease of about 19,000 persons during that time. Under the high-growth centralized plan, each county in the Region would experience significant growth, with the increases ranging from 37,000 persons in Walworth County to 176,000 persons in Walworth County to 176,000 persons in Waukesha County. Under this plan, Milwaukee County would experience the second largest increase, about 159,000 persons, between 1985 and 2010.

Under the intermediate-growth decentralized plan, the resident population of the Region would increase by about 129,000 persons, or by just over 7 percent, to a level of about 1,872,000 persons by the year 2010. Among the seven counties, Waukesha County would experience the largest absolute increase in population, about 136,000 persons. Ozaukee, Walworth, and Washington Counties would experience population increases of 26,000, 25,000, and 47,000 persons, respectively. Kenosha and Racine Counties would experience slight increases in population, while Milwaukee County would experience a substantial decrease in population, more than 108,000 persons.

Under the low-growth decentralized plan, the resident population of the Region would decline to a level of about 1,517,000 persons by the year 2010, a loss of about 226,000 persons, or about 13 percent, from the 1985 level. Most of the anticipated population loss would occur in Kenosha, Milwaukee, and Racine Counties. Walworth County would experience a modest decrease of about 2,000 persons, while Ozaukee County would experience virtually no population change. Washington and Waukesha Counties would experience increases in population of about 4,000 persons and 15,000 persons, respectively.

The changes in population anticipated under the alternative futures plans and the adopted land use plan would significantly alter the relative distribution of population among the counties within the Region (see Table 141). The greatest change in this respect would occur in Milwaukee and Waukesha Counties. Between 1985 and 2010, Milwaukee County's share of the regional population would decrease from about 54 percent to about 49 percent under the low-growth decentralized plan, to about 48 percent under the highgrowth centralized plan, to about 44 percent under the intermediate-growth decentralized plan, and to about 40 percent under the highgrowth decentralized plan. Conversely, Waukesha County's share of the regional population would increase from about 16 percent in 1985 to about 20 percent under the low-growth decentralized and high-growth centralized plans, to about 22 percent under the intermediate-growth decentralized plan, and to about 23 percent under the high-growth decentralized plan.

Similar information regarding the number and distribution of households in the Region anticipated under the alternative futures plans and the adopted land use plan is presented in Table 142 and Figure 58. As indicated in Table 142, between 1985 and 2010 the number of households in the Region would increase by about 202,600, or about 32 percent, to about

EXISTING AND PROPOSED HOUSEHOLDS IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

			_					House	holds							
		Re	commended	Plan	D	Low-Growt ecentralized	h Plan	Inte De	rmediate-Gro centralized f	awth Plan	Dec	High-Growth centralized f	n Plan	c	High-Growth entralized Pl	ı an
County	Existing	Plani Increr 1985-	ned nent 2010	Total	Plan Incre 1985	med ment -2010	Total	Plan Increr 1985-	ned nent 2010	Total	Plan Incren 1985-	ned nent 2010	Total	Planı Incren 1985-	ned nent 2010	Total
	1985	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010
Kenosha	44,200	14,900	33.7	59,100	3,300	7.5	47,500	5,700	12.9	49,900	17,100	38.7	61,300	17,500	39.6	61,700
Milwaukee	368,200	32,800	8.9	401,000	-21,600	-5.9	346,600	-13,600	-3.7	354,600	-9,000	-2.4	359,200	62,400	16.9	430,600
Ozaukee	22,900	7,600	33.2	30,500	5,600	24.5	28,500	12,700	55.5	35,600	29,800	130.1	52,700	14,100	61.6	37,000
Racine	61,200	12,700	20.8	73,900	1,700	2.8	62,900	7,700	12.6	68,900	20,500	33.5	81,700	20,900	34.2	82,100
Walworth	25,600	10,000	39.1	35,600	5,700	22.3	31,300	13,900	54.3	39,500	26,200	102.3	51,800	14,700	57.4	40,300
Washington	28,500	13,100	46.0	41,600	8,900	31.2	37,400	21,600	75.8	50,100	34,000	119.3	62,500	21,800	76.5	50,300
Waukesha	93,200	39,400	42.3	132,600	28,700	30.8	121,900	61,000	65.5	154,200	84,000	90.1	177,200	61,100	65.6	154,300
Region	643,800	130,500	20.3	774,300	32,300	5.0	676,100	109,000	16.9	752,800	202,600	31.5	846,400	212,500	33.0	856,300

Figure 58

EXISTING AND PROPOSED HOUSEHOLDS IN THE REGION: 1985, 2010 RECOMMENDED



Source: SEWRPC.

	Percentage Distribution of Households within the Region											
Existing County 1985		Existing 1990	Recommended Plan: 2010	Low-Growth Decentralized Plan: 2010	Intermediate-Growth Decentralized Plan: 2010	High-Growth Decentralized Plan: 2010	High-Growth Centralized Plan: 2010					
Kenosha Milwaukee Ozaukee	6.9 57.2 3.5	6.9 55.2 3.8	7.6 51.8 3.9	7.0 51.3 4.2	6.6 47.1 4.7	7.3 42.4 6.2	7.2 50.3 4 3					
Racine	9.5 4.0	9.4 4.1	9.6 4.6	9.3	9.2	9.7	9.6					
Washington Waukesha	4.4	4.9 15.7	5.4	5.6 18.0	6.7 20.5	7.4	5.9 18.0					
Region	100.0	100.0	100.0	100.0	100.0	100.0	100.0					

PERCENTAGE DISTRIBUTION OF HOUSEHOLDS IN THE REGION BY COUNTY: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

Source: SEWRPC.

846,400 under the high-growth decentralized plan; by about 212,500 households, or about 33 percent, to about 856,300 under the highgrowth centralized plan; by about 109,000, or about 17 percent, to about 752,800 under the intermediate-growth decentralized plan; and by 32,300, or about 5 percent, to 676,100 under the low-growth decentralized plan.¹⁶ Differences in relative rates of growth in population and households, are primarily attributable to anticipated changes in household types and related changes in household sizes, including a continued increase in the relative proportion of single-parent and single-person households.

The change in the relative distribution of households among the seven counties envisioned under the alternative futures plans and the recommended plan is indicated in Table 143. These changes generally parallel the anticipated changes in the relative distribution of the population described above, with Milwaukee and Waukesha Counties being the most affected.

The population density of the developed urban area of the Region would continue to decline under each of the alternative futures plans and also under the adopted plan. As indicated in Table 144, between 1985 and 2010 the urban population density would decrease from about 3,600 persons per square mile to about 3,100 persons per square mile under the high-growth centralized plan, to about 2,900 persons per square mile under the high-growth decentralized plan, and to about 2,600 persons per square mile under the intermediate-growth decentralized and low-growth decentralized plans.

Employment Distribution: Employment levels anticipated under the alternative futures plans and the recommended land use plan are presented by county in Table 145 and Figure 59. Under both the high-growth decentralized and high-growth centralized plans, total employment in the Region would increase by 380,000 jobs, or about 44 percent, from about 872,000 jobs in 1985 to about 1,252,000 jobs by the year 2010. Under the high-growth decentralized plan, Waukesha County would experience the largest absolute increase in employment, about 116,000 jobs. The other six counties in the Region would experience increases ranging from 27,000 jobs in Walworth County to 95,000 jobs in Milwaukee County. Under the high-growth centralized plan,

¹⁶The number of households anticipated under the high-growth centralized plan differs from that anticipated under the high-growth decentralized plan even though the plans are based on the same design year regional population level. This situation is due to the differences in the anticipated distribution of population under the respective plans combined with anticipated differences in household size within the Region. The high-growth decentralized plan envisions fewer households than the high-growth centralized plan because it envisions higher population levels in outlying areas, where household sizes are typically larger, and lower population levels in older urban areas, where household sizes are typically smaller.

POPULATION DENSITY IN THE REGION: SELECTED YEARS 1850-1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

	Urban Population		Rural Population ^a			Area (square miles)		Persons per Square Mile	
Condition	Number	Percent	Number	Percent	Total	Urbon	Total	Linkan	Tetal
Contaition	Number		Number		ropulation	Ulball	IOLAI	Urban	IO(a)
Actual									
1850	28,623	25.2	84,766	74.8	113,389	4	2,689	7,156	42
1880	139,509	50.3	137,610	49.7	277,119	18	2,689	7,751	103
1900	354,082	70.6	147,726	29.4	501,808	37	2,689	9,570	187
1920	635,376	81.1	148,305	18.9	783,681	56	2,689	11,346	291
1940	991,535	92.9	76,164	7.1	1,067,699	90	2,689	11,017	397
1950	1,179,084	95.0	61,534	5.0	1,240,618	146	2,689	8,076	461
1963	1,634,200	97.6	40,100	2.4	1,674,300	282	2,689	5,795	623
1970	1,728,946	98.5	27,137	1.5	1,756,083	338	2,689	5,115	653
1980	1,749,238	99.1	15,558	0.9	1,764,796	444	2,689	3,940	656
1985	1,730,500	99.3	12,200	0.7	1,742,700	477	2,689	3,628	648
Planned									
Low-Growth									
Decentralized: 2010	1,508,000	99.4	9,100	0.6	1,517,100	582	2,689	2,591	564
Intermediate-Growth									
Decentralized: 2010	1,865,400	99.6	6,800	0.4	1,872,200	708	2,689	2,635	696
High-Growth									
Decentralized: 2010	2,310,000	99.7	6,100	0.3	2,316,100	805	2,689	2,870	861
High-Growth									
Centralized: 2010	2,309,300	99.7	6,800	0.3	2,316,100	741	2,689	3,116	861
Recommended: 2010	1,902,800	99.6	8,200	0.4	1,911,000	668	2,689	2,849	711

^a The rural population has been divided into rural farm and rural nonfarm by the U. S Bureau of the Census since 1930. The rural population shown on this table for the years 1850 to 1920 includes the total rural population as enumerated by the Census Bureau. The rural population for 1940 and after includes the rural farm population only, the rural nonfarm population being included in the urban population.

Source: U. S. Bureau of the Census and SEWRPC.

Milwaukee County would experience the largest absolute gain in employment, 135,000 jobs, while the other six counties would experience increases ranging from 18,000 in Walworth County to 90,000 in Waukesha County.

Under the intermediate-growth decentralized plan, total employment within the Region would increase to about 1,051,000 jobs by the year 2010, an increase of about 179,000 jobs, or about 21 percent, over 1985. Under that plan, Milwaukee County would experience a modest increase in employment of about 5,000 jobs, or about 1 percent. The other six counties in the Region would experience larger employment increases ranging from about 15,000 jobs in Walworth County to about 79,000 jobs in Waukesha County. Under the low-growth decentralized plan, total employment within the Region in the year 2010 would be about the same as in 1985, although the distribution of jobs in the Region would continue to change. Among the seven counties the largest absolute changes anticipated under the low-growth decentralized plan would be an increase of about 41,000 jobs in Waukesha County and a decrease of about 77,000 jobs in Milwaukee County.

The relative distribution of jobs among the seven counties in the Region called for under the alternative futures plans and the adopted land use plan is indicated in Table 146. As indicated in that table, between 1985 and 2010, Milwaukee County's share of the total regional employment would decrease from about 60 percent to about

Figure 59



EXISTING AND PROPOSED EMPLOYMENT IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

Source: SEWRPC.

Table 145

EXISTING AND PROPOSED EMPLOYMENT IN THE REGION: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

County	Employment															
	Existing 1985	Recommended Plan			Low-Growth Decentralized Plan		Intermediate-Growth Decentralized Plan			High-Growth Decentralized Plan			High-Growth Centralized Plan		h lan	
		Planned Increment 1985-2010		Total	Planned Increment 1985-2010		Total	Planned Increment 1985-2010		Total	Planned Increment 1985-2010		Total	Planned Increment 1985-2010		Total
		Number	Percent	2010	Number Percent	Percent	nt 2010	Number	Percent	2010	Number	Percent	2010	Number	Percent	2010
Kenosha	42,500	20,500	48.2	63.000	7,900	18.6	50,400	23,200	54.6	65,700	38,300	90.1	80,800	46,400	109.2	88,900
Milwaukee	527,300	85,400	16.2	612,700	-76,500	14.5	450,800	4,800	0.9	532,100	94,900	18.0	622,200	135,300	25.7	662,600
Ozaukee	26,900	11,800	43.9	38,700	8,900	33.1	35.800	17,400	64.7	44,300	29,900	111.2	56,800	20,400	75.8	47,300
Racine	74,500	17,600	23.6	92,100	1,200	1.6	75,700	19,100	25.6	93,600	38,100	51.1	112,600	43,500	58.4	118,000
Walworth	28,100	12,400	44.1	40,500	6,700	23.8	34,800	14,600	52.0	42,700	27,400	97.5	55,500	18,300	65.1	46,400
Washington	31,300	16,600	53.0	47,900	9,700	31.0	41,000	21,400	68.4	52,700	34,800	111.2	66,100	25,600	81.8	56,900
Waukesha	141,300	58,800	41.6	200,100	41,100	29.1	182,400	78,900	55.8	220,200	116,300	82.3	257,600	90,200	63.8	231,500
Region	871,900	223,100	25.6	1,095,000	-1,000	-0.1	870,900	179,400	20.6	1.051.300	379,700	43.5	1,251,600	379,700	43.5	1,251,600

(1929-1927)												
	Percentage Distribution of Employment within the Region											
County	Existing 1985	Existing 1990	Recommended Plan: 2010	Low-Growth Decentralized Plan: 2010	Intermediate-Growth Decentralized Plan: 2010	High-Growth Decentralized Plan: 2010	High-Growth Centralized Plan: 2010					
Kenosha	4.9 60.5	4.7 54.8	5.7 56.0	5.8 51.8	6.3 50.6	6.5 49.7	7.1 52.9					
Ozaukee	3.1	3.2	3.5	4.1	4.2	4.5	3.8					
Racine	8.5	8.3	8.4	8.7	8.9	9.0	9.4					
Walworth	3.2	3.7	3.7	4.0	4.1	4.4	3.7					
Washington	3.6	4.2	4.4	4./	5.0	5.3	4.6					
Waukesha	16.2	17.4	18.3	20.9	20.9	20.6	18.5					
Region	100.0	100.0	100.0	100.0	100.0	100.0	100.01					

PERCENTAGE DISTRIBUTION OF EMPLOYMENT IN THE REGION BY COUNTY: 1985, 2010 RECOMMENDED LAND USE PLAN, AND 2010 ALTERNATIVE FUTURES LAND USE PLANS

Source: SEWRPC.

53 percent under the high-growth centralized plan; to about 52 percent under the low-growth decentralized plan; to about 51 percent under the intermediate-growth decentralized plan; and to about 50 percent under the high-growth decentralized plan. Under the recommended plan, Milwaukee County would account for about 56 percent of the total regional employment. Waukesha County's share of the total regional employment would increase from about 16 percent in 1985 to about 19 percent by the year 2010 under the high-growth centralized plan; and to about 21 percent under the low-growth decentralized, the intermediate-growth decentralized. and the high-growth decentralized plans. The other five counties would maintain or experience modest increases in their shares of total regional employment under the respective plans.

SUMMARY

Forecasting future population and economic activity levels is a necessary step precedent to the design of physical development plans. Since surveillance activities point to increasing uncertainty with regard to future social and economic conditions in Southeastern Wisconsin, the Commission has adopted an "alternative futures" approach to forecast preparation. This approach involves the postulation of alternative future growth scenarios for the Region and the preparation of related projections of population and employment, thereby providing a broader basis for plan design and evaluation. Three alternative futures growth scenarios were postulated for the Region. The sets of conditions postulated for each "future" are intended to represent consistent, reasonable scenarios of future change in population and economic activity within the Region through the year 2010. A "high-growth" scenario and a "lowgrowth" scenario are intended to represent reasonable extremes; a third scenario, an "intermediate-growth" scenario, is intended to represent a likely future.

The projections of population, household, employment, and personal income levels and related land use demand projections were made prior to the release of the 1990 U. S. Census data. Consequently, actual conditions in 1980 were used as a basis upon which future conditions were projected. A comparison of the recently released 1990 Federal data for the Region is included in the first section of this chapter. A summary of changes in population, households, employment, personal income, and land use projected under the three regional growth scenarios follows.

1. Under the high-growth scenario, the resident population of the Region would increase by about 551,300 persons, or 31 percent, from 1,764,800 persons in 1980 to 2,316,100 persons in the year 2010. The intermediate-growth scenario envisions a population increase of 107,400 persons, or 6 percent, to a level of 1,872,200 persons in 2010. Conversely, the low-growth scenario envisions a decrease in the regional population of 247,700 persons, or 14 percent, to a level of 1,517,100 persons in the year 2010.

- The anticipated changes in the number of 2. households in the Region would not necessarily parallel the anticipated changes in population levels under the respective scenarios because of changing household types and related changes in household size. Under the high-growth scenario, the number of households in the Region would increase by 224,700, or 36 percent, from 628,000 in 1980 to 852,700 in 2010. The projected relative increase in households of 36 percent is slightly greater than the relative increase of 31 percent in the regional population projected for the highgrowth scenario. Under the intermediategrowth scenario, the number of households would increase by 140,600, or 22 percent, to a level of 768,600 in 2010. The projected 22 percent increase in households is significantly greater than the projected increase of 6 percent in the regional population for this scenario. Under the lowgrowth scenario, the number of households would increase by 73,900, or 12 percent, to about 701,900 in the year 2010. This is in contrast to the projected 14 percent decrease in the regional population under this scenario and is a result of the anticipated increase in smaller households, particularly single-person and singleparent households.
- 3. Under the high-growth scenario, total regional employment would increase by 367,400 jobs, or 42 percent, from 884,200 jobs in 1980 to about 1,251,600 jobs in 2010. Under the intermediate-growth scenario, employment would increase by 167,100 jobs, or 19 percent, to about 1,051,300 jobs in 2010. Under the low-growth scenario, total employment would approximate 870,900 jobs in 2010, about 13,300 jobs, or about 2 percent, less than the 1980 level.
- 4. Anticipated trends in personal income vary significantly for the three regional growth scenarios.¹⁷ Under the high-

growth scenario, per capita income would recover rapidly from the depressed levels of the 1979 to 1983 recession and continue rising to about \$20,700 in the year 2010, an increase of about \$7,800, or 60 percent, over 1980. The intermediate-growth scenario anticipates a slower but nevertheless significant increase, with per capita income rising to about \$15,800 in the year 2010, an increase of \$2,900, or 22 percent, over 1980. Under the low-growth scenario, recovery from the depressed recessionary income levels would be very slow, with per capita income only expected to approximate \$12,700 in the year 2010, about the same as the 1980 level.

5. Changes in the levels and distribution of population, households, and employment may be expected to generate additional demand for urban land development in the Region. Under the high-growth scenario, urban lands, consisting of areas devoted to residential, commercial, industrial, governmental and institutional, transportation, and recreational uses, could be expected to increase by about 159,400 acres, or 42 percent. from about 378,100 acres in 1980 to 537,500 acres in 2010. Increases of 98,000 acres, or 26 percent, and 50,900 acres, or 14 percent, could be expected under the intermediate- and low-growth scenarios, respectively. By the year 2010, lands devoted to urban uses would account for about 31 percent of the total area of the Region under the high-growth scenario, nearly 28 percent under the intermediategrowth scenario and about 25 percent under the low-growth scenario.

The adopted regional land use plan was designed to accommodate the intermediategrowth forecasts, modestly adjusted, however, to reflect trends indicated by 1990 U.S. Census data. Accordingly, the plan envisions an increase of about 168,000 persons in the resident population, an increase of 130,000 in the number of households, and an increase of about 223,000 in total regional employment between 1985 and 2010. The plan proposes to accommodate the anticipated growth in population, households, and employment by converting about 54,800 acres of land from rural to urban uses. Under the plan, urban population density would decrease from about 3,600 persons per square mile in 1985

¹⁷All historic and projected income data presented in this chapter are expressed in constant 1990 dollars.

to about 2,800 persons per square mile in the year 2010.

The adopted year 2010 regional land use plan incorporates the basic concepts of the year 2000 regional land use plan and updates and extends that plan to a new design year. The year 2010 plan recommends a relatively compact, centralized regional settlement pattern, with urban development occurring generally in concentric rings along the full periphery of, and outward from, existing urban centers. Within urban areas the plan envisions a total of 19 major commercial centers and 25 major industrial centers by the plan design year, including five new commercial centers and three new industrial centers. The plan also envisions a total of 31 major park sites, only two of which remain to be publicly acquired.

The adopted plan recommends that new urban development occur primarily in those areas of the Region which are covered by soils suitable for such development; which are not subject to special hazards such as flooding and shoreline erosion; and which can be readily served by essential municipal facilities and services, including public sanitary sewerage, water supply, and public transit systems. The plan recommends the preservation in essentially natural, open uses of primary environmental corridors and the preservation in agricultural and related use of most of the remaining prime agricultural lands in the Region.

Because of the possibility of future growth and change in the Region at variance with the rates assumed in the adopted land use plan, four alternative land use plans were prepared. The four alternative plans differ from the recommended plan in terms of the overall scale and distribution of urban development. While the recommended plan seeks to centralize urban development, three of the alternative plans assume that the historic trend toward decentralized urban development in the Region will continue. A high-growth decentralized plan was designed to accommodate future population and economic activity levels that could be anticipated under a high-growth scenario. An intermediate-growth decentralized plan and a low-growth decentralized plan were designed to accommodate population and economic activity levels anticipated under the intermediate- and low-growth scenarios, respectively. A fourth alternative plan, high-growth centralized, was designed to accommodate population and economic activity levels anticipated under the high-growth scenario, and unlike the other alternative plans emphasized a centralized development pattern for the Region.

Consideration of alternative land use plans broadens the scope of the regional transportation system plan. The ability of various alternative transportation system plans to accommodate travel demand generated under different development conditions can be determined and a plan identified as most adaptable to various possible futures can then be selected for adoption. It may be that no transportation system plan can be designed to serve cost-effectively each identified future land use scenario. Nevertheless, such plan evaluation permits the identification of potential problems under alternative development futures. Having identified such potential problems, local project level planning can be executed to solve or mediate potential problems should they occur.

It is in this regard that the alternative futures approach validates the cyclical nature of planning recognized by the Commission as essential to plan implementation (see Chapter II). It is recommended that local detailed transportation project planning, county jurisdictional highway system planning, and local transit development programs give consideration to the full range of possible future conditions to ensure that any proposal is sufficiently adjusted to accommodate travel demand under various scenarios. Such analyses undertaken at the local project level may prevent costly future problems associated with, among other things, insufficient capacity and service. (This page intentionally left blank)

Chapter VII

TRAVEL SIMULATION MODELS

INTRODUCTION

One of the seven basic principles, set forth in Chapter II of this report, upon which the regional transportation plan reevaluation is based, is:

Transportation facilities and management proposals must be planned as an integrated system. The capacities of each link in the system must be carefully fitted to traffic loads and the effects of each proposed facility and management proposal on the remainder of the system quantitatively tested.

This principle is particularly important because, unless transportation system plans are subject to quantitative test and evaluation permitting preparation of forecasts of the amount of travel and traffic the system must carry, the adequacy of the plans must remain in doubt from an engineering standpoint. Transportation system plans prepared without quantitative test and evaluation are little more than intuitively created street patterns and transit networks rather than engineered transportation system designs. Such untested plans cannot provide a sound basis for project design, capital investment, or operating funding decisions. Nor can the implementation of such plans be expected to provide the desired long-range solution to traffic and transportation problems. Indeed, such untested plans may create more problems than they purport to solve. The lack of quantitative system test and evaluation is one of the major factors which has in the past contributed to the ineffectiveness of many transportation planning programs carried on at the local level of government, since State and Federal transportation agencies have, quite properly, refused to implement plans which were unsupported by carefully prepared travel and traffic forecasts and assignments.

The Role of Quantitative Analysis

Quantitative traffic analysis is a fundamental requirement for any transportation planning effort but is a particularly important and complex requirement of the urban transportation planning process. Three basic questions must be confronted in the planning and design of a new transportation facility, an improvement to an existing transportation facility, or a new transportation management action.

- 1. How will the existing travel and traffic patterns be distributed on the proposed facility or affected by the management proposal? That is, how many persons and vehicles will use the proposed facility upon its completion or be affected by the management proposal?
- 2. How will future traffic patterns be distributed on the proposed facility or affected by the management proposal? That is, how many persons and vehicles will use the proposed facility at the end of its physical and economic life or be affected by the management proposal?
- 3. How will the proposed facility or management proposal affect traffic on the total transportation system? That is, how many persons and vehicles will be diverted by the proposed facility or management proposal to or from other facilities comprising the total transportation system?

No transportation facility service or management action can be soundly planned, designed, or implemented without answers to these three basic transportation planning questions. It should be noted that these three questions not only recognize the need for an understanding of present travel and traffic patterns but also recognize that future travel and traffic conditions may differ from present conditions. Existing patterns change in form and intensity as new land use activities and transportation facilities are added to the regional complex and as established land use activities and transportation facilities are changed or relocated. Thus, sound transportation system planning must recognize the need to consider both existing and future travel and traffic patterns and must do so in an explicit, quantitative manner.

Originally, the only quantitative traffic information available for use in the transportation planning process consisted of traffic volume counts. Such counts can indicate only the distribution of traffic patterns on the existing transportation network. Alone, such counts are of little value for long-range transportation planning since they do not provide an answer to even the first of the three transportation planning questions. For many years, however, the application of growth factors to traffic volume counts was the only traffic forecasting technique available for planning purposes.

About 50 years ago, the origin and destination study was developed to collect factual information about existing travel habits and patterns. This survey provided the travel data necessary to answer the first of the three transportation planning questions on existing travel habits and patterns, but not the data necessary to answer directly the second or third transportation planing questions which involve the characteristics of future travel behavior.

About 35 years ago, new transportation planning techniques were developed which for the first time provided quantitative answers to the second and third of the three basic transportation planing questions. These techniques made it possible to calculate future travel demand quantitatively as a function of regional development patterns instead of deriving such demand, as was necessary in the past, from simple expansions of existing traffic patterns. The techniques were based on the concept that those aspects of regional development which affect the magnitude and pattern of regional travel demand could be identified, quantified, and correlated with such travel through the analysis of origin and destination and land use survey data. The concept, further, was that the relationships between land use and travel so established would remain reasonably stable over time, enabling the simulation of future travel patterns on the basis of projected or planned future land use patterns. By considering the future distribution of land use activity within an urban region as the major factor influencing future traffic patterns, integrated transportation system designs could be developed which would not only serve the existing traffic patterns within an urban region but which would also serve the entirely new patterns that will evolve with changing development.

The first time that these new transportation planning techniques were used on a regional scale in Southeastern Wisconsin was during the initial land use-transportation study conducted by the Commission from 1963 to 1966. The historic design year 1990 regional transportation plan was in large part based upon the quantitative analysis of the performance of alternative highway and transit systems permitted by the battery of traffic simulation models developed for the 1963 to 1966 study.

Since the need for quantitative analysis of the performance of alternative transportation systems has, if anything, increased over the last 30 years, it has been necessary to review and, as needed, refine the initial battery of traffic simulation models developed by the Commission for use in the regional transportation planning effort. As part of the preparation of the secondgeneration regional transportation plan in the 1970s, such refinement consisted of an analysis of the adequacy of each individual model used in the initial study. The effectiveness of this analysis was greatly enhanced by the fact that the findings of major travel inventories were available for two points in time, 1963 and 1972, thus permitting examination of the temporal stability of the traffic simulation models. In this manner those initial traffic simulation procedures which were shown to be reliable could be retained for use in the plan reevaluation effort. Significant advances in the state of the art of traffic simulation occurred since the conduct of the initial transportation study. An investigation of newer modeling strategies was conducted and some of these strategies were incorporated into the simulation process for use in the secondgeneration plan reevaluation. A complete description of the initial study procedures can be found in SEWRPC Planning Report No. 7, The Regional Land Use-Transportation Study, Volume Two, Forecasts and Alternative Plans: 1990. The refinements to the initial procedures incorporated in the battery of traffic simulation models used in the second-generation plan preparation are documented in SEWRPC Planning Report No. 25, A Regional Land Use and a Regional Transportation Plan for Southeastern Wisconsin: 2000, Volume Two, Alternative and Recommended Plans. A similar review and refinement of those second-generation models was conducted as part of this preparation of a third-generation transportation plan, and is documented in this chapter. This refinement again has the benefit of not only the findings of a new major travel inventory conducted in 1991, but also of consistent inventory data for three points in time, 1963, 1972, and 1991.

Basic Modeling Concepts

The simulation of existing and future travel demand is a complex procedure requiring application of a variety of mathematical and statistical techniques. Although the basic formats of the traffic simulation process used in all major urban transportation studies today are similar, the specific needs and characteristics of each study area require a certain amount of individualization for the precise procedures used.

The simulation of travel and traffic is based upon the premise that the magnitude and pattern of travel is a stable function of the characteristics of the land use pattern and of the transportation system, with the term land use referring to not only land use type and intensity, but also to population and employment levels and characteristics. In travel simulation modeling, those aspects of the land use pattern and of the highway and public transit system characteristics which affect the magnitude and distribution of travel demand are identified, quantified, and correlated through the analysis of detailed travel origin-and-destination and land use and transportation system survey data. It has been demonstrated that the relationships between travel and land use and transportation system characteristics remain reasonably stable over time, thus enabling the forecast of future travel and traffic patterns based upon postulated future land use patterns and transportation system configurations.

Given the necessary land use and transportation system data, the complete sequence of travel simulation occurs in four steps:

- 1. Trip generation, in which the total number of person trips generated in each subarea of the planning area for the time period under analysis is determined by using relationships found to exist between land use and travel by analyses of the land use and travel inventory data. The output from this step is the total number of person trip ends, that is, trips entering and leaving each subarea of the study area.
- 2. Trip distribution, in which the person trips generated in each subarea are linked with trip ends in other subareas, thereby defining the universe of person trips by point of origin and point of destination. The output from this step is the number of person trips made between each subarea pair.

- 3. Modal choice, in which the number of person trips between each subarea pair is divided among the travel modes, primarily public transit and automobile. The output of this step is the number of person trips made between each subarea pair by automobile, including those driving alone and those using carpools and vanpools. The person trips made by automobile are correlated to vehicle trips by the application of automobile occupancy factors or by estimating the number of automobile trips made by driving alone or by carpools.
- 4. Traffic assignment, in which the subarea transit trips are assigned to the existing or to proposed alternative future transit system networks and the subarea vehicle trips are assigned to the existing or proposed alternative arterial street and highway facility networks. The output of this step is the number of people utilizing the routes and facilities of the existing or proposed public transit system and the number of vehicles utilizing each segment of the existing or proposed public transit and arterial street and highway systems.

The result of the four-step travel simulation process is a complete description of the use of an existing or proposed transportation system consisting of both arterial streets and highways and transit lines. Although the four steps set forth above provide a general description of the travel simulation modeling process, some variations are necessary and desirable for the various types of trips to be simulated.

As noted earlier, a battery of travel simulation models was first developed and applied on a regional scale in Southeastern Wisconsin during the initial regional land use-transportation study conducted by the Commission in 1963.¹ These models were developed by using an extensive data base developed from specific land use, socio-economic, and travel inventories conducted by the Commission. A massive travel survey was conducted in the late spring of 1963 to obtain data describing the amount, kind, and

¹See SEWRPC Planning Report No. 7, <u>The</u> <u>Regional Land Use-Transportation Study</u>, Volume Two, <u>Forecasts and Alternative Plans: 1990</u>.

distribution of travel occurring throughout the Region on an average weekday and the characteristics of the tripmakers. Four separate travel surveys were conducted, including a home interview survey, a truck and taxi survey, and a postal questionnaire survey used to obtain information on travel internal to the Region and an external (roadside) survey used to obtain information on travel between the Region and other areas of the Country.

The home interview survey was the most comprehensive of the four surveys, providing an inventory of the travel habits and patterns of 18,000 households in the Kenosha, Milwaukee, and Racine urbanized areas. It was conducted at a 3 percent sample rate in the Milwaukee area and at a 10 percent sample rate in both the Kenosha and Racine areas. The truck and taxi survey was conducted at a sample rate of 8.5 percent in the Milwaukee area and 25 percent in the Racine and Kenosha areas and included the survey of trips generated by over 4,300 trucks.

The postal questionnaire survey was conducted to provide basic travel data relating to households, trucks, and taxis in the less populated areas of the Region outside of the Milwaukee, Racine, and Kenosha home interview areas. The survey, which was sent to all households and commercial vehicle operators in the rural areas of the Region, had a usable sample rate of 16 percent of the 69,000 households and 18 percent of the 18,400 commercial vehicle operators in the rural areas. In the external cordon survey, nearly 75,000 of the 101,500 vehicles crossing the boundaries of the Region, or about 74 percent, were stopped and interviewed.

In addition to gathering information on actual travel on the day preceding the receipt of the survey, these surveys also obtained census-type socio-economic data for households in the Region, including total population, income, household size, and migration patterns, which were also used in travel model development. The extensive travel inventory data were then combined for the purposes of travel model development with land use type and density data for each of the 10,800 quarter-sections in the Region and with detailed 1960 U. S. Census data on populating and housing.

The development of the Commission's original, design year 1990, regional transportation system plan was in part based upon quantitative analyses of the performance of alternative highway and transit systems permitted by the battery of travel simulation models developed to forecast average weekday travel under that study. These models were subsequently applied in detailed jurisdictional highway planning studies for each of the seven counties in the Region, freeway and other arterial highway location and design studies, a preliminary engineering study of a busway in the Milwaukee east-west corridor, and other transportation studies.

Between the completion of that initial regional land use-transportation planning effort in 1966 and the initiation of the major land use and transportation plan reevaluation effort in 1972, the emphasis on, and consequent need for, quantitative analysis of the performance of alternative transportation systems increased. In addition, between 1963 and 1972, substantial changes had occurred within the Region, including substantial growth in population and employment, increases in personal income and automobile availability, and further decentralization of land use. As a consequence, the Commission found it desirable to validate, and, as necessary, recalibrate, the initial battery of traffic simulation models for use in the regional transportation system plan reevaluation effort in 1972.²

This review of the initial models was possible because the Commission conducted a new regional inventory of travel in 1972. The 1972 surveys included all the basic origin-destination surveys conducted in 1963, including household travel, truck and taxi, and external cordon surveys, and also a special public transit user survey. The home interview survey involved 17,500 households, or about 3 percent of the Region's households. About 5 percent of all trucks within the Region and about 82 percent of all licensed taxis were sampled under the truck and taxi survey. In the external cordon survey, interviewers at roadside stations stopped and interviewed about 80,300 of the 130,300 motor vehicles crossing the cordon lines, or about

²See SEWRPC Planning Report No. 25, <u>A</u> <u>Regional Land Use Plan and a Regional Trans-</u> portation Plan for Southeastern Wisconsin: 2000, Volume Two, <u>Alternative and Recommended</u> <u>Plans.</u>

59 percent. Sampling rates of transit riders varied among the systems, ranging from about 25 percent in the Milwaukee area to about 50 percent in the Racine and Kenosha areas. Land use data for the Region's 10,800 quartersections were updated and 1970 U.S. Census data were collated and analyzed, as was done in 1963, for use in the reevaluation of the models.

The first step in this review process consisted of an analysis of the adequacy of the travel simulation model battery and of each individual model used in the initial study. The analysis of the temporal stability of the individual models, trip generation, trip distribution, and mode choice, was possible only because major travel inventories were available for two points in time, 1963 and 1972. The Commission's original travel simulation models, calibrated using 1963 home interview survey data, were shown through this testing, which used as input socio-economic and land use survey data gathered in 1972, to estimate accurately travel in Southeastern Wisconsin in 1972. This successful testing of the initial study procedures constituted an important validation of the accuracy of those procedures. The testing included a test of each individual model, as well as a test of the final model estimate of transit ridership and highway traffic to actual counts of transit ridership and highway traffic. Although the validity of the initial study procedures was proven through these analyses, an investigation of newer modeling strategies was conducted and, as applicable, newer techniques were incorporated into a refined battery of the travel and traffic simulation models.

This refined, second-generation, battery of travel simulation models was used in the regional land use-transportation plan reevaluation efforts, completed in mid-1978, and in several short- and long-range highway and transit planning studies since the completion of the 1978 plan reevaluation. One such study which involved extensive application of travel simulation models was the areawide rapid transit system planning study, completed in 1982, which extensively evaluated alternative rapid transit systems for the Milwaukee area.³ Another such study which involved extensive application of the models was the Northwest Corridor Rapid Transit Study, an in-depth analysis of light-rail and express bus alternatives in the corridor and which constituted a transit alternatives analysis-draft environmental impact statement under the review of the Federal Transit Administration.⁴ The battery of travel simulation models was also applied in many arterial street and highway planning studies.

A third regional inventory of travel was conducted in the fall of 1991 and spring of 1992. The 1991 survey included all the basic origindestination surveys conducted in 1972, including household travel, truck, external cordon, and public transit user surveys. The household travel and public transit user surveys were conducted in the fall of 1991; the external cordon and truck and taxi surveys were conducted in the spring of 1992. The home interview survey again involved about 17,500 households, or about 2.5 percent of the Region's households. About 5 percent of all commercial trucks within the Region and about 35 percent of all licensed taxis were sampled under the truck and taxi survey. In the external cordon survey, interviewers at roadside stations stopped and distributed postal card interview forms to about 160,000 of the 230,000 motor vehicles crossing the cordon lines. Approximately 30 percent of the vehicles surveyed returned the survey forms, providing about a 20 percent sample. Sample rates of transit riders varied among the systems, ranging from about 10 percent in the Milwaukee area to about 20 to 30 percent in the Racine, Kenosha, and Waukesha areas. Land use and employment data for the Region's 10,800 quarter-sections were updated to the year 1990 and 1990 U.S. Census data were collated and analyzed for use in the reevaluation of the models.

The first step in this review process again consisted of an analysis of the adequacy of the travel simulation model battery and of each individual model used in the initial study. The analysis of the temporal stability of the indi-

³See SEWRPC Planning Report No. 33, <u>A Pri-</u> mary Transit System Plan for the Milwaukee <u>Area</u>, June 1982.

⁴See SEWRPC Community Assistance Planning Report No. 150, <u>A Rapid Transit Facility Plan</u> for the Milwaukee Northwest Corridor, January 1988, and <u>The Milwaukee Northwest Corridor</u> <u>Rapid Transit Study</u>, Report No. 2, "Travel Simulation Models," 1986.

vidual models, including trip generation, trip distribution, and mode choice, was again possible only because comparable findings of major travel inventories were now available for three points in time: 1963, 1972, and 1991. The Commission's second-generation travel simulation models, calibrated on the basis of 1972 survey data, were shown through this testing, which used as input socio-economic and land use survey data gathered in 1990, to estimate accurately travel in Southeastern Wisconsin in 1991. This successful testing of the travel modeling procedures constituted an important validation of the accuracy of those procedures. Although the validity of the second-generation travel modeling procedures developed in the mid-1970s was proven through these analyses, an investigation of newer modeling strategies was again conducted and, as applicable, newer techniques were incorporated into a refined battery of the travel and traffic simulation models. The travel simulation modeling to be conducted for the preparation of the third-generation regional transportation system plan utilizes this refined battery of travel and traffic simulation models.

Classification of Travel

An important aspect of the Commission's battery of travel and traffic simulation models is the classification of the different components of travel within the Region. Such classification of trips is necessary because different types of trips exhibit different characteristics and, as a consequence, require different simulation techniques. In addition, some of these types of trips represent very small proportions of total travel in an urban region. The classification of trips and the determination of the relative proportion of total travel they represent allow travel simulation modeling resources to be focused on those types of trips which represent the greater proportions of travel.

As shown in Table 147, the first major division of trips involves the distinction between internal and external trips. Internal trips are defined as those trips which have both ends within the Southeastern Wisconsin Region. External trips are defined as those trips which have one or both ends outside of the Region. As intraregional travel by residents of the Region has consistently since 1963 accounted for over 96 percent of the vehicle trips and 97 percent of the person trips observed on an average weekday, the primary emphasis in the modeling process is on internal trips. It should be noted, however, that external trips do have important effects in the use of facilities in certain major travel corridors, particularly near the boundaries of the Region.

Among internal trips a further classification is made based upon mode of travel. The major transportation modes in Southeastern Wisconsin include personal vehicle, such as automobile, van, or light truck; public mass transit; school bus; commercial truck; and taxicab. The manner in which these modes are combined for simulation purposes depends not only upon modal characteristics but also upon the type of travel inventory survey used for calibrating relationships.

The vast majority of trips belong to the category of internal bus and automobile trips. These trips are further classified by the living quarters of the persons making the trip. Essentially trips by group-quartered persons and nonresidents of the Region are separated for special consideration because of the limitations of the trip generation procedure and to the unique travel habits and patterns exhibited by these persons. Groupquartered persons are defined as those persons residing in dormitories, convents, homes for the aged, and similar group residences. Together, group-quartered trips and nonresident trips have consistently accounted for about 1 percent of the total person trips within the Region since 1963.

Because of the sheer volume of travel involved, the primary emphasis of the traffic simulation modeling process focuses on internal automobile and mass transit trips made by nongroupquartered residents. These trips represent over 90 percent of total trips made within the Region on an average weekday. This group of trips is further subdivided by trip purpose into: homebased work; home-based shopping; home-based other (excluding school); nonhome-based (excluding school); and school trips. Home-based trips are defined as those trips having one end located at the residence of the tripmaker. The purpose of a home-based trip is thus determined by the nonhome end of the trip as either work, shopping, or other, the last representing an aggregation of personal business, medical-dental, socialeatmeal, recreation, and serve passenger purposes. Nonhome-based trips are defined as those trips having neither end located at the place of residence of the tripmaker and can be made for any purpose except school. Separate considera-

TRIP CLASSIFICATION AND TRAVEL SIMULATION PROCEDURE

		I fip Classification		Simulation Procedure							
					Trip Ger	neration					
Internal or External	Mode ^a of Travel	Types of Quarters	Trip Purpose	Percent of Total Trips ^b	Production	Attraction	Trip Distribution	Modal Split	Traffic Assignment		
Internal	Automobiles, transit bus, and school bus	Residents in households	Home-based work Home-based shopping Home-based other (excluding school)	22 13 28	Cross-classification analysis	Cross-classification analysis	Gravity model	Logit analysis	Minimum path		
			Nonhome-based (excluding school)	18					· · ·		
			School Central Milwaukee area colleges and universities ^C Other	9	Factor existing total person trip levels		Existing patterns Existing patterns	Logit analysis N/A	Minimum path		
							by mode				
		Group-quartered residents	All	0.5	Factor existing trip levels by mode		Existing patterns by mode	N/A			
		Nonresidents	All	0.5					-		
	Heavy truck	All	All	3	Multiple-regression a	nalysis	Fratar model	1			
	Light truck and taxi	All	All	3]						
External	All	All	All	3	1						

NOTE: N/A indicates not applicable.

^aNot including the lesser modes of railroad, bicycle, motorcycle, air travel, water travel, and charter bus.

^bPercentage of total travel based upon 1991 origin-destination survey trip information.

^CIncludes University of Wisconsin-Milwaukee, Marquette University, Milwaukee Area Technical College, and Milwaukee School of Engineering.

Source: SEWRPC.

tion of home-based and nonhome-based school trips is necessary because of the arbitrary constraints imposed upon travel patterns by school service area boundaries, constraints which cannot be adequately represented in the normal travel simulation sequence in current trip distribution models. In addition, most schools serve local neighborhoods or communities so that tripmaking to or from such schools has a relatively small impact on arterial street and highway travel. Trips to and from all schools, elementary, junior and senior high, vocational and technical schools, and colleges and universities. have consistently represented approximately 9 percent of all person trips observed in the Region on an average weekday.

On the basis of the classification of trips noted above, Table 147 indicates the specific modeling techniques used for each type of trip in the plan reevaluation effort. Each of these techniques will be more fully described in the appropriate sections of this chapter.

Geographic Aggregation System

Another aspect of the traffic simulation modeling process that warrants discussion is the method of spatially aggregating trips. In the description of the four-step simulation process, trips are described as being generated within, distributed between, and split by mode among subareas of the study area. The manner in which these subareas are determined is critical to the travel simulation process. One objective of the geographic aggregation is to achieve maximum homogeneity of land use, broadly defined, within each subarea. Another objective is to replicate as closely as possible the way in which traffic in each zone gains access to the transportation system, that is, the movement of vehicular traffic between local land-access, collector, and arterial streets and the use by transit riders of pedestrian, bicycle, automobile, and feeder bus access to transit service. Although not all models are calibrated on the basis of average subarea characteristics, all models must eventually be applied by subarea. The greater the degree of homogeneity of the land uses in the subareas and the closer the replication of subarea access to the transportation system, the better able the models are to accurately simulate actual traffic conditions.

The basic unit of geographic identification used by the Commission for the collection and analysis of land use, demographic and economic, and travel inventory data is the U.S. Public Land Survey quarter-section, consisting of an approximately one-half mile on a side rectilinear area containing approximately 160 acres. Inventory data are summed to the quarter-section level for all parts of the Region except the central business district (CBD) of the Cities of Milwaukee. Racine, and Kenosha. In these areas, quartersections are divided into smaller areas representing individual city blocks or aggregations of city blocks. Using these basic geographic units, special analytical units can be developed for use in the travel demand modeling process.

Essentially, two related systems of analytical areas are used in the modeling effort to define internal subareas of the Region, as shown on Map 78. A system of 1,431 traffic analysis zones. composed of entire quarter-sections or of combinations of quarter-sections outside the major CBDs and of groupings of city blocks within the major CBD area, is the first such system delineated for the travel demand simulation process. These 1,431 zones are typically used in the calibration of the traffic simulation models and for most applications of the travel simulation models. Except for the major CBDs, where groups of city blocks are delineated as zones, the zone system provides an analytical unit equal to, or larger than, the quarter-section and, thus, decreases the number of geographic units for which data must be handled and forecasts made regarding the approximately 10,800 quartersections within the Region. Within the Region. the traffic analysis zones range in area from 0.04 square mile in the Milwaukee central business district to 36 square miles in the more sparsely settled portions of the Region.

The second system of subareas used in the simulation modeling effort consists of a set of 60 planning analysis areas. These analysis areas are intended to represent rational areas for comprehensive urban planning purposes and, as such, are generally intended to be composed of a number of "neighborhoods." The neighborhoods can be grouped to form "communities," which may consist of groups of minor civil divisions, cities, villages, and towns or subareas of such minor civil divisions. In cases where the minor civil division was considered too large to constitute a viable analysis area, subcommunity areas were delineated within the civil divisions as planning analysis areas. Each planning analysis area represents an aggregation of traffic analysis zones. The 60 planning analysis areas within the Region are shown on Map 78. Another set of aggregated analysis areas divides the Region into 37 areas; yet another set into 110 areas, as shown on Map 79. Both are used principally for travel pattern analyses based on the household survey sample size.

Format of Presentation

This chapter is intended to document the simulation models used in the regional transportation plan reevaluation effort to forecast future travel demand and the manner in which that demand may be expected to impact alternative transportation systems. The documentation includes not only a description of the specific model formats but also a description of the methods used to develop, apply, and test the models.

In the first section of this chapter, it was noted that land use data comprise the basic input to the traffic simulation process. Although Chapter VII of the 1992 land use plan contains a description of most land use characteristics and the methodology used in the land use plan reevaluation to develop probable future land use patterns, there are certain critical land use variables which were specifically developed for the individual traffic simulation models. In this section these variables are identified and the procedures for their forecasting described. Subsequent sections of this chapter are then devoted to describing the development and application of each of the following four steps in the traffic simulation process: trip generation, trip distribution, modal split, and traffic assignment. For each step a more complete explanation of basic modeling concepts is presented followed by a description and assessment of the specific models developed and applied in the initial land use-transportation planning effort. On the basis of this assessment, refinements made in the simulation modeling procedures for the plan reevaluation are described. Finally, the results of various statistical and mathematical testing



It is necessary to relate travel data to geographic areas within the Region for analytical purposes. The Commission has identified a series of geographical subareas of the region for such analytical purposes. This map identifies the boundaries of the 1,431 traffic analysis zones within the Region used in the transportation system modeling effort and for related analytical purposes. Outside the major central business districts in the Region, each traffic analysis zone is composed of one or more U. S. Public Land Survey quarter sections. Within the major central business districts, each traffic analysis zone is composed of city blocks. Traffic analysis zones are used in the traffic simulation models in many applications of those models and in related analyses. The zones range in size from 0.04 square mile in the Milwaukee central business district to 36 square miles in the most rural portions of the region. This map also identifies a set of 60 planning analysis areas. Each planning analysis areas, in most cases comprised of groups of minor civil divisions, are used by the Commission for comprehensive and land use planning purposes.

LEGEND ---- BOUNDARIES FOR 37 AREAS MEQUON svilez VASHI 51 24 12 14 18 38 18 37 EASLE MUSKEDO FRAMELIN 201 0 0 10 10 10 10 40,000 MI CAN CREEK Se ! 19 18 18 WHITEWATER AP mor Whitesh merren 22 17 park (2) 21

53.



For purposes of data compilation and analysis, it was found useful to create two additional geographic subareas of the Region comprised of aggregations of individual traffic analysis zones. One aggregation results in 37 geographic subareas; the other results in 110 geographic subareas. Both sets of subareas are identified on this map. These particular aggregations of traffic analysis zones were used in conducting analyses of subarea travel characteristics, including number and characteristics of trips within each subarea, and between subareas. The subareas were defined to allow statistically valid analyses to be conducted from the findings of the origin and destination travel surveys.

Source: SEWRPC.

WISCONSIN

SUPPLEMENTAL SUBAREAS USED FOR TRAVEL PATTERN ANALYSIS WITHIN THE SOUTHEASTERN WISCONSIN REGION
procedures used to validate each individual model are presented. A test of the adequacy of the entire four-step simulation process is also presented in the traffic assignment section. In this section the simulated traffic volumes derived from using the initial and revised simulation models are compared with actual observed vehicle ground counts.

TRANSPORTATION-SPECIFIC LAND USE FORECASTS

As noted earlier, current traffic simulation models are based upon relationships between land use and travel. As such, the development and application of traffic simulation models requires a detailed description of existing land uses to calibrate the models and a similar description of future land uses to apply the models. As noted earlier, the term "land use" as used in this context has a special meaning, including demographic and economic variables.

Ideally, it would be desirable, from a planning standpoint, to be able to express trip end generation rates in terms of the broad, generalized land use categories used in the existing land use inventories and in the preparation of land use plans. Each of these broad land use categories, however, represents a complex of related and unrelated human activities comprising the reasons for which trips are actually made. Broad land use categories, therefore, usually are inadequate as a basis for analyzing and forecasting trip generation. Refinement of these broad land use categories is, therefore, required for trip generation analysis and forecast purposes; and this refinement is usually best expressed in terms of more detailed characteristics of the activities taking place on the land, such as resident population, population density, employment, income level, and automobile availability. These more detailed socio-economic characteristics of land use, however, must be capable of derivation from a forecast of future land use, or from a land use plan. Since the ability to predict or control the future location and intensity of land use activity tends to diminish rapidly as the detail of classification increases, a balance must be struck between the transportation planning needs and the land use planning capabilities.

Chapter VI of this report includes a description of the methodologies used to establish forecast and planned levels of future land use variables, including land use, population, households, and employment. In addition, there are certain land use variables which are specifically required by and thus derived for the travel simulation process. The next section describes the specific procedures used to forecast and allocate the transportation-specific land use variables of vehicle availability in the land use-transportation plan reevaluation effort.

Personal Vehicles

The availability of a personal vehicle, be it automobile, van, or truck, is a very significant variable in the travel simulation modeling process, influencing not only the ability to make a trip, or trip generation, but also the decision of which travel mode to use in making the tripmode choice. For traffic simulation purposes, it is necessary to forecast vehicle availability not only on the regional and county level, but it is also necessary to allocate regional totals to the traffic analysis zone level.

An important criterion in the selection of the model to be used to forecast vehicle ownership was that it realistically reflect the decision process of owning a vehicle through the variables specified in its formulation to represent the decision process of households. In this respect, vehicle availability, or ownership, within an area is influenced, for the most part, by the number of households or persons within the area and by the residential density of the area, and the characteristics of household income and family size of the persons residing in the area. These relationships are rational, because the number of vehicles per household would logically be affected by household income, indicating ability of ownership, and influenced by residential density, indicating ability to use public transit or walking as a means of transportation. Analyses of 1990 U.S. Census and origin and destination survey data indicated, however, that in central Milwaukee County, where vehicle ownership and incomes are relatively low, vehicle ownership was principally affected by income.

Therefore, under the Commission's battery of travel simulation models, average household vehicle availability was determined with two linear equations developed through multiple regression analyses with 1990 U.S. Bureau of Census socio-economic data and 1990 Commission land use data and applied at the traffic analysis zone level. Reflecting the above discussion, the model for household vehicle availability in central Milwaukee County is a function of household income, and for the remainder of the Region it is a function of household income, household size, and residential density. Specifically, the equation used to derive average zonal household vehicle availability is:

Central Milwaukee County

VAH = 0.0466 HI - 0.0622

Remainder of Region

 $VAH = 0.445 \log_{e}(HI) + 0.200 PH - 0.163 \log_{e}(HGRA) - 0.144$

 Where:
 VAH
 = number of vehicles available per household

 HI
 = average household income in \$1,000s

 PH
 = number of persons per household

 HGRA
 = number of households per developed gross

residential acre

Because the cross-classification models used to forecast total person trip productions require knowledge of the distribution of households within each zone with zero, one, or two or more vehicles available by household size, one person, two persons, three or four persons, and five or more persons, a household stratification model was developed to forecast the probability of auto availability by household size. The household stratification model is actually a combination of a household size stratification model and vehicle ownership stratification model. The household size stratification model provides estimates of the proportion of households within a zone for each specific household size category, given the zonal average household size as derived from forecast population and household levels. Similarly, the vehicle ownership stratification model provides estimates of the proportion of households within a zone for each vehicle ownership category, given the zonal average household vehicle ownership as derived by the above equation. The household size and vehicle ownership stratification models were verified and calibrated by using zonal population, household, and vehicle data from the 1991 origin-destination survey.

In order to determine the number of households within each category by both household size and vehicle availability, the results of the two individual stratification models were combined. This was done by defining the proportions of households in each category as probabilities and then applying elementary probability theory to determine the joint probability of a household possessing a specific family size and a specific vehicle availability. If the two characteristics were considered independent, then the joint probability of a particular household size category K (HHS_k), and a particular vehicle availability category j, (AA_j) for a zone i would merely be the product of the two probabilities, or:

$$P(HHS_k - AA_j)i = P(HHS_k)i \times P(AA_j)i$$

Examination of the available data, however, indicated that there was a certain degree of dependence between the two characteristics. The joint probability thus includes a conditional probability, that is:

$$P(HHS_k - (AA_i)i = P(AA_i)i \times P(HHS_k/AA_i)i.$$

The value of the conditional probability $(PAA_j/HHS_k/)i$, representing the probability of vehicle availability category j given the household size category K, was determined by the calibration of the equation:

$$P(HHS_k/AA_i)i = C_1(HHS_k)i + C_2P(HHS_k/AA_i)$$
region

Where: C_1 and C_2 are the calibrated weights reflecting the amount of independence and dependence, respectively, of the two characteristics, and C_1 and $C_2 = 1$. P(HHS_k/AA_j)region is the observed conditional probability for the Region.

A number of combinations of C_1 and C_2 were examined to find that combination which best replicated the observed household classification. The final version of the household stratification model, using equal dependent and independent weights, is summarized as:

 $P(HHS_{k} - AA_{i})i = P(AA_{i})i \times [0.5 P(HHS_{k})i + 0.5 P(HHS_{k}/AA_{i})region]$

Travel survey-estimated and model-estimated distributions of households by size and automobile availability are shown in Table 148.

TRIP GENERATION

Basic Concepts

The first major step in the travel simulation process is trip generation whereby the total number of trip ends generated within each zone of the study area is determined through the identification and quantification of relationships between travel and land use. As such, trip

PERCENTAGE OF HOUSEHOLD BY VEHICLES AVAILABLE AND HOUSEHOLD SIZE: 1991

Household Travel Survey Estimate						
		Household Size				
Vehicles Available	One	Two	Three to Four	Five or More	Total	
None One Two or More	5.4 17.8 1.8	1.5 10.2 20.0	1.5 5.0 26.1	0.7 1.6 8.4	9.1 34.6 56.3	
Total	25.0	31.7	32.6	10.7	100.0	

Model Estimate							
Vehicles Available	One	Two	Three to Four	Five or More	Total		
None One Two or More	4.6 13.5 7.0	2.6 10.8 18.3	2.6 8.1 21.9	1.0 2.5 7.1	10.8 34.9 54.3		
Total	25.1	31.7	32.6	10.6	100.0		

Source: SEWRPC.

generation occupies a critical position between land use and transportation planning. The inputs to this step include a detailed accounting of relevant existing and forecast or planned future land use characteristics. The output of the trip generation step is an estimate of the probable future number of trips, usually by trip purpose, entering or leaving each zone.

Within trip generation, then, the travel data are expressed in terms of trip ends, which may be conveniently represented as points in space with no regard to the direction, length, or duration of the trip. By convention, one end of each trip is termed the "production end" while the other end is termed the "attraction end." For trips beginning or ending at home, termed "home-based trips," the production end is always considered as the home end of the trip while the attraction end is always considered as the nonhome end, irrespective of the actual direction of the trip. For trips having neither end at home, termed "nonhome-based trips," the production end is defined as the origin of the trip while the attraction end is defined as the destination.

Two sets of trip generation relationships are consequently developed. One set is developed for the production end of trips generated within the planning area and involves relating trip ends to primarily residential land uses. The other set is developed for the attraction end of trips generated within the planning area and involves relating trip ends to primarily nonresidential land uses.

There are four basic approaches to trip generation: factoring of existing trip patterns, land area trip analysis, regression analysis, and cross-classification analysis. The most basic approach to trip generation is the application of growth factors to existing travel patterns. Since the growth factor is applied to observed trip interchanges, this approach actually represents a combination of the trip generation and trip distribution steps. Usually this approach is used only for relatively small trip categories, such as external trips, or for trip purposes for which trip distribution would be artificially constrained, such as school trips.

The second approach deals explicitly with land areas and has as its objective the establishment of trip rates which reflect the character, location, and intensity of land use. Trip rates, in terms of trips per acre, are determined for different kinds of land uses, such as residential, commercial, and industrial. To project future trip generation, these rates are simply applied to the projected land use distribution in each zone.

The third approach involves the development of trip generation relationships by using linear regression analysis. In this procedure, the dependent variable, the number of trips generated per zone or household, is determined by a linear combination of independent variables representing the land use and socio-economic characteristics of the zone or household. To account for any potential nonlinear effects of an independent variable on the dependent variable, a monotonic transform, such as a logarithmic or exponential transform, may be made on an independent variable.

The fourth approach involves cross-classification, whereby the trip generation relationships are developed by establishing a multi-dimensional matrix. Each dimension of the matrix represents an independent or explanatory variable of tripmaking, such as household income, household size, or automobile availability, which is stratified into several classes. Values of the dependent variables, number of trips, are accumulated for each matrix cell from the household survey data; thus the mean trip generation rate for that cell is determined. Thus, the value of each matrix cell represents the average trip production rate of households which possess the cell's independent variable characteristics. The model can then be applied to any geographic subarea of the planning area to obtain trip production estimates by linking the characteristics of the households within the subarea to the appropriate trip generation rate and summing the total number of trips made by all households residing in that area.

The selection of approaches to determine trip generation is dependent upon the type of trip to be simulated, the data available for calibration, and the particular needs and character of the study area.

Validation of Second-Generation Trip Generation Procedures: The second-generation trip generation models developed by the Commission with 1972 origin and destination survey data for internal regional travel by resident households of the Region use cross-classification analysis for trip productions and multiple regression analysis for trip attractions for all trip purposes except school trips. School trip generation involved factoring of existing travel patterns for school trips. These second-generation trip generation models are fully documented in SEWRPC Planning Report No. 25, A Regional Land Use Plan and a Regional Transportation Plan for Southeastern Wisconsin: 2000, Volume Two, Alternative and Recommended Plans, May 1978.

Internal home-based and nonhome-based trips by the residents of households in the Region for all purposes except school constitute the vast majority of daily trips made within the Region, over 80 percent. The production of these homebased and nonhome-based trips was analyzed and forecast under the second-generation models through the use of cross-classification analysis. Home-based trips were stratified into categories of home-based work, home-based shopping, and home-based other. The home-based trip purposes of personal business, medical-dental, socialeatmeal, and recreation, as well as trips made solely to transport passengers, were combined into a single category, home-based other, since such trips were found in the travel inventories to have reasonably similar trip production, attraction, and length characteristics. Within each trip purpose used in the modeling process, four cross-classification models were developed for forecasting future household trip production by geographic location. These models were developed for the Milwaukee urban area, the Racine urban area, the Kenosha urban area, and the remaining rural area within the Region.

An important consideration in the development of the trip production cross-classification models was the selection of the variables to be used to explain household trip frequency. Among the variables considered and investigated were automobile availability, household size, household income, structure type, neighborhood density, and stage in family life cycle. Household automobile availability and household size were selected as the independent variables to explain tripmaking in the trip generation model, since these variables exhibited a high correlation with tripmaking frequency and explained most of the variation in household tripmaking as observed in the 1963 and 1972 surveys.

The ability of the second-generation trip production and attraction models to simulate 1991 internal person tripmaking was investigated by comparing the result of the application of the trip production and attraction equations using 1990 land use data to actual 1991 travel survey data. As shown in Table 149, the travel surveys conducted by the Commission indicated that trip generation within Southeastern Wisconsin increased substantially in all trip purpose categories from 1972 to 1991. The ability of the trip generation models developed in 1972 to predict these changes in regional trip generation accurately is also demonstrated in Table 149. As can be seen, the equations were able to predict regional trip generation quite closely considering the nature of the phenomena involved. In this respect, it must be recognized that the actual 1991 regional trip generation used the 1972 trip generation data used to calibrate the original equations as the basis for comparison and much of the data necessary to prepare predictions of 1991 trip generation, such as household socioeconomic characteristics, are in themselves estimates. Thus, considering the limitations inherent in the data being compared, it may be

COMPARISON OF OBSERVED AND ESTIMATED INTERNAL PERSON TRIP GENERATION WITHIN THE SOUTHEASTERN WISCONSIN REGION: 1972 AND 1991

Trip Purpose	1972	1991	Percent Change	1991	Percent
	Observed	Observed	1972-1991	Estimated ^a	Difference ^b
Home-Based Work	1,055,500	1,302,700	23.4	1,355,400	4.1
Home-Based Shopping	673,600	798,100	18.4	802,300	0.5
Home-Based Other	1,532,600	1,687,400	10.1	1,960,300	16.1
Nonhome-Based	779,800	1,127,500	44.6	1,075,100	-4.9
Total	4,041,500	4,915,700	21.6	5,193,600	5.6

^aEstimated with 1972 models using 1990 land use inputs.

^bPercent difference = <u>(Estimated-Observed)</u> X 100 Observed

Source: SEWRPC.

concluded that the total trips generated within the Region in 1991 were predicted by the equations developed under the 1972 secondgeneration planning effort with a high degree of accuracy even though some divergence exists within specific trip purpose categories.

As already noted, the generation of school trips was measured in the second-generation travel models by factoring of existing travel patterns. Such separate consideration of home-based school trips was necessitated by the limitations imposed by fixed school service area boundaries. Since trips to and from all schools, including elementary, junior and senior high, vocational and technical schools, and colleges and universities, amounted to less than 6 percent of total person trips generated within the Region on an average weekday in 1972 and to less than 3 percent of total vehicle trips, and since most schools serve local neighborhoods or communities so that tripmaking to and from such schools has a very small impact on arterial street and highway travel, it was considered adequate for planning purposes to let the 1972 pattern of vehicle trips for school purposes represent the probable future pattern of such trips. A factor based upon population growth was applied to the observed 1972 automobile school trip interchanges to derive the future-year interchange volumes.

Transit trips to and from schools comprised about 31 percent of total transit trips made within the Region on an average weekday in 1972. Transit trips to or from schools were forecast by initially applying a growth factor based upon population and by further adjusting the trips in those zones where significant changes in school enrollment were anticipated. Adjustments were made in those areas where school service boundaries were expected to change in accordance with the recommended land use plan, where new construction of educational institutions was anticipated, and where improvements in public transit service were programmed. The most significant of these adjustments reflected the improvement of transit service to the University of Wisconsin-Milwaukee (UWM) resulting from the 1974 initiation of the UBUS program and from the opening of three new vocational school extensions: Milwaukee Area Technical College in Oak Creek and Mequon and Waukesha County Technical Institute in Pewaukee.

The ability of the initial study procedures to predict the future generation of home-based school trips can be discerned by a comparison of the estimated and observed trip generation for 1991. The travel inventories indicated that the number of school trips in the Region increased 49.9 percent between 1972 and 1991, from 418,400 to 627,100, respectively. These totals represent all school trips made on an average weekday by all modes of travel, including the school bus. Using the second-generation procedures, a 1991 estimate of 480,000 school trips was derived, a total approximately 23 percent lower than the observed figure largely as a result of underestimation of school purpose trips by automobile. Considering the nature and significance of the home-based school trips, it may be concluded that the second-generation modelling procedure was adequate for traffic analysis purposes. Consequently, the same procedure was used in the third-generation models.

The generation of internal trips by groupquartered persons and nonresidents was forecast in the second-generation travel models by the application of uniform growth factors. Trip generation by group-quartered persons, persons residing in dormitories, convents, homes for the aged, and other similar residences, and by nonresidents was considered separately from that for persons living in households because the survey data indicated that the travel behavior and characteristics of group-quartered persons and of nonresidents in the Region were significantly different from those of persons living within the Region in households. Accounting for less than 1 percent of the total person trips within the Region in 1972, travel by groupquartered and nonresident persons was accordingly analyzed and then forecast through use of a growth factor. This approach assumed that the 1972 pattern of trips could be adjusted through application of zonal growth factors reflecting the increases in travel of group-quartered persons by zone, expressed as a function of the change in the number of group-quartered persons by zone of residence. The travel surveys indicated that. from 1972 to 1991, average weekday travel by group-quartered persons increased from 25,400 trips to 53,400 trips, or a 110.2 percent increase, which compares favorably to the 60,500 trips by group-quartered persons forecast by the secondgeneration models developed in 1972.

Truck trips within the Region constitute a relatively small proportion of total vehicle trips, approximately 12 percent of all such trips generated within the Region on an average weekday. In the second-generation regional transportation planning effort, the generation of truck trips was accomplished by the use of multiple linear regression analysis. Taxi trips, comprising less than one-third of 1 percent of total trips within the Region on an average weekday, were included with truck trips because they displayed trip characteristics similar to those of truck trips. Analysis of the 1972 travel inventory data had indicated that both truck and taxi trips were primarily of a nonhome-based nature with short average trip lengths; trips made by both types of vehicles were made as essentially pickup and delivery functions.

The ability of the truck and taxi trip generation technique utilized in the second-generation planning program to predict 1991 truck and taxi travel was evaluated by using inventory data from the 1991 origin and destination survey. These travel inventories indicated that truck and taxi travel within the Region had increased substantially from 385,300 to 536,500 trips per average weekday over the past two decades. The travel forecasting techniques, as developed in 1972, predicted this increase in regional truck and taxi travel within 14 percent of the actual number of truck and taxi trips observed in the 1991 travel survey. In view of the demonstrated accuracy of this technique employed in the second-generation planning effort, the same technique was utilized for forecasting truck and taxi trips within the Region in the plan reevaluation process.

External trips comprised less than 4 percent of total vehicular travel within the Region in 1972. In the second-generation land use-transportation planning effort, external travel was forecast by extrapolating the existing 1972 pattern of external tripmaking by applying growth factors to the 1972 trips on the basis of the forecast changes in population over employment at the production and attraction ends of the trips, respectively.

The ability of this forecasting technique to forecast external tripmaking within the Region in 1991 was tested by using the travel inventory data collected by the Commission in 1991. Over the past two decades, external travel affecting the Region has increased by approximately 117 percent, from 125,700 vehicle trips per average weekday in 1972 to 273,300 such trips in 1991. Comparison of observed and forecast 1991 external travel indicated that the secondgeneration study procedures underestimated the increase in external travel, predicting an increase to 160,000 trips. The underestimation was a result of underestimation of the decline in vehicle occupancy and of basing the prediction of the increase in travel on population change rather than change in households and employment.

Third-Generation Trip Generation Procedures

The third-generation trip generation relationships developed with the new 1991 origin and destination survey data again use crossclassification analysis for major internal resident household home-based trip productions, cross-classification analysis for major internal resident household home-based trip attractions, and factoring of existing travel patterns for school trips.

The cross-classification, or category trip production models calibrated for the Milwaukee, Racine, and Kenosha urban areas and the remaining area of the Region, are presented by trip purpose in Tables 150 through 153. Like the second-generation models, the third-generation models provide unique trip production rates for each of the Region's three urban areas and for the rural areas of the Region as those areas are shown on Map 80. The models base the household trip production rate on household size and on household automobile availability. As shown in these tables, the trip generation rates vary with the areas concerned, generally exhibiting higher tripmaking frequencies in the urban than in the rural area of the Region, particularly for nonwork trip purposes. Tables 150 through 153 also indicate the generally direct relationships that exist between the rate of household tripmaking and household size and household automobile availability.

While Table 153 includes a cross-classification trip rate model for nonhome-based trip productions, the forecast of nonhome-based trip production by zone must be accomplished through an additional step which allocates the forecast regional total production of nonhome-based trips to zones. The production of nonhome-based trips cannot be directly simulated by zone with the cross-classification approach since neither end of the trip represents the place of residence of the tripmaker. Cross-classification, however, can provide an estimate of total regional nonhomebased trip productions based on the total number of regional households and their characteristics. The allocation of the regional totals of nonhomebased trip productions to each zone is accomplished by use of a trip rate, which is also developed through cross-classification analysis. This trip rate relates the number of nonhomebased trip productions in a zone to the number of households, retail employment, and other employment in a zone, based upon analyses of the 1991 household travel survey data with respect to nonhome-based trip production from residential land uses, retail land uses, and other land uses.

NHBP = 0.29 X HH + 2.38 X REMP + 0.76 X OEMP

Where:	NHBP	=	number of nonhome-based productions
	нн	=	number of households
	REMP	=	number of retail employees
	OEMP	=	number of other employees

The other set of trip end relationships developed in the trip generation process was for trip attractions, which are primarily a function of the nonresidential land use activity within the subareas of the Region. Person trip attraction relationships were developed through the calibration of four cross-classification, or trip rate models, representing home-based work, homebased shopping, home-based other, and nonhome-based trip purposes. The models relate home-based person trip attractions to employment by type and households on the basis of analysis of the 1991 household travel survey with respect to the trip attractions by trip purpose to residential and nonresidential land uses. The calibrated trip attraction equations are presented in Figure 60.

Two adjustments are made to the forecast trip attractions derived through application of the calibrated trip attraction model. The first adjustment is necessary to account for special trip attraction rate to major commercial centers and to General Mitchell International Airport.

The second adjustment is made to balance the total number of forecast trip attractions and the total number of forecast trip productions. Because trip productions and attractions are forecast independently, the total number of forecast trip productions do not always match the total number of forecast trip attractions. Work trip productions are factored so that the regional sum of the productions equals the regional total work trip attractions. The latter are based directly upon the number of jobs in the Region and are, therefore, a better estimate of total work trips made within the Region.

INTERNAL PERSON TRIP PRODUCTION CROSS-CLASSIFICATION MODELS: HOME-BASED WORK TRIPS

Milwaukee Urbanized Area						
		T	rips per Ho	ousehold		
		Household Size				
Vehicles Available	One	Two	Three to Four	Five or More	Average	
None	0.40	0.74	0.94	0.97	0.60	
One	1.01	1.09	1.84	2.04	1.21	
Two or More	1.16	2.30	3.00	3.21	2.72	

Racine Urbanized Area							
		Т	rips per Ho	ousehold			
		Hous	ehold Size)			
Vehicles			Three	Five			
Available	One	Two	to Four	or More	Average		
None	0.18	0.40	0.89	0.99	0.29		
One	1.03	1.09	1.74	2.13	1.20		
Two or More	1.33	2.11	2.87	3.30	2.62		
Average					2.04		

Kenosha Urbanized Area Trips per Household Household Size Three Vehicles Five Available to Four One Two or More Average 0.77 0.35 0.19 0.31 0.86 None 0.84 0.69 0.76 1.34 1.53 One Two or More 1.16 1.49 2.35 2.49 2.03 1.52 Average - -- -- -- -

Rural Area							
		Trips per Household					
		Household Size					
Vehicles Available	One	Two	Three to Four	Five or More	Average		
None One Two or More	0.20 0.77 1.07	0.40 0.79 1.99	1.20 1.53 2.75	1.30 1.98 3.04	0.31 0.91 2.48		
Average					2.01		

Source: SEWRPC.

Table 151

INTERNAL PERSON TRIP PRODUCTION CROSS-CLASSIFICATION MODELS: HOME-BASED SHOPPING TRIPS

Milwaukee Urbanized Area						
		Т	rips per Ho	ousehold		
		Hous	ehold Size))		
Vehicles Available	One	Two	Three to Four	Five or More	Average	
None One Two or More	0.32 0.67 0.68	0.34 1.13 1.27	0.51 1.33 1.63	0.57 1.54 1.85	0.39 0.95 1.50	
Average					1.17	

Racine Urbanized Area						
		Т	rips per Ho	ousehold		
		Hous	ehold Size	9		
Vehicles Available	One	Two	Three to Four	Five or More	Average	
None One Two or More	0.33 0.70 0.89	0.39 1.18 1.39	0.69 1.26 1.69	0.89 1.38 1.97	0.38 0.94 1.60	
Average					1.32	

Kenosha Urbanized Area Trips per Household Household Size Vehicles Three Five Available One Two to Four or More Average 0.24 0.33 0.66 0.75 0.36 None One 0.69 1.17 1.24 1.41 0.93 Two or More 0.83 1.41 1.66 1.82 1.57 1.28 Average - -- -- -- -

Rural Area							
		Trips per Household					
		Household Size					
Vehicles Available	One	Two	Three to Four	Five or More	Average		
None One Two or More	0.22 0.69 0.73	0.45 0.94 1.09	0.73 1.13 1.24	0.94 1.35 1.77	0.30 0.84 1.26		
Average					1.12		

INTERNAL PERSON TRIP PRODUCTION CROSS-CLASSIFICATION MODELS: HOME-BASED OTHER TRIPS

Milwaukee Urbanized Area						
		T	rips per Ho	ousehold		
		Household Size				
Vehicles Available	One	Two	Three to Four	Five or More	Average	
None	0.45	0.92	1.19	1.34	0.73	
Two or More	1.65	2.42	3.19	5.02	3.13	
Average					2.35	

Racine Urbanized Area								
		Trips per Household						
		Household Size						
Vehicles			Three	Five				
Available	One	Two	to Four	or More	Average			
None	0.52	0.78	1.17	1.94	0.63			
One	1.03	1.09	1.74	2.13	2.06			
Two or More	1.33	2.11	2.87	3.30	3.70			
Average					3.00			

Source: SEWRPC.

Kenosha Urbanized Area						
		T	rips per Ho	ousehold		
		Hous	ehold Size	9		
Vehicles			Three	Five		
Available	One	Two	to Four	or More	Average	
None	0.42	0.84	1.50	2.53	0.31	
One	1.32	2.49	3.29	4.13	2.05	
Two or More	2.25	2.73	4.04	5.45	3.72	
Average					2.98	

Rural Area						
		Т	rips per Ho	ousehold		
		Hous	ehold Size)		
Vehicles			Three	Five		
Available	One	Two	to Four	or More	Average	
None	0.41	0.71	1.22	2.54	0.58	
One	1.30	1.93	3.05	3.49	1.78	
Two or More	1.44	2.08	3.41	5.25	3.19	
Average					2.75	

Table 153

INTERNAL PERSON TRIP PRODUCTION CROSS-CLASSIFICATION MODELS: NONHOME-BASED TRIPS

Milwaukee Urbanized Area						
		T	rips per Ho	ousehold		
		Hous	ehold Size	•		
Vehicles Available	One	Two	Three to Four	Five or More	Average	
None	0.16	0.37	0.49	0.54	0.28	
One	1.06	1.31	1.36	1.46	1.20	
Two or More	1.31	2.09	2.17	2.42	2.15	
Average					1.59	

Racine Urbanized Area						
		Ti	rips per Ho	ousehold		
		Hous	ehold Size)		
Vehicles			Three	Five		
Available	One	Two	to Four	or More	Average	
None	0.18	0.28	0.37	0.45	0.22	
One	1.11	1.34	1.48	1.86	1.26	
Two or More	1.94	2.03	2.67	3.23	2.51	
Average					1.98	

Kenosha Urbanized Area Trips per Household Household Size Vehicles Three Five Average Available One Two to Four or More 0.26 0.21 0.31 0.38 0.43 None 0.93 1.01 1.27 1.54 1.03 One Two or More 1.81 2.02 2.32 2.47 2.22 Average - -- -1.69 - -- -

Rural Area						
		T	rips per Ho	ousehold		
		Hous	ehold Size)		
Vehicles			Three	Five		
Available	One	Two	to Four	or More	Average	
None	0.20	0.32	0.40	0.49	0.24	
One	0.93	1.29	1.72	1.91	1.17	
Two or More	1.47	1.97	2.30	2.36	2.17	
Average					1.85	



Individual trip production rates were developed for each of the three major urban areas of the Region, Milwaukee, Racine, and Kenosha, as well as for the remaining rural areas. This map identifies those four individual areas. Trip generation rates vary within the areas concerned on the basis of the number and characteristics of the households and upon household automobile availability, generally exhibiting higher tripmaking frequencies in the urban than in the rural areas of the Region.

Figure 60

		Trins per Job or Household	
		The per set of Household	
Home-Based Work Trips	Region		
Total Employment	1.32		
	Central	Remainder	
Home-Based Shopping Trips	<u>Milwaukee County^a</u>	of Region	
Retail Employment	2.74	5.38	
	Central	Remainder of	Remainder
Home-Based Other Trips	<u>Milwaukee County^a</u>	Milwaukee_County	of Region
Households	0.25	0.48	0.64
Retail Employment	1.25	1.41	1.49
Other Employment	0.66	1.53	2.26
	Central	Remainder of	Remainder
<u>Nonhome-Based Trips</u>	<u>Milwaukee County^a</u>	Milwaukee_County	of Region
Households	0.15	0.21	0.24
Retail Employment	1.94	3.11	3.27
Other Employment	0.39	0.66	0.68
• • •			

INTERNAL PERSON TRIP ATTRACTION CROSS-CLASSIFICATION MODELS

^aCentral Milwaukee County includes areas 29, 30, and 32 as shown on Map 79.

Source: SEWRPC.

For the other trip purposes, home-based shopping, home-based other, and nonhome-based, the zonal trip attractions derived from application of the trip attraction model are factored so that the sum of the zonal trip attractions for each nonwork trip purpose equals the total regional cross-classification estimate of trip productions for that trip purpose. This procedure for the nonwork trip purposes is applied because it was considered that for nonwork trip purposes the cross-classification models used to forecast trip productions provide a better estimate of the total number of nonwork trips made within the Region than do the trip attraction models.

As already noted, the exclusion of trips to or from all schools, be they elementary, junior high, senior high, vocational, and technical, college, or university, from the other internal trip production, attraction, and distribution analyses is necessary because a trip distribution modeling procedure would inherently treat all schools as possible attractions, not being able to account for limitations imposed by school service boundaries. The generation of school trips was, therefore, measured separately through the application of growth factors to the observed 1991 travel patterns for school purpose trips made by each mode, automobile, school bus, and public transit. In 1991, vehicle trips made for school purposes by automobile and school bus accounted for approximately 3 percent of all regional vehicle trips. Trips made for school purposes by public transit accounted for about 40 percent of all transit person trips in the Region.

Essentially, school purpose trips are generated in two steps. In the first step, growth factors are applied separately by mode to the observed 1991 trip tables of school trips. The growth factors are based upon the forecast changes in population observed at the planning analysis level between 1990 and the plan design year. The average growth factor used for the Milwaukee area for Grades K through 12 is 1.16. In the second step, the initial forecasts of auto and transit person school trips are then adjusted at the traffic analysis zone level to account for potential changes in school service boundaries and the construction of new educational institutions. With respect to trips to such universities and colleges within central Milwaukee County as the University of Wisconsin-Milwaukee, Marquette University, Milwaukee Area Technical College, and Milwaukee School of Engineering, the growth factor procedure is applied to total person, rather than to individual mode, travel patterns and the growth factor is based on projected enrollment. A mode choice model, described later in this report then is used to divide the total person trips into those using public transit and those using the automobile. Public transit school trips to these colleges and universities represent about 20 percent of all school purpose public transit trips within Southeastern Wisconsin.

School trips made by public transit on special bus routes are operated only on schooldays. These special schoolday-only routes are designed to serve only elementary and secondary school students, providing only a very limited level of service each weekday. They typically consist of bus runs to school in the morning and from school in the evening. These routes are not included in the transit network used in the traffic assignment process. School purpose transit trips served by special bus routes are also not included in the forecast transit travel demand; they are removed from existing school transit trip tables. It is estimated that about 5 percent of the public transit school purpose trips made in the Milwaukee urbanized area in 1991 use such special schoolday-only bus routes.

Accounting for less than 1 percent of total regional travel, the generation of internal trips by group-quartered persons and nonresidents was accomplished by the application of uniform growth factors, similar to the approach used in the second-generation travel models. This approach assumes that the 1991 pattern of trips can be adjusted through application of zonal growth factors reflecting the increases in travel of group-quartered persons by zone, expressed as a function of the change in the number of groupquartered persons by zone of residence. It was assumed that the trip generation rate of groupquartered persons as observed in the 1991 travel survey would remain stable.

The generation of truck and taxi trips was accomplished using the same technique as in the initial and second-generation studies, namely, multiple regression analysis. Using the 1991 travel survey data, a new analysis of the relationship between land use and truck and taxi tripmaking was undertaken. The results of the analysis indicated that truck and taxi trip generation could be best expressed on a zonal level as a function of employment, population, and retail and service land, the same variables used to explain truck and taxi trip generation in the initial and second-generation studies. The new truck and taxi trip generation equation as derived from this analysis is:

TTRP = 0.05 X POP + 0.11 X REMP + 0.17 X OTHEMP + 234

Where:	TTRP	=	Number of truck and taxi trip ends
	REMP	=	Number of persons Number of retail employees
	OTHEMP	=	Number of other employees

Unlike the equation used in 1963 and 1972, which was based on traffic analysis zone data from 619 and 1,220 subareas within the Region, respectively, the 1991 equation was developed for a zone system of 1,431 subareas within the Region. As a result of this change in the aggregation of the data used in the calibration of the model and because slight changes have occurred over time in the relationship between truck and taxi trip generation and population and employment, and particularly, retail and service land use, the terms and factors in the equation defined with 1991 data vary modestly from those in the equation used in the previous planning efforts.

The generation of the final trip category, external trips, was accomplished through the use of a growth factor approach. Growth factors were developed for defined external and internal areas which relate the forecast change in trips to the change in the number of households at the production trip end and employment at the attraction trip end.

TRIP DISTRIBUTION

Basic Concepts

The second major step in the travel simulation process is trip distribution whereby the number of trips between each zonal pair is determined. The input to this step from trip generation is the number of trip ends produced by, or attracted to, each zone. Additional inputs, such as the travel times between zones, may also be required for calibration and application depending on the type of trip distribution model used. There are currently three basic types of trip distribution models in general use: growth factor models, the gravity model, and the opportunity model. Growth factor models derive the number of trips between two zones for some projection year as a function of the number of trips observed between those two zones in the base year and some growth factor. The growth factor may simply represent an annual increment of number of trips or may be an annual percentage increase similar to a compounding interest function. The actual growth factors used, however, are usually determined by some other characteristic, such as forecast population growth. As such, this type of formulation represents a combined trip generation-trip distribution model and is used for relatively small trip categories, which cannot be easily simulated by more sophisticated techniques. The most popular growth factor model is the Fratar method, in which growth factors are applied to both the trip productions and attractions. A trip generation model must precede the Fratar method since the growth factors are calculated as the ratio of projection year productions and attractions to base year productions and attractions. Calibration of the Fratar model is necessary to ensure that the number of projection year trips entering or leaving each zone matches the generated productions and attractions for that zone, respectively.

The gravity model is the most widely accepted and used trip distribution model. The basic premise of the gravity model is that a trip interchange, or the number of trips between two zones in the study area, is a direct function of the number of trip ends in each zone and some inverse function of their spatial separation. This function of spatial separation adjusts the relative attraction of each zone by the ability, desire, or necessity of the tripmaker to overcome the travel distance or travel time involved.

Mathematically, the gravity model may be stated as follows:

$$\Gamma_{ij} = P_i \frac{\frac{A_j}{d_{ij}^b}}{\left(\frac{A_1}{d_{i1}^b} + \frac{A_2}{d_{i2}^b} + \dots + \frac{A_n}{d_{in}^b}\right)}$$

Where: $T_{ij} = trips produced in zone i and attracted to zone j$

- P_i = trips produced by zone i
- $A_{j} = trips$ attracted by zone j
- d_{ij} = the spatial separation between zones i and j, generally expressed in terms of door-to-door travel time
- an empirically determined exponent which expresses the average areawide effect of spatial separation between zones on trip interchange
- n = the number of traffic analysis zones within the planning area

The exponent b has been observed to vary with trip purpose, assuming values of about 3.0 for social trips, 2.0 for shopping trips, and 1.0 for work trips, when spatial separation has been expressed as in-vehicle travel time.

The decrease in the exponent implies that spatial separation is a less restrictive factor on a trip interchange or that people generally are willing to travel farther for a purpose such as work than for purposes such as shopping or social functions. The exponent b also has been observed to increase as the separation increased, indicating that the effect of spatial separation increases as the separation itself increases. Moreover, the value of the exponent has been found to vary from urban area to urban area, particularly for nonwork purpose trips.

As a consequence of the variability in the exponent b, it is necessary to develop and calibrate gravity work models for each region under study and for each trip purpose category considered. Moreover, since past experience has demonstrated that the exponent of travel time is not necessarily constant for all intervals of time and that travel patterns are affected by various social and economic characteristics of the travelers, it has become common practice to express the gravity model formula in the following form:

$$T_{ij} = \frac{P_i A_j F_{ij} K_{ij}}{\sum_{\substack{\Sigma \\ j = 1}} A_j F_{ij} K_{ij}}$$

Where: F_{ij} = an empirically derived travel time friction factor which expresses the average areawide effect of spatial separation on trip interchange between zones which are tij minutes apart

- K_{ij} = an adjustment factor applied on a zoneto-zone basis to allow for the incorporation of the effect on travel patterns of social, economic, political, or historic characteristics not otherwise accounted for in the model formulation
- T_{ii}, P_i, and A_i are as previously defined

The use of the set of travel time friction factors to express the effect of spatial distribution on zonal trip interchange as a modification of the classic inverse exponential function serves to provide for the consideration that the effect of spatial separation generally increases as the separation itself increases. Derived from the characteristics of the origin and destination zones, the zonal adjustment factor is essentially the ratio necessary to adjust the model so as to match computed travel patterns with the travel patterns observed between subareas of the Region in origin and destination surveys. This factor accounts quantitatively for effects of biases which can generally be identified qualitatively through experienced knowledge about the areas affected.

In order to apply the gravity model to forecast future travel patterns, it is necessary to calibrate the model to reflect existing travel patterns and characteristics within a Region accurately. This calibration process actually determines the numerical values of the travel time friction factors and the zonal adjustment factors so that the gravity model accurately simulates the trip length characteristics determined in the travel inventory. These numerical values are assumed to remain constant over time, thereby providing a model which can be used to simulate the future trip interchange patterns, given future trip productions, attractions, and travel times between subareas of the Region.

The available evidence indicates that the assumption that the friction factors are stable over time is reasonable. However, research studies have shown that drastic changes in the level of service provided by the transportation system or radical changes in the distribution of land use activity throughout a Region will invalidate this assumption. The assumption of constant zonal adjustment factors is more difficult to justify since social, economic, historic, or political effects or biases which exist in the base year may not exist in the future. The final type of trip distribution model is the opportunity model, either the intervening opportunity model or the competing opportunity model. Although both of these models have a strong theoretical basis, there is no standardized procedure for calibration, operation, or application which is as efficient or as well tested as the gravity model. Moreover, little research has been done to verify the ability of the opportunity model to either replicate existing travel patterns or remain stable over time. Consequently, while investigated, the opportunity model was not considered a viable alternative for the trip distribution stage in the plan reevaluation effort.

Validation of Second-Generation

Trip Distribution Procedures

The distribution of major internal person trips was accomplished in the second-generation models developed with 1972 origin and destination survey data by using the gravity model. Individual gravity models were calibrated for home-based work, home-based shopping, homebased other, and nonhome-based trip purposes for combined total automobile and transit person trips.

The distribution of internal school trips, groupquartered person trips, internal truck and taxi trips, and external trips was accomplished by using the Fratar model.

The ability of the second-generation gravity trip distribution models, calibrated by using 1972 travel inventory data, to predict changes in trip distribution, as measured by average trip lengths and trip length frequency distribution, was determined through the application of the models with 1991 data; that is, the gravity model was applied by using the second-generation study friction factors, 1991 observed productions and attractions, and 1991 observed travel times. It should be noted that average trip lengths measured in terms of travel time changed by only about 5 percent between 1972 and 1991. Work purpose trips exhibited a 5 percent increase and trips for all other purposes exhibited about a 5 percent decrease. In terms of trip length measured in miles, work trips exhibited about a 20 percent increase in length and trips for other purposes exhibited about a 10 percent increase in length. As shown in Table 154, the average trip lengths for all purposes were predicted with a quite adequate degree of accu-

COMPARISON OF OBSERVED AND ESTIMATED AVERAGE TRIP LENGTH FOR TOTAL PERSON TRAVEL WITHIN THE REGION: 1972 AND 1991

	1972 Observed 19 Trip Length		1991 Ob Trip Le	1991 Observed Trip Length		Percent Change 1972-1991		1991 Estimated ^a Trìp Length		Percent Difference ^b	
Trip Purpose	Minutes	Miles	Minutes	Miles	Minutes	Miles	Minutes	Miles	Minutes	Miles	
Home-Based Work	16.1	7.5	16.9	9.1	5.0	21.3	14.1	7.4	-16.6	-18.7	
Home-Based Shopping	9.6	4.0	9.1	4.3	-5.5	7.5	9.1	4.5		4.7	
Home-Based Other	11.6	4.9	10.9	5.4	-6.0	10.2	10.3	5.3	-5.6	-1.9	
Nonhome-Based	12.4	4.9	11.6	5.7	-6.5	16.3	9.4	4.5	-19.0	-21.1	

^aEstimated with 1991 models and 1991 inputs.

^bPercent difference = <u>(Estimated-Observed)</u> X 100 Observed

Source: SEWRPC.

racy, considering that all data utilized to establish both the actual and estimated trip distribution were estimates derived from travel surveys. Thus, although changes in the land use pattern and the transportation system within the Region resulted in changes in average trip lengths, the basic relationships determined by the gravity model between relative travel times and trip distribution have remained relatively stable over time.

Third-Generation Trip Distribution Procedures

The third-generation trip distribution model developed with 1991 origin and destination survey data again accomplishes the distribution of major internal person trips through use of the gravity model. Gravity models were calibrated for total internal person travel for home-based work, home-based shopping, home-based other, and nonhome-based trip purposes. Peak-hour highway travel times were used in the calibration of the work trip purpose model, and off-peak highway travel times were used in the calibration of the gravity models for the other trip purposes. Indicating the effect of spatial separation on trip interchanges observed in the 1991 travel surveys, the calibrated friction factors for each trip purpose are shown in Figure 61. Since friction factors are relative, of greater importance than their absolute magnitudes is the slope of the smoothed friction factor curve. For this reason, the friction factor curves in Figure 61 were normalized and plotted on logarithmic scales to facilitate a comparison of trip purposes. As shown in Figure 61, the friction

Figure 61

TRAVEL TIME FRICTION FACTORS FOR INTERNAL TOTAL PERSON TRIPS IN THE REGION: 1991





factor curve with the smallest negative slope is that for home-based work, indicating the smaller effects of spatial separation on the distribution of work trips as the travel time increases. Conversely, the curve for home-based shopping trips shows the greatest sensitivity to spatial separation, as the travel time increases. The distribution of internal person school purpose trips, group-quartered person trips, internal truck and taxi trips, and external trips was accomplished through the use of the Fratar model.

Modal Split

Basic Concepts: The third major step in the travel simulation process is modal split, or mode choice, whereby the total number of trips traveling between each pair of traffic analysis zones by trip purpose is divided on the basis of travel mode used. Primarily, this step involves the division of internal person trips between the two major modes of travel, public transit and the private automobile. The determination of modal split is essentially an evaluation of the potential demand for public transit service. The aggregate demand for public transit service is determined by many individual decisions; many factors operate to influence each individual choice of public transportation over the private automobile. For analytical purposes, however, the factors affecting individual modal choice may be grouped into three categories: factors relating to the characteristics of the tripmaker, factors relating to the characteristics of the trip, and factors relating to the characteristics of the transportation system. The purpose of modal split modeling is to select from within these three general categories those quantifiable factors that best explain modal choice.

The modal split step must also determine for each zone-to-zone interchange those auto trips which will drive alone and those which will carpool, or "share a ride," and thus determine the average number of persons per automobile trip. Both automobile vehicle trips and transit person trips are determined in this step as necessary inputs to the traffic assignment step of the travel simulation process.

Validation of Second-Generation Modal Split Procedures

The second-generation modal split models were developed from the 1972 origin and destination survey data for the trip purposes of home-based work, home-based shopping and other purposes combined, and nonhome-based. These models were applied in the second-generation regional plan reevaluation effort and in subsequent public transit studies conducted by the Commission, including the Milwaukee Northwest Corridor transit alternative analysis conducted in the 1980s. The method used to define the mode choice models mathematically was logit analysis, one of the three techniques that have been devised to calibrate disaggregate, behavioral, probabilistic models of mode choice. The use of logit analysis in the development of modal split or choice models is consistent with the economic theory of consumer behavior, specifically, that maximization of utility establishes choice decisions.

The models calibrated for home-based work and home-based shopping and other trip purposes express the probability of mode choice as a function of household automobile availability, invehicle travel time, out-of-vehicle travel time, and out-of-pocket cost differences between automobile and transit modes. The nonhome-based mode choice model expresses the probability of mode choice as a function of in-vehicle time, out-ofvehicle travel time, and out-of-pocket cost differences between automobile and transit modes. Representing that part of door-to-door travel time spent outside a vehicle, out-of-vehicle time includes for automobile travel, any automobile access and parking time; for public transit, travel waiting and transfer time. In-vehicle time represents that part of door-to-door travel time spent inside the public transit vehicle or the automobile. A distinction was made in the mode choice model between the in-vehicle and out-of-vehicle portions of total travel time, because available data indicate that travelers find time spent walking, waiting, or transferring, that is, time spent outside the vehicle, more inconvenient than time spent in the vehicle. The other transportation system variable considered, out-of-pocket cost, was intended to represent those costs of travel that individuals normally consider in the choice of mode. Out-of-pocket costs for transit were considered to consist of fares and the costs of driving an automobile to, and parking at, a public transit station. Out-of-pocket costs associated with automobile travel were considered to consist of parking costs and fuel costs.

In applying the modal choice models, two sets of transportation system variables are prepared: one to represent the in-vehicle and out-of-vehicle travel times and out-of-pocket costs of the automobile and transit modes during average weekday peak travel periods and one to represent the in-vehicle and out-of-vehicle travel times and out-of-pocket costs of the automobile and transit modes during average weekday off-peak periods. The transportation system variables for the peaktravel periods are used to forecast home-based work trips made by automobile and transit. Data collected under the Commission 1991 travel surveys indicated that approximately 70 percent of all home-based work trips made on an average weekday are made during the peak-travel periods. Forecasts of home-based shopping and other trips and nonhome-based trips made by automobile and transit were based upon the set transportation system variables representative of the average weekday off-peak travel periods. Data collected under the Commission 1991 travel surveys indicated that total trips made for these purposes during the off-peak periods accounted for 70 percent of the home-based shopping trips, home-based other trips, and nonhome-based trips made on an average weekday.

The second major part of the modal split step of the second-generation models is the determination of automobile occupancy. An automobile occupancy model converts automobile person trips into automobile vehicle trips by determining the number of persons, driver and passenger(s), per automobile. Average automobile occupancies were calculated for all zone-to-zone interchanges by trip purpose based on the Commission's 1972 travel survey, and these automobile occupancies were assumed to remain stable over time. The average automobile occupancies, while varying according to trip purposes, were relatively similar for all interchanges within a particular trip purpose with the exception of higher occupancies for trips to and from the Milwaukee CBD, and for trips of longest length. The average regional automobile occupancy, as observed in the 1972 travel surveys, was 1.17 for home-based work, 1.47 for home-based shopping, 1.54 for home-based other, and 1.38 for nonhome-based trips.

A major validation of the second-generation modal split model was conducted by the Commission in 1984 under the Milwaukee Northwest Corridor transit alternatives analysis. This validation is documented in the Commission publication entitled, <u>The Milwaukee Northwest</u> <u>Corridor Rapid Transit Study</u>, Report No. 2, "Travel Simulation Models," 1986. The validation entailed applying the second-generation models with inventoried 1980 demographic, economic, and land use data, and 1984 highway and transit transportation system data. The model-estimated 1984 travel demand was then assigned to the 1984 transportation system to produce simulated volumes of vehicle trips and transit passengers. The results of the transit and highway traffic assignments were then used to evaluate the performance of the travel simulation models by comparing the model-estimated volumes of transit passenger trips to observed transit passenger counts.

A comparison of the average weekday unlinked passenger trips, or boarding passengers, observed during 1984 versus those estimated for 1984 by the traffic assignment is presented in Table 155. About 239,700 boarding passengers were simulated for 1984 by the models, or about 8 percent fewer than the 261,600 average weekday boarding passengers observed during 1984 for the routes simulated in the transit network. The model-estimated transfer ratio of 0.45 transfers per linked revenue trip simulated on the transit network for 1984 was found to be slightly below the transfer ratio of 0.50 transfers per trip observed for the same routes in 1984.

A comparison of the transit ridership by time period as observed in 1984 and estimated for 1984 by the models is shown in Table 156. The number of total daily trips estimated to be made in the midday and afternoon peak periods were about equal to the trips actually made, while the number of trips made in the morning peak period was slightly overstated and the number made in the evening nonpeak period was slightly understated.

With respect to automobile occupancy, a significant decline in automobile occupancy rates has been observed within the Region from the 1972 occupancy rates assumed and proposed to remain stable on the basis of implementation of travel demand management measures, including an aggressive carpool promotion program, an areawide freeway ramp-metering system with high-occupancy vehicle bypasses, parking rate reversal and supply control in the Milwaukee CBD, and implementation of the regional land use plan. Table 157 compares the 1972 and 1991 survey-estimated automobile occupancy by trip purpose.

Third-Generation Modal Split Procedures: The third-generation modal choice models developed with the new origin and destination survey data again use logit analysis for the trip purposes of home-based work, home-based shopping and

COMPARISON OF AVERAGE WEEKDAY TRANSIT RIDERSHIP IN THE MILWAUKEE URBANIZED AREA: 1984 OBSERVED VERSUS 1984 MODEL-ESTIMATED FIGURES

	Average Weekday Transit Passengers					
	1984	1984	Difference Estimated ar	between nd Observed		
Trip Category	Observed	Estimated	Number	Percent		
Linked Revenue Passenger Trips Unlinked Passenger Trips (boarding passengers) Boards per Passenger Trip	174,100 ^a 261,600 ^b 1.50	165,600 239,700 1.45	-8,500 -21,900 -0.05	-5.1 -8.4 -3.3		

^aBased on a total annual ridership of 51,175,000 linked revenue passengers for the Milwaukee urbanized area in 1984. A factor of 294 was used to convert annual ridership to average weekday ridership.

^bBased on actual counts taken on the routes included in the simulated transit network.

Source: SEWRPC.

Table 156

COMPARISON OF 1984 OBSERVED AND 1984 MODEL-ESTIMATED AVERAGE WEEKDAY TRANSIT RIDERSHIP BY TIME PERIOD

	Average Weekday Unlinked Passenger Trips (boarding passengers)								
	1984 Observed		1984 E	stimated	Difference between Estimated and Observed				
Time Period	Number	Percent of Daily Total	Number	Percent of Daily Total	Number	Percent			
Morning Peak Period	65,000	24.9	65,400	27.2	400	0.5			
Midday Nonpeak Period	88,500	33.8	80,200	33.5	-8,300	-8.3			
Evening Peak Period	75,700	28.9	69,400	29.0	-6,300	-8.6			
Night Nonpeak Period	32,400	12.4	24,700	10.3	-7,700	-24.5			
Total	261,600	100.0	239,700	100.0	-21,900	-8.4			

Source: SEWRPC.

other purposes combined and nonhome-based. The model calibrated for the home-based work trip purpose, the home-based shopping and other trip purpose, the nonhome-based trip purpose, and the home-based school purpose, for university or college trips, are shown in Table 158. The modal choice model calibrated for the homebased work trip purpose expresses the probability of mode choice as a function of household automobile availability, in-vehicle travel time, out-of-vehicle travel time, and out-of-pocket cost differences between three modes: public transit, drive alone, and shared ride. This modal choice model incorporated these three alternatives into

Table 157

COMPARISON OF 1972 AND 1991 SURVEY-ESTIMATED AUTOMOBILE OCCUPANCY BY TRIP PURPOSE

Trip Purpose	1972 Persons per Automobile	1991 Persons per Automobile	Percentage Change 1972-1991
Home-Based Work	1.17	1.06	9.4
Home-Based Shopping	1.47	1.27	13.6
Home-Based Other	1.54	1.33	13.6
Nonhome-Based	1.38	1.20	13.0
Average	1.38	1.21	12.3

MODE-CHOICE MODELS

Home-Bas	ed Wo	<u>rk</u>	
P _T ≕ _		e	<u>, UT</u>
	е ^U T	+ e	U _{DA + e} U _{SR}
Where:	U _T U _{SR} UDA	=	0045 TC0157 TI0412 TO 0045 SC0157 SI0412 SO - 1.207 0045 DC0157 DI0412 DO + 1.359 A8276
Llomo Ros			
Home-bas	seu sna	эрри	g and Other
P _T = _	<u>e</u> 0	Т	$\overline{\mathbf{u}}^{-}$
	е ^U T	+ (, ⁰ A
Where:	UT UA	=	0133 TC0094 TI0692 TO 0133 AC0094 AI0692 AO + 3.067 A - 2.447
Nonhomo	Baad		
- NOTHORNE	Jaseu		
	11	_	$\frac{1}{2}$, $\frac{1}$
^P T = -	<u> </u>	<u>T</u>	$\frac{1}{11}$ and $\frac{1}{10}$ and $\frac{1}{10}$
	е ^О Т	+ (• A
Where:	U _T	=	0031 TC0106 TT 0031 AC0106 AT + 1.815
	• <u>A</u>		
Home-Bas	sed Sch	<u>lool</u> e	
P _T = _	e ^U	Т	
	е ^U T	+	_e U _A
W/barat			0122 TC 0080 TT
where:	UT	=	0133 AC0090 TT + .5762 A1344
Where:	UT	=	transit utility drive alone utility
		_	shared ride or carpool utility
		=	automobile person utility (driver and/or passenger)
	тĉ	=	transit cost
	TI	=	transit in-vehicle time
	10	_ =	transit out-of-vehicle time
	SU	=	shared ride in-vehicle time
	so	=	shared ride out-of-vehicle time
	DC	=	drive alone cost
	DI	=	drive alone in-vehicle time
	DO	=	drive alone out-of-vehicle time
		=	automobile person cost automobile person in-vehicle time
· · · ·	AO	=	automobile person out-of-vehicle time
	TT	=	transit total travel time (in-vehicle time plus out-of-vehicle time factored by 7.0 for nonhome-
	_		based trips and not factored for home-based school trips)
	A	=	number of autos in household (0 for 0 autos, 1 for 1 auto, 2 for 2 or more autos)
	AT	=	automobile person total travel time (in-vehicle time plus out-of-vehicle time factored by 7.0 for nonhome-based trips and not factored for home-based school trips)
L			

^a The home-based school trip model is applied to trips to and from colleges and universities in the study area and is applied to trips by households with one or more automobiles.

a single model since some of the alternatives to be considered in the third-generation regional transportation system planning effort were anticipated to include high-occupancy vehicle lanes for use by both carpools and buses. The majority of home-based work trips within the Region are made during peak-travel periods, when high-occupancy vehicle lanes provide travel time savings over the single-occupancy automobile may encourage the use of transit and car- and vanpools. The models calibrated for home-based shopping and other trip purposes express the probability of mode choice as a function of household automobile availability, in-vehicle travel time, out-of-vehicle travel time. and out-of-pocket cost differences between automobile and transit modes. The nonhome-based mode choice model expresses the probability of mode choice as a function of in-vehicle time, outof-vehicle travel time, and out-of-pocket cost differences between automobile and transit modes. An automobile occupancy model is used to convert total automobile person trips for home-based shopping and other trips based on household size and automobile availability of the zone of trip production.

In applying the modal choice models, two basic sets of transportation system variables are prepared: one to represent the in-vehicle and outof-vehicle travel times and out-of-pocket costs of the automobile and transit modes during average weekday peak travel periods and one to represent the in-vehicle and out-of-vehicle travel times and out-of-pocket costs of the automobile and transit modes during average weekday offpeak-travel periods. The transportation system variables for the peak-travel period are used to forecast home-based work trips made by automobile and transit.

The mode choice models were reviewed for both reasonableness and validity. The model coefficients are all of correct sign and significantly different from zero at a 99 percent level of confidence. In addition, the model coefficients are reasonable with respect to relative magnitude, indicating out-of-vehicle time is more onerous than in-vehicle time by a factor of 2.6 for work trips and 7.4 for other trips, and the value of time is approximately \$2.00 per hour for work trip in-vehicle time and \$5.20 per hour for out-of-vehicle time as measured in 1991 dollars. The models, calibrated with observations of individual tripmaker behavior, were also determined to estimate total regional and subregional transit trips accurately.

The second major part of the modal split step is the determination of automobile occupancy. For home-based work trips within the greater Milwaukee transit service area, the mode choice model also estimates automobile occupancy. For home-based work trips outside the Milwaukee area and for home-based shopping and other trips, an auto occupancy model was developed to convert auto person trips into auto vehicle trips based upon the household size and automobile availability characteristics of the zone of trip production as shown in Table 159. Analyses of automobile occupancy of trips for all purposes, be they home-based work, home-based shopping, home-based other or nonhome-based, indicated that travel distance and out-of-pocket cost had little or no effect on automobile occupancy. Household automobile availability and household size of the tripmaker, however, had significant effect. By definition, model predicted nonhome-based trips cannot be related to the characteristics of the tripmaker on a zone-byzone basis. Therefore, for the nonhome-based trip purpose, average automobile occupancies were calculated for all zone-by-zone interchanges based on the origin and destination survey, and these automobile occupancies are adjusted based upon forecast regional changes in automobile availability and household size.

Traffic Assignment

The fourth and final major step in the traffic forecasting and analysis process is the assignment of the zone-to-zone trip volumes forecast in the trip distribution and modal split phases to specific routes of existing and proposed alternative transportation systems. The output of traffic assignments for the arterial street and highway system is a forecast of the number of vehicles on an annually averaged weekday that may be expected to use each segment of the arterial street and highway system by direction, complete with turning movements at intersections. The output of traffic assignment for the transit system is an estimate of the number of passengers on an annual average weekday that may be expected to use each segment of the transit system by direction, complete with transfers at route intersections. The assignment

-	Home-Base	d Work Tri	ps	
		Househ	old Size	
Vehicles Available	One	Two	Three to Four	Five or More
None One Two or More	3.55 1.01 1.02	2.74 1.21 1.04	2.45 1.24 1.04	3.10 1.29 1.07

F	lome-Based S	Shopping	Trips	
		Housel	nold Size	
Vehicles Available	One	Two	Three to Four	Five or More
None One Two or More	14.20 1.03 1.02	3.28 1.46 1.17	4.89 1.43 1.26	9.28 1.59 1.32

VEHICLE OCCUPANCY BY TRIP PURPOSE: AUTOMOBILE PERSON TRIPS PER VEHICLE TRIP

Vehicles

Available

...

None

One

I WO OF MORE	1.02	1.20	1.32	1.40
<u></u>				
	Nonhome-	Based Trip	s	
		Househ	old Size	
Vehicles Available	One	Two	Three to Four	Five or More
None One Two or More	6.63 1.05 1.02	5.60 1.46 1.12	5.98 1.49 1.19	5.29 1.49 1.30

Home-Based Other Trips

Two

3.31

1.44

1 20

One

11.50

1.06

1 00

Household Size

Three

to Four

2.42

1.48

Five

or More

4.20

1.46

4 40

Source: SEWRPC.

of traffic demand to the transportation system is accomplished separately for the highway and transit systems and in several steps.

The first step in the assignment process is the preparation of highway and transit network maps to provide a definitive description of the arterial street and highway and the transit systems to be tested. The definitive description of the highway and transit systems involves, in addition to the preparation of highway and transit network maps, the collection, coding, and transfer to computer-usable form of data describing the location, capacity, and operating speeds of each link in the two networks so that the operation of the overall transportation system can be simulated. Inasmuch as the transit and highway networks are the source of the zonal travel time information used in the trip distribution and modal split steps, the initial preparation of highway and transit networks must actually be completed near the beginning of the entire travel simulation process.

The first step in the preparation of the highway network is to define in detail the highway system for the alternative plan to be tested, identifying all freeways and standard surface arterial constituting the system. The highway network includes all arterial streets in the Region, which represents about 35 percent of total street mileage and on which about 90 percent of vehicle miles of travel occur on an average weekday. The highway network is constructed using the battery of urban transportation planning programs known as TRANPLAN. This battery of microcomputer-based programs is similar to the battery of mainframe-computer-based urban transportation programs developed by the Federal Highway and Transit Administrations. An example of a computer encoded highway network map is shown in Figure 17.

The transfer of information on the highway system to computer-usable form requires the assignment of node numbers to all intersections of, and access points to, the arterial street and highway system. Each arterial street segment between two nodes is defined as an arterial link. All freeways and freeway ramps, all surface arterial streets, and some nonarterial local roads are represented in the highway network for the Region as different types of arterial links. For each arterial link, attendant data pertinent to system analysis are also encoded. The types of data which are encoded for each arterial link are listed in Table 160. Of the data encoded, link operating speed and 24-hour average weekday capacity are of particular importance. Link operating speeds are calculated by direction of travel as a function of the speed limit and the number and type of traffic controls at interme-

Type of Information	Description
Distance	Distance in miles by direction
Speed or Time	Speed in miles per hour by direction for arterial link; time in minutes by direction for centroid connector links
24-Hour Capacity	Capacity expressed in vehicles per 24-hours by direction
24-Hour Traffic Count	Average daily vehicles by direction (coded for existing or base year networks only)
Administrative Classification	Jurisdiction
Assignment Group Code	Type of link in network
Number of Lanes	Number of existing lanes
County	County in Region in which link is located
Functional Classification	Classification under U.S. Department of Transportation functional classification system
Urban Area	Urban area of Region in which link is located
Zone Number	Traffic analysis zone in which link is located

TYPE OF DATA ENCODED FOR LINKS IN HIGHWAY NETWORK

Source: SEWRPC.

diate and terminating intersections. Hourly and 24-hour average weekday capacities for each link are calculated as a function of the typical crosssection of the facility, which includes consideration of area type, functional classification, number of through traffic lanes, and number of turn lanes.

A second type of link is used in the highway network to connect the land uses served to the arterial street and highway system. For each traffic analysis zone in the Region, the center of activity is determined and marked by load nodes, or centroids, representing the points from which all trips originate, and to which they are destined, the zones enter or leave the highway network. The centroids are connected to the access points on the highway network by access links termed centroid connector links, representing nonarterial collector streets. Access times coded on the centroid connector links are a function of the time required to access the vehicle and the time required to access the arterial system over the collector streets within each zone.

As for the highway network, the first step in the preparation of the transit network is to define in detail the transit system to be tested. This involves the identification of all existing and planned routes in the public transit system to be simulated, along with the street and highway facilities and special rapid and express transit facilities over which the routes of the system are to operate. The transit networks include all existing transit routes, with the exception of special school routes. The transit network is also constructed using the aforementioned urban transportation planning programs known as TRANPLAN. An example of a computer encoded transit network is shown in Figure 25.

After the transit routes and facilities have been identified, node numbers are assigned to all terminal and transfer points for each route. Each section of street or special transit facility between two nodes is defined as a transit link. Different types of transit links are encoded in the network to represent the different types of transit service provided on the transit system, including local and feeder bus transit service,

Mode and Link Type	Designated Mode Number	Information Code
Access		
Walk Access Link	1	Distance and speed
Automobile Access Link	2	Distance and speed
Special Access/Transfer Link	3,5,6	Distance and time
Dummy Park/Ride Transfer Link	4	Distance and time
Transit		
Local Transit Link	14,15	Distance and time
Express Transit Link: Freeway Flyer	16	Distance and time
Suburban Commuter Bus	. 17	Distance and time
Express Transit Link: Arterial Street	18	Distance and time
Guideway (light rail/busway) Link	21	Distance and time

MODE DESIGNATIONS FOR TRANSIT AND ACCESS LINKS USED IN ENCODED TRANSIT NETWORKS

Source: SEWRPC.

express transit service, and rapid transit service. In addition, where different types of transit service with different operating characteristics and speeds, such as local bus service and skipstop express bus service, are provided over the same street or route, a separate set of parallel transit links is encoded to represent each type of service. The types of data which are encoded each transit link are listed in Table 161.

Each transit link is encoded with travel times by direction of travel for both peak and off-peak travel periods. The travel times are initially based upon current schedules for the existing routes operating over each transit link; for future, networks are modified to reflect future conditions on the arterial street and highway and transit systems. In this step, forecast future traffic volumes and capacity of arterial streets carrying transit routes are compared to existing traffic volumes and capacities. If the comparison indicates a significant decline in the level of service, the effects on the arterial travel speeds are estimated and the transit route travel times during peak periods are adjusted accordingly. Specifically, existing 1991 peak and off-peak travel speeds were plotted for each segment of the arterial street system and travel times between key activity centers within the corridor determined. Forecast plan design year 2010 highway traffic volumes under the no-build alternative were utilized to forecast future peak and off-peak travel speeds and travel times. These estimated future speeds were also plotted by segment of the arterial street system with average travel times between key activity centers recalculated. These speeds and travel times were then used to describe the highway element of the transportation system under the no-build alternative. The forecast 2010 travel speeds and travel times were then compared to the estimated 1991 travel speeds and travel times to identify changes in speed and travel time. These changes were reviewed to identify those transit routes which required adjustment to the remaining speeds to reflect increased traffic congestion. All changes of 5 percent or more in highway operating speeds resulted in an adjustment of transit running speeds. The adjusted transit speeds were defined by segment of the transit system. A similar review of highway and transit speeds was conducted for each transit build alternative; that is, a comparison was made of the year 2010 no-build network to each transit build alternative network.

Generally, current local bus speeds, including dwell time in the outlying portions of the urban areas of the Region, range from 15 to 18 miles per hour; in the areas surrounding the central business districts, from 12 to 15 miles per hour; and in the central business districts, from five to 10 miles per hour. Some arterial streets in the urban areas of the Region may experience future increases in traffic sufficient to result in weekday traffic volumes approaching maximum arterial capacity and, thereby, result in congestion severe enough to reduce transit running speeds. For planned local routes, such as for extensions of routes into areas presently unserved by the existing transit system, transit running speeds were based upon scheduled times observed for transit routes operating in similar areas over similar streets. Travel times for express bus services provided on arterial streets and for bus-on-busway and light-rail transit service were specifically developed as part of the operational plan for each such alternative considered.

The transit network is more complex than the highway network in that access to the transit system both by walking and by automobile must be allowed for in the simulation, as must the transfer between the different types of transit service. To allow for the different access methods and transfer capabilities, four different types of nontransit access links are used: walk access links, special service access and transfer links, auto drive access links, and auto drive dummy transfer links.

Walk access links are used to represent walk access to the transit system from the areas which each route serves. Each traffic analysis zone centroid is connected to the network so that the zone is directly connected by walk access only to those routes which directly serve the zone. Exceptions to this rule are made if over 50 percent of the development or the major traffic generator within the zone is within a reasonable walking distance, that is, one-quarter mile, of a route not directly serving the zone. Special attention is given to the presence of any natural or man-made geographic barriers, such as rivers, railways, and freeways, which would prevent direct walk access from a zone to transit service. In cases where a zone is divided by such a barrier, walk access is not provided for unless pedestrian bridges are available. Each walk link is encoded with the distance measured from the center of the activity of the zone at the zone centroid to the access node on the network at an average walking speed of three miles per hour.

To allow for a more accurate simulation of the use of the various types of transit service, walk access links are generally used only to connect zone centroids to access nodes on transit links representing local transit service, with the times resulting from the encoded speed and distances representing times associated with accessing local transit service. Access from a zone centroid to express transit service is allowed for through the use of special access links encoded between access nodes on the local transit links and access nodes on the express transit links. Such links also allow for the simulation of the transfer of passengers between the different types of transit service. Walk access for rapid transit service is provided for directly between the zone centroid and the rapid transit station or stop. However, the walk access times are generally longer than those encoded for access to local transit service. This method of coding access to express and rapid transit service is used to reflect differences between the walk access times for these transit services and local transit service.

Because walk access times are a function of transit stop spacing and because a different stop spacing is associated with each type of transit service, walk access times for rapid and express transit service should reflect the fact that it may take longer to access such transit services than to access local transit service. This is because stops on rapid and express transit services are spaced farther apart than stops on local transit service. Accordingly, the times encoded on the special access links for express service and the walk access links for rapid transit service reflect the additional walk time required to access rapid and express transit service.

In addition, because additional time is used in traveling by rapid or express transit service for access to, and egress from, the service, a corresponding decrease in travel time spent on the vehicle, or in-vehicle time, should be provided in order to induce travelers to choose rapid or express transit service over local transit service for a particular trip. Generally, the in-vehicle time savings required to offset the increased walk access time can only be realized on long trips. The encoding of additional walk access time for rapid and express transit service permits the calibration of the network so that the shortest trips are made using local transit service, the longest trips are made using rapid and express transit services. Calibration of the network in this manner is important to obtain an accurate assignment of trips in travel corridors where two different types of transit service compete for trips made between the same origins and destinations, as is the case where both local and skip-stop express bus service is provided over the same route and is simulated separately in the network.

The third and fourth types of nontransit links, auto drive access and auto drive transfer links, are used to represent access to transit service by automobile to loading points on the transit system where such access is feasible. For the existing and planned public transit system networks serving the Milwaukee area, auto drive access is limited to rapid transit service at both formally designated park-ride lots, and at some stops where patrons are known to "park-andride" or "kiss-and-ride," although a formal parkride lot may not be designated.

Rapid transit service, by Commission definition, includes all service other than local bus, including freeway flyer bus service, express or limitedstop bus service, and includes express and rapid rail transit services as well. Furthermore, automobile drive access is only considered to be available where park-and-ride or kiss-and-ride facilities are provided. The only exception would be where it is explicitly known that sufficient onstreet parking is available in the vicinity of the transit stops. In areas with extensive local bus or feeder bus service to the park-ride lot, auto drive access is generally limited to an area within three miles of the station. In outlying areas with little or no feeder bus service to the park-ride lot, auto drive access is generally limited to an area within five miles of a station. Exceptions for longer distances are made where a zone has access to a park-ride lot via a freeway or major arterial street.

In defining the service areas for auto drive access of each park-ride lot, the presence of competing lots is taken into consideration, as well as the reluctance of tripmakers to "backtrack" to access a particular transit route or type of service if another route or type of service is available which would allow a more direct travel route. As a result, the auto drive service area for each parkride lot generally reflects the logical service area of the lot and includes only those zones which could be reasonably expected to use the rapid transit service provided at the lot. As a general rule, automobile access links from a particular zone are coded to only one park-ride lot, primarily to those lots where parking is available. The coding of modes of access to rapid transit stations in the transit networks utilizes all four of the above access links. Basically, three potential access modes to rapid transit stations may be provided: walk, feeder bus, and automobile. Walk access is provided for by encoding walk access links between the stations and the zones located within one-half mile of the stations. It is anticipated that most, if not all, transit stations will have some walk access provided for in this manner. The transfer of passengers between feeder bus lines and rapid transit lines at a station is provided for by encoding transfer links between the terminal or transfer node on the feeder bus line and the station node. Times are encoded on the transfer link which reflect walking times spent, reaching the rapid transit boarding platform. Finally automobile access is provided for by encoding automobile access links between zones located more than one-half mile from the transit station. Automobile access links are encoded only to those transit stations where parking is available. The encoding of automobile access links utilizes an auto drive transfer link between all automobile access links and the transit station node. A time is encoded on the transfer link which reflects the time spent in walking from the automobile to the boarding platform at each station.

The automobile access connection includes a oneminute connection penalty between the park-andride lot and the transit station to reflect walking time. The automobile access connection also includes the travel time of automobile access between the zone and the park-ride lot, including walking between home and the automobile, driving over the local street system within the zone to access the arterial street system, and driving over the arterial street system to the park-ride lots. The typical total encoded travel time for an automobile access connection is about 50 percent, or three minutes, more than the overthe-road arterial street travel time. This automobile access travel time is usually less than the travel time which would result if local bus service were available as an alternative.

Transit access travel times reflect the actual travel times of the potential feeder bus or connecting bus service. Introduction of feeder bus transit service under a new alternative to an area previously served only by automobile drive access to park-ride lots, therefore, cannot reduce transit ridership under the new alternative, but may have the potential to increase transit ridership.

As the final step in encoding the transit network, the routes of the transit system are converted to transit lines on the network, defined by the series of node numbers along a given route and the operating headways associated with the route. Where a route operates regular service with specific branches and turnback points, separate lines are encoded on the network to represent each branch or truncated portion of the route. Similarly, separate lines are encoded where routes utilize different streets by direction of travel, as with the use of one-way pairs of arterial streets or where headways on a route or portion thereof vary by direction of travel or during peak periods, to reflect a higher level of service in the peak travel direction. All routes which regularly provide a significant level of transit service on an average weekday are encoded on the transit system.

Special routes providing a limited level of service, such as special school routes operating only one or two bus trips per day, are not included in the transit network. As was done in coding transit links, different types of transit lines are encoded to represent the different types of transit service provided on the transit system. Operating headways are encoded for each line to reflect the level of service provided during each of four times: a morning peak period, a midday off-peak period, an afternoon peak hour, and a night off-peak period. The routes and operating headways encoded in the transit network are defined by the operational plans prepared for each alternative.

The second step in the traffic assignment process involves the computation, from the descriptions of the transportation networks, of two sets of minimum time paths from each traffic analysis zone within the Region to all other such zones, one for automobile travel and one for transit travel. For the highway network, the minimum time paths are computed by systematically comparing travel time for all links in the system in successfully outward steps from the starting centroid until the shortest time path to all centroids has been computed. The shortest travel time and routes through the system between the starting centroid and all other centroids are systematically recorded and mapped. The resulting minimum time path routes are referred to as "trees," and represent the shortest door-to-door travel times between any two zones within the Region, including walk times at either end of the trip and park and unpark times for automobile trips.

A similar process is used to define the zone-tozone travel paths for transit except that the computed time paths are weighted to reflect the different coefficients assigned to work and nonwork trips in the modal split model. In this respect, the out-of-vehicle times, such as walk and drive access times, initial waiting times, and waiting times incurred in transferring between bus routes, along the transit travel path are factored by the ratio of the out-of-vehicle and invehicle coefficients in the modal split model. The minimum time path for a particular zonal pair, consequently, reflects the path with the lowest combined in-vehicle and weighted out-of-vehicle time. While this path may not reflect the shortest absolute time path for a zonal pair on the transit network, it is believed to be more representative of the path a transit patron would take since outof-vehicle time is viewed as the most onerous part of a transit trip.

From minimum time paths for the highway and transit networks, the zone-to-zone travel times for automobile and transit trips can be determined. These zone-to-zone travel times are used as inputs to the trip distribution and modal split steps in the travel simulation process. Total highway travel time is used as the measure of spatial separation of the zones input into the gravity model used for the trip distribution. The in-vehicle and out-of-vehicle portions of total travel time on the highway and transit networks are used as inputs into the modal choice models.

In the third step in the traffic assignment process, matrices, or tables, of both vehicle trip interchanges and transit passenger trip interchanges are prepared from the matrices of average weekday trip interchange volumes created by the process of trip generation, trip distribution, and modal split. For assignment of traffic demand to the highway network system, approximately 11 individual trip interchange tables which are direct outputs of the application of the modal split and trip distribution models must be combined to provide total zonal trip interchanges volumes. The individual trip inter-

	_																
	6:00 A	.M. to 9:0	00 A.M.	9:00 A.	M. to 3:0	0 Р.М. ^а	3:00 P.	M. to 6:0	0 P.M. ^a	6:00 P.M. to 6:00 A.M.							
		Morning			Midday			Evening		Night							
Trip Purpose	1972	1984	1991	1972	1984	1991	1972	1984	1991	1972	1984	1991					
Home-Based Work	0.37	0.38	0.33	0.16	0.19	0.16	0.39	0.34	0.34	0.08	0.09	0.17					
Home-Based Shopping	0.00	0.02	0.02	0.59	0.58	0.59	0.34	0.28	0.25	0.07	0.12	0.14					
Home-Based Other	0.08	0.12	0.11	0.48	0.45	0.45	0.27	0.22	0.27	0.17	0.21	0.17					
Nonhome-Based	0.08	0.07	0.04	0.45	0.45	0.50	0.41	0.32	0.33	0.06	0.16	0.13					
School	0.42	0.30	0.35	0.19	0.38	0.22	0.39	0.24	0.37	0.00	0.08	0.06					

FACTORS TO CONVERT AVERAGE WEEKDAY TRANSIT TRAVEL BY TRIP PURPOSE TO TRAVEL BY TIME PERIOD

^aFor 1991, the midday period is defined as 9:00 a.m. to 2:30 p.m. and the evening period is defined as 2:30 p.m. to 6:00 p.m.

Source: SEWRPC.

change tables which are thus combined include tables for internal vehicle trips by automobile for each of the four trip purposes derived from the modal split phase plus those made for school purposes, tables for internal automobile truck trips made by nonresidents of the Region, tables for automobile trips made by persons residing in group quarters, tables for external vehicle trips made by automobile, and tables for internal and external truck and taxi trips. For assignment of trips to the transit system, the individual trip interchange tables for transit passenger trips for each of the four trip purposes derived from the modal split phase plus those transit passenger trips made for school purposes are factored to create tables of transit passenger trips by purpose made during specific time periods of each weekday and then combined by time period. Four tables are thus created to represent transit passenger trips made during each of the aforementioned time periods using specific factors and on the basis of origin-destination surveys. The factors obtained in the Commission 1972 large-scale travel survey, the 1984 smallscale origin and destination survey, and the 1991 large-scale travel survey are shown in Table 162. The factors display considerable stability; the most current factors from the most recent survey are those used to convert average weekday transit passenger trips made by time period for current year transit ridership estimation and future year ridership forecasts.

In the final step of the traffic assignment process, the tables of zone-to-zone trip volumes for vehicles and transit passengers created in the previous step are assigned to all the individual arterial street segments and route segments comprising the minimum time paths for all zonal interchanges on the highway and transit networks. Thus, traffic volumes are accumulated on the links for all zonal interchanges, resulting in a complete assignment of traffic demand to the network.

In assigning travel to the highway network, an iterative process is used to account for the impact of assigned traffic volumes and congestion on link speeds and zone-to-zone travel times. In this respect, since vehicle trips are assigned to the shortest time paths on the highway network, some of the volumes on the individual links of the network may exceed the actual design capacity of the arterial street facilities being simulated, thus affecting the travel time used initially to determine the minimum time paths. The output of the assignment program at this stage is termed an "unrestrained" assignment. The ratios of the assigned volumes to the capacity of each link in the highway network are then calculated. The travel times are then reduced for those links having a volume-todesign capacity ratio of less than one and increased for those links having a ratio greater than one. Minimum time paths are reassigned on the basis of the revised minimum time path through the highway network. This iterative process is continued until the assigned volumes are observed to stabilize. Thus, the operating speed at which each segment of the transportation system can be traveled is modified to simulate the affect of increasing congestion in the system. The resulting capacity restraint serves to modify the unrestrained assignment volumes and provide a more accurate distribution of vehicular traffic over the highway system and more accurate travel times and travel speeds by simulating the manner in which vehicle operator will seek less congested arterial routes in tripmaking.

The travel time-volume-to-capacity ratio relationship used to establish 24-hour average weekday travel times is shown on Figure 62. The 24-hour average weekday average travel times include about one-third peak travel period travel time and two-thirds off-peak travel period travel time. These travel times are used for highway traffic assignment. Such travel times have been shown to permit accurate 24-hour assignments within the Region, including the assignment of the 1991 origin and destination survey vehicle trip data to the 1991 highway networks. Also shown in Figure 62 is the travel time, volume-to-capacity ratio relationship used to establish peak hour travel times based on average weekday total traffic volume-to-capacity ratios. The estimated peak hour travel times are used for home-based work trip distribution and modal choice.

As noted above, one output of the highway traffic assignment is an adjusted set of capacityrestrained travel times. As travel times are also input to trip distribution and mode choice steps. the resulting travel times from traffic assignment must be reviewed and compared to those input to trip distribution and modal choice. If differences are found, then the trip distribution and modal choice steps in the simulation process must be repeated with the revised travel times. The percentile difference considered significant for this purpose is 5 percent. The travel simulation model estimated 1990 travel speeds under peak-hour and off-peak-hour conditions were found to be within 5 percent of observed 1990 travel speeds.

It should be noted that the vehicle traffic loadings determined with the above procedure are expressed in terms of 24-hour average weekday vehicle traffic volumes, which are comparable to the highway network capacities derived from the transportation system inventories conducted by the Commission. These 24-hour average weekday traffic volumes can be converted to peak hourly volumes by applying the appropriate factors shown in Table 163. With respect to high occupancy vehicle (HOV) lanes, a different procedure is used to convert shared-

Figure 62

TRAVEL TIME TO VOLUME/CAPACITY RATIO RELATIONSHIPS: 24-HOUR AND PEAK-HOUR



Source: SEWRPC.

ride trips from a daily travel forecast to a peakhour travel forecast. The potential ridership on the HOV lanes has two components. One is the incremental number of ridesharers projected by the home-based work trip purpose modal choice model as using the HOV lane. These ridesharers are assumed to use the lane during the morning and afternoon peak periods, with 50 percent using the lane during the peak travel hours. A second component is nonwork trips. It is assumed that 7 percent of these trips occur during the morning peak period and 15 percent during the evening peak travel period, with 50 percent of the peak- period trips occurring during the peak hour.

The assignment of transit passenger travel to the transit network involves the assignment of the four tables of transit passenger trips by time period, using minimum time paths created for each of the time periods. Unlike the highway assignment, the capacity of the transit system is

	Arteria	Proportion of Total V al Street and Highwa	/ehicle Travel on the y System in the Peak	Hour								
	Milwauke Busines	of Region										
Arterial Type	Peak-Hour Both Directions	Peak-Hour Both Directions	Peak-Hour Both Directions	Peak-Hour Peak Direction								
Freeways Standard Surface Arterials	. 0.080 0.040 0.080 0.048 . 0.100 0.050 0.100 0.060											

FACTORS TO CONVERT AVERAGE WEEKDAY HIGHWAY TRAVEL TO PEAK-PERIOD TRAVEL

Source: SEWRPC.

not restrained as additional transit capacity can be readily provided by the provision of additional transit vehicles and the attendant reduction of headways. Accordingly, after the transit passenger trips for a particular time period have been assigned to the transit network, each transit line in the network is reviewed to determine how the initial operating headways and passenger-carrying capacity of the line relate to the assigned volume of trips. If more trips are assigned to the line than can be accommodated by the initial capacity of the line, the operating headway for the line is reduced, thereby providing additional vehicles and passenger-carrying capacity for the line. If the trips assigned indicate a significant amount of surplus capacity, operating headways for the line are increased to policy headway levels, thereby reducing the number of vehicles operated on, and the passenger-carrying capacity of, the line. The adjustment of headways based upon assigned passenger volumes is performed to balance the supply of transit service with the simulated demand, providing for realistic estimates of equipment requirements and operating characteristics for the transit system and consistent assumptions for ridership and cost estimation. Significant changes in operating headways require repetition of simulation of the modal split with the modification headways.

Estimates of the volume of transit trips using the specific types of transit services and facilities being simulated in the network can be made directly from the assigned link and line volumes. With regard to rapid transit services and facilities, estimates of the volume of trips made using the various access and egress modes available at each rapid transit station must also be derived in order to provide information needed to estimate the size and amenities needed for each transit station and the attendant capital costs. Although the transit network is encoded with links between zone centroids and public transit stations, representing the major access and egress modes: walk, automobile, and local and feeder bus access, the volume of passengers for each zone which may be expected to use each access and egress mode available at each station sometimes cannot be derived directly from the assigned link volumes. This particular problem occurs whenever both automobile and feeder bus access are encoded in the network between a zone and transit station. This is because the algorithm used to determine the minimum time path between a zone and a rapid transit station assigns all trips to the access or egress mode encoded in the network which has the shortest travel time, which usually is the automobile access mode. Consequently, the algorithm does not allow for the possibility that, for some zones and transit stations, the passenger volumes assigned between the zone and the station should be divided between the automobile and feeder bus modes.

Accordingly, a special model is used to determine the access and egress modes of transit passengers at public transit stations. This special model was developed primarily to obtain estimates of parking demand at public transit stations, utilizing information on the characteristics of transit passenger trips made using rapid transit services provided in the Milwaukee area,

PERCENT OF BOARDING AND ALIGHTING PASSENGERS BY MODE OF ACCESS/EGRESS AT PUBLIC TRANSIT STATIONS

		Type of 1	Fransit Station ^a	
Access/Egress Mode	Stations Serving High- to Medium-Density Residential Areas and with Good Local Feeder-Bus Service	Stations Serving Medium- to Low-Density Residential Areas with Poor Local Feeder-Bus Service	Stations Serving Low- to Rural-Density Residential Areas with Demand-Responsive Feeder-Bus Service	Stations Serving Low- to Rural-Density Residential Areas with Poor Local Feeder-Bus Service
Walk	20	15	10	10
Feeder Bus	20	10	20	
Auto Park and Ride Kiss and Ride	30 30	41 34	56 14	72 18
Subtotal	60	75	70	90
Total	100	100	100	100

^a High density is 7.0 to 17.9 dwelling units per net residential acre; medium density is 2.3 to 6.9 dwelling units per net residential acre; and, low density is 0.7 to 2.2 dwelling units per net residential acre.

Poor local bus service to a transit station is considered to be bus service which has headways of 30 minutes or more, and only one to two local bus routes serve the station. Good bus service is considered to be bus service with headways of about 15 minutes, and multiple routes serve the station, including local and express routes.

Source: SEWRPC.

which consist solely of "freeway flyer" bus service operated from outlying park-ride lots to the Milwaukee CBD. On bus survey data were analyzed regarding the mode used by freeway flyer passengers to access freeway flyer bus service at park-ride lots, along with current operational information for each freeway flyer route to determine the number of boarding passengers per parked car at each park-ride lot.

Information was also reviewed concerning the density of residential development within the service areas of each of the existing park-ride lots and concerning how well the park-ride lot was provided with local bus service. From this information, estimates of the percent of total boarding passengers that may be expected to use different access modes were developed as a function of the density of residential development in the service area of the park-ride lot and the level of local bus service provided to each park-ride lot. To estimate the number of trips made by each mode of access and egress for each park-ride lot, the percentages for each mode of access are applied to the total simulated volume of passengers boarding rapid transit service at each park-ride lot. The percentages used to estimate access and egress modes are shown in Table 164. The model assumes that the percentage of passengers using the automobile for access to, or egress from, a park-ride lot will be highest in areas of low residential density with poor local feeder bus service, lowest in areas of high residential density with good local feeder bus service.

The mode of access model is applied to all boarding passenger volumes at a transit station, with no distinction between households which do, or do not, own automobiles. Existing and planned future park-and-ride lots are generally located in outlying parts of the Milwaukee urbanized area where there are few households with no automobile available for travel. The mode of access model does include an equilibration step which assures that the automobile parking demand at any given station, as determined by the mode of access model, does not exceed the proposed parking supply at the station.

The access trips determined to use feeder bus access are not directly included in the transit trip table and assigned to transit feeder bus routes; they are not accounted for in system service level equilibration and fare box revenue estimation. On the basis of the transit fare schedule assumed for planning purposes, however, no fare would be received for feeder bus route use. Transit access to stations is encoded only for those zones within walking distance of a feeder bus route. Walk access to stations is coded only for zones within walking distance of a station and these zones are generally located immediately adjacent to the station. The area of all zones within the study area encoded with access to park-ride lots generally approximates a U. S. Public Land survey one-quarter section, or 160 acres.

All transit alternatives have nearly identical coding for automobile access, for transit access, and for walk access. In addition, the zone sizes are all small enough so that access opportunities are homogeneous across each zone. Therefore, the proposed mode of access procedure permits a valid comparison to be made between transit alternatives with respect to total ridership and mode of access. It also permits an estimation of feeder bus ridership which can then be utilized in feeder bus system equilibration and costing analyses.

The access-egress model may be used in the detailed corridor as well as areawide planning efforts to obtain estimates of passengers using the various access modes available at transit stations where parking is supplied. For stations without parking, a different method of estimating the access-egress modes at such stations would presumably be limited to the walk, feeder bus, and kiss-and-ride modes. The proportion of total boarding stations at each station using the kiss-and-ride mode is estimated on the basis of the proportions found for stations with parking. The proportion of the remaining boarding passengers who would use the walk and feeder bus modes is estimated on the basis of the proportion of passengers using these accessegress modes as derived directly from the assigned link volumes in the transit network. This methodology assumes that stations without parking will be located within the service area of a station with parking, or within densely developed areas served by feeder bus routes. Consequently, the primary access-egress modes at such stations would be the walk and feeder bus modes, rather than the automobile mode.

Traffic Assignment Calibration

The traffic assignment process was calibrated by assigning trip interchanges from the 1991 Commission travel survey to the existing highway network and comparing link volumes with corresponding actual volumes determined by ground counts. If the comparisons so indicated, appropriate modifications were coded into the network describing the highway system so that the simulated traffic volumes would correspond with the observed volumes. Such modifications included, as necessary, adjustments in link capacities or operating speeds, addition or deletion of loading links and modification in the location of load nodes. Such adjustments were required for less than 1 percent of the arterial street system and included increasing arterial segment speed or capacity by approximately 5 to 10 percent to account for increased capacity or increased speeds on arterial street segments to reflect traffic signal progression.

Calibration of the transit assignment process was accomplished through a detailed analysis of the minimum time paths and associated zone-tozone travel times on the simulated transit network utilizing data such as current route schedules. Where necessary, modifications were made in the transit network so that simulated travel paths would reflect the most logical path followed by the trip concerned. These modifications included changes similar to those made in the highway network regarding adjustment of link speeds and location of access links, and also included adjustments made to the wait and transfer times on the network. Minimal adjustments were made to transit route segments and access links in the transit network to bring the 1990 simulated transit passenger volumes into conformance with actual 1990 transit system ridership. The minimal changes necessary included limited changes in access links to provide more direct access from selected zones to selected routes, and adjustments in route speeds for selected routes by about 5 percent.

TRAVEL SIMULATION MODEL VALIDATION

The third-generation travel simulation models developed by the Commission using the new 1991 travel survey data to forecast design year travel demand were described in the preceding sections of this chapter. This section of the chapter presents the findings of the validation effort for the third-generation travel simulation models.

The model validation entailed applying the full battery of simulation models with inventoried

1990 demographic, economic, and land use data and 1991 transportation system data to estimate year 1990 travel demand and traffic flows. First, automobile availability and trip production and attraction models were applied to estimate total travel demand in 1990. Then, the trip distribution model was applied to estimate zone-to-zone travel demand and the mode choice model was applied to estimate zone-to-zone travel demand by individual mode. The estimated year 1990 travel demand was then assigned to the 1991 transportation system to produce simulated volumes of vehicle trips and transit passengers. The result of the transit and highway traffic assignments were then used to evaluate the performance of the travel simulation models by comparing the simulated volumes of highway vehicle and transit passenger trips to observed vehicle and transit passenger counts. In this respect, it should be recognized that the observed counts of vehicle and transit trips to which the model estimates are compared do in fact themselves represent "estimates" which contain their own errors. Many of the counts were taken on only one or two days of the entire year and, therefore, reflect the effects of the daily and monthly variations in travel, requiring adjustment to reflect average weekday conditions.

Total Travel Demand—Model Validation

The year 1990 was selected as the base year for model validation. Major inventories of population, employment, and land use within the Region were undertaken by the Commission in 1990 as part of its continuing efforts to maintain an accurate and up-to-date planning data base for the Region. Also, the latest Federal census of population and housing was conducted in 1990. Computer encoded networks representing the 1991 arterial street and highway system within the Region and the 1991 public transit systems in the Milwaukee, Racine, and Kenosha urbanized areas were also available. In addition, a comprehensive survey of travel was conducted in the Region by the Commission in 1991.

Starting with the trip generation and trip distribution steps in the travel simulation process, the basic socio-economic and land use inventory data were input into the automobile availability and trip generation models to obtain estimates of the total number of trips made by trip purpose. The results of the application of these steps are presented in Table 165. The results of the application of the trip distribution

Table 165

COMPARISON OF OBSERVED AND MODEL-ESTIMATED AVERAGE WEEKDAY INTERNAL PERSON TRIP GENERATION WITHIN THE SOUTHEASTERN WISCONSIN REGION: 1990

Trip Purpose	1990 Observed ^a	1990 Estimated ^b	Percent Difference
Home-Based Work Home-Based Shopping Home-Based Other Nonhome-Based	1,302,700 798,100 1,687,400 1,127,500	1,307,200 774,800 1,601,000 1,127,700	0.3 -2.9 -5.1
Total	4,915,700	4,810,700	-2.1

^aBased upon 1991 origin and destination travel survey.

^b The necessary adjustment of estimated trip production to equal estimated trip attraction was less than five percent for each trip purpose.

Source: SEWRPC.

step are presented in Table 166 and on Figure 63, which compares model-estimated to survey-estimated trip length frequency distributions and travel patterns for the 37 planning analysis areas of the Region.

Transit Travel Demand

The estimated 1990 travel demand was divided into automobile and transit person trips by applying the modal split and auto occupancy travel simulation models. The procedure followed to validate the simulation of transit passenger travel within the Milwaukee urbanized area began with the conversion of the transit trips by trip purpose estimated by the modal choice models into trip tables. These trip tables, including school purpose transit trips, were then assigned by time period to the 1991 transit network for the Milwaukee urbanized area. The transit passenger volumes derived from the traffic assignment were then compared with 1990 passenger count information obtained from the local transit operators. The comparisons used to evaluate the transit assignment and the overall travel simulation process included systemwide comparisons of observed versus estimated passenger trips in total and by time period. Comparisons were also made between observed and estimated transit ridership on selected major transit routes.

A comparison of the average weekday linked passenger trips, as determined from the Commission travel survey, with those estimated by the application of the simulation models for the





Figure 63 (continued)

HOME-BASED SHOPPING TRIPS



Source: SEWRPC.

Milwaukee urbanized area, is presented in Table 167. A comparison of average weekday unlinked passenger trip, that is, of boarding passengers observed during 1990 by the transit operator with those estimated by the application of the simulation models, is presented in Table 168. A comparison of the transit ridership by time period as observed by operator counts in 1990 with that estimated by application of the simulation models is presented in Table 169. The Figure 63 (continued)

HOME-BASED OTHER TRIPS







factors applied to convert the estimated 1990 transit trips by trip purpose to trips by time period are set forth in Table 163.

Table 170 presents a comparison of the 1990 average weekday transit ridership for selected routes as observed through passenger counts with that estimated by application of the simulation models. Map 81 shows the routes considered. The passenger count information is based

TOTAL PERSON TRIP TABLE: 1991 SURVEY ESTIMATE AND MODEL ESTIMATE

Production		<u> </u>							Attractio	n Zone: 199	1 Survey								
Zone	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19
1	107,138	6,804	2,668	1,684	2,483	809	230	132	448	2,301	1,032	0	492	102	64	0	0	0	0
2	8,570	31,130	3,195	366	1,575	840	40	309	223	2,703	520	0	690	32	0	20	42	55	0
3	1,283	4,270	24,861	192	593	1,246	206	719	632	14,805	2,103	0	1,444	671	57	0	0	24	104
4	1,764	185	105	37,775	7,952	2,201	0	57	0	360	285	0	119	0	0	0	0	0	0
5	761	1,090	438	5,368	51,130	7,044	0	54	50	857	472	0	256	308	105	29	0	0	0
6	261	77	547	1,010	4,196	29,125	42	74	0	828	943	0	328	273	0	0	36	58	0
7	36	0	217	0	0	0	42,745	5,848	373	358	3,411	2,191	3,679	448	64	104	122	118	0
8	43	414	313	0	70	0	6,782	28,689	746	981	5,576	2,738	9,718	740	479	195	103	48	56
9	30	134	1,196	59	0	144	229	896	9,421	6,376	3,096	59	4,013	353	24	162	0	0	0
10	310	523	6,585	112	395	700	0	181	2,326	51,895	10,592	127	2,436	581	141	64	0	0	80
11	102	69	913	84	194	672	675	1,027	594	6,924	97,693	303	17,383	6,269	612	93	49	92	186
12	19	0	38	0	0	٥	5,147	5,495	311	639	4,029	16,398	14,648	1,051	5,137	675	0	0	38
13	207	192	781	63	109	498	2,300	4,193	2,117	2,496	28,090	3,069	160,574	6,217	2,588	482	337	119	148
14	52	0	106	42	57	230	60	408	134	956	14,495	242	6,556	36,987	2,474	0	88	307	46
15	72	0	34	0	48	57	103	289	189	265	2,698	1,412	5,354	3,046	27,925	1,144	40	856	55
16	0	0	0	0	0	66	108	31	113	31	1,239	776	2,017	872	4,383	52,588	8,719	2,222	122
17	0	21	0	0	0	0	0	43	0	0	60	0	276	0	497	11,781	84,047	4,123	204
18	0	39	24	39	29	0	0	164	0	0	810	156	543	834	2,836	1,603	2,579	68,193	3,482
19	0	20	0	0	24	0	0	0	0	174	72	76	221	140	0	0	0	1,457	16,976
20	160	0	98	212	0	240	0	212	60	230	550	0	313	458	85	364	377	4,843	26,850
21	0	0	0	0	0	0	0	0	0	0	64	0	233	80	0	211	698	5,058	384
22	0	37	0	0	0	37	0	0	0	24	0	0	110	0	0	0	0	331	990
23	0	0	0	0	0	78	0	0	24	72	81	0	0	0	0	/2	511	583	1,225
24	/85	332	1,/3/	281	1,300	5,024	303	221	329	12,562	5,526		1,239	479	96	1/9	0	133	0
25	200		201	1/3	1,120	5,467		244	274	1,020	1,599	226	2 012	1 706	671			150	96
20	567	200	2 2 2 0	195	1 169	2 042	419	244 542	900	3,030	21,539	230	3,012	3 350	671	29	0	204	100
27	299	405	2,370	331	570	2,045	266	J42 77	173	1 1 1 4	20,410	36	4,320	469	127	0	108	35	102
20	59	50 69	59	430	582	1 596	236	260	23	2,193	5 010		1 316	871	273	ő	186	36	217
30	157	34	231	241	518	911	124	105	331	846	3,544	24	1,338	196	76	0	0	0	64
31	155	115	397	94	187	353	20	272	220	1.067	8 664	324	3,932	7 487	1 556	222		317	0
32	66		55	45	261	841	84	89	0	1,455	2.631	0	1.044	919	1,000	39	0	67	295
33	164	0	251	0	278	413	338	310	110	970	3.548	149	2.633	2.514	2.030	408	0	1.121	124
34	75	508	347	392	145	808	169	89	59	1.042	4.025	182	2.004	3,555	910	308	845	758	445
35	72	144	112	34	120	0	69	141	189	332	1,269	70	282	453	166	0	0	418	480
36	0	0	68	34	0	122	0	34	43	235	849	138	421	689	1,044	146	Ó	435	445
37	27	0	o	0	0	101	o	18	65	162	497	0	350	100	214	о	26	228	804
Total	123,692	47,369	49,299	49,401	75,420	63,392	61,001	51,223	20,634	131,293	259,586	28,706	254,609	82,429	55,294	70,918	99,020	92,569	54,098

Table 166 (continued)

Braduation								Attr	action Zone	: 1991 Surv	ey								
Zone	20	21	22	23	24	25	26	27	28	29	30	31	32	33	34	35	36	37	Total
1	64	0	0	0	2,072	369	1,334	1,473	644	947	1,023	310	254	207	427	55	164	128	135,858
2	o	46	0	0	1,379	363	827	918	269	371	541	238	247	86	403	0	0	0	55,998
3	27	0	0	84	4,825	1,174	1,733	3,842	769	1,484	806	1,176	562	220	840	62	107	166	71,087
4	32	0	0	0	1,637	891	216	912	1,085	1,092	1,166	430	102	43	81	40	0	0	58,530
5	o	0	0	0	3,119	2,373	959	1,831	1,815	1,570	2,199	534	571	220	626	128	46	46	83,999
6	128	112	46	0	5,410	6,655	1,650	2,080	3,403	2,199	4,024	588	949	306	894	64	42	34	66,382
7	38	0	0	0	382	241	1,474	1,025	23	1,402	405	627	122	314	357	92	0	54	66,270
8	0	0	44	0	846	135	1,880	976	267	1,151	775	1,049	281	202	488	36	23	0	65,844
9	0	0	0	0	1,689	168	1,440	729	256	865	200	434	219	155	225	0	23	0	32,595
10	60	0	0	44	7,865	1,262	5,170	4,727	1,262	2,257	1,142	1,913	812	1,145	734	271	207	0	105,919
11	476	35	70	0	3,739	782	18,587	8,307	1,381	4,798	3,650	8,208	1,102	2,391	2,681	548	367	773	191,829
12	58	56	38	0	516	0	1,643	249	212	773	627	893	546	177	648	57	191	87	60,396
13	497	282	100	63	777	458	5,023	3,286	742	3,213	2,123	6,016	757	1,616	1,921	499	456	249	242,658
14	300	177	56	29	738	459	4,667	2,000	396	2,718	2,339	14,883	1,158	8,133	3,725	684	650	803	107,155
15	304	64	32	46	405	132	1,694	492	334	1,573	518	3,300	993	5,993	2,619	571	1,149	739	64,545
16	392	297	58	155	0	62	593	245	0	93	281	485	141	821	477	97	64	166	77,714
17	285	2,570	138	554	0	0	0	80	0	60	45	513	138	252	188	0	0	0	105,875
18	7,031	3,173	1,158	728	77	115	1,247	533	46	1,388	516	1,884	737	3,275	2,000	/2/	934	1,062	107,962
19	34,621	409	914	1,643	54	88	54	168	84	453	226	220	505	537	1,167	905	402	2,248	64,019
20	276,082	1,488	4,591	9,027	280	372	1,201	405	3//	2,543	1,341	/19	567	1,409	3,779	1,064	/64	3,701	52 407
21	1,792	32,597	3,343	8,052 29,705	0	02	177	104	40	208	102	19	0	76	143	60	0	107	16 210
22	15 059	5 4 9 9	20.682	195 796	162	0	331	261	135	871	96	230	416	525	496	80	130	152	233 556
23	15,058	0,400	20,002	100,700	60 281	10 767	11 276	25 883	7 373	7 5 3 1	5 2 2 5	3 364	2 267	1 193	2 011	474	68	372	166 771
25	209	ő	0	ů O	10.881	24.553	1.822	4.520	11.658	3.683	5.666	747	1.001	293	1.191	183	24	0	78.314
26	228	152	218	0	5.535	1,365	67,136	24,291	2.396	10.324	6.025	11.028	2.083	3.007	4.921	463	487	249	174,546
27	668	0	422	184	42,500	9,471	58,200	166,111	17,577	51,862	15,866	20,865	9,344	7,135	12,534	2,529	855	1,386	466,245
28	253	o	0	130	6,117	10,512	3,490	8,706	53,317	12,719	16,939	1,633	1,746	734	1,214	210	244	180	128,231
29	375	ō	35	40	6,547	2,458	11,535	40,973	10,524	88,080	26,240	7,641	12,176	3,731	7,557	1,086	379	1,006	233,799
30	383	o	0	0	2,172	3,050	4,780	5,613	13,370	19,270	40,080	3,732	4,359	2,838	5,028	1,146	382	493	115,436
31	475	0	0	225	2,680	1,198	13,645	11,187	1,269	8,495	4,152	101,316	5,775	18,987	15,806	. 954	1,823	798	214,216
32	261	o	0	39	856	551	3,273	5,252	1,581	13,887	5,772	11,431	36,843	9,305	25,733	2,780	668	1,318	127,441
33	977	0	80	427	1,262	104	7,050	3,770	1,000	6,370	5,085	20,031	5,343	84,574	32,087	5,731	7,795	4,960	202,007
34	1,721	99	220	170	3,131	1,205	8,177	8,438	2,791	17,935	13,743	26,373	23,100	42,481	112,503	16,062	4,386	9,211	308,412
35	859	92	100	124	682	402	2,573	2,027	683	4,916	3,717	2,843	4,428	6,578	19,408	75,415	2,468	9,646	141,312
36	750	59	91	34	363	38	1,621	1,332	267	2,824	1,671	3,748	1,412	15,508	7,227	1,780	13,503	4,369	61,300
37	1,133	96	211	42	186	240	845	642	305	2,058	1,495	960	1,649	4,052	7,625	8,681	2,474	18,684	53,970
Total	351,811	49,217	39,418	236,431	179,165	82,095	247,323	343,548	137,659	282,149	176,094	260,411	122,866	228,648	279,854	123,534	41,275	63,187	4,914,638

Table 166 (continued)

													_									_						_						_					_
	19	25	18	38	24	30	26	24	26	15	48	85	55	124	134	206	131	160	2,989	12,751	24,509	1,116	1,662	3,135	55	34	105	287	109	296	161	215	234	565	934	1,304	571	1,550	53.751
	18	35 -	26	51	21	32	38	98	130	42	64	238	711	943	563	2,221	2,955	5,174	57,942	1,701	3,295	4,642	831	1,699	109	33	185	374	79	264	134	358	238	994	881	531	871	517	89,020
	17	11	11	17	6	4	9	74	74	16	25	80	562	327	150	546	9,220	79,518	2,224	46	138	1,132	71	210	31	10	56	87	15	55	27	95	40	123	139	53	46	30	95,278
	16	17	26	36	9	17	6	226	176	26	43	128	1,963	607	238	1,578	48,431	10,891	1,795	38	107	318	30	78	47	11	72	150	23	88	40	140	59	197	193	73	73	33	67,983
	15	77	62	105	32	47	50	339	471	94	158	767	4,791	3,463	3,347	20,479	4,458	678	4,083	188	411	190	52	128	215	56	491	734	109	419	180	1,156	343	2,863	1,537	434	1,526	315	54,848
	14	261	203	466	80	185	187	644	985	548	876	8,275	1,275	9,559	21,963	3,631	685	295	1,473	168	407	105	50	129	984	190	2,888	3,619	338	1,580	609	8,589	1,014	4,553	4,010	795	1,108	381	83,108
	13	1,151	1,058	2,592	193	357	370	5,488	11,150	4,893	3,239	21,757	14,159	132,881	10,833	6,614	2,202	877	2,937	244	576	246	74	209	2,035	382	3,896	5,838	662	2,414	957	5,290	1,387	3,033	3,153	934	896	441	255,418
odel	12	45	64	06	4	16	16	1,933	2,315	91	82	399	14,316	3,471	261	2,209	1,314	249	304	17	45	28	e	16	58	18	105	181	18	95	37	168	57	123	155	49	38	16	28,406
stribution M	11	1,408	1,072	3,269	354	766	832	2,782	5,332	3,921	10,976	78,948	3,627	34,899	15,936	2,786	889	417	1,524	318	802	152	97	253	7,388	1,017	19,417	22,055	1,467	5,694	1,935	13,838	2,893	5,475	6,031	1,454	1,354	685	262,063
1991 Trip Di	10	3,009	2,671	13,517	645	1,856	2,441	839	1,820	4,676	39,363	8,752	570	3,587	1,100	424	171	82	312	102	244	44	31	86	14,304	2,322	4,743	13,629	1,538	2,002	639	2,254	716	1,258	1,448	414	336	196	132,141
tion Zone:	თ	349	388	1,584	50	125	120	530	1,625	5,703	1,818	1,726	287	2,672	287	121	55	34	75	19	30	с Л	e	12	646	104	332	851	106	174	62	270	78	152	165	54	46	18	20,676
Attrac	8	322	610	1,367	48	101	69	6,784	22,099	1,423	637	1,640	5,039	5,539	462	519	332	106	151	21	63	16	9	21	343	75	395	743	92	271	106	423	160	251	306	85	71	30	50,750
	7	171	317	507	26	45	41	39,059	7,800	428	250	746	4,333	2,395	228	364	406	96	105	20	41	19	9	10	161	33	224	415	49	192	82	242	96	150	201	68	42	40	59,408
	9	1,190	925	2,581	2,466	7,064	20,879	103	179	301	1,541	493	69	332	179	86	41	19	89	58	130	21	20	49	5,775	6,394	598	4,116	2,775	1,708	734	412	479	315	674	238	97	66	63,229
	5	3,212	2,185	1,409	8,134	44,558	4,988	99	121	175	721	251	37	187	80	46	15	10	53	25	82	12	6	32	1,530	1,181	283	1,321	819	690	330	203	217	151	324	127	47	50	73,681
	4	2,942	480	317	32,570	6,479	1,220	28	40	50	175	78	16	68	36	19	6	e	25	18	41	9	-	11	400	458	96	557	409	377	187	89	122	80	177	78	29	27	47,717
	з	2,435	3,535	18,672	397	1,347	1,747	391	950	2,288	6,281	1,098	176	920	197	86	51	22	67	23	50	7	7	19	2,620	708	676	2,141	329	437	148	409	146	232	280	96	60	37	49,097
	2	8,331	26,883	3,608	430	1,319	318	392	645	402	664	245	157	360	67	56	33	16	26	11	27	ى ك	2	:	440	121	175	528	93	145	60	106	52	77	66	25	26	22	45,977
	-	99,993	8,102	1,897	3,167	2,150	427	155	243	315	806	318	77	321	84	45	18	10	45	14	34	2 2	5	12	554	181	235	753	144	226	100	182	77	117	150	56	41	22	121,081
Production	Zone	-	2	e	4	5	9	7	8	6	10	11	12	13	14	15	16	17	18	19	20	21	22	23	24	25	26	27	28	29	30	31	32	33	34	35	36	37	Total
L		1													_				_																				Í.
Table 166 (continued)

Braduction]							Attraction Z	one: 1991	Trip Distribu	tion Model								
Zone	20	21	22	23	24	25	26	27	28	29	30	31	32	33	34	35	36	37	Total
1	57	9	4	14	2,155	756	1,463	1,752	627	1,104	677	815	346	409	419	131	54	82	135,858
2	37	3	4	9	1,657	533	1,030	1,234	366	666	360	577	199	265	253	83	43	43	55,998
3	70	7	7	21	5,581	1,782	2,563	3,090	821	1,278	715	1,231	412	544	530	140	85	87	71,087
4	65	8	6	15	1,263	1,201	480	1,134	1,239	1,616	1,075	378	469	201	444	145	52	83	58,530
5	94	11	9	30	3,663	2,405	1,040	1,977	1,914	2,296	1,518	624	671	310	582	188	59	110	83,999
6	103	11	7	34	8,437	8,247	1,226	3,375	3,713	2,754	1,722	750	729	373	701	213	68	111	66,382
7	72	14	2	24	491	118	1,115	810	156	765	449	920	310	404	419	114	53	79	66,270
8	83	12	12	28	890	208	1,816	1,313	260	1,053	577	1,423	427	611	601	156	92	101	65,844
9	40	7	8	10	1,428	302	1,304	1,171	260	588	316	769	216	309	295	66	49	50	32,595
10	123	13	13	36	11,346	2,051	6,963	7,503	1,365	2,002	1,035	2,568	630	1,050	938	220	146	150	105,919
11	299	34	29	99	4,927	741	21,208	9,967	1,026	4,182	2,034	11,556	1,712	3,837	2,789	570	415	380	191,829
12	151	79	24	56	383	93	1,286	750	153	943	539	1,347	397	813	714	183	134	131	60,396
13	394	95	55	120	2,088	456	7,507	4,330	720	4,027	2,211	7,958	1,683	3,768	2,903	704	512	472	242,658
14	320	65	35	93	1,120	240	6,341	3,061	467	2,963	1,584	16,341	1,438	9,794	4,789	803	960	596	107,155
15	515	150	52	180	450	120	1,783	1,060	233	1,437	836	3,059	760	6,683	2,481	577	1,477	644	64,545
16	345	440	73	208	176	46	584	389	98	537	336	770	272	848	697	181	159	139	77,714
17	471	2,967	230	718	87	36	301	214	52	286	183	394	155	422	383	113	95	111	105,875
18	6,271	3,780	1,027	2,481	362	121	1,216	814	287	1,588	1,046	1,742	864	3,501	2,542	975	1,681	1,445	107,962
19	31,295	546	1,262	2,689	133	74	389	395	186	1,117	742	616	591	1,033	2,003	1,831	731	2,604	64,019
20	267,393	1,192	5,453	14,412	313	187	967	902	461	2,501	1,665	1,358	1,303	2,015	3,795	4,139	1,218	4,456	344,762
21	2,414	31,530	2,290	6,069	50	35	146	155	74	372	255	218	205	365	518	254	161	287	53,467
22	6,953	1,649	7,586	24,672	44	21	123	110	49	306	200	155	175	222	401	201	125	266	46,218
23	16,967	3,672	19,699	181,252	119	46	296	308	134	735	501	426	394	583	957	508	276	563	233,556
24	200	17	20	66	52,573	13,993	11,353	25,242	7,753	5,368	2,854	4,083	1,418	1,628	1,638	401	213	256	166,771
25	113	10	14	34	15,323	21,976	1,528	6,395	9,398	3,942	2,540	8/4	993	443	929	256	/3	145	/8,314
26	301	36	33	99	7,010	994	52,700	30,603	2,091	10,525	4,043	17,098	3,188	4,985	4,161	719	513	4/6	1/4,546
27	/53	64	90	245	30,942	9,818	00,244	152,354	25,535	15 200	14 957	20,402	2 941	1,004	2 666	2,007	207	1,221	400,245
28	252	2/	27	84	0,995	0,171	2,043	29.264	45,085	70 765	14,007	1,302	14 546	5 217	12 281	2 6 1 5	207	1 5 2 0	120,231
25	438	10	30	148	1 475	1 302	3.017	6 857	9 4 7 9	25 527	40.830	3 699	6.325	2 1 9 9	5.079	1 3 2 3	377	753	115 436
21	430	40 54		190	2 850	604	20,629	12 643	1 453	10.954	4 987	75 157	7 383	20 161	17 871	1 583	1 468	1 174	214 216
22	619	54 60	63	183	1 108	741	4 643	6 152	2 531	20.505	10 100	11 730	25 989	6 260	23 087	3 085	773	1 404	127 441
33	1 277	126	136	356	1,524	438	7,483	4.521	1.203	7.863	4.616	20.063	5.619	73,103	34,282	5.078	8.477	5,133	202 007
34	2 240	192	215	580	1.990	1.027	8.653	8,156	3,215	21,173	12.323	26,929	22,361	43,502	101.028	19,134	5.001	9.836	308.412
35	3.595	132	158	476	578	345	1.881	1.989	1,091	6,367	4.439	3.223	5,272	6,856	22,099	62.896	2,309	10,738	141.312
36	1,177	109	.00	292	412	143	1,754	1,111	372	2,343	1,446	3,563	1,418	15,449	7,552	2,660	8.913	5,215	61,300
37	3.579	116	152	402	237	138	867	848	407	2,420	1,532	1,477	1,462	3,980	8,200	8,795	2,779	12,070	53,970
Total	350,482	47,350	39,078	236,681	180,734	82,425	250,684	347,584	138,279	283,966	177,038	263,040	123,521	230,331	281,342	123,755	41,393	63,323	4,914,638

ယ Source: SEWRPC. ယ

COMPARISON OF AVERAGE WEEKDAY TRANSIT RIDERSHIP IN THE MILWAUKEE URBANIZED AREA: 1990 OBSERVED VERSUS 1990 MODEL-ESTIMATED FIGURES

Linked Transit Revenue Trips Observed Estimated Percent Trip Category 1990 1990 Difference Milwaukee Urbanized Area Home-Based Work 40,900 45,300 10.8 14,900 Home-Based Shopping . . . 15,100 1.3 Home-Based Other 26,900 29,100 8.2 Nonhome-Based 10 800 13.700 26.9 School^a 61,700 61,800 0.2 Total 155,200 165,000 6.3

^aExcludes trips made on yellow school bus.

Source: SEWRPC.

Table 168

COMPARISON OF AVERAGE WEEKDAY UNLINKED TRANSIT PASSENGER TRIPS (BOARDING PASSENGERS) CARRIED BY TRANSIT OPERATORS SERVING THE FOUR-COUNTY MILWAUKEE METROPOLITAN STATISTICAL AREA: ACTUAL 1990 OPERATOR COUNTS AND 1990 MODEL ESTIMATES

Transit Provider	1990 Actual ^a	1990 Model Estimated
Milwaukee County Transit System	228,900	236,500
Coach Lines, Inc.)	600	2,800
City of Waukesha Transit System Utility—Metro Transit	2,800	4,400
Total	232,300	243,700

^aBased on actual operator counts taken during the months of September through June during 1989 through 1991 by the Milwaukee County Transit System, the City of Waukesha Transit System Utility, and Wisconsin Coach Lines, Inc., for Waukesha County. Counts were adjusted to account for monthly and annual variations in ridership so as to be more representative of average weekday 1990 volumes.

Source: SEWRPC.

Table 169

COMPARISON OF 1990 OBSERVED AND 1990 MODEL-ESTIMATED AVERAGE WEEKDAY TRANSIT RIDERSHIP BY TIME PERIOD FOR TRANSIT OPERATORS SERVING THE FOUR-COUNTY MILWAUKEE METROPOLITAN STATISTICAL AREA

	Average Weekday Unlinked Passenger Trips (boarding passengers)					
	1990 Observed		1990	Estimated		
Time Period	Number	Percent of Daily Total	Number	Percent of Daily Total		
Morning Peak Period Midday Nonpeak Period Afternoon Peak Period Night Nonpeak Period	52,700 69,800 76,100 33,700	22.7 30.0 32.8 14.5	62,700 71,200 82,600 27,200	25.7 29.2 33.9 11.2		
Total	232,300	100.0	243,700	100.0		

Source: SEWRPC.

upon actual weekday passenger counts taken by the transit operators between 1989 and 1991 as adjusted to account for monthly and annual variations in ridership levels so as to be representative of average weekday 1990 ridership.

Highway Travel Demand

The estimated 1990 arterial street average weekday traffic volumes derived from applica-

tion of the traffic simulation models were compared to estimated 1990 average weekday traffic volumes derived from actual traffic counts. Table 171 presents a comparison of estimated vehicle-miles of travel from the traffic models and traffic counts for each county within Southeastern Wisconsin. Map 82 presents a comparison of the estimated average weekday traffic for the freeway system and selected major

Table 170

COMPARISON OF ACTUAL TRANSIT RIDERSHIP COUNTS TO MODEL-ESTIMATED TRANSIT RIDERSHIP ON SELECTED BUS ROUTES WITHIN THE REGION: 1990

	Average Weekday Unlinked Trips (boarding passengers)				
	1000 A -tu-lâ	1990 Model	Difference	Percent	
Bus Route	1990 Actual*	Estimated	(estimated to actual)	Difference	
Milwaukee County Transit System					
Route No. 10	6,400	8,310	1,910	30	
Route No. 12	13,000	12,560	-400	-3	
Route No. 14, 14X	7,950	7,080	-870	-11	
Route No. 15	11,930	15,040	3,110	26	
Route No. 18	7,430	6,700	-730	-10	
Route No. 19	16,130	16,150	20	0	
Route No. 23, 23X	10,510	13,930	3,420	32	
Route No. 27	16,530	14,190	-2,340	-14	
Route No. 30, 30X	24,690	29,870	5,180	21	
Route No. 31	8,050	8,190	140	2	
Route No. 62	10,900	7,790	-3,110	-28	
Route No. 76	9,430	12,550	3,120	-33	
Route No. 80	9,620	10,890	1,270	-13	
Total	152,570	163,250	10,680	7	

^aBased on actual operator counts taken during the months of September through June during 1989 through 1991 by the Milwaukee County Transit System. Counties were adjusted to account for monthly and annual variations in ridership so as to be more representative of average weekday 1990 volumes.

Source: SEWRPC.

Table 171

COMPARISON OF ESTIMATED VEHICLE MILES OF TRAVEL ON AN AVERAGE WEEKDAY IN THE REGION: 1990

County	Estimated Average Weekday Vehicle-Miles of Travel from Traffic Counts	Estimated Average Weekday Forecast Vehicle-Miles of Travel from Travel Simulation Models	Percent of Estimated Travel
Kenosha	2,500,000	2,755,000	110.2
Milwaukee	14,391,000	14,691,000	102.1
Ozaukee	1,942,000	2,057,000	105.9
Racine	2,966,000	3,273,000	110.4
Walworth	1,913,000	2,179,000	113.9
Washington	2,379,000	2,436,000	102.4
Waukesha	6,981,000	7,693,000	110.2
Region	33,072,000	35,054,000	106.0

Source: SEWRPC.

arterials within Southeastern Wisconsin and Map 83 presents such a comparison for all arterial facilities within the East-West Corridor in Southeastern Wisconsin.

It may be concluded that the traffic simulation models have the ability to forecast traffic volume with adequate accuracy for transportation planning and facility design purposes. This is particularly true when it is recognized that the actual traffic counts are estimates themselves, having been taken over a triennial period centered on the base year. As such, the counts reflect yearly as well as seasonal variations in traffic flow and random errors that occur in the counting process itself.





Map 81 Inset



GRAPHIC SCALE 400 800 1600 FEET

The major local and express bus routes operated by the Milwaukee County Transit System in 1991 are shown on this map. These routes were selected for purposes of comparing passenger counts with ridership estimates obtained by application of the travel simulation models. A comparison of the passenger counts with the ridership estimates obtained through simulation modeling is set forth in Table 170 for these routes. That comparison demonstrates that the models accurately simulate actual loadings on the transit system.

Source: SEWRPC.

ISSUES RELATING TO TRAVEL SIMULATION MODELS

Nationally, criticisms have from time to time been leveled against the travel simulation models now widely used in transportation system planning. These criticisms have usually been general and leveled primarily by academicians. Some of these criticisms are valid. Others, however, are not valid, particularly with respect to the travel simulation models which have been developed and refined by the Commission over the past 30 years for application in Southeastern Wisconsin. This section of the report identifies and responds to these criticisms.

One of the principal general criticisms of traffic simulation models is that the models have been designed only to produce forecasts of vehicular traffic for use in highway planning and engineering. This is not true with respect to the Commission travel simulation models. For over 30 years the Commission travel simulation models have been developed to assist in both transit and

COMPARISON OF SIMULATED AVERAGE WEEKDAY TRAFFIC VOLUMES AND TRAFFIC VOLUME COUNTS AT SELECTED LOCATIONS ON THE ARTERIAL STREET AND **HIGHWAY SYSTEM: 1990**



WASHINGTON

OZAUKEE

Portrayed on this map are comparisons of the estimated average weekday travel volumes as obtained by traffic counts and by travel simulation model application at selected locations on the regional freeways and surface arterial streets and highways within the Region. These comparisons between traffic counts and simulation model results indicate that the models have the ability to simulate traffic volumes with an accuracy fully adequate for transportation planning and facility design purposes.

Source: SEWRPC.

COMPARISON OF ESTIMATED AVERAGE WEEKDAY TRAFFIC VOLUMES AT SELECTED LOCATIONS ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE EAST-WEST TRAVEL CORRIDOR IN SOUTHEASTERN WISCONSIN: 1991



This map, like Map 82, provides a comparison of average weekday traffic volumes obtained through traffic counting programs and through the application of the regional travel simulation models. This particular comparison demonstrates the validity of the models for use in conducting the East-West corridor study, a study sponsored by the Wisconsin Department of Transportation and carried out simultaneously with the preparation of the new regional transportation system plan. The comparison indicates that the models can adequately simulate traffic flows on the arterial street and highway system.

337

highway system planning and facility engineering, as well as to facilitate consideration of the implications of potential public transit and arterial street and highway improvements on transit ridership and on arterial street and highway traffic volumes and operating conditions. In this respect, it may be noted that the Commission second-generation and the new third-generation travel simulation models presented in this chapter have been reviewed and approved by the Federal Transit Administration for use in detailed corridor-level evaluation of transit alternatives. The Commission travel simulation models have also been designed and applied to facilitate consideration of the implications of alternative regional land use patterns and of the effects of differing levels of population and economic activity and of socio-economic change on travel patterns and on transit and arterial street and highway traffic volumes.

Another general criticism typically leveled against traffic simulation models is that such models are often developed utilizing travel data which may be 10 to 20 years old and substantially out of date or that such models are often developed on the basis of a very small sample survey of existing travel habits and patterns. The Commission travel simulation models, however, have been developed, calibrated, and validated using current travel data. The initial model development took place in 1963 using then current travel and related land use data derived from a comprehensive travel survey. The models were re-calibrated and validated in 1972 based upon the findings of a second "full scale" comprehensive travel survey. The models were again re-calibrated and revalidated in 1991 based upon the findings of a third full-scale comprehensive travel survey.

Another general criticism typically leveled against travel simulation models is that the models reflect neither the implications of traffic congestion nor of substantial highway improvements on trip length and modal choice. This criticism sometimes notes that some models do reflect the effect of traffic congestion and substantial highway improvements on route choice. The Commission travel simulation models, however, have been designed and applied to reflect the effect of congestion, as well as of substantial highway improvements, on travel patterns and trip lengths through use of the trip distribution model, and the effect on modal choice through use of the modal split model, as well as on route choice. The Commission travel 338

simulation models do not, however, assume that traffic congestion or substantial highway improvements will substantially alter trip generation, that is, the number of trips made by households on an average weekday for work, shopping, and other trip purposes. As shown by the Commission travel surveys conducted in a consistent, comprehensive manner at three points in time over a 30-year period, household trip generation rates have been remarkably stable, particularly when the households are categorized by automobile availability and household size.

Another general criticism of travel simulation models relates to the application of the models in the preparation of a transportation system plan. The criticism is that the model applications result in "self-fulfilling" projections of substantially increasing highway traffic and congestion in outlying areas, and this leads to recommendations for excessive highway improvements. This criticism is not valid with respect to the Commission use of its travel simulation models in the regional transportation system planning process. The criticized projections of increasing highway congestion in outlying areas are the result of assumed continuation of existing trends toward the areawide dispersion of urban development and the attendant inputs to the travel models. The Commission, however, has never based its transportation system planning on a forecast of a land use pattern resulting from an assumed continuation of existing land use trends; rather, it derives its travel and traffic forecasts from a normative land use plan which seeks to modify existing land use development trends by a return to a centralized pattern of development.⁵ The

⁵It should be noted in this respect that, in the preparation of the first generation regional transportation system plan, the then State Highway Commission of Wisconsin asked that the Regional Planning Commission prepare an "unplanned" existing trends projection of land use development trends to test the sensitivity of the then recommended highway element of the regional transportation system to a land use pattern substantially different from the recommended regional land use plan. In the preparation of the third generation regional transportation system plan, the Wisconsin Departments of Transportation and Natural Resources made a similar request.

application of the Commission travel simulation models, therefore, indicates the potential travel patterns and highway and transit traffic volumes attendant to a centralized development plan and recommends transportation improvements which would provide the implementation of that land use plan.

Another general criticism typically leveled at the application of travel simulation models in transportation system planning is that, following the application of the travel simulation models in the design and test of alternative transportation system plans and in the development of a recommended transportation system plan, there is no test through application of a land use simulation model of the recommended transportation system plan with respect to its potential to promote desirable land use development and redevelopment. The Commission has always conducted such a test of its recommended transportation system plan. The Commission, however, has conducted such tests through direct measurement of the accessibility provided by the recommended transportation system plan and by comparing the overall level of accessibility and change in accessibility levels to the desired land use pattern in the adopted regional land use plan. The Commission transportation planning objectives and standards explicitly call for higher levels of accessibility to be provided to areas proposed for urban development and redevelopment. The Commission believes that this is a better test than the use of land use simulation models which the Commission has investigated but rejected as unreliable. Land use simulation models typically do not forecast the location of basic employment, which is an important determinant of a future land use pattern, but distribute such employment intuitively. Land use simulation models typically also do not incorporate the effects of natural resource features on the location of land use development. Historically, natural resource features and the availability of sewer service rather than transportation facilities have been the most important factors influencing land use development patterns within Southeastern Wisconsin. The Commission review of accessibility nevertheless allows direct analysis of the potential influence of transportation improvements on land use development and redevelopment.

Another criticism typically leveled against travel simulation models is that the models have

not been developed and applied at a level of detail sufficient to provide reasonable forecasts of traffic for individual highway and transit facilities, or even for broader multi-facility corridors. The Commission has historically developed and maintained a highly detailed travel model with respect to size of traffic analysis zones and level of detail of the highway and transit networks. This has been done specifically to permit the models to be used in the development of traffic forecasts for the second-level planning and engineering of individual highway and transit facilities. The Commission travel simulation models utilize 1,431 internal traffic analysis zones within the 2.689-square-mile Region. The zone size varies with the type and density of land use and ranges down to individual city blocks of about five acres in size. The Commission transportation networks include all arterial street and highway facilities in the Region, totaling approximately 3,300 miles. The Commission transportation networks also include every public transit route operating in the Region. Traffic analysis zones are carefully delineated to permit accurate determination of transit walk access distances and times for each zone, which is necessary to estimate potential transit ridership accurately. The Commission simulation models are calibrated and validated against actual highway traffic and transit ridership volumes and travel times in order to assure that the model accurately represents highway and transit travel times between subareas of the Region, as well as highway and transit traffic on individual facilities.

A related common criticism of travel simulation models is that the highway networks used do not separate street segment and street intersection capacities to permit consideration of the potential effects of both on highway travel time. The Commission has carefully investigated this issue and found that the refinement implied by the criticism is not essential to producing accurate arterial street and highway travel times and volumes in Southeastern Wisconsin. The investigation involved the comparison of modelestimated traffic volumes to actual ground counts and model-estimated travel times to actual measured travel times in both peak and off-peak travel periods. The comparisons consistently indicate that model-estimated traffic volumes and travel times are within 5 to 10 percent of ground traffic counts and measured travel times.

Another general criticism typically leveled against travel simulation models is that the models do not accurately estimate future changes in trip generation or account for varying trip generation rates within a planning area. Commission validations of its travel simulation models, however, have demonstrated that the models do indeed accurately estimate trip production over time, even during periods of substantial socio-economic, lifestyle, and land use changes. To this end, the Commission trip production model incorporates considerations of household size, vehicle ownership, residential density, and household income.

Another general criticism of travel models is that the models may accurately forecast average weekday traffic but do not produce accurate estimates of peak-hour traffic. The Commission travel simulation models provide forecasts of peak-hour traffic volumes from 24-hour average weekday traffic volumes, using factors developed for each individual arterial facility. Within the Region, peak-hour traffic generally represents about 10 percent of the average weekday traffic on an arterial facility, almost always falling within a range of 8 to 12 percent, with the lower percentage typical of facilities serving areas devoted to retail land uses and/or subject to substantial traffic congestion and the higher percentage typical of lower-volume facilities and facilities which serve areas devoted to residential, industrial, and office land use development. Because of the relatively small variation in peak-hour traffic on arterial facilities, the Commission intends to maintain this procedure.

A related criticism is that work and school travel trip distribution and modal choice should be based on peak-hour travel times and not on average daily travel times. However, while the Commission travel simulation models are designed to normally produce forecasts of average weekday traffic volumes for all trip purposes, the models have a feature which permits the incorporation in the trip distribution and modal choice steps of peak-hour travel times for the modeling of work and school travel.

Another general criticism of travel simulation models is that the models cannot directly incorporate the effects of substantial changes in the price, or cost, of travel which may be postulated as potential public policy for demand management purposes. The Commission travel simulation models do permit the incorporation of such pricing measures directly into the modal choice element. The Commission has also in the past utilized trip distribution models which can be used to determine the probable effects of substantial changes in travel costs. In the consideration of pricing alternatives as part of the third-generation regional transportation system plan preparation, the Commission will directly incorporate the pricing alternatives in the modal choice model and will utilize a trip distribution model which considers both travel time and cost in the determination of travel patterns and trip length.

Another general criticism of the travel simulation models is that the models do not directly incorporate the potential for reduced trip production and reduced automobile travel as a result of higher-density land use development and site designs which promote travel by modes other than the automobile. The Commission travel simulation models indirectly incorporate the effect of higher-density development by including residential density in the automobile ownership model, automobile ownership being a key variable in the forecast of trip production. Modifications to the models to consider site design considerations directly are described in the next section of this chapter.

The Commission's third-generation travel simulation models are entirely adequate for use in travel and traffic forecasting within the Region. As has been the Commission's practice, these models will be reviewed and, as may be found desirable, further refined in the years ahead to incorporate any potential improvements developed in national research efforts, just as the Commission's third-generation models reflect refinements and improvements with respect to the Commission's first- and second-generation travel simulation models. Indeed, these thirdgeneration models reflect refinement and improvement with respect to the Commission's first- and second-generation travel simulation models. Such refinements, however, are not essential to the development of accurate travel and traffic forecast within the Region.

Modifications to Travel Simulation Models

As noted in the previous section of this chapter, modifications were made to the travel simulation models to permit the test and evaluation of the impacts of policies that would significantly increase the cost of operating an automobile and of policies that would promote the use of land use site-design practices for new development that would encourage the use of bicycles and walking rather than the automobile for travel. In addition, the vehicle ownership model was modified to include the potential effect of improved and expanded transit service on vehicle ownership. The modifications made to each travel simulation model and applied in the testing of alternative transportation system plans are described below.

Vehicle Ownership: The vehicle ownership model as described in the preceding section of this chapter was modified to include consideration of the effect of transit accessibility on household vehicle ownership. The original and revised models are linear multiple regression equations which were calibrated with 1990 U.S. Bureau of the Census socio-economic data, 1990 Commission land use data, and 1990 Commission transportation network data. The original model expresses vehicle availability to be a function of household income, household size, and residential density, while the revised model also includes transit accessibility. Transit accessibility is a measure of the availability and quality of transit service.⁶ The effect of the addition of transit accessibility to the vehicle ownership model is that improvements in transit service have the potential to reduce vehicle ownership, which in turn has the potential to reduce total trip generation and increase transit ridership. The revised vehicle ownership model may be expressed as:

Central Milwaukee County

 $VAH = 0.0436HI - (0.402 \times 10^{-11})TA + 0.0803$

Remainder of Region

VAH = 0.04	146log _e (HI)	+ 0.190PH -	0.145log _e (HGRA) - (0.351 X 10-11)TA - 0.164
Where:	VAH	=	number of vehicles available per household

н	=	average household income in \$1,000s
PH		number of persons per household
HGRA	=	number of households per developed gross residential acre
ТА	=	transit accessibility factor

<u>Trip Generation</u>: The results of the application of the Commission trip production and attraction models were adjusted to reflect the potential impacts of the adoption of land use site-design practices for new development to the year 2010 that would better accommodate bicycle and pedestrian travel, thereby promoting travel by those modes, rather than by automobile. A reduction factor was applied to total number of person trips made, by personal vehicle and transit, as estimated by the Commission trip production and attraction models, the reduction factor representing the proportion of trips which may be expected to be made by bicycle and walking, rather than by personal vehicle or transit. The reduction factor was applied only to trip productions from, and attractions to, analysis zones which were envisioned in the regional land use plan to have an increase of 50 or more households, or 50 or more jobs, by the year 2010. The percentage reduction was applied across all trip purposes.

The reduction factor applied was 5 percent. A review of the literature indicated that little information existed concerning the potential reduction in vehicle trips per household that may be attributed to bicycle- and pedestrianfriendly land use site design. Two studies indicated that the potential reduction in vehicle trips per household which may be attributed to bicycle- and pedestrian-friendly land use site design as compared to a typical site design would be a maximum of about 5 percent. That is, a maximum of about 5 percent of total trips generated may be expected to use the bicycle or walking modes for travel rather than personal vehicle. It should be noted that this modification was only applied to those transportation system plan alternatives which included policies to implement bicycle and pedestrian friendly land use site designs.

⁶Transit accessibility is a quantitative measure of the relative availability and quality of transit service. It is expressed in the form of an index which for any zone, i, within the Region is defined as the product of the person trip attractions in zone j times the friction factor for the transit travel time value of the interchange from zone i to zone j summed for all interchanges with zone i. The friction factor is an empirically derived relationship defined as the inverse of the door-to-door travel time raised to some power which varies with the travel time. Further clarification of the variables used in the accessibility measure is found in the trip generation, trip distribution, and modal choice sections of this chapter.

Trip Distribution

The trip distribution model was modified by considering both automobile travel time and travel costs, as opposed to only automobile travel times, in the determination by the trip distribution model of the number of trips made between each pair of traffic analysis zones. The premise of the trip distribution model is that the number of trips between two zones in the Region is a direct function of the trips in each zone and a measure of the spatial separation of the zones. The original third-generation trip distribution model only utilized automobile travel time to measure spatial separation, as did the first- and second-generation models. The revised thirdgeneration model was modified to include both automobile travel time and travel cost to permit testing of the implications on trip distribution and trip length of policies which would significantly increase automobile operating cost.

Air Quality-Related Issues

Some concerns have also been raised nationally with respect to the validity of transportationrelated air pollution emissions estimates, which are the product of the travel simulation models and air pollutant emission factor models. Based on the foregoing, it is apparent that in Southeastern Wisconsin the travel simulating models have been shown to be totally adequate for highway and transit system planning and engineering purposes. With respect to air pollutant emission estimation, the Commission utilizes the pollutant emission factor models, maintained by the U.S. Environmental Protection Agency. These models have been criticized as having some shortcomings. These include an inability to reflect the effect of vehicle pollutant emission enrichment which results from rapid vehicle acceleration and deceleration and to differentiate between emissions attendant to starting and stopping from emissions attendant to motion. The Commission must look to the Federal and State agencies concerned to address these issues since the Commission has not been responsible for air quality management planning within the Region since the completion of the first regional air quality maintenance plan in 1982.

SUMMARY AND CONCLUSIONS

This chapter describes the travel simulation models used in the design, test, and evaluation of the alternative transportation system plans under the third-generation regional transportation system plan preparation process. These models and the relationships and techniques incorporated in these models are important, not only because they provide the technical basis for the design of a regional transportation system plan which is properly related to the travel patterns that the planned system must serve, but also because these techniques provide the necessary link between land use and transportation system planning.

The Commission has over 30 years of experience in travel simulation modeling. The initial travel simulation models were developed in 1963, utilizing the findings of comprehensive travel survey and applied in the initial regional land use-transportation study. The initial models were validated in 1972, utilizing the results of a second full-scale travel survey and demonstrated to simulate accurately 1972 travel patterns, arterial street and highway traffic volumes, and transit ridership. Some refinements were made to the models in 1972 before their application in the second-generation transportation study. The refined set of models was revalidated in 1991, utilizing the findings of a third comprehensive travel survey. They were again demonstrated to accurately simulate 1991 conditions. Again, some refinements were made to produce a thirdgeneration set of travel models. The Commission staff thus has substantially demonstrated expertise and substantial experience in the development and application of travel simulation models, as well as intimate knowledge of the travel habits and patterns of the Region and of changes in these habits and patterns over time. The Commission believes that the general criticisms of travel models leveled materially by academicians are not applicable to the Commission travel models. The Commission believes that the most rigorous test possible of a set of travel simulation models consists of applying the full set of models to socio-economic and transportation system data from a year, be it past or future, other than the year of the data from which the models were calibrated and then comparing the model-estimated travel and traffic to survey-estimated travel and actual measured traffic counts. The Commission models have passed this test, not once, but twice, as demonstrated by comparisons indicating modelestimated traffic volumes are within 10 percent of ground traffic counts.

The travel simulation process used by the Commission consists of four major steps: trip generation, trip distribution, modal split, and traffic assignment. The first step in the development of these models for the third-generation planning effort consisted of an assessment of the models developed and used in the second-generation land use-transportation planning effort. Such an assessment was possible because the findings of three identical large-scale comprehensive travel surveys were available for 1963, 1972, and 1991, allowing the testing of the temporal stability of the models. Despite the major changes in socioeconomic conditions, in land use development, and in transportation system development that occurred in the interims involved, the travel simulation models from the first- and secondgeneration planning efforts demonstrated an ability to simulate current travel habits and patterns with remarkable accuracy.

In spite of the excellent performance of the travel and traffic forecasting models developed under the first- and second-generation transportation planning efforts and the conclusions that these models could continue to be used with confidence in either their original form or in a refined form through re-calibration with more recent travel survey data, certain refinements in the models were determined to be desirable.

A summary description of the travel simulation models used in the third-generation planning effort can be found in Table 147. Each of these models was individually validated by using travel survey data; the entire model process chain validated by comparing the outputs of the models to observed ground counts, transit and highway. These analyses clearly demonstrated the validity of the calibrated models to predict future travel and traffic conditions under a broad range of development conditions with accuracies adequate for transportation system planning and engineering. The Commission models are believed to provide the Region with a technically sound transportation systems planning tool which can be used with confidence in the planning and design of surface transportation facilities within the Region.

(This page intentionally left blank)

OBJECTIVES, PRINCIPLES, AND STANDARDS

INTRODUCTION

Planning is a rational process for formulating and meeting objectives. The formulation of objectives, therefore, is an essential task which must be undertaken before plans can be prepared. The seven-county Southeastern Wisconsin Planning Region is composed of many diverse and often divergent interests. Consequently, formulating objectives for the preparation of advisory comprehensive regional development plans is a difficult task.

Soundly conceived regional development objectives should incorporate the combined knowledge of many people who are informed about the Region and should be established by elected or appointed representatives legally assigned this task, rather than by planning technicians. This consideration is particularly important because of the value-system implications inherent in any set of development objectives. Active participation by elected or appointed public officials and by citizen leaders in the regional planning program is implicit in the structure and organization of the Southeastern Wisconsin Regional Planning Commission. Moreover, the Commission, since its inception, has provided for the establishment of advisory committees to assist the Commission and its staff in conducting the regional planning program and in developing objectives, principles, and standards in particular.

The objectives, principles and standards set forth in this chapter reflect the insight of advisory committees operating, since the adoption of the second-generation regional transportation system plan, within the framework of the continuing regional land use-transportation study. Such advisory committees have guided all major transportation planning efforts including the following: the jurisdictional highway system plans prepared for all seven counties in the Region; the transit system development plans and programs prepared for the Racine, Kenosha and Waukesha areas; A Primary Transit System Plan for the Milwaukee Urbanized Area; A Land Use and Transportation System Development Plan for the IH 94 South Freeway Corridor; A Transportation System Plan for the Milwaukee <u>Northwest Side/Ozaukee County Study Area;</u> and <u>A Regional Transportation System Plan for</u> the Transportation Handicapped. The advisory committees involved have had a combined membership of elected and appointed officials and concerned citizens.

The use of advisory committees has been, and still appears to be, the most practical and effective procedure available for actively involving interested and knowledgeable public officials, technicians, and private citizens in the regional planning process and of openly arriving at decisions and programs which can shape the future physical development of the Region. Only by combining the accumulated knowledge and experience about the Region which the various advisory committee members possess can a meaningful expression of the desired direction, magnitude, and quality of future regional development be obtained. One of the major tasks of these committees in the initial regional land usetransportation planning effort was, therefore, to assist in the formulation of regional development objectives and supporting planning principles and standards.

Because of the passage of time, the attainment of additional knowledge about the Region, the formulation of additional development objectives under other related regional and subregional planning programs, the degree of attainment of each of the various adopted regional development objectives, and both adverse and favorable public reaction to plan implementation proposals, careful review of the regional development objectives and supporting principles and standards was deemed essential. This chapter sets forth the results of that review in the form of revised regional transportation system development objectives, principles, and standards. These have been adopted by the Commission after careful review and upon recommendations by the Commission staff and the Technical Coordinating and Advisory Committee on **Regional Transportation System Planning.**

BASIC CONCEPTS AND DEFINITIONS

Definitions for the term "objective," as well as for the terms "principle," "standard," "plan," "policy," and "program," were established for use as a common frame of reference in the initial land use-transportation study. These definitions have remained valid over time and for convenience are set forth below as originally established:

- 1. Objective: a goal or end toward the attainment of which plans and policies are directed.
- 2. Principle: a fundamental, primary, or generally accepted tenet used to support objectives and prepare standards and plans.
- 3. Standard: a criterion used as a basis of comparison to determine the adequacy of plan proposals to attain objectives.
- 4. Plan: a design which seeks to achieve agreed upon objectives.
- 5. Policy: a rule or course of action used to ensure plan implementation.
- 6. Program: a coordinated series of policies and actions to carry out a plan.

Although this chapter deals with only the first three of these terms, an understanding of the interrelationship between the foregoing definitions and the basic concepts which they represent is essential to the following discussion of objectives, principles, and standards.

OBJECTIVES

In order to be useful in the regional transportation system planning process, objectives must be logically sound and related in a demonstrable and measurable way to alternative system development proposals. Only if the objectives can be clearly related to physical development and only if they are subject to objective tests can an intelligent choice be made from among alternative plans so as to select the one plan or combination of plans which best meets the agreed upon objectives.

Recognizing that: 1) various public and private interest groups within a Region as large and diverse as Southeastern Wisconsin may have varying, and at times conflicting, objectives, 2) many of these objectives are of a qualitative nature and, therefore, difficult to quantify, and 3) many objectives which may be held to be important by the various interest groups within the Region may not be related in a demonstrable manner to physical development plans, the Commission identified two basic types of objectives under the first and second regional land use-transportation system planning efforts. These are general development objectives, often referred to by other agencies as "goals," which are by their very nature either qualitative or difficult to relate directly to development plans and to specific development objectives which can be directly related to physical development plans and at least crudely quantified. The other type of objectives, the specific objectives, will be dealt with below. The rationale for using these two types of objectives remains valid; for the most part, the general and specific regional development objectives which have been adopted for use in the transportation system plan reevaluation are quite similar to those formulated and adopted in the first and second land use-regional transportation system planning efforts. Thus, the broad needs which the regional transportation system plan is to be designed to satisfy, as expressed in the form of development objectives, have remained essentially the same.

General Objectives

The following general development objectives have been adopted by the Commission after careful review and recommendation by the Technical Coordinating and Advisory Committee on Regional Transportation System Planning:

- 1. Economic growth at a rate consistent with regional resources, including land, labor, and capital, coupled with a primary dependence on free enterprise, in order to provide needed employment opportunities for the expanding labor force of the Region.
- 2. A wide range of employment opportunities through a broad, diversified economic base.
- 3. Conservation and protection of desirable existing residential, commercial, industrial, and agricultural development in order to maintain desirable social and economic values; renewal of obsolete and deteriorating residential, commercial, and industrial areas in the rural, as well as in the urban, areas of the Region; and prevention of slums and blight.
- 4. A broad range of choice among housing designs, sizes, types, and costs, recogniz-

ing changing trends in age-group composition, income, and family living habits.

- 5. An adequate, flexible, and balanced level of community services and facilities.
- 6. An efficient and equitable allocation of fiscal resources within the public sector of the economy.
- 7. An attractive and healthful physical and social environment with ample opportunities for high-quality education, cultural activities, and outdoor recreation.
- 8. Protection, wise use, and enhancement of the natural resource base.
- 9. Development of communities having distinctive individual character, based on physical conditions, historical factors, and local desires.

The foregoing general development objectives are proposed as goals which public policy within the Region should promote over time. They are all necessarily general but, nevertheless, provide the broad framework within which regional planning can take place and the more specific goals of the various functional elements and component parts of the Region stated and pursued. No ranking is implied by the order in which these objectives are listed. The statement of these objectives is concerned entirely with ends and not with means; the principal emphasis of these general objectives is on those aspects of regional development which relate either to the expenditure of public funds or to the effects of government actions and regulations. With respect to these general development objectives, it was deemed sufficient to arrive at a consensus among the Advisory Committees and the Commission itself that the plan proposals do not conflict with the objectives. Such a consensus represents the most practical evaluation of the ability of the alternative plan proposals to meet the general development objectives.

Specific Development Objectives

Within the framework established by the general development objectives, a secondary set of more specific objectives can be postulated which is directly relatable to physical development plans and can be at least crudely quantified. The quantification is facilitated by complementing each specific objective with a set of quantifiable planning standards which can, in turn, be related to a planning principle supporting the chosen objective. The planning principles thus augment each specific objective by asserting its inherent validity as an objective.

The specific objectives adopted for the regional transportation system plan are largely selfdescriptive. They are concerned primarily with providing a flexible multi-modal transportation system; alleviating traffic congestion; reducing travel time and accident exposure; and minimizing costs and disruptive effects upon communities and upon the natural resource base. The following specific transportation development objectives have been adopted by the Commission after careful review and recommendation by the Technical Coordinating and Advisory Committee on Transportation System Planning:

- 1. A multi-modal transportation system which, through its location, capacity, and design, will effectively serve the existing regional land use pattern and promote the implementation of the regional land use plan, meeting and managing the anticipated travel demand generated by existing and proposed land uses.
- 2. A transportation system which is economical and efficient and best meets all other objectives at the lowest possible cost.
- 3. A multi-modal transportation system which provides appropriate types of transportation needed by all residents of the various subareas of the Region regardless of race, color, national origin, age, physical ability or income status at an adequate level of service; choices among transportation modes; and intermodal connectivity. The transportation system shall also permit ready adaptation to changes in travel demand, transportation technology, modal use, and new transportation system management measures.
- 4. A transportation system which minimizes disruption of existing neighborhood and community development, including adverse effects upon the property tax base.
- 5. A transportation system which serves to protect the overall quality of the natural environment, which promotes the public

health, and which helps achieve ambient air quality standards.

- 6. A transportation system which facilitates the movement of people between component parts of the Region.
- 7. A transportation system which reduces accident exposure and provides for increased travel safety and personal security.
- 8. A transportation system which minimizes the amount of energy consumed, especially of such nonrenewable energy sources as fossil fuels.
- 9. A transportation system which facilitates linked tripmaking, providing facilities and services necessary for efficient, fast, safe, and convenient intermodal connections.

These transportation development objectives are very similar, but not identical, to those adopted under the two previous regional transportation planning efforts. The review and evaluation of the objectives by the Commission staff, Advisory Committee, and the Commission itself indicated that the basic needs which a transportation system should seek to satisfy in the Region have not changed appreciably in the past three decades.

PRINCIPLES AND STANDARDS

A planning principle and a set of planning standards complements each of the foregoing specific transportation system development objectives, as shown in Table 172. Each set of standards is directly related to the planning principle, as well as to the objective, and serves to facilitate quantitative application of the objectives in plan design, test, and evaluation.

In the preparation of the necessary planning principles for the initial regional land usetransportation planning effort, a careful search of the planning literature failed to reveal a documented set of comprehensive principles which were universally accepted as tenets basic to the physical planning process. It was necessary, therefore, to adapt such principles as could be found to the regional planning effort and then to draw upon the collective experience of the practitioners of the many technical disciplines represented on the Technical Coordinating and Advisory Committee to formulate additional principles to augment those adapted from the literature. Thus, through the combined knowledge of experienced technicians, a set of comprehensive planning principles was formulated. These principles were used as guidelines in the first and second regional land use-transportation planning processes and were incorporated, largely unchanged, into the current planning effort. For a comparison of the year 2000 and the year 2010 development standards, see Table 173.

The planning standards herein adopted fall into two groups: comparative and absolute. Comparative standards can be applied only through a comparison of alternative plan proposals, as in the example, minimizing the total vehicle-miles of travel within the Region. No desirable value can be realistically assigned to this standard, therefore, its application must be a comparative one, in which the alternative plan resulting in the lowest vehicle-miles of travel is deemed to best meet this standard. Absolute standards can be applied individually to each alternate plan proposal since they are expressed in terms of maximum, minimum, or desirable values, as, for example, a maximum overall travel time of 35 minutes to three major retail and service centers.

OVERRIDING CONSIDERATIONS

In applying planning standards and in preparing the regional transportation system plan, several overriding considerations must be recognized.

- 1. Standards cannot be used to determine the effect of individual facilities on each other or on the system as a whole. Traffic simulation models are used in this respect to perform a quantitative test of the ability of a proposed system to accommodate the travel demand derived from the land use plan.
- 2. An overall evaluation of each transportation plan must be made on the basis of cost. Such an analysis may show that the attainment of one or more of the standards is beyond the economic capability of the Region.
- 3. It is unlikely that any one plan proposal will meet all the standards completely.

The extent to which each standard is met, exceeded, or violated must serve as a measure of the ability of each alternative plan proposal to achieve the specific objectives which the given standard complements.

- 4. Certain objectives and standards may be complementary; the achievement of one objective or standard may support the achievement of others. Conversely, some objectives and standards may be conflicting, requiring resolution through compromise. Meaningful plan evaluation can take place only through a comprehensive assessment of each of the alternative plans against all of the objectives and standards.
- 5. Standards must be judiciously applied to areas or facilities which are already partially or fully developed. Application of standards in such cases may require extensive renewal or reconstruction programs. In this connection it should be noted that standards concerned with natural resource protection, use, or development or with neighborhood and community development relate primarily to those areas of the Region where the resource base has not as been significantly vet diminished, depleted, or destroyed and where neighborhood and community development has not yet been significantly disrupted. In areas where such disruption, deterioration, depletion, or destruction has already occurred, application of the standards may make it necessary to inaugurate programs which would restore neighborhoods and the resource base to a higher level of quality as well as quantity.¹

The formulation of objectives continues throughout the planning process. As alternative plan proposals are designed and evaluated and as plan implementation is considered, objectives may change. In the planning process, the various alternative plans proposed are ranked according to ability to meet objectives. If the best plan so identified nevertheless falls short of meeting the chosen objectives, either a better plan must be designed or the objectives must be compromised. The plan evaluation process provides the basis for deciding which objectives to compromise. The compromises may take three forms: certain objectives may be dropped because their satisfaction has been proven unrealistic, new objectives may be suggested, or conflicts between inconsistent objectives may be balanced out.

The continued validity of the objectives set forth in this chapter is ultimately derived from community values. Such values are expressed within the framework of the political system as actions to implement plan proposals are advanced over time. Continued adverse public reaction or response to plan implementation actions may indicate a need to reevaluate specific objectives, principles, and/or standards for continued relevance. Conversely, favorable public reaction may indicate that the objectives involved continue to be valid.

Care must be exercised in assessing public reaction to plan implementation proposals to ensure that such reaction, adverse or favorable. truly reflects the values of the citizen body as a whole within the Region and not the values of small special-interest groups. Care must also be exercised to ensure that public reaction reflects long-term, stable community values and not ephemeral opinions. To this end, the plan reevaluation process incorporated attitudinal and personal opinion surveys to provide information on public preferences for various types and levels of transportation facilities and services and for various housing types and locations. When considered in relation to the results of accompanying behavioral studies of travel habits and patterns and of housing types and locations, these surveys probably provide the best available measure of not only current public opinion but also of underlying attitudes and of changes in these attitudes. Therefore, continued assessment of public reaction to plan implementation proposals and of public attitudes and opinions provides a sound basis for considering needed revisions in regional development objectives and standards.

¹Such programs are specifically recommended for surface water resources in the adopted comprehensive watershed plans and in the regional water quality management plan, for air resources in the regional air quality attainment and maintenance plan, and for certain recreational resources in the park and open space plan.

Table 172

TRANSPORTATION SYSTEM DEVELOPMENT OBJECTIVES, PRINCIPLES, AND STANDARDS

OBJECTIVE NO. 1

A multi-modal transportation system which, through its location, capacity, and design, will effectively serve the existing regional land use pattern and promote the implementation of the regional land use plan, meeting and managing the anticipated travel demand generated by the existing and proposed land uses.

PRINCIPLE

An integrated multi-modal regional transportation system connects major land use activities within the Region, providing the accessibility essential to the support of these activities. Through its effect on accessibility, the regional transportation system can be used to induce development in desirable locations and to discourage development in undesirable locations.

STANDARDS

1. The transportation system should provide service by highway and public transit modes within each urbanized area of the Region so that all residents of an urbanized area, without regard to color, race, or national origin, are:

- a. within 30 minutes' overall travel time^a through travel by personal vehicle on the arterial street and highway system and 45 minutes' overall travel time through travel on the public transit system of 40 percent of that urbanized area's employment opportunities;
- b. within 35 minutes' overall travel time of three major retail and service centers in the Milwaukee urbanized area and one such center in the Kenosha and Racine urbanized areas;
- c. within 40 minutes' overall travel time of a major medical center and/or 30 minutes' overall travel time of a hospital and/or medical clinic;
- d. within 40 minutes' overall travel time of a major park or outdoor recreation area;
- e. within 40 minutes' overall travel time of a vocational school, college, or university; and
- f. within 60 minutes' overall travel time of a scheduled air transport terminal.

2. The regional transportation system should be adjusted to the regional land use plan so that a higher relative accessibility is provided to areas in which higher-density development is planned than to areas in which low-density development is planned or to areas which should be protected from development.

3. Urban rapid transit service should connect and serve:

- a. major retail and service centers;
- b. major industrial centers;
- c. major medical centers;
- d. major park and outdoor recreation areas;
- e. vocational schools, colleges, and universities;
- f. scheduled air transport terminals; and
- g. high-density residential areas.

OBJECTIVE NO. 2

A transportation system which is economical and efficient and best meets all other objectives while minimizing public and private costs.

PRINCIPLE

The total resources of the Region are limited, and any undue investment in transportation facilities and services must occur at the expense of other public and private investment; therefore, total transportation costs for the desired level of service should be minimized.

STANDARDS

1. The sum of transportation system operating and capital investment costs^b should be minimized.

2. The direct benefits derived from transportation system improvements should exceed the direct costs of such improvements.

3. Full use of all existing major transportation facilities should be encouraged through low-capital-intensive and noncapital-intensive transportation system management measures^C cooperatively fostered by government, business, and industry, prior to any capital-intensive or disruptive construction or provision of new facilities and services.

OBJECTIVE NO. 3

A multi-modal transportation system which provides appropriate types of transportation needed by all residents of the Region regardless of race, color, national origin, age, physical ability, or income status, at an adequate level of service; choices among transportation modes; and intermodal connectivity. The transportation system shall also permit ready adaptation to changes in travel demand, transportation technology, modal use, and new transportation management measures.

PRINCIPLE

A flexible, intermodal regional transportation system, functionally integrated into the larger urban complex, is necessary to provide an adequate level of transportation service to all segments of the population and to support essential economic and social activities. A regional transportation system consisting, as may be found appropriate, of arterial street and highway facilities, public transit facilities, bicycle and pedestrian facilities, associated terminal facilities, and transportation system management measures can be located and designed to be readily adaptable to changes in transportation technology and to the major socio-economic changes that affect travel demand. Arterial streets and highways provide for the movement of persons utilizing automobiles, taxicabs, buses, and bicycles and for the major transport of goods utilizing trucks and buses. Public transit provides passenger service utilizing rail vehicles, buses, vans, and taxicabs. Public transit supplies additional passenger transportation system capacity which can alleviate peak loadings on highway facilities and assist in reducing the demand for additional highways and for land necessary for parking facilities at major regional land use activities. Bicycle and pedestrian facilities which may provide for the sole movement of bicyclists and pedestrians may share the rights-of-way of arterial streets and can be designed to promote connectivity between various modes of travel. Transportation system management can facilitate safe and efficient travel on highway and public transit facilities, can influence travel demand, and reduce peak loadings on the transportation system.

STANDARDS

1. ARTERIAL STREET AND HIGHWAY SYSTEM

- a. Arterial streets and highways should be provided at intervals of no more than one-half mile in each direction in urban high-density areas, at intervals of no more than one mile in each direction in urban medium-density areas, at intervals of no more than two miles in each direction in urban low-density and suburban-density areas, and at intervals of no less than two miles in each direction in rural areas.
- b. Freeways or expressways should be considered for those travel corridors^d within the Region which meet all of the following criteria:
 - 1) The corridor provides intercommunity service;
 - 2) The desired speeds or a volume-to-design capacity ratio of 1.0 requires control of access and an uninterrupted flow; and

3) Potential average weekday traffic exceeds 45,000 vehicles per day in urban areas and 25,000 vehicles per day in rural areas.

2. PUBLIC TRANSIT

- a. Urban public transit facilities should be provided to connect^e noncontiguous urban development with the urban center^f of an urbanized area and within urbanized areas local transit should be provided to serve^g only high- and medium-density residential neighborhoods and to connect such neighborhoods to the following land areas:
 - 1) Transportation terminal facilities, including interregional and urban rapid and express transit service loading and unloading points and scheduled air transport terminals;
 - 2) Major and community retail and service centers;
 - 3) Major and community industrial centers;
 - 4) Major parks and such special-use areas as zoological and botanical gardens, civic centers, senior citizen centers, fairgrounds, arenas, and stadiums; and
 - 5) Such institutions as universities, colleges, vocational schools, secondary schools, community libraries, hospitals, mental health centers and sanitariums, and seats of State, county, and local governments.
- b. Urban rapid transit service should be provided in travel corridors where service will save a minimum of one minute per mile of travel over alternative local transit service and where in-vehicle trip length is four miles or longer.
- c. Rapid or express transit service should be provided as necessary to reduce peak loadings on arterial streets and highways so as to maintain a desirable level of transportation service between component parts of the Region.
- d. Rapid and express transit service should be extended as warranted to perform a collection and distribution function so as to maximize the convenience of transit service.
- e. Urban residential land shall be considered as served by urban public transit when such land is within the distance or time of the various types of service set forth in the following table:

Type of Urban	Maximum Distance or Time				
Public Transit Service	Walking	Driving	Feeder Bus or Van		
Rapid ^h	0.50 mile	3.0 miles	15 minutes		
Express ¹	0.50 mile	1.5 miles			
Local ^J	0.25 mile	1.5 miles			

- f. The number of residents of an urbanized area served by rapid transit should be maximized.
- g. The number of jobs in an urbanized area served by rapid transit should be maximized. A job shall be considered served by rapid transit if it is within a one-half mile walking distance or a 15-minute feeder bus ride of a rapid transit stop.
- h. Public transit routes should be direct in alignment, with a minimum number of turning movements, and arranged to minimize duplication of service and minimize transfers which would discourage transit use.
- i. Operating headways^k for local transit service within urban areas shall be designed to provide service at intervals capable of accommodating passenger demand at the recommended load standards, but shall not exceed 30 minutes during weekday peak periods nor 60 minutes during weekday off-peak periods and weekends.
- j. Operating headways for rapid transit service should be designed to provide service at intervals capable of accommodating demand at the recommended load standards, but shall not exceed 30 minutes. Operating headways shall be less than 30 minutes if necessary to meet transit demand during weekday peak periods.

k. Urban fixed-route public transit stops within urban areas should be located as follows:

Type of Urban Public Transit Service	Location of Stops
Rapid	At terminal areas and one-half mile or more on line-haul sections
Express	At terminal areas, intersecting public transit routes, intersecting arterial streets, and major traffic generators
Local	From 600 to 1,200 feet apart

- I. Express and local public transit routes should be located sufficiently near concentrations of demand, including within the central business districts, so that 90 percent of transit users need walk no more than one block,¹ or 600 feet, to a stop.
- m. Rapid transit routes should be located sufficiently near concentrations of demand, including the central business districts, so as to maximize the number of users who need walk no more than one-quarter mile to a stop.
- n. The proportion of total trips to the Milwaukee central business district by public transit should be increased to at least 30 percent.
- Public transit stops should be located and designed to minimize walking distance to and from major trip generators; to provide protection from inclement weather; to promote ready access by feeder bus service where appropriate; and to provide, to the greatest extent practicable, modal interface with other forms of personal and public transportation service.
- p. Paratransit service should be available within transit service areas to meet the transportation needs of the elderly and of those persons who because of a mental or physical disability are unable to avail themselves of conventional transit service. Specialized transportation service should be available within the rural portions of the Region to provide a level of transit service at least one day per week.

3. BICYCLE AND PEDESTRIAN FACILITIES

Appropriate bicycle and pedestrian facilities should be provided on those arterial streets and highways, on which bicyclists and pedestrians are legally permitted to operate, identified in the Regional Bicycle and Pedestrian Facilities Plan for Southeastern Wisconsin.

OBJECTIVE NO. 4

A transportation system which minimizes disruption of existing neighborhood and community development, including adverse affects upon the property tax base.

PRINCIPLE

The social and economic costs attendant to the disruption and dislocation of homes, businesses, industries, and communication and utility facilities as well as the adverse affects on the natural resource base can be minimized through the proper location, design, and operation of transportation facilities and terminals.

STANDARDS

1. The penetration of neighborhood units and of neighborhood facility service areas by arterial streets and highways and primary rapid transit routes should be minimized.

2. The dislocation of households, businesses, industries, and public and institutional buildings caused by the reconstruction of existing or the construction of new transportation facilities and terminals should be minimized.

3. The total amount of land used for transportation facilities and terminals should be minimized.

4. The reduction of the property tax base caused by the reconstruction of existing or the construction of new transportation facilities and terminals should be minimized.

5. The destruction of historic buildings and of historic, scenic, scientific, archaeological, and cultural sites caused by the reconstruction of existing or the construction of planned transportation facilities and terminals should be minimized.

6. The proper use of land for, and adjacent to, transportation facilities should be maximized and the disruption of future development minimized through advance reservation of rights-of-way for transportation facilities.

7. Transportation facility construction plans should be developed which use sound geometric, structural, and landscape design standards which consider the aesthetic quality of the transportation facilities and the areas through which they pass and which consider any environmental enhancement activities likely to be required.

8. Transportation facilities should be so located as to avoid destruction of visually pleasing buildings, structures, and natural features and to enhance vistas to such features.

OBJECTIVE NO. 5

A transportation system which serves to protect the overall quality of the natural environment, which promotes the public health, and which helps to achieve ambient air quality standards.

PRINCIPLE

Adverse effects on the natural environment, air pollution, water pollution, and the loss of natural habitat and biological diversity in particular, can be minimized through the proper location, design, and operation of the transportation system. The relationship of the residents of the Southeastern Wisconsin Region to the natural environment should be one of stewardship.

STANDARDS

1. The location of transportation facilities in or through primary environmental corridors, particularly through the woodland and wetland portions of such corridors, should be minimized.

2. Any damaging effects on the natural resource base caused by the construction of transportation facilities should be minimized.

3. The amount of air pollutants emitted through the operation of the transportation system should be minimized.^m

4. The loss of prime agricultural farmland to transportation facility construction should be minimized.

OBJECTIVE NO. 6

A transportation system which facilitates the movement of people and goods between component parts of the Region.

PRINCIPLE

To support the everyday economic and social activities, a transportation system which provides for reasonably fast, convenient travel is essential. Personal vehicle travel, while offering a high degree of mobility, comfort, and convenience, can result, particularly in corridors of high travel demand, in traffic congestion, excessive air pollutant emissions, and unnecessary motor fuel consumption. Effective and attractive high-quality public transit service and bicycle and pedestrian facilities may have the potential to directly reduce traffic congestion and associated personal delay, energy consumption, and air pollution when used by previous automobile users. Traffic congestion increases the costs of transportation and can adversely affect the attractiveness of an area for residential use and for the location and operation of businesses and industries.

STANDARDS

1. Total passenger-hours of travel, by highway and public transit modes, within the Region should be minimized.

2. Total vehicle-hours of highway travel within the Region should be minimized.

3. Total vehicle-miles of travel within the Region should be minimized.

4. Highway transportation facilities should be located and designed so as to provide adequate capacity, that is, a volumeto-design capacity ratioⁿ equal to, or less than, 1.0 on the basis of 24-hour average weekday traffic volumes, to meet the existing and potential travel demand.

5. Urban public transit facilities should be designed, implemented, and operated so as to attract the maximum number of travelers currently operating single occupancy vehicles and to provide adequate transit vehicle capacity to meet existing and potential travel demand. The average maximum load factor⁰ shall not exceed 1.00 in rapid, express, and local transit service in off-peak periods or beyond the 10-minute point^p in peak periods. The load factor should not exceed 1.00 in rapid and express transit service provided by bus in peak periods or 1.25 in rapid and express transit provided by rail in peak periods. The load factor should not exceed 1.25 in local transit service in peak periods.

6. Bicycle and pedestrian facilities should be located and designed to attract the greatest number of travelers currently operating single-occupancy vehicles.

7. Adequate capacity and a sufficiently high level of geometric design should be provided to achieve the following overall travel speeds based on average weekday conditions for the arterial street and highway and the urban public transit components of the transportation system:

	Overall Tr	avel Speed by Area ^q (m	niles per hour)
Transportation System Component	CBD	Urban	Rural
Arterial Street and Highway			1
Freeway	35-55	40-55	50-55
Expressway	25-40	30-50	50-55
Standard Arterial			
Divided	15-25	25-45	45-55
Undivided	15-25	20-40	40-50
Urban Public Transit			an a
Rapid			
Rail	20-30	40-60	40-60
Bus	10-20	40-50	40-50
Express	10-20	20-35	40-50
Local	5-15	10-20	40-50

8. The use of transportation system management measures should be maximized in travel corridors to achieve the desired level of service for both arterial street and highway and public transit facilities and services.

OBJECTIVE NO. 7

A transportation system which reduces accident exposure and provides for increased travel safety and personal security.

PRINCIPLE

Accidents take a heavy toll in life, property damage, and human suffering; contribute substantially to overall transportation costs; and increase public costs for police, emergency medical services, and other social services. Therefore, every attempt should be made to reduce both the incidence and severity of accidents. Crime and the perception of crime hamper the mobility of persons who must travel within areas deemed unsafe, especially those persons dependent on public transportation; promotes urban blight and unsafe and difficult living and working conditions for those individuals and business who cannot move away from high-crime areas; promotes the costly dispersion of urban development as business and residents seek safer commercial and residential arrangements; increases public costs for police, emergency medical services, and other social services. Therefore, every attempt should be made to reduce the incidence of crime where it hampers mobility and access to basic opportunities the transportation system would otherwise provide in the absence of crime and to increase personal security in the operation of the transportation system.

STANDARDS

1. Travel on facilities which exhibit the lowest accident exposure should be maximized.

2. Traffic congestion and vehicle conflicts should be reduced by maintaining a volume-to-design capacity ratio equal to, or less than, 1.0, on the basis of 24-hour average weekday traffic volumes.

3. Railroad grade separations should be provided at all crossings involving the provision of intercity passenger and commuter train service. For all other crossings, the decision whether or not to provide grade separations should be made at the project planning stage.

OBJECTIVE NO. 8

A transportation system which minimizes the amount of energy consumed, especially such nonrenewable energy sources as fossil fuels.

PRINCIPLE

The environmental costs attendant to the widespread consumption, as well the mining, drilling, and transport, of fossil fuels used in the operation of the transportation system can include air and water pollution and the despoiling of natural landand water-based wildlife habitats. The long-term efficiency of the transportation system depends on the conservation of existing nonrenewable energy sources and the increased application of renewable energy sources to fuel transportation.

STANDARDS

1. The total amount of nonrenewable energy consumed in the operation of the transportation system, particularly petroleum-based fuels, should be minimized.

OBJECTIVE NO. 9

A transportation system which facilitates linked tripmaking, providing facilities and services necessary for efficient, fast, safe, and convenient intermodal connections.

PRINCIPLE

An intermodal transportation system provides for efficient interaction among appropriate modes of transportation to facilitate effective passenger and freight movement. Where the use of more than one transportation mode is essential for travel between two points or is best able to achieve transportation related objectives, proper modal access, terminal capacity, coordination among transportation providers and between route and schedule information and services are necessary to prevent travel delays and unwanted transportation movements. Time spent waiting for transfers between or among modes raises the costs of travel and may discourage the use of certain modes.

1. The time individuals spend waiting at any modal transfer point for connecting modes of transportation should be minimized.

2. Parking should be provided at park-and-ride transit stations to accommodate the total parking demand generated by trips which change from auto and bicycle to public transit at each station and at carpool lots to accommodate the total parking demand generated by carpool and other ridesharing participants.

NOTE: Standard 1a of Objective No. 1, as initially approved by the Advisory Committee, indicated that both highway and transit modes should provide, within 30 minutes overall travel time, access to 40 percent of an urbanized area's employment opportunities in an urbanized area. In the preparation of alternative system plans, it was found difficult to meet this standard by travel on transit even with significant improvements in transit service. Two factors were found to contribute to this difficulty: the dispersed nature of employment and resident locations within the urbanized areas; and the fact that nearly one-half of overall transit travel time was spent out-of-vehicle. Accordingly, the standard public transit travel time was increased to 45 minutes.

^aOverall travel time is defined as the total door-to-door time of travel from origin to destination, including the time required to arrive at the vehicle and leave the vehicle as well as over-the-road travel time.

^bThe costs to be considered may be termed "life-cycle costs" and include capital, maintenance, and operational costs for facilities over the projected physical and economic life of the facility.

^CLow-capital-intensive and noncapital-intensive alternatives to the construction and provision of new transportation facilities and services may include, but are not limited to, the following transportation management measures:

- 1. Such traffic engineering improvements as left- and right-turn lanes, channelization, one-way streets, reversible traffic lanes, intersection widening, bus turnout bays, and improved signing and pavement markings.
- 2. Such traffic control improvements as coordination of traffic signals, use of bus-priority signal control systems, and computer-based traffic control and freeway traffic management.
- 3. Such freeway operational control as advisory information, incident management, on-ramp metering and monitoring, and high-occupancy-vehicle (HOV) lanes and preferential access.
- 4. Ride-sharing programs.
- 5. Such parking management measures as pricing of off-street parking to encourage ride-sharing for short-term parking, preferential carpool/vanpool parking, and increased rates for weekday parking in the central business district.
- 6. Such transit service improvements as special bus lanes, transfer centers, bus turnout bays, shelters, reduced transit fare programs, shuttle service between retail and employment sites, and computer-based interactive scheduling and routing systems.
- 7. Employer-designed trip reduction strategies.
- 8. Staggered work hours.
- 9. Liberal licensing of taxicabs.
- 10. Banning private vehicles from sections of central business districts during weekdays.

^dThe term travel corridor is defined as a relatively long and narrow geographic area centered on an existing or proposed arterial highway or rapid transit facility along which a substantial volume of persons or goods are, or are expected to be, transported.

^eUrban public transit facilities shall be considered to connect noncontiguous urban development with the urban center of an urbanized area when the transit vehicle provides immediate access to the urban center and to a public transit system serving the urbanized area.

^fThe term "urban center" is defined as the largest concentrated complex of commercial activities within a single urbanized area.

^gUrban residential land when shall be considered served by public transit when such land is within the distances of a transit route set forth in Standard 2e of Objective 3.

^hRapid transit is intended to facilitate relatively fast and convenient transportation along heavily traveled corridors and between major activity centers and high-density residential communities. Rapid transit has relatively high average operating speeds and relatively low accessibility, with station spacings located one-half mile or more apart. Rapid transit service can be provided by commuter rail and "heavy" rail operating over exclusive, grade-separated rights-of-way or by buses operating over exclusive, grade-separated busways. Rapid transit can also be provided by buses operating in mixed traffic on freeways and by "light" rail operating over exclusive, though unseparated-grade, rights-of-way.

¹Express transit service is provided over arterial streets and highways with stops generally located less than 1,200 feet apart at intersecting transit routes, intersecting arterial streets, and major traffic generators. Express transit serves trips of moderate length and can be provided by bus or by light rail operating in mixed traffic on shared rights-of-way. Express mass transit service provides a greater degree of accessibility at somewhat slower operating speeds than rapid transit; it may provide "feeder" service to the rapid transit system.

^JLocal transit service is characterized by a high degree of accessibility and low operating speeds. Local service is provided over arterial and collector streets, with stops located no more than 1,200 feet apart. Such service can be provided by bus, trolley, or light rail vehicles. Local transit also provides a passenger collection-circulation-distribution function within major

activity centers. The collection-circulation-distribution function of local transit service may include the use of buses, vans, trolleys, light rail vehicles, automated guideway vehicles, and other types of people movers, such as moving ramps.

^kThe term "operating headway." is defined as the time between any two vehicles operating with fixed routes and schedules.

¹The percent of urban public transit users walking less than one block from transit stop to destination within the Kenosha, Milwaukee, and Racine central business districts in 1991 is set forth below.

Central Business District	Percent of Transit Users Walking Less than One Block (1991)
Kenosha	87
Milwaukee	81
Racine	90

^mAn analysis, based upon guidelines promulgated by the U. S. Environmental Protection Agency, will be undertaken to demonstrate conformity of the final recommended regional transportation system to the objectives of the Federal Clean Air Act as reflected in the State Implementation Plan for Air Quality.

ⁿVolume-to-design capacity ratio is defined as the relationship between the average weekday traffic volume on a particular section of the arterial system and the design capacity of that section, with volume and design capacity expressed in terms of number of vehicles per 24 hours. The design capacity of arterial facilities is set forth in the following table.

Facility Type	Average Daily Traffic Volumes (vehicles per 24 hours)
Urban Freeway	
Four-Lane	60,000
Six-Lane	90,000
Rural Freeway	· · ·
Four-Lane	52,500
Six-Lane	85,000
Urban Standard Arterial	
Two-Lane	13,000
Four-Lane Undivided	17,000
Four-Lane Divided	25,000
Six-Lane Divided	35,000
Eight-Lane Divided	45,000
Rural Standard Arterial	
Two-Lane	7,000
Four-Lane Divided	25,000

Arterial facilities operating at or under design capacity will generally permit the following average speeds to be achieved during peak-traffic periods:

	Average Traffic Speed		
Facility Type	Urban	Rural	
Freeway			
Posted Speed 50 mph	40-50		
Posted Speed 55 mph	45-55	45-55	
Posted Speed 65 mph		55-65	
Standard Arterial		-	
Posted Speed 30 mph	18-27		
Posted Speed 40 mph	27-37	30-40	
Posted Speed 55 mph		40-55	

Urban standard arterial streets operating over design capacity will provide substantial delays at signalized intersections. During peak-traffic periods, vehicles may have to wait through more than one traffic signal red phase. The average delay to each vehicle at controlled intersections will be at least 35 seconds and may approach 120 seconds. The average travel speed along such urban arterials will generally be less than 15 to 20 miles per hour (mph). In addition, the potential for accidents is increased on arterials carrying traffic volumes over design capacity. Standard arterials operating at design capacity will have average speeds of about 20 to 30 mph and average delays at signalized intersections of about 25 seconds. Urban standard arterials operating below design capacity will have average speeds of 25 to 40 mph and average delays at signalized intersections of five to 15 seconds.

Rural arterials (with a 55-mph speed limit) operating over design capacity will have average speeds of 30 to 45 mph with significant restrictions on lane changing on multi-lane facilities and on passing on two-lane facilities. Rural arterials operating at design capacity will have average speeds of 40 to 50 mph with some restrictions on lane changing or passing. Rural arterials operating under design capacity will have average speeds of 50 to 55 mph with minimal restrictions on lane changing or lane changing on lane changing or
Freeways operating over design capacity will have average speeds of 30 to 45 mph with significant restrictions on lane changing. Stop-and-go traffic at speeds below 30 mph may occur behind over-design-capacity freeway sections. Freeways operating at design capacity will have speeds of 40 to 50 mph with some restrictions on lane changing. Freeways operating under design capacity (with 55-mph speed limit) will have average speeds of 55 mph with no restrictions on lane changing.

⁰The average maximum load factor is defined as the ratio of the number of passengers carried on public transit vehicles past the maximum load point of any route to the seating capacity of vehicles past that point in the peak-flow direction during the operating period.

^pThe 10-minute point is a point located 10 minutes of travel time from the maximum load point on any public transit route. Application of this standard would provide that no passenger would have to stand on board the public transit vehicle for longer than 10 minutes.

^qOverall travel speed is defined as the over-the-road travel distance divided by the overall travel time.

Source: SEWRPC.

Table 173

COMPARISON OF TRANSPORTATION SYSTEM DEVELOPMENT STANDARDS: DESIGN YEAR 2000 AND REVISED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLANS

Objective and Standard Numbers Standard					
Year 2000	Revised Year 2010	Name	Year 2000	Revised Year 2010	Rationale for Change
3(1b)	(3)1b	Traffic volume threshold for freeway consideration	30,000 vehicles per day, urban areas; 15,000 vehicles per day, rural areas	45,000 vehicles per day, urban areas; 25,000 vehicles per day, rural areas	Revised threshold more accurately reflects volumes standard arterials currently carry
	(3)3	Bicycle and pedestrian facility		Should be provided along arterials identified in the Regional Bicycle and Pedes- trian Facilities Plan for Southeastern Wisconsin	New standard emphasizes multi-modal approach to regional transportation planning
(3)3b		Parking in central business district (CBD)	Within central business district parking should be provided at various levels		Standard dropped because the regional planning effort does not explicitly consider CBD parking
(3)3c		Short-term parking in central business districts	Parking provided so that 90 percent of short term parkers need walk no more than one block		Standard dropped because the regional planning effort does not explicitly consider CBD parking
(3)4a		System adaptability	Transportation system should be capable of ready adapta- tion to changes in travel demand and transportation technology		Standard dropped, since experience has shown standard cannot be applied as comparative criterion; emphasis is better addressed in other objectives and standards
4(7)		Harmful and annoying noise levels	Transportation system should be designed and located to minimize exposure of residents to harmful and annoying noise levels		Standard dropped, since experience has shown stand- ard to be more suitably applied in local/project level planning
4(8)	(4)5	Destruction of important sites	Destruction of historic buildings and of historic, scenic, scientific, and cultural sites should be minimized	Destruction of historic buildings and of historic, scenic, scientific, archaeological, and cultural sites should be minimized	Revised standard reflects need to preserve archaeological sites in the Region
	5(3)	Air pollution		The transportation system should be located, designed, and operated so as to minimize air pollution	New standard to recognize importance of minimizing mobile sources of air pollution
	5(4)	Minimize loss of prime agricultural land		Loss of prime agricultural land to transportation facility construction should be minimized	New standard to reinforce one of the basic recommendation of the regional land use plan, preservation of prime farmland
(5)5&6	(6)4	Highway facilities, adequate design capacity	Volume to design capacity equal to, or less than, 1.1	Volume to design capacity equal to, or less than, 1.0	Revised standard reflects more precise calculation of volume to capacity ratios as set forth in Highway Research Board: <u>Highway Capacity Move-</u> <u>ment</u> , 1985
	(6)6	Location and design of bicycle and pedestrian facilities		Located and designed to attract the maximum proportion of travelers currently using automobile	New standard reflects need to offer bicycling and walking as attractive alternatives to the automobile for certain trips

Table 173 (continued)

Objective and Standard Numbers		Standard			
Year 2000	Revised Year 2010	Name	Year 2000	Revised Year 2010	Rationale for Change
	6(8)	Transportation system manage- ment in travel corridors		Use of transportation system management should be maximized in travel corridors to achieve desired levels of highway and public transit service	New standard reflects need to promote more efficient travel in certain corridors in light of severe traffic congestion and attendant problems, including air pollution
(3)3a	(9)2	Provision of parking	At park-and-ride public transit stations to accommodate demand generated by trips changing from auto to transit	At park-and-ride public transit stations to accommodate demand generated by trips changing from auto and bicycle to transit and at carpool lots to accommodate demand generated by ridesharing participants	Add criteria for bicycle parking demand to emphasize inter- modal connections and new criteria for provision of parking at carpool lots

Source: SEWRPC.

(This page intentionally left blank)

Chapter IX

ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM MANAGEMENT PLANS

INTRODUCTION

The travel demand generated by a given land use pattern can be met by various combinations of transportation facilities, services, and demand management measures. With respect to arterial street and highway facilities, the travel loads unconstrained by demand management measures can be distributed in relatively low volumes over a large number of surface arterials, providing a relatively low level of service. Those travel loads can also be concentrated in relatively high volumes on a small number of high-capacity freeways, providing a relatively high level of service. A similar wide range in the configuration of transit facilities can be envisioned. In addition, the various highway and transit facility configurations can be combined with various travel demand management measures which serve to constrain demand and limit the loads on transportation facilities as well and to reduce the need for investment in new or expanded facilities. Actual transportation systems will inevitably consist of a complex combination of these extreme approaches to capacity provisions, modal split, and demand management.

Ultimately, the attainment of a balance between traffic loads and service levels in system design will be dictated by an assortment of factors, including the transportation facilities that already exist or have been committed to construction, the transportation system development objectives and standards to be met, considerations of system integration and continuity, the pattern of land uses to be served, and related benefits and costs. Some of these considerations may be conflicting, and as a result, the transportation system plan finally selected for adoption and implementation will represent a compromise between the theoretical and the practical, the desirable and the possible, and between the demand for transportation facilities and services and the ability to pay for those facilities and services. The design of the year 2010 alternative regional transportation system plans required a methodology permitting the thoughtful incorporation of these design considerations. This chapter presents that methodology.

This chapter also describes the no-build transportation system plan alternative, evaluates the operation of the no-build system under the travel demand derived from the adopted regional land use plan, describes the three alternative transportation system management plans designed and considered, and describes the results of testing these alternatives under the adopted regional land use plan.

PLAN DESIGN METHODOLOGY

It is in the system design phase of the transportation planning process that alternative future transportation networks are synthesized to satisfy regional transportation system development objectives and standards while meeting overriding criteria of system integration and continuity. As already noted, the design of transportation networks is a highly complex process, requiring not only the assimilation of large amounts of information and the development and application of traffic simulation models, but also the exercise of experienced engineering judgment. To a considerable extent, the process is one of finding successive approximations of the best design solution, with specific solutions being proposed to specific system problems in each iteration, then tested through application of the traffic simulation models. The more comprehensive and detailed the knowledge and understanding of regional traffic patterns to be served, the more readily sound design solutions can be found to satisfy development objectives.

Proper utilization of traffic assignments, derived from application of traffic simulation models, requires careful analysis of the resulting network volumes to find possible design solutions to indicated problems. Such utilization also requires the conversion of these volumes into a form usable in plan evaluation, so that the degree to which plan objectives and standards are met can be determined. The analyses made for system design and evaluation purposes involve application of certain well-developed engineering techniques, the most important of which warrant brief description here. (This page intentionally left blank)

causing traffic overloads nor of the tripmakers located in areas inadequately served. Future traffic assignments alone will not effectively suggest possible design solutions to eliminate these deficiencies most effectively.

To provide the necessary information about the characteristics of trips, special screenline, trip origin-destination, and selected link analyses were made using the traffic simulation models. In the screenline analyses, traffic distribution within major corridors was examined along sections across the corridors which cut all of the major transportation facilities. The sections, or screenlines, were delineated on the basis of an analysis of the results of the initial traffic assignments. The distribution and characteristics of the traffic crossing the screenline on the major facilities within the corridor were then determined and compared with the distribution of the physical capacity of the various facilities serving each such corridor, the total transportation system capacity in the corridor evaluated against the loads, and the possible diversion of traffic between overloaded and underloaded facilities within the corridor analyzed.

Since highway service is virtually ubiquitous within the Region, if direct routes do not exist between two subareas of the Region, trips desired to be made between such areas will still be made. albeit by less direct routes. Also, as more desirable transportation facilities between two subareas of the Region become overloaded, additional trips between these areas will still be made, but on less direct routes, utilizing facilities that have available capacity. Traffic loads, therefore, are continuously redistributed as existing facilities become overloaded and new facilities are constructed until a state of equilibrium is approximated in the system. If the volume of future trips between certain concentrations of trip origins and destinations within the Region is sufficiently large, the use of demand management, the provision of direct transit service, or the construction of a direct transportation facility may be justified to assist in achieving the desired equilibrium in the system. The need for such transportation improvements becomes particularly acute if the circuitous movement of heavy traffic volumes between portions of the Region results in the overloading of facilities required to serve other travel demands.

To facilitate identification of the demand for direct movement between subareas of the Region, the Region was divided into planning analysis areas consisting of combinations of adjacent traffic analysis zones. In this way, shorter local trips could be treated as intra-area trips, leaving considerably fewer traffic movements to be studied to ascertain major future travel desires. Similar results can also be achieved by assigning selected trip length categories to the existing plus committed transportation network, a technique which was also used in plan design.

Finally, a better understanding of the characteristics of trips utilizing overloaded links in the network was gained through selected link analyses. This involved the identification of heavily overloaded links in the existing plus committed network and identification of the origins and destinations of all trips passing through these links as well as of the land use activities at the origins and destinations. Thus, it was possible to identify the specific interzonal trips which may be expected to utilize heavily overloaded facilities and to analyze the feasibility of rerouting these trips over other portions of the transportation system. It was also possible to consider travel demand management measures to reduce travel demand on specified overloaded facilities. This technique provided a particularly powerful tool to identify circuitous travel paths as well as facilities requiring transportation system management and additional capacity to relieve overloads on more direct routings.

To provide the necessary information about characteristics of tripmakers located in residential areas inadequately served by the transportation system, trip pattern, household income, and automobile availability data were analyzed to determine the type of transportation improvements most appropriate to meet Standard No. 1 of Objective No. 1. Public transit service is not ubiquitous within the Region, nor even within the urbanized areas of the Region. Therefore, if transit routes between subareas of an urbanized area do not exist, trips desired to be made between such subareas may be made by alternatives to transit, by less direct transit routes if available, involving increased travel times and inconvenience, or such trips may not be made at all. If the volume of desired trips between certain concentrations of trip origin and destination is large and these trips cannot be conveniently made or cannot be made at all because of the lack of transit service, the provision of direct and

frequent transit service linking such concentrations may be justified. The need for such service becomes particularly acute when its absence contributes to a lack of access to employment, educational, shopping, recreational, and social opportunities otherwise provided to those able to make private use of highway facilities.

Plan Synthesis

On the basis of the analyses of the assignment of future trips to the existing plus committed transportation facilities, utilizing the techniques described, the need for new facilities was identified and alternative transportation system plans synthesized. The assignment of future trips was then made to alternative transportation system management plans. These alternative plans consisted of the existing plus committed arterial street and highway system and the proposed transit systems and demand management measures. This analysis procedure was repeated until all possible system deficiencies could be found to be resolved. Finally, as described in Chapter X of this report, arterial street and highway improvements were postulated to resolve remaining identified deficiencies until a practical and workable transportation system design had been evolved. Thus, proposals advanced to overcome indicated deficiencies in the transportation system were tested and evaluated.

Thus, the plan design process consisted of sequentially postulating, evaluating, and selecting alternative combinations of transportation system management measures (including demand management measures, traffic management measures, and public transit improvements) and arterial street and highway improvements. The degree to which such combinations resolved expected deficiencies was tested until a combination of solutions deemed best able to achieve transportation development objectives was found. Arterial street and highway improvements were postulated only to resolve residual system deficiencies, that is, traffic congestion problems not resolved through postulated transportation system management measures.

It is important to note that the alternative travel demand management measures selected for consideration provided alternative frameworks for the subsequent construction of the public transit and the arterial street and highway improvement alternatives. This concept becomes clear when it is considered that travel demand management measures that substantially increase the cost of single-occupancy vehicle travel tend to reduce travel demand and affect mode choice more than measures that do not increase the cost of travel. Since travel demand and modal choice are affected to varying degrees through this process, the consideration of alternative combinations of public transit and arterial street and highway improvements is required. A mix of transportation improvements well suited to one demand management alternative may not be well suited to another.

As previously indicated, transportation system deficiencies expected to occur under future land use patterns can be met by various combinations of transportation system improvements. The methodology employed in the design of the year 2010 alternative transportation system plans allowed for explicit consideration of a variety of such combinations and enabled the most costeffective alternative plans to be designed. Because the design process started with the consideration of transportation system management measures, relatively inexpensive and nondisruptive measures to resolve expected deficiencies and to meet development objectives and standards were considered before more costly and potentially disruptive improvements. The plan design methodology thus explicitly considered travel demand management measures, traffic management measures, and public transit improvements to address the transportation needs of the Region prior to considering arterial highway improvements and expansions.

The traffic analyses used in plan design required application of the full battery of travel and traffic simulation models described in Chapter VII of this report. Application of these models included the generation of person trips at the zonal level, distribution of trips generated by trip purpose between traffic analysis zones, subdivision of distributed trips by personal vehicle and public transit modes, assignment of travel demand to the arterial street and highway network and the transit network, and evaluation of the assigned travel demand against the supply of transportation capacity provided by each alternative plan analyzed. Other transportation service and facility improvements were added to the alternative system plans until the demand for transportation service could be met and the agreed-upon transportation system development objectives and standards largely satisfied.
The plan design methodology permitted the implications of a broad range of actions concerning transportation system development to be evaluated and presented for critical public review and reaction. The Technical Coordinating and Advisory Committee assisted the Commission staff in identifying the base year system, reviewing the traffic analyses made to identify the operating conditions associated with that system in the design year, developing and analyzing the initial alternatives postulated to address deficiencies identified in the existing system under future traffic loadings, and in designing the final alternative regional transportation system plans.

Sources of Design Solutions

In designing the year 2010 alternative regional transportation system plans, preliminary design solutions to be tested and evaluated were drawn from three major sources. The first source consisted of travel demand management measures and traffic management measures drawn from the Clean Air Act Amendments of 1990 and relevant literature. It was recognized that, upon consideration and incorporation of these measures in the final recommended plan, the Wisconsin Department of Natural Resources may have classified them as transportation control measures for inclusion in the State Implementation Plan for Air Quality and in the multi-state plan for Southeastern Wisconsin, Northeastern Illinois, Northeastern Indiana, to attain ambient air quality standards for ozone.

The second source of preliminary design solutions consisted of a detailed reconsideration of the adopted regional transportation plan undertaken by public transit operators, the Wisconsin Department of Transportation, the seven county highway agencies concerned, local municipal planning and public works agencies, and county and local elected officials and concerned citizens serving on jurisdictional highway planning and transit development planning advisory committees. Such reconsideration of the adopted plan was undertaken in the preparation of transit development programs for the urban areas of Kenosha, Milwaukee, Racine, and Waukesha and the preparation of the seven county jurisdictional highway plans. The preliminary design solutions from the second source thus originated with the experienced professional engineers and planners in the employ of the State, county, and local units of government, who had a knowledge

of, and long-standing experience with, the transportation system within the Region and county and local elected officials who had intimate knowledge of public attitudes and concerns within their political jurisdictions.

The third source of preliminary design solutions was developed directly from analyses of the traffic assignments and subsequent network analyses, in which solutions to correct system deficiencies became apparent through the knowledge acquired of the existing and probable future traffic patterns within the Region and the manner in which these were being distributed on the existing plus committed transportation network.

All proposed design solutions developed from these three sources were carefully reviewed by the Technical Coordinating and Advisory Committee. As approved or modified by the Committee process, the proposed improvements were synthesized into a potential transportation system and the resulting system was tested and refined as heretofore described.

Consideration of Federal

Provisions in Plan Design

The methodology used in the design of the year 2010 alternative regional transportation system plans reflected careful consideration of the requirements of the Clean Air Act Amendments of 1990 (CAAA) and the Intermodal Surface Transportation Efficiency Act of 1991 (ISTEA). ISTEA in particular requires that the metropolitan transportation plan address specific elements. Appendix B of this report documents the Commission's explicit consideration of these elements.

NO-BUILD TRANSPORTATION SYSTEM PLAN

One possible course of action for the provision of transportation services and facilities in the Region would be to make no major improvements to the existing transportation system. This course would attempt to serve existing and probable future travel demand entirely with existing arterial street and highway and transit facilities. This "do-nothing," or no-build, alternative not only represents a possible policy alternative for the Region but, from a technical standpoint, also becomes the point of departure for initial testing and design of alternative transportation plans which do incorporate facility and service improvements. Thus, the nobuild alternative comprises both the final system to result from pursuing a course of action that would minimize capital investment in transportation facilities and the point of departure for investigating a range of possible transportation system improvements that would serve travel demand through additional capital investment and offering increased system capacity and improved levels of service.

Determination of a No-Build Transportation System

In developing the transportation system which would constitute the no-build alternative plan, three possible definitions were considered by the Advisory Committee.

The first definition identified the no-build system as composed only of those transportation facilities in use as of January 1, 1994. The advantages of this approach lie principally in the simplicity of identification and the potential that it offers for agreement on the part of all concerned. The principal disadvantage of this approach is that it ignores improvements to the regional transportation system that may currently be in a stage of implementation. Some of these improvements or modifications may be so nearly complete that to ignore them would not only be unrealistic but could lead to misinterpretation of any system test results.

Since transportation system development is a dynamic process and since certain transportation facility improvements within an urbanizing region are normally in some stage of implementation and will become available to system users in the near future, there may be certain proposed, as well as partially completed, facilities that should, as a practical matter, be incorporated into the no-build transportation system. While no absolute certainty exists that these improvements will be completed and placed in operation, most reasonable persons probably would agree that certain transportation improvement projects will indeed be implemented and should, therefore, properly be considered a part of the no-build system.

It is evident, then, that a second definition of the no-build system is one composed of all facilities actually in use in the base year plus certain facilities under construction whose completion may reasonably be regarded as irreversible. The advantage of this expanded definition that it provides a basis for identifying system deficiencies by recognizing and incorporating improvement projects likely to be available for use in the short-term future. The disadvantage of this definition lies principally in the difficulty of all concerned in reaching agreement upon those projects currently in various stages of implementation that can be expected to be completed and available for use.

Yet a third definition of the no-build transportation system recognizes that certain political and administrative decisions have been made which, in effect, should commit resources to facility improvements, and that such committed improvements should be added to the transportation network already in use. This was the course of action followed in preparing the firstand second-generation regional transportation system plans.

The principal advantage of the third definition is that it is consistent with official actions and decisions undertaken to commit public resources to transportation facilities, facilities which may have been proposed in a legally adopted longrange plan. Any decision to reverse such a commitment must be made only after the most careful reappraisal if public funds are not to be wasted and great harm done to the public interest originally intended to be served. The principal disadvantage of this alternative is that, while some projects may appear to be committed by administrative and political actions, these projects may not be completed because of litigation or change in administrative or political decisions. This approach to defining the no-build system for the third-generation plan was selected by the Technical Coordinating and Advisory Committee.

Arterial Street and Highway System: Following extended consideration of the alternative ways to define the no-build transportation system, the Technical Coordinating and Advisory Committee recommended that the no-build arterial street and highway system be defined as those arterial facilities completed and opened to traffic as of January 1, 1994, with the addition of those facilities programmed for construction in the annual element, the element applicable to calendar years 1994 and 1995, of the 1993 through 1998 Federally approved <u>Transportation Improvement</u> <u>Program for Southeastern Wisconsin</u> prepared by the Regional Planning Commission.

The arterial street and highway component of the no-build alternative regional transportation system plan is shown on Map 84. The number of miles of existing and committed arterial street and highway facilities in the Region is defined by arterial facility type on a county-by-county basis in Table 174. On a regional basis, the arterial street and highway system would increase from 3,274 miles in 1991 to 3,480 miles, an increment of about 206 miles, or 6 percent. The additional 206 miles consist of three miles of new arterial streets and highways opened to traffic by January 1, 1994; 11 miles of committed new facilities scheduled for completion in 1994 and 1995; and 192 miles of existing nonarterial streets which could be expected to function as arterial streets or highways by 2010.

<u>Transit System</u>: The Technical Coordinating and Advisory Committee recommended that the no-build regional public transit system be defined as the existing intraregional urban transit systems operating within the Region on January 1, 1994, with the addition of those facility and service improvements programmed for implementation in the annual element of the 1993 through 1998 <u>Transportation Improvement</u> Program for Southeastern Wisconsin.

The committed transit improvements consisted primarily of minor route extensions and modifications. The no-build regional transit system is identified on Map 85. As shown in Table 175, rapid transit service within the Region would consist of 519 round trip route miles, representing an increase of 70 miles, or about 16 percent, over the 1991 level. The number of round trip route miles provided by express transit would decrease between 1991 and 2010 by 16 miles, or 4 percent, from 393 miles to 377 miles. This reduction would be the result of the conversion of certain express routes to rapid transit service. Table 175 also indicates that the number of round trip route miles of local service within the Kenosha and Milwaukee urbanized areas would increase by 21 and 68 miles, respectively, and the number would be reduced in the Racine urbanized area by three miles.

Maps 86 through 88 show the local transit service area in the Kenosha, Milwaukee, and Racine urbanized areas, respectively. The maps show the additional areas that would be served under the no-build alternative plan, as well as the areas which would no longer be served by local transit. These minor changes in service area will have relatively insignificant impacts on the number of people served by local transit. The local transit service area is that area located one-quarter mile on either side of a local transit route, the maximum convenient walking distance to a transit stop.

The population served by local transit within urbanized areas may be anticipated to increase, largely because of the planned distribution of future population growth to those areas served by local transit. Through the land use settlement pattern envisioned under the adopted regional land use plan and the minor transit changes envisioned under the no-build plan, the number of persons residing within the Kenosha local transit service area may be expected to decline by about 3 percent, from 86,600 in 1991 to 84,200 by 2010. The proportion of the Kenosha urbanized area population served by transit may be expected to decrease, as well, from 92 percent to 79 percent. The number of persons residing within the Milwaukee local transit service area may be expected to decline from 935,400 in 1991 to 926,500 by 2010. The proportion of the Milwaukee urbanized area population served by local transit may be expected to decrease from 76 percent to 73 percent. The number of people residing within the Racine local transit service area may be expected to decrease by 5 percent, from 114,600 in 1991 to 108,900 by 2010. As within the Kenosha and Milwaukee urbanized areas, the proportion of the population within the Racine urbanized area served by local transit may be expected to decline, from 94 percent to 82 percent.

Transportation System Management: The Technical Coordinating and Advisory Committee recommended that the travel demand management measures considered part of the no-build transportation system be defined as those measures in operation within the Region as of January 1, 1994, with the addition of those measures for which funding has been committed in the annual element of the 1993 through 1998 Transportation Improvement Program for Southeastern Wisconsin. Such measures include, among others, freeway traffic management system implementation and the construction and reconstruction of bikeways.



The no-build street and highway system plan includes those surface arterial streets and highways proposed to be open to traffic during the period from 1991 through 1994, the committed facilities identified in the annual element of the 1993 through 1998 Transportation Improvement Program for Southeastern Wisconsin, and certain existing streets and highways which did not function as arterials in 1991 but which would function as arterials by the year 2010. On a regional basis, the arterial street and highway system would increase from 3,274 miles in 1991 to 3,480 miles by 2010, an increment of about 206 miles, or 6 percent.

ARTERIAL STREET AND HIGHWAY FACILITIES IN THE REGION BY ARTERIAL FACILITY TYPE AND BY COUNTY: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Arterial Streets and Highways						
	Base Ye	ar 1991	Planned	ncrement	20	10	
Arterial Facility Type	Miles	Percent of Total	Miles	Percent Change	Miles	Percent of Total	
Kenosha County						,	
Freeway							
Six-Lane	12.0	3.8			12.0		
Eight-Lane				·			
Subtotal	12.0	3.8			12.0	3.5	
Standard Arterial		_	_				
Two-Lane	256.1	80.6	19.0	7.4	275.1	79.7	
Four-Lane	49.6	15.6	8.5	17.1	58.1	16.8	
Six-Lane							
Subtotal	305.7	96.2	27.5	9.0	333.2	96.5	
	317.7	100.0	27.5	8.7	345.2	100.0	
Milwaukee County							
Four-Lane	13.9	1.8			13.9	1.8	
Six-Lane	53.9	6.9			53.9	6.8	
Eight-Lane	1.4	0.2			1.4	0.2	
Subtotal	69.2	8.9			69.2	8.8	
Standard Arterial							
Two-Lane	379.0	48.9	-1.4	-0.4	377.6	47.8	
Four-Lane	288.9	37.3	12.3	4.3	301.2	38.2	
	706.3	4.3	12.0	7.8	710.1	01.0	
Subtotal	706.2	91.1	13.9	2.0	720.1	91.2	
	//5.4	100.0	13.9	1.8	/89.3	100.0	
Freeway				•		·	
Four-Lane	27.4	9.5			27.4	8.3	
Six-Lane						1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	
Eight-Lane	• •						
Subtotal	27.4	9.5			27.4	9.5	
Standard Arterial	044.0						
Four-Lane	241.3	83.0 69			241.3	83.6	
Six-Lane			·				
Subtotal	261.1	90.5			261.1	90.5	
County Subtotal	288.5	100.0			288.5	100.0	
Racine County Freeway							
Four-Lane						·	
Six-Lane	12.0	3.4			12.0	2.9	
	12.0	3.4			12.0	2.9	
Standard Arterial	301.6	867	57.8	10.2	360 1	90.2	
Four-Lane	31.1	9.0	1.8	5.8	32.9	8 1	
Six-Lane	3.2	0.9			3.2	0.8	
Subtotal	335.9	96.6	59.6	17.7	395.5	97.1	
County Subtotal	347.9	100.0	59.6	17.1	407.5	100.0	

Table 174 (continued)

	Arterial Streets and Highways					
	Base Ye	ar 1991	Planned I	ncrement	20	10
Arterial Facility Type	Miles	Percent of Total	Miles	Percent Change	Miles	Percent of Total
Walworth County Freeway						
Four-Lane	50.0	11.7			50.0	11.2
Six-Lane						
Subtotal	50.0	117			50.0	11.2
Standard Arterial	00.0				30.0	11.2
Two-Lane	370.5	86.3	14.6	3.9	385.1	86.6
Four-Lane	8.7	2.0	1.1	12.6	9.8	2.2
Six-Lane						
	379.2	88.3	15.7	4.1	394.9	88.8
Washington County	429.2	100.0	15.7	3.7	444.9	100.0
Freeway						
Four-Lane	15.1	3.8			15.1	3.4
Six-Lane	6.5	1.6			6.5	1.4
Subtotal	21.6	5.4			21.6	
Standard Arterial	21.0	<u></u>			21.0	7.0
Two-Lane	340.0	85.2	49.5	14.6	389.5	86.8
Four-Lane	37.6	9.4			37.6	8.4
Subtotal	377.6	94.6	49.5	13.1	427.1	95.2
County Subtotal	399.2	100.0	49.5	12.4	448.7	100.0
Waukesha County Freeway						
Four-Lane	45.6	6.3			45.6	6.0
Eight-Lane	13.4					1.8
Subtotal	59.0	8.3			59.0	7.8
Standard Arterial						
I wo-Lane	600.2 52.2	83.8	25.0	4.2	625.2 66 5	82.7
Six-Lane	4.9	0.7			4.9	0.7
Subtotal	657.3	91.8	39.3	6.0	696.6	92.2
County Subtotal	716.3	100.0	39.3	5.5	755.6	100.0
Southeastern Wisconsin Region						
Four-Lane	152.0	4.6			152.0	4.3
Six-Lane	97.8	3.0			97.8	2.8
Light-Lane	1.4	0.1			1.4	0.1
Subtotal	251.2	7.7			251.2	7.2
Two-Lane	2,488.7	76.0	164.5	6.6	2,653.2	76.3
Four-Lane	487.9	14.9	38.0	7.8	525.9	15.1
Six-Lane	46.4	1.4	3.0	6.5	49.4	1.4
Subtotal	3,023.0	92.3	205.5	6.8	3,228.5	92.8
Region Total	3,274.2	100.0	205.5	6.3	3,479.7	100.0



The no-build regional public transit system may be defined as the transit system operating within the Region on January 1, 1994, with the addition of those facility and service improvements programmed for implementation in the annual element of the 1993 through 1998 Transportation Improvement Program for Southeastern Wisconsin. Under the no-build plan, the urbanized area population to be served by public transit would remain stable at about 1.1 million.

	Base Ye	ar 1991	Planned I	ncrement	20	10
Transit Service	Number	Percent of Total	Number	Percent	Number	Percent
			Number	Change	Number	
Round-Trip Route Length (miles)						
Rapid Routes	449	19.6	70	15.6	519	21.3
Express Routes	393	17.1	-16	-4.1	377	15.5
Local Routes						
Kenosha Urbanized Area	171	7.4	21	12.3	192	7.9
Milwaukee Urbanized Area	1,112	48.5	68	6.1	1,180	48.4
Racine Urbanized Area	171	7.4	-3	-1.8	168	6.9
Subtotal	1,454	63.3	86	5.9	1,540	63.2
Total	2,296	100.0	140	6.1	2,436	100.0
Special Facilities (miles)						
Exclusive Right-of-Way	0	0.0	0	0.0	0	0.0
Exclusive Lanes on Streets	2	100.0	0	0.0	2	100.0
Total	2	100.0	0	0.0	2	100.0
Average Weekday Vehicle Requirements ^a						· · · · · · · · · · · · · · · · · · ·
Peak Period	530		-3	-0.6	527	
Midday to Off-Peak Period	285		6	2.1	291	

TRANSIT SYSTEM OPERATING CHARACTERISTICS AND FACILITIES IN THE REGION: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup vehicles.

Source: SEWRPC.

Assignment of Travel to Highway and Transit Networks

As previously indicated, an analysis of the performance of the no-build alternative transportation system plan, under present and probable future travel demand, was the first step in the plan design process. This analysis indicated the location and magnitude of deficiencies in the nobuild transportation system; information essential to the development of alternative system plans, implementation of which could overcome identified deficiencies. The analysis of the performance of the no-build system was undertaken with the aid of the travel simulation models described in Chapter VII of this report.

Personal Vehicle Availability: Given the postulated no-build alternative transportation system plan and land use development in conformance with the adopted regional land use plan, it may be expected that the number of personal vehicles available in the Region will increase by nearly 177,300, or 16 percent, from about 1.13 million in 1991 to about 1.31 million in 2010 (see Table 176). This compares with a forecast increase of approximately 6 percent in resident population, 8 percent in households, and 11 percent in employment. Under the no-build plan, the number of vehicles per person would decrease from 1.6 in 1991 to 1.5 by 2010, and the number of vehicles per household would remain at 1.7.

<u>Person Trip Generation</u>² If fully developed, the land use pattern postulated under the adopted land use plan, together with the transportation

²In deriving future travel demand for all transportation plan alternatives presented in this chapter, the simulation models were applied at the traffic analysis zone level. To present the resulting detailed traffic demand data in conventional report format is impractical. For the purpose of presenting the travel demand data in this report, it was necessary to aggregate the zonal data to obtain regional totals which could be used to present and analyze the major traffic characteristics under each alternative plan at the regional scale. It is important to note that all travel demand figures are estimates of probable future conditions and, while expressed as exact numbers, in reality reflect the variability and uncertainties inherent in any forecasting procedure.



AREA SERVED BY LOCAL FIXED-ROUTE TRANSIT IN THE KENOSHA URBANIZED AREA: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

Map 86 shows the changes in the Kenosha local transit service area which would come about under the no-build alternative transportation system plan. Through the land use settlement pattern envisioned under the adopted regional land use plan and the minor transit changes proposed, the number of persons residing within the Kenosha local transit service area may be expected to decrease by about 3 percent, from 86,600 in 1991 to 84,200 by 2010.



AREA SERVED BY LOCAL FIXED-ROUTE TRANSIT IN THE MILWAUKEE URBANIZED AREA: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

Map 87 shows the changes in the Milwaukee local transit service area which would come about under the no-build alternative transportation system plan. The population of the local transit service area may be expected to decrease by about 1 percent, from 935,400 in 1991 to 926,500 by 2010.



AREA SERVED BY LOCAL FIXED-ROUTE TRANSIT IN THE RACINE URBANIZED AREA: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

Map 88 shows the changes in the Racine local transit service area which would come about under the no-build alternative transportation system plan. The number of people residing within the Racine local transit service area may be expected to decrease by about 5 percent, from 114,600 in 1991 to 108,900 by 2010.

SELECTED SOCIO-ECONOMIC AND TRANSPORTATION CHARACTERISTICS OF THE REGION: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Base Year		Forecast	Change
Characteristic	1991	No-Build	Number	Percent
PopulationHouseholdsEmploymentVehicles AvailableInternal Person Trips	1,810,400	1,911,000	100,600	5.6
	714,900	774,300	59,400	8.3
	990,300	1,095,000	104,700	10.6
	1,132,000	1,309,300	177,300	15.7
	5,541,000	6,116,100	575,200	10.4
Persons per Vehicle	1.6	1.5	-0.1	-6.3
	1.7	1.7	0.0	0.0
	3.1	3.2	0.1	3.2
	7.8	7.9	0.1	1.3

Source: SEWRPC.

Table 177

DISTRIBUTION OF INTERNAL PERSON TRIPS MADE BY HOUSEHOLD RESIDENT IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Internal Person Trips Generated on an Average Weekday							
	Base Year 1991		Forecast I	Forecast Increment		0		
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total		
Home-Based WorkHome-Based ShoppingHome-Based OtherNonhome-BasedSchool	1,302,700 798,000 1,687,300 1,127,400 625,600	23.5 14.4 30.5 20.3 11.3	152,100 88,200 143,500 124,600 66,700	11.7 11.1 8.5 11.1 10.7	1,454,800 886,200 1,830,800 1,252,000 692,300	23.8 14.5 29.9 20.5 11.3		
Total	5,541,000	100.0	575,100	10.4	6,116,100	100.0		

Source: SEWRPC.

system postulated under the no-build transportation system plan, may be expected to generate a total of nearly 6.12 million internal person trips on an average weekday. This represents an increase of about 10 percent over the nearly 5.54 million internal person trips generated within the Region on an average weekday in 1991, as shown in Table 176. The average number of internal person trips generated per capita within the Region on an average weekday may be expected to increase under this alternative from about 3.1 in 1991 to about 3.2 by 2010, while the average number of internal person trips generated per household may be expected to increase from 7.8 to about 7.9. The distribution of internal person trips by trip purpose is set forth in Table 177.

<u>Mode of Travel</u>: The distribution of internal person trips within the Region by mode of travel under the adopted land use plan and the no-build transportation system plan is summarized in Table 178. Average weekday transit trip production may be expected to decline by about 4 percent, from 172,200 trips in 1991 to about 168,000 trips by 2010. Trips made by automobile drivers may be expected to increase by about 16 percent, from 4.06 million in 1991 to 4.72

	Internal Person Trips Generated on an Average Weekday							
	Base Year 1991		Forecast Ir	Forecast Increment		2010		
Mode of Travel	Number	Percent of Total	Number	Percent Change	Number	Percent of Total		
Automobile Driver Automobile Passenger Transit Passenger School Bus Passenger	4,060,900 1,080,300 172,200 227,600	73.3 19.5 3.1 4.1	660,700 -138,900 -4,200 57,500	16.3 -12.9 -2.4 25.3	4,721,600 941,400 168,000 285,100	77.2 15.4 2.7 4.7		
Total	5,541,000	100.0	575,100	10.4	6,116,100	100.0		

DISTRIBUTION OF INTERNAL PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY MODE OF TRAVEL: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

Source: SEWRPC.

million in 2010. The relative utilization of transit and personal vehicles is indicated by trip purpose categories in Tables 179, 180, and 181.

Average weekday internal vehicle trips may be expected to increase, as will person trips (see Table 182). If fully developed, the land use pattern recommended under the adopted land use plan may be expected to generate about 5.3 million internal vehicle trips, including automobile and truck trips, by the year 2010. This represents an increase of about 16 percent over the nearly 4.58 million internal vehicle trips generated on an average weekday in 1991.

System Performance

Allocation of the vehicle travel demand generated under the adopted land use plan to the nobuild alternative transportation system plan indicates that vehicle miles of travel on the arterial street and highway system may be expected to increase from about 33.07 million per average weekday in 1991 to nearly 44.52 million by 2010, an increase of about 35 percent. The most significant increase in vehicle miles of travel may be expected to occur on the regional freeway system, where such travel would increase from about 11.6 million vehicle-miles in 1991 to about 17.3 million vehicle-miles in 2010, an increase of about 49 percent. The anticipated level of vehicle miles of travel on an average weekday under the no-build plan is identified by county and facility type in Table 183. The anticipated performance of the regional transit system under the no-build plan is summarized in Table 184. Under the no-build plan, the number of revenue vehicle-miles of transit service provided on an average weekday would increase by 4 percent, from 63,300 in 1991 to 65,400 in 2010. The number of revenue vehicle-hours of service would also increase, by 8 percent, from 5,220 in 1991 to 5,600 in 2010. Despite the slight increase in transit service provided, annual transit ridership may be expected to decline by about 4 percent, from about 50.2 million passengers in 1991 to about 48.0 million by 2010.

System Deficiencies

The impact of the anticipated increase in travel on the no-build arterial street and highway system is reflected in the number of arterial miles that may be expected to operate over design capacity by the year 2010 (see Table 185 and Map 89). In 1991, about 279 miles, or nearly 9 percent of the 3,274-mile arterial street and highway system, were operating over design capacity.

By 2010, the number of miles operating over design capacity may be expected to increase to about 681, an increase of 401 miles, or about 144 percent. Thus, by 2010, about 20 percent of total arterial mileage may be expected to operate over design capacity. By 2010, the number of miles expected to operate at design capacity may be expected to increase from 264 in 1991 to about 510 by the year 2010. Thus, the proportion of arterial mileage that may be expected to operate at design capacity would increase from 8 percent in 1991 to about 15 percent in 2010.

DISTRIBUTION OF INTERNAL TRANSIT TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Internal Transit Trips Generated on an Average Weekday								
	Base Ye	ar 1991	Forecast	Increment	2010				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Home-Based Work	44,000	25.5	-100	-0.2	43,900	26.1			
Home-Based Shopping	15,800	9.2	-2,400	-15.2	13,400	8.0			
Home-Based Other	28,200	16.4	-2,400	-8.5	25,800	15.4			
Nonhome-Based	11,700	6.8	3,800	32.5	15,500	9.2			
School	72,500	42.1	-3,100	-4.3	69,400	41.3			
Total	172,200	100.0	-4,200	-2.4	168,000	100.0			

Source: SEWRPC.

Table 180

DISTRIBUTION OF INTERNAL AUTOMOBILE PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Internal Automobile Person Trips Generated on an Average Weekday							
	Base Year 1991		Forecast I	ncrement	201	2010		
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total		
Home-Based WorkHome-Based ShoppingHome-Based OtherNonhome-BasedSchool	1,258,700 782,200 1,659,100 1,115,700 325,500	24.5 15.2 32.3 21.7 6.3	152,200 90,600 145,900 120,800 12,300	12.1 11.7 8.8 10.8 3.8	1,410,900 872,800 1,805,000 1,236,500 337,800	24.9 15.4 31.9 21.8 6.0		
Total	5,141,200	100.0	521,800	10.1	5,663,000	100.0		

Source: SEWRPC.

Table 181

DISTRIBUTION OF INTERNAL AUTOMOBILE DRIVER TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Internal Person Trips Generated on an Average Weekday								
	Base Year 1991		Forecast I	ncrement	2010				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Home-Based WorkHome-Based ShoppingHome-Based OtherNonhome-BasedSchool	1,149,500 616,600 1,246,900 928,500 119,400	28.3 15.2 30.7 22.9 2.9	217,000 98,500 162,600 146,100 36,500	18.9 16.0 13.0 15.7 30.6	1,366,500 715,100 1,409,500 1,074,600 155,900	28.9 15.1 29.9 22.8 3.3			
Total	4,060,900	100.0	660,700	16.3	4,721,600	100.0			

	1								
		Total Vehicle Trips Generated on an Average Weekday							
	Base Ye	ar 1991	Forecast	Increment	20	2010			
		Percent		Percent		Percent			
Vehicle Class	Number	of Total	Number	Change	Number	of Total			
Automobile									
Internal	4,060,900	83.0	660,700	16.3	4,721,600	82.4			
External	229,200	4.7	77,100	33.6	306,300	5.3			
Other	39,300	0.8	26,200	66.7	65,500	1.2			
Subtotal	4,329,400	88.5	764,000	17.6	5,093,400	88.9			
Truck									
Internal	520,100	10.6	58,900	11.3	579,000	10.1			
External	44,100	0.9	14,100	32.0	58,200	1.0			
Subtotal	564,200	11.5	73,000	12.9	637,200	11.1			
Total	4,893,600	100.0	837,000	17.1	5,730,600	100.0			

DISTRIBUTION OF TOTAL VEHICLE TRIPS IN THE REGION BY VEHICLE CLASS: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

Source: SEWRPC.

The identification of deficiencies under the no-build transportation system alternative plan and adopted regional land use plan also included the identification of existing and proposed urban residential areas of the Region inadequately served by the highway and public transit modes. The criteria for making this determination consisted of the travel time standards accompanying Objective No. 1 as set forth in Chapter VIII of this report. Table 186 summarizes data on the resident population within the urbanized areas of the Region adequately served by the street and highway system. Table 187 provides similar data for the public transit system.

Table 186 and Map 90 indicate that adequate highway service is nearly ubiquitous throughout the urbanized portions of the Region and may be expected to remain so through the year 2010 under the no-build plan. The percent of total urbanized population in the Region able to meet the employment travel time standard through highway travel may be expected to remain at about 94 percent. The population anticipated to be able to meet the other travel time standards in each of the three urbanized areas is shown on Table 186. Maps showing the location of the resident population expected to meet these other travel time accessibility standards are included in Appendix C of this report. Table 187 indicates that under the no-build plan residents of significant portions of the urbanized area in the Region are unable, and may continue to be unable, to access in a timely manner important employment, shopping, recreational, educational, and health-related activity centers by transit. The urbanized area population able to meet the employment travel time standard through travel on the transit system may be expected to decrease from 131,300 in 1991, or 9 percent of the urbanized area population, to 108,400 in 2010, or 7 percent of the urbanized area population (see Map 91). Maps showing the location of the resident population expected to meet the travel time accessibility standards are included in Appendix C of this report.

When comparing the accessibility provided by transit and automobile modes, it should be noted that nearly one-half of the travel time entailed in making a local transit trip is "out-of-vehicle" time, that is, time spent walking to a transit stop, waiting for a transit vehicle, and walking to the trip destination. Significantly less out-ofvehicle travel time is usually involved in making a trip by automobile. Consequently, more activities can typically be accessed within a given period of time by automobile than by transit.

VEHICLE-MILES OF TRAVEL ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY COUNTY: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

		Arterial Vehicle Miles of Travel on an Average Weekday (thousands)					
	Base Yea	ar 1991	Forecast I	ncrement	20	10	of Increase in Arterial Travel
County	Number	Percent of Total	Number	Percent Change	Number	Percent of Total	1991-2010 (percent)
Kenosha Freeway	675	27.0	410	60.7	1,085	26.6	2.5
Standard Arterial	1,825	73.0	1,170	64.1 63.2	2,995	73.4	2.6
Mihwaukoo	2,300		1,000	03.2	4,000	100.0	2.0
Freeway	5,945 8,446	41.3 58.7	1,849 859	31.1 10.2	7,794 9,305	45.6 54.4	1.4
Subtotal	14,391	100.0	2,708	18.8	17,099	100.0	0.9
Ozaukee		_				· .	:
Freeway	762	39.2	280	36.7	1,042	44.5	1.7
Standard Arterial	1,180	60.8	118	10.0	1,298	55.5	0.5
Subtotal	1,942	100.0	398	20.5	2,340	100.0	1.0
Racine	708	23.0	5/1	76 /	1 2/0	20.8	20
Standard Arterial	2,258	76.1	687	30.4	2,945	70.2	1.4
Subtotal	2,966	100.0	1,228	41.4	4,194	100.0	1.8
Walworth							
Freeway	540	28.2	454	84.1	994	37.8	3.3
Standard Arterial	1,373	71.8	260	18,9	1,633	62.2	0.9
Subtotal	1,913	100.0	714	37.3	2,627	100.0	1.7
Washington	-		0.07				
Freeway	1 833	23.0	267	48.9	813 2414	25.2	2.1
Subtotal	2,379	100.0	848	35.6	3 2 2 7	100.0	1.5
Waukasha	2,070			00.0	<u> </u>		1.0
Freeway	2,421	34.7	1.896	78.3	4,317	39.4	3.1
Standard Arterial	4,560	65.3	2,078	45.6	6,638	60.6	2.0
Subtotal	6,981	100.0	3,974	56.9	10,955	100.0	2.4
Southeastern Wisconsin Region							
Freeway	11,597	35.1	5,697	49.1	17,294	38.8	2.1
Standard Arterial	21,475	64.9	5,753	26.8	27,228	61.2	1.3
Total	33,072	100.0	11,450	34.6	44,522	100.0	1.6

TRANSIT SYSTEM	PERFORMANCE I	N THE REGION: 1991
AND 2010 NO-BUI	LD TRANSPORTA	TION SYSTEM PLAN

			· · · · ·	
			Forecast In	ncrement
	Base Year			Percent
Transit System Characteristics	1991	2010	Number	Change
Service Provided, Average Weekday				
Revenue Vehicle-Miles				
Rapid	3,400	3,100	-300	-8.8
Express	3,300	6,000	2,700	81.8
Local	56,600	56,800	200	0.4
Total	63,300	65,900	2,600	4.1
Revenue Vehicle-Hours		· · ·		
Rapid	170	160	-10	-5.9
Express	170	360	190	111.8
Local	4,880	5,130	250	5.1
Total	5,220	5,650	430	8.2
Seat-Miles	2,975,000	3,028,300	53,300	1.8
Service Utilization				
Ridership				
Average Weekday Revenue Passengers	172,200	168,000	-4,200	-2.4
Annual Revenue Passengers	50,222,900	48,998,000	-1,224,900	-2.4
Revenue Passengers per Revenue Vehicle-Hour	33.0	29.7	-3.3	-9.9
Average Weekday Passenger Miles	609,100	728,800	119,700	19.7

Source: SEWRPC.

CONSIDERATION OF TRAVEL DEMAND MANAGEMENT

Travel demand management refers to a series of measures or strategies which are intended to reduce or eliminate personal and vehicular travel or to shift such travel to alternative times and routes. The general intent of such measures, in combination with public transit and arterial street and highway improvements, is to reduce traffic congestion, air pollution, and fuel consumption and, importantly, to assist in implementing the adopted regional land use plan. Travel demand management, in particular, implicitly emphasizes reducing the rate of urban decentralization and encouraging the development of more compact and interconnected neighborhoods, as recommended in the adopted year 2010 regional land use plan.

The achievement of the regional transportation system development objectives through the use of demand management depends on four factors: 1) the degree to which the appropriate variables influencing travel demand are identified for management, 2) the degree to which travel demand management measures are combined into coherent packages, 3) the degree to which travel demand management measures are integrated with public transit and arterial street and highway improvements, and 4) the extent to which demand management measures are practicable and implementable.

Many variables influence individual travel behavior; these have been described in detail in previous chapters of this report. These variables can generally be divided into three categories: the characteristics of the transportation system, the characteristics of the travelers, and the physical and temporal distribution of human activities as reflected in the land use pattern. In designing travel demand management measures it is essential that only those variables amenable to control, in the context of the Southeastern Wisconsin Region, be identified for manage-

ARTERIAL MILEAGE IN THE REGION OPERATING UNDER, AT, AND OVER DESIGN CAPACITY: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Base Year 1991							
	Under Desig	n Capacity	At Desig	At Design Capacity		Over Design Capacity		
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles	
Kenosha Milwaukee Ozaukee Racine Walworth Washington	286.0 543.2 254.2 288.7 411.1 356.5	90.0 70.1 88.1 83.0 95.8 89.3	16.6 110.3 21.0 24.9 15.0 23.2	5.2 14.2 7.3 7.2 3.5 5.8	15.1 121.9 13.3 34.3 3.1 19.5	4.8 15.7 4.6 9.9 0.7 4.9	317.7 775.4 288.5 347.9 429.2 399.2	
Total	591.3 2,731.0	82.5	264.0	8.1	279.2	8.5	3,274.2	

	Year 2010 No-Build								
	Under Design Capacity		At Desig	At Design Capacity		Over Design Capacity			
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles		
Kenosha	230.1	66.7	74.2	21.5	40.9	11.8	345.2		
Milwaukee	490.0	62.1	122.6	15.6	176.7	22.3	789.3		
Ozaukee	233.9	81.1	26.3	9.1	28.3	9.8	288.5		
Racine	241.7	59.3	80.5	19.8	85.3	20.9	407.5		
Walworth	387.2	87.0	40.2	9.0	17.5	4.0	444.9		
Washington	337.7	75.3	54.1	12.1	56.9	12.6	448.7		
Waukesha	369.1	48.8	111.6	14.7	274.9	36.5	755.6		
Total	2,289.7	65.8	509.5	14.6	680.5	19.6	3,479.7		

Source: SEWRPC.

ment. Household income, for example, is a variable characteristic of travelers which influences individual travel behavior and demand. Household income, however, is not practicably amenable to control by measures that seek to change travel behavior. Measures that seeks to manage other variables that influence travel demand, such as travel time or costs, may prove more practicable.

The second factor crucial to the successful design and implementation of travel demand management is the degree to which measures are combined into coherent, and possibly synergistic, packages. The impact on highway use and traffic congestion may be minimal, for instance, if only a single measure, such as congestion pricing, is implemented. Should such a single measure be advanced in the absence of complementary strategies, commuters may not make intended shifts to alternative modes but, instead, may simply select alternative highway routes, shifting congestion to other routes without impacting vehicle-miles traveled. Combining measures may enhance the chances of achieving intended results. The impact of travel demand management is enhanced when a variety of coordinated measures are considered.

The third factor to be considered in the design of travel demand management measures is the degree to which such measures are integrated

OPERATING UNDER, AT, AND OVER DESIGN CAPACITY: 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN



The traffic congestion impacts of anticipated increases in travel on the no-build arterial street and highway system are shown on the above map. By the year 2010, the number of miles operating over design capacity may be expected to increase to about 681, an increase of 401 miles, or about 144 percent. Thus, by the year 2010, about 20 percent of total arterial mileage may be expected to operate over design capacity.

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON ARTERIAL STREETS AND HIGHWAYS: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

Urbanized Area	Urbanized Area Population		Base Year 1991		No-Build Transportation System Plan: 2010	
and Activity Center	Base Year 1991	2010	Number	Percent	Number	Percent
Kenosha Urbanized Area	94,300	107,300				
Employment-Related ^a			90,900	96.4	107,300	100.0
Major Retail-Service ^b			0	0.0	0	0.0
Medical Facility ^C			94,300	100.0	107,300	100.0
Major Park ^d			94,300	100.0	107,300	100.0
Higher-Education Facility ^e			94,300	100.0	107,300	100.0
Scheduled Air Transport ^T			94,300	100.0	107,300	100.0
Milwaukee Urbanized Area	1,226,300	1,277,100				
Employment-Related			1,147,900	93.6	1,180,500	92.4
Major Retail-Service			1,161,400	94.7	1,277,100	100.0
Medical Facility			1,226,300	100.0	1,277,100	100.0
Major Park			1,226,300	100.0	1,277,100	100.0
Higher-Education Facility			1,226,300	100.0	1,277,100	100.0
Scheduled Air Transport			1,226,300	100.0	1,277,100	100.0
Racine Urbanized Area	121,800	132,100				
Employment-Related			121,800	100.0	132,100	100.0
Major Retail-Service			32,200	26.4	132,100	100.0
Medical Facility		- -	121,800	100.0	132,100	100.0
Major Park			121,800	100.0	132,100	100.0
Higher-Education Facility			121,800	100.0	132,100	100.0
Scheduled Air Transport			121,800	100.0	132,100	100.0

^aStandard: 30 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of three major retail and service centers.

^cStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

Source: SEWRPC.

with public transit and arterial street and highway improvements. Because travel demand management consists largely of measures intended to affect individual travel behavior, alternative means of transportation must be available that can effectively and efficiently serve the altered behavioral patterns. Where such alternatives are not provided, demand management may be perceived simply as an attempt to usurp personal freedom and to restrict mobility.

Finally, the importance of setting must be understood. Any demand management measure proposed must be practicable, technically and politically, within the complex governmental setting presented by a large metropolitan region.

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON TRANSIT: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLANS

Urbanized Area	Urbanized Area Population		Base Yea	ar 1991	No-Build Transportation System Plan: 2010	
and Activity Center	Base Year 1991	2010	Number	Percent	Number	Percent
Kenosha Urbanized Area	94,300	107,300				
Employment-Related ^a			58,000	61.5	51,400	47.9
Major Retail-Service ^b			0	0.0	1,400	1.3
Medical Facility ^C			60,600	64.3	53,300	49.7
Major Park ^d			12,600	13.4	20,700	19.3
Higher-Education Facility ^e			33,200	35.2	51,400	47.9
Scheduled Air Transport [†]			0	0.0	0	0.0
Milwaukee Urbanized Area	1,226,300	1,277,100				
Employment-Related			14,200	1.2	600	0.0
Major Retail-Service ^g			10,100	0.8	35,300	2.8
Medical Facility			700,400	57.1	648,000	50.7
Major Park			636,300	51.9	617,300	48.3
Higher-Education Facility			775,700	63.2	740,900	58.0
Scheduled Air Transport			338,600	27.6	316,800	24.8
Racine Urbanized Area	121,800	132,100				
Employment-Related			59,100	48.5	56,400	42.7
Major Retail-Service			21,500	17.7	18,300	13.9
Medical Facility			53,700	44.1	46,700	35.4
Major Park			24,000	19.7	37,000	28.0
Higher-Education Facility			79,600	65.4	88,400	66.9
Scheduled Air Transport			18,300	15.0	18,600	14.1

^aStandard: 45 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of one major retail and service center.

^cStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

^gStandard: 35 minutes' overall travel time of three major retail and service centers.

Source: SEWRPC.

An attempt to promote travel demand management measures that are perceived as highly impractical will have negative value.

Table 188 sets forth a description of some of the various travel demand management measures and a qualitative assessment of the potential impacts of these measures, in isolation, on vehicle trips and vehicle miles traveled. The table also assesses the current potential for such measures to be implemented and the actions necessary for implementation. The measures considered in Table 188 fall under the following five headings: pricing, parking control, land use, work-schedule changes, and other measures. It was assumed in qualitatively estimating the

AREAS MEETING THE TRAVEL TIME STANDARDS FOR EMPLOYMENT OPPORTUNITIES BY ARTERIAL STREET AND HIGHWAY: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN



In terms of timely access to employment opportunities, adequate highway service is nearly ubiquitous throughout the urbanized area of the Region and may be expected generally to remain so through the year 2010 under the no-build alternative. The percent of the total urbanized area population in the Region able to access 40 percent of urbanized area jobs within 30 minutes through arterial travel may be expected to remain at about 94 percent.

AREAS MEETING TRAVEL TIME STANDARDS FOR EMPLOYMENT OPPORTUNITIES THROUGH TRAVEL ON TRANSIT: 1991 AND 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN



EXISTING 1991

2010 NO-BUILD TRANSPORTATION SYSTEM PLAN



Generally, public transit service currently is, and may be expected to remain, inadequate in providing timely access to employment opportunities within the Region. The urbanized area population able to access 40 percent of urbanized area jobs within 45 minutes by transit may be expected to decrease from 131,300 in 1991, or 9 percent, to 108,400 in 2010, or 7 percent.

TRANSPORTATION SYSTEM DEMAND MANAGEMENT MEASURES POTENTIALLY APPLICABLE TO THE SOUTHEASTERN WISCONSIN REGION

Dema	and Management		Potential Impact On		
Category	Specific Measure	Description	Vehicle Trips and Vehicle-Miles Traveled	Actions Required to Implement	Probability of Implementation
Pricing	Cash-out of employer- paid parking	Require employers offering free/subsidized parking to employees to give cash value of parking to those employees and to charge those employees the market value of parking. Such a program would apply initially to leased parking and new employer-owned parking. (Such a program could apply to schools and universities)	Potential reduction in vehicle trips through increased use of transit and ridesharing on theory that some employees will "pocket" cash payment and use other mode of travel	Employer coopera- tion, Federal legislation, and State legislation	Medium; potential resistance by employers
	Pay-as-you-drive auto insurance	Offset of automobile insurance premiums by per-gallon insurance fee collected at the pump. Recognize that portion of risk is related to vehicle use	Potential reduction in vehicle trips and vehicle-miles of travel	State legislation	Low; high degree of resistance by insurance companies and retail fuel dealers; potential to purchase cheaper fuel in nonparticipating adjacent jurisdictions
	Public sector parking pricing	Local and county governments can alter the price of off- street parking to discourage long-term commuter-related parking and encourage use of other modes	Limited potential reduction in single-occupancy work- related vehicle trips	Local and county policies	Low; does not affect cost of parking provided in private sector; potential adverse effect on land use
	Parking excise tax	A State or local parking excise tax would be imposed on all parking spaces. The tax could be increased incrementally and revenues could be used to defray the cost of transportation services	Potential reduction in vehicle trips and vehicle-miles of travel if fees encourage a reduction in nonwork vehicle trips	State legislation, county and local policies	Low; potential high degree of resistance from the general public
	Parking pricing for nonwork-related destinations	A charge would be levied through use of Intelligent Vehicle Highway System (IVHS)-related electronic "smart cards." Low-cost sensors which interact with in-vehicle smart cards would be installed on the public right-of-way which abuts driveways to shopping and other activity centers to enable the imposition of modest trip destination charges. Charges could vary depending on location and time of day	Potential reduction in vehicle- miles of travel and nonwork vehicle trips, particularly if the policy is implemented uniformly on a regionwide basis	State, county and local policies	Low; potential high degree of resistance from the general public
	Road pricing	Users of designated highway segments charged a fee; fees may vary according to time of day and congestion levels	Potential reduction in vehicle trips and vehicle-miles of travel if alternative routes are not attractive and if more distant land use destinations are not available	State legislation	Low; high degree of resistance from general public; potential adverse effect on land use
	Area pricing	Personal vehicles entering designated areas would be required to purchase a permit. Permit fees may be varied by time of day, ambient air quality conditions, and traffic congestion levels	Potential for significant reductions in vehicle trips to and within designated areas	State legislation	Low; enforcement difficulty; potential adverse effects on land use

Table 188 (continued)

Dema	nd Management	.	Potential Impact On	Actions Based	Beckshiller
Category	Specific Measure	Description	Venicle Trips and Vehicle-Miles Traveled	to Implement	Implementation
Pricing (continued)	Graduated registration fees	A graduated registration fee related to annual miles traveled and vehicle emission production	Potential reduction in trip- making and vehicle-miles of travel	State legislation	Medium
	Motor fuel tax	Per gallon tax	Potential reduction in trip- making and vehicle-miles of travel	Federal or State legislation	Medium
	Employee commuter option (ECO) pass systems	The ECO Pass is a program allowing employers and institutions to offer low-cost annual bus passes as a benefit to their employees and students. The pass would also include a guaranteed ride home in the event an	Potential reduction in trip- making and vehicle-miles of travel as the cost of transit is reduced	Federal or State legislation	High
		employee works late or encounters a personal emergency. The cost of the program varies by the employer's location, number of employees, and level of transit service			· · · · · · · · · · · · · · · · · · ·
Parking Control Measures	Parking management; control of parking supply	The use of local zoning regu- lations to limit the number of parking spaces provided for new development; the number of spaces provided may be varied depending on the location of the development and alternative transportation options provided to the property. (Parking supply restrictions can also be applied to schools and universities)	Potential reduction in vehicle trips	Local ordinance	Low; potential adverse effects on land use
	Preferential parking for carpools and vanpools	Reservation of most conveni- ent parking spaces for carpool and vanpool vehicles	Potential reduction in vehicle trips by encouraging ridesharing	Employer cooperation	High
	Major activity center fringe park and ride parking	Parking facilities constructed in locations remote to major activity centers to promote use of other modes	Potential reduction in vehicle- miles of travel	State, county, and local governments policy	High
Land Use Measures	Major activity center and neighborhood site planning	Use of zoning, official mapping, subdivision control, site plan review, and permitting to achieve appropriate urban design that reduces dependence on automobile; provide good circulation systems for bicyclists and pedestrians and associated amenities, a mix of land use activities, higher residential densities, public transit access, and traffic calming	Potential reduction in vehicle trips and vehicle-miles of travel	Revision of land use regulations, State legislation, changes in funding to provide financial incentives	Medium
	Growth management	The use of land use regulation to direct the extent, timing, rate, and location of new urban development in accord with the adopted regional land use plan	Potential reduction in vehicle trips and vehicle-miles of travel	Adoption of growth management policy, revision of land use regulations, State regulation or law, changes in funding to provide financial incentives	Medium

Table 188 (continued)

Demai	nd Management		Potential Impact On		
Category	Specific Measure	Description	Vehicle Trips and Vehicle-Miles Traveled	Actions Required to Implement	Probability of Implementation
Work Schedule Charges	Telecommuting	Employees work at remote locations and avoid trip to central office. The tele- commuting office could be at home, at a satellite work center, or a neighborhood work center	Potential reduction in vehicle trips and vehicle-miles of travel	Employer cooperation	Medium
	Flextime and compressed work week	Flextime allows employees to set work starting and ending times to avoid peak-period congestion and to coordinate with transit scheduling. A compressed work week allows employees to work fewer days per week	Potential reduction in vehicle trips made during peak hours; may increase off-peak vehicle trips and vehicle-miles of travel	Employer cooperation	Medium
Other	Trip-reduction ordinance	Public land use regulations promoting employer efforts to reduce the number of single occupancy vehicle trips made by employees; ordinances may permit increased densities in return for measures to reduce vehicle tripmaking	Potential reduction in trip- making and vehicle-miles of travel	Local ordinance	Medium
	Areawide rideshare	Public coordination of private efforts to encourage ridesharing	Potential reduction in vehicle trips and vehicle-miles of travel	State, county, and local governments	High
	Transportation management associations	Promotion of geographically related transportation demand management measures; associations may be policy- and/or service-oriented	Potential reduction in vehicle trips and vehicle-miles of travel	Employer cooperation	High
	Education/marketing	Measures to heighten public interest and knowledge of the Region's transportation and air quality problems to promote responsible trip- making and reduce single occupancy travel. Possible measures include, but are not limited to, expanded driver education, increased driver license educational requirements, and regular public service announcements	Potential reduction in vehicle trips and vehicle-miles of travel	State, county, and local government	High
	IVHS advanced transit information systems	IVHS to provide transit and paratransit information regarding departures, arrivals and specialized services designed to help system users better structure their time and reduce waiting time	Limited potential reduction in vehicle trips and vehicle- miles of travel	State, county, and local policies	Medium

Source: SEWRPC.

potential impacts of the demand management measures that all measures which increase the cost of vehicle operation may be expected to reduce the number of vehicle-miles traveled, but not necessarily the number of vehicle trips.

Pricing Measures

As indicated in Table 188, there are a number of pricing measures which have a potential to reduce vehicular travel, including "cash-out" of employer-paid parking, pay-as-you-drive automobile insurance, road pricing, area pricing, parking pricing, graduated registration fees, fuel taxes, and Employee Commute Option (ECO) pass programs. Collectively, such pricing measures have the potential to reduce vehicular travel demand on par with major transit improvements. Placing more of the full costs of construction, maintenance, and operation of transportation facilities and services directly on the user of the system at the time of use induces the user to more closely evaluate the utility achieved in travel choices and to adjust behavior accordingly. Congestion pricing, in particular, takes the assignment of costs one step further by imposing on the users of congested facilities those costs currently shifted to others.

Experience with congestion pricing in Europe and Asia, identified as road and area pricing in Table 188, has shown that reductions in vehicle miles traveled, traffic congestion, and motor vehicle emissions may be effected. Issues remaining to be addressed with respect to congestion pricing in the United States include legal impediments, feasibility and acceptance, and enforcement. However, recent technological innovations have increasingly made enforcement more feasible. Congestion pricing measures have not been implemented to date anywhere in the United States.

Other measures that directly increase the cost of vehicle operation can affect travel behavior. Commission analysis of the utilization of carpool parking and public transit ridership has indicated that both carpooling and transit use increased during the late 1970s and early 1980s as motor fuel prices increased. Indeed, Commission findings indicated that the major factor contributing to the transit ridership increases of 1979 and 1980 was the increase in gasoline prices.³ Conversely, the decline in gasoline prices since 1981 has contributed to reductions in transit ridership. Since the cost of travel significantly influences travel behavior and is within the purview of governmental action to affect, the cost of travel constitutes a viable variable to consider.

Parking Measures

Table 188 identifies three parking management measures: control of parking supply, preferential parking for high occupancy vehicles, and central business district (CBD)-fringe park and ride facilities. Vehicle parking management has long been recognized by the Commission as a measure capable of reducing work related vehicle travel and increasing transit ridership. The second-generation regional transportation system plan adopted by the Commission in 1978 recommended that the City of Milwaukee evaluate the effects of changing the parking rate structure in the CBD to eliminate discounts for long-term parking. The parking measures listed in Table 174 focus on managing the amount of parking provided through the use of land use regulation and the provision of preferential parking for high-occupancy vehicles.

Land Use Measures

Table 188 identifies land use-related travel demand management measures intended to reduce vehicle miles of travel, vehicle trips, and average trip lengths by reducing the need for vehicular travel and by enhancing the attractiveness of alternatives to the use of a personal vehicle. Since land use development practices significantly influence travel behavior and fall within the legal purview of local governmental police power, land use development is a variable that should be managed to achieve transportation system development objectives.

The land use-related measures set forth in Table 188 are divided into two categories: growth management and land use site design. Growth management refers to the use of land use regulation to direct the type, extent, location, timing, and rate of urban development within the Region. In its most basic sense, within the context of the regional transportation planning effort, growth management constitutes a policy framework designed to facilitate the implementation of the adopted regional land use plan. Within the Region, appropriate growth management policy would encourage urban development to occur only in areas suitable and planned for such development, within sanitary sewer and public transit service areas in particular, and only when necessary supporting facilities and services were in place.

The second category, land use site design, emphasizes the enforcement of local land development regulations, within existing and planned

³See SEWRPC, Memorandum Report No. 7, <u>Public Transit Ridership Trends in Southeastern</u> <u>Wisconsin: 1975-1986, May 1987.</u>

public transit service areas in particular, to encourage mixed-use urban development, integrated pedestrian and bicycle circulation systems, and to promote residential densities that permit the efficient provision of urban services such as public transit. Such measures are intended to promote urban settlement patterns that are amenable to the provision of public transit service, especially within existing and planned transit corridors. Conventional land development practices and regulations within much of the Region generally have promoted large, single-use office parks, shopping centers, and residential subdivisions well served only by highway transportation. Travel to, and within, such development is, as a practical matter, limited to automobiles and trucks.

Work Schedule Changes

As shown in Table 188, some changes in work scheduling, such as telecommuting, flextime, and compressed work weeks, have limited potential for managing travel demand. Such changes are becoming more feasible with certain technological and organizational changes. Such advances in telecommunications as fiber optics and computers, structural changes in the economy, and increasing employee demands for quality-oflife benefits are making the typical fiveday-weekly commute unnecessary for some employees and are redefining managerialemployee relationships. Accordingly, changes in work schedules and in the travel behavior related to work scheduling may be expected to occur regardless of governmental action, and may be expected to impact travel demand. Reducing work tripmaking, however, does not necessarily result in a reduction in total tripmaking by an individual. Shifting to a four-day work week for example, will reduce by 20 percent the number of work trips. That reduction, however, may be more than offset by increased nonwork tripmaking on the "extra" day off.

The Federal Clean Air Act Amendments of 1990 mandate that, for the six severe ozone nonattainment counties in the Region, Employer Commute Option programs be designed and implemented by businesses employing 100 or more workers at a single site. This provision is likely to promote the use of changes in work schedules, including telecommuting. Because of the potentially favorable transportation impacts of these measures and the minimal costs imposed upon employees, these measures were considered for inclusion in the alternative plans.

Other Measures

The travel demand management measures listed in Table 188 under the heading "other" include: trip reduction ordinances, areawide rideshare programs, transportation management associations, and education and marketing campaigns. These measures are largely publicly sponsored efforts to coordinate private-sector activities intended to reduce travel demand. Transportation control measures which are likely to assist in reducing air pollution, but which are not expected to impact travel demand, such as the use of alternative-fuel vehicles, are being considered by the Wisconsin Department of Natural Resources as part of its preparation of the State Implementation Plan for Air Quality.

ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

With an understanding of the probable magnitude and distribution of future travel demand within the Region and of the probable impact of that demand on the existing transportation system, as made possible by analyses of the assignment of traffic to the no-build arterial street and highway and transit networks, alternative transportation system plans were postulated to overcome system deficiencies and to serve the adopted regional land use plan.

The process initially included the development and analyses of three alternative transportation system management plans: Alternative Plan 1, Alternative Plan 2, and Alternative Plan 3. Each plan consisted mainly of travel demand management measures, transit improvements, and traffic management measures.

Each of the alternative transportation system management plans contained several major elements in common: comprehensive freeway traffic management; traffic management; promotion of areawide carpooling and work schedule changes, including telecommuting; the recognition of the eventual potential to apply intelligent vehicle highway systems technology to the management of travel in the Region; a focus on land use planning and site design; the use of measures to increase the perceived cost of driving an automobile or light truck; and improvement in public transit service. These measures are intended to ensure that maximum use is made of existing transportation facilities before commitments are made to new capital

investment. The implementation of such measures would have the effect of encouraging use of high-occupancy vehicles; reducing vehicle use, especially within congested areas and during peak travel periods; reducing motor fuel consumption; and facilitating efficient traffic flow.

The three alternative transportation system management plans differ, however, in the degree to which pricing measures and land use planning and site design would influence travel demand. The plans also differ in the degree to which the existing public transit system in the Region would be improved and expanded. The traffic management measures included in each of the alternative plans are described below, then the land use planning and site design recommendations and pricing measures of each plan are described, and finally, the public transit recommendations of each plan are described.

Comprehensive Freeway Traffic Management

Each of the alternative third-generation transportation system management plans includes the freeway traffic management plan for the Greater Milwaukee Area prepared by the Commission in 1988. The freeway traffic management plan is composed of recommendations concerning incident management, advisory information, traffic monitoring and control, and freeway operational control. The proposed freeway traffic management system is intended to encourage the use of carpools and transit vehicles, reduce congestion owing to accidents and disabled vehicles, encourage motorists to make better route decisions through advance information signing, and reduce peak freeway travel demand by managing access to the freeway system.

The 1988 freeway traffic management plan called for a greatly expanded system of on-ramp meters and corresponding traffic detectors to manage freeway access throughout the Milwaukee area, preferential access by buses and other high-occupancy vehicles, provision of driver advisory information signs, and the establishment of a central control center to monitor, on a real-time basis, traffic volumes and flow conditions and to adjust access accordingly to achieve reasonable operating conditions.

Traffic Management

ized traffic signalization, the provision of rightand left-hand turn lanes at major intersections, and the proper spacing of driveway and local street intersections. Arrangements to facilitate pedestrian and bicycle movements and transit use must be considered in all traffic engineering activities and in the design of new or widened arterial streets and highways.

Intelligent Vehicle Highway Systems (IVHS)⁴ The aforereferenced traffic management measures are currently applicable within the Region. It should be realized, however, that by the year 2010 newer elements of Intelligent Vehicle Highway Systems technology currently under development may become applicable within the Region. Indeed, aspects of advanced traffic management systems (ATMS) technology are currently being applied by the Wisconsin Department of Transportation as part of the Department efforts to implement the freeway traffic management system recommended in the second-generation regional transportation system plan. Other elements of IVHS that may have future application within the Region include: advanced travel information systems (ATIS), providing travelers with real-time information on route-specific traffic conditions; advanced public transportation systems (APTS), facilitating transit use by providing, among other things, real-time transit vehicle location and scheduling information to transit users; commercial vehicle operations (CVO), technology automating commercial vehicle licensing, registrations, and toll collection; advanced rural transportation systems (ARTS), providing up-todate weather and road condition reports to rural highway users; and advanced vehicle control systems (AVCS) and in-vehicle technology that improve the safety and efficiency of automobile travel. The potential impacts of these and other

⁴The Commission, beginning in the early 1970s, provided technical assistance on a study of the application of technology that would permit vehicles to be operated both fully automatically on designated guideways and manually over the conventional street system. The <u>Dual Mode</u> <u>Planning Case Study</u> was funded by the United States Department of Transportation, Urban Mass Transit Administration, and executed by Cambridge Systematics, Inc., with Barton-Aschman Associates, August 1977.

The traffic management recommendations emphasize the use of traffic engineering practices to assist in achieving efficient traffic flow. Each of the alternative plans includes synchron-

technological innovations cannot be estimated at this time.

Land Use Planning and Site Design Measures Each of the three alternative plans proposes that new urban development occur in neighborhood units with a mix of land uses to promote travel by transit, to facilitate bicycle and pedestrian travel, and to provide for shorter automobile tripmaking.

Alternative Plan 1 places a much greater emphasis on land use planning and site design as a measure to bring about reductions in travel demand than Alternative Plans 2 and 3. Under Alternative Plan 1, as new neighborhoods are developed and as older neighborhoods are redeveloped, detailed land use plans would be prepared and implemented that would seek to orient, within the overall density framework of the regional land use plan, higher-density residential development and employment locations along transit lines and around transit stations. In the design of Alternative Plan 1, transit walk-access times were reduced by onethird to reflect the increase in the convenience of transit travel made possible through the use of higher residential densities along transit lines. This reduction was incorporated in the mode choice model and consequently indicated that higher levels of transit ridership could be achieved under Alternative Plan 1 than under the other alternatives. Also, in the design of Alternative Plan 1, the land use planning and site design measures were assumed to reduce total trip generation by both automobile and transit by 5 percent and the trip generation model was accordingly adopted.

Pricing Measures

Commission travel simulation models indicated that travel behavior is, in part, a function of the costs perceived by travelers to be directly associated with daily vehicle operation, specifically fuel and parking costs. In 1991, the perceived cost of operating or driving an automobile, as determined by Commission analyses, was about \$0.055 per mile, and approximated the combined effect of the 1991 motor fuel price of \$1.10 per gallon and average motor fuel efficiency of 20 miles per gallon. It should be noted that this perceived cost of driving an automobile is substantially less than the true total cost of owning and operating an automobile, which includes depreciation, insurance, garaging, repairs, fees, and maintenance. These costs are

generally fixed, or upfront, costs that do not vary significantly with the amount of driving.

Only fuel and parking costs are "perceived" at the time of driving. In 1991, the true cost approximated \$0.28 per mile. It is the perceived cost, that is, the cost that varies directly with driving, of \$0.055 per mile, which is assumed to be increased under the transportation system management plans.

Alternative Plan 1 proposes to increase the perceived cost of driving substantially as a measure to moderate vehicle travel demand and make more efficient use of the arterial street and highway system. Under Alternative Plan 1, the perceived cost of driving would increase by 100 percent, from \$0.055 per mile to \$0.11 per mile. This increase assumes no change in the average motor fuel efficiency of the vehicle fleet. For perspective, should this entire cost increase be achieved through use of the motor fuel tax, the addition to that tax would approximate \$1.10 per gallon.

Under Alternative Plan 2, the perceived costs of driving an automobile would be increased by 50 percent, from \$0.055 per mile to \$0.0825 per mile. If the full cost increase were to be achieved through the use of the motor fuel tax, the addition to that tax would approximate \$0.55 per gallon.

Unlike the other alternatives, Alternative Plan 3 would not increase the perceived costs of automobile operation to offset travel demand. Rather, this alternative would use pricing measures, increasing the perceived cost per mile of driving only enough to cover revenue shortfalls incurred in implementing the plan.

One way to ameliorate the impacts of the cost increases would be to use the revenues generated by the measures to replace those revenues generated through property taxes. Hence, both Alternative Plans 1 and 2 propose to shift the local share of funding for the construction, maintenance, and operation of arterial streets and highways and public transit away from the property tax to transportation system user fees. The extensive package of pricing measures under Alternative Plan 1 may be expected to yield approximately \$512 million per year in additional revenue. The moderate package of pricing measures under Alternative Plan 2 may be expected to yield approximately \$102 million per year.

The level of pricing proposed under Alternative Plan 2 would be sufficient to permit removal of local transit and arterial funding from the local property tax and permit funding of full implementation of the second-generation regional transportation system plan. Of the measures listed in Table 188, those applied could be a motor fuel tax or a graduated registration fee, or some combination thereof. The level of pricing under Alternative Plan 1 would be sufficient to accomplish the above, as well as to permit the removal of some nonarterial transportation costs from the property tax.

To ensure that individual travelers retain the capacity to adjust to the pricing measures, a series of short-term and longer-term complementary nonpricing travel demand management measures are included in Alternative Plans 1 and 2. Short-term measures include the provision of preferential parking at reduced rates for highoccupancy vehicles in parking facilities serving work trips; the immediate use of public service announcements, marketing, and educational campaigns which would promote the changes in travel behavior intended through the use of the measures; and the public transit improvements described in the next section of this chapter. Longer-term measures of this type would include promoting telecommunications and sound local land use planning and site design.

Public Transit Improvement and Expansion

As previously indicated, the three alternative transportation system management plans differ in the degree to which the existing public transit system is proposed to be improved and expanded. The transit elements of each plan, however, were designed to meet, to the extent practicable, the transportation system planning development objectives and design standards, listed in Chapter VIII, including the desired levels of accessibility to be provided by transit, cost-effectiveness of service, frequency of service, and location and spacing of routes, stops, and stations.

Alternative Plan 1 emphasized the full achievement of the accessibility standards, relative to the cost-effectiveness standards. It proposed a substantial expansion of rapid, express, and local transit service and a significant improvement in the frequency of existing transit service (see Map 92). These extensions of service and improvements in frequency of service may exceed what is warranted by potential transit ridership. This transit alternative was thus similar to the transit element of the adopted year 2000 regional transportation system plan.

As shown in Table 189, the transit system envisioned under Alternative Plan 1 would consist of 4,190 round-trip route miles, some 1,894 miles, or 82 percent, greater than that provided in 1991. The transit system would require 914 vehicles operating during peak travel periods; the system would provide about 141,900 revenue vehicle-miles and 9,200 revenue vehiclehours of service per average weekday. In 1991, the transit system provided 530 buses during peak travel periods and 63,300 revenue vehiclemiles and 5,200 revenue vehicle-hours of service per average weekday.

The transit element of Alternative Plan 2, shown in Map 93, was also designed to meet the transportation system planning objectives and design standards. It emphasized, however, the full achievement of the cost-effectiveness standards and proposed to provide only the extent and frequency of service warranted by potential transit ridership. As shown in Table 190, the transit system envisioned under this alternative plan would consist of 3,680 round-trip route miles, some 1,400 miles, or 61 percent, greater than what was provided in 1991. The transit system would require 791 vehicles operating during peak travel periods; the system would provide about 114,700 revenue vehicle-miles and 7,800 revenue vehicle-hours of service per average weekday.

The transit element of Alternative Plan 3, shown in Map 94, focused primarily on providing weekday rapid and express transit service in selected portions of major travel corridors within the Region. This alternative proposed minimal expansion of the local transit service areas and no improvement in the frequency of local service. As shown in Table 191, the transit system envisioned under Alternative Plan 3 would consist of 3,260 round-trip route miles, which would be 964 miles, or about 42 percent, greater than that provided in 1991. The transit system would require 667 vehicles operating during peak periods; the system would provide 96,900 revenue vehicle-miles and 6,700 revenue vehicle-hours of service per average weekday.

The rapid, express, and local transit service elements of each of the alternative transportation system management plans are described below.



Alternative Plan 1 proposes significant extensions of transit service and improvements in the frequency of service. The transit system envisioned under Alternative Plan 1 would consist of 4,190 round-trip route-miles, which would be about 1,894 miles, or 82 percent, greater than that provided in 1991. The transit system would require that 914 vehicles operate during peak travel periods; the system would provide about 141,900 revenue vehicle-miles and 9,200 revenue vehicle-hours of service per average weekday.

	Base Ye	ar 1991	Planned I	ncrement	2010	
		Percent		Percent		Percent
Transit Service	Number	of Total	Number	Change	Number	of Total
Round-Trip Route Length (miles)						
Rapid Routes	449	19.6	1,131	251.9	1,580	37.7
Express Routes	393	17.1	97	24.7	490	11.7
Local Routes						
Kenosha Urbanized Area	171	7.4	69	40.4	240	5.7
Milwaukee Urbanized Area	1,112	48.5	548	49.3	1,660	39.6
Racine Urbanized Area	171	7.4	49	28.7	220	5.3
Subtotal	1,454	63.3	666	45.8	2,120	50.6
Total	2,296	100.0	1,894	82.5	4,190	100.0
Special Facilities (miles)						
Exclusive Right-of-Way	0	0.0	30		30	93.8
Exclusive Lanes on Streets	2	100.0	0	0.0	2	6.2
Total	2	100.0	30	1,500.0	32	100.0
Average Weekday Vehicle Requirements ^a						
Peak Period	530		384	72.5	914	
Midday to Off-Peak Period	285	,	216	75.8	501	

TRANSIT SYSTEM OPERATING CHARACTERISTICS AND FACILITIES IN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLAN 1

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup vehicles.

Source: SEWRPC.

Rapid Transit: Within each major travel corridor, each alternative plan proposes to provide bus rapid transit service either over operationally controlled freeway lanes in mixed traffic, or over exclusive busways. In each bus rapid transit corridor, a busway capable of accommodating buses and high-occupancy vehicles would be provided if the freeway within the corridor is determined to be operating under severe traffic congestion and if the busway is determined to have the potential to accommodate more travel than the open freeway lanes.⁵ A busway would carry more traffic than a regular freeway lane if it carried more than 2,200 passengers in the peak traffic direction in the peak traffic hour. Buses operating in mixed traffic on freeways may be expected to operate at speeds of 40 miles per hour or more during peak traffic periods.

Consideration was given as to whether each busway facility should be developed through the preemption of an existing freeway lane or through the development of a new facility. Preemption of existing freeway lanes, however, was rejected. Busways were proposed only for freeway stretches exhibiting severe traffic congestion. Consequently, a removal of a freeway lane on severely congested freeway facility would substantially increase traffic congestion on the remaining lanes. The substantial increase in traffic congestion could be expected even given the anticipated shift of single-occupancy automobile traffic to carpools and public transit. The resultant level of congestion could be expected to far exceed levels currently experienced in the Region.

⁵Congested facilities carry existing or forecast future traffic volumes which exceed the design capacity of the facility by more than 30 percent, or by 80,000 vehicles per average weekday for a four-lane freeway and 120,000 vehicles per average weekday for a six-lane freeway. Under such traffic volumes, freeway speeds during peak hours may be expected to drop to 30 miles per hour or less and the freeway may be expected to experience stop-and-go traffic conditions.





WASHINGTON

KEWASK

OZAUKEE

(43)

The transit system envisioned under Alternative Plan 2 would consist of 3,680 round-trip route-miles, which would be about 1,384 miles, or 60 percent, greater than that provided in 1991. The transit system would require that 804 vehicles operate during peak travel periods; the system would provide about 114,700 revenue vehicle-miles and 7,800 revenue vehicle-hours of service per average weekday.

	·					
	Base Ye	ar 1991	Planned I	ncrement	2010	
		Percent		Percent		Percent
Transit Service	Number	of Total	Number	Change	Number	of Total
Round-Trip Route Length (miles)						
Rapid Routes	449	19.6	761	169.5	1,210	32.9
Express Routes	393	17.1	27	6.9	420	11.4
Local Routes						
Kenosha Urbanized Area	171	7.4	69	40.4	240	6.5
Milwaukee Urbanized Area	1,112	48.5	478	43.0	1,590	43.2
Racine Urbanized Area	171	7.4	49	28.7	220	6.0
Subtotal	1,454	63.3	596	41.0	2,050	55.7
Total	2,296	100.0	1,384	60.3	3,680	100.0
Special Facilities (miles)						
Exclusive Right-of-Way	0	0.0	30		30	93.8
Exclusive Lanes on Streets	2	100.0	0	0.0	2	6.2
Total	2	100.0	30	1,500.0	32	100.0
Average Weekday Vehicle Requirements ^a						
Peak Period	530		274	51.7	804	
Midday to Off-Peak Period	285		217	41.1	402	

TRANSIT SYSTEM OPERATING CHARACTERISTICS AND FACILITIES IN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLAN 2

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup vehicles.

Source: SEWRPC.

Table 191

Base Year 1991 Planned Increment 2010 Percent Percent Percent **Transit Service** Number of Total Number Change Number of Total Round-Trip Route Length (miles) Rapid Routes 449 19.6 761 169.5 1,210 37.1 393 17.1 -3 -0.8 390 12.0 Local Routes Kenosha Urbanized Area 7:4 171 19 190 5.8 11.1 Milwaukee Urbanized Area 48.5 1,112 178 16.0 1,290 39.6 Racine Urbanized Area 171 7.4 5.3 180 5.5 9 Subtotal 1,454 63.3 206 14.2 1,660 50.9 Total 2,296 100.0 964 42.0 3.260 100.0 Special Facilities (miles) 0 0.0 30 30 93.8 - -Exclusive Lanes on Streets 100.0 0.0 6.2 2 0 2 Total 2 100.0 30 1,500.0 32 100.0 Average Weekday Vehicle Requirements^a Peak Period 530 - -137 25.8 667 - -Midday to Off-Peak Period 285 - -47 16.5 332 - -

TRANSIT SYSTEM OPERATING CHARACTERISTICS AND FACILITIES IN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLAN 3

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup vehicles.



Alternative Plan 3 would focus primarily on providing weekday rapid and express transit service in selected travel corridors within the Region and would minimally expand local transit service, making no improvements in the frequency of local transit service. The transit system envisioned under Alternative Plan 3 would consist of 3,260 round-trip route-miles, which would be 964 miles, or about 42 percent, greater than that provided in 1991. The transit system would require that 667 vehicle operate during peak periods; the system would provide 96,900 revenue vehicle-miles and 6,700 revenue vehicle-hours of service per average weekday.
Under a preemption of existing freeway lanes, single-occupancy automobile traffic demand may be expected at a minimum to exceed maximum freeway capacity by more than 10 to 20 percent and to exceed design capacity by more than 45 to 60 percent. In addition, preemption of existing freeway lanes for busway use would require reconstruction of freeway-tofreeway interchanges and freeway-to-arterialstreet interchanges to eliminate the left-hand onand off-ramps which would make such facilities discontinuous.

Under each alternative, commuter railway passenger service could be considered as a rapid transit alternative to a busway facility. Consideration of commuter rail as an option to busways would occur during the conduct of major investment studies. Such studies are required by current Federal regulations prior to programming fixed-guideway transit facilities for implementation.

As previously indicated, rapid transit service under each of the alternative plans would be provided within the major travel corridors emanating from the Milwaukee CBD. Under Alternative Plan 1, such service would be extended southward to the Cities of Racine and Kenosha, to the southwest to the Village of East Troy, to the west to the Cities of Waukesha and Oconomowoc, to the northwest to the City of West Bend, and northward to the Cities of Cedarburg and Port Washington. The extent of rapid transit service would be more limited, however, under Alternative Plans 2 and 3. Under these plans, the existing service would be extended to the southwest only to the Village of Mukwonago, to the northwest only to the Village of Germantown, and to the north only to the City of Cedarburg and the Village of Grafton. Each of the alternative plans includes the busway and high-occupancy vehicle facility in the East-West corridor between STH 164 near the City of Waukesha and the City of Milwaukee currently being studied by the Wisconsin Department of Transportation.

Under Alternative Plan 1, rapid transit buses would operate over 28 routes, totalling 1,500 weekday round-trip route-miles and serving 74 transit stations. Under Alternative Plans 2 and 3, 28 routes would be operated over 1,210 roundtrip route-miles and 65 transit stations would be served. Under each of the alternatives, three routes would provide direct service to the University of Wisconsin-Milwaukee campus. Also under all the alternatives, rapid transit routes from public transit stations in Milwaukee County to the Milwaukee CBD would be operated with a limited number of intermediate stops as necessary to connect and coordinate with feeder express and local bus services, thereby providing access to major travel generators other than the Milwaukee CBD. Rapid transit routes originating at locations outside Milwaukee County, but within the anticipated future limits of the Milwaukee urbanized area, would generally serve two or three outlying park-ride lots prior to proceeding nonstop to downtown Milwaukee.

Under each of the alternatives, rapid transit routes originating at locations outside of the Milwaukee urbanized area would stop at two to six outlying public transit stations prior to proceeding essentially nonstop to the Milwaukee CBD. The public transit stations would be located, to the extent practicable, within, or near, freeway interchanges to minimize travel times. The rapid transit networks proposed under Alternative Plans 1, 2, and 3 are shown in red on Maps 92, 93, and 94, respectively.

Each alternative envisions a significant improvement in rapid transit service levels over those provided in 1991. Rapid transit service under Alternative Plan 1 would operate on weekdays between 6:00 a.m. and 6:00 p.m., on weekend evenings until 10:00 p.m., and on weekends over selected routes. Peak-period headways would range from five to 30 minutes, and off-peak headways would range from 30 to 60 minutes. Under Alternative Plans 2 and 3, weekday service would be provided over all routes during morning and afternoon peak periods, or from 6:00 a.m. to 9:00 a.m., and from 3:00 p.m. to 6:00 p.m. Under Alternative Plan 2, service would be provided over selected routes in Milwaukee County and over routes in the west and south corridors connecting the Cities of Waukesha, Racine, and Kenosha to the Milwaukee CBD during weekday off-peak periods, weekday evenings until 10:00 p.m., and on weekends. No evening or weekend service would be provided. Like Alternative Plan 1, Alternative Plan 2 would provide service in both directions of travel at peak-period headways of five to 30 minutes and nonpeak-period headways of 30 to 60 minutes. Peak-period headways under Alternative Plan 3 would range from 10 to 30 minutes.

Express Transit: Each alternative plan also proposed to develop a comprehensive express transit system. Express, or limited-stop, transit service would be provided in major travel corridors in the Milwaukee urbanized areas to connect concentrations of urban development and activity centers with each other and with the Milwaukee CBD. The express service would operate with preferential treatment over other vehicles in one of three ways: with buses operating over exclusive lanes on arterial streets, with buses on guideways, or with light-rail transit vehicles over exclusive arterial street lanes or guideways. For system evaluation purposes, the express transit service was designed to reflect stop spacings, operating speeds, and alignments typical of these three options. All three alternatives recommended that express service be provided with buses, in mixed traffic and over exclusive lanes, as soon as practicable. Consideration of bus-on-guideway or light-rail transit would coincide with Federally required major investment studies for each corridor concerned. The light-rail facilities in the East-West and North corridors of Milwaukee County currently under study by the Wisconsin Department of Transportation are included in each of the alternative plans.

The express transit network proposed under Alternative Plans 1, 2, and 3 are shown in blue on Maps 92, 93, and 94, respectively. Under Alternative Plan 1, express transit service would be provided over 15 routes, totalling 450 roundtrip route-miles. Under Alternative Plan 2, express transit service would be provided over 12 routes, totalling 420 round-trip route-miles. Under Alternative Plan 3, express transit service would also be provided over 12 routes, totaling 390 round-trip route-miles. The express routes proposed under each of the alternatives within Milwaukee County would provide crosstown service as well as service oriented to the Milwaukee CBD.

Express transit service would be operated on a regular basis on weekdays between 6:00 a.m. and 6:00 p.m. under each of the alternatives. Service would also be provided over certain routes until 10:00 p.m. and on weekends. Peakperiod headways would range from five to 15 minutes under Alternative Plan 1, and from 10 to 15 minutes under the other two alternatives. Nonpeak-period headways would range from 15 to 30 minutes under Alternative Plan 1 and 20 to 30 minutes under the other two alternatives.

Within the Racine and Kenosha urbanized areas, the express transit service plan under each of the alternatives would directly connect the Cities of Racine and Kenosha. Under Alternative Plans 2 and 3, such service would also connect the Cities of Racine and Kenosha with new and expanding commercial, office, and employment centers located along IH 94 in the western portion of each urbanized area. In all areas, buses providing express transit service would, where appropriate, provide collection and distribution services beyond the proposed termini of the express route.

Local Transit: Alternative Plans 1 and 2 propose a significant expansion of the existing local transit service provided within the Milwaukee, Racine, and Kenosha urbanized areas. Alternative Plan 3, however, proposes a limited expansion of such service when compared to the other plans. Under Alternative Plans 1 and 2, local transit service would be extended to all contiguous areas of high- and medium-density urban development within the Milwaukee, Kenosha, and Racine urbanized areas. Within the Milwaukee urbanized area, in particular, local bus service would be extended, primarily into northern and southern Milwaukee County, southern Ozaukee County, southeastern Washington County, and eastern Waukesha County. Under Alternative Plan 3, service would be extended only within northern and southern Milwaukee County and the densely developed eastern Waukesha County area.

Each of the alternative plans envisions the possibility of developing a transit route system based on the transit center concept within those newly developing areas to be provided with local transit service. Transit centers would serve as the focus for the local bus routes serving a particular area, much like the CBDs of the Cities of Kenosha, Racine, and Waukesha serve as the focus for local bus routes serving these areas. The transit centers would be located at rapid transit stations, the termini of express bus transit routes, or at major activity centers. Buses on all routes would connect at these locations simultaneously, at which time necessary transfers between buses could be effected. This approach to the provision of local transit service would permit transit operators to operate fewer vehicle-miles of service and yet achieve virtually the same levels of transit service as provided under a more traditional grid system.

In the Kenosha and Racine urbanized areas, each of the alternative plans envision that local transit service would reflect the changes to the Racine and Kenosha transit systems identified in the current transit system development plans for each system, along with extensions into contiguous areas anticipated to be fully developed, where cost-effective. Under Alternative Plans 1 and 2, local service would be extended west from the Cities of Kenosha and Racine to serve the areas of commercial and industrial development expected to occur around the major interchanges along IH 94.

The proposed local transit service areas under Alternative Plans 1, 2, and 3 are shown on Maps 92, 93, and 94, respectively. The number of round-trip route-miles provided within the Milwaukee urbanized area would be 1,660 under Alternative Plan 1, 1,590 under Alternative Plan 2, and 1,290 under Alternative Plan 3. Under Alternative Plans 1 and 2, service levels on existing routes in the Milwaukee urbanized area would be substantially improved over 1991 levels, particularly within the portion of Milwaukee County south of Silver Spring Drive, east of 76th Street, and north of Layton Avenue. In corridors in this area where express transit is not provided, headways would be 10 minutes during peak travel periods, 20 minutes during midday off-peak travel periods, and 30 minutes during all other times. Outside this area, headways on local transit routes would range from 15 to 30 minutes during weekday peak period and 30 to 60 minutes during off-peak periods. Peak-period headways under Alternative Plan 3 would range from 10 to 40 minutes in Milwaukee County.

Within the Kenosha and Racine urbanized areas, local transit service would consist of 240 and 220 round-trip route-miles, respectively, under Alternative Plan 1 and Alternative Plan 2 and 190 and 180 under Alternative Plan 3. Peakperiod headways within these areas would range from 15 to 30 minutes under Alternative Plans 1 and 2 and 20 to 30 minutes under Alternative Plan 3.

<u>Transit Fares</u>: Transit fares under Alternative Plans 1 and 2 were assumed to be consistent with fares charged in the Region in 1991, while fares under Alternative Plan 3 were assumed to

be consistent with fares charged in the Region in 1994. Accordingly, the base adult fare for local and express bus services within Milwaukee County was assumed to be \$1.00 per ride under Alternative Plans 1 and 2 and \$1.25 per ride under Alternative Plan 3, expressed in constant 1994 dollars. The rapid transit fare within Milwaukee County would be \$1.25 under Alternative Plans 1 and 2 and \$1.50 under Alternative Plan 3. The fares for rapid transit between Milwaukee County and surrounding counties would increase with the distance traveled outside Milwaukee County and would generally range from \$1.75 per ride at the outer limits of the future urbanized area to about \$2.50 at the extreme limit of service from the Milwaukee CBD under Alternative Plans 1 and 2, \$2.00 at the outer limits of the urbanized area, and \$3.00 at the extreme limits of service from the Milwaukee CBD under Alternative Plan 3. Local transit fares would range from \$0.50 per ride on the Racine transit system to \$0.60 per ride on the Kenosha and Waukesha transit systems under Alternative Plans 1 and 2 and from \$0.60 per ride on the Racine transit system to \$0.75 per ride on the Kenosha and Waukesha transit systems under Alternative Plan 3.

<u>Summary of Alternative Transportation</u> System Management Plans

As previously indicated, the successful design and implementation of measures to reduce the demand for personal vehicle travel depends on several factors including the degree to which travel demand management measures are integrated with public transit improvements. Because one of the purposes of travel demand management is to promote changes in modal use, the provision of modes of travel alternative to the personal vehicle is essential. The combinations of public transit improvements, travel demand management measures, and traffic management measures considered are intended to ensure that such alternatives are available both to promote and to serve altered travel habits and patterns.

Three alternative regional transportation system management plans were developed (see Table 192). Alternative Plan 1 proposes to increase the perceived costs of vehicle operation by 100 percent, from \$0.055 per mile to \$0.11 per mile, and to approximately double the level of transit service by 2010. Alternative Plan 2 proposes to increase the perceived cost of vehicle

SUMMARY OF ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

		Alternative Transportation System Plans	
Plan Elements	Alternative Plan 1	Alternative Plan 2	Alternative Plan 3
Public Transit ^a	Expansion of rapid and express service in all major corridors during week- days, weekday evenings, and week- ends. Expansion of local service throughout urban areas. Provision of weekday evening service throughout local transit service areas. Substan- tial areawide improvements in fre- quency of transit service. Transit fares reduced by approximately 25 percent to 1991 level. Estimated 124 percent increase in transit vehicle-miles and 76 percent increase in vehicle-hours of service	Provision of rapid and express service in selected portions of all major travel corridors. Limited weekday evening and weekend rapid and express service. Limited expansion of local service. Improvements in frequency of service limited to major Milwaukee area routes and Keno- sha, Racine, and Waukesha transit systems. Estimated 81 percent increase in transit vehicle-miles and 50 percent increase in vehicle-hours of service	Provision of rapid and express service in selected portions of all major travel corridors. No weekday evening and weekend rapid and express service. Minimal expansion in local service. No improvement in frequency of local service. Estimated 53 percent increase in transit vehicle-miles and 28 percent increase in vehicle-hours of service
Travel Demand Management			
Pricing ^b	Increase in perceived cost of operating an automobile from \$0.055 per mile to \$0.11 per mile	Increase in perceived cost of operating an automobile from \$0.055 per mile to \$0.0825 per mile	Increase in perceived cost of operating an automobile from \$0.055 to an amount necessary to cover antici- pated revenue shortfalls to implement the plan ^C
Land Use Site Design	Design of new urban development to promote bicycle and pedestrian travel. ^d Higher residential densities along all transit lines and stations. ^e New urban development in neighbor- hood units with mix of land uses to encourage pedestrian travel and shorter automobile trips	New urban development in neighbor- hood units with mix of land uses to encourage bicycle and pedestrian travel and shorter automobile trips	New urban development in neighbor- hood units with mix of land uses to encourage bicycle and pedestrian travel and shorter automobile trips
Other ^f	Other measure, including areawide ridesharing, telecommuting, and work time rescheduling programs, as well as the promotion of transporta- tion management associations and educational campaigns to encourage ridesharing and transit use	Other measures, including areawide ridesharing, telecommuting, and work time rescheduling programs, as well as the promotion of transportation management associations and educational campaigns to encourage ridesharing and transit use	Other measures, including areawide ridesharing, telecommuting, and work time rescheduling programs, as well as the promotion of transporta- tion management associations and educational campaigns to encourage ridesharing and transit use
Traffic Management ^g	Completion of Milwaukee comprehen- sive freeway traffic management system; general traffic management activities, including synchronized traffic signalization, the provision of right- and left-hand turn lanes, and proper spacing of driveways and local street intersections; use of intelligent transportation systems technology as such technology becomes applicable	Completion of Milwaukee comprehen- sive freeway traffic management system; general traffic management activities, including synchronized traffic signalization, the provision of right- and left-hand turn lanes, and proper spacing of driveways and local street intersections; use of intelligent transportation systems technology as such technology becomes applicable	Completion of Milwaukee comprehen- sive freeway traffic management system; general traffic management activities, including synchronized traffic signalization, the provision of right- and left-hand turn lanes, and proper spacing of driveways and local street intersections; use of intelligent transportation systems technology as such technology becomes applicable

^a The quality of public transit service is directly incorporated in the automobile ownership and mode choice travel simulation models and improvements in service quality serve to reduce automobile ownership and increase transit ridership.

^bThe cost of automobile travel is directly incorporated in the trip distribution and mode choice models and cost increases serve to reduce trip length and increase transit ridership.

^CBased upon past similar analysis results, perceived automobile operating costs necessary to cover the revenue shortfall tentatively envisioned under a fully developed plan may be expected to have to increase to \$0.06 per mile.

^dThe design of new development to promote bicycle and pedestrian travel is assumed to have the potential to reduce total trip generation by automobile and transit by 5 percent, and model forecast trip generation is accordingly adjusted. All travel forecasts assume implementation of the adopted regional land use plan which proposes that new urban development occur in neighborhood units with a mix of land uses. This would be directly incorporated in the trip distribution model, and may be expected to result in reducing automobile trip length.

^eThe postulated higher-residential density along all transit routes was assumed to reduce all transit walk-access by 33 percent, which was incorporated in the mode choice model and resulted in an increase in transit ridership.

^fOther measures were assumed to have minimal impact on travel.

^gTraffic management not assumed to impact travel in this analysis.

Source: SEWRPC.

406

SELECTED SOCIO-ECONOMIC	AND TRANSPORTATION	CHARACTERISTICS O	F THE REGION: 1991
AND 2010 NO-BUILD AND A	ALTERNATIVE TRANSPOR	TATION SYSTEM MA	NAGEMENT PLANS

		Year 2010 Alternative Plans									
Characteristic	Base Year 1991	No-Build	Percent Change	Plan 1	Percent Change	Plan 2	Percent Change	Plan 3	Percent Change		
Population	1,810,400	1,911,000	5.6	1,911,000	5.6	1,911,000	5.6	1,911,000	5.6		
Households	714,900	774,300	8.3	774,300	8.3	774,300	8.3	774,300	8.3		
Employment	990,300	1,095,000	10.6	1,095,000	10.6	1,095,000	10.6	1,095,000	10.6		
Vehicles Available	1,132,000	1,309,300	15.7	1,287,400	13.7	1,291,900	14.1	1,296,400	14.5		
Internal Person Trips	5,541,000	6,116,100	10.4	6,055,700	9.3	6,098,300	10.1	6,103,500	10.2		
Persons per Vehicle	1.6	1.5	-6.3	1.5	-6.3	1.5	-6.3	1.5	-6.3		
Vehicles per Household	1.7	1.7	0.0	1.7	0.0	1.7	0.0	1.7	0.0		
Trips Per Capita	3.1	3.2	3.2	3.2	3.2	3.2	3.2	3.2	3.2		
Trips per Household	7.8	7.9	1.3	7.8	0.0	7.9	1.3	7.9	1.3		

Source: SEWRPC.

operation by 50 percent, to about \$0.08 per mile, and to increase to the level of transit service by about 50 percent by 2010. Alternative Plan 3 does not propose to increase the perceived cost of driving to affect changes in travel demand, like the other alternatives, but instead proposes to increase perceived vehicle operating costs by an amount necessary to cover anticipated revenue shortfall in funding to implement the plan. Alternative Plan 3 proposes to increase transit service in the Region by about 35 percent.

TESTING OF ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

The second step in the design of alternative transportation system plans was to evaluate the three alternative transportation system management plans that were designed. It should be noted that at this critical stage of the plan design process, no improvements to the arterial street and highway system were considered. Each alterative transportation system management plan was thus combined with the no-build arterial street and highway system for the purpose of testing. As previously indicated, it was only after the ability of each of the transportation system management plans to resolve transportation system deficiencies was determined that arterial improvements were considered.

Assignment of Travel to

Highway and Transit Networks

The performance of each of the three alternative transportation system management plans under probable future land use and travel conditions was analyzed with the assistance of the traffic simulation models described in Chapter VII of this report. By applying these models, the probable future travel demand generated under the adopted regional land use plan was assigned to the no-build arterial street and highway network and to the alternative transit networks pertaining to each plan.

Personal Vehicle Availability: Given the transportation system demand management measures and transit system improvements envisioned under each alternative plan, it is estimated that by the year 2010 the number of personal vehicles available in the Region may be expected to increase from about 1.13 million in 1991 to about 1.29 million, or by about 14 percent, under Alternative Plans 1 and 2 and to 1.30 million, or by nearly 15 percent, under Alternative Plan 3 (see Table 193). Personal vehicles are defined to include automobiles and those trucks weighing 8,000 pounds or less which are used for personal travel rather than for commercial reasons. Under the no-build plan, the number of personal vehicles is estimated to increase to 1.31 million, or by about 16 percent, by the year 2010.

The resident population of the Region is expected to increase by about 6 percent during the same period, the number of households by about 15 percent, and employment by about 11 percent. The number of persons per vehicle may then be expected to continue to decline from 1.6 in 1991 to about 1.5 under each of the alternative plans and the no-build plan, while the number of vehicles per household may be expected to increase from 1.6 in 1991 to 1.7 under each of the alternative plans and the no-build plan.

DISTRIBUTION OF INTERNAL PERSON TRIPS WITHIN THE REGION BY MODE: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

i.				Yea	r 2010 Alt	ternative Plan	s		
Mode of Travel	Base Year 1991	No-Build	Percent Change	Plan 1	Percent Change	Plan 2	Percent Change	Plan 3	Percent Change
Automobile Driver Automobile Passenger ^a Transit Passenger School Bus Passenger	4,060,900 1,080,300 172,200 227,600	4,721,600 941,400 168,000 285,100	16.3 -12.9 -2.4 25.3	4,583,600 915,400 271,600 285,100	12.9 -15.3 57.7 25.3	4,673,500 932,000 207,700 285,100	15.1 -13.7 20.6 25.3	4,694,500 937,000 186,900 285,100	15.6 -13.2 8.5 25.3
Total	5,541,000	6,116,100	10.4	6,055,700	9.3	6,098,300	10.1	6,103,500	10.2

^a The cost of travel was incorporated into the forecast of nonwork trips with respect to potential use of transit and trip length. It was not incorporated into the forecast of nonwork trip vehicle occupancy. Analyses of the factors influencing vehicle occupancies for nonwork trips indicated that travel time, distance, and cost had little effect on vehicle occupancy and that the number of persons per household and the number of automobiles available per household established the differences observed in vehicle occupancy.

Source: SEWRPC.

Person Trip Generation: Table 193 also shows the increases in internal tripmaking that may be expected under each of the alternative plans. Given the travel demand management measures and transit system improvements envisioned under each of the three alternative plans, average weekday internal person trips may be expected to increase from nearly 5.54 million trips in 1991 to about 6.06 million by 2010, or by about 9 percent, under Alternative Plan 1, and to about 6.1 million, or by 10 percent, under Alternative Plans 2 and 3. Under the no-build plan internal person trips may be expected to increase to nearly 6.12 million. As shown in Table 193, the average number of internal person trips per capita may be expected to increase from about 3.1 in 1991 to 3.2 by 2010 under each of the alternative plans and the nobuild plan. The average number of internal person trips per household may be expected to remain at 7.8 under Alternative Plan 1 and to increase to 7.9 under the other alternatives.

<u>Mode of Travel</u>: The distribution of internal person trips within the Region by mode of travel under each of the alternative transportation system management plans and the no-build plan is summarized in Table 194. Average weekday transit trip production within the Region may be expected to increase from about 172,200 trips in 1991 to about 271,600 trips by 2010, or by about 58 percent, under Alternative Plan 1; to about 207,700 trips, or by about 21 percent, under Alternative Plan 2; and to about 186,900 trips,

or by about 9 percent, under Alternative Plan 3. The proportion of total internal travel generated within the Region served by transit may be expected to increase from 3.1 percent in 1991 to 4.5 percent by 2010 under Alternative Plan 1, to increase to 3.4 percent under Alternative Plan 2, and to remain at about 3.1 percent under Alternative Plan 3. The proportion of trips made within Milwaukee County by transit may be expected to be 7 percent in 2010 under Alternative Plan 1 and 4.3 percent under Alternative Plan 3. Under the no-build plan, average weekday transit trip production may be expected to decrease to 168,000 trips by 2010, or by about 2 percent, and the proportion of total internal person trips made by transit within the Region may be expected to decline to 2.7 percent. The proportion of total trips within Milwaukee County made by transit may be expected to be 3.3 percent in the year 2010 under the no-build plan.

Table 194 also shows that average weekday internal automobile driver trips, including personal-use light truck driver trips, may be expected to increase from nearly 4.06 million trips in 1991 to about 4.58 million trips by 2010, or by about 13 percent, under Alternative Plan 1; to about 4.67 million trips, or by 15 percent, under Alternative Plan 2; and to about 4.69 million trips, or by 16 percent, under Alternative Plan 3. The proportion of all internal person trips made as an automobile driver may be expected to increase from 73.3 percent in 1991 to 75.5 percent

· · · · · · · · · · · · · · · · · · ·		Year 2010 Alternative Plans									
Vehicle Class	Base Year 1991	No-Build	Percent Change	Plan 1	Percent Change	Plan 2	Percent Change	Plan 3	Percent Change		
Automobile Internal External ^a Other ^a Subtotal	4,060,900 229,200 39,300 4 329 400	4,721,600 306,300 65,500	16.3 33.6 66.7	4,583,600 306,300 65,500 4,955,400	12.9 33.6 66.7	4,673,500 306,300 65,500	15.1 33.6 66.7 16.5	4,694,500 306,300 65,500	15.6 33.6 66.7 17.0		
Truck Internal ^a External ^a	520,100 44,100	579,000 58,200	11.3 32.0	579,000 58,200	11.3 32.0	579,000 58,200	11.3 32.0	579,000 58,200	11.3 32.0		
Subtotal	564,200	637,200	12.9	637,200	12.9	637,200	12.9	637,200	12.9		
Total	4,893,600	5,730,600	17.1	5,592,600	14.3	5,682,500	16.1	5,703,500	16.6		

DISTRIBUTION OF VEHICLE TRIPS WITHIN THE REGION BY VEHICLE CLASS: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

^a The number of external automobile, other automobile, external commercial truck, and internal commercial truck trips are shown to be the same in the year 2010 for each of the alternative plans since the availability of transit service and the promotion of bicycle and pedestrian friendly development may not be expected to significantly affect the generation of external travel and internal truck travel.

Source: SEWRPC.

by 2010 under Alternative Plan 1, to 76.7 percent under Alternative Plan 2, and to 76.9 percent under Alternative Plan 3. Under the no-build alternative, average weekday automobile driver trips would increase to 4.72 million trips, or by 16 percent. The proportion of all trips made as an automobile driver may be expected to increase to 77.2 percent under the no-build plan.

The anticipated increases in internal and external vehicle trips by vehicle class are shown in Table 195. Internal vehicle trip production, including trips made by both automobile and truck, on an average weekday is estimated to increase from 4.58 million trips in 1991 to 5.16 million trips by 2010, or by about 13 percent, under Alternative Plan 1; to 5.25 million trips, or by about 15 percent, under Alternative Plan 2; and to 5.27 million trips, or also by 15 percent, under Alternative Plan 3. Under the no-build plan, internal vehicle trips may be expected to increase to 5.30 million trips, or by about 16 percent, over the 1991 level. Whereas the transportation system management measures would probably impact internal vehicle tripmaking under each plan differently, the variance in the impact of such measures on external tripmaking among the plans may be expected to be minimal. Accordingly, the number of external vehicle trips may be expected to remain the same for each of the alternative plans. increasing by 33 percent, from 273,300 in 1991 to 364.500 in 2010.

System Performance

Arterial vehicle-miles of travel generated within the Region on an average weekday under each of the alterative plans may be expected to increase from about 33.07 million in 1991 to about 38.87 million by 2010, or by 18 percent, under Alternative Plan 1: to about 41.57 million, or by 26 percent, under Alternative Plan 2; and to about 44.09 million, or by 33 percent, under Alternative Plan 3. Under the no-build plan, vehicle-miles of travel may be expected to increase to about 44.52 million by 2010, or by 35 percent (see Table 196). Thus, a 13 percent reduction in vehicle-miles of travel from that which may be expected to occur under the nobuild plan may be realized under Alternative Plan 1. Similarly, a 7 percent reduction may be realized under Alternative Plan 2. No meaningful reduction in vehicle-miles of travel from that expected to occur under the no-build plan may be realized under Alternative Plan 3.

Allocation of the transit travel demands to each of the alternative transportation system management plans indicates that the number of annual revenue passengers in the Region may be expected to increase from about 50.22 million in 1991 to about 73.75 million by 2010, or by 47 percent, under Alternative Plan 1; to 60.65 million, or by 21 percent, under Alternative Plan 2; and to 54.11 million, or by 8 percent, under Alternative Plan 3 (see Table 197). Under the no-

VEHICLE-MILES OF TRAVEL ON THE ARTERIAL STREET AND HIGHWAY SYSTEM WITHIN THE REGION BY COUNTY: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

		Year 2010 Alternative Plans										
County	Base Year 1991	No-Build	Percent Change	Plan 1	Percent Change	Plan 2	Percent Change	Plan 3	Percent Change			
Kenosha	2,500,000	4,080,000	63.2	3,658,000	46.3	3,813,000	52.5	3,930,000	57.2			
Milwaukee	14,391,000	17,099,000	18.8	14,497,000	0.7	15,576,000	8.2	16,585,000	15.2			
Ozaukee	1,942,000	2,340,000	20.5	2,143,000	10.4	2,249,000	15.8	2,360,000	21.5			
Racine	2,966,000	4,194,000	41.4	3,500,000	18.0	3,776,000	27.3	4,036,000	36,1			
Walworth	1,913,000	2,627,000	37.3	2,373,000	24.0	2,554,000	33.5	2,754,000	44.0			
Washington	2,379,000	3,227,000	35.6	2,800,000	17.7	2,998,000	26.0	3,161,000	32.9			
Waukesha	6,981,000	10,955,000	56.9	9,895,000	41.7	10,604,000	51.9	11,262,000	61.3			
Total	33,072,000	44,522,000	34.6	38,866,000	17.5	41,570,000	25.7	44,088,000	33.3			

Source: SEWRPC.

Table 197

TRANSIT SYSTEM PERFORMANCE IN THE REGION: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

				Yea	ar 2010 Alt	ernative Plans			4 1
		No-Bu	ild	Plan	1	Plan 2		Plan	3
	Base Year		Percent		Percent		Percent		Percent
Transit System Characteristics	1991	Number	Change	Number	Change	Number	Change	Number	Change
Service Provided, Average Weekday						n in the state		9	4. ¹
Revenue Vehicle-Miles						l de la companya de l		1.125	
Rapid	3,400	3,100	-8.8	25,600	652.9	18,400	441.2	15,900	367.6
Express	3,300	6,000	81.8	29,800	803.0	21,100	539.4	20,400	518.2
Local	56,600	56,800	0.4	86,500	52.8	75,200	32.9	60,600	7.1
Total	63,300	65,900	4.1	141,900	124.2	114,700	81.2	96,900	53.1
Revenue Vehicle-Hours								en an	n sali
Rapid	170	160	-5.9	1,100	547.1	700	311.8	600	252.9
Express	170	360	111.8	1,500	782.4	1,000	488.2	1,000	488.2
Local	4,880	5,130	5.1	6,600	35.2	6,100	25.0	5,100	4.5
Total	5,220	5,650	8.2	9,200	76.2	7,800	49.4	6,700	28.4
Seat-Miles	2,975,000	3,028,000	1.8	6,403,000	115.2	5,235,000	76.0	4,425,000	48.7
Service Utilization			_			na i di di	4		
Ridership					1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	· · · · ·	1990 - Albert		1.1
Average Weekday Revenue Passengers	172,200	168,000	-2.4	271,600	57.7	207,700	20.6	186,900	.8.5
Annual Revenue Passengers	50,222,900	48,998,000	-2.4	80,122,000	59.5	60,648,000	20.8	54,575,000	8.7
Revenue Passengers per		· · · ·	÷			- · · ·	1999 - 1999 1999 - 1999		
Revenue Vehicle-Hour	33.0	29.7	-9.9	29.5	-10.5	26.6	-19.3	25.3	-23.4
Average Weekday Passenger Miles	609,100	728,800	19.7	1,376,000	125.9	1,004,900	65.0	903,400	48.3

Source: SEWRPC.

build plan the number of revenue passengers may be expected to decline to about 49.00 million per year, or by 2 percent. Thus, a 50 percent increase in annual transit ridership over that expected to occur under the no-build transportation system plan may be realized under Alternative Plan 1. Similarly, a 24 percent increase in annual transit ridership may be expected under Alternative Plan 2. Under Alternative Plan 3, a 10 percent increase in annual transit ridership over that expected to occur under the no-build plan may be realized. Table 197 also compares the anticipated utilization of the transit service proposed to be provided under each of the alternative plans. The highest utilization of transit service among the alternative plans, as measured in revenue passengers per revenue vehicle hour, may be expected to occur under Alternative Plan 1. Under that plan, about 30 passengers per vehicle hour may be expected to be carried. Because the proposed increase in vehicle hours of service provided is proportionately greater than the anticipated ridership under Alternative Plan 1,

POPULATION SERVED BY LOCAL PUBLIC TRANSIT IN THE REGION BY URBANIZED AREA: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

			•	Year 2010 Alt	ernative Tra	Insportation S	ystem Plans	5	
	Base Year		Percent		Percent	1. State 1.	Percent		Percent
Urbanized Area	1991	No-Build	Change	Plan 1	Change	Plan 2	Change	Plan 3	Change
Kenosha									1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Urbanized Area Population	94,300	107,300	13.8	107,300	13.8	107,300	13.8	107,300	13.8
Service Area Population	86,600	84,200	-2.7	99,000	14.3	99,000	14.3	84,600	97.7
Percent Served	91.8	78.5	-14.5	92.3	0.5	92.3	0.5	78.8	-14.2
Milwaukee									
Urbanized Area Population	1,226,300	1,277,100	4.1	1,277,100	4.1	1,277,100	4.1	1,277,100	4.1
Service Area Population	935,400	926,500	-1.0	1,147,300	22.7	1,089,800	16.5	935,600	0.0
Percent Served	76.2	72.5	-4.9	89.8	17.8	85.3	11.9	73.3	-3.8
Racine									
Urbanized Area Population	121,800	132,100	8.5	132,100	8.5	132,100	8.5	132,100	8.5
Service Area Population	114,600	108,900	-5.0	125,300	9.3	125,300	9.3	111,100	-3.9
Percent Served	94.1	82.4	-12.4	94.9	0.9	94.9	0.9	84.1	-10.6
Total									
Urbanized Area Population	1,442,400	1,516,500	5.1	1,516,500	5.1	1,516,500	5.1	1,516,500	5.1
Service Area Population	1,136,600	1,119,600	-1.5	1,371,600	20.7	1,314,100	15.6	1,130,300	-0.6
Percent Served	78.8	73.8	-6.3	90.4	14.7	86.7	10.0	74.5	-5.5

Source: SEWRPC.

the anticipated utilization of transit is actually lower than the existing utilization of 33 passengers per vehicle hour.

Table 198 shows the proportion of the resident population that may be expected to be served by local public transit in the urbanized areas by the vear 2010 under each of the alternative plans. As shown in Table 198, the resident population within the Kenosha urbanized area expected to reside in areas served by local transit would increase from about 86,600 persons in 1991, or about 92 percent of the urbanized area population, to about 99,000 persons, or 92 percent, under Alternative Plans 1 and 2 and decrease to about 84,600 persons or 79 percent under Alternative Plan 3. Under Alternative Plan 3, by the year 2010 a smaller proportion of the Kenosha urbanized area population would be served by local transit than was served in 1991.

In the Milwaukee urbanized area the population expected to reside in areas served by local transit would increase from about 935,400 persons in 1991, or about 76 percent of the urbanized area population, to about 1.15 million persons, or 90 percent, under Alternative Plan 1; to about 1.09 million persons, or 85 percent, under Alternative Plan 2; and decrease to about 934,600 persons, or 73 percent, under Alternative Plan 3. Under Alternative Plan 3, by the year 2010 a smaller proportion of the Milwaukee urbanized area population would be served by local transit than was served in 1991.

In the Racine urbanized area the population expected to reside in areas served by local transit would increase from 114,600 persons in 1991, or 94 percent of the urbanized area population, to about 125,300 persons, or 95 percent, under Alternative Plans 1 and 2; and decrease to about 111,100 persons, or 84 percent, under Alternative Plan 3. Under Alternative Plan 3, by the year 2010 a smaller proportion of Racine urbanized area population would be served by local transit than was served in 1991.

System Deficiencies

The impact of the anticipated increase in travel on the arterial street and highway system is reflected in the anticipated number of arterial miles projected to operate at or over design capacity in the year 2010. As shown in Table 199 by 2010, the number of miles operating over design capacity may be expected to increase from 279 in 1991 to about 601, or by 115 percent, under Alternative Plan 1; to 628, or by 125 percent, under Alternative Plan 2; and to about 679, or 143 percent, under Alternative Plan 3. Under the no-build plan, the number of miles operating over design capacity may be expected to increase to 681 miles, or by

ARTERIAL MILEAGE OPERATING UNDER, AT, AND OVER DESIGN CAPACITY: 1991 AND 2010 NO-BUILD ALTERNATIVE AND TRANSPORTATION SYSTEM MANAGEMENT PLANS

	Base Year 1991							
	Under Design Capacity		At Desig	n Capacity	Over Desi] .		
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles	
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	286.0 543.2 254.2 288.7 411.1 356.5 591.3	90.0 70.1 88.1 83.0 95.8 89.3 82.5	16.6 110.3 21.0 24.9 15.0 23.2 53.0	5.2 14.2 7.3 7.2 3.5 5.8 7.4	15.1 121.9 13.3 34.3 3.1 19.5 72.0	4.8 15.7 4.6 9.9 0.7 4.9 10.1	317.7 775.4 288.5 347.9 429.2 399.2 716.3	
Total	2,731.0	83.4	264.0	8.1	279.2	8.5	3,274.2	

	Year 2010 No-Build Plan						
	Under Design Capacity		At Desig	n Capacity	Over Desi		
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	230.1 490.0 233.9 241.7 387.2 337.7 369.1	66.7 62.1 81.1 59.3 87.0 75.3 48.8	74.2 122.6 26.3 80.5 40.2 54.1 111.6	21.5 15.6 9.1 19.8 9.0 12.1 14.7	40.9 176.7 28.3 85.3 17.5 56.9 274.9	11.8 22.3 9.8 20.9 4.0 12.6 36.5	345.2 789.3 288.5 407.5 444.9 448.7 755.6
Total	2,289.7	65.8	509.5	14.6	680.5	19.6	3,479.7

	2010 Alternative Plan 1							
	Under Design Capacity		At Desig	n Capacity	Over Desi			
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles	
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	233.8 496.1 248.5 251.2 387.2 348.5 377.9	67.7 62.9 86.1 61.6 87.0 77.7 50.0	76.9 136.4 18.0 89.2 40.2 49.3 125.8	22.3 17.3 6.2 21.9 9.0 11.0 16.6	34.5 156.8 22.0 67.1 17.5 50.9 251.9	10.0 19.9 7.6 16.5 3.9 11.3 33.3	345.2 789.3 288.5 407.5 444.9 448.7 755.6	
Total	2,343.2	67.3	535.8	15.4	600.7	17.3	3,479.7	

			2010 Alter	native Plan 2			
	Under Desi	Under Design Capacity At Design Capacity Over Design Capacity					
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	232.8 493.9 244.5 246.9 392.8 337.7 376.7	67.4 62.6 84.7 60.6 88.3 75.3 49.9	74.5 133.0 19.0 85.4 38.5 58.1 117.7	21.6 16.9 6.6 21.0 8.7 12.9 15.6	37.9 162.4 25.0 75.2 13.6 52.9 261.2	11.0 20.6 8.7 18.5 3.1 11.8 34.6	345.2 789.3 288.5 407.5 444.9 448.7 755.6
Total	2,325.3	66.8	526.2	15.1	628.2	18.1	3,479.7

	2010 Alternative Plan 3							
	Under Design Capacity		At Desig	n Capacity	Over Desi			
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Miles	
Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha	230.1 490.0 236.2 243.0 337.7 387.7 369.1	66.7 62.1 81.9 59.6 87.0 75.3 48.8	74.2 122.6 25.5 79.2 40.2 54.1 111.6	21.5 15.5 8.8 19.4 9.0 12.0 14.8	40.9 176.7 26.8 85.3 17.5 56.9 274.9	11.8 22.4 9.3 20.9 3.9 12.7 36.4	345.2 789.3 288.5 407.5 444.9 448.7 755.6	
Total	2,293.3	65.9	507.4	14.6	679.0	19.5	3,479.7	

Source: SEWRPC.

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON ARTERIAL STREETS AND HIGHWAYS: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

						Urbanize	ed Area Popula	tion Meetir	g Travel				
	Lirbania	ad Area				Time Stand	lard on Arteria	l Streets an	d Highways				
	Popu	lation				Proposed 2010							
					No-Bi	uild							
			Base	Base Year		rtation	Altern	ative	Altern	ative	Alternative		
Urbanized Area and	Basa Vaar		199	91	System	Plan	Pian	1	Plan	2	Plan	3	
Activity Center	1991	2010	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent	
Kenosha Urbanized Area	94 300	107.300											
Employment-Related ^a			90.000	96.4	107.300	100.0	107 300	100.0	107 300	100.0	107 300	100.0	
Major Retail-Service ^b			0	0.0	0	0.0	0	0.0	0	0.0	107,500	0.0	
Medical Facility ^C			94.300	100.0	107.300	100.0	107.300	100.0	107.300	100.0	107 300	100.0	
Major Park ^d			94,300	100.0	107.300	100.0	107.300	100.0	107.300	100.0	107.300	100.0	
Higher-Education Facility ^e			94,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0	
Scheduled Air Transport ^f			94,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0	
Milwaukee Urbanized Area	1,226,300	1,277,100											
Employment-Related			1,147,900	93.6	1,180,500	92.4	1,180,500	92.4	1,180,500	92.4	1,180,500	92.4	
Major Retail-Service		••	1,161,400	94.7	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	
Medical Facility			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	
Major Park	••		1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	
Higher-Education Facility			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	
Scheduled Air Transport			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0	
Racine Urbanized Area	121,800	132,100				• •							
Employment-Related			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0	132,100	100.0	
Major Retail-Service			32,200	26.4	132,100	100.0	132,100	100.0	132,100	100.0	132,100	100.0	
Medical Facility			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0	132,100	100.0	
Major Park			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0	132,100	100.0	
Higher-Education Facility			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0	132,100	100.0	
Scheduled Air Transport			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0	132,100	100.0	

^aStandard: 30 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of three major retail and service centers.

^CStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

Source: SEWRPC.

144 percent, by the year 2010. Similarly, the number of arterial miles operating at design capacity, which totaled about 264 in 1991, may be expected to increase to about 536, or by 103 percent, under Alternative Plan 1; to 526, or by 99 percent, under Alternative Plan 2; and to about 507, or by 92 percent, under Alternative Plan 3. Under the no-build plan such mileage may be expected to increase to about 510 miles, or by 93 percent, by the year 2010.

To further identify the transportation system deficiencies that remained after the alternative transportation system management plans were postulated, the previously indicated travel time standards were applied to identify existing and proposed urban residential areas inadequately served by the highway mode and by the public transit mode. As shown in Table 200, the alternative plans do not differ in their ability to provide timely access to employment and major activity centers through travel on the arterial system. A comparison of the findings set forth in Table 201 indicates, however, that the alternative plans vary significantly in the degree to which they provide timely access to important activity centers and employment opportunities through travel on the public transit system.

While each of the alternative plans generally provides an improvement in access over that provided by the no-build plan, Alternative Plans 1 and 2 generally provide the greatest level of access. For instance, the proportion of the

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON TRANSIT: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

	Urbaniz	ed Area	Area			Area Popu	lation Meeting	Travel Tim	e Standard on	Transit		
	Popu	lation			2010							
Urbanized Area and Base Year		Base Year 1991		No-Build Transportation System Plan		Alterna Plan	itive 1	Alternative Plan 2		Alternative Plan 3		
Activity Center	1991	2010	Number	Percent	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Kenosha Urbanized Area Employment-Related ^a Major Retail-Service ^b	94,300 	107,300 	 58,000 0	 61.5 0.0	 51,400 1,400	47.9 1.3	97,700 9,500	91.1 8.9	 97,700 9,500	91.1 8.9	 53,200 1,400	49.6 1.3
Medical Facility ^c Major Park ^d Higher-Education Facility ^e Scheduled Air Transport ^f			60,600 12,600 33,200 0	64.3 13.4 35.2 0.0	53,300 20,700 51,400 0	49.7 19.3 47.9 0.0	63,600 77,500 80,800 51,900	59.3 72.2 75.3 48.4	63,600 77,500 80,800 51,900	59.3 72.2 75.3 48.4	53,300 24,700 68,100 51,900	49.7 23.0 63.5 48.4
Milwaukee Urbanized Area Employment-Related Major Retail-Service ⁹ Medical Facility Major Park Higher-Education Facility Scheduled Air Transport	1,226,300 	1,277,100 	14,200 10,100 700,400 636,300 775,600 338,600	1.2 0.8 57.1 51.9 63.2 27.6	600 35,300 648,000 617,300 740,900 316,800	0.0 2.8 50.7 48.3 58.0 24.8	619,900 282,800 870,200 994,200 1,168,000 776,800	48.5 22.1 68.1 77.8 91.5 60.8	276,000 107,900 770,000 790,500 1,092,200 665,600	21.6 8.4 60.2 61.9 85.5 52.1	243,600 66,200 728,900 701,000 931,900 569,400	19.1 5.2 57.1 54.9 73.0 44.6
Racine Urbanized Area Employment-Related Major Retail-Service Medical Facility Major Park Higher-Education Facility Scheduled Air Transport	121,800 	132,100 	59,100 21,500 53,700 24,000 79,600 18,300	48.5 17.7 44.1 19.7 65.4 15.0	56,400 18,300 46,700 37,000 88,400 18,600	42.7 13.9 35.4 28.0 66.9 14.1	107,500 52,500 61,300 59,800 110,700 63,800	81.4 29.7 46.4 45.3 83.8 48.3	107,500 52,500 61,300 59,800 110,700 63,800	81.4 29.7 46.4 45.3 83.8 48.3	94,400 32,100 49,200 38,300 102,700 60,100	71.5 24.3 37.2 29.0 77.7 45.5

^aStandard: 45 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of one major retail and service center.

^CStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

^gStandard: 35 minutes' overall travel time of three major retail and service centers.

Source: SEWRPC.

Kenosha urbanized area population expected to be within 45 minutes of 40 percent of jobs within the Kenosha urbanized area may be expected to increase from about 62 percent in 1991 to about 91 percent under Alternative Plans 1 and 2 and to decrease to about 50 percent under Alternative Plan 3. The proportion of the Milwaukee urbanized area population expected to be within 45 minutes of 40 percent of the jobs in the Milwaukee urbanized area may be expected to increase from about 1 percent in 1991 to about 49 percent under Alternative Plan 1, to about 22 percent under Alternative Plan 2, and to about 19 percent under Alternative Plan 3. The proportion of the Racine urbanized area population expected to be within 45 minutes of 40 percent of jobs within the Racine urbanized area may be expected to increase from about 49 percent in 1991 to about 81 percent under Alternative Plans 1 and 2 and to increase to about 72 percent under Alternative Plan 3.

Conclusions from the Testing of Alternative

Transportation System Management Plans As indicated in Table 202, the alternative transportation system management plans may be expected to vary in the degree to which they impact travel within the Region by the year 2010. The most significant difference is evident with regard to use of the arterial street and highway system, as measured in vehicle-miles of travel. Under both the no-build alternative and Alternative Plan 3, daily vehicle-miles of travel in the Region may be expected to increase between 1991 and 2010 by about 35 percent, to about 44.52 million vehicle miles. Under Alternative Plans 1 and 2, however, vehicle-miles of travel may be expected to increase by only 18 percent, to 38.87 million per day, and by 26 percent, to 41.57 million per day, respectively. Thus, both Alternative Plans 1 and 2 may be expected to reduce the amount of arterial travel forecast under the no-build alternative.

	and the second					
				19. J. 1		
Category	Measure	Base Year 1991	No-Build Alternative	Alternative Plan 1	Alternative Plan 2	Alternative Plan 3
Fravel	Average weekday internal person trips (millions)	5.54	6.12	6.06	6.10	6.10
	Average weekday internal vehicle-trips (millions)	4.58	5.30	5.16	5.25	5.27
	Average weekday vehicle-miles of travel (millions)	33.07	44.52	38.87	41.57	44.09
:	Average weekday transit ridership	172,200	168,000	271,600	207,700	186,900
Fraffic Congestion	Average weekday arterial traffic congestion Amount and proportion of arterial mileage at design capacity	264 miles 8.1 percent	509.5 miles 14.9 percent	535.8 miles 15.4 percent	526.2 miles 15.1 percent	507.4 miles 14.6 percent
	Amount and proportion of arterial mileage over design capacity	279 miles 8.5 percent	681 miles 19.6 percent	601 miles 17.3 percent	628 miles 18.1 percent	679 miles 19.5 percent
Accessibility by Transit	Percent population meeting employment access standard by transit ^a Kenosha Urbanized Area	61.5	47.9	91.1	91.1	49.6
	Milwaukee Urbanized Area	1.2	0.0	48.5	21.6	19.1
	Racine Urbanized Area	48.5	42.7	81.5	81.5	71.5

1.14

COMPARISON OF MAJOR TRANSPORTATION IMPACTS OF THE YEAR 2010 ALTERNATIVE TRANSPORTATION SYSTEM MANAGEMENT PLANS

^aWithin a given urbanized area: 45 minutes' overall travel time of 40 percent of that urbanized area's jobs.

Source: SEWRPC.

The reduction in the use of the arterial system from what can be expected to occur under the nobuild alternative, 12 percent under Alternative Plan 1 and 6 percent under Alternative Plan 2, is attributable to the impacts of the proposed transportation system management measures, including travel pricing, land use site design, and public transit improvements. The impact of the pricing measures on trip length would appear to account for the principal differences between the alternative plans in vehicle-miles of travel within the Region. Indeed, the alternative plans differ by less than one percent in the number of vehicle trips generated. Further, the transit improvements proposed under Alternative Plan 1, while having the potential to significantly impact weekday transit ridership, have an estimated impact of less than 2 percent on weekday vehicle-miles of travel.

Population served by local transit (millions)

It can be concluded from the above that the significant increases in the perceived costs of automobile travel proposed under Alternative Plan 1 tend to reduce the distance people are willing to travel by automobile. This impact may be manifest in the decisions of households to locate closer to employment centers or to pursue shopping and recreational opportunities closer to home.

SUMMARY AND CONCLUSIONS

1.12

The travel demand generated by a given land use pattern can be met by various combinations of transportation facilities, services, and demand management measures. This chapter has summarized a complex process of plan design that allowed for the explicit consideration of a variety of improvements and enabled cost-effective alternative transportation system management plans to be developed.

1.37

1.31

1.13

The plan design process began by identifying a no-build transportation system which included existing and committed transportation system facilities, services, and management measures. A forecast of future travel demand derived from the adopted year 2010 regional land use plan was then assigned to the no-build system. The results of this assignment indicated potential problems and deficiencies expected to occur in the absence of significant improvements in transportation facilities, services, and demand management.

On the basis of the results of the assignment of future travel demand to the no-build plan, three transportation system management plans were prepared. The plans combined measures to resolve deficiencies identified in the no-build plan, traffic congestion, and inadequate transit service, in particular. The plans include measures to manage and reduce vehicular travel demand and to manage traffic flow, as well as public transit improvements. These plans did not include arterial street and highway improvements and, for the purpose of system analysis, included the arterial street and highway system element of the no-build alternative. Forecasts of future regional travel were assigned to the three alternative plans and their ability to resolve the transportation problems and deficiencies expected to occur under the no-build plan was tested.

As is described in Chapter X, arterial street and highway improvements were postulated to resolve remaining transportation system deficiencies. Such improvements were proposed only after the full potential transportation-related impacts of the transportation system management measures were understood. The major findings of the foregoing plan design process are summarized below.

No-Build Transportation System Plan

The no-build plan represents both a policy alternative for the Region and a point of departure for the initial testing and design of alternative transportation plans which incorporate facility and service improvements and travel demand management measures. The arterial street and highway and public transit systems under the no-build plan include those facilities and services completed and open to traffic as of January 1, 1994, and those facilities programmed for construction in the annual element, that is, the element applicable to the calendar years 1994 and 1995, of the 1993 through 1998 Federally approved Transportation Improvement Program for Southeastern Wisconsin. The no-build plan also includes those existing streets and highways which did not function as arterials in 1991 but would function as arterials by the year 2010. The arterial street and highway system would increase from 3,274 miles in 1991 to 3,480 miles by 2010, an increment of about 206 miles, or 6 percent.

The no-build regional transit system was defined as the existing transit facilities and services in operation on January 1, 1994, plus those improvements programmed for implementation in the annual element of the transportation improvement program. Under the no-build plan,

rapid transit service within the Region would consist of 519 round-trip route-miles, representing an increase of 70 miles, or 16 percent, over the 1991 level. Express transit service would consist of 377 round-trip route-miles, a decrease of 16 route-miles, or 4 percent, from the level provided in 1991. Local transit service within the Region would consist of 1.540 round-trip routemiles, an increase of 86 route-miles, or 6 percent, over the 1991 level. The no-build transportation system also includes those transportation system management measures included in the annual element of the transportation improvement program. Such measures include, among others, freeway traffic management system implementation and the construction and reconstruction of bikeways.

Under the no-build plan, the number of personal automobiles and light trucks available in the Region may be expected to increase by about 16 percent, from 1.13 million in 1991 to 1.31million by 2010. Nearly 6.12 million internal person trips may be expected to be made on an average weekday, an increase of about 10 percent over the nearly 5.54 million trips generated within the Region in 1991. Average weekday transit ridership may be expected to decline by about 4 percent, from 172,200 trips in 1991 to about 168,000 trips by 2010. Trips made as an automobile driver may be expected to increase by about 16 percent, from 4.06 million in 1991 to 4.72 million in 2010. Under the no-build plan, the number of average weekday internal vehicle trips may be expected to increase by about 16 percent, from 4.58 million in 1991 to 5.3 million by the year 2010.

Allocation of the vehicle travel demand generated under the adopted land use plan to the nobuild alternative indicates that vehicle-miles of travel on the arterial system may be expected to increase by about 35 percent, from 33.07 million in 1991 to nearly 44.52 million by 2010. A significant increase in the use of the regional freeway system may be expected to occur between 1991 and the year 2010 under the nobuild plan. Vehicle-miles of travel on the freeway system may be expected to increase by 49 percent, from 11.60 million vehicle-miles in 1991 to 17.29 vehicle-miles in 2010. The number of arterial miles operating over design capacity may be expected to increase by about 401 miles. from 279 in 1991, or about 9 percent of the total system mileage to 681 miles, or about 20 percent

of the total system mileage. The number of arterial miles operating at design capacity may be expected to increase by about 93 percent, from 264 miles in 1991, or 8 percent of the total system mileage in 1991, to about 510 miles, or about 15 percent of the total system mileage in 2010.

Under the no-build plan, the number of urbanized area residents within the Region able to access 40 percent of the urbanized area jobs within 45 minutes by public transit may be expected to decrease from 131,300, or 9 percent of the urbanized area population in 1991, to 108,400, or 7 percent of the urbanized area population in 2010.

Alternative Transportation

System Management Plans

Each of the alternative transportation system management plans contain several elements in common: comprehensive freeway traffic management, general traffic management, the recognition of the potential to eventually apply intelligent vehicle highway system technology to the management of travel within the Region, and travel demand management measures including work-schedule changes, telecommuting, and transportation management associations. The three alternatives differ in the degree to which they emphasize pricing measures and land use planning and site design to bring about reductions in travel demand. The plans also differ in the degree to which public transit service would be improved and expanded.

In an effort to manage travel demand, Alternative Plan 1 would, by the year 2010, increase perceived vehicle operating costs by 100 percent, from \$0.055 in 1991 to \$0.11 in 2010. This proposed cost increase would be the equivalent to a \$1.10 per gallon increase in the cost of motor fuel. The transit system envisioned under Alternative Plan 1 would consist of 4,190 round-trip route-miles, which would be about 1,894 miles, or 82 percent, greater than that provided in 1991. The transit system would require that 914 vehicles operate during peak travel periods, and that 141,900 revenue vehicle-miles and 9,200 revenue vehicle-hours of service be provided per average weekday. In 1991, 530 buses provided 63,300 revenue vehicle-miles and 5,220 revenue vehicle-hours of service per average weekday.

In an effort to manage travel demand, Alternative Plan 2 would, by the year 2010, increase the perceived vehicle operating cost by 50 percent, from \$0.055 cents per mile in 1991 to \$0.0825 per mile by 2010. The 50 percent increase in the perceived cost of vehicle operation proposed under Alternative Plan 2 would be equivalent to a \$0.55 per gallon increase in the cost of motor fuel. The transit system envisioned under Alternative Plan 2 would consist of 3,680 round-trip route-miles, about 1,384 miles, or 61 percent, greater than that provided in 1991. The transit system would require 804 vehicles operating during peak periods and providing 114,700 revenue vehicle-miles and 7,800 revenue vehiclehours of service per average weekday.

Alternative Plan 3, unlike the other alternatives, does not propose to increase perceived vehicle operating costs in an effort to manage travel demand, but rather proposes to increase such costs only to the extent necessary to cover the revenue shortfall anticipated to be incurred in implementing the plan. The transit system envisioned under Alternative Plan 3 would consist of 3,260 round-trip route-miles, some 964 miles, or about 42 percent, greater than that provided in 1991. The system would require 667 vehicles operating during peak periods and providing 96,900 vehicle revenue miles and 6,700 revenue vehicle-hours of service per average weekday.

<u>Testing of Alternative Transportation</u> System Management Plans

The performance of each of the three alternative transportation system management plans under future land use and travel conditions was analyzed with the assistance of the Commission traffic simulation models, as previously indicated. It is estimated that, by the year 2010, the number of vehicles available in the Region may be expected to increase from 1.13 million in 1991 to 1.29 million, or by about 14 percent, under Alternative Plan 1; to 1.29 million, or by about 14 percent, under Alternative Plan 2; and to 1.30 million, or by nearly 15 percent, under Alternative Plan 3.

Average weekday internal person trips may be expected to increase from nearly 5.54 million trips in 1991 to about 6.06 million by 2010, or by 9 percent, under Alternative Plan 1 and to about 6.10 million, or by nearly 10 percent, under Alternative Plans 2 and 3. The average number of internal person trips per capita may be expected to increase from about 3.1 in 1991 to 3.2 by 2010, or by about 3 percent, under each of the alternative plans. The average number of internal person trips per household may be expected to remain at 7.8 under Alternative Plan 1 and to increase to 7.9 under the other alternatives.

Average weekday transit trip production may be expected to increase from about 172,200 trips in 1991 to about 271,600 by 2010, or by about 58 percent, under Alternative Plan 1; to about 207,700 trips, or by about 21 percent, under Alternative Plan 2; and to about 186,900 trips, or by about 9 percent, under Alternative Plan 3. The proportion of all internal person trips within the Region served by transit may be expected to increase from 3.1 percent in 1991 to 4.5 percent by 2010 under Alternative Plan 1; to increase to 3.4 percent under Alternative Plan 2; and to remain at about 3.1 percent under Alternative Plan 3.

Internal automobile driver trips, including personal-use light truck driver trips, may be expected to increase from nearly 4.06 million trips in 1991 to about 4.58 million trips by 2010, or by about 13 percent, under Alternative Plan 1; to about 4.67 million trips, or by 15 percent, under Alternative Plan 2; and to about 4.69 million trips, or by about 16 percent, under Alternative Plan 3. The proportion of all internal person trips made as an automobile driver may then be expected to increase from 73.3 percent in 1991 to 75.5 percent by 2010 under Alternative Plan 1, to 76.7 percent under Alternative Plan 2, to 76.9 percent under Alternative Plan 3.

Internal vehicle trip production, including internal trips made by both automobiles and trucks, on an average weekday, may be expected to increase from 4.58 million trips in 1991 to about 5.16 million trips by 2010, or by about 13 percent, under Alternative Plan 1; to 5.25 million trips, or by about 15 percent, under Alternative Plan 2; and to 5.27 million trips, or also by about 15 percent, under Alternative Plan 3.

Arterial vehicle miles of travel within the Region on an average weekday may be expected to increase from about 33.07 million in 1991 to about 38.87 million by 2010, or by 18 percent, under Alternative Plan 1; to about 41.57 million, or by 26 percent, under Alternative Plan 2; and to about 44.09 million, or by 33 percent, under Alternative Plan 3.

The number of arterial miles operating over design capacity may be expected to increase from 279 in 1991 to about 601, or by 115 percent, under Alternative Plan 1; to 628 miles, or by 125 percent, under Alternative Plan 2; and to about 679 miles, or 143 percent, under Alternative Plan 3. The number of arterial miles operating at design capacity may be expected to increase from about 264 in 1991 to 536, or by 103 percent, under Alternative Plan 1; to 526 miles, or by 99 percent, under Alternative Plan 2; and to about 507 miles, or by 92 percent, under Alternative Plan 3.

The proportion of the total urbanized area population within the Region able to access 40 percent of jobs within 45 minutes by public transit may be expected to increase from 131,300 persons, or 9 percent, in 1991, to 825,100 persons, or about 54 percent, under Alternative Plan 1; to 481,200, or about 32 percent, under Alternative Plan 2; and to 391,200, or about 26 percent, under Alternative Plan 3. The proportion of the population within each urbanized area expected to meet the standard is shown in Table 202.

After careful consideration of the alternatives set forth in this chapter, the Technical Coordinating and Advisory Committee selected Alternative Plans 1 and 3 for further refinement and evaluation, including the incorporation of necessary arterial street and highway improvements to resolve identified transportation deficiencies, particularly traffic congestion. In selecting Alternative Plans 1 and 3, the Technical Coordinating and Advisory Committee recommended the elimination of Alternative Plan 2 from further consideration because its performance was not sufficiently different from the performances of Alternative Plans 1 and 3. This set of options would not provide the Commission, elected officials, and the public with information sufficient to permit selection of a recommended plan and because the proposed pricing mechanism assumed for demand management purposes under Plan 2 was insufficient to effect the degree of change in travel behavior sought by proponents of demand management in the Advisory Committee. The Advisory Committee determined that Alternative Plans 1 and 3 presented a reasonable range of alternatives upon which to base the preliminary selection of a recommended plan. A detailed description of the alternatives selected, including proposed arterial street and highway improvements, is set forth in Chapter X of this report.

Chapter X

ALTERNATIVE PLAN COMPARISON AND EVALUATION

INTRODUCTION

As described in Chapter IX, the design of alternative transportation system plans proceeded from an analysis of how effectively transportation system management measures alone could serve future travel demand. This chapter describes the alternative plans selected by the Advisory Committee for further analysis complete with attendant improvements to the arterial street and highway system. This chapter also describes the results of an evaluation of the alternatives plans and compares the plans against the adopted development objectives and design standards.

Upon careful consideration and extensive deliberation, the Technical Coordinating and Advisory Committee on Regional Transportation System Planning advised the Commission to evaluate Alternative Transportation System Management Plans 1 and 3 fully. The Committee directed that the full evaluation of these plan alternatives follow the inclusion of arterial street and highway system improvements necessary to resolve the remaining deficiencies in each plan. As described in Chapter IX, the Advisory Committee reasoned that Alternative Plans 1 and 3 presented appropriate ranges of alternatives upon which to base a selection of a preliminary recommended plan. Alternative Plans 1 and 3 are fully described in this chapter and summarized in Table 203. The no-build alternative transportation system plan is fully described in the previous chapter of this report.

ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM PLAN 1

Alternative Plan 1 represents a significant departure from the historic and existing transportation development policy within the Region. The plan emphasizes the management of travel demand. Because Commission studies have shown that travel behavior is largely a function of the perceived cost of travel, the plan proposes to increase those costs to bring about changes in travel behavior which have been shown to help meet transportation system development objectives. Additionally, the plan proposes specific changes in current land use practices within the Region, consistent with the adopted regional land use plan. Such changes would promote the development of compact communities and encourage the use of bicycling, walking, and transit. Alternative Plan 1 also proposes other travel demand management measures, including, telecommuting, ridesharing, trip reduction ordinances, and transportation management associations.

To accommodate anticipated changes in travel patterns, the travel demand management measures may engender and to encourage further reductions in single-occupancy vehicle travel, Alternative Plan 1 also proposes to nearly double the level of public transit service. More specifically, the plan proposes to develop comprehensive rapid and express transit systems connecting major concentrations of urban development to the Milwaukee central business district (CBD). The plan proposes to expand local transit service throughout the urbanized areas of the Region and to improve the frequency of all levels of transit service.

Alternative Plan 1 also includes recommendations to maintain and improve the arterial street and highway system in the region. The plan proposes to expand arterial capacity where it was determined that the full implementation of transit improvement recommendations and transportation system management resources, including pricing, could not be anticipated to alleviate future traffic congestion. Because of the extensive reliance on transit improvement and transportation system management measures, most notably pricing, fewer improvements to the arterial system were determined to be needed under Alternative Plan 1 than under Alternative Plan 3.

Transportation System Management

Alternative Plan 1 proposes a number of transportation system management measures to reduce the demand for travel, travel demand management, and to ensure that full use is achieved of existing transportation facilities before commitments are made to new capital investment. The proposed measures may be grouped into four categories: pricing, land use

SUMMARY OF MAJOR PLAN ELEMENTS COMPRISING THE NO-BUILD AND ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM PLANS FOR SOUTHEASTERN WISCONSIN: 2010

Major Pl	an Element			
Category	Specific Measure	No-Build Alternative	Alternative Plan 1	Alternative Plan 3
Transportation System Management	Traffic management	 Implementation of Milwaukee-area freeway traffic management system 	 Implementation of Milwaukee-area freeway traffic management system 	 Implementation of Milwaukee-area freeway traffic management system
		 Curb parking restrictions on 291 arterial route-miles during peak travel periods 	 Curb parking restrictions on 458 arterial route-miles during peak travel periods 	 Curb parking restrictions on 464 arterial route-miles during peak travel periods
		 Application of traffic engineering techniques, e.g., turning lanes, traffic control signals, and coordinated timing of signals 	 Application of traffic engineering techniques, e.g., turning lanes, traffic control signals, and coordinated timing of signals 	 Application of traffic engineering techniques, e.g., turning lanes, traffic control signals, and coordinated timing of signals
	Pricing	 No increase in perceived cost of operating an automobile 	 Significant increase in perceived cost of automobile operation from 5.5 cents per mile to 11.0 cents per mile; could be achieved by a variety of measures equivalent to motor fuel tax increase of \$1.10 per gallon; cost increase intended to reduce automobile travel 	 Insignificant increase in perceived cost of operating an automobile from 5.5 cents per mile to 6.0 cents per mile; equivalent to motor fuel tax increase of 10 cents per gallon; cost increase not intended to reduce automobile travel
Areawide promotional tatior assoc		 Minimal effort to promote ridesharing and transpor- tation management associations 	 Aggressive areawide effort to promote ridesharing, transit use, bicycle use, telecommuting, work-time rescheduling, and trans- portation management associations 	 Aggressive areawide effort to promote ridesharing, transit use, bicycle use, telecommuting, work-time rescheduling, and trans- portation management associations
	Land use	 Modest effort to promote travel by transit and to facilitate bicycle and pedestrian travel through detailed site-specific neighborhood land use planning, including appropriate mixtures of land use 	 Aggressive effort to promote travel by transit and to facilitate bicycle and pedestrian travel through detailed site- specific neighborhood land use planning, including provision of appropriate mixtures of land use, efficient and direct pedestrian and bicycle pathways, and higher land use densities along transit lines 	 Modest effort to promote travel by transit and to facilitate bicycle and pedestrian travel through detailed site-specific neighborhood land use planning, including appropriate mixtures of land use
Public Transit System Mainte- nance and Improvement	Rapid transit	 Continue service within Milwaukee County in major corridors and to Milwaukee central business district from Oak Creek, Wauke- sha, Mukwonago, Ocono- mowoc, and Menomonee Falls, and from sites in River Hills and Glendale in the IH 43-North Corridor 	• Expand service in all major corridors to Milwaukee central business district from Racine, Kenosha, East Troy, Waukesha, Oconomowoc, West Bend, Cedarburg, Grafton, and Port Washington	 Provide service in major corridors to Milwaukee central business district from Racine, Kenosha, Waukesha, Oconomowoc, Mukwonago, Germantown, Cedarburg, and Grafton

Table 203 (continued)

Major Pla	an Element			
Category	Specific Measure	No-Build Alternative	Alternative Plan 1	Alternative Plan 3
Public Transit System Mainte- nance and Improvement (continued)	Rapid transit (continued)	• <u>Service Hours</u> ^a Weekdays – 6:00 a.m. to 8:30 a.m., 3:30 p.m. to 6:00 p.m.	• <u>Service Hours</u> Weekdays-6:00 a.m. to 6:00 p.m. (evening and weekend service on some routes)	• <u>Service Hours</u> Weekdays – 6:00 a.m. to 8:30 a.m., 3:30 p.m. to 6:00 p.m. (midday service over some routes; no weekend or evening service)
		• <u>Headways</u> Peak—15 to 60 minutes	 <u>Headways</u> Peak—five to 30 minutes Off-peak—30 to 60 minutes 	 <u>Headways</u> Peak—five to 30 minutes Off-peak—30 to 60 minutes
		 Fares Within Milwaukee County, \$1.50; Milwaukee County to limits of Milwaukee urbanized area, \$2.00; Milwaukee central business district to outer limits, \$2.50 	 <u>Fares</u> Within Milwaukee County, \$1.25; Milwaukee County to limits of urbanized area, \$1.75; Milwaukee central business district to outer limits, \$2.50 	 <u>Fares</u> Within Milwaukee County, \$1.50; Milwaukee County to limits of urbanized area, \$2.00; Milwaukee central business district to outer limits, \$3.00
	Express transit	 Continue Milwaukee-area central business district- oriented service and special UW-Milwaukee service; service between Oconomowoc, Waukesha, and Milwaukee central business district; and between Milwaukee central business district, Racine, and Kenosha 	• Expand Milwaukee-area central business district- oriented service; provide cross-town service in Milwaukee County; provide service between Racine and Kenosha, and to connect Racine and Kenosha with urbanizing areas along IH 94	• Expand Milwaukee central business district-oriented service; provide cross- town service in Milwaukee County; provide service to connect Cities of Racine and Kenosha; no connection of such cities to urbanizing areas along IH 94
		• <u>Service Hours</u> ^a Weekdays-6:00 a.m. to 10:00 p.m. (weekend service on some routes)	• <u>Service Hours</u> Weekdays-6:00 a.m. to 6:00 p.m. (weekdays until 10:00 p.m. and weekends on some routes)	• <u>Service Hours</u> Weekdays-6:00 a.m. to 6:00 p.m. (weekdays until 10:00 p.m. and weekends on some routes)
• <u>Heady</u> Peak- (60 m Racini areas) Off-pe minut minut Racini areas)		 Headways Peak—10 to 15 minutes (60 minutes in Waukesha, Racine, and Kenosha areas) Off-peak—15 to 20 minutes (60 to 120 minutes in Waukesha, Racine, and Kenosha areas) 	 Headways Peak—five to 15 minutes (30 minutes in Racine and Kenosha areas) Off-peak—15 to 30 minutes (60 minutes in Racine and Kenosha areas) 	 <u>Headways</u> Peak-10 to 15 minutes (30 minutes in Racine and Kenosha areas) Off-peak-20 to 30 minutes (60 minutes in Racine and Kenosha areas)
		• Fares Milwaukee County, \$1.25; Milwaukee County to limits of Milwaukee urbanized area, \$2.00; Milwaukee central business district to outer limits, \$2.50 to \$4.00	 Fares Milwaukee County, \$1.00; Racine and Kenosha, \$0.60 	 <u>Fares</u> Milwaukee County, \$1.25; Racine and Kenosha, \$0.75

Table 203 (continued)

Major Pla	an Element			
Category	Specific Measure	No-Build Alternative	Alternative Plan 1	Alternative Plan 3
Public Transit System Mainte- nance and Improvement (continued)	Local transit	• Continue fixed-route service in Milwaukee County and in Cities of Waukesha, Racine, and Kenosha; continue shared- ride taxi service in Cities of Hartford, West Bend, Whitewater, and Port Washington	• Extend fixed-route service to all high- and medium- density development within Milwaukee, Racine, and Kenosha urbanized areas; new service to be provided to northern and southern Milwaukee County, southern Ozaukee County, southeastern Washington County, and eastern Waukesha County; continue shared-ride taxi services	• Extend fixed-route service to all high- and medium- density development with densities of five dwelling units per acre or greater; new service to be provided to portions of northern and southern Milwaukee County, and eastern Waukesha County contiguous to existing service areas; continue shared-ride taxi services
		 Headways Peak-10 to 40 minutes in Milwaukee County; 30 to 60 minutes in Waukesha; 30 minutes in Kenosha; 20 to 30 minutes in Racine Off-peak-20 to 45 minutes in Milwaukee County; 30 to 60 minutes outside Milwaukee County 	Headways Peak—10 minutes in central Milwaukee County; 15 to 30 minutes outside central Milwaukee County Off-peak—20 to 60 minutes in Milwaukee County; 30 to 60 minutes outside Milwaukee County	 <u>Headways</u> Peak-10 to 40 minutes in Milwaukee County; 30 minutes in Waukesha; 20 to 30 minutes in Kenosha and Racine Off-peak-30 to 45 minutes in Milwaukee County; 60 minutes outside Milwaukee County
	Total transit	• <u>Average Weekday</u> Round-trip route- miles: 2,440	<u>Average Weekday</u> Round-trip route- miles: 4,190	 <u>Average Weekday</u> Round-trip route- miles: 3,260
		Vehicle-miles of service: 65,990	Vehicle-miles of service: 141,900	Vehicle-miles of service: 96,900
		Vehicle-hours of service: 5,700	Vehicle-hours of service: 9,200	Vehicle-hours of service: 6,700
		Vehicles required in peak period: 527	Vehicles required in peak period: 915	Vehicles required in peak period: 670
	Rail transit considerations	 No provision of transit service by light rail or commuter rail 	 Potential provision of commuter rail service in four travel corridors dependent upon detailed corridor studies 	 Potential provision of commuter rail service in four travel corridors dependent upon detailed corridor studies
			 Potential provision of light rail express transit service in six travel corridors dependent upon detailed corridor studies 	 Potential provision of light rail express transit service in six travel corridors dependent upon detailed corridor studies
Arterial Street and Highway System Mainte- nance and Improvement	New facilities	 11 route-miles of newly constructed facilities 	 112 route-miles of newly constructed facilities 	 124 route-miles of newly constructed facilities
	Widened facilities	• 47 route-miles of widened arterial street and highway facilities	 382 route-miles of widened arterial street and highway facilities 	 410 route-miles of widened arterial street and highway facilities
	Preserved facilities	• 3,422 route-miles of facilities to be preserved through resurfacing or reconstruction for same capacity	 3,104 route-miles of facilities to be preserved through resurfacing or reconstruction for same capacity 	 3,075 route-miles of facilities to be preserved through resurfacing or reconstruction for same capacity
	Total facilities	 Total system of 3,480 route-miles of arterial streets and highways 	 Total system of 3,598 route-miles of arterial streets and highways 	 Total system of 3,609 route-miles of arterial streets and highways

^aThe Milwaukee County Transit System would continue to operate one freeway flyer rapid transit bus route between the Milwaukee central business district and Northridge Shopping Center during nonpeak periods at 60-minute headways.

Source: SEWRPC.

planning and site design, areawide promotional programs, and traffic management.

Pricing: Alternative Plan 1 assumes a substantial increase in the perceived cost of driving a personal vehicle as measure to moderate vehicle travel demand. Under Alternative Plan 1, the perceived cost of driving would double, to \$0.11 per mile. The cost increase could result from a combination of pricing related travel demand management measures, such as cash-out of employer paid parking, pay-as-you-drive automobile insurance, vehicle registration fees, and motor fuel taxes. For perspective, should the entire 100 percent cost increase be achieved through use of the motor fuel tax, the addition to that tax would approximate \$1.10 per gallon. It is envisioned under Alternative Plan 1 that the revenue generated through these measures would reduce or eliminate the need for transportation funding from the property tax and provide adequate funding for full transportation system plan implementation.

Land Use Planning and Site Design Measures: Alternative Plan 1 also emphasizes significant land use-related travel demand management measures to reduce vehicle-miles of travel, vehicle trips, and average trip lengths by reducing the need for vehicular travel and by enhancing the attractiveness of alternatives to personal vehicle use. Since land use development practices significantly influence travel behavior and fall within the legal purview of local governmental police power, land use development is a variable that is proposed to be significantly managed under Alternative Plan 1 to achieve transportation system development objectives.

Implementation of appropriate land use measures by county and local units of government under Alternative Plan 1 would provide: that all newly developing areas be designed to provide integrated bicycle and pedestrian systems, that building and sites be oriented to promote effective transit provision and use, and that an appropriate mix of land use activities be encouraged in newly developing areas. In addition, Alternative Plan 1 proposes that high-density residential development and employment locations be provided along all existing and new transit routes. Such land use measures would permit travellers in areas affected to travel more easily and safely by walking, bicycle, or transit and to make shorter automobile trips than are required under more conventional recent development patterns.

In the design of Alternative Plan 1, the travel forecasting models were adjusted to reflect the probable impact of the land use measures on travel demand. Specifically, assumed transit walk-access was reduced by one-third to reflect the planned increase in the convenience of transit travel made possible by the development of higher residential densities along transit lines. This reduction was incorporated into the mode choice model. Further, the land use planning and site design measures were assumed to reduce, by 5 percent, total trip generation within areas so affected. The trip generation model was adjusted accordingly.

Areawide Promotional Measures: Alternative Plan 1 recommends a coordinated areawide effort to promote travel through ridesharing, transit use, and bicycle use. This program would aggressively promote carpooling and vanpooling, telecommuting, and rescheduling of work time to reduce travel by automobile, particularly by single-occupancy automobile, particularly in the peak periods. In addition, this program would include the promotion and support of transportation management associations at major employment centers throughout the Region to provide an institutional structure for helping to achieve the goal of reducing travel by single-occupancy vehicles.

Traffic Management: Alternative Plan 1 also proposes the use of traffic management measures to accommodate vehicular travel demand not abated through the use of travel demand management measures and public transit improvements. Traffic management measures are intended to ensure that available highway capacity is used prior to the construction of additional capacity. The traffic management measures of the Alternative Plan 1 include restrictions on peak-period curb-lane parking; the implementation of the Milwaukee area comprehensive freeway traffic management system plan; and the extensive application of traffic engineering measures by State, county, and municipal agencies responsible for plan implementation. The plan recommends that curb restrictions be placed on a total of 458 arterial route-miles during peak periods. The plan also recommends the use of advanced traffic management technology, Intelligent Transportation Systems (ITS), as such technology becomes practical within the Region.

	Existing	g 1991	Planned I	ncrement	20	10	
		Percent		Percent	-	Percent	
Transit Service	Number	of Total	Number	Change	Number	of Total	
Round-Trip Route Length (miles)							
Rapid Routes	449	19.6	1,131	251.9	1,580	37.7	
Express Routes	393	17.1	97	24.7	490	11.7	
Local Routes							
Kenosha Urbanized Area	171	7.4	69	40.4	240	5.7	
Milwaukee Urbanized Area	1,112	48.5	548	49.3	1,660	39.6	
Racine Urbanized Area	171	7.4	49	28.7	220	5.3	
Subtotal	1,454	63.3	666	45.8	2,120	50.6	
Total	2,296	100.0	1,894	82.5	4,190	100.0	
Special Facilities (miles)					. 1		
Exclusive Right-of-Way	• O	0.0	30		30	93.8	
Exclusive Lanes on Streets	2	100.0	0	0.0	2	6.2	
Total	2	100.0	30	1,500.0	32	100.0	
Average Weekday Vehicle Requirements ^a						, ,	
Peak Period	530		384	72.5	914		
Midday Off-Peak Period	285		216	75.8	501	- -	

TRANSIT SYSTEM OPERATING CHARACTERISTICS AND FACILITIES IN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup vehicles.

Source: SEWRPC.

Public Transit System

Maintenance and Improvement

Alternative Plan 1 proposes a near doubling of existing transit service within the Region. The public transit element of this alternative is shown on Map 92 on page 398 in Chapter IX and is summarized in Table 204. The transit system envisioned under Alternative Plan 1 would consist of about 4,190 round-trip routemiles, 1,894 miles, or 82 percent, more than provided in 1991. The transit system would require that 914 vehicles operate during peak periods and would provide about 141,900 revenue vehicle-miles and 9,200 revenue vehicle-hours of service per weekday. The base year 1991 system required that 530 buses operate during peak periods and provided about 63,300 revenue vehicle-miles and 5,220 revenue vehicle-hours of service per weekday.

<u>Rapid Transit Service</u>: Under Alternative Plan 1, a rapid transit system would be developed to connect the major outlying activity centers with the Milwaukee CBD. The existing network of bus rapid transit routes would be expanded to provide service in all major travel corridors emanating from the Milwaukee CBD: to the south to the Cities of Racine and Kenosha, to the southwest to the Village of East Troy, to the west to the Cities of Waukesha and Oconomowoc, to the northwest to the City of West Bend, and to the north to the Cities of Cedarburg and Port Washington and the Village of Grafton.

Within each of these major travel corridors, bus rapid transit service would be provided over operationally controlled freeway lanes in mixed traffic, or over exclusive busways. In each rapid transit corridor, a busway capable of accommodating buses and high-occupancy vehicles (HOVs) would be considered if the freeway within the corridor is determined to be operating under severe traffic congestion and if the busway is determined to have the potential to accommodate more travel than the existing freeway lanes, as described in Chapter IX of this report. The extent of potential busway facilities in the Milwaukee area is shown on Map 95. Map 95



EXTENT OF POTENTIAL BUSWAY AND LIGHT-RAIL/EXPRESS BUS GUIDEWAY FACILITIES IN THE MILWAUKEE AREA

Under both Alternative Plans 1 and 3, rapid transit busway facilities and express transit light-rail facilities would be considered as an alternative to providing transit service by motor bus over arterial highway lanes. Consideration of such fixed-guideway transit service would be initiated as part of Federally required major investment studies for each of the identified corridors.

The bus rapid transit system would consist of 35 routes totaling 1,580 round-trip route-miles. The system would serve 74 public transit stations or stops, located to the extent practicable within or near freeway interchanges to minimize travel times. Three of the 35 rapid transit routes envisioned under the plan would provide special service directly to the University of Wisconsin-Milwaukee campus. The other 32 routes would be focused on the Milwaukee CBD.

In addition, commuter railway passenger train service is recommended as a potential alternative to bus-on-freeway rapid transit service in four corridors emanating from the Milwaukee CBD: to the south to the Cities of Racine and Kenosha, to the west to the City of Oconomowoc, to the northwest to the City of West Bend, and to the north to the City of Cedarburg. Pursuant to Federal law, the ultimate decision to provide such facilities would be made following the completion of major investment studies in each corridor. The corridor studies would compare the costeffectiveness of commuter rail to bus-on-freeway service and would also establish the financial feasibility of the commuter railway passenger train service in each identified corridor.

Alternative Plan 1 also envisions a significant improvement in existing rapid transit service levels: the planned weekday hours of operation for all routes, 6:00 a.m. to 6:00 p.m. Service would also be provided on weekday evenings until 10:00 p.m., as well as on weekends over selected routes within corridors connecting the Cities of Waukesha, Racine, and Kenosha to the Milwaukee CBD. Headways, the time interval between the departure of successive buses at transit stops or stations, during weekday peak periods would range from five to 30 minutes. At all other times, headways would range from 30 to 60 minutes. Service would be provided in both travel directions over all routes.

<u>Express Transit</u>: The development of a comprehensive express transit system is also proposed under Alternative Plan 1. Express transit in the form of express bus service in mixed arterial traffic and over exclusive lanes would be provided in major urbanized area travel corridors. Consideration of bus on guideway or light rail transit service would be initiated as part of major investment studies for those express transit corridors shown on Map 95. Express transit service totaling about 490 roundtrip route-miles would be provided over 15 routes within the Milwaukee, Racine, and Kenosha urbanized areas. Within Milwaukee County, the routes would be located so as to provide crosstown service, as well as service oriented to the Milwaukee CBD. Within Milwaukee County, express transit routes would be operated with headways ranging from five to 15 minutes during weekday peak periods and 15 to 30 minutes during off-peak periods. Express transit service over some routes would replace local bus service. Basic hours of operation would be similar to those proposed for rapid transit service.

Within the Racine and Kenosha urbanized areas, express service would provide a direct connection between the Cities of Racine and Kenosha. Express routes would also connect these cities with new and expanding commercial centers located along IH 94 in the western portion of each urbanized area. Express transit service in these areas would be operated at headways of 30 minutes during weekday peak periods and 60 minutes during weekday midday off-peak periods. In all areas, buses providing express transit service would, where appropriate, also provide a collection and distribution function beyond the proposed termini of the express routes.

Local Transit Service: The local transit service proposed under Alternative Plan 1 also represents a significant expansion over existing service. Within the Milwaukee, Kenosha, and Racine urbanized areas, local transit service would be extended into all contiguous areas of high- and medium-density urban development. Within the Milwaukee urbanized area, local bus service would thus be extended into northern and southern Milwaukee County, southern Ozaukee County, southeastern Washington County, and eastern Waukesha County.

Alternative Plan 1 envisions that within portions of Milwaukee County currently served by local bus service, a grid system of local bus routes would continue to be operated. Where local transit service would be extended into portions of northern and southern Milwaukee County, as well as adjacent portions of Waukesha, Washington, and Ozaukee Counties, the plan envisions that either a grid system of routes or a transit center-based route system would be developed.

Should a transit-center system be established, the centers would serve as focal points for buses serving a particular area, much like the CBDs of the Cities of Kenosha, Racine, and Waukesha serve as the focal points for the local bus routes serving these areas. The transit centers would be located at rapid transit stations, at the termini of express transit routes, or at major activity centers. These locations would allow local bus routes serving each center to provide convenient collection and distribution transit services.

While the majority of expanded local transit service would be within Milwaukee County and eastern Waukesha County, the plan envisions that a limited level of service would be extended into the Germantown area of Washington County and the Mequon, Cedarburg, and Grafton areas of Ozaukee County. Additionally, special local circulator transit services would connect various land uses within areas of highintensity development which are envisioned to be developed along IH 94 in western Waukesha County in the future.

Within the Kenosha and Racine urbanized areas, transit service would be extended to contiguous areas which are anticipated to be fully developed in the future. These areas include future commercial, office, and industrial development around major interchanges along IH 94. The westward expansion of local transit service would be facilitated by the development of additional transit centers for the Racine and Kenosha transit systems located in the central portion of each urbanized area closer to areas of new urban development along IH 94.

The areas served by the expanded local transit services proposed under this alternative are shown on Map 92 on page 398 of Chapter IX. The proposed local transit service in the Milwaukee urbanized area would consist of approximately 1,660 round-trip route-miles. Service levels on existing local transit routes in the Milwaukee urbanized area would be substantially improved over existing 1991 levels, particularly within the portion of Milwaukee County south of Silver Spring Drive, east of 76th Street, and north of Layton Avenue. In corridors where express service is not also provided, headways on all local transit routes within this central area would be established at maximums of 10 minutes during the morning and afternoon peak periods. 20 minutes during the midday off-peak periods, and 30 minutes during all other times of operation. Outside of this area, headways on local transit routes would range from 15 to 30 minutes during weekday peak periods and from 30 to 60 minutes during all other off-peak periods of operation. The hours of operation for local transit routes serving the City of Waukesha would also be extended to include evening service.

Within the Racine and Kenosha urbanized areas, the expanded local transit systems would consist of a total of approximately 240 and 220 round-trip route-miles, respectively. Headways on local routes serving these areas would range from 15 to 30 minutes during peak periods and 30 to 60 minutes during off-peak periods. Hours of operation in the Cities of Racine and Kenosha would be extended to include evening service.

Alternative Plan 1 also proposes to maintain the existing demand responsive shared-ride taxi service within the Cities of Hartford, Port Washington, West Bend, and Whitewater. The plan proposes to provide similar local transit service in other urban areas of like size and in some lower-density portions of the Milwaukee urbanized area as demand for such service may warrant.

Transit Fares: Transit fares under this transit alternative were assumed to be consistent with fares charged by transit operators within the Region in 1991. The base adult fare, expressed in constant 1994 dollars, within Milwaukee County was assumed to be \$1.00 per ride for local and express bus service and \$1.25 per ride for rapid transit service. Fares for rapid transit service between Milwaukee County and surrounding counties would increase with distance traveled outside of Milwaukee County, generally ranging from about \$1.75 per ride at the outer limits of the future urbanized area to about \$2.50 at the extreme limits of service from the Milwaukee CBD. Local transit fares in the Cities of Racine, Kenosha, and Waukesha and their immediate environs were assumed to be consistent with 1991 fares expressed in 1994 dollars. Accordingly, such fares would range from \$0.50 per ride on the Racine transit system to \$0.60 per ride on the Kenosha and Waukesha transit systems.

Arterial Street and Highway System Maintenance and Improvement

Despite an extensive commitment to transportation systems management and public transit improvement, the large majority of travel in the Region in the year 2010 may be expected to occur on the arterial street and highway system. That system under Alternative Plan 1 would total 3,599 route-miles and 10,071 lane-miles of facilities (see Table 205). The arterial street and highway maintenance and improvement recommendations are graphically presented on Map 96 and summarized below.

New Arterial Facilities: The plan would provide for the construction of 112 route- and 266 lanemiles of new arterial street and highway facilities. As shown on Map 96, these include such long-planned facilities as the Lake Arterial highway south from the Hoan Bridge to E. Layton Avenue, the USH 12 bypass of Whitewater, the STH 16 Freeway bypass of Oconomowoc, the Waukesha bypass, and the STH 36 bypass of Burlington. Excluded from this alternative are previously planned new arterial facilities, including the USH 12 Freeway extension in Walworth County, the STH 83 bypass of Beaver Lake in Waukesha County, and the Lake Arterial through southern Milwaukee County and eastern Racine and Kenosha Counties. In all, the proposed new arterial highway facilities would represent about 3 percent of the total planned arterial route-miles and nearly 3 percent of the total planned arterial lane-miles.

<u>Widening of Existing Arterial Facilities</u>: Also under Alternative Plan 1, a total of 382 routemiles of facilities would be widened to provide an additional 1,571 arterial lane-miles. This represents about 11 percent of the total route-miles and about 16 percent of the total lane-miles. As shown on Map 96, this includes many long-planned arterial widening projects, such as the STH 36 widening in Milwaukee, Waukesha, and Racine Counties, the STH 31 widening in Kenosha and Racine Counties, the STH 57 widening in Ozaukee County, and the completion of the STH 50 widening in Kenosha and Walworth Counties.

<u>Preserving Arterial Streets and Highways</u>: With respect to the remaining 3,105 route-miles and 8,234 lane-miles of facilities envisioned under Alternative Plan 1, representing about 86 percent of the total route-miles and 82 percent of the total lane-miles, this alternative plan calls only for resurfacing or, as may be necessary because of pavement deterioration, reconstruction to provide the same capacity. These facilities are also shown on Map 96. Important among such preservation actions would be the resurfacing and bridge replacement work needed to maintain the Milwaukee-area freeway system.

<u>Assignment of Travel to</u> Highway and Transit Networks

An analysis of the performance of Alterative Plan 1 under probable future land use and travel conditions was undertaken with the aid of the traffic simulation models described in Chapter VII of this report. Through application of these models, travel demand for the design year 2010 was developed based upon the land use pattern proposed under the adopted regional land use plan and assigned to the arterial street and highway and transit networks.

Vehicle Availability: Given transportation system development in accordance with Alternative Plan 1 and land use development in accordance with the adopted regional land use plan, it may be expected that the number of automobiles and light trucks available for personal use within the Region may increase from about 1.13 million in 1991 to about 1.29 million in 2010, an increase of about 14 percent (see Table 206). This increase compares with an approximate 6 percent increase in resident population, an 8 percent increase in the number of households, and 11 percent increase in the number of jobs in the Region between 1991 and 2010. The number of persons per vehicle may be expected to continue to decline by 8 percent, from 1.60 in 1991 to 1.48 in 2010. The number of vehicles per household may be expected in increase by 5 percent, from 1.58 in 1991 to 1.66 in 2010.

Person Trip Generation: Given transportation system development in accordance with Alternative Plan 1 and land use development in accordance with the adopted regional land use plan, internal person trips may be expected to increase from nearly 5.54 million trips per day in 1991 to about 6.06 million trips by 2010, or by 9 percent. The average number of trips per capita may then be expected to increase from about 3.1 in 1991 to about 3.2 in 2010. The average number of internal person trips per household may be expected to remain at about 7.8. Future internal person trip production within the Region by trip purpose under this alternative plan is indicated in Table 207 and compared to existing trip production.

<u>Mode of Travel</u>: A comparison of the distribution of internal trips in the Region by mode of travel under the adopted regional land use plan and Alternative Plan 1 is set forth in Table 208. Average weekday transit trip production within



The arterial street and highway system under Alternative Plan 1 would total 3,599 route-miles and 10,071 lane-miles of facilities. The plan would provide for the construction of 112 route-miles and 266 lane-miles to widen 382 route-miles, providing an additional 1,571 lane-miles, and also for the preservation of 3,105 route-miles and 8,234 lane-miles of arterial street and highway facilities. Important among the preservation activities would be the resurfacing and interchange and bridge replacement work needed to maintain the Milwaukee area freeway system.

ARTERIAL STREET AND HIGHWAY FACILITIES IN THE REGION BY COUNTY 1991 AND 2010 ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM PLAN 1

	Arterial Streets and Highways							
	Existin	g 1991	Planned I	ncrement	Total	2010		
Arterial Facility Type	Miles	Percent of Total	Miles	Percent Change	Miles	Percent of Total		
Kenosha County								
Freeway Four-Lape								
Six-Lane	12.0	3.8			12.0	3.4		
Eight-Lane								
Subtotal	12.0	3.8			12.0	3.4		
Standard Arterial	0504			0.7	040 5			
I wo-Lane	49.6	15.6	-9.6	-3.7 87.1	246.5 92.8	69.2 26.0		
Six-Lane			4.9		4.9	1.4		
Subtotal	305.7	96.2	38.5	12.6	344.2	96.6		
County Subtotal	317.7	100.0	38.5	12.1	356.2	100.0		
Milwaukee County Freeway								
Four-Lane	13.9	1.8	-3.0	-21.6	10.9	1.4		
Six-Lane	53.9	6.9	3.0	5.6	56.9 1 4	7.1 0.2		
Subtotal	69.2	8.9	0.0	0.0	69.2	8.7		
Standard Arterial	00.2		0.0		00.2	0.7		
Two-Lane	379.0	48.9	-63.2	-16.7	315.8	39.7		
Four-Lane	288.9	37.3	52.6	-18.2	341.5	43.0		
Six-Lane	38.3	4.9	28.7	/4.9	67.0 1 4	8.4		
Subtotal	706.2	91 1	19.5	2.8	725.7	91.3		
County Subtotal	775.4	100.0	19.5	2.5	794.9	100.0		
Ozaukee County					70.10			
Freeway								
	27.4	9.5	0.0	0.0	27.4	8.8		
Eight-Lane								
Subtotal	27.4	9.5	0.0	0.0	27.4	8.8		
Standard Arterial								
Two-Lane	241.3	83.6	-11.4	-4.7	229.9	74.0		
Four-Lane	19.8	6.9	33.4	168.7	53.2	17.1		
Subtotal	261.1	90.5	22.0	8.4	283.1	91.2		
County Subtotal	288.5	100.0	22.0	7.6	310.5	100.0		
Racine County								
Four-Lane								
Six-Lane	12.0	3.4	0.0	0.0	12.0	2.9		
Eight-Lane	•••			-,-				
Subtotal	12.0	3.4	0.0	0.0	12.0	2.9		
Standard Arterial Two-Lane	301.6	86.7	43	14	305 9	73.0		
Four-Lane	31.1	9.0	59.9	192.6	91.0	21.7		
Six-Lane	3.2	0.9	6.9	215.6	10.1	2.4		
Subtotal	335.9	96.6	71.1	21.2	407.0	97.1		
County Subtotal	347.9	100.0	71.1	20.4	419.0	100.0		

Table 205 (continued)

	Arterial Streets and Highways							
	Existing	1991	Planned I	ncrement	Total	2010		
		Percent	· · ·	Percent	· · ·	Percent		
Arterial Facility Type	Miles	of Total	Miles	Change	Miles	of Total		
Walworth County Freeway	-							
Four-Lane	50.0	11.7	0.0	0.0	50.0	10.6		
	50.0	11.7	0.0	0.0	50.0	10.6		
Subtotal	50.0		0.0	0.0	50.0	10.0		
	370.5	86.3	5.5	1.5	376.0	79.4		
Four-Lane	8.7	2.0	38.7	444.8	47.4	10.0		
Six-Lane								
Subtotal	379.2	88.3	44.2	11.7	423.4	89.4		
County Subtotal	429.2	100.0	44.2	10.3	473.4	100.0		
Washington County Freeway		· .			. ,	· ·		
Four-Lane	15.1	3.8	21.1	139.7	36.2	7.7		
Six-Lane	0.5	1.6	0.0	0.0	6.5	1.4		
Subtotal	21.6	5.4	21.1	97.7	42.7	9.1		
Standard Arterial					· _ · ·			
Two-Lane	340.0	85.2	37.6	11.1	377.6	80.4		
Four-Lane	37.6	9.4	11.9	31.6	49.5	10.5		
Subtotal	377.6	94.6	49.5	13.1	427.1	90.9		
County Subtotal	399.2	100.0	70.6	17.7	469.8	100.0		
Waukesha County Freeway								
Four-Lane	45.6	6.3	6.3	13.8	51.9	6.7		
Six-Lane	13.4	1.9	0.0	0.0	13.4	1.7		
Subtotal	59.0	8.3	63	10.7	65.3	8.4		
Standard Arterial	0010	0.0						
Two-Lane	600.2	83.8	-79.0	13.2	521.2	67.2		
Four-Lane	52.2	7.3	118.4	226.8	170.6	22.0		
Six-Lane	4.9	0.7	10.0	204.1	14.9	1.9		
Subtotal	657.3	91.8	52.5	8.0	709.8	91.6		
County Subtotal	716.3	100.0	58.8	8.2	775.1	100.0		
Southeastern Wisconsin Begion	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,							
Freeway		-						
Four-Lane	152.0	4.6	24.4	16.1	176.4	4.9		
Six-Lane	97.8	3.0	3.0	3.1 0.0	100.8	2.8		
Subtotal	251.2	77	27 4	10.9	278.6	7.7		
Standard Arterial	201.2		<u> </u>	+	270.0			
Two-Lane	2,488.7	76.0	-115.8	-4.7	2,372.9	65.9		
Four-Lane	487.9	14.9	358.1	73.4	846.0	23.5		
Six-Lane	46.4	1.4	50.5 4.5	108.8	96.9 4.5	2.7		
Subtotal	3,023.0	92.3	297.3	9.8	3,320.3	92.3		
Region Total	3,274.2	100.0	324.7	9.9	3,598.9	100.0		

Source: SEWRPC.

VEHICLE AVAILABILITY WITHIN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

<u></u>		and the second	A sector part of the sector of the	the second se	
	Base Year	Forecast In	Forecast Increment		
Characteristic	1991	Number	Percent	2010	
Vehicles Available	1,132,000	155,400	13.7	1,287,400	
Population	1,810,400	100,600	5.6	1,911,000	
Persons per Vehicle	1.60	-0.12	-7.5	1.48	
Households	714,900	59,400	8.3	774,300	
Vehicles per Household	1.58	0.08	5.1	1.66	

Source: SEWRPC.

Table 207

DISTRIBUTION OF INTERNAL PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ADOPTED LAND USE PLAN AND ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

	Internal Person Trips Generated on an Average Weekday									
	Base Yea	Base Year 1991		ncrement	2010					
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total				
Home-Based Work	1,302,700	23.5	141,900	10.9	1,444,600	23.9				
Home-Based Shopping	798,000	14.4	75,200	9.4	873,200	14.4				
Home-Based Other	1,687,300	30.5	114,900	6.8	1,802,200	29.8				
Nonhome-Based	1,127,400	20.3	116,000	10.3	1,243,400	20.5				
School	625,600	11.3	66,700	10.7	692,300	11.4				
Total	5,541,000	100.0	514,700	9.3	6,055,700	100.0				

Source: SEWRPC.

Table 208

DISTRIBUTION OF INTERNAL PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY MODE OF TRAVEL: 1991 AND 2010 ADOPTED LAND USE PLAN AND ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

	Internal Person Trips Generated on an Average Weekday									
	Base Yea	r 1991	Forecast In	ncrement	2010					
Mode of Travel	Number	Percent of Total	Number	Percent Change	Number	Percent of Total				
Automobile Driver Automobile Passenger Transit Passenger School Bus Passenger	4,060,900 1,080,300 172,200 227,600	73.3 19.5 3.1 4.1	522,700 -164,900 99,400 57,500	12.9 -15.3 57.7 25.3	4,583,600 915,400 271,600 285,100	75.7 15.1 4.5 4.7				
Total	5,541,000	100.0	514,700	9.3	6,055,700	100.0				

Source: SEWRPC.

DISTRIBUTION OF INTERNAL AUTOMOBILE DRIVER TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ADOPTED LAND USE PLAN AND ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

	Internal Person Trips Generated on an Average Weekday								
	Base Yea	ir 1991	Forecast	ncrement	2010				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Home-Based Work Home-Based Shopping Home-Based Other Nonhome-Based School	1,149,500 616,600 1,246,900 928,500 119,400	28.3 15.2 30.7 22.9 2.9	163,600 76,500 118,800 127,400 36,400	14.2 12.4 9.5 13.7 30.5	1,313,100 693,100 1,365,700 1,055,900 155,800	28.6 15.1 29.8 23.0 3.4			
Total	4,060,900	100.0	522,700	12.9	4,583,600	100.0			

Source: SEWRPC.

Table 210

DISTRIBUTION OF INTERNAL AUTOMOBILE PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ADOPTED LAND USE PLAN AND ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

	Internal Automobile Person Trips Generated on an Average Weekday								
	Base Year 1991		Forecast	increment	2010				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Home-Based Work	1,258,700	24.5	95,900	7.6	1,354,600	24.6			
Home-Based Shopping	782,200	15.2	63,800	8.2	846,000	15.4			
Home-Based Other	1,659,100	32.3	90,300	5.4	1,749,400	31.8			
Nonhome-Based	1,115,700	21.7	99,000	8.9	1,214,700	22.1			
School	325,500	6.3	8,800	2.7	334,300	6.1			
Total	5,141,200	100.0	357,800	7.0	5,499,000	100.0			

Source: SEWRPC.

the Region may be expected to increase from about 172,200 trips in 1991, to about 271,600 trips by 2010, or by about 58 percent. The proportion of total internal travel generated within the Region served by transit could be expected to increase from 3.1 percent in 1991, to about 4.5 percent by 2010.

Table 208 also shows that internal automobile driver trips may be expected to increase from nearly 4.06 million trips per day in 1991 to about 4.58 million trips by 2010, or by about 13 percent. The proportion of all trips made as an automobile driver may be expected to increase from 73.3 percent in 1991 to 75.7 percent under Alterative Plan 1. Anticipated increases in automobile driver trips by trip purpose category are identified in Table 209, while anticipated increases in internal automobile person trips are shown in Table 210.

The use of transit under this combination of plans is indicated by trip purpose category in Table 211. Significant increases in transit trip production of 105 percent and 145 percent may be anticipated to occur in home-based work and nonhome-based tripmaking categories, respectively. Home-based work transit trips may be expected to increase from about 44,000 trips per average weekday, or 26 percent of daily transit trips in 1991, to 90,000 trips per average weekday, or 33 percent of daily transit trips in 2010. Trips not involving the home as either origin or destination, or nonhome-based trips, may be

	Internal Transit Trips Generated on an Average Weekday									
	Base Year 1991 Forecast Increm			Increment	20)10				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total				
Home-Based Work	44,000	25.6	46,000	104.5	90,000	33.1				
Home-Based Shopping	15,800	9.2	11,400	72.2	27,200	10.0				
Home-Based Other	28,200	16.4	24,600	87.2	52,800	19.4				
Nonhome-Based	11,700	6.8	17,000	145.3	28,700	10.6				
School	72,500	42.0	400	0.6	72,900	26.8				
Total	172,200	100.0	99,400	57.7	271,600	100.0				

DISTRIBUTION OF INTERNAL TRANSIT TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ADOPTED LAND USE PLAN AND ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

Source: SEWRPC.

expected to increase from 11,700 trips per average weekday, or about 7 percent of transit trips in 1991, to 28,700 trips per average weekday, or about 11 percent of daily transit trips in 2010. The significant increase in nonhome-based transit tripmaking may be expected to occur under Alternative Plan 1 because of the proposed reduced headways and wait time. Such improvements enhance the relative attractiveness of transit, especially within high-density areas.

The anticipated increases in internal vehicle trips by vehicle class are shown in Table 212. Total vehicle trip production on an average weekday is estimated to increase from about 4.89 million trips in 1991 to about 5.59 million trips by 2010, or by about 14 percent.

System Performance: Allocation to Alternative Plan 1 of the vehicle travel demand generated under the adopted land use plan indicates that vehicle-miles of travel on the arterial street and highway system may be expected to increase by about 5.3 million vehicle-miles, or 16 percent, from about 33.07 million per average weekday in 1991 to about 38.38 million per average weekday in 2010. About one-half of the total increase in regional vehicle-miles of travel may be expected to occur on the regional freeway system, where such travel is estimated to increase by 22 percent, from about 11.60 million vehicle-miles per average weekday in 1991 to 14.19 million vehiclemiles per average weekday in 2010. Anticipated increases in vehicle-miles of travel identified by county and facility type are set forth in Table 213.

Allocation to Alternative Plan 1 of transit demand generated under the adopted regional land use plan indicates that annual transit ridership within the Region may be expected to increase by about 60 percent from about 50.22 million revenue passengers in 1991 to about 80.12 million revenue passengers in 2010, approximating the annual ridership level of 1967. Most of this increase in transit ridership may be expected to occur within the Milwaukee urbanized area as the combined rapid, express, and local transit service area is expanded to cover nearly 100 percent of the Milwaukee urbanized area population.

As indicated in Chapter IX, under the no-build plan the number of annual revenue passengers may be expected to decrease by 2 percent to 49.0 million. The performance of the regional public transit system under Alternative Plan 1 is summarized in Table 214. Anticipated utilization of the planned transit service as measured in revenue passengers per revenue vehicle-hour may be expected to decline by about 11 percent between 1991 and 2010. In 1991, the transit system carried 33 passengers for every hour it operated. By 2010, because of proportionately larger increases in service provided than in

	-	Total Vehicle	e Trips Generat	ed on an Ave	age Weekday		
	Base Yea	ar 1991	Forecast	Increment	2010		
		Percent		Percent		Percent	
Vehicle Class	Number	of Total	Number	Change	Number	of Total	
Automobile							
Internal	4,060,900	83.0	522,700	12.9	4,583,600	82.0	
External	229,200	4.7	77,100	33.6	306,300	5.5	
Other	39,300	0.8	26,200	66.7	65,500	1.2	
Subtotal	4,329,400	88.5	626,000	14.5	4,955,400	88.6	
Truck							
Internal	520,100	10.6	58,900	11.3	579,000	10.4	
External	44,100	0.9	14,100	32.0	58,200	1.0	
Subtotal	564,200	11.5	73,000	12.9	637,200	11.4	
Total	4,893,600	100.0	699,000	14.3	5,592,600	100.0	

DISTRIBUTION OF TOTAL VEHICLE TRIPS IN THE REGION BY VEHICLE CLASS: 1991 AND 2010 ADOPTED LAND USE PLAN AND ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

Source: SEWRPC.

transit ridership, the regional transit system may be expected to carry about 30 passengers every hour of its operation.

Table 215 shows the population of the combined rapid, express, and local transit service area. Up to this point in the planning report, only the population served by local public transit has been noted in the context of evaluating the performance of the existing transit system and that of alternative plans. The total service area population shown in Table 215 represents the population located within one-quarter mile of a local transit route plus the population located within one-quarter mile of an express transit route and within a three-mile radius of a rapid transit station. The population served by local demand-responsive shared-ride taxi service is not included in the total service area population figures. The total transit service area population may be expected to increase by 20 percent from 1.36 million in 1991 to 1.63 million by 2010. As is indicated in Table 215, as a result of the significant planned expansion of rapid transit, the population located outside the urbanized area served by transit may be expected to comprise almost 9 percent of the total population served, up from 3 percent in 1991. Conversely,

the population located within the urbanized areas served by transit, while likely to increase by about 14 percent under Alternative Plan 1, will decrease as a proportion of total population served, from 97 percent in 1991 to about 91 percent in 2010.

System Deficiencies

As indicated in other chapters of this report, the volume-to-design-capacity (V/C) ratio exhibited by an arterial facility is a useful means of identifying and quantifying possible future imbalances between street usage and supply. The V/C ratio is defined as the relationship between the average daily weekday traffic utilizing a particular section of an arterial facility and the design capacity of that particular section. In all prior chapters of this report, V/C ratios were grouped into three categories: under design capacity (V/C ratio of 0.90 or less), at design capacity (V/C ratio of 0.91 to 1.10), and over design capacity (V/C ratio of 1.11 or more). Segments of arterial facilities operating over design capacity are said to be congested.

After considerable deliberation, the Advisory Committee determined that the minimum V/C ratio which would define a facility operating over design capacity should be lowered to 1.01.

VEHICLE-MILES OF TRAVEL ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE **REGION BY COUNTY: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1**

		Average Annual Rate					
	Base Ye	ar 1991	Forecast I	ncrement	2010		of Increase in Arterial Travel
County	Number	Percent of Total	Number	Percent Change	Number	Percent of Total	1991-2010 (percent)
Kenosha	675	27.0	287	42.5	962	26.8	1 90
Standard Arterial	1,825	73.0	803	44.0	2,628	73.2	1.90
Subtotal	2,500	100.0	1,090	43.6	3,590	100.0	1.90
Milwaukee				_			
Freeway	5,945	41.3	-66	-1.1	5,879	41.2	-0.05
	8,446	58.7	-59	-0.7	8,387	58.8	-0.04
Subtotal	14,391	100.0	-125	-0.9	14,266	100.0	-0.05
Ozaukee	762	20.2	45	5.9	907	20.0	0.30
Standard Arterial	1,180	60.8	35	3.0	1,215	60.1	0.30
Subtotal	1,942	100.0	80	4.1	2,022	100.0	0.20
Racine							
Freeway	708	23.9	202	28.5	910	26.2	1.30
Standard Arterial	2,258	76.1	307	13.6	2,565	73.8	0.70
Subtotal	2,966	100.0	509	17.2	3,475	100.0	0.80
Walworth							· · · ·
Freeway	540 1 373	28.2	302	55.9	842	35.2	2.40
Subtotal	1,012	100.0	190	25.2	2 205	100.0	0.00
Weshington	1,913	100.0	402	25.2	2,395	100.0	1.20
	546	23.0	1 1 50	210.6	1 696	517	6 1 0
Standard Arterial	1,833	77.0	-251	-13.7	1,582	48.3	-0.70
Subtotal	2,379	100.0	899	37.8	3,278	100.0	1.70
Waukesha							
Freeway	2,421	34.7	669	27.6	3,090	33.0	1.30
Standard Arterial	4,560	65.3	1,705	37.4	6,265	67.0	1.70
Subtotal	6,981	100.0	2,374	34.0	9,355	100.0	1.60
Southeastern Wisconsin Region							
Freeway	11,597	35.1	2,588	22.3	14,185	37.0	1.10
	21,475	04.9	2,/21	12.7	24,196	63.0	0.60
lotal	33,072	100.0	5,309	16.1	38 <u>,</u> 381	100.0	0.80

Source: SEWRPC.

The Committee determined that this modification would more accurately reflect the actual traffic conditions on the arterial system. This modification resulted in the classification of more arterial mileage as congested. In the remainder of the planning report, the volume-todesign-capacity ratios are grouped into the 436

following three categories: under design capacity (V/C ratio of 0.90 or less), at design capacity (V/C ratio of 0.91 to 1.0), and over design capacity (V/C ratio of 1.01 or more).

The arterial mileage which operated under, at, and over design capacity in 1991 is shown on

TRANSIT SYSTEM PERFORMANCE IN THE SOUTHEASTERN WISCONSIN REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

			Forecast In	crement
	Base Year			Percent
Transit System Characteristics	1991	2010	Number	Change
Service Provided, Average Weekday				
Revenue Vehicle-Miles				
Rapid	3,400	25,600	22,200	652.9
Express	3,300	29,800	26,500	803.0
Local	56,600	86,500	29,900	52.8
Total	63,300	141,900	78,600	124.2
Revenue Vehicle-Hours				
Rapid	170	1,100	930	547.1
Express	170	1,500	1,330	782.4
Local	4,880	6,600	1,720	35.2
Total	5,220	9,200	3,980	76.2
Seat-Miles	2,975,000	6,507,600	3,532,600	118.7
Service Utilization				
Ridership				
Average Weekday Revenue Passengers	172,200	271,600	99,400	57.7
Annual Revenue Passengers	50,222,900	80,122,000	29,899,100	59.5
Revenue Passengers per Revenue Vehicle-Hour	33.0	29.5	-3.5	-10.5
Average Weekday Passenger-Miles	609,100	1,375,900	766,800	125.9

Source: SEWRPC.

Table 215

1991 2010 Population Population Percent Percent Location Served of Total Served of Total 87.400^b 99.300^c 6.4 Kenosha Urbanized Area 6.1 1,114,500^d 1,267,600^e 82.0 77.6 Milwaukee Urbanized Area 114,300^f 127,200^g Racine Urbanized Area 8.4 7.8 1,494,100 1,316,200 96.8 91.5 Subtotal 3.2 Outside Urbanized Areas 43,900 138,700 8.5 1,360,100 100.0 1,632,800 100.0 Total

POPULATION SERVED BY PUBLIC TRANSIT IN THE REGION 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1^a

^aDoes not include that population served by demand-responsive shared-ride taxi service.

^bRepresents about 93 percent of the 1991 Kenosha urbanized area population.

^cRepresents about 93 percent of the 2010 Kenosha urbanized area population.

^dRepresents about 91 percent of the 1991 Milwaukee urbanized area population.

^eRepresents about 99 percent of the 2010 Milwaukee urbanized area population.

^fRepresents about 94 percent of the 1991 Racine urbanized area population.

^gRepresents about 96 percent of the 2010 Racine urbanized area population.

Source: SEWRPC.

	Design	At Design		Over Design Capacity							
	Сара	city ^a	Cap	Capacity ^b		Moderate ^C		Severe ^d		Extreme ^e	
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Mileage
Kenosha	286.0	90.0	8.1	2.5	8.5	2.7	10.4	3.3	4.7	1.5	317.7
Ozaukee	254.2	88.1	10.5	3.6	42.7	3.6	7.0	2.4	20.5 6.3	2.6	288.5
Racine	288.7	83.0 95.8	16.9 8.4	4.9	8.0	2.3	28.3 3.1	8.1	6.0	1.7	347.9 429.2
Washington	356.5	89.3	20.2	5.1	3.0	0.7	12.5	3.1	7.0	1.8	399.2
Waukesha	591.3	82.5	25.9	3.6	27.1	3.8	54.5	7.6	17.5	2.5	716.3
Total	2,731.0	83.4	157.6	4.8	106.4	3.2	217.2	6.6	62.0	2.0	3,274.2

TRAFFIC CONGESTION ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY COUNTY: 1991

^aVolume-to-design-capacity ratio: 0.00-0.90

^bVolume-to-design-capacity ratio: 0.91-1.00

^CVolume-to-design-capacity ratio: 1.01-1.10

^dVolume-to-design-capacity ratio: 1.11-1.30

^eVolume-to-design-capacity ratio: over 1.30

Source: SEWRPC.

Map 97 and summarized by county on Table 216. In 1991, 12 percent of 3,274-mile arterial system, or 385 miles, operated over design capacity. To better refine the identification of traffic congestion problems, the Advisory Committee grouped the mileage classified as over design capacity into three categories: moderately congested (V/C ratio of 1.01 to 1.10), severely congested (V/C ratio of 1.11 to 1.30), and extremely congested (V/C ratio of 1.31 or more). In 1991, about 106 miles, or 3 percent, of the total arterial mileage was moderately congested; 217 miles, or 7 percent, was severely congested; and 62 miles, or 2 percent, was extremely congested.

As shown on Map 98 and in Table 217 under the no-build alternative, traffic congestion may be expected to increase significantly. Indeed, by the year 2010, about 20 percent of the 3,480-mile arterial system, or 712 miles, may be expected to operate over design capacity. About 31 miles, or 1 percent of planned arterial mileage, may be expected to be moderately congested; 538 miles, or 15 percent, may be expected to be severely congested; and 143 miles, or 4 percent, may be expected to be extremely congested. As a result of the transportation system management measures, public transit improvements, and arterial street and highway improvements recommended under Alternative Plan 1, the level of traffic congestion may be expected to be substantially below that which would occur under the no-build plan. Indeed, under Alternative Plan 1, only 4 percent of the planned 3,599mile arterial system, or 131 miles, would operate over design capacity. As shown in Table 218, about 87 miles, or 2 percent of planned arterial mileage, would be moderately congested; 39 miles, or 1 percent, severely congested; and five miles, or less than 1 percent, extremely congested. The locations of those arterial facilities which would operate under, at, and over design capacity under Alternative Plan 1 are shown on Мар 99.

As shown on Map 99, the freeway system in the Milwaukee urbanized area may be expected to carry traffic volumes exceeding its design capacity and to operate under congested conditions through the year 2010, even with the transportation development proposals advanced under this alternative plan. The east-west
Map 97



In the base year 1991, 12 percent of the 3,274-mile arterial system, or 385 miles, operated over design capacity, with a volume-to-design-capacity ratio of 1.01 or greater. About 106 miles, or 3 percent of the arterial mileage, was moderately congested; 217 miles, or 7 percent of arterial mileage, was severely congested; and 62 miles, or about 2 percent of the arterial mileage, was extremely congested.

Source: SEWRPC.

TRAFFIC CONGESTION ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY COUNTY: 2010 NO-BUILD TRANSPORTATION SYSTEM PLAN

	Under [Design	Δ+ Γ	At Design Capacity ^b		Over Design Capacity					
	Сара	city ^a	Сар			erate ^C	Severed		Extreme ^e		
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Mileage
Kenosha	230.1 490.0	66.7 62.1	70.9 120.6	20.5 15.3	3.3 2.0	1.0 0.2	33.4 132.4	9.7 16.8	7.5 44.3	2.1 5.6	345.2 789.3
Racine	233.9 241.7 387.2	59.3 87.0	77.0	18.9 7.1	3.5 8.5	0.8 1.9	62.6 9.2	9.8 15.4 2.1	22.7 8.3	5.6 1.9	288.5 407.5 444.9
Washington Waukesha	337.7 369.1	75.3 48.8	54.1 103.8	12.1 13.7	 7.8	1.1	49.8 222.0	11.1 29.4	7.1 52.9	1.5 7.0	448.7 755.6
Total	2,289.7	65.8	477.9	13.7	31.6	0.9	537.7	15.5	142.8	4.1	3,479.7

^aVolume-to-design-capacity ratio: 0.00-0.90

^bVolume-to-design-capacity ratio: 0.91-1.00

^CVolume-to-design-capacity ratio: 1.01-1.10

^dVolume-to-design-capacity ratio: 1.11-1.30

^eVolume-to-design-capacity ratio: over 1.30

Source: SEWRPC.

freeway between the Milwaukee CBD and CTH Y in Waukesha County would experience severe traffic congestion, with volume-to-designcapacity ratios in the range of 1.11 to 1.30. The number of arterial miles in each of the counties of the Region that may be expected to operate under, at, and over design capacity in the year 2010 is shown in Table 218.

System Development Cost: The capital costs of preserving, improving, and expanding the arterial street and highway system is estimated at \$4.8 billion, or \$298 million annually. The estimated average annual operating and maintenance cost for the planned arterial street and highway system is \$68 million. The total capital costs of preserving, improving, and expanding the regional public transit system is estimated at \$1.0 billion, or \$65 million annually. The average annual operating and maintenance cost for the planned transit system is estimated at \$158 million. With the subtraction of anticipated average annual farebox revenues of \$43 million, the net average annual transit operating and maintenance cost would be \$115 million.

ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM PLAN 3

Alternative Plan 3 seeks to accommodate future travel demand primarily through the improvement and expansion of the arterial street and highway and public transit systems. Alternative Plan 3 proposes the use of measures to increase the perceived cost of vehicle travel but only to the extent necessary to cover shortfalls in plan funding. The revenue necessary to cover revenue shortfalls experienced in implementing the transportation improvements envisioned under Alternative Plan 3 could be raised with an increase in the motor fuel tax. The magnitude of the motor fuel tax increase, however, would not be significant enough to impact travel demand within the Region.

Alternative Plan 3 proposes to expand the public transit system by approximately 35 percent, focusing primarily on providing weekday rapid and express transit service in selected portions of major travel corridors within the Region. The plan proposes minimal expansion of the local transit service areas and no improvement in the



Under the "no-build" alternative, traffic congestion may be expected to increase significantly between 1991 and 2010. By the year 2010, about 20 percent of the 3,480-mile arterial system, or 712 miles, may be expected to operate over design capacity. About 31 miles, or about 1 percent, of the arterial mileage, may be expected to be moderately congested; 538 miles, or about 15 percent, of the arterial mileage may be expected to be severely congested; and 143 miles, or about 4 percent, of the arterial mileage may be expected to be extremely congested.

Under Design			Δ+ Γ	At Design		Over Design Capacity						
	Сара	city ^a	Сар	Capacity ^b		erate ^C	Severe ^d		Extreme ^e			
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Mileage	
Kenosha	327.1	91.8	25.8	7.2			3.3	0.9			356.2	
Milwaukee	673.2	84.7	57.0	7.2	37.9	4.8	21.7	2.7	5.1	0.6	794.9	
Ozaukee	308.3	99.3	0.8	0.3	1.4	0.5					310.5	
Racine	405.0	96.7	12.7	3.0	0.6	0.1	0.7	0.1			419.0	
Walworth	449.8	95.0	11.9	2.5	11.7	2.5					473.4	
Washington	469.1	99.9	0.7	0.1							469.8	
Waukesha	681.5	87.9	45.3	5.8	35.3	4.6	13.0	1.7			775.1	
Total	3,314.0	92.1	154.2	4.3	86.9	2.4	38.7	1.1	5.1	0.1	3,598. 9	

TRAFFIC CONGESTION ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY COUNTY: 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 1

^aVolume-to-design-capacity ratio: 0.00-0.90

^bVolume-to-design-capacity ratio: 0.91-1.00

^CVolume-to-design-capacity ratio: 1.01-1.10

^dVolume-to-design-capacity ratio: 1.11-1.30

^eVolume-to-design-capacity ratio: over 1.30

Source: SEWRPC.

frequency of local transit service. Significant improvements to the arterial street and highway system are proposed under Alternative Plan 3. Such improvements would be necessary to accommodate the significant increases in demand for automobile travel within the Region in the future.

Transportation System Management

Alternative Plan 3, like Alternative Plan 1, proposes a number of transportation system management measures to ensure that full use of existing transportation facilities is achieved before commitments are made to new capital investment. Unlike Alternative Plan 1, however, this plan does not propose to reduce a moderate travel demand through the use of pricing measures. The proposed measures included in Alternative Plan 3 may be grouped in three categories: land use planning and site design, areawide promotional programs, and traffic management.

Land Use Planning and Site Design: As does Alternative Plan 1, this plan recommends that local units of government design land use plans that encourage urban development to occur in neighborhood units developed at medium or higher densities in areas adjacent to existing urban development. In such areas transit service can be provided economically and shorter automobile trips can be made. Alternative Plan 3 would not, however, provide specifically that high-density residential and commercial development occur along transit lines. Consequently, in the design of Alternative Plan 3, neither the mode choice model nor the trip generation model were adjusted.

Areawide Promotional Measures: Like Alternative Plan 1, Alternative Plan 3 recommends a coordinated areawide effort to promote travel through ridesharing, transit use, and bicycle use. This program would aggressively promote carpooling and vanpooling, telecommuting, and rescheduling of work time to reduce travel by automobile, particularly by single-occupancy automobile, and particularly in the peak periods. In addition, this program would include the promotion and support of transportation management associations at major employment



Under Alternative Plan 1, the level of traffic congestion may be expected to be substantially below that which would occur under the no-build alternative. By the year 2010, only about 4 percent of the planned 3,599-mile arterial system, or 131 miles, would operate over design capacity. About 87 miles, or 2 percent, of planned arterial mileage would be moderately congested; 39 miles, or about 1 percent, would be severely congested; and 5 miles, or less than 1 percent, would be extremely congested. While the transportation development proposals included in Alternative Plan 1 serve to reduce the amount of congested facilities throughout the entire Region, the Milwaukee area freeway system may be expected to carry traffic volumes exceeding its design capacity and to operate with congested conditions through the year 2010.

	Base Ye	ar 1991	Planned I	ncrement	20	10
		Percent		Percent		Percent
Transit Service	Number	of Total	Number	Change	Number	of Total
Round-Trip Route Length (miles)					_	
Rapid Routes	449	19.6	761	169.5	1,210	37.1
Express Routes	393	17.1	-3	-0.8	390	12.0
Local Routes						
Kenosha Urbanized Area	171	7.4	19	11.1	190	5.8
Milwaukee Urbanized Area	1,112	48.5	178	16.0	1,290	39.6
Racine Urbanized Area	171	7.4	9	5.3	180	5.5
Subtotal	1,454	63.3	206	14.2	1,660	50.9
Total	2,296	100:0	964	42.0	3,260	100.0
Special Facilities (miles)						
Exclusive Right-of-Way	0	0.0	30		30	93.8
Exclusive Lanes on Streets	2	100.0	0	0.0	2	6.2
Total	2	100.0	30	1,500.0	32	100.0
Average Weekday Vehicle Requirements ^a						
Peak Period	530		137	25.8	667	
Midday Off-Peak Period	285		47	16.5	332	

TRANSIT SYSTEM OPERATING CHARACTERISTICS AND FACILITIES IN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup vehicles.

Source: SEWRPC.

centers throughout the Region to provide an institutional structure for helping to achieve the goal of reducing travel by single-occupancy vehicles.

Traffic Management: Alternative Plan 3 proposes the use of traffic management measures, as does Alternative Plan 1, to accommodate vehicular travel demand not abated through travel demand management and public transit improvements. Traffic management measures are intended to make full use of existing arterial capacity prior to the construction of new capacity. The traffic management measures included in the Alternative Plan 3 include restrictions on peak-period curb-lane parking; comprehensive freeway traffic management; and the use of traffic engineering by State, county, and municipal agencies responsible for plan implementation. The plan recommends that curb parking restrictions be placed on a total of 464 arterial route-miles during peak periods.

Public Transit System

Maintenance and Improvement

Alternative Plan 3 envisions that public transit service within the Milwaukee, Racine, and Kenosha urbanized areas would be improved over existing services, but to a lesser degree than proposed under Alternative Plan 1. The public transit element of Alternative Plan 3 is shown on Map 94 on page 402 in Chapter IX and summarized in Table 219.

The transit system envisioned under Alternative Plan 3 would consist of 3,260 round-trip routemiles, which would be 964 miles, or about 42 percent, greater than that provided in 1991. The system would require that 667 vehicles operate during peak periods and would provide about 96,900 revenue vehicle-miles and 6,700 revenue vehicle-hours of service per weekday. The base year 1991 system required that 530 buses operate during peak periods and provided about 63,300 revenue vehicle-miles and 5,220 revenue vehicle-hours of service per average weekday. Rapid Transit Service: Alternative Plan 3 proposes to expand the rapid transit system to connect major activity centers with central Milwaukee County and the Milwaukee CBD. The existing network of bus rapid transit routes would be expanded but not to the same degree proposed under Alternative Plan 1. The rapid transit routes would extend from the Milwaukee CBD south to the Cities of Racine and Kenosha, southwest to the Village of Mukwonago, west to the City of Oconomowoc, northwest to the Village of Germantown, and north to the City of Cedarburg and the Village of Grafton.

Alternative Plan 3 proposes to operate bus rapid transit service either over operationally controlled freeway lanes in mixed traffic or over exclusive busways. Such busways, capable of accommodating buses and high-occupancy vehicles should be provided only in segments of corridors where the freeway in those corridors is determined to be operating under severe traffic congestion and only if the busway is found to have potential to serve more travel than freeway lanes open to all traffic. As with Alternative Plan 1, commuter railway passenger train service would be considered as an alternative option in selected travel corridors for providing rapid transit service.

The proposed rapid transit system would consist of 28 routes totaling 1,210 round-trip route miles. Three routes would provide direct service to the University of Wisconsin-Milwaukee campus. The routes would serve a total of 65 public transit stops or stations.

Alternative Plan 3 proposes to improve service, but not to the level envisioned under Alternative Plan 1. Under Alternative Plan 3, service would be provided over all rapid transit routes during the morning and afternoon peak periods of each weekday, or from 6:00 a.m. to 9:00 a.m. and from 3:00 p.m. to 6:00 p.m. Service would also be provided during weekday midday off-peak periods over routes within Milwaukee County which provide midday service. Service would be provided in both directions of travel at headways ranging from 10 to 30 minutes during weekday peak periods to between 30 to 60 minutes during weekday off-peak periods. No evening or weekend service is proposed to be provided under Alternative Plan 3.

Express Transit Service: The development of a comprehensive express transit system is also proposed under this alternative plan. Alternative Plan 3 recommends that express service be implemented as bus service operating in mixed traffic and upgraded to exclusive lanes as warranted. Consideration of bus on guideway or light rail transit service would be initiated as part of major investment studies for those express transit corridors shown on Map 95.

Express transit service would be provided over 12 transit routes within the Milwaukee, Racine, and Kenosha urbanized areas. As compared to Alternative Plan 1, the express route system of Alternative Plan 3 is less dense and extensive, totaling about 390 round trip route-miles. About 490 round trip route-miles would be provided under Alternative Plan 1. Express routes proposed to be located within Milwaukee County would provide crosstown service, as well as service oriented to the Milwaukee CBD. The planned service would be operated on weekdays between 6:00 a.m. and 6:00 p.m. Weekend express transit service would be provided over some routes. Operating headways for the express routes would range from 10 to 15 minutes during weekday peak periods and from 20 to 30 minutes during off-peak periods.

Express transit service would directly connect the Cities of Racine and Kenosha but would not connect these cities to newly developing areas west of STH 31 as is proposed under Alternative Plan 1. Service over the express routes between Racine and Kenosha would be operated at headways of 30 minutes during weekday peak periods and 60 minutes during weekday midday off-peak periods. In all areas, buses providing express transit service would, where appropriate, also provide collection and distribution service beyond the proposed termini of the express routes.

Local Transit Service: Alternative Plan 3 also proposes to expand local transit service within the Milwaukee, Racine, and Kenosha urbanized areas. Local transit services would be extended into residential areas of high- and mediumdensity (5.0 dwelling units per acre or greater) which are contiguous to existing transit service areas. Local bus service within the Milwaukee urbanized area would be extended principally to northern and southern Milwaukee County and around the City of Waukesha. Alternative Plan 3 proposes to continue the grid system of local bus routes within the portion of Milwaukee County currently served by local transit. This alternative, like Alternative Plan 1, envisions the potential development of special local circulator transit services to connect various land uses within areas of high-intensity development along IH 94 in western Waukesha County. The proposed local transit services in the Milwaukee urbanized area would consist of about 1,290 round-trip route-miles. Service levels, including headways, on existing local transit routes in the Milwaukee urbanized area would be maintained at existing 1991 levels.

Proposed local transit service within the Racine and Kenosha urbanized areas would include extensions of service to areas contiguous to existing local transit service areas anticipated to develop at medium- and high-densities as previously described. Special, local circulator transit service is proposed to serve areas of commercial, office, and industrial development along IH 94. Local transit would also connect to the rapid and express transit routes proposed to be operated between the Milwaukee, Racine, and Kenosha urbanized areas.

The local transit system in the Kenosha and Racine urbanized areas would consist of 190 and 180 round-trip route-miles, respectively. Headways on local routes serving these areas would be maintained at existing levels, generally 20 to 30 minutes during peak periods and 60 minutes during off-peak periods. The areas served by the expanded local transit services proposed under this alternative are shown on Map 94 on page 402 in Chapter IX.

Transit Fares: Transit fares under Alternative Plan 3 were assumed to remain consistent with fares charged by transit operators within the Region in 1994, increasing only with general price inflation. Accordingly, the base adult fare within Milwaukee County, expressed in constant 1994 dollars, would remain at \$1.25 per ride for local and express bus service and at \$1.50 per ride for rapid transit bus service. Similarly, fares for rapid transit service between Milwaukee County and surrounding counties would increase with distance traveled outside of Milwaukee County and would generally range from about \$2.00 per ride at the outer limits of the future urbanized area to about \$3.00 at the extreme limits of service from the Milwaukee CBD. Local transit fares in the Cities of Racine, Kenosha, and Waukesha and their immediate environs are proposed to be consistent with 1994 fares, expressed in 1994 dollars. Accordingly, such fares would range from \$0.60 per ride on the Racine transit system to \$0.75 per ride on the Kenosha and Waukesha transit systems.

Arterial Street and Highway System Maintenance and Improvement

As under Alternative Plan 1, under Alternative Plan 3 the large majority of travel in the Region in the year 2010 will continue to have to be accommodated on the arterial street and highway system. That system, under Alternative Plan 3 would total 3,609 route-miles and 10,182 lane-miles, or 111 route-miles more than Alternative Plan 1 (see Table 220). The arterial street and highway maintenance and improvement recommendations are graphically presented on Map 100 and summarized below. Alternative Plan 3 envisions the following with respect to the arterial street and highway system:

<u>New Arterial Facilities</u>: The plan would provide for the construction of 124 route-miles and 323 lane-miles of new arterial street and highway facilities. In addition to all of the planned new facilities included in Alternative Plan 1, Alternative Plan 3 includes one new major facility, the completion of the STH 12 freeway in Walworth County, as well as several relatively minor new facilities. The proposed new arterial highway facilities would represent 3 percent of the total planned arterial route-miles and 3 percent of the total planned arterial lane-miles.

Widening of Existing Arterial Facilities: Under Alternative Plan 3, a total of 410 route-miles of facilities would be widened to provide an additional 1,676 arterial lane-miles. This represents about 11 percent of the total route-miles and 16 percent of the total lane-miles. This may be compared with the widening of 382 route-miles to provide an additional 1,571 lane-miles envisioned under Alternative Plan 1. In addition to the arterial widening projects included under Alternative Plan 1, Alternative Plan 3 includes a series of arterial widenings in the outlying portions of the Region, including the widening of STH 60 in Ozaukee and Washington Counties and STH 67 in Waukesha County.

<u>Preserving Arterial Streets and Highways</u>: With respect to the remaining 3,075 route-miles and 8,183 lane-miles of facilities envisioned under Alternative Plan 3, representing about 85 per-



The arterial street and highway system under Alternative Plan 3 would total 3,609 route-miles and 10,182 lane-miles. The plan would provide for the construction of 124 route-miles and 323 lane-miles of new arterial street and highway facilities to widen 410 route-miles of facilities to provide an additional 1,676 arterial lane-miles and for the preservation of 3,075 route-miles and 8,183 lane-miles of arterial street and highway facilities. Like Alternative Plan 1, Alternative Plan 3 includes the pavement resurfacing and interchange and bridge replacement work on the Milwaukee area freeway system.

ARTERIAL STREET AND HIGHWAY FACILITIES IN THE REGION BY COUNTY: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

	Arterial Streets and Highways							
	Existin	g 1991	Planned I	ncrement	Total	2010		
Arterial Facility Type	Miles	Percent of Total	Miles	Percent Change	Miles	Percent of Total		
Kenosha County								
Freeway Four-Lane								
Six-Lane	12.0	3.8	0.0	0.0	12.0	3.4		
Eight-Lane								
Subtotal	12.0	3.8			12.0	3.4		
Standard Arterial								
	256.1	80.6	-11.8	-4.6	244.3	68.6		
Six-Lane			4.9		4.9	1.4		
Subtotal	305.7	96.2	38.5	12.6	344.2	96.6		
County Subtotal	317.7	100.0	38.5	12.1	356.2	100.0		
Milwaukee County								
Freeway								
⊢our-Lane Six-Lane	13.9 53.9	1.8	-3.0	-21.6	10.9	1.4		
Eight-Lane	1.4	0.2	0.0	0.0	1.4	0.2		
Subtotal	69.2	8.9	0.0	0.0	69.2	8.7		
Standard Arterial								
Two-Lane	379.0	48.9	-63.2	-16.7	315.8	39.7		
Four-Lane	288.9	37.3	52.6	18.2	341.5	43.0		
Eight-Lane	0.0	0.0	1.4		1.4	0.2		
Subtotal	706.2	91.1	19.5	2.8	725.7	91.3		
County Subtotal	775.4	100.0	19.5	2.5	794.9	100.0		
Ozaukee County								
Freeway	07.4	0.5			07.4			
Six-Lane	27.4	9.5	0.0	0.0	27.4	8.8		
Eight-Lane								
Subtotal	27.4	9.5	0.0	0.0	27.4	8.8		
Standard Arterial						_		
Two-Lane	241.3	83.6	-17.5	-7.3	223.8	72.1		
Six-Lane	19.8	6.9	39.5	199.5	59.3	19.1		
Subtotal	261.1	90.5	22.0	84	283.1	91.2		
County Subtotal	288.5	100.0	22.0	7.6	310.5	100.0		
Racine County						100.0		
Freeway								
Eight-Lane		3.4	0.0	0.0	12.0	2.9		
Subtotal	12.0	3.4	0.0	0.0	12.0	2.9		
Standard Arterial								
Two-Lane	301.6	86.7	-0.2	-0.1	301.4	71.9		
Four-Lane	31.1	9.0	64.4	207.1	95.5	22.8		
Subtotal	325.0	96.6	71 1	210.0	10.1	2.4		
County Subtotal	347 9	100.0	71.1	20.4	407.0	97.1		
	577.3		/ 1.1	20.4	419.0	100.0		

Table 220 (continued)

	· .	A	Arterial Streets	and Highways		
	Existing	1991	Planned I	ncrement	Total	2010
Arterial Facility Type	Miles	Percent of Total	Miles	Percent Change	Miles	Percent of Total
Walworth County						
Freeway	50.0	11.0	107	00.4	007	10.0
Four-Lane	50.0	11.0		33.4		
Eight-Lane						
Subtotal	50.0	11.6	16.7	33.4	66.7	13.8
Standard Arterial						
Two-Lane	370.5	86.4	1.0	0.3	371.5	76.7
Four-Lane	8.7	2.0	37.2	427.0	45.9	9.5
Subtotal	379.2	88.4	38.2	10.1	417.4	86.2
County Subtotal	429.2	100.0	54.9	12.8	484.1	100.0
Washington County						
Freeway	·					
Four-Lane	15.1	3.8	21.0	139.7	36.2	7.7
Six-Lane	6.5	1.6	0.0	0.0	c.o 	1.4
Subtotal	21.6	5.4	21 1	97.7	42.7	9.1
Standard Arterial	21.0	0.1		0,1,1		
Two-Lane	340.0	85.2	32.1	9.4	372.1	79.2
Four-Lane	37.6	9.4	17.4	46.3	55.0	11.7
Six-Lane			· · · · · ·			
Subtotal	377.6	94.6	49.5	13.1	427.1	90.9
County Subtotal	399.2	100.0	70.6	17.7	469.8	100.0
Waukesha County						
Four-Lane	45.6	6.3	6.3	13.8	51.9	6.7
Six-Lane	13.4	1.9	0.0	0.0	13.4	1.7
Eight-Lane						
Subtotal	59.0	8.2	6.3	10.7	65.3	8.4
Standard Arterial	600.2	00.0	04.0	14.1	E1E /	66 5
Four-Lane	52.2	7.3	123.7	237.0	175.9	22.7
Six-Lane	4.9	0.7	10.5	214.3	15.4	2.0
Eight-Lane		·	3.1		3.1	0.4
Subtotal	657.3	91.8	52.5	8.0	709.8	91.6
County Subtotal	716.3	100.0	58.8	8.2	775.1	100.0
Southeastern Wisconsin Region					× '	
Freeway Four-Lane	152.0	4.6	41.1	27.0	193.1	5.3
Six-Lane	97.8	3.0	3.0	3.1	100.8	2.8
Eight-Lane	1.4	0.1	0.0	0.0	1.4	0.0
Subtotal	251.2	7.7	44.1	17.6	295.3	8.2
Standard Arterial	0.400.7	70.0			2 244 2	64.0
	2,488.7 <u>1</u> 87 9	76.0 14 9	-144.4	-5.8 77 9	2,344.3 868 1	24.0
Six-Lane	46.4	1.4	51.0	109.9	97.4	2.7
Eight-Lane			4.5		4.5	0.1
Subtotal	3,023.0	92.3	291.3	9.6	3,314.3	91.8
Region Total	3,274.2	100.0	335.4	10.2	3,609.6	100.0

Source: SEWRPC.

	Base Year	Forecast In	crement	
Characteristic	1991	Number	Percent	2010
Vehicles Available	1,132,000	164,400	14.5	1,296,400
Population	1,810,400	100,600	5.6	1,911,000
Persons per Vehicle	1.60	-0.13	-8.1	1.47
Households	714,900	59,400	8.3	774,300
Vehicles per Household	1.58	0.09	5.7	1.67

VEHICLE AVAILABILITY WITHIN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

Source: SEWRPC.

cent of the total route-miles and 80 percent of the total lane-miles, this alternative calls only for resurfacing or reconstruction. Like Alternative Plan 1, Alternative Plan 3 includes important preservation actions involving the resurfacing and bridge replacement work on the Milwaukeearea freeway system.

Assignment of Travel to

Highway and Transit Networks

An analysis of the performance of Alterative Plan 3 under probable future land use and travel conditions was undertaken with the aid of the traffic simulation models described in Chapter VII of this report. Through application of these models, travel demand for the design year 2010 was developed for the land use pattern proposed under the adopted regional land use plan and assigned to the arterial street and highway and transit networks.

Vehicle Availability: Given transportation system development in accordance with Alternative Plan 3 and land use development in accordance with the adopted regional land use plan, it may be expected that the number of automobiles and light trucks available within the Region may increase from about 1.13 million in 1991 to about 1.30 million in 2010, an increase of about 15 percent, (see Table 221). This increase compares with an approximate 6 percent increase in resident population, 8 percent increase in the number of households, and 11 percent increase in the number of jobs over the same time period. The number of persons per automobile then may be expected to continue to decline by 8 percent between 1991 and 2010, from 1.60 to 1.47. The number of automobiles per household may be expected to increase by about 6 percent, from 1.58 in 1991 to 1.67 in 2010.

Person Trip Generation: Given transportation system development in accordance with Alternative Plan 3 and land use development in accordance with the adopted regional land use plan. internal person trips may be expected to increase from 5.54 million trips per day in 1991, to about 6.11 million trips by 2010, or by 10.2 percent. The average number of trips per capita may then be expected to increase from about 3.1 in 1991 to 3.2 in 2010. The average number of internal person trips per household may be expected to increase from 7.8 in 1991 to 7.9 in 2010. Future internal person trip production within the Region by trip purpose under this alternative plan is shown in Table 222 and compared to existing 1991 trip production.

<u>Mode of Travel</u>: A comparison of the distribution of internal trips in the Region by mode of travel under the adopted regional land use plan and Alternative Plan 3 is summarized in Table 223. Average weekday transit trip production within the Region may be expected to increase from about 172,200 trips in 1991, to about 186,900 trips by 2010, or by nearly 9 percent. The proportion of total internal travel generated within the Region served by transit could be expected to remain at about 3.1 percent between 1991 and 2010.

Table 223 also shows that internal automobile driver trips may be expected to increase from 4.06 million trips per average weekday in 1991, to about 4.70 million trips per average weekday by 2010, or by about 16 percent. The proportion of all trips made as an automobile driver may be expected to increase from 73.3 percent in 1991 to 76.9 percent under Alterative Plan 3. Anticipated increases in automobile driver trips by trip purpose category are identified in Table 224,

DISTRIBUTION OF INTERNAL PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

	Internal Person Trips Generated on an Average Weekday							
	Base Yea	Base Year 1991		Forecast Increment		0		
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total		
Home-Based WorkHome-Based ShoppingHome-Based OtherNonhome-BasedSchool	1,302,700 798,000 1,687,300 1,127,400 625,600	23.5 14.4 30.5 20.3 11.3	152,100 85,000 136,100 124,600 66,700	11.7 10.7 8.1 11.1 10.7	1,454,800 883,000 1,823,400 1,252,000 692,300	23.8 14.5 29.9 20.5 11.3		
Total	5,541,000	100.0	564,500	10.2	6,105,500	100.0		

Source: SEWRPC.

Table 223

DISTRIBUTION OF INTERNAL PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY MODE OF TRAVEL: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

	Internal Person Trips Generated on an Average Weekday								
	Base Yea	r 1991	Forecast In	crement	2010				
Mode of Travel	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Automobile Driver	4,060,900	73.3	634,700	15.6	4,695,600	76.9			
Transit Passenger	172,200	3.1	14,700	8.5	186,900	3.1			
School Bus Passenger	227,600	4.1	57,500	25.3	285,100	4.7			
Total	5,541,000	100.0	564,500	10.2	6,105,500	100.0			

Source: SEWRPC.

Table 224

DISTRIBUTION OF INTERNAL AUTOMOBILE DRIVER TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

	Internal Person Trips Generated on an Average Weekday								
	Base Yea	r 1991	Forecast Ir	ncrement	2010				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Home-Based Work Home-Based Shopping Home-Based Other Nonhome-Based School	1,149,500 616,600 1,246,900 928,500 119,400	28.3 15.2 30.7 22.9 2.9	205,500 94,800 153,500 144,500 36,400	17.9 15.4 12.3 15.6 30.5	1,355,000 711,400 1,400,400 1,073,000 155,800	28.9 15.2 29.8 22.9 3.3			
Total	4,060,900	100.0	634,700	15.6	4,695,600	100.0			

Source: SEWRPC.

	Internal Automobile Person Trips Generated on an Average Weekday								
	Base Yea	ar 1991	Forecast I	ncrement	2010				
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total			
Home-Based Work	1,258,700 782,200	24.5 15.2	139,400 86,300	11.1 11.0	1,398,100 868,500	24.8 15.4			
Home-Based Other	1,659,100 1,115,700 325,500	32.3 21.7	136,100 119,200 11 300	8.2 10.7 3.5	1,795,200 1,234,900 336,800	31.9 21.9 6.0			
Total	5,141,200	100.0	492,300	9.6	5,633,500	100.0			

DISTRIBUTION OF INTERNAL AUTOMOBILE PERSON TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

Source: SEWRPC.

while anticipated increases in internal automobile person trips within the Region are shown in Table 225.

The use of transit under this combination of plans is indicated by trip purpose category in Table 226. Home-based work transit trips may be expected to increase by nearly 29 percent, from 44,000 trips in 1991 to 56,700 trips in 2010. Nonhome-based transit trips may be expected to increase by about 46 percent from 11,700 trips in 1991 to about 17,100 trips in 2010. Home-base shopping transit trips may be expected to decrease by about 8 percent, from 15,800 trips in 1991 to 14,500 trips in 2010. School trips made by transit may also be expected to decline by 2,100 trips, or about 3 percent.

The anticipated increases in internal vehicle trips by vehicle class are shown in Table 227. Total vehicle trip production, on an average weekday, is estimated to increase from about 4.9 million trips in 1991 to about 5.7 million trips by 2010, or by about 15 percent.

System Performance: Allocation of vehicle travel demand generated under the adopted regional land use plan to Alternative Plan 3 indicates that vehicle-miles of travel on the arterial street and highway system may be expected to increase by about 9.78 million vehicle-miles or 30 percent, from about 33.07 million per average weekday in 1991 to about 42.85 million per average weekday in 2010. About 45 percent of this increase may be expected to occur on the regional freeway system, where such travel is estimated to increase from about 11.6 million vehicle-miles per average weekday in 1991 to 16 million vehicle-miles per average weekday in 2010. The anticipated increases in vehicle-miles of travel identified according to county and facility type are set forth in Table 228.

Allocation of transit demand generated under the adopted regional land use plan to Alternative Plan 3 indicates that annual transit ridership within the Region may be expected to increase by about 9 percent, from about 50.22 million revenue passengers in 1991 to about 54.58 million revenue passengers in 2010. As with Alternative Plan 1, most of this increase in transit ridership may be expected to occur within the Milwaukee urbanized area as the combined rapid, express, and local transit service area is expanded to cover some 98 percent of the Milwaukee urbanized area population.

The performance of the regional public transit system under Alternative Plan 3 is summarized in Table 229. Under Alternative Plan 3 the number of revenue vehicle-miles of service provided may be expected to increase by 53 percent from, 63,300 in 1991 to 96,900 by 2010. Over 88 percent of this increase will come about through expanded rapid and express transit service reflecting the plan's emphasis on providing fast and convenient transit service over long distances. Local transit vehicle-miles would be increased by about 7 percent, a relatively minor

DISTRIBUTION OF INTERNAL TRANSIT TRIPS MADE BY HOUSEHOLD RESIDENTS IN THE REGION BY TRIP PURPOSE: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

	Internal Transit Trips Generated on an Average Weekday							
	Base Ye	Base Year 1991		Forecast Increment		10		
Trip Purpose Category	Number	Percent of Total	Number	Percent Change	Number	Percent of Total		
Home-Based Work	44,000	25.6	12,700	28.9	56,700	30.3		
Home-Based Shopping	15,800	9.2	-1,300	-8.2	14,500	7.8		
Home-Based Other	28,200	16.4	0	0.0	28,200	15.1		
Nonhome-Based	11,700	6.8	5,400	46.2	17,100	9.1		
School	72,500	42.0	-2,100	-2.9	70,400	37.7		
Total	172,200	100.0	14,700	8.5	186,900	100.0		

Source: SEWRPC.

Table 227

DISTRIBUTION ON TOTAL VEHICLE TRIPS IN THE REGION BY VEHICLE CLASS: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

	Total Vehicle Trips Generated on an Average Weekday									
	Base Yea	ar 1991	Forecast	Increment	20	10				
Vehicle Class	Number	Percent of Total	Number	Percent Change	Number	Percent of Total				
Automobile Internal External Other	4,060,900 229,200 39,300	83.0 4.7 0.8	634,700 77,100 26,200	15.6 33.6 66.7	4,695,600 306,300 65,500	82.3 5.4 1.1 88.8				
Truck Internal	520,100 44,100	10.6 0.9	58,900 14,100	11.3 32.0	579,000 58,200	10.1 1.0				
Subtotal	564,200	11.5	73,000	12.9	637,200	11.2				
Total	4,893,600	100.0	811,000	16.6	5,704,600	100.0				

Source: SEWRPC.

increase compared to the 53 percent increase envisioned under Alternative Plan 1. The same emphasis on the expansion of rapid and express transit is evident with respect to revenue vehicle hours of service as shown in Table 229.

Transit service utilization may be expected to decline under Alternative Plan 3, from 33 revenue passengers per revenue vehicle-hour in 1991 to 28 by 2010. The transit system under Alternative Plan 3 would carry nearly 2 fewer passengers per vehicle hour than the system under Alternative Plan 1. The combined population served by rapid, express, and local transit is shown for 1991 and 2010 under Alternative Plan 3 in Table 230. The total service area population may be expected to increase by about 14 percent, from 1.36 million in 1991 to 1.55

VEHICLE-MILES OF TRAVEL ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY COUNTY: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

		Arte on an A	erial Vehicle- Average Wee	Miles of Tr ekday (thou	avel usands)		Average Annual Rate
	Base Ye	ar 1991	Forecast I	ncrement	20	10	of Increase in Arterial Travel
County	Number	Percent of Total	Number	Percent Change	Number	Percent of Total	1991-2010 (percent)
Kenosha Freeway Standard Arterial	675 1,825	27.0 73.0	319 1,014	47.3 55.6	994 2,839	25.9 74.1	2.1 2.4
Subtotal	2,500	100.0	1,333	53.3	3,833	100.0	2.3
Milwaukee Freeway	5,945 8,446	41.3 58.7	472 1,159	7.9 13.7	6,417 9,605	40.1 59.9	0.4 0.7
Subtotal	14,391	100.0	1,631	11.3	16,022	100.0	0.7
Ozaukee Freeway Standard Arterial	762 1,180	39.2 60.8	111 169	14.6 14.3	873 1,349	39.3 60.7	0.7 0.7
Subtotal	1,942	100.0	280	14.4	2,222	100.0	0.7
Racine Freeway Standard Arterial Subtotal	708 2,258 2,966	23.9 76.1	330 648 978	46.6 28.7 33.0	1,038 2,906 3,944	26.3 73.7	2.0 1.3
Walworth Freeway Standard Arterial Subtotal	540 1,373	28.2 71.8	754 151 905	139.6 11.0 47.3	1,294 1,524 2,818	45.9 54.1	4.7 0.5 2.1
Washington Freeway	546 1,833 2,379	23.0 77.0	1,254 -63	229.7 -3.4 50.1	1,800 1,770 3,570	50.4 49.6	6.5 -0.2 2.2
Waukesha		100.0	1,101	00.1	0,070	100.0	2.2
Freeway	2,421 4,560	34.7 65.3	1,142 2,320	47.2 50.9	3,563 6,880	34.1 65.9	2.1 2.2
Subtotal	6,981	100.0	3,462	49.6	10,443	100.0	2.1
Southeastern Wisconsin Region Freeway Standard Arterial	11,597 21,475	35.1 64.9	4,382 5,398	37.8 25.1	15,979 26,873	37.3 62.7	1.7 1.2
Total	33,072	100.0	9,780	29.6	42,852	100.0	1.4

Source: SEWRPC.

million by 2010. About 85,700 fewer people would be served by transit under this plan than under Alternative Plan 1. As shown in Table 230, the population located outside of the urbanized area which may be expected to be served by transit would increase from 43,900 to 77,100, or by 76 percent. Under this plan, 5 percent of the population served would be located outside of urbanized areas, as compared to about 9 percent under Alternative Plan 1.

System Deficiencies: Like Alternative Plan 1, Alternative Plan 3 may be expected to result in substantially less traffic congestion than the no-

TRANSIT SYSTEM PERFORMANCE IN THE REGION: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

			Forecast In	crement
Transit System Characteristics	Base Year 1991	2010	Number	Percent Change
Service Provided, Average Weekday Revenue Vehicle-Miles				
Rapid	3,400	15,900	12,500	367.6
Express	3,300	20,400	17,100	518.2
Local	56,600	60,600	4,000	7.1
Total	63,300	96,900	33,600	53.1
Revenue Vehicle-Hours				
Rapid	170	600	430	252.9
Express	170	1,000	830	488.2
Local	4,880	5,100	220	4.5
Total	5,220	6,700	1,480	28.4
Seat-Miles	2,975,000	4,460,900	1,485,900	49.9
Service Utilization				
Ridership		100.000	4 4 7 9 9	
Average Weekday Revenue Passengers	172,200	186,900	14,700	8.5
Annual Revenue Passengers	50,222,900	54,575,000	4,352,100	8.7
Revenue Passengers per Revenue Vehicle-Hour	33.0	27.9	-5.1	-15.4
Average Weekday Passenger-Miles	609,100	903,400	294,300	48.3

Source: SEWRPC.

Table 230

2010 1991 Population Percent Population Percent Served of Total Served of Total Location 87,400^b Kenosha Urbanized Area 90,400^c 6.4 5.8 1,114,500^d 82.0 1.252.700^e 81.0 114,300^f Racine Urbanized Area 8.4 126,900^g 8.2 Subtotal 1,316,200 96.8 1,470,000 95.0 43,900 3.2 77,100 5.0 Outside Urbanized Areas 1,547,100 Total 1,360,100 100.0 100.0

POPULATION SERVED BY PUBLIC TRANSIT IN THE REGION 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3^a

^aDoes not include that population served by demand-responsive shared-ride taxi service.

^bRepresents about 93 percent of the 1991 Kenosha urbanized area population.

^cRepresents about 84 percent of the 2010 Kenosha urbanized area population.

^dRepresents about 91 percent of the 1991 Milwaukee urbanized area population.

^eRepresents about 98 percent of the 2010 Milwaukee urbanized area population.

^fRepresents about 94 percent of the 1991 Racine urbanized area population.

^gRepresents about 96 percent of the 2010 Racine urbanized area population.

Source: SEWRPC.

					_						
	Under [Desian	At D	At Design Capacity ^b		Over Design Capacity					
	Capao	city ^a	Сар			Moderate ^C		Severe ^d		Extreme ^e	
		Percent		Percent		Percent		Percent		Percent	Total
County	Miles	of Total	Miles	of Total	Miles	of Total	Miles	of Total	Miles	of Total	Mileage
Kenosha	310.3	87.1	42.6	12.0			3.3	0.9			356.2
Milwaukee	654.5	82.3	56.1	7.1	53.3	6.7	24.2	3.1	6.8	0.8	794.9
Ozaukee	306.2	98.6	2.9	0.9							310.5
Racine	387.8	92.6	25.0	6.0	1.4	0.3	0.7	0.1			419.0
Walworth	478.0	98.7	6.1	1.3	5.5						484.1
Washington	469.1	99.9	0.7	0.1							469.8
Waukesha	636.9	82.2	65.3	8.4	47.3	6.1	25.6	3.3			775.1
Total	3,242.8	89.8	198.7	5.5	107.5	3.0	53.8	1.5	6.8	0.2	3,609.6

TRAFFIC CONGESTION ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION BY COUNTY: 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLAN 3

^aVolume-to-design-capacity ratio: 0.00-0.90

^bVolume-to-design-capacity ratio: 0.91-1.00

^CVolume-to-design-capacity ratio: 1.01-1.10

^dVolume-to-design-capacity ratio: 1.11-1.30

^eVolume-to-design-capacity ratio: over 1.30

Source: SEWRPC.

build plan. As indicated in Table 231, only 168 miles, or 5 percent of the planned arterial street and highway system mileage, may be expected to operate over design capacity by the year 2010. Under the no-build plan, 712 miles or nearly 21 percent of arterial mileage would be congested (see Map 98). Under Alternative Plan 3, about 107 miles, or 3 percent of planned arterial mileage may be expected to be moderately congested; 54 miles, or about 2 percent, severely congested; and seven miles, or less than 1 percent, extremely congested. The location of those arterial facilities which would operate under, at, and over design capacity under Alternative Plan 3 are shown on Map 101.

System Development Cost: The capital costs of preserving, improving, and expanding the arterial street and highway system is estimated at \$4.8 billion, or \$302 million annually. The estimated average annual operating and maintenance costs of the planned arterial street and highway system is \$68 million. The total capital costs of preserving, improving, and expanding the regional public transit system is estimated at \$0.9 billion, or \$57 million annually. The average annual operating and maintenance costs attendant to the planned transit system is estimated at \$129 million. With the subtraction of anticipated average annual farebox revenues of \$36 million, the net average annual operating and maintenance costs would total \$93 million.

ALTERNATIVE TRANSPORTATION SYSTEM PLAN EVALUATION, SATISFACTION OF OBJECTIVES, AND STANDARDS

The transportation system development objectives to be met by the regional transportation system plans and the plan design standards have been set forth in Chapter VIII of this report. To determine the ability of the alternative transportation plans to meet the development objectives, the plans were scaled against the standards supporting each development objective. Results of this evaluation are presented in the following section of this chapter.

Objective No. 1: Serve Regional Land Use Pattern Efficiently

The first transportation system development objective refers to the achievement of an integrated transportation system which, through its location, capacity, and design, will effectively





Like Alternative Plan 1, Alternative Plan 3 may be expected to result in substantially less traffic congestion on the regional arterial street and highway system than the no-build plan. Under Alternative Plan 3, about 168 miles, or 5 percent, of the planned arterial street and highway mileage may be expected to operate over design capacity by the year 2010. About 107 miles, or about 3 percent, of the planned arterial mileage may be expected to be moderately congested; 54 miles, or about 2 percent, of planned arterial mileage may be expected to be severely congested; and 7 miles, less than 1 percent, may be expected to be extremely congested.

serve the existing regional land use pattern and promote the implementation of the regional land use plan. This objective is supported by three standards. Each standard relates to measures of accessibility provided by the transportation system.

Standard No. 1 of Objective No. 1 indicates that the regional transportation system should provide service within each urbanized area of the Region so that all residents of those areas are within 30 and 45 minutes' overall travel time of at least 40 percent of the employment opportunities in the urbanized area by arterial street and highway and transit modes, respectively; 35 minutes' overall travel time of at least three major retail and service centers (one major center within the Kenosha and Racine urbanized areas); 40 minutes' overall travel time of a major medical center and/or 30 minutes' overall travel time of a hospital or medical clinic; 40 minutes' overall travel time of a major park or outdoor recreation area; 40 minutes' overall travel time of a vocational school, college, or university; and 60 minutes' overall travel time of an airport offering scheduled air transportation service. that is, General Mitchell International Airport.

Overall travel time is defined as the total doorto-door time of travel from origin to destination. This includes the time required to arrive at the transportation vehicle, as well as the over-theroad travel time. The results of these analyses for the arterial street and highway components of the alternative transportation system plans are shown in Table 232. The results of the analyses for the transit components of the alternative plans are shown in Table 233.

When comparing the accessibility provided by transit with that provided by the automobile, it should be noted that nearly one-half of travel time entailed in making a local transit trip is "out-of-vehicle" time, that is, time spent walking to a transit stop, waiting for a transit vehicle, and walking to a trip destination. Significantly less out-of-vehicle travel time is usually involved in making the same trip by automobile. Consequently, typically more activities can be accessed within a given time by automobile than by transit.

As indicated in Table 232, the alternative plans may be expected to meet the highway component of Standard No. 1 equally well. As indicated in Table 233, however, the plans differ significantly in the degree to which they meet the standard when applied to transit. Alternative Plan 1 may be expected to provided a greater degree of accessibility to the residents of the urbanized areas who use transit than Alternative Plan 3 or the no-build plan. Indeed, 91 percent, 49 percent, and 81 percent of the Kenosha, Milwaukee and Racine urbanized area populations, respectively, may be expected to meet the employment standard by transit under Alternative Plan 1. Alternative Plan 3 provides accessibility in this regard to about 50 percent, 19 percent, and 72 percent of the Kenosha, Milwaukee, and Racine urbanized area populations, respectively. Despite the differences in the level of accessibility provided by Alternative Plans 1 and 3, the implementation of either plan would contribute significantly to improving the access of transit users to employment and major activity centers. As shown in Table 233, less than 1 percent of the Milwaukee urbanized area populations may be expected to access 40 percent of urbanized area jobs within 45 minutes under the no-build plan.

It is also apparent from the above analysis that the provision of transit services within a large urbanized area such as Milwaukee at a level sufficient to meet land use service standards such as those set forth in Objective No.1 is difficult because of the extent to which a diffused, low-density land use pattern has already been allowed to develop. The analysis indicates that persons dependent upon transit may not have good access to as many opportunities within a given travel time as they would if an automobile were available. Map 102 shows the portions of the urbanized areas of the Region that may be expected to be within 30 minutes' travel time of 40 percent of the urbanized area jobs by arterial street and highway under the alternative plans. Map 103 shows the portions of the urbanized area that may be expected to be within 45 minutes' travel time of 40 percent of the jobs within the particular urbanized area by transit under the alternative plans. Maps showing the results of applying the remaining travel time standards of Standard No. 1 are included in Appendix C of this report.

Standard No. 2 indicates that the transportation system should be adjusted to the regional land use plan so that a higher relative accessibility is provided to areas in which higher-density development is planned than to areas in which

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON ARTERIAL STREETS AND HIGHWAYS: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

· · · · · · · · · · · · · · · · · · ·												
					Urbanized Area Population Meeting Travel							
	Urbaniz	ed Area		Time Standard on Arterial Streets and Highways								
	Popu	lation		Proposed 2010								
			- Base Year		No-Build Transportation		Alternative		Alternative			
Urbanized Area and	Existing		199	1	System	Plan	Plan	1.	Plan	3		
Activity Center	1991	2010	Number	Percent	Number	Percent	Number	Percent	Number	Percent		
Kenosha Urbanized Area	94,300	107,300								·		
Employment-Related ^a		••	90.000	96.4	107,300	100.0	107.300	100.0	107.300	100.0		
Major Retail-Service ^b			, 0	0.0	0	0.0	. 0	0.0	· 0	0.0		
Medical Facility ^C		·	94,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0		
Major Park ^d			94,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0		
Higher-Education Facility ^e			94,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0		
Scheduled Air Transport ^f	••		94,300	100.0	107,300	100.0	107,300	100.0	107,300	100.0		
Milwaukee Urbanized Area	1,226,300	1,277,100										
Employment-Related			1,147,900	93.6	1,180,500	92.4	1,230,800	94.3	1,230,800	94.3		
Major Retail-Service			1,161,400	94.7	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0		
Medical Facility			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0		
Major Park			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0		
Higher-Education Facility			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0		
Scheduled Air Transport			1,226,300	100.0	1,277,100	100.0	1,277,100	100.0	1,277,100	100.0		
Racine Urbanized Area	121,800	132,100					·		, ,			
Employment-Related			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0		
Major Retail-Service			32,200	86.4	132,100	100.0	132,100	100.0	132,100	100.0		
Medical Facility			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0		
Major Park			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0		
Higher-Education Facility			121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0		
Scheduled Air Transport		• -	121,800	100.0	132,100	100.0	132,100	100.0	132,100	100.0		

^aStandard: 30 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of three major retail and service centers.

^CStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

Source: SEWRPC.

low-density development is planned or areas which should be protected from development. In order to determine the relative accessibility of the various subareas of the Region under base year 1991 conditions and under proposed transportation system plans, combined accessibility indices were computed for all traffic analysis zones within the Region.

The combined accessibility index measures the ease with which any land use activity can be reached from any other land use activity within the Region. Under this approach a high index indicates a greater degree of accessibility than a low index. Land use activity is defined as the location to which a trip destination can be satisfied for home-based work, home-based shopping, home-based other, and nonhome-based trip purposes. The combined index for any zone was calculated by summing the four indices computed for the zone, one for each of four trip purposes. These combined indices were plotted as iso-accessibility lines for each of the alternative transportation system plans and for base year 1991 conditions, and are shown on Map 104.

In essence, the accessibility index measures the relative ease with which a traveller living at a given point can reach all possible trip destinations. If the travel times to destinations are

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON TRANSIT 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

	Urbaniz	ad Aron	Urbanized Area Population Meeting Travel Time Standard on Transit							
	Popu	lation			Proposed 2010					
Urbanized Area and	Base Year		Base Year 1991		No-Build Transportation System Plan		Alternative Plan 1		Alternative Plan 3	
Activity Center	1991	2010	Number	Percent	Number	Percent	Number	Percent	Number	Percent
Kenosha Urbanized Area Employment-Related ^a Majaz Rataji Savijas ^b	94,300	107,300 	58,000	61.5	51,400	47.9	97,700	91.1	53,200	49.6
Major Peran-Service			60,600 12,600	64.3 13.4	53,300 20,700	49.7 19.3	63,600 77,500	59.3 72.2	53,300 24,700	49.7 23.0
Scheduled Air Transport ^f	·		33,200 0	35.2 0.0	51,400	47.9	80,800 51,900	75.3 48.4	51,900	48.4
Milwaykaa Urbanizad Area	1 226 200	1 277 100								
Employment-Related	1,220,300	1,277,100	14 200	1 2	600	0.0	610 000	48.5	242 600	10.1
Major Betail-Service9			10,100	0.8	35 300	2.8	282 800	22.1	66 200	52
Medical Facility			700 400	57 1	648,000	50.7	870,200	68 1	728,900	57.1
Major Park			636,300	51.9	617.300	48.3	994,200	77.8	701.000	54.9
Higher-Education Facility			775,600	63.2	740,900	58.0	1,168,000	91.5	931,900	73.0
Scheduled Air Transport			338,600	27.6	316,800	24.8	776,800	60.8	569,400	44.6
Racine Urbanized Area	121,800	132,100								
Employment-Related			59,100	48.5	56,400	42.7	107,500	81.4	94,400	71.5
Major Retail-Service			21,500	17.7	18,300	13.9	52,500	29.7	32,100	24.3
Medical Facility			53,700	44.1	46,700	35.4	61,300	46.4	49,200	37.2
Major Park			24,000	19.7	37,000	28.0	59,800	45.3	38,300	29.0
Higher-Education Facility			79,600	65.4	88,400	66.9	110,700	83.8	102,700	77.7
Scheduled Air Transport			18,300	15.0	18,600	14.1	63,800	48.3	60,100	45.5

^aStandard: 45 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of one major retail and service center.

^CStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

 $g_{\rm Standard:}$ 35 minutes' overall travel time of three major retail and service centers.

Source: SEWRPC.

relatively low because of the proximity of the points to the trip origin, or because of the efficiency of the transportation system, a high value for the index will be evident. A high accessibility index value will also be evident if the destination points nearest a traveller's origin have large population or employment bases. On the other hand, if a traveller is located at a point relatively distant in travel time from major trip destinations, either because of low speeds or indirect routes on the transportation system, a low accessibility index will be evident.

As indicated by the accompanying maps, rural areas of the Region generally have lower accessibility than urbanized areas. The areas of highest accessibility within the Region, both under existing and alternative plan conditions. are generally located in western and northwestern Milwaukee County and in eastern Waukesha County. From these locations of relatively high accessibility, accessibility levels decrease in irregular concentric rings to the fringe areas of the Region, with certain exceptions of high relative accessibility in the Cities of Kenosha and Racine. Changes in accessibility levels due to transportation system improvements can be identified by comparing alternative plans with the no-build plan. Such comparisons indicate that the transportation improvements will only marginally change accessibility, as measured by the index, within the Region.

Objective No. 2: Minimize Costs

The second transportation system development objective concerns the achievement of a transportation system which is economical and efficient, satisfying all other objectives at the lowest possible cost. This objective is supported by three standards relating to transportation system costs and benefits and utilization of existing transportation facilities.

Standard No. 1 indicates that the sum of the transportation system operating and capital investment costs should be minimized. Accordingly, an analysis was made to determine the estimated total costs of implementing each alternative transportation system plan. These estimates were prepared by applying unit improvement costs to the estimated mileage of proposed future facility improvements, including acquisition of rights-of-way and construction of new arterial street and highway facilities, and by preparing special estimates of the cost of constructing proposed transit facilities, including transitways and preferential transit lanes, transit parking stations, transit vehicles, and transit vehicle storage and maintenance facilities. In addition, the costs of operating and maintaining the existing and proposed arterial street and highway and transit systems were estimated. Operating cost estimates included not only the costs to public implementing agencies with responsibilities for operating and maintaining the transportation system, but also the operating cost to system users in terms of time, vehicle operating costs, and accident costs. All cost estimates have been developed in constant 1994 dollars.

Rights-of-way costs include the cost to acquire land and structures necessary to provide a rightof-way capable of accommodating the required improvement cross-section, as well as the costs associated with relocation and assistance payments. The right-of-way and relocation costs were based upon recent experience of local units of government and the Wisconsin Department of Transportation in the Southeastern Wisconsin Region.

Construction costs, which include engineering and utility relocation, also reflect capital cost experiences of highway agencies within the Region and have been developed for each street and highway cross-section type. Where applicable, the construction costs reflect current estimates developed for those projects. In addition to the initial investment required to provide new or expanded facility capacity, resurfacing of all other streets and highways was assumed during the plan implementation period, as well as the additional resurfacing of some arterial facilities a second time to achieve adequate pavement life over the 16-year plan design period. The cost of reconstructing and resurfacing collector and minor streets during the plan implementation period was calculated on the basis of unit cost data for each crosssection type and the amount of such mileage to be preserved.

Unit operation and maintenance cost data for arterial streets were developed from information provided by the Wisconsin Department of Transportation and local units of government in the Region. Similarly, unit operation and maintenance cost data for transit facilities were developed from information provided by local transit agencies in the Region.

User costs include the value of time spent in travel, the cost of accidents, and, in the case of auto and truck users, the out-of-pocket costs of vehicle operation and parking, depreciation, and that portion of insurance costs not represented in accident costs. Vehicle operating costs were developed on the basis of gasoline used at \$1.10 per gallon as a function of vehicle and facility types used plus a cost of \$0.15 per mile. Heavy duty truck operating costs were developed to include estimates of the cost of fuel used plus \$0.25 per mile, a function of vehicle speed. The value of time spent in automobile and transit travel was estimated at \$3.00 per hour. The trip value attendant to light commercial truck operations was estimated at \$9.00 per hour. The trip value attendant to heavy duty truck operations was estimated at \$15 per hour. Although transit fares are out-of-pocket costs to the transit user, in an economic analysis such fares represent revenue to offset transit operating and maintenance costs. Transit fares are not, therefore, included as a transit user cost since such inclusion would represent a double accounting. Transit fares, therefore, were treated in the same manner as road user taxes.

The resulting total cost estimates for each of the alternative transportation system plans are set forth in summary form in Table 234. Alternative Plan 3, with an estimated total cost of \$98.1 billion, would be the least expensive "build" plan. Of this total, about \$5.74 billion would

AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY ARTERIAL STREET AND HIGHWAY



2010 NO-BUILD ALTERNATIVE PLAN

LEGEND URBANIZED AREA BOUNDARY AREA MEETING TRAVEL TIME STANDARD ⁹30 MINUTES OVERALL TRAVEL TIME OF 40 PERCENT OF URBANIZED Contraine State AREA EMPLOYMENT OPPORTUNITIES. washin hand have άi. RENOSHA CO. 2.4 WISCONSIN COL

2010 ALTERNATIVE PLAN 1

Map 102 (continued)

2010 ALTERNATIVE PLAN 3



In terms of timely access to employment opportunities, adequate highway service is nearly ubiquitous throughout the urbanized areas of the Region and may be expected to remain so, as indicated on the accompanying maps. Nearly 100 percent of the population of the Kenosha and Racine urbanized areas and about 95 percent of the population the Milwaukee urbanized area may be expected to be able to access 40 percent of the urbanized area jobs within 30 minutes through arterial travel under each alternative plan.

Source: SEWRPC,

AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY TRANSIT

2010 NO-BUILD ALTERNATIVE PLAN



2010 ALTERNATIVE PLAN 1



Map 103 (continued)

2010 ALTERNATIVE PLAN 3



In terms of providing timely access to employment opportunities throughout the urbanized areas, public transit service is not as adequate as highway service, though some substantial improvements in such service could be expected under Alternative Plans 1 and 3, as is indicated on the accompanying maps. In the Milwaukee urbanized area in particular, the percent of urbanized area population able to access 40 percent of urbanized area employment opportunities within 45 minutes through transit travel may be expected to increase from 0 percent to nearly 49 percent under Alternative Plan 1 and to about 20 percent under Alternative Plan 3.

Source: SEWRPC.

Map 104

ACCESSIBILITY TO LAND USE ACTIVITIES IN THE REGION





2010 NO-BUILD ALTERNATIVE PLAN

466

2010 ALTERNATIVE PLAN 1

2010 ALTERNATIVE PLAN 3





This series of four maps graphically portrays the relative accessibility in the Region under base year 1991 land use and transportation system conditions and under year 2010 land use and proposed transportation system conditions. The areas of highest relative accessibility, both under existing conditions and under the alternative plan, are generally located in western and northwestern Milwaukee County and in eastern Waukesha County. From these locations, the accessibility levels decrease in irregular concentric circles toward the fringe areas of the Region, excepting only small areas of relatively high accessibility in the Cities of Kenosha and Racine. Accessibility levels in the Region could be expected to increase throughout the Region under all of the alternative plans. The major cause of change in accessibility levels would be population and associated land use redistribution, rather than transportation system improvements.

Source: SEWRPC.

TRANSPORTATION SYSTEM CAPITAL, OPERATION AND MAINTENANCE, AND USER COSTS IN THE REGION OVER THE PERIOD 1994-2010: NO-BUILD AND ALTERNATIVE REGIONAL TRANSPORTATION SYSTEM PLANS

	· · · · · · · · · · · · · · · · · · ·		
	Cu	mulative Cost: 1995-20	10
Cost Element	No-Build Transportation System Plan	Alternative Plan 1	Alternative Plan 3
System Element Costs Arterial Streets and Highways			
Construction	\$ 3,040,000,000 1,072,000,000	\$ 4,768,000,000 1,088,000,000	\$ 4,832,000,000 1,088.000.000
Subtotal	\$ 4,112,000,000	\$ 5,856,000,000	\$ 5,920,000,000
Public Transit Construction Operation and Maintenance Subtotal	\$ 304,000,000 1,344,000,000 \$ 1,648,000,000	\$ 1,040,000,000 1,840,000,000 \$ 2,880,000,000	\$ 912,000,000 1,488,000,000 \$ 2,400,000,000
Subtotal	\$ 5,760,000,000	\$ 8,736,000,000	\$ 8,320,000,000
User Costs Street and Highway			
Time	\$27,129,300,000 44,156,900,000	\$ 24,715,200,000 49,617,000,000	\$26,298,800,000 43,307,000,000
Accident	19,544,000,000 \$90,830,200,000	18,192,000,000 \$ 92,524,200,000	19,288,000,000 \$88,893,800,000
Transit			
Time	\$ 700,800,000 152,000,000	\$ 855,000,000 224,000,000	\$ 707,900,000 176,000,000
Subtotal	\$ 852,800,000	\$ 1,079,000,000	\$ 883,900,000
Subtotal	\$91,683,000,000	\$ 93,603,200,000	\$89,777,700,000
Total	\$97,443,000,000	\$102,339,200,000	\$98,097,700,000

Source: SEWRPC.

represent capital or construction costs, about \$2.58 billion operation and maintenance costs, and \$89.78 billion users costs. Total costs under Alternative Plan 1 would approximate \$102.34 billion. Of this total, about \$5.81 billion would represent capital costs, about \$2.93 operation and maintenance costs, and about \$93.60 billion user costs. The no-build plan would have an estimated total cost of \$97.44 billion. Of this total, about \$3.34 billion would represent capital costs, about \$2.42 billion operation and maintenance costs, and about \$91.68 billion user costs.

Standard No. 2 indicates that the direct benefits derived from transportation system improvements should exceed the direct costs of such improvements. Application of this standard permits a comparative analysis of the "build" alternatives, those that include transportation system improvements, with the no-build alterna-

tive plan. The direct benefits derived from transportation system improvements include a reduction in the cost of travel time, of vehicle operation, and of accidents that are achieved through improvements to the transportation system. The direct costs of such improvements are composed of the necessary capital investments and the cost to public agencies of operating and maintaining the physical facilities and transportation services. A benefit-cost ratio was accordingly derived for the "build" plans by computing the difference in transit and highway user costs between the no-build alternative and the "build" alternatives and dividing that number by the difference in transportation system costs between the no-build alternative and the "build" alternatives.

The benefit-cost analysis is intended to demonstrate the economic value of the transportation

Alternative Plan	1994-2035 Road User Costs ^a (millions)	Construction, Operation, and Maintenance Costs ^a 1994-2035 (millions)	Incremental Benefits (millions)	Incremental Costs (millions)	Benefit-Cost Ratio
No-Build	\$135,222.6	\$6,869.1			
Alternative Plan 1	129,979.8	9,984.1	\$5,242.8	\$3,115.0	1.68
Alternative Plan 3	132,538.7	9,138.7	2,683.9	2,269.6	1.18

COMPARISON OF THE USER AND SYSTEM COSTS AND BENEFIT-COST RATIOS: 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLANS

^aPresent worth in 1994 costs incurred between 1994 and 2035 under proposed transportation system plan alternatives.

It should be noted that a 3 percent rate of return was used in the calculation of the present worth of both plan benefits and costs. An analysis of an alternative rate of return, namely 4 percent, indicated that the benefit-cost ratio for Alternative Plan 1 would be 1.55 and for Alternative Plan 3 would be 1.06.

Source: SEWRPC.

system plan proposals. In presenting the results of the benefit-cost analysis, it should be noted that benefits and costs were calculated as accruing over a period of time extending from 1994 to 2035 in order to bring the salvage value of each staged facility recommended in the plans to zero. It should also be noted that the benefitcost ratios set forth in Table 235 apply to the aggregation of system improvements proposed in the regional transportation plans and do not imply that each individual project within this aggregation will have a similar benefit-cost ratio.

The economic analysis resulted in a benefit-cost ratio of 1.68 for Alternative Plan 1. This means that for every dollar invested in implementing the plan, transportation system users, on the average, could be expected to enjoy \$1.68 in benefits in terms of reduced accidents, travel time and vehicle operation costs. The benefitcost ratio of Alternative Plan 3 is 1.18.

The next standard of Objective No. 2 was used as a fundamental input in the design of both Alternative Plans 1 and 3. Standard No. 3 indicates that the full use of all existing transportation facilities should be encouraged through low-capital- and noncapital-intensive transportation system management measures, cooperatively fostered by government, business, and industry prior to any capital-intensive or disruptive construction or provision of new facilities and services. In essence, this standard specifies that full use should be made of all existing transportation facilities.

Standard No. 3 was used to constrain the plan design process. For example, in the computation of system capacity, curb parking was assumed to have been eliminated in the peak direction during the peak hours on all arterial streets in order to minimize traffic congestion and the need for additional capital investment. Similarly, every effort was made in the plan design to facilitate a higher level of transit service by utilizing available arterial street capacity for the exclusive or preferential movement of transit vehicles in major transportation corridors. It should be understood that the specific development and implementation of the traffic management techniques assumed in system plan design will require local planning and engineering and plan implementation actions. Such actions are essential if congestion is to be reduced and kept at tolerable levels.

Objective No. 3: Provide a

Multimodal Transportation System

The third transportation system development objective calls for the achievement of a multimodal transportation system which will provide appropriate types transportation needed by all residents of the various subareas of the Region at an adequate level of service and which will

Location	No-Build Alternative	Alternative Plan 1	Alternative Plan 3
Urbanized Area Kenosha	84,900 1,175,800 108,800	99,300 1,267,600 127,200	90,400 1,252,700 126,900
Subtotal	1,369,500	1,494,100	1,470,000
Nonurbanized Area	67,300	138,700	77,100
Total	1,436,500	1,632,800	1,547,100

TOTAL POPULATION SERVED BY PUBLIC TRANSIT: 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS^a

^aDoes not include the population served by demand-responsive shared-ride taxi service.

Source: SEWRPC.

permit ready adaptation to both changes in travel demand and in transportation technology, modal use, and new transportation management measures. This objective is supported by two arterial street and highway standards, sixteen public transit standards, and one bicycle and pedestrian facility standard.

Standard No. 1 indicates that arterial streets and highways should be provided at intervals of no more than one-half mile in each direction in urban high-density areas, at intervals of no more than one mile in each direction in urban medium-density areas, at intervals of no more than two miles in each direction in urban lowdensity and suburban residential areas, and at intervals of no less than two miles in each direction in rural areas. This standard has been essentially met under each of the alternative plans because it served as an input to the plan design process. It is important to note that the initial preparation and subsequent amendments to the seven county jurisdictional highway system plans serve to ensure, to the maximum extent practical, the designation of an arterial street and highway system properly related to this standard.

Standard No. 2 states that freeways or expressways should be considered for those travel corridors within the Region which meet all of the following criteria: the corridor provides intercommunity service, the desired speed or a volume-todesign-capacity ratio of 1.0 requires control of access and uninterrupted flow, and potential average weekday traffic exceeds 45,000 vehicles per day in urban areas and 25,000 vehicles per day in rural areas. This standard was also applied as an input to the plan design process.

Objective No. 3, as it relates to public transit service, is supported by sixteen standards. Three of the standards are explicitly comparative in nature and thus are set forth here. The remaining standards under this objective either have all been met in the design of the alternative plans or could be met by the plans if those plans are properly implemented.

The first comparative transit-related standard indicates that the number of persons served by public transit should be maximized. The Region's population which would be located within the combined service areas or rapid. express, and local transit in the year 2010 totals about 1.63 million, or about 85 percent of the Region's population, under Alternative Plan 1; about 1.55 million, or about 81 percent of the Region's population, under Alternative Plan 3; and about 1.44 million, or about 75 percent of the Region's population, under the no-build alternative plan (see Table 236). Thus, Alternative Plan 1 best meets this standard as it proposes to provide service to about 85,600 more residents than Alternative Plan 3 and to about 196,000 more residents than no-build plan.

The second comparative standard indicates that the number of jobs served by rapid transit should be maximized. The standard states that a job shall be considered served by rapid transit

Location	No-Build Alternative	Alternative Plan 1	Alternative Plan 3
Urbanized Area Kenosha	0 146,700 0	24,400 328,000 26,500	24,400 335,600 26,500
Subtotal	146,700	378,900	386,500
Nonurbanized Area	0	34,000	21,800
Total	146,700	412,900	408,300

NUMBER OF JOBS SERVED BY RAPID TRANSIT: 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

Source: SEWRPC.

if it is located within one-half mile walking distance, or a 15-minute feeder-bus ride, from a rapid transit stop. As shown in Table 237, the jobs which would be served by rapid transit total 412,900 under Alternative Plan 1, or about 38 percent of the jobs in the Region; 408,300 under Alternative Plan 3, or about 37 percent of the jobs in the year 2010 in the Region; and 146,700 under the no-build plan, or about 13 percent of the jobs in the Region. Though Alternative Plan 1 best meets this standard, both "build" alternatives represent significant improvements over the no-build plan; indicative of the expansion of rapid transit routes within the Region proposed under these plans.

The third comparative standard indicates that the proportion of total person trips to the Milwaukee CBD made by public transit should be increased to at least 30 percent. Under Alternative Plan 1, 54,000 of the 360,200 total person trips to the Milwaukee CBD on an average weekday in the year 2010, or 15 percent, may be expected to be made by transit. Under Alternative Plan 3, 40,800 trips of the 364,200 total trips to the Milwaukee CBD on an average weekday in the year 2010, or 11 percent, may be expected to be made by transit. Under the no-build alternative, 37,900 of the 364,700 total trips to the Milwaukee CBD on an average weekday in the year 2010, or 10 percent, may be expected to be made by public transit.

The last standard supporting Objective No. 3 indicates that appropriate bicycle and pedestrian facilities should be provided on those arterial streets and highways identified in the regional bicycle and pedestrian facilities plan for Southeastern Wisconsin. This standard was assumed to be met equally well by each of the two "build" alternative plans considered.

Objective No. 4: Minimize Disruption

The fourth transportation objective is to minimize disruption by the transportation system of existing neighborhoods and community development including adverse affects upon the property tax base. This objective is supported by eight standards.

Standard No. 1 indicates that the penetration of neighborhood units and facility service areas by arterial streets and highways and rapid transit routes should be minimized. A precise quantitative evaluation of the extent to which this standard is met depends, in large part, upon the completion of project planning, including a determination of the precise alignment of new transportation facilities. In addition, it is necessary for local units of government to delineate neighborhood boundaries.

Standard No. 2 provides that the dislocation of households, businesses, industries, and public and institutional buildings caused by the reconstruction of existing, or the construction of new, transportation facilities and terminals should be minimized. In order to estimate the extent to which the alternative plans met this standard, an estimate was made of the number of residential and nonresidential units that would have to be displaced and relocated because of transpor-

COMPARISON OF LAND-TAKING REQUIREMENTS FOR TRANSPORTATION SYSTEM IMPROVEMENTS IN THE REGION BY COUNTY: 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

County and Taking	No-Build	Alternative	Alternative
	Alternative	Plan 1	Plan 3
Kenosha County Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost		62	87
		32	33
	\$ 198,000	\$ 15,996,000	\$ 16,864,000
Milwaukee County	11	61	61
Number of Residential Units	8	43	43
Number of Nonresidential Structures	\$1,768,000	\$ 11,231,000	\$ 9,532,000
Ozaukee County Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost	9 1 \$1,210,000	40 2 \$ 12,344,000	53 5 \$ 13,721,000
Racine County Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost	 \$ 68,000	191 27 \$ 23,761,000	198 27 \$ 25,236,000
Walworth County Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost	 \$ 92,000	44 8 \$ 25,786,000	46 4 \$ 49,084,000
Washington County Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost	8	128	138
	1	10	12
	\$ 369,000	\$ 19,302,000	\$ 20,058,000
Waukesha County Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost	11	125	137
	4	41	41
	\$ 999,000	\$ 37,023,000	\$ 36,180,000
Southeastern Wisconsin Region Number of Residential Units Number of Nonresidential Structures Acquisition, Demolition, and Relocation Cost	39 14 \$4,704,000	651 163 \$145,393,000	720 165 \$170,675,000

Source: SEWRPC.

tation system improvements. The results of this analysis are set forth in Table 238. The number of residential structures estimated to be displaced would total 651 under Alternative Plan 1 and 720 under Alternative Plan 3. The number of nonresidential structures estimated to be displaced would total 163 under Alternative Plan 1, 165 under Alternative Plan 3.

Standard No. 3 indicates that the amount of land used for new transportation facilities should be minimized. To estimate the relative extent to which each of the alternative plans meets the standard, an estimate was made of the amount of land required for new facilities. This estimate, expressed in terms of total acquisition, demolition, and relocation costs required to carry out each alternative plan, is also set forth in Table 238. This cost would total about \$4.70 million under the no-build plan, about \$145.39 million under Alternative Plan 1, and about \$170.68 million under Alternative Plan 3.

Standard No. 4 indicates that the reduction of the property tax base caused by the reconstruction of existing or the construction of new transportation facilities and terminals should be minimized. The analysis set forth above for minimizing the total amount of land needed to provide new and improved transportation facilities under the alternative plans would hold equally true for this standard on the assumption that the direct impact upon the property tax base of land and building acquisition for transportation facility improvements would be directly proportional to the cost of such land and building acquisition. Accordingly, since under Alternative Plan 1, less land-taking would be required than under Alternative Plan 3, it may be concluded that the impact upon the property tax base would be less under Alternative Plan 1 than under Alternative Plan 3. Thus, Alternative Plan 1 may be considered to meet this transportation system development standard better.

While this analysis would indicate that the acquisition of land and buildings results in a direct tax base loss, it can be established that there may, in fact, be no tax base loss to the Region if the property owners relocate elsewhere in the Region. Accordingly, a presumed cost due to tax base loss is not used in the benefit-cost analysis under system level planning. Indeed, what may appear as a loss as a result of improvements in one area, or to one community, may result in a gain in the tax base elsewhere. In addition, improvements to transportation facilities have often increased the value of abutting properties so that a tax base gain follows a facility improvement.

Standard No. 5 states that the destruction of historic buildings and of historical, scenic, scientific, archaeological, and cultural sites caused by the reconstruction of existing or the construction of planned transportation facilities and terminals should be minimized. Attainment of this standard cannot be measured at the system level of planning since the standard can only be met through attention to historic preservation at the project level of planning. Accordingly, this standard was assumed to be met equally well by each alternative plan considered.

Standard No. 6 states that the proper use of land for, and adjacent to, transportation facilities should be maximized and the disruption of future development minimized through advance reservation of rights-of-way for transportation facilities. This standard was assumed to be met equally well by each alternative transportation system plan considered. The proper use of land adjacent to new transportation facilities, arterial streets and highways, public transit lines and stations, as well as the preservation of land for such future public works, can be effectively ensured through vigorous attention to plan implementation and proper use of available land use regulatory techniques by the State and local units and agencies of government concerned.

Standard No. 7 indicates that transportation facility construction plans should be developed using sound geometric, structural, and landscape design standards which consider the aesthetic quality of the transportation facilities and the areas through which they pass and which consider any environmental enhancement activities likely to be required. This standard can be met through proper facility design during plan implementation. Accordingly, this standard was assumed to be met equally well by each alternative plan considered.

Standard No. 8 indicates that transportation facilities should be located so as to avoid destruction of visually pleasing buildings, structures, and natural features and to enhance vistas to such features. This standard can be met through careful facility design during plan implementation by each alternative plan considered. Accordingly, this standard was assumed to be met equally well by each alternative plan considered.

Objective No. 5: Protect Natural

Environment, Promote Public Health, and Achieve Ambient Air Quality Standards

The fifth transportation system development objective is a transportation system which serves to protect the overall quality of the natural environment, which promotes the public health, and which helps to achieve ambient air quality standards. Four specific standards support this objective.

Standard No. 1 states that the location of transportation facilities in, or through, primary environmental corridors, particularly through the woodland and wetland portions of such corridors, should be minimized. To determine the relative extent to which each of the alternative plans met the standard, an analysis was made of the location of all new, or substantially improved, transportation facilities proposed under the plan. The results of this analysis are summarized in Table 239.

	Wetland	s (acres)	Woodlands (acres)		Other	(acres)	Total Primary Environmental Corridor Lands (acres)	
County	Alternative Plan 1	Alternative Plan 3	Alternative Plan 1	Alternative Plan 3	Alternative Plan 1	Alternative Plan 3	Alternative Plan 1	Alternative Plan 3
Kenosha	3.5	3.9	2.7	3.5	3.4	4.5	9.6	11.9
Milwaukee	7.8	7.8	0.9	0.9	7.8	7.8	16.5	16.5
Ozaukee	6.5	6.5	2.3	2.3	10.9	10.9	19.7	19.7
Racine	6.5	6.5	5.6	5.6	4.6	4.6	16.7	16.7
Walworth	15.0	19.8	8.5	34.9	14.4	22.7	37.9	77.4
Washington	141.8	141.8	19.1	19.1	61.3	61.3	222.2	222.2
Waukesha	69.1	70.4	18.1	18.2	22.1	22.4	109.3	111.0
Region	250.2	256.7	57.2	84.5	124.5	134.2	431.9	475.4

LOSS OF PRIMARY ENVIRONMENTAL CORRIDORS IN THE REGION TO IMPROVEMENT AND EXPANSION OF ARTERIAL STREET AND HIGHWAY FACILITIES: 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLANS 1 AND 3

Source: SEWRPC.

As shown in Table 239, the number acres of primary environmental corridor land, including upland and wetland areas, that may be expected to be lost to new or substantially improved transportation facilities would approximate 432 acres under Alternative Plan 1 and 475 acres under Alternative Plan 3. Under Alternative Plan 1, 250 acres of primary environmental corridor wetlands would be lost and under Alternative Plan 3, 256 acres of primary environmental corridor wetlands would be lost. About one-tenth of 1 percent of the total 303,200 planned acres of primary environmental corridor in the Region would be lost as a result of the construction or improvement of transportation facilities envisioned in the alternative plans.

It is important to note that the foregoing analysis had been made at the system planning level and is necessarily general in nature. A more detailed examination during the project planning and design phases may indicate that the new or improved transportation facilities could be placed outside the environmental corridor lands. This is particularly possible where transportation facilities are to be constructed along an alignment that approximately parallels to a primary environmental corridor.

Standard No. 2 states that any damaging affects on the natural resource base caused by the construction of transportation facilities should be minimized. This standard can be met through proper facility design and proper use of appropriate construction techniques during plan implementation. Accordingly, this standard was assumed to be met equally well by each alternative plan considered.

Standard No. 3 states that the amount of air pollutants emitted through the operation of motor vehicles on the arterial street and highway system should be minimized. Table 240 shows estimates of the amount of volatile organic compounds, nitrogen oxide, and carbon monoxide pollutant emissions generated in 1990 and expected to be generated in 2010 under the no-build plan, and Alternate Plans 1 and 3. As shown in Table 240, Alternative Plan 1 best meets this standard. Under Alternative Plan 1, about 25.6 tons of volatile organic compound may be expected to be generated per day, 3.1 tons per day fewer than under Alternative Plan 3; 71.3 tons of nitrogen oxides may be expected to be generated per day, 8.6 tons per day fewer than under Alternative Plan 3; and 211.5 tons of carbon monoxide may be expected to be generated per day, 26.7 tons per day fewer than under Alternative Plan 3.

Significant reductions in volatile organic compounds, nitrogen oxides, and carbon monoxide emissions may be expected by the year 2010. Such reductions will occur despite forecast increases in travel within the Region over time. These reductions are expected to come about
AVERAGE WEEKDAY AIR-POLLUTANT EMISSIONS IN THE REGION BY TRANSPORTATION SYSTEM PLAN: 1990 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

	and the second	and the second	and the second
Transportation System	Volatile Organic	Nitrogen Oxides	Carbon Monoxide
	Compounds (tons per day)	(tons per day)	(tons per day)
1990 Existing2010 No-Build Alternative2010 Alternative Plan	119.2	129.8	815.1
	29.8	81.4	248.4
	25.6	71.3	211.5
2010 Alternative Plan 3	28.7	79.9	238.2

Source: SEWRPC.

because of Federally mandated changes in vehicle emission standards and through the required use of cleaner-burning motor fuels.

Table 240 also indicates that both of the "build" alternative plans result in lower levels of air pollutant emissions than the no-build plan. This is directly related to the lower levels of vehiclemiles of travel anticipated under Alternative Plans 1 and 3. This potential reduction in transportation-related air pollutant emissions is significant because, to the extent to which mobile-source emissions can be reduced, fewer restrictions on industrial source emissions will be necessary to meet national ambient air quality standards within the Region.

Standard No. 4 indicates that the loss of prime agricultural land to transportation facility construction should be minimized. To determine the relative extent to which each of the alternative plans met this standard, an analysis was made of the location of all new or substantially improved transportation facilities proposed under each plan. The results of this analysis are summarized in Table 241. The number of acres of prime agricultural areas within the Region that may be lost to the improvement and expansion of the arterial street and highway system approximates 1,039 acres under Alternative Plan 1 and 1,644 acres under Alternative Plan 3. About 95 percent of the difference in acreage lost can be attributed to the construction of the longplanned USH 12 freeway in Walworth County included in Alternative Plan 3.

Objective No. 6: Facilitate Movement

between Component Parts of the Region

The principle for Objective No. 6 indicates that to support everyday economic and social activities, a transportation system which provides for

Table 241

LOSS OF PRIME AGRICULTURAL LAND IN THE REGION TO IMPROVEMENT AND EXPANSION OF ARTERIAL STREET AND HIGHWAY FACILITIES: 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLANS 1 AND 3

County	Alternative Plan 1 (acres)	Alternative Plan 3 (acres)
Kenosha	39.1	39.1
Milwaukee	0.0	0.0
Ozaukee	40.1	49.3
Racine	119.0	120.2
Walworth	320.1	895.5
Washington	302.8	311.1
Waukesha	217.6	228.5
Region	1,038.7	1,643.7

Source: SEWRPC.

reasonably fast, convenient travel is essential. It further implies that traffic congestion increases the cost of transportation and can adversely affect the attractiveness of an area for residential use and for the location and operation of businesses and industries. Eight specific standards support Objective No. 6.

Standard Nos. 1, 2, and 3 indicate that the total passenger-hours and vehicle-hours of travel and the total miles of travel within the Region, respectively, should be minimized. Data pertaining to these standards are set forth in Table 242. Alternative Plan 1 may be expected to generate about 38.38 million total highway and about 142,000 total transit vehicle-miles and 1.02 million highway vehicle-hours and 9,200 transit vehicle-hours on an average weekday. By contrast, Alternative Plan 3 may be expected to

COMPARISON OF THE AMOUNT OF TOTAL TRAVEL IN THE REGION: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

		·			20	10	- <u> </u>	
	Base 19	Base Year 1991		Build ortation m Plan	Alterr Pla	native	Alterr	native n 3
Travel Characteristics	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total
Vehicle-Miles of Travel (thousands) Arterial Streets and Highways								2
Freeway	11,597 21,475	35.1 64.9	17,294 27,228	38.8 61.2	14,185 24,196	37.0 63.0	15,979 26,873	37.3 62.7
Total	33,072	100.0	44,522	100.0	38,381	100.0	42,852	100.0
Transit Rapid	3 3 57	5.4 5.2 89.4	3 6 57	4.7 9.1 86.2	26 30 85	18.0 21.0 61.0	16 20 61	16.4 21.1 62.5
Total	63	100.0	66	100.0	142	100.0	97	100.0
Vehicle-Hours of Travel (thousands) Arterial Streets and Highways Freeway	236.0	26.2	329.0	27.4	269.3	26.5	304.6	26.8
Standard Arterial	665.2	73.8	873.5	72.6	746.0	73.5	833.5	73.2
Total	901.2	100.0	1,202.5	100.0	1,015.3	100.0	1,138.1	100.0
Transit Rapid Express Local	0.2 0.2 4.8	3.3 3.3 93.4	0.2 0.4 5.1	2.8 6.4 90.8	1.1 1.5 6.6	12.0 16.3 71.7	0.6 1.0 5.1	9.0 14.9 76.1
Total	5.2	100.0	5.7	100.0	9.2	100.0	6.7	100.0
Passenger-Hours of Travel (thousands) Arterial Streets and Highways Transit	1,120 49	95.8 4.2	1,431	96.6 3.4	1,208	94.3 5.7	1,354 51	96.4 3.6
Total	1,169	100.0	1,481	100.0	1,280	100.0	1,405	100.0

Source: SEWRPC.

generate 42.85 million highway vehicle-miles of travel and 97,000 transit vehicle-miles of travel in the year 2010, and 1.14 million highway vehicle-hours of travel and 6,700 transit vehiclehours of travel. The no-build plan may be expected to generate 44.52 million highway vehicle-miles and 66,000 transit vehicle-miles of travel and 1.2 million highway vehicle hours and 5,700 transit vehicle-hours of travel within the Region in the year 2010.

Standard No. 4 states that highway transportation facilities should be located and designed so as to provide adequate capacity; that is, a volume-to-design-capacity-ratio equal to or less than 1.0 based on 24-hour average weekday traffic volumes to meet the existing and potential travel demand. To determine the degree to which each alternative transportation plan 476 meets this standard, the anticipated travel demand expected to be generated under each plan was assigned to the alternative transportation networks to evaluate the adequacy of the proposed improvements. The levels of service of each of the alternative plans, as measured by arterial street and highway volume-to-designcapacity ratios, are summarized in Table 243. Under Alternative Plan 1, about 131 arterial miles would have volume-to-design-capacity ratios exceeding 1.0 and would thus experience traffic congestion. Under Alternative Plan 3, about 168 arterial miles would have volume-todesign-capacity ratios which exceed 1.0.

Standard No. 5 indicates that public transit facilities should be located and designed so as to provide adequate transit vehicle capacity to meet existing and potential travel demand. Average

ARTERIAL STREET AND HIGHWAY FACILITIES OPERATING UNDER, AT, AND OVER DESIGN CAPACITY IN THE REGION: 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

			,		No-E	uild Altern	ative				
	Under Design		At [Design		Over Design Capacity					
	Capa	city ^a	Сар	Capacity ^b		Moderate ^C		Severed		Extreme ^e	
County	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Miles	Percent of Total	Total Mileage
Kenosha	230.1	66.7	70.9	20.5	3.3	1.0	33.4	9.7	7.5	2.1	345.2
Milwaukee	490.0	62.1	120.6	15.3	2.0	0.2	132.4	16.8	44.3	5.6	789.3
Ozaukee	233.9	81.1	19.8	6.9	6.5	2.2	28.3	9.8			288.5
Racine	241.7	59.3	77.0	18.9	3.5	0.8	62.6	15.4	22.7	5.6	407.5
Walworth	387.2	87.0	31.7	7.1	8.5	1.9	9.2	2.1	8.3	1.9	444.9
Washington	337.7	75.3	54.1	12.1			49.8	11.1	7.1	1.5	448.7
Waukesha	369.1	48.8	103.8	13.7	7.8	1.1	222.0	29.4	52.9	7.0	755.6
Total	2,289.7	65.8	477.9	13.7	31.6	0.9	537.7	15.5	142.8	4.1	3,479.7

	Alternative Plan 1												
	Under I	Design	At	At Design		Over Design Capacity							
	Сара	city ^a	Capacity ^b		Mod	lerate ^C	Severe ^d		Extreme ^e				
		Percent		Percent		Percent		Percent		Percent	Total		
County	Miles	of Total	Miles	of Total	Miles	of Total	Miles	of Total	Miles	of Total	Mileage		
Kenosha	327.1	91.8	25.8	7.2			3.3	0.9			356.2		
Milwaukee	673.2	84.7	57.0	7.2	37.9	4.8	21.7	2.7	5.1	0.6	794.9		
Ozaukee	308.3	99.3	0.8	0.3	1.4	0.5			·		310.5		
Racine	405.0	96.7	12.7	3.0	0.6	0.1	0.7	0.1	·		419.0		
Walworth	449.8	95.0	11.9	2.5	11.7	2.5					473.4		
Washington	469.1	99.9	0.7	0.1							469.8		
Waukesha	681.5	87.9	45.3	5.8	35.3	4.6	13.0	1.7			775.1		
Total	3,314.0	92.1	154.2	4.3	86.9	2.4	38.7	1.1	5.1	0.1	3,598.9		

		Alternative Plan 3												
	Under I	Design	At [)esian		. (ver Desi	gn Capacit	y					
	Capa	city ^a	Cap	Capacity ^b		Moderate ^C		vere ^d	Extreme ^e		1			
		Percent		Percent		Percent		Percent		Percent	Total			
County	Miles	of Total	Miles	of Total	Miles	of Total	Miles	of Total	Miles	of Total	Mileage			
Kenosha	310.3	87.1	42.6	12.0			3.3	0.9			356.2			
Milwaukee	654.5	82.3	56.1	7.1	53.3	6.7	24.2	3.1	6.8	0.8	794.9			
Ozaukee	306.2	98.6	2.9	0.9		·	- <u>-</u>		·		310.5			
Racine	387.8	92.6	25.0	6.0	1.4	0.3	0.7	0.1			419.0			
Walworth	478.0	98.7	6.1	1.3	5.5					[]	484.1			
Washington	469.1	99.9	0.7	0.1							469.8			
Waukesha	636.9	82.2	65.3	8.4	47.3	6.1	25.6	3.3			775.1			
Total	3,242.8	89.8	198.7	5.5	107.5	3.0	53.8	1.5	6.8	0.2	3,609.6			

^aVolume-to-design-capacity ratio: 0.00-0.90

^bVolume-to-design-capacity ratio: 0.91-1.00

^CVolume-to-design-capacity ratio: 1.01-1.10

maximum load factors are specified in the standard for rapid, express, and local levels of service in both peak and off-peak travel periods. This standard has been met under each of the alternative plans considered because it served as an input to the plan design process. ^dVolume-to-design-capacity ratio: 1.11-1.30

eVolume-to-design-capacity ratio: over 1.30

Source: SEWRPC.

Standard No. 6 states that bicycle and pedestrian facilities should be located and designed to attract the maximum number of travelers currently operating single-occupancy vehicles. This standard can be met through proper facility design during the implementation of the 477 regional bicycle and pedestrian facilities plan and through the proper consideration of bicycle travel in traffic engineering and management activities. Accordingly, this standard is assumed to be met equally well by each alternative plan considered.

Standard No. 7 indicates that adequate capacity and a sufficiently highly level of geometric design should be provided to achieve specific overall travel speeds based on average weekday conditions for the highway and public transit components of the transportation system. The overall travel speeds specified in the standard were utilized in the design of the alternative transportation plans. Accordingly, this standard was assumed to be met equally well by each alternative plan considered.

Standard No. 8 provides that the use of transportation system management be maximized in travel corridors to achieve the desired level of service for both arterial street and highway and public transit facilities and services. Since the plan design process emphasized the use of transportation system management to resolve all practicable transportation deficiencies prior to the consideration of arterial street and highway system improvement or expansion, each alternative plan essentially meets this standard.

Objective No. 7: Reduce Accident Exposure

The seventh transportation system development objective relates to the reduction of accident exposure and the provision of increased travel safety and personal security on the regional transportation system. This objective is supported by three specific standards.

Standard No. 1 indicates that travel on transportation facilities which exhibit the lowest accident rates should be maximized. Freeways have been found to experience significantly lower accident rates than standard surface arterial facilities. Further, travel on transit is generally safer for passengers than travel in other vehicles. From the foregoing, it can be concluded that an alternative plan which provides for the greatest amount of travel on freeways and transit would best meet this standard. Both Alternative Plans 1 and 3 would accommodate about 37 percent of average weekday arterial vehicle miles of travel on the freeway system. Under the no-build plan, about 36 percent of vehicle miles of travel would occur on the freeway system. Additionally, under Alternative Plan 1, 4.5 percent of total travel would occur on the transit system, while under Alternative Plan 3, 3.1 percent regional travel would occur on the transit system. Under the nobuild plan the public transit system would serve 2.7 percent of the trips made on an average weekday within the Region. Thus, Alternative Plan 1 would maximize the total amount of travel within the Region made by transit and both Alternative Plans 1 and 3 would serve about the same amount of travel in the Region on the freeway system.

In an effort to identify the safest alternative transportation system plan, an analysis was made of the distribution of passenger-miles of travel by load and facility type, assuming that the alternative plan which maximizes passenger travel on a combination of freeways and mass transit would best meet the standard. The results of this analysis are identified in Table 244. The no-build plan represents the safest plan, since it would provide that 40 percent of total passenger miles of travel occur on a combination of freeways and mass transit. Alternative Plan 1 and Alternative Plan 3 would provide for about 38 percent.

A comparison of anticipated traffic accident experience and cost on the transportation system in the Region under the alternative plans is set forth in Table 245. The total cost of accidents in the Region in the plan design year 2010 would approximate \$1.39 billion for the no-build transportation plan, \$1.23 billion for Alternative Plan 1, and \$1.37 billion for Alternative Plan 3.

Standard No. 2 states that traffic congestion and vehicle conflicts should be reduced by maintaining a volume-to-design-capacity ratio equal to or less than 1.0, based on a 24-hour average weekday traffic volume. As shown in the previously cited Table 243, Alternative Plan 1 best meets this standard, in that 92 percent of the total arterial street and highway mileage in the Region would be expected to operate under design capacity. Under Alternative Plan 3, about 90 percent of arterial street and highway mileage in the Region would be expected to operate under design capacity. Under the no-build plan, about 66 percent of arterial street and highway mileage would be expected to operate under design capacity.

Standard No. 3 provides that railway grade separations be provided at all crossings involving the provision of intercity passenger and commuter train service. This standard also provides that, for all other crossings, the deci-

	Passenger-Miles of Travel on an Average Weekday (thousands)											
	2010											
	Base Year 1991		No-Build Transportation System Plan		Alternative Plan 1		Alternative Plan 3					
Mode and Facility Type	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total				
Freeways	18,555.2 609.1	34.7 1.1	25,941.0 728.8	38.4 1.1	20,993.8 1,375.9	36.1 2.3	23,489.1 903.4	36.8 1.4				
Subtotal	19,164.3	35.8	26,669.8	39.5	22,369.7	38.4	24,392.5	38.2				
Standard Arterials	34,360.0	64.2	40,842.0	60.5	35,810.1	61.6	39,503.3	61.8				
Total	53,524.3	100.0	67,511.8	100.0	58,179.8	100.0	63,895.8	100.0				

DISTRIBUTION OF PASSENGER-MILES OF TRAVEL IN THE REGION BY MODE AND FACILITY TYPE: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

Source: SEWRPC.

sion whether or not to provide a grade separation should be made at the project planning stage. This particular standard could be met through local planning and plan implementation activities under each of the alternative transportation system plans considered.

Objective No. 8: Minimize Energy Consumption

The eighth transportation system development objective relates to achieving a transportation system which minimizes the amount of energy consumed, especially nonrenewable energy. One specific standard supports this objective. Standard No. 1 states that the total amount of nonrenewable energy consumed in the operation of the transportation system, particularly petroleumbased fuels, should be minimized. To determine the relative degree to which each of the alternative plans meet this standard, an analysis was made of the estimated total annual consumption of motor vehicle fuel. The results of this analysis are summarized in Table 246.

For automobiles and trucks, the indicated motor fuel consumption data were determined as a function of the vehicle type, operating speed, the amount of travel on each link in the transportation network, and the type of highway facility traveled. Rates of fuel consumption for gasolineand diesel engine-powered vehicles at different speeds, representative of the efficiency of these vehicles in 1991, were used in the analysis. For gasoline-powered vehicles, different rates of fuel consumption were used for freeway and arterial facilities; travel on freeway facilities being somewhat more efficient than travel on standard arterials. For diesel-powered trucks, fuel usage was determined by applying a variable rate of fuel consumption depending upon vehicle speed to the total amount of vehicle miles of travel by diesel-powered vehicles. Transit diesel fuel usage was determined by applying an average rate of fuel consumption for all vehicle speeds to the total amount of vehicle-miles of travel by transit vehicles.

As indicated in Table 246, the total annual motor fuel consumption, including gasoline and diesel fuel, within the Region approximated 359 million gallons in 1991. Annual motor fuel consumption, under Alternative Plan 1 could be expected to increase by 53 million gallons to about 412 million gallons by 2010, an increase of about 15 percent. Annual motor fuel consumption under Alternative Plan 3 could be expected to increase by 96 million gallons to about 455 million gallons by 2010, an increase of about 27 percent. Annual motor fuel consumption under the no-build plan could be expected to increase by about 109 million gallons to about 468 million gallons by 2010, an increase of about 30 percent.

Objective No. 9: Facilitate Linked Tripmaking

The ninth transportation system development objective calls for the achievement of a transportation system which facilitates linked tripmaking, providing facilities and services necessary for efficient, fast, safe, and convenient intermodal connections. This objective recognizes that

TRAFFIC ACCIDENT EXPERIENCE AND COST ON THE TRANSPORTATION SYSTEM IN THE REGION: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

	Arterial Street and History Sustem Cumulating Assident Sussiance and Ocet									
		Arterial Stre	et and Highw	ay System Cu	mulative Acc	ident Experie	nce and Cost			
		Base Y	'ear 1991		No-Build Alternative					
County	Property Damage Accidents	Number of Injuries	Number of Fatalities	Cost of Accidents (million dollars)	Property Damage Accidents	Number of Injuries	Number of Fatalities	Cost of Accidents (million dollars)		
Kenosha County										
Freeway	303 3,677	220 1,556	1 8	9 77	488 6,035	353 2,553	1 14	14 127		
Subtotal	3,980	1,776	9	86	6,523	2,906	15	141		
Milwaukee County					1					
Freeway	2,672 17,019	1,935 7,200	6 38	79 357	3,503 18,750	2,537 7,933	7 42	104 393		
Subtotal	19,691	9,135	44	436	22,253	10,470	49	497		
Ozaukee County Freeway	343 2,378	248 1,006	1 5	10 50	468 2,615	339 1,107	1 6	14 55		
Subtotal	2,721	1,254	6	60	3,083	1,446	7	69		
Racine County Freeway	318 4,550	230 1,925 2 155	1	9 95	561 5,934	407 2,511 2,918	1 13	17 124		
	4,000	2,100		104	0,435	2,510	14			
Freeway	243 2,767	176 1,170	1 6	7 58	447 3,290	324 1,392	1 7	13 69		
Subtotal	3,010	1,346	· 7	65	3,737	1,716	· · · 8 …	82		
Washington County Freeway	245 3,693	178 1,563	1 8	7	365 4,864	265 2,058	1× 11	11 102		
Subtotal	3,938	1,741	9	84	5,229	2,323	12	113		
Waukesha County Freeway Standard Arterial	1,088 9,188	788 3,887	2 21	32 193	1,940 13,376	1,405 5,659	4 30	58 280		
Subtotal	10,276	4,075	23	225	15,310	7,064	- 34	338		
Southeastern Wisconsin Region Freeway Standard Arterial	5,212 43,272	3,775 18,307	11 98	153 907	7,772 54,864	5,630 23,213	16 123	231 1,150		
Subtotal	48,484	22,082	109	1,060	62,636	28,843	139	1,381		
Transit Systems within South- eastern Wisconsin Region	1,015	302	1	9	1,047	362	1	10		
Total	49,499	22,384	110	1,069	63,683	29,205	140	1,391		

where the use of more than one transportation mode is essential for travel between two points, proper modal access, terminal capacity, coordination among transportation providers concerning, among other things, route and schedule information, are necessary to prevent travel delays or unwanted transportation movements. Two specific standards support Objective No. 9.

Standard No. 1 indicates that the time passengers spend waiting at public transit stops for connecting modes of transportation should be minimized. This standard could be expected to be met better under Alternative Plan 1 than under Alternative Plan 3 because Alternative Plan 1 generally proposes lower headways, which translate into lower waiting times at public transit stops. In this regard, the transit element of Alternative Plan 1 proposed the development of transit centers in the western portions of the Kenosha and Racine urbanized areas, and in certain portions of Waukesha

Table 245 (continued)

		Arterial Street and Highway System Cumulative Accident Experience and Cost									
		Alternat	tive Plan 1			Alternat	ive Plan 3				
County	Property Damage Accidents	Number of Injuries	Number of Fatalities	Cost of Accidents (million dollars)	Property Damage Accidents	Number of Injuries	Number of Fatalities	Cost of Accidents (million dollars)			
Kenosha County Freeway	432 5,295	313 2,240	1 12	13 111	454 5,719	328 2,419	1 13	13 120			
Subtotal	5,728	2,553	13	124	6,173	2,747	14	133			
Milwaukee County Freeway	2,643 16,900 19,543	1,914 7,150 9,064	5 38 43	78 354 432	2,859 19,455 22,314	2,070 8,231 10.301	6 44 50	85 408 493			
Freeway	363 2,448	263 1,036	1 6	11 51	422 2,809	306 1,188	1 6	13 59			
Subtotal	2,811	1,299	7	62	3,231	1,494	7	72			
Racine County Freeway	409 5,168 5.577	296 2,187 2,483	1 12 13	12 108 120	471 5,868 6.339	341 2,482 2.823	1 13 14	14 123 137			
Walworth County						-,					
Freeway	378 3,129	274 1,324	1 7	11 66	617 3,133	447 1,326	1 7	18 66			
Subtotal	3,507	1,598	8	77	3,750	1,773	8	84			
Washington County Freeway Standard Arterial	762 3,188	552 1,349	2 7	23 67	563 3,718	408 1,573	1 8	17 78			
Subtotal	3,950	1,901	9 .	90	4,281	1,981	9	95			
Waukesha County Freeway Standard Arterial	1,389 12,624	1,006 5,341	3 29	41 265	1,638 13,865	1,186 5,866	3 31	49 291 340			
Subjotal	14,013	0,347	32	300	13,503	7,052		040			
Southeastern Wisconsin Region Freeway Standard Arterial	6,376 48,752	4,618 20,627	14 111	189 1,022	7,022 54,566	5,086 23,085	14 122	209 1,145			
Subtotal	55,128	25,245	125	1,211	61,588	28,171	136	1,354			
Transit Systems within South- eastern Wisconsin Region	2,174	640	2	19	1,682	473	1	15			
Total	57,302	25,885	127	1,230	63,270	28,644	137	1,369			

Source: SEWRPC.

County. The transit-center approach to the provision of local transit would permit timed transfers, that is, the simultaneous arrival of all buses at a center and transfer of all passengers.

Standard No. 2 indicates that parking should be provided at park-and-ride transit stations to accommodate the total parking demand generated by trips which change from auto and bicycle to public transit at each station and at carpool lots to accommodate the total parking demand generated by carpool and other ridesharing participants. This standard was assumed to be equally met by each alternative plan as it served as an input to the design of the plans.

Satisfaction of Objectives and Standards

The foregoing discussion concerning the scaling of the alternative regional transportation system plans against the transportation system development objectives and standards is summarized in tabular form in Table 247. This table indicates,

	Annu	al Motor Fuel Consum	ption (millions of g	allons)				
		Proposed 2010						
Vehicle Type	Base Year 1991	No-Build Transportation System Plan	Alternative Plan 1	Alternative Plan 3				
Street and Highway	353.2 5.7	462.3 5.8	400.0	446.5 8.2				
Total	358.9	468.1	412.1	454.7				

COMPARISON OF MOTOR FUEL CONSUMPTION BY VEHICLES TRAVELING IN THE REGION: 1991 AND 2010 NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS

NOTE: A 50 percent improvement over the existing average motor fuel efficiency would result in an annual motor fuel consumption of 314.2 million gallons under the no-build plan, 278.9 million gallons under Alternative Plan 1, and 306.0 million gallons under Alternative Plan 3.

The Clean Air Act Amendments of 1990 require that certain centrally fueled fleets in severe ozone nonattainment areas use alternatively fueled vehicles. Therefore, some reductions in the annual amount of motor fuel consumed set forth in the table under each of the alternative plans may be anticipated through the plan design period.

^aAn average fuel efficiency for transit vehicles of four miles per gallon was assumed in the forecast of annual motor fuel consumption for transit vehicles.

Source: SEWRPC.

based on the foregoing discussion, the relative ability of each of the alternative plans, and the no-build alternative to meet the system development objectives. For the purpose of plan comparison, only the quantitative standards are included in Table 247.

Comparison of Alternative Plan Costs to Estimated Available Revenues

Estimates of the average annual costs and revenues associated with the no-build alternative and Alternative Plans 1 and 3 are set forth in Table 248. The estimates include both the capital and operating costs associated with the arterial street and highway system and with the transit system. The costs and revenues are expressed on an average annual basis for the 16-year period from 1995 through 2010. The estimated average annual revenues set forth in Table 248 reflect assumptions attendant to the continuation of Federal, State, and local revenue streams for highway capital and operating and transit operating and capital purposes (see footnotes of Table 248). All figures are set forth in constant 1994 dollars.

The estimated average annual cost of the nobuild alternative is \$360 million. The equivalent costs of Alternative Plans 1 and 3 are, respectively, \$546 million and \$520 million. The principal reason for the difference between the two alternative plans is the greater annual operating cost associated with the more extensive transit system proposed under Alternative Plan 1.

Under the no-build alternative, average annual revenues could be expected to total about \$352 million. This results in an estimated average annual shortfall over the plan design period of \$8 million. That shortfall occurs in the area of transit capital and operating costs. For perspective, the Advisory Committee directed that an estimate be made of the additional motor fuel tax that would be required to fund any such shortfall. On that basis, an additional one-centper-gallon motor fuel tax levied on gasoline and diesel fuel in the Region was found to be sufficient to cover the anticipated shortfall.

Under Alternative Plan 1, average annual revenues could be expected to total about \$389 million. This results in an estimated average annual shortfall over the plan design period of \$157 million. About 60 percent of the shortfall occurs in the highway area, with the remaining 40 percent occurring in the transit area. An additional 20-cents-per-gallon motor fuel tax would be required to cover the anticipated shortfall. This amount could be raised as part of the assumed doubling of the perceived cost of automobile operation that is an important

SUMMARY OF RELATIVE ABILITY OF THE NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS TO MEET TRANSPORTATION DEVELOPMENT OBJECTIVES

	Year 2010 Alternative Transportation System Plans								
and Supporting Standards ^a	No-I	Build	Pia	n 1	Pla	in 3			
Objective No. 1: Effectively Serve Regional Land Pattern Percent of Urbanized Area Population within	Highway	Transit	Highway	Transit	Highway	Transit			
30 Minutes of 40 Percent of Jobs	93.6	71	95.2	54.4	95.2	25.8			
35 Minutes of Three Major Retail and Service Centers ^b	92.9	3.6	88.1	22.7	88.1	6.6			
or 30 Minutes of a Hospital or Clinic	100.0	49.3	100.0	65.6	100.0	54.8			
40 Minutes of a Major Outdoor Recreation Center	100.0	44.5 58.1	100.0	74.6 89.6	100.0	50.4			
60 Minutes of General Mitchell International Airport	100.0	22.1	100.0	58.9	100.0	44.9			
Objective No. 2: Minimize Costs Sum of Transportation System Capital and Operating Costs Benefit-Cost Ratio	\$360 millio	\$360 million annually		on annually 68	\$520 million annually 1.18				
Objective No. 3: Provide Flexible, Balanced Transportation System									
Residents Served by Transit	1,430	3,500 700	1,493	3,900 900	1,46	9,700			
Local Transit Headways-Peak (Milwaukee County)	10 to 45	minutes	10 to 30	minutes	10 to 4	5 minutes			
Rapid Transit Headways-Peak	15 to 60	minutes	Five to 3	0 minutes	Five to 30 minutes				
District Made by Transit ^C	1	0	1	5	11				
Objective No. 4: Minimize Disruption	Desider		Pasidan	tial CE1	Residential: 720				
Residential and Nonresidential Structures to be Dislocated	Nonresid	ential: 39	Nonreside	ntial: 051	Nonreside	ential: 720			
Value of Land Used for Transportation and Terminal Facilities Amount of Property Tax Base Reduction	\$4.7 million \$3.7 million		\$145.4 \$129.1	million million	\$170. \$152.	7 million 7 million			
Objective No. 5: Protect Natural Environment		_							
Acres of Primary Environmental Corridor Land Lost Tons of Air Pollution Emitted per Day	1	5	4	32	4	/5			
Volatile Organic Compounds	2	9.8 1 4	2	5.6 1 3	28.7				
	24	B.4	21	1.5	238.2				
Acres of Prime Agricultural Land Lost		3	1,0)39	1,	644			
Objective No. 6: Facilities Traffic Flow			1						
Arterial Street and Highway	1.8 r	nillion	1.5 r	nillion	1.7	million			
Transit	50,	000	72,	000	51,	,000			
Average Weekday Vehicle-Hours of Travel	1.2 r	nillion	1.0 r	nillion	1.1	million			
Transit	5,7	00	9,:	200	6,	700			
Average Weekday Vehicle-Miles of Travel	44.5	million	38.4	million	42.9	million			
	66,	000	142	,000	97	,000			
Percent of Arterial Mileage Operating at	7	5	96	3.4	9	5.3			
Objective No. 7: Reduce Accident Exposure	,,		+						
Percent of Passenger Miles of Travel on Facilities with Lowest Accident Exposure, Transit and Freeways	39	9.5	38	3.4	3	8.2			
Objective No. 8: Minimize Energy Consumption									
Gallons of Motor Fuel Consumed Annually	468.1	million	412.0	million	454.7 million				
Objective No. 9: Facilitate Linked Tripmaking Average Wait Time at Transit Stops	11.6	minutes	9.1 m	ninutes	10.4	minutes			

^aThe standards which are listed in this table are all comparative in nature. Standards not listed include those that have been completely met because they served as input to the design of the plans, or those that could be met through proper local or project level planning and plan implementation activities. The standards listed in this table are easily quantifiable and provide a sound basis for determining the relative ability of the alternative plans to meet the transportation development objectives.

^bOr one major retail or service center by transit within the Kenosha and Racine urbanized areas.

^CDoes not include those trips made by persons residing in group quarters, trips made by nonresidents of the Region, or trips made by persons in school buses.

^dVolume-to-design-capacity ratio equal to or less than 1.0.

Source: SEWRPC.

AVERAGE ANNUAL COSTS AND REVENUES ASSOCIATED WITH NO-BUILD AND ALTERNATIVE TRANSPORTATION SYSTEM PLANS IN THE REGION: 1995 THROUGH 2010

Cost or Revenue Item	No-Build Alternative	Alternative Plan 1	Alternative Plan 3
Transportation System Cost ^a (average annual 1995-2010 expressed as millions of dollars)	÷		
Capital	190 67	298 68	302 68
Subtotal	257	366	370
Transit System Capital Operating	19 84	65 115	57 93
Subtotal	103	180	150
Total	360	546	520
Transportation System Revenue (average annual 1995-2010 expressed as millions of dollars) Highway Capital			
Federal	126 66 11	132 66 11	132 66 11
Subtotal	203	209	209
Highway Operating State Local	36 28	36 28	36 28
Subtotal	64	64	64
Transit Capital Federal	10 3 16	23 3 26	23 3 26
Transit Operating Federal State Local	6 54 12	6 72 12	6 61 12
Subtotal	72	90	79
Total	352	389	378
Cost-Revenue Comparison Average Annual Difference between Cost and Revenue (millions of dollars) Motor Fuel Tax Required to Fund Shortfall (cents per gallon)	-8	-157 20	-142

^aThe costs for Alternative Plans 1 and 3 include a busway/high-occupancy vehicle (HOV) facility and a light-rail line in the East-West Corridor. The busway/HOV cost is part of the highway operating cost (\$18 million annually) and the light-rail cost is part of the transit capital cost (\$32 million annually). The costs for the system of busway/HOV facilities have also been included in the highway capital costs of these two plans (\$58 million annually).

The transit operating costs presented in this table are net operating costs, that is, total operating costs less farebox revenues. Estimated current 1993 annual transit farebox revenues in Southeastern Wisconsin are \$34 million, total annual operating costs are \$97 million, and net annual operating costs are \$63 million. Upon full implementation of the paratransit service needed to meet the requirements of the Federal Americans with Disabilities Act of 1990, it is estimated that total and net annual operating costs would substantially increase. This is reflected in the No-Build alternative plan year 2010 total annual operating cost of \$119 million in 1994 constant dollars, compared to the current 1993 total operating cost of \$97 million.

The estimated year 2010 annual total operating costs in constant 1994 dollars are \$197 million for Alternative Plan 1 and \$143 million for Alternative Plan 3.

The estimated year 2010 annual farebox revenues in 1994 constant dollars are \$38 million for the No-Build plan, \$51 million for Alternative Plan 1, and \$42 million for Alternative Plan 3. These farebox revenues were adjusted for this plan cost and revenue comparison to reflect the historic differential impact of general price ridership of increasing fares. The adjusted year 2010 farebox revenues are \$32 million for the No-Build plan, \$43 million for alternative Plan 1, and \$36 million for Alternative Plan 3. The year 2010 annual net operating costs in 1994 constant dollars are \$87 million for the No-Build plan, \$154 million for Alternative Plan 1, and \$107 million for Alternative Plan 3.

^bHighway operating costs include those annual costs associated with snow and ice control, cleaning of arterial street and highway facilities, lighting and signalization, and routine maintenance such as crack sealing and pavement patching.

Source: SEWRPC. 484 element of Alternative Plan 1. A 20-cents-pergallon tax increase would represent about onefifth of the cost increase assumed under Alternative Plan 1.

Under Alternative Plan 3, average annual revenues could be expected to total about \$378 million. This results in an estimated average annual shortfall over the plan design period of \$142 million. About two-thirds of the shortfall occurs in the highway area with the remaining one-third occurring in the transit area. An additional 18-cents-per-gallon motor fuel tax would be required to cover the anticipated shortfall.

It is important to note that the funding shortfalls attendant to the alternative plans are more substantial than the shortfalls estimated by the Commission in its recent regional transportation authority feasibility study.¹ This is primarily because, unlike the analysis conducted previously, this analysis includes consideration of State trunk highway costs, including reconstruction of the Milwaukee area freeway system busway/HOV facility costs. It also considers a plan implementation period of 16 years, rather than 20 years.

ADVISORY COMMITTEE RECOMMENDATION FOR PRELIMINARY RECOMMENDED PLAN

After significant deliberation, the Advisory Committee determined that Alternative Plan 3 should be further refined and detailed and advanced as the preliminary recommended plan to be taken to public hearing. The Advisory Committee made that decision based on consideration of the relatively low probability of achieving the significant increases in automobile pricing assumed under Alternative Plan 1 within the plan design period; the additional cost of Alternative Plan 1 and attendant shortfall of available revenues for plan implementation in the absence of significant increases in nonproperty-tax-based revenues for transportation purposes; and the marginal differences between Alternative Plan 1 and Alternative Plan 3 with respect to required arterial highway improvement and expansion.

Compared to the no-build alternative. Alternative Plan 3 provides substantial improvements in transit service and accessibility in the Region and potential substantial reduction in traffic congestion. Alternative Plan 3, however, also has a substantial revenue shortfall with respect to plan implementation. In proposing that Alternative Plan 3 be refined and taken to public hearing as the preliminary recommended plan, the Advisory Committee also recommended that, subject to the exceptions noted below, the busway/HOV lanes in the plan, together with the express light-rail/bus guideway and commuter rail lines in the plan, be identified as potential future guideway facilities. These potential future facilities would be confirmed as integral parts of the plan, and subsequently implemented, only upon the completion of corridor major investment studies as required by Federal regulations. An important part of such corridor studies is the identification of funding sources for any additional guideway facilities recommended to be added to the plan and then implemented. The exceptions to the foregoing consist of the following guideway facilities:

- 1. A busway/HOV lane guideway in the eastwest freeway corridor, from about N. 4th Street in the City of Milwaukee west to the STH 164 interchange in the City of Waukesha and Town of Pewaukee.
- 2. An express light-rail guideway in the eastwest travel corridor, from the Milwaukee CBD to the Milwaukee Regional Medical Center.
- 3. An express light-rail guideway in the northeast travel corridor, from the Milwaukee CBD to the University of Wisconsin-Milwaukee campus and Village of Glendale area.

These guideway facilities were included as integral parts of the preliminary recommended plan based upon the preliminary results of a corridor major investment study being conducted by the Wisconsin Department of Transportation at the time of the preparation of the plan.²

¹See SEWRPC Memorandum Report No. 38, <u>A</u> <u>Regional Transportation Authority Feasibility</u> Study for Southeastern Wisconsin, 1990.

²The formal name of the referenced major investment study sponsored by the Wisconsin Department of Transportation is "Milwaukee East/West Corridor Transit Study." The study had proceeded to the point of a draft Environmental Impact Statement as of summer 1994.

SUMMARY AND CONCLUSIONS

This chapter has described two complete alternative regional transportation system plans considered in the preparation of a new regional transportation system plan for Southeastern Wisconsin. The chapter presents the results of an evaluation and a comparison of these plans against agreed-upon transportation system development objectives and supporting standards. Both alternatives contain three elements: transportation system management, public transit system maintenance and improvement, and arterial street and highway system maintenance and improvement. The arterial improvements were added to resolve remaining transportation deficiencies identified through the testing of the alternative transportation system management plans described in Chapter IX of this report.

In the category of transportation system management, each plan includes traffic management measures, such as peak-period curb-lane parking restrictions and freeway operational control, and areawide promotional measures to promote carpooling, vanpooling, and telecommuting. In addition, Alternative Plan 1 proposes to approximately double the perceived cost of driving to \$0.11 per mile, the equivalent of a \$1.10 increase in the motor fuel tax, and assumes the implementation by local units of government of land use and site design plans to moderate travel demand. Both alternative plans would improve and expand the public transit system, albeit Alternative Plan 1 envisions a much greater increase in transit service: a 124 percent increase as measured in vehicle-miles of service, compared to a 53 percent increase under Alternative Plan 3.

Despite the substantial increase in transit service and the emphasis on travel demand management, the bulk of travel under Alternative Plan 1, as under Alternative Plan 3, may be expected to occur on the arterial street and highway system. Consequently, the envisioned scale of arterial system improvement is similar under both plans. Under Alternative Plan 1, 112 new route-miles would be constructed, compared to 124 route-miles under Alternative Plan 3; 382 route-miles would be widened for additional capacity, compared to 410 route-miles under Alternative Plan 3. A summary of the elements comprising the alternative plans and the costs and revenues associated with these alternatives is set forth in Table 203 and 248, respectively.

The evaluation and comparison of the alternative plans indicated that the principal difference between the plans was the use of the highway system: Alternative Plan 1 would generate over 10 percent less vehicle-miles of travel than Alternative Plan 3. This difference is principally the result of the pricing measures assumed under Alternative Plan 1. Alternative Plan 1 would also result in the emission of 11 percent fewer tons of volatile organic compounds per day, 11 percent fewer tons of nitrogen oxides per day, and 11 percent fewer tons of carbon monoxide per day, and about 9 percent fewer gallons of motor fuel consumed annually. Alternative Plan 1 would also have a substantially higher benefit-cost ratio than Alternative Plan 3, providing \$1.68 in benefits for every dollar invested in implementing the plan. This compares to \$1.18 under Alternative Plan 3. Because Alternative Plan 1 proposes similar arterial street and highway improvements as Alternative Plan 3 and substantially greater transit system improvements, its total cost is about 5 percent higher than the cost of Alternative Plan 3.

Finding a low probability of achieving the automobile pricing increases of Alternative Plan 1 within the plan design period, finding Alternative Plan 1 to have higher costs and higher attendant shortfalls of available revenues, and finding marginal differences between Alternative Plans 1 and 3 with respect to needed arterial highway improvements, the Advisory Committee selected Alternative Plan 3 for refinement and advancement to public hearing as the preliminary recommended plan.

RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN

INTRODUCTION

As described in Chapter X of this report, the Commission's Technical Coordinating and Advisory Committee on Regional Transportation System Planning recommended that three alternative plans be fully evaluated and tested as a basis for the selection of a preliminary recommended plan to be taken to public hearing. Upon careful consideration of the three alternative plans, the Committee determined that the alternative known as Alternative Plan 3 should become the basis for the design of a recommended plan which would be submitted to public review. The public review process began at a Regional Planning Conference held on June 27, 1994, and extended through the summer of 1994. This chapter describes the preliminary recommended plan taken to public hearing, summarizes the public review comments received regarding the preliminary plan, describes the changes made to the preliminary plan in response to the comments received, and describes the final year 2010 regional transportation system plan recommended for adoption. In selecting Alternative Plan 3 as the basis for the preliminary recommended plan, the Advisory Committee did not select the no-build plan and the alternative known as Alternative Plan 1. The no-build plan proposed no significant improvements in transportation facilities and services over those improvements which were considered to be committed, that is, programmed for implementation in the annual element of the 1993 through 1998 Transportation Improvement Program for Southeastern Wisconsin. The Committee rejected the no-build plan, finding the limited scope of improvements proposed therein inadequate to provide the levels of highways and transit service deemed essential to the continuous sound social and economic development and redevelopment of the Region.

The Committee also chose not to select Alternative Plan 1. Alternative Plan 1 proposed substantial increases in automobile operating costs as a measure to moderate travel demand, particularly the use of the single-occupancy automobile, as well as substantial improvements to the public transit system in the Region. The Committee made the decision not to select Alternative Plan 1 because of the relatively low probability of achieving, within the plan design period, the significant increases in automobile pricing assumed under that plan; the additional costs of the plan and attendant shortfall with respect to available revenues for plan implementation in the absence of a significant increase in nonproperty tax-based revenues for transportation purposes; and the marginal differences between Alternative Plans 1 and 3 with respect to required arterial highway improvement and expansion.

Upon selecting Alternative Plan 3 as the basis for the preliminary recommended plan, the Advisory Committee recommended that certain changes be made to the level of transit improvement and expansion proposed in that alternative plan to improve the accessibility provided by local transit service. The following changes were made to the local public transit service initially envisioned under Alternative Plan 3.

- Peak-period headways on major Milwaukee County transit routes in the area south of Silver Spring Drive, east of 76th Street, and north of Layton Avenue were reduced from the 10 to 40 minutes assumed under Alternative Plan 3 to 10 minutes.
- Peak-period headways in the Kenosha and Racine urban areas were reduced from the 20 to 30 minutes assumed under the Alternative Plan 3 to 15 to 30 minutes.
- Peak-period headways in the Waukesha urban area were reduced from the 30 minutes on some routes and 60 minutes on other routes assumed under Alternative Plan 3 to 30 minutes on all routes.
- Local transit routes within all three urbanized areas of the Region were expanded. Within the Milwaukee urbanized area, about 160 route-miles would be added to the system initially proposed under Alternative Plan 3; within the Racine urbanized area, about 20 route-miles would be added; and within the Kenosha urbanized area, about 20 route-miles would be added.

The changes made in the frequency and extent of local transit service in the Milwaukee, Kenosha, Racine, and Waukesha areas have served to increase the level of vehicle-miles and vehiclehours of transit service provided on an average weekday. Under the preliminary recommended plan, therefore, 72,300 revenue vehicle-miles of local transit service would be provided within the Region, an increase of 11,700 vehicle-miles, or 19 percent, over that proposed under Alternative Plan 3. Similarly, 5,800 revenue vehiclehours of local transit service would be provided within the Region, an increase of 700 vehiclehours, or 14 percent, over that proposed under Alternative Plan 3.

As a result of these improvements in transit service levels, transit ridership may be expected to reach about 197,000 revenue passengers per average weekday, an increment of about 5 percent over the 187,000 revenue passengers per weekday forecast under Alternative Plan 3. The expansion and improvement of local transit service envisioned under the preliminary recommended plan would require that 760 vehicles operate during peak travel periods within the Region, or 93 more than would be required to operate under Alternative Plan 3.

THE PRELIMINARY RECOMMENDED YEAR 2010 REGIONAL TRANSPORTATION SYSTEM PLAN

The preliminary year 2010 regional transportation system plan consists of recommendations in three major component areas: transportation systems management, public transit, and arterial street and highways.

Transportation Systems Management

The preliminary plan proposes the use of transportation systems management measures to ensure more efficient use of existing transportation facilities before commitments are made to new capital investment; to encourage use of highoccupancy vehicles, including buses, vanpools, and carpools; to reduce vehicle use, primarily in congested areas; to effect motor fuel savings; and to modify travel demand through reductions in vehicular travel during peak periods, thereby better adjusting travel demand to the available transportation system capacity. The preliminary recommended plan also proposes transit improvements and areawide coordinated efforts to promote carpooling, vanpooling, telecommuting, and work schedule changes which would assist in implementation of the Employee Commute Options program.

<u>Comprehensive Freeway Traffic Management</u> <u>System</u>: The preliminary plan includes the implementation of the long-planned Milwaukee area freeway traffic management system. This system would provide the means to monitor traffic flows system-wide and to regulate access to the freeways on the basis of current "realtime" traffic conditions. The system would also provide preferential access to buses and other high-occupancy vehicles, a central traffic management center, incident management procedures, and a motorist advisory information and assistance service.

<u>Curb-Lane Parking Restrictions</u>: The preliminary plan recommends that, as necessary, curblane parking on arterial streets be prohibited during peak hours of travel in order to ensure that all available arterial capacity is effectively used. The plan identifies 464 miles of the 3,614mile arterial system where it is envisioned that curb-lane parking restrictions will be necessary. In some cases such restrictions have already been placed into effect. In other cases it will be necessary for municipalities to restrict on-street parking as traffic volumes increase over time.

<u>Traffic Management</u>: The preliminary plan recommends the use of traffic engineering practices, including synchronized traffic signalization, the provision of right- and left-turn lanes at major intersections, and proper spacing of driveway and local street intersections, to assist in achieving efficient traffic flow on arterial facilities. The plan also recognizes, in accordance with the regional bicycle and pedestrian facilities plan, that arrangements to facilitate pedestrian and bicycle movements must be considered in all traffic engineering activities and in the design of new or widened arterial streets and highways.

Intelligent Transportation Systems (ITS): While the aforereferenced traffic management measures are currently applicable within the Region, by the year 2010 the application of more advanced traffic management technology known as Intelligent Transportation Systems may become practical. Advanced traffic management systems (ATMS) technology is currently being applied by the Wisconsin Department of Transportation as part of the Department efforts to implement the freeway traffic management system recommended in the second-generation regional transportation system plan. Other elements of ITS that may have future application within the Region include advanced travel information systems (ATIS), which provide travelers with "real-time" information on routespecific traffic conditions; advanced public transportation systems (APTS), which facilitate transit use by providing "real-time" transit vehicle location and scheduling information to transit users: commercial vehicle operations (CVO) technology, which automates commercial vehicle licensing, registration, and toll collection; advanced rural transportation systems (ARTS), which provide up-to-date reports of weather and road condition to rural highway users; and advanced vehicle control systems (AVCS), the invehicle technology which may improve the safety and efficiency of automobile travel.

Areawide Carpooling Promotion Measures: The preliminary plan recommends a coordinated areawide effort to promote travel through ridesharing, transit use, and bicycle use. This effort would aggressively promote carpooling and vanpooling, as well as telecommuting and work time rescheduling programs, to reduce automobile travel, particularly during peak periods and by single-occupancy vehicles. In addition, the plan would include the promotion and support of transportation management associations throughout the Region to provide an institutional structure for helping to achieve the goal of reducing travel by single-occupancy vehicles during peak travel periods.

Land Use Measures: The preliminary plan recommends that local units of government in the Region give careful attention to detailed, sitespecific land use planning in an effort to reduce the number of automobile trips, to reduce the length of automobile trips, to promote travel by transit, and to facilitate bicycle and pedestrian travel. The plan, therefore, proposes the use of appropriate land use regulation to promote a coordinated mix of land use activities in newly developing areas, higher-density development near transit lines and stations, orientation of buildings on sites in a manner which facilitates transit use, and integrated bicycle and pedestrian circulation systems. The plan envisions that as new neighborhoods are developed and older neighborhoods are redeveloped, detailed land use and community facility plans would be prepared and implemented that would be in accord with the aforementioned recommendations.

In recognition of the relationship between regional patterns of land use development and travel demand, the plan further proposes that county and local units of government within the Region employ the regional land use plan as a guide in making decisions regarding urban development. In this regard, the plan envisions that county and local units of government would institute land use policies that would serve to implement the adopted regional land use plan, helping to direct the type, extent, location, timing, and rate of urban development in accordance with that plan. Such development policies would provide that urban development occur in planned neighborhood units or in planned major activity centers and only in areas covered by soils suitable for urban use, not subject to such special hazards as flooding, and readily provided with essential urban services including sanitary sewerage, public water supply, and public mass transit.

Public Transit

The preliminary recommended plan calls for major increases in the levels of rapid and express transit service, as well as increases in the level of local service provided (see Table 249). The plan proposes the development of a true system of rapid and express routes integrated with local transit service. Rapid transit routes would operate within all major travel corridors oriented to the Milwaukee central business district (CBD) and express transit would operate in a grid pattern of routes within Milwaukee County. In total, the plan proposes a 72 percent increase in transit service as measured by vehicle-miles of transit service provided, from 63,300 in 1991 to 108,600 in 2010. This increase embodies the combined effects of proposed improvements in the frequency of operation of rapid and express transit and the additions and extensions of rapid and express transit routes. The plan also proposes a 28 percent increase in local transit service as measured by vehicle miles of service provided. The transit recommendations shown in graphic summary form on Map 105 include the following.

<u>Rapid Transit</u>: The preliminary recommended plan proposes that the existing freeway flyer bus service within the Region continue to be operated from the Milwaukee CBD to the southwest to the Village of Mukwonago and to the west to

TRANSIT SYSTEM OPERATING CHARACTERISTICS IN THE REGION: 1991 AND 2010 PRELIMINARY RECOMMENDED PLAN

	Base Year Planned Increment		ncrement	
Transit Service	1991	Number	Percent	2010
Round-Trip Route Length (miles)				
Rapid Routes	449	761	169.5	1,210
Express Routes	393	-3	-0.8	390
Local Routes	. – .			
Kenosha Urbanized Area	171	39	22.8	210
Milwaukee Urbanized Area	1,112	338	30.4	1,450
Racine Urbanized Area	171	29	17.0	200
Subtotal	1,454	406	27.9	1,860
Total	2,296	1,164	50.7	3,460
Average Weekday Vehicle Requirements ^a				
Peak Period	530	230	43.4	760
Midday Off-Peak Period	285	62	21.8	347
Revenue Vehicle-Miles (average weekday)				
Rapid	3,400	12,500	367.6	15,900
Express	3,300	17,100	518.2	20,400
Local	56,600	15,700	27.7	72,300
Total	63,300	45,300	71.6	108,600
Revenue Vehicle-Hours (average weekday)				
Rapid	170	430	252.9	600
Express	170	830	488.2	1,000
Local	4,880	920	18.9	5,800
Total	5,220	2,180	41.8	7,400

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup.

Source: SEWRPC.

the Cities of Waukesha and Oconomowoc. The plan also calls for the expansion of such service in the south corridor to the Cities of Racine and Kenosha, in the northwest corridor from its current terminus at the Pilgrim Road transit station to the Village of Germantown, and in the IH 43 north corridor from its current terminus at the Brown Deer Road transit station in River Hills to the City of Cedarburg and Village of Grafton. The rapid transit system envisioned would provide relatively fast and convenient service in all major travel corridors connecting major activity centers in the Region with the Milwaukee CBD. The network of rapid transit routes is shown in red on Map 105.

In all, 28 rapid transit freeway flyer routes, 25 oriented to the Milwaukee CBD and three to the University of Wisconsin-Milwaukee campus, would be operated over 1,210 round-trip routemiles. The rapid transit system would be served by 65 transit stations, spaced about every three to five miles. In 1991, the base year for the new plan, 13 rapid transit freeway flyer routes were operated over 449 round-trip route-miles within the Region. The planned service also differs from existing service in that the planned rapid transit system would serve intermediate stations and would provide service in both directions during both peak periods.

Central to the expansion of rapid transit service proposed under the recommended plan is the planned increase in the number of miles and hours transit vehicles would operate on an average weekday. In this regard, the plan recommends that the number of rapid transit revenue vehicle-miles of service provided be increased by 12,500 vehicle-miles, from 3,400 in 1991 to 15,900 by 2010. Similarly, the plan



The preliminary recommended plan proposes to extend rapid transit service within all major Milwaukee-oriented travel corridors; to provide a comprehensive system of express transit routes, particularly within Milwaukee County, and to expand local transit service areas as urban development proceeds in conformance with the adopted regional land use plan. By the year 2010, the transit system envisioned under the preliminary recommended plan would consist of about 3,460 round-trip route-miles, about 51 percent greater than the level provided in 1991. The planned transit system would provide 108,600 revenue vehicle-miles of service per average weekday, or about 72 percent more than in 1991, and 7,400 revenue vehicle-hours of service per average weekday, or about 30 percent more than in 1991. 491

recommends that the number of rapid transit revenue hours be increased by 430 vehicle-hours, from 170 in 1991 to 600 by 2010.

The plan also identifies a potential system of almost 49 miles of exclusive busway and highoccupancy vehicle (HOV) facilities. These facilities would be located within, or parallel to, the most heavily congested freeway corridors. The ultimate decisions concerning the provision of exclusive busways or combined busway and HOV facilities would be made through detailed corridor studies. If corridor studies determine that the implementation of busway facilities should not be pursued, rapid transit service would continue to be provided in mixed traffic over existing freeway lanes. The corridor studies would also determine whether or not highoccupancy vehicles would operate on the busways that might be constructed. One such study for the east-west corridor between the Cities of Waukesha and Milwaukee is now underway under the sponsorship of the Wisconsin Department of Transportation. Only this busway and its attendant costs have been included in the preliminary plan.

The rapid transit service provided under the preliminary plan would operate primarily during peak periods, that is, between 6:00 a.m. and 8:30 a.m. and 3:30 p.m. and 6:00 p.m. on weekdays. Midday service over some routes would be provided. No weekend or evening service would be provided. Headways on the rapid transit system would range from five to 30 minutes during peak periods, and 30 to 60 minutes during off-peak periods over those routes provided with service during the midday.

The fares for rapid transit service would remain at 1994 levels. The freeway flyer rapid transit bus fare for a trip within Milwaukee County would be \$1.50. The fare charged for a trip between points within Milwaukee County and the limits of the Milwaukee urbanized area would be \$2.00. The fare charged for a trip between the Milwaukee CBD and the outer limits of the rapid transit system would be \$3.00. It is assumed that these fares would increase over the plan design period only with general price inflation.

The preliminary plan also recognizes the potential to establish commuter railway passenger train service as an alternative to freeway flyer or exclusive busway rapid transit service in four major travel corridors: from Milwaukee to Racine and Kenosha, from Milwaukee to Oconomowoc, from Milwaukee to West Bend, and from Milwaukee to Cedarburg and Grafton (see Map 105). Detailed corridor studies would be required to determine whether or not to add such commuter rail service to the regional plan for implementation, as well as to identify funding sources.

Express Transit

The express level of transit service envisioned in the plan would operate over arterial streets with stops generally located at intersecting arterial streets and at intersecting rapid and express transit routes. Express service is distinguished from rapid service in that it provides a greater degree of accessibility at somewhat slower overall operating speeds. The preliminary plan proposes that 12 regular express transit bus routes be provided in a grid pattern, largely within Milwaukee County. In 1991, seven regular express transit routes were in place. Within the Milwaukee urbanized area, the express transit would be provided in major travel corridors to connect major activity centers to the Milwaukee CBD. One express transit route would also connect the Cities of Racine and Kenosha, although this route could be replaced by commuter rail service. The planned express routes are shown in blue on Map 105.

The preliminary plan also proposes light-rail express transit in both the east-west and northeast travel corridors in Milwaukee County, as shown on Map 105. The necessary corridor studies are currently being conducted by the Wisconsin Department of Transportation. Four other travel corridors have been identified with potential for light-rail express transit service. These include the Milwaukee CBD to the South Milwaukee area, the Milwaukee CBD to the General Mitchell International Airport, the 27th Street crosstown corridor to the Southridge Shopping Center, and the Milwaukee CBD to the Northridge Shopping Center. The feasibility of light rail in these corridors would be determined through Federally required major investment studies.

The frequency of operation of transit vehicles over express routes would be significantly increased. As shown in Table 249, the plan envisions that the number of vehicle-miles provided on an average weekday would increase by 17,100 vehicle-miles, from about 3,300 in 1991 to about 20,400. Similarly, vehicle-hours of express service provided on an average weekday would increase by 830 vehicle-hours, from 170 in 1991 to 1000 in 2010. Express transit service would be provided on weekdays from 6:00 a.m. to 6:00 p.m. on all routes and during weekday evenings and weekends on some routes. Peak-period headways would range from five to 15 minutes in the Milwaukee urbanized area to 30 minutes on the route connecting Racine and Kenosha. Off-peak headways would range from 10 to 30 minutes within the Milwaukee urbanized area to 60 minutes on the Racine-Kenosha route. Express transit fares would remain at 1994 levels, that is, \$1.25 in Milwaukee County and \$0.75 on the Racine-Kenosha route. It is assumed that these fares would increase over the plan design period only with general price inflation.

Local Transit: The local level of service envisioned in the plan consists of buses operating over arterial and collector streets with frequent stops for passenger boarding and alighting. Local fixed-route service would continue to be provided within Milwaukee County and the Cities of Waukesha, Racine, and Kenosha and their immediate environs. The preliminary recommended plan proposes to expand local transit service into residential areas of high and medium density. As shown on Map 105, these areas of expanded service are generally located in southern and northern Milwaukee County and in the most heavily developed portions of Waukesha County. As proposed under the preliminary plan, local transit service would operate over 1,860 round-trip route-miles within the Region, an increase of 406 route-miles, or 28 percent, over the approximately 1,454 routemiles provided in 1991.

The frequency of local transit service would be improved over 1991 levels. Within Milwaukee County peak-period headways on the major routes in the area south of Silver Spring Drive, east of 76th Street, and north of Layton Avenue would be improved from 10 to 40 minutes to 10 minutes. Peak-period headways in the Racine and Kenosha urban areas would be improved from 20 to 30 minutes to 15 to 30 minutes. Peakperiod headways in the Waukesha urban area would be improved such that all routes would operate at 30 minute headways.

As already indicated, under the preliminary recommended plan, local transit fares would remain at 1994 levels. Accordingly, fares within Milwaukee County would be \$1.25; within Racine, \$0.60; and within Waukesha and Kenosha, \$0.75, increasing only with general price inflation. The preliminary recommended plan also recognizes the need to provide local transit service in the small urban communities of the Region, particularly through shared-ride taxi service.

The recommended plan also includes a paratransit service component which is consistent with the Americans with Disabilities Act (ADA) of 1990. Accordingly, the plan assumes that all transit vehicles that provide conventional fixed-route transit service would be accessible to persons with disabilities, including persons using wheelchairs. This assumption is reflected in the capital cost estimate for transit vehicle fleet replacement and expansion under the recommended plan. The plan also ensures that all public entities operating fixed-route transit systems will provide comparable paratransit service to those disabled persons within local transit service areas who are unable to use fixed-route transit services. Accordingly, the complementary paratransit services currently provided within the Region would continue to be operated and expanded, consistent with the planned expansion of local transit service areas within the Kenosha, Milwaukee, and Racine urbanized areas.

Like existing complementary paratransit services provided within the Region, the planned paratransit services would meet Federally specified ADA eligibility and service requirements. In this respect, the complementary paratransit services would serve any person with a permanent or temporary disability who is unable independently to board, ride, or disembark from an accessible vehicle used to provide fixed-route transit service, who is capable of using an accessible vehicle, but one is not available for the desired trip, or who is unable to travel to or from the boarding or disembarking location of the fixed-route transit service. Within a given area, the planned paratransit service would be available during the same hours and on the same days as the fixed-route transit service, would be provided to eligible persons on a "next day" trip reservation basis, would not limit service to eligible persons based on restrictions or priorities relative to trip purpose, and would not be operated under capacity constraints which might limit the ability of eligible persons to receive service for a particular trip. The paratransit service fares assumed under the preliminary recommended plan would range from about \$1.20 to \$2.50 per one-way trip.

Arterial Street and Highway System

In 1991, the arterial street and highway system in the Region consisted of about 3.274 route-miles of facilities. Under the preliminary recommended plan, the arterial system would be increased by about 340 route-miles, to 3,614 route-miles and 10,208 lane-miles, by the year 2010 (see Map 106). The additional arterial mileage reflects primarily the conversion of existing nonarterial facilities to arterial status and function as urban development proceeds within the Region. The proposed arterial street and highway improvements are shown in summary form for each county on Maps 107 through 113 and are set forth by county and by arterial facility type in Table 250. The preliminary recommended plan includes the following improvements to the arterial street and highway system.

System Expansion: Constructing New Facilities: System expansion includes all projects which would significantly increase the capacity of the existing system through construction of new facilities. The plan would provide for the construction of 128 route-miles and 331 lane-miles of new arterial facilities. These include such longplanned facilities as the Lake Parkway south from the Hoan Bridge to E. Layton Avenue, the STH 16 freeway bypass of Oconomowoc, the completion of the Waukesha bypass, the STH 36 bypass of Burlington, and the completion of the USH 12 freeway in Walworth County. Excluded from the plan are the previously planned STH 83 bypass of the Village of Chenequa in Waukesha County and the Lake Arterial through southern Milwaukee County and eastern Racine and Kenosha Counties. In all, proposed new arterial street and highway facilities would represent about 3 percent of both the total planned arterial route-miles and lane-miles.

System Improvement: Widening Existing Facilities: System improvement includes all projects which would significantly increase the capacity of the existing system through street widening or relocation. Under the preliminary plan, 418 routemiles of facilities would be widened to provide an additional 1,713 arterial lane-miles. Proposed improvements would include the widening of STH 36 in Milwaukee, Waukesha, and Racine Counties; Pewaukee Road (CTH J) in Washington and Waukesha Counties; Cleveland Avenue (CTH D) and Racine Avenue (CTH Y) in Waukesha County; STH 31 and CTH Y in Kenosha and Racine Counties; Northwestern Avenue (CTH K) and Spring Street (CTH C) in Racine County; STH 57 and Port Washington Road (CTH W) in Ozaukee County; STH 33 in Ozaukee and Washington Counties; Sherman Boulevard (CTH G) and Rawson Avenue (CTH BB) in Milwaukee County; and the completion of the widening of STH 50 in Kenosha and Walworth Counties. The system improvement activities would comprise about 12 percent of the total planned route-miles and 17 percent of the total planned lane-miles.

System Preservation: Maintaining Existing Facilities: System preservation includes all arterial improvement projects required to maintain the structural adequacy and serviceability of the existing arterial system without significantly increasing the capacity of that system. This would include all projects classified as resurfacing and reconstruction for the same capacity. The preliminary plan proposes system preservation activities for about 3,068 route-miles and 8,164 lane-miles, representing about 85 percent of the total planned route-miles and 80 percent of the total planned lane-miles. Important preservation actions would include the pavement, bridge, and intersection restoration and reconfiguration work needed to maintain and modernize the Milwaukee area freeway system, which will reach the end of its physical life during the plan implementation period.

PLAN PERFORMANCE AND COSTS

Selected characteristics of the preliminary recommended regional transportation system plan for 2010 are identified in Tables 251 and 252. The number of internal person trips generated within the Region on an average weekday is expected to increase under the plan from 5.54 million in 1991 to about 6.10 million in the year 2010, or by about 10 percent. The number of transit trips made on an average weekday is expected to increase from about 172,200 in 1991 to about 196,600 in the year 2010, or by about 14 percent, assuming the transit plan recommendations are implemented. Despite this increase in daily transit tripmaking, the proportion of total internal person trips made by transit would remain at about 3 percent over the plan design period.

The number of vehicle-miles of travel within the Region on an average weekday is expected to increase by about 29 percent, from about 33.07 million in 1991 to about 42.70 million in 2010. Of the latter total, about 16 million vehicle-miles of



The arterial street and highway system under the preliminary recommended plan would total 3,614 route-miles and 10,208 lane-miles of facility. The plan would provide for the construction of 128 route-miles and 331 lane-miles; for the widening of 418 route-miles, which would provide an additional 1,713 lane-miles; and for the preservation of 3,068 route-miles and 8,164 lane-miles of arterial street and highway facilities.

Map 107

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN KENOSHA COUNTY: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN



LEGEND

ARTERIAL STREET OR HIGHWAY





Under the preliminary recommended plan, the arterial street and highway system in Kenosha County would increase by 39 miles, or 12 percent, to 357, miles by the year 2010. The increase in arterial mileage would come about largely through the conversion of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of about seven miles, for the widening of about 43 miles, and for the preservation of about 307 miles of arterial facilities within the County.



Under the preliminary recommended plan, the arterial street and highway system in Milwaukee County would be increased by 21 miles, or 3 percent, to 796 miles, by the year 2010. The increase in arterial mileage would come about largely through the conversion of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for construction of about 12 miles, for the widening of about 49 miles, and for the preservation of about 735 miles of arterial facilities within the County.



FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN OZAUKEE COUNTY: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN

Map 109

Source: SEWRPC.

KATHERINA BR

MILWAUKEE

preservation of 256 miles of arterial facilities within the County.

Map 110

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN RACINE COUNTY: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN



Under the preliminary recommended plan, the arterial street and highway system in Racine County would increase by 72 miles, or by about 21 percent, to about 420 miles, by the year 2010. The increase in arterial mileage would come about largely through the conversion of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of about 16 miles, for the widening of about 62 miles, and for the preservation of about 342 miles of arterial facilities within the County.



EAST TROY AND Commine D AFAYETTE GENEVA SPRING PRAIR WISCONSIN LINK

LEGEND

ARTERIAL STREET OR HIGHWAY

NEW

WIDENING AND/OR OTHER IMPROVEMENT TO PROVIDE SIGNIFICANT ADDITIONAL CAPACITY

RESURFACING OR RECONSTRUCTION TO PROVIDE ESSENTIALLY THE SAME CAPACITY

4 NUMBER OF TRAFFIC LANES FOR NEW OR WIDENED AND/OR IMPROVED FACILITY (2 LANES WHERE UNNUMBERED) Under the preliminary recommended plan, the arterial street and highway system in Walworth County would increase by about 55 miles, or by about 13 percent, to 484 miles, by the year 2010. The increase in arterial mileage would come about largely through the conversion of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of about 36 miles, for the widening of nearly 28 miles, and for the preservation of about 420 miles of arterial facilities within the County.

Source: SEWRPC.

Map 111





FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WASHINGTON COUNTY: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN

Under the preliminary recommended plan, the arterial street and highway system in Washington County would be increased by 71 miles, or by about 18 percent, to 470 miles, by the year 2010. The increase in arterial mileage would come about largely through the conversion of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of about 25 miles, for the widening of about 70 miles, and for the preservation of about 375 miles of arterial facilities.

Source: SEWRPC.

Map 112



FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WAUKESHA COUNTY: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN

LEGEND

ARTERIAL STREET OR HIGHWAY

- NEW
 WIDENING AND ZOR OTHER IMPROVEMENT TO
 PROVIDE SIGNIFICANT ADDITIONAL CAPACITY
 RESURFACING OR RECONSTRUCTION TO PROVIDE
 ESSENTIALLY THE SAME CAPACITY
 - 4 NUMBER OF TRAFFIC LANES FOR NEW OR WIDENED AND/OR IMPROVED FACILITY (2 LANES WHERE UNNUMBERED)

Under the preliminary recommended plan, the arterial street and highway system in Waukesha County would be increased by 60 miles, or by about 8 percent, to 776 miles, by the year 2010. The increase in arterial mileage would come about largely through the conversion of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of about 25 miles, for the widening of about 119 miles, and for the preservation of 632 miles of arterial facilities. *Source: SEWRPC.*

Î

Map 113

ARTERIAL STREET AND HIGHWAY SYSTEM PRESERVATION, IMPROVEMENT, AND EXPANSION BY ARTERIAL FACILITY TYPE BY COUNTY: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN

	System Preservation		System Improvement		System Expansion		
		Percent		Percent		Percent	
County	Miles	of Total	Miles	of Total	Miles	of Total	Total
Kenosha							
Freeway	12.0	100.0	0.0	0.0	0.0	0.0	12.0
Standard Arterial	295.5	85.6	42.6	12.3	7.3	2.1	345.4
Subtotal	307.5	86.1	42.6	11.9	7.3	2.0	357.4
Milwaukee							
Freeway	66.2	95.7	3.0	4.3	0.0	0.0	69.2
Standard Arterial	669.3	92.0	45.7	6.3	12.1	1.7	727.1
Subtotal	735.5	92.4	48.7	6.1	12.1	1.5	796.3
Ozaukee							
Freeway	27.4	100.0	0.0	0.0	0.0	0.0	27.4
Standard Arterial	228.7	.80.8	48.5	17.1	5.9	2.1	283.1
Subtotal	256.1	82.5	48.5	15.6	5.9	1.9	310.5
Racine							
Freeway	12.0	100.0	0.0	0.0	0.0	0.0	12.0
Standard Arterial	329.7	80.8	62.2	15.2	16.1	4.0	408.0
Subtotal	341.7	81.4	62.2	14.8	16.1	3.8	420.0
Walworth							
Freeway	50.0	75.0	0.0	0.0	16.7	25.0	66.7
Standard Arterial	370.3	88.7	27.4	6.6	19.7	4.7	417.4
Subtotal	420.3	86.8	27.4	5.7	36.4	7.5	484.1
Washington							
Freeway	21.6	50.6	21.1	49.4	0.0	0.0	42.7
Standard Arterial	353.2	82.7	48.9	11.4	25.0	5.9	427.1
Subtotal	374.8	79.8	70.0	14.9	25.0	5.3	469.8
Waukesha							
Freeway	57.6	88.2	2.0	3.1	5.7	8.7	65.3
Standard Arterial	574.7	80.9	116.7	16.4	19.2	2.7	710.6
Subtotal	632.3	81.5	118.7	15.3	24.9	3.2	775.9
Region							
Freeway	246.8	83.6	26.1	8.8	22.4	7.6	295.3
Standard Arterial	2,821.4	85.0	392.0	11.8	105.3	3.2	3,318.7
Total	3,068.2	84.9	418.1	11.6	127.7	3.5	3,614.0

Source: SEWRPC.

travel, or about 37 percent, is expected to be made on freeways which would comprise about 8 percent of the total arterial system.

Arterial street and highway congestion, evidenced by the number of arterial miles operating over capacity, is expected to decrease from about 385 miles, or about 12 percent of the total system, in 1991 to about 167 miles, or about 5 percent of the total system, in 2010. The number of arterial miles operating at design capacity, however, is expected to increase from about 158 miles, or about 5 percent of the total system in 1991, to about 218 miles, or about 6 percent of the total system in 2010. The location of the facilities expected to operate under, at, and over design

SUMMARY OF TRANSPORTATION PERFORMANCE CHARACTERISTICS: 1991 AND 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN

Perf	ormance Characteristic			Percent
Category	Specific Measure	Base Year 1991	Preliminary Plan: 2010	Change
Travel	Internal person trips (average weekday)	5.54 million	6.10 million	10.1
	Internal vehicle trips (average weekday)	4.58 million	5.27 million	15.1
	Vehicle-miles of travel ^a (average weekday)	33.07 million	42.70 million	29.1
	Transit ridership (average weekday)	172,200	196,600	14.2
	Relative distribution of trips by mode of travel (average weekday) Auto driver ^b Auto passenger Transit passenger School bus passenger	73.3 percent 19.5 percent 3.1 percent 4.1 percent	76.8 percent 15.3 percent 3.2 percent 4.7 percent	
	Proportion of trips made by transit within Milwaukee County	3.8 percent	3.8 percent	
	Proportion of trips made by transit to Milwaukee central business district	13 percent	12 percent	
	Proportion of passenger-miles of travel made on transit (average weekday)	1.1 percent	1.5 percent	
Traffic Congestion ^C	Amount and proportion of arterial street and highway system over design capacity Moderately congested (V/C ratio 1.01 to 1.10) Severely congested (V/C ratio 1.11 to 1.30) Extremely congested (V/C ratio over 1.31)	106 miles 3.2 percent 217 miles 6.6 percent 62 miles 2.0 percent	83 miles 2.3 percent 48 miles 1.3 percent 36 miles 1.0 percent	-21.7 -77.9 -41.9
	Total	385 miles 11.8 percent	167 miles 4.6 percent	-56.6
Energy Use	Annual motor fuel consumption (millions of gallons) By auto and truck By motor bus Total	353.2 5.7	444.2 9.2 453.4	25.8 61.4
Air Quality	Air pollutant emission from vehicles on the arterial street and highway system (tons per average weekday) Volatile organic compounds	119.2	28.6	-76.0
	Nitrogen oxides Carbon monoxide	129.8 815.1	79.1 237.8	-39.1 -70.8

^aWithin Walworth County, the number of vehicle-miles of travel may be expected to increase from 1.91 million in 1991 to 2.81 million in 2010, or by 47 percent, under the preliminary recommended plan.

^bThe average vehicle occupancy for work trips within the Region may be expected to decline from 1.06 in 1991 to 1.03 under the preliminary recommended plan.

^CLevel of service designations are used to qualitatively measure the operational characteristics of a given link in the arterial network. Under this approach, there are six levels of service corresponding to letters "A" through "F"; "A" describing free flow, unrestricted traffic conditions, and "F" describing a breakdown in traffic flow. Arterial facilities operating under design capacity exhibit travel conditions representative of levels of service "A" or "B." Facilities operating at design capacity exhibit travel conditions representative of level of service "C." Facilities operating moderately over design capacity, with volume to design capacity ratios of 1.01 to 1.10, exhibit travel conditions representative of level of service of 1.11 to 1.30, exhibit travel conditions representative of levels of service of 1.11 to 1.30, exhibit travel conditions representative of levels of 1.31 or higher, exhibit travel conditions representative of level of service "F."

Source: SEWRPC.

capacity are shown on Map 114. Annual motor fuel consumption in the year 2010 is expected to approximate 453 million gallons, or about 94 million gallons more than the amount consumed in the Region in 1991.

The average annual public cost of carrying out the recommended plan, including the construction of new facilities and the operation and maintenance of the arterial street and highway and transit systems, is estimated at nearly \$473 million. The anticipated average annual public revenues, excluding transit fare box revenues, is estimated at \$381 million. Thus, the difference between anticipated costs and expected revenues is \$92 million per year over the plan design period. The equivalent of an \$0.11 per gallon increase in the motor fuel tax within the Region would be necessary to eliminate the estimated \$92 million annual shortfall.

Anticipated Impact on Travel Attendant to a 100 Percent Increase in Perceived Automobile Operating Costs

The Technical Coordinating and Advisory Committee determined that the potential impacts associated with a policy of substantially increasing automobile operating costs should be presented for public discussion. The Advisory Committee, while recognizing the potential benefits attributable to such a demand management policy, considered implementation of the degree of pricing measures necessary to significantly affect travel demand impractical within the 16-year plan implementation period. The following review of the potential impacts attendant to such a policy is based upon the Commission analyses of Alternative Plan 1. It should be noted that the information provided in Table 253 assumes not only the institution of significant pricing measures to discourage the use of the automobile for personal travel but also assumes full implementation of substantial improvements and expansions of public transit service within the Region included in Alternative Plan 1.

The pricing measures proposed would increase perceived costs of automobile use by 100 percent over the 1991 level. This doubling would mean that the cost of automobile operation per mile would increase from about \$0.055 to about \$0.11 per mile. This would be the equivalent of a \$1.10 per gallon increase in the cost of motor fuel. Several measures could be used in combination to increase the cost; the Advisory Committee suggested for consideration motor fuel taxes, travel-sensitive automobile insurance payments, travel-sensitive vehicle registration fees, and the elimination of employer-paid parking, among others.

Should the perceived cost of automobile operation be increased by 100 percent, 38.38 million vehiclemiles of travel could be expected to take place on

AVERAGE ANNUAL PUBLIC COSTS AND REVENUES ASSOCIATED WITH THE PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN: 1995 THROUGH 2010

Cost or Revenue Item	Preliminary Plan 2010
Transportation System Cost (average annual 1995-2010	
Arterial Street and Highway System	
Capital	228 68
Subtotal	296
Transit System Capital	76 101
Subtotal	177
Total	473
Transportation System Revenues (average annual 1995-2010 expressed as millions of dollars) Highway Capital ^b	
Federal	132 66 11
Subtotal	209
Highway Operating State	36 28
Subtotal	64
Transit Capital ^b Federal	23 3 26
Federal State Local Local	6 64 12
Subtotal	82
Total	381
Cost-Revenue Comparison Average Annual Difference between Cost and Revenue (millions of dollars) Motor Fuel Tax Required to Fund Shortfall (cents per gallon)	-92 11

^aThe transit operating costs presented in this table represent net operating costs, that is, total operating costs less farebox revenues. Estimated current 1993 annual transit farebox revenues in Southeastern Wisconsin were \$34 million, total annual operating costs were \$97 million, and net operating costs were \$63 million. It is estimated that upon full implementation of the paratransit service needed to meet the requirements of the Federal Americans with Disabilities Act of 1990, total and net annual operating costs would substantially increase. Indeed, under the no-build plan, total annual operating costs would be expected to increase to \$119 million, expressed in 1994 constant dollars. The estimated year 2010 annual total operating costs and farebox revenues adjusted to reflect the historic differential impact of general price inflation on transit operating costs and farebox revenues, including the effect on ridership of increasing fares, is \$36 million, expressed in 1994 constant dollars. The year 2010 annual net operating cost, accordingly, is \$107 million, also expressed in 1994 constant dollars.

^bFederal funds for capital assistance may, in general, be used for either arterial street and highway or transit capital projects.

Source: SEWRPC.



Under the preliminary recommended plan, the level of traffic congestion may be expected to be substantially below that which would occur under the no-build plan. By the year 2010, only about 5 percent of the 3,614-mile arterial highway system would operate over design capacity. About 83 miles, or more than 2 percent, of the planned arterial mileage, would be moderately congested; 48 miles, or about 1 percent, would be severely congested; and about 36 miles, or about 1 percent, would be extremely congested. While the transportation development proposals included in the preliminary recommended plan serve to reduce the amount of congested facilities throughout the entire region, the Milwaukee area freeway system may be expected to carry traffic volumes which exceed its design capacity and to operate with congested conditions through the year 2010.

COMPARISON OF THE PERFORMANCE OF ALTERNATIVE PLANS: 2010 PRELIMINARY RECOMMENDED TRANSPORTATION SYSTEM PLAN AND PLAN WITH PRICING MEASURES

Performance Characteristic		Alternative Plan with		
Category	Specific Measure	Pricing Measures: ^a 2010	Preliminary Plan: 2010	
Travel	Vehicle-miles of travel (average weekday)	38.38 million	42.70 million	
Traffic Congestion	Amount and proportion of arterial street and highway system over design capacity Moderately congested (V/C ratio 1.01 to 1.10) Severely congested (V/C ratio 1.11 to 1.30) Extremely congested (V/C ratio over 1.31)	87 miles 2.4 percent 39 miles 1.1 percent 5 miles 0.1 percent	107 miles 3.0 percent 54 miles 1.5 percent 7 miles 0.2 percent	
	Total	131 miles 3.6 percent	168 miles 4.5 percent	
Energy Use	Annual motor fuel consumption (millions of gallons) By auto and truck By motor bus	400.0 12.1	444.2 9.2	
	Total	412.1	453.4	
Air Quality	Air pollutant emission from vehicles on the arterial street and highway system (tons per average weekday) Volatile organic compounds Nitrogen oxides Carbon monoxide	25.6 71.3 211.5	28.6 79.1 237.8	
Natural Resource Land Lost	Primary Environmental Corridors (acres) Prime Agriculture Lands (acres)	431.9 1,038.7	475.4 1,643.7	

^a This alternative assumes a 100 percent increase in the cost of automobile operation, equivalent to a motor fuel tax increase of \$1.10 per gallon.

Source: SEWRPC.

the arterial street and highway system on an average weekday in the plan design year 2010. This would be about 4.32 million fewer vehiclemiles than expected under the preliminary recommended plan. By way of comparison, in 1991 about 33.07 million vehicle-miles of travel took place on the arterial street and highway system of the Region.

About 131 miles of arterial facilities could be expected to operate over design capacity during peak travel periods should the perceived cost of driving be increased by 100 percent as proposed under Alternative Plan 1. This would be about 37 miles, or 22 percent, fewer than could be expected under the preliminary recommended plan. The lower levels of travel and traffic congestion could be expected to result in less air pollutants emitted and motor fuel consumed within the Region; 25.6 tons of volatile organic compounds, 71.3 tons of nitrogen oxides, and 211.5 tons of carbon monoxide would be generated, representing 3.0, 7.8, and 26.3 fewer tons, respectively. Also, 412 million gallons of fuel could be expected to be consumed annually within the Region, about 41 million gallons, or 9 percent, less per year than could be expected under the preliminary recommended plan. Should significant pricing measures be implemented, the planned extent of the arterial street and highway system could be reduced by about 15 route-miles from the 3,614-mile system proposed under the preliminary recommended plan. The lower levels of arterial improvement and expansion necessary would mean that nearly 44 fewer acres of primary environmental corridor land and 605 fewer acres of prime farmland would be lost to highway construction.

Commission analyses of the impacts of doubling automobile operating costs indicate that residents of the Region would tend to reduce the distance they are willing to travel by automobile. This change in travel behavior would be manifest in the decisions of households to locate closer to employment centers and to pursue shopping and recreational opportunities closer to home. Such a change in travel behavior would reduce motor vehicle related air pollution and motor fuel consumption. The change in travel behavior would also slightly reduce the number of arterial street and highway improvements and expansions necessary over the plan design period. Though the Commission analyses did not specifically focus on the potential land use impacts attendant to such pricing measures, the findings indicated that, given the strong relationship between travel behavior and urban development patterns, the potential for such measures to reduce urban sprawl regionwide exists.

PUBLIC REACTION TO THE PRELIMINARY RECOMMENDED PLAN

The preliminary recommended regional transportation system plan was the subject of a Regional Planning Conference held on June 27, 1994; a series of public informational meetings and hearings held in each of the seven counties of the Region in July and August of 1994; meetings of the Commission's Kenosha, Ozaukee, Racine, Walworth, Washington, and Waukesha County Technical Coordinating and Advisory Committees on Jurisdictional Highway System Planning; and a joint meeting of the Intergovernmental Coordinating and Advisory Committees on Transportation System Planning and Programming for the Kenosha, Milwaukee, and Racine Urbanized Areas. The conference and public hearings were attended by a total of 760 persons;

143 letters were received for the record following the public hearings. Membership of the jurisdictional highway system planning committees totals 157 persons; membership of the three transportation system planning and programming committees totals 42 persons. It is important to note that every local unit of government within a county, as well as the county government, is represented on the county jurisdictional highway planning committees.

For use in connection with the meetings and hearings, the Commission prepared and widely distributed three SEWRPC Newsletters. These included Vol. 33, No. 5, "Work Begins on a New Regional Transportation System Plan"; Vol. 34, No. 2, "Alternative Regional Transportation System Plans Designed and Evaluated"; and Vol. 34, No. 3, "New Regional Transportation System Plan Readied for Public Review." Together, these Newsletters provide background on the transportation system planning process and the existing transportation system; descriptive and evaluative information on the plan alternatives; and, in summary form, a description of the preliminary recommended plan. In addition, copies were made available of SEWRPC Newsletter, Vol. 32, No. 2, which summarized the design year 2010 regional land use plan that formed the basis for the new regional transportation system plan.

Through the Newsletters, the Regional Planning Conference, the public hearings, and the various Advisory Committee meetings, information on the findings of the inventories and analyses, on alternative plans, and on the preliminary plan was provided in considerable depth and detail to the elected officials of each county and local unit of government within the Region, as well as to officials of the State and Federal governments, representatives of business and industry, and interested citizens. The planning work received substantial attention from the mass media in the form of numerous newspaper articles and editorials and radio and television announcements. some based upon Commission news releases. The record of the public hearings, together with attendant correspondence and supporting materials, was published by the Commission and provided for review to each member of the Commission Technical Coordinating and Advisory Committee on Regional Transportation

System Planning and to each member of the Regional Planning Commission. The full record is on file at the Commission offices.¹

The following sections of this chapter summarize the public reaction to the preliminary recommended regional transportation system plan as expressed at the County jurisdictional highway system planning advisory committee meetings; at the joint transportation system planning and programming advisory committee meetings; and at the public hearings, including the written comments received by the Commission following the hearings. In addition, the Advisory Committee response to the public reaction is documented.

<u>County Jurisdictional Highway System</u> <u>Planning Advisory Committee and</u> <u>Transportation System Planning and</u> Programming Advisory Committee Meetings

The Commission Technical Coordinating and Advisory Committees on Jurisdictional Highway System Planning for Kenosha, Ozaukee, Racine, Walworth, Washington, and Waukesha Counties each met to consider and each acted to approve the preliminary recommended regional transportation system plan with the following requested changes:

- <u>Kenosha County</u>
 - 1. The addition to the plan of the improvement of CTH KR from two to four traffic lanes from IH 94 to STH 32
 - 2. The addition to the plan of the improvement of IH 94 from six to eight traffic lanes within Kenosha County
 - 3. The addition to the plan of the longplanned extension of CTH KD and the long-planned realignments of CTH F, CTH Q, and CTH AH
 - 4. The modification of the plan to reroute STH 50 between 39th Avenue and STH 32 over Roosevelt Road and 63rd Street rather than the existing route of

75th Street and the improvement of this new route from two to four traffic lanes

- Ozaukee County
 - 1. The addition to the plan of the improvement of IH 43 from four to six traffic lanes from Brown Deer Road in Milwaukee County to Highland Road in Ozaukee County
 - 2. The addition to the plan of the longplanned extension of Granville Road from Freistadt Road to Highland Road
 - 3. The modification of the plan to retain Belgium-Kohler Road between CTH E and STH 57 as a local arterial facility
- <u>Racine County</u>
 - 1. The modification of the plan to retain the existing state trunk highway routing of STH 11 and to retain CTH KR as a county trunk highway facility
 - 2. The addition to the plan of recommendations to protect the potential for future widening of IH 94 from six to eight lanes and for the development of a new interchange on IH 94 at a relocated CTH C
 - 3. The modification of the plan to retain Four Mile Road as a local, rather than county, arterial between IH 94 and STH 31
- Walworth County
 - 1. The addition to the plan of a proposed commuter railway passenger train service from the Village of Walworth through Fox Lake, Illinois, to Chicago
- Washington County
 - 1. The addition to the plan of a new half interchange on USH 41-USH 45 with Freistadt Road
 - 2. The modification of the plan to incorporate an alignment for the proposed eastwest northern arterial route connecting the communities of Hartford and Slinger to USH 41 over Arthur Road between USH 41 and CTH U, then over new alignment between CTH U and Independence Drive, and then over Independence Drive to STH 60

¹See <u>Record of Public Informational Meetings</u> <u>and Public Hearings, Preliminary New Regional</u> <u>Transportation System Plan for Southeastern</u> <u>Wisconsin: 2010</u>, July 11 through August 3, 1994.
• Waukesha County

- 1. The addition to the plan of the improvement of CTH X from two to four traffic lanes from STH 59 to CTH H
- 2. The addition to the plan of the improvement of STH 164 from four to six traffic lanes from IH 94 to USH 18
- 3. The addition to the plan of the extension of River Crest Drive from STH 175 to CTH Q
- 4. The addition to the plan of the expansion of the interchange of IH 94 with CTH P from a half to a full interchange
- 5. The addition to the plan of the improvement of Barker Road from two to four traffic lanes between Racine Avenue and STH 59 and between North Avenue and STH 74 to provide a continuous facility with four traffic lanes from Racine Avenue and CTH Q

In addition, the Transportation System Planning and Programming Advisory Committees for the Kenosha, Milwaukee, and Racine Urbanized Areas acted unanimously to endorse the preliminary plan and to recommend that the proposals for implementing the plan include not only the full funding of the attendant shortfall in revenues needed to implement the plan, but also the substitution of alternative funding for the existing property tax support of public transit services and county and municipal arterial street and highway facilities.

Public Hearing Comments and Correspondence

The comments received at the public hearings and in correspondence following the hearings may be divided into six categories, that is, comments regarding the overall plan, the transit element of the plan, the highway element of the plan, land use, funding, and other comments.

A number of comments were received expressing support and approval of the preliminary plan and, in particular, endorsing the plan as a balanced approach to the use of transit, highway, system management, and land use actions in addressing the future transportation needs of Southeastern Wisconsin. A number of comments were made that the plan should incorporate greater improvement and expansion of transit service and less expansion of highway capacity. However, some comments were also received that the plan should incorporate less transit improvement and expansion.

With respect to the transit element of the plan, a number of statements were made in support of the proposed express and rapid transit service in particular. Support was expressed for the proposed extensions of rapid transit service into Ozaukee, Washington, and Waukesha Counties. A substantial number of comments were received in support of rail transit, including light-rail and particularly commuter rail transit. With respect to commuter-rail passenger service, comments were made expressing the need to improve existing Chicago-based commuter-rail service to Kenosha and to extend such service to the Cities of Racine and Milwaukee. Support was also expressed for the proposed consideration of commuter-rail service in corridors in Waukesha, Washington, and Ozaukee Counties. Comments were made that Chicago-based commuter-rail service to western Kenosha County and Walworth County and Milwaukee-based service to the City of Waukesha should be included in the final plan. Concerning light-rail transit, comments were made that the final plan should recognize that light-rail service could function to provide both express and rapid transit service, depending upon station spacing and facility location, either on reserved street lanes or on such exclusive rights-of-way as abandoned railway corridors. Other technologies were also suggested as alternatives to light-rail service, including personal rapid transit and "pipe rail" technology, used for amusement park rides.

With respect to the bus element of the plan, some comments were received which requested that the final plan incorporate even greater improvements in the frequency of bus service in both peak and off-peak periods. Other statements encouraged that a greater geographic expansion of transit service, including that an areawide grid of express bus service, be included in the final plan. Some statements suggested modifying the plan to provide improved frequency of transit by reducing the geographic expansion of transit service proposed in the preliminary plan; other statements asked that the final plan propose a greater geographic expansion of transit service and a reduction in improved service frequency. Suggestions were made that transit service improvements should be implemented prior to any improvement in highway facilities.

Statements, however, were also made in opposition to the improvement and expansion of transit service proposed in the preliminary plan. Statements were made that the plan placed an inordinate emphasis on public transit, given the proportion of the existing and potential future travel which is, and may be expected to be, carried by public transit in Southeastern Wisconsin and the attendant small impacts on arterial street and highway system traffic volumes and congestion and on air pollutant emissions. Comments were made that the plan should not include light-rail or commuter-rail service because of the high capital costs involved compared to bus alternatives. Suggestions were made that busways and HOV lanes be built prior to any rail transit construction in order to demonstrate the potential market for transit. Opposition was also expressed to the construction of any transit facility through the Veterans Administration Center at Wood, Wisconsin.

With respect to the highway element of the plan, most of the statements made expressed opposition to a specific arterial highway improvement and expansion project proposed in the preliminary plan. There were also some statements providing support for specific arterial highway improvements. Some comments were received which indicated overall support of the highway element in the plan and comments were received expressing overall opposition to the highway element of the plan. With respect to individual improvements recommended in the plan, many comments were received in opposition to the long-proposed extension of 30th Avenue as an arterial street between 89th Street and STH 165 in Kenosha County. Some statements of opposition suggested that other alternatives be considered, including the extension of 22nd Avenue or the widening of 39th Avenue.

Comments were also received in opposition to the following arterial street improvements proposed in the preliminary plan: the proposed extension of Cold Spring Road over the Milwaukee River in Ozaukee County, the proposed widening to four traffic lanes of STH 59 between 124th Street and STH 164 in Waukesha County, the construction of a new interchange on IH 94 at Calhoun Road in Waukesha County, the widening of CTH J to four traffic lanes between IH 94 and STH 60 in Washington and Waukesha Counties, the widening of STH 38 between

CTH K and the Racine-Milwaukee county line and of CTH K between STH 38 and IH 94 in Racine County, the widening of STH 50 to six lanes between IH 94 and 39th Avenue and to four lanes between 39th Avenue and STH 32 in Kenosha County, the widening of STH 158 to four lanes between IH 94 and STH 31 in Kenosha County, the expansion of the CTH ML interchange on IH 94 and the proposed extension of CTH ML in Kenosha County, the widening of Washington Road to four traffic lanes between 30th Avenue and STH 32 in the City of Kenosha, the widening of STH 165 to four traffic lanes between STH 31 and STH 32, the widening of STH 60 to four traffic lanes between USH 41 and IH 43 in Ozaukee and Washington Counties, the widening of STH 33 to four traffic lanes between West Bend and IH 43 in Ozaukee and Washington Counties, the construction of the West Waukesha Bypass in Waukesha County, the widening of STH 164 to four traffic lanes between STH 59 and IH 43 in Waukesha County, the widening of STH 67 to four traffic lanes between USH 18 and IH 94 in Waukesha County, the widening of Racine Avenue to four traffic lanes between STH 164 and IH 43 in Waukesha County, and the widening of STH 100 to eight traffic lanes in the Village of Hales Corners between IH 43 and STH 24. Opposition was expressed to all highway capacity expansion projects connecting urban areas and within urban fringe areas because of their perceived potential to promote urban sprawl. Statements were also made in opposition to the proposed HOV facility recommended in the east-west corridor and the proposed consideration of similar facilities in future corridor studies of the Milwaukee area freeway system.

Comments were also made in support of a number of highway improvement and expansion projects recommended in the preliminary plan, including a HOV facility in the east-west corridor and other such facilities in other corridors of the Milwaukee area freeway system. Support was also expressed for the proposed modernization of IH 94 in Waukesha and Milwaukee Counties, including the reconstruction of freeway interchanges to modern design standards, utilizing all right-hand on- and off-ramps and auxiliary lanes. Support was also expressed for the widening of STH 31 to four traffic lanes between CTH MM and STH 38 in Racine County, the widening of CTH J to four traffic lanes between IH 94 and STH 60 in Waukesha and Washington Counties, and the construction of extensions of Barker Road to provide a second continuous north-south arterial facility in Waukesha County between IH 43 and the Waukesha-Washington county line.

Also, with respect to the highway element, a number of statements were made opposing modification of the preliminary plan, including the addition to the plan of the long-planned realignments of CTH F and CTH KD in Kenosha County, the retention on the plan of the current routing of STH 11 between STH 75 and STH 32 in Racine County, the addition to the plan of the widening of USH 14 between CTH O and the Walworth-Rock county line in Walworth County, and the replacement of the proposed CTH ML interchange in Kenosha County with a new interchange at 60th Street (CTH K) and the widening of 60th Street to four traffic lanes between IH 94 and 39th Avenue. Statements were also made in support of the widening of STH 158 and CTH S to four traffic lanes as proposed in the plan to provide relief to STH 50; statements were made in support of the deletion in the preliminary plan of the proposed extension of CTH W in Kenosha County.

With respect to land use, comments were received which proposed that the transportation system plan should be based upon a land use plan which assumes all existing and proposed land use be configured and reconfigured to provide a compact high-density land use pattern around transit stations and that legislation strengthening land use control be provided. Other suggestions proposed legislation that would relate highway funding to the implementation of land use plans. Comments were received supporting a no-growth policy in Southeastern Wisconsin. In addition, comments were received stating that urban sprawl was desirable as well as inevitable and needed to be accommodated in transportation planning.

With respect to issues regarding the funding of the estimated \$92 million annual shortfall attendant to the plan, some comments were received which suggested that addressing the shortfall through an additional 11 cent per gallon motor fuel tax was acceptable. A number of comments were made in support of significantly raising automobile operating costs by the equivalent of a dollar per gallon added tax on motor fuel to reflect more appropriately the full social and environmental cost attendant to automobile use. An equal number of comments were made in opposition to such increases in the costs of driving. A number of comments were made in support of developing a new dedicated funding source for public transit and removing transit costs from the local property tax. A statement was made in support of a proposed regional transportation authority and in support of taxing parking facilities.

Other suggestions made with respect to the transportation plan included comments in support of the construction of bicycle ways to promote the use of the bicycle as an alternative to the automobile. A comment was made in support of annual vehicle inspections for compliance with air pollutant emission standards. Comments were made that the increase in vehicle-miles of travel within the Region envisioned in the plan was too high, given the forecast increase in population: it was suggested that vehicle-miles of travel should be reduced by increased transit service. However, a comment was also made that the impact of transit improvements on the transportation system and vehicle-miles of travel was very small and that such transit improvements cannot be justified on the basis of their impacts on air quality. A comment was also made that additional alternatives should be examined with respect to land use and travel demand management and that the Regional Planning Commission should work with citizen groups to facilitate evaluation of such additional alternatives. A comment was made that the plan should accept the occurrence of traffic congestion, not propose measures to reduce congestion.

The City Engineer of the City of Milwaukee expressed support for the preliminary plan, noting that the plan was generally in accord with the development and redevelopment goals and objectives of the City of Milwaukee and appropriately emphasized public transit and travel demand management actions. The City Engineer, however, also expressed concern over the feasibility and desirability of implementing the additional curb-lane parking restrictions proposed in the plan, the need for adequate funding of proposed implementation of good traffic engineering practices, and the need to conduct the light-rail and commuter-rail major investment studies in the corridors identified in the preliminary plan in a timely manner. The City Engineer also requested that the following changes be made to the arterial street and highway system element of the preliminary plan: the addition, as an arterial facility, of Canal Street between USH 41 and S. 2nd Street; the addition of the widening to four traffic lanes of W. Fond du Lac Avenue between N. 20th Street and N. 35th Street; the addition of the widening to four traffic lanes of E. Oklahoma Avenue from S. Kinnickinnic Avenue to Lake Parkway; the addition of the provision of four traffic lanes on N. 124th Street between STH 145 and Brown Deer Road: the addition of W. Metro Boulevard from USH 45 to N. 107th Street as a four-lane arterial; and the deletion of N. 68th Street between N. Industrial Road and W. Brown Deer Road.

The Mayor of the City of Kenosha requested that the following changes be made to the plan: the addition of 50th Street between Sheridan Road and 30th Avenue as an arterial facility; the addition of the long-planned extension of 85th Street between Sheridan Road to 7th Avenue as an arterial facility; the modification of the plan to recommend six traffic lanes, rather than the proposed four traffic lanes, on STH 158 between IH 94 and STH 31; and the addition of a proposed improvement of 60th Street between Sheridan Road and 39th Avenue to four traffic lanes and between IH 94 and STH 31 to six traffic lanes. In addition, the Mayor requested that, following completion and adoption of the new regional plan, a special study be conducted of the special transportation needs between the Kenosha urbanized area and northeastern Illinois. The requested study would consider particularly commuter movements between these areas and the potential transit services and facilities that would serve those movements.

The Kenosha County Executive expressed support for the preliminary plan and particularly its balanced approach to public transit and arterial highway improvement and expansion. The County Executive also requested that the plan be modified to include the following long-planned highway improvements in Kenosha County: the realignments of CTH F, CTH Q, CTH AH, and CTH KD; the long-planned extension of CTH W; and the improvement of IH 94 from the Wisconsin-Illinois state line to Milwaukee to provide eight traffic lanes. The Mayor of the City of New Berlin expressed support of the preliminary plan as a realistic proposal to meet existing and future transportation needs and supported the balance proposed in the plan with respect to highways, transit, land use, and travel demand management. The Mayor further commented that there was no need for the extension of commuter-rail or light-rail service into Waukesha County, expressing concern that the plan devoted too much of the limited available capital resources to public transit.

The President of the Village of Grafton commented that the final plan must provide needed transportation services for the Region as a whole and that needed transportation improvements should not be eliminated from the preliminary plan because of opposition of individual local governments, citing as an example the opposition of the City of Mequon to the long-proposed improvement of Wauwatosa Road in Ozaukee County between STH 167 and STH 60. The Village President also requested the extension in the final plan of rapid transit bus service from the Village of Grafton to the City of Port Washington.

The Mayor of the City of Port Washington also requested that the planned rapid transit bus service be extended from the Village of Grafton to the City of Port Washington. The Mayor also supported the proposal in the plan for commuter-rail service into Ozaukee County, requesting that the potential for the extension of such service to Port Washington be considered. The Mayor also expressed opposition to the longplanned extension of Walter Street between Grant Street and CTH LL in the City and Town of Port Washington.

The Chairman of the Town of Saukville expressed opposition to the proposed extension of Cold Springs Road over the Milwaukee River, suggesting that a new bridge over the Milwaukee River instead be provided at some point north of Cold Springs Road. The Town Chairman also expressed support of the proposed improvement of STH 33 through the Village of Saukville from two to four traffic lanes.

The Planner for the Town of Mt. Pleasant suggested that the preliminary plan be modified by replacing the proposed incorporation of Willow Road, Stuart Road, and Airline Road as an arterial facility with an arterial facility consisting of Airline Road, Oakes Road, and extensions of Oakes Road between STH 11 and CTH K. The Town Planner noted that the Town was attempting to preserve right-of-way for this facility and that such a facility would not require any railway crossings, unlike the proposal in the preliminary regional plan.

The Director of Community Services of the City of West Bend requested that the preliminary plan be modified to incorporate the extension of rapid transit bus service from the Village of Germantown to the City of West Bend.

The City of Waukesha, through comments made at a briefing on the plan for the Mayor and Common Council and through a letter from the City Director of Public Works, requested that the plan be modified to incorporate the widening to four traffic lanes of CTH X from CTH H to STH 59. The Mayor and Common Council, as well, indicated their support of the plan and, in particular, noted that the West Waukesha bypass was a particularly needed facility.

The Planner for the Village of Pleasant Prairie requested that the extension of 30th Avenue between 89th Street and STH 165 be deleted from the plan and that the widening of STH 50 between 60th Avenue and STH 32 be deleted from the plan.

State Senator Joseph F. Andrea, through his comments in a newspaper article appearing in the <u>Kenosha News</u>, indicated his overall support of the plan but questioned the ability to fund fully the financial shortfall attendant to the plan. He also expressed concern over the extent of public transit service improvement and expansion in the plan, given the small proportion of regional travel which is now carried and which may be expected to be carried by public transit.

The Mayor and Common Council of the City of Oak Creek requested that the preliminary plan be modified to include long-planned new freeway interchanges on IH 94 at Drexel Avenue and Puetz Road and that the proposed widening of Pennsylvania Avenue within the City be deleted from the plan.

ADVISORY COMMITTEE RESPONSE TO PUBLIC COMMENT ON PRELIMINARY PLAN

In response to the public comments concerning the modal balance struck in the preliminary new regional transportation system plan, the Advisory Committee determined that the plan provided an appropriate balance with respect to meeting the existing and probable future transportation needs within Southeastern Wisconsin. The Advisory Committee noted that in the plan design process careful consideration was given first to the potential of land use site design, public transit, and transportation system management measures to address the identified transportation needs. Highway system capacity improvement and expansion were considered only as a measure of last resort, that is, as a measure to address traffic congestion problems which the analyses indicated could not be resolved by expansion of public transit services and the implementation of travel and traffic demand management measures.

It was also noted that the plan proposed more than a 74 percent expansion of transit service, measured in terms of vehicle-miles of service, or about a 70 percent expansion in terms of seatmiles of service, and about a 26 percent expansion of highway capacity, measured in terms of added lane-miles. In addition, the plan proposes a significant increase in the proportion of total transportation funding to be devoted to public transit. Under the plan, the proportion of total transportation system funding devoted to public transit would increase from a current 22 percent to nearly 38 percent. The public transit element of the plan would be expected to carry about 3 percent of total person trips generated within Southeastern Wisconsin. The Advisory Committee did note that the relative timing of implementation of transit and highway capacity improvement and expansion was an important issue and that this should be explicitly addressed in the implementation schedule for the planned highway and transit system improvement.

In regard to the transit system element, the Advisory Committee concluded that only modest modifications to the preliminary plan were warranted in response to the public comments. As modified, the plan would extend bus rapid transit service from the Village of Germantown to the City of West Bend and from the Village of Grafton to the City of Port Washington. The plan would also, as modified, recognize the potential to provide commuter passenger train service from the City of Burlington through Silver Lake and Antioch, Illinois, to Chicago and from the Village of Walworth through Fox Lake, Illinois, to Chicago. As modified, the final plan would recommend that feasibility studies, followed by a major investment analyses, be conducted to explore the feasibility and desirability of implementing these extensions of Chicagobased commuter railway passenger train service. As modified, the recommended plan would also include a study to consider the feasibility of providing commuter transit service between the City of Kenosha and Lake County, Illinois.

The Advisory Committee also acted to identify specific measures to improve transit convenience and facilitate transit use. Such measures are of a transportation system management nature and were thus proposed to be part of the transportation system management element of the recommended plan.

The Advisory Committee determined that further modification of the transit element of the plan was not warranted, including extension of additional commuter-rail lines, for example, from the City of Kenosha into western Kenosha County or from the City of Milwaukee to the City of Waukesha, as suggested in some comments on the preliminary plan. With respect to local transit service, the Advisory Committee concluded that the preliminary plan proposed a proper balance between the geographic extension of transit service and the improvement of the frequency of such service and noted that the proposed local transit service component of the plan could be provided as fixed-route local transit service, or, in the alternative, route deviation or demand-responsive service.

With respect to the arterial street and highway element, the Advisory Committee also concluded that only modest modifications to the preliminary plan were warranted in response to the public comments. It was concluded that the level of congestion used to define the need for improvements was appropriate, given that the highway capacity improvements were proposed as a measure of last resort to address traffic congestion problems. The timing of the highway capacity improvements would be coordinated with the timing of transit improvements in a schedule for plan implementation. This schedule would define the highway system improvements needed in the short term to address existing traffic problems or, in the long term, to address potential future traffic problems.

In response to the public comment, the Advisory Committee did recommend the following specific changes with respect to the arterial street and highway element of the plan:

- <u>Kenosha County</u>
 - 1. The deletion from the plan of the extension of 30th Avenue from 89th Street to STH 165
 - 2. The addition to the plan of the longplanned extension of CTH KD, and the long-planned realignments of CTH F, CTH Q, and CTH AH
 - 3. The modification of the plan to reroute STH 50 between 39th Avenue and STH 32 over Roosevelt Road and 63rd Street rather than the existing route of 75th Street and the improvement of this new route from two to four traffic lanes
 - 4. The addition to the plan as an arterial facility of 50th Street from Sheridan Road to 30th Avenue
 - 5. The addition to the plan of the extension of 85th Street from Sheridan Road to 7th Avenue
 - 6. The addition to the plan of the preservation of the potential for future widening, including acquisition of right-of-way, of STH 158 to six traffic lanes from IH 94 to STH 31, of 60th Street to six traffic lanes from STH 31 to IH 94, and of CTH KR from two to four traffic lanes from IH 94 to STH 32
 - 7. The addition to the plan of the improvement of 60th Street from two to four traffic lanes from Sheridan Road to 39th Avenue and the preservation for future widening to six lanes of 60th Street from IH 94 to STH 31
 - 8. The addition to the plan of the preservation of the potential for future widening of IH 94 from six to eight lanes within Kenosha County, including preservation of the potential to provide HOV

lanes, and of the potential for the construction of a new interchange on IH 94 at 60th Street

- Milwaukee County
 - 1. The addition to the plan of Canal Street from USH 41 to S. 2nd Street as an arterial street
 - 2. The addition to the plan of the widening from two to four traffic lanes of W. Fond du Lac Avenue from N. 20th Street to N. 35th Street
 - 3. The addition to the plan of the widening to four traffic lanes of E. Oklahoma Avenue from S. Kinnickinnic Avenue to the Lake Parkway
 - 4. The addition to the plan of the provision of four traffic lanes on N. 124th Street from STH 145 to W. Brown Deer Road
 - 5. The addition to the plan of a new W. Metro Boulevard from USH 45 to N. 107th Street
 - 6. The deletion from the plan of a new N. 68th Street from N. Industrial Road to W. Brown Deer Road
 - 7. The addition to the plan as an arterial of existing W. Cold Spring Road from S. 60th Street to S. 84th Street
- Ozaukee County
 - 1. The addition to the plan of the improvement of IH 43 from four to six traffic lanes from Brown Deer Road in Milwaukee County to Highland Road in Ozaukee County
 - 2. The addition to the plan of the longplanned extension of Granville Road from Freistadt Road to Highland Road
 - 3. The modification of the plan to drop Belgium-Kohler Road from CTH E to STH 57 from the arterial system
 - 4. The deletion from the plan of the longplanned extension of Walters Street from Grant Street to CTH LL
- <u>Racine County</u>
 - 1. The modification of the plan to retain the existing routing of STH 11 and to

retain CTH KR as a county trunk highway facility

- 2. The modification of the plan to retain Four Mile Road as a local, rather than county, arterial from IH 94 to STH 31
- 3. The modification of the plan to replace the designation of Willow Road, Stuart Road, and Airline Road as an arterial facility with Oakes Road and its extensions from STH 11 to CTH K
- 4. The modification of the plan to include CTH S and STH 75 from CTH K to STH 50 as county, rather than State, facilities
- 5. The addition to the plan of the preservation of the potential for future widening of IH 94 from six to eight lanes, including preservation of the potential to provide HOV lanes, and of a potential new interchange on IH 94 at a relocated CTH C
- Walworth County
 - 1. The addition to the plan of the improvement of USH 14 from two to four traffic lanes from CTH O to the Walworth-Rock county line
 - 2. The addition to the plan of the improvement of STH 67 from Lincoln Street to USH 12 from two to four traffic lanes
- Washington County
 - The modification of the plan to incorporate an alignment for the proposed eastwest northern arterial route connecting the communities of Hartford and Slinger to USH 41 over Arthur Road from USH 41 to CTH U, then over new alignment from CTH U to Independence Drive, and then over Independence Drive to STH 60
- Waukesha County
 - 1. The addition to the plan of the improvement of CTH X from two to four traffic lanes from STH 59 to CTH H
 - 2. The addition to the plan of the improvement of STH 164 from four to six traffic lanes from IH 94 to USH 18

- 3. The addition to the plan of the extension of River Crest Drive from STH 175 to CTH Q
- 4. The addition to the plan of the expansion of the interchange of IH 94 with CTH P from a half to a full interchange
- 5. The addition to the plan of the improvement of Barker Road from two to four traffic lanes from Racine Avenue to STH 59 and from North Avenue to STH 74 to provide a continuous four traffic lanes from Racine Avenue and CTH Q

With respect to the consideration of the interrelationship between land use and transportation, the Advisory Committee determined that the transportation plan was appropriately based upon the adopted regional land use plan and that the transportation plan as proposed would serve to promote implementation of the land use plan and, therefore, a desirable pattern for future land use. The Advisory Committee did recommend that sensitivity of the performance of the regional transportation system plan, particularly with respect to vehicle-miles of travel, be identified with respect to potential high-growth as well as intermediate-growth futures for the Region and with respect to continuing trends and decentralized land use scenarios, as well as to centralized land use scenarios, including the adopted regional land use plan.

The Advisory Committee further recommended that a subsequent study be conducted by the Commission in an attempt to determine the extent to which transportation travel habits and patterns and transportation needs would differ under a land use scenario which may propose a more radical departure from continued land use development trends than recommended in the adopted regional land use plan. Such a plan may entail the substantial restructuring of existing, as well as future, development and incorporate significantly higher densities along transit corridors.

With respect to funding, the Advisory Committee concluded that the final plan should generally incorporate the recommendations of the preliminary plan. Accordingly, the final plan would entail a substantial funding shortfall. How best to address that shortfall will be set forth in the plan implementation chapter. The Advisory Committee also determined that the plan implementation chapter would need to address the funding attendant to the removal of the existing costs of public transit and arterial street and highway capital and operating costs from local property tax.

FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

In accordance with the Committee determinations noted above, a final recommended regional transportation plan was synthesized and evaluated. The final plan reflects the consensus of the Kenosha, Ozaukee, Racine, Walworth, Washington, and Waukesha County Jurisdictional Highway System Planning Committees and the Milwaukee, Kenosha, and Racine Urbanized Area Transportation System Planning and Programming Committees, as well as the Technical Coordinating and Advisory Committee on Regional Transportation System Planning. The final plan is largely identical to the preliminary plan, differing only with respect to a relatively few recommended facilities and services. Each of the three major elements of the final plan, transportation system management, public transit maintenance and improvement, and arterial street and highway system maintenance and improvement, is described below.

Transportation Systems Management

The final recommended plan, like the preliminary plan, emphasizes transportation systems management. The plan proposes measures within six separate categories to ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment; to encourage use of highoccupancy vehicles, including buses, vanpools, and carpools; to reduce vehicle use; to effect motor fuel savings; and to modify travel behavior through reductions in vehicular travel during peak travel periods and thereby better adjust travel demand to the available transportation system capacity. The six categories of management measures are:

• Freeway Traffic Management System

The final plan includes implementation of the long-proposed Milwaukee-area freeway traffic management system. This system has the following components:

- 1. A freeway incident management program, including an electronic freeway traffic data-gathering and analysis system, a citizens' band radio monitoring system, a closed-circuit television monitoring system, an emergency service patrol, and a major-incident response team
- 2. A motorist advisory information program, consisting of fixed changeable message signs, transportable changeable message signs, and improved provision of information to commercial radio broadcasting stations
- 3. A freeway operational control program, consisting of ramp meters, highoccupancy-vehicle preferential access lanes, and computer-executed strategies for managing freeway operations by controlling access through an integrated areawide system of ramp meters. The control strategy would seek to provide for average operating speeds of approximately 35 miles per hour on all freeway segments during peak travel periods, although growing traffic volumes and a specified two-minute maximum delay at ramp meters may prevent achieving that objective on certain freeway segments. Preferential access would be provided at the metered ramps for buses and other HOVs
- 4. A central traffic management center, established to coordinate the incident management program, operate the realtime data collection system, and adjust operations through operation of the ramp meters to carry out the control strategy

The Wisconsin Department of Transportation is in the process of implementing this important management recommendation, which was first proposed by the Commission in 1978.

• Curb-Lane Parking Restrictions

The plan identifies 442 miles of the planned 3,607-mile arterial system where it is envisioned that curb-lane parking would be prohibited during peak hours of travel. Maps 115 through 121 identify for each county those arterial facilities where curblane parking restrictions may be advisable. In some cases such restrictions have already been placed into effect. In other cases it will probably be necessary for municipalities to restrict on-street parking as traffic volumes increase over time. It should be noted that such restrictions are essential to providing good transit service as well as to reducing traffic congestion on arterial streets and highways to acceptable levels.

• Traffic Management

The plan recommends the use of state-ofthe-art traffic engineering practices to assist in achieving efficient traffic flow on arterial facilities. The plan also recommends that arrangements to facilitate pedestrian and bicycle movements be considered in all traffic engineering activities and in the design of new or widened arterial streets and highways.

• Intelligent Transportation Systems

The plan recommends the application of advanced traffic management technology known as Intelligent Transportation Systems (ITS) as such technology becomes practical within the Region. Advanced traffic management systems (ATMS) technology is currently being applied by the Wisconsin Department of Transportation as part of the Department efforts to implement the freeway traffic management system. Other elements of ITS include: advanced travel information systems (ATIS) that provide travelers with real-time information on route-specific traffic conditions; advanced public transportation systems (APTS) that facilitate transit use by providing real-time transit vehicle location and scheduling information to transit users; commercial vehicle operations (CVO) technology that automates commercial vehicle licensing, registration, and fee collection; advanced rural transportation systems (ARTS) that provide up-to-date weather and road condition reports to rural highway users; and advanced vehicle control systems (AVCS) or the in-vehicle technology that may improve the safety and efficiency of automobile travel.

• Areawide Promotional Measures

The plan recommends coordinated areawide programs to promote travel through ride-

LOCATION OF ANTICIPATED CURB-LANE PARKING RESTRICTIONS ON ARTERIAL STREETS AND HIGHWAYS IN KENOSHA COUNTY: 2010



To help ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment, it is recommended that concerned local units of government restrict curb-lane parking on certain arterial streets and highways during peak travel periods. The final recommended plan identified about 34.0 route-miles of arterial streets and highways in Kenosha County where it would be advisable for municipalities to restrict parking.



To help ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment, it is recommended that concerned local units of government restrict curb-lane parking on certain arterial streets and highways during peak travel periods. The final recommended plan identified about 212.5 route-miles of arterial streets and highways in Milwaukee County where it would be advisable for municipalities to restrict parking.





LOCATION OF ANTICIPATED CURB-LANE PARKING RESTRICTIONS ON ARTERIAL STREETS AND HIGHWAYS IN OZAUKEE COUNTY: 2010



ROLFSON RO MILWAUKEE CO. A 22 E WAUKESHA CO. CO. CALEBON ROINE TRAYMOND 80 RACINE WATE VEN WIND LAN MATTREFESE 8 FIVE AND ONE HALF FIVE MILE BURMEISTER BUEN OLSON # FOUR NIND BOOSE LAKE NORTH KEG I DOVER WATERFORD ROCHESTER 80 -MJ PLEASANT ST. - GOOLD ST -YOUT ST. -MARQUETTE ST. -DR. MARTIN LUTHER XMS JR. OR -STATE ST. HESTER -SIXTH ST -SIATH ST. -SEVENTH ST. -SEVENTH ST -WISCONSIN AVE. -FOURTEENTH ST CHURCH RD WASHINGTON AVE LAKE KETTER RIFEV ELMWOOD PARK BRAITS . 2 UNION GROVE COLE RA DVER KENOSHA CREEK LEGEND FREEWAY SURFACE ARTERIAL WHERE CURB PARKING RESTRICTION NOT REQUIRED END SURFACE ARTERIAL WHERE CURB PARKING RESTRICTION IS ADVISABLE

LOCATION OF ANTICIPATED CURB-LANE PARKING RESTRICTIONS ON ARTERIAL STREETS AND HIGHWAYS IN RACINE COUNTY: 2010

To help ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment, it is recommended that concerned local units of government restrict curb-lane parking on certain arterial streets and highways during peak travel periods. The final recommended plan identified about 36.5 route-miles of arterial streets and highways in Racine County where it would be advisable for municipalities to restrict

GRAPHIC SCALE

8000

iuu i

4000

2 MILES

12000FEE

ပာ parking. လ

1.2.5

RACINE

RACINE CO. BURLINGTON

KENOSHA CO.

6 19



LOCATION OF ANTICIPATED CURB-LANE PARKING RESTRICTIONS ON ARTERIAL STREETS AND HIGHWAYS IN WALWORTH COUNTY: 2010

To help ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment, it is recommended that concerned local units of government restrict curb-lane parking on certain arterial streets and highways during peak travel periods. The final recommended plan identified about 9.5 route-miles of arterial streets and highways in Walworth County where it would be advisable for municipalities to restrict parking.

2 MILES

CHER.

Source: SEWRPC.

Map 119





LOCATION OF ANTICIPATED CURB-LANE PARKING RESTRICTIONS ON ARTERIAL STREETS AND HIGHWAYS IN WASHINGTON COUNTY: 2010

To help ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment, it is recommended that concerned local units of government restrict curb-lane parking on certain arterial streets and highways during peak travel periods. The final recommended plan identified about 18.5 route-miles of arterial streets and highways in Washington County where it would be advisable for municipalities to restrict parking.



LOCATION OF ANTICIPATED CURB-LANE PARKING RESTRICTIONS ON ARTERIAL STREETS AND HIGHWAYS IN WAUKESHA COUNTY: 2010



To help ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment, it is recommended that concerned local units of government restrict curb-lane parking on certain arterial streets and highways during peak travel periods. The final recommended plan identified about 117.5 route-miles of arterial streets and highways in Waukesha County when it would be advisable for municipalities to restrict parking.

÷

120

sharing, transit use, bicycle use and pedestrian movement, as well as telecommuting and rescheduling of work time. The plan recommends that the State and local units of government promote and support transportation management associations. Such associations would provide an institutional structure for helping to achieve the goals of the Federal Clean Air Act and the mandated Federally Employee Commute Options program.

• Land Use Measures

The plan recommends that local units of government prepare detailed, site-specific land use plans to facilitate travel by transit, bicycle, and walking movement. The plan recommends the use of zoning, subdivision control, official mapping, and other land use plan implementation devices to promote the development over time of a coordinated mix of land use activities in newly developing areas, higher-density development near transit lines and stations, the orientation of buildings on sites in a manner which facilitates transit use, and integrated bicycle and pedestrian circulation systems linked to transit stops and stations.

In recognition of the relationship between regional patterns of land use development and travel demand, the plan recommends that county and local units of government within the Region employ the regional land use plan as a guide in making decisions regarding urban development. In this regard, the plan envisions that county and local units of government would institute land use policies that would help to direct the type, extent, location, timing, and rate of urban development in accordance with the regional land use plan. Such development policies would provide that urban development occur in planned neighborhood units or in planned major activity centers and only in areas covered by soils suitable for urban use, not subject to such special hazards as flooding, and readily provided with essential urban services including sanitary sewerage, public water supply, and public mass transit.

<u>Transit Systems Management and Service</u> <u>Enhancement Measures</u>: The plan recommends a range of activities to be undertaken by the transit agencies in the Region to enhance the quality of transit services and to facilitate transit use. The plan recommends that transit agencies take steps to improve transit information systems, including real-time vehicle locators to reduce waiting time; to improve facilities for transfers and waiting; to improve bicycle storage at stops and to consider the use of onboard carriers as appropriate; to work with local governments to provide transit only street connections to facilitate through movement of transit vehicles; to use marketing/public education activities to promote the use of transit; to improve transit security; to consider improved vehicle design, including low-floor vehicles, alternative fuels, improved noise and vibration control, and comfort improvement; to investigate methods for increased bus speeds through priority systems and signal preemption; and to promote innovative fare payment systems such as the UPASS, commuter check, smart card, employer treatment of transit as a fringe benefit, and other means to provide alternative ways to pay for transit services.

Public Transit Maintenance and Improvement

The final recommended plan calls for major increases in the levels of rapid and express transit service provided within the Region, as well as increases in the level of local service provided (see Table 254). The plan proposes the development of a true system of rapid and express transit routes integrated with local transit service. Rapid transit routes would operate within all major travel corridors oriented to the Milwaukee CBD and be coordinated with express transit operating over a grid pattern of routes within Milwaukee County. The recommended year 2010 public transit system for the Region is presented in graphic summary form on a one inch equals 8,000 feet scale regional map available from the Regional Planning Commission upon request.

Over all, the plan proposes a 75 percent increase in transit service as measured in vehicle-miles of service provided, from 63,300 in 1991 to 110,600 in 2010. This increase embodies the combined effects of proposed improvements in the frequency of operation of rapid and express transit and the additions and extensions of rapid, express, and local transit routes. The transit recommendations are graphically summarized on Map 122.

<u>Rapid Transit</u>: Like the preliminary recommended plan, the final plan recommends that existing freeway flyer bus service within the Region continue to be operated from the Milwau-

	Base Year	Planned I	ncrement	
Transit Service	1991	Number	Percent	2010
Round-Trip Route Length (miles)				
Rapid Routes	449	911	202.9	1,360
Express Routes	393	27	6.9	420
Local Routes				
Kenosha Urbanized Area	171	39	22.8	210
Milwaukee Urbanized Area	1,112	338	30.4	1,450
Racine Urbanized Area	171	29	17.0	200
Subtotal	1,454	406	27.9	1,860
Total	2,296	1,344	58.5	3,640
Average Weekday Vehicle Requirements ^a				
Peak Period	530	249	47.0	779
Midday Off-Peak Period	285	65	22.8	350
Revenue Vehicle-Miles (average weekday)				
Rapid	3,400	13,500	397.1	16,900
Express	3,300	18,100	548.5	21,400
Local	56,600	15,700	27.7	72,300
Total	63,300	47,300	74.7	110,600
Revenue Vehicle-Hours (average weekday)				
Rapid	170	530	311.8	700
Express	170	930	547.1	1,100
Local	4,880	920	18.9	5,800
Total	5,220	2,380	45.6	7,600

TRANSIT SYSTEM OPERATING CHARACTERISTICS IN THE REGION: 1991 AND 2010 FINAL RECOMMENDED TRANSPORTATION SYSTEM PLAN

^aRepresents only the vehicles required for daily system operation. Excludes vehicles needed as spare or backup.

Source: SEWRPC.

kee CBD to the south to the Cities of Racine and Kenosha, to the southwest to the Village of Mukwonago, and to the west to the Cities of Waukesha and Oconomowoc. The plan also calls for the expansion of such service in the northwest corridor from its current terminus at the Pilgrim Road transit station to the City of West Bend and in the IH 43 north corridor from its current terminus at the Brown Deer Road transit station in River Hills through Cedarburg, Grafton and Saukville to Port Washington. The network of rapid transit routes is shown in red on Map 122.

Under the final plan, 30 rapid transit routes, 27 oriented to the Milwaukee CBD and three to the University of Wisconsin-Milwaukee campus, would be operated over 1,360 round-trip routemiles. The rapid transit system would be served by 73 transit stations, spaced about every three to five miles. In 1991, the base year for the new plan, 13 rapid transit freeway flyer routes were operated over 449 round-trip route-miles within the Region. These routes served 25 transit stations.

The planned rapid transit system would serve intermediate stations and would provide service in both directions during both peak periods. The final plan recommends that the number of rapid transit revenue vehicle-miles of service provided be increased by 13,500 vehicle-miles, from 3,400 in 1991 to 16,900 by 2010. Similarly, the plan recommends that the number of rapid transit revenue vehicle-hours of service be increased by 530 vehicle-hours, from 170 in 1991 to 700 by 2010.

The rapid transit service provided under the recommended plan would operate primarily during peak periods, that is, between the hours



The year 2010 final recommended regional transit system consists of an extensive rapid transit system serving all major Milwaukee central business district travel corridors, an extensive grid system of express transit routes particularly in Milwaukee County, and an expansion of local transit service areas with enhancements to accompanying paratransit services. The plan also incorporates the continuation of local shared-ride taxi service currently provided in certain smaller urban areas of the Region. The regional public transit system envisioned under the final recommended plan would consist of 3,640 round-trip route-miles, which would be about 59 percent greater than that provided in 1991. The planned transit system would provide 110,600 revenue vehicle-miles of service per average weekday, or 75 percent more than in 1991, and 7,600 revenue vehicle-hours of service per average weekday, or 46 percent more than in 1991.

of 6:00 a.m. and 8:30 a.m. and 3:30 p.m. and 6:00 p.m. on weekdays. Midday service over some routes would be provided. No weekend or evening service would be provided. Headways on the rapid transit system would range from five to 30 minutes during peak periods, and 30 to 60 minutes during off-peak periods over those routes provided with service during the midday.

The fares for rapid transit service would remain at 1994 levels, adjusted only for future general price inflation. The freeway flyer rapid transit bus fare for a trip within Milwaukee County would be \$1.50. The fare charged for a trip between points within Milwaukee County and the limits of the Milwaukee urbanized area would be \$2.00. The fare charged for a trip between the Milwaukee CBD and the outer limits of the rapid transit system would be \$3.00.

Like the preliminary plan, the final plan identifies a potential system of almost 49 miles of exclusive busway and HOV facilities, including a 15-mile-long busway/HOV facility in the eastwest corridor between the Cities of Waukesha and Milwaukee (see Map 123). These facilities would be located within, or parallel to, the most heavily congested freeway corridors. The ultimate decision concerning the inclusion of such facilities in the regional transportation system plan would be made following the completion of detailed major investment studies for each corridor.

The plan also recognizes the potential to establish commuter passenger train service as an alternative to freeway flyer or exclusive busway rapid transit service in four major Milwaukeeoriented travel corridors: from Milwaukee through St. Francis, Cudahy, South Milwaukee, Oak Creek, Racine, and Kenosha over CP Rail System (former Chicago, Milwaukee, St. Paul & Pacific Railroad Company) and Chicago & North Western Transportation Company lines; from Milwaukee through Wauwatosa, Elm Grove, Brookfield, Pewaukee, Hartland, Delafield, and Nashotah to Oconomowoc over CP Rail System (former Chicago, Milwaukee, St. Paul & Pacific Railroad Company) lines; from Milwaukee through Germantown and Jackson to West Bend over the CP Rail System (former Chicago, Milwaukee, St. Paul & Pacific Railroad Company), Chicago & North Western Transportation Company, and Wisconsin Central Transportation Corporation (former Chicago & North Western Transportation Company) lines; and from Milwaukee through Brown Deer, Cedarburg and Grafton to Saukville over CP Rail System and Wisconsin Central Transportation Corporation (former Chicago, Milwaukee, St. Paul & Pacific Railroad Company) lines. The plan also recognizes the potential to provide commuter passenger train service in two Chicago-oriented corridors: from the Village of Walworth through Fox Lake, Illinois, to the Chicago area over Wisconsin and Southern Railroad Company and Metra railway lines (former Chicago, Milwaukee, St. Paul & Pacific Railroad Company); and from the City of Burlington through Silver Lake and Antioch, Illinois, to Chicago over Wisconsin Central Transportation Company lines (former Soo Line Railroad Company) (see Map 122). Detailed corridor studies would be required for these proposals.

Express Transit

Like the preliminary plan, the final plan recommends that 12 regular express transit bus routes be provided in a grid pattern, largely within Milwaukee County. In 1991, seven regular express transit routes were in operation. Within the Milwaukee urbanized area, the express transit would be provided in major travel corridors to connect major activity centers to the Milwaukee CBD. One express transit route would also connect the CBDs of the Cities of Racine and Kenosha. The planned express routes are shown in blue on Map 122.

Like the preliminary plan, the final plan identifies a potential 15-mile-long light-rail express transit line connecting the City of Glendale, the University of Wisconsin-Milwaukee, the Milwaukee CBD, and the Milwaukee Regional Medical Center and Research Park (see Map 123). These facilities would serve heavily traveled corridors in Milwaukee County. The ultimate decision concerning the inclusion of such facilities in the regional transportation system plan would be made following the completion of the major investment study now underway for the eastwest travel corridor.

Three other travel corridors are identified in the plan as having potential for light-rail express transit service. These include the corridors that extend from the Milwaukee CBD to the General Mitchell International Airport, along the 27th Street crosstown corridor from W. Silver Spring Drive to the Southridge Shopping Center, and from the Milwaukee CBD to the Northridge

EXTENT OF POTENTIAL BUSWAY AND LIGHT-RAIL/EXPRESS BUS GUIDEWAY FACILITIES IN THE MILWAUKEE AREA: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN



as alternatives to motor bus transit service over arterial highway lanes. Consideration of such fixed-guideway transit service facilities would be initiated as part of Federally required major investment studies for each of the identified corridors. The busway facility, which extends along the IH 94 corridor from the City of Milwaukee to the STH 164 interchange in Waukesha County, shown on the accompanying map, and the light-rail facility, which extends from the Village of Glendale through the central business district of Milwaukee to the Milwaukee County Institutions Grounds, have been explicitly included in the final recommended plan. It is recognized that the implementation of these fixed-guideway transit facilities depends upon the ultimate outcome of the corridor study currently being conducted by the Wisconsin Department of Transportation. Upon completion of that study, the local units of government concerned, the Wisconsin Department of Transportation, and the Regional Planning Commission would have to affirm the study findings and, if necessary, amend the regional transportation system plan.

Shopping Center. The feasibility of effectively providing light-rail service in these corridors would be determined through Federally required major investment studies. The potential lightrail facilities are envisioned to operate with preferential treatment over reserved street lanes within street rights-of-way or over such exclusive rights-of-way as along railway or former electric interurban railway rights-of-way. Light-rail operating characteristics may be expected to vary depending upon the type of right-of-way and adjacent development and attendant station spacing, and may approach rapid transit operating characteristics.

Under the plan, the extent of express transit service would be significantly expanded through the provision of a grid of express routes. The frequency of operation of transit vehicles over the express routes would also be significantly increased. As shown in Table 254, the number of vehicle-miles provided on an average weekday would increase by 18,100 vehicle-miles, from about 3,300 in 1991 to about 21,400. Similarly, vehicle-hours of express service provided on an average weekday would increase by 930 vehiclehours, from 170 in 1991 to 1,100 in 2010.

Express transit service would be provided on weekdays from 6:00 a.m. to 6:00 p.m. on all routes and during weekday evenings and weekends on some routes. Peak-period headways would range from five to 15 minutes in the Milwaukee urbanized area to 30 minutes on the route connecting Racine and Kenosha. Off-peak headways would range from 10 to 30 minutes within the Milwaukee urbanized area to 60 minutes on the Racine-Kenosha route. Express transit fares would remain at 1994 levels, that is, \$1.25 in Milwaukee County and \$0.75 on the Racine-Kenosha route. It is assumed that these fares would increase with general price inflation over the plan design period.

Local Transit: The level of local service envisioned in the plan consists of buses operating over arterial and collector streets with frequent stops for passenger boarding and alighting. Local fixed-route service would continue to be provided and would be extended within Milwaukee County and the Cities of Waukesha, Racine, and Kenosha and their immediate environs. The plan recommends that the local transit operators undertake detailed implementation studies to identify the best way to provide for the service extensions, holding open the possibility of transit center-oriented local route systems to replace, in some areas, the grid route systems. As shown on Map 122, these areas of expanded service are generally located in southern and northern Milwaukee County and in the most heavily developed portions of Waukesha County. Under the final plan, local transit service would operate over 1,860 round-trip route-miles within the Region, representing an increase of 406 route-miles, or 28 percent, over the approximately 1,454 route-miles provided in 1991.

The frequency of local transit service would be substantially improved over 1991 levels. Within Milwaukee County peak-period headways on the major routes in the area south of Silver Spring Drive, east of 76th Street, and north of Layton Avenue would be improved from 10 to 40 minutes to 10 minutes. Peak-period headways in the Racine and Kenosha urban areas would be improved from 20 to 30 minutes to 15 to 30 minutes. Peak-period headways in the Waukesha urban area would be improved so that all routes would operate at 30 minute headways.

Under the final plan, local transit fares would remain at 1994 levels, adjusted only for the effects of general price inflation. Accordingly, fares within Milwaukee County would be \$1.25; within Racine, \$0.60; and within Waukesha and Kenosha, \$0.75. Transit fares would increase only with general price inflation. The final recommended plan also recognizes the need to provide local transit service in the smaller urban communities of the Region, particularly through shared-ride taxi service. Explicitly incorporated into the plan are the continuation of the sharedride taxi services provided in the Cities of Hartford, Port Washington, West Bend, and Whitewater.

The recommended plan also includes a paratransit service component which is consistent with the Americans with Disabilities Act (ADA) of 1990. The plan assumes that all transit vehicles that provide conventional fixed-route transit service would be accessible to persons with disabilities, including those persons using wheelchairs. This assumption is reflected in the capital cost estimate for transit vehicle fleet replacement and expansion under the recommended plan. The plan also ensures that all public entities operating fixed-route transit systems will provide comparable paratransit service to those disabled persons within local transit service areas who are unable to use fixed-route transit services. Accordingly, the complementary paratransit services currently provided within the Region would continue to be operated and expanded consistent with the planned expansion of local transit service areas within the Kenosha, Milwaukee, and Racine urbanized areas.

Like existing complementary paratransit services provided within the Region, the planned paratransit services would meet Federally specified ADA eligibility and service requirements. The complementary paratransit services would serve any person with a permanent or temporary disability who is unable independently to board, ride, or disembark from an accessible vehicle used to provide fixed-route transit service; who is capable of using an accessible vehicle, but one is not available for the desired trip; or who is unable to travel to or from the boarding or disembarking location of the fixed-route transit service. Within a given area, the planned paratransit service would be available during the same hours and on the same days as the fixedroute transit service, would be provided to eligible persons on a "next day" trip reservation basis, would not limit service to eligible persons based on restrictions or priorities relative to trip purpose, and would not be operated under capacity constraints which might limit the ability of eligible persons to receive service for a particular trip. The paratransit service fares assumed under the recommended plan would range from about \$1.20 to \$2.50 per one-way trip.

Relationship between the East-West Corridor Transit Study and the Recommended Plan

The final recommended plan provisionally includes an east-west travel corridor busway and light-rail express transit lines emanating from the Milwaukee CBD west to the Milwaukee Regional Medical Center and Research Park and northeast to the City of Glendale through the University of Wisconsin-Milwaukee campus. It is recognized that, when the East-West Corridor Transit Study is completed, it will be necessary for the Regional Planning Commission to appropriately affirm or amend the regional transportation system plan before any projects that would implement these potential facilities could be placed into the regional transportation improvement program.

Arterial Street and Highway System Maintenance and Improvement

The planned additions and changes to the arterial street and highway system in the Region

between 1991 and 2010 are summarized in Table 255. The proposed arterial street and highway system is shown on Map 124. In 1991, the arterial street and highway system in the Region consisted of about 3,274 route-miles of facilities. Under the final recommended plan, by the year 2010 the arterial system would be increased by about 333 route-miles and would total 3.607 route-miles and 10.303 lane-miles. The additional arterial mileage reflects primarily the conversion of existing nonarterial facilities to arterial status and function as urban development proceeds within the Region. About 131 route-miles, or 4 percent of the proposed additional arterial mileage, would be added through new construction.

The recommended year 2010 arterial street and highway system for the Region is presented in graphic summary form on a one inch equals 8,000 feet scale regional map available from the Regional Planning Commission upon request. Each link in the arterial street and highway system is identified on this map, together with the proposed number of through travel lanes to be provided on that link. Freeways are identified as having either four, six, or eight lanes. Standard surface arterials are identified as having either two, four, or six lanes.

There are no typical cross-sections identified on the aforementioned planned arterial system map. Rather, only the number of lanes recommended to be provided on each link in the arterial network is indicated. The number of lanes identified refers to through travel lanes, that is, those lanes that would carry traffic directly through intersections. Thus, the number does not include any auxiliary traffic lanes to be provided for left- and right-turning movements, for vehicle parking, or for use by distressed vehicles. It was assumed in the regional systems analysis that such right- and left-turn lanes will be provided where the volumes of turning vehicles would adversely affect the movement of vehicles through the intersection. The provision of turn lanes would, therefore, follow a design investigation in connection with a given improvement project. In addition to determining whether or not right- and/or left-turn lanes should be provided at intersections, the design investigation should determine whether or not a given arterial street improvement should be made using a divided or an undivided roadway cross-section. Thus, the precise cross-section to

Table 255

ARTERIAL STREET AND HIGHWAY SYSTEM PRESERVATION, IMPROVEMENT, AND EXPANSION BY ARTERIAL FACILITY TYPE BY COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

County	System Preservation (miles)	System Improvement (miles)	System Expansion (miles)	Total Miles
Kenosha				
Freeway	12.0	0.0	0.0	12.0
Standard Arterial	288.9	44.8	9.4	343.1
Subtotal	300.9	44.8	9.4	355.1
Milwaukee			· · ·	
Freeway	69.2	0.0	0.0	69.2
Standard Arterial	666.3	50.2	11.3	727.8
Subtotal	735.5	50.2	11.3	797.0
Ozaukee				
Freeway	27.4	0.0	0.0	27.4
Standard Arterial	222.6	48.5	5.9	277.0
Subtotal	250.0	48.5	5.9	304.4
Racine				
Freeway	12.0	0.0	0.0	12.0
Standard Arterial	330.5	62.2	19.1	411.8
Subtotal	342.5	62.2	19.1	423.8
Walworth				
Freeway	50.0	0.0	16.7	66.7
Standard Arterial	359.5	38.2	19.7	417.4
Subtotal	409.5	38.2	36.4	484.1
Washington				
Freeway	21.6	21.1	0.0	42.7
Standard Arterial	354.1	48.9	22.6	425.6
Subtotal	375.7	70.0	22.6	468.3
Waukesha				
Freeway	57.6	2.0	5.7	65.3
Standard Arterial	556.3	132.1	20.5	710.0
Subtotal	613.9	134.1	26.2	774.2
Region				
Freeway	249.8	23.1	22.4	295.3
Standard Arterial	2,778.2	424.9	108.5	3,311.6
Total	3,028.0	448.0	130.9	3,606.9

Source: SEWRPC.

be selected for a given improvement project should be determined by the State, county, and local implementing agencies following appropriate design study.

The proposed arterial street and highway improvements are shown in summary form for each county on Maps 125 through 131 and are set forth by county and by arterial facility type in Table 255. The final recommended plan includes arterial street and highway system expansion, improvement and preservation recommendations.

System Expansion: Constructing New Facilities: System expansion includes all projects which would significantly increase the capacity of the existing system through construction of new facilities. The plan would provide for the construction of 131 route-miles and 337 lane-miles of



The year 2010 final recommended regional arterial street and highway system consists of 3,607 route-miles and 10,303 lane-miles of facilities. The recommended plan would provide for the construction of 131 route-miles and 337 lane-miles of new facilities within the Region and the widening or other improvement of 448 route-miles, adding 854 lane-miles to the 1,023 lane-miles already provided on those routes. The plan also calls for pavement resurfacing and bridge and interchange restoration and reconfiguration work necessary to maintain and appropriately modernize the remaining 3,028 route-miles and 8,089 lane-miles of planned arterial facilities.



RAGINE 00 KENOSHA 4 RACINE CO. KENOSHA CO. WHEATLAND 1 KENNEDY MEMORIAL а 45TH ST BOTH ST 4 4 BRIGHTO 4 PADDOC 4157.51 CLARE LORLE SALEM BR 6 6 6 6 6 28/2010/07 HOOKER 121 4 ANDALL POWERS LAKE (0)NESNEDICT S LAKE HENER 4 4 4 4 4 100 H 122ND S DENET WISCONSIN 4 SALE 17 TOL KENOSHA CO RANIDA MC HENRY 10.21 LAKE CO

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN KENOSHA COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

LEGEND

- ARTERIAL STREET OR HIGHWAY
- NEW

 WIDENING AND /OR OTHER IMPROVEMENT TO PROVIDE SIGNIFICANT ADDITIONAL CAPACITY

 RESURFACING OR RECONSTRUCTION TO PROVIDE ESSENTIALLY THE SAME CAPACITY
 - 4 NUMBER OF TRAFFIC LANES FOR NEW OR WIDENED AND / OR IMPROVED FACILITY (2 LANES WHERE UNNUMBERED)

anaPHIC SCALE 2 HILES 0 4000 B000 12000FEET

Under the final recommended regional transportation system plan, the arterial street and highway system in Kenosha County would be expanded by 37 miles, or 12 percent, from 318 miles in 1991 to 355 miles in the year 2010. The increase in arterial mileage would come about through the construction of nine miles of facilities and through the conversion of 28 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of nine miles of new arterial facilities, for the widening of 45 miles, and for the preservation of 301 miles of facilities within the County.



Under the final recommended regional transportation system plan, the arterial street and highway system in Milwaukee County would be expanded by 22 miles, or 3 percent, from 775 miles in 1991 to 797 miles in the year 2010. The increase in arterial mileage would come about through the construction of 11 miles of new facilities and through the conversion of 11 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of 11 miles of new arterial facilities, for the widening of 50 miles, and for the preservation of 736 miles of facilities within the County.



FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN OZAUKEE COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN





FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN RACINE COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

Under the final recommended regional transportation system plan, the arterial street and highway system in Racine County would be expanded by 76 miles, or 22 percent, from 348 miles in 1991 to 424 miles in the year 2010. The increase in arterial mileage would come about through the construction of 19 miles of new facilities and through the conversion of 57 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of 19 miles of new arterial facilities, for the widening of 62 miles, and for the preservation of 343 miles of facilities within the County.



FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WALWORTH COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

LEGEND

ARTERIAL STREET OR HIGHWAY

- NEW
- WIDENING AND/OR OTHER IMPROVEMENT TO PROVIDE SIGNIFICANT ADDITIONAL CAPACITY
- RESURFACING OR RECONSTRUCTION TO PROVIDE ESSENTIALLY THE SAME CAPACITY
- NUMBER OF TRAFFIC LANES FOR NEW OR WIDENED AND / OR IMPROVED FACILITY (2 LANES WHERE UNNUMBERED) 4

Under the final recommended regional transportation system plan, the arterial street and highway system in Walworth County would be expanded by 55 miles, or 13 percent, from 429 miles in 1991 to 484 miles in the year 2010. The increase in arterial mileage would come about through the construction of 36 miles of new facilities and through the conversion of 19 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of 36 miles of new arterial facilities, for the widening of 38 miles, and for the preservation of 410 miles of facilities within the County.

Concession of the local division of the loca



FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WASHINGTON COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

Under the final recommended regional transportation system plan, the arterial street and highway system in Washington County would be expanded by 69 miles, or 17 percent, from 399 miles in 1991 to 468 miles in the year 2010. The increase in arterial mileage would come about through the construction of 23 miles of new facilities and through the conversion of 46 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of 23 miles of new arterial facilities, for the widening of 70 miles, and for the preservation of 375 miles of facilities within the County.



FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WAUKESHA COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

LEGEND

ARTERIAL STREET OR HIGHWAY

- NEW
 WIDENING AND /OR OTHER IMPROVEMENT TO
 PROVIDE SIGNIFICANT ADDITIONAL CAPACITY
 RESURFACING OR RECONSTRUCTION TO PROVIDE
 ESSENTIALLY THE SAME CAPACITY
 NUMBER OF TRAFFIC LANES FOR NEW OR WIDEN
 - 4 NUMBER OF TRAFFIC LANES FOR NEW OR WIDENED AND/OR IMPROVED FACILITY (2 LANES WHERE UNNUMBERED)

Under the final recommended regional transportation system plan, the arterial street and highway system in Waukesha County would be expanded by 58 miles, or 8 percent, from 716 miles in 1991 to 774 miles in the year 2010. The increase in arterial mileage would come about through the construction of 26 miles of new facilities and through the conversion of 32 miles of previously nonarterial facilities to arterial status in order to accommodate expected traffic volumes and to provide the arterial spacing necessary to properly structure planned urban development. The plan would provide for the construction of 26 new miles of arterial facilities, for the widening of 134 miles, and for the preservation of 614 miles of facilities within the County.

RACINE

new arterial facilities. These include such longplanned facilities as the Lake Parkway south from the Hoan Bridge to E. Layton Avenue, the STH 16 freeway bypass of Oconomowoc, the completion of the Waukesha bypass, the STH 36 bypass of Burlington, and the completion of the USH 12 freeway in Walworth County. Excluded from the plan are such previously planned new facilities as the STH 83 bypass of the Village of Chenequa in Waukesha County and the Lake Arterial through southern Milwaukee County and eastern Racine and Kenosha Counties. In all, proposed new arterial street and highway facilities would represent about 4 percent of the total planned arterial route-miles and 3 percent of the total planned lane-miles.

System Improvement: Widening Existing Facilities: System improvement includes all projects which would significantly increase the capacity of the existing system through street widening or relocation. Under the final plan, a total of 448 route-miles of facilities would be widened or otherwise improved, representing a total of 1,877 lane-miles after widening. Proposed improvements would include the widening of STH 36 in Milwaukee, Waukesha, and Racine Counties; Pewaukee Road (CTH J) in Washington and Waukesha Counties; Cleveland Avenue (CTH D) and Racine Avenue (CTH Y) in Waukesha County; STH 31 and CTH Y in Kenosha and Racine Counties; Northwestern Avenue (CTH K) and Spring Street (CTH C) in Racine County; STH 57 and Port Washington Road (CTH W) in Ozaukee County; STH 33 in Ozaukee and Washington Counties; and Rawson Avenue (CTH BB) in Milwaukee County; and the completion of the widening of STH 50 in Kenosha and Walworth Counties. The system improvement activities would comprise about 12 percent of the total planned route system and 18 percent of the total planned lane-miles.

System Preservation: Maintaining Existing <u>Facilities</u>: System preservation includes all arterial improvement projects required to maintain the structural adequacy and serviceability of the existing arterial system without significantly increasing the capacity of that system. This would include all projects classified as resurfacing and reconstruction for the same capacity. The final plan proposes system preservation activities for about 3,028 route-miles and 8,089 lane-miles, representing about 84 percent of the total planned arterial system and about 79 percent of the total lane-miles.

System Preservation: The Milwaukee Area Freeway System: The plan recommends that the Milwaukee area freeway system be reconstructed and modernized to current freeway design standards. About \$1.5 billion has been included in the plan for this purpose. It should be noted that any project to reconstruct the east-west freeway must be coordinated with decisionmaking under the East-West Corridor Transit Study being conducted by the Wisconsin Department of Transportation. As noted above, a busway facility in the east-west corridor is being considered under that study. Should the eastwest corridor study ultimately conclude that a busway be provided, then the design of projects to rebuild the east-west freeway must take that decision into account.

It is accordingly recommended that the Wisconsin Department of Transportation, working closely with Milwaukee and Waukesha Counties, the City of Milwaukee, other local units of government in the corridor, and the Commission bring the East-West Corridor Transit Study to a conclusion in 1995. This will make it possible for a final decision to be made on the precise way in which the east-west freeway should be reconstructed. This may require adaptation of eastwest freeway projects currently included in the regional transportation improvement program.

Plan Performance and Costs

All of the data and analyses necessary properly to relate the final recommended plan to the objectives and standards set forth in Chapter VIII of this report were developed so that comparisons could be made between the final recommended regional transportation plan and all of the alternative transportation plans considered and previously presented in this report. Summary plan performance data are presented in Table 256. Summary cost and revenue data are presented in Table 257.²

²The costs and revenues set forth for the final recommended plan differ significantly from the costs and revenues reported earlier in this chapter for the preliminary recommended plan. The differences may be accounted for by the following three factors: 1) changes to the recommendations for improvement of the regional arterial street and highway and transit systems made by the Advisory Committee in response to the results of the public participation process,

(Footnote 2 continued on page 544)

Under the final recommended plan, the number of internal person trips generated within the Region on an average weekday may be expected to increase from 5.54 million in 1991 to about 6.10 million in the year 2010, or by about 10 percent. The number of transit trips made on an average weekday may be expected to increase from about 172,200 in 1991 to about 196,400 in the year 2010, or by about 14 percent, assuming the transit plan recommendations are implemented. Despite this increase in daily transit tripmaking, the proportion of total internal person trips made by transit would remain at about 3 percent of the total person trips generated within the Region on an average weekday through the plan design period.

The number of vehicle-miles of travel within the Region on an average weekday is expected to increase by about 28 percent, from about 33.07 million in 1991 to about 42.42 million in 2010. Of this total, about 15.40 million vehicle-miles of

(Footnote 2 continued from page 543)

2) more refined cost estimates attendant, in particular, to projects associated with the State trunk highway system submitted as a result of careful review of the preliminary plan by the Wisconsin Department of Transportation, and 3) the selection of a different revenue accounting procedure relative to certain sources of Federal and State revenues for both highway and transit plan implementation projects. With respect to the latter, for example, significant amounts of potential Federal and State revenue associated with implementation of the provisionally recommended buswav and light-rail facilities in the east-west travel corridor, amounting to about \$45 million annually, were in the financial analyses related to the preliminary plan counted as revenue prior to the calculation of the anticipated funding shortfall. Those same Federal and State revenues were factored into the costrevenue analyses associated with the final recommended plan after calculation of the anticipated shortfall. This change in technique, which had the practical effect of significantly increasing the anticipated revenue shortfall. became necessary in order to correlate the final plan cost-revenue analyses with similar costrevenue analyses associated with the State Intermodal Transportation Plan presented simultaneously by the Wisconsin Department of Transportation as part of the TRANSLINKS 21 statewide planning process.

travel, or about 36 percent, are expected to be made on freeways, which would comprise about 8 percent of the total arterial system.

Arterial street and highway congestion, evidenced by the number of arterial miles operating over design capacity, is expected to decrease from about 385 miles, or about 12 percent of the total system in 1991, to about 165 miles, or about 5 percent, in 2010. The number of miles anticipated to operate as moderately congested would decrease from 106 to 82, as severely congested would decrease from 217 to 47, and as extremely congested would decrease from 62 to 36. The number of arterial miles operating at design capacity, however, is expected to increase from about 158 miles, or about 5 percent of the total system in 1991, to about 218 miles, or about 6 percent of the total system in 2010. The location of the facilities expected to operate under, at, and over design capacity are shown on Map 132.

Annual motor fuel consumption in the year 2010 is expected to approximate 444 million gallons, or about 85 million gallons more than the amount consumed in the Region in 1991 and 24 million gallons fewer than the amount consumed under the no-build plan.

The average annual public cost of carrying out the recommended plan, including the construction of new facilities and the operation and maintenance of the arterial street and highway and transit systems, is estimated at nearly \$521.9 million. Base-year existing revenue trends, the anticipated average annual public revenues, excluding transit fare box revenues, is estimated at \$299.9 million. Thus, the difference between anticipated costs and expected revenues is \$222.0 million per year over the plan design period. The equivalent of an approximate \$0.26 per gallon increase in the motor fuel tax would be necessary to fund this \$222.0 million annual shortfall.

From an economic analysis viewpoint, the investment proposed in the final recommended plan in the Region's arterial street and highway and transit systems may be expected to generate benefits in excess of the costs incurred. One measure of this is the benefit-cost ratio of the plan. That ratio is derived by computing the difference in transit and highway user costs between the no-build alternative and the final recommended plan and dividing that number by the difference in transportation system costs

Table 256

SUMMARY OF TRANSPORTATION PERFORMANCE CHARACTERISTICS OF THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN: 2010

	Performance Characteristic			2010	
Category	Specific Measure	Base Year 1991	No-Build Alternative	Preliminary Recommended Plan	Final Recommended Plan
Travel	Automobile and light truck availability (millions)	1.13	1.31	1.30	1.30
	Average weekday internal person-trips (millions)	5.54	6.12	6.10	6.10
	Average weekday internal vehicle-trips (millions)	4.58	5.33	5.27	5.27
	Average weekday vehicle-miles of travel (millions) ^a	33.07	44.52	42.70	42.42
	Average weekday transit ridership	172,200	168,000	196,600	196,400
	Average weekday relative distribution of trips by mode of travel (percent) ^b Auto driver Auto passenger Transit passenger School bus passenger	73.3 19.5 3.1 4.1	77.6 15.0 2.7 4.7	76.8 15.3 3.2 4.7	76.8 15.3 3.2 4.7
	Proportion of trips made by transit within Milwaukee County (percent)	3.8	3.3	3.8	3.8
	Proportion of trips made by transit to Milwaukee central business district (percent)	13	10	12	12
	Average weekday proportion of passenger- miles of travel made on transit (percent)	1.1	1.1	1.5	1.5
Traffic Congestion	Amount and proportion of arterial street and highway system over design capacity Moderately congested (V/C ratio 1.01 to 1.10) Severely congested {V/C ratio 1.11 to 1.30} Extremely congested {V/C ratio over 1.31}	106 miles 3.2 percent 217 miles 6.6 percent 62 miles 2.0 percent	31 miles 0.9 percent 538 miles 15.5 percent 143 miles 4.1 percent	83 miles 2.3 percent 48 miles 1.3 percent 36 miles 1.0 percent	82 miles 2.3 percent 47 miles 1.3 percent 36 miles 1.0 percent
	Total	385 miles 11.8 percent	712 miles 20.5 percent	167 miles 4.6 percent	165 miles 4.6 percent
Energy Use	Annual motor fuel consumption (millions of gallons) By auto and truck By motor bus	353.2 5.7	462.3 5.8	444.2 9.2	434.9 9.4
	Total	358.9	468.1	453.4	444.3
Air Quality	Air pollutant emission from vehicles on the arterial street and highway system (tons per average weekday) Volatile organic compounds Nitrogen oxides Carbon monoxide	119.2 129.8 815.1	29.8 81.4 248.4	28.6 79.1 237.8	27.7 78.9 226.0

^aWithin Walworth County, the number of vehicle-miles of travel may be expected to increase from 1.91 million in 1991 to 2.90 in 2010, or by 52 percent, under the final recommended plan.

^bThe average vehicle occupancy for work trips within the Region may be expected to decline from 1.06 in 1991 to 1.03 under the final recommended plan.

Table 257

AVERAGE ANNUAL COSTS AND REVENUES ASSOCIATED WITH THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 1995 THROUGH 2010

Cost or Revenue Item	Final Recommended Plan: 2010
Transportation System Cost (average annual 1995-2010 expressed as millions of dollars) Arterial Street and Highway System Capital	277.3
Subtotal	343.4
Transit System Capital	76.8 101.7 ^a
Subtotal	178.5
Total	521.9
Transportation System Revenues (average annual 1995-2010 expressed as millions of dollars) Highway Capital Federal State Local	78.3 67.8 13.3
Subtotal	159.4
Highway Operating State	30.0 27.6
Subtotal	57.6
Transit Capital Federal	12.2 2.5
Subtotal	14.7
Transit Operating Federal State Local	6.0 45.1 17.1
Subtotal	68.2
Total	299.9
Cost-Revenue Comparison Average Annual Difference between Cost and Revenue (millions of dollars) Motor Fuel Tax Required to Fund Shortfall (cents per gallon)	222.0 26.1

^aThis figure represents the anticipated operating deficit. The total average annual cost of operating the recommended transit system is estimated at \$139.4 million. Farebox revenues are estimated at \$37.7 million.

Source: SEWRPC.

between the no-build alternative and the final recommended plan. The transit and highway user cost consists of accident, travel time, and vehicle operation costs, including motor fuel, maintenance, and vehicle depreciation. This economic analysis resulted in a benefit-cost ratio attendant to the recommended plan of 1.55. This means that for every dollar invested in implementing the plan, the transportation system users, on the average, could be expected to enjoy \$1.55 in benefits in terms of reduced accidents, travel time, and vehicle operational costs.
Map 132

ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN



Under the final recommended plan, the level of traffic congestion may be expected to be substantially below that which would occur under the no-build alternative plan. By the year 2010, only about 5 percent of the planned 3,607-mile arterial system, or 165 miles, would operate over design capacity. About 82 miles, or over 2 percent, of planned arterial mileage would be moderately congested; 47 miles, or about 1 percent, would be severely congested; and 36 miles, or about 1 percent, would be extremely congested. While the transportation development proposals included in the final recommended plan serve to reduce traffic congestion throughout the entire Region, the Milwaukee area freeway system may be expected to carry traffic volumes exceeding its design capacity and to operate with congested conditions through the year 2010.

ANALYSIS OF SENSITIVITY OF THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN TO ALTERNATIVE FUTURE REGIONAL GROWTH SCENARIOS

The following section of this chapter describes the results of an analysis of the sensitivity of the final recommended plan to alternative future land use patterns and related levels of population, households, and employment within the Region. The analysis was requested by the Technical Coordinating and Advisory Committee on Regional Transportation System Planning. The purpose of the analysis was to investigate the differences in the amount of regional travel that could be expected to attend future land use patterns that were significantly different from the pattern envisioned in the adopted land use plan.

It should again be noted that the recommended regional transportation system plan was based upon the adopted regional land use plan. This means that the transportation system plan was designed to accommodate and serve the travel demand expected to result from a centralized land use pattern and modest forecast increases in population, households, and economic activity through the year 2010. Accordingly, throughout this chapter, the description of the performance of the recommended transportation system plan has assumed full implementation of the adopted regional land use plan.

A detailed description of the adopted land use plan as well as the alternative forecasts of growth considered is provided in Chapter VI of this report and in SEWRPC Planning Report No. 40, A Regional Land Use Plan for Southeastern Wisconsin-2010, 1992. The adopted regional land use plan proposes to reverse past trends within the Region toward the decentralization of urban development; thus, use of the term "centralized" to describe the plan. For the purpose of testing the sensitivity of the recommended transportation system plan, three other future land use patterns were considered. These alternatives were: an intermediate-growth existing trends, a high-growth centralized, and a high-growth decentralized land use pattern.

These alternative future land use patterns differ from the adopted regional land use plan in fundamental ways. The existing trends alternative was designed at the request of the Advisory Committee to investigate the potential impacts upon travel patterns and the recommended transportation system plan of a land use pattern that could be expected to result under intermediate population, household, and employment growth levels and under a continuation of existing trends toward the areawide dispersion of urban development. This alternative is described in Appendix A of this report. Both of the high-growth scenarios were considered by the Commission as alternatives during the preparation of the adopted regional land use plan. These alternative scenarios are described in the aforereferenced SEWRPC Planning Report No. 40. The high-growth centralized plan is similar to the adopted plan, except that it was designed to accommodate higher levels of future population, households, and economic activity than the adopted plan. The high-growth decentralized plan was also based upon higher regional growth levels. It envisioned, however, that much of the attendant land use development would occur in and around outlying urban communities of the Region rather than in and around the older central cities of Kenosha, Milwaukee, and Racine. It should be noted that this alternative was not a "sprawl plan" in that the growth in the outlying urban communities would be located in areas that could be served by sewer, water, and transit.

Table 258 compares the demographic and economic characteristic of each of the alternative land use plans. As shown, the high-growth plans would seek to accommodate over 20 percent more population, 9 percent more households, and 14 percent more employment than the two intermediate-growth plans. The greater levels of automobile availability under the high-growth plans are attributable to the higher population and household levels. Table 258 indicates that the demographic and economic characteristics of the existing trends land use pattern would not differ significantly from those of the adopted regional land use plan. Indeed, the table indicates that the minor differences between the two land use patterns are related to automobile availability; this difference is attributable to observed differences in the lifestyles of people residing in urban and in suburban and rural areas of the Region.

Table 259 compares the travel characteristics related to each future land use scenario assum-

Table 258

DEMOGRAPHIC AND ECONOMIC CHARACTERISTICS: 1970, 1980, 1990, AND 2010 ADOPTED REGIONAL LAND USE PLAN AND ALTERNATIVE FUTURE SCENARIOS

				2010			
				Intermediate-Growth Scenario		High-Growth Scenario	
Characteristics	1970	1980	1990	Adopted Regional Plan	Existing Trends Land Use Alternative	Decentralized Land Use Plan	Centralized Land Use Plan
Demographic							
Households	536,500	628,000	676,100	774,300	772,600	846,400	856,300
Total Population	1,756,100	1,764,800	1,810,400	1,911,000	1,912,700	2,316,100	2,316,100
Household Size (household population/households)	3.2	2.8	2.6	2.4	2.4	2.7	2.7
Percent Population in Central County	60.0	54.7	53.0	48.9	46.7	39.8	47.4
Households per Net Residential Acre	3.8	3.5	3.5	3.5	3.1	3.3	3.5
Vehicles (includes personal-use trucks)	648,300	886,200	1,114,000	1,307,200	1,311,600	1,597,500	1,566,100
Average Household Income (1990 dollars)	\$40,400	\$41,000	\$38,500	\$43,300	\$43,300	\$51,900	\$51,900
Average Vehicles per Household	1.2	1.4	1.6	1.7	1.7	1.9	1.8
Economic							
Employment	753,700	884,200	990,300	1,102,100	1,102,100	1,251,600	1,251,600
Percent Employment in Central County	67.3	61.3	58.4	55.6	51.9	49.7	52.9

Source: SEWRPC.

Table 259

TRAVEL CHARACTERISTICS IN THE REGION: 1970, 1980, 1990, AND 2010 ADOPTED REGIONAL LAND USE PLAN AND ALTERNATIVE FUTURE SCENARIOS

				2010			
				Intermediate-Growth Scenario		High-Growth Scenario	
Characteristics	1970 ^a	1980	1990 ^b	Adopted Regional Plan	Existing Trends Land Use Alternative	Decentralized Land Use Plan	Centralized Land Use Plan ^C
Auto Occupancy, Work	1.17		1.06	1.03	1.03	1.03	1.04
Auto Occupancy, Nonwork	1.48		1.31	1.22	1.22	1.23	1.23
Average Trip Length, Work (miles)	7.5		9.1	8.9	9.0	9.5	9.0
Average Trip Length, Nonwork (miles)	4.7		5.2	5.9	6.0	6.2	6.1
Average Weekday Trips ^d	4.3 million		5.3 million	5.8 million	5.8 million	6.7 million	6.7 million
Transit Trips	184,200	210,000	172,200	196,600	186,000	169,300	193,500
Percent Transit Use	4.3		3.2	3.4	3.2	2.5	2.9
Vehicle-Miles of Travel (VMT)	20.1 million	24.9 million	33.1 million	42.7 million	42.8 million	48.5 million	47.1 million
Average Annual VMT Growth Rate (percent) ^e	•••			1.4	1.4	1.9	1.9

^aTravel data are presented for the year 1972.

^bTravel data are presented for the year 1991.

^C Two modifications of this alternative were also analyzed at the request of the Wisconsin Department of Natural Resources, both assuming household sizes and incomes similar to the intermediate-growth future and one assuming high regional growth to the year 1999 and intermediate growth from the year 2001 to the year 2010. The latter future was determined to result in 44.4 million vehicle-miles of highway travel in the year 2010 at an average annual rate of 1.5 percent, and the other future in 46.1 vehicle-miles of highway travel in the year 2010 at an annual growth rate of 1.8 percent.

^dHome-based work, home-based shop, home-based other, nonhome-based trips, and school trips; does not include trips by school bus.

^eGrowth rate is calculated on the basis of the 19-year period, 1991 to 2010. The growth rates do not incorporate any reduction in vehicle-miles of travel which may result from the implementation of the federally mandated employee commute options program. The potential maximum impact of the employee commute options program would be a reduction of 0.1 percent in the annual percentage growth rate in vehicle-miles of travel under any of the alternative futures. This potential maximum reduction assumes full implementation by all affected employers and that such implementation is achieved by employees modifying their travel in ways which eliminate auto travel such as using transit, carpools, vanpools, bicycle, or walking and not modifications which would not eliminate auto travel such as alternative fueled vehicles, shifted work start times, and employer shuttles from remote-park-ride lot.

Source: SEWRPC.

ing that the recommended transportation system plan has been fully implemented. The data set forth in the table indicate that the most significant differences exist between the two intermediate-growth plans and the two highgrowth plans with respect to three measures: total weekday trips, weekday transit trips, and vehicle-miles of travel. The high-growth plans, with significantly greater numbers of resident population, households, and employment, have, as would be expected, higher tripmaking levels than the intermediate-growth scenarios. The higher growth levels are the primary determinants of the differences in vehicle-miles of travel between the two sets of scenarios. The decentralized plan may be expected to generate nearly 1.5 million more vehicle-miles of travel per average weekday, or about 3 percent more than the highgrowth centralized plan.

With regard to transit ridership, both of the highgrowth plans show less ridership than the adopted regional land use plan. This is to be expected under the high-growth plans since these plans would accommodate population growth in areas not proposed to be well served by transit under the recommended transportation system plan. The recommended transportation system plan is based upon the intermediate-growth scenario and, accordingly, is not designed to provide transit service to the more extensive growth areas of the higher-growth alternatives. Transit ridership would be lowest under the high-growth decentralized plan, since much of the forecast population growth would be distributed almost entirely to areas not served under the recommended transportation system plan by public transit. In addition, the high-growth future scenarios assume higher household incomes, in part due to the higher employment levels.

Table 259 also indicates that there is little difference with respect to transportation impacts between the adopted land use plan and the continued trends land use pattern. Indeed, the travel that the recommended transportation system plan is expected to accommodate does not appreciably differ from the travel likely to occur under a scenario representing the continuation of existing urban development trends. The only significant difference is in the level of transit ridership. The adopted land use plan is likely to generate about 196,600 transit trips on an average weekday, while the existing trends land use alternative is likely to generate about 186,000 such trips on an average weekday. The sensitivity analysis conducted indicates that rather insignificant differences may be expected in the character of travel between the adopted regional land use plan and the existing trends land use alternative. Two factors serve to explain this conclusion. First, both future land use scenarios were designed to serve over the plan design period a growth in resident population of about 6 percent, in households of about 14 percent, and in employment of about 11 percent. Second, the major difference between the two plans is the distribution of this forecast growth throughout the Region. Since monitoring of recent development trends by the Commission has indicated that nearly 80 percent of all new housing units constructed within the Region from 1970 to 1985 were located within the planned urban service areas, it may be expected that under the existing trends alternative a sizable proportion of forecast household growth will, under any conditions, occur in conformance with the adopted plan. The household being the primary determinant of internal tripmaking, it may be concluded that the travel demand that the recommended transportation plan was designed to serve will not significantly differ from the travel demand that may be expected to occur should the regional land use pattern continue to develop according to recent trends.

The differences in the character of travel between the adopted regional land use plan and the high-growth plans are largely related to the greater increases in resident population and economic activity levels forecast under these plans, and, in the case of the high-growth decentralized plan, the distribution of that anticipated growth. Should population and economic activity increase as forecast under the high-growth plans, the planned arterial street and highway system may be expected to carry higher volumes of traffic and, in some locations, experience greater levels of traffic congestion. Under the high-growth decentralized plan, the planned transit system may be expected to carry fewer passengers.

SUMMARY

This chapter has described the preliminary recommended year 2010 regional transportation system plan and summarized the findings of the public review of that plan held throughout the Region during the summer of 1994. It has also described the revisions made to the preliminary plan in response to the comments received through the public review process and described the final recommended plan proposed for adoption by the Commission and the concerned constituent units and agencies of government.

The Advisory Committee recommended that three alternative plans be fully evaluated as a basis for selecting a preliminary recommended plan for public presentation and review. The no-build plan proposed no significant improvements in transportation facilities and services over those which were programmed for implementation in the annual element of the 1993 through 1998 Transportation Improvement Program for Southeastern Wisconsin. Alternative Plan 1 proposed substantial increases in automobile operating costs as a measure to moderate travel demand and substantial improvements to the regional public transit system. Alternative Plan 3, rather than emphasizing the management of travel demand, as did Alternative Plan 1, proposed to accommodate future travel demand largely through improvements to the public transit and arterial street and highway systems. Upon careful consideration of these alternatives, the Advisory Committee recommended that Alternative Plan 3 be further refined, detailed, and advanced to public review as a preliminary recommended plan.

The Committee rejected the no-build plan, finding the scope of improvements proposed therein inadequate to address the identified transportation problems, including severe traffic congestion and a relatively poor level of transit service provided. The Committee rejected Alternative Plan 1, finding the proposed doubling of automobile operating costs determined necessary to affect travel demand in the Region significantly impractical to implement within the plan design period. The Committee also recognized that the substantial pricing measures proposed under Alternative Plan 1 would not significantly reduce the need for arterial street and highway improvements through the year 2010. Indeed, under Alternative Plan 1, about 338 miles would need to be reconstructed for additional capacity, only 24 miles less than under Alternative Plan 3. Also, about 101 new miles of new arterial facility would need to be constructed, only 11 miles less than under Alternative Plan 3.

In recommending Alternative Plan 3 as a preliminary recommended plan, the Advisory Committee directed that additional improvements be proposed in the local transit service to be provided within the Milwaukee, Kenosha, Racine, and Waukesha areas. Accordingly, changes were made to the local public transit service initially envisioned under Alternative Plan 3, which increased the vehicle-miles of local transit service that would be provided an average weekday to 72,300, or by about 19 percent over that proposed under Alternative Plan 3, and increased the vehicle-hours of local transit service that would be provided on an average weekday to 5,800, or by 14 percent over that proposed under Alternative Plan 3.

Public Reaction to Preliminary Recommended Plan and Advisory Committee Response

The preliminary recommended regional transportation system plan was the subject of a Regional Planning Conference held on June 27, 1994; a series of public informational meetings and hearings held in each of the seven counties of the Region in July and August of 1994; meetings of the Commission's Kenosha, Ozaukee, Racine, Walworth, Washington, and Waukesha County Technical Coordinating and Advisory Committees on Jurisdictional Highway System Planning; and of a joint meeting of the Intergovernmental Coordinating and Advisory Committees on Transportation System Planning and Programming for the Kenosha, Milwaukee, and Racine urbanized areas. The conference and public hearings were attended by a total of 760 persons and 143 letters were received for the record following the public hearings. Membership of the jurisdictional highway system planning committees totaled 157 persons, while membership of the three transportation system planning and programming committees totaled 42 persons. Based upon the findings of this extensive public review process, the Technical Coordinating and Advisory Committee recommended certain modifications to the preliminary plan.

With respect to the transit system element, the Advisory Committee concluded that only minor modifications to the preliminary plan were warranted in response to the public comment. These modifications included the proposed extension of bus rapid transit service from the Village of Germantown to the City of West Bend and from the Village of Grafton to the Village of Saukville and City of Port Washington. These modifications also included the addition of commuter passenger train service from the City of Burlington through Silver Lake and Antioch, Illinois, to Chicago and from the Village of Walworth through Fox Lake, Illinois, to Chicago. The final plan recommends that feasibility studies, followed by major investment analyses, be conducted to determine the feasibility and desirability of implementing these proposed extensions into the Region of Chicago-based commuter passenger train service.

With respect to the arterial street and highway element, the Advisory Committee also concluded that only modest modifications to the preliminary plan were warranted in response to the public comment. It was concluded that the level of congestion used to define the need for improvements was appropriate, given that the highway capacity improvements were proposed as a measure of last resort to address traffic congestion problems.

In response to the public comment, the Advisory Committee recommended certain specific changes with respect to the arterial street and highway element of the plan in each County of the Region. These changes are described in a previous section of this chapter.

Recommended Year 2010 Regional Transportation System Plan

Like the preliminary plan, the final recommended plan consists of recommendations in three component areas: transportation system management, public transit maintenance and improvement, and arterial street and highway maintenance and improvement.

The recommended plan proposed the use of transportation system management measures to, among other things, ensure that maximum use is made of existing transportation facilities before commitments are made to new capital investment. The plan envisions the completion of the comprehensive freeway traffic management system within the Milwaukee area; the imposition of peak-hour curb-lane parking restrictions on approximately 442 miles of urban arterial streets; the use of appropriate traffic management and engineering techniques to assist in achieving efficient traffic flow on urban arterial streets; the eventual application of intelligent transportation systems technology; areawide promotional measures to encourage carpooling, vanpooling, telecommuting, and rescheduling of work time; and transit management and operational measures that have the potential to make transit use more convenient.

The plan also envisions that, as new urban neighborhoods are developed and older neighborhoods are redeveloped, detailed land use and community facility plans would be prepared and implemented which would promote a mix of land use activities, higher-density development near transit lines and stations, the orientation of buildings on sites in a manner facilitating transit use; and the use of bicycle and pedestrian as well as transit facilities. Importantly, the plan recommends that the county and local units of government in the Region work to implement the adopted regional land use plan.

The plan also proposes that an integrated system of rapid, express, and local transit facilities be developed within the Region. The plan seeks to provide bus rapid transit service within the major travel corridors emanating from the Milwaukee CBD. The plan calls for the provision of such service south to the Cities of Racine and Kenosha, southwest to the Village of Mukwonago, and west to the Cities of Waukesha and Oconomowoc. The plan would also provide such service in the northwest corridor to the City of West Bend and in the IH 43 north corridor to the Village of Saukville and City of Port Washington. In all, 30 rapid transit bus routes would be operated within the Region, providing over 1,360 round-trip route-miles of service. In 1991, 13 rapid transit routes were in place, with service provided over 449 round-trip route-miles. The planned rapid transit system would be served by 73 transit stations. There were 25 such stations within the Region in 1991.

The plan also envisions the provision of a busway-HOV facility in the east-west corridor between the Cities of Waukesha and Milwaukee, pending the outcome of a major investment study conducted by the Wisconsin Department of Transportation. The plan recognizes the need to provide additional busway-HOV facilities within the Region as freeway traffic congestion increases to unacceptable levels.

The plan also envisions the provision of commuter passenger train service in four Milwaukeeoriented corridors: north to Saukville, northwest to West Bend, west to Oconomowoc, and south through Racine and Kenosha to Chicago and two Chicago-oriented corridors: Walworth to Chicago and Burlington to Chicago. The implementation of these proposed commuter transit facilities would necessitate the conduct of Federally required major investment studies prior to their implementation.

The plan also proposes that 12 regular express transit bus routes, or five more routes than in 1991, be provided within the Region. Within the Milwaukee urbanized area, the express transit routes would be provided in certain major travel corridors connecting major activity centers to the Milwaukee CBD, as well as in a grid pattern of crosstown routes. An express transit route would also connect the Cities of Racine and Kenosha.

The plan also provides for the development of a light-rail express transit line between the City of Glendale and the University of Wisconsin-Milwaukee campus, the Milwaukee CBD, and the Milwaukee County Institutions grounds. The necessary corridor study for this fixed-guideway facility is currently being undertaken by the Wisconsin Department of Transportation. The implementation of the light-rail proposal, as well as of the busway proposal in the east-west corridor, would depend upon confirmation by the study and the concurrence of the implementing units of government concerned in the final findings and recommendations of the corridor study. The plan identifies three other travel corridors with potential for the development of light-rail lines. The feasibility of light-rail service in these corridors would also have to be confirmed through the conduct of major investment studies.

The plan also proposes the expansion and improvement of local public transit service within Milwaukee County and the Cities of Waukesha, Racine, and Kenosha and their immediate environs. Local transit service would be expanded generally in southern and northern Milwaukee County and in the most heavily developed portions of eastern Waukesha County. The number of round-trip route-miles of local service would be increased from 171 in 1991 to 210 by 2010 within the Kenosha urbanized area; from 1,112 in 1991 to 1,450 by 2010 within the Milwaukee urbanized area; and from 171 in 1991 to 200 by 2010 within the Racine urbanized area. The plan also recognizes the need to provide local transit service in the smaller outlying urban communities of the Region, particularly through shared-ride taxi service.

The recommended plan envisions that the arterial street and highway system would, by the plan design year 2010, consist of about 3,607 route-miles and 10.303 lane-miles of facility. In 1991, the arterial system consisted of about 3,274 route-miles and 8,420 lane-miles of facility. The plan provides for the construction of 131 routemiles and 337 lane-miles of new arterial facilities, for the widening or otherwise improvement of 448 miles of existing arterial facilities to provide an additional 1,877 lane-miles, and for the preservation of 3,028 route-miles and 8,089 lane-miles. The recommended plan includes important preservation actions, including the reconstruction to current design standards of the failing Milwaukee area freeway system.

Recommended Plan Performance and Costs

The number of internal person trips generated within the Region on an average weekday may be expected to increase from 5.54 million in 1991 to about 6.10 million in the year 2010, or by 10 percent. The number of transit trips made on an average weekday may be expected to increase from about 172,200 in 1991 to about 196,400 by the year 2010, or by 14 percent. The proportion of total internal person trips made by transit may be expected to remain, however, at about 3 percent.

Vehicle-miles of travel within the Region on an average weekday may be expected to increase from about 33.07 million in 1991 to about 42.42 million by the year 2010, or by about 28 percent. Arterial street and highway congestion, as indicated by the number of arterial miles expected to operate over design capacity, may be expected to decrease from about 385 miles, or 12 percent of the total arterial mileage in 1991, to about 165 miles, or 5 percent, by 2010. The arterial mileage operating at design capacity, however, may be expected to increase from about 158 miles, or about 5 percent of the total system in 1991, to 218 miles, or about 6 percent in 2010.

The public cost of carrying out the recommended plan, including the construction of new facilities, and the operation and maintenance of the arterial street and highway and transit systems is estimated at \$521.9 million per year over the 16-year plan implementation period. The public revenues anticipated to be available based on existing trends are estimated at \$299.9 million per year. The difference between anticipated costs and revenues is approximately \$222.0 million per year. An equivalent of a \$0.26 per gallon motor fuel tax would be necessary to cover this \$222.0 million annual shortfall in order to fully implement the recommended regional transportation system plan for the year 2010.

Analysis of the Sensitivity of the Recommended Plan to Alternative Future Growth Scenarios

The Advisory Committee requested that the sensitivity of the recommended plan to alternative future land use patterns within the Region and associated levels of population, households, and employment be analyzed. The purpose of the analysis was to investigate the differences in the amount of travel that would attend future land use patterns different than that envisioned under the adopted regional land use plan. Three alternative land use patterns were used: an intermediate-growth existing trends, a high-growth centralized, and a high-growth decentralized.

It may be concluded that little difference may be expected in the character of travel within the Region between the adopted regional land use plan and the existing trends land use alternative. Two factors serve to explain this conclusion. First, both land use scenarios were designed to serve a growth in population of 6 percent, in households of about 14 percent, and in employment of about 11 percent. Second, a sizable portion of urban growth, household growth in particular, under the existing trends land use pattern may be expected to occur in conformance with the adopted land use plan. Since the household is the primary determinant of internal trip making, the travel demand the transportation system plan was designed to serve would not differ appreciably from the travel demand that may be expected to occur should the regional land use pattern continue to develop according to recent trends.

It may also be concluded that the difference in travel between the adopted regional land use plan and the high-growth scenarios is related to the greater increases in resident population and economic activity assumed under the highgrowth scenarios. Should population and economic activity increase as forecast under the high-growth scenarios, the planned arterial street and highway system may be expected to carry higher traffic volumes and in some locations experience greater levels of traffic congestion. Under a high-growth decentralized plan, the planned transit system may be expected to carry substantially fewer passengers.

Anticipated Impact on Travel Attendant to a 100 Percent Increase

in Automobile Operating Costs

The Technical Coordinating and Advisory Committee also recommended that the potential impacts associated with a policy of substantially increasing automobile operating costs should be presented for informational purposes during the public review of the preliminary recommended plan. Alternative Plan 1 was used as a basis for this analysis as it assumed a 100 percent increase in the cost of driving. The preliminary recommended plan is compared to Alternative Plan 1 previously in this chapter but in this concluding section, the final recommended plan is compared to Alternative Plan 1. The pricing measures assumed under Alternative Plan 1 for the analyses would double the perceived costs of operating an automobile over the 1991 level. This doubling would mean that the perceived cost of driving an automobile would increase from about \$0.055 to about \$0.11 per mile.

Should the perceived cost of automobile operation be doubled. 38.38 million vehicle-miles of travel could be expected to take place on the arterial street and highway system on an average weekday in the plan design year 2010. This would be about 4.04 million vehicle-miles fewer than expected under the recommended plan. By way of comparison, in 1991 about 33.07 million vehicle-miles of travel took place on the arterial street and highway system of the Region. About 131 miles of arterial facilities could be expected to operate over design capacity during peak-hour travel periods, representing 34 miles, or 21 percent, less than could be expected under the recommended plan. The lower levels of travel and traffic congestion could be expected to result in the emission of fewer air pollutants and in the consumption of fewer gallons of motor fuel within the Region. About 25.6 tons of volatile organic compounds, 71.3 tons nitrogen oxides, and 211.5 tons carbon monoxide would be generated, representing 2.1, 7.6, and 14.5 fewer tons, respectively than under the recommended plan. Also, 412 million gallons of fuel could be expected to be consumed annually within the Region, or about 32 million gallons, or 7 percent, less per year than could be expected under the recommended plan.

Concluding Remarks

The recommended regional transportation system plan for the year 2010 is a sound plan,

worthy of consideration for adoption and implementation by all parties concerned. Among the reasons this is so are the following:

- Implementation of the recommended transportation system plan would provide the Region with an integrated transportation system that can both effectively serve and promote a desirable regional land use pattern. That pattern is defined in the adopted regional land use plan. The recommended transportation system plan was explicitly designed to serve the future travel demand that may be expected to be generated by the regional land use plan. Thus, the transportation system plan calls for the development of highway and transit facilities to serve those areas where land use development and redevelopment are recommended to occur.
- The plan explicitly proposes arterial street and highway system capacity expansion only as a measure of last resort in addressing traffic congestion problems. The potential ability for land use planning and site design, public transit service, travel demand management, and traffic management measures to address traffic congestion were first determined and the problems remaining after the assumed application of these measures explicitly identified. Only those residual problems were addressed by capacity expansion.
- The recommended transportation system plan will provide the facilities needed to meet anticipated future travel demand at an adequate level of service. The plan will abate traffic congestion, reduce travel time and costs between component parts of the Region, and reduce accident exposure. In 1991, 385 miles of arterial facilities, or 12 percent of the total arterial system, carried traffic volumes on an average weekday which exceeded their design capacity and therefore experienced congestion. Under the no-build plan, arterial mileage subject to congestion may be expected to increase to 712 miles, or about 21 percent of total arterial system mileage. Under the final recommended plan, that mileage may be expected to decline to 165 miles, or 5 percent of total arterial system mileage, principally as a result of the recommended highway system capacity improvement and expansion.
- The recommended transportation system plan will provide a balanced transportation system, with appropriate types of highway and transit facilities and services provided for the various subareas of the Region. For an increased proportion of the resident population of the Region, implementation of the plan would enable trips to be made either by automobile or by transit, thus enhancing choice in travel-related decision making. Transit service would be extended and improved to enhance access to employment opportunities and to major educational, medical, shopping, and recreational facilities and services. The plan would provide a system of rapid transit, express transit, and paratransit facilities and services, as well as conventional mass transit facilities and services, and seek to halt the historic decline in the use of transit and the funding for the operation and maintenance of transit service within the Region. As an indication of the emphasis in the plan on the improvement in transit service, it may be noted that less than 4 percent of total travel demand is now, or may in the future be expected to be accommodated on public transit facilities. Nevertheless, the plan proposes that 38 percent of the total financial resources available for transportation be allocated to transit. The transit system would be expanded by about 75 percent, as measured in vehicle-miles of service, while the highway capacity would be expanded by only 22 percent as measured by lane-miles of facility.
- The benefits to be secured by implementation of the recommended transportation system plan exceed the costs associated with that plan, indicating that the plan is economically sound and cost effective, as well as technically feasible and efficient.
- The plan is environmentally sound. It provides for a reduction in energy consumption and air pollutant emissions, compared to the no-build alternative. The loss in primary environmental corridor lands, wetlands, and primary agricultural lands attendant to construction proposed under the plan is less than three tenths of a percent of the existing total of such lands within the Region. Importantly, in this respect, the transportation system plan is based upon and serves to help implement

the adopted regional land use plan, which plan seeks to preserve the environmental corridors and remaining prime agricultural areas of the Region and which seeks to discourage the further diffusion of lowdensity urban development and the decentralization of jobs within the Region.

- The plan would provide for the proper maintenance of the existing transit and arterial street and highway infrastructure within the Region, including provision for the rehabilitation of the existing greater Milwaukee area freeway system. The level of transit service within the Region has, over the last decade, been steadily reduced, measured by vehicle-miles of service. Similarly, the needed resurfacing and reconstruction of county and local arterials has been delayed with the rates of such maintenance being inadequate to maintain current pavement conditions into the future.
- The plan provides for the long-planned and long-needed improvement of the public transit and arterial street and highway systems. With respect to public transit, the plan proposes the development of rapid and express transit system elements. With respect to arterial streets, the plan proposes the widening to provide additional traffic lanes of 448 miles of arterial streets, or 3 percent of the total system, and the construction of 131 miles of new arterial streets, representing about a 4 percent expansion of arterial system capacity.
- The plan is realistic and achievable. The existing and new revenue sources necessary to implement the plan are identified and a proposal to secure the resources to resolve estimated funding shortfalls is set forth in Chapter XII.

PLAN IMPLEMENTATION

INTRODUCTION

The recommended regional transportation system plan, described in the preceding chapter of this report, provides a design for the attainment of specific transportation development objectives. The recommended plan proposes transportation system management measures to ensure that the most efficient use is made of existing transportation facilities. The plan also proposes needed improvements to the public transit and arterial street and highway systems within the Region. In a practical sense, however, the plan is not complete until the steps required to implement the plan are specified.

This chapter specifies the steps required to implement the recommended plan. It identifies the various units and agencies of government which have plan adoption and implementation responsibilities germane to the recommended plan and specifies necessary plan adoption and implementation actions. It prioritizes the transit and highway improvements included in the plan, providing the basis for the preparation of future regional transportation improvement programs. It describes the detailed implementation planning that will need to be conducted during the plan implementation period, identifying in particular transportation improvements that warrant Federally required major investment studies. The chapter also recommends how the identified funding shortfall should be addressed, as well as the manner in which the Federally required transportation management systems should operate over the plan implementation period. Finally, the chapter outlines a series of proposed special studies relating to transportation implementation matters.

BASIC PRINCIPLES AND CONCEPTS

Plan implementation measures must grow out of adopted plans. Thus, action policies and programs must emphasize the most important and essential elements of the plan. This is particularly important in planning for the orderly and economical development of a large urban region. The task is so highly complex that care must be taken not to become lost in plan implementation detail, the effects of which may be meaningless at the regional scale. Accordingly, regional transportation system plan implementation should focus on those facilities having areawide significance. The regional plan will be largely achieved if: 1) the regional freeway system is completed and appropriately rehabilitated and modernized, 2) improvements to the major surface arterials, particularly those included in the recommended National Highway System, are implemented, and 3) the rapid and express transit expansion and improvement recommendations are carried out.

There are three main areas through which regional plan implementation may be achieved; these parallel the three functions of the Regional Planning Commission: areawide inventory or monitoring, preparation of a framework of longrange plans for the physical development of the Region, and provision of a center for the coordination of Federal, State, county, and local planning and plan implementation activities. All require a receptive attitude and, preferably, active planning and plan implementation programs at the local, county, State and Federal levels of government.

A great deal can be achieved in guiding areawide development along better lines if all concerned units and agencies of government, along with the Regional Planning Commission, perform the simple task of collecting, analyzing, and disseminating basic planning and engineering data on a continuing, uniform, areawide basis. Experience within the Region has shown that if this important inventory, or monitoring, function is properly carried out, the resulting information will be used and acted upon by local, county, State, and Federal agencies of government and by private investors. If these same data were used as a primary input into the regional plan preparation, their utilization in arriving at public and private development decisions on a day-to-day basis would tend to contribute toward implementation of the regional plans.

With respect to plan preparation it is essential that the regional plans, although confined to those functional elements having areawide

significance, be prepared in sufficient depth and detail to provide a sound basis for plan implementation. This means that for necessary public works facilities, such as freeways, major surface arterials, and rapid and express transit facilities, the plans must be carried to a stage wherein the location and alignment has been determined with sufficient accuracy and precision to provide an adequate basis for right-of-way reservation. Given such detailed plans, implementation will further require the continuation of close working relationships between the Commission, the seven county boards, the local units of government, and such State agencies as the Wisconsin Departments of Transportation and Natural Resources.

Finally, it will be highly desirable, although not essential, to achieve an even finer degree of plan implementation than would be attainable through concern with the major plan elements alone through the Commission function of serving as a center for the coordination of local, county, State, and Federal planning and plan implementation activities within the Region. The Commission's community assistance program, which actively assists counties and local municipalities in the preparation of plans and plan implementation devices, is an important factor in this respect and will make possible the close integration of regional, county, and local plans, adjusting the details of the latter to the broad framework of the former.

PLAN IMPLEMENTATION ORGANIZATIONS

Because the Commission is an advisory agency, implementation of the recommended plans will be entirely dependent upon the actions taken by certain local, county, areawide, State, and Federal agencies of government. Examination of the various agencies that are available to implement the recommend plan under existing enabling legislation reveals an array of departments, commissions, councils, boards, districts, and authorities at all levels of government. These agencies range from general-purpose units of local government, such as common councils and village boards, to such agencies as the Federal Highway and Transit Administrations.

The agencies whose actions will significantly impact, either directly or indirectly, the successful implementation of the recommended regional transportation system plan and whose full cooperation in plan implementation will be essential are listed and discussed below. For convenience, these agencies are arranged by level of government; however, the interdependence between the various levels, as well as between the various agencies, of government and the need for close intergovernmental cooperation cannot be overemphasized.

Local-Level Agencies

Local Plan Commissions: Sections 62.23, 61.35, and 60.10(2)(c) of the Wisconsin Statutes permit municipalities to create plan commissions. Plan commissions, among other functions, are charged with the responsibility of making and adopting a master plan for the physical development of the community, including recommendations relating to streets and highways, routes for railways and buses, and terminals. Moreover, the location, extension, alteration, and acquisition of land for any street or other public way must be referred to the plan commission for recommendation prior to any action by the municipal governing body.

<u>Boards of Public Works</u>: Sections 62.14, 61.35, and 60.10(2)(c) of the Wisconsin Statutes permit municipalities, under the direction of the common council, village board, or town board, to form boards of public works to superintend the construction and maintenance of all public works. Such boards have primary responsibility for local arterial streets and highways and are able to take on public transit responsibilities as well.

Committees of the County Boards: Highway, Transit, and Public Works: These committees are responsible for the administration and expenditure of county funds for highway construction and maintenance. They are empowered to establish and change the county trunk highway systems, subject to the approval of the Wisconsin Department of Transportation; to acquire land for county highway purposes by purchase or condemnation; and to give direction to the operation and maintenance of public transit systems. All seven counties within the Southeastern Wisconsin Region have established such committees: in Kenosha County, a Highway and Parks Committee; in Milwaukee County, a Mass Transit Committee and a Transportation and Public Works Committee; in Ozaukee County, a Highway Committee; in Racine County, a Public Works Committee; in Walworth County, a Highway Committee; in Washington County, a Highway Committee; and in Waukesha County, a Public Works Committee.

<u>Transit Commissions and Boards</u>: Transit commissions can be established by cities and are empowered to establish, maintain, and operate a public transportation system, the major portions of which are located within the city. The Cities of Kenosha, Racine, and Waukesha have created such governmental bodies to provide urban public transit services. Transit boards may be established by counties and are empowered to create, maintain, and operate a public transportation system within that county and any contiguous or cornering counties. There currently are no county transit boards in the Region.

Areawide Agencies

<u>Cooperative Contract Commissions</u>: Section 66.30 of the Wisconsin Statutes provides that municipalities¹ may contract with each other to provide jointly any services or exercise jointly any powers that such municipalities may be authorized to provide or exercise separately. While no transportation-related cooperative contract commissions currently exist within the Region, there is potential to achieve significant economies through providing transportation services and facilities on a cooperative, areawide basis. Moreover, the nature of certain transportation problems often requires that solutions be approached on an areawide basis.

<u>Metropolitan Transit Authority</u>: Such an authority, if created pursuant to Section 66.94 of the Wisconsin Statutes, would have the power to acquire, construct, and operate a public transportation system and would have the power of eminent domain within a district which would include all of Milwaukee County and those units of government located in adjacent counties through, and into, which the transit system would extend. Such an authority does not have any powers of taxation. It can, however, issue revenue bonds. No such authority has been activated within the Region at present.

Regional Transportation Authority: The Regional Planning Commission studied the feasibility of creating a regional transportation authority (RTA) within Southeastern Wisconsin.² Following that study, State legislation was enacted to create an RTA encompassing all seven counties in the Region and directing that the RTA conduct its own study and recommend whether or not it should continue in existence after September 30, 1993.³ Over an approximately 15-month period during 1992 and 1993, the RTA Board carried out its own study. The results of that study were set forth in a report to the Governor and the Legislature.⁴ In that report, the RTA Board developed a proposal for a permanent authority, the essence of which consisted of the following:

1. Geographic Scope

The study proposed a seven-county RTA providing, however, that during the first six months of existence, a county could exercise a withdrawal option. Absent such a withdrawal, the county would be a permanent member of the RTA. Any county which withdrew in the initial six months could petition later to rejoin. The RTA Board would be permitted to impose conditions for rejoining.

2. Board Structure

The study proposed that the RTA be governed by an 11-member board, assuming all seven counties participated, including, on a ex-officio basis, the State Secretary of Transportation. Each participating county would have one representative residing in that county. There would be three at-large members residing in the Region, with one of those appointed residing within the City of Milwaukee. All members would be appointed by the Governor and confirmed

²See SEWRPC Memorandum Report No. 38, <u>A</u> <u>Regional Transportation Authority Feasibility</u> <u>Study for Southeastern Wisconsin</u>, November 1990.

³See <u>Wisconsin Statutes</u>, Section 59.966.

⁴See <u>Southeastern Wisconsin Regional Transpor-</u> tation Authority Report to Governor Thompson and the Wisconsin Legislature, May 1993.

¹Under this section of the Statutes, the term municipality is defined to include the State and any agency thereof, cities, villages, towns, counties, school districts, and regional planning commissions.

by the State Senate. The Governor would designate the Board chair.

3. Functions and Responsibilities

The study proposed that the RTA be empowered as a funding and plan implementation agency. All transportation projects supported with RTA funds would have to be drawn from the adopted regional transportation system plan. The RTA would not be enabled to construct and maintain arterial highway systems; however, the RTA would be enabled to provide funds to county and local governments for arterial highway construction, operation, and maintenance. The RTA would also be enabled to fund county and local governments that deliver transit services as well as to directly sponsor and provide transit services on a contractual basis either with public transit agencies or with private providers. The RTA would also be empowered to assume responsibilities to provide county and local transit services where county and local governments want to transfer that function to the RTA. Finally, the RTA would be given responsibility to carry out areawide transportation demand management programs, such as carpooling and vanpooling promotional efforts.

4. Revenues

The study proposed that the RTA be funded through two additional taxes levied in the Region by the RTA: a 0.4 percent general sales tax and a five-cent-per-gallon motor fuel tax. The motor fuel tax would not be levied on diesel fuel. These two taxes could be expected to raise a minimum of \$90 million annually in the Region.

5. Revenue Allocation

The study proposed that the legislation guarantee that over a six-year period every county would receive a minimum of 98 percent of the revenue raised in the county. In addition, every county would be guaranteed to receive annually at least 80 percent of the revenue raised in the county.

The RTA Board delivered its study recommendations to the seven counties in the Region early in 1993. Resolutions supporting the study recommendations were defeated by the County Boards of Kenosha, Ozaukee, Racine, Walworth, Washington, and Waukesha Counties. The Milwaukee County Board approved the supporting resolution on the condition that the regional taxes envisioned be levied instead Statewide and be confined to motor fuel taxes. On the strength of these County Board actions, the RTA Board recommended to the Governor and the Legislature that the Board be disbanded and that a permanent authority not be created at that time. Accordingly, the RTA went out of existence on September 30, 1993.

Regional Planning Commission: Although not a direct plan implementation agency, one other areawide agency warrants description herein: the Regional Planning Commission. The Commission, created under Section 66.945 of the Wisconsin Statutes, is empowered to prepare and adopt a master plan, of which the transportation system plan is a part, for the physical development of the Region. It has no statutory plan implementation powers. A special designation assigned to the Regional Planning Commission under Federal law is that of Metropolitan Planning Organization. This designation means that the Commission provides a forum for cooperative decision making concerning the preparation and adoption of transportation system plans and improvement programs. Under Federal law, the Commission, as the Metropolitan Planning Organization, is given the responsibility to program certain Surface Transportation Program monies allocated for use in the urbanized areas. The Commission has chosen to exercise this responsibility through its Intergovernmental Coordinating and Advisory Committee on Transportation System Planning and Programming for the Milwaukee Urbanized Area.

State-Level Agencies

Wisconsin Department of Transportation: The Wisconsin Department of Transportation is responsible for the planning of all transportation modes within the Region. The Department is authorized to provide the State with an integrated and intermodal transportation system and to administer State and Federal aids for highway and transit improvements. The Department is also responsible for planning, designing, constructing, and maintaining all State trunk highways and for planning, laying out, revising, constructing, reconstructing, and maintaining the National Highway System and the Interstate System, subject to Federal review and regulation.⁵

Wisconsin Department of Natural Resources: Pursuant to the provisions of the Federal Clean Air Act, the Wisconsin Department of Natural **Resources must prepare a State Implementation** Plan for the attainment and maintenance of the National Ambient Air Quality Standards. Under the Clean Air Act Amendments of 1990, the seven-county Southeastern Wisconsin Region has been designated a nonattainment area with respect to ozone. Therefore, the recommended regional transportation system plan, together with subsequently prepared transportation improvement programs, must be found to conform to the State Implementation Plan for air quality. If a conformity finding cannot be made, then the plan must be revised until a conformity finding can be made. These requirements have made close cooperation between the Commission and the Wisconsin Department of Natural Resources essential both in the preparation and implementation of the plan.

Wisconsin Department of Administration: The Wisconsin Department of Administration provides the Governor with the information and policy alternatives necessary for the preparation of the State biennial budget. In addition, the Department acts as the State clearinghouse for intergovernmental review of Federally aided programs and projects. Through this review process the Department may comment on all Federally aided transportation projects.

Wisconsin Department of Development: The Department administers the State's economic development programs and policies. Though the Department has no direct role in the implementation of the regional transportation system plan, its activities, especially related to the retention or expansion of existing business and the attraction of new business, may significantly impact the use of the regional transportation system, particularly if such activities should run counter to the adopted land use plan for the Region.

University of Wisconsin-Extension

As a part of the University of Wisconsin System, the University of Wisconsin-Extension is the institution principally charged with implementing the "Wisconsin Idea" of extending the knowledge and resources of the University of Wisconsin System to the citizens of the State, thereby helping them to make more informed decisions. The Regional Planning Commission and the University of Wisconsin-Extension have entered into a contractual agreement for the provision of educational services, focused upon transportation, land use, and environmental protection planning and plan implementation issues. Each county in the Region has a University of Wisconsin-Extension office that can be used to expand the network of education related to plan implementation. It is recognized that educating public officials and the citizens at large in the Region about the plan findings and recommendations will contribute significantly to, and indeed is a necessary element of, plan implementation.

Federal-Level Agencies

<u>U. S. Department of Transportation</u>: The <u>U. S. Department of Transportation</u>, in cooperation with the Regional Planning Commission, as the Metropolitan Planning Organization for the Southeastern Wisconsin Region, must make the necessary determinations to ensure that the recommended regional transportation system plan and subsequently prepared improvement programs conform with the State Implementation Plan for air quality. When making the conformity determinations the Federal Highway and Federal Transit Administrations will require certain Federally specified technical

⁵The Wisconsin Department of Transportation's recommendations for major highway projects, defined in Section 13.489 of the Wisconsin Statutes as highway reconstruction or reconditioning costing \$5 million or more and involving either relocation of 2.5 miles or more, or construction of five or more miles of additional lanes to an existing highway, are reviewed by the Wisconsin Transportation Projects Commission. This Commission, which is composed of the Governor, five Senators, five Assembly Representatives, three citizen members, and the Secretary of the Wisconsin Department of Transportation (a nonvoting member), reviews Department recommendations and, in turn, recommends to the Legislature highway projects proposed to be enumerated in the Statutes. The enumeration of major highway projects in the Statutes signifies a firm State commitment to funding such projects in future years.

work to be completed by the Commission in order to demonstrate that the conformity criteria are met. Importantly, the plan implementation responsibilities of the two Federal agencies also extend to the funding of transportation improvement projects.

<u>U. S. Department of Transportation, Federal</u> <u>Highway Administration</u>: The Federal Highway Administration administers all Federal Highway aid programs, working through the Wisconsin Department of Transportation. The Federal Highway Administration must approve all changes in the National Highway System and Interstate System and will, in this respect, have an important role in implementation of the highway element of the recommended transportation system plan for the Region.

U.S. Department of Transportation, Federal Transit Administration: The Federal Transit Administration administers a comprehensive set of programs offering Federal funds to eligible local recipients in partial support of the preservation, improvement, and expansion of public transit service. Federal Transit Administration programs germane to the recommended plan include Section 3, Section 9, and Section 18. Federal funds made available for transit projects under the Surface Transportation Program and the Congestion Mitigation and Air Quality Improvement Program are transferred from the Federal Highway Administration to the Federal Transit Administration Section 9 and Section 18 programs and thus become subject to the administrative requirements of the Federal Transit Administration.

<u>U. S. Environmental Protection Agency</u>: The U. S. Environmental Protection Agency (EPA) is responsible for approving the State Implementation Plan for air quality and for imposing sanctions on a State for failing to meet certain requirements of the Clean Air Act Amendments of 1990. If the EPA finds a failure on the part of the State to submit a State Implementation Plan for air quality, or a portion thereof, to implement the provisions of such an approved plan, or to conform to any other provision required by the Federal Clean Air Act, it must, after a period of grace during which the deficiency can be corrected, impose mandated sanctions. The EPA may cause Federal highway funds to be withheld and/or may require a twoto-one offset for major stationary air pollutant sources.⁶

PLAN ADOPTION AND INTEGRATION

Upon adoption of the new regional transportation system plan by formal resolution of the Southeastern Wisconsin Regional Planning Commission, in accordance with Section 66.945(10) of the Wisconsin Statutes, the Commission will transmit a certified copy of the resolution and adopted plan to all local legislative bodies within the Region and to all of the aforementioned existing State, local, county, areawide, and Federal agencies. Endorsement, adoption, or formal acknowledgment and integration of these plans by the local legislative bodies and the existing local, county, areawide, State, and Federal agencies involved is highly desirable, and in some cases necessary, to assure a common understanding between the several government levels and to enable their staffs to program the necessary implementation work. Adoption of the new regional transportation system plan by units and agencies of government that have adopted the initial design year 1990 and the second-generation design year 2000 regional transportation plans will serve to substitute the new plan for the old.

Adoption of the recommended plan by any unit or agency of government pertains only to the statutory duties and functions of the adopting unit or agency. Such adoption does not and cannot in any way preempt action by another unit or agency of government within its jurisdiction. Thus, adoption of the regional transportation system plan by a county would make the plan applicable as a guide, for example, to county highway development but not to municipal street development. The plan would have to be adopted by the municipality concerned to make it applicable as a guide to municipal street development.

⁶In an area under such a sanction, each ton of air pollutant emissions created by a new stationary source must be offset by a two-ton reduction through additional control measures on existing stationary sources.

Finally, while the adoption and endorsement of the recommended regional transportation system plan is important, the need to also to adopt or endorse the companion year 2010 regional land use plan cannot be overlooked. The successful implementation of the regional transportation system plan is interrelated with the successful implementation of the regional land use plan. The development of the transportation system in accord with the recommended plan may not be sufficient to provide a high level of transportation service throughout the Region should urban development occur in a manner significantly contrary to the recommendations of the regional land use plan. Plan adoption, endorsement, and integration recommendations are listed below.

Local Agencies:

- It is recommended that the cities, villages, and towns in the Region, upon recommendation of their plan commissions and boards of public works, adopt the recommended regional transportation system plan as authorized by Section 66.945(12) of the Wisconsin Statutes and integrate the plan into their comprehensive plans and capital improvement programs.
- It is recommended that the seven county boards, upon recommendation of their highway, transit, and/or public works committees, formally adopt the recommended regional transportation system plan as authorized by Section 66.945(12) of the Wisconsin Statutes and integrate the plan into their comprehensive plans and capital improvement programs.
- It is recommended that the Cities of Kenosha, Hartford, Port Washington, Racine, Waukesha, West Bend, and Whitewater and the Counties of Milwaukee and Waukesha, as well as any local unit of government that may in the future begin to operate public transit service, adopt the recommended transportation system as a guide to future transit system development and integrate the plan into their transit development programs.

Areawide Agencies:

• It is recommended that any transportationrelated cooperative contract commission subsequently created formally acknowledge the recommended transportation system plan in regard to the exercise of its specific powers and duties.

• It is recommended that, should a regional or metropolitan transportation authority be established, it, as one of its initial actions, adopt the recommended regional transportation system plan.

State Agencies:

- It is recommended that the Wisconsin Department of Transportation endorse the recommended regional transportation system plan. It is further recommended that the Department integrate the recommendations into the State long-range transportation plan, as authorized by Sections 84.01, 84.02, and 84.025 of the Wisconsin Statues, as a functional guide to highway and transit system development within the Region.
- It is recommended that the Wisconsin Natural Resources Board endorse the recommended regional transportation system plan and that the Wisconsin Department of Natural Resources continue to cooperate with the Regional Planning Commission to ensure its conformance with the State Implementation Plan for air quality.
- It is recommended that the Wisconsin Department of Administration endorse the recommended regional transportation system plan and use the plan in discharging its responsibilities in reviewing and commenting on Federally funded transportation programs and projects.
- It is recommended that the Wisconsin Department of Development endorse the recommended regional transportation system plan and support implementation of the plan through its economic development activities, considering the long-term transportation-related and environmental impacts of its decisions.
- It is recommended that the University of Wisconsin-Extension acknowledge the recommended regional transportation system plan and promote implementation of the plan through its educational programming.

Federal Agencies:

• It is recommended that the U.S. Department of Transportation, Federal Highway Administration and Federal Transit Administration, endorse the recommended regional transportation system plan and find it to conform with the State Implementation Plan for air quality and to have been prepared in a manner consistent with the Intermodal Surface Transportation Efficiency Act of 1991.

• It is recommended that the U.S. Environmental Protection Agency endorse the recommended regional transportation system plan.

Subsequent Adjustment of the Plan

No plan can be permanent in all its aspects or precise in all its elements. The very definition and characteristics of "regional planning" suggest that a regional plan, to be viable and useful to local, State, and Federal units and agencies of government, be continually adjusted through formal amendments, extensions, additions, and refinements to reflect changing conditions. The Wisconsin Legislature foresaw this when it gave to regional planning commissions the power to "amend, extend, or add to the master plan or carry any part or subject matter into greater detail" under Section 66.945(9) of the Wisconsin Statutes.

Amendments, extensions, and additions to the regional transportation system plan will be forthcoming, not only from the work of the Commission under the continuing regional planning program, but also from Statewide plans and from Federal agencies as national policies are established or modified, new programs created, or existing programs expanded or curtailed. Adjustments may come from State, subregional, district, and county and local planning programs which, of necessity, must be prepared in greater detail and may result in refinement and adjustment of the regional plan. All refinements and adjustments will require cooperation between local, areawide, State, and Federal agencies, as well as coordination by the Southeastern Wisconsin Regional Planning Commission, which is empowered under Section 66.945(8)of the Wisconsin Statutes to act as a coordinating agency for programs and activities of the county and local units of government concerned. To achieve this coordination between local, areawide, State, and Federal programs most effectively and efficiently and, therefore, assure the timely adjustment of the regional transportation system plan, it is recommended that all the aforesaid agencies having various plan and plan implementation powers transmit all subsequently prepared planning studies, plan proposals and amendments, and plan implementation devices to the Southeastern Wisconsin Regional Planning Commission for consideration as to integration into the adopted regional plan.

PLAN IMPLEMENTATION RECOMMENDATIONS

The recommended year 2010 regional transportation system plan has three major elements: transportation system management, public transit system maintenance and improvement, and arterial street and highway system maintenance and improvement. The specific actions and the agencies responsible for those actions required to implement each of these elements are described in the following sections of this chapter.

Transportation System Management

The planned transportation system management element includes the Milwaukee area freeway traffic management system; consideration of peak-period curb-lane parking restrictions on 464 route-miles of arterial facilities in the Region; areawide promotion of measures to encourage travel through ridesharing, transit use, and bicycle use, as well as to encourage telecommuting and work time rescheduling; and the use of site-specific land use planning to facilitate travel by transit, bicycle, and pedestrian modes. The following are specific implementation responsibilities with respect to these matters:

- 1. It is recommended that the Wisconsin Department of Transportation continue to develop and operate the Milwaukee area freeway traffic management system. The completed system should include a coordinated areawide system of ramp controls that will achieve the highest possible level of service on the freeways and that will encourage travel by transit and by carpools and vanpools.
- 2. It is recommended that the Wisconsin Department of Transportation develop an expanded ridesharing promotional campaign in the greater Milwaukee area; assist area businesses and industries to implement the Federally required Employee Commute Options program, including the formation of private section transportation

management associations (TMAs); promote employer-based transportation demand management strategies; review income tax policies to ensure that there is no modal bias in terms of employer-subsidized parking and employer-provided transit subsidies; explore the feasibility and desirability of influencing local parking pricing and management policies by directly linking State transportation aids to the enactment of local policies that discourage travel by single-occupant vehicles; fund a program to promote travel by bicycle and walking as well as telecommuting; and construct telecommuting centers and carpool lots as more detailed planning efforts and demand may indicate and warrant. It is recommended that upon endorsement of the regional transportation system plan, the Wisconsin Department of Transportation prepare a program plan to identify in detail how best to address the various transportation demand management activities recommended in the regional system plan. It is recognized that in preparing that program plan, the Department may determine to contract with others, including the **Regional Planning Commission**, to provide assistance in carrying out certain activities. It is further recommended that the work plan be subject to review and comment by a Task Force created by the Department for this purpose, the membership of which should include representatives of all interests concerned.

- 3. It is recommended that upon referral to, and recommendation of, the local plan commission, and upon appropriate further study by responsible local traffic engineering staffs, each common council and village board consider appropriately restrict curb-lane parking during peak travel periods on the 464 miles of arterial streets and highways designated as candidates for such restriction in the recommended plan.
- 4. It is recommended that upon referral to, and recommendation of, the local plan commission, each common council, village board, and town board within the Region evaluate zoning, subdivision, and other site planning and development ordinances and practices; identify the manner in which such ordinances and practices may

discourage the development of integrated neighborhoods, transit use, bicycling, and walking; and institute measures to correct land development practices that promote dependence on the automobile.

In this respect it is recommended that within the urban density framework provided by the adopted regional land use plan, higher-density, transit- and pedestrian-friendly urban development be promoted along transit lines and around transit stops and stations through neighborhood development and redevelopment plans. Further, it is recommended that appropriate land development incentives be used to promote high-density, mixed use development around fixed-guideway transit stations, as such fixed-guideway transit facilities are developed within the plan implementation period. It is recommended that local units of government strive to implement the regional bicycle and pedestrian facilities plan by providing both the on-street and off-street bicycle ways recommended in that plan. The former should be provided as the arterial streets and highways concerned are constructed and reconstructed.

Finally, in this respect, it is recommended that the Southeastern Wisconsin Regional Planning Commission, with funding provided by the Wisconsin Department of Transportation, prepare a local planning guide designed to illustrate transit- and pedestrian-friendly land use development practices. It is further recommended that the University of Wisconsin-Extension use the guide in its educational programming.

Public Transit

Under the plan, rapid transit service by buses would operate over freeway lanes providing relatively fast and convenient commuter transit service in the major travel corridors of the Region. A grid pattern of express transit routes by buses operating in mixed traffic over arterial streets would be provided primarily within Milwaukee County. The plan recognizes that both rapid and express transit services could be provided in the future over fixed-guideway facilities pending the outcome of major investment studies. The preliminary recommended plan also proposes to expand and improve local public transit service within the Kenosha, Milwaukee, and Racine urbanized areas, including eastern Waukesha County. A suggested prioritization of the transit improvement and expansion recommendations is set forth in SEWRPC report <u>Assessment of Conformity of the New Year 2010 Regional Transportation System Plan and the 1995 through 1997 Transportation Improvement Program with Respect to the State of Wisconsin Air Quality Implementation Plan.</u>

To the extent that the recommended transit services are to be provided within the geographic limits of a single county, the county itself provides an adequate institutional structure for the provision of such services. Counties, working cooperatively with local units of government in the county, have all the authority needed to assume plan implementation responsibilities with respect to transit. A significant number of the rapid transit facilities and services recommended in the plan, however, extend geographically over two or more counties. This is true in that bus rapid transit facilities and services and potential commuter-rail passenger service are proposed in the plan to extend from Milwaukee County into Ozaukee, Washington, and Waukesha Counties and also south along the Lake Michigan shoreline from Milwaukee County through Racine and Kenosha Counties to potential connections with Chicago-oriented commuter-rail passenger service. The potential commuter-rail passenger service extension between Burlington and Antioch, Illinois, would also involve two counties within the Region, Racine and Kenosha.

The proposed Southeastern Wisconsin Regional Transportation Authority, as described earlier in this chapter, would have provided an ideal institutional structure for providing these multicounty rapid transit services. There was, however, little political support evidenced in 1993 among the seven County Boards for the creation of such a regional transportation authority. Consequently, absent a change in that political position among at least several of the County Boards concerned and absent a position by the Wisconsin Department of Transportation that it would assume responsibility for the engineering, management, and operation of multi-county rapid transit services, the only alternative for implementation of the rapid transit elements of the regional transportation plan appears to be through intergovernmental agreements on the part of the counties concerned. Some relatively

modest examples of this type of approach to providing transit services that extend across county boundaries existed in 1991 within the Region in the form of agreements between Milwaukee and Waukesha Counties to extend local Milwaukee County bus routes into eastern Waukesha County and to provide freeway flyer bus services between communities in Waukesha County and the Milwaukee central business district (CBD). Similarly, in the Lake Michigan shoreline south corridor, the City of Racine has worked cooperatively with the City of Kenosha and Milwaukee County in providing express bus transit service between Kenosha, Racine, and Milwaukee.

The structure of the transit services recommended in the regional plan makes it important that, if a regional transportation authority is not created, and if the Wisconsin Department of Transportation does not assume engineering, management, and operational responsibilities for the multi-county rapid transit services recommended in the plan, the seven counties in the Region assume basic responsibility for the provision of public transit services. This is to be expected, given the areawide nature of the services provided not only within the Milwaukee, Racine, and Kenosha urbanized areas, but also between those urbanized areas and outlying communities as well. Milwaukee and Waukesha Counties have already assumed public transit responsibilities. Transit services in the other five counties to date, however, have been sponsored not by the counties but by individual communities.

Recent State legislation has made it more difficult for individual communities, as opposed to counties, to provide transit services on an areawide basis.⁷ That legislation prohibits a local public transit operator, such as a city, from providing transit services outside its corporate limits without formal agreements between the city and the neighboring local units of government. Any such services provided as of the effective date of that legislation, April 28, 1994, are exempted from this new law. Clearly, then, bringing about the areawide public transit service recommendations contained in the regional plan has been made more difficult by the enactment of this new law, which augurs badly for providing even local public transit services at the county level of government.

⁷See 1993 Wisconsin Act 279.

The issue of the lack of an appropriate institutional structure to provide the engineering, management, and operational capabilities necessary to provide a true rapid transit system in Southeastern Wisconsin must be addressed as the conduct of the several major investment studies attendant to the development of the rapid transit system envisioned in the plan are undertaken. For example, one of the major investment studies pertains to the provision of rapid transit services in the south travel corridor extending from Milwaukee to Racine and Kenosha with a potential interchange with the Chicago-oriented commuter-rail service that currently terminates in Kenosha. This major investment study will have to examine in detail how best to provide rapid transit services in this rapidly urbanizing corridor and, in particular, whether such services should be provided by buson-freeway, bus-on-busway, or commuter-rail service. Should that major investment study conclude that bus-on-freeway or bus-on-busway is the appropriate rapid transit alternative for the travel corridor, then the lack of an areawide institutional structure to implement the services is less problematic than if commuter-rail service were to be found to be the preferred alternative. This is because the rapid transit bus service would be provided over either the freeways or over busways attendant to the freeways and there is an appropriate institutional structure. the Wisconsin Department of Transportation, in place to engineer, manage, and operate all of the fixed-way facilities. The rolling stock and services could be provided through an intergovernmental contract between the three counties concerned. Should that major investment study, however, identify commuter-rail service as the alternative for implementation, then it will be necessary to either assign the responsibilities for engineering, managing, and operating the commuter-rail system to the Wisconsin Department of Transportation or to create a new areawide institutional structure in the form of a regional transit or transportation authority to assume those important functions.

It should be noted that the plan implementation problems relating to institutional structure pertain exclusively to the areawide rapid transit services which are proposed to extend into all seven counties of the Region. These services are identified on the recommended plan maps in Chapter XI in red lines, akin to the red lines on the maps which depict the recommended State trunk highway system. The express level of transit services recommended in the plan are located largely, although not entirely, within the limits of a single county and, accordingly, can be implemented by using the county level of government as the transit institutional structure. The few exceptions to this rule consist of express bus service proposed between Milwaukee and Waukesha Counties and between Racine and Kenosha Counties. Those services could be provided by intergovernmental contracts between the counties concerned. All the recommended potential light-rail express transit services are located within Milwaukee County. Consequently, Milwaukee County is the logical provider of those services.

Given the foregoing, recognizing that there will be uncertainty at least over the near-term future as to the institutional structure which the State and local elected officials determine should be used for providing needed public transit services in Southeastern Wisconsin, the following specific plan implementation recommendations are recommended. In considering these recommendations, it should be understood that the conduct of the major investment studies recommended later in this chapter may result in changes to these plan implementation recommendations. Moreover, it should be understood that the following recommendations, which would provide for greater county responsibilities so as to more efficiently implement the public transit element of the plan, may be expected to evolve over time. In particular, the larger cities within the Region that currently provide transit service may be expected to continue to do so until such time as they and their respective counties have cooperatively conducted appropriate studies and have found that the counties are ready, willing, and able to assume responsibility for the transit function.

1. It is recommended that Milwaukee County continue in its lead role as the major provider of public transit services in Southeastern Wisconsin. Milwaukee County should continue to provide rapid, express, and local transit services as recommended in the plan. In addition, it is recommended that Milwaukee County enter into appropriate agreements with Kenosha and Racine Counties for the joint provision of rapid transit services, consisting initially of bus service, south along the Lake Michigan shoreline; with Ozaukee County for the joint provision of such rapid transit services north along the Lake Michigan shoreline; with Washington County for the joint provision of such rapid transit services in the northwest travel corridor; and with Waukesha County for the joint provision of bus express and bus rapid transit services in the northwest, east-west, and southwest travel corridors.

- 2. It is recommended that Ozaukee County enter into an agreement with Milwaukee County for the joint provision of bus rapid transit services north along the Lake Michigan shoreline. It is also recommended that Ozaukee County, after preparing a detailed transit development plan to refine and detail the recommendations of the regional system plan, work cooperatively with the local units of government in the County to implement the transit services recommended in that plan.
- 3. It is recommended that Washington County enter into an agreement with Waukesha and Milwaukee Counties for the joint provision of rapid transit services in the northwest travel corridor. It is also recommended that Washington County, after preparing a detailed transit development plan to refine and detail the recommendations of the regional system plan to refine and detail the recommendations of the regional system plan, work cooperatively with the local units of government in the County to implement the transit services recommended in that plan.
- 4. It is recommended that Waukesha County continue to provide rapid and express transit services as recommended in the plan, entering into agreements with Milwaukee and Washington Counties for the joint provision of rapid transit services in the northwest travel corridor and agreements with Milwaukee County for the joint provision of rapid transit services in the east-west and southwest travel corridors. It is also recommended that Waukesha County, working cooperatively with the local units of government in the County, implement the local transit services recommended in the plan. In this respect, it is further recommended that Waukesha County, working cooperatively with the

City of Waukesha, conduct a study to explore the proposed transfer of the public transit function from the city to the county level of government.

- 5. It is recommended that Racine County enter into an agreement with Milwaukee and Kenosha Counties for the joint provision of bus rapid transit services south along the Lake Michigan shoreline and an agreement with Kenosha County to explore the potential for commuter-rail passenger service in the corridor from Burlington to Chicago through Antioch, Illinois. It is also recommended that Racine County, working cooperatively with the local units of government in the County, implement the local transit services recommended in the plan. Finally, it is recommended that Racine County, working cooperatively with the City of Racine, conduct a study to explore the proposed transfer of the public transit function from the city to the county level of government. Should that transfer become a reality, it is recommended that Racine County become the designated recipient of Federal transit funds in the County.
- 6. It is recommended that Kenosha County enter into an agreement with Racine and Milwaukee Counties for the joint provision of bus rapid transit services south along the Lake Michigan shoreline and an agreement with Racine County to explore the potential for commuter-rail passenger service in the corridor from Burlington to Chicago through Antioch, Illinois. It is also recommended that Kenosha County, working cooperatively with the local units of government in the County, implement the local transit services recommended in the plan. Toward this end, it is recommended that Kenosha County become the designated recipient of Federal transit funds in the County. Finally, it is recommended that Kenosha County, working cooperatively with the City of Kenosha, conduct a study to explore the proposed transfer of the public transit function from the city to the county level of government. Should that transfer become a reality, it is recommended that Kenosha County become the designated recipient of Federal transit funds in the County.

- 7. It is recommended that Walworth County participate in a study to explore the potential for commuter-rail passenger service in the corridor from Walworth to Chicago through to Fox Lake, Illinois, and, depending upon the outcome of that study, assume responsibility for sponsorship of any recommended services.
- 8. It is recommended that the three cities in the Region that currently provide fixedroute public transit services, the Cities of Kenosha, Racine, and Waukesha, continue to provide those services, but, over time, as the city transit systems are increasingly called upon to serve areas beyond their municipal limits, enter into cooperative studies with their respective counties to explore the transfer of the transit function to the county level of government and, in so doing, determine the manner in which the transit function could be most costefficiently administered.
- 9. It is recommended that the Cities of Hartford, Port Washington, West Bend, and Whitewater continue to provide demandresponsive shared-ride taxicab public transit services in their communities, seeking to coordinate these efforts with any county transit programs which may be established. Depending upon the outcome of detailed transit development planning by individual counties, it may be desirable to transfer responsibility for operation of the shared-ride taxicab services to the county level of government.
- 10. It is recommended that the Wisconsin Department of Transportation advise the Governor, after consultation with the counties and communities concerned and upon assumption by the counties of the transit function, to designate Kenosha and Racine Counties as the recipients of Federal transit funds for the Kenosha and Racine urbanized areas, respectively.
- 11. It is recommended that each of the seven counties, as well as any affected local transit operators, work cooperatively with the Wisconsin Department of Transportation and the Regional Planning Commission in the conduct of the major investment

studies attendant to rapid and express transit services identified in a later section of this chapter.

Arterial Streets and Highways

The arterial street and highway system envisioned in the recommended plan would consist of 3,607 route-miles and 10,303 lane-miles of facilities. The plan recommends the construction of 131 route-miles and 337 lane-miles of new facilities within the Region. The plan also recommends the widening or other improvement of 448 route-miles, adding 854 lane-miles to the 1,023 lane-miles already provided on those routes. The plan also calls for pavement resurfacing and bridge and interchange restoration and reconfiguration work necessary to maintain and appropriately modernize the remaining 3,028 route-miles and 8,089 lane-miles of planned arterial facilities, including, importantly, the Milwaukee area freeway system.

<u>Jurisdictional Recommendations</u>: Jurisdictional classification is important to arterial street and highway plan implementation. It establishes which level of government, be it State, county, or local, has, or should have, responsibility for the design, construction, maintenance, and operation of each segment of the total street and highway system. Jurisdictional classification is intended to group all streets and highways logically into subsystems under the jurisdiction of a given level of government.

Jurisdictional classification is the first step in implementing arterial street and highway recommendations. Upon completion of the initial regional transportation system plan in 1966, detailed county jurisdictional highway system plans were prepared. These plans were updated as part of the second-generation regional transportation system plan, completed in 1978. As an integral part of the preparation of this, the thirdgeneration plan, the county jurisdictional highway system plans have again been updated and revised. The recommended jurisdictional arterial street and highway systems for the seven counties are shown on Maps 133 through 139.

Table 260 sets forth the distribution of planned arterial street and highway mileage among each jurisdictional subsystem within the Region and within each county of the Region. By the year 2010, about 1,167 miles, or about 32 percent of the planned arterial system, are recommended to

Table 260

	St	State County		Local		Total		
County	Miles	Percent of Total						
Kenosha	103	8.8	203	12.8	49	5.7	355	9.8
Milwaukee	220	18.9	184	11.6	393	45.9	797	22.1
Ozaukee	72	6.2	155	9.8	78	9.1	305	8.5
Racine	160	13.7	156	9.8	108	12.6	424	11.7
Walworth	223	19.1	239	15.1	22	2.6	484	13.4
Washington	159	13.6	234	14.8	75	8.8	468	13.0
Waukesha	230	19.7	413	26.1	131	15.3	774	21.5
Region	1,167	100.0	1,584	100.0	856	100.0	3,607	100.0

DISTRIBUTION OF ARTERIAL STREET AND HIGHWAY MILEAGE WITHIN IN THE REGION BY COUNTY AND JURISDICTIONAL CLASSIFICATION: 2010 RECOMMENDED PLAN

Source: Wisconsin Department of Transportation and SEWRPC.

be classified as State trunk highways, including connecting streets; about 1,584 miles, or 44 percent, are recommended to be classified as county trunk highways; and the remaining 856 miles, or about 24 percent, are recommended to be classified as local arterials. The jurisdictional transfers proposed under the recommended plan in each county are summarized in Tables 261 through 267 and are displayed on Maps 140 through 146. The planned distributions among the three jurisdictional classifications of arterial mileage within each county are described below.

<u>Kenosha County</u>: By the year 2010, about 103 miles, or nearly 29 percent, of the planned arterial system in Kenosha County are recommended to be classified as State trunk highways, including connecting streets; about 203 miles, or 57 percent, are recommended to be classified as county trunk highways; and the remaining 49 miles, or about 14 percent, are recommended to be classified as local arterials. Map 133 shows the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Kenosha County. Table 261 and Map 140 show the recommended transfers in highway system jurisdictional responsibility in Kenosha County.

<u>Milwaukee County</u>: By the year 2010, about 220 miles, or nearly 28 percent, of the planned arterial system in Milwaukee County are recommended to be classified as State trunk highways, including connecting streets; about 184 miles, or 23 percent, are recommended to be classified as county trunk highways; and the remaining 393 miles, or about 49 percent, are recommended to be classified as local arterials. Map 134 shows the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Milwaukee County. Table 262 and Map 141 show the recommended transfers in highway system jurisdictional responsibility in Milwaukee County.

<u>Ozaukee County</u>: In the year 2010, about 72 miles, or nearly 24 percent, of the planned arterial system in Ozaukee County are recommended to be classified as State trunk highways, including connecting streets; about 155 miles, or 51 percent, are recommended to be classified as county trunk highways; and the remaining 78 miles, or about 25 percent, are recommended to be classified as local arterials. Map 135 shows the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Ozaukee County. Table 263 and Map 142 show the recommended transfers in highway system jurisdictional responsibility in Ozaukee County.

<u>Racine County</u>: In the year 2010, about 160 miles, or nearly 38 percent, of the planned arterial system in Racine County are recommended to be classified as State trunk highways, including connecting streets; about 156 miles, or 37 percent, are recommended to be classified as county trunk highways; and the remaining 108 miles, or about 25 percent, are recommended to be classified as local arterials. Map 136 shows Map 133



PROPOSED JURISDICTION OF ARTERIAL STREETS AND HIGHWAYS INCLUDED IN THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO KENOSHA COUNTY: 2010

The level of government recommended to have the responsibility for the design, construction, maintenance, and operation of each segment of the arterial street and highway system in Kenosha County is shown on the accompanying map. By the year 2010, the arterial street and highway system in Kenosha County may be expected to total 355 miles. About 103 miles, or nearly 29 percent of planned arterial mileage, are recommended to be classified as State trunk highways, including connecting streets; about 203 miles, or 57 percent, are recommended to be classified as county trunk highways; and the remaining 49 miles, or about 14 percent, are recommended to be classified as local arterials.

571



Map 134 Inset



The level of government recommended to have the responsibility for the design, construction, maintenance, and operation of each segment of the arterial street and highway system in Milwaukee County is shown on the accompanying map. By the year 2010, the arterial street and highway system in Milwaukee County may be expected to total 797 miles. About 220 miles, or 28 percent of planned arterial mileage, are recommended to be classified as State trunk highways, including connecting streets; about 184 miles, or 23 percent, are recommended to be classified as county trunk highways; and the remaining 393 miles, or about 49 percent, are recommended to be classified as local arterials.

Source: SEWRPC.

the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Racine County. Table 264 and Map 143 show the recommended transfers in highway system jurisdictional responsibility in Racine County.

<u>Walworth County</u>: In the year 2010, about 223 miles, or nearly 46 percent, of the planned arterial system in Walworth County are recommended to be classified as State trunk highways, including connecting streets; about 239 miles, or 49 percent, are recommended to be classified as county trunk highways; and the remaining 22 miles, or about 5 percent, are recommended to be classified as local arterials. Map 137 shows the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Walworth County. Table 265 and Map 144 show





Map 135

Summer Sugar

MILWAUKEE

CO

R. 21 F

Map 133



PROPOSED JURISDICTION OF ARTERIAL STREETS AND HIGHWAYS INCLUDED IN THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO KENOSHA COUNTY: 2010

The level of government recommended to have the responsibility for the design, construction, maintenance, and operation of each segment of the arterial street and highway system in Kenosha County is shown on the accompanying map. By the year 2010, the arterial street and highway system in Kenosha County may be expected to total 355 miles. About 103 miles, or nearly 29 percent of planned arterial mileage, are recommended to be classified as State trunk highways, including connecting streets; about 203 miles, or 57 percent, are recommended to be classified as county trunk highways; and the remaining 49 miles, or about 14 percent, are recommended to be classified as local arterials.

571



PROPOSED JURISDICTION OF ARTERIAL STREETS AND HIGHWAYS INCLUDED IN THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO WALWORTH COUNTY: 2010

LEGEND

FREEWAY





The level of government recommended to have the responsibility for the design, construction, maintenance, and operation of each segment of the arterial street and highway system in Walworth County is shown on the accompanying map. By the year 2010, the arterial street and highway system in Walworth County may be expected to total 484 miles. About 223 miles, or 46 percent of planned arterial mileage, are recommended to be classified as State trunk highways, including connecting streets; about 239 miles, or 49 percent, are recommended to be classified as county trunk highways; and the remaining 22 miles, or about 5 percent, are recommended to be classified as local arterials.

WALWORTH CO.

WISCONSIN LINN

4

Source: SEWRPC.



PROPOSED JURISDICTION OF ARTERIAL STREETS AND HIGHWAYS INCLUDED IN THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO WASHINGTON COUNTY: 2010

Map 138

The level of government recommended to have the responsibility for the design, construction, maintenance, and operation of each segment of the arterial street and highway system in Washington County is shown on the accompanying map. By the year 2010, the arterial street and highway system in Washington County may be expected to total 468 miles. About 159 miles, or 34 percent of planned arterial mileage, are recommended to be classified as State trunk highways, including connecting streets; about 234 miles, or 50 percent, are recommended to be classified as county trunk highways; and the remaining 75 miles, or about 16 percent, are recommended to be classified as local arterials.



The level of government recommended to have the responsibility for the design, construction, maintenance, and operation of each segment of the arterial street and highway system in Waukesha County is shown on the accompanying map. By the year 2010, the arterial street and highway system in Waukesha County may be expected to total 774 miles. About

230 miles, or 30 percent of planned arterial mileage, are recommended to be classified as State trunk highways, including connecting streets; about 413 miles, or 53 percent, are

recommended to be classified as county trunk highways; and the remaining 131 miles, or about 17 percent, are recommended to be classified as local arterials.

Source: SEWRPC,

578

Map 139

PROPOSED JURISDICTION OF ARTERIAL STREETS AND HIGHWAYS INCLUDED IN THE

Map 140



PROPOSED CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS UNDER THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO KENOSHA COUNTY

LEGEND

TRANSFERS TO:

COUNTY TRUNK HIGHWAY SYSTEM

LOCAL (NONARTERIAL) SYSTEM



The proposed changes in jurisdictional responsibility for arterial streets and highways in Kenosha County are shown on the accompanying map. In 1991, the State trunk highway system totalled 140 miles, and the local arterial system totalled 59 miles. By the year 2010, through the transfer of those facilities shown on the accompanying map and listed in Table 261, the State trunk highway system would total 103 miles, the county trunk highway system would total 103 miles, the county trunk highway system would total 49 miles.

Source: SEWRPC.

Table 261

CHANGES IN JURISDICTIONAL RESPONSIBILITY OF ARTERIAL STREETS AND HIGHWAYS IN KENOSHA COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

	lurisdictional	Responsibility			
	Evisting	Blenned	Enaility	Erom	То
Civil Division	Existing	Planned	Facility	From	10
Town of Brighton	State trunk highway	County trunk highway	STH 75	North town line	South town line
1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	Local nonarterial	County trunk highway	1st Street	224th Avenue	East town line
	County trunk highway	Local nonarterial	СТНВ	North town line	
	County trunk highway	Local nonarterial			STH JB
	County trunk highway	Local nonarterial	CTH EW	CTHUB	стн к
	County trunk highway	Local nonarterial	CTH NN	стн к	East town line
	County trunk highway	Local nonarterial	СТН РН	CTH JB	STH 75
Taura of Briatal	Least truck artarial	County trunk highway	128th Street	A paint about 1.0 mile	
I OWN OF Bristol	Local trunk arterial	County trunk highway	128th Street	west of USH 45	050 45
	New facility	County trunk highway	CTH O extension	184th Avenue extended	104th Street
	County trunk highway	Local poparterial	CTH D	стн к	стн с
	County trunk highway	Local nonarterial	СТН U	стн с	South town line
	County trunk highway	Local nonarterial	стн v	West town line	USH 45
	County trunk highway	Local nonarterial	СТН АН	USH 45	CTH D
	County trunk highway	Local nonarterial	стн сј	USH 45	СТН U
	County trunk highway	Local nonarterial	CTH JS	West town line	USH 45
	County trunk highway	Local nonarterial	СТН МВ	North town line	South town line
Town of Paris	Local nonarterial	County trunk arterial	1st Street	West town line	USH 45
	County trunk highway	Local nonarterial	СТН А	STH 142	IH 94
	County trunk highway	Local nonarterial	СТН D	STH 142	стн к
	County trunk highway	Local nonarterial	СТН МВ	СТН А	стн к
	County trunk highway	Local nonarterial	CTH NN	West town line	USH 45
	County trunk highway	Local nonarterial	CTH UE	STH 142	CTH N
Town of Randall	Local trunk highway	County trunk highway	Benedict Street	West town line	СТН Р
	New facility	County trunk highway	CTH KD extension	CTH EM	СТН F
	New facility	County trunk highway	CTH F extension	СТН О	СТН F
	County trunk highway	Local nonarterial	СТН F	СТНО	CTH F extension
	County trunk highway	Local nonarterial	СТН О	CTH F extension	North town line
	County trunk highway	Local nonarterial	CTH W	СТН НМ	322nd Avenue
	County trunk highway	Local nonarterial	CTH EM	CTH W	CTH Z
	County trunk highway	Local nonarterial			CIH KD extension
	County trunk highway	Local nonarterial			North town line
	County trunk highway	Local nonarterial		CTH W	CTH F
	county trank nightray		0711 AU		0711.04
Town of Salem	New facility	County trunk highway			
	Least trunk highway	County trunk nighway	SIH 75 Reak Lake Read	South town line	
	Local trunk highway	County trunk arterial	264th Avenue		
	County trunk arterial	Local poparterial	CTH F	CTH SA	STH 50
	County trunk arterial	Local nonarterial	CTH F	CTH AH extension	CTH SA
	County trunk arterial	Local nonarterial	CTH V	стн с	East town line
	County trunk arterial	Local nonarterial	СТН АН	97th Street	89th Street
	County trunk arterial	Local nonarterial	CTH FR	СТН F	стн w
	County trunk arterial	Local nonarterial	CTH JF	Rock Lake Road	STH 83
	County trunk arterial	Local nonarterial	CTH JS	СТН V	East town line
	County trunk arterial	Local nonarterial	CTH SA	264th Avenue	STH 83
	County trunk arterial	Local nonarterial	CTH SA	CIH F	СІНАН
Town of Somers	County trunk highway	Local trunk highway	CTH L	СТН Н	СТН G
	County trunk arterial	Local nonarterial	CTH A	IH 94	STH 32
	County trunk highway	Local nonarterial	CTH EA	CTH KR	STH 142
	County trunk highway	Local nonarterial	CTH G	CTH E	CTH KR
	County trunk highway	Local nonarterial		IH 94	CTH S
	County trunk highway	Local nonarterial		51H 31	
Town of Wheatland	Local nonarterial	County trunk highway	Karcher Road	Fish Hatchery Road	СТН КД
	County trunk highway	Local nonarterial	стн о	South town line	STH 50
	County trunk highway	Local nonarterial	CTH W	СТН К	CTH JB
	County trunk highway	Local nonarterial	CTH FR	South town line	CTH W
1	County trunk highway	Local nonarterial		South town line	CTHW

Table 261 (continued)

	Jurisdictional	Responsibility				
Civil Division	Existing	Planned	Facility	From	То	
Village of Pleasant Prairie	Local trunk highway Local trunk highway	County trunk highway County trunk highway	Bain Station Road 128th Street	CTH C A point about 0.4 mile west of STH 31	STH 31 STH 32	
	New Facility County trunk highway County trunk highway New Facility County trunk highway	County trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial	122nd Street extension CTH EZ CTH EZ 51st Avenue extension CTH ML	CTH H 128th Street CTH ML STH 165 CTH H	CTH ML CTH ML North corporate limit 93rd Street 122nd Street extension	
City of Kenosha	Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway State trunk highway New facility	State trunk highway State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway	Roosevelt Road 63rd Street Washington Road 22nd Avenue 30th Avenue 60th Street 75th Street 85th Street extension	75th Street Roosevelt Road 39th Avenue South corporate limit Roosevelt Road West corporate limit Roosevelt Road 39th Avenue	63rd Street STH 32 STH 32 North corporate limit Washington Road STH 32 STH 32 30th Avenue	

Source: SEWRPC.

the recommended transfers in highway system jurisdictional responsibility in Walworth County.

<u>Washington County</u>: In the year 2010, about 159 miles, or nearly 34 percent, of the planned arterial system in Washington County are recommended to be classified as State trunk highways, including connecting streets; about 234 miles, or 50 percent, are recommended to be classified as county trunk highways; and the remaining 75 miles, or about 16 percent, are recommended to be classified as local arterials. Map 138 shows the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Washington County. Table 266 and Map 145 show the recommended transfers in highway system jurisdictional responsibility in Washington County.

<u>Waukesha County</u>: In the year 2010, about 230 miles, or nearly 30 percent, of the planned arterial system in Waukesha County are recommended to be classified as State trunk highways, including connecting streets; about 413 miles, or 53 percent, are recommended to be classified as county trunk highways; and the remaining 131 miles, or about 17 percent, are recommended to be classified as local arterials. Map 139 shows the recommended plan year 2010 jurisdictionally classified arterial street and highway system for Waukesha County. Table 267 and Map 146 show the recommended transfers in highway system jurisdictional responsibility in Waukesha County.

Jurisdictional Transfers: It is recommended that the county boards of the seven constituent counties in the Region, upon recommendation of the county highway and transportation committees and in cooperation with the Wisconsin Department of Transportation, seek realignment of the State trunk, county trunk, and local trunk systems to the recommended regional transportation system plan. It is further recommended that the Wisconsin Department of Transportation seek, in cooperation with the seven county boards and appropriate local officials, realignment of the State trunk, county trunk, and local trunk systems consistent with the recommended regional transportation system plan.

<u>Functional Improvement Recommendations</u>: The plan provides for three types of functional improvement: system expansion, or the construction of new arterial facilities; system improvement, or the widening of facilities to provide significant additional capacity; and system preservation, or the resurfacing and reconstruction necessary to properly maintain and modernize existing arterial facilities. As previously indicated, the plan would provide for the con-



The proposed changes in jurisdictional responsibility for arterial streets and highways in Milwaukee County are shown on the accompanying map. In 1991, the State trunk highway system in Milwaukee County totalled 251 miles, the county trunk highway system totalled 82 miles, and the local arterial system totalled 442 miles. By the year 2010, through the jurisdictional transfers identified on the accompanying map and listed on Table 262, the State trunk highway system would total 220 miles, the county trunk highway system would total 184 miles, and the local arterial system would total 393 miles.

Source: SEWRPC.
CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS IN MILWAUKEE COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

	Jurisdictional Responsibility				
Civil Division	Existing	Planned	Facility	From	То
Village of Bayside	County trunk highway	Local trunk highway	Port Washington Road	Ozaukee county line	South corporate limits
Village of Brown Deer	County trunk highway State trunk highway County trunk highway	State trunk highway County trunk highway Local trunk highway	CTH D (Teutonia Avenue) STH 57 (Green Bay Road) CTH G (Sherman Boulevard)	STH 57 STH 57 Green Bay Road	South corporate limits South corporate limits South corporate limits
City of Cudahy	Local trunk highway Local trunk highway	County trunk highway State trunk highway	Layton Avenue Pennsylvania Avenue	STH 62 Layton Avenue	STH 32 (Lake Drive) South corporate limits
Village of Fox Point	County trunk highway	Local trunk highway	Port Washington Road	North corporate limits	South corporate limits
City of Franklin	State trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway New facility Local trunk highway Local trunk highway Local trunk highway County trunk highway	County trunk highway State trunk highway Local nonarteriał Local runk highway County trunk highway County trunk highway County trunk highway Local nonarterial County trunk highway Local nonarterial	27th Street Rawson Avenue CTH J (Northcape Road) CTH K (Crystal Ridge Lane) CTH MM (St. Martins Road) Puetz Road Puetz Road Puetz Road Hunting Park Drive 51st Street CTH A	North corporate limits STH 100 Waukesha county line Rawson Avenue CTH J STH 100 CTH U Hunting Park Drive 51st Street North corporate limits Milwaukee County House of Corrections	Racine county line East corporate limits Forest Home Avenue 76th Street STH 100 76th Street Hunting Park Drive East corporate limits Puetz Road STH 100 (Ryan Road) STH 100 (Ryan Road)
City of Glendale	State trunk highway County trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway	County trunk highway Local trunk highway Local trunk highway County trunk highway County trunk highway County trunk highway	STH 57 (Green Bay Road) Port Washington Road Port Washington Road Mill Road Silver Spring Drive Hampton Avenue	Range Line Road North corporate limits Daphne Road West corporate limits West corporate limits West corporate limits	South corporate limits Daphne Road Hampton Avenue STH 57 East corporate limits East corporate limits
Village of Greendale	Local trunk highway	County trunk highway	51st Street	North corporate limits	South corporate limits
City of Greenfield	Local trunk highway State trunk highway State trunk highway State trunk highway County trunk highway State trunk highway Local trunk highway County trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway County trunk highway Local trunk highway	124th Street STH 100 27th Street Layton Avenue 92nd Street STH 24 (Forest Home Avenue) 51st Street CTH T (Beloit Road)	North corporate limits North corporate limits Layton Avenue Waukesha county line North corporate limits South corporate limits STH 36 (Loomis Road) Waukesha county line	Layton Avenue IH 43 South corporate limits STH 100 Forest Home Avenue North corporate limits South corporate limits East corporate limits
Village of Hales Corners	State trunk highway State trunk highway	County trunk highway County trunk highway	STH 24 (Janesville Road) STH 24 (Forest Home Avenue)	Waukesha county line STH 100	Forest Home Avenue East corporate limits
City of Milwaukee	Local trunk highway New facility Local trunk highway State trunk highway County trunk highway Local trunk highway State trunk highway State trunk highway Local trunk highway	County trunk highway County trunk highway County trunk highway State trunk highway State trunk highway County trunk highway	Boundary Road Boundary Road extension 107th Street 76th Street CTH D (Teutonia Avenue) CTH D (Teutonia Avenue) STH 57 (Green Bay Road) STH 175 (Appleton Avenue) USH 41 (Appleton Avenue) USH 41 (Appleton Avenue) Lisbon Avenue Silver Spring Drive Hampton Avenue Burleigh Street Hopkins Street Locust Street North Avenue State Street Lover's Lane 27th Street Lincoln Avenue Forest Home Avenue	County Line Road STH 100 STH 100 USH 41 North corporate limits Good Hope Road Silver Spring Drive Waukesha county line USH 45 North Avenue 68th Street 60th Street West corporate limits Burleigh Street Hopkins Street West corporate limits West corporate limits Silver Spring Drive Teutonia Avenue West corporate limits 27th Street	STH 100 STH 145 USH 41 South corporate limits Good Hope Road Hampton Avenue Capitol Drive USH 45 76th Street 27th Street East corporate limits Hopkins Street Locust Street IH 43 USH 41 35th Street South corporate limits Highland Boulevard Bay Street Lincoln Avenue
	Local trunk highway New facility New facility Local trunk highway Local trunk highway State trunk highway State trunk highway State trunk highway State trunk highway Local trunk highway Local trunk highway Local nonarterial State trunk highway Local trunk highway Local trunk highway	County trunk highway Local trunk highway Local trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway County trunk highway Local nonarterial	Muskego Avenue Canal Street extension Canal Street extension Greenfield Avenue Howell Avenue STH 38 (Howell Avenue) Oklahoma Avenue STH 38 (Chase Avenue) 6th Street 27th Street Layton Avenue 92nd Street 76th Street 84th Street Becher Street	Lincoln Avenue 6th Street USH 41 16th Street Lincoln Avenue Howell Avenue West corporate limits 6th Street National Avenue Layton Avenue West corporate limits Oklahoma Avenue Blue Mound Road Glenview Avenue Muskego Avenue	Greenfield Avenue Water Street 27th Street 1st Street STH 38 South corporate limits Kinnickinnic Avenue Howell Avenue Chase Avenue South corporate limits East corporate limits Howard Avenue South corporate limits Schlinger Avenue Forest Home Avenue

Table 262 (continued)

	Jurisdictional Responsibility		· · · ·		
Civil Division	Existing	Planned	Facility	From	То
City of Milwaukee (continued)	State trunk highway County trunk highway New facility County trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway New facility	County trunk highway Local trunk highway State trunk highway Local trunk highway Local trunk highway Local nonarterial County trunk highway County trunk highway County trunk highway	STH 24 (Forest Home Avenue) CTH T (Beloit Road) Lake Parkway CTH G (Sherman Boulevard) STH 57 (20th Street) Good Hope Road County Line Road 124th Street 124th Street	South corporate limits West corporate limits Lincoln Avenue North corporate limits Capitol Drive North Avenue Waukesha county line Waukesha county line Silver Spring Drive STH 100	Lincoln Avenue North corporate limits South corporate limits Mill Road North Avenue Highland Boulevard 107th Street East corporate limits South corporate limits STH 145
City of Oak Creek	State trunk highway State trunk highway State trunk highway County trunk highway Local trunk highway New facility	County trunk highway County trunk highway Local trunk highway State trunk highway County trunk highway Local trunk highway	STH 38 (Howell Avenue) 27th Street 13th Street CTH BB (Rawson Avenue) Puetz Road 10th/15th Avenue extension	North corporate limits North corporate limits North corporate limits West corporate limits West corporate limits STH 100	Rawson Avenue Racine county line Racine county line East corporate limits STH 32 Racine county line
Village of River Hills	State trunk highway	County trunk highway	STH 57 (Green Bay Road)	Bradley Road	South corporate limits
City of St. Francis	New facility New facility	State trunk highway Local trunk highway	Lake Parkway Howard Avenue extension	North corporate limits Thompson Avenue	South corporate limits Iowa Avenue
City of South Milwaukee	Local trunk highway Local trunk highway	State trunk highway State trunk highway	Rawson Avenue Nicholson Avenue	West corporate limits North corporate limits	STH 32 Rawson Avenue
City of Wauwatosa	State trunk highway Local trunk highway Local trunk highway Local trunk highway State trunk highway Local trunk highway Local trunk highway New facility Local trunk highway State trunk highway State trunk highway State trunk highway	County trunk highway County trunk highway Local trunk highway Local trunk highway	76th Street (Wauwatosa Avenue) Burleigh Street North Avenue Watertown Plank Road Glenview/Harwood/Harmonee Avenues State Street 124th Street 124th Street extension 124th Street STH 100 Glenview Avenue Glenview Avenue	North corporate limits Waukesha county line Waukesha county line Watertown Plank Road Harmonee Avenue North corporate limits Knoll Road Blue Mound Road North corporate limits Blue Mound Road Harwood Avenue	Menomonee River Parkway East corporate limits East corporate limits Glenview Avenue Menomonee River Parkway East corporate limits Knoll Road Blue Mound Road South corporate limits South corporate limits South corporate limits 84th Street
City of West Allis	New facility Local trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial County trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway	124th Street 124th Street STH 100 National Avenue Cleveland Avenue Lincoln Avenue 76th Street 76th Street CTH T (Beloit Road)	North corporate limits Greenfield Avenue North corporate limits Waukesha county line Waukesha county line National Avenue Greenfield Avenue North corporate limits South corporate limits	Greenfield Avenue South corporate limits South corporate limits Greenfield Avenue National Avenue East corporate limits South corporate limits Greenfield Avenue Oklahoma Avenue
Village of Whitefish Bay	Local trunk highway	County trunk highway	Silver Spring Drive	West corporate limits	STH 32
	Local truthe highway		nampton Avenue	weat corporate innits	311 32

Source: SEWRPC.

struction of 131 route-miles and 337 lane-miles of new facilities within the Region; and the widening or other improvement of 448 route-miles, adding 854 lane-miles to the 1,023 lane-miles already provided on those routes. The plan also calls for pavement resurfacing and bridge and interchange restoration and reconfiguration work necessary to maintain and appropriately modernize the remaining 3,028 route-miles and 8,089 lane-miles of planned arterial facilities. The planned functional improvements to the regional arterial street and highway system are shown on a county-by-county basis on Maps 125 through 131 in Chapter XI of this report. The mileage of planned system expansion, system improvement, and system preservation within each county are described below.

<u>Kenosha County</u>: The arterial street and highway system in Kenosha County is recommended to be expanded by 37 miles, or 12 percent, from 318 miles in 1991 to 355 miles in the year 2010 (see Map 125, page 536). The increase in arterial mileage would come about through the construction of nine miles of new facility and through the conversion of 28 miles of previously nonarterial



CEDARRORS

EASANT

PIGEON

REISTA

STIRC ST T 10 N

(CD)

20

CO. E.E ů,

OLD PORT

MILWAUKEE

CÓ

ZEDLER

KARANA STATISTICS

SAYSIDE

PROPOSED CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS UNDER THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO OZAUKEE COUNTY



streets and highways in Ozaukee County are shown on the accompanying map. In 1991, the State trunk highway system in Ozaukee County totalled 101 miles, the county trunk highway system totalled 97 miles, and the local arterial system totalled 91 miles. By the year 2010, through the jurisdictional transfers identified on the accompanying map and listed on Table 263, the State trunk highway system would total 72 miles, the county trunk highway system would total 155 miles, and the local arterial system would total 78 miles.

CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS IN OZAUKEE COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

	lurisdictional	Besnonsibility			
Civil Division	Evisting	Planned	Facility	From	То
Town of Polgium	State truck bigbway	County trunk highway		West town line	South town line
	Local nonarterial	County trunk highway	CTH KW	North town line	South town line
	County trunk highway	Local nonarterial	СТНВ	South town line	СТНА
Town of Cedarburg	County trunk highway	State trunk highway	CTH N	STH 143	Pioneer Road
	State trunk highway	County trunk highway	STH 143 Mante Road extension	Cedar Creek Boad	Village of Grafton
	State trunk highway	Local trunk highway	STH 57	City of Cedarburg	East town line
	State trunk highway	Local trunk highway	STH 143	CTH N	City of Cedarburg
	County trunk highway	Local trunk highway	СТНС	Granville Road City of Cedarburg	STH 60
	County trunk highway	Local trunk highway	СТНТ	East town line	City of Cedarburg
	County trunk highway	Local trunk highway	СТНТ	CTH N	City of Cedarburg
Town of Fredonia	State trunk highway	County trunk highway	STH 84	West town line	Village of Fredonia
	State trunk highway	County trunk highway	STH 84 CTH AA	Village of Fredonia CTH I	East town line
Town of Grafton	State truck biobway	Coupty trunk highway	STH 57	IH 43	Village of Grafton
Town of Granton	New facility	Local trunk highway	Maple Road extension	Cedar Creek Road	Village of Grafton
	County trunk highway	Local trunk highway	СТНТ	West town line	стн w
Town of Port Washington	State trunk highway	County trunk highway	STH 84	City of Port Washington	North town line
	Local trunk highway	County trunk highway	CTH CC	City of Port Washington	STH 32
	County trunk highway	Local nonarterial	СТНВ	CTHIL	North town line
	County trunk highway	Local nonarterial	СТНКК	Spring Street	City of Port Washington
Town of Saukville	New facility County trunk highway	Local trunk highway Local nonarterial	Cold Springs Road extension CTH O	стн о стн і	Village of Saukville Village of Saukville
Village of Fredonia	State trunk highway County trunk highway	County trunk highway Local nonarterial	STH 84 CTH AA	STH 57 STH 84	CTH AA STH 84
Village of Grafton	State trunk highway	Local trunk highway	STH 57	West corporate limit	East corporate limit
Village of Saukville	New facility County trunk highway	Local trunk highway Local nonarterial	Cold Springs Road CTH O	STH 33 STH 33	North corporate limit North corporate limit
Village of Thiensville	State trunk highway	Local trunk highway	STH 57	South corporate limit	North corporate limit
City of Cedarburg	County trunk highway	State trunk highway	CTH N	South corporate limit	North corporate limit
	Local trunk highway	County trunk highway	Pioneer Road	STH 57 Pioneer Boad	East corporate limit
	State trunk highway	Local trunk highway	STH 57	Washington Avenue	Keup Road
	State trunk highway	Local trunk highway	STH 57	City of Mequon	Pioneer Road
	State trunk highway	Local trunk highway	STH 143	STH 57	North corporate limit
	County trunk highway County trunk highway	Local trunk highway	СТНТ	Webster Avenue	Evergreen Boulevard
City of Meguon	Local trunk highway	State trunk highway	Wauwatosa Road	STH 167	Pioneer Road
	Local trunk highway	County trunk highway	County Line Road	Wasaukee Road	STH 57
	Local trunk highway	County trunk highway	Granville Road	County Line Road	Freistadt Road
	Local trunk highway	County trunk highway	Granville Road	Highland Road	Pioneer Road
	Local trunk highway	County trunk highway	Highland Road	Granville Road	IH 43
	Local trunk highway	County trunk highway	Pioneer Road	Granville Road	Davis Road
	County trunk highway	Local trunk highway	Pioneer Road	Wasaukee Road	Granville Road
	New facility	Local trunk highway	River Road	Highland Road	Bonniwell Road
	New facility	Local trunk highway	River Road	Grace Avenue	Freistadt Road
	State trunk highway	Local trunk highway	STH 57	Village of Thiensville	North corporate limit
City of Port Washington	Local trunk highway	State trunk highway	Wisconsin Street	Jackson Street	Grand Avenue
	Local trunk highway	State trunk highway	Swing Street	Franklin Street	Jackson Street
	Local trunk highway	State trunk highway County trunk highway	Jackson Street Chestnut Street	Division Street	Wisconsin Street
	Local trunk highway	County trunk highway	Division Street	South corporate limit	Chestnut Street
	Local trunk highway	County trunk highway	Spring Street	STH 33 Chaptout Street	CTH KK
	Local trunk highway County trunk highway	Local trunk highway	CTH CC	STH 32	CTH C
	County trunk highway	Local nonarterial	СТНКК	Spring Street	North corporate limit

Map 143

PROPOSED CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS UNDER THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO RACINE COUNTY



The proposed changes in jurisdictional responsibility for arterial streets and highways in Racine County are shown on the accompanying map. In 1991, the State trunk highway system in Racine County totalled 159 miles, the county trunk highway system totalled about 125 miles, and the local arterial system totalled 65 miles. By the year 2010, through the jurisdictional transfers identified on the accompanying map and listed on Table 264, the State trunk highway system would total 160 miles, the county trunk highway system would total 160 miles.

CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS IN RACINE COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

	kurindiational P				1
Civil Division	Existing	Planned	Facility	From	To
Town of Burlington	New facility Local nonarterial Local nonarterial State trunk highway County trunk highway State trunk highway State trunk highway State trunk highway State trunk highway State trunk highway	State trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	Burlington bypass Fish Hatchery Road Karcher Road STH 36/83 CTH W STH 11 STH 11 STH 11 STH 142 STH 36 STH 83	Town of Rochester CTH P Fish Hatchery Road North town line City of Burlington City of Burlington Mormon Road City of Burlington West town line City of Burlington	STH 36 Karcher Road CTH KD City of Burlington CTH A Burlington bypass City of Burlington Bypass route City of Burlington Bypass route
Town of Caledonia	County trunk highway Local trunk highway Local trunk highway Local trunk highway New facility New facility County trunk highway County trunk highway County trunk highway New facility	State trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	CTH K Seven Mile Road Four Mile Road Three Mile Road Five Mile Road extension Five Mile Road extension CTH G CTH V CTH V Emmertsen Road	IH 94 West town line STH 32 Middle Road Charles Street STH 32 North town line Seven Mile Road Three Mile Road	STH 38 STH 32 STH 31 CTH G Five Mile Road Erie Street Three Mile Road Seven Mile Road Seven Mile Road Sth 38
Town of Dover	State trunk highway Local trunk highway State trunk highway County trunk highway County trunk highway	County trunk highway County trunk highway County trunk highway Local nonarterial Local nonarterial	STH 75 Schroeder Road STH 20 CTH B CTH N	STH 20 STH 75 West town line STH 11 STH 20	South town line East town line East town line South town line CTH A
Town of Mt. Pleasant	County trunk highway New facility County trunk highway Local trunk highway New facility County trunk highway New facility New facility New facility New facility	State trunk highway County trunk highway Local nonarterial County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	CTH K CTH MM realignment CTH V Ohio Street 21st Street CTH X S. Memorial Drive Oakes Road extension Oakes Road extension Rapids Court extension	Kraut Road West of STH 38 Town of Caledonia CTH C Oakes Road extension STH 31 CTH KR Oakes Road Spring Street Rapids Drive	North town line STH 38 STH 20 City of Racine CTH of Racine CTH T Chicory Road STH 11 STH 20 STH 38
Town of Norway	County trunk highway New facility Local nonarterial County trunk highway	State trunk highway State trunk highway County trunk highway Local nonarteriał	CTH K CTH K extension Denoon Road CTH K	West town line Britton Road West town line Apple Road	Britton Road USH 45 CTH Y East town line
Town of Raymond	County trunk highway New facility Local trunk highway County trunk highway County trunk highway	State trunk highway State trunk highway County trunk highway Local nonarterial Local nonarterial	CTH K CTH K extension Seven Mile Road CTH G CTH K	108th Street 108th Street East town line USH 45 108th Street	IH 94 USH 45 West town line IH 94 West town line
Town of Rochester	New facility State trunk highway State trunk highway	State trunk highway County trunk highway County trunk highway	Burlington bypass STH 20 STH 36/83	STH 36/83 STH 36/83 Burlington bypass	Town of Burlington East town line South town line
Town of Waterford	County trunk highway Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial	State trunk highway County trunk highway	CTH K Bridge Drive Buena Park Road Denoon Road Fox River Road Honey Creek Road Marsh Road North Lake Drive Ranke Road	STH 36 Marsh Road Ranke Road STH 164 Bridge Drive West town line North town line Fox River Road Marsh Road	East town line Fox River Road STH 20 Town of Norway North Lake Drive STH 20 Ranke Road STH 164 Buena Park Road
Town of Yorkville	Local trunk highway	County trunk highway	1st Street	West town line	USH 45
Village of Elmwood Park Village of Rochester	County trunk highway County trunk highway County trunk highway	Local trunk highway	CTH T CTH W (Front Street)	North corporate limits Main Street-CTH D	South corporate limits North corporate limits
Village of Waterford	Local trunk highway County trunk highway	State trunk highway Local trunk highway	Main Street CTH W	First Street Main Street	East corporate limits South corporate limits
Village of Wind Point	County trunk highway	Local trunk highway	СТН G	Four Mile Road	Three Mile Road
City of Burlington	Local trunk highway State trunk highway State trunk highway State trunk highway State trunk highway State trunk highway	County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	McHenry Street STH 36/83 STH 11 STH 142 STH 36 STH 83	STH 36 North corporate limits East corporate limits STH 11 West corporate limits Milwaukee Avenue	South corporate limits McHenry Street West corporate limits South corporate limits McHenry Street South corporate limits
City of Racine	Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway New facility County trunk highway County trunk highway New facility	State trunk highway State trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	Douglas Avenue Yout Street Main Street Spring Street Three Mile Road 21st Street CTH T CTH X Rapids Court extension	Yout Street Douglas Avenue Gould Street CTH C STH 32 West corporate limits STH 11 STH 11 Rapids Drive	Gould Street Main Street North corporate limits STH 38 CTH G STH 31 South corporate limits South corporate limits STH 38

CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS IN WALWORTH COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Civil Division	Existing	Planned	Facility	From	То	
Town of Bloomfield	New facility	State trunk highway	New facility	Town Line Road	West town line	
	Local trunk highway	County trunk highway	N. Bloomfield Road	CTH H	Hafs Road	
	Local nonarterial	County trunk highway	Clover Road	Lake Geneva Road	Lake Shore Drive	
	Local trunk highway	County trunk highway	Darling Road	CTH H	Twin Lakes Road	
	Local trunk highway	County trunk highway	Hafs Road	N. Bloomfield Road	CTH U	
	Local trunk highway	County trunk highway	Lake Geneva Road	CTH H	Clover Road	
	Local trunk highway	County trunk highway	Lake Shore Drive	Clover Road	Orchid Drive	
	Local trunk highway	County trunk highway	Orchid Drive	Lake Shore Drive	Pell Lake Drive	
	Local trunk highway	County trunk highway	Pell Lake Drive	Orchid Drive	CTH U	
	Local trunk highway	County trunk highway	Powers Lake Road	CTH U	East town line	
	Local trunk highway	County trunk highway	South Road	N. Bloomfield Road	North town line	
	Local nonarterial	County trunk highway	Twin Lakes Road	Darling Road	CTH U	
Town of Darien	Local trunk highway	County trunk highway	Darien-Sharon Town Line Road	CTH X	East town line	
	Local nonarterial	County trunk highway	Foundry Road	CTH X	New facility	
	New facility	County trunk highway	Foundry Road extension	Foundry Road	Village of Darien	
	County trunk highway	Local nonarterial	CTH C	USH 14	North town line	
	County trunk highway	Local nonarterial	CTH M	City of Delavan	North town line	
Town of Delavan	County trunk highway Local nonarterial New facility State trunk highway State trunk highway Local nonarterial Local trunk highway Local nonarterial New facility County trunk highway	State trunk highway State trunk highway State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local nonarterial	CTH F Bailey's Road Bailey's Road extension STH 11 STH 67 Briggs Road Darien-Sharon Town Line Road Town Hall Road New facility CTH O	Bailey's Road CTH F Bailey's Road City of Delavan New facility North town line West town line STH 50 North town line North town line	South town line New facility STH 67 City of Elkhorn Village of Williams Bay STH 11 CTH 0 East town line City of Elkhorn City of Elkhorn	
Town of East Troy	Local trunk highway	County trunk highway	Booth Lake Road	STH 20	New facility	
	New facility	County trunk highway	Booth Lake Road extension	Booth Lake Road	St. Peter's Road	
	Local trunk highway	County trunk highway	Booth Lake Road	CTH J	St. Peter's Road	
	Local trunk highway	County trunk highway	Honey Creek Road	Racine County	South town line	
	County trunk highway	Local trunk highway	CTH G	Village of East Troy	IH 43	
Town of Geneva	State trunk highway	County trunk highway	STH 67	New facility	Village of Williams Bay	
	State trunk highway	County trunk highway	STH 120	Sheridan Street	City of Lake Geneva	
	Local nonarterial	County trunk highway	Palmer Road	STH 67	CTH H	
	New facility	Local trunk highway	New facility	City of Lake Geneva	STH 50	
Town of Lafayette	State trunk highway	County trunk highway	STH 11	IH 43	City of Elkhorn	
	Local trunk highway	County trunk highway	Bowers Road	IH 43	CTH D	
Town of LaGrange	New facility	State trunk highway	USH 12 freeway	West town line	South town line	
	Local trunk highway	County trunk highway	Kettle Moraine Drive	West town line	CTH H	
	Local nonarterial	County trunk highway	Palmyra Road	North town line	STH 67	
	County trunk highway	Local nonarterial	CTH O	USH 12	Jackson Road	
Town of Linn	New facility	State trunk highway	STH 120 bypass	East town line	STH 120	
	Local nonarterial	County trunk highway	Willow Road	CTH BB	STH 120	
	State trunk arterial	Local trunk highway	STH 120	City of Lake Geneva	STH 120 bypass	
	County trunk highway	Local nonarterial	CTH BB	Willow Road	STH 120	
Town of Lyons	New facility	State trunk highway	STH 120 bypass	City of Lake Geneva	South town line	
	Local trunk highway	County trunk highway	Amity Street	STH 36	Spring Valley Road	
	Local trunk highway	County trunk highway	South Road	Spring Valley Road	South town line	
Town of Richmond	New facility	County trunk highway	CTH P extension	Territorial Road	CTH A	
	County trunk highway	Local nonarterial	CTH M	STH 89	South town line	
	County trunk highway	Local nonarterial	CTH M	West town line	South town line	
	County trunk highway	Local nonarterial	CTH P	Territorial Road	CTH A	
Town of Sharon	Local trunk highway	County trunk highway	Darien-Sharon Town Line Road	CTH X	East town line	
	County trunk highway	Local nonarterial	CTH B	West town line	CTH C	
Town of Spring Prairie	New facility	State trunk highway	Burlington bypass	STH 11	STH 36	
	Local trunk highway	County trunk highway	Honey Creek Road	North town line	CTH D	
	State trunk highway	Local trunk highway	STH 11	Burlington bypass	East town line	
Town of Sugar Creek	New facility Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial New facility New facility County trunk arterial	State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway County trunk highway Local nonarterial	USH 12 freeway Briggs Road Cobbie Road Granville Road Hazel Ridge Road Sugar Creek Road New facility CTH H extension CTH O	North town line Hazel Ridge Road CTH H Sugar Creek Road Briggs Road Granville Road CTH H CTH H North town line	STH 67 South town line Sugar Creek Road Hazel Ridge Road Granville Road Cobbie Road Town of Delavan STH 67 South town line	

Table 265 (continued)

	1				· ·
	Jurisdictional Re	sponsibility		· · · · · ·	
Civil Division	Existing	Planned	Facility	From	То
Town of Troy	Local trunk highway	County trunk highway	Booth Lake Road	CTH J	St. Peter's Road
	Local trunk highway	County trunk highway	Booth Lake Road	St. Peter's Road	STH 20
	Local nonarterial	County trunk highway	Palmyra Road	North town line	STH 67
	Local trunk highway	County trunk highway	Town Line Road	STH 20	CTH ES
	County trunk highway	Local nonarterial	CTH N	CTH ES	STH 20
Town of Walworth	New facility	State trunk highway	STH 67 bypass	STH 67	USH 14
	County trunk highway	State trunk highway	CTH F	North town line	STH 67 bypass
	State trunk highway	County trunk highway	STH 67	CTH F	Theatre Road
	Local trunk highway	County trunk highway	N. Walworth Road	CTH O	STH 67
	State trunk highway	Local trunk highway	STH 67	Village of Walworth	Village of Fontana
	State trunk highway	Local trunk highway	STH 67	STH 67 bypass	Village of Fontana
Town of Whitewater	New facility State trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway New facility New facility Local trunk highway County trunk highway	State trunk highway County trunk highway Local trunk highway	USH 12 freeway USH 12 Anderson Road Clover Valley Road Howard Road Kettle Moraine Drive Howard Road extension Main Street extension Warner Road CTH S	West town line North town line STH 89 Anderson Road North town line Clover Valley Road USH 12 West town line North town line West town line	East town line CTH S Clover Valley Road Kettle Moraine Drive USH 12 East town line CTH P USH 12 freeway CTH S City of Whitewater
Village of Darien	Local nonarterial	County trunk highway	Madison Street	West corporate limit	USH 14
Village of East Troy	Local trunk highway	County trunk highway	Town Line Road	STH 20	CTH ES
	County trunk arterial	Local trunk arteriał	CTH G	CTH ES	South corporate limit
Village of Fontana	State trunk highway	Local trunk highway	STH 67	North corporate limit	Village of Walworth
Village of Genoa City	Local nonarterial	County trunk highway	Fellows Road	CTH B	South corporate limit
	State trunk highway	County trunk highway	USH 12	Freeway terminus	South corporate limit
	New facility	State trunk highway	USH 12 freeway extension	CTH H	South corporate limit
Village of Walworth	New facility	State trunk highway	STH 67 bypass	USH 14	West corporate limit
	State trunk highway	County trunk highway	USH 14	West corporate limit	STH 67
	State trunk highway	County trunk highway	STH 67	USH 14	CTH B
	State trunk highway	Local trunk highway	USH 14	STH 67	South corporate limit
	State trunk highway	Local trunk highway	STH 67	Village of Fontana	CTH B
Village of Williams Bay	State trunk highway	County trunk highway	STH 67	West corporate limit	North corporate limit
City of Delavan	State trunk highway	County trunk highway	STH 11	STH 50	East corporate limit
	Local trunk highway	County trunk highway	Beloit Street	STH 11	Creek Road
	Local trunk highway	County trunk highway	Richmond Road	North corporate limit	STH 11
	Local trunk highway	County trunk highway	2nd Street	STH 11	South corporate limit
	County trunk highway	Local nonarterial	CTH M	West corporate limit	CTH P
	County trunk highway	Local nonarterial	CTH O	North corporate limit	A point 0.01 mile north
City of Elkhorn	State trunk highway	County trunk highway	STH 11	Town of Delavan	STH 67
	State trunk highway	County trunk highway	STH 11	Lincoln Street	East corporate limit
	State trunk highway	Local trunk highway	STH 11	STH 67	CTH H
	New facility	Local trunk highway	New facility	STH 67	STH 11
	New facility	Local trunk highway	New facility	Town of Delavan	STH 67
City of Lake Geneva	Local nonarterial	State trunk highway	Edwards Boulevard	STH 50	End of Edwards Boulevard
	New facility	State trunk highway	STH 120 bypass	End of Edwards Boulevard	South corporate limit
	New facility	State trunk highway	STH 120 bypass	North corporate limit	N. Bloomfield Road
	State trunk highway	County trunk highway	STH 120	USH 12	STH 50
	State trunk highway	Local trunk highway	STH 120	STH 50	South corporate limit
	New facility	Local trunk highway	Grant Street extension	West corporate limit	CTH H
City of Whitewater	New facility	State trunk highway	Main Street extension	USH 12 freeway	Frontage Road
	Local trunk facility	State trunk highway	Whitewater Street	Fremont Street	USH 12
	Local trunk highway	County trunk highway	Tratt Street	Jefferson County	USH 12
	County trunk highway	Local trunk highway	CTH S	West corporate limit	Pleasant Street
	New facility	Local trunk highway	New facility	USH 12	CTH S

Map 144



PROPOSED CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS UNDER THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO WALWORTH COUNTY

The proposed changes in jurisdictional responsibility for arterial streets and highways in Walworth County are shown on the accompanying map. In 1991, the State trunk highway system in Walworth County totalled 214 miles, the county trunk highway system totalled 168 miles, and the local arterial system totalled 47 miles. By the year 2010, through the jurisdictional transfers identified on the accompanying map and listed on Table 265, the State trunk highway system would total 223 miles, the county trunk highway system would total 239 miles, and the local arterial system would total 22 miles.

CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS IN WASHINGTON COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

	Jurisdictional Responsibility					
Civil Division	Existing	Planned	Facility	From	То	
Town of Addison	New facility State trunk highway State trunk highway Local trunk highway State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	State trunk highway County trunk highway County trunk highway County trunk highway Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial	STH 33 STH 175 STH 175 Aurora Road, Deer Road, Indian Drive STH 33 CTH U CTH S CTH K CTH DW CTH W	Rock River STH 83 West town line STH 33 Rock River STH 33 CTH U STH 83 USH 41 STH 175	USH 41 CTH K STH 33 CTH K USH 41 South town line CTH W Turtle Road West town line North town line	
Town of Barton	New facility New facility Local trunk highway Local trunk highway County trunk highway New facility New facility County trunk highway County trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial Local nonarterial	Kettle View Drive extension N. River Road extension Kettle View Drive Kettle View Drive Newark Road/Lighthouse Drive CTH B Schuster Drive extension 18th Avenue CTH B CTH B	Schuster Drive City of West Bend North town line CTH D CTH D Schuster Drive extension Schuster Drive City of West Bend CTH D CTH D	STH 33 STH 144 CTH D Schuster Drive STH 144 City of West Bend Beaver Dam Road CTH D North town line Schuster Drive extension	
Town of Erin	County trunk highway County trunk highway County trunk highway	Local nonarterial Local nonarterial Local nonarterial	СТН Q СТН 00 СТН Е	STH 83 СТН О STH 83	СТН К STH 83 СТН К	
Town of Farmington	State trunk highway Local trunk highway State trunk highway County trunk highway County trunk highway	County trunk highway County trunk highway Local nonarterial Local nonarterial Local nonarterial	STH 84 Trading Post Trail STH 84 CTH HH CTH DD	CTH X CTH H STH 144 STH 28 Along STH 144	East town line South town line CTH X STH 144	
Town of Germantown	County trunk highway	Local nonarterial	СТН Ү	STH 145	North town line	
Town of Hartford	New facility State trunk highway Local trunk highway Local trunk highway New facility New facility New facility New facility State trunk highway County trunk highway County trunk highway	State trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local onarterial Local nonarterial Local nonarterial	New STH 83 STH 175 Kettle Moraine Drive Arthur Road Arthur Road extension Monroe Avenue extension Taylor Road extension Wacker Drive extension STH 83 CTH U CTH K CTH E	City of Hartford CTH K CTH K CTH U Independence Avenue Pond Road STH 60 STH 60 City of Hartford City of Hartford CTH N STH 83 STH 83	CTH E Village of Slinger STH 60 East town line Arthur Road Monroe Avenue Pond Road Lee Road CTH E North town line City of Hartford CTH K	
Town of Jackson	State trunk highway State trunk highway Local trunk highway County trunk highway	County trunk highway County trunk highway County trunk highway Locał trunk highway	STH 143 STH 143 Jackson Road CTH M	Town of Polk CTH P STH 143 Country Aire Drive	CTH P East town line Village of Jackson East town line	
Town of Kewaskum	New facility New facility Local trunk highway Local trunk highway County trunk highway County trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway Local nonarterial Local nonarterial	CTH H extension Kettle View Drive Kettle View Drive Badger Road CTH B CTH H	USH 45 STH 28 CTH H Kettle View Drive CTH H Town of Wayne	Badger Road CTH H South town line Prospect Drive South town line Kettle View Road	
Town of Polk	County trunk highway Local trunk highway New facility State trunk highway State trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway County trunk highway County trunk highway County trunk highway	State trunk highway State trunk highway County trunk highway Local trunk highway Local nonarterial Local nonarterial	CTH J Lovers Lane Pioneer Road extension STH 144 STH 175 STH 175 STH 143 STH 175 Arthur Road Pioneer Road Scenic Drive Pleasant Valley Road CTH C ⁻ CTH AA CTH E CTH C	STH 175 STH 60 Pioneer Road CTH K Village of Slinger USH 45 STH 60 STH 144 USH 41 CTH C CTH Z Lilly Road STH 144 CTH CC STH 60	South town line STH 175 CTH CC Village of Slinger West town line East town line South town line West town line Pioneer Road extension STH 60 USH 45 CTH Z USH 41 CTH J CTH P	
Town of Richfield	County trunk highway State trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	CTH J STH 175 STH 175 Pioneer Road Scenic Drive Willow Creek Road Colgate Road	North town line Village of Germantown STH 167 Pioneer Road extension STH 167 Scenic Drive Willow Creek Road	CTH Q STH 167 North town line USH 41 Willow Creek Road Colgate Road CTH Q	

Table 266 (continued)

	Jurisdictional Responsibility				
Civil Division	Existing	Planned	Facility	From	То
Town of Trenton	State trunk highway Local trunk highway Local trunk highway New facility New facility County trunk highway	County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local nonarterial	STH 143 Trading Post Trail S. River Road Jefferson Street extension Trenton Road/Maple Road CTH M	CTH G North town line STH 33 West town line STH 33 CTH M	CTH M CTH M CTH I Trenton Road Maple Road end CTH MY
Town of Wayne	New facility New facility State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	State trunk highway County trunk highway Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial	STH 28 realignment CTH D realignment STH 28 CTH W CTH W CTH H CTH H CTH H	Mullen Lane USH 41 USH 41 CTH D STH 28 USH 41 North town line USH 41	West town line W. Beechnut Drive Mullen Lane South town line North town line East town line West town line CTH D
Town of West Bend	State trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway County trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway	STH 144 STH 143 18th Avenue Decorah Road S. River Road Paradise Drive Paradise Drive CTH NN	STH 33 CTH P CTH NN 18th Avenue STH 33 18th Avenue City of West Bend 18th Avenue	CTH K CTH G City of West Bend City of West Bend City of West Bend City of West Bend CTH G CTH P
Village of Germantown	New facility State trunk highway State trunk highway Local trunk highway	County trunk highway County trunk highway	Division Road extension STH 175 STH 175 Pleasant View Drive County Line Road County Line Road Bonniwell Road Freistadt Road Division Road Country Aire Drive Division Road	Mequon Road Maple Road North corporate limit CTH F Pilgrim Road STH 145 Pleasant View Drive Division Road STH 167 Bonniwell Road Freistadt Road	Freistadt Road South corporate limit Maple Road Bonniwell Road STH 145 East corporate limit Country Aire Drive STH 145 CTH Q CTH C STH 145
	Local trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial Local nonarterial Local nonarterial	Lannon Road CTH F CTH M CTH M CTH Y CTH Y CTH Y CTH F	STH 175 Pleasant View Road Country Aire Drive CTH C Hill Top Drive STH 145 Mequon Road Mequon Road	USH 41-45 East corporate limit East corporate limit End Goldendale Road Mequon Road STH 175 STH 175
Village of Jackson	Local trunk highway	County trunk highway	Jackson Road	STH 60	North corporate limit
Village of Kewaskum	New facility	County trunk highway	Kettle View Drive extension	STH 28	South corporate limit
Village of Slinger	State trunk highway State trunk highway County trunk highway	County trunk highway County trunk highway Local trunk highway	STH 175 STH 144 CTH AA	North corporate limit North corporate limit STH 144	South corporate limit STH 60 USH 41
City of Hartford	New facility Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway New facility New facility State trunk highway State trunk highway County trunk highway	State trunk highway State trunk highway State trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial	New STH 83 N. Wilson Avenue S. Wilson Avenue Arthur Road State Street Wacker Drive Monroe Avenue extension Taylor Road extension Grand Avenue, Main Street, Union Street Branch Street CTH U CTH K	Monroe Avenue STH 83 Monroe Avenue CTH U State Street West corporate limit STH 60 North corporate limit Main Street Arthur Road North corporate limit	North corporate limit Sumner Street South corporate limit East corporate limit Wacker Drive Sumner Street Willow Lane CTH N South corporate limit Lincoln Avenue CTH N South corporate limit
City of West Bend	New facility New facility Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway New facility	County trunk highway County trunk highway Local trunk highway	S. River Road extension N. River Road extension Island Avenue S. River Road N. River Road N. Main Street Paradise Drive 18th Avenue Main Street Decorah Road 18th Avenue extension	STH 33 Creek Road STH 33 CTH I STH 33 Green Tree Road 18th Avenue South corporate limit Island Avenue 18th Avenue North corporate limit	South corporate limit North corporate limit Main Street North corporate limit Creek Road Barton Avenue East corporate limit STH 33 Paradise Drive CTH I Park Avenue
City of Milwaukee	Local trunk highway	County trunk highway	County Line Road	West corporate limit	Wausaukee Road



PROPOSED CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS UNDER THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO WASHINGTON COUNTY

The proposed changes in jurisdictional responsibility for arterial streets and highways in Washington County are shown on the accompanying map. In 1991, the State trunk highway system in Washington County totalled 186 miles, the county trunk highway system totalled 148 miles, and the local arterial system totalled 65 miles. By the year 2010, through the jurisdictional transfers identified on the accompanying map and listed on Table 266, the State trunk highway system would total 159 miles, the county trunk highway system would total 234 miles, and the local arterial system would total 75 miles.

PROPOSED CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS UNDER THE RECOMMENDED REGIONAL TRANSPORTATION PLAN AS APPLIED TO WAUKESHA COUNTY



LEGEND

TRANSFERS TO:

- STATE TRUNK HIGHWAY SYSTEM
- COUNTY TRUNK HIGHWAY SYSTEM
- LOCAL TRUNK HIGHWAY SYSTEM
- LOCAL (NONARTERIAL) SYSTEM

NOTE: THE RECOMMENDATION TO PLACE CALHOUN ROAD ON COUNTY TRUNK HIGHWAY SYSTEM FROM CTH ES TO CTH K IS CONTINGENT UPON THE CONSTRUCTION OF A NEW INTERCHANGE ON IH 94 AT CALHOUN ROAD.



The proposed changes in jurisdictional responsibility for the arterial streets and highways in Waukesha County are shown on the accompanying map. In 1991, the State trunk highway system in Waukesha County totalled 231 miles, the county trunk highway system totalled 320 miles, and the local arterial system totalled 165 miles. By the year 2010, through the jurisdictional transfers identified on the accompanying map and listed on Table 267, the State trunk highway system would total 230 miles, the county trunk highway system would total 131 miles.

CHANGES IN JURISDICTIONAL RESPONSIBILITY FOR ARTERIAL STREETS AND HIGHWAYS IN WAUKESHA COUNTY UNDER THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

	Jurisdictional	Responsibility			
Civil Division	Existing	Planned	Facility	From	То
Town of Brookfield	Local trunk highway Local nonarterial	County trunk highway County trunk highway Local trunk highway	Springdale Road Extension of Barker Road Brookfield Road	CTH JJ STH 190 North corporate limits	STH 190 Village of Menomonee Falls South corporate limits
Town of Delafield	State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	County trunk highway County trunk highway Local nonarterial Local nonarterial Local nonarterial Local nonarterial	Frontage Road (Golf Road) Extension of CTH KE CTH E CTH G (Silvernail Road) CTH G (Elmhurst Road) CTH G (Brandy Brook Road) CTH GR	Town of Pewaukee STH 83 CTH KE CTH G (Elmhurst Road) CTH G (Silvernail Road) USH 18 CTH KE	CTH E CTH E USH 18 CTH SS USH 18 Town of Genesee Town of Pewaukee
Town of Eagle	State trunk highway Local nonarterial County trunk highway County trunk highway County trunk highway County trunk highway	County trunk highway County trunk highway Local nonarterial Local nonarterial Local nonarterial Local nonarterial	STH 99 Little Prairie Road CTH N CTH S CTH NN CTH ZC CTH ZZ	STH 67 CTH NN Jefferson county line STH 59 Jefferson county line Town of Ottawa Jefferson county line	Town of Mukwonago Walworth county line STH 59 STH 67 STH 67 CTH ZZ Town of Ottawa
Town of Genesee	County trunk highway County trunk highway	Local nonarterial Local nonarterial	CTH C CTH G (Brandy Brook Road)	USH 18 Town of Delafield	CTH G Town of Ottawa
Town of Lisbon	County trunk highway Local trunk highway	State trunk highway County trunk highway	CTH J Plain View Road	Washington county line Oak Road	Town of Pewaukee Town Line Road
Town of Merton	Local trunk highway State trunk highway County trunk highway	State trunk highway State trunk highway State trunk highway County trunk highway Local nonarterial Local nonarterial	STH 83 STH 83 STH 83 Oak Road Extension of CTH KE STH 83 CTH KE	Oconomowoc River Point 950 feet north of Little Oconomowoc River STH 16 Plain View Road CTH K Village of Chenequa 0.5 mile north of CTH K	Little Oconomowoc River CTH CW Village of Chenequa Village of Merton CTH KE STH 16 CTH K
Town of Mukwonago	 State trunk highway County trunk highway	State trunk highway County trunk highway Local nonarterial	STH 83 (Mukwonago bypass) STH 99 CTH I	IH 43 Village of Mukwonago Town of Vernon	CTH NN Town of Eagle STH 83
Town of Oconomowoc	State trunk highway State trunk highway Local trunk highway	State trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway	STH 16-STH 67 (Oconomowoc bypass) STH 16 STH 67 Wisconsin Avenue Oconomowoc Parkway	STH 67 at STH 16 City of Oconomowoc STH 16 Oconomowoc bypass CTH R STH 16	Jefferson county line Jefferson county line City of Oconomowoc CTH P Town of Summit
Town of Ottawa	County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial Local nonarterial	CTH C CTH D CTH C (Brandy Brook Road) CTH ZC CTH ZC CTH ZZ	USH 18 CTH Z East corporate limits CTH Z CTH C Town of Eagle	CTH G STH 67 CTH D CTH C TOWN of Eagle STH 67
Town of Pewaukee	County trunk highway County trunk highway Local trunk highway Local trunk highway State trunk highway Local trunk highway Local trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	State trunk highway State trunk highway State trunk highway State trunk highway State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway	CTH TT CTH J Meadowbrook Road Meadowbrook Road Waukesha western bypass STH 164 Frontage Road (Golf Road) Northview Road Springdale Road Extension of CTH SS CTH J CTH FT CTH TJ CTH GR	USH 18 Town of Lisbon IH 94 City of Waukesha Northview Road STH 190 City of Waukesha Meadowbrook Road CTH JJ CTH G City of Waukesha A point about 0.7 mile west of CTH J City of Waukesha Town of Delafield	Town of Waukesha IH 94 City of Waukesha Northview Road USH 18 City of Waukesha Town of Delafield City of Waukesha STH 190 CTH T IH 94 CTH J CTH JJ CTH JJ Village of Pewaukee
Town of Summit	State trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	County trunk highway Local nonarterial Local nonarterial	STH 16 Valley Road CTH B CTH B CTH BB CTH Z Oconomowoc Parkway CTH B CTH B	CTH P STH 67 Jefferson county line CTH P CTH DR CTH B Town of Oconomowoc CTH Z STH 67	City of Oconomowoc CTH P CTH Z City of Delafield City of Oconomowoc City of Oconomowoc STH 67 City of Oconomowoc CTH P
Town of Vernon	Local trunk highway County trunk highway County trunk highway	County trunk highway Local nonarterial Local nonarterial	Center Drive CTH U (Guthrie Road) CTH I	CTH L Town of Waukesha Town of Waukesha	Racine county line STH 164 Town of Mukwonago

Table 267 (continued)

	Jurisdictional	Responsibility			
Civil Division	Existing	Planned	Facility	From	То
Town of Waukesha	County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	State trunk highway State trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial Local nonarterial Local nonarterial	CTH TT Waukesha western bypass CTH D (Broadway) CTH D (Sunset Drive) CTH J (Racine Avenue) CTH I CTH U (Guthrie Road) CTH U (Guthrie Road)	Town of Pewaukee Macarthur Road City of Waukesha CTH TT City of Waukesha Lawnsdale Road CTH Y City of Waukesha	Macarthur Road STH 59 STH 59 CTH X City of Waukesha Town of Vernon City of Waukesha Town of Vernon
Village of Butler	Local trunk highway	County trunk highway	124th Street	North corporate limits	South corporate limits
Village of Chenequa	Local trunk highway State trunk highway	State trunk highway County trunk highway Local nonarterial	STH 83 Vettelson Drive STH 83	CTH K CP Rail System Thompson Lane	Thompson Lane City of Delafield Town of Merton
Village of Elm Grove	Local trunk highway Local trunk highway Local trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway	North Avenue Pilgrim Parkway 124th Street Extension of 124th Street	City of Brookfield North Avenue City of Brookfield City of Brookfield	Milwaukee county line USH 18 Knoll Road Knoll Road
Village of Hartland	Local trunk highway Local trunk highway 	County trunk highway County trunk highway County trunk highway	Vettelson Drive Capitol Drive Extension of CTH KE	City of Delafield Vettelson Drive STH 83	Capitol Drive CTH KC Town of Delafield
Village of Menomonee Falls	State trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local nonarterial Local trunk highway Local trunk highway Local trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway Local trunk highway Local onnarterial Local nonarterial	STH 175 Boundary Road County Line Road Lisbon Road Hampton Avenue Pilgrim Road Extension of Boundary Road Extension of Lannon Road River Crest Drive Extension of River Crest Drive Extension of River Crest Drive Extension of Shady Lane Roosevelt Drive Water Street Richfield Way	Milwaukee county line County Line Road Pilgrim Road Calhoun Road Lisbon Road County Line Road STH 100 CTH K STH 175 Shady Lane STH 175 Shady Lane STH 175 Shady Lane STH 174 Water Street	Washington county line STH 100 Boundary Road Hampton Avenue Village of Butler STH 175 STH 175 STH 145 City of Brookfield Shady Lane Village of Germantown Thomas Drive Pilgrim Road Richfield Way Village of Germantown
Village of Merton	Local trunk highway	County trunk highway	Oak Road	СТН VV	Town of Merton
Village of Mukwonago	County trunk highway State trunk highway State trunk highway	State trunk highway State trunk highway County trunk highway County trunk highway	CTH NN STH 83 (Mukwonago bypass) STH 83 (Rochester Street) STH 99	Mukwonago bypass North corporate limits CTH NN Town of Mukwonago	STH 83 South corporate limits Mukwonago bypass STH 83 (Rochester Street)
Village of Nashotah	Local trunk highway	County trunk highway	Rasmus Drive	СТНС	City of Delafield
Village of Pewaukee	Local trunk highway Local trunk highway Local trunk highway Local trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway County trunk highway	Capitol Drive Oakton Avenue Prospect Avenue Wisconsin Avenue Extension of CTH SS	Oakton Avenue Capitol Drive Wisconsin Avenue East corporate limits East corporate limits	STH 16 Wisconsin Avenue Town of Pewaukee West corporate limits West corporate limits
Village of Sussex	County trunk highway Local trunk highway	State trunk highway County trunk highway	CTH J Main Street	North corporate limits Locust Street	South corporate limits STH 164
Village of Wales	County trunk highway	Local nonarterial	CTH G (Brandy Brook Road)	East corporate limits	West corporate limits
City of Brookfield	Local trunk highway Local trunk highway	County trunk highway County trunk highway	Barker Road Calhoun Road Lisbon Road Hampton Avenue Moorland Road North Avenue Pilgrim Parkway Pilgrim Road 124th Street 124th Street Springdale Road Extension of 124th Street	North corporate limits STH 59 Calhoun Road Lisbon Road USH 18 Town of Pewaukee North Avenue Lisbon Road Village of Elm Grove Robinwood Street STH 190 North corporate limits	South corporate limits CTH K Hampton Avenue Village of Butler STH 59 Milwaukee county line USH 18 North Avenue Village of Butler Village of Elm Grove South corporate limits Robinwood Street
City of Delafield	Local trunk highway Local trunk highway Local trunk highway Local trunk highway	County trunk highway County trunk highway County trunk highway County trunk highway	Main Street Milwaukee Street Vettelson Drive Vettelson Drive	Town of Summit Main Street Village of Nashotah Village of Chenequa	Milwaukee Street STH 83 CP Rail System Village of Hartland
City of Muskego	Local trunk highway Local nonarterial Local nonarterial Local nonarterial	County trunk highway County trunk highway County trunk highway County trunk highway Local trunk highway Local trunk highway Local trunk highway	S. Denoon Road Extension of Moorland Road Extension of Moorland Road Durham Road Martin Drive Lannon Drive Extension of Sunnyslope Road	Crowbar Drive CTH HH CTH L Woods Road CTH HH Martin Drive CTH HH	CTH Y CTH L Durham Road CTH OO Lannon Drive CTH L CTH L

Table 267 (continued)

					F
	Jurisdictional	Responsibility		and the second se	
Civil Division	Existing	Planned	Facility	From	То
City of New Berlin	Local trunk highway	County trunk highway	Calhoun Road	CTH ES	STH 59
	Local trunk highway	County trunk highway	Johnson Road	STH 59	0.4 mile south of STH 59
	Local trunk highway	County trunk highway	Johnson Road	Lincoln Avenue	Coffee Road
	Local trunk highway	County trunk highway	Moorland Road	IH 43	Grange Avenue
		County trunk highway	Extension of Johnson Road	0.4 mile south of STH 59	Lincoln Avenue
		County trunk highway	Extension of Johnson Road	Coffee Road	CTH Y
	'	County trunk highway	Extension of Moorland Road	South corporate limits	Grange Avenue
City of Oconomowoc		State trunk highway	STH 16-STH 67 (Oconomowoc bypass)	South corporate limits	North corporate limits
	State trunk highway	County trunk highway	STH 16	East corporate limits	West corporate limits
	State trunk highway	County trunk highway	STH 67	North corporate limits	STH 16
	Local trunk highway	County trunk highway	Summit Avenue	Thackeray Trail	STH 16
	Local nonarterial/	County trunk highway	Lake Drive/Fairview Road	STH 67	Lapham Street
	County trunk highway	Local trunk highway	стн z	Marigold Street	Lake Drive
		Local trunk highway	Oconomowoc Parkway	North corporate limits	East corporate limits
	County trunk highway	Local nonarterial	СТНВ	East corporate limits	West corporate limits
City of Waukesha	County trunk highway	State trunk highway	СТНТТ	North corporate limits	South corporate limits
	Local trunk highway	State trunk highway	Meadowbrook Road	North corporate limits	South corporate limits
	State trunk highway	County trunk highway	STH 164	USH 18	Town of Pewaukee
	State trunk highway	County trunk highway	Frontage Road (Golf Road)	СТН Т	Town of Pewaukee
	Local trunk highway	County trunk highway	Grandview Boulevard	USH 18	Northview Road
	Local trunk highway	County trunk highway	North Street	USH 18	St. Paul Avenue
	Local trunk highway	County trunk highway	Northview Road	Town of Pewaukee	СТНТ
	Local trunk highway	County trunk highway	St. Paul Avenue	Harris Highland Drive	USH 18
	County trunk highway	Local trunk highway	CTH D (Sunset Drive)	стн х	West corporate limits
	County trunk highway	Local trunk highway	СТНЈ	Gascoigne Drive	Town of Pewaukee
	County trunk highway	Local trunk highway	СТН FT	A point about 0.7 mile west of CTH J	СТН Ј
	Local nonarterial	Local trunk highway	Main Street	Hartwell Avenue	Moreland Boulevard
	County trunk highway	Local nonarterial	CTH U (Guthrie Road)	North corporate limits	South corporate limits
	County trunk highway	Local trunk highway	СТН ТЈ	СТНТ	Town of Pewaukee
	County trunk highway	Local nonarterial	СТН ТЈ	Meadow Lane	СТНТ

Source: SEWRPC.

facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development properly. The plan would provide for the construction of nine new miles of arterial facilities, for the widening of 45 miles, and for the preservation of 301 miles of facilities within the County. The location and description of each of the planned functional improvements in Kenosha County is set forth in Table 268.

<u>Milwaukee County</u>: The arterial street and highway system in Milwaukee County is recommended to be expanded by 22 miles, or 3 percent, from 775 miles in 1991 to 797 miles in the year 2010 (see Map 126, page 537). The increase in arterial mileage would come about through the construction of 11 new miles of facility and through the conversion of 11 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development properly. The plan would provide for the construction of 11 new miles of arterial facilities, for the widening of 50 miles, and for the preservation of 736 miles of facilities within the County. The location and description of each of the planned functional improvements in Milwaukee County is set forth in Table 269.

<u>Ozaukee County</u>: The arterial street and highway system in Ozaukee County is recommended to be expanded by 16 miles, or 6 percent, from 289 miles in 1991 to nearly 305 miles in the year 2010 (see Map 127, page 538). The increase in arterial mileage would come about through the construction of six new miles of facility and through the conversion of nine miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN KENOSHA COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Recommonded	Improvement			
Jurisdiction	Type	Facility	Termini	Improvement Description
State	Widening	STH 31 STH 32 STH 50 STH 50 STH 83 STH 158 STH 165 STH 165 STH 165 STH 165 G3rd Street Roosevelt Road	31st Street to CTH KR 128th Street to CTH T Walworth county line to 381st Avenue IH 94-USH 41 to 39th Avenue 128th Street to STH 50 104th Avenue to STH 31 CTH H to STH 31 STH 31 to STH 32 IH 94-USH 41 to a point approximately one mile west of CTH H 22nd Avenue to STH 32 39th Avenue to 63rd Street	Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from four to six traffic lanes Widen from two to four traffic lanes
County	Widening	CTH E 30th Avenue 60th Avenue CTH S 22nd Avenue 22nd Avenue Washington Road	STH 31 to STH 32 27th Street to CTH E 39th Avenue to STH 32 IH 94-USH 41 to STH 31 CTH L to E CTH E to CTH KR 39th Avenue to STH 32	Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	IH 94-USH 41 CTH AH extension CTH F extension CTH KD extension CTH ML extension CTH Ω extension	CTH ML CTH F to CTH SA CTH O to 89th Street CTH EM to CTH F CTH H to STH 31 184th Street extended to 168th Street	Construct new interchange Construct two lanes on new alignment Construct two lanes on new alignment
Local	Widening	39th Avenue 104th Avenue	Van Buren Road to STH 50 STH 50 to STH 158	Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	39th Avenue extension 51st Avenue extension 85th Street extension 85th Street extension	24th Street to 15th Street 93rd Street to STH 165 39th Avenue to 32nd Avenue Sheridan Road to 7th Avenue	Construct two lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment

Source: SEWRPC.

properly. The plan would provide for the construction of six new miles of arterial facilities, for the widening of 49 miles, and for the preservation of 250 miles of facilities within the County. The location and description of each of the planned functional improvements in Ozaukee County is set forth in Table 270.

<u>Racine County</u>: The arterial street and highway system in Racine County is recommended to be expanded by 76 miles, or 22 percent, from 348 miles in 1991 to 424 miles in the year 2010 (see Map 128, page 539). The increase in arterial mileage would come about through the construction of 19 new miles of facility and through the conversion of 57 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development properly. The plan would provide for the construction of 19 new miles of arterial facilities, for the widening of 62 miles, and for the preservation of 343 miles of facilities within the County. The location and description of each of the planned functional improvements is set forth in Table 271.

<u>Walworth County</u>: The arterial street and highway system in Walworth County is recommended to be expanded by 55 miles, or 13 percent, from 429 miles in 1991 to 484 miles in the year 2010 (see Map 129, page 540). The increase in arterial mileage would come about through the construction of 36 new miles of

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN MILWAUKEE COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Recommended Jurisdiction	Improvement Type	Facility	Termini	Improvement Description
State	Widening	USH 45-STH 36 STH 32 STH 38 STH 100 STH 100 STH 100 STH 100 CTH BB Fond du Lac Avenue Pennsylvania Avenue Pennsylvania Avenue 124th Street	Waukesha County line to STH 100 County Line Road to STH 100 County Line Road to Oakwood Road IH 43 to STH 24 CTH H to USH 41 STH 36 to CTH H STH 38 to STH 32 CTH U to USH 41 35th Street to 20th Street College Avenue to Layton Avenue Rawson Avenue to College Avenue STH 145 to USH 41-USH 45	Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from six to eight traffic lanes Widen from two to four traffic lanes
	Expansion	Lake Parkway	Lincoln Avenue to CTH Y	Construct four lanes on new alignment
County	Widening	107th Street CTH U CTH U CTH Y CTH ZZ CHVELAND AVENUE Good Hope Road Layton Avenue Oklahoma Avenue 124th Street 124th Street 124th Street	Good Hope Road to STH 145 Grange Avenue to Rawson Avenue Rawson Avenue to Puetz Road Pennsylvania Avenue to STH 32 STH 36 to USH 41 STH 38 to Pennsylvania Avenue Waukesha County line to 113th Street Waukesha County line to USH 41-USH 45 108th Street to 84th Street Clement Avenue to Kinnickinnic Avenue STH 190 to Hampton Avenue Hampton Avenue to CTH VV North Avenue to Watertown Plank Road	Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	Puetz Road extension 124th Street extension 124th Street extension	CTH U to Hunting Park Drive STH 100 to STH 145 Watertown Plank Road to STH 59	Construct two lanes on new alignment Construct four lanes on new alignment Construct two lanes on new alignment
Local	Widening	CTH G Drexel Avenue Howard Avenue Morgan Avenue Pennsylvania Avenue Pennsylvania Avenue Whitnall Avenue Whitnall Avenue 84th Street 91st Street	Mill Road to Bradley Road STH 38 to Pennsylvania Avenue Thompson Avenue to STH 32 Clement Avenue to Lake Parkway Forest Home Avenue to 43rd Street STH 100 to Drexel Avenue Drexel Avenue to Rawson Avenue CTH Y to Packard Avenue Clement Avenue to Pennsylvania Avenue Whitaker Avenue to Howard Avenue Ozaukee County line to STH 100	Widen from two to four traffic lanes Widen from two to four traffic lanes Reconstruction of two traffic lanes Widen from two to four traffic lanes
	Expansion	Canal Street extension Canal Street extension Howard Avenue extension 15th Avenue extension Metro Boulevard	6th Street to 2nd Street USH 41 to 21st Street Thompson Avenue to Kansas Avenue STH 100 to Elm Road 115th Street to 107th Street	Construct two lanes on new alignment Construct two lanes on new alignment Construct four lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment

Source: SEWRPC.

facility and through the conversion of 19 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development properly. The plan would provide for the construction of 36 new miles of arterial facilities, for the widening of 38 miles, and for the preservation of 410 miles of facilities within the County. The location and description of each of the planned functional improvements is set forth in Table 272.

<u>Washington County</u>: The arterial street and highway system in Washington County is recommended to be expanded by 69 miles, or

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN OZAUKEE COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Recommended	Improvement			
Jurisdiction	Туре	Facility	Termini	Improvement Description
State	Widening	STH 32 STH 33 STH 33 STH 33 STH 57 STH 57 STH 60 STH 60 STH 60 STH 143 STH 167 STH 167	Spring Street to Franklin Street Washington county line to Progress Drive Progress Drive to Foster Street IH 43 to Spring Street IH 43 to Sheboygan county line Milwaukee county line to STH 167 STH 143 to STH 57 STH 57 to IH 43 Washington county line to STH 143 CTH N to STH 60 Washington county line to Wauwatosa Road	Widen from two to four traffic lanes Widen from two to four traffic lanes
	Evenesion	UL 42	Highland Road	Construct new interchange
	Expansion	IH 43		Construct new interchange
County	Widening	CTH C/Pioneer Road CTH C/Pioneer Road CTH W CTH W	CTH N to McKinley Boulevard McKinley Boulevard to IH 43 Port Washington Lane to a point about 0.5 mile north of Donges Bay Road STH 167 to Highland Road	Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	Granville Road	Highland Road to Freistadt Road	Construct two lanes on new alignment
Local	Widening	STH 57	Bridge Street to Chateau Drive	Widen from two to four traffic lanes
	Expansion	Cold Springs Road extension Maple Road extension River Road extension River Road extension	CTH O to STH 33 Cedar Creek Road to Rose Street at the Village of Grafton north city limits Bonniwell Road to Highland Road Freistadt Road to Grace Avenue	Construct two lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment

Source: SEWRPC.

17 percent, from 399 miles in 1991 to 468 miles in the year 2010 (see Map 130, page 541). The increase in arterial mileage would come about through the construction of 23 new miles of facility and through the conversion of 46 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development properly. The plan would provide for the construction of 23 new miles of arterial facilities, for the widening of 70 miles, and for the preservation of 375 miles of facilities within the County. The location and description of each of the planned functional improvements is set forth in Table 273.

<u>Waukesha County</u>: The arterial street and highway system in Waukesha County is recommended to be expanded by 58 miles, or 8 percent, from 716 miles in 1991 to 774 miles in the year 2010 (see Map 131, page 542). The increase in arterial mileage would come about through the construction of 26 new miles of facility and through the conversion of 32 miles of previously nonarterial facilities to arterial status to accommodate expected traffic volumes and to provide the spacing of arterial facilities necessary to structure planned urban development properly. The plan would provide for the construction of 26 new miles of arterial facilities, for the widening of 134 miles, and for the preservation of 614 miles of facilities within the County. The location and description of each of the planned functional improvements is set forth in Table 274.

<u>Functional Improvements</u>: It is recommended that the Wisconsin Department of Transportation act to expand, improve, and maintain, in accordance with the plan recommendations, the arterial street and highway facilities under State jurisdiction. It is also recommended that the county boards of the seven constituent counties

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN RACINE COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Becommended	Improvement			
Jurisdiction	Туре	Facility	Termini	Improvement Description
State	Widening	STH 20 STH 20 STH 20 STH 31 STH 31 STH 32 STH 32 STH 36-STH 83 STH 36 STH 38 CTH K CTH K STH 11 STH 11 STH 11	IH 94-USH 41 to Sunnyslope Drive USH 45 to a point 0.73 mile west of CTH C CTH KR to STH 11 CTH MM to STH 32 Milwaukee County to Five Mile Road A point about 0.3 mile north of CTH G to Three Mile Road Wegge Road to STH 20 STH 20 to Waukesha County Milwaukee County to CTH K IH 94 to CTH H Kraut Road to STH 38 71st Street in the Village of Union Grove to IH 94 IH 94 to CTH H 86th Street in the Village of Sturtevant to Willow Road	Widen from four to six traffic lanes Widen from two to four traffic lanes
	Expansion	Burlington bypass Burlington bypass CTH K extension	STH 36 (Milwaukee Avenue) to STH 11 STH 11 to STH 36 (State Street) Britton Road to 108th Street	Construct two lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment
County	Widening	CTH C CTH C CTH Y Three Mile Road	CTH V to Airline Road Airline Road to Sunnyslope Road CTH KR to CTH X STH 32 to CTH G	Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from two to four traffic lanes Widen from two to four traffic lanes
Local	Expansion	21st Street extension Emmertsen Road extension Five Mile Road extension Memorial Drive extension Oakes Road extension Oakes Road extension	STH 31 to Oakes Road Three Mile Road to STH 38 STH 32 to Erie Street Chicory Road to CTH KR STH 11 to 16th Street STH 20 to Airline Road	Construct two lanes on new alignment Construct two lanes on new alignment

Source: SEWRPC.

in the Region, upon recommendation of the county public works, highway, and transportation committees, act to expand, improve, and maintain, in accordance with the plan recommendations, the arterial street and highway facilities under county jurisdiction. It is further recommended that the common councils, village boards, and town boards within the Region, upon recommendation of plan commissions and boards of public works, act to expand, improve, and maintain, in accordance with the plan recommendations, the arterial street and highway facilities under local jurisdiction.

Recommended Revisions to Proposed National <u>Highway System</u>: The Federal Intermodal Surface Transportation Efficiency Act of 1991 requires that a National Highway System be nominated by the Federal Highway Administration and confirmed by the U.S. Congress by September 30, 1995. The time schedule for nominating routes to be included on the proposed National Highway System required that the Wisconsin Department of Transportation consult with the Commission and the local units of governments concerned early in the process of preparing the new regional transportation system plan. That consultation resulted in a proposed National Highway System for the Southeastern Wisconsin Region shown on Map 147. Approval of this system is currently pending Congressional action. The system as initially proposed would include most of the existing freeways in the Region, totaling about

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WALWORTH COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Recommended	Improvement			
Jurisdiction	Туре	Facility	Termini	Improvement Description
State	Widening	USH 12 (Main Street) USH 14 USH 14 USH 14 STH 50	Frontage Road to Fremont Street CTH O to proposed STH 67 bypass Proposed STH 67 bypass to McHenry county line Rock county line to CTH O STH 11 to Wisconsin Street	Widen from two to four traffic lanes Widen from two to four traffic lanes
		STH 50 STH 50 STH 50 STH 50 STH 50 STH 67 STH 67 STH 67 STH 89 STH 120	IH 43 to STH 67 STH 67 to Geneva Street Pearson Drive to Madison Street CTH H to Edwards Boulevard USH 12 to the Kenosha county line IH 43 to the proposed STH 67 bypass at STH 50 USH 12 to Lincoln Avenue Willis Ray Road to Whitewater Street STH 36 to USH 12	Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	IH 43 USH 12 freeway USH 12 freeway STH 67 bypass (Walworth, Fontana, and Williams Bay) STH 120 bypass Burlington bypass	CTH O Whitewater to Elkhorn ^a CTH H to McHenry county line Existing STH 67 at Village of Walworth south corporate limits to existing STH 67 at STH 50 Townline Road to existing STH 120 at Willow Road STH 11 to Mormon Road	Construct new interchange Construct four lanes on new alignment Construct four lanes on new alignment Construct four lanes on generally new alignment Construct two lanes on existing and new alignment Construct two lanes on generally new alignment
County	Widening Expansion	STH 11 Willow Road extension CTH P realignment	CTH O to 7th Street West Side Road to CTH H Territorial Road to CTH A	Widen from two to four traffic lanes Construct two lanes on new alignment Construct two lanes on new alignment
Local	Expansion	New Facility Grant Street extension Main Street extension New Facility New Facility	CTH H east to STH 67 CTH H to STH 50 in Lake Geneva Frontage Road to Rock county line STH 67 south to STH 11 west STH 11 north to CTH H STH 67 north to STH 11 east	Construct two lanes on new alignment Construct two lanes on new alignment

^a The initial segment of the USH 12 freeway between the City of Whitewater and the City of Elkhorn is anticipated to be the segment bypassing the City of Whitewater from existing USH 12 at approximately Howard Road southeast of the City to existing USH 12 at approximately Cold Spring Road northwest of the City. Initially, only two travel lanes are anticipated to be constructed and are anticipated to be open to traffic by the year 2001.

Source: SEWRPC.

242 route-miles, together with 400 route-miles of surface arterial streets and highways, resulting in a total National Highway System in the Region of about 642 route-miles.

With the completion of a new recommended regional transportation system plan, including updated jurisdictional highway system plans for each of the seven counties, consideration should be given to making revisions to the National Highway System at an appropriate future date. These proposed revisions relate largely to proposed changes in jurisdiction of a number of important surface arterial facilities in the Region. By definition, it is proposed that all routes in the Region that are ultimately included on the National Highway System be existing or planned State trunk highways. Accordingly, it is recommended that the changes to the National Highway System identified in Table 275 and on Map 148 be endorsed by the Wisconsin Department of Transportation and forwarded to the Federal Highway Administration. With these changes, the recommended National Highway

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WASHINGTON COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Recommended Jurisdiction	Improvement Type	Facility	Termini	Improvement Description
State	Widening	USH 41 USH 45 STH 33 STH 33 STH 33 STH 33 STH 33 STH 60 STH 60 STH 60 STH 60 STH 167 STH 167 CTH J	STH 145 to Dodge County CTH D to Prospect Drive USH 41 to CTH Z CTH Z to Valley Avenue 18th Avenue to STH 144 Schmidt Road to Trenton Road Oakes Road to Ozaukee county line STH 83 to Wilson Avenue USH 41 to CTH P Wilshire Drive to Ozaukee county line River Lane to Pilgrim Road Pilgrim Road to Ozaukee county line CTH Q to STH 60	Convert expressway to freeway Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	USH 41 USH 41 USH 41 USH 41 STH 33 STH 33 STH 83 STH 83	CTH K STH 33 CTH D STH 28 Trenton Road to Oakes Road Newplat Street to USH 41 CTH E to Monroe Avenue Monroe Avenue to STH 60	Construct new interchange Construct new interchange Construct new interchange Construct new interchange Construct four lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment Construct two lanes on new alignment
County	Widening	CTH Q CTH Q CTH Q CTH Y Decorah Road Main Street Paradise Drive	Division Road to Pilgrim Road CTH V to STH 175 STH 175 to USH 41-45 CTH Q to USH 41-45 7th Avenue to Indiana Avenue Vine Street to Walnut Street A point 1,250 feet east of USH 45 to Main Street	Widen from two to four traffic lanes Widen from two to four traffic lanes
	Expansion	Arthur Road extension Division Road extension Kettleview Road extension Kettleview Road extension Pioneer Road extension	CTH N to Arthur Road STH 167 to Freistadt Road CTH H to STH 28 STH 33 to Schuster Drive CTH J to CTH CC	Construct two lanes on new alignment Construct two lanes on new alignment
Local	Widening	Maple Road	STH 167 to Freistadt Road	Widen from two to four traffic lanes
	Expansion	Independence Avenue extension Jefferson Street extension Monroe Avenue extension River Crest Drive extension N. River Road extension Schuster Drive extension Taylor Road extension Trenton Road extension Wacker Drive extension 18th Avenue extension	STH 60 to CTH N Trenton Road to N. River Road Monroe Avenue to Pond Road CTH Q to Waukesha county line N. River Road to STH 144 Schuster Drive to Beaver Dam Road Pond Road to STH 60 STH 33 to Maple Road STH 60 to Lee Road Park Avenue to Jefferson Street Jefferson Street to CTH D	Construct two lanes on new alignment Construct two lanes on new alignment

Source: SEWRPC.

System for the Region would approximate 613 route-miles, including 242 route-miles of freeway and 371 route-miles of surface arterial streets and highways (see Map 149).⁸

PLAN IMPLEMENTATION PRIORITIES

To ensure that the areawide transportation system plans conform to the State Implementation Plan for air quality, the Federal Clean Air Act Amendments of 1990 require that the imple⁸The recommended National Highway System shown on Map 149 reflects a requirement that the facilities exist at this time. Accordingly, as the regional transportation system plan is implemented, further adjustments in the recommended National Highway System would be required. These adjustments include the planned route of USH 12 as a freeway in Walworth County, the planned extension of future STH 164 along the west side of Waukesha, and the planned STH 36/STH 11 bypass of Burlington.

FUNCTIONAL IMPROVEMENTS TO THE ARTERIAL STREET AND HIGHWAY SYSTEM IN WAUKESHA COUNTY UNDER THE FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN: 2010

Recommended	Improvement			
Jurisdiction	Туре	Facility	Termini	Improvement Description
State	Widening	IH 94	CTH G to STH 16	Widen from four to six traffic lanes
		STH 36	Racine county line to Milwaukee county line	Widen from two to four traffic lanes
		STH 59	CTH X to Sunset Drive	Widen from two to four traffic lanes
		STH 59	STH 164 to Moorland Road	Widen from two to four traffic lanes
		STH 59	Moorland Road to Milwaukee county line	Widen from two to four traffic lanes
		STH 67		Widen from four to six traffic lanes
		SIH 0/ 6TU 02	CTH KE extension to USH 16	Widen from two to four traffic lanes
		STH 03 STH 83	Mariner Drive to CTH KE extension	Widen from two to four traffic lanes
	1	STH 83	IH 94 to USH 18	Widen from two to four traffic lanes
		STH 145	Milwaukee county line to	Widen from two to four traffic lanes
			Washington county line	
		STH 164	STH 74 to STH 190	Widen from two to four traffic lanes
		STH 164	STH 59 to CTH ES	Widen from two to four traffic lanes
		STH 190	STH 164 to CTH Y	Widen from four to six traffic lanes
		STH 190	CTH Y to Brookfield Road	Widen from four to six traffic longs
		CTH J	CTH M to Weshington county line	Widen from two to four traffic lanes
			MacArthur Road to USH 18	Widen from two to four traffic lanes
		Meadowbrook Boad	Northview Road to IH 94	Widen from two to four traffic lanes
		124th Street	STH 145 to USH 41-USH 45	Widen from two to four traffic lanes
	Expansion	14 94	Calboun Boad	Construct new interchange
	Expansion	IH 94	CTH P	Construct new interchange
		STH 16-STH 67 bypass	Wisconsin Avenue to Jefferson county line	Construct four lanes on new alignment
		STH 83	STH 16 to Thompson Lane	Construct two lanes on new alignment
		STH 83	Kilbourne Road to CTH CW	Construct two lanes on new alignment
		Mukwonago bypass	IH 43 to CTH ES	Construct two lanes on new alignment
		Waukesha west bypass	CTH X to Macarthur Road	Construct four lanes on new alignment
		Waukesha west bypass		Construct rour laries of new alignment
County	Widening	STH 164	Moreland Boulevard to IH 94	Widen from four to six traffic lanes
		STH 1/5	STU 59 STU 164 to Moorland Road	Widen from two to four traffic lanes
			Moorland Road to Milwaukee county line	Widen from two to four traffic lanes
		СТНК	CTH Y to Calhoun Road	Widen from two to four traffic lanes
		CTHL	CTH Y to CTH HH	Widen from two to four traffic lanes
		СТН Q	CTH V to STH 175	Widen from two to four traffic lanes
		стн о	Division Road to Pilgrim Road	Widen from two to four traffic lanes
		СТН Т	Northview Road to Silvernail Road	Widen from two to four traffic lanes
		СТН Т	Golf Road to proposed CTH SS extension	Widen from two to four traffic lanes
		CTH W	Pilgrim Road to STH 175	Widen from two to four traffic lanes
				Widen from two to four traffic lanes
			STH 59 to Moreland Boulevard	Widen from two to four traffic lanes
		СТНУ	Hillendale Drive to CTH HH	Widen from two to four traffic lanes
		СТН Ү	IH 43 to Coffee Road	Widen from two to four traffic lanes
		стн ү	STH 59-STH 164 to Coffee Road	Widen from two to four traffic lanes
		СТН Ү	STH 59 to USH 18	Widen from two to four traffic lanes
		СТН Ү	North Avenue to STH 190	Widen from two to four traffic lanes
		CTH Y	USH 18 to North Avenue	Widen from two to four traffic lanes
		СТН Ү	CTH K to STH 74	Widen from two to four traffic lanes
				Widen from two to four traffic lanes
				Widen from two to four traffic lanes
	1		Lisbon Boad to CTH VV	Widen from two to four traffic lanes
		СТН ҮҮ	CTH VV to CTH W	Widen from two to four traffic lanes
		Calhoun Road	CTH ES to CTH D	Widen from two to four traffic lanes
]	Calhoun Road	CTH D to STH 59	Widen from two to four traffic lanes
		Calhoun Road	STH 59 to Gebhardt Road	Widen from two to four traffic lanes
		Calhoun Road	Gebhardt Road to North Avenue	Widen from two to four traffic lanes
		Calhoun Road	North Avenue to STH 190	Widen from two to four traffic lanes
		Hampton Road	Lisbon Road to 132nd Street	Widen from two to four traffic lanes
		Johnson Road	Coffee Road to Lincoln Avenue	Widen from two to four traffic lanes
		Johnson Road	A point about 2,000 feet south of	Widen from two to four traffic lanes
			STH 59 to STH 59	
		Lisbon Road	Calhoun Road to Hampton Road	Widen from two to four traffic lanes

Table 274 (continued)

Recommended	Improvement			
Jurisdiction	Туре	Facility	Termini	Improvement Description
County	Widening	North Avenue	Barker Road to 147th Street	Widen from two to four traffic lanes
	(continued)	North Avenue	Lilly Road to 124th Street	Widen from two to four traffic lanes
		Pilgrim Road	STH 175 to USH 41-USH 45	Widen from two to four traffic lanes
		Pilgrim Road	USH 41-USH 45 to Washington county line	Widen from two to four traffic lanes
		Pilgrim Road	USH 18 to North Avenue	Widen from two to four traffic lanes
		Pilgrim Road	A point about 700 feet north of	Widen from two to four traffic lanes
			North Avenue to Lisbon Road	· · · · · · · · · · · · · · · · · · ·
		124th Street	STH 190 to Hampton Avenue	Widen from two to four traffic lanes
		124th Street	Hampton Avenue to CTH VV	Widen from two to four traffic lanes
		124th Street	North Avenue to Watertown Plank Road	Widen from two to four traffic lanes
	Expansion	CTH Y extension	STH 190 to CTH K	Construct four lanes on new alignment
		CTH KE extension	CTH E to STH 83	Construct two lanes on new alignment
		CTH KE realignment	CTH K to a point about 800 feet north	Construct two lanes on new alignment
		CTH SS extension	CTH G to CTH T	Construct two lanes on new alignment
		Johnson Road extension	A point about 2,000 feet north of STH 59 to Lincoln Avenue	Construct four lanes on new alignment
		Johnson Road extension	Coffee Road to CTH Y	Construct four lanes on new alignment
		Lake Drive extension	Lapham Street to STH 67	Construct four lanes on new alignment
		Moorland Road extension	Woods Road to CTH L	Construct two lanes on new alignment
1		Moorland Road extension	CTH L to IH 43	Construct two lanes on new alignment
		Pilgrim Road realignment	North Avenue to a point about 700 feet north	Construct two lanes on new alignment
		124th Street extension	STH 100 to STH 145	Construct four lanes on new alignment
		124th Street extension	Watertown Plank Road to STH 59	Construct two lanes on new alignment
Local	Widening	Brookfield Road	USH 18 to a point approximately	Widen from two to four traffic lanes
		Main Street		Widen from two to four traffic lanes
	l	Main Street	Downing Drive to STU 50 STU 164	Widen from two to four traffic lanes
		Racine Avenue	Toppy Avenue to STH 59-STH 164	Widen from two to four traffic lanes
		Sunset Drive		Widen nom two to toll thank takes
	Expansion	Brookfield Road extension	Davidson Road to STH 59	Construct two lanes on new alignment
		Oconomowoc Parkway	STH 16 to CTH Z	Construct two lanes on new alignment
		Oconomowoc Parkway	CIH Z to SIH 67	Construct two lanes on new alignment
		River Crest Drive extension	Shady Lane to Washington county Line	Construct two lanes on new alignment
		Shady Lane extension	St. Inomas Drive to STH 175	Construct two lanes on new alignment
		Sunnyslope Road extension	CTH HH to CTH L	Construct two lanes on new alignment
		Valley Road	SIH 67 to CTH P	Construct two lanes on new alignment

Source: SEWRPC.

mentation of areawide transportation system plans be accompanied by analyses of attendant transportation-related air pollutant emissions. The Clean Air Act sets forth criteria for identifying analysis years, in effect, implementation priority years, and, as applied to the Southeastern Wisconsin Region, the following years have been designated as these analysis years: 1996 as the benchmark year relating directly to the 1996 "target" year for the November 1993 State Implementation Plan submittal by the Wisconsin Department of Natural Resources; 2001, because it is at least five years beyond the benchmark year and serves to meet the criteria that analysis years be no more than 10 years apart; 2007, because it is the year in which the Region must demonstrate attainment of the National Ambient Air Quality Standards for ozone; and 2010, because it is the design year of the new regional transportation system plan.

The Clean Air Act Amendments require that a "baseline" scenario and an "action" scenario be defined for each of the analysis years. The baseline scenario is defined as the existing transportation system plus the completion of those significant transportation projects which are in final stages of implementation. The action

RECOMMENDED CHANGES IN THE NATIONAL HIGHWAY SYSTEM DERIVED FROM THE RECOMMENDED 2010 REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN

Recommended Changes to the National Highway System	Facility	From	То
Additions	CTH BB	Pennsylvania Avenue STH 190 CTH J Oklahoma Avenue CTH ZZ	STH 32 IH 94 STH 164 IH 43/IH 894 CTH BB
Deletions	CTH Y CTH ZZ STH 32 STH 74 STH 164 S. 27th Street Moorland Road Oklahoma Avenue Pilgrim Parkway Pilgrim Road	S. 27th Street IH 94 CTH ZZ Pilgrim Road STH 190 IH 43/IH 894 USH 18 IH 894/USH 45 North Avenue STH 74	STH 62 STH 32 CTH BB USH 41/USH 45 IH 94 CTH Y IH 43 Lake Parkway USH 18 North Avenue

Source: SEWRPC.

scenario is defined as the transportation system represented by the baseline scenario and improvements proposed for implementation by the analysis year: 1996, 2001, 2007, or 2010.

The purpose of defining the action and baseline scenarios is to provide a method by which the estimated reduction in the emission of ozone causing pollutants attributable to the implementation of transportation projects can be assessed. This assessment is made by estimating the anticipated emissions generated under both the baseline and the action scenarios for each analysis year. For any given analysis year, the transportation plan will be found to be in conformance with the State Implementation Plan for air quality if the level of emissions forecast for the action scenario is less than the level forecast for the baseline scenario. It must be demonstrated that the action scenario results in a reduction of ozone causing emissions after discounting all other factors that may also contribute to a reduction in such emissions, such as the Federally-mandated actions attendant to alternative fuels and enhanced motor vehicle inspection and maintenance programs.

The Federally required analyses demonstrating that the regional transportation system plan conforms to the State Implementation Plan for air quality are provided in SEWRPC report Assessment of Conformity of the New Year 2010 Regional Transportation System Plan and the 1995 through 1997 Transportation Improvement Program with Respect to the State of Wisconsin Air Quality Implementation Plan. Tables 276 and 277 summarize those analyses in terms of identifying incremental arterial street and highway and transit improvements, respectively, for each of the four analysis years. More specifically, Table 276 identifies the amount of additional lane-miles of arterial streets and highways to be constructed within each county of the Region for each analysis period. Table 277 provides information on the incremental vehiclemiles of revenue transit services to be provided within each analysis period by transit service type, rapid, express, and local.

For analytical purposes, the findings of conformity between the regional transportation system plan and the Wisconsin State Implementation Plan for air quality are predicated upon the existence in each of the four analysis years, 1996, 2001, 2007, and 2010, of specific arterial street and highway and transit networks. The detailed configuration of those networks is addressed in SEWRPC report <u>Assessment of Conformity of the New Year 2010 Regional Transportation System Plan and the 1995 through 1997 Transportation Improvement Program with Respect to the State of Wisconsin Air Quality Implementation Plan. It is recom-</u>

RECOMMENDED ADDITIONS TO, AND DELETIONS FROM, THE PROPOSED NATIONAL HIGHWAY SYSTEM WITHIN SOUTHEASTERN WISCONSIN



Map 147

THE PROPOSED NATIONAL HIGHWAY SYSTEM WITHIN SOUTHEASTERN WISCONSIN: JUNE 1994



Source: Wisconsin Department of Transportation.

Source: SEWRPC.

Map 149

RECOMMENDED NATIONAL HIGHWAY SYSTEM WITHIN SOUTHEASTERN WISCONSIN



Source: SEWRPC.

The Federal Intermodal Surface Transportation Efficiency Act of 1991 requires that a National Highway System be nominated by the Federal Highway Administration and confirmed by the U.S. Congress by September 30, 1995. Consultation between the Wisconsin Department of Transportation, the Regional Planning Commission, and concerned local units of government has resulted in the proposed National Highway System shown on Map 147. The system, as initially proposed, would include most of the existing freeways in the Region, totalling about 242 route-miles, together with 400 route-miles of surface arterial streets and highways. The National Highway System in the Southeastern Wisconsin Region would total 642 route-miles.

Recommended revisions to the initially proposed National Highway System in the Region are shown on Map 148. These revisions reflect the changes in jurisdictional responsibility of significant arterial street and highways in the Region recommended in the year 2010 plan. All arterial routes ultimately included on the National Highway System are existing or planned State trunk highways.

As a result of the revisions to the initially proposed National Highway System made necessary by jurisdictional changes of certain arterial facilities recommended in the year 2010 plan, the National Highway System for the Region would total 613 routemiles (see Map 149). The system would include 242 route-miles of freeway and 371 route-miles of surface arterial streets and highways.

NUMBER AND PERCENT OF ADDITIONAL LANE-MILES OF ARTERIAL STREETS AND HIGHWAYS IN THE REGION BY COUNTY AND IMPLEMENTATION PRIORITY SCHEDULE: 1996, 2001, 2007, AND 2010

	Proposed State Trunk Highway Incremental Arterial Lane-Miles											
	19	96	20	2001 2007		20	2010		Total			
County	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total		
Kenosha	2	4	18	32	20	36	16	28	56	100		
Milwaukee	4	6	29	46	14	22	16	26	63	100		
Ozaukee			24	30	33	40	24	30	81	100		
Racine			48	36	23	17	64	47	135	100		
Walworth	3	2	25	14	24	13	128	71	180	100		
Washington	6	6	17	18	44	46	28	30	95	100		
Waukesha	9	6	33	23	73	50	31	21	146	100		
Region	24	3	194	26	231	30	307	41	756	100		

		Proposed County and Local Trunk Highway Incremental Arterial Lane-Miles										
	19	96	20	2001		07	20	2010		Total		
County	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total		
Kenosha			10	19	16	30	27	51	53	100		
Milwaukee	14	20	17	25	20	29	18	26	69	100		
Ozaukee			12	55	10	45			22	100		
Racine			2	7	9	32	17	61	28	100		
Walworth					4	21	15	79	19	100		
Washington	2	4	13	25	11	22	25	49	51	100		
Waukesha	15	8	26	13	54	28	98	51	193	100		
Region	31	7	80	18	124	29	200	46	435	100		

		Proposed Total Incremental Arterial Lane-Miles											
	19	1996		2001		007	2010		Total				
County	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total			
Kenosha	2	2	28	26	36	33	43	39	109	100			
Milwaukee	18	14	46	34	34	26	34	26	132	100			
Ozaukee			36	35	43	42	24	23	103	100			
Racine			50	31	32	19	81	50	163	100			
Walworth	3	1	25	13	28	14	143	72	199	100			
Washington	8	5	30	21	55	38	53	36	146	100			
Waukesha	24	7	59	17	127	38	129	38	339	100			
Region	55	5	274	23	355	30	507	42	1,191	100			

Source: SEWRPC.

mended that, as the target years approach, the basic test of plan implementation, insofar as conformity with the air quality plan is concerned, be measured differently for the principal regional arterial highway facilities than for the minor arterial street facilities and local transit services. The former category, all of which consists of State trunk highway facilities, can be programmed with greater certainty since a simple agency exists that is staffed to program expeditiously the design and construction of the facilities concerned. The latter category of projects depends for implementation on a diffused decision-making structure that consists of seven counties and numerous cities, villages, and towns. Consequently, the programming of this latter group of facilities and services is far more difficult and requires the provision for greater flexibility in the programming of improvements. This should not, however, detract from the overall attainment of plan implementation levels which are reflected in Tables 276 and

NUMBER AND PERCENT OF ADDITIONAL REVENUE VEHICLE-MILES OF TRANSIT SERVICE IN THE REGION BY TYPE OF SERVICE AND IMPLEMENTATION PRIORITY SCHEDULE: 1996, 2001, 2007, AND 2010

	Existing Transit Vehicle-Miles		Proposed Incremental Transit Vehicle-Miles of Revenue Service										
		1996		2001		20	2007		10	Total			
Transit Service Type	of Revenue Service: 1991	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total	Number	Percent of Total		
Rapid Express Local	3,400 3,300 56,600	-300 2,700 200	-2.2 14.9 1.3	5,000 3,500 7,700	37.0 19.3 49.0	6,100 5,800 7,800	45.2 32.1 49.7	2,700 6,100 	20.0 33.7	13,500 18,100 15,700	100.0 100.0 100.0		
Total	63,300	2,600	5.5	16,200	34.3	19,700	41.6	8,800	18.6	47,300	100.0		

Source: SEWRPC.

277 in terms of the number of additional lanemiles of arterial streets and highways, the incremental vehicle-miles of transit services, and the impacts on air pollutant emissions.

FUNDING PLAN IMPLEMENTATION

As noted in Chapter XI, an analysis of plan implementation costs and of potential revenues for such implementation, the latter based upon current revenue levels and patterns, indicates that an average annual shortfall in revenues over the 16-year period 1995 through 2010 of about \$222 million may be expected. This section of this chapter addresses that funding deficit, setting forth a recommended course of action for securing the additional revenues necessary to implement the plan. If such revenues cannot be secured in a timely manner, a restructuring of the plan to reduce costs to match the available revenues may become necessary.⁹

State Trunk Highway

Revenues and Expenditures

The average annual cost of implementing the State trunk highway preservation, improvement, and expansion element of the recommended plan is about \$229 million (see Table 278). This includes a nearly \$1.5-billion program to reconstruct and modernize the Milwaukee-area freeway system, as well as a series of projects to preserve, improve, and expand the remainder of the State trunk highway system in the Region. This also includes the estimated costs of operating and maintaining that system over the plan implementation period.

An analysis of the existing revenue sources that support the State trunk highway program indicates that those revenues, for both capital and operating and maintenance purposes, currently total about \$144 million annually. Thus, there is an estimated revenue shortfall of \$85 million annually to implement the State trunk highway element of the plan. Expressed in terms of an equivalent motor fuel tax, that shortfall approximates 10.0 cents per gallon on the basis of the amount of motor fuel estimated to be sold within the Region.

The recommended regional transportation system plan was completed almost simultaneously with, and was fully coordinated with, the Wisconsin Department of Transportation Intermodal Transportation Plan developed through the long-range Statewide transportation planning process termed TRANSLINKS 21.¹⁰ The draft State plan recognizes a need, not only in Southeastern Wisconsin, but also throughout the remainder of the State, to provide increased funding for all modes of transportation. It proposes that over the period 1995 through 2020 Statewide transportation user fees be increased about every five years to pay for implementing the multi-modal State plan.

Full funding of the TRANSLINKS 21 proposals would yield about \$95 million annually in

⁹The cost entailed in implementing the Federallymandated conversion of signage to the metric system of measurement has not been included in the cost estimates set forth in this chapter.

¹⁰See report, <u>Wisconsin TRANSLINKS 21 Draft</u> <u>Intermodal Transportation Plan</u>, published by the Wisconsin Department of Transportation in September 1994.

AVERAGE ANNUAL COSTS AND REVENUE ASSOCIATED WITH IMPLEMENTATION OF THE STATE TRUNK HIGHWAY PRESERVATION, IMPROVEMENT, AND EXPANSION PROJECTS INCLUDED IN THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN: 1995 THROUGH 2010

	Millions of Dollars
	(average annual
Cost or Revenue Item	1995-2010)
State Trunk Highway Cost	1
Capital	
Preservation	157.4
Improvement	35.1
Expansion	14.1
Subtotal	206.6
Operation and Maintenance	22.6
Total	229.2
State Trunk Highway Revenues	
Capital ^a	
Federal (modified existing trend)	64.0
State (modified existing trend)	60.0
Subtotal	124.0
Operation and Maintenance (existing trend)	20.0
Total	144.0
Cost-Revenue Comparison	
Average Annual Difference between Costs and Revenues (existing trend)	-85.2
Equivalent Motor Fuel Tax in Cents per Gallon of the Cost-Revenue Shortfall	10.0
Resolving the Funding Shortfall	
Estimated Additional State Revenue from TRANSLINKS 21 Proposals ^b	85.2
Equivalent Motor Fuel Tax in Cents per Gallon of Additional TRANSLINKS 21 Funds	10.0

^aThe Federal capital revenue estimate assumes continuation of Federal highway funding to the State at the levels authorized in the Intermodal Surface Transportation Efficiency Act of 1991, as well as the full appropriation of those funds, and the receipt by the Region of a fair share of those funds defined as 30 percent of State funding, which is the Region's share of Statewide vehicle-miles of travel. This includes \$10 million of Interstate Maintenance funds; \$17 million of National Highway System funds; \$33 million of Surface Transportation Program, discretionary and other funds; and \$4 million of Surface Transportation Program, Milwaukee area funds. The State capital revenue estimate assumes continuation of State trunk highway construction funding at historic levels of about \$200 million Statewide annually, and the receipt by the Region of 30 percent, or a fair share, of that funding. Historically, the Region has received about 23 percent of such funding, or about \$47 million annually. The difference between the assumed funding level of \$60 million and the historic level of \$47 million is \$13 million annually.

^bThe TRANSLINKS 21 proposals for increased funding of the State trunk highway system, including the Milwaukee-area freeway system, would provide an additional \$95.0 million annually to the seven-county Southeastern Wisconsin Region. The difference between the shortfall and the estimated additional revenue, about \$9.8 million annually, represents the revenue needed to provide State trunk highway funding in the Region at a "fair-share" level.

Source: SEWRPC.

additional funding in the Region for State trunk highway purposes. This revenue enhancement would be sufficient to fund the anticipated shortfall attendant to implementation of the recommended State trunk highway improvements within the Region, including the major rehabilitation of the Milwaukee area freeway system. In addition, there would be an opportunity to secure revenue for regional projects from the proposed major investment studies described later in this chapter, particularly including any busways and high-occupancy vehicle (HOV) lanes that might be identified for implementation in such studies.

County and Local Highway and Transit Revenues and Expenditures¹¹

The average annual cost to implement the county and local arterial highway preservation, improvement, and expansion element of the recommended plan is about \$114 million (see Table 279). This includes the estimated costs of operating and maintaining the county trunk highways and local arterial street systems over the plan implementation period. This cost does not include the capital and operating costs attendant to local land-access and collector streets.

An analysis of the existing revenue sources that support the county and local arterial street and highway programs indicates that those revenues, for both capital and operating and maintenance purposes, currently total about \$73

¹¹The expenditure restraint program, formerly known as the tax rate disparity program, provides State aid to cities, villages, and towns as an incentive for limiting increases in local expenditures. The program applies to communities with a property tax rate for general purposes exceeding five mills. Such communities are eligible for aid under the expenditure restraint program if the annual percentage increase in its municipal budget, exclusive of principal and interest on long term debt, is less than the sum of the annual percentage increase in the consumer price index plus a growth factor percentage. The growth factor percentage is calculated for each community based upon the change in equalized value due to new construction and other factors; it may not exceed 2 percent. Qualifying communities share in funds budgeted for the program by the State Legislature; forty two million dollars is currently budgeted. The formula for allocation of the aid among qualifying communities takes into account the equalized value of property in each community and each community's property tax rate. This program constitutes an additional source of uncertainty attendant to implementation of the local transportation improvements recommended in the plan.

million annually within the Region. Thus, there is an estimated shortfall of \$41 million annually to implement the county and local arterial street and highway element of the recommended regional system plan. Expressed in terms of a motor fuel tax, that shortfall approximates five cents per gallon on the basis of the amount of motor fuel estimated to be sold within the Region. As shown in Table 279, that shortfall varies throughout the Region, ranging from a low of the equivalent of a 1.4 cents per gallon motor fuel tax in Milwaukee County to a high of 12.5 cents per gallon in Walworth County.

Also as shown in Table 279, the draft State Intermodal Transportation Plan recognizes the need to provide additional State revenues to help county and local governments fund solutions to the increasing needs associated with the arterial street and highway system. Based on the TRANSLINKS 21 proposal, an additional approximately \$13 million annually may be expected over the plan implementation period through the Wisconsin Local Road Improvement Program (LRIP). This proposed infusion of additional State aid would reduce the anticipated average annual funding deficit for county and local arterial streets and highways attendant to regional plan implementation from about \$41 million annually to about \$28 million annually. The additional State funds would represent the equivalent of nearly two cents per gallon in terms of a motor fuel tax levied within the Region. The resulting remaining deficit of about \$28 million annually would represent the equivalent of about three cents per gallon in terms of such a motor fuel tax. As shown in Table 279, that equivalent motor fuel tax would range from a low of no tax at all in Milwaukee County to a high of about 11 cents per gallon in Walworth County.

The average annual cost of implementing the transit element of the recommended regional transportation system plan is about \$179 million (see Table 280). This includes the estimated costs of building transit stations and acquiring rolling stock, as well as operating the transit system after taking into account anticipated farebox revenues. This amount also includes the capital and operating costs associated with the busway and light-rail facilities included in the recommended plan pending the results of the east-west corridor major investment study being conducted by the Wisconsin Department of Transportation.

AVERAGE ANNUAL COSTS AND REVENUE ASSOCIATED WITH IMPLEMENTATION OF THE COUNTY AND LOCAL TRUNK HIGHWAY PRESERVATION, IMPROVEMENT, AND EXPANSION ELEMENT OF THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN BY COUNTY: 1995 THROUGH 2010

	Millions of Dollars (average annual 1995-2010)								
County									
Cost or Revenue Item	Kenosha	Waukesha	Bogier						
County/Local Trunk Highway Costs	Kentosna	IVIII WEALKEE	OZaukee		waiworth	washington	vvaukesna	Region	
Capital									
Preservation	3.8	14.8	3.9	4.2	3.4	4.3	7.9	42.3	
Improvement	2.1	2.4	0.7	1.0	1.1	0.7	8.8	16.8	
Expansion	0.9	3.9	0.6	0.6	1.1	2.3	2.2	11.6	
Subtotal	6.8	21.1	5.2	5.8	5.6	7.3	18.9	70.7	
Operation and Maintenance	3.2	15.3	2.9	5.3	3.0	4.1	9.7	43.5	
Total	10.0	36.4	8.1	11.1	8.6	.11.4	28.6	114.2	
County/Local Trunk Highway Revenues Capital			-						
Federal (existing trend)	0.7	7.5	0.9	0.7	0.3	0.5	3.7	14.3	
State (existing trend)	0.5	3.6	0.3	0.6	0.1	0.7	2.0	7.8	
County/Local (existing trend)	0.7	6.0	0.4	1.0	0.1	1.3	3.8	13.3	
Subtotal	1.9	17.1	1.6	2.3	0.5	2.5	9.5	35.4	
Operation and Maintenance							-		
State (existing trend)	0.8	3.5	0.7	1.2	0.7	0.9	2.2	10.0	
County/Local	2.0	10.4	1.7	3.2	1.9	2.5	5.9	27.6	
Subtotal	2.8	13.9	2.4	4.4	2.6	3.4	8.1	37.6	
Total	4.7	31.0	4.0	6.7	3.1	5.9	17.6	73.0	
Cost-Revenue Comparison Average Annual Difference between		_							
Costs and Revenues (existing trend) Equivalent Motor Fuel Tax in Cents per Gallon	-5.3	-5.4	-4.1	-4.4	-5.5	-5.5	-11.0	-41.2	
	9.3	1.4	9.1	6.0	12.5	10.4	6.2	4.8	
Resolving the Funding Shortfall Estimated Additional State Revenue				-					
from TRANSLINKS 21 Proposals	1.4	5.4 ⁰	0.7	1.4	0.7	0.7	2.8	13.1	
of Additional TRANSLINKS 21 Funds	2.5	1.4	1.5	1.9	1.6	1.3	1.6	1.5	
Local-County-Regional Sources	3.9		3.4	3.0	4.8	4.8	8.2	28.1	
of Remaining Required Revenue	6.8		7.6	4.1	10.9	9.1	4.6	3.3	

^aBecause of many factors that affect the revenue yield of a motor fuel tax, including the efficiency of the motor vehicle fleet in terms of fuel utilization, the amount of vehicle-miles of travel, transit use, general price inflation, and the general condition of the National, State, and regional economies, it was determined for the purpose of this analysis to make simplifying assumptions that would tend to yield conservative estimates of motor fuel tax revenues in future years. More specifically, it was assumed that any change in the average fuel utilization efficiency of the motor vehicle fleet over the plan implementation period would be offset by anticipated increases in vehicle-miles of travel. Presently, the fleet efficiency approximates 21 miles per gallon. At that efficiency level, and at current travel levels, a one-cent-per-gallon motor fuel tax yields the following estimated revenues annually: Kenosha County, \$0.57 million; Milwaukee County, \$4 million; Ozaukee County, \$0.44 million; Washington County, \$0.53 million; and Waukesha County, \$1.77 million. The regional average is \$8.5 million annually.

^b Milwaukee County would receive an additional \$2.3 million annually under the TRANSLINKS 21 proposals. This additional amount could be applied to the \$6 million in county and local funds now being spent for this purpose and would represent property tax relief.

Source: SEWRPC.

An analysis of the existing revenue sources supporting county and local public transit programs in the Region indicates that those revenues, for both capital and operating and maintenance purposes, currently total nearly \$83 million annually. Thus, there is an estimated

shortfall of \$96 million annually to implement the transit element of the plan. Expressed in terms of a motor fuel tax, that shortfall approximates 11 cents per gallon based upon the amount of motor fuel estimated to be sold in the Region. As shown in Table 280, that shortfall

AVERAGE ANNUAL COSTS AND REVENUE ASSOCIATED WITH IMPLEMENTATION OF THE TRANSIT ELEMENT OF THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN BY COUNTY: 1995 THROUGH 2010

Millions of Dollars (average annual 1995-2010) County	
County Co	
Cost or Revenue Item Kenosha Milwaukee Ozaukee Racine Walworth Washington Waukesha Regi	gion
Transit Costs	
Capital ^a	6.8
Operating ^b	1.7
Total 9.6 139.7 1.1 9.4 0.8 17.9 178	8.5
Transit Revenues Capital	
Federal (existing trend) 1.2 9.8 0.5 0.7 12	2.2
County/Local (existing trend) 0.1 2.2 0.1 0.1 2	2.5
Subtotal 1.3 12.0 0.6 0.8 14	4.7
Operating	
Federal (existing trend)	6.0
State (existing trend)	5.1
County/Local (existing trend) 0.5 15.5 0.5 17	7.1
Subtotal 2.1 60.7 2.9 2.5 68	8.2
Total 3.4 72.7 3.5 3.3 82	2.9
Cost-Revenue Comparison Average Annual Difference between Costs and	
Revenues (existing trend)	5.6
of the Cost-Revenue Shortfall	1.2
Resolving the Funding Shortfall	
Estimated Additional Federal Revenue ^C	9.5
Estimated Additional State Operating Revenue Assumed Prior to TRANSLINKS 21 Proposals ^d 3.7 1.9 0.7 2.7 0.6 3.7 13	3.3 ^e
Estimated Additional State Operating Revenue 1.2 8.0 1.2 1.3 11 Form TRANSLINKS 21 Proposals ¹ 1.2 8.0 1.2 1.3 11	1.7 ^e
Proposed New State Transit Capital Program 1.1 9.9 0.2 0.8 0.1 1.7 13	3.8
Foregoing Additional Federal and State Funds 10.5 13.5 2.0 6.4 1.3 6.9 9	9.2
Remaining Revenue Required from	
Local-County-Regional Sources	7.3
of Remaining Required Revenue	2.0

^a Transit capital costs include \$792 million in expenditures associated with the proposed busway in the east-west travel corridor and the proposed light-rail transit lines in the east-west and Northeast travel corridors. The following assumptions were made attendant to the funding of these facilities, which are subject to confirmation in the major investment study being sponsored by the Wisconsin Department of Transportation: 1) all of the Interstate Conversion Estimate (ICE) monies reserved for Wisconsin, about \$289 million, would be made available for these projects; 2) Wisconsin would receive a previously reserved Federal Transit Administration Section 3 grant of \$200 million for this purpose; 3) Wisconsin would ultimately receive an additional \$145 million in Federal funds for this purpose, thus raising the total Federal contribution to an 80 percent participation rate; and 4) the required 20 percent match to Federal funds would be divided equally between the State of Wisconsin and the local governments concerned. In addition, for the purpose of allocating costs within this table to the counties, it was assumed that all costs associated with the lightrail facilities would be incurred within Milwaukee County, and that the costs of the busway would be apportioned 60 percent to Milwaukee County and 40 percent to Waukesha County. Detailed funding arrangements for the proposed busway and light-rail facilities are to be determined as the Wisconsin Department of Transportation completes the east-west corridor study.

^bOperating costs are net of transit fares. It was assumed in this analysis that transit fares would over the plan implementation period approximate the existing 1994 fare levels, adjusted as may be necessary over time to account for general price inflation. The current fare levels are: Milwaukee County, local and express service, \$1.25, rapid service, \$1.50; City of Waukesha, local service, \$0.75; City of Kenosha, local service, \$0.75; City of Racine, local service, \$0.60; and rapid service in the remainder of the Region, a distance based fare schedule ranging from \$1.50 to \$3.00.

^cAll of this additional revenue is associated with the proposed busway and light-rail transit facilities in the east-west and northeast travel corridors.

^dRepresents incremental State operating revenues needed to provide for State operating assistance equal to 42 percent of the projected total transit operating expenses within each county up to the amount of the projected operating deficits exceeding the existing trend operating revenues. Assumes that a fixed level of local matching funds will no longer be required for use of State operating assistance funds.

^e Together these amounts total \$25.0 million annually. This total represents about two-thirds of the statewide total average annual new urban transit funds proposed in the TRANSLINKS 21 plan. The assumed apportionment to Southeastern Wisconsin reflects historic experience. Programmatic proposals and constraints outlined in the TRANSLINKS 21 plan are assumed to be subject to change to meet plan implementation needs.

^f The additional State operating revenues were assumed to be used first to increase the combined Federal and State operating levels within Racine and Kenosha Counties to cover up to 65 percent of the projected total transit operating expenses up to the amount of the projected operating deficits exceeding the existing trend revenues. The remaining funds were used to increase the combined Federal and State operating assistance levels within Milwaukee and Waukesha Counties to cover up to about 54 percent of the projected total transit operating expenses. Assumes that a fixed level of local matching funds will no longer be required for use of State transit operating assistance funds. varies throughout the Region, ranging from no shortfall at all in Walworth County to a high of nearly 17 cents per gallon in Milwaukee County.

Also as shown in Table 280, substantial additional Federal and State monies may be available to reduce the transit-related shortfall. It is assumed that, over the plan implementation period, nearly \$40 million annually in supplemental Federal funds will be made available for the purposes of constructing the busway and light-rail transit facilities identified in the aforereferenced east-west corridor major investment study. This includes an estimated \$289 million in Federal Interstate Conversion Estimate (ICE) monies; an estimated \$200 million in Federal Transit Administration Section 3 monies reserved in the present Federal transportation legislation for Milwaukee; and an additional \$145 million in Federal funds presumed to be secured in subsequent Federal transportation legislation.

In addition, the draft State Intermodal Transportation Plan recognizes the need to provide additional State revenues to help county and local governments fund improved transit service, including the extension of transit service to provide greater geographic service in metropolitan regions and more frequent service reflected in shorter vehicle headways. Based on the TRANSLINKS 21 proposal, an additional approximately \$25 million could be expected over the plan implementation period through the operation of various State transit operating assistance programs.¹² In addition, it is tentatively proposed that a State transit capital program be established, subject to confirmation in a special study outlined later in this chapter. The envisioned State transit capital aid program would provide State funding for up to 50 percent of the nonFederal share of transit capital costs. This is consistent with the position being taken by the Wisconsin Department of Transportation in the east-west travel corridor major investment study. As shown in Table 280, it is estimated that such a new program could bring an additional \$13.8 million annually in State aid to county and local governments in the Region which sponsor the transit systems.

The foregoing assumed total infusion of additional Federal and State aid would reduce the anticipated average annual funding deficit for county and local transit systems from about \$96 million annually to about \$17 million annually. The additional Federal and State funds would represent the equivalent of about nine cents per gallon in terms of a motor fuel tax levied within the Region. The resulting deficit of about \$17 million annually would represent the equivalent of about two cents per gallon in terms of such a motor fuel tax. As shown in Table 280, that equivalent motor fuel tax would range from a low of no tax at all in Walworth County to a high of about three cents per gallon in Milwaukee County.

Concluding Recommendations

Based upon the foregoing analyses of anticipated expenditures and revenues attendant to implementing the regional transportation system plan, the following conclusions may be drawn:

- 1. Assuming acceptance of the supplemental State funding proposals identified in the Statewide Intermodal Transportation Plan prepared by the Wisconsin Department of Transportation, there should be sufficient Federal and State revenues becoming available over the plan implementation period to fund the plan implementation recommendations attendant to the State trunk highway system, including rehabilitation and reconstruction of the Milwaukee area freeway system.
- 2. Also assuming that the proposed supplemental Federal transit funding, together with the supplemental funding recommendations for county and local arterial highways and transit identified in the draft Intermodal Transportation Plan for Wisconsin, are forthcoming, the anticipated funding deficits with respect to implementation of those two elements of the regional transportation system plan should be significantly reduced. Nevertheless, there is an anticipated funding deficit of about \$28 million annually attendant to the county and local arterial street and highway system, and an additional funding deficit of about \$17 annually attendant

¹²This estimate assumes that certain existing programmatic constraints would ultimately be adjusted to reflect needs, including a cap of 50 percent State operating assistance in the Milwaukee area and a requirement that there be fixed levels of local matching funds in order to obtain State operating assistance funds.

SUMMARY OF ADDITIONAL LOCAL REVENUE REQUIRED TO IMPLEMENT THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN BY COUNTY: AVERAGE ANNUAL, 1995 THROUGH 2010

	Average Ar Local Fund (m	nnual Plan Impl ing Shortfall: 1 illions of dolla	ementation 995-2010 rs)	Equiva Required (c	alent Motor Fu I to Fund the S cents per gallo	el Tax Shortfall ^a n)	Equivalent General Sales Tax Required to Fund the Shortfall ^b (percent)			
County	Arterial Highways	Transit	Total	Arterial Highways	Transit	Total	Arterial Highways	Transit	Total	
Kenosha	3.9	0.2	4.1	6.8	0.4	7.2	0.5	c	0.5	
Milwaukee		13.2	13.2		3.3	3.3		0.2	0.2	
Ozaukee	3.4	0.2	3.6	7.6	0.4	8.0	0.7	C	0.7	
Racine	3.0	1.2	4.2	4.1	1.6	5.7	0.3	0.1	0.4	
Walworth	4.8		4.8	10.9		10.9	1.0	• • ·	1.0	
Washington	4.8	0.1	4.9	9.1	0.2	9.3	0.8	°C	0.8	
Waukesha	8.2	2.4	10.6	4.6	1.4	6.0	0.4	0.1	0.5	
Region	28.1	17.3	45.4	3.3	2.0	5.3	0.2	0.1	0.3	

^aFor the assumptions attendant to the calculation of these estimates, see footnote "a" to Table 279.

^bFor analysis purposes, it was assumed that existing sales tax yields would be maintained over the plan implementation period. Presently, a 1 percent sales tax yields the following amounts of revenue annually: Kenosha County, \$8.5 million; Milwaukee County, \$63.9 million; Ozaukee County, \$4.9 million; Racine County, \$11.7 million; Walworth County \$4.9 million; Washington County, \$6.4 million; and Waukesha County \$20.3 million. The regional total is \$120.6 million.

^CLess than 0.1 percent.

Source: SEWRPC.

to the public transit systems. Together these two anticipated funding deficits total about \$45 million annually and represent a significant challenge to be addressed as plan implementation proceeds.

The creation of a regional transportation authority, which proved to be politically infeasible in 1993, would have provided a good structure for securing and allocating the additional revenues required to implement the county and local arterial street and highway and transit elements of the plan. The authority would have been empowered to raise sufficient revenues to fund the anticipated shortfalls through a combination of transportation user fees and a general sales tax. Most transportation systems in large metropolitan areas in the Nation are funded in part by a dedicated areawide revenue source, typically a sales tax.

Assuming that there continues to be insufficient political support for the creation of a regional transportation authority and further assuming that county and local property taxes can no longer be viewed as either an adequate or an appropriate funding source for major transportation system improvement, operation, and maintenance activities, then the only other alternative appears to be to seek State enabling legislation that would provide permissive authority to counties to impose, within limits, either supplemental transportation user fees or a supplemental general sales tax, or a combination of such taxes, dedicated to transportation purposes. The local plan implementation funding shortfall by county is identified in Table 281, together with an estimate of the magnitude of a motor fuel tax or general sales tax required within each county to raise the revenue to fund that shortfall. A supplemental motor fuel tax required to fund the shortfall would range from a low of about three cents in Milwaukee County to a high of nearly 11 cents in Walworth County, averaging, throughout the Region, about five cents per gallon. A supplemental general sales tax required to fund the shortfall would range from a low of about 0.2 percent in Milwaukee County to a high of about 1 percent in Walworth County, averaging about 0.3 percent throughout the Region.

Given the foregoing, it is recommended that the seven counties in the Region, in cooperation with the cities, villages, and towns and the Wisconsin Department of Transportation, collectively work

SUMMARY OF ADDITIONAL LOCAL REVENUE REQUIRED TO IMPLEMENT THE RECOMMENDED	
REGIONAL TRANSPORTATION SYSTEM PLAN FOR SOUTHEASTERN WISCONSIN AND TO RELIEVI	E
THE PROPERTY TAX OF COSTS ASSOCIATED WITH ARTERIAL STREETS AND HIGHWAYS BY COUN	ITY

		Current Annual Estimate of Property Tax Funded Expenditures (millions of dollars)							it Motor Fuel T is per gallon)	Equivalent General Sales Tax (percent)			
	Arteri	al Streets and Hig	ghways	Transit				Bequired to	- Required		Bequired to	Required	
County	Capital	Operation and Maintenance	Subtotal	Capital	Operation and Maintenance	Subtotal	Total	Relieve the Property Tax	to Fund Shortfall	Total	Relieve the Property Tax	to Fund Shortfall	Total
Kenosha	0.7	2.0	2.7	0.1	0.5	0.6	3.3	5.8	7.2	13.0	0.4	0.5	0.9
Milwaukee	6.0	10.4	16.4	2.2	15.5	17.7	34.1	8.5	3.3	11.8	0.5	0.2	0.7
Ozaukee	0.4	1.7	2.1				2.1	4.7	8.0	12.7	0.4	0.7	1.1
Racine	1.0	3.2	4.2	0.1	0.5	0.6	4.8	6.5	5.7	12.2	0.4	0.4	0.8
Walworth	0.1	1.9	2.0				2.0	4.5	10.9	15.4	0.4	1.0	1.4
Washington	1.3	2.5	3.8				3.8	7.2	9.3	16.5	0.6	0.8	1.4
Waukesha	3.8	5.9	9.7	0.1	0.6	0.7	10.4	5.9	6.0	11.9	0.5	0.5	1.0
Region	13.3	27.6	40.9	2.5	17.1	19.6	60.5	7.1	5.3	12.4	0.5	0.3	0.8

Source: SEWRPC.

through the State legislative delegation from the Region to secure enabling legislation that would permit the counties to impose either a transportation user fee, such as the motor fuel tax, or a general sales tax, or some combination thereof. at a level sufficient in each county to raise the revenue required to fund the anticipated plan implementation shortfall. Such legislation should ensure that the revenues obtained are dedicated to the sole purpose of implementing the recommendations included in the regional transportation system plan. The legislation must also require the establishment of a technique, perhaps unique to each county, whereby revenues raised within each county are apportioned to the county and the local units of government therein in proportion to their need for such revenue as determined by the implementation recommendations included in the regional plan. State-level oversight would likely be required in this respect as, for example, a determination by the Wisconsin Department of Transportation that a proposed revenue allocation scheme within a given county is consistent with the regional transportation system plan implementation recommendations. The Regional Planning Commission could assist the Department in such matters by making findings appropriate to any given proposal for revenue allocation within a county. It is further recommended that the Wisconsin Department of Transportation, as part of its TRANSLINKS 21 funding initiative, make known to the State Legislature the need to provide the counties in Southeastern Wisconsin with the authority to raise nonproperty-tax revenue dedicated for transportation purposes.

A proposal set forth later in this chapter, which was derived from the public hearing process, calls for a study of the potential for removing from county and local property taxes all existing funding burdens associated with building and maintaining the arterial street and highway system and providing public transit services in each county. An estimate of the current amount of county and local property taxes expended for that purpose in the Region is set forth in Table 282. Over \$60 million of property taxes annually are used in the Region for purposes related to implementation of the regional transportation system plan, about two-thirds of which are expended for arterial streets and highways and about one-third for public transit. Estimates of the equivalent motor fuel tax and general sales tax that would be required within each county and within the Region as a whole to remove these funding burdens from the property tax are also set forth in Table 282. At the regional level, such property tax relief would require about a 7-cent-per-gallon motor fuel tax or a 0.5 percent general sales tax.

Table 282 also identifies the total motor fuel tax or general sales tax that would be required in the Region and within each county of the Region if it were ultimately determined to be desirable to
enable either a regional transportation authority, should the re-creation of such an authority become politically feasible, or individual counties both to fund the plan implementation shortfall and to remove all major transportation expenditures from the property tax. At the regional level, accomplishing these dual objectives would require about a 12-cent-per-gallon motor fuel tax or a 0.8 percent general sales tax.

In any consideration of the financial analysis presented herein, it is important to note that extensive opportunity was provided for public review of, and comment on, the plan and its attendant costs. Two of three issues of the Commission Newsletter published during the planning process presented plan cost-related data, including the equivalent increases in motor fuel taxes that would be required to cover the revenue shortfalls attendant to plan implementation. Further, during the Regional Planning Conference held on June 27, 1994, by the Commission for the purpose of presenting the recommended plan for public review, a session devoted to transportation plan financing was conducted to receive public input on costs. revenues, revenue shortfalls, and potential means of raising additional needed revenues. The Commission also held public hearings on the plan in all seven counties of the Region. During such hearings, the costs associated with the plan, as well as the equivalent increases in the motor fuel tax required to eliminate anticipated funding shortfalls, were addressed. There was no widespread opposition expressed to the findings that additional revenue would be required to implement the plan, nor any suggestions whatsoever made that the plan be scaled back to fit current revenues.

Finally, it is recognized that the Federal Intermodal Surface Transportation Efficiency Act of 1991 requires that transportation system plans be constrained to reasonable estimates of available revenue. Accordingly, it is recommended that monitoring activities be undertaken by the Regional Planning Commission in the near-term plan implementation period with respect to the actions needed to secure the additional funding required to implement the plan. Should that monitoring indicate a determination not to make available sufficient supplemental funding to implement the plan, then it is recommended that such a finding be taken into account as a fourthgeneration regional transportation system plan is prepared for the design year 2020.

DETAILED IMPLEMENTATION PLANNING

More detailed planning will be required prior to the programming of certain elements of the recommended regional transportation system plan. This includes the conduct of work identified under the Federal Intermodal Surface Transportation Efficiency Act as "major investment studies," as well as of the more detailed State, county, or local planning efforts required to refine the basic transit and highway improvement recommendations contained in the plan.

Major Investment Studies

Under Federal law, a major investment study (MIS) is required as an integral part of the implementation of major highway and transit improvement projects recommended in the system plan. While the regional transportation system plan may identify a need for such a project in a travel corridor, a major investment study is required to confirm that need and to provide the basis for more detailed consideration of alternatives before a final decision on the major transportation investment concerned is made. The Federal Highway and Transit Administrations have promulgated rules and guidance under which major investment studies are to be conducted. These studies are intended to focus on travel corridors.

Under the Federal rules, the following types of transportation facility improvements may be included on a provisional basis in a regional transportation system plan, but are subject to further study and confirmation in a major investment study:

- 1. New freeways.¹³
- 2. The addition of through travel lanes to existing freeways for a distance of one mile or more.
- 3. New expressways.¹⁴

¹³A freeway is defined as a fully gradeseparated, divided highway with access permitted only at interchanges with other arterial facilities.

¹⁴An expressway is defined as a partially grade separated, partially divided highway with access permitted only at interchanges and intersections with other arterial facilities.

- 4. The addition of through travel lanes to existing expressways for a distance of one mile or more.
- 5. Any type of fixed-guideway transit facility, including busways, light-rail lines, heavy-rail lines, and commuter-rail lines.

Within Southeastern Wisconsin the Federal rules governing the conduct of major investment studies apply only to the six-county Milwaukee Transportation Management Area (TMA) defined by the U. S. Department of Transportation. That area, which excludes Walworth County, is coterminous with the six-county "severe" ozone nonattainment area for air quality management designated by the U. S. Environmental Protection Agency.

The recommended regional transportation system plan recognizes the potential for 11 major highway or transit facilities that are provisionally included in the plan. For some facilities, initial feasibility studies may first be undertaken to evaluate the project potential further before proceeding to the more expensive and timeconsuming major investment study. Where the project potential is confirmed, a major investment study will be conducted by the Wisconsin Department of Transportation or the Regional Planning Commission, in cooperation with all interested parties, and will evaluate all feasible technologies. Thereafter, the regional plan will be amended to specifically include the project proposed by the major investment study, as well as the financial provisions necessary for its development and operation.

As each major investment study is completed, it is recommended that appropriate consideration be given to establishing a transit corridor overlay zoning district and attendant regulations. Such an ordinance would be intended to be adopted cooperatively by the affected municipalities along whatever rapid transit lines ultimately may be constructed and would be designed to help ensure the proper development and redevelopment of areas near transit stops and stations.

1. Current WisDOT East-West Travel Corridor Study

A corridor study sponsored by the Wisconsin Department of Transportation was underway at the time of the completion of the new regional transportation system plan in 1994. As noted in Chapter XI, a decision was made to incorporate into the new system plan the preliminary recommendations formulated by the advisory committee created by the Department to guide that study. While the current corridor study was conceived prior to the promulgation of rules and guidance by the Federal government attendant to the conduct of major investment studies, it is assumed that the completed study will meet all of the requirements of a major investment study.

The following major transportation facilities have been included in the recommended regional transportation system plan subject to further evaluation in the current major investment study (see Map 150):

- a. A busway/HOV facility in the east-west travel corridor extending for about 15 miles from the Milwaukee CBD west to STH 164 in Waukesha County. That busway would be used by 16 of the 30 rapid transit bus routes recommended in the regional transportation system plan.
- b. A light-rail transit line in the east-west travel corridor extending for about 7.5 miles from the Milwaukee CBD west to the Milwaukee County Research Park located west of the Zoo Freeway and south of Watertown Plank Road.
- c. A light-rail transit line in the northeast travel corridor extending for about 7.5 miles from the CBD of Milwaukee through the University of Wisconsin-Milwaukee campus to a terminus in the City of Glendale in the vicinity of the IH 43 Freeway and E. Hampton Avenue.
- d. Rehabilitation of the east-west freeway, defined for reconstruction purposes as that segment extending along IH 794 from S. Carferry Drive to, and along, IH 94, west to the interchange with STH 16. The rehabilitation of the eastwest freeway would include the redesign and reconstruction of the Marquette, Stadium, and Zoo interchanges; the redesign and reconstruction of the major interchanges connecting the freeway to surface arterials; and the replacement of pavements and bridges.

Map 150





The fixed-guideway transit facilities shown on the accompanying map have been provisionally included in the recommended regional transportation system plan. A corridor study sponsored by the Wisconsin Department of Transportation is currently underway to determine the feasibility of these proposed facilities. Upon completion of that study, the local units of government concerned, the Regional Planning Commission, and the Wisconsin Department of Transportation plan if necessary.

Source: SEWRPC.

2. <u>North Travel Corridor</u> Major Investment Study¹⁵

A major investment study will be required for proposed facilities in the north travel corridor extending from the Milwaukee CBD to the Saukville-Port Washington area of Ozaukee County. The following major transportation facilities have been provisionally included in the recommended regional transportation system plan and are subject to further evaluation in the proposed north travel corridor study (see Map 151):

- a. The widening of the IH 43 freeway from four to six lanes from W. Bender Road in the City of Glendale to Highland Road in the City of Mequon, a distance of about eight miles.
- b. A busway/high occupancy vehicle facility along IH 43 extending for about seven miles from the Milwaukee CBD north to W. Bender Road. That busway would be used by six of the 30 rapid transit bus routes recommended in the regional transportation system plan.

¹⁵This major investment study is proposed to be initiated in 1995.



Map 151

TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED IN A NORTH TRAVEL CORRIDOR MAJOR INVESTMENT STUDY

ĸ.

-	WIDENING OF IH-43 TO SIX LANES
-	NORTH SHORE BUSWAY /HIGH OCCUPANCY VEHICLE LANE
	COMMUTER RAIL LINE TO SAUKVILLE.
	PROPOSED RAPID SERVICE BUS ROUTE
^	PROPOSED RAPID TRANSIT STATION



A major investment study will be required to determine the scope and content of the proposed transportation facilities in the north travel corridor, which extends from the central business district of Milwaukee to the Saukville-Port Washington area of Ozaukee County. This major investment study is proposed to be initiated in 1995.

Source: SEWRPC.

- c. A commuter-rail passenger line extending from the Milwaukee AMTRAK station north for about 28 miles to the Village of Saukville.
- 3. <u>Northwest Travel Corridor</u> Major Investment Study

A major investment study will be required for proposed facilities in the northwest travel corridor extending from the Milwaukee CBD to the City of West Bend in Washington County. The following major transportation facilities have been provisionally included in the recommended regional transportation system plan and are subject to further evaluation in the proposed northwest travel corridor study (see Map 152):

- a. A busway/HOV facility along the Zoo Freeway extending for about eight miles from the Zoo Interchange north to W. Mill Road. That busway would be used by eight of the 30 rapid transit routes recommended in the regional transportation system plan.
- b. A commuter-rail passenger line extending from the Milwaukee AMTRAK station for about 35 miles to a terminal in the City of West Bend.
- 4. <u>South Travel Corridor</u> <u>Major Investment Study</u>¹⁶

A major investment study will be required for proposed facilities in the south travel corridor extending from the Milwaukee CBD to Racine and Kenosha. The following major transportation facilities have been provisionally included in the recommended regional transportation system plan and are subject to further evaluation in the proposed south travel corridor study (see Map 153):

a. A busway/HOV lane extending along IH 43/IH 94 for about six miles from the Milwaukee CBD south to the Mitchell Interchange, along IH 43/IH 894 for about four miles from the Mitchell Interchange to W. Forest Home Avenue, and another along IH 94 south for about three miles from the Mitchell Interchange to W. Rawson Avenue. These busways, which together would total about 13 miles in length, would be used by seven of the 30 rapid transit routes recommended in the regional transportation system plan. The busway along IH 43/IH 894 would be used by two of the rapid transit routes. The busway along IH 94 would be used by five of the rapid transit bus routes.

b. A commuter-rail passenger line extending from the Milwaukee AMTRAK station south through St. Francis, Cudahy, South Milwaukee, and Racine to the Kenosha Metra station in the Kenosha CBD, a distance of about 33 miles. In 1994, Chicago-oriented commuter-rail service was provided by Metra, the Chicago based public agency created to deliver commuter-rail services, from the Kenosha station. Accordingly, this particular major investment study will need to examine potential relationships between the proposed commuter-rail passenger service in Southeastern Wisconsin and the existing Chicagooriented commuter-rail service.

5. Southwest Corridor

Major Investment Study

A major investment study will be required in the southwest travel corridor for a proposed facility provisionally included in the recommended regional transportation system plan: a busway/HOV lane along the Zoo Freeway, extending for about six miles from the Zoo Interchange south through the Hale Interchange to S. 116th Street (see Map 154). This busway would be used by three of the 30 rapid transit routes recommended in the regional transportation system plan.

6. <u>Milwaukee Crosstown Corridor</u> Major Investment Study

> A major investment study will be required in the Milwaukee crosstown travel corridor for a proposed facility provisionally included in the recommended regional transportation system plan: a light-rail transit facility extending from a terminal

 $^{^{16}}$ This major investment study is proposed to be initiated in 1995.



A major investment study will be required to determine the scope and content of the proposed transportation facilities in the northwest travel corridor, which extends from the central business district of Milwaukee to the City of West Bend in Washington County.

Source: SEWRPC.



WG

Map 153

TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED IN A SOUTH TRAVEL CORRIDOR MAJOR INVESTMENT STUDY

LEGEND

SOUTH SHORE BUSWAY / HIGH OCCUPANCY VEHICLE LANE

COMMUTER RAIL LINE TO RACINE AND KENOSHA WITH LINK TO CHICAGO METRA SYSTEM LINE

PROPOSED RAPID SERVICE BUS ROUTE

PROPOSED RAPID TRANSIT STATION



A major investment study will be required to determine the scope and content of proposed facilities in the south travel corridor, which extends from the central business district of Milwaukee to the Racine and Kenosha areas. This corridor study is proposed to be initiated during 1995.

Source: SEWRPC.

TO CHICAGO



TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED IN A SOUTHWEST TRAVEL CORRIDOR MAJOR INVESTMENT STUDY

A major investment study will be required in the southwest travel corridor attendant to a southwest busway/high-occupancy vehicle lane. This facility would extend along the Zoo Freeway for about six miles from the Zoo interchange south through the Hale interchange to 116th Street.

Source: SEWRPC,

near the Southridge Shopping Center in the Village of Greendale and City of Greenfield to and along S. and N. 27th Street to a terminus near the intersection of Teutonia Avenue and W. Florist Avenue in the City of Milwaukee, a distance of about 14 miles (see Map 155). In addition to potential crosstown light-rail transit service, this facility would serve potential light-rail routes to and from the Milwaukee CBD.

7. Mitchell Field Corridor

Major Investment Study

A major investment study will be required in the Mitchell Field travel corridor for a single proposed facility provisionally included in the recommended regional transportation system plan: a light-rail transit facility extending from the Milwaukee CBD south to General Mitchell International Airport and to a terminal near the interchange of IH 94 and W. College Avenue, a distance of about 10 miles (see Map 156).

8. <u>Milwaukee Northridge Corridor</u> Major Investment Study

> The regional transportation system plan recognizes the potential to establish an express transit service in the Milwaukee Northridge travel corridor, extending for about 12 miles from the Milwaukee CBD to

Map 156

TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED IN A MILWAUKEE CROSSTOWN TRAVEL CORRIDOR MAJOR INVESTMENT STUDY



A major investment study will be required in the Milwaukee crosstown travel corridor attendant to a crosstown light-rail line. The light-rail transit line would extend from a terminal near the Southridge Shopping Center in the Village of Greendale and the City of Greenfield to and along North and South 27th Street, ending near the intersection of Teutonia Avenue and W. Florist Avenue in the City of Milwaukee. The facility would extend a distance of about 14 miles.

Source: SEWRPC.

a terminal near the Northridge Shopping Center (see Map 157). This particular service, which could be provided by bus or light-rail service, was the subject of a corridor study conducted by the Commission for Milwaukee County. The findings and recommendations of this study are documented in SEWRPC Community Assistance Planning Report No. 150, <u>A</u> Rapid Transit Facility Plan for the Milwaukee Northwest Corridor, January 1988. Based upon that study, Milwaukee County determined to implement an express bus system in the corridor. It was recognized, however, that, should transit ridership





A major investment study will be required attendant to a light-rail transit facility extending from the Milwaukee central business district south to the General Mitchell International Airport and to a terminal near the interchange of IH 94 and W. College Avenue. This facility, would extend for about 10 miles.

Source: SEWRPC.

increase significantly in the corridor following initiation of the express bus service, it may be desirable to again examine the potential for light-rail service. The recommended regional transportation system plan thus holds open the possibility for such a reexamination.

9. East-West Corridor

Major Investment Study

Because the scope of the current Wisconsin Department of Transportation East-West Corridor Transit Study did not include consideration of commuter-rail passenger service, a future major investment study





TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED

A major investment study will be required attendant to a Northridge light-rail line, which would extend a distance of about 12 miles from the Milwaukee central business district to a terminal near the Northridge Shopping Center.

Source: SEWRPC.

will be required in the east-west travel corridor for the proposed commuter-rail passenger line extending from the Milwaukee AMTRAK station west to the City of Oconomowoc, a distance of about 32 miles (see Map 158). Historically, commuter-rail passenger service was provided on the line in this corridor, the former Chicago, Milwaukee, St. Paul & Pacific Railroad, until its termination in 1972. Because this major investment study will follow the current study in the east-west travel corridor, the future study should be designed to accept as committed decisions those major transportation facilities which ultimately are recommended for implementation through the Wisconsin Department of Transportation East-West Corridor Transit Study.

Map 158



A major investment study will be required attendant to commuter-rail passenger service in the East-West travel corridor between the Milwaukee central business district and the City of Oconomowoc. It is necessary to conduct an additional study in this corridor because the scope of the current Wisconsin Department of Transportation East-West corridor study did not include consideration of commuter-rail passenger service. The commuter-rail line would extend a distance of about 32 miles.

Source: SEWRPC.

629

10. <u>Burlington-Antioch, Illinois,</u> Corridor Major Investment Study¹⁷

A major investment study will be required for the proposed commuter-rail passenger line extending from the City of Burlington in Racine County through the Village of Silver Lake in Kenosha County to Antioch. Illinois, and on into Chicago. This commuter-rail line would represent an extension of Chicago-oriented service of Metra, which is scheduled to begin in 1996. The extension from Antioch to Burlington would represent about 17 miles of additional service, with the Antioch terminal located about 1.5 miles south of the Wisconsin-Illinois state line (see Map 159). It is recommended that this major investment study begin with a limited-scope feasibility analysis and, depending upon the findings and recommendations of that analysis, be expanded to complete the requirements for a major investment study.

11. <u>Walworth-Fox Lake, Illinois,</u> <u>Corridor Major Investment Study¹⁸</u>

A major investment study is required for the proposed commuter-rail passenger line extending from the Village of Walworth in Walworth County to Fox Lake, Illinois, and on into Chicago. This commuter-rail line would represent an extension of the current Chicago-oriented service of Metra. The extension from Fox Lake to Walworth would represent about 26 miles of additional service, about nine of which would be located within Walworth County (see Map 160). A major investment study would be required even though Walworth County is outside of the Milwaukee Transportation Management Area because the two counties in Illinois through which the extension would run, McHenry and Lake, lie within a Chicago-based transportation management area. It is recommended that this major investment study begin with a limited scope feasibility analysis and, depending upon the findings of that analysis, be expanded to complete the requirements for a major investment study.

The only other project identified in the recommended regional transportation system plan that might require a major investment study and that lies within the six-county Milwaukee Transportation Management Area is the extension of the STH 16 freeway to bypass Oconomowoc. That project will not require the conduct of a new major investment study. An abbreviated study of this proposed improvement, completed by the Wisconsin Department of Transportation in 1994, meets the requirements of the Federal regulations implementing the Intermodal Surface Transportation Efficiency Act of 1991. The work included a final Environmental Impact Statement (EIS). A formal record of decision by the Federal and State governments concerned was pending late in 1994. The only other freeway improvement projects included in the recommended regional transportation system plan lie in Walworth County, where the major investment study requirements do not apply and where, therefore, the normal preliminary engineering-environmental impact statement process will be used as a decision making mechanism. The recommended regional transportation system plan does not call for the construction of any expressways.

The Federal rules attendant to major investment studies envision a cooperative and collaborative process involving the Federal and State Departments of Transportation; affected local public works agencies: transit operators, both public and private; environmental agencies and interest groups; and the Regional Planning Commission as the Metropolitan Planning Organization. A major investment study can be initiated by the affected public transit operator, the Regional Planning Commission, or the Wisconsin Department of Transportation. Regardless of sponsorship, upon completion of a major investment study, the Regional Planning Commission must formally act to either confirm or revise the adopted regional transportation system plan if the transportation facilities identified for implementation in the major investment study are to be eligible for inclusion in the regional transportation improvement program.

Transit Development Planning

In addition to the major investment studies identified above, it is recommended that each of the public transit operators in the Region undertake the preparation of transit development plans and programs as a basis for refining and detailing the recommendations of the regional

¹⁷This major investment study is proposed to be initiated in 1995.

¹⁸This major investment study is proposed to be initiated in 1995.



TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED IN A BURLINGTON-ANTIOCH TRAVEL CORRIDOR MAJOR INVESTMENT STUDY

A major investment study will be required attendant to commuter-rail passenger service from the City of Burlington in Racine County through the Village of Silver Lake in Kenosha County to Antioch, Illinois, and on to Chicago. It is recommended that this major investment study begin with a limited-scope feasibility analysis and, depending upon the findings and recommendations of that analysis, be expanded to complete the requirements of a major investment study. This major investment study is proposed to be initiated in 1995.

Source: SEWRPC.





TRANSPORTATION FACILITIES PROPOSED TO BE CONSIDERED IN A WALWORTH-FOX LAKE TRAVEL CORRIDOR MAJOR INVESTMENT STUDY

A major investment study is required attendant to commuter-rail passenger service from the Village of Walworth in Walworth County to Fox Lake, Illinois, and on to Chicago. It is recommended that this major investment study begin with a limited-scope feasibility analysis and, depending upon the findings of that analysis, be expanded to complete the requirements of a major investment study. This major investment study is proposed to be initiated in 1995.

Source: SEWRPC.

plan and for programming projects to implement that plan. Typically, such plans and programs are prepared with a relatively short-term, fiveyear, time horizon. These plans and programs provide the basis for day-to-day decision making on new transit system starts and on modifications to existing transit services. These plans provide the basis, then, for the programming of transit projects by each operator in terms of their individual agency budgets, as well as for submittal of projects to be included in the regional transportation improvement program.

In carrying out the transit development planning, it is further recommended that each transit operator give attention to the need to identify ways to improve nonmotorized access to and from transit stops and stations, to the desirability of encouraging the development of multipurpose land use activity centers at major transit stations, to improve transit signage and route information materials, and to the feasibility of offering an Employee Commute Options (ECO) Pass program, whereby major employers would be encouraged to purchase, at discounted rates, transit passes for use by all its employees. It also recommended that each transit operator explore the desirability of providing neighborhood-based transit vans to "shuttle" individuals between their homes and major transit stations and activity centers. Finally, it is further recommended that each transit operator continue to plan for improved paratransit services to elderly and disabled individuals, looking in particular toward the improved coordination of service delivery.

Arterial Street and Highway Planning

County and local public works agencies may also undertake detailed implementation planning attendant to the recommended regional arterial street and highway system. Such planning can serve as a basis for amendment of the regional transportation system plan, but more frequently is used as a basis for refining and detailing that plan and, in particular, for identifying recommended urban and rural arterial street and highway cross-sections and for the determination of right-of-way requirements attendant to such sections.¹⁹ It is recommended that each county and public works agency consider preparing a plan that would refine and detail the arterial street and highway element of the regional transportation system plan.

Upon completion of any needed detailing and refinement of the arterial street and highway element of the system plan, including preliminary engineering studies, it is recommended that, as appropriate, the Wisconsin Department of Transportation, each county highway and public works agency, and each local public works agency take steps to reserve the required future rights-of-way by means of official mapping, building setback line ordinances, land division ordinances, and private deed restrictions. Such prior reservation of right-of-way serves as an expression of governmental intent to acquire land for highway purposes in advance of actual facility construction and thereby not only achieves economies in right-of-way acquisition, but also permits land adjacent to the rightof-way to be privately purchased and developed or redeveloped with full knowledge of the future highway development proposals. The most effective and efficient means of prior reservation of right-of-way is the use of official mapping powers granted to the Wisconsin Department of Transportation, as well as to counties, cities, villages, and towns, in Wisconsin.

MANAGEMENT SYSTEMS

The Intermodal Surface Transportation Efficiency Act requires the State of Wisconsin to

develop six transportation management systems: pavement, bridge, safety, congestion, public transportation, and intermodal management. These management systems, as defined by the applicable Federal rules, are intended to assist decision makers in identifying cost effective actions to improve the efficiency and safety of the transportation infrastructure. The congestion, public transportation, and intermodal management systems, in particular, are important to the successful implementation of metropolitan transportation system plans. These latter three management systems must be integrated and incorporated into the ongoing regional planning process conducted within Southeastern Wisconsin. The objective of such integration is to ensure the provision of the information necessary to implement and to periodically review and revise, as may be found necessary, recommended areawide plans throughout the implementation periods.

The management systems are viewed by the Federal government as ongoing processes. The congestion, public transportation, and intermodal management systems are to be designed to include: 1) the continuing collection of data concerning congestion levels, public transit assets, and the use of intermodal facilities and services, 2) the analyses of alternatives to address problems associated with traffic congestion, transit assets, and intermodal transportation, 3) the selection and implementation of preferred alternatives to achieve the resolution of existing and future potential problems, and 4) monitoring to ascertain if the preferred alternatives have been implemented and the identified problems resolved. It is envisioned that the analyses and conclusions drawn from the management systems will provide the basis for periodically reevaluating and, as may be necessary, amending the regional transportation system plan.

The State of Wisconsin may enter into agreements with local units of government, as well as with the Regional Planning Commission, to implement parts of any management system, but the State is ultimately responsible for overseeing and coordinating the implementation of all of the required management systems. The Federal regulations require that the State of Wisconsin develop a work program identifying the major activities attendant to each of the required management systems and the parties to be responsible for such activities. The manage-

¹⁹For an example of this type of implementation planning, see SEWRPC Community Assistance Planning Report No. 210, <u>City of West Bend</u> <u>Transportation System Plan: 2010, Washington</u> County, Wisconsin, March 1994.

ment systems within the Southeastern Wisconsin Region are to be fully operational by October 1, 1995.

It is recommended that the Regional Planning Commission be assigned a significant role in carrying out the requirements of the congestion management system. Indeed, the transportation system planning process employed by the Commission was designed to meet the requirements of a congestion management system; the new required transportation system plan is intended to comprise a congestion management plan for Southeastern Wisconsin. As described in Chapter IX of this report, the plan design process explicitly addressed existing and anticipated future traffic congestion problems in a disciplined way so as to ensure that highway capacity expansions were proposed only as a last resort to resolve traffic congestion problems. The Commission has the capability to monitor traffic congestion problems within the Region and to evaluate the effectiveness of implemented measures proposed to resolve such problems.

It is important to note that requirements relating to the congestion management system extend to the project level of transportation planning. In this regard, the Federal regulations require that, where the provision of additional highway capacity is determined to be an appropriate measure to relieve traffic congestion, noncapacity improvements which would serve to maintain the functional integrity of expanded highway capacity must be considered. It is recommended, therefore, that the Wisconsin Department of Transportation, in its preparation of the congestion management system work program, include land use planning and detailed site design among the demand management and facility operational measures that must be incorporated into facility design.

It is also likely that the Regional Planning Commission will have a role in fulfilling the requirements of the public transportation management system. The Commission, in its preparation of public transit development plans within the Region, is involved in the collection and analysis of data concerning, among other things, the quantity of transit capital stock and transit ridership. The Commission will not have as great a role in the intermodal management system, since the State Department of Transportation has historically dealt with intercity passenger and freight movements and seaport and other terminal and intermodal facilities, though the Commission does conduct airport system planning within the Region.

SPECIAL STUDIES

During the conduct of the third-generation regional transportation system planning program, and in particular through the work of the Commission's Advisory Committees and through the extended public participation process, a number of proposals were advanced that require, for their proper consideration, additional work efforts. From among many proposals, the following five were found by the Advisory Committee to have significant potential to enhance and improve the regional transportation system plan and its implementation potential over time:

1. USH 14 Study

USH 14, from Chicago to Janesville, traverses the extreme southwest corner of Walworth County. The new regional transportation system plan recommends that four travel lanes be provided on this facility within the Region. Because this facility is interregional in character, it is recommended that the Wisconsin Department of Transportation specifically reexamine facility improvement needs on this arterial highway route and confirm the need for four through travel lanes before proceeding to program an improvement project.

2. Oversize Truck Routing Study

One of the issues raised through the public hearing process involved the need for a designated route for oversized truck movements through the Southeastern Wisconsin Region. Because of the interregional and interstate nature of this matter, it is recommended that the Wisconsin Department of Transportation conduct a special study that would identify a network of arterial highway facilities that could be designated for oversized truck movements and that would connect important intermodal terminal facilities, such as the Port of Milwaukee, with important industrial areas.

3. Kenosha Area Commuter Study

It is recommended that a special study of the daily commuting patterns between the Kenosha area and Northeastern Illinois be undertaken. This study, which would be coordinated with the Northeastern Illinois Planning Commission and the Chicago Area Transportation Study, would have as its objective a detailed examination of daily travel habits and patterns across the Wisconsin-Illinois state line, which are perceived as becoming increasingly important over time. The study may lead to an amendment to the regional transportation system plan that would, for example, propose special transit services to accommodate daily tripmaking from homes in the Kenosha area to jobs in northeastern Illinois.

4. State Transit Capital Program

As noted earlier in this chapter, there is a substantial funding deficit in terms of implementing the transit element of the regional transportation system plan. One way to address in part that deficit is to institute a State transit capital matching fund program. Under such a program, the State would participate in the funding of transit capital projects, with participation ranging from 50 to 100 percent of the nonFederal share of such projects, depending upon whether the projects were local, express, or rapid transit in nature. Capital transit projects that are rapid in nature, such as major transit stations, busways, and commuter fail facilities, serve an areawide function much like the State trunk highway system, help to relieve traffic on State trunk highways, and are deserving of 100 percent State funding of the nonFederal share of such projects.

This particular recommendation was first made by the SEWRPC Regional Transportation Authority Study Committee in November 1990 and documented in SEWRPC Memorandum Report No. 38, A **Regional Transportation Authority Feasi**bility Study for Southeastern Wisconsin. It is recommended that, as the Wisconsin Department of Transportation completes its TRANSLINKS 21 Intermodal Transportation Plan, this particular funding recommendation be considered and that an appropriate State capital grant program be established. It is further suggested that the Wisconsin Department of Transportation examine the feasibility and desirability of funding this new program, from among a number of options, through the transfer of the present sales tax revenue on motor vehicles from the State's general fund to the State's transportation fund and/or through the establishment of a dedicated one cent motor fuel tax for this purpose.

5. <u>Reduction of Reliance on</u>

Property Tax to Fund Arterial Streets and Highways and Transit It is recommended that the counties comprising the Region take the lead in seeking legislation that would be directed at enabling county and local governments to reduce or eliminate reliance on property taxes to fund the costs of building, maintaining, and operating arterial streets and highways and public transit systems. More specifically, it is recommended that a State-level study be undertaken and consider, perhaps among a number of alternatives, the feasibility and desirability of providing enabling legislation that would permit a county to impose a county motor fuel tax or a county sales tax, or a combination of such taxes. The revenue received from such a tax should be used to reduce or eliminate the use of property taxes for arterial streets and highway and transit purposes. The legislation should also provide a means by which such a county tax would be fairly and equitably shared between the county imposing the tax and the municipalities within the county which are also levving property taxes for arterial street and highway and transit purposes.

6. <u>Linking of Enhanced General</u> <u>Transportation Aids to Implementation</u> <u>of the Regional Land</u> Use Plan

It is recommended that the Wisconsin Department of Transportation undertake a study that would seek to bring about a greater degree of implementation of the regional land use plan by linking the level of general transportation aids provided to implementation of that plan by individual counties and local units of government. At present, general transportation aids are provided to counties and local units of government and provide reimbursement for about 25 percent of the transportation costs incurred. It is specifically proposed that the study examine the desirability of raising that reimbursement level to 50 percent for those counties and local units of government that formally adopt the regional land use plan and enact appropriate local land use controls found to implement that plan substantially by providing for an appropriate distribution and density of new urban land use development, the protection and preservation of primary environmental corridors, and the preservation of prime agricultural lands.

A number of other proposals were advanced during the planning process but were not specifically endorsed by the Advisory Committee. These included such proposals as the enactment of State legislation to create a parking "cash out" program, legislation to enable local governments to place an excise tax on parking spaces, a broad policy study to comprehensively examine other transportation pricing issues, legislation to require that public transit facilities and services be required concurrently with the development of major land use projects, legislation that would limit the tax incremental financing technique to geographic areas served by public transit, legislation that would explore how State tax and other policies might be changed so as to encourage infill and redevelopment of urban areas. State policies that would withhold economic development grants and loans for projects at variance with regional plans, the creation of light-rail development corporations, and mandatory State policies to locate new government buildings within geographic areas publicly served by public transit. The Advisory Committee concluded that while some of these proposals may have merit, the recommended regional transportation plan was not the proper vehicle for their promotion. Moreover, in some cases the proposals would be contrary to the recommended plan, which does not include pricing mechanisms specifically intended to modify travel behavior.

SUMMARY AND CONCLUSIONS

This chapter has described the various means available, and has recommended specific procedures, for implementation of the recommended regional transportation system plan. The most important recommended plan implementation actions are summarized in the following paragraphs by level of government and responsible agency or unit of government.

Local Level

<u>County Boards of Supervisors</u>: It is recommended that the county boards of the seven counties comprising the Region, upon recommendation of the appropriate highway, transit, and public works committees:

- 1. Adopt the recommended regional transportation system plan as that plan affects each respective county.
- 2. Work cooperatively with the Wisconsin Department of Transportation and the cities, villages, and towns in the county in effecting changes in jurisdictional responsibility for portions of the arterial street and highway system as recommended in the plan.
- 3. Act to expand, improve, and maintain the arterial street and highway facilities designated in the plan for county jurisdiction in accordance with the functional plan recommendations, including undertaking, as may be appropriate, detailed planning, preliminary engineering, and official mapping work efforts.
- 4. Cooperate with the Wisconsin Department of Transportation, the Regional Planning Commission, and adjoining counties as necessary to conduct the major investment studies attendant to freeways and fixedguideway transit facilities identified in the plan, and carry out, as appropriate, detailed countywide transit planning programs to refine and detail the transit element of the regional transportation plan.
- 5. Take the lead in working through the State legislative delegation from the Region to secure enabling legislation that would permit the counties to impose either a transportation user fee or a general sales tax, or some combination thereof, at a level sufficient to raise the revenue required to fund the anticipated plan implementation shortfall attendant to county and local arterial streets and highways and transit, as well as to provide property tax relief.
- 6. Continue to provide public transit services in accordance with the recommendations set forth in the transit element of the plan including entering into agreements with

adjoining counties to provide transit services that cross county lines (Milwaukee and Waukesha Counties).

- 7. Enter into agreements with adjoining counties to effect the provision of recommended transit services which cross county lines (Kenosha, Ozaukee, Racine, Washington, and Walworth Counties).
- 8. Conduct studies to explore the transfer of the public transit function from the city to the county level of government (Kenosha, Racine, and Waukesha Counties).

<u>Common Councils, Village Boards, and Town</u> <u>Boards</u>: It is recommended that the common councils, village boards, and town boards in the Region, upon recommendations, as appropriate, of their plan commissions, boards of public works, and transit commissions:

- 1. Adopt the recommended regional transportation system plan as that plan affects each respective civil division.
- 2. Work cooperatively with the Wisconsin Department of Transportation and their respective counties in effecting changes in jurisdictional responsibility for portions of the arterial street and highway system as recommended in the plan.
- 3. Act to expand, improve, and maintain any arterial street or highway facility designated in the plan for local jurisdiction in accordance with the functional recommendations, including undertaking, as may be appropriate, detailed planning, preliminary engineering, and official mapping work efforts.
- 4. As appropriate, and upon consideration by, and recommendation of, local plan commissions and traffic engineering staffs, restrict curb-lane parking during peak travel periods on those arterial streets designated in the recommended plan as candidates for such restriction.
- 5. As appropriate, and upon consideration by and recommendation of the local plan commissions, integrate into the local planning and development practices and ordinances transit- and pedestrian-friendly land use development concepts.

- 6. As appropriate, cooperate with the Wisconsin Department of Transportation, the Regional Planning Commission, and any concerned counties in conducting major investment studies attendant to freeways and fixed-guideway transit facilities identified in the plan.
- 7. Support the efforts of the seven counties to secure State enabling legislation whereby either a transportation user fee or a general sales tax, or some combination thereof, is imposed at the county level to raise the revenue required to fund the anticipated plan implementation shortfall for county and local arterial streets and highways and transit.
- 8. Participate in studies attendant to the potential transfer over time of the public transit function to the county level of government (Cities of Kenosha, Racine, and Waukesha).
- 9. Continue to provide demand-responsive public transit services and cooperate with counties in developing detailed plans to consider the expansion of such services (Cities of Hartford, Port Washington, West Bend, and Whitewater).
- 10. Cooperate with the Regional Planning Commission in conducting a detailed study of Kenosha area commuting patterns, focusing on daily work-purpose travel between the Kenosha area and job locations in Northeastern Illinois (City of Kenosha, Village of Pleasant Prairie, and Town of Somers).

Areawide Level

Regional Planning Commission: It is recommended that the Southeastern Wisconsin Regional Planning Commission:

- 1. Adopt the recommended regional transportation system plan, acting not only in its capacity as a regional planning agency but also as the Federally recognized Metropolitan Planning Organization for the Kenosha, Milwaukee, and Racine urbanized areas.
- 2. Conduct a continuing regional transportation planning program to review, revise

and amend, and update and extend the adopted regional transportation system plan from time to time.

- 3. Work cooperatively with the county and local governments concerned, the Wisconsin Department of Transportation and the U. S. Department of Transportation, Federal Highway and Transit Administrations, in conducting the major investment studies attendant to freeways and fixedguideway transit facilities identified in the plan.
- 4. Cooperate with the Wisconsin Department of Natural Resources in assuring the continued conformity of the regional transportation system plan to the State Implementation Plan for air quality.
- 5. Provide assistance upon request to county and local governments in conducting highway and transit plan implementation efforts, including the preparation of detailed county and local highway and transit development plans and such implementation devices as official mapping.
- 6. Prepare a local planning guide designed to illustrate transit- and pedestrian-friendly land use development practices and provide assistance upon request to county and local governments in incorporating those practices into county and local plans and ordinances.
- 7. Provide assistance, upon request, to the Wisconsin Department of Transportation and county and local governments in carrying out cooperative efforts to effect changes in jurisdictional responsibility for portions of the arterial street and highway system as recommended in the plan.
- 8. Work cooperatively with the Wisconsin Department of Transportation as that Department discharges its responsibilities attendant to the development and operation, over time, of the transportation management systems called for in the Federal Intermodal Surface Transportation Efficiency Act, with particular focus on joint work efforts attendant to the congestion, public transportation, and intermodal management systems.

9. Undertake a special study of daily commuting patterns between the Kenosha area and Northeastern Illinois, working cooperatively with the City of Kenosha, the Village of Pleasant Prairie, and the Town of Somers.

State Level

Wisconsin Department of Transportation: It is recommended that the Wisconsin Department of Transportation:

- 1. Adopt the recommended regional transportation system plan and integrate that plan into the Wisconsin TRANSLINKS 21 intermodal transportation plan as a functional and jurisdictional guide to transit and highway system development within the Region.
- 2. Work cooperatively with the counties, cities, villages, and towns in the Region in effecting changes in jurisdictional responsibility for portions of the arterial street and highway system as recommended in the plan.
- 3. Act to expand, improve, and maintain any arterial street or highway facility designated in the plan for State jurisdiction in accordance with the functional recommendations, including undertaking necessary preliminary engineering and official mapping efforts.
- 4. Cooperate with the Regional Planning Commission and the concerned counties and local governments in the Region in conducting the major investment studies related to freeways and fixed-guideway transit facilities identified in the plan.
- 5. Provide financial support for the preparation by the Regional Planning Commission of a local planning guide designed to illustrate transit- and pedestrian-friendly land use development practices.
- 6. Develop and carry out in a cooperative manner with the Regional Planning Commission the congestion, public transportation, and intermodal management systems called for in the Federal Intermodal Surface Transportation Efficiency Act of 1991.

- 7. Continue to develop and operate the Milwaukee area freeway traffic management system so as to achieve the highest possible level of service on the freeways and help encourage travel and transit and carpools and vanpools.
- 8. Undertake an expanded transportation demand management program, including ridesharing promotion, assistance to transportation management associations, promotion of employee-based transportation demand management strategies, promotion of travel by bicycle and walking, and construction of telecommuting centers and carpool lots.
- 9. Advise the Governor to designate Kenosha and Racine Counties as recipients of Federal transit funds for the Kenosha and Racine urban areas upon the transfer of the transit function to the county level in those areas.
- 10. Conduct a special study of the USH 14 travel corridor in the southwest corner of Walworth County.
- 11. Conduct a special study to identify an oversize truck route through the Region.
- 12. Work with the Governor and the Wisconsin Legislature to secure, over time, the supplemental State funding outlined in the Wisconsin TRANSLINKS 21 Intermodal Transportation Plan so as to help provide an adequate level of financial support for implementation of the regional transportation system plan.
- 13. Support the efforts of the seven counties to secure State enabling legislation whereby either a transportation user fee or a general sales tax, or some combination thereof, is imposed at the county level to raise the revenue required to fund the anticipated plan implementation shortfall for county and local arterial streets and highways and transit.
- 14. Consider the establishment of a State transit capital aid program to fund at least 50 percent of the nonFederal share of capital transit projects.

- 15. Undertake a study that would seek to help implement the regional land use plan by linking a greater amount of general transportation aids to a finding that a local unit of government has taken appropriate steps to implement that plan.
- 16. Endorse the recommended changes to the proposed National Highway System within the Region and seek Federal approval of such changes.

Wisconsin Department of Natural Resources: It is recommended that the Wisconsin Natural Resources Board endorse the regional transportation system plan and direct its staff to complete the State air quality implementation plan in a manner consistent with the transportation plan.

<u>Wisconsin Department of Administration</u>: It is recommended that the Wisconsin Department of Administration endorse the regional transportation system plan and use that plan as a basis for reviewing and commenting on Federally funded transportation projects.

<u>Wisconsin Department of Development</u>: It is recommended that the Wisconsin Department of Development endorse the regional transportation system plan and consider that plan as it makes economic development-related decisions.

<u>University of Wisconsin-Extension</u>: It is recommended that the University of Wisconsin-Extension acknowledge the regional transportation system plan and promote implementation of the plan in its ongoing educational programs.

Federal Level

U. S. Department of Transportation, Federal Highway Administration: It is recommended that the Federal Highway Administration endorse the regional transportation plan, find that the plan meets the requirements of the Intermodal Surface Transportation Efficiency Act of 1991 and is consistent with the State implementation plan for air quality, and use the plan in the administration of its various Federal grant programs.

<u>U.S. Department of Transportation, Federal</u> <u>Transit Administration</u>: It is recommended that the Federal Transit Administration endorse the regional transportation plan, find that the plan meets the requirements of the Intermodal Surface Transportation Efficiency Act of 1991 and is consistent with the State Implementation Plan for air quality, and use the plan in the administration of its various Federal grant programs.

<u>U.S. Environmental Protection Agency</u>: It is recommended that the U.S. Environmental Protection Agency endorse the regional transportation plan and use the plan as it carries out its day-to-day regulatory activities.

General Considerations

Several particularly significant aspects of regional transportation system plan implementation warrant restatement here in summary form. First, the recommended regional transportation plan presented in this report, like the companion recommended regional land use plan documented in a separate report, is intended to comprise a guide to certain important aspects of the sound physical development of the Region. As such, the plan is advisory to the local, State, and Federal units and agencies of government concerned as these public bodies consider transportation facility development matters in the Region. The regional transportation system plan is not to be considered as an inflexible mold to which all future transportation system development within the Region must precisely conform. Rather, the regional transportation system plan is to be regarded as a point of departure against which transportation system development proposals can be evaluated as they arise and in the light of which better development decisions can be made by all parties concerned. The regional plan is intended to be used as a framework for more detailed county and local planning. As such, the plan is subject to refinement, detailing, and amendment as plan implementation proceeds, over time, within the Region.

Second, the adoption or endorsement of the recommended regional transportation plan as a guide to the sound development of the Region by the local units of government and the various State and Federal agencies concerned is highly desirable. Indeed, in some cases, that adoption or endorsement is essential in order to ensure a common understanding of the areawide development objectives and to permit the necessary plan implementation work to be cooperatively programmed and jointly executed. Moreover, the Federal Intermodal Surface Transportation Efficiency Act of 1991 envisions the appropriate incorporation of metropolitan transportation plans and programs into Statewide transportation plans and programs.

Third, plan implementation action policies and programs should not only be preceded by plan adoption or endorsement, but should also emphasize the most important and essential elements of the plan and those areas of action which will have the greatest impact on guiding and shaping transportation system development in accordance with the recommended plan. Regional transportation system plan implementation should focus on those facilities and activities having areawide significance. Implementation of the regional transportation plan will be largely achieved if the transportation management measures recommended in the plan are implemented, particularly the proper implementation of the Milwaukee area freeway system traffic management system and the promotion of demand management activities by the Wisconsin Department of Transportation, if the freeway system serving the greater Milwaukee area is appropriately rehabilitated and modernized, if improvements to the major surface arterials, particularly those identified on the recommended National Highway System, are implemented, and if the rapid and express transit expansion and improvement recommendations are carried out.

Fourth, the importance of close coordination and cooperation between the local units of government and between those units of government and the State and Federal agencies concerned to plan implementation cannot be overemphasized. **Responsibilities for achieving such coordination** and cooperation on a voluntary basis within the traditional framework of government in Wisconsin have been assigned to the Commission by the State Legislature through the regional planning enabling act. In addition, the Federal Intermodal Surface Transportation Efficiency Act further provides a basis for coordinating planning and plan implementation efforts by the Commission as the designated Metropolitan Planning Organization. In its capacity as the coordinating agency under both State and Federal law, advisory review of proposed transportation facilities by the Commission is essential for the effective development over time of the regional transportation plan. That system must be put in place to properly serve and promote the desired regional land use pattern. The proper vehicle for the review of proposed transportation facilities is the regional transportation improvement program compiled annually by the Commission in accordance with the requirements of the Federal transportation legislation.

Fifth, implementation of the regional transportation plan will not be brought about by a single massive action on the part of one unit or agency of government. Rather, implementation of that plan will be brought about through many individual development decisions made on a day-to-day basis over a period of many years by public administrators and elected officials operating at the local, areawide, State, and Federal levels of government. It is extremely important that the individuals and agencies making these decisions be aware of and understand the development proposals set forth in the recommended regional transportation plan so that those proposals receive proper consideration as development decisions are made.

Finally, regional transportation plan implementation can only be achieved within the context of a continuing, comprehensive, areawide planning effort wherein the planning inventories and forecasts on which the regional plans are based are updated, monitored, and revised; the plans are reappraised and, as necessary, revised to accommodate changing conditions; and through which the plans are interpreted on a day-to-day basis to the local, State, and Federal units and agencies of government concerned as the need to make development decisions arises. In this respect, planning does not and cannot concern itself with future decisions; that is, with "things that should be done in the future." Rather, it must be recognized that decisions exist only in the present and that planning is necessary just because decisions can be made only in the present, yet cannot be made for the present alone. The question, therefore, that faces elected officials and concerned citizens throughout the Region concerning implementation of the recommended regional transportation system plan is not what should be done tomorrow to bring about that plan, but, rather, what must be done today in light of the plan to get ready for an uncertain tomorrow.

(This page intentionally left blank)

Chapter XIII

SUMMARY AND CONCLUSIONS

INTRODUCTION

Because of its important impacts on daily life and on regional development, transportation is one of the principal areas of public policy determination facing public officials and concerned citizens within the Region. Although large amounts of public capital are expended for improving transportation facilities and services within the Region, there are never enough funds for all projects proposed. Precisely how this capital should be invested, how much should be allocated to transit facilities and how much to highway facilities and what the spatial location and capacities of these facilities should be, involves public policy determinations that are as difficult as they are important.

Recognizing the foregoing, and in light of Federal legislation set forth in the Clean Air Act Amendments of 1990 and the Intermodal Surface Transportation Efficiency Act of 1991, the Southeastern Wisconsin Regional Planning Commission undertook a multi-year planning effort to prepare a third-generation regional transportation system plan. That plan builds upon two earlier regional transportation system plans, one adopted in December 1966 and documented in SEWRPC Planning Report No. 7, Land Use-Transportation Study, and a second adopted in June 1978 and documented in SEWRPC Planning Report No. 25, A Regional Land Use Plan and a Regional Transportation Plan for Southeastern Wisconsin: 2000. As conceived by the Commission, the regional planning process requires a periodic review of the major elements of the comprehensive regional plan, including transportation. Such major reviews should occur at about 10-year intervals. The review process of major regional plan elements is intended to take into account changes which may have occurred since the preparation of the prior plan with respect to, for example, population, household, and employment levels and land use patterns, and in light of any changes in regional development objectives. Given the cyclical nature of the planning process, a new regional transportation system plan would also take into account decisions concerning transportation facility development made by State, county, and local implementing agencies since the preparation of the prior plan.

Significant trends affecting the movement of people and goods have taken hold in the Region since completion of the second-generation regional transportation plan. The Southeastern Wisconsin Region has witnessed many changes in recent years, including substantial increases in employment and households with little change in resident population; increasing participation in the labor force, particularly by females; declining costs of automobile operation; increasing per capita and per household vehicle ownership; continuing local land development policies and practices that foster urban sprawl and automobile dependency; and reductions in transit service. The effects of these and related socioeconomic trends have included increasing trip lengths, declining ridesharing and transit use, increasing highway traffic, and more severe traffic congestion. The regional transportation plan reappraisal and updating process needs to reflect appropriately and respond to these trends.

The new regional transportation system plan was designed to serve and help implement a new companion regional land use plan. That plan is set forth in SEWRPC Planning Report No. 40, <u>A</u> <u>Regional Land Use Plan for Southeastern Wisconsin—2010</u>, and was adopted by the Commission in September 1992 and by each county within the Region shortly thereafter. Like the new regional transportation plan, the companion regional land use plan was a thirdgeneration plan, prepared in light of the changes that have occurred since preparation of a secondgeneration regional land use plan and its adoption in December 1977.

This report describes the findings and recommendations coming out of the regional transportation plan reappraisal process. That process was guided by a Technical Coordinating and Advisory Committee on Regional Transportation System Planning. The report presents, in summary form, the results of reinventories of the factors influencing sound long-range, areawide transportation system planning and development in Southeastern Wisconsin, including 643 information on the implementation status of the second-generation transportation plan, on existing transportation facilities and services, and on travel habits and patterns. This report also presents a set of revised regional transportation system development objectives, principles, and standards formulated as a part of the plan reevaluation effort; summarizes the forecasts of future growth and change in the Region in terms of resident population, household, and economic activity levels as reflected in the companion regional land use plan; presents the results of the evaluation of alternative regional transportation system plans; and presents a recommended regional transportation system plan for a new plan design year 2010, together with specific plan implementation recommendations.

INVENTORY FINDINGS

In the process of preparing the third-generation regional transportation system plan, three major inventory efforts were undertaken: 1) an inventory of the implementation status of the secondgeneration regional transportation system plan, 2) inventories of the existing arterial street and highway and transit facilities and services, and the degree of utilization of those facilities and services, and 3) a major reinventory of regional travel to ascertain current travel habits and patterns and provide the basis for determining any changes in those habits and patterns that have occurred since the last major inventory of travel in 1972.² The results of these three inventories are set forth in Chapters III, IV, and V, respectively.

¹The new regional transportation system plan presented in this report includes elements related to transportation demand management, arterial street and highway facility development and management, and transit system development and management. The new regional plan also includes plan elements attendant to the bicycle and pedestrian modes of transportation. For these latter modal elements, the reader is referred to SEWRPC Planning Report No. 43, <u>A Regional Bicycle and Pedestrian Facilities</u> System Plan for Southeastern Wisconsin: 2010.

²In a companion work effort, a new inventory of land use in the Region was undertaken in 1990.

Year 2000 Regional Plan Implementation

The following briefly summarizes the findings and conclusions from the inventory efforts related to the status of implementation of the second-generation regional transportation system plan, which had a plan design year of 2000:

- There has been modest implementation of the year 2000 regional transportation system plan. At the time of the preparation of the third-generation regional transportation plan, about halfway through the design period of the year 2000 plan, plan implementation should have approximated 50 percent. The adopted plan envisioned an expanded and improved public transit system within the Region that would serve a variety of trips and provide an attractive alternative to the automobile. In particular, the plan envisioned a doubling of transit service, as measured by transit vehiclemiles provided and by the average speed of a transit trip. The number of transit vehiclemiles provided on an average weekday in the Region increased from 1972 through 1980 by about 22 percent, but by 1991 had returned to slightly less than the level recorded in 1972. The average speed of a transit trip has remained about the same as in 1972.
- Public transit plan recommendations included the provision of rapid transit service in the northwest corridor of Milwaukee County, the expansion of freeway flyer service throughout the Milwaukee metropolitan area to 24 routes, and, depending upon the results of detailed corridor studies, the provision of light-rail transit in five Milwaukee-area travel corridors. The plan also identified three travel corridors for the provision of demonstration commuter-rail passenger service within the Region and recommended the expansion of express and local bus service within the Kenosha, Milwaukee, and Racine urbanized areas. On the basis of the findings of a corridor facilities planning study, Milwaukee County decided to provide express bus rapid transit service in the northwest corridor instead of light-rail service. Freeway flyer service has been expanded to 15 routes serving 25 transit stations. Provision of light-rail transit in the east-west and northeast corridors is currently being considered.

No action has been taken to implement the recommended demonstration commuter-rail service. Local and express transit routes have since 1972 been provided and expanded within the urbanized areas of the Region. Local shared-ride taxicab service has been initiated in the Cities of Whitewater, Hartford, and West Bend.

- The transportation system management element of the adopted plan was aimed at managing traffic demand and enlarging existing arterial capacity with minimal cost and disruption of traffic. To these ends the plan recommendations included the following: the expansion of freeway traffic management, the provision of carpool parking lots in outlying metropolitan areas and a program to market carpooling to Milwaukeearea commuters, studies to determine the effect of parking policies on commuter mode choice, the prohibition of curb-lane parking on arterial streets, and the use of traffic engineering and access management. As of 1991, significant progress had been made in the provision of carpool parking lots, with 12 lots constructed between 1978 and 1991. The program to promote carpooling, however, has not operated since 1981. The Wisconsin Department of Transportation has begun the installation of new on-ramp metering equipment and corresponding traffic detectors as part of freeway traffic management efforts.
- The plan recommended three types of improvements to the arterial street and highway system: system expansion and system improvement, both aimed at increasing arterial capacity, and system preservation. By 1991, work on 192 miles, or 30 percent, of the mileage planned for system expansion and improvement had been completed. Work on 706 miles, or 24 percent, of the mileage planned for system preservation had been completed.
- Several factors have contributed to the lack of more complete plan implementation. Inadequate funding has been a substantial impediment. Federal funding for highway improvement has declined and no significant replacement State program funding has been forthcoming. The necessary reliance on the property tax to fund county and local arterial improvement and preservation

work has resulted in lower levels of implementation than planned. Federal funding for transit improvements has also declined, but this decline has been in part offset by increased State funding. It is apparent that without such State funding the level of transit service would be even lower than it is. The lack of public transit plan implementation may also be attributed to escalating transit unit costs, declining costs of motor fuel and the attendant decline in public interest in transit, increasing transit fares, and the lack of more complete land use plan implementation.

• Under existing transportation funding methods, to implement only the arterial street and highway and transit capital improvements contained in the current plan, the Region would face an annual shortfall of about \$57.3 million. A new transportation system plan must recognize this financially constrained context. This means that plan recommendations must be scaled to fit reasonable expectations of future available revenue and/or that methods to finance transportation improvements, as well as the operating and maintenance costs of the existing system, must be proposed in the plan.

Regional Transportation

System Facilities and Use

The following briefly summarizes the findings of the reinventories of the arterial street and highway and transit systems in the Region and of the use of those systems:

• In 1991, the arterial street and highway system consisted of 3,274 miles, representing 29 percent of the total street and highway mileage within the Region. Use of the arterial system has increased over time. In 1963, approximately 13 million vehicle-miles were traveled on the arterial system. Vehicle-miles increased to 20 million in 1972 and to about 33 million by 1991. The significant increase in arterial utilization increased traffic congestion. In both 1963 and 1972, about 10 percent of the arterial mileage was operating at or over design capacity. In 1991, about 17 percent of arterial mileage, or 543 miles, was operating at or over design capacity.

- Between 1963 and 1991, the number of public transit round-trip route-miles provided increased by about 171 percent, from 847 in 1963 to 2,296 in 1991. As a result, about 47 additional square miles were brought into the boundaries of local transit service areas between 1963 and 1991. This expansion did not, however, increase the number of people served by local transit. In fact, regionwide, in 1991, 56,700 fewer persons were served by local transit than were served in 1963. In providing service to smaller, less concentrated populations over greater route-mileage, transit operators within the Region provided fewer vehiclemiles of service. Average daily transit vehicle-miles provided within the Region decreased by about 25 percent between 1963 and 1991.
- Public transit ridership within the Region declined by nearly 47 percent between 1963 and 1991 and by 7 percent between 1972 and 1991. Between 1972 and 1991, ridership within the Kenosha and Racine urbanized areas increased more than twofold and threefold, respectively, but did not attain levels set in 1963. Unlike in the smaller urbanized areas, the ridership level in the Milwaukee area in 1991 was significantly, some 5.6 million, below the level set in 1972. Public transit travel within the Region on an average weekday has decreased as a proportion of total tripmaking from 9 percent of total regional travel in 1963 to 4 percent in 1972 and to 3 percent in 1991.

Travel Habits and Patterns

The following briefly summarizes the findings and conclusions of the major reinventory of travel in the Region conducted in 1991 and of an analysis of the changes in those habits and patterns compared to similar travel inventories conducted by the Commission in 1963 and 1972:

• On an average weekday in 1991, nearly 5.9 million person-trips were made within the Region. This represents increases of 26 percent between 1972 and 1991 and of 56 percent between 1963 and 1991. Growth in regional tripmaking reflects increases in the number of households within the Region of 26 percent from 1972 to 1991 and of 46 percent from 1963 to 1991 and of increases in employment of 31 percent from 1972 to 1991 and of 56 percent from 1963 to 1991.

- On an average weekday in 1991, about 4.9 million vehicle-trips were made within the Region, resulting in 33 million vehicle-miles of travel. About half the increase in vehiclemiles of travel between 1972 and 1991 may be attributed to the concurrent 26 percent increase in person-trips. The other half of the increase in vehicle-miles of travel may be attributed primarily to two factors: an 11 percent decline in automobile occupancy and a 15 percent increase in average trip length. Accordingly, about half the growth in highway traffic over the past 20 years may be attributed to factors that influence increased tripmaking: growth in the number of households and of employment.
- Increased tripmaking in the Region between 1963 and 1991 is in part linked to household and employment growth in the suburban and exurban portions of the Region. The increase in tripmaking within the suburban areas of the Region accounted for 38 percent of the increase in total regional tripmaking. Nearly 44 percent of regional household growth and 53 percent of regional employment growth occurred in suburban locations. The increase in tripmaking within the exurban area accounted for 13 percent of the regional growth in tripmaking. Nearly 22 percent of regional household growth and 14 percent of regional employment growth occurred in exurban locations. Should the growth in households and employment continue to occur in a highly dispersed pattern among the suburban and exurban subareas of the Region, tripmaking within and between these areas may be expected to increase further, as will the demand for investments in new transportation services and infrastructure in those areas.

OBJECTIVES, PRINCIPLES, AND STANDARDS

In preparing the third-generation regional transportation system plan, the Advisory Committee reviewed and modified, as found to be necessary, the set of transportation system development objectives, principles, and standards that had been formulated in the first-generation regional transportation system planning effort and modified once before in the second-generation regional transportation system planning effort. A total of nine transportation system development objec-

tives were recommended by the Advisory Committee to guide the design of the 2010 regional transportation system plan. These objectives and their supporting planning principles and sets of design standards are set forth in Chapter VIII and are concerned primarily with the provision of a flexible, multi-modal regional transportation system, alleviating traffic congestion, reducing travel time and accident exposure, and minimizing costs and disruptive effects upon communities and the natural environment. The objectives and standards were used both to identify problems and to evaluate alternative plans proposed to resolve those problems, providing the basis for the selection of a recommended plan that best achieved the specified objectives.

ANTICIPATED REGIONAL GROWTH AND CHANGE

The recommended system of transportation facilities and services included in the new regional transportation system plan was based upon, and indeed was designed to serve, the adopted regional land use plan for the year 2010. Thus, the transportation system plan was designed to accommodate the travel and transportation demands anticipated to be generated by the pattern of land use development identified in the regional land use plan. The land use plan contains the following three important recommendations:

• The creation of a relatively compact, centralized regional settlement pattern by encouraging the placement of new urban development generally in concentric rings along the full periphery of, and outward from, existing urban centers. To accomplish this centralized settlement pattern will require modification of existing development trends toward decentralization of urban land uses in a highly diffused, lowdensity pattern throughout the Region. Under the land use plan, new urban development would occur only in areas covered by soils suitable for urban development, which are not subject to such special hazards as flooding, and which can be readily and efficiently provided with essential facilities and services, including, particularly, public sanitary sewerage, water supply, and transit.

- The preservation in essentially open, natural uses of the remaining primary environmental corridors in the Region. Those corridors contain most of the remaining important natural resources of the Region, such as wetlands, woodlands, and wildlife habitat. The preservation of these corridors is essential to maintaining a high level of environmental quality in the Region, protecting its natural heritage and beauty, and providing scientific, educational, and recreational opportunities. The exclusion of urban development from these corridors will also help to prevent the creation of such serious and costly problems as air and water pollution, wet and flooded basements, building and pavement foundation failures, and excessive infiltration and inflow of clear water into sanitary sewerage systems.
- The preservation in agricultural and related uses of most of the remaining prime farmlands in the Region, keeping those lands relatively free from intrusion by urban land uses.

Changes in population and economic activity levels within the Region envisioned under the regional land use plan are as follows:

- The population of the Region would be expected to increase by about 6 percent, from 1.81 million persons in 1990 to 1.91 million persons by 2010.
- The number of households would be expected to increase by about 15 percent, from 676,000 in 1990 to 774,300 by 2010.
- The number of jobs would be expected to increase by about 11 percent, from 990,300 in 1990 to 1.10 million in 2010.

The planned distribution of households within the Region based upon the regional land use plan is especially critical to the determination of transportation system needs, since the place of residence is the generator of about 80 percent of all trips made within the Region.

DESIGN AND EVALUATION OF ALTERNATIVE PLANS

Given the regional land use plan and the underlying forecasts of population, households, and economic activity, as well as the set of

transportation system development objectives and supporting design standards, alternative regional transportation system plans were designed, tested, and evaluated. This involved the application of the full battery of Commission travel simulation models to estimate the magnitude and pattern of future travel and traffic. These models were first developed by the Commission in 1963 on the basis of an initial comprehensive regional travel survey, and include trip generation, trip distribution, modal split, and traffic assignment models. In 1972, these models were validated and, as found necessary, recalibrated using data from a second comprehensive regional travel survey conducted in that year. The models were again validated and recalibrated based upon data collected in the third comprehensive regional travel survey conducted in 1991. The models were demonstrated to accurately simulate travel conditions on the regional transit and arterial street and highway systems, and to have a demonstrated ability to simulate future travel and traffic conditions even given significant changes in lifestyles accompanied by changing land use development and travel habits and patterns within the Region.

The methodology used to design alternative regional transportation system plans was explicitly structured to ensure that before any proposal was brought forward to widen existing arterial streets and highways or to build new arterial streets and highways, full and adequate consideration was given to resolving existing and anticipated future transportation problems through demand management, traffic management, and public transit measures. This approach to plan design, which is the same basic approach followed in the two prior Commission regional transportation planning efforts, is fully consistent with the planning approach envisioned under the Federal Intermodal Surface Transportation Efficiency Act of 1991.

More specifically, the approach to designing alternative regional transportation system plans consisted of the following steps:

1. Definition of a "no-build" highway and transit transportation system and the assignment of forecast future travel demand to that system in order to identify potential problems and deficiencies under a scenario where no significant additional capital investment would be made in new and improved transportation facilities. The no-build alternative plan consisted of the arterial street and highway and public transit systems open to traffic and operating as of January 1, 1994, together with those arterial street and highway and transit facility and service improvements identified for implementation in calendar years 1994 and 1995 in the regional transportation improvement program.

- 2. Based upon the results of the analysis of the no-build option, the postulation of a series of alternative plans that included only actions that by definition were of a transportation system management nature. These actions included measures to manage and reduce vehicular travel demand, measures to manage traffic flow, and public transit service improvements.
- 3. The assignment of forecast future travel demand to a range of alternative transportation system management plans, with the range based upon the demand management measures considered in the second step, in order to ascertain the extent to which such actions would meet current and future transportation needs, and alone resolve traffic congestion problems.
- 4. Based upon the findings of the analyses conducted under the third step, the postulation of arterial street and highway improvements to resolve, to the extent considered feasible, identified remaining transportation system deficiencies.

In following this process, it was determined to fully evaluate the no-build alternative as well as two alternative regional transportation system plans that would provide improved public transit and arterial street and highway systems. The two alternative improvement plans had a number of common elements, including: 1) implementation of the long-planned Milwaukee-area freeway traffic management system, 2) placing curb-lane parking restrictions during peak periods on key arterial street facilities, 3) widespread application of traffic engineering techniques to improve traffic flow, 4) aggressive promotional efforts attendant to ridesharing, bicycle use, transit use, telecommuting, and the establishment of private transportation management associations, 5) detailed attention to land use planning at the community and neighborhood levels to provide appropriate mixtures of land uses, efficient and direct pedestrian and bicycle pathways, and clustering of higher-density land uses along transit lines and at transit stations and stops, and 6) the potential, depending upon the results of forthcoming major transportation investment studies, to provide light-rail and commuter-rail transit services in selected travel corridors found to have potential to support such services.

The two alternative improvement plans also had the following two significant differences:

- 1. Under one alternative, termed Alternative Plan 1, it was assumed that the perceived cost of automobile operation would be doubled from about 5.5 cents per mile to 11 cents per mile. While this cost increase could be achieved by a variety of measures, it was equivalent to an increase in the motor fuel tax of \$1.10 per gallon, representing a significant increase intended to reduce automobile travel. No such significant increase in the perceived cost of operating an automobile was assumed under the second alternative improvement plan evaluated, termed Alternative Plan 3.
- 2. Public transit service under Alternative Plan 1 would, in terms of revenue vehiclemiles of service, be increased by nearly 125 percent over the level of base year 1991. This was assumed in order to provide an effective alternative mode of transportation for those who would choose not to use the automobile under the assumed increased cost. By contrast, transit service improvements under Alternative Plan 3 represented a 53 percent increase in revenue vehicle-miles of service over the baseyear level.

In carrying out the plan design process described above, the transportation simulation models were applied to Alternative Plans 1 and 3 prior to incorporating any improvements to the arterial street and highway system. This step was undertaken to determine the extent to which the assumed transportation demand management and public transit improvement measures might resolve existing and anticipated future transportation system problems. Upon completion of that analysis, the residual transportation system problems were addressed by proposing the construction of new arterial streets and the widening of existing arterial streets and highways where necessary and feasible to resolve the identified residual traffic congestion problems. This resulted in the following:

- 1. The number of route-miles of proposed new arterial streets and highways under Alternative Plan 1, which sought to decrease automobile travel by doubling the perceived cost of automobile use and which provided a significantly greater increase in transit service than Alternative Plan 3, was only about 10 percent less than under Alternative Plan 3: 112 new route-miles under Alternative Plan 1, as opposed to 124 new route-miles under Alternative Plan 3.
- The number of route-miles of needed widening of arterial streets and highways under Alternative Plan 1 was only about 7 percent less than under Alternative Plan 3: 382 route-miles of widened facilities under Alternative Plan 1, as opposed to 410 route-miles of widened facilities under Alternative Plan 3.

One of the major findings of the evaluation of the alternative plans, then, was that an assumed doubling of the perceived cost of automobile use, when combined with a significantly better regional transit system, only modestly reduced the need for new and widened arterial streets and highways. While total travel in the Region on an average weekday in the design year 2010 under Alternative Plan 1, as measured by the number of person-trips generated, was about 1 percent less than under Alternative Plan 3, and as measured by vehicle-miles of travel was about 10 percent less than under Alternative Plan 3, and while transit ridership under Alternative Plan 1 was about 45 percent greater than under Alternative Plan 3, those changes in travel behavior did not result in particularly significant reductions in the need for arterial street and highway extensions and improvements. Consequently, while travel demand management and public transit measures represent legitimate public policy options, neither one nor both in combination can be viewed in isolation as the solution to existing and probable future transportation problems in the Region. New arterial street and highway construction and the widening of existing arterial streets and highways will also be required if the transportation needs of the Region are to be met. Moreover, the magnitude of transportation pricing changes needed to bring about a significant change in travel behavior so that the substantial addition of capacity in the arterial system could be avoided is such that, barring any major social or economic upheavals, such changes can only be brought about over a long period of time, if at all, and through a national. not local, approach to demand management. A full performance comparison of the alternative plans considered is set forth in Chapter X.

PRELIMINARY RECOMMENDED PLAN AND PUBLIC PARTICIPATION

Based upon the performance evaluations of the alternative plans considered, a modified version of Alternative Plan 3 was developed and taken to public hearing as a preliminary recommended new regional transportation system plan. That preliminary recommended plan was prepared and fully evaluated in terms of its performance relative to the transportation system development objectives and standards. The preliminary recommended plan is fully described in Chapter XI, and includes three basic elements: 1) transportation systems management, 2) public transit system maintenance and improvement, and 3) arterial street and highway system maintenance and improvement.

The following efforts were made to secure effective public participation in the review of, and comment on, the preliminary recommended plan:

- 1. The preliminary plan was described in an issue of the widely distributed Commission <u>Newsletter</u>. This issue, together with two previous issues of the <u>Newsletter</u>, one presenting the results of the inventory findings and the other presenting a description and evaluation of the alternative plans, provided in-depth descriptions of the plan and the alternatives thereto for consideration by all who seriously desired to learn about and react to the preliminary plan.
- 2. The preliminary recommended plan was the subject of a Regional Planning Confer-

ence well attended by elected officials, key local and State agency staff members, and concerned citizens.

- 3. The preliminary plan was presented to nine Commission advisory committees dealing with subregional transportation matters, including six county jurisdictional highway system planning committees and the three intergovernmental coordinating and advisory committees on transportation system planning and programming for the Kenosha, Racine, and Milwaukee urbanized areas. These committees reviewed the preliminary plan and made numerous recommendations for changes to the plan.
- 4. Seven formal public hearings were held on the preliminary plan, one in each of the seven counties of the Region. Each public hearing was preceded by a staffed "open house" at which individuals could examine and discuss the preliminary recommended plan with Commission staff and submit comments without attending the formal hearings.

Full documentation of the public participation process was provided in published Conference proceedings and the record of the public informational meetings and public hearings. The record of the public informational meetings and public hearings is referenced in Chapter XI, which also includes a summary of the public reaction to the preliminary plan.

Following the intensive period of public participation in the planning process, the Advisory Committee reviewed the reaction to the plan and guided the Commission staff in the preparation of a final recommended regional transportation system plan.

FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

The final recommended regional transportation system plan, briefly described below, like the preliminary plan, has three major elements: transportation systems management, public transit maintenance and improvement, and arterial street and highway maintenance and improvement. A more complete description of the plan is contained in Chapter XI.

Transportation Systems Management Element

The transportation systems management element of the plan consists of the following seven measures:

Freeway Traffic Management 1.

Full implementation of the Milwaukee-area freeway traffic management system, including an operational control strategy that would seek to provide, through restricted access of single-occupancy vehicles at ramp meters, for average operating speeds of about 35 miles per hour on all freeway segments during peak periods. Buses and high-occupancy vehicles would receive preferential access at the ramps.

- 2. Arterial Curb-Lane Parking Restrictions Restrictions of curb-lane parking during peak periods along 442 miles, or about 12 percent, of the planned 3,607-mile arterial street and highway system in order to reduce congestion and help provide good transit service. Local governmental units will need to carefully review the recommended curb-lane parking restrictions for feasibility and desirability in light of local conditions.
- 3. Traffic Engineering

The use of state-of-the-art traffic engineering practices to assist in achieving efficient traffic flow on arterial facilities and to facilitate pedestrian and bicycle movements as arterial streets and highways are constructed, reconstructed, and widened.

Traffic Management Technology 4.

The application of advanced traffic management technology, known as Intelligent Transportation Systems (ITS), as such technology becomes practicable and available over the plan implementation period. The Milwaukee-area freeway traffic management system is an example of the application of such technology. Other technologies with the potential for application in the Region include systems that provide real-time transit-vehicle location and arrival information to transit users, as well as systems that provide real-time route-specific traffic-condition information to drivers.

Travel Demand Management Promotion 5. A regionwide program to promote travel through ridesharing, transit use, bicycle use, and pedestrian movement, together with telecommuting and work-time rescheduling as may be found feasible. In addition, it is recommended that State and local governments promote and support private-sector transportation management associations to help achieve reductions in single-occupancy-vehicle travel.

6. Detailed Land Use

Planning and Site Design The preparation and implementation by local governmental units of detailed, sitespecific neighborhood land use plans to facilitate travel by transit, bicycle, and pedestrian movement.

7. Transit Systems Management

and Service Enhancement Measures The undertaking by the transit agencies in the Region of a range of activities to enhance the quality of transit services and to facilitate transit use. These activities include steps to improve transit information systems, including use of real-time vehicle locators to help maintain schedules; improvement of facilities for transfers and waiting; improvement of bicycle storage at stops and consideration of the use of onboard carriers as appropriate; working with local governments to provide transit-only street connections to facilitate through movement of transit vehicles: conduct of marketing/public education activities to promote the use of transit; improvement of transit security; improvement of vehicle design, which includes lowfloor vehicles, alternative fuels, improved noise and vibration control, and comfort enhancements; investigation of methods for increased bus speeds through priority systems and signal preemption; and promotion of innovative fare-payment systems, such as the UPASS, commuter check, smart card, employer treatment of transit as a fringe benefit, and other means of providing alternative ways to pay for transit services.

Public Transit Maintenance

and Improvement Element

The recommended plan calls for significant improvements to the public transit system in the Region. The improvements would include both expansion of the geographic extent of public transit service provided in the Region and improvement in the frequency of service on many of the transit routes in the system. Altogether, service on the regional transit system would be increased by about 75 percent from the 1991 level, measured in terms of revenue vehicle-miles of service provided.

More specifically, the public transit element of the plan consists of the following measures:

1. <u>Rapid Transit Service</u>

A significant expansion of the freeway flyer bus service in the Region to provide a truly areawide rapid transit system is proposed, including extending such service south to Racine and Kenosha, southwest to Mukwonago, west to Waukesha and Oconomowoc, northwest to West Bend, and north to Cedarburg, Grafton, Saukville, and Port Washington. A total of 30 such rapid transit routes are envisioned. 27 of which would be oriented to the Milwaukee central business district and three to the University of Wisconsin-Milwaukee campus. The rapid transit system would be served by 73 transit stations. Service would be provided in both directions during peak periods.

Initially, all service would be provided over the regional freeway system, with service extensions on selected surface arterial streets and highways. Ultimately, depending upon the results of major transportation investment studies recommended in Chapter XII, the rapid transit routes could operate over exclusive busway facilities in the most congested freeway travel corridors in the Region. One of those busway facilities, along the east-west freeway from the Marquette interchange in the City of Milwaukee to the STH 164 interchange in the City of Waukesha, has been included in the final recommended plan subject to confirmation upon completion of an eastwest corridor major investment study now under way, sponsored by the Wisconsin Department of Transportation.

The plan also recognizes the potential to establish commuter-rail passenger service as an alternative form of rapid transit

service to bus-on-freeway or bus-on-busway service in four major travel corridors, from Milwaukee through St. Francis, Cudahy, South Milwaukee, Oak Creek, Racine, and Kenosha to a connection with existing Chicago-oriented service; from Milwaukee through Wauwatosa, Elm Grove, Brookfield, Pewaukee, Hartland, Delafield, and Nashotah to Oconomowoc; from Milwaukee through Germantown and Jackson to West Bend; and from Milwaukee through Brown Deer, Cedarburg, and Grafton to Saukville. In addition, two potential extensions of Chicago-oriented commuter-railway passenger train service to termini in the Region are recommended to be studied: from Walworth in Walworth County through Fox Lake, Illinois, to Chicago, and from Burlington through Silver Lake and Antioch, Illinois, to Chicago. The plan recommends that the commuter-rail passenger service potential be evaluated through the conduct of major investment studies for the travel corridors concerned. Those studies, as appropriate, should begin with feasibility analyses and, depending upon the results of those analyses, proceed to complete the requirements of a major investment study. Through these corridor studies, then, final decisions would be made as to whether to provide the rapid transit service through bus-on-freeway, bus-on-busway, or commuter-rail passenger service. Pending the conduct of these studies, all rapid transit service would be provided by the bus-on-freeway mode.

2. Express Transit Service

The plan recommends that a total of 12 express transit bus routes be provided in a grid pattern largely within Milwaukee County in major travel corridors. The express routes would provide a highquality transit service, accommodating shorter trips than those made on the rapid transit system. Initially, all service would be provided by buses operating in mixed traffic over surface arterial streets and highways with limited stops. Ultimately, depending upon the results of major transportation investment studies recommended in Chapter XII, certain of the express transit service could be provided by buses operating over reserved lanes on arterial

streets, as well as in mixed traffic, or could be converted to the light-rail transit mode. With respect to two such express routes, from the Milwaukee central business district west to the Milwaukee Regional Medical Center and the Milwaukee County Research Park and from the Milwaukee central business district north to the University of Wisconsin-Milwaukee campus and Glendale, light-rail transit lines have been included in the final recommended plan subject to confirmation upon completion of the aforereferenced east-west corridor major investment study presently being conducted by the Wisconsin Department of Transportation. The other express transit service corridors identified in the plan as having potential for light-rail service include the corridors running from the central business district of Milwaukee to General Mitchell International Airport, from the central business district of Milwaukee to the Northridge Shopping Center, and along the 27th Street crosstown corridor from about W. Florist Avenue south and west to the Southridge Shopping Center. Pending the conduct of major investment studies for each of the potential light-rail transit lines, all express transit service would be provided by buses operating in mixed traffic over arterial streets.

3. Local Transit Service

The plan recommends the continued operation of local bus transit service over arterial and collector streets with frequent stops throughout the Kenosha, Milwaukee, and Racine urbanized areas. The plan calls for substantial improvements, however, in the frequency of local transit service provided, particularly on the major local routes. In addition, the plan holds open the potential to restructure local transit services to provide for transit-center-oriented local systems to replace grid route systems, depending upon detailed local plan implementation studies. The plan also recommends the continuation of local transit services through shared-ride taxis in the smaller urban areas of the Region. Finally, the plan recommends the continuation of appropriate paratransit services to help meet the transportation needs of disabled individuals in the Region.

Arterial Street and Highway

Maintenance and Improvement Element

The recommended plan calls for extensions and improvements to the arterial street and highway system in the Region. In 1991, there were 3,274 route-miles and 8,420 lane-miles of arterial streets and highways open to traffic in the Region. Under the plan, that system would, by the year 2010, total 3,607 route-miles and 10,303 lane-miles. Of the total increase of 1,883 arterial lane-miles, 692 lane-miles, or 37 percent, represent a reclassification of existing nonarterial facilities to arterial status as urban growth continues. The remaining 1,191 lane-miles, or 63 percent, represent proposals for new capacity in terms of widening of existing arterial facilities and construction of new facilities. The true increment in arterial capacity, measured in lanemiles of new construction, then, is about 1,191 lane-miles, or 14 percent, over 1991 conditions. The plan identifies the number of through travel lanes to be provided on each link in the arterial street and highway system. More detailed studies by the implementing agencies are required to determine the precise cross-section to be selected for a given improvement project, which would in turn define right-of-way requirements.

More specifically, the arterial street and highway element of the plan consists of the following:

1. <u>New Arterial Streets and Highways</u>

The plan recommends that 131 route-miles of new arterial streets and highways be constructed. These new facilities would provide an additional 337 arterial lanemiles. Included are the Lake Arterial Parkway south from the Daniel Webster Hoan Memorial Bridge to E. Layton Avenue; the STH 16 freeway bypass of Oconomowoc; the USH 12 freeway in Walworth County, including the Whitewater bypass; the STH 36 bypass of Burlington; and the completion of the western leg of the Waukesha circumferential arterial facility.

2. <u>Widening and Improving Existing</u> Arterial Streets and Highways

The plan recommends that widening and other improvements be undertaken along a total of 448 route-miles of existing arterial streets and highways. Such projects would provide an additional 854 arterial lanemiles. Among the proposed improvement projects are: the widening of STH 36 in Milwaukee, Waukesha, and Racine Counties; Pewaukee Road (CTH J) in Washington and Waukesha Counties; Cleveland Avenue (CTH D) and Racine Avenue (CTH Y) in Waukesha County; STH 31 and CTH Y in Kenosha and Racine Counties; Northwestern Avenue (CTH K) and Spring Street (CTH C) in Racine County; STH 57 and Port Washington Road (CTH W) in Ozaukee County; STH 33 in Ozaukee and Washington Counties; and Rawson Avenue (CTH BB) in Milwaukee County; and the completion of the widening of STH 50 in Kenosha and Walworth Counties.

3. <u>Maintaining Existing</u>

Arterial Streets and Highways

The plan recommends that all other arterial streets and highways in the proposed regional system be maintained over the plan implementation period through resurfacing and reconstruction to provide the same essential capacity. This particular proposal applies to 3,028 route-miles of existing arterial facilities. This particular plan recommendation incorporates a proposal to reconstruct and modernize to current freeway design standards the Milwaukee-area freeway system, coordinating any such freeway rehabilitation and modernization efforts with the results of major investment studies that are to examine the potential for providing busways and high-occupancy-vehicle lanes in the most congested corridors.

Performance and Costs of Recommended Plan

A detailed evaluation of the final recommended regional transportation system plan was undertaken and is reported in Chapter XI. Among the findings of that evaluation are the following:

1. Travel Characteristics

Under the recommended plan, automobile and light truck availability may be expected to increase by about 15 percent, from 1.13 million vehicles in 1991 to 1.30 million by 2010. Over that same period, the number of person-trips internal to the Region may be expected to increase by about 10 percent, from 5.54 million in 1991 to 6.10 in 2010. The number of vehicle-trips on an average weekday internal to the Region may be expected to increase by about 15 percent, from 4.58 million in 1991 to 5.27 million by 2010. Average weekday highway travel in the Region may be expected to increase by about 28 percent, from 33.07 million vehicle-miles in 1991 to 42.42 million by 2010. Transit ridership over the same period is expected to increase by about 14 percent, from 172,200 to 196,400 passengers per average weekday. The share of travel by transit may be expected to increase slightly, from 3.1 percent of all trips made on an average weekday in 1991 to 3.2 percent in 2010.

2. <u>Traffic Congestion</u>

In 1991, nearly 12 percent of the arterial street and highway system in the Region, or 385 route-miles, was found to be operating under congested conditions during peak periods, most of which was found to be operating under the "severely congested" and "extremely congested" conditions. If the transit and highway improvements recommended in the plan are carried out, the number of miles of congested facilities in 2010 may be expected to total less than 5 percent of the arterial street and highway system.

3. Environmental Assessment

A complete environmental assessment of the final recommended plan is set forth in Appendix D. Furthermore, a demonstration of the conformity of the final recommended plan to the State Air Quality Implementation Plan is set forth in SEWRPC report Assessment of Conformity of the New Year 2010 Regional Transportation System Plan and the 1995 through 1997 Transportation Improvement Program with Respect to the State of Wisconsin Air Quality Implementation Plan. In terms of air pollutant emissions from vehicles operating over the arterial street and highway system, significant reductions in the emission of volatile organic compounds, nitrogen oxides, and carbon monoxide are expected, these reductions being approximately 77 percent, 39 percent, and 72 percent, respectively. These significant reductions are expected to come about largely because of Federally mandated requirements for "cleaner" vehicles and fuels and only marginally because of improvements to the transit and highway systems.

The total estimated cost of implementing the recommended plan on an average annual basis
Table 283

FUNDING IMPLEMENTATION OF THE MAJOR ELEMENTS OF THE RECOMMENDED REGIONAL TRANSPORTATION PLAN FOR SOUTHEASTERN WISCONSIN: 2010^a

	Ave Co	rage Annual P osts and Rever (millions	lan Impleme nues: 1995-2 of dollars)	ntation 2010	Anticipated Addi Annual Revenue f State TRANSLINK	tional Average rom Federal and S 21 Programs ^b	Required Supplemental Average Annual Revenue from County/Local Sources		
			Shortfall						
Major Plan Element	Estimated Costs ^C	Existing Trend Revenues ^d	Amount	Equivalent Motor Fuel Tax (cents per gallon)	Amount (millions of dollars)	Equivalent Motor Fuel Tax (cents per gallon)	Amount (millions of dollars)	Equivalent Motor Fuel Tax (cents per gallon)	Equivalent Sales Tax (percent)
State Trunk Highway	229.2	144.0	85.2	10.0	85.2	10.0			
Trunk Highways	114.2 178.5	73.0 82.9	41.2 95.6	4.8 11.2	13.1 78.3	1.5 9.2	28.1 17.3	3.3 2.0	0.2 0.1
Total	521.9	299.9	222.0	26.0	176.6	20.7	45.4	5.3	0.3

^aAll figures are expressed in constant 1994 dollars.

^bIncludes a State transit capital funding program not proposed in TRANSLINKS 21 documents. The regional plan recommends that a study be conducted to examine the desirability and feasibility of such a program. Pending the results of that study, the plan tentatively includes revenue from such a program.

^C Includes capital and operation and maintenance costs; transit costs are net after subtracting farebox revenue.

^dBased upon currently available revenue sources and funding levels, modified as necessary to reflect a "fair share" of State revenues made available to the Region.

Source: SEWRPC.

over the 16-year plan implementation period 1995 through 2010 is \$522 million (see Table 283). All figures in that table and in this discussion are expressed in constant 1994 dollars. Of the total estimated cost, about \$229 million, or 44 percent, would be required for the State trunk highway system; about \$114 million, or 22 percent, for the county and local trunk highway systems; and the remaining \$179 million, or 34 percent, for the recommended transit systems. Existing revenues at Federal, State, county, and local levels approximate \$300 million annually, resulting in a shortfall of about \$222 million annually (see Table 283). This shortfall approximates 26 cents per gallon in terms of an equivalent motor fuel tax.

Given anticipated increases in Federal funding, in part through reservations made in the current Federal transportation legislation and in part through assumed additional reservations in forthcoming Federal transportation legislation, and further given the additional State revenue proposed to be sought under the State transportation plan, about \$177 million annually would be made available for Southeastern Wisconsin to help fund the shortfall. This represents the equivalent of an increase of nearly 21 cents per gallon in the motor fuel tax in the Region. Given these supplemental Federal and State funding assumptions, there would remain, then, a shortfall of about \$45 million annually. This represents the additional revenue that would need to be secured from within the Region if the plan is to be fully implemented within the 16-year plan implementation period. In terms of a motor fuel tax, this would be the equivalent of about five cents per gallon. In terms of a retail sales tax, this would be the equivalent of 0.3 percent. The plan recommends that the county and local governments in the Region seek enabling legislation to be able to impose the additional fees and taxes required to fund this shortfall, as well as to relieve the property tax burden currently associated with funding the local share of building and maintaining arterial streets and highways and providing transit services.

PLAN IMPLEMENTATION

Specific plan implementation recommendations directed at the concerned Federal, State, county, and local units and agencies of government operating within the Region are set forth in Chapter XII. The following summarizes the major implementation recommendations made:

1. Formal Plan Adoption

Formal adoption or endorsement of the recommended regional transportation system plan is highly desirable by all levels, units, and agencies of government concerned to assure a common understanding between the governmental levels and agencies as to the policies and directions to be pursued in building an integrated regional transportation system. It is important that the new regional plan be integrated into the detailed transportation planning and project implementation programs to be conducted by the many agencies concerned over the next two decades.

2. Arterial Street and

Highway Jurisdictional and Functional Improvements

The Wisconsin Department of Transportation, the seven counties, and many of the cities, villages, and towns in the Region need to work cooperatively to effect changes in jurisdictional responsibility for portions of the arterial street and highway system as recommended in the plan. Upon assumption of the proper jurisdiction, the implementing agencies concerned should program the necessary arterial street and highway improvement and maintenance activities.

3. <u>Transit Jurisdiction and</u> Functional Improvements

> Given the demonstrated lack of political support for a regional transportation authority in Southeastern Wisconsin, it is recommended that, over time, each of the counties in the Region assume responsibility for providing public transit services in accordance with the plan recommendations. This will facilitate not only the development of the functional rapid and express transit systems which provide services across county lines, but also the development of expanded local transit systems which cross municipal lines.

4. Major Investment Studies

A number of major investment studies of freeways and fixed-guideway transit facilities identified in the plan need to be undertaken. By Federal law, these studies are to be collaborative in nature, with initiation of a particular study possible by either the Wisconsin Department of Transportation, the Regional Planning Commission, or a transit operator. Such major investment studies are envisioned to lead to final decisions as to whether or not to move forward with a particular freeway expansion or major transit project. Upon completion of each major investment study, the Regional Planning Commission is to formally act to either confirm or revise the adopted regional transportation system plan with respect to such major transportation facilities.

5. <u>Transportation System</u>

and Demand Management

The Wisconsin Department of Transportation has been assigned particularly important responsibilities regarding the transportation system and demand management elements of the plan. Not only does the Department need to complete installation, and begin operation, of the Milwaukee-area freeway management system in accordance with the plan recommendations: the plan also calls for the Department to take the lead in creating an expanded areawide demand management program, including ridesharing promotion, assistance to transportation management associations, promotion of employee-based demand management strategies, and promotion of travel by bicycle and walking. In addition, the urban local governments in the Region need to prepare, adopt, and implement detailed land use plans that facilitate travel by bicycle and pedestrian movement and promote transit use through higher-density development along transit lines.

6. Funding of Plan Implementation

Collectively, the Wisconsin Department of Transportation and the county and local units of government in the Region need to work toward securing additional financial resources for transportation during the plan implementation period if any significant level of plan implementation is to be achieved. Not only will it be necessary for the Wisconsin Department of Transportation to work closely with the Governor and the Wisconsin Legislature in obtaining the additional Federal and State revenues required, as outlined in the State Intermodal Transportation Plan, but the counties in the Region will need to take the lead in seeking enabling legislation to permit them to raise additional revenues for transportation through dedicated nonproperty-taxbased fees or taxes, and to share those resources fairly with local units of government that also have plan implementation responsibilities. If there is not sufficient support for creating an areawide authority for the purposes of raising sufficient revenue to fund county and local transportation improvements, then the county and local governments will have to individually assume that responsibility, after working collectively for the needed permissive enabling legislation. Over the long run, the alternative to not securing additional revenue for transportation will be to significantly scale back the planned county and local arterial street and highway and transit improvements now included in the recommended regional transportation system plan.

CONCLUSION

This report has described the recommended regional transportation system plan for the year 2010. The plan has been carefully crafted to meet the transportation needs of the Region over approximately the next two decades. It is anticipated that person-trip generation within the Region will increase over this time by about 10 percent: that internal vehicle-trip generation will increase by about 15 percent; and that vehicle-miles of travel will increase by about 28 percent. During this period of time, transit ridership may be expected to increase by about 14 percent if the recommended extent and frequency of service is provided, although the share of travel by transit, about 3 percent, is not expected to change significantly. If implemented, the plan would meet these changing travel needs in the Region while reducing congestion on the arterial street and highway system from about 12 percent of the total of arterial mileage within the Region to about 5 percent.

To meet the future travel and transportation demands, the plan seeks to expand arterial street and highway system capacity, as measured in newly constructed arterial lane-miles, by about 14 percent, while expanding the transit system, as measured by vehicle-miles of revenue service, by about 75 percent. To accomplish these improvements to the transportation system will require not only additional financial resources, but the expenditure of a greater share of those resources for the transit mode. In 1991, about 22 percent of the total resources available for the combined arterial street and highway and transit systems in the Region were allocated to transit; under the plan, about 34 percent of the resources would be allocated to transit.

Thus, implementation of the recommended plan may be expected to provide the Region with an integrated transportation system that will effectively serve and promote a desirable regional land use pattern, meeting anticipated future travel demand at an adequate level of service through transportation system management measures, as well as transit and highway improvements. In terms of modes, the plan is as balanced as is practicable, with appropriate types of both highway and transit facilities provided for the various subareas of the Region. Implementation of the plan would abate traffic congestion, reduce travel time and costs, and reduce accident exposure. As such, implementation of, or the failure to implement, the recommended plan will affect not only the efficiency of the regional transportation system, and thereby directly affect the cost of living and doing business in the Region, but will also affect the overall quality of life in the Region for many years. It is critical, therefore, that government, business and industry, labor, and concerned citizens in the Region take an active interest in securing implementation of the plan recommendations.

Planning is, by definition, expected to deal with an uncertain future. As the governmental agencies concerned consider the recommended plan and its implementation, however, a number of these uncertainties need to be kept specifically in mind. First, there is uncertainty over the future levels of population and economic activity within the Region. A more vigorous economy could lead to greater-than-anticipated levels of growth and change in the Region. Alternatively, a less vigorous economy could return the Region to periods of modest changes in population and employment levels. The variables which relate to this uncertainty need to be closely monitored as plan implementation proceeds.

Second, there is an uncertainty with respect to the degree to which county and local governments in the Region will take appropriate and effective steps to implement the regional land use plan. The recommended regional transportation system plan is robust in the sense that analyses have shown that the recommended system will serve well not only the regional settlement pattern identified in the adopted regional land use plan, but a more decentralized pattern as well. Significant deviations from the recommended land use pattern, however, could prove to be more problematic. Accordingly, this is another variable which bears close scrutiny as plan implementation proceeds.

Third, there is great uncertainty over the feasibility of fully implementing the recommended regional transportation system plan owing to the substantial additional financial resources that will be required. Monitoring activities over the past two decades have demonstrated that it has become increasingly difficult for county and local governments to discharge fully their transportation responsibilities relying totally upon the property tax as the source of needed local revenue for that purpose. This has led to less-than-expected implementation of prior regional transportation system plans with respect to county and local arterial highways and public transit systems. At the State level, the monitoring has demonstrated substantially less fiscal uncertainty, since a single State agency exists with dedicated nonproperty-tax revenue sources available to foster more complete plan implementation. While the State has moved in recent years to provide greater revenues to county and local governments for both transit and highway plan implementation, and while even greater efforts are proposed in the new State Intermodal Transportation Plan, the level of uncertainty with respect to plan implementation remains far greater at the county and local level than at the State level. Inevitably, it would appear that Southeastern Wisconsin must, like most large metropolitan areas in the Nation, come to grips with this uncertainty by identifying and securing a dedicated nonproperty-tax revenue source for county and local transportation purposes. This uncertainty is perhaps the greatest of all in terms of plan implementation and will need to be carefully monitored as the implementation period proceeds.

Fourth, there is uncertainty with respect to plan implementation that derives from the Federal Clean Air Act Amendments of 1990 and the Federal rulemaking attendant thereto. This particular uncertainty can only be addressed by the Wisconsin Department of Natural Resources as it prepares and submits to the Federal government its State Implementation Plan for air quality. From a technical perspective, there is no reason why air quality considerations should delay implementation of the regional plan. The rulemaking by the Federal government attendant to the air quality legislation may operate, however, to effectively delay plan implementation if the State Air Quality Implementation Plan is not submitted in a complete and timely manner. This represents an additional uncertainty which must be addressed and monitored as plan implementation proceeds.

Finally, all these uncertainties need to be taken into account in a continuing regional land usetransportation planning program. As the Federally recognized Metropolitan Planning Organization, and as the regional planning agency for Southeastern Wisconsin, the Regional Planning Commission bears the responsibility for conducting that planning process. The process must include efforts to monitor all the foregoing uncertainties, and others as well. The process must continually survey and monitor many factors, must provide for amendment of the regional plan over time, and must provide for the extension of the plan to provide a continual 20-year planning horizon. Consequently, it should be anticipated that the Commission will work collaboratively with the Federal, State, county, and local units and agencies of government concerned and move forward with a work program designed to discharge its continuing planning responsibilities properly.

APPENDICES

(This page intentionally left blank)

Appendix A

URBAN SPRAWL REGIONAL LAND USE PLAN ALTERNATIVE: 2010

INTRODUCTION

At its meeting on September 14, 1993, the SEWRPC Technical Coordinating and Advisory Committee on Regional Transportation System Planning directed that the Commission staff prepare, as a companion to the adopted year 2010 Regional Land Use Plan, a regional land use alternative that would be differentiated from the adopted plan in two important respects:

- 1. The alternative plan would provide for a greater level of decentralization of regional population, household, job, and attendant land use than envisioned under the adopted regional land use plan.
- 2. The alternative plan would provide for a greater degree of "urban sprawl" in the Region, reflecting land use development trends made possible by the accommodation of urban development served by onsite sewage disposal systems, which are evident in portions of the Region.

The Committee reasoned that, in view of the continuing uncertainty surrounding future social and economic conditions in the Region, it would be imprudent to dismiss the possibility that future growth and change in the Region could occur in a manner at variance with that envisioned under the adopted 2010 land use plan. Accordingly, the Committee requested that the Commission staff prepare an existing trends urban sprawl land use alternative that was consistent with the adopted plan in terms of the overall population, household, and employment levels envisioned within the Region by 2010 but that would differ with respect to the manner in which such households and jobs would be accommodated in the Region. This alternative plan was intended to supplement the adopted 2010 land use plan and thereby broaden the framework within which planning and decision making regarding transportation system development and redevelopment within the Region could be carried out. It was intended within this framework, for example, that proposals for major public facilities, especially transportation facilities, be evaluated to determine how well they would perform under a land use pattern different from that envisioned in the adopted regional land use plan. Through such sensitivity analyses, more "robust" plan elements which could be expected to remain viable under greatly varying land use development conditions could be identified.

ASSUMPTIONS USED IN PREPARING AN EXISTING TRENDS URBAN SPRAWL LAND USE PLAN ALTERNATIVE

Unlike the adopted 2010 regional land use plan, which is a "normative" land use plan, setting forth a regional development pattern that seeks to modify historic development trends in order to promote public health, safety, and general welfare and a more efficient and economical development pattern, the existing trends urban sprawl alternative sets forth one possible regional development pattern that could evolve, assuming historic development trends in the Region were to continue. In this respect, it should be understood that since such land use development under this alternative would be almost entirely market driven, a large number of somewhat different patterns could evolve.

Thus, while the number of persons, households, and jobs under both the existing trends urban sprawl alternative and the adopted regional land use plan are essentially the same, the resultant development patterns are significantly different because of the following assumptions utilized in the preparation of the existing trends urban sprawl alternative.

1. Under the existing trends urban sprawl alternative, the number of persons, households, and jobs within each county and each subcounty planning analysis area was determined by examining the existing trends in each county and planning analysis area over the period from 1970 to 1990. Such trends were then projected to 2010, with the results adjusted as necessary to match regional control totals.

- 2. The existing trends urban sprawl alternative, insofar as land use distribution is concerned, assumed the proportion of new households created over the design period within the Region that would occur on scattered locations utilizing onsite sewage disposal system would be similar to the proportion experienced within the Region in the 1970 to 1985 period.
- 3. The existing trends urban sprawl alternative further assumed that the location of development utilizing onsite sewage disposal systems would be spatially distributed within each county and planning analysis area on the basis of the historic distribution pattern of onsite sewage disposal system development which occurred in that county between 1970 and 1985.
- 4. The existing trends urban sprawl alternative also assumed that the distribution of new residential development within each county and planning analysis area would not be constrained by withholding such development from the upland portions of the primary environmental corridor. Rather, the design process would assume only that the lowland portions of the environmental corridors, the floodlands and wetlands, would be protected and preserved.
- 5. Finally, the existing trends urban sprawl alternative did not constrain new development from occurring within the prime agricultural areas since a significant amount of the urban development historically occurred on such lands.

Utilization of the foregoing assumptions resulted in significantly greater decentralization of urban activity and significantly higher levels of urban development occurring in outlying areas of the Region than envisioned in the adopted regional land use plan. Such development relied heavily on the utilization of onsite sewage disposal systems, encroached into the upland portions of the primary environmental corridors, and converted significantly greater areas of prime agricultural lands. A summary of the key findings of the existing trends urban sprawl regional land use plan alternative, illustrated on Maps A-1 and A-2, compared to the adopted 2010 regional land use plan, are presented in the following section.

COMPARISON OF THE EXISTING TRENDS URBAN SPRAWL REGIONAL LAND USE PLAN ALTERNATIVE AND THE YEAR 2010 ADOPTED REGIONAL LAND USE PLAN

Population

The population of the Region under the existing trends urban sprawl alternative is envisioned to increase by 170,000 persons, or about 10 percent, from 1,742,700 in 1985 to 1,912,700 by 2010. While the increase in population levels under this alternative is identical to that envisioned under the 2010 adopted land use plan, the envisioned distribution of this incremental population is significantly different.

As indicated in Table A-1, when compared to the adopted land use plan, the existing trends urban sprawl alternative plan would, as a result of assumed continuation of past development trends, accommodate about 36,000 fewer persons in the highly urbanized Kenosha, Milwaukee, and Racine Counties and significantly more persons in the outlying counties of the Region, especially Washington and Waukesha Counties.

Households

Household levels under the existing trends urban sprawl alternative are envisioned to increase by about 128,800, or about 20 percent, from 643,800 in 1985 to 772,600 in 2010. As indicated in Table A-2, the existing trends urban sprawl alternative would, by the year 2010, accommodate about 2,300 fewer households than the adopted plan as a result of the assumed larger household size in the outlying counties, where higher population levels were envisioned to occur. Similar to the distribution of population, the existing trends urban sprawl alternative, when compared to the adopted land use plan, would accommodate about 19,000 fewer households in the urbanized Kenosha, Milwaukee, and Racine Counties and more households in the outlying counties of the Region, especially Washington and Waukesha Counties.

TOTAL POPULATION IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		Total Population											
		20	10 Adopted P	lan	201 Urbai	10 Existing Tre	ends native	2010 Existing					
		Planned I 1985	Planned Increment 1985-2010		Planned I 1985	d Increment 85-2010		Trends Urban Sprawl Alternative Compared					
County	1985	Number	Percent	Total	Number	Percent	Total	(number)					
Kenosha	121,100	26,800	22.1	147,900	18,900	15.6	140,000	-7,900					
Milwaukee	939,600	-10,300	-1.1	929,300	-33,600	-3.6	906,000	-23,300					
Ozaukee	67,500	12,300	18.2	79,800	17,500	25.9	85,000	5,200					
Racine	169,200	16,800	9.9	186,000	11,800	7.0	181,000	-5,000					
Walworth	72,200	15,100	20.9	87,300	14,800	20.5	87,000	-300					
Washington	87,200	24,500	28.1	111,700	32,800	37.6	120,000	8,300					
Waukesha	285,900	84,800	29.7	370,700	107,800	37.7	393,700	23,000					
Region	1,742,700	170,000	9.8	1,912,700	170,000	9.8	1,912,700	0					

Source: SEWRPC.

Table A-2

TOTAL HOUSEHOLDS IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		Total Households											
		20	10 Adopted P	lan	201 Urbar	0 Existing Tre	ends native	2010 Existing					
	Planned Increment 1985-2010			Planned I 1985	ncrement -2010		Trends Urban Sprawl Alternative Compared						
County	1985	Number	Percent	Total	Number	Percent	Total	(number)					
Kenosha	44,200	14,900	33.7	59,100	11,800	26.7	56,000	-3,100					
Milwaukee	368,200	30,800	8.4	399,000	16,800	4.6	385,000	-14,000					
Ozaukee	22,900	7,600	33.2	30,500	10,100	44.1	33,000	2,500					
Racine	61,200	12,700	20.8	73,900	10,800	17.6	72,000	-1,900					
Walworth	25,600	10,000	39.1	35,600	10,400	40.6	36,000	400					
Washington	28,500	13,100	46.0	41,600	16,500	57.9	45,000	3,400					
Waukesha	93,200	42,000	45.1	135,200	52,400	56.2	145,600	10,400					
Region	643,800	131,100	20.4	774,900	128,800	20.0	772,600	-2,300					

Source: SEWRPC.

Employment

Employment levels under the existing trends urban sprawl alternative are envisioned to increase by about 230,200 jobs, or about 26 percent, from 871,900 jobs in 1985 to 1,102,100 jobs in the year 2010. As indicated in Table A-3 and as a result of the assumed continuation of the decentralization of economic activity levels, the existing trends urban sprawl alternative, when compared to the adopted land use plan, would accommodate significantly more jobs in the outlying areas of the Region, especially in Washington and Waukesha Counties. In this regard, Waukesha County would accommodate almost 21,000 more jobs under the existing trends plan than it would under the adopted plan.

Land Use

As indicated in Table A-4, urban land uses under the existing trends urban sprawl alternative are envisioned to increase by about 144 square miles, or almost 24 percent, from about 606 square miles in 1985 to about 750 square miles by the year 2010. As further indicated in Table A-4, the existing trends urban sprawl alternative, when compared to the adopted land use plan, would convert about 58 more square miles of lands from rural to urban use. Almost all of this envisioned increase in urban

Map A-1

EXISTING TRENDS URBAN SPRAWL LAND USE PLAN FOR SOUTHEASTERN WISCONSIN: 2010



Map A-2

COMPARISON OF 2010 ADOPTED LAND USE PLAN AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE



TOTAL EMPLOYMENT IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		Total Employment											
		20	10 Adopted P	lan	201 Urbar	0 Existing Tre Sprawl Alter	ends native	2010 Existing					
		Planned I 1985	Planned Increment 1985-2010		Planned Increment 1985-2010			Trends Urban Sprawl Alternative Compared					
County	1985	Number	Percent	Total	Number	Percent	Total	(number)					
Kenosha	42,500	20,500	48.2	63,000	17,500	41.2	60,000	-3,000					
Milwaukee	527,300	85,400	16.2	612,700	52,700	10.0	580,000	-32,700					
Ozaukee	26,900	11,800	43.9	38,700	15,100	56.1	42,000	3,300					
Racine	74,500	17,600	23.6	92,100	18,500	24.8	93,000	900					
Walworth	28,100	12,400	44.1	40,500	15,900	56.6	44,000	3,500					
Washington	31,300	16,600	53.0	47,900	23,700	75.7	55,000	7,100					
Waukesha	141,300	65,900	46.6	207,200	86,800	61.4	228,100	20,900					
Region	871,900	230,200	26.4	1,102,100	230,200	26.4	1,102,100	0					

Source: SEWRPC.

Table A-4

TOTAL LAND USE IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

	2010 Add			dopted Plan		L	2010 Exis Irban Sprav	ting Trends vI Alternative		2010 Existing	
	Existing	Existing 1985		Planned Increment 1985-2010			Planned In 1985-2	crement 2010			Trends Urban Sprawl Alternative Compared to
Land Use Category	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total	Acres	Percent	Acres	Percent of Total	2010 Adopted Plan (acres)
Urban											
Residential											
Urban High-Density	27,797	1.6	1,817	6.5	29,614	1.7	572	2.1	28,369	1.6	-1,245
Urban Medium-Density	54,153	3.1	30,802	56.9	84,955	4.9	28,695	53.0	82,848	4.8	-2,107
Urban Low-Density	94,618	5.5	3,128	3.3	97,746	5.7	30,216	31.9	124,834	7.3	27,088
Suburban-Density	8,035	0.5	1,338	16.7	9,373	0.5	3,650	45.4	11,685	0.7	2,312
Subtotal	184,603	10.7	37,085	20.0	221,688	12.8	63,133	34.2	247,736	14.4	26,048
Commercial	9 714	0.5	1 4 4 4	16.6	10.159	0.6	1 757	20.2	10 471		212
	12 080	0.5	5 295	10.0	17.465	0.8	6,006	20.2	10,471	0.6	513
Transportation	12,000	0.7	5,365	44.0	17,405	1.0	5,630	40.0	17,976	1.0	511
Communication											
and Litilitios a	120.270	7.0	14 010	12.2	125.000	7.0	00.205	10.0	140.004	~ ~	5 500
	120,275	7.0	14,019	12.3	135,098	7.8	20,385	10.9	140,664	8.2	5,566
	17.240	10	1.044	61	10 204	1 1	707	4.6	19.027	1.0	247
Becreational b	25 564	1.0	4 1 9 5	16.4	20 740	1.1	2 0 2 4	4.0	18,037	1.0	-247
	10.215	1.5	4,100	10.4	29,749	1.7	3,934	10.4	29,498	1.7	-251
	13,210	1,1	-0,397	-43.7	10,818	0.6	-3,554	-18.5	15,661	0.9	4,843
Subtotal	387,695	22.5	55,565	14.3	443,260	25.6	92,348	23.8	480,043	27.8	36,783
Rural											
Residential ^C			721		721	d	1,368		1.368	0.1	647
Agricultural ^e	989,180	57.5	-56,014	-5.7	933,166	54.3	-92,821	-9.4	896,359	52.1	-36.807
Other Open Lands ^f	344,238	20.0	-272	-0.1	343,966	20.1	-895	-0.3	343,343	20.0	-623
Subtotal	1,333,418	77.5	-55,565	-4.2	1,277,853	74.4	-92,348	-6.9	1,241,070	72.2	-36,783
Total	1,721,113	100.0	0	0.0	1,721,113	100.0	0	0.0	1,721,113	100.0	0

^aIncludes off-street parking areas.

^b1985 data includes net site area of public and nonpublic recreation sites while 2010 data includes only that net site area recommended for public recreation use.

^CIncluded in 1985 land use inventory as part of urban residential land use.

d_{Less} than 0.1 percent.

^eIncludes agricultural and unused rural land.

^fIncludes woodlands, water, wetlands, landfill sites, and quarries.

Source: SEWRPC.

666

TOTAL URBAN LANDS IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

[n	Tot	al Urban Lanc	ls			
	(*****	2010 Adopted Plan			20 ² Urbai	10 Existing Tre	ends native	2010 Existing	
	1985 Total		Planned Increment 1985-2010		Planned 1985	Increment -2010	Total	Trends Urban Sprawl Alternative Compared	
County	(acres)	Acres	Percent (acre		Acres	Percent	(acres)	(acres)	
Kenosha	31,971	7,533	23.6	39,504	8,248	25.8	40,219	715	
Milwaukee	116,795	6,081	5.2	122,876	8,051	6.9	124,846	1,970	
Ozaukee	27,292	5,163	18.9	32,455	6,912	25.3	34,204	1,749	
Racine	40,341	6,179	15.3	46,520	6,804	16.9	47,145	625	
Walworth	38,080	4,675	12.3	42,755	7,711	20.2	45,791	3,036	
Washington	33,671	6,659	19.8	40,330	15,328	45.5	48,999	8,669	
Waukesha	99,545	19,275	19.4	118,820	39,294	39.5	138,839	20,019	
Region	387,695	55,565	14.3	443,260	9:2,348	23.8	480,043	36,783	

Source: SEWRPC.

Table A-6

TOTAL PRIME AGRICULTURAL LANDS IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		-		Total Prin	ne Agricultura	l Lands			
		2010 Adopted Plan			201 Urbar	0 Existing Tre Sprawl Alter	ends native	2010 Existing	
	1095 Total	Planned Increment 1985-2010		Total	Planned I 1985	ncrement -2010	Total	Trends Urban Sprawl Alternative Compared to 2010 Adopted Plan (acres)	
County	(acres)	Acres	Acres Percent		Acres	Percent	(acres)		
Kenosha	76,471	-2,042	-2.7	74,429	-4,636	-6.1	71,835	-2,594	
Milwaukee	1,351	0	0.0	1,351	-132	-9.8	1,219	-132	
Ozaukee	73,335	-1,425	-1.9	71,910	-3,049	-4.2	70,286	-1,624	
Racine	98,626	-526	-0.5	98,100	-1,452	-1.5	97,174	-926	
Walworth	208,941	-917	-0.4	208,024	-2,125	-1.0	206,816	-1,208	
Washington	108,256	-1,463	-1.4	106,793	-5,952	-5.5	102,304	-4,489	
Waukesha	103,078	-4,543	-4.4	98,535	-20,478	-19.9	82,600	-15,935	
Region	670,058	-10,916	-1.6	659,142	-37,824	-5.6	632,234	-26,908	

Source: SEWRPC.

lands would occur in the low-density residential category as a result of the assumed proliferation of scattered, low-density residential subdivisions in the outlying counties of the Region. As indicated in Table A-5, the increment in urban lands under the existing trends urban sprawl alternative was significantly higher in the outlying counties compared to the 2010 adopted plan. In this regard, the envisioned increase of about 61 square miles in urban lands in Waukesha County under the existing trends urban sprawl alternative is more than double that of about 30 square miles envisioned under the adopted plan.

Prime Agricultural Lands

Prime agricultural lands under the existing trends urban sprawl alternative are envisioned to decrease by about 59 square miles, or almost 6 percent, from 1,047 square miles in 1985 to about 988 square miles in 2010. As indicated in Table A-6, this envisioned decrease in prime agricultural lands is significantly higher than that envisioned under the adopted plan, only about 17 square miles, during the period from 1985 to 2010. It is interesting to note that the loss of prime agricultural lands under the existing trends urban sprawl alternative is greater in every county than under the adopted plan.

TOTAL PRIMARY ENVIRONMENTAL CORRIDOR AREA IN THE REGION BY COUNTY: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		Total Primary Environmental Corridor Area											
- -		2010 Adopted Plan			20 <i>°</i> Urba	10 Existing Tre n Sprawl Alter	ends native	2010 Existing					
1985 Tot		Planned Increment 1985-2010		Tetel	Planned Increment 1985-2010		Total	Trends Urban Sprawl Alternative Compared					
County	(acres)	Acres	res Percent (acres		Acres	Percent	(acres)	(acres)					
Kenosha	28,600	300	1.0	28,900	-160	-0.6	28,440	-460					
Milwaukee	9,800	500	5.1	10,300	460	4.7	10,260	-40					
Ozaukee	19,900	0	0.0	19,900	-550	-2.8	19,350	-550					
Racine	23,600	200	0.8	23,800	-230	-1.0	23,370	-430					
Walworth	65,200	300	0.5	65,500	-900	-1.4	64,300	-1,200					
Washington	60,300	600	1.0	60,900	-1,340	-2.2	58,960	-1,940					
Waukesha	92,200	1,700	1.8	93,900	-930	-1.0	91,270	-2,630					
Region	299,600	3,600	1.2	303,200	-3,650	-1.2	295,950	-7,250					

Source: SEWRPC.

Table A-8

POPULATION DENSITY IN THE REGION: SELECTED YEARS 1850-1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

	Urban Population		Rural Population			Area (square mile		Persons per Square Mile	
		Percent		Percent	Total			_	
Year	Number	of Total	Number	of Total	Population	Urban	Total	Urban	Total
1850	28,623	25.2	84,766	74.8	113,389	4	2,689	7,156	42
1880	139,509	50.3	137,610	49.7	277,119	18	2,689	7,751	103
1900	354,082	70.6	147,726	29.4	501,808	37	2,689	9,570	187
1920	635,376	81.1	148,305	18.9	783,681	56	2,689	11,346	291
1940	991,535	92.9	76,164	7.1	1,067,699	90	2,689	11,017	397
1950	1,179,084	95.0	64,534	5.0	1,243,618	146	2,689	8,076	462
1963	1,634,200	97.6	40,100	2.4	1,674,300	282	2,689	5,795	623
1970	1,728,946	98.5	27,137	1.5	1,756,083	338	2,689	5,115	653
1980	1,749,238	99.1	15,558	0.9	1,764,796	444	2,689	3,940	656
1985	1,730,500	99.3	12,200	0.7	1,742,700	477	2,689	3,628	648
2010 Adopted Land Use Plan	1,904,500	99.6	8,200	0.4	1,912,700	670	2,689	2,843	711
2010 Existing Trends Urban									
Sprawl Land Use Plan	1,904,500	99.6	8,200	0.4	1,912,700	743	2,689	2,563	711

Source: SEWRPC.

Primary Environmental Corridor

Primary environmental corridor lands, under the existing trends urban sprawl alternative, were envisioned to decrease by about six square miles, or by about 1.2 percent, from about 468 square miles in 1985 to about 462 square miles in the year 2010. As indicated in Table A-7, every county in the Region, with the exception of Milwaukee County, was envisioned to lose some environmental corridor lands during the 25-year period. This is in marked contrast to the adopted plan, which envisioned no change or an actual gain in primary environmental corridor lands during the plan design period. The gain in primary environmental corridor lands under the adopted plan was envisioned to occur primarily as a result of protection of floodplain areas and restoration of such areas to natural conditions, thereby increasing primary environmental corridor lands. The envisioned loss of primary environmental corridor lands under the existing trends urban sprawl alternative would occur primarily as a result of the conversion of the upland portions of such corridors to urban residential uses.

EXISTING AND PROPOSED DEVELOPED AREA AND POPULATION SERVED BY PUBLIC SANITARY SEWER AND WATER SUPPLY SERVICE IN THE REGION: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		r	· · ·	2010 Ad	opted Plan		2010 Existing Trends Urban Sprawl Alternative			
	1985 Existing Service		Planned Increment Existing Service 1985-2010		Total Service		Planned Increment 1985-2010		Total Service	
Area and Population	Public Sanitary Sewer	Public Water Supply	Public Sanitary Sewer	Public Water Supply	Public Sanitary Sewer	Public Water Supply	Public Sanitary Sewer	Public Water Supply	Public Sanitary Sewer	Public Water Supply
Developed Area										
Total Square Miles	476.5	476.5	193.9	193.9	670.4	670.4	266.9	266.9	743.4	743.4
Square Miles Served	319.8	262.9	253.1	312.4	572.9	575.3	255.0	314.3	574.8	577.2
Percent of Total Served	67.1	55.2	°		85.5	85.8			77.3	77.6
Population										
Total Population	1,742,700	1,742,700	170,000	170,000	1,912,700	1,912,700	170,000	170,000	1,912,700	1,912,700
Population Served	1,507,800	1,389,700	232,400	353,900	1,740,200	1,743,600	165,800	286,800	1,673,600	1,676,500
Percent of Total Served	86.5	79.7			91.0	91.2			87.5	87.7

Source: SEWRPC.

Population Density

As indicated in Table A-8, the population density, measured in persons per square mile, is envisioned to decrease under the existing trends urban sprawl alternative, from about 3,630 persons per square mile in 1985 to about 2,560 persons per square mile in the year 2010. This again is in contrast to the adopted plan, in which the envisioned 2010 population density approximated 2,840 persons per square mile. The envisioned decrease in population density under the existing trends urban sprawl alternative is due to the increase in urban lands, primarily as low-density residential development in outlying areas of the Region.

Urban Services

It is envisioned that public sanitary sewer and public water supply services would be extended to about 77 percent of the total developed area of the Region by the year 2010 under the existing trends urban sprawl alternative (see Table A-9). In addition, about 88 percent of the total population would be served by such facilities under this alternative. The adopted 2010 land use plan envisions about the same area to be served with such public services, about 575 square miles, but because the adopted plan envisions more compact development and thus fewer total square miles of land being used for urban purposes, about 86 percent of the total developed area of the Region would be provided with public sanitary sewer and water supply and over 91 percent of the population would be so served under the adopted plan.

Major Centers

There were 14 existing major retail or office centers in the Region in 1985. Between 1985 and the year 2010, it is envisioned that five additional major retail or office centers would be provided under both the adopted regional plan, and the existing trends urban sprawl alternative. As indicated in Table A-10, the adopted regional plan and the existing trends urban sprawl alternative both envision the provision of a new retail center at the intersection of IH 94 and STH 50 in Kenosha County; the provision of two new office centers in Milwaukee County, one at Milwaukee County Research Park and the other at Park Place, in the northwest portion of Milwaukee County; and a new office center in the Town of Pewaukee, at the intersection of IH 94 and CTH J. As indicated in Map A-2, the existing trends urban sprawl alternative envisioned a new office complex at the intersection of IH 94 and CTH Q in Kenosha County, not so identified under the adopted land use plan, while the adopted land use plan envisioned an additional office center to be located in the City of Mequon, not so identified under the existing trends urban sprawl alternative.

		Existing 1985		2010 Ado	pted Plan	2010 Exist Urban Spraw	ing Trends I Alternative
County	Major Commercial Center	Retail	Office	Retail	Office	Retail	Office
Kenosha	Kenosha CBD Kenosha-southwest Kenosha-west		X	 X	X 	 X	X X
Milwaukee	Bay Shore Capitol Court Mayfair Milwaukee CBD Milwaukee County Research Park Northridge Park Place Southgate-Point Loomis Southridge West Allis	× × × × × × ×	× × ×	× × × × × ×	× × × × × × × × × × × × × × × × × × ×	× × × × × ×	× × × × × × × × × × × × × × × × × × ×
Ozaukee	Mequon				х		
Racine	Racine CBD Regency Mall	x	X	x	X	x	X
Washington	West Bend		х		х		×
Waukesha	Blue Mound Road Pewaukee Waukesha CBD	X 	x 	×	X X X	×	X X X

EXISTING AND PROPOSED MAJOR COMMERCIAL CENTERS IN THE REGION: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

NOTE: To qualify as a major retail center, a site must accommodate at least 2,000 retail jobs. To qualify as a major office center, a site must accommodate at least 3,500 office and service-related jobs.

Source: SEWRPC.

There were 22 existing major industrial centers in the Region in 1985. It is envisioned that three new industrial centers will be provided in the Region under both the existing trends urban sprawl alternative and the 2010 adopted plan. As indicated in Table A-11, these three new centers would be located in the Village of Pleasant Prairie, Kenosha County; in the City of Burlington, Racine County; and in the City of Hartford, Washington County. It should be noted that two of the major industrial centers which existed in the Region in 1985 were envisioned under the existing trends urban sprawl alternative, unlike the adopted land use plan, to decline in importance so as to not meet the criteria as a major industrial center by the year 2010. These two centers, Milwaukee-south and West Allis-east, are both located in Milwaukee County (see Map A-2).

There were 27 existing major outdoor recreation centers and seven special-use outdoor recreation centers in the Region in 1985. As indicated in Table A-12, it is envisioned that four additional major outdoor recreation centers would be provided in the Region under both the existing trends urban sprawl alternative and the adopted 2010 regional land use plan. These centers would be located in the Village of Pleasant Prairie in Kenosha County, in the Town of Sugar Creek in Walworth County, in the Town of West Bend in Washington County, and in the City of Brookfield in Waukesha County.

EXISTING AND PROPOSED MAJOR INDUSTRIAL CENTERS IN THE REGION: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

County	Major Industrial Center	Existing 1985	2010 Adopted Plan	2010 Existing Trends Urban Sprawl Alternative
Kenosha	Kenosha Pleasant Prairie	X	X X	X X
Milwaukee	Cudahy-South Milwaukee Milwaukee-Glendale Milwaukee-Granville Milwaukee-Menomonee Valley east Milwaukee-Menomonee Valley west Milwaukee-near north Milwaukee-near south Milwaukee-north Milwaukee-south Oak Creek West Allis-east West Allis-west West Milwaukee	X X X X X X X X X X X X X X	X X X X X X X X X X X X X X	X X X X X X X X X X X
Racine	Burlington Mt. Pleasant Racine-east	X X	X X X	X X X
Washington	Hartford West Bend-north	×	XX	X X
Waukesha	Butler New Berlin Pewaukee Waukesha-north Waukesha-south	X X X X X	X X X X X	X X X X X

NOTE: To qualify as a major industrial center, a site must accommodate at least 3,500 industrial jobs.

EXISTING AND PROPOSED MAJOR OUTDOOR RECREATION CENTERS IN THE REGION: 1985, 2010 ADOPTED LAND USE PLAN, AND 2010 EXISTING TRENDS URBAN SPRAWL ALTERNATIVE

		Existing 1985		2 Adop	010 ted Plan	2010 Existing Trends Urban Sprawl Alternative		
County	Major Outdoor Recreation Center	Park	Special-Use Site	Park	Special-Use Site	Park	Special-Use Site	
Kenosha	Bong Recreation Area Brighton Dale Petrifying Springs Pleasant Prairie Silver Lake	× × ×	× 	X X X X	×	X X X X X	X 	
Milwaukee	Bender Brown Deer Dretzka Greenfield Lake Michigan-north Lake Michigan-south Lincoln Maier Festival Park Milwaukee County Stadium Milwaukee County Zoo Mitchell Conservatory Oakwood State Fair Park Whitnall	x ^a x x x x x x x x x x x x x	 X X X X X	× × × × × × · · · · · · · · · · · · · ·		× × × × × × × × × × × × × × × × × × ×	 X X X X X	
Ozaukee	Harrington Beach Hawthorne Hills Mee-Kwon	X X X		X X X		X X X		
Racine	Cliffside Ela Johnson	X X ^a X		X X X		X X X		
Walworth	Big Foot Beach Sugar Creek Whitewater Lake	x x		X X X		X X X		
Washington	Paradise Valley Pike Lake	x		X X		x x		
Waukesha	Menomonee Minooka Mitchell Monches Mukwonago Naga-Waukee Old World Wisconsin Ottawa Lake	X X X X X X	 X	× × × × × ×	 X	× × × × ×		

^aSite substantially acquired but not yet developed for recreational use in 1985.

Appendix B

CONSIDERATION IN THE REGIONAL TRANSPORTATION SYSTEM PLANNING PROCESS FOR SOUTHEASTERN WISCONSIN OF THE FIFTEEN SPECIFIC METROPOLITAN PLANNING CONSIDERATIONS SET FORTH IN THE FEDERAL INTERMODAL SURFACE TRANSPORTATION EFFICIENCY ACT (ISTEA)

	Federal ISTEA Metropolitan Planning Consideration	
Number	Description	Consideration in SEWRPC Planning Process
1	Preservation of existing transportation facilities and, where practical, ways to meet transportation needs by using existing transportation facilities more efficiently	The planning process emphasized the full and efficient use of existing transportation facilities and also explicitly identified the resources needed to preserve existing transportation facilities. In addressing existing and potential future traffic congestion, full consideration was given to strategies to obtain greater efficiency from existing transportation facilities, including intersection improvement and channelization, traffic signalization and signal progression, removal of on-street parking during peak travel periods, and freeway traffic management, including ramp metering and high-occupancy vehicle (HOV) bypasses of ramp meters. These transportation system management actions were incorporated in the regional transportation system plan prior to consideration of any highway system capacity improvement and expansion. With respect to consideration of the costs attendant to the preservation of the existing transportation system, the financial analysis of the recommended plan explicitly estimated the costs associated with the resurfacing and reconstruction of existing highway facilities and the rehabilitation and replacement of existing transit facilities
2	The consistency of transportation planning with applicable Federal, State, and local energy conservation programs, goals, and objectives	The energy use implications of the transportation system were explicitly considered in the planning process. An objective and standard utilized in the design and evaluation of alternative transportation system plans was the minimization of the energy use attendant to transportation systems. The recommended transportation system plan may be expected to result in a reduction in energy use compared to a "no-build" alternative. The plan recommends actions expected to result in an increase in carpooling, vanpooling, and transit use, including significantly improved and expanded transit services and actions such as freeway ramp meter HOV bypasses, which provide preferential treatment to high-occupancy vehicles. In addition, the transportation system plan is designed to serve an adopted regional land use plan, which proposes a compact, energy-efficient, urban development pattern
3	The need to relieve congestion and avoid congestion where it does not yet occur	The planning process explicitly addressed the need to relieve existing and potential future traffic congestion. One of the planning objectives and standards utilized in the design and evaluation of alternative transportation system plans was the need to facilitate traffic flow by relieving existing and future traffic congestion. Inventories and analyses were conducted to identify existing levels of traffic congestion and historic trends in levels of traffic congestion. One of the measures utilized in the evaluation of alternative transportation system plans was the level of traffic congestion under each plan. The development of the recommended transportation system plan explicitly considered the potential for land use, travel demand management, traffic management, and public transit measures to reduce traffic congestion prior to any consideration of highway improvement and expansion. Highway capacity improvement and expansion was considered as a measure of last resort in addressing traffic congestion problems. The final recommended transportation system plan included travel demand management, traffic management, land use, public transit, and highway capacity improvement and expansion actions as measures to address identified existing and projected future traffic congestion. The recommended transportation system plan is projected to provide substantial reduction in traffic congestion with respect to both existing levels of traffic congestion and projected future traffic congestion.

	Federal ISTEA Metropolitan Planning Consideration	
Number	Description	Consideration in SEWRPC Planning Process
4	The probable effect of transportation policy decisions on land use and development and the consistency of transportation plans and programs with the provisions of all applicable short- and long-term land use and development plans	The planning process results in a transportation system plan which is fully consistent with land use and development plans as the transportation system plan is designed to serve the adopted regional land use plan and not to forecast land use trends. The transportation system plan therefore serves to promote the implementation of the adopted long-range regional land use plan, which is cooperatively developed with local units of government within Southeastern Wisconsin. The planning process also provides for the assessment of the probable effect of the transportation system plan on land use through the evaluation under the recommended plan and all alternative plans of the relative accessibility provided by the plans throughout Southeastern Wisconsin and the comparison of that accessibility to the proposed development pattern contained within the adopted regional land use plan. The accessibility as provided by the transportation system plan is reviewed with respect to both highway and transit elements of the transportation system
5	The programming of expenditures on transportation enhancement activities	The bicycle and pedestrian system element of the regional transportation system plan is documented in SEWRPC Planning Report No. 43, <u>A Regional Bicycle and</u> <u>Pedestrian Facilities System Plan for Southeastern Wisconsin: 2010</u> . This plan element provides recommendations for the development of bicycle ways and routes and pedestrian facilities which would provide transportation system enhancement and as well assist in meeting future transportation needs
6	The effects of all transportation projects to be undertaken within the metropolitan area, without regard to whether such projects are publicly funded	The transportation planning process has identified and considered the implications of all transportation projects which may affect the regional transportation system within Southeastern Wisconsin, including all arterial highway capacity improvement and expansion projects, all urban transit improvement and expansion projects, all travel demand management, and all traffic management projects
7	International border crossings and access to ports, airports, intermodal transportation facilities, major freight distribution routes, national parks, recreation areas, monuments and historic sites, and military installations	The planning process and the recommended transportation system plan explicitly considered the need to provide appropriate transportation service to, and the transportation system implications of, these facilities as appropriate to Southeastern Wisconsin, including ports, particularly the City of Milwaukee Port; airports, particularly General Mitchell International Airport, but also general aviation airports, including both public and privately-owned airports; intermodal transportation facilities for freight and passenger travel; regional parks and recreation areas, historic sites, and military installations. The plan also addressed the needs of freight distribution with respect to both intraregional freight travel and interregional freight traffic
8	The need for connectivity of roads within the metropolitan area with roads outside the metropolitan area	The planning process and the recommended plan provide for the connectivity of arterial facilities both within Southeastern Wisconsin and between Southeastern Wisconsin and adjacent areas. For such connections along the southern boundary of the Region coordination was achieved by coordinating the regional transportation system plan in Southeastern Wisconsin with that plan as prepared by the Chicago Area Transportation Study, which is the Metropolitan Planning Organization for those counties lying to the south of Southeastern Wisconsin in the State of Illinois. Connectivity of the arterial street system with respect to counties to the west and north of the Southeastern Wisconsin Region boundary was addressed by coordinating the regional system plan of Southeastern Wisconsin with the Wisconsin Department of Transportation to provide for the connectivity of the State trunk highway system, including proposed improvements on that system at the region boundary, including the STH 16 bypass and the USH 12 freeway and the USH 41 freeway. The Wisconsin Department of Transportation with respect to arterial county trunk highway connections across the Region boundary

	Federal ISTEA Metropolitan	
Number	Description	Consideration in SEWRPC Planning Process
9	Transportation needs identified through use of the management systems required by ISTEA	The planning process has been designed to provide those elements of the congestion management system for Southeastern Wisconsin which would provide the definition of performance indicators to measure congestion, the identification of existing and future congestion problems, the definition and evaluation of alternative congestion resolution actions, and the recommendation of strategies to address existing and future congestion. The planning process explicitly considered highway capacity improvement and expansion as a last resort in addressing such congestion problems. The other management systems of pavement, bridges, safety, public transportation and intermodal transportation in cooperation with the Regional Planning Commission and will be incorporated, as appropriate, into future regional transportation system plans
10	Preservation of rights-of-way for construction of future transportation projects, including identification of unused rights-of-way which may be needed for future transportation corridors and identification of those corridors for which action is most needed to prevent destruction or loss	The recommended transportation system plan identifies the conceptual design and alignment of proposed new arterial street and highway and public transit facilities. Such recommended conceptual design and alignment provides sufficient information for local units of government and State government to preserve rights-of-way for such future projects in advance of the conduct of preliminary engineering studies for such highway and transit facilities
11	Methods to enhance the efficient movement of freight	The recommended transportation system plan considered in great detail the intraregional movement of freight, which is largely by truck. Inventories were conducted of intraregional truck travel through a survey of registered trucks within Southeastern Wisconsin and analyses were conducted of truck trip characteristics, travel patterns, and facility utilization. In addition, the transportation system plan included actions to address existing congestion problems and potential future congestion problems, which actions should enhance the movement of both freight and passenger travel. In addition, with respect to interregional freight movement, the SEWRPC plan is fully coordinated with the Statewide transportation system plan being developed by the Wisconsin Department of Transportation with respect to inventories and forecasts of interregional freight traffic
12	The use of life-cycle costs in the design and engineering of bridges, tunnels, or pavement	The planning process considered both capital costs and operating and maintenance costs of both highway and transit system elements of the regional transportation system plan. The costs of constructing new highway system projects were considered, along with the necessary costs of resurfacing and reconstructing existing pavements. In addition, annual operating and maintenance costs attendant to both the highway system and the transit system were considered in preparing the estimated costs of the recommended plan and its alternatives
13	The overall social, economic, energy, and environmental effects of transportation decisions	The planning process explicitly considered the overall social, economic, energy, and environmental effects of transportation decisions and alternatives. A number of the objectives and standards developed to guide the design and evaluation of the alternative transportation system plans addressed the social, economic, energy, and environmental implications of transportation actions. The alternative transportation system plans which were defined as part of the planning process were each evaluated according to their achievement of the defined objectives and standards, which included their social, economic, energy and environmental impacts. A systems-level environmental assessment of the recommended regional transportation system plan was also conducted, which addressed its social, economic, energy, and environmental impacts

	Federal ISTEA Metropolitan Planning Consideration	
Number	Description	Consideration in SEWRPC Planning Process
14	Methods to expand and enhance transit services and to increase the use of such services	The planning process considered alternative means to expand and enhance transit services and to increase the use of such services. Alternatives considered a wide range of potential transit service improvement and expansion, including expansion of rapid, express, and local transit services. In addition, the potential of travel demand management measures, including preferential treatment for freeway access, reserved street lanes, fixed guideways, and pricing measures to increase transit use, were all evaluated under the planning process. The recommended regional transportation system plan includes a 75 percent expansion in transit vehicle-miles of service, including the development of rapid and express transit services and the expansion of existing local transit service. Service improvement and expansion includes extension of service to areas not currently served, improvement of existing service through increased frequency of service and improved speed of service by limiting service stops and through development of fixed guideways and preferential treatment of transit
15	Capital investments that would result in increased security in transit systems	The recommended transportation system plan does not explicitly include any capital investments which would result in increased security in transit systems. The Advisory Committee guiding the preparation of the plan, which includes existing transit operators in the Region and existing local units of government served by public transit, could not identify specific measures to provide such increased security which should be incorporated in the transit system. The transit element of the recommended transportation system plan, however, does include a number of measures which should enhance system security, including improved frequency of transit service; the development of limited-stop transit service, which should concentrate transit activity at a more limited number of stops; and an increase in transit ridership

Appendix C

ACCESSIBILITY PROVIDED TO EMPLOYMENT OPPORTUNITIES AND MAJOR ACTIVITY CENTERS THROUGH TRAVEL ON ARTERIAL STREETS AND HIGHWAYS AND PUBLIC TRANSIT IN THE REGION IN 2010 UNDER THE NO-BUILD ALTERNATIVE AND ALTERNATIVE PLANS 1 AND 3

Table C-1

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON ARTERIAL STREETS AND HIGHWAYS: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLANS

	Urbanized Area Population		Urbanized Area Population Meeting Travel Time Standard on Arterial Streets and Highways								
					Proposed 2010						
Urbanized Area and	Existing		Base Year 1991		No-Build Transportation System Plan		Alternative Plan 1		Alternative Plan 3		
Activity Center	1991	2010	Number	Percent	Number	Percent	Number	Percent	Number	Percent	
Kenosha Urbanized Area Employment-Related ^a Major Retail-Service ^b	94,300 	107,300 	 90,900 0	96.4 0.0	 107,300 0	100.0 0.0	107,300 0	100.0 0.0	107,300 0	100.0 0.0	
Medical Facility ⁶	 		94,300 94,300 94,300 94,300	100.0 100.0 100.0 100.0	107,300 107,300 107,300 107,300	100.0 100.0 100.0 100.0	107,300 107,300 107,300 107,300	100.0 100.0 100.0 100.0	107,300 107,300 107,300 107,300	100.0 100.0 100.0 100.0	
Milwaukee Urbanized Area Employment-Related Major Retail-Service Medical Facility Major Park Higher-Education Facility Scheduled Air Transport	1,226,300 	1,277,100 	1,147,900 1,161,400 1,226,300 1,226,300 1,226,300 1,226,300	93.6 94.7 100.0 100.0 100.0 100.0	1,180,500 1,277,100 1,277,100 1,277,100 1,277,100 1,277,100	92.4 100.0 100.0 100.0 100.0 100.0	1,230,800 1,277,100 1,277,100 1,277,100 1,277,100 1,277,100	94.3 100.0 100.0 100.0 100.0 100.0	1,230,800 1,277,100 1,277,100 1,277,100 1,277,100 1,277,100	94.3 100.0 100.0 100.0 100.0 100.0	
Racine Urbanized Area Employment-Related Major Retail-Service Medical Facility Major Park Higher-Education Facility Scheduled Air Transport	121,800 	132,100 	121,800 32,200 121,800 121,800 121,800 121,800	100.0 86.4 100.0 100.0 100.0 100.0	132,100 132,100 132,100 132,100 132,100 132,100 132,100	100.0 100.0 100.0 100.0 100.0 100.0	132,100 132,100 132,100 132,100 132,100 132,100 132,100	100.0 100.0 100.0 100.0 100.0 100.0	132,100 132,100 132,100 132,100 132,100 132,100 132,100	100.0 100.0 100.0 100.0 100.0 100.0	

^aStandard: 30 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of three major retail and service centers.

^CStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

 $^d {\it Standard:}$ 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

Table C-2

URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON TRANSIT: 1991 AND 2010 ALTERNATIVE TRANSPORTATION SYSTEM PLANS

			Urbanized Area Population Meeting Travel Time Standard on Transit								
	Urbanized Area Population			Proposed 2010							
Urbanized Area and	Base Year		Base Year 1991		No-Build Transportation System Plan		Alternative Plan 1		Alternative Plan 3		
Activity Center	1991	2010	Number	Percent	Number	Percent	Number	Percent	Number	Percent	
Kenosha Urbanized Area Employment-Related ^a Major Retail-Service ^b	94,300 	107,300 	58,000 0	61.5 0.0	51,400 1,400 52,200	 47.9 1.3	97,700 9,500	91.1 8.9	53,200 1,400	 49.6 1.3	
Major Park ^d	* 		12,600 33,200 0	13.4 35.2 0.0	20,700 51,400 0	19.3 47.9 0.0	77,500 80,800 51,900	72.2 75.3 48.4	24,700 68,100 51,900	23.0 63.5 48.4	
Milwaukee Urbanized Area Employment-Related Major Retail-Service ⁹ Medical Facility Major Park Higher-Education Facility Scheduled Air Transport	1,226,300 	1,277,100 	14,200 10,100 700,400 636,300 775,700 338,600	1.2 0.8 57.1 51.9 63.2 27.6	600 35,300 648,000 617,300 740,900 316,800	0.0 2.8 50.7 48.3 58.0 24.8	619,900 282,800 870,200 994,200 1,168,000 776,800	48.5 22.1 68.1 77.8 91.5 60.8	243,600 66,200 728,900 701,000 931,900 569,400	19.1 5.2 57.1 54.9 73.0 44.6	
Racine Urbanized Area Employment-Related Major Retail-Service Medical Facility Major Park Higher-Education Facility Scheduled Air Transport	121,800 	132,100 	59,100 21,500 53,700 24,000 79,600 18,300	48.5 17.7 44.1 19.7 65.4 15.0	56,400 18,300 46,700 37,000 88,400 18,600	42.7 13.9 35.4 28.0 66.9 14.1	107,500 52,500 61,300 59,800 110,700 63,800	81.4 29.7 46.4 45.3 83.8 48.3	94,400 32,100 49,200 38,300 102,700 60,100	71.5 24.3 37.2 29.0 77.7 45.5	

^aStandard: 45 minutes' overall travel time of 40 percent of urbanized area employment opportunities.

^bStandard: 35 minutes' overall travel time of one major retail and service center.

^cStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

 $^d\mathit{Standard:}$ 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.

^gStandard: 35 minutes' overall travel time of three major retail and service centers.

Map C-1

AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY ARTERIAL STREET AND HIGHWAY: 2010 NO-BUILD ALTERNATIVE PLAN



Map C-2

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RETAIL AND SERVICE CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 NO-BUILD ALTERNATIVE PLAN





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR MEDICAL CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 NO-BUILD ALTERNATIVE PLAN



Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RECREATION CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 NO-BUILD ALTERNATIVE PLAN



AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR EDUCATIONAL CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 NO-BUILD ALTERNATIVE PLAN



Map C-6

AREAS MEETING THE TRAVEL TIME STANDARD FOR SCHEDULED AIR TRANSPORT TERMINALS BY ARTERIAL STREET AND HIGHWAY: 2010 NO-BUILD ALTERNATIVE PLAN





AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY TRANSIT: 2010 NO-BUILD ALTERNATIVE PLAN





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RETAIL AND SERVICE CENTERS BY TRANSIT: 2010 NO-BUILD ALTERNATIVE PLAN



AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR MEDICAL CENTERS BY TRANSIT: 2010 NO-BUILD ALTERNATIVE PLAN





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RECREATION CENTERS BY TRANSIT: 2010 NO-BUILD ALTERNATIVE PLAN



Map C-11

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR EDUCATIONAL CENTERS BY TRANSIT: 2010 NO-BUILD ALTERNATIVE PLAN









AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 1



Map C-14

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RETAIL AND SERVICE CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 1





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR MEDICAL CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 1



Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RECREATION CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 1



AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR EDUCATIONAL CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 1





Map C-18

AREAS MEETING THE TRAVEL TIME STANDARD FOR SCHEDULED AIR TRANSPORT TERMINALS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 1





AREAS MEETING THE TRAVEL

TIME STANDARD FOR EMPLOYMENT BY

Map C-20

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RETAIL AND SERVICE CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 1







Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR MEDICAL CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 1





Map C-22

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RECREATION CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 1



AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR EDUCATIONAL CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 1



Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR SCHEDULED AIR TRANSPORT TERMINALS BY TRANSIT: 2010 ALTERNATIVE PLAN 1


Map C-26

AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 3



Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RETAIL AND SERVICE CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 3





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR MEDICAL CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 3



Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RECREATION CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 3



AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR EDUCATIONAL CENTERS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 3





Map C-30

AREAS MEETING THE TRAVEL TIME STANDARD FOR SCHEDULED AIR TRANSPORT TERMINALS BY ARTERIAL STREET AND HIGHWAY: 2010 ALTERNATIVE PLAN 3





AREAS MEETING THE TRAVEL TIME STANDARD FOR EMPLOYMENT BY TRANSIT: 2010 ALTERNATIVE PLAN 3





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RETAIL AND SERVICE CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 3



AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR MEDICAL CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 3



Map C-34

AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR RECREATION CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 3





AREAS MEETING THE TRAVEL TIME STANDARD FOR MAJOR EDUCATIONAL CENTERS BY TRANSIT: 2010 ALTERNATIVE PLAN 3



Source: SEWRPC.

AREAS MEETING THE TRAVEL TIME STANDARD FOR SCHEDULED AIR TRANSPORT TERMINALS BY TRANSIT: 2010 ALTERNATIVE PLAN 3



Appendix D

ENVIRONMENTAL ASSESSMENT OF THE RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

INTRODUCTION

Federal planning guidelines issued jointly by the Federal Highway Administration and the Federal Transit Administration of the U.S. Department of Transportation call for the preparation of environmental assessments in connection with the development of areawide transportation plans. These guidelines envision that the environmental assessments provide the basis, in part, for the preparation of any Federally required environmental impact statements by State and Federal implementing agencies. The Commission believes that the inventory data, analyses, alternative plans, and recommended plans presented in this planning report constitute, in effect, a comprehensive environmental assessment. However, in response to explicit Federal planning guidelines, a separate section of the planning report, termed "environmental assessment," has been prepared. In so doing, the Commission has attempted to bring together all the data and analyses concerning the potential environmental impacts of full implementation of the recommended regional transportation system plan for convenient reference purposes. Accordingly, this appendix sets forth an environmental assessment of the recommended regional transportation system plan.

In considering the material included in this appendix, which has largely been drawn from data previously presented in this report, it should be kept in mind that the regional transportation plan is an areawide, system level plan. As such, an environmental assessment can only be done at a systems level; consequently, done in a relatively generalized manner. More detailed project level planning designed to implement the regional transportation plan will result in more specific environmental assessments. In this report, however, it might be agreed that the more generalized areawide systems level assessment is in fact more meaningful than the more detailed project level assessments limited to isolated corridors of the overall system.

TRANSPORTATION SYSTEM PLAN ASSESSMENT

One of the necessary considerations in the regional transportation system plan reevaluation process is the environmental impact of the recommended regional transportation plan. The importance of environmental considerations in the development of the recommended transportation system plan is reflected by the following, one of the seven basic principles guiding the system planning process, as stated in Chapter II of this report.

Transportation system planning must recognize the existence of a limited natural resource base to which urban and rural development must be properly adjusted to ensure a pleasant and habitable environment. Land, water, and air resources are limited and subject to great misuse through improper land use and transportation system development. Such misuse may lead to serious environmental problems that may be difficult or impossible to correct.

The importance of the consideration of environmental impacts in the planning of transportation facilities and services and in other public actions is further reflected by the National Environmental Policy Act and the Wisconsin Environmental Policy Act, which require the assessment of public projects which may significantly affect the quality of the human environment and involve Federal and State agencies of government. Because past public actions at the project level have in many cases been drawn from long-range systemwide plans and future public actions can be expected to be precipitated by future systemwide long-range planning efforts, there is a need, also, to include an environmental assessment of the recommended transportation plan in the transportation plan reevaluation process.

This report has provided nearly all the information necessary for a regional environmental assessment of the recommended transportation system plan. The regional framework within which

the transportation plan was developed was presented through extensive descriptions of the existing characteristics of the Region, including the previously adopted regional land use and transportation plans; the regional demographic and economic base; the regional natural resource and public utility base; the land use pattern of the Region; and the transportation facilities and travel habits and patterns of the Region. Also in this report, this regional framework was expanded upon thorough analyses and presentation of the anticipated growth and change in the Region to the year 2010. The adopted regional land use plan which the recommended regional transportation system plan was designed to serve was described. Also presented were a number of alternative regional transportation system plans, one of which, with some refinement, became the recommended transportation plan and another of which was a "no-build" option. As documented in Chapter X of this report, the plan alternatives were compared and evaluated against a comprehensive set of transportation system planning objectives and their supporting principles and standards. The objectives, principles, and standards against which the plans were measured included a wide range of considerations directly related to the development of a systems level environmental assessment, as well as considerations related to accessibility, cost, mobility, safety, and system flexibility.

A summary environmental assessment of the recommended regional transportation plan is provided in the following sections of this appendix. The environmental assessment includes a description of the recommended plan and the no-build plan and a summary of their principal environmental impacts. The environmental implications of the recommended and no-build transportation plans to be summarized include land use impacts, energy impacts, vehicle air pollutant emission impacts, and community and local government impacts. Particular attention is given to adverse environmental effects and to the resource commitments attendant to the recommended and no-build plans. The regional framework within which the recommended plan was developed and the characteristics and impacts of transportation plan alternatives other than the no-build plan are not repeated in this section, having been fully described in the body of the report.

The Recommended and the No-Build Transportation System Plans

The recommended regional transportation system plan consists of three basic elements: transportation system management, public transit system maintenance and improvement, and arterial street and highway system maintenance and improvement. A regional bicycle way and pedestrian facilities system plan constitutes a supplementary element of the plan. The no-build transportation system plan represents the alternative of maintaining and preserving the existing transportation system with no improvement to, or expansion of, that system.

<u>Transportation System Management Recommendations</u>: The recommended regional transportation system plan for the year 2010 includes significant transportation system management recommendations. The plan recommends the implementation of the long-planned greater Milwaukee area freeway traffic management system. In general, this system would constrain access by single-occupancy vehicles to the freeway system during peak periods while providing preferential access to highoccupancy vehicles, including transit vehicles, to help ensure high rates of traffic flow at reasonable speeds. The plan recommends that local units of government restrict curb-lane parking on about 442 route-miles of arterial streets during peak periods to accommodate increased traffic flow without investing in the expansion of arterial street capacity. Finally, the plan recommends that local units of government apply traffic engineering techniques to help ensure efficient use of the existing arterial street system and to provide safe and efficient travel ways for bicyclists and pedestrians.

The plan also proposes areawide promotional measures, including an aggressive effort by the Wisconsin Department of Transportation to promote travel through ridesharing, transit use, and bicycle use, as well as telecommuting and work-time rescheduling to reduce personal travel by automobile, particularly during peak periods. This recommendation also includes promoting the creation, and supporting the efforts, of private-sector transportation management associations at major employment centers throughout the Region to help employers meet the requirements of the Employee Commute Options program mandated by the Federal government in the Clean Air Act Amendments of 1990.

The plan also recommends that local units of government facilitate travel by transit, bicycle, and pedestrian modes through detailed, site-specific land use planning as land use development and redevelopment take place within the Region and as communities review private-sector development proposals. The plan recommends that attention be given to ensuring that appropriate mixes of land uses and higher densities of urban development are provided along transit lines and around transit stops and stations. The plan recommends that efficient and direct pedestrian and bicycle pathways to transit stops and stations be provided and that careful attention should be given to the location and orientation of proposed buildings on development sites to facilitate convenient transit use.

<u>Public Transit System Maintenance and Improvement</u>: Transit system development proposals for the year 2010 were prepared for the Region under the recommended and no-build transportation system plans. As shown in Table D-1, under the recommended plan, the regional transit system would, by the year 2010, provide 3,640 round-trip route-miles of service, which would be about 1,344 miles, or 59 percent, greater than that provided in 1991. The transit system would operate 779 vehicles during peak periods by the plan design year 2010. On an average weekday, the recommended plan would provide about 110,600 revenue vehicle-miles and 7,600 revenue vehicle-hours of service, about 75 percent and 46 percent, respectively, greater than provided in 1991. In 1991, the transit system provided 530 buses during peak periods and 63,300 revenue vehicle-miles and 5,220 revenue vehicle-hours of service per average weekday.

Under the no-build plan, the transit system would provide 2,440 round-trip route-miles of service, a 6 percent increase over that provided in 1991, operating 527 vehicles during peak periods. On an average weekday, the system would provide 65,900 revenue vehicle-miles and 5,650 revenue vehicle-hours of service, about 4 percent and 8 percent greater than provided in 1991, respectively.

<u>Rapid Transit Service</u>: Under the recommended plan, a total of 30 rapid transit routes would be provided, of which 27 would be focused on the Milwaukee central business district (CBD) and three on the University of Wisconsin-Milwaukee campus. Rapid transit service would be extended from the City of Milwaukee to the Cities of Racine and Kenosha to the south, to the Village of Mukwonago to the southwest, to the Cities of Waukesha and Oconomowoc to the west; to the Village of Germantown and the City of West Bend to the northwest, and to the City of Cedarburg, the Villages of Saukville and Grafton and the City of Port Washington to the north. Service would be initially provided by buses operating over the freeway system, with route extensions over surface arterial streets for the provision of collector and distributor service. A total of 73 transit stations would be served.

The plan recommends conducting major investment studies to determine the feasibility of eventually upgrading the planned bus rapid transit to bus-on-exclusive-busway or commuter-rail passenger service. The plan explicitly recommends that rapid transit service in the East-West Corridor between N. 4th Street in the City of Milwaukee to STH 164 in the City of Waukesha be provided over a busway and high-occupancy vehicle facility, pending the outcome of current studies being conducted by the Wisconsin Department of Transportation.

<u>Express Transit Service</u>: Under the recommended plan, a total of 12 regular express transit bus routes would be provided. Within Milwaukee County, the routes would be oriented to the Milwaukee CBD, connecting that district to other major trip generators and to provide needed crosstown service. Within the Kenosha and Racine areas, express service would provide direct connections from the central business district to major employment concentrations and between such concentrations. The plan also identifies a potential 15-mile-long light-rail express transit line connecting the City of Glendale, the University of Wisconsin-Milwaukee, the Milwaukee CBD, and the Milwaukee Regional Medical Center and Research Park. These facilities would serve heavily traveled corridors in Milwaukee County. The ultimate decision concerning the inclusion of such facilities in the regional transportation system plan would be made following the completion of the major investment study now underway for the eastwest travel corridor.

Three other travel corridors are identified in the plan as having potential for light-rail express transit service. These include the corridors that extend from the Milwaukee CBD to the General Mitchell

TRANSIT SYSTEM OPERATING CHARACTERISTICS, FACILITIES, AND PERFORMANCE: 1991 AND 2010 NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

			No-Build Plan			Recommended Plan		
		Planned	Increment		Planned			
Transit Service	Base Year 1991	Number	Percent Change	2010	Number	Percent Change	2010	
Round-Trip Route Length (miles)								
Rapid Routes	449	70	15.6	519	911	202.9	1,360	
Express Routes	393	-16	-4.1	377	27	6.9	420	
Kenosha Urbanized Area	171	21	12.3	192	39	22.8	210	
Milwaukee Urbanized Area	1,112	68	6.1	1,180	338	30.4	1,450	
Racine Urbanized Area	171	-3	-1.8	168	29	17.0	200	
Subtotal	1,454	86	5.9	1,540	406	27.9	1,860	
Total	2,296	140	6.1	2,436	1,344	58.5	3,640	
Average Weekday Vehicle Requirements ^a								
Peak Period	530	-3	-0.6	527	249	47.0	779	
Midday Off-Peak Period	285	6	2.1	291	65	22.8	350	
Service Provided on Average Weekday Revenue Vehicle-Miles						×		
Rapid	3,400	-300	-8.8	3,100	13,500	397.1	16,900	
Express	3,300	2,700	81.8	6,000	18,100	548.5	21,400	
Local	56,600	200	0.4	56,800	15,700	27.7	72,300	
Total	63,300	2,600	4.1	65,900	47,300	74.7	110,600	
Revenue Vehicle-Hours								
Rapid	170	-10	-5.9	160	530	311.8	700	
Express	170	190	111.8	360	930	547.1	1,100	
Local	4,880	250	5.1	5,130	920	18.9	5,800	
Total	5,220	430	8.2	5,650	2,380	45.6	7,600	
Seat-Miles	2,975,000	53,300	1.8	3,028,300	2,098,600	70.5	5,073,600	
Service Utilization Ridership								
Average Weekday Revenue Passengers	172,200	-4,200	-2.4	168,000	24,200	14.1	196,400	
Annual Revenue Passengers	50,222,900	-1,224,900	-2.4	48,998,000	7,126,100	14.2	57,349,000	
Revenue Passengers per Revenue Vehicle-Hour	33.0	-3.3	-9.9	29.7	-7.1	-21.7	25.8	
Average Weekday Passenger-Miles	609,100	119,700	19.7	728,800	342,300	56.2	951,400	

Source: SEWRPC.

International Airport; along the 27th Street crosstown corridor from W. Silver Spring Drive to the Southridge Shopping Center; and from the Milwaukee CBD to the Northridge Shopping Center. The feasibility of effectively providing light-rail service in these corridors would be determined through Federally required major investment studies.

Local Transit Service: The local transit service included in the recommended plan represents an expansion and improvement of existing services. The local transit service element of the plan assumes the continued operation and extension of the present grid system of local bus routes in Milwaukee County. That system would extend across the Milwaukee-Waukesha county line to provide service to the eastern tier of urban communities in Waukesha County. The plan also provides for extensions of the systems of local bus routes in the Cities of Waukesha, Racine, and Kenosha. The plan recommends that the local transit operators undertake more detailed implementation studies to identify the best means of providing the needed local service extensions, holding open the possibility of a transit-center-based route system to replace, in some areas, the grid route system.

The recommended plan also recognizes the need to provide transit service in the smaller urban communities of the Region. Explicitly incorporated into the plan are the continuation of the sharedride taxi services now being provided in the Hartford, Port Washington, West Bend, and Whitewater areas of the Region. The plan also recognizes that it may be desirable to establish such demandresponsive services in other smaller urban areas of the Region.

ARTERIAL STREET AND HIGHWAY SYSTEM PRESERVATION, IMPROVEMENT, AND EXPANSION BY ARTERIAL FACILITY TYPE BY COUNTY: 2010 FINAL RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLAN

County	System Preservation (miles)	System Improvement (miles)	System Expansion (miles)	Total (miles)
Kenosha Freeway	12.0 288.9	0.0 44.8	0.0 9.4	12.0 343.1
Subtotal	300.9	44.8	9.4	355.1
Milwaukee Freeway	69.2 666.3	0.0 50.2	0.0 11.3	69.2 727.8
Subtotal	735.5	50.2	11.3	797.0
Ozaukee Freeway	27.4 222.6 250.0	0.0 48.5	0.0 5.9	27.4 277.0
Basing	250.0	40.5	5.9	304.4
Freeway	12.0 330.5	0.0 62.2	0.0 19.1	12.0 411.8
Subtotal	342.5	62.2	19.1	423.8
Walworth Freeway Standard Arterial	50.0 359.5	0.0 38.2	16.7 19.7	66.7 417.4
	409.5	38.2	36.4	484.1
Washington Freeway Standard Arterial	21.6 354.1	21.1 48.9	0.0 22.6	42.7 425.6
Subtotal	375.7	70.0	22.6	468.3
Waukesha Freeway Standard Arterial Subtotal	57.6 556.3	2.0 132.1	5.7 20.5 26.2	65.3 710.0
Region	013.3	134.1	20.2	//4.2
Freeway Standard Arterial	249.8 2,778.2	23.1 424.9	22.4 108.5	295.3 3,311.6
Total	3,028.0	448.0	130.9	3,606.9

Source: SEWRPC.

<u>Arterial Streets and Highways Maintenance and Improvement</u>: Summary data on the arterial street and highway systems under the recommended regional transportation system plan is given by county and arterial facility type in Table D-2. The arterial street improvements have been categorized as system preservation, system improvement, or system expansion. System preservation includes all arterial improvement projects of resurfacing and reconstruction which are required to maintain structural adequacy and serviceability and do not significantly increase existing capacity. System improvement includes all projects which would significantly increase capacity through street widening or relocation. System expansion includes all projects involving construction of new facilities.

Under the recommended plan for the year 2010, the standard arterial system would increase from 3,274 miles of streets and highways in 1991 to approximately 3,607 miles. This additional mileage consists primarily of the addition of existing nonarterial facilities to the arterial system. The

County and Taking	No-Build Plan	Recommended Plan
Kenosha County		80
Number of Nerrosidential Chits		80
Acquisition Demolition and Polosetian Cost	÷ 199.000	53
	\$ 196,000	\$ 17,084,000
Milwaukee County		
Number of Residential Units	11	87
Number of Nonresidential Structures	8	95
Acquisition, Demolition, and Relocation Cost	\$1,768,000	\$ 22,470,000
Ozaukee County		
Number of Residential Units	9	56
Number of Nonresidential Structures	1	5
Acquisition, Demolition, and Relocation Cost	\$1,210,000	\$ 15,071,000
Bacine County		
Number of Besidential Units		202
Number of Nonresidential Structures		27
Acquisition, Demolition, and Relocation Cost	\$ 68.000	\$ 27.578.000
Welworth County	,	
Number of Residential Unite		62
Number of Nonrocidential Structures		
Acquisition Demolition and Polocation Cost	è 92.000	¢ 52 258 000
	9 92,000	\$ 53,358,000
Washington County	-	
Number of Residential Units	8	146
Number of Nonresidential Structures	1	3
Acquisition, Demolition, and Relocation Cost	\$ 369,000	\$ 22,462,000
Waukesha County		
Number of Residential Units	11	164
Number of Nonresidential Structures	4	45
Acquisition, Demolition, and Relocation Cost	\$ 999,000	\$ 43,286,000
Southeastern Wisconsin Region		
Number of Residential Units	39	788
Number of Nonresidential Structures	14	215
Acquisition, Demolition, and Relocation Cost	\$4,704,000	\$201,289,000

COMPARISON OF LAND-TAKING REQUIREMENTS FOR TRANSPORTATION SYSTEM IMPROVEMENTS IN THE REGION BY COUNTY: 2010 NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS

Source: SEWRPC.

construction of new surface arterial facilities would total only about 131 miles under the recommended plan. More specifically, under the plan, about 3,028 miles, or 84 percent, of the arterial street and highway system is recommended for system preservation only; 448 miles, or 12 percent, for system improvement; and the remaining 131 miles, or 4 percent, for system expansion.

Under the no-build plan for the year 2010, the arterial street and highway system would increase by about 6 percent, from 3,274 miles in 1991 to 3,480 miles. The additional 206 miles consist of three miles of new arterial facilities opened to traffic by January 1, 1994, 11 miles of committed new facilities scheduled for construction in 1994 and 1995, and 192 miles of existing nonarterial streets which could be expected to function as arterials by 2010. More than 98 percent of the arterial system, or about 3,422 miles, would be in the system preservation category; only 47 miles would be in the system improvement category. The improvement and expansion categories would be composed entirely of committed projects.

Land Use Impacts

Direct land use impacts of the recommended regional transportation system plan and the no-build plan include the acquisition of land for proposed transportation system development, the dislocation of residential and nonresidential structures by proposed improvement and expansion projects, and the penetration of primary environmental corridors by proposed transportation facilities.

The recommended regional transportation system plan would require significantly more land for system development than would the no-build plan. As shown in Table D-3, the cost of land for new or widened transportation facilities for terminals under the recommended plan, including acquisition, demolition and relocation, would approximate \$201.3 million. The no-build plan would only require \$4.70 million for this purpose. The recommended plan would also entail substantially more dislocation of households, businesses, and industries than would the no-build plan, as shown in Table D-3. Under the recommended plan, the reconstruction of existing, and the construction of new, transportation facilities would displace about 788 residential units and 215 nonresidential structures. Under the no-build plan, about 39 residential units and 14 nonresidential structures would be displaced.

Implementation of the recommended transportation system plan would also require the location of nearly 47.5 miles of new or improved arterial facilities in or through primary environmental corridors. The no-build plan would require the location of about 3.2 miles of arterial facilities in environmental corridors.

Primary environmental corridors contain the best remaining park lands, open space, woodlands, wetlands, wildlife habitat areas, and surface water resources of the Region, and the location of transportation facilities through such environmentally sensitive areas generally should be avoided. However, total avoidance of the penetration or crossing of environmental corridors by transportation facilities is impossible, because both transportation facilities and environmental corridors comprise linear components of the landscape and inevitably must cross at some locations. Under the recommended plan, about 199 acres of environmental corridor may be lost as a result of the construction of proposed new transportation facilities and about 306 acres may be lost as a result of the reconstruction for additional capacity of existing facilities.

It should be noted that this estimate of environmental corridor penetration has been made at the regional system planning level and is necessarily general in nature. More detailed examination during project planning and design phases may indicate that the planned new or improved transportation facilities could be placed outside of environmental corridor lands. This is particularly true, of course, where transportation facilities are proposed to be constructed approximately parallel to the primary environmental corridors.

An indirect land use impact of the recommended and no-build transportation system plans is the relative accessibility to subareas of the Region provided by each plan. Accessibility provides an overall measure of the relative ease with which work, shopping, social, and recreational activities within the Region can be reached from the various subareas of the Region. The accessibility provided under each plan thus influences the relative attractiveness of subareas within the Region with respect to supporting existing land use development and inducing new land use development. The recommended and no-build arterial street and highway plan elements provide nearly the same degree of accessibility to all parts of the Region, and as a consequence, may be expected to support the recommended land use plan to the same degree. The recommended transit plan element provides a higher level of accessibility than the no-build transit plan element to, and within, the central urban areas of the Region. As a consequence, it may be expected to support the regional land use plan to a higher degree than the no-build transit plan element.

Energy Impacts

Another potential environmental impact of the recommended and no-build regional transportation system plans is the fuel consumption by motor vehicles under each plan. The recommended plan may be expected to result in the annual consumption of about 444.3 million gallons of motor fuel. The no-build plan may be expected to result in the annual consumption of 468.1 million gallons of motor fuel (see Table D-4).

As shown in Table D-5, the reasons the recommended plan will result in a lower level of motor fuel consumption than the no-build plan include: a lower level of automobile availability and tripmaking owing primarily to the improved transit service envisioned in the plan and the accommodation of a larger proportion of total travel on more fuel-efficient transportation facilities, transit and uncongested arterial streets and highways.

Noise Impacts

While noise is a transportation-related impact, a proper assessment of this impact cannot be made at the systems level of planning. Although anticipated traffic volumes on the arterial street and highway system may indicate the relative magnitude of transportation-related noise, such volumes cannot be relied upon alone as an accurate measure of potential noise impacts. Transportation-related noise impacts are properly estimated at the facilities level of planning and preliminary engineering, since the degree to which noise may present a problem depends largely upon the horizontal and vertical location and alignment of the facility, the type and location of the land uses adjacent to the facility, pavement types, the use of such noise abatement measures as walls and plantings, and upon the volume, speed, and type of vehicular traffic on the facility.

Table D-4

COMPARISON OF MOTOR FUEL CONSUMPTION BY VEHICLES TRAVELING IN THE REGION BY FUEL TYPE: 2010 NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS

	Annual Motor F (millions	uel Consumption of gallons)
Vehicle Type	No-Build Plan	Recommended Plan
Street and Highway Gasoline Diesel	429.5 32.8	404.0 30.9
Subtotal	462.3	434.9
Transit (diesel)	5.8	9.4
Total	468.1	444.3

Source: SEWRPC.

Vehicle Air Pollutant Emission Impacts

Significant reductions in three air pollutant emissions from motor vehicles operating on the arterial street and highway system, volatile organic compounds, nitrogen oxides, and carbon monoxide, may be expected by the year 2010. These reductions in emissions, which reflect estimates of tons emitted on an average summer weekday, are expected to come about, in fact, through Federally mandated changes in vehicle emission standards, through the required use of cleaner-burning motor fuels, and through travel and traffic management measures, the arterial street and highway and transit improvements included in the plan. As compared to the no-build alternative, the emissions expected under the preliminary recommended plan are, in all three cases, somewhat less. These smaller quantities are, in part, the direct result of the fewer vehicle-miles of travel anticipated under the preliminary recommended plan, compared with the no-build alternative.

As shown in Table D-6, the amount of air pollutant emissions will be reduced as a result of the implementation of the final recommended regional transportation system plan. Under the no-build plan, 29.8 tons of organic compounds, 81.4 tons of nitrogen oxide, and 248.4 tons of carbon monoxide may be expected to be generated on an average summer weekday in 2010. Under the final recommended plan, a total of 27.7 tons of volatile organic compounds, 78.9 tons of nitrogen oxides, and 226.0 tons of carbon monoxide may be expected to be generated on an average summer weekday in 2010.

Community and Local Government Impacts

The recommended and no-build regional transportation system plans also have significant social, economic, and environmental impacts on the communities and local governments within the Region. Community-level impacts include the penetration of neighborhoods by new or improved transportation facilities and the provision of greater accessibility to areas outside the community. Impacts on local government include the reduction of the property tax base and the increase of the local share of transportation system costs envisioned under the plans.

<u>Neighborhood Penetration</u>: The penetration of neighborhood units and neighborhood facility service areas by arterial streets and highways and rapid transit routes may be a community impact of both the recommended and no-build transportation plans. However, the extent of this impact is dependent to a large degree upon project level planning. It is only in such planning that the precise alignments and cross-sections of new transportation facilities would be determined, and the impact on neighborhoods quantitatively determined. It should be noted, however, that the regional land use plan is structured so that arterial streets and highways are to form the exterior boundaries of both existing

CHARACTERISTICS OF THE NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS WHICH RELATE TO ENERGY CONSUMPTION

		the second s
Year 2010 Characteristic	No-Build Plan	Recommended Plan
Automobile and Light Truck Availability	1,310,000	1,300,000
Average Weekday Internal Trip Production Vehicle Transit Person	5,330,000 168,000	5,270,000 196,400
Total Person	6,120,000	6,100,000
Average Weekday Vehicle-Miles of Travel		
Freeway	17,290,000 27,230,000	15,400,000 27,020,000

Source: SEWRPC.

Table D-6

AIR POLLUTANT EMISSIONS FROM VEHICLES ON THE ARTERIAL STREET AND HIGHWAY SYSTEM IN THE REGION: 1991 AND 2010 NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS

	Air Pollutant Emissions (tons per average weekday)					
Transportation System	Volatile Organic Compounds	Nitrogen Oxides	Carbon Monoxide			
Base Year 1991	119.2	129.8	815.1			
No-Build Alternative 2010	29.8	81.4	248.4			
Recommended Plan 2010	27.7	78.9	226.0			

Source: SEWRPC.

and proposed urban neighborhoods; the penetration of such neighborhoods by transportation facilities should, therefore, be minimized.

<u>Property Tax Base Impacts</u>: The reconstruction of existing transportation facilities and the construction of new transportation facilities under the recommended and no-build regional transportation system plans may be expected to impact the property tax base within the Region. An estimate of the property tax base reduction under both plans is shown in Table D-3 as the total value of land consumed and buildings acquired for proposed new and improved transportation facilities. The impact of the recommended plan, \$181.6 million, is substantially greater than that of the no-build plan, \$3.70 million. It should be noted, however, that the relocation of property owners displaced by new or improved transportation facilities to communities within the Region, possibly even within the same local unit of government, will correspondingly lessen any regional tax base loss.

<u>Transportation System Costs</u>: The recommended and no-build regional transportation system plans also have an impact on the Region in terms of the attendant construction, maintenance, and operation costs of the proposed facilities and services. As shown in Table D-7, the recommended regional plan has a total system construction, operation, and maintenance cost of nearly \$522 million per year over the 16-year plan implementation period, about 45 percent greater than the total annual cost of the no-build alternative. This difference in total cost is primarily the result of the greater degree of transportation system improvement envisioned under the plan. Accessibility Impacts: Another important impact of the recommended and no-build transportation plans is the accessibility which the plans propose to provide. Regionwide, the recommended plan provides a higher level of accessibility than the no-build plan. Passenger-hours per average weekday in the year 2010 would total 1.41 million under the recommended plan and 1.85 million under the no-build plan, indicative of the greater level of accessibility provided by the recommended plan. Implementation of the recommended plan would result in only 162 miles of congested arterial facilities within the Region, or 550 miles fewer than under the nobuild plan.

Further social, economic, and environmental implications of the regional transportation system plans are indicated by the impact of accessibility under each plan on subareas of the

Table D-7

TRANSPORTATION SYSTEM CAPITAL AND OPERATION AND MAINTENANCE COSTS IN THE REGION OVER THE PERIOD 1994-2010: NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS

Cost Element	No-Build Plan	Recommended Plan
Arterial Streets and Highways Capital	190 67	277 66
Subtotal	257	343
Public Transit Capital	19 84	77 102
Subtotal	103	179
Total	360	522

Source: SEWRPC.

Region. Accessibility by public transit is a measurement of the availability of transit service. Table D-8 identifies the urbanized area and population that may be expected to be served by the regional transit system. The total transit service area is composed of the area within one-quarter mile of a local or express transit route plus the land outside of this local/express service area which is located within a three-mile radius of a rapid transit station.

As shown in Table D-8, the transit service area for the Milwaukee urbanized area under the recommended plan is larger than the service area under the no-build plan. More than 98 percent of the resident population of the Milwaukee urbanized area would be served by transit under the recommended plan, compared to 92 percent under the no-build plan. In the Kenosha and Racine urbanized areas, 79 and 82 percent of the population, respectively, would be served by transit under the no-build plan. The recommended plan would serve larger proportions of the population in these areas as well. By the year 2010, about 88 percent of the Kenosha urbanized area and 95 percent of the Racine urbanized area would be served by transit.

Accessibility to the major employment, retail service, medical, outdoor recreation centers; higher education facilities; and to the General Mitchell International Airport was also analyzed. As shown in Table D-9, the recommended and no-build plans are very similar with respect to meeting accessibility standards within the urbanized area through travel on arterial streets and highways. Nearly all of the population within the Kenosha, Milwaukee, and Racine urbanized areas may be expected to have access to employment opportunities and to two or more of each of the various types of activity centers in a timely manner through use of the arterial street and highway system. In the Milwaukee urbanized area in particular, the recommended plan provides access by arterial streets and highways under 30 minutes' travel time to 40 percent of the employment opportunities within the urbanized area for 1,230,800 persons, or 94 percent of urbanized area population. The no-build plan would provide such access to 92 percent of the Milwaukee urbanized area population. The relative achievement of the other arterial street and highway travel time standards by the recommended and no-build plans is shown in Table D-9.

Also, as shown in Table D-9, the recommended plan provides for better accessibility to employment opportunities and activity centers through use of the transit system than the no-build plan. In the Milwaukee urbanized area, in particular, implementation of the recommended plan would provide access by transit to 40 percent of urbanized area jobs within 45 minutes for 277,500 persons, or 22 percent of urbanized area population. The no-build plan would provide such access to only 600 persons, or less than one percent of the Milwaukee urbanized area population. The relative achievement of the other transit travel time standards by the recommended and no-build plans is shown in Table D-9.

TOTAL POPULATION SERVED BY PUBLIC TRANSIT BY AREA: 1991 AND 2010 NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS

	Base Year 1991			2010						
					No-Build Plan			Recommended Plan		
Area	Total Population	Population Served	Percent of Total	Total Population	Population Served	Percent of Total	Total Population	Population Served	Percent of Total	
Kenosha Urbanized Area Milwaukee Urbanized Area	94,300 1,226,300 121,800 1,442,400	87,400 1,114,500 114,300 1,316,200	92.7 90.9 93.8 91.2	107,300 1,277,100 132,100 1,516,500	84,900 1,176,700 108,800 1,370,400	79.1 92.1 82.4 90.4	107,300 1,277,100 132,100 1,516,500	94,100 1,255,100 125,900 1,475,100	87.7 98.3 95.3 97.3	
Nonurbanized Area ^a	368,000	43,900	11.9	394,500	66,100	16.8	394,500	128,900	32.7	
Region	1,810,400	1,360,100	75.1	1,911,000	1,436,500	75.2	1,911,000	1,604,000	83.9	

NOTE: The population served by transit is defined as the population located within a quarter-mile walking distance of a local or express transit route and the population located within three miles driving distance of a rapid transit station.

^aDoes not include population served by local demand-responsive shared-ride taxi service.

Source: SEWRPC.

Table D-9

COMPARISON OF URBANIZED AREA POPULATION MEETING TRAVEL TIME STANDARDS TO EMPLOYMENT OPPORTUNITIES AND SELECTED ACTIVITY CENTERS THROUGH TRAVEL ON TRANSIT AND ON ARTERIAL STREETS AND HIGHWAYS: 2010 NO-BUILD AND RECOMMENDED REGIONAL TRANSPORTATION SYSTEM PLANS

		Urbanized Area Population Meeting Travel Time Standards								
		By Arterial Streets					By Transit			
Urbanized Area	Anticipated Area	No-Build	l Plan	Recommen	ded Plan	No-Buil	d Plan	Recommer	ided Plan	
and Activity Center	Population	Number	Percent	Number	Percent	Number	Percent	Number	Percent	
Kenosha Urbanized Area	107,300									
Employment-Related ^a		107,300	100.0	107,300	100.0	51,400	47.9	93,100	86.7	
Major Retail Service ^b		0	0.0	0	0.0	1,400	1.3	6,700	6.2	
Medical Facility ^C		107,300	100.0	107,300	100.0	53,300	49.7	57,400	53.5	
Major Park ^d		107,300	100.0	107,300	100.0	20,700	19.3	24,700	23.0	
Higher-Education Facility ^e		107,300	100.0	107,300	100.0	51,400	47.9	68,100	63.5	
Scheduled Air Transport ^f		107,300	100.0	107,300	100.0	0	0.0	51,900	48.4	
Milwaukee Urbanized Area	1,277,100									
Employment-Related		1,180,500	92.4	1,230,800	94.3	600	0.0	277,500	21.7	
Major Retail-Service		1,277,100	100.0	1,277,100	100.0	35,300	2.8	78,600	6.2	
Medical Facility		1,277,100	100.0	1,277,100	100.0	648,000	50.7	747,000	58.5	
Major Park		1,277,100	100.0	1,277,100	100.0	617,300	48.3	759,900	59.5	
Higher-Education Facility		1,277,100	100.0	1,277,100	100.0	740,900	58.0	959,700	75.1	
Scheduled Air Transport		1,277,100	100.0	1,277,100	100.0	316,800	24.8	630,100	49.3	
Racine Urbanized Area	132,100									
Employment-Related		132,100	100.0	132,100	100.0	56,400	42.7	105,900	80.2	
Major Retail-Service		132,100	100.0	132,100	100.0	18,300	13.9	32,100	24.3	
Medical Facility		132,100	100.0	132,100	100.0	46,700	35.4	53,000	40.1	
Major Park		132,100	100.0	132,100	100.0	37,000	28.0	57,000	43.1	
Higher-Education Facility		132,100	100.0	132,100	100.0	88,400	66.9	102,700	77.7	
Scheduled Air Transport		132,100	100.0	132,100	100.0	18,600	14.1	60,100	45.5	

^aStandard: 30 minutes' overall travel time of 40 percent of urbanized area employment opportunities (45 minutes by transit).

^bStandard: 35 minutes' overall travel time of three major retail and service centers in the Milwaukee urbanized area and one major retail and service center in the Kenosha and Racine urbanized areas.

^CStandard: 40 minutes' overall travel time of a major regional medical center and/or 30 minutes' overall travel time of a hospital or medical clinic.

^dStandard: 40 minutes' overall travel time of a major public outdoor recreation center.

^eStandard: 40 minutes' overall travel time of a vocational school, college, or university.

^fStandard: 60 minutes' overall travel time of a scheduled air transport airport.